ARC MONOGRAPHS

PERFLUOROOCTANOIC ACID (PFOA) AND PERFLUOROOCTANESULFONIC ACID (PFOS)

LINE DIS 1

IARC MONOGRAPHS ON THE IDENTIFICATION OF CARCINOGENIC HAZARDS TO HUMANS

International Agency for Research on Cancer

VOLUME 135



ARC MONOGRAPHS

PERFLUOROOCTANOIC ACID (PFOA) AND PERFLUOROOCTANESULFONIC ACID (PFOS)

THE REAL PROPERTY IN

VOLUME 135

This publication represents the views and expert opinions of an IARC Working Group on the Identification of Carcinogenic Hazards to Humans, which met in Lyon, France, 7–14 November 2023

LYON, FRANCE - 2025

IARC MONOGRAPHS ON THE IDENTIFICATION OF CARCINOGENIC HAZARDS TO HUMANS

International Agency for Research on Cancer



IARC MONOGRAPHS

In 1969, the International Agency for Research on Cancer (IARC) initiated a programme on the evaluation of the carcinogenic hazard of chemicals to humans, involving the production of critically evaluated monographs on individual chemicals. The programme was subsequently expanded to include evaluations of carcinogenic hazards associated with exposures to complex mixtures, lifestyle factors and biological and physical agents, as well as those in specific occupations. The objective of the programme is to elaborate and publish in the form of monographs critical reviews of data on carcinogenicity for agents to which humans are known to be exposed and on specific exposure situations; to evaluate these data in terms of cancer hazard to humans with the help of international working groups of experts in carcinogenesis and related fields; and to identify gaps in evidence. The lists of IARC evaluations are regularly updated and are available on the internet at https://monographs.iarc.who.int/.

This programme has been supported since 1982 by Cooperative Agreement U01 CA33193 with the United States National Cancer Institute, Department of Health and Human Services. Additional support has been provided since 1986 by the European Commission Directorate-General for Employment, Social Affairs, and Inclusion, initially by the Unit of Health, Safety and Hygiene at Work, and since 2014 by the European Union Programme for Employment and Social Innovation "EaSI" (for further information please consult: https://ec.europa.eu/social/easi). Support has also been provided since 1992 by the United States National Institute of Environmental Health Sciences, Department of Health and Human Services. The contents of this volume are solely the responsibility of the Working Group and do not necessarily represent the official views of the United States National Cancer Institute, the United States National Institute of Environmental Health Sciences, the United States Department of Health and Human Services, or the European Commission.



Co-funded by the European Union

Published by the International Agency for Research on Cancer, 25 avenue Tony Garnier, CS 90627, 69366 Lyon Cedex 07, France ©International Agency for Research on Cancer, 2025 Online publication, February 2025

Publications of the World Health Organization enjoy copyright protection in accordance with the provisions of Protocol 2 of the Universal Copyright Convention. All rights reserved.

IARC Monographs (and Corrigenda) are published online at <u>https://publications.iarc.who.int</u>. To report an error, please contact: <u>imo@iarc.who.int</u>.

Distributed by WHO Press, World Health Organization, 20 Avenue Appia, 1211 Geneva 27, Switzerland (tel.: +41 22 791 3264; fax: +41 22 791 4857; website: https://apps.who.int/bookorders; email: bookorders@who.int).

Permissions and rights: Some rights reserved. This work is available under the Creative Commons Attribution-NonCommercial-NoDerivs 3.0 IGO licence (CC BY-NC-ND 3.0 IGO; https://creativecommons.org/licenses/by-nc-nd/3.0/igo/). Under the terms of this licence, you may copy and redistribute the work for non-commercial purposes, provided the work is appropriately cited, as indicated below. In any use of this work, there should be no suggestion that WHO endorses any specific organization, products, or services. The use of the WHO logo is not permitted. Any mediation relating to disputes arising under the licence shall be conducted in accordance with the mediation rules of the World Intellectual Property Organization.

To submit requests for adaptations or commercial use and queries on rights and licensing, see the IARC Publications website (<u>https://publications.iarc.who.int/Rights-And-Permissions</u>).

Third-party materials: If you wish to reuse material from this work that is attributed to a third party, such as tables, figures or images, it is your responsibility to determine whether permission is needed for that reuse and to obtain permission from the copyright holder. The risk of claims resulting from infringement of any third-party-owned component in the work rests solely with the user.

General disclaimers: The designations employed and the presentation of the material in this publication do not imply the expression of any opinion whatsoever on the part of the Secretariat of the World Health Organization concerning the legal status of any country, territory, city, or area or of its authorities, or concerning the delimitation of its frontiers or boundaries.

The mention of specific companies or of certain manufacturers' products does not imply that they are endorsed or recommended by WHO in preference to others of a similar nature that are not mentioned. Errors and omissions excepted, the names of proprietary products are distinguished by initial capital letters.

All reasonable precautions have been taken by WHO to verify the information contained in this publication. However, the published material is being distributed without warranty of any kind, either expressed or implied. The responsibility for the interpretation and use of the material lies with the reader. In no event shall WHO or contributing agencies be liable for damages arising from its use.

The *IARC Monographs* Working Group alone is responsible for the views expressed in this publication.



About the cover: Worker at a wastewater treatment plant. PFOA and PFOS are ubiquitous in the environment and may contaminate drinking-water.

Source: © APchanel/Adobe Stock

How to cite: IARC (2025). Perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS). *IARC Monogr Identif Carcinog Hazards Hum.* 135:1–754.

IARC Library Cataloguing-in-Publication Data

Names: IARC Working Group on the Identification of Carcinogenic Hazards to Humans.

Title: Perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS)

Description: Lyon: International Agency for Research on Cancer, 2025. | Series: IARC monographs on the identification of carcinogenic hazards to humans, ISSN 1017-1606; v. 135. | "This publication represents the views and expert opinions of an IARC Working Group on the Identification of Carcinogenic Hazards to Humans, which met in Lyon, France, 7–14 November 2023." | Includes bibliographical references.

Identifiers: ISBN 9789283201748 (pbk.) | ISBN 9789283202929 (ebook)

Subjects: MESH: Neoplasms--etiology. | Environmental Exposure. | Alkanesulfonic Acids. | Fluorocarbons. | Risk Factors.

Classification: NLM W1



The *IARC Monographs* Working Group and Secretariat for Volume 135, Perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS), which met in Lyon, France, on 7–14 November 2023.

CONTENTS

NOTE TO THE READER				
LIST OF	PARTICIPANTS			
PREAMB	JLE	9		
A. GE	NERAL PRINCIPLES AND PROCEDURES	9		
1.	Background			
2.	Objective and scope	10		
3.	Selection of agents for review	11		
4.	The Working Group and other meeting participants.	11		
5.	Working procedures.	13		
6.	Overview of the scientific review and evaluation process	14		
7.	Responsibilities of the Working Group			
B. SC	IENTIFIC REVIEW AND EVALUATION	17		
1.	Exposure characterization			
2.	Studies of cancer in humans			
3.	Studies of cancer in experimental animals			
4.	Mechanistic evidence			
5.	Summary of data reported			
6.	Evaluation and rationale			
Referen	nces.	37		
GENERA	L REMARKS	41		
1. EXPOS	URE CHARACTERIZATION	49		
1.1	Identification of the agent			
1.2	Production and uses	55		
1.3	Detection and quantification	57		
1.4	Occurrence and exposure	68		
1.5	Regulations and guidelines			
1.6	Quality of exposure assessment in key epidemiological studies of cancer and mechanistic	c		
-	studies in humans	116		
Referen	nces	131		

2. CANCE	R IN HUMANS	157
2.1	Cohort descriptions	158
2.2	Cancers of the urinary tract	213
2.3	Cancers of the male genital tract	252
2.4	Cancers of the breast and thyroid gland	280
2.5	Cancers of the digestive tract	347
2.6	Cancers of the brain and lymphatic and haematopoietic tissue	358
2.7	Cancer of all sites combined.	366
2.8	Evidence synthesis for cancer in humans	370
Referen	nces	390
3. CANCE	R IN EXPERIMENTAL ANIMALS	397
3.1	Perfluorooctanoic acid (PFOA)	397
3.2	Perfluorooctanesulfonic acid (PFOS)	425
3.3	Evidence synthesis for cancer in experimental animals	435
Referen	1ces	438
4. MECHA	ANISTIC EVIDENCE	441
4.1	Absorption, distribution, metabolism, and excretion	441
4.2	Evidence relevant to key characteristics of carcinogens	454
4.3	Evaluation of high-throughput in vitro screening data	658
Referen	1ces	662
5. SUMMA	ARY OF DATA REPORTED	705
5.1	Exposure characterization	705
5.2	Cancer in humans	706
5.3	Cancer in experimental animals.	710
5.4	Mechanistic evidence	711
6. EVALU	ATION AND RATIONALE	717
6.1	Cancer in humans	717
6.2	Cancer in experimental animals.	717
6.3	Mechanistic evidence	.717
6.4	Overall evaluation	.717
6.5	Rationale	717
LIST OF A	ABBREVIATIONS	719
		/1/
ANNEX 1.	Supplementary material for Section 1, Exposure Characterization	727
ANNEX 2.	Actions and regulations for the elimination of PFAS worldwide	729
ANNEX 3.	. Supplementary analyses used in reviewing evidence on cancer in humans	733
ANNEX 4.	. Supplementary material for Section 2, Cancer in Humans	.747
ANNEX 5.	Supplementary material for Sections 4.1 and 4.2. Mechanistic Evidence	.749

ANNEX 6. Supplementary material for Section 4.3, Evaluation of high-throughput in vitro toxicity		
screening data		
SUMMARY OF FINAL EVALUATIONS		

NOTE TO THE READER

The evaluations of carcinogenic hazard in the *IARC Monographs on the Identification of Carcinogenic Hazards to Humans* series are made by international working groups of independent scientists. The *IARC Monographs* classifications do not indicate the level of risk associated with a given level or circumstance of exposure. The *IARC Monographs* do not make recommendations for regulation or legislation.

Anyone who is aware of published data that may alter the evaluation of the carcinogenic hazard of an agent to humans is encouraged to make this information available to the *IARC Monographs* programme, International Agency for Research on Cancer, 25 avenue Tony Garnier, CS 90627, 69366 Lyon Cedex 07, or via email at <u>imo@iarc.who.int</u>, in order that the agent may be considered for re-evaluation by a future Working Group.

Although every effort is made to prepare the monographs as accurately as possible, mistakes may occur. Readers are requested to communicate any errors to the *IARC Monographs* programme. Corrigenda are published online on the relevant webpage for the volume concerned (IARC Publications: <u>https://publications.iarc.who.int/</u>).

LIST OF PARTICIPANTS

Members ¹

Mohamed A-E. Abdallah

School of Geography, Earth and Environmental Sciences University of Birmingham Birmingham UK

Chad R. Blystone

Division of Translational Toxicology National Institute of Environmental Health Sciences Morrisville, North Carolina USA

Jens Peter Bonde (Subgroup Chair, Cancer in Humans)

University of Copenhagen Copenhagen Denmark

Miriam Calkins

United States Centers for Disease Control and Prevention, National Institute for Occupational Safety and Health Cincinnati, Ohio USA

Weihsueh Albert Chiu (Subgroup Chair, Mechanistic Evidence)

School of Veterinary Medicine and Biomedical Sciences Texas A&M University College Station, Texas USA

Guang-Hui Dong

School of Public Health Sun Yat-sen University Guangzhou China

¹ Working Group Members and Invited Specialists serve in their individual capacities as scientists and not as representatives of their government or any organization with which they are affiliated. Affiliations are provided for identification purposes only. Invited Specialists do not serve as Meeting Chair or Subgroup Chair, draft text that pertains to the description or interpretation of cancer data, or participate in the evaluations. Each participant was asked to declare potentially relevant research, employment, and financial interests that are current or that have occurred during the past 4 years. Minimal interests are not disclosed here, and include stock valued at no more than US\$ 1000 overall, grants that provide no more than 5% of the research budget of the expert's organization and that do not support the expert's research or position, and consulting or speaking on matters not before a court or government agency that does not exceed 2% of total professional time or compensation. All other nonpublicly funded grants that support the expert's research or position and all consulting or speaking on behalf of an interested party on matters before a court or government agency are disclosed as potentially significant conflicts of interests.

David Dorman

Department of Molecular Biosciences North Carolina State University Raleigh, North Carolina USA

Rebecca Fry

Gillings School of Public Health University of North Carolina at Chapel Hill Chapel Hill, North Carolina USA

Huan Guo [unable to attend]

Tongji Medical College Huazhong University of Science and Technology Wuhan China

Line Småstuen Haug

Norwegian Institute of Public Health Oslo Norway

Jonathan Hofmann

Division of Cancer Epidemiology and Genetics National Cancer Institute Bethesda, Maryland USA

Jane Hoppin (Subgroup Chair, Exposure Characterization)

North Carolina State University Raleigh, North Carolina USA

Motoki Iwasaki

Division of Epidemiology National Cancer Center Institute for Cancer Control Tokyo Japan

Jun Kanno (Subgroup Chair, Cancer in Experimental Animals)

National Institute of Health Sciences and Nissan Tamagawa Hospital Tokyo Japan

Miroslav Machala

Department of Pharmacology and Toxicology Veterinary Research Institute Brno Czechia

Francesca Romana Mancini

Paris-Saclay University and Institut national de la santé et de la recherche médicale (INSERM) Paris France

Peter Møller

Department of Public Health University of Copenhagen Copenhagen Denmark

Jack Chakmeng Ng

Queensland Alliance for Environmental Health Sciences The University of Queensland Woolloongabba Australia

Marc Pallardy

Université Paris-Saclay and Institut national de la santé et de la recherche médicale (INSERM) Paris France

Gloria Post

New Jersey Department of Environmental Protection Trenton, New Jersey USA

Samira Salihovic

School of Medical Sciences Örebro University Örebro Sweden

Jennifer Schlezinger

School of Public Health Boston University Boston, Massachusetts USA

Anatoly Soshilov

Office of Environmental Health Hazard Assessment California Environmental Protection Agency Sacramento, California USA

Kyle Steenland [not present for final evaluations]

Emory University Atlanta, Georgia USA

Inger-Lise K. Steffensen

Department of Food Safety Norwegian Institute of Public Health Oslo Norway

Silvya Stuchi Maria Engler

School of Pharmaceutical Sciences University of São Paulo São Paulo Brazil

Volodymyr Tryndyak

National Center for Toxicological Research United States Food and Drug Administration Jefferson, Arizona USA

Alexandra White

National Institute of Environmental Health Sciences National Institutes of Health Durham, North Carolina USA

Susan Woskie

Zuckerberg College of Health Sciences University of Massachusetts Lowell Lowell, Massachusetts USA

Shelia Zahm² (Meeting Chair)

National Cancer Institute (retired) Bangor, Maine USA

 2 Dr Zahm is employed by Shelia K. Zahm Consulting, but the work conducted by this company does not involve competing interests for the topics covered by the present *IARC Monographs* meeting.

Invited Specialist

Antony Fletcher ³ [not present for final evaluations]

London School of Hygiene and Tropical Medicine London UK

Representatives

François Pouzaud

Agence nationale de sécurité sanitaire de l'alimentation, de l'environnement et du travail (ANSES) Maisons-Alfort France

Andrea Winquist

National Center for Environmental Health Centers for Disease Control and Prevention Atlanta, Georgia USA

Observers ⁴

Amina Amadou

Centre Léon Bérard Lyon France

Valentin Thomas

Sciences Po Lyon (Institut d'études politiques de Lyon) Lyon France

Danny Whu⁵

International Association of Firefighters Washington, District of Columbia USA

IARC Secretariat

Nahid Ahmadi
Ayat Ahmadi (*Rapporteur*, *Cancer in Humans*)
Lamia Benbrahim-Tallaa (*Rapporteur*, *Cancer in Experimental Animals*)
Wendy Bijoux
Shirisha Chittiboyina (*Rapporteur*, *Cancer in Experimental Animals*)
Aline de Conti (*Rapporteur*, *Mechanistic Evidence*)
Caterina Facchin (*Rapporteur*, *Mechanistic Evidence*)

³ Dr Fletcher reported providing expert opinion for plaintiffs in cases of exposure to PFOA in drinking-water in Italy and in the USA, and for the Government of Australia in defending against a class action lawsuit on behalf of communities seeking damages for drinking-water contaminated from use of aqueous film-forming foam (AFFF) on military airfields, and for the Government of Jersey in his capacity of member of the PFAS Scientific Advisory Panel.

⁴ Each Observer agreed to respect the Guidelines for Observers at *IARC Monographs* meetings. Observers did not serve as Working Group members, draft any part of a monograph, or participate in the evaluations. They also agreed not to contact participants before the meeting, not to lobby them at any time, not to send them written materials, and not to offer them meals or other favours. IARC asked and reminded Working Group members to report any contact or attempt to influence that they may have encountered, either before or during the meeting.

 $^{^{\}rm 5}$ Dr Whu reported receiving support for travel from the International Association of Firefighters.

Federica Madia (Rapporteur, Mechanistic Evidence)
Heidi Mattock (Scientific Editor)
Mira Merdas
Elisa Pasqual (Rapporteur, Cancer in Humans)
Mary Schubauer-Berigan (Programme Head; Rapporteur, Cancer in Humans)
Eero Suonio (Rapporteur, Exposure Characterization)
Susana Viegas (Rapporteur, Exposure Characterization)
Roland Wedekind (Responsible Officer and Rapporteur, Exposure Characterization) ⁶
Ljubica Zupunski

Administrative and Technical Assistance

Noëmi Joncour Jennifer Nicholson Mathieu Rose Sandrine Ruiz

Production Team

Niree Kraushaar Solène Quennehen

Pre- and Post-Meeting Assistance

Scott Bartell Liacine Bouaoun Mark Cleasby Nathan DeBono Fatiha El Ghissassi Yann Grosse Misty Hein John Kaldor Simone Kühnle Alison Lees Karen Müller (*Managing Editor*)

⁶ Dr Wedekind reports that his life partner is a salaried employee of the SEB group working for the Rowenta brand. Tefal, a former user of PFOA, is another brand of the SEB group. This was determined not to present a conflict of interest for the present *IARC Monographs* meeting, since Rowenta has no reported involvement with the agents under review, and Tefal ceased using PFOA more than a decade ago.

PREAMBLE

The Preamble to the *IARC Monographs* describes the objective and scope of the programme, general principles and procedures, and scientific review and evaluations. The *IARC Monographs* embody principles of scientific rigour, impartial evaluation, transparency, and consistency. The Preamble should be consulted when reading a *Monograph* or a summary of a *Monograph's* evaluations. Separate Instructions for Authors describe the operational procedures for the preparation and publication of a volume of the *Monographs*.

A. GENERAL PRINCIPLES AND PROCEDURES

1. Background

Soon after the International Agency for Research on Cancer (IARC) was established in 1965, it started to receive frequent requests for advice on the carcinogenicity of chemicals, including requests for lists of established and suspected human carcinogens. In 1970, an IARC Advisory Committee on Environmental Carcinogenesis recommended "that a compendium on carcinogenic chemicals be prepared by experts. The biological activity and evaluation of practical importance to public health should be referenced and documented." The next year, the IARC Governing Council adopted a resolution that IARC should prepare "monographs on the evaluation of carcinogenic risk of chemicals to man", which became the initial title of the series.

In succeeding years, the scope of the programme broadened as *Monographs* were developed for complex mixtures, occupational exposures, physical agents, biological organisms, pharmaceuticals, and other exposures. In 1988, "of chemicals" was dropped from the title, and in 2019, "evaluation of carcinogenic risks" became "identification of carcinogenic hazards", in line with the objective of the programme.

Identifying the causes of human cancer is the first step in cancer prevention. The identification of a cancer hazard may have broad and profound implications. National and international authorities and organizations can and do use information on causes of cancer in support of actions to reduce exposure to carcinogens in the workplace, in the environment, and elsewhere. Cancer prevention is needed as much today as it was when IARC was established, because the global burden of cancer is high and continues to increase as a result of population growth and ageing and upward trends in some exposures, especially in low- and middle-income countries (https://publications.iarc.who.int/Non-Series-Publications/ World-Cancer-Reports).

IARC's process for developing *Monographs*, which has evolved over several decades, involves

the engagement of international, interdisciplinary Working Groups of expert scientists, the transparent synthesis of different streams of evidence (exposure characterization, cancer in humans, cancer in experimental animals, and mechanisms of carcinogenesis), and the integration of these streams of evidence into an overall evaluation and classification according to criteria developed and refined by IARC. Since the *Monographs* programme was established, the understanding of carcinogenesis has greatly deepened. Scientific advances are incorporated into the evaluation methodology. In particular, strong mechanistic evidence has had an increasing role in the overall evaluations since 1991.

The Preamble is primarily a statement of the general principles and procedures used in developing a *Monograph*, to promote transparency and consistency across *Monographs* evaluations. In addition, IARC provides Instructions for Authors (<u>https://monographs.iarc.who.int/</u> <u>preamble-instructions-for-authors/</u>), which specify more detailed working procedures. IARC routinely updates these Instructions for Authors to reflect advances in methods for cancer hazard identification and accumulated experience, including input from experts.

2. Objective and scope

The objective of the programme is to prepare, with the engagement of international, interdisciplinary Working Groups of experts, scientific reviews and evaluations of evidence on the carcinogenicity of a wide range of agents.

The *Monographs* assess the strength of the available evidence that an agent can cause cancer in humans, based on three streams of evidence: on cancer in humans (see Part B, Section 2), on cancer in experimental animals (see Part B, Section 3), and on mechanistic evidence (see Part B, Section 4). In addition, the exposure to each agent is characterized (see Part B, Section 1). In this Preamble, the term "agent" refers to any

chemical, physical, or biological entity or exposure circumstance (e.g. occupation as a painter) for which evidence on the carcinogenicity is evaluated.

A cancer *hazard* is an agent that is capable of causing cancer, whereas a cancer *risk* is an estimate of the probability that cancer will occur given some level of exposure to a cancer hazard. The *Monographs* assess the strength of evidence that an agent is a cancer hazard. The distinction between hazard and risk is fundamental. The *Monographs* identify cancer hazards even when risks appear to be low in some exposure scenarios. This is because the exposure may be widespread at low levels, and because exposure levels in many populations are not known or documented.

Although the *Monographs* programme has focused on hazard identification, some epidemiological studies used to identify a cancer hazard are also used to estimate an exposure-response relationship within the range of the available data. However, extrapolating exposure-response relationships beyond the available data (e.g. to lower exposures, or from experimental animals to humans) is outside the scope of *Monographs* Working Groups (<u>IARC, 2014</u>). In addition, the *Monographs* programme does not review quantitative risk characterizations developed by other health agencies.

The identification of a cancer hazard should trigger some action to protect public health, either directly as a result of the hazard identification or through the conduct of a risk assessment. Although such actions are outside the scope of the programme, the *Monographs* are used by national and international authorities and organizations to inform risk assessments, formulate decisions about preventive measures, motivate effective cancer control programmes, and choose among options for public health decisions. *Monographs* evaluations are only one part of the body of information on which decisions to control exposure to carcinogens may be based. Options to prevent cancer vary from one situation to another and across geographical regions and take many factors into account, including different national priorities. Therefore, no recommendations are given in the *Monographs* with regard to regulation, legislation, or other policy approaches, which are the responsibility of individual governments or organizations. The *Monographs* programme also does not make research recommendations. However, it is important to note that *Monographs* contribute significantly to the science of carcinogenesis by synthesizing and integrating streams of evidence about carcinogenicity and pointing to critical gaps in knowledge.

3. Selection of agents for review

Since 1984, about every five years IARC convenes an international, interdisciplinary Advisory Group to recommend agents for review by the Monographs programme. IARC selects Advisory Group members who are knowledgeable about current research on carcinogens and public health priorities. Before an Advisory Group meets, IARC solicits nominations of agents from scientists and government agencies worldwide. Since 2003, IARC also invites nominations from the public. IARC charges each Advisory Group with reviewing nominations, evaluating exposure and hazard potential, and preparing a report that documents the Advisory Group's process for these activities and its rationale for the recommendations.

For each new volume of the *Monographs*, IARC selects the agents for review from those recommended by the most recent Advisory Group, considering the availability of pertinent research studies and current public health priorities. On occasion, IARC may select other agents if there is a need to rapidly evaluate an emerging carcinogenic hazard or an urgent need to re-evaluate a previous classification. All evaluations consider the full body of available evidence,

not just information published after a previous review.

A *Monograph* may review:

(a) An agent not reviewed in a previous *Monograph*, if there is potential human exposure and there is evidence for assessing its carcinogenicity. A group of related agents (e.g. metal compounds) may be reviewed together if there is evidence for assessing carcinogenicity for one or more members of the group.

(b) An agent reviewed in a previous *Monograph*, if there is new evidence of cancer in humans or in experimental animals, or mechanistic evidence to warrant re-evaluation of the classification. In the interests of efficiency, the literature searches may build on previous comprehensive searches.

(c) An agent that has been established to be carcinogenic to humans and has been reviewed in a previous *Monograph*, if there is new evidence of cancer in humans that indicates new tumour sites where there might be a causal association. In the interests of efficiency, the review may focus on these new tumour sites.

4. The Working Group and other meeting participants

Five categories of participants can be present at *Monographs* meetings:

(i) Working Group members are responsible for all scientific reviews and evaluations developed in the volume of the Monographs. The Working Group is interdisciplinary and comprises subgroups of experts in the fields of (a) exposure characterization, (b) cancer in humans, (c) cancer in experimental animals, and (d) mechanistic evidence. IARC selects Working Group members on the basis of expertise related to the subject matter and relevant methodologies, and absence

of conflicts of interest. Consideration is also given to diversity in scientific approaches and views, as well as demographic composition. Working Group members generally have published research related to the exposure or carcinogenicity of the agents being reviewed, and IARC uses literature searches to identify most experts. Since 2006, IARC also has encouraged public nominations through its Call for Experts. IARC's reliance on experts with knowledge of the subject matter and/or expertise in methodological assessment is confirmed by decades of experience documenting that there is value in specialized expertise and that the overwhelming majority of Working Group members are committed to the objective evaluation of scientific evidence and not to the narrow advancement of their own research results or a pre-determined outcome (Wild and Cogliano, 2011). Working Group members are expected to serve the public health mission of IARC, and should refrain from consulting and other activities for financial gain that are related to the agents under review, or the use of inside information from the meeting, until the full volume of the Monographs is published.

IARC identifies, from among Working Group members, individuals to serve as Meeting Chair and Subgroup Chairs. At the opening of the meeting, the Working Group is asked to endorse the selection of the Meeting Chair, with the opportunity to propose alternatives. The Meeting Chair and Subgroup Chairs take a leading role at all stages of the review process (see Part A, Section 7), promote open scientific discussions that involve all Working Group members in accordance with normal committee procedures, and ensure adherence to the Preamble.

(ii) *Invited Specialists* are experts who have critical knowledge and experience but who also have a conflict of interest that warrants

exclusion from developing or influencing the evaluations of carcinogenicity. Invited Specialists do not draft any section of the *Monograph* that pertains to the description or interpretation of cancer data, and they do not participate in the evaluations. These experts are invited in limited numbers when necessary to assist the Working Group by contributing their unique knowledge and experience to the discussions.

(iii) *Representatives of national and international health agencies* may attend because their agencies are interested in the subject of the meeting. They do not draft any section of the *Monograph* or participate in the evaluations.

(iv) Observers with relevant scientific credentials may be admitted in limited numbers. Attention is given to the balance of Observers from constituencies with differing perspectives. Observers are invited to observe the meeting and should not attempt to influence it, and they agree to respect the Guidelines for Observers at IARC Monographs meetings. Observers do not draft any section of the *Monograph* or participate in the evaluations. (v) The IARC Secretariat consists of scientists who are designated by IARC and who have relevant expertise. The IARC Secretariat coordinates and facilitates all aspects of the evaluation and ensures adherence to the Preamble throughout development of the scientific reviews and classifications (see Part A, Sections 5 and 6). The IARC Secretariat organizes and announces the meeting, identifies and recruits the Working Group members, and assesses the declared interests of all meeting participants. The IARC Secretariat supports the activities of the Working Group (see Part A, Section 7) by searching the literature and performing title and abstract screening, organizing conference calls to coordinate the development of pre-meeting

Category of participant	Role			
	Prepare text, tables, and analyses	Participate in discussions	Participate in evaluations	Eligible to serve as Chair
Working Group members	\checkmark	\checkmark	\checkmark	\checkmark
Invited Specialists	√a	\checkmark		
Representatives of health agencies		√b		
Observers		√b		
IARC Secretariat	√c	\checkmark	√ ^d	

Table 1 Roles of participants at IARC Monographs meetings

^a Only for the section on exposure characterization.

^b Only at times designated by the Meeting Chair and Subgroup Chairs.

^c When needed or requested by the Meeting Chair and Subgroup Chairs.

^d Only for clarifying or interpreting the Preamble.

drafts and discuss cross-cutting issues, and reviewing drafts before and during the meeting. Members of the IARC Secretariat serve as meeting rapporteurs, assist the Meeting Chair and Subgroup Chairs in facilitating all discussions, and may draft text or tables when designated by the Meeting Chair and Subgroup Chairs. Their participation in the evaluations is restricted to the role of clarifying or interpreting the Preamble.

All participants are listed, with their principal affiliations, in the front matter of the published volume of the *Monographs*. Working Group members and Invited Specialists serve as individual scientists and not as representatives of any organization, government, or industry (Cogliano et al., 2004).

The roles of the meeting participants are summarized in <u>Table 1</u>.

5. Working procedures

A separate Working Group is responsible for developing each volume of the *Monographs*. A volume contains one or more *Monographs*, which can cover either a single agent or several related agents. Approximately one year before the meeting of a Working Group, a preliminary list of agents to be reviewed, together with a Call for Data and a Call for Experts, is announced on the *Monographs* programme website (<u>https://</u><u>monographs.iarc.who.int/</u>).

Before a meeting invitation is extended, each potential participant, including the IARC Secretariat, completes the WHO Declaration of Interests form to report financial interests, employment and consulting (including remuneration for serving as an expert witness), individual and institutional research support, and non-financial interests such as public statements and positions related to the subject of the meeting. IARC assesses the declared interests to determine whether there is a conflict that warrants any limitation on participation (see <u>Table 2</u>).

Approximately two months before a *Monographs* meeting, IARC publishes the names and affiliations of all meeting participants together with a summary of declared interests, in the interests of transparency and to provide an opportunity for undeclared conflicts of interest to be brought to IARC's attention. It is not acceptable for Observers or third parties to contact other participants before a meeting or to lobby them at any time. Meeting participants are asked to report all such contacts to IARC (Cogliano et al., 2005).

The Working Group meets at IARC for approximately eight days to discuss and finalize the scientific review and to develop summaries

Approximate timeframe	Engagement		
Every 5 years	IARC convenes an Advisory Group to recommend high-priority agents for future review		
~1 year before a <i>Monographs</i> meeting	IARC selects agents for review in a new volume of the <i>Monographs</i> IARC posts on its website: Preliminary List of Agents to be reviewed Call for Data and Call for Experts Request for Observer Status WHO Declaration of Interests form		
~8 months before a <i>Monographs</i> meeting	Call for Experts closes		
~4 months before a <i>Monographs</i> meeting	Request for Observer Status closes		
~2 months before a <i>Monographs</i> meeting	IARC posts the names of all meeting participants together with a summary of declared interests, and a statement discouraging contact of the Working Group by interested parties		
~1 month before a <i>Monographs</i> meeting	Call for Data closes		
~2-4 weeks after a <i>Monographs</i> meeting	IARC publishes a summary of evaluations and key supporting evidence		
~9 months after a <i>Monographs</i> meeting	IARC Secretariat publishes the verified and edited master copy of plenary drafts as a <i>Monographs</i> volume		

Table 2 Public engagement during Monographs development

and evaluations. At the opening of the meeting, all participants update their Declaration of Interests forms, which are then reviewed by IARC. Declared interests related to the subject of the meeting are disclosed to the meeting participants during the meeting and in the published volume (Cogliano et al., 2004). The objectives of the meeting are peer review and consensus. During the first part of the meeting, subgroup sessions (covering exposure characterization, cancer in humans, cancer in experimental animals, and mechanistic evidence) review the pre-meeting drafts, develop a joint subgroup draft, and draft subgroup summaries. During the last part of the meeting, the Working Group meets in plenary session to review the subgroup drafts and summaries and to develop the consensus evaluations. As a result, the entire volume is the joint product of the Working Group, and there are no individually authored sections. After the meeting, the master copy is verified by the IARC Secretariat and is then edited and prepared for publication. The aim is to publish the volume within approximately nine months of the Working Group meeting. A summary of the

evaluations and key supporting evidence is prepared for publication in a scientific journal or is made available on the *Monographs* programme website soon after the meeting.

In the interests of transparency, IARC engages with the public throughout the process, as summarized in <u>Table 2</u>.

6. Overview of the scientific review and evaluation process

The Working Group considers all pertinent epidemiological studies, cancer bioassays in experimental animals, and mechanistic evidence, as well as pertinent information on exposure in humans. In general, for cancer in humans, cancer in experimental animals, and mechanistic evidence, only studies that have been published or accepted for publication in the openly available scientific literature are reviewed. Under some circumstances, materials that are publicly available and whose content is final may be reviewed if there is sufficient information to permit an evaluation of the quality of the methods and results of the studies (see Step 1, below). Such materials may include reports and databases publicly available from government agencies, as well as doctoral theses. The reliance on published and publicly available studies promotes transparency and protects against citation of premature information.

The principles of systematic review are applied to the identification, screening, synthesis, and evaluation of the evidence related to cancer in humans, cancer in experimental animals, and mechanistic evidence (as described in Part B, Sections 2–4 and as detailed in the Instructions for Authors). Each *Monograph* specifies or references information on the conduct of the literature searches, including search terms and inclusion/exclusion criteria that were used for each stream of evidence.

In brief, the steps of the review process are as follows:

Step 1. Comprehensive and transparent identification of the relevant information: The IARC Secretariat identifies relevant studies through initial comprehensive searches of literature contained in authoritative biomedical databases (e.g. PubMed, PubChem) and through a Call for Data. These literature searches, designed in consultation with a librarian and other technical experts, address whether the agent causes cancer in humans, causes cancer in experimental systems, and/or exhibits key characteristics of established human carcinogens (in humans or in experimental systems). The Working Group provides input and advice to IARC to refine the search strategies, and identifies literature through other searches (e.g. from reference lists of past Monographs, retrieved articles, and other authoritative reviews).

For certain types of agents (e.g. regulated pesticides and pharmaceuticals), IARC also provides an opportunity to relevant regulatory authorities, and regulated parties through such authorities, to make pertinent

unpublished studies publicly available by the date specified in the Call for Data. Consideration of such studies by the Working Group is dependent on the public availability of sufficient information to permit an independent evaluation of (a) whether there has been selective reporting (e.g. on outcomes, or from a larger set of conducted studies); (b) study quality (e.g. design, methodology, and reporting of results), and (c) study results. Step 2. Screening, selection, and organization of the studies: The IARC Secretariat screens the retrieved literature for inclusion based on title and abstract review, according to pre-defined exclusion criteria. For instance, studies may be excluded if they were not about the agent (or a metabolite of the agent), or if they reported no original data on epidemiological or toxicological end-points (e.g. review articles). The Working Group reviews the title and abstract screening done by IARC, and performs full-text review. Any reasons for exclusion are recorded, and included studies are organized according to factors pertinent to the considerations described in Part B, Sections 2-4 (e.g. design, species, and endpoint). Inclusion of a study does not imply acceptance of the adequacy of the study design or of the analysis and interpretation of the results.

Step 3. Evaluation of study quality: The Working Group evaluates the quality of the included studies based on the considerations (e.g. design, methodology, and reporting of results) described in Part B, Sections 2–4. Based on these considerations, the Working Group may accord greater weight to some of the included studies. Interpretation of the results and the strengths and limitations of a study are clearly outlined in square brackets at the end of study descriptions (see Part B).

Step 4: Report characteristics of included studies, including assessment of study

quality: Pertinent characteristics and results of included studies are reviewed and succinctly described, as detailed in Part B, Sections 1–4. Tabulation of data may facilitate this reporting. This step may be iterative with Step 3.

Step 5: Synthesis and evaluation of strength of evidence: The Working Group summarizes the overall strengths and limitations of the evidence from the individual streams of evidence (cancer in humans, cancer in experimental animals, and mechanistic evidence; see Part B, Section 5). The Working Group then evaluates the strength of evidence from each stream of evidence by using the transparent methods and defined descriptive terms given in Part B, Sections 6a-c. The Working Group then develops, and describes the rationale for, the consensus classification of carcinogenicity that integrates the conclusions about the strength of evidence from studies of cancer in humans, studies of cancer in experimental animals, and mechanistic evidence (see Part B, Section 6d).

7. Responsibilities of the Working Group

The Working Group is responsible for identifying and evaluating the relevant studies and developing the scientific reviews and evaluations for a volume of the *Monographs*. The IARC Secretariat supports these activities of the Working Group (see Part A, Section 4). Briefly, the Working Group's tasks in developing the evaluation are, in sequence:

(i) Before the meeting, the Working Group ascertains that all appropriate studies have been identified and selected, and assesses the methods and quality of each individual study, as outlined above (see Part A, Section 6). The Working Group members

prepare pre-meeting working drafts that present accurate tabular or textual summaries of informative studies by extracting key elements of the study design and results, and highlighting notable strengths and limitations. They participate in conference calls organized by IARC to coordinate the development of working drafts and to discuss cross-cutting issues. Pre-meeting reviews of all working drafts are generally performed by two or more subgroup members who did not participate in study identification, data extraction, or study review for the draft. Each study summary is written or reviewed by someone who is not associated with the study.

(ii) At the meeting, within subgroups, the Working Group members critically review, discuss, and revise the pre-meeting drafts and adopt the revised versions as consensus subgroup drafts. Subgroup Chairs ensure that someone who is not associated with the study leads the discussion of each study summary. A proposed classification of the strength of the evidence reviewed in the subgroup using the *IARC Monographs* criteria (see Part B, Sections 6a–c) is then developed from the consensus subgroup drafts of the evidence summaries (see Part B, Section 5).

(iii) During the plenary session, each subgroup presents its drafts for scientific review and discussion to the other Working Group members, who did not participate in study identification, data extraction, or study review for the drafts. Subgroup Chairs ensure that someone who is not associated with the study leads the discussion of each study summary. After review, discussion, and revisions as needed, the subgroup drafts are adopted as a consensus Working Group product. The summaries and classifications of the strength of the evidence, developed in the subgroup in line with the *IARC Monographs* criteria (see Part B, Sections 6a–c), are considered, revised as needed, and adopted by the full Working Group. The Meeting Chair proposes an overall evaluation using the guidance provided in Part B, Section 6d.

The Working Group strives to achieve consensus evaluations. Consensus reflects broad agreement among the Working Group, but not necessarily unanimity. The Meeting Chair may poll the Working Group to determine the diversity of scientific opinion on issues where consensus is not apparent.

Only the final product of the plenary session represents the views and expert opinions of the Working Group. The entire *Monographs* volume is the joint product of the Working Group and represents an extensive and thorough peer review of the body of evidence (individual studies, synthesis, and evaluation) by an interdisciplinary expert group. Initial working papers and subsequent revisions are not released, because they would give an incomplete and possibly misleading impression of the consensus developed by the Working Group over a full week of deliberation.

B. SCIENTIFIC REVIEW AND EVALUATION

This part of the Preamble discusses the types of evidence that are considered and summarized in each section of a *Monograph*, followed by the scientific criteria that guide the evaluations. In addition, a section of General Remarks at the front of the volume discusses the reasons the agents were scheduled for evaluation and any key issues encountered during the meeting.

1. Exposure characterization

This section identifies the agent and describes its occurrence, main uses, and production locations and volumes, where relevant. It also summarizes the prevalence, concentrations in relevant studies, and relevant routes of exposure in humans worldwide. Methods of exposure measurement and analysis are described, and methods of exposure assessment used in key epidemiological studies reviewed by the Working Group are described and evaluated.

Over the course of the Monographs programme, concepts of exposure and dose have evolved substantially with deepening understanding of the interactions of agents and biological systems. The concept of exposure has broadened and become more holistic, extending beyond chemical, physical, and biological agents to stressors as construed generally, including psychosocial stressors (National Research Council, 2012; National Academies of Sciences, Engineering, and Medicine, 2017). Overall, this broader conceptualization supports greater integration between exposure characterization and other sections of the Monographs. Concepts of absorption, distribution, metabolism, and excretion are considered in the first subsection of mechanistic evidence (see Part B, Section 4a), whereas validated biomarkers of internal exposure or metabolites that are routinely used for exposure assessment are reported on in this section (see Part B, Section 1b).

(a) Identification of the agent

The agent being evaluated is unambiguously identified. Details will vary depending on the type of agent but will generally include physical and chemical properties relevant to the agent's identification, occurrence, and biological activity. If the material that has been tested in experimental animals or in vitro systems is different from that to which humans are exposed, these differences are noted.

For chemical agents, the Chemical Abstracts Service Registry Number is provided, as well as the latest primary name and other names in common use, including important trade names, along with available information on the composition of common mixtures or products containing the agent, and potentially toxic and/or carcinogenic impurities. Physical properties relevant to understanding the potential for human exposure and measures of exposure used in studies in humans are summarized. These might include physical state, volatility, aqueous and fat solubility, and half-life in the environment and/ or in human tissues.

For biological agents, taxonomy and structure are described. Mode of replication, life-cycle, target cells, persistence, latency, and host responses, including morbidity and mortality through pathologies other than cancer, are also presented.

For foreign bodies, fibres and particles, composition, size range, relative dimensions, and accumulation, persistence, and clearance in target organs are summarized. Physical agents that are forms of radiation are described in terms of frequency spectrum and energy transmission.

Exposures may result from, or be influenced by, a diverse range of social and environmental factors, including components of diet, sleep, and physical activity patterns. In these instances, this section will include a description of the agent, its variability across human populations, and its composition or characteristics relevant to understanding its potential carcinogenic hazard to humans and to evaluating exposure assessments in epidemiological studies.

(b) Detection and analysis

Key methods of detection and quantification of the agent are presented, with an emphasis on those used most widely in surveillance, regulation, and epidemiological studies. Measurement methods for sample matrices that are deemed important sources of human exposure (e.g. air, drinking-water, food, residential dust) and for validated exposure biomarkers (e.g. the agent or its metabolites in human blood, urine, or saliva) are described. Information on detection and quantification limits is provided when it is available and is useful for interpreting studies in humans and in experimental animals. This is not an exhaustive treatise but is meant to help readers understand the strengths and limitations of the available exposure data and of the epidemiological studies that rely on these measurements.

(c) Production and use

Historical and geographical patterns and trends in production and use are included when they are available, to help readers understand the contexts in which exposures may occur, both within key epidemiological studies reviewed by the Working Group and in human populations generally. Industries that produce, use, or dispose of the agent are described, including their global distribution, when available. National or international listing as a high-production-volume chemical or similar classification may be included. Production processes with significant potential for occupational exposure or environmental pollution are indicated. Trends in global production volumes, technologies, and other data relevant to understanding exposure potential are summarized. Minor or historical uses with significant exposure potential or with particular relevance to key epidemiological studies are included. Particular effort may be directed towards finding data on production in low- and middle-income countries, where rapid economic development may lead to higher exposures than those in high-income countries.

(d) Exposure

A concise overview of quantitative information on sources, prevalence, and levels of exposure in humans is provided. Representative data from research studies, government reports and websites, online databases, and other citable, publicly available sources are tabulated. Data from low- and middle-income countries are sought and included to the extent feasible; information gaps for key regions are noted. Naturally occurring sources of exposure, if any, are noted. Primary exposure routes (e.g. inhalation, ingestion, skin uptake) and other considerations relevant to understanding the potential for cancer hazard from exposure to the agent are reported.

For occupational settings, information on exposure prevalence and levels (e.g. in air or human tissues) is reported by industry, occupation, region, and other characteristics (e.g. process, task) where feasible. Information on historical exposure trends, protection measures to limit exposure, and potential co-exposures to other carcinogenic agents in workplaces is provided when available.

For non-occupational settings, the occurrence of the agent is described with environmental monitoring or surveillance data. Information on exposure prevalence and levels (e.g. concentrations in human tissues) as well as exposure from and/or concentrations in food and beverages, consumer products, consumption practices, and personal microenvironments is reported by region and other relevant characteristics. Particular importance is placed on describing exposures in life stages or in states of disease or nutrition that may involve greater exposure or susceptibility.

Current exposures are of primary interest; however, information on historical exposure trends is provided when available. Historical exposures may be relevant for interpreting epidemiological studies, and when agents are persistent or have long-term effects. Information gaps for important time periods are noted. Exposure data that are not deemed to have high relevance to human exposure are generally not considered.

(e) Regulations and guidelines

Regulations or guidelines that have been established for the agent (e.g. occupational exposure limits, maximum permitted levels in foods and water, pesticide registrations) are described in brief to provide context about government efforts to limit exposure; these may be tabulated if they are informative for the interpretation of existing or historical exposure levels. Information on applicable populations, specific agents concerned, basis for regulation (e.g. human health risk, environmental considerations), and timing of implementation may be noted. National and international bans on production, use, and trade are also indicated.

This section aims to include major or illustrative regulations and may not be comprehensive, because of the complexity and range of regulatory processes worldwide. An absence of information on regulatory status should not be taken to imply that a given country or region lacks exposure to, or regulations on exposure to, the agent.

(f) Critical review of exposure assessment in key epidemiological studies

Epidemiological studies evaluate cancer hazard by comparing outcomes across differently exposed groups. Therefore, the type and quality of the exposure assessment methods used are key considerations when interpreting study findings for hazard identification. This section summarizes and critically reviews the exposure assessment methods used in the individual epidemiological studies that contribute data relevant to the *Monographs* evaluation.

Although there is no standard set of criteria for evaluating the quality of exposure assessment methods across all possible agents, some concepts are universally relevant. Regardless of the agent, all exposures have two principal dimensions: intensity (sometimes defined as concentration or dose) and time. Time considerations include duration (time from first to last exposure), pattern or frequency (whether continuous or intermittent), and windows of susceptibility. This section considers how each of the key epidemiological studies characterizes these dimensions. Interpretation of exposure information may also be informed by consideration of mechanistic evidence (e.g. as described in Part B, Section 4a), including the processes of absorption, distribution, metabolism, and excretion.

Exposure intensity and time in epidemiological studies can be characterized by using environmental or biological monitoring data, records from workplaces or other sources, expert assessments, modelled exposures, job-exposure matrices, and subject or proxy reports via questionnaires or interviews. Investigators use these data sources and methods individually or in combination to assign levels or values of an exposure metric (which may be quantitative, semi-quantitative, or qualitative) to members of the population under study.

In collaboration with the Working Group members reviewing human studies (of cancer and of mechanisms), key epidemiological studies are identified. For each selected study, the exposure assessment approach, along with its strengths and limitations, is summarized using text and tables. Working Group members identify concerns about exposure assessment methods and their impacts on overall quality for each study reviewed (see Part B, Sections 2d and 4d). In situations where the information provided in the study is inadequate to properly consider the exposure assessment, this is indicated. When adequate information is available, the likely direction of bias due to error in exposure measurement, including misclassification (overestimated effects, underestimated effects, or unknown) is discussed.

2. Studies of cancer in humans

This section includes all pertinent epidemiological studies (see Part B, Section 2b) that include cancer as an outcome. These studies encompass certain types of biomarker studies, for example, studies with biomarkers as exposure metrics (see Part B, Section 2) or those evaluating histological or tumour subtypes and molecular signatures in tumours consistent with a given exposure (<u>Alexandrov et al., 2016</u>). Studies that evaluate early biological effect biomarkers are reviewed in Part B, Section 4.

(a) Types of study considered

Several types of epidemiological studies contribute to the assessment of carcinogenicity in humans; they typically include cohort studies (including variants such as case-cohort and nested case-control studies), case-control studies, ecological studies, and intervention studies. Rarely, results from randomized trials may be available. Exceptionally, case reports and case series of cancer in humans may also be reviewed. In addition to these designs, innovations in epidemiology allow for many other variants that may be considered in any given *Monographs* evaluation.

Cohort and case-control studies typically have the capacity to relate individual exposures under study to the occurrence of cancer in individuals, and provide an estimate of effect (such as relative risk) as the main measure of association. Well-conducted cohort and case-control studies provide most of the evidence of cancer in humans evaluated by Working Groups. Intervention studies are much less common, but when available can provide strong evidence for making causal inferences.

In ecological studies, the units of investigation are usually whole populations (e.g. in particular geographical areas or at particular times), and cancer frequency is related to a summary measure of the exposure in the population under study. In ecological studies, data on individual exposure and outcome are not available, which renders this type of study more prone to confounding and exposure misclassification. In some circumstances, however, ecological studies may be informative, especially when the unit of exposure is most accurately measured at the population level (see, for example, the *Monograph* on arsenic in drinking-water; <u>IARC</u>, 2004).

Exceptionally, case reports and case series may provide compelling evidence about the carcinogenicity of an agent. In fact, many of the early discoveries of occupational cancer hazards came about because of observations by workers and their clinicians, who noted a high frequency of cancer in workers who share a common occupation or exposure. Such observations may be the starting point for more structured investigations, but in exceptional circumstances, when the risk is high enough, the case series may in itself provide compelling evidence. This would be especially warranted in situations where the exposure circumstance is fairly unusual, as it was in the example of plants containing aristolochic acid (IARC, 2012a).

The uncertainties that surround the interpretation of case reports, case series, and ecological studies typically make them inadequate, except in rare instances as described above, to form the sole basis for inferring a causal relationship. However, when considered together with cohort and case-control studies, these types of study may support the judgement that a causal relationship exists.

Epidemiological studies of benign neoplasms, pre-neoplastic lesions, malignant precursors, and other end-points are also reviewed when they relate to the agents reviewed. On occasion they can strengthen inferences drawn from studies of cancer itself. For example, benign brain tumours may share common risk factors with those that are malignant, and benign neoplasms (or those of uncertain behaviour) may be part of the causal path to malignancies (e.g. myelodysplastic syndromes, which may progress to acute myeloid leukaemia).

(b) Identification of eligible studies of cancer in humans

Relevant studies of cancer in humans are identified by using systematic review principles as described in Part A, further elaborated in the Instructions for Authors, and as detailed below. Eligible studies include all studies in humans of exposure to the agent of interest with cancer as an outcome. Multiple publications on the same study population are identified so that the number of independent studies is accurately represented. Multiple publications may result, for example, from successive follow-ups of a single cohort, from analyses focused on different aspects of an exposure-disease association, or from inclusion of overlapping populations. Usually in such situations, only the most recent, most comprehensive, or most informative report is reviewed in detail.

(c) Assessment of study quality and informativeness

Epidemiological studies are potentially susceptible to several different sources of error, summarized briefly below. Qualities of individual studies that address these issues are also described below.

Study quality is assessed as part of the structured expert review process undertaken by the Working Group. A key aspect of quality assessment is consideration of the possible roles of chance and bias in the interpretation of epidemiological studies. Chance, which is also called random variation, can produce misleading study results. This variability in study results is strongly influenced by the sample size: smaller studies are more likely than larger studies to have effect estimates that are imprecise. Confidence intervals around a study's point estimate of effect are used routinely to indicate the range of values of the estimate that could easily be produced by chance alone.

Bias is the effect of factors in study design or conduct that lead an association to erroneously appear stronger or weaker than the association that really exists between the agent and the disease. Biases that require consideration are varied but are usually categorized as selection bias, information bias (e.g. error in measurement of exposure and diseases), and confounding (or confounding bias) (Rothman et al., 2008). Selection bias in an epidemiological study occurs when inclusion of participants from the eligible population or their follow-up in the study is influenced by their exposure or their outcome (usually disease occurrence). Under these conditions, the measure of association found in the study will not accurately reflect the association that would otherwise have been found in the eligible population (Hernán et al., 2004). Information bias results from inaccuracy in exposure or outcome measurement. Both can cause an association between hypothesized cause and effect to appear stronger or weaker than it really is. Confounding is a mixing of extraneous effects with the effects of interest (Rothman et al., 2008). An association between the purported causal factor and another factor that is associated with an increase or decrease in incidence of disease can lead to a spurious association or absence of a real association of the presumed causal factor with the disease. When either of these occurs, confounding is present.

In assessing study quality, the Working Group consistently considers the following aspects:

- **Study description:** Clarity in describing the study design and its implementation, and the completeness of reporting of all other key information about the study and its results.
- **Study population:** Whether the study population was appropriate for evaluating the

association between the agent and cancer. Whether the study was designed and carried out to minimize selection bias. Cancer cases in the study population must have been identified in a way that was independent of the exposure of interest, and exposure assessed in a way that was not related to disease (outcome) status. In these respects, completeness of recruitment into the study from the population of interest and completeness of follow-up for the outcome are essential measures.

- Outcome measurement: The appropriateness of the cancer outcome measure (e.g. mortality vs incidence) for the agent and cancer type under consideration, outcome ascertainment methodology, and the extent to which outcome misclassification may have led to bias in the measure(s) of association.
- Exposure measurement: The adequacy of the methods used to assess exposure to the agent, and the likelihood (and direction) of bias in the measure(s) of association due to error in exposure measurement, including misclassification (as described in Part B, Section 1f).
- Assessment of potential confounding: To what extent the authors took into account in the study design and analysis other variables (including co-exposures, as described in Part B, Section 1d) that can influence the risk of disease and may have been related to the exposure of interest. Important sources of potential confounding by such variables should have been addressed either in the design of the study, such as by matching or restriction, or in the analysis, by statistical adjustment. In some instances, where direct information on confounders is unavailable, use of indirect methods to evaluate the potential impact of confounding on exposure-disease associations is appropriate (e.g. Axelson and Steenland, 1988; Richardson et al., 2014).

- Other potential sources of bias: Each epidemiological study is unique in its study population, its design, its data collection, and, consequently, its potential biases. All possible sources of bias are considered for their possible impact on the results. The possibility of reporting bias (i.e. selective reporting of some results and the suppression of others) should be explored.
- Statistical methodology: Adequacy of the • statistical methods used and their ability to obtain unbiased estimates of exposure-outcome associations, confidence intervals, and test statistics for the significance of measures of association. Appropriateness of methods used to investigate confounding, including adjusting for matching when necessary and avoiding treatment of probable mediating variables as confounders. Detailed analyses of cancer risks in relation to summary measures of exposure such as cumulative exposure, or temporal variables such as age at first exposure or time since first exposure, are reviewed and summarized when available.

For the sake of economy and simplicity, in this Preamble the list of possible sources of error is referred to with the phrase "chance, bias, and confounding", but it should be recognized that this phrase encompasses a comprehensive set of concerns pertaining to study quality.

These sources of error do not constitute and should not be used as a formal checklist of indicators of study quality. The judgement of experienced experts is critical in determining how much weight to assign to different issues in considering how all of these potential sources of error should be integrated and how to rate the potential for error related to each of these considerations.

The informativeness of a study is its ability to show a true association, if there is one, between the agent and cancer, and the lack of an association, if no association exists. Key determinants of informativeness include: having a study population of sufficient size to obtain precise estimates of effect; sufficient elapsed time from exposure to measurement of outcome for an effect, if present, to be observable; presence of an adequate exposure contrast (intensity, frequency, and/ or duration); biologically relevant definitions of exposure; and relevant and well-defined time windows for exposure and outcome.

(d) Meta-analyses and pooled analyses

Independent epidemiological studies of the same agent may lead to inconsistent results that are difficult to interpret or reconcile. Combined analyses of data from multiple studies may be conducted as a means to address this ambiguity. There are two types of combined analysis. The first involves combining summary statistics such as relative risks from individual studies (meta-analysis), and the second involves a pooled analysis of the raw data from the individual studies (pooled analysis) (Greenland and O'Rourke, 2008).

The strengths of combined analyses are increased precision because of increased sample size and, in the case of pooled analyses, the opportunity to better control for potential confounders and to explore in more detail interactions and modifying effects that may explain heterogeneity among studies. A disadvantage of combined analyses is the possible lack of comparability of data from various studies, because of differences in population characteristics, subject recruitment, procedures of data collection, methods of measurement, and effects of unmeasured covariates that may differ among studies. These differences in study methods and quality can influence results of either meta-analyses or pooled analyses. If published meta-analyses are to be considered by the Working Group, their adequacy needs to be carefully evaluated, including the methods used to identify eligible studies

and the accuracy of data extracted from the individual studies.

The Working Group may conduct ad hoc meta-analyses during the course of a *Monographs* meeting, when there are sufficient studies of an exposure–outcome association to contribute to the Working Group's assessment of the association. The results of such unpublished original calculations, which would be specified in the text by presentation in square brackets, might involve updates of previously conducted analyses that incorporate the results of more recent studies, or de novo analyses.

Irrespective of the source of data for the meta-analyses and pooled analyses, the following key considerations apply: the same criteria for data quality must be applied as for individual studies; sources of heterogeneity among studies must be carefully considered; and the possibility of publication bias should be explored.

(e) Considerations in assessing the body of epidemiological evidence

The ability of the body of epidemiological evidence to inform the Working Group about the carcinogenicity of the agent is related to both the quantity and the quality of the evidence. There is no formulaic answer to the question of how many studies of cancer in humans are needed from which to draw inferences about causality, although more than a single study in a single population will almost always be needed. The number will depend on the considerations relating to evidence described below.

After the quality of individual epidemiological studies of cancer has been assessed and the informativeness of the various studies on the association between the agent and cancer has been evaluated, a judgement is made about the strength of evidence that the agent in question is carcinogenic to humans. In making its judgement, the Working Group considers several aspects of the body of evidence (e.g. <u>Hill, 1965</u>; Rothman et al., 2008; Vandenbroucke et al., 2016).

A strong association (e.g. a large relative risk) is more likely to indicate causality than is a weak association, because it is more difficult for confounding to falsely create a strong association. However, it is recognized that estimates of effect of small magnitude do not imply lack of causality and may have impact on public health if the disease or exposure is common. Estimates of effect of small magnitude could also contribute useful information to the assessment of causality if level of risk is commensurate with level of exposure when compared with risk estimates from populations with higher exposure (e.g. as seen in residential radon studies compared with studies of radon from uranium mining).

Associations that are consistently observed in several studies of the same design, or in studies that use different epidemiological approaches, or under different circumstances of exposure are more likely to indicate a causal relationship than are isolated observations from single studies. If there are inconsistent results among investigations, possible reasons are sought (e.g. differences in study informativeness because of latency, exposure levels, or assessment methods). Results of studies that are judged to be of high quality and informativeness are given more weight than those of studies judged to be methodologically less sound or less informative.

Temporality of the association is an essential consideration: that is, the exposure must precede the outcome.

An observation that cancer risk increases with increasing exposure is considered to be a strong indication of causality, although the absence of a graded response is not necessarily evidence against a causal relationship, and there are several reasons why the shape of the exposure–response association may be non-monotonic (e.g. <u>Stayner</u> et al., 2003). The demonstration of a decline in risk after cessation of or reduction in exposure

in individuals or in whole populations also supports a causal interpretation of the findings.

Confidence in a causal interpretation of the evidence from studies of cancer in humans is enhanced if it is coherent with physiological and biological knowledge, including information about exposure to the target organ, latency and timing of the exposure, and characteristics of tumour subtypes.

The Working Group considers whether there are subpopulations with increased susceptibility to cancer from the agent. For example, molecular epidemiology studies that identify associations between genetic polymorphisms and inter-individual differences in cancer susceptibility to the agent(s) being evaluated may contribute to the identification of carcinogenic hazards to humans. Such studies may be particularly informative if polymorphisms are found to be modifiers of the exposure–response association, because evaluation of polymorphisms may increase the ability to detect an effect in susceptible subpopulations.

When, in the process of evaluating the studies of cancer in humans, the Working Group identifies several high-quality, informative epidemiological studies that clearly show either no positive association or an inverse association between an exposure and a specific type of cancer, a judgement may be made that, in the aggregate, they suggest evidence of lack of carcinogenicity for that cancer type. Such a judgement requires, first, that the studies strictly meet the standards of design and analysis described above. Specifically, the possibility that bias, confounding, or misclassification of exposure or outcome could explain the observed results should be considered and ruled out with reasonable confidence. In addition, all studies that are judged to be methodologically sound should (a) be consistent with an estimate of relative effect of unity (or below unity) for any observed level of exposure, (b) when considered together, provide a combined estimate of relative risk that is at or below unity, and (c) have a narrow confidence interval. Moreover, neither any

individual well-designed and well-conducted study nor the pooled results of all the studies should show any consistent tendency that the relative risk of cancer increases with increasing level of exposure. It must be noted that evidence of lack of carcinogenicity obtained from several epidemiological studies can apply only to the type(s) of cancer studied, to the exposure levels reported and the timing and route of exposure studied, to the intervals between first exposure and disease onset observed in these studies, and to the general population(s) studied (i.e. there may be susceptible subpopulations or life stages). Experience from studies of cancer in humans indicates that the period from first exposure to the development of clinical cancer is sometimes longer than 20 years; therefore, latency periods substantially shorter than about 30 years cannot provide evidence of lack of carcinogenicity. Furthermore, there may be critical windows of exposure, for example, as with diethylstilboestrol and clear cell adenocarcinoma of the cervix and vagina (IARC, 2012a).

3. Studies of cancer in experimental animals

Most human carcinogens that have been studied adequately for carcinogenicity in experimental animals have produced positive results in one or more animal species. For some agents, carcinogenicity in experimental animals was demonstrated before epidemiological studies identified their carcinogenicity in humans. Although this observation cannot establish that all agents that cause cancer in experimental animals also cause cancer in humans, it is biologically plausible that agents for which there is *sufficient evidence* of carcinogenicity in experimental animals (see Part B, Section 6b) present a carcinogenic hazard to humans. Accordingly, in the absence of additional scientific information, such as strong evidence that a given agent causes cancer in

experimental animals through a species-specific mechanism that does not operate in humans (see Part B, Sections 4 and 6; <u>Capen et al., 1999</u>; <u>IARC, 2003</u>), these agents are considered to pose a potential carcinogenic hazard to humans. The inference of potential carcinogenic hazard to humans does not imply tumour site concordance across species (<u>Baan et al., 2019</u>).

(a) Types of studies considered

Relevant studies of cancer in experimental animals are identified by using systematic review principles as described in Part A, further elaborated in the Instructions for Authors, and as detailed below. Consideration is given to all available long-term studies of cancer in experimental animals with the agent under review (or possibly metabolites or derivatives of the agent) (see Part A, Section 7) after a thorough evaluation of the study features (see Part B, Section 3b). Those studies that are judged to be irrelevant to the evaluation or judged to be inadequate (e.g. too short a duration, too few animals, poor survival; see below) may be omitted. Guidelines for conducting long-term carcinogenicity experiments have been published (e.g. OECD, 2018).

In addition to conventional long-term bioassays, alternative studies (e.g. in genetically engineered mouse models) may be considered in assessing carcinogenicity in experimental animals, also after a critical evaluation of the study features. For studies of certain exposures, such as viruses that typically only infect humans, use of such specialized experimental animal models may be particularly important; models include genetically engineered mice with targeted expression of viral genes to tissues from which human cancers arise, as well as humanized mice implanted with the human cells usually infected by the virus.

Other types of studies can provide supportive evidence. These include: experiments in which the agent was administered in the presence of factors that modify carcinogenic effects (e.g. initiation-promotion studies); studies in which the end-point was not cancer but a defined precancerous lesion; and studies of cancer in non-laboratory animals (e.g. companion animals) exposed to the agent.

(b) Study evaluation

Considerations of importance in the interpretation and evaluation of a particular study include: (i) whether the agent was clearly characterized, including the nature and extent of impurities and contaminants and the stability of the agent, and, in the case of mixtures, whether the sample characterization was adequately reported; (ii) whether the dose was monitored adequately, particularly in inhalation experiments; (iii) whether the doses, duration and frequency of treatment, duration of observation, and route of exposure were appropriate; (iv) whether appropriate experimental animal species and strains were evaluated; (v) whether there were adequate numbers of animals per group; (vi) whether animals were allocated randomly to groups; (vii) whether the body weight, food and water consumption, and survival of treated animals were affected by any factors other than the test agent; (viii) whether the histopathology review was adequate; and (ix) whether the data were reported and analysed adequately.

(c) Outcomes and statistical analyses

An assessment of findings of carcinogenicity in experimental animals involves consideration of (i) study features such as route, doses, schedule and duration of exposure, species, strain (including genetic background where applicable), sex, age, and duration of follow-up; (ii) the spectrum of neoplastic response, from pre-neoplastic lesions and benign tumours to malignant neoplasms; (iii) the incidence, latency, severity, and multiplicity of neoplasms and pre-neoplastic
lesions; (iv) the consistency of the results for a specific target organ or organs across studies of similar design; and (v) the possible role of modifying factors (e.g. diet, infection, stress).

Key factors for statistical analysis include: (i) number of animals studied and number examined histologically, (ii) number of animals with a given tumour type or lesion, and (iii) duration of survival.

Benign tumours may be combined with malignant tumours in the assessment of tumour incidence when (a) they occur together with and originate from the same cell type as malignant tumours in an organ or tissue in a particular study and (b) they appear to represent a stage in the progression to malignancy (Huff et al., 1989). The occurrence of lesions presumed to be preneoplastic may in certain instances aid in assessing the biological plausibility of any neoplastic response observed.

Evidence of an increased incidence of neoplasms with increasing level of exposure strengthens the inference of a causal association between the exposure and the development of neoplasms. The form of the dose–response relationship can vary widely, including non-linearity, depending on the particular agent under study and the target organ. The dose–response relationship can also be affected by differences in survival among the treatment groups.

The statistical methods used should be clearly stated and should be the generally accepted techniques refined for this purpose (Peto et al., 1980; Gart et al., 1986; Portier and Bailer, 1989; Bieler and Williams, 1993). The choice of the most appropriate statistical method requires consideration of whether there are differences in survival among the treatment groups; for example, reduced survival because of non-tumour-related mortality can preclude the occurrence of tumours later in life and a survival-adjusted analysis would be warranted. When detailed information on survival is not available, comparisons of the proportions of tumour-bearing animals among the effective number of animals (alive at the time that the first tumour was discovered) can be useful when significant differences in survival occur before tumours appear. The lethality of the tumour also requires consideration: for rapidly fatal tumours, the time of death provides an indication of the time of tumour onset and can be assessed using life-table methods; non-fatal or incidental tumours that do not affect survival can be assessed using methods such as the Mantel-Haenszel test for changes in tumour prevalence. Because tumour lethality is often difficult to determine, methods such as the poly-*k* test that do not require such information can also be used. When results are available on the number and size of tumours seen in experimental animals (e.g. papillomas on mouse skin, liver tumours observed through nuclear magnetic resonance tomography), other, more complicated statistical procedures may be needed (Sherman et al., 1994; Dunson et al., 2003).

The concurrent control group is generally the most appropriate comparison group for statistical analysis; however, for uncommon tumours, the analysis may be improved by considering historical control data, particularly when betweenstudy variability is low. Historical controls should be selected to resemble the concurrent controls as closely as possible with respect to species, sex, and strain, as well as other factors, such as basal diet and general laboratory environment, which may affect tumour response rates in control animals (Haseman et al., 1984; Fung et al., 1996; Greim et al., 2003). It is generally not appropriate to discount a tumour response that is significantly increased compared with concurrent controls by arguing that it falls within the range of historical controls.

Meta-analyses and pooled analyses may be appropriate when the experimental protocols are sufficiently similar.

4. Mechanistic evidence

Mechanistic data may provide evidence of carcinogenicity and may also help in assessing the relevance and importance of findings of cancer in experimental animals and in humans (Guyton et al., 2009; Parkkinen et al., 2018) (see Part B, Section 6). Mechanistic studies have gained in prominence, increasing in their volume, diversity, and relevance to cancer hazard evaluation, whereas studies pertinent to other streams of evidence evaluated in the Monographs (i.e. studies of cancer in humans and lifetime cancer bioassays in rodents) may only be available for a fraction of agents to which humans are currently exposed (Guyton et al., 2009, 2018). Mechanistic studies and data are identified, screened, and evaluated for quality and importance to the evaluation by using systematic review principles as described in Part A, further elaborated in the Instructions for Authors, and as detailed below.

The Working Group's synthesis reflects the extent of available evidence, summarizing groups of included studies with an emphasis on characterizing consistencies or differences in results within and across experimental designs. Greater emphasis is given to informative mechanistic evidence from human-related studies than to that from other experimental test systems, and gaps are identified. Tabulation of data may facilitate this review. The specific topics addressed in the evidence synthesis are described below.

(a) Absorption, distribution, metabolism, and excretion

Studies of absorption, distribution, metabolism, and excretion in mammalian species are addressed in a summary fashion; exposure characterization is addressed in Part B, Section 1. The Working Group describes the metabolic fate of the agent in mammalian species, noting the metabolites that have been identified and their chemical reactivity. A metabolic schema may indicate the relevant metabolic pathways and products and whether supporting evidence is from studies in humans and/or studies in experimental animals. Evidence on other adverse effects that indirectly confirm absorption, distribution, and/or metabolism at tumour sites is briefly summarized when direct evidence is sparse.

(b) Evidence relevant to key characteristics of carcinogens

A review of Group 1 human carcinogens classified up to and including IARC Monographs Volume 100 revealed several issues relevant to improving the evaluation of mechanistic evidence for cancer hazard identification (Smith et al., 2016). First, it was noted that human carcinogens often share one or more characteristics that are related to the multiple mechanisms by which agents cause cancer. Second, different human carcinogens may exhibit a different spectrum of these key characteristics and operate through distinct mechanisms. Third, for many carcinogens evaluated before Volume 100, few data were available on some mechanisms of recognized importance in carcinogenesis, such as epigenetic alterations (Herceg et al., 2013). Fourth, there was no widely accepted method to search systematically for relevant mechanistic evidence, resulting in a lack of uniformity in the scope of mechanistic topics addressed across IARC Monographs evaluations.

To address these challenges, the key characteristics of human carcinogens were introduced to facilitate systematic consideration of mechanistic evidence in *IARC Monographs* evaluations (Smith et al., 2016; Guyton et al., 2018). The key characteristics described by Smith et al. (2016) (see Table 3), such as "is genotoxic", "is immunosuppressive", or "modulates receptor-mediated effects", are based on empirical observations of the chemical and biological properties associated with the human carcinogens identified by

Table 3 The key characteristics of carcinogens

Ten key characteristics of carcinogens			
1.	Is electrophilic or can be metabolically activated to an electrophile		
2.	Is genotoxic		
3.	Alters DNA repair or causes genomic instability		
4.	Induces epigenetic alterations		
5.	Induces oxidative stress		
6.	Induces chronic inflammation		
7.	Is immunosuppressive		
8.	Modulates receptor-mediated effects		
9.	Causes immortalization		
10.	Alters cell proliferation, cell death, or nutrient supply		

From Smith et al. (2016).

the IARC Monographs programme up to and including Volume 100. The list of key characteristics and associated end-points may evolve, based on the experience of their application and as new human carcinogens are identified. Key characteristics are distinct from the "hallmarks of cancer", which relate to the properties of cancer cells (Hanahan and Weinberg, 2000, 2011). Key characteristics are also distinct from hypothesized mechanistic pathways, which describe a sequence of biological events postulated to occur during carcinogenesis. As such, the evaluation approach based on key characteristics, outlined below, "avoids a narrow focus on specific pathways and hypotheses and provides for a broad, holistic consideration of the mechanistic evidence" (National Academies of Sciences, Engineering, and Medicine, 2017).

Studies in exposed humans and in human primary cells or tissues that incorporate endpoints relevant to key characteristics of carcinogens are emphasized when available. For each key characteristic with adequate evidence for evaluation, studies are grouped according to whether they involve (a) humans or human primary cells or tissues or (b) experimental systems; further organization (as appropriate) is by endpoint (e.g. DNA damage), duration, species, sex, strain, and target organ as well as strength of study design. Studies investigating susceptibility related to key characteristics of carcinogens (e.g. of genetic polymorphisms, or in genetically engineered animals) can be highlighted and may provide additional support for conclusions on the strength of evidence. Findings relevant to a specific tumour type may be noted.

(c) Other relevant evidence

Other informative evidence may be described when it is judged by the Working Group to be relevant to an evaluation of carcinogenicity and to be of sufficient importance to affect the overall evaluation. Quantitative structure-activity information, such as on specific chemical and/or biological features or activities (e.g. electrophilicity, molecular docking with receptors), may be informative. In addition, evidence that falls outside of the recognized key characteristics of carcinogens, reflecting emerging knowledge or important novel scientific developments on carcinogen mechanisms, may also be included. Available evidence relevant to criteria provided in authoritative publications (e.g. Capen et al., 1999; IARC, 2003) on thyroid, kidney, urinary bladder, or other tumours in experimental animals induced by mechanisms that do not operate in humans is also described.

(d) Study quality and importance to the evaluation

Based on formal considerations of the quality of the studies (e.g. design, methodology, and reporting of results), the Working Group may give greater weight to some included studies.

For observational and other studies in humans, the quality of study design, exposure assessment, and assay accuracy and precision are considered, in collaboration with the Working Group members reviewing exposure characterization and studies of cancer in humans, as are other important factors, including those described above for evaluation of epidemiological evidence (García-Closas et al., 2006, 2011; Vermeulen et al., 2018) (Part B, Sections 1 and 2).

In general, in experimental systems, studies of repeated doses and of chronic exposures are accorded greater importance than are studies of a single dose or time-point. Consideration is also given to factors such as the suitability of the dosing range, the extent of concurrent toxicity observed, and the completeness of reporting of the study (e.g. the source and purity of the agent, the analytical methods, and the results). Route of exposure is generally considered to be a less important factor in the evaluation of experimental studies, recognizing that the exposures and target tissues may vary across experimental models and in exposed human populations. Non-mammalian studies can be synthetically summarized when they are considered to be supportive of evidence in humans or higher organisms.

In vitro test systems can provide mechanistic insights, but important considerations include the limitations of the test system (e.g. in metabolic capabilities) as well as the suitability of a particular test article (i.e. because of physical and chemical characteristics) (Hopkins et al., 2004). For studies on some end-points, such as for traditional studies of mutations in bacteria and in mammalian cells, formal guidelines, including those from the Organisation for Economic Co-operation and Development, may be informative in conducting the quality review (OECD, 1997, 2016a, b). However, existing guidelines will not generally cover all relevant assays, even for genotoxicity. Possible considerations when evaluating the quality of in vitro studies encompass the methodology and design (e.g. the end-point and test method, the number of replicate samples, the suitability of the concentration range, the inclusion of positive and negative controls, and the assessment of cytotoxicity) as well as reporting (e.g. of the source and purity of the agent, and of the analytical methods and results). High-content and high-throughput in vitro data can serve as an additional or supportive source of mechanistic evidence (Chiu et al., 2018; Guyton et al., 2018), although large-scale screening programmes measuring a variety of end-points were designed to evaluate large chemical libraries in order to prioritize chemicals for additional toxicity testing rather than to identify the hazard of a specific chemical or chemical group.

The synthesis is focused on the evidence that is most informative for the overall evaluation. In this regard, it is of note that some human carcinogens exhibit a single or primary key characteristic, evidence of which has been influential in their cancer hazard classifications. For instance, ethylene oxide is genotoxic (IARC, 1994), 2,3,7,8-tetrachlorodibenzo-para-dioxin modulates receptor-mediated effects (IARC, 1997), and etoposide alters DNA repair (IARC, 2012a). Similarly, oncogenic viruses cause immortalization, and certain drugs are, by design, immunosuppressive (IARC, 2012a, b). Because non-carcinogens can also induce oxidative stress, this key characteristic should be interpreted with caution unless it is found in combination with other key characteristics (Guyton et al., 2018). Evidence for a group of key characteristics can strengthen mechanistic conclusions (e.g. "induces oxidative stress" together with "is electrophilic or can be metabolically activated to an

electrophile", "induces chronic inflammation", and "is immunosuppressive"); see, for example, 1-bromopropane (<u>IARC, 2018</u>).

5. Summary of data reported

(a) Exposure characterization

Exposure data are summarized to identify the agent and describe its production, use, and occurrence. Information on exposure prevalence and intensity in different settings, including geographical patterns and time trends, may be included. Exposure assessment methods used in key epidemiological studies reviewed by the Working Group are described and evaluated.

(b) Cancer in humans

Results of epidemiological studies pertinent to an evaluation of carcinogenicity in humans are summarized. The overall strengths and limitations of the epidemiological evidence base are highlighted to indicate how the evaluation was reached. The target organ(s) or tissue(s) in which a positive association between the agent and cancer was observed are identified. Exposureresponse and other quantitative data may be summarized when available. When the available epidemiological studies pertain to a mixed exposure, process, occupation, or industry, the Working Group seeks to identify the specific agent considered to be most likely to be responsible for any excess risk. The evaluation is focused as narrowly as the available data permit.

(c) Cancer in experimental animals

Results pertinent to an evaluation of carcinogenicity in experimental animals are summarized to indicate how the evaluation was reached. For each animal species, study design, and route of administration, there is a statement about whether an increased incidence, reduced latency, or increased severity or multiplicity of neoplasms or pre-neoplastic lesions was observed, and the tumour sites are indicated. Special conditions resulting in tumours, such as prenatal exposure or single-dose experiments, are mentioned. Negative findings, inverse relationships, dose– response patterns, and other quantitative data are also summarized.

(d) Mechanistic evidence

Results pertinent to an evaluation of the mechanistic evidence on carcinogenicity are summarized to indicate how the evaluation was reached. The summary encompasses the informative studies on absorption, distribution, metabolism, and excretion; on the key characteristics with adequate evidence for evaluation; and on any other aspects of sufficient importance to affect the overall evaluation, including on whether the agent belongs to a class of agents for which one or more members have been classified as carcinogenic or probably carcinogenic to humans, and on criteria with respect to tumours in experimental animals induced by mechanisms that do not operate in humans. For each topic addressed, the main supporting findings are highlighted from exposed humans, human cells or tissues, experimental animals, or in vitro systems. When mechanistic studies are available in exposed humans, the tumour type or target tissue studied may be specified. Gaps in the evidence are indicated (i.e. if no studies were available in exposed humans, in in vivo systems, etc.). Consistency or differences of effects across different experimental systems are emphasized.

6. Evaluation and rationale

Consensus evaluations of the strength of the evidence of cancer in humans, the evidence of cancer in experimental animals, and the mechanistic evidence are made using transparent criteria and defined descriptive terms. The Working Group then develops a consensus overall evaluation of the strength of the evidence of carcinogenicity for each agent under review.

An evaluation of the strength of the evidence is limited to the agents under review. When multiple agents being evaluated are considered by the Working Group to be sufficiently closely related, they may be grouped together for the purpose of a single and unified evaluation of the strength of the evidence.

The framework for these evaluations, described below, may not encompass all factors relevant to a particular evaluation of carcinogenicity. After considering all relevant scientific findings, the Working Group may exceptionally assign the agent to a different category than a strict application of the framework would indicate, while providing a clear rationale for the overall evaluation.

When there are substantial differences of scientific interpretation among the Working Group members, the overall evaluation will be based on the consensus of the Working Group. A summary of the alternative interpretations may be provided, together with their scientific rationale and an indication of the relative degree of support for each alternative.

The categories of the classification refer to the strength of the evidence that an exposure is carcinogenic and not to the risk of cancer from particular exposures. The terms *probably carcinogenic* and *possibly carcinogenic* have no quantitative significance and are used as descriptors of different strengths of evidence of carcinogenicity in humans; *probably carcinogenic* signifies a greater strength of evidence than *possibly carcinogenic*.

(a) Carcinogenicity in humans

Based on the principles outlined in Part B, Section 2, the evidence relevant to carcinogenicity from studies in humans is classified into one of the following categories:

Sufficient evidence of carcinogenicity: A causal association between exposure to the agent and human cancer has been established. That is, a positive association has been observed in the body of evidence on exposure to the agent and cancer in studies in which chance, bias, and confounding were ruled out with reasonable confidence.

Limited evidence of carcinogenicity: A causal interpretation of the positive association observed in the body of evidence on exposure to the agent and cancer is credible, but chance, bias, or confounding could not be ruled out with reasonable confidence.

Inadequate evidence regarding carcinogenicity: The available studies are of insufficient quality, consistency, or statistical precision to permit a conclusion to be drawn about the presence or the absence of a causal association between exposure and cancer, or no data on cancer in humans are available. Common findings that lead to a determination of inadequate evidence of carcinogenicity include: (a) there are no data available in humans; (b) there are data available in humans, but they are of poor quality or informativeness; and (c) there are studies of sufficient quality available in humans, but their results are inconsistent or otherwise inconclusive.

Evidence suggesting lack of carcinogenicity: There are several high-quality studies covering the full range of levels of exposure that humans are known to encounter, which are mutually consistent in not showing a positive association between exposure to the agent and the studied cancers at any observed level of exposure. The results from these studies alone or combined should have narrow confidence intervals with an upper limit below or close to the null value (e.g. a relative risk of unity). Bias and confounding were ruled out with reasonable confidence, and the studies were considered informative. A conclusion of *evidence suggesting lack of carcinogenicity* is limited to the cancer sites, populations and life stages, conditions and levels of exposure, and length of observation covered by the available studies. In addition, the possibility of a very small risk at the levels of exposure studied can never be excluded.

When there is *sufficient evidence*, a separate sentence identifies the target organ(s) or tissue(s) for which a causal interpretation has been established. When there is limited evidence, a separate sentence identifies the target organ(s) or tissue(s) for which a positive association between exposure to the agent and the cancer(s) was observed in humans. When there is evidence suggesting lack of carcinogenicity, a separate sentence identifies the target organ(s) or tissue(s) where evidence of lack of carcinogenicity was observed in humans. Identification of a specific target organ or tissue as having sufficient evidence or limited evidence or evidence suggesting lack of carcinogenicity does not preclude the possibility that the agent may cause cancer at other sites.

(b) Carcinogenicity in experimental animals

The evidence relevant to carcinogenicity from studies in experimental animals is classified into one of the following categories:

Sufficient evidence of carcinogenicity: A causal relationship has been established between exposure to the agent and cancer in experimental animals based on an increased incidence of malignant neoplasms

or of an appropriate combination of benign and malignant neoplasms in (a) two or more species of animals or (b) two or more independent studies in one species carried out at different times or in different laboratories and/or under different protocols. An increased incidence of malignant neoplasms or of an appropriate combination of benign and malignant neoplasms in both sexes of a single species in a well-conducted study, ideally conducted under Good Laboratory Practices (GLP), can also provide *sufficient evidence*.

Exceptionally, a single study in one species and sex may be considered to provide *sufficient evidence of carcinogenicity* when malignant neoplasms occur to an unusual degree with regard to incidence, site, type of tumour, or age at onset, or when there are marked findings of tumours at multiple sites.

Limited evidence of carcinogenicity: The data suggest a carcinogenic effect but are limited for making a definitive evaluation because, for example, (a) the evidence of carcinogenicity is restricted to a single experiment and does not meet the criteria for sufficient evidence; (b) the agent increases the incidence only of benign neoplasms or lesions of uncertain neoplastic potential; (c) the agent increases tumour multiplicity or decreases tumour latency but does not increase tumour incidence; (d) the evidence of carcinogenicity is restricted to initiation-promotion studies; (e) the evidence of carcinogenicity is restricted to observational studies in non-laboratory animals (e.g. companion animals); or (f) there are unresolved questions about the adequacy of the design, conduct, or interpretation of the available studies.

Inadequate evidence regarding carcinogenicity: The studies cannot be interpreted as showing either the presence or the absence of a carcinogenic effect because of major qualitative or quantitative limitations, or no data are available on cancer in experimental animals.

Evidence suggesting lack of carcinogenicity: Well-conducted studies (e.g. conducted under GLP) involving both sexes of at least two species are available showing that, within the limits of the tests used, the agent was not carcinogenic. The conclusion of *evidence suggesting lack of carcinogenicity* is limited to the species, tumour sites, age at exposure, and conditions and levels of exposure covered by the available studies.

(c) Mechanistic evidence

Based on the principles outlined in Part B, Section 4, the mechanistic evidence is classified into one of the following categories:

Strong mechanistic evidence: Results in several different experimental systems are consistent, and the overall mechanistic database is coherent. Further support can be provided by studies that demonstrate experimentally that the suppression of key mechanistic processes leads to the suppression of tumour development. Typically, a substantial number of studies on a range of relevant end-points are available in one or more mammalian species. Quantitative structure-activity considerations, in vitro tests in non-human mammalian cells, and experiments in non-mammalian species may provide corroborating evidence but typically do not in themselves provide strong evidence. However, consistent findings across a number of different test systems in different species may provide strong evidence.

Of note, "strong" relates not to potency but to strength of evidence. The classification applies to three distinct topics: (a) Strong evidence that the agent belongs, based on mechanistic considerations, to a class of agents for which one or more members have been classified as carcinogenic or probably carcinogenic to humans. The considerations can go beyond quantitative structure-activity relationships to incorporate similarities in biological activity relevant to common key characteristics across dissimilar chemicals (e.g. based on molecular docking, –omics data).

(b) Strong evidence that the agent exhibits key characteristics of carcinogens. In this case, three descriptors are possible:

- 1. The strong evidence is in exposed humans. Findings relevant to a specific tumour type may be informative in this determination.
- 2. The strong evidence is in human primary cells or tissues. Specifically, the strong findings are from biological specimens obtained from humans (e.g. ex vivo exposure), from human primary cells, and/or, in some cases, from other humanized systems (e.g. a human receptor or enzyme).
- 3. The strong evidence is in experimental systems. This may include one or a few studies in human primary cells and tissues.

(c) Strong evidence that the mechanism of carcinogenicity in experimental animals does not operate in humans. Certain results in experimental animals (see Part B, Section 6b) would be discounted, according to relevant criteria and considerations in authoritative publications (e.g. Capen et al., 1999; IARC, 2003). Typically, this classification would not apply when there is strong mechanistic evidence that the agent exhibits key characteristics of carcinogens.

Limited mechanistic evidence: The evidence is suggestive, but, for example, (a) the studies cover a narrow range of experiments, relevant end-points, and/or species; (b) there are unexplained inconsistencies in the studies of similar design; and/or (c) there is unexplained incoherence across studies of different endpoints or in different experimental systems.

Inadequate mechanistic evidence: Common findings that lead to a determination of inadequate mechanistic evidence include: (a) few or no data are available; (b) there are unresolved questions about the adequacy of the design, conduct, or interpretation of the studies; (c) the available results are negative.

(d) Overall evaluation

Finally, the bodies of evidence included within each stream of evidence are considered as a whole, in order to reach an overall evaluation of the carcinogenicity of the agent to humans. The three streams of evidence are integrated and the agent is classified into one of the following categories (see <u>Table 4</u>), indicating that the Working Group has established that:

The agent is carcinogenic to humans (Group 1)

This category applies whenever there is *sufficient evidence of carcinogenicity* in humans.

In addition, this category may apply when there is both *strong evidence in exposed humans that the agent exhibits key characteristics of carcinogens* and *sufficient evidence of carcinogenicity* in experimental animals.

The agent is probably carcinogenic to humans (Group 2A)

This category generally applies when the Working Group has made at least *two of the following* evaluations, *including at least one* that

involves either exposed humans or human cells or tissues:

- *Limited evidence of carcinogenicity* in humans,
- *Sufficient evidence of carcinogenicity* in experimental animals,
- Strong evidence that the agent exhibits key characteristics of carcinogens.

If there is *inadequate evidence regarding carcinogenicity* in humans, there should be *strong evidence in human cells or tissues that the agent exhibits key characteristics of carcinogens*. If there is *limited evidence of carcinogenicity in humans*, then the second individual evaluation may be from experimental systems (i.e. *sufficient evidence of carcinogenicity* in experimental animals or *strong evidence in experimental systems that the agent exhibits key characteristics of carcinogens*).

Additional considerations apply when there is strong evidence that the mechanism of carcinogenicity in experimental animals does not operate in humans for one or more tumour sites. Specifically, the remaining tumour sites should still support an evaluation of sufficient evidence in experimental animals in order for this evaluation to be used to support an overall classification in Group 2A.

Separately, this category generally applies if there is strong evidence that the agent belongs, based on mechanistic considerations, to a class of agents for which one or more members have been classified in Group 1 or Group 2A.

The agent is possibly carcinogenic to humans (Group 2B)

This category generally applies when only one of the following evaluations has been made by the Working Group:

- *Limited evidence of carcinogenicity* in humans,
- *Sufficient evidence of carcinogenicity* in experimental animals,

Table 4 Integration of streams of evidence in reaching overall classifications (the evidence in
<i>bold italic</i> represents the basis of the overall evaluation)

	Classification based on			
Evidence of cancer in humans ^a	Evidence of cancer in experimental animals	Mechanistic evidence	strength of evidence	
Sufficient	Not necessary	Not necessary	Carcinogenic to humans	
Limited or Inadequate	Sufficient	Strong (b)(1) (exposed humans)	(Group 1)	
Limited	Sufficient	Strong (b)(2-3), Limited, or Inadequate	Probably carcinogenic to	
Inadequate	Sufficient	Strong (b)(2) (human cells or tissues)	humans (Group 2A)	
Limited	Less than Sufficient	<i>Strong (b)(1–3)</i>		
Limited or Inadequate	Not necessary	Strong (a) (mechanistic class)		
Limited	Less than Sufficient	Limited or Inadequate	Possibly carcinogenic to	
Inadequate	Sufficient	Strong (b)(3), Limited, or Inadequate	humans (Group 2B)	
Inadequate	Less than Sufficient	<i>Strong (b)(1–3)</i>		
Limited	Sufficient	Strong (c) (does not operate in humans) ^b		
Inadequate	Sufficient	Strong (c) (does not operate in humans) ^b	Not classifiable as to its	
	carcinogenicity to humans (Group 3)			

^a Human cancer(s) with highest evaluation.

^b The strong evidence that the mechanism of carcinogenicity in experimental animals does not operate in humans must specifically be for the tumour sites supporting the classification of sufficient evidence in experimental animals.

• Strong evidence that the agent exhibits key characteristics of carcinogens.

Because this category can be based on evidence from studies in experimental animals alone, there is **no** requirement that the strong mechanistic evidence be in exposed humans or in human cells or tissues. This category may be based on *strong evidence in experimental systems that the agent exhibits key characteristics of carcinogens.*

As with Group 2A, additional considerations apply when there is *strong evidence that the mechanism of carcinogenicity in experimental animals does not operate in humans* for one or more tumour sites. Specifically, the remaining tumour sites should still support an evaluation of *sufficient evidence in experimental animals* in order for this evaluation to be used to support an overall classification in Group 2B.

The agent is not classifiable as to its carcinogenicity to humans (Group 3)

Agents that do not fall into any other group are generally placed in this category.

This includes the case when there is strong evidence that the mechanism of carcinogenicity in experimental animals does not operate in humans for one or more tumour sites in experimental animals, the remaining tumour sites do not support an evaluation of sufficient evidence in experimental animals, and other categories are not supported by data from studies in humans and mechanistic studies.

An evaluation in Group 3 is not a determination of non-carcinogenicity or overall safety. It often means that the agent is of unknown carcinogenic potential and that there are significant gaps in research.

If the evidence suggests that the agent exhibits no carcinogenic activity, either through *evidence suggesting lack of carcinogenicity* in both humans and experimental animals, or through *evidence suggesting lack of carcinogenicity* in experimental animals complemented by strong negative mechanistic evidence in assays relevant to human cancer, then the Working Group may add a sentence to the evaluation to characterize the agent as well-studied and without evidence of carcinogenic activity.

(e) Rationale

The reasoning that the Working Group used to reach its evaluation is summarized so that the basis for the evaluation offered is transparent. This section integrates the major findings from studies of cancer in humans, cancer in experimental animals, and mechanistic evidence. It includes concise statements of the principal line(s) of argument that emerged in the deliberations of the Working Group, the conclusions of the Working Group on the strength of the evidence for each stream of evidence, an indication of the body of evidence that was pivotal to these conclusions, and an explanation of the reasoning of the Working Group in making its evaluation.

References

- Alexandrov LB, Ju YS, Haase K, Van Loo P, Martincorena I, Nik-Zainal S, et al. (2016). Mutational signatures associated with tobacco smoking in human cancer. *Science*. 354(6312):618–22. doi:<u>10.1126/science.aag0299</u> PMID:<u>27811275</u>
- Axelson O, Steenland K (1988). Indirect methods of assessing the effects of tobacco use in occupational studies. Am J Ind Med. 13(1):105–18. doi:10.1002/ ajim.4700130107 PMID:3344750
- Baan RA, Stewart BW, Straif K, editors (2019). Tumour site concordance and mechanisms of carcinogenesis (IARC Scientific Publication No. 165). Lyon, France: International Agency for Research on Cancer. Available from: https://publications.iarc.who.int/578.
- Bieler GS, Williams RL (1993). Ratio estimates, the delta method, and quantal response tests for increased carcinogenicity. *Biometrics*. 49(3):793–801. doi:10.2307/2532200 PMID:8241374
- Capen CC, Dybing E, Rice JM, Wilbourn JD, editors (1999). Species differences in thyroid, kidney and urinary bladder carcinogenesis (IARC Scientific Publication No. 147). Lyon, France: IARC Press.

Available from: <u>https://publications.iarc.who.int/302</u> PMID:<u>10627184</u>

- Chiu WA, Guyton KZ, Martin MT, Reif DM, Rusyn I (2018). Use of high-throughput in vitro toxicity screening data in cancer hazard evaluations by IARC Monograph Working Groups. *ALTEX*. 35(1):51–64. doi:10.14573/altex.1703231 PMID:28738424
- Cogliano V, Baan R, Straif K, Grosse Y, Secretan B, El Ghissassi F, et al. (2005). Transparency in IARC Monographs. *Lancet Oncol.* 6(10):747 doi:10.1016/ <u>S1470-2045(05)70380-6</u>
- Cogliano VJ, Baan RA, Straif K, Grosse Y, Secretan MB, El Ghissassi F, et al. (2004). The science and practice of carcinogen identification and evaluation. *Environ Health Perspect*. 112(13):1269–74. doi:<u>10.1289/ehp.6950</u> PMID:<u>15345338</u>
- Dunson DB, Chen Z, Harry J (2003). A Bayesian approach for joint modeling of cluster size and subunit-specific outcomes. *Biometrics*. 59(3):521–30. doi:<u>10.1111/1541-</u> 0420.00062 PMID:<u>14601753</u>
- Fung KY, Krewski D, Smythe RT (1996). A comparison of tests for trend with historical controls in carcinogen bioassay. *Can J Stat.* 24(4):431–54. doi:10.2307/3315326
- García-Closas M, Vermeulen R, Cox D, Lan Q, Caporaso N, Hunter D, et al. (2011). Population-based study designs in molecular epidemiology. In: Rothman N, Hainaut P, Schulte P, Smith M, Boffetta P, Perera FP, editors. Molecular epidemiology: principles and practices. Lyon, France: International Agency for Research on Cancer; pp. 241–59.
- García-Closas M, Vermeulen R, Sherman ME, Moore LE, Smith MT, Rothman N (2006). Application of biomarkers in cancer epidemiology. In: Schottenfeld D, Fraumeni JF Jr, editors. Cancer epidemiology and prevention. 3rd ed. New York (NY), USA: Oxford University Press; pp. 70–88. doi:10.1093/acprof: oso/9780195149616.003.0006
- Gart JJ, Krewski D, Lee PN, Tarone RE, Wahrendorf J (1986). Statistical methods in cancer research, Volume III: The design and analysis of long-term animal experiments (IARC Scientific Publication No. 79). Lyon, France: International Agency for Research on Cancer.
- Greenland S, O'Rourke K (2008). Meta-analysis. In: Rothman KJ, Greenland S, Lash TL, editors. Modern epidemiology. 3rd ed. Philadelphia (PA), USA: Wolters Kluwer Health/Lippincott Williams & Wilkins; pp. 652–682.
- Greim H, Gelbke H-P, Reuter U, Thielmann HW, Edler L (2003). Evaluation of historical control data in carcinogenicity studies. *Hum Exp Toxicol*. 22(10):541–9. doi:<u>10.1191/0960327103ht3940a</u> PMID:<u>14655720</u>
- Guyton KZ, Kyle AD, Aubrecht J, Cogliano VJ, Eastmond DA, Jackson M, et al. (2009). Improving prediction of chemical carcinogenicity by considering multiple mechanisms and applying toxicogenomic approaches. *Mutat*

Res. 681(2–3):230–40. doi:<u>10.1016/j.mrrev.2008.10.001</u> PMID:<u>19010444</u>

- Guyton KZ, Rusyn I, Chiu WA, Corpet DE, van den Berg M, Ross MK, et al. (2018). Application of the key characteristics of carcinogens in cancer hazard identification. *Carcinogenesis*. 39(4):614–22. doi:<u>10.1093/carcin/ bgy031</u> PMID:<u>29562322</u>
- Hanahan D, Weinberg RA (2000). The hallmarks of cancer. *Cell*. 100(1):57–70. doi:<u>10.1016/S0092-8674(00)81683-9</u> PMID:<u>10647931</u>
- Hanahan D, Weinberg RA (2011). Hallmarks of cancer: the next generation. *Cell*. 144(5):646–74. doi:<u>10.1016/j.</u> <u>cell.2011.02.013</u> PMID:<u>21376230</u>
- Haseman JK, Huff J, Boorman GA (1984). Use of historical control data in carcinogenicity studies in rodents. *Toxicol Pathol.* 12(2):126–35. doi:10.1177/019262338401200203 PMID:11478313
- Herceg Z, Lambert MP, van Veldhoven K, Demetriou C, Vineis P, Smith MT, et al. (2013). Towards incorporating epigenetic mechanisms into carcinogen identification and evaluation. *Carcinogenesis*. 34(9):1955–67. doi:10.1093/carcin/bgt212 PMID:23749751
- Hernán MA, Hernández-Díaz S, Robins JM (2004). A structural approach to selection bias. *Epidemiology*. 15(5):615–25. doi:<u>10.1097/01.ede.0000135174.63482.43</u> PMID:<u>15308962</u>
- Hill AB (1965). The environment and disease: association or causation? *Proc R Soc Med.* 58:295–300. PMID:14283879
- Hopkins AL, Groom CR, Alex A (2004). Ligand efficiency: a useful metric for lead selection. *Drug Discov Today*. 9(10):430–1. doi:10.1016/S1359-6446(04)03069-7 PMID:15109945
- Huff JE, Eustis SL, Haseman JK (1989). Occurrence and relevance of chemically induced benign neoplasms in long-term carcinogenicity studies. *Cancer Metastasis Rev.* 8(1):1–22. doi:10.1007/BF00047055 PMID:2667783
- IARC (1994). Some industrial chemicals. *IARC Monogr Eval Carcinog Risks Hum*. 60:1–560. Available from: <u>https://publications.iarc.who.int/78</u> PMID:<u>7869568</u>
- IARC (1997). Polychlorinated dibenzo-para-dioxins and polychlorinated dibenzofurans. IARC Monogr Eval Carcinog Risks Hum. 69:1–631. Available from: <u>https://</u> <u>publications.iarc.who.int/87</u> PMID:9379504
- IARC (2003). Predictive value of rodent forestomach and gastric neuroendocrine tumours in evaluating carcinogenic risks to humans (IARC Technical Publication No. 39). Lyon, France: IARC Press. Available from: <u>https://</u> <u>publications.iarc.who.int/362</u>.
- IARC (2004). Some drinking-water disinfectants and contaminants, including arsenic. IARC Monogr Eval Carcinog Risks Hum. 84:1–477. Available from: <u>https:// publications.iarc.who.int/102</u> PMID:<u>15645577</u>
- IARC (2012a). Pharmaceuticals. IARC Monogr Eval Carcinog Risks Hum. 100A:1–437. Available from: <u>https://publications.iarc.who.int/118</u> PMID:23189749

- IARC (2012b). Biological agents. *IARC Monogr Eval Carcinog Risks Hum.* 100B:1–441. Available from: https://publications.iarc.who.int/119 PMID:23189750
- IARC (2014). Report of the IARC Advisory Group to Recommend on Quantitative Risk Characterization (IARC Internal Report 14/001). Available from: <u>https://monographs.iarc.who.int/wp-content/</u><u>uploads/2018/06/14-001.pdf</u>.
- IARC (2018). Some industrial chemicals. *IARC Monogr Eval Carcinog Risks Hum*. 115:1–292. Available from: https://publications.iarc.who.int/563.
- National Academies of Sciences, Engineering, and Medicine (2017). Using 21st century science to improve risk-related evaluations. Washington (DC), USA: National Academies Press. doi:<u>10.17226/24635</u> PMID:<u>28267305</u>
- National Research Council (2012). Exposure science in the 21st century: a vision and a strategy. Washington (DC), USA: National Academies Press. doi:<u>10.17226/13507</u> PMID:<u>24901193</u>
- OECD (1997). Test no. 471: Bacterial reverse mutation test, OECD Guidelines for the Testing of Chemicals, Section 4. Paris, France: Organisation for Economic Co-operation and Development. doi:10.1787/9789264071247-en
- OECD (2016a). Test no. 476: In vitro mammalian cell gene mutation tests using the *Hprt* and *xprt* genes, OECD Guidelines for the Testing of Chemicals, Section 4. Paris, France: Organisation for Economic Co-operation and Development. doi:<u>10.1787/9789264264809-en</u>
- OECD (2016b). Test no. 490: In vitro mammalian cell gene mutation tests using the thymidine kinase gene, OECD Guidelines for the Testing of Chemicals, Section 4. Paris, France: Organisation for Economic Co-operation and Development. doi:10.1787/9789264264908-en
- OECD (2018). Test no. 451: Carcinogenicity studies, OECD Guidelines for the Testing of Chemicals, Section 4: Health effects. Paris, France: Organisation for Economic Co-operation and Development. doi:10.1787/9789264071186-en
- Parkkinen VP, Wallmann C, Wilde M, Clarke B, Illari P, Kelly MP, et al. (2018). Evaluating evidence of mechanisms in medicine. Principles and procedures. Switzerland: Springer Cham. doi:10.1007/978-3-319-94610-8
- Peto R, Pike MC, Day NE, Gray RG, Lee PN, Parish S, et al. (1980). Guidelines for simple, sensitive significance tests for carcinogenic effects in long-term animal experiments. In: Long-term and short-term screening assays for carcinogens: a critical appraisal. *IARC Monogr Eval Carcinog Risk Chem Hum*. Suppl. 2. Lyon, France: International Agency for Research on Cancer; pp. 311–426. Available from: https://publications.iarc. who.int/134 PMID:6935176

- Portier CJ, Bailer AJ (1989). Testing for increased carcinogenicity using a survival-adjusted quantal response test. *Fundam Appl Toxicol*. 12(4):731–7. doi:<u>10.1016/0272-0590(89)90004-3</u> PMID:<u>2744275</u>
- Richardson DB, Laurier D, Schubauer-Berigan MK, Tchetgen Tchetgen E, Cole SR (2014). Assessment and indirect adjustment for confounding by smoking in cohort studies using relative hazards models. *Am J Epidemiol.* 180(9):933–40. doi:10.1093/aje/kwu211 PMID:25245043
- Rothman KJ, Greenland S, Poole C, Lash TL (2008). Causation and causal inference. In: Rothman KJ, Greenland S, Lash TL, editors. Modern epidemiology. 3rd ed. Philadelphia (PA), USA: Wolters Kluwer Health/Lippincott Williams & Wilkins; pp. 5–31.
- Sherman CD, Portier CJ, Kopp-Schneider A (1994). Multistage models of carcinogenesis: an approximation for the size and number distribution of late-stage clones. *Risk Anal.* 14(6):1039–48. doi:<u>10.1111/j.1539-6924.1994.</u> <u>tb00074.x</u> PMID:<u>7846311</u>
- Smith MT, Guyton KZ, Gibbons CF, Fritz JM, Portier CJ, Rusyn I, et al. (2016). Key characteristics of carcinogens as a basis for organizing data on mechanisms of carcinogenesis. *Environ Health Perspect*. 124(6):713–21. doi:10.1289/ehp.1509912 PMID:26600562

- Stayner L, Steenland K, Dosemeci M, Hertz-Picciotto I (2003). Attenuation of exposure-response curves in occupational cohort studies at high exposure levels. *Scand J Work Environ Health*. 29(4):317–24. doi:10.5271/ sjweh.737 PMID:12934726
- Vandenbroucke JP, Broadbent A, Pearce N (2016). Causality and causal inference in epidemiology: the need for a pluralistic approach. *Int J Epidemiol*. 45(6):1776– 86. doi:10.1093/ije/dyv341 PMID:26800751
- Vermeulen R, Bell DA, Jones DP, Garcia-Closas M, Spira A, Wang TW, et al. (2018). Application of biomarkers in cancer epidemiology. In: Thun MJ, Linet MS, Cerhan JR, Haiman CA, Schottenfeld D, editors. Schottenfeld and Fraumeni cancer epidemiology and prevention. 4th ed. New York (NY), USA: Oxford University Press; pp. 77–96.
- Wild CP, Cogliano VJ (2011). A plea on behalf of expert evaluation and the experts involved. *Int J Epidemiol*. 40(1):253–61. doi:10.1093/ije/dyq038 PMID:20223796

GENERAL REMARKS

This one-hundred-and-thirty-fifth volume of the *IARC Monographs* contains evaluations of the carcinogenic hazard to humans of perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS), and their corresponding isomers and salts.

PFOA was considered previously by the *IARC Monographs* programme in 2014 (<u>IARC</u>, 2016), when it was evaluated as *possibly carcinogenic to humans* (Group 2B). PFOS has not been evaluated previously by the *IARC Monographs* programme.

The Advisory Group to Recommend Priorities for the *IARC Monographs* that met in 2019 recommended that some perfluorinated compounds, such as PFOA, be evaluated with high priority (<u>IARC</u>, 2019a; <u>Marques et al.</u>, 2019), largely on the basis of emerging evidence of carcinogenicity in experimental animals and of mechanistic evidence related to the key characteristics of carcinogens (KCs). A summary of the findings of this volume appears in *The Lancet Oncology* (Zahm et al., 2023).

PFOA and PFOS in the context of the broader class of PFAS

PFOA and PFOS are part of a class of thousands of synthetic per- and polyfluoroalkyl substances (PFAS) that are used widely throughout the world. The Working Group noted that the carbon-fluorine bond is one of the strongest in organic chemistry and is responsible for the environmental and biological persistence, long-range environmental transport, as well as bioaccumulation and biomagnification of this class of chemicals. In the present monograph, the Working Group assessed the carcinogenic hazard of only two PFAS, the uses of both of which have restricted or eliminated under the Stockholm Convention to which more than 180 countries are parties (UNEP, 2023). Information is limited regarding exposure to precursors of PFOA and PFOS and to PFAS used to replace PFOA and PFOS, and few studies in humans have examined these substances as the primary exposure metric when evaluating health outcomes.

Lack of comprehensive exposure data for PFOA and PFOS

The Working Group noted major gaps in the existing literature that hampered the understanding of PFOA and PFOS exposure worldwide. When stratified by location or exposure source, including country (e.g. within the USA and Europe versus outside, as well as in communities with a known source of contamination versus those without), this gap in knowledge was exacerbated by the absence of surveillance initiatives. Although many countries have phased-out the production and/or use of PFOA or PFOS, the Working Group identified studies indicating that certain precursor PFAS are known to break down or transform into PFOA and PFOS in the environment and biological systems, including in humans. This suggests that ongoing exposure may be expected, even if production and use of PFOA and PFOS compounds were to cease entirely around the world.

Although the workplace is often the source of highest exposure to PFAS, characterization of occupational exposure to PFOA and PFOS was limited to only a few occupations. The majority of studies focused on biomonitoring of fluorochemical-production workers (including perfluoroalkyl polymer-production workers) and first responders (especially firefighters), and other occupations that produce, use, or dispose of products that have been treated with or contain PFAS have been examined to a lesser degree (if at all). Female workers are largely absent in the available literature, limiting potential epidemiological analyses of occupational exposure sources among women. Additionally, the relative contribution of different exposure routes, namely dermal absorption versus inhalation, in these settings is poorly understood.

New evidence on cancer in humans published since the previous *IARC Monographs* evaluation

When PFOA was evaluated by the *IARC Monographs* programme in 2014, the epidemiological evidence consisted of studies on three occupationally exposed populations, one population exposed to drinking-water that was highly contaminated via a nearby industrial facility, and three case-control studies of members of the general population in communities without a PFOA pollution point source (this ambient exposure is referred to in the present monograph as "background" exposure). The present evaluations of PFOA and PFOS are based on 36 epidemiological studies, including further reports on the same three occupationally exposed populations, two additional populations in highly contaminated areas, and many case-control studies in the general population. The latter included nested case-control studies using prospectively collected biospecimens and less-informative, non-nested case-control studies using biospecimens collected after diagnosis of cancer and, in some instances, after treatment for cancer. Ecological studies, with the exception of one with an extremely high contrast in environmental exposure to PFOA relative to exposure to other PFAS, were excluded from the review. Despite these additional studies, data gaps and limitations remain, including low exposure contrasts in the studies of "background" exposure, potential healthy-worker survivor bias in most of the occupational studies, and, in case-control studies, uncertainties surrounding the measurement of PFOA and PFOS after diagnosis and, possibly, treatment for cancer. Additionally, there were few studies that addressed cancer subtypes defined by histology, genotype, receptor status, and other characteristics. Another data gap was the lack of studies among additional populations known to have occupational or substantial environmental exposure, such as workers in fluorochemical production or residents in communities with substantial pollution, e.g. in Italy, France, or Australia. Such studies might help address the data gaps noted above related to the carcinogenicity of PFOS and to the specific cancer types linked to PFOA and PFOS exposure in populations with high exposure contrast.

One challenge in the epidemiological literature is the difficulty in evaluating the effects of individual PFAS compounds, because there is widespread co-exposure to many highly correlated PFAS compounds. The evaluation of the cancer hazard resulting from exposure to mixtures of PFAS compounds, although important, was beyond the scope of the present volume and may require the development of new statistical analytical approaches.

The Working Group conducted three new analyses of existing epidemiological data, which assisted in their evaluation (see Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans): (i) an analysis based on summary statistics for repeated serological measurements of PFOA that were available from subsets of participants in two nested case-control studies, which were used to evaluate the representativeness of serum PFOA measurements from a single time point as a surrogate for longer term levels; (ii) a meta-analysis of PFOA exposure and kidney cancer; and (iii) an ecological analysis of the correlation between serum concentration measurements of PFOA and the rates of orchiectomies (a strong correlate of testicular cancer incidence in this region) within 21 municipalities in the Veneto region of Italy where there had been industrial contamination of drinking-water with PFOA.

Extensive mechanistic evidence

Since the previous *IARC Monographs* evaluation of PFOA in 2014, by far the greatest increase in the amount of research available has occurred with respect to toxicokinetic data and mechanistic evidence, including data relevant to the KCs (see Section 4). Particularly noteworthy is the extent of evidence related to epigenetic alterations (Section 4.2.4) and immunosuppression (Section 4.2.7) in exposed humans. The Working Group noted that there are only a few agents evaluated by the *IARC Monographs* programme (e.g. occupational exposure as a firefighter) for which there are such extensive data from multiple studies in multiple populations supporting these KCs. These data, combined with the data from cancer bioassays in experimental animals, underpin the rationale for the evaluation of PFOA as *carcinogenic to humans*, Group 1. Moreover, this is the first time that mechanistic evidence from a variety of test systems specifically for these two KCs has supported a Group 1 evaluation, particularly in the absence of strong evidence in exposed humans for either genotoxicity (KC2) or modulation of receptor-mediated effects (KC8).

It should also be noted that the contribution of mechanistic evidence in exposed humans to a Group 1 evaluation does not require a PFOAspecific mechanism of carcinogenicity to be identified. Thus, although empirical data directly linking PFOA-specific effects on the epigenome and immune system to increased cancer risk in humans were not available, it was the judgement of the Working Group that the observed effects in exposed humans, supported by evidence in human primary cells and in experimental systems, were sufficiently linked to carcinogenic processes to support a Group 1 evaluation, in combination with the positive results in cancer bioassays in animals.

Challenges in using PFOA and PFOS to define a mechanistic class of carcinogens

For the present volume, the Working Group identified overall similar mechanistic evidence for PFOA and PFOS across the KCs on the basis of data obtained in exposed humans, in human primary cells, and in experimental systems; as reported above, this included consistent and coherent mechanistic evidence for the KCs "induces epigenetic alterations" (KC4) and "is immunosuppressive" (KC7). In addition, it was reported that both agents have long half-lives in humans and both bind to multiple relevant protein targets, including nuclear receptors, membrane transporters, and carrier proteins. However, it remained unclear whether PFOA might represent a mechanistic class of carcinogens to which PFOS (or other PFAS compounds) may belong. Despite a rich mechanistic database (as reviewed in Section 4 of the present monograph), the Working Group could not identify a common specific mechanism by which exposure to PFOA and PFOS leads to carcinogenesis; it is possible or even likely that multiple mechanisms are in play.

It is worth noting that additional data streams that could be helpful in establishing a chemical class, including studies on the non-cancerrelated toxicity of PFOA and PFOS independent of the KCs, relative potency considerations, and mixture effects, were beyond the scope of an IARC Monographs evaluation. Although both PFOA and PFOS appear to activate a similar suite of nuclear receptors in human primary cells and experimental systems in vivo (e.g. there was consistent and coherent evidence for activation of peroxisome proliferator-activated receptor alpha (PPARa), constitutive androstane receptor (CAR), and pregnane X receptor (PXR) and suggestive evidence for activation of PPARy in human primary cells), the degree of activation differs between the receptors, potentially influencing the strength of receptor-driven carcinogenic effects. Additionally, technical challenges hinder the ascertainment of whether there is modulation of these receptor pathways in exposed humans, since accessible and specific biomarkers of these pathways are not readily available in humans. This data gap is compounded by the fact that there are species differences in the events associated with modulation of these pathways by PFOA and PFOS. As yet, no studies have been conducted in mice expressing human genes for these receptors.

Overall, the Working Group was not able to conclude whether PFOA could represent a mechanistic class to which PFOS (or other PFAS) belong, based on considerations described in the Preamble to the *IARC Monographs* (see present volume; <u>IARC, 2019b</u>).

Relevance to humans of PFOA and PFOS effects on altered lipid metabolism in rodents

The Working Group noted that nuclear receptor activation and deregulation of lipid metabolism are relevant KC-related end-points and mechanisms that might contribute to the hepatocarcinogenicity of PFOA and PFOS in rodents.

The activation of hepatic PPARα and CAR/ PXR in rodents has been reported to: transiently increase the activity of liver enzymes such as acyl coenzyme A (CoA) oxidase, and cytochrome P450s CYP4A, CYP2B, and CYP3A; increase the liver proliferative index and decrease the liver apoptotic index; decrease the frequency of hepatocellular glycogen-induced vacuoles; increase the frequency of centrilobular hepatocellular hypertrophy (Elcombe et al., 2012a, b), but also cause alterations in plasma cholesterol level, and increase centrilobular hepatocellular hypertrophy. Also, induction of hepatic steatosis has been observed in mice after dietary exposure to PFOS (Bagley et al., 2017).

The molecular mechanisms by which PFOA or PFOS can cause hepatotoxicity (e.g. fatty liver disease and other hepatotoxic effects) have not been fully described either in experimental animals or in humans. However, an accumulation of fatty acids and triglycerides and deregulation of the expression of genes related to the metabolism of fatty acids and triglycerides has been reported in a series of in vitro studies (as well as several epidemiological studies) (e.g. <u>Wan et al.</u>, <u>2012; Louisse et al.</u>, 2020). These and other effects independent of the PPARα receptor that cause deregulation of gene expression, resulting in a substantial shift from carbohydrate metabolism to fatty acid oxidation and hepatic triglyceride accumulation, have been also observed in human and rat primary liver cells (<u>Vanden Heuvel et al.,</u> <u>2006; Bjork et al., 2011; Das et al., 2017; Rosen et al., 2017; Behr et al., 2018</u>).

Other mechanisms that could be related to PFOA/PFOS-induced hepatocarcinogenicity through alterations in lipid metabolism, identified in studies in human hepatoma cells, include the activation of specific endoplasmic reticulum stress (ERS)-response genes (e.g. ATF4, DDIT3, ATF3) and enzymes involved in lipid metabolism, e.g. cholesterol (HMGCR), upon PFOS exposure (Louisse et al., 2023). After exposure to PFOA, activation of the unfolded protein response (UPR) pathway, induction of steatosis and fibrosis and expression of TNFa and IL6 inflammatory markers, increased production of endogenous reactive oxygen species in liver cells (Qi et al., 2023), and deregulation of the genes controlling lipid homeostasis (Das et al., 2017) were observed. In addition, ERS/UPR stress was also induced by PFOA in pancreatic acinar cells (Hocevar et al., 2020); and induction of cell proliferation and migration and invasion upon exposure to PFOA or PFOS were reported for various in vitro models (Matkowskyj et al., 2014; Pierozan et al., 2020; Hu et al., 2022).

Consistent with a potential involvement of both PFOA and PFOS in metabolic alterations, the Working Group identified data from metabolomic analyses in exposed humans suggesting increased activities of glycolytic pathways. Transcriptomic analyses have also indicated alterations in cell proliferation and in lipid metabolism pathways in human primary and experimental systems, respectively (see Section 4.2.11).

Data gaps for PFOS and other PFAS

The evaluation of the carcinogenicity of PFOS was hampered by a relative paucity of studies of cancer in humans and also by the existence of only one study in experimental animals that complied with Good Laboratory Practice and gave positive results. However, the strength of the mechanistic evidence for PFOS, together with its relatively potent toxic effects, suggest that additional carcinogenicity studies may fill this data gap.

Many "novel" or emerging PFAS are currently used, but the toxic and carcinogenic characteristics of most of these have not been tested systematically. Some emerging PFAS have a chemical structure similar to that of PFOA and PFOS, and a similar pattern of effects has been reported for PFOA, PFOS, and several other PFAS, e.g. perfluorononanoic acid (PFNA). In view of the large number of PFAS in past and present use, it has been suggested that grouping these compounds on the basis of chemical structure or exposure levels would facilitate the choice of suitable candidates for future research on their impact on human health. However, there is a gap in data to support such potential groupings of PFAS. Similarly, the effects of typical mixtures of PFAS have not been characterized systematically and remain a subject of ongoing research.

Scope of the systematic review

Standardized searches of the PubMed database (NCBI, 2023) were conducted for PFOA and PFOS for each outcome (cancer in humans, cancer in experimental animals, and mechanistic evidence, including the KCs). For cancer in humans, searches were also conducted in the Web of Science (Clarivate, 2023) and Embase (Elsevier, 2023) databases. The literature tree for PFOA and PFOS, including the full set of search terms for the agent name and each outcome type, is available online.^a

As described in the current Preamble to the *IARC Monographs* (last revised in 2019; <u>IARC</u>, 2019b; see present volume), the Working Group reviews publicly available scientific data, such as peer-reviewed papers in the scientific literature, and may also review unpublished reports, if made available in their final form by governmental agencies and if they contain enough detail for critical review. A public Call for Data was opened on the *IARC Monographs* website 1 year ahead of the meeting for Volume 135. Eligible studies were only those published or accepted for publication in the openly available scientific literature by the time of the Working Group meeting.

References

- Bagley BD, Chang SC, Ehresman DJ, Eveland A, Zitzow JD, Parker GA, et al. (2017). Perfluorooctane sulfonate-induced hepatic steatosis in male Sprague Dawley rats is not attenuated by dietary choline supplementation. *Toxicol Sci.* 160(2):284–98. doi:10.1093/toxsci/kfx185 PMID:28973659
- Behr AC, Lichtenstein D, Braeuning A, Lampen A, Buhrke T (2018). Perfluoroalkylated substances (PFAS) affect neither estrogen and androgen receptor activity nor steroidogenesis in human cells in vitro. *Toxicol Lett.* 291:51–60. doi:<u>10.1016/j.toxlet.2018.03.029</u> PMID:<u>29601859</u>
- Bjork JA, Butenhoff JL, Wallace KB (2011). Multiplicity of nuclear receptor activation by PFOA and PFOS in primary human and rodent hepatocytes. *Toxicology*. 288(1-3):8-17. doi:<u>10.1016/j.tox.2011.06.012</u> PMID: <u>21723365</u>
- Clarivate (2023). Web of Science [online database]. Available from: <u>https://www.webofscience.com/wos/</u><u>woscc/basic-search</u>.
- Das KP, Wood CR, Lin MT, Starkov AA, Lau C, Wallace KB, et al. (2017). Perfluoroalkyl acids-induced liver steatosis: Effects on genes controlling lipid homeostasis. *Toxicology.* 378:37–52. doi:10.1016/j.tox.2016.12.007 PMID:28049043

- Elcombe CR, Elcombe BM, Foster JR, Chang SC, Ehresman DJ, Butenhoff JL (2012b). Hepatocellular hypertrophy and cell proliferation in Sprague-Dawley rats from dietary exposure to potassium perfluorooctanesulfonate results from increased expression of xenosensor nuclear receptors PPARa and CAR/PXR. *Toxicology*. 293(1–3):16–29. doi:10.1016/j. tox.2011.12.014 PMID:22245121
- Elcombe CR, Elcombe BM, Foster JR, Chang SC, Ehresman DJ, Noker PE, et al. (2012a). Evaluation of hepatic and thyroid responses in male Sprague Dawley rats for up to eighty-four days following seven days of dietary exposure to potassium perfluorooctanesulfonate. *Toxicology*. 293(1–3):30–40. doi:10.1016/j. tox.2011.12.015 PMID:22239858
- Elsevier (2023). Embase [online database]. Elsevier. Available from: <u>https://www.embase.com/</u>.
- Hocevar SE, Kamendulis LM, Hocevar BA (2020). Perfluorooctanoic acid activates the unfolded protein response in pancreatic acinar cells. *J Biochem Mol Toxicol.* 34(11):e22561. doi:<u>10.1002/jbt.22561</u> PMID:<u>32578922</u>
- Hu WY, Lu R, Hu DP, Imir OB, Zuo Q, Moline D, et al. (2022). Per- and polyfluoroalkyl substances target and alter human prostate stem-progenitor cells. *Biochem Pharmacol.* 197:114902. doi:<u>10.1016/j.bcp.2021.114902</u> PMID:<u>34968493</u>
- IARC (2016). Some chemicals used as solvents and in polymer manufacture. *IARC Monogr Eval Carcinog Risks Hum.* 110:1–276. Available from: <u>https://</u> publications.iarc.who.int/547. PMID:<u>31829531</u>
- IARC (2019a). Report of the Advisory Group to Recommend Priorities for the *IARC Monographs* during 2020–2024. Lyon, France: International Agency for Research on Cancer. Available from: <u>https://monographs.iarc.who.</u> <u>int/wp-content/uploads/2019/10/IARCMonographs-AGReport-Priorities_2020-2024.pdf</u>.
- IARC (2019b). Preamble to the *IARC Monographs* (amended January 2019). Lyon, France: International Agency for Research on Cancer. Available from: <u>https://monographs.iarc.who.int/iarc-monographs-preamble-preamble-to-the-iarc-monographs/</u>.
- Louisse J, Fragki S, Rijkers D, Janssen A, van Dijk B, Leenders L, et al. (2023). Determination of in vitro hepatotoxic potencies of a series of perfluoroalkyl substances (PFASs) based on gene expression changes in HepaRG liver cells. *Arch Toxicol.* 97(4):1113–31. doi:10.1007/s00204-023-03450-2 PMID:36864359
- Louisse J, Rijkers D, Stoopen G, Janssen A, Staats M, Hoogenboom R, et al. (2020). Perfluorooctanoic acid (PFOA), perfluorooctane sulfonic acid (PFOS), and perfluorononanoic acid (PFNA) increase triglyceride

^a The literature tree for the monograph in the present volume is available at: <u>https://hawcproject.iarc.who.int/assessment/664</u>.

levels and decrease cholesterogenic gene expression in human HepaRG liver cells. *Arch Toxicol*. 94(9):3137–55. doi:10.1007/s00204-020-02808-0 PMID:32588087

- Marques MM, Berrington de Gonzalez A, Beland FA, Browne P, Demers PA, Lachenmeier DW, et al.; *IARC Monographs* Priorities Group (2019). Advisory Group recommendations on priorities for the *IARC Monographs. Lancet Oncol.* 20(6):763–4. doi:10.1016/ S1470-2045(19)30246-3 PMID:31005580
- Matkowskyj KA, Bai H, Liao J, Zhang W, Li H, Rao S, et al. (2014). Aldoketoreductase family 1B10 (AKR1B10) as a biomarker to distinguish hepatocellular carcinoma from benign liver lesions. *Hum Pathol.* 45(4):834–43. doi:10.1016/j.humpath.2013.12.002 PMID:24656094
- NCBI (2023). PubMed [online database]. Bethesda (MD), USA: National Library of Medicine. Available from: <u>https://pubmed.ncbi.nlm.nih.gov/</u>.
- Pierozan P, Cattani D, Karlsson O (2020). Perfluorooctane sulfonate (PFOS) and perfluorooctanoic acid (PFOA) induce epigenetic alterations and promote human breast cell carcinogenesis in vitro. Arch Toxicol. 94(11):3893–906. doi:10.1007/s00204-020-02848-6 PMID:32700164
- Qi Q, Niture S, Gadi S, Arthur E, Moore J, Levine KE, et al. (2023). Per- and polyfluoroalkyl substances activate UPR pathway, induce steatosis and fibrosis in liver cells. *Environ Toxicol.* 38(1):225–42. doi:<u>10.1002/tox.23680</u> PMID:<u>36251517</u>
- Rosen MB, Das KP, Rooney J, Abbott B, Lau C, Corton JC (2017). PPARα-independent transcriptional targets of perfluoroalkyl acids revealed by transcript profiling.

Toxicology. 387:95–107. doi:<u>10.1016/j.tox.2017.05.013</u> PMID:<u>28558994</u>

- UNEP (2023). PFASs listed under the Stockholm Convention. Stockholm Convention. Nairobi, Kenya: United Nations Environment Programme. Available from: <u>http://chm.pops.int/Implementation/Industrial</u> <u>POPs/PFAS/Overview/tabid/5221/Default.aspx</u>, accessed March 2023.
- Vanden Heuvel JP, Thompson JT, Frame SR, Gillies PJ (2006). Differential activation of nuclear receptors by perfluorinated fatty acid analogs and natural fatty acids: a comparison of human, mouse, and rat peroxisome proliferator-activated receptor-alpha, -beta, and -gamma, liver X receptor-beta, and retinoid X receptor-alpha. *Toxicol Sci.* 92(2):476–89. doi:10.1093/toxsci/kfl014 PMID:16731579
- Wan HT, Zhao YG, Wei X, Hui KY, Giesy JP, Wong CKC (2012). PFOS-induced hepatic steatosis, the mechanistic actions on β-oxidation and lipid transport. *Biochim Biophys Acta*. 1820(7):1092–101. doi:10.1016/j. bbagen.2012.03.010 PMID:22484034
- Zahm S, Bonde JP, Chiu WA, Hoppin J, Kanno J, Abdallah M, et al. (2023). Carcinogenicity of perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS). *Lancet Oncol.* 25(1):16–17. doi:<u>10.1016/S1470-2045(23)</u> <u>00622-8</u> PMID:<u>38043561</u>

1. EXPOSURE CHARACTERIZATION

1.1 Identification of the agent

Because of their acidic nature, perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS) exist in the environment, and in aqueous solutions, in equilibrium with their conjugated bases, perfluorooctanoate and perfluorooctane sulfonate. Salts of PFOA and PFOS will dissociate in solution and in the human body (except the stomach) to produce the respective anions perfluorooctanoate and perfluorooctane sulfonate.

The terms "PFOA" and "PFOS" are used for both the acid and the deprotonated form in environmental or biological samples, if not otherwise specified.

All isomeric forms of PFOA and PFOS and their salts should be considered to be part of the definition of the agents considered in the present monograph.

1.1.1 Nomenclature and molecular information

(a) PFOA and its salts

The agents considered in the present monograph include PFOA and its salts (see <u>Table 1.1</u> for a non-exhaustive list). PFOA and its salts exist as linear and branched isomers (see <u>Fig. 1.1</u>). Depending on the production method used, PFOA is present primarily as the linear isomer or as a mixture of linear (*n*-isomer) and branched isomers (see Section 1.2).

(b) PFOS and its salts

The agents considered in this present monograph include PFOS and its salts. Linear and branched isomers of PFOS and its salts exist (see <u>Table 1.2</u>). Depending on the production method, PFOS is present primarily as the linear isomer or as a mixture of linear and branched isomers (see <u>Fig. 1.2</u>).

1.1.2 Chemical and physical properties of the pure substances

Selected chemical and physical properties of PFOA and PFOS are presented in Table 1.3. The Working Group noted that there is some inconsistency in the data reported for these agents. This may be attributed to a combination of factors, including the purity of the acid form used to conduct the measurement; the low water solubility of the pure acid forms; and their strong surface active properties, resulting in sorption to interfaces such as the water surface or the walls of a glass vessel to an extent that is unknown for other substances (Goss, 2008).] The salts of PFOA and PFOS are more soluble in water than are their acid forms. For example, the water solubility of PFOA is 9.5 g/L, whereas the water solubility of ammonium perfluorooctanoate (APFO) is

Chemical name	CAS No.	IUPAC name and synonyms	Molecular formula	Relative molecular mass
<i>n</i> -Perfluorooctanoic acid	335-67-1 (<u>NCBI, 2023a</u>)	2,2,3,3,4,4,5,5,6,6,7,7,8,8,8-Pentadecafluoro- octanoic acid PFOA; <i>n</i> -Perfluorooctanoic acid; Pentadecafluoro-1-octanoic acid; Pentadecafluoro- <i>n</i> -octanoic acid; Pentadecafluorooctanoic acid; Perfluorocaprylic acid; Perfluoroctanoic acid; Perfluoroheptanecarboxylic acid (<u>NCBI, 2023a</u>)	C ₈ HF ₁₅ O ₂ (<u>NCBI, 2023a</u>)	414.07 (<u>NCBI,</u> <u>2023a</u>)
branched- Perfluorooctanoic acid	207678-51-1 705240-04-6 1144512-18-4 909009-42-3 15166-06-0 1144512-35-5 1192593-79-5 1144512-36-6 1144512-34-4 35605-76-6 (<u>Nielsen, 2012</u>)	sb-Perfluorooctanoic acid (CDC, 2022), br-Perfluorooctanoic acid (e.g. Jin et al., 2020) [The Working Group noted that different sums of isomers have been used. The exact definition varies between studies and might include all or just some of the isomers.] (See Fig. 1.1 for the names of some isomers)	C ₈ HF ₁₅ O ₂ (<u>NCBI, 2023a</u>)	414.07 (<u>NCBI,</u> <u>2023a</u>)
Ammonium perfluorooctanoate	3825-26-1 207678-62-4 19742-57-5 13058-65-5 (<u>Nielsen, 2012</u>)	Ammonium perfluorocaprylate; Pentadecafluorooctanoic acid ammonium salt; Octanoic acid, pentadecafluoro-, ammonium salt, APFO (<u>NCBI, 2023c</u>)	C ₈ H ₄ F ₁₅ NO ₂ (<u>NCBI, 2023c</u>)	431.10 (<u>NCBI,</u> 2023c)
Sodium perfluorooctanoate	335-95-5 207678-72-6 646-84-4 18017-22-6 1195164-59-0 (<u>Nielsen, 2012</u>)	Sodium perfluorocaprylate; Octanoic acid, pentadecafluoro-, sodium salt; Perfluorooctanoic acid sodium salt (<u>NCBI, 2023d</u>)	C ₈ F ₁₅ NaO ₂ (<u>NCBI, 2023d</u>)	436.05 (<u>NCBI,</u> <u>2023d</u>)
Potassium perfluorooctanoate	2395-00-8 207678-65-7 29457-73-6 (<u>Nielsen, 2012</u>)	Potassium, 2,2,3,3,4,4,5,5,6,6,7,7,8,8,8-pentadecafluoro- octanoate; Octanoic acid, pentadecafluoro-, potassium salt (NCBI, 2023e)	C ₈ F ₁₅ KO ₂ (<u>NCBI, 2023e</u>)	452.16 (<u>NCBI,</u> <u>2023e</u>)

Table 1.1 Nomenclature and molecular information for PFOA isomers and selected salts

br, branched; CAS, Chemical Abstracts Service Registry; IUPAC, International Union of Pure and Applied Chemistry; PFOA, perfluorooctanoic acid; sb, sum of branched isomers.

> 500 g/L, at 20 °C (<u>OECD, 2008</u>). [The Working Group noted that other properties of the salts might be different from those of the acid form, but data are lacking.]

[The Working Group noted that even though the data on the pK_a of PFOA and PFOS were inconsistent, the values were in the range of that for weak to strong acids. In aqueous samples of low concentrations (e.g. drinking-water, biospecimen), it can be assumed that both agents are mainly present in the deprotonated form.]

Fig. 1.1 Main salts and isomers of PFOA

a. PFOA isomers		b. Ammonium salts of PFOA isomers (APFO)		
Structure of carbon chain	CAS No.	Structure of carbon chain	CAS No.	
Соон	335-67-1 Linear PFOA	C00 ⁻ NH ₄ ⁺	3825-26-1	
Соон	207678-51-1 Perfluoro-2-methyl- bentanoic acid	C00 ⁻ NH ₄ ⁺	207678-62-4	
Соон	705240-04-6 Perfluoro-3-methyl- heptanoic acid	C00 ⁻ NH ₄ ⁺	19742-57-5	
Соон	1144512-18-4 Perfluoro-4-methyl- hentanoic acid	COO ⁻ NH ₄ ⁺	13058-06-5	
		c. Sodium salts of PFOA isomers		
Соон	909009-42-3 Perfluoro-5-methyl- heptanoic acid	Structure of carbon chain	CAS No.	
Соон	15166-06-0 Perfluoro-6-methyl-	COO-Na+	335-95-5	
1	heptanoic acid	COO"Na ⁺	207678-72-6	
	1144512-35-5			
	Perfluoro-3,5-dimethyl- hexanoic acid	COO"Na ⁺	646-84-4	
Соон	1192593-79-5 Perfluoro-4,4-dimethyl- hexanoic acid	- COO-Na*	18017-22-6	
Соон	1144512-36-6 Perfluoro-4,5-dimethyl- hexanoic acid	COO-Na*	1195164-59-0	
Соон	1144512-34-4			
	Perfluoro-5,5-dimethyl- hexanoic acid	d. Potassium salts of PFOA isome	ers	
	25005 50 5	_ Structure of carbon chain	CAS NO.	
СООН	35605-76-6 Perfluoro-2-ethyl- hexanoic acid	COO^-K+	2395-00-8	
	invanore and	C00-K+	207678-65-7	
		CO0⁻K⁺	29457-73-6	

APFO, ammonium perfluorooctanoate; CAS, Chemical Abstracts Service; PFOA, perfluorooctanoic acid. From <u>Nielsen (2012)</u>, as cited in <u>IARC (2016)</u>.

Chemical name	CAS No.	IUPAC name and synonyms	Molecular formula	Relative molecular mass
<i>n</i> -Perfluorooctane- sulfonic acid	1763-23-1 (<u>NCBI, 2023b</u>)	1,1,2,2,3,3,4,4,5,5,6,6,7,7,8,8,8-Heptadecafluorooctane- 1-sulfonic acid PFOS, <i>n</i> PFOS; Heptadecafluorooctane-1-sulfonic acid; Perfluorooctane sulfonate; Perfluorooctane-1-sulfonic acid; Perfluorooctylsulfonic acid; Heptadecafluoro-1- octanesulfonic acid; Heptadecafluorooctane sulfonic acid; 1-Perfluorooctanesulfonic acid (NCBI, 2023b; Royal Society of Chemistry, 2023)	C ₈ HF ₁₇ O ₃ S (<u>NCBI,</u> 2023b)	500.13 (<u>NCBI,</u> <u>2023b</u>)
Branched- Perfluorooctane- sulfonic acid	255831-20-0 747385-21-3 775554-63-7 740777-79-1 765246-09-1 927670-12-0 950669-24-6 950669-23-5 950669-22-4 950669-21-3 927670-09-5 (CAS, 2023)	sm-Perfluorooctanesulfonic acid (CDC, 2022), br-Perfluorooctanesulfonic acid (EFSA Panel on Contaminants in the Food Chain, 2018) [The Working Group noted that different sums of isomers have been used. The exact definition varies between studies and might include all or just some of the isomers.] See Fig. 1.2 for the names of some isomers.	C ₈ HF ₁₇ O ₃ S (<u>NCBI,</u> <u>2023b</u>)	500.13 g/mol (<u>NCBI, 2023b</u>)
Ammonium perfluorooctane- sulfonate	29081-56-9 (<u>NCBI, 2023f</u>)	Ammonium heptadecafluoro-1-octanesulfonate (<u>NCBI, 2023f</u>)	$C_{8}H_{4}F_{17}NO_{3}S$ (<u>NCBI,</u> <u>2023f</u>)	517.16 g/mol (<u>NCBI, 2023f</u>)
Potassium perfluorooctane- sulfonate	2795-39-3 (<u>NCBI, 2023h</u>)	Potassium heptadecafluoro-1-octanesulfonate 1-Octanesulfonic acid, heptadecafluoro-, potassium salt (<u>NCBI, 2023h</u>)	$C_{8}F_{17}KO_{3}S$ (NCBI, 2023h)	538.22 g/mol (<u>NCBI, 2023h</u>)
Lithium perfluorooctane- sulfonate	29457-72-5 (<u>NCBI, 2023g</u>)	Lithium heptadecafluorooctanesulfonate Heptadecafluoro-1-octanesulfonic acid lithium salt (NCBI, 2023g)	$C_{8}F_{17}LiO_{3}S$ (NCBI, 2023g)	506.10 g/mol (<u>NCBI, 2023g</u>)

Table 1.2 Nomenclature and molecular information for PFOS isomers and selected salts

br, branched; CAS, Chemical Abstracts Service Registry; IUPAC, International Union of Pure and Applied Chemistry; PFOS, perfluorooctanesulfonic acid; sm, sum of perfluoromethylheptane sulfonate isomers.

Fig. 1.2 Main salts and isomers of PFOS

a. PFOS isomers





CAS, Chemical Abstracts Service; PFOS, perfluorooctanesulfonic acid. From Langlois and Oehme (2006); Miralles-Marco and Harrad (2015).

Property	PFOA	PFOS
Boiling-point	192 °C (<u>US EPA, 2017a; NCBI, 2023a</u>)	258–260 °C (<u>US EPA, 2017a</u>)
Melting-point	54.3 °C (<u>IARC, 2016; ATSDR, 2021</u>)	84 °C [The Working Group noted that these are predicted data (<u>US EPA, 2023a</u>)]
Vapour pressure	[0.0421 hPa] at 25 °C (<u>ATSDR, 2021; NCBI,</u> <u>2023a</u>), [0.700 hPa] at 25 °C (<u>US EPA, 2017a</u>)	[0.003 hPa] at 25 °C (<u>US EPA, 2017a; NCBI,</u> 2023b)
Water solubility	9.5 g/L at 25 °C (<u>IARC, 2016; US EPA, 2017a</u> ; <u>ATSDR, 2021</u>)	680 mg/L at 25 °C (<u>US EPA, 2017a;</u> <u>NCBI, 2023b</u>)
Density	1.8 g/cm ³ at 20 °C (<u>IARC, 2016; ATSDR, 2021</u>)	1.84 g/cm ³ [The Working Group noted that these are predicted data (<u>US EPA, 2023a</u>)]
log K _{ow} (octanol/water partition coefficient, <i>P</i>)	Not measurable, since PFOA forms multiple layers in an octanol/water mixture (<u>ATSDR, 2021</u>)	Not measurable, since PFOS forms multiple layers in an octanol/water mixture (<u>ATSDR</u> , <u>2021</u> ; <u>NCBI</u> , <u>2023b</u>)
$\log K_{\rm oc}$ (organic carbon/water partition coefficient)	2.06 (<u>US EPA, 2017a</u>)	2.57–3.14 (<u>US EPA, 2017a; ATSDR, 2021</u>)
Conversion factor	1 ppm = 16.94 mg/m³, 1 mg/m³ = 0.059 ppm, at 25 °C and 101 kPa	1 ppm = 20.45 mg/m ³ ; 1 mg/m ³ = 0.049 ppm, at 25 °C and 101 kPa
Physical description	White to off-white powder (<u>ATSDR, 2021</u>)	White powder (<u>ACS, 2019</u>); also reported as off- white to grey liquid (<u>NCBI, 2023b</u>)
Stability	When heated to decomposition, it emits toxic vapours of hydrogen fluoride. Perfluoroalkyl carboxylates are resistant to direct photolysis and reaction with acids, bases, oxidants, and reductants (<u>IARC, 2016; ATSDR, 2021</u>).	When heated to decomposition, it emits toxic vapours of sulfur oxides and fluorine (<u>NCBI</u> , <u>2023b</u>). Perfluoroalkyl sulfonates are resistant to direct photolysis and reaction with acids, bases, oxidants, and reductants (<u>ATSDR, 2021</u>).

Table 1.3 Chemical and physical properties of pure PFOA and PFOS in acid form

PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; ppm, parts per million.

1.1.3 Technical grade and impurities

(a) PFOA

Before 2002, PFOA, which was produced mainly by the electrochemical fluorination (ECF) method, was reported to have a consistent isomer composition of $78\% \pm 1.2\%$ linear isomers and 22% ± 1.2% branched-chain isomers in 18 production lots tested over a 20-year period. PFOA produced by ECF was reported to contain the following impurities: perfluorohexanoate, 0.73%; perfluoroheptanoate, 3.7%; perfluorononanoate, 0.2%; perfluorodecanoate, 0.0005%; perfluoroundecanoate, 0.0008%; and perfluorododecanoate, 0.0008%. From 2002 onwards, PFOA, which is produced mainly by the telomerization method, is typically an isomerically pure, linear product (Benskin et al., 2010a; IARC, 2016).

(b) PFOS

PFOS and its salts are mainly produced by ECF. This ECF-produced PFOS comprises 11 different isomers, including the linear isomer (approximately 70%) and various branched isomers (approximately 30%) (Naile et al., 2010). Some of these isomers (specifically those with branched chains) are chiral, and the environmental fate and behaviour of PFOS may vary according to its isomeric and enantiomeric composition (Miralles-Marco and Harrad, 2015). The following impurities were reported in a commercial sample of potassium perfluorooctanesulfonate (purity, 86.9%): homologues with fewer carbons (C2-C7, predominantly C6), 9.38%; metals (calcium, magnesium, sodium, nickel, and iron), 1.45%; inorganic fluoride, 0.59%; molecules containing perfluorinated sulfur(VI) (sulfur hexafluoride), 0.68%; PFOA,

0.33%; nonafluoropentanoic acid, 0.28%; hydrocarbon sulfonate salts, 0.15%; terminal cyclopentyl PFOS, 0.11%; heptafluorobutyric acid, 0.1%; and trifluoroacetic acid, 0.015% (Seacat et al., 2003).

1.2 Production and uses

1.2.1 Production process

PFOA and PFOS have been manufactured by ECF and fluorotelomerization. During ECF, an organic acyl or sulfonyl fluoride backbone structure is dissolved in a solution of aqueous hydrogen fluoride (<u>Buck et al., 2011; ATSDR, 2021; ITRC, 2022c</u>). All the hydrogens on the molecule are then replaced with fluorines when a direct electrical current is passed through the solution. Perfluoroacyl fluorides produced by ECF are hydrolysed to form the perfluorocarboxylic acid, which is then separated via distillation. The ECF process results in a mixture of linear and branched isomers, with 78% and 70% linear forms of PFOA and PFOS, respectively (<u>Buck et al., 2011; ATSDR, 2021; ITRC, 2022c</u>).

Fluorotelomerization produces primarily linear perfluorocarboxylic acids with an even number of carbon atoms, which includes PFOA. The process begins with the preparation of pentafluoroiodoethane from tetrafluoroethene. Tetrafluoroethene is then added to the product at a molar ratio that gives a product of the desired chain length, before the product is oxidized to form the carboxylic acid (<u>Buck et al., 2011</u>; <u>ATSDR, 2021; ITRC, 2022c</u>).

1.2.2 Production volume

Production of perfluoroalkyl carboxylates began in 1947, initially by ECF. By 2000, ECF was still the leading process, accounting for the majority (80–90%) of the production of APFO – a salt of PFOA – worldwide, which was approximately 260 tonnes in 1999 (Prevedouros et al., 2006). Global production of perfluorooctane sulfonyl fluoride (POSF) – a production precursor of PFOS – was estimated to be 96 000 tonnes (or 122 500 tonnes, including wastes, largely disposed of through land farming/landfilling or incineration) between 1970 and 2002. One major company based in the United States of America (USA) manufactured most of the POSF, using ECF, accounting for about 78% of global production in 2000 (Paul et al., 2009). [The Working Group noted that data on production volumes were limited, particularly after 2002 (see below).]

In the USA, the manufacture and import of PFOA and PFOS has been phased out; however, some existing stocks may remain. PFOS was phased out of production by its primary manufacturer between 2000 and 2002 (US EPA, 2016) and was not reported in the 2006 or 2012 Chemical Data Reporting effort (US EPA, <u>2023e</u>). Before 2006, production volume ranges in the USA were reported as follows: PFOA, [5-227] tonnes in 1986, 1994, 1998, and 2002; APFO, [5-227] tonnes in 1986, 1990, 1994, and 1998, and [227-454] tonnes in 2002; and PFOS, [5-227] tonnes in 1994 and 2002 (ATSDR, 2021). In 2006, the United States Environmental Protection Agency (US EPA) invited eight major leading companies manufacturing PFOA to join the 2010/2015 PFOA Stewardship Program. All participating companies reported meeting the goals of this programme, which included eliminating emissions by 2015 (US EPA, 2022). As of November 2016, PFOA and PFOS are no longer used in food contact applications sold in the USA (US FDA, 2023). For regulatory agency guidelines on the production and use of PFOA and PFOS that might explain changes over time, see Section 1.5.

Since 2002 there has been a geographical shift in industrial production (particularly fluoropolymer-production sites) from North America, Europe, and Japan to some countries in Asia, especially China (<u>Wang et al., 2014</u>). <u>Zhang et al. (2012</u>) report PFOS production in China of 247 tonnes in 2006 and about 100 tonnes in 2008, with the majority used in metal plating (30-40 tonnes/ year) and aqueous film-forming foam (AFFF) (25-35 tonnes/year), as well as the production of sulfluramid insecticides (4-8 tonnes/year). During 2004-2012, 480 tonnes of PFOA and its salts were produced in China using the ECF process (Li et al., 2015). China has also implemented a phase-out of PFOA and PFOS, with the Chinese Ministry of Environmental Protection restricting and banning different uses (OECD, 2023a). Brazilian imports of N-ethyl perfluorooctane sulfonamide (N-EtFOSA), a PFOS precursor, for the production of sulfluramid between 2005 and 2015 were almost exclusively from China; imports of *N*-EtFOSA peaked at > 1.3 tonnes in 2012, and exports increased to around 2 tonnes per year in 2012 (Löfstedt Gilljam et al., 2016). [The Working Group noted that these data were not for PFOS itself, but might give some indication of use or production in these geographical regions where data for PFOS itself are lacking.]

1.2.3 Uses

The unique properties of per- and polyfluoroalkyl substances (PFAS), including PFOA and PFOS, have led to extensive uses in a wide variety of diverse applications. These properties, including the "ability to lower the aqueous surface tension, high hydrophobicity, high oleophobicity, non-flammability, high capacity to dissolve gases, high stability, extremely low reactivity, high dielectric breakdown strength, good heat conductivity, low refractive index, low dielectric constant, ability to generate strong acids, operation at a wide temperature range, low volatility in vacuum, and impenetrability to radiation" (Glüge et al., 2020) facilitate nearly 300 different uses and functions. For the more than 1400 PFAS evaluated by Glüge et al. (2020), uses fell within 20 industry branches (e.g. chemical industry and electroplating) and 44 other use categories (e.g. cleaning compositions and personal care

products). [However, the Working Group noted that the uses identified by Glüge et al. (2020) are summarized across all 1400 PFAS; some uses may not be applicable to PFOA and PFOS.] PFOA and PFOS may be present in industrial and consumer products as main ingredients, or as unreacted raw materials, undesired reaction by-products, or cross-contaminants along production and supply chains (OECD, 2015a; Glüge et al., 2020). PFOA and APFO are used in chemical manufacturing processes, industrial products and processes, and consumer products. As a processing aid, APFO has been used extensively to manufacture fluoropolymers, such as polytetrafluoroethylene (PTFE) (Buck et al., 2011). Applications for fluoropolymers containing PFOA, as well as direct uses for PFOA, include household products with non-stick coatings (e.g. cookware); textiles for outdoor or personal protection applications (e.g. firefighter turnout gear); personal care products (e.g. cosmetics, sunscreens, dental floss); seals and gaskets used in the aviation and aerospace industries; coatings for cables and wires; electronics, solar panels and electrolyte fuel cells; fluoropolymer fabrication materials used in food processing (e.g. liners for grills and ovens); carpets; cleaning and impregnating agents; construction materials (e.g. chipboard and oriented strand board); and surface coatings conferring stain-, oil- and water resistance on carpets, textiles, leather products, and paper or cardboard packaging used in food and feed contact paper and board (e.g. popcorn bags, pizza boxes, fast food containers) (Kotthoff et al., 2015; Bečanová et al., 2016; ATSDR, 2021; Ramírez Carnero et al., 2021; ITRC, 2022a). [The Working Group noted that the concentration of PFOA varied by application and product. For example, in fluoropolymer-based consumer products (e.g. non-stick cookware or textiles) PFOA may be present in a chemically bound form or at lower concentrations than in products in which PFOA is an intentionally added ingredient.]

With some applications that overlap those of PFOA, such as waxes (e.g. car, shoe, floor, ski), carpets, and packaging used for food and feed (Kotthoffetal., 2015; Nordic Council of Ministers, 2017), PFOS has additionally been used in the semiconductor industry; as a hydraulic fluid additive in the aviation and aerospace industries; as an etchant and antireflective coating in photolithography processes; and in the fabrication of imaging devices (e.g. cameras, mobile phones, and printers) (ITRC, 2022a). During electroplating processes in metal finishing and plating operations, PFOS has been used as a mist-suppressing agent to prevent workers' exposure to aerosols and mists; however, in the USA, the US EPA National Emissions Standards for Hazardous Air Pollutants (NESHAP) mandated that use of PFOS-based mist-suppressants in chromium electroplating be discontinued by 2015 (Office of the Federal Register, 2012). Similar phase-outs of PFOS for this application have occurred in other countries (Ramírez Carnero et al., 2021; ITRC, 2022a); however, this application is still permitted in the European Union (EU) (Swedish Chemicals Agency, 2020). PFOS is also present in a variety of building and construction materials, including paints and varnishes; insulation (phenolic foam); dyes and ink; and in wetting, levelling, and dispersing agents (<u>ITRC, 2022a</u>).

PFOS together with other PFAS have been used extensively in class B firefighting foams known as AFFFs. [The Working Group noted that AFFFs were designed to meet firefighting performance criteria; formulations of PFAS have changed over time and by manufacturer (Leeson et al., 2021).] These foams were developed in the 1960s to extinguish liquid fuel fires by efficiently suppressing flammable liquid vapour, suffocating the fire hazard, and preventing re-ignition (Rosenfeld et al., 2023). AFFF containing PFOS was manufactured in the USA from the late 1960s until 2002; however, other fluorotelomer-based AFFF manufactured from the 1970s until 2016 contained precursors of PFOA. Although newer formulations of class B foams exist, the legacy products have been used during fire response, training, and equipment maintenance activities by the military, airport and municipal fire departments, and oil and gas production and refining industries worldwide (<u>Prevedouros et al., 2006</u>; <u>ITRC, 2022b</u>).

PFAS that are known to convert into PFOA and PFOS, frequently referred to as "precursors", are used in a variety of settings (see Section 1.4(d)). Although a detailed description of these uses and functions is beyond the scope of the present monograph, examples include the semiconductor and electronics industry; personal care products, coatings for medical devices, apparel, pharmaceutical equipment; and the pesticide sulfluramid (Löfstedt Gilljam et al., 2016; Glüge et al., 2020; ITRC, 2022a).

1.3 Detection and quantification

General considerations

(a) Analytical method terminology

Analytical methods used for PFAS consist of targeted, non-targeted, and total fluorine analysis approaches. Targeted analyses refer to methods for a pre-defined, known list of analytes for which authentic chemical standards exist. Non-targeted analyses are capable of identifying suspect and unknown analytes in a sample, often through mass spectrometry. Analyte identity can then be confirmed using authentic chemical standards, and unknown analytes can be tentatively identified through matching to existing chemical libraries (US EPA, 2023b). Total fluorine methods quantify the fluorine (often organic fluorine) present in a sample, regardless of chemical structure, and thus are unable to differentiate between chemical structures of analytes (Schultes et al., 2019).

Some methods are able to differentiate between linear and branched isomers. [The Working Group noted that some recent studies of PFOA and PFOS differentiate between linear and branched isomers. For example, linear, secondary-branched, and tertiary-branched isomers of PFOA and PFOS can be resolved by high-resolution differential ion mobility-mass spectrometry (DMS-MS) (Ahmed et al., 2019).] Isomer profiling can be used in the quantitative assessment of manufacturing source (Benskin et al., 2010b).

The Working Group noted that, in the papers reviewed for PFOA and PFOS exposure, multiple approaches were used to report the lowest concentration of a chemical analyte. These commonly include the limit of detection (LOD), which describes the lowest concentration identifiable by the analytical instrumentation, and the limit of quantification (LOQ), which describes the lowest concentration that can be determined by means of a given analytical procedure with the established accuracy, precision, and uncertainty. The Working Group noted that some studies reported lowest measurable concentrations as LODs, whereas others reported LOQs. This makes comparison between studies more challenging at the lower end of the concentration range studied.]

Liquid chromatography-mass spectrometry (LC-MS), commonly used for the analysis of PFOA and PFOS, is a sophisticated analytical technique that requires the purchase and maintenance of an expensive instrument. [The Working Group noted that, consequently, access to PFOA and PFOS analyses can be challenging for regions or populations with limited resources, such as low- or middle-income countries (LMICs), and may explain the paucity of available data in some regions of the world.]

(b) Potential for cross-contamination

Consideration of numerous potential sources of cross-contamination (also referred to as "background interference") of PFOA and PFOS have been documented in the context of sample collection and analysis (Method 533, <u>US EPA</u>, <u>2023b</u>; <u>MDEQ</u>, <u>2018</u>). Potential sources of PFAS cross-contamination in the typical sampling environment include water used for washing or decontamination and materials used within the sampling environment (<u>MDEQ</u>, <u>2018</u>).

In a laboratory setting, analytical instrumentation (e.g. mass spectrometry) and laboratory equipment or materials often have fluoropolymer (e.g. PTFE) components that may contain PFOA (Method 533, <u>US EPA, 2023b; MDEQ, 2018</u>). [The Working Group noted that cross-contamination issues may affect the concentrations of PFOA and PFOS in samples and blanks alike. This may contribute to the high LOQs reported in some studies.]

1.3.1 Air

Several methods have been reported for the quantification of PFOA and PFOS in indoor and/or outdoor air using both active and passive air-sampling techniques, and some examples are presented in Table 1.4. These methods generally rely on a combination of sampling media to collect both gas and particle-bound PFOA and PFOS. Most reported active air sampling methods apply a filter (glass fibre or quartz) to capture the particle phase, followed by an adsorbent resin to bind the gaseous-phase PFAS. Few active air-sampling methods reported the use of filters only to capture particle-bound PFAS, or sorbent only to capture both gas- and particle-bound PFAS on the same sampling medium. The passive sampling methods use a compact-design sampler containing a sorbent-impregnated, polyurethane foam (PUF) disc to sample PFAS from both the gaseous and particle phases. Therefore, active sampling methods with two independent sampling media can differentiate between gas- and particle-bound PFAS concentrations, whereas passive sampling methods can only provide PFAS concentrations as the sum of concentrations in the two phases. In general, sampling media (filters and PUF discs)

Sample matrix	Sampler type	Sample collection method	Instrument (LOD)ª	Reference
Air emissions from stationary sources	Active (flow rate not specified – minimum sample of 3 m ³)	Gas- and particle-bound PFAS collected on a sampling train of GFF or QFF, a packed column of adsorbent material	HPLC-MS/MS (PFOA, 0.35 ng/m ³ ; PFOS, 0.43 ng/m ³)	EPA-OTM-45 <u>US EPA (2021)</u>
Indoor and outdoor air	Active (flow rate of 6.4 m ³ /h)	Gas- and particle-bound analytes collected using GFFs (particle phase) and glass columns with a PUF–XAD-2–PUF sandwich (gaseous phase)	HPLC-TOF/MS (1 pg/m ³)	<u>Barber et al.</u> (2007)
Outdoor air	Active (flow rate of 1.1 m ³ /h)	Particle-bound analytes collected using GFF	HPLC-TOF/MS (PFOA, 0.2 pg/m ³ ; PFOS, 0.4 pg/m ³)	<u>Jahnke et al.</u> (2007)
Outdoor air, PM _{2.5}	Active (flow rate of 30 m ³ /h)	$\mathrm{PM}_{\scriptscriptstyle 2.5}\text{-}\mathrm{bound}$ analytes collected on QFF	HPLC-MS/MS (0.14 pg/m ³)	<u>Beser et al.</u> (2011)
Indoor and outdoor air	Passive	Gas- and particle-bound analytes collected on sorbent (XAD-4)-impregnated PUF disc samplers	HPLC-MS/MS (PFOA, 0.47 pg/m ³ ; PFOS, 0.02 pg/m ³)	<u>Shoeib et al.</u> (2010, 2011)
Indoor air and personal breathing zone	Active (flow rate of 0.12 m ³ /h)	ISOLUTE ENV+ sorbent (hydroxylated polystyrene– divinylbenzene copolymer) cartridge	HPLC-MS/MS (PFOA, 73 pg/g extract; PFOS, 38 pg/g extract)	<u>Nilsson et al.</u> (2013b)
Outdoor air, PM _{2.5}	Active (flow rate of 30 m ³ /h)	PM _{2.5} -bound analytes collected on QFF	HPLC-HRMS (PFOA, 0.18 pg/mL extract; PFOS, 0.11 pg/mL extract)	<u>Kourtchev et al.</u> (2022)

Table 1.4 Selected analytical methods for the measurement of PFOA and PFOS in air

GFF, glass-fibre filters; h, hour(s); HPLC, high-performance liquid chromatography; HRMS, high-resolution mass spectrometry; ISOLUTE ENV+, commercial solid-phase extraction column; LOD, limit of detection; MS, mass spectrometry; MS/MS, tandem mass spectrometry; NR. not reported; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PM_{2.5}, particulate matter with diameter < 2.5μ m; PUF, polyurethane foam; QFF, quartz fibre filters; SIP, sorbent-impregnated polyurethane; TOF, time-of-flight; XAD, commercial resin.

^a Using electrospray ionization in negative ion mode.

are extracted with an organic solvent (mostly methanol), followed by clean-up using filtration, centrifugation, or solid-phase extraction (SPE). Instrumental analysis is usually carried out using LC-MS with an electrospray ionization (ESI) source, operated in negative ion mode. [The Working Group noted that the LC-MS methods reported for the analysis of PFOA and PFOS in air had low instrumental LODs and were appropriate for trace level detection of these chemicals in air samples.]

EPA-OTM-45 is a standardized method that can be used to measure air emissions of PFOA and PFOS from stationary emission sources. In this method, a sampling train of glass fibre or quartz filter is applied, followed by a packed column of adsorbent material to collect both gaseous-phase and particulate-bound target analytes. The samples are then extracted with methanol/5% ammonium hydroxide, cleaned-up and concentrated using SPE and quantified using LC-MS/MS. The method detection limits (MDLs) for PFOA and PFOS were 0.43 ng/m³ and 0.35 ng/m³, respectively (<u>US EPA, 2021</u>).

[The Working Group noted that most of these methods have been developed for environmental measurements, and there has been no validated method using personal samplers developed for occupational exposure measurements.]

1.3.2 Water

Several methods have been developed to measure PFOA and PFOS concentrations in water. Some selected methods are summarized in <u>Table 1.5</u>.

The US EPA Methods 537.1 (published in 2009) and 533 (published in 2019) describe methods to analyse PFOA and PFOS in drinking-water (US EPA, 2019; Shoemaker and Tettenhorst, 2020). Water samples are fortified with surrogate standards and passed through a solid-phase sorbent cartridge to extract the PFAS and surrogates. The extract is concentrated, and isotopically labelled performance standards are added. Extracts are analysed by LC-MS/MS. LODs were reported as 0.53 and 1.1 ng/L for PFOA and PFOS, respectively. Interlaboratory comparisons have reported coefficients of variation (CVs) between laboratories of 23% for PFOA and 33–40% for PFOS isomers (van der Veen et al., 2023). An earlier interlaboratory comparison reported substantially higher CVs: 118% for PFOA and 95% for PFOS in water samples (van Leeuwen et al., 2006).

The US EPA has also validated SW-846 Method 8327 using external standard calibration and LC-MS/MS for the analysis of PFOA and PFOS (and other PFAS) in surface water, groundwater, and wastewater effluent (<u>US EPA, 2023c</u>).

In 2023, a draft version was published of US EPA Method 1633, which had already been finalized for the aqueous matrices wastewater, surface water, and groundwater (<u>US EPA 2023d</u>).

Some examples of low detection limits reported for PFOA and PFOS detected via various methods were: PFOA, 0.3 ng/L in demineralized water and 0.5 ng/L in natural spring water (Janda et al., 2019); 0.10 ng/L (Song et al., 2023); PFOA, 0.1 ng/L, and PFOS, 0.5 ng/L (Chen et al., 2016); and PFOA, 0.01 ng/L, and PFOS, 0.01 ng/L (Zheng et al., 2023). [The Working Group noted that detection limits have changed as the methodology for sample processing and detection has improved over time. Differences in LODs might also be explained by the use of different methods to derive these LODs.]

1.3.3 Soil, sediment, consumer products, and foods

Several analytical methods for the quantification of PFOA and PFOS in soil, sediment, dust, and consumer products have been reported. Because of the large variability in sample matrices, the analytical methods involved various extraction techniques, including solvent extraction, ultrasonic extraction, ion-pair

Table 1.5 Selected analytical methods for the measurement of 11 OA and 11 OS in water

Sample matrix	Sample preparation	Instrument	LOD	Reference
Drinking-water	Adsorb on polystyrene divinylbenzene; elute with methanol; reconstitute in water/methanol with ¹³ C-PFOA internal standards	HPLC-MS/MS	PFOA, 0.53 ng/L; PFOS, 1.1 ng/L	<u>Shoemaker and</u> <u>Tettenhorst (2020)</u> US EPA Method 537.1
Drinking-water	Adsorb on polystyrene divinylbenzene; elute with methanol containing ammonium hydroxide; reconstitute in water/methanol with ¹³ C-PFOA internal standards	HPLC-MS/MS	PFOA, 3.4 ng/L; PFOS, 4.4 ng/L	<u>US EPA (2019)</u> US EPA Method 533
Reagent water, surfacewater, groundwater, and wastewater effluent	Uses US EPA Method 3512 – dilute and filter; does not use SPE or carbon clean-up steps, which is a significant difference from the other US EPA methods	LC-MS/MS	PFOA, 10 ng/L; PFOS, 10 ng/L (LOQ)	<u>US EPA (2023c)</u> US EPA Method 8327
Drinking-water, ground water and surface water (fresh water and sea water)	No pretreatment; adsorb on WAX SPE cartridges, elute with methanol, evaporate with nitrogen gas	HPLC-MS/MS	PFOA, 10 ng/L; PFOS, 2.0 ng/L (LOQ)	<u>ISO (2009)</u> ISO Method 25101
Wastewater, surface water, groundwater, landfill leachate	Glass fibre filtration of total suspended solids; aqueous samples with \leq 50 mg of suspended solids must not be filtered; aqueous sample: spiking with isotopically labelled standards, SPE, and carbon clean-up	HPLC-MS/MS	PFOA, 0.54 ng/L; PFOS, 0.63 ng/L	US EPA (2023d) US EPA Method 1633 (draft version as of November 2023, finalized for the aqueous matrices: wastewater, surface water, and groundwater) ^a
Non-filtered waters, e.g. drinking-water, natural water (fresh water and sea water) and wastewater	Adsorb on high-purity mixed-mode WAX sorbent; elute with methanol	LC-MS/MS	PFOA, 0.31 ng/L; PFOS, 0.29 ng/L	ISO (2019) ISO Method 21675; Jones and Harden (2022)
Drinking-water	Adsorb on WAX SPE cartridges, elute with 1% ammonium hydroxide in methanol; concentrate to dryness; reconstitute in methanol	LC-MS/MS	PFOA, 0.01 ng/L; PFOS, 0.01 ng/L	<u>Zheng et al. (2023)</u>

HPLC, high-performance liquid chromatography; ISO, International Organization for Standardization; LC, liquid chromatography; LOD, limit of detection; LOQ, limit of quantification; MS/MS, tandem mass spectrometry PFAS, per- and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; SPE, solid-phase extraction; US EPA, United States Environmental Protection Agency; WAX, weak anion exchange.

^a US EPA recommends the use of Method 1633, and it is currently the only PFAS method that has been validated in multiple laboratories for aqueous matrices that include wastewater, surface water, groundwater, and landfill leachate, as well as for soil, sediment, biosolids, and fish and shellfish tissue.

extraction and dispersive SPE. Sample clean-up methods also varied, from filtration after pH control, to SPE and QuEChERS ("Quick, Easy, Cheap, Effective, Rugged, Safe"). LC-MS-ESI was the method of choice for the analysis of PFOA and PFOS. A summary of these methods is provided in <u>Table 1.6</u>.

The standard test method ASTM D7968 can be used for the determination of PFOA and PFOS in soil samples. The method uses solvent extraction with methanol:water (50:50) under basic conditions, followed by filtration, acidification, and then LC-MS/MS analysis. The MDLs were 6.2 and 18.8 ng/kg for PFOA and PFOS, respectively (<u>ASTM International, 2017</u>).

The United States Food and Drug Administration (US FDA) published a validated method C-010.02 for the analysis of 16 PFAS chemicals, including PFOA and PFOS, in various food items. Target PFAS are extracted from the food samples using acetonitrile and formic acid. After extraction, a modified QuEChERS technique is performed for clean-up, and further SPE is required for clean-up of complex samples. The cleaned extracts are then analysed using LC-MS/MS, with MDLs of 12–24 ng/kg for PFOA and 7–28 ng/kg for PFOS, in the different food items tested (US FDA, 2021b).

1.3.4 Human biospecimens

In early studies on exposed workers, total serum fluorine was used as a surrogate variable for PFOA exposure (e.g. Gilliland and Mandel, 1996). [The Working Group noted that using a total fluorine approach as a surrogate for PFOA is not an accurate quantification method for an individual analyte.] In 2001, LC-MS/MS was used for the first time for the analysis of PFOA and PFOS in biological samples (Hansen et al., 2001). At present, mainly targeted methods are used for the analysis of PFOA and PFOS in whole blood, serum, and plasma. Non-targeted mass spectrometry-based methods, lacking the ability

to quantify concentrations, are also used (e.g. Chang et al., 2023). [However, these methods provide semiquantitative intensity levels that allow ranking of participants within a study.] A selection of methods for the analysis of PFOA and PFOS in human biospecimens is shown in Table 1.7. The usual sample preparation step before extraction is protein precipitation (for example, with acetonitrile). An aliquot of the supernatant is analysed using LC-MS/MS. Isotopically labelled internal standards may be used. Typical instrumental LODs are < 0.1 ng/mL for PFOA and PFOS, although higher values were reported in earlier publications (e.g. 10 ng/mL for PFOA; Sottani and Minoia, 2002), and lower values in more recent ones (e.g. 0.023 ng/mL for PFOA and 0.033 ng/mL for PFOS; Gao et al., 2018).

A method for determination of PFOA and PFOS (and other PFAS) in human serum, plasma, and whole blood described the use of methanol for protein precipitation and online SPE-LC-MS/MS. LODs for PFOA in serum, plasma, and whole blood were 0.018, 0.009, and 0.045 ng/mL, respectively, whereas the corresponding LODs for PFOS were 0.009 ng/mL for all three matrices (Poothong et al., 2017).

Earlier interlaboratory comparisons indicated quite large CVs, for example, 51% and 20% for PFOA and 24% and 32% for PFOS, in plasma samples (van Leeuwen et al., 2006; Longnecker et al., 2008). More recently, one interlaboratory comparison reported CVs ranging from 9% for PFOA and from 9% to 38% for PFOS isomers (van der Veen et al., 2023). An interlaboratory comparison and training exercise carried out for four rounds, involving 21 laboratories across Europe, included several PFAS (Nübler et al., 2022). For PFOA, the relative standard deviation improved from 12% to 6% from the second to the fourth round, and the relative standard deviation for PFOS was 11–12% in both rounds. [The study by van Leeuwen et al. (2006) was nearly 20 years old and involved the use of different extraction
Table 1.6 Selected analytical methods for the measurement of PFOA and PFOS in soil, sediment, dust, consumer products, and foods

Sample matrix	Sample preparation	Instrument (LOD) ^a	Reference
Soil	Solvent extraction with methanol:water (50:50) under basic conditions (pH ~9–10, adjusted with ~20 μ L NH ₄ OH), followed by filtration, and acidification (pH ~3–4, adjusted with ~50 μ L acetic acid)	HPLC-MS/MS (PFOA, 6.2 ng/g; PFOS, 18.8 ng/g)	<u>ASTM International</u> (2017)
Soil, sediment, and sludge (PFOA)	Solvent extraction with acetonitrile/0.2 M NaOH	HPLC-MS/MS (PFOA, 1 ng/g)	<u>Powley et al. (2005)</u>
Soil (PFOA)	Ultrasonic extraction with acetonitrile/water mixture	HPLC-MS/MS (PFOA, 180 fg on column)	<u>Washington et al.</u> (2008)
Soil and biosolids	Ultrasonic extraction with methanol containing 1% $\rm NH_4OH$	HPLC-MS/MS (0.02-0.5 ng/g)	<u>Sepulvado et al.</u> <u>(2011)</u>
Soil and riverine sediment	Ion pair extraction with 0.5 M TBAS and 0.25 M sodium carbonate buffer (pH 10)	HPLC-MS/MS (soil: PFOA, 0.34 ng/g; PFOS, 0.32 ng/g; sediment: PFOA, 0.30 ng/g)	<u>Lorenzo et al. (2015)</u>
Sediment and sludge	Ultrasonic extraction with methanol and 1% acetic acid	HPLC-MS/MS (sediment: PFOA, 0.01 ng/g; PFOS, 0.1 ng/g; sludge: PFOA, 1.0 ng/g; PFOS, 0.9 ng/g)	<u>Higgins et al. (2005)</u>
Marine sediment	Ultrasonic extraction with methanol	HPLC-MS/MS (PFOA, 0.01 ng/g; PFOS, 0.05 ng/g)	<u>Wang et al. (2018b)</u>
Lake sediment	Solvent extraction with acetonitrile/0.2 M NaOH	HPLC-MS/MS (PFOA, 0.02 ng/g; PFOS, 0.05 ng/g)	<u>Guo et al. (2016)</u>
Marine plastic litter	Ultrasonic extraction with hexane	HPLC-MS/MS (PFOA, 0.03 ng/g; PFOS, 0.01 ng/g)	<u>Gómez et al. (2021)</u>
Sewage sludge	Ion pair extraction with 0.5 M TBAS and 0.25 M sodium carbonate buffer (pH 10)	HPLC-MS/MS (PFOA, 0.6 ng/g; PFOS, 5 ng/g)	<u>Zhang et al. (2010)</u>
Asphalt	Ultrasonic extraction with methanol and 1% $\rm NH_4OH$	HPLC-MS/MS (PFOA, 0.6 ng/g; PFOS, 0.7 ng/g)	<u>Srivastava et al.</u> (2022)
Indoor dust	Ultrasonic extraction with acetonitrile	HPLC-MS/MS (PFOA, 2.3 ng/g; PFOS, 4.6 ng/g)	<u>Kubwabo et al.</u> (2005)
Indoor dust	Solvent extraction with methanol followed by filtration	Online SPE-HPLC-TOF/MS (PFOA, 0.03 ng/g; PFOS, 0.01 ng/g)	<u>Padilla-Sánchez and</u> <u>Haug (2016)</u>
Home garden produce (e.g. tomato, pepper, apples)	Dispersive SPE using magnesium sulfate and acetonitrile with 1% $\rm NH_4OH$	HPLC-MS/MS (PFOA, 0.03 ng/g; PFOS, 0.01 ng/g)	<u>Scher et al. (2018)</u>
Food (various items)	Solvent extraction with acetonitrile plus formic acid, followed by QuEChERS clean- up; further SPE clean-up on WAX sorbent cartridges is required for complex food matrices	HPLC-MS/MS (PFOA, 0.012–0.024 ng/g; PFOS, 0.007–0.028 ng/g)	US FDA (2021b) (Validated US FDA method number C-010.02)
Food (various items)	Ultrasonic extraction with acetonitrile plus NaOH, followed by clean-up on WAX sorbent cartridges	NanoLC – Orbitrap MS (PFOA, 0.001–0.3 ng/g; PFOS, 0.001–0.3 ng/g)	<u>Zacs et al. (2023)</u>

Tab	le 1	.6	(con	tinu	ed)
IUN		••	(2011		,

Sample matrix	Sample preparation	Instrument (LOD) ^a	Reference
Microwave paper packaging	FUSLE with ethanol	HPLC-QTOF/MS (PFOA, 1.53 ng/g; PFOS, 0.63 ng/g)	<u>Monge Brenes et al.</u> (2019)
Consumer products (papers and textiles)	Solvent extraction with methanol	HPLC-MS/MS (papers: PFOA, 0.040 μg/m ² , PFOS, 0.038 μg/m ² ; textiles: PFOA, 0.12 μg/m ² ; PFOS, 0.15 μg/m ²)	<u>Robel et al. (2017)</u>
Consumer products (e.g. waterproofing agents, textiles, paints, cookware, waterproofing agents, firefighting foams, electronics)	Ultrasonic extraction with methanol	HPLC-QTOF/MS (NR)	<u>Herzke et al. (2012)</u>
Consumer products (e.g. textiles (outdoor materials), carpets, cleaning and impregnating agents, leather samples, baking and sandwich papers, paper baking forms and ski waxes)	Depending on the matrix procedures using ion pair extraction, acidic–alkaline sequential extraction or SPE with WAX were applied	HPLC-MS/MS (0.1–0.5 ng/g)	Kotthoff et al. (2015)

FUSLE, focused ultrasonic liquid extraction; HPLC, high-performance liquid chromatography; LOD, limit of detection; MS, mass spectrometry; MS/MS, tandem mass spectrometry; NaOH, sodium hydroxide; NH₄OH, ammonium hydroxide; NR, not reported; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; QTOF, quadrupole time-of-flight; QuEChERS, Quick, Easy, Cheap, Effective, Rugged, and Safe; SPE, solid-phase extraction; TBAS, tetrabutylammonium hydrogen sulfate; TOF, time-of-flight; WAX, weak anion exchange. ^a Using electrospray ionization (ESI) in negative ion mode.

and instrumental techniques, which led to the large variation and high z-scores.]

Reported serum-to-plasma ratios for PFOA and PFOS were approximately 1:1, whereas serum- or plasma-to-whole blood ratios were approximately 2:1 (Ehresman et al., 2007; Poothong et al., 2017). In the past, total PFOS was normally presented, but in more recent publications PFOS isomers have been distinguished, separating linear and the sum of branched forms; LODs in serum have also improved (e.g. Li et al., 2022c) (see Table 1.7). The method reported by Li et al. (2022c) can be applied for the analysis of PFOA and PFOS in urine; the resulting LODs were 0.01 ng/mL for PFOA and 0.01–0.02 ng/mL for PFOS isomers (Li et al., 2022c). Similar methods are used for breast milk or colostrum. Existing methods for sample preparation and analysis of PFAS concentrations in human breast milk were reviewed by Macheka-Tendenguwo et al. (2018). SPE is more popular, owing to higher recovery, shorter analysis times, simpler procedures, and less use of solvents (e.g. Kärrman et al., 2007; Abdallah et al., 2020) than in other techniques, such as liquid–liquid extraction (LLE). The LOQ for each PFAS in colostrum and breast milk has been reported as 0.01 ng/mL. In two replication sets with in-house controls (n = 6 each), relative standard deviations were 28% and 11.1% for PFOA, and 20.2% and 8.8% for PFOS, respectively (Blomberg et al., 2023).

Sample matrix	Sample preparation	Instrument (LOD)	Comments	Reference
Whole blood	Adding of labelled internal standards; solvent extraction with acetonitrile; carbon–acetic acid, filtration; addition of performance standards ${}^{13}C_{8}$ - PFOA and ${}^{13}C_{8}$ -PFOS, with 2 mM ammonium acetate	HPLC-MS/MS (PFOA, 0.4–0.7 ng/mL; PFOS, 0.01–0.1 ng/mL)		<u>Hardell et al. (2014)</u>
Plasma and serum	Protein precipitation with acetonitrile; ¹³ C-labelled PFOA internal standards	LC-MS/MS (PFOA LOQ, 0.5 ng/mL)	Validated to meet US FDA guidelines for bioanalytical methods	<u>Flaherty et al. (2005)</u>
Plasma	Labelled internal standards; protein precipitation with acetonitrile; shaking, centrifugation	LC-MS/MS (PFOA, 0.4 ng/mL; PFOS, 0.5 ng/mL)		<u>Li et al. (2018)</u>
Plasma	Protein precipitation with acetonitrile; reconstitution in MeOH; filtration	HPLC-ESI-MS/MS (PFOA LOQ, 0.5 ng/mL; PFOS LOQ, 0.1 ng/mL)		<u>Tsai et al. (2020)</u>
Plasma	Addition of ¹³ C-labelled PFAS compounds; addition of acetonitrile to precipitate proteins; vortex mixing, centrifugation	LC-HRMS (PFOA, 0.01 μg/L; PFOS, 0.43 μg/L)		<u>Goodrich et al. (2022)</u>
Plasma, serum, and whole blood	Protein precipitation with MeOH, mixing, centrifugation	HPLC-MS/MS (PFOS, 0.009 ng/mL; PFOA, 0.009 ng/mL plasma; 0.018 ng/mL serum; 0.045 ng/mL whole blood)	Validated for human plasma, serum, and whole blood	<u>Poothong et al. (2017)</u>
Serum	Sample with internal standard and TBAS solution mixed; MTBE added and shaken; centrifugation; separation ×2; reconstitution in MeOH; vortex mixing; filtration	HPLC-MS/MS (PFOA, 1.0 ng/mL; PFOS, 1.7 ng/mL)		<u>Hansen et al. (2001)</u>
Serum	Ion-pair extraction	HPLC-MS/MS (PFOA, 10 ng/mL)		<u>Sottani and Minoia</u> (2002)
Serum	Proteins precipitated with formic acid; SPE clean-up	HPLC-MS/MS (PFOA, 0.2 ng/mL; PFOS, 0.2 ng/mL)		<u>Kuklenyik et al. (2005)</u>
Serum	Acidification with HCl, addition of hexanoic acid and THF; vortex-shaking, centrifugation	LC/QQQ MS/MS (2–20 pg/mL)		<u>Luque et al. (2012)</u>
Serum	Dilution with ultrapure water and isotope internal standards in MeOH, centrifugation	HPLC-MS/MS (PFOA, 0.023 ng/mL; PFOS, 0.033 ng/mL)		<u>Gao et al. (2018)</u>
Serum	Alkaline digestion followed by two-stage SPE purification using polymeric HLB and graphitized non- porous carbon cartridges	LC-MS/MS (NR)	Fully validated (2002/657/CE decision) and accredited (ISO 17025 standard)	<u>Mancini et al. (2020)</u>

Table 1.7 Selected analytical methods for the measurement of PFOA and PFOS in human biospecimens

Table 1.7 (continued)

Sample matrix	Sample preparation	Instrument (LOD)	Comments	Reference
Serum	Precipitation using acetonitrile by vigorous shaking; all sample batches include chemical blanks and three quality control samples	LC-MS/MS (PFOA, 0.09 ng/mL; <i>n</i> -PFOS, 0.2 ng/mL; 3/4/5m-PFOS, 0.01 ng/mL)		<u>Li et al. (2022c)</u>
Breast milk, serum	Proteins precipitated with formic acid; SPE clean-up	HPLC-MS/MS (PFOA, 0.2 ng/mL milk; 0.1 ng/mL serum; PFOS, 0.3 ng/mL milk; 0.4 ng/mL serum)		<u>Kuklenyik et al. (2004)</u>
Breast milk	LLE; purification by two successive SPE; reconstitution in fluorometholone solution as external standard in MeOH/water	LC-HRMS (PFOA, 0.003 ng/mL; PFOS, 0.002 ng/mL)		<u>Kadar et al. (2011)</u>
Breast milk	LLE with acetonitrile; purification by dispersive SPE using C18 sorbent; shaking and centrifugation; reconstitution in MeOH; filtration	HPLC-MS/MS (PFOA LOQ, 0.006 ng/mL; <i>n</i> -PFOS LOQ, 0.005 ng/mL; br-PFOS LOQ, 0.010 ng/mL)		<u>Lankova et al. (2013)</u>
Semen, serum	Samples were spiked with mass-labelled extraction standard, TBAS solution, NaHCO ₃ /Na ₂ CO ₃ buffer solution and MTBE; shaking; extraction $\times 2$ with MTBE; all three extracts combined, evaporated to dryness under nitrogen at 40 °C, and reconstituted with MeOH	HPLC-MS/MS (PFOA LOQ, 0.004–0.010 ng/mL semen; 0.020 ng/mL serum; PFOS LOQ, 0.004–0.010 ng/mL semen; 0.020 ng/mL serum)		<u>Pan et al. (2019)</u>
Urine and serum	For urine, add isotope-labelled internal standard and ammonium acetate buffer including β-glucuronidase, and subsequently formic acid; for serum, isotope- labelled internal standard was added, and formic acid; samples vortexed	SPE-HPLC-MS/MS (PFOA, 0.1 ng/mL; PFOS, 0.1 ng/mL)		<u>Kato et al. (2018)</u>
Urine	Precipitation using acetonitrile by vigorous shaking for 30 min; all sample batches include chemical blanks and three quality control samples	LC-MS/MS (PFOA, 0.01 ng/mL; <i>n</i> -PFOS; 0.01 mL; 3/4/5m-PFOS, 0.02 ng/mL)		<u>Li et al. (2022c)</u>
Hair, nail, urine, serum	For hair and nails: soaking in water, washing twice with acetone, air-drying, grinding to powder, extraction by various organic solvents, cleaning by WAX cartridge, elution with 9% NH_4OH in MeOH, concentration to dryness under nitrogen gas and reconstitution in water/MeOH (v/v; 1/1), filtration	HPLC-MS/MS (PFOA, 0.03 ng/g hair; 0.04 ng/g nail; 0.02 ng/mL serum; 1.07 ng/L urine; PFOS, 0.03 ng/g hair; 0.05 ng/g nail; 0.02 ng/mL serum; 2.09 ng/L urine)		<u>Wang et al. (2018a)</u>

Table 1.7 (continued)

Sample matrix	Sample preparation	Instrument (LOD)	Comments	Reference
Dried blood spots	Punch samples desorbed in ultrapure water, sonicated, and extracted into MTBE with labelled internal standards; dried then reconstituted in MeOH	HPLC-ESI-MS/MS (PFOA, 0.4 ng/mL; PFOS, 0.2 ng/mL)	LODs expressed in units of whole blood equivalents	<u>Spliethoff et al. (2008)</u>
Dried blood spots	Punch samples desorbed into MeOH with labelled internal standards; mixed, sonicated, centrifuged	SPE-HPLC-MS/MS (PFOA, 0.0075 ng/mL; PFOS, 0.03 ng/mL)		<u>Poothong et al. (2019)</u>
Placental tissue	Shaking with MeOH and MPFOA for 5 min, freeze- drying, homogenized with acetonitrile, centrifuged	HPLC-MS/MS (PFOA, 0.03 ng/g; PFOS, 0.03 ng/g)	Linearity, selectivity, accuracy (trueness and precision) and sensitivity validated according to US FDA guidelines	<u>Martín et al. (2016)</u>

br-, branched chain; C18, octadecyl alkyl substituent; ESI, electrospray ionization; HCl, hydrochloric acid; HLB, hydrophilic-lipophilic-balanced; HPLC, high-performance liquid chromatography; ISO, International Organization for Standardization; LC-HRMS, liquid chromatography-high-resolution mass spectrometry; LC-MS/MS, liquid chromatography-tandem mass spectrometry; LC/QQQ-MS/MS, liquid chromatography/triple quadrupole-tandem mass spectrometry; LOD, limit of quantification; LLE, liquid–liquid extraction; MeOH, methanol; min; minute(s); MPFOA, perfluoro-*n*-[1,2,3,4-¹³C₄]octanoic acid; MS/MS, tandem mass spectrometry; MTBE, methyl *tert*-butyl ether; NaHCO₃/Na₂CO₃, sodium bicarbonate/sodium carbonate; NH₄OH, ammonium hydroxide; NR, not reported; PFAS, per- and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; 3/4/5m-PFOS, corresponds to the sum of branched isomers 3m-PFOS, 4m-PFOS, and 5m-PFOS; SPE, solid-phase extraction; TBAS, tetra-*n*-butylammonium hydrogen sulfate; THF, tetrahydrofuran; US FDA, United States Food and Drug Administration; v/v, volume per volume; WAX, weak anion exchange.

Thomsen et al. (2010) reported a different method, with internal standards and acetonitrile added, followed by mixing and centrifugation. After the addition of formic acid, the supernatant is analysed by online column-switching LC-MS/MS.

Dried blood spots have been used to assess PFOA and PFOS exposure (<u>Spliethoff et al.</u>, <u>2008</u>). Detection limits as low as 0.0075 ng/mL for PFOA and 0.030 ng/mL for PFOS, estimated for the corresponding serum concentrations, have been reported.

PFOA and PFOS concentrations have also been measured in placental tissue (<u>Martín et al., 2016</u>), hair, and nails (<u>Wang et al., 2018a</u>).

1.4 Occurrence and exposure

Introduction to occurrence and exposure

(a) Life cycle and practices involved in end-oflife and disposal

The occurrence of PFOA and PFOS in the environment is influenced by the chemical life cycle, including during fluorochemical production; secondary manufacturing processes (e.g. products containing fluorochemicals); product use; and management of waste (industrial waste, products containing PFOA and PFOS, and materials contaminated with PFOA or PFOS) (see Fig. 1.3). The presence of PFOA and PFOS in consumer and industrial products, as well as environmental media subject to remediation, creates avenues for inadvertent, repeated cycles of contamination (Stoiber et al., 2020).

The available approaches to managing large quantities of PFAS wastes include land-filling, incineration, and wastewater treatment (<u>US EPA, 2020</u>). Landfills have been used historically for disposal at perfluorochemical facilities (<u>ATSDR, 2021</u>). The presence of PFAS in leachate from landfills has been documented in several countries, including Australia, China, Germany, and the USA (Stoiber et al., 2020). Incineration of products containing PFOA or PFOS generally requires temperatures of > 800 °C, using a scrubber to remove hydrogen fluoride. Although limited, experimental studies have indicated that incineration can break down PFOA and PFOS (Stoiber et al., 2020; ATSDR, 2021). Liquid wastes are treated with precipitation, decanting, or filtering to separate solids, followed by landfill or incineration of the solids and discharge of the liquids to a wastewater treatment facility (ATSDR, 2021). The US EPA interim guidance also lists underground injection as a possible means of disposal (US EPA, 2020).

In some settings, PFOA- or PFOScontaminated waste products, including food wastes and sludge from municipal wastewater treatment, have been dispersed over land, for example, by land application of biosolids or composts (Kenny, 2021; ITRC, 2022a). Land application of these products may contribute to the contamination of crops and livestock and the continued cycle of contamination (Stoiber et al., 2020; Kenny, 2021).

[The Working Group acknowledged that in geographical regions with restrictions and phase-out of PFOA and PFOS production and use (e.g. Europe and the USA), trends towards decreases in PFOA and PFOS concentrations in human biospecimens (mainly in serum) have been observed (see Section 1.4.3); however, no clear patterns of declining trends have been observed for abiotic and environmental samples from the same regions. Decreasing concentrations in humans may be influenced by the removal of certain PFAS from consumer products and associated reductions in direct exposure (Land et al., 2018). Persistent levels in the environment may reflect the re-circulation of historically manufactured and released PFOA and PFOS and potentially the breakdown of their precursors.]

Fig. 1.3 Life cycle of PFOA and PFOS



PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid. Adapted from <u>European Environment Agency (2021)</u>.

(b) Persistence and mobility

The carbon-fluorine bond is one of the strongest bonds known in nature and makes PFAS extremely resistant to degradation in the natural environment. PFOA and PFOS are among the most environmentally persistent organic chemicals and are, therefore, under the Regulation on the Registration, Evaluation, Authorisation and Restriction of Chemicals (REACH) definition for persistence, classified as "very persistent (vP)" (Cousins et al., 2020).

Unlike other known persistent organic pollutants, PFOA and PFOS are highly mobile in the environment. They are quite soluble in water, and thus can be carried to remote regions through oceanic currents and long-range atmospheric currents. They can also vertically infiltrate sediment layers and move across the water column (ECHA, 2023).

(c) Global and temporal trends

PFOA and PFOS have been detected in environmental media worldwide, including in remote areas like the Arctic, Antarctic, and Mount Everest (Cai et al., 2012; ATSDR, 2021; Miner et al., 2021; Garnett et al., 2022). Estimations of total global annual emissions of PFOA-based products show that emissions steadily increased from 1960 to 2002 and quickly decreased from 2002 to 2012, followed by an increase from 2012 to 2015. The same trend was observed for PFOSbased products (OECD, 2015a). The estimated oceanic transport of PFOA to the Arctic for the period 1951–2004 was greater than the estimated atmospheric transport (Prevedouros et al., 2006). The deposition into soil from the atmosphere and subsequent transport pathways, such as leaching, also contribute to the widespread distribution of these substances in the environment (ATSDR, <u>2021</u>). Retention by soil is expected to be low (Prevedouros et al., 2006). In the environment, most PFOA and PFOS are estimated to be in ocean water, and smaller amounts are present in freshwater and sediments. The presence of PFOA and PFOS in groundwater is widespread (Johnson et al., 2022).

The presence of PFOA and PFOS in snow and ice core samples indicates their atmospheric deposition from production and/or use (see Section 1.4(c)(i) below). Likewise, sediment cores reflect time trends corresponding to initial production and subsequent changes in patterns of production and use (Section 1.4(c)(ii)).

(i) Snow and ice cores

In a snow core from the Mount Muztagata glacier (western Tibet, China) showed a steady increase in PFOA and PFOS from 1983 to 1999. A more recent (1996–2007) core from Mount Zuoqiupo glacier (south-eastern Tibet) contained lower concentrations of PFOA and PFOS, with no clear trend. Differences in concentrations were attributed to different upwind sources affecting the respective study sites (e.g. sources in Europe or central Asia for Mount Muztagata and sources in India for Mount Zuoqiupu) (Wang et al., 2014).

In glacial ice cores from Svalbard, Norway, representing deposition from 1990 to 2005, higher concentrations of PFOA and PFOS were detected in the layers representing 1997–2000, the period that coincides with the peak production of these compounds (Kwok et al., 2013).

In the eastern Antarctic, a firn core representing the period from 1958 to 2017, showed PFOA levels peaking in 1997–2000. Subsequently there was a short decline, then an increase from 2003 to 2013 with no sign of a decrease, despite recent global restrictions on PFOA production and use (<u>Garnett et al., 2022</u>). [The Working Group noted this may be attributed to increasing production of fluorochemicals in emerging Asian economies, which probably offsets emission reduction in North America and Europe, and may account for the higher concentrations observed in the later years represented in the firn core. PFOS was not detected in any of the studied samples.]

(ii) Sediment cores

In a sediment core containing deposits from the 1950s to 2004, in Tokyo Bay, Japan, concentrations of PFOA increased consistently from 1994 to 2004, which is generally consistent with the PFOA production and usage profile during this period in Japan (Zushi et al., 2010). PFOS concentrations decreased gradually after the early 1990s, whereas concentrations of some PFOS precursors decreased rapidly in the late 1990s. This trend could reflect the shift in PFOS industrial production processes after the phase-out of POSF-based products in 2001 (Zushi et al., 2010). Another study on three sediment cores from Lake Ontario, Canada, (1952-2005) reported a marked increase in PFOA and PFOS concentrations from the mid-1970s to 2005, which is generally in line with PFOA and PFOS production and usage profiles (Yeung et al., 2013). In a sediment core from the Bering Sea, covering almost 70 years of deposition, PFOS concentrations generally showed an upward trend since 1952 and peaked in about 2003, after which concentrations dropped to a lower level until 2015. This largely coincides with the production and usage history of PFOS. Conversely, PFOA concentrations showed a more fluctuating pattern among layers, which was explained by its vertical mobility in pore water (Lin et al., 2020a). [The Working Group noted that although the temporal trends in PFOS concentrations in dated sediment cores reflect PFOS production and usage history, temporal trends in PFOA concentrations can be influenced by its vertical mobility in pore water and thus may not adequately reflect changes in its production and use in certain geographical areas.]

(d) Precursor compounds

In the present monograph, "precursor compounds" refers to PFAS that are known to break down or transform into PFOA or PFOS in the environment or biota, including humans. Precursors include, but are not limited to, fluorotelomer alcohols (FTOH) and polyfluoroalkyl phosphate diesters (diPAP) for PFOA; and perfluorooctane sulfonamides (e.g. N-EtFOSA), perfluorooctane sulfonamidoacetic acids (e.g. N-EtFOSAA) and perfluorooctane sulfonamidoethanols (e.g. N-EtFOSE) for PFOS (Gebbink et al., 2015). While estimates vary by exposure scenario, it has been estimated that a substantial proportion of the body burden of PFOA and PFOS may originate from intake of precursors (Vestergren et al., 2008; Gebbink et al., 2015) (see also Section 4.1). While direct exposure to PFOA and PFOS may decline as a result of regulation or voluntary efforts, production and use of precursors may contribute to ongoing exposure from the breakdown of precursors. Breakdown of precursors has also resulted in PFOA and PFOS contamination in remote areas with no direct sources of pollution (ATSDR, 2021).

1.4.1 Environmental occurrence

(a) Air and dust

The atmospheric environment is not only an important compartment for the transport of PFOA and PFOS, but it is also an exposure pathway for PFOA and PFOS (Liu et al., 2018a). Air is a mixture of particles, gases, and dust. The sources and levels of PFOA and PFOS in outdoor and indoor air differ, and the characteristics of PFOA and PFOS are described here for outdoor air, indoor air, and settled dust separately.

(i) Outdoor air

The sources of PFOA and PFOS in outdoor air include direct emissions from the fluorochemical industry and products containing fluorochemicals (<u>Butt et al., 2010</u>), long-range transport via the gas phase, and degradation of PFAS precursors (<u>McMurdo et al., 2008</u>). Sampling time and location varied among multiple studies; representative concentrations are presented in <u>Table 1.8</u>.

1
4
2
()
>
\geq
\circ
7
$\overline{}$
ğ
ር)
R
\geq
P
T
S
1
<u> </u>
ω
01

Reference	Location and	Characteristics	Analytical method (reporting limits)	PFC	A	PF	OS	Unit	Comments
	collection date	illection of sampling ite (number, sites, sampler, time, flow, and duration)		Mean (range)	Median (IQR)	Mean (range)	Median (IQR)		
Camoiras González et al. (2021)	15 countries in Africaª, 2017–2019	118, meteorological station, PAS, 3 mo/sample, 2 yr	LC-MS/MS, (LOQ: PFOA, 13 pg/PUF disc; PFOS, 12 pg/PUF disc)	207 (< 13-1190)	148	185 (< 12–2480)	97.7	pg/ PUF disc	Long sampling time. Good reflection of PFAS levels in Africa.
	7 countries in Asia ^b , 2017–2019	46, meteorological station, PAS, 3 mo/sample, 2 yr		271 (83.1–965)	183	139 (27.3–634)	101		Long sampling time. Good reflection of PFAS levels in Asia.
	10 countries in Group of Latin America and Caribbean ^c , 2017–2019	101, meteorological station, PAS, 3 mo/sample, 2 yr		257 (58.9–655)	233	376 (< 12-2260)	192		Long sampling time. Good reflection of PFAS levels in Group of Latin America and Caribbean countries.
	9 countries in Pacific Islands subregion ^d , 2017– 2019	43, meteorological station, PAS, 3 mo/sample, 2 yr		181 (< 13-417)	165	297 (< 12-827)	266		Long sampling time. Good reflection of PFAS levels in Pacific Islands subregion.

Ta Re

Table 1.8	Table 1.8 (continued)								
Reference	Location and	Characteristics	Analytical method	PFC	DA	PF	OS	Unit	Comments
	collection date	llection of sampling te (number, sites, sampler, time, flow, and duration)	(reporting limits)	Mean (range)	Median (IQR)	Mean (range)	Median (IQR)		
<u>Chaemfa</u> <u>et al.</u> (2010)	UK, July to October 2007	15, background and city centre area, PAS, 2–3 mo/sample	LC-TOF-MS (LOD: PFOA, 27 pg/sample; PFOS, 3.9 pg/sample)	[2657 (< 27–27 000)]	[400]	[53.5 (< 3.9–720)]	[6.5]	pg/ sample per day	Long sampling time. Good reflection of PFAS levels in north-western England.
	UK–Norway, June to October 2006	11, background and semi-rural/ rural area, PAS, 2–3 mo/sample		[139 (< 27–1200)]	[< 27]	[3.0 (< 3.9-7.7)]	[< 3.9]		Long sampling time. Good reflection of PFAS levels in UK–Norway transect.
	Europe, June to November 2006	23, ranged from background to city centre area, PAS, 2–3 mo/sample		[117 (< 27–540.0)]	[< 27]	[10 (< 3.9–69.0)]	[< 3.9]		Long sampling time. Good reflection of PFAS levels in Europe.
<u>Dreyer</u> <u>et al.</u> (2015)	Geesthacht, Germany, December 2007 to May 2008	11, semirural, AAS, 14– 21 days/sample, 2 sample/mo	HPLC-MS/ MS, (LOQ: PFOA, 10 pg/sample; PFOS, 10 pg/sample)	0.7 (0.1-4.8)	NR	0.65 (0.2–3.5)	NR	pg/m³	Sampling time in each month was relatively limited. Partially reflects PFAS levels in semirural area.

$\overline{>}$
R
\cap
\geq
0
2
0
G
R
\geq
Ō
Ť
S
T
135
.

Table 1.8 (continued)

Reference	Location and	Characteristics	Analytical method	PFOA		PF	OS	Unit	Comments
	date	of sampling (number, sites, sampler, time, flow, and duration)	(reporting limits)	Mean (range)	Median (IQR)	Mean (range)	Median (IQR)		
<u>Guo et al.</u> (2018)	Shanghai, China, December 2013 to January 2015	18, urban area (reflects long-range transported PFAS from northern or eastern continental China and surrounding seas), AAS, 24 h/sample, 28.3 L/min	HPLC-MS/ MS, (LOD: PFOA, 0.35 pg/L; PFOS, 1.30 pg/L)	145.6	101.0 (71.3–230.0)	24.2	24.1 (14.2–29.0)	pg/m³	Sampling time collected in every month was limited. Partially reflects PFAS levels in urban area in winter.
<u>Lin et al.</u> (2020a)	Xiamen, China, December 2016 to September 2018	13, eastern coastal China and commercial/ residential area, AAS, 2–3 days/sample, 0–1 sample/mo, 20 L/min	LC-MS/MS (minimum MQL: PFOA, 0.089 pg/m ³ ; PFOS, 0.174 pg/m ³)	[3.21] (0.211–7.47)	[0.72]	[2.79] (< 0.315-15.7)	[1.50]	pg/m³	Sampling time in each month was relatively limited. Partially reflects PFAS levels in commercial/ residential area in Xiamen.
	Delhi, India, December 2017 to May 2018	2, commercial and residential area, AAS, 2–3 days/sample, 0–1 sample/mo, 20 L/min		[0.42] (< 0.367–1.07)	[0.38]	[0.63] (ND to 1.33)	[0.61]		Sampling time in each month was relatively limited. Partially reflects PFAS levels in commercial/ residential area in Delhi.

Table 1.8 (continued)									
Reference	Location and	Characteristics	Analytical method	PFO	A	PF	OS	Unit	Comments
	collection date	of sampling (number, sites, sampler, time, flow, and duration)	(reporting limits)	Mean (range)	Median (IQR)	Mean (range)	Median (IQR)		
Lin et al. (2020a) (cont.)	Beijing, China, May 2017 to January 2018	7, rural (surrounded by forest; near some residents), AAS, 2–3 day/ sample, 0–1 sample/ month, 20 L/min		[0.68] (< 0.182–2.81)	[0.41]	[0.53] (< 0.350-1.16)	[0.41]		Sampling time in each month was relatively limited. Partially reflects PFAS levels in rural area in Beijing.
	Yuxi, China, August 2016 to April 2017	7, rural (fewer residents and low traffic density), AAS, 5 days/sample, 0–1 sample/mo, 20 L/min		[0.12] (< 0.091–0.393)	[0.07]	[0.11] (ND to 0.209)	[0.09]		Sampling time in each month was relatively limited. Partially reflects PFAS levels in rural area in Yuxi.
	Wenchuan, China, May 2017 to October 2017	3, rural (mountain areas), AAS, 2–3 days/sample, 0–1 sample/mo, 20 L/min		[0.70] (0.365–1.22)	[0.66]	[0.57] (ND to 1.37)	[0.47]		Sampling time in each month was relatively limited. Partially reflects PFAS levels in mountain area in Wenchuan.
	Tsukuba, Japan, July to December 2017	5, rural (fewer residents and low traffic density), AAS, 4 days/sample, 0–1 sample/mo, 20 L/min		[0.51] (< 0.124–3.01)	[0.28]	[0.19] (< 0.24-0.709)	[0.14]		Sampling time in each month was relatively limited. Partially reflects PFAS levels in rural Japan.

Þ
R
\cap
2
\geq
$\underline{\circ}$
~
0
G
R
\geq
Ρ
Т
S
Ι
<u> </u>
ß
••••

Table 1.8 (continued)

Reference	Location and	Characteristics	Analytical method	PFOA		PFOS		Unit	Comments
	collection date	of sampling (number, sites, sampler, time, flow, and duration)	(reporting limits)	Mean (range)	Median (IQR)	Mean (range)	Median (IQR)		
Lin et al. (2020a) (cont.)	Jinju, Republic of Korea, April 2017 to January 2018	6, rural (fewer residents and low traffic density), AAS, 3-4 days/sample, 0-1 sample/mo, 20 L/min		[1.47] (0.212–7.84)	[0.65]	[0.39] (ND to 1.16)	[0.38]		Sampling time in each month was relatively limited. Partially reflects PFAS levels in rural Jinjiu.
	Nanjing, China, September 2017 to July 2018	7, urban (industrial area), AAS, 3–5 day/sample, 0–1 sample/mo, 20 L/min		[5.71] (0.695–26.8)	[2.73]	[2.10] (ND to 17.1)	[0.76]		Sampling time in each month was relatively limited. Partially reflects PFAS levels in industrial area in Nanjing.
	Gujarat, India, December 2016 to November 2017	12, urban (western coastal India, residential area), AAS, 2–5 days/sample, 0–1 sample/mo, 20 L/min		[0.28] (ND to 2.06)	[0.17]	[0.37] (ND to 1.81)	[0.35]		Sampling time in each month was relatively limited. Partially reflects PFAS levels in residential area in Gujarat.
<u>Lin et al.</u> (2022)	Karachi, Pakistan, December 2012 to January 2013	18, urban (near industrial area and garbage dumping sites), AAS, 24 h/sample, 16.7 L/min	LC-MS/MS, (MQL: PFOA, 1.0 pg/m ³ ; PFOS, 0.2 pg/m ³)	2.01 (0.85-8.70)	1.6	1.69 (0.64–3.17)	1.55	pg/m³	Sampling time was relatively long. Reflects PFAS levels in urban areas.

Table 1.0											
Reference	Location and	Characteristics	Analytical method (reporting limits)	PFOA		PF	OS	Unit	Comments		
	date	(number, sites, sampler, time, flow, and duration)		Mean (range)	Median (IQR)	Mean (range)	Median (IQR)				
<u>Liu et al.</u> (2023)	Pearl River Delta, China, May to July and October to December 2018	186, urban, AAS, 24 h/sample, 100 /min in summer and 1.05 m ³ /min in winter	HPLC-MS/MS (LOD: PFOA, 0.0025 ng/mL; PFOS, 0.0003–0.0016 ng/mL)	10.80 (1.02–56.53)	6.05 (3.71–13.04)	45.19 (3.90–378.06)	24.18 (11.94-44.18)	pg/m ³	Sampling time for each sample was limited, but sample sites were representative and sample size was large. Partially reflects PFAS levels in an urban area in the Pearl River Delta.		
<u>Seo et al.</u> (2019)	Hyung- san River, Gyeongju and Pohang, Republic of Korea, September 2014	8, urban (near wastewater treatment plants), AAS, 18–24 h/sample, 700 L/min	LC-MS/MS (MDL: PFOA, 0.13 pg/m ³ ; PFOS, 0.13 pg/m ³)	48.66	43.09	90.52	99.03	pg/m³	Total sampling time was long. Reflects PFAS levels near wastewater treatment plants in the Republic of Korea.		
<u>Wang</u> <u>et al.</u> (2021)	Shandong, China, November 2017	12, urban (fluorochemical industry park), AAS, 20 h/sample, 800 L/min	LC-MS (LOD: PFOA, 0.06 pg/m ³ ; PFOS, 0.13 pg/m ³ ; LOQ: PFOA, 0.31 pg/m ³ ; PFOS, 0.31 pg/m ³	1610 (42.8–9730)	451	1.24 (< 0.31–2.74)	1.01	pg/m³	Long total sampling time. Reflects PFAS levels at source.		

Table 1.8 (continued)

Reference	Location and	Characteristics	Analytical method	PFOA		PFO	DS	Unit	Comments
	collection date	of sampling (number, sites, sampler, time, flow, and duration)	(reporting limits)	Mean (range)	Median (IQR)	Mean (range)	Median (IQR)		
<u>Yu et al.</u> (2018)	Coastal areas of the Bohai Sea, China, May 2015 to April 2016	48, urban (large emission of PFAS, economic zones), AAS, 48 h/sample, 2 sample/mo, 300 L/min	HPLC-MS/MS (LOD: PFOA, 0.01 pg/m ³ ; PFOS, 0.02 pg/m ³ ; LOQ: PFOA, 0.05 pg/m ³ ; PFOS, 0.05 pg/m ³)	27.0 (0.1–362.9)	[26.2] [(15.0–34.8)]	[1.8] (< 0.05–11.1)	[1.4] [(1.0–2.1)]	pg/m³	Sampling time in every month was relatively limited, but number of samples was large. Partially reflects PFAS levels in an urban area.
	Coastal areas of the Yellow Sea, China, May 2015 to April 2016	35, urban (large emission of PFAS, economic zones), AAS, 48 h/sample, 2 sample/month, 300 L/min		30.5 (0.6–524.8)	[18.3] [(13.3–46.1)]	[0.6] (< 0.05-8.6)	[0.8]) [(0.5–0.9)]		Sampling time in each month was relatively limited, but number of samples was large. Partially reflects PFAS levels in an urban area.
<u>Zhou</u> <u>et al.</u> (2021)	North Carolina, USA, 2018–2019	60, suburban residential areas and on or near university campuses, AAS, 6 days/sample, 3 mo, 10.0 L/min	HPLC-MS/ MS (LOD: PFOA, 0.0067 pg/m ³ ; PFOS, 0.0047 pg/m ³ ; MDL:PFOA,2.86pg/m ³ ; PFOS, 0.18 pg/m ³)	(< 0.005-14.06)	NR	(< 0.004-4.75)	NR	pg/m³	Long sampling time. Good reflection of PFAS levels in North Carolina.

AAS, active air sampler; h, hour(s); HPLC, high-performance liquid chromatography; IQR, interquartile range; LC, liquid chromatography; LOD, limit of detection; LOQ, limit of quantification; MDL, method detection limit; min, minute(s); mo, month(s); MQL, method quantification limit; MS/MS, tandem mass spectrometry; ND, not detected; NR, not reported; PAS, passive air sampler; PFAS, per- and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PUF, polyurethane foam; TOF, time-of-flight; UK, United Kingdom; USA, United States of America.

^a Including Democratic Republic of the Congo, Egypt, Ethiopia, Ghana, Kenya, Mali, Mauritius, Morocco, Nigeria, Senegal, Togo, Tunisia, Uganda, United Republic of Tanzania, Zambia.

^b Including Cambodia, Indonesia, Lao People's Democratic Republic, Mongolia, Philippines, Thailand, Viet Nam.

^c Including Antigua and Barbuda, Argentina, Barbados, Brazil, Chile, Colombia, Ecuador, Mexico, Peru, Uruguay.

^d Including Fiji, Kiribati, Marshall Islands, Niue, Palau, Samoa, Solomon Islands, Tuvalu, Vanuatu.

In general, PFOA and PFOS levels differ according to the surroundings of the sampling sites. Air collected near fluorochemical industrial sites (Yu et al., 2018; Wang et al., 2021), wastewater treatment plants (Seo et al., 2019), and in industrial areas (Lin et al., 2020a) was highly contaminated by PFOA and PFOS, whereas air collected from areas that were remote from exposure sources and had fewer residents and less traffic had lower levels of PFOA and PFOS (Lin et al., 2020a). For example, the average concentration of PFOA in air samples from the fluorochemical industry park in Shangdong, China, was 1610 pg/m³ (Wang et al., 2021); however, in coastal areas of the Bohai Sea, China, which is more than 100 km from the industry park, the average PFOA concentration was 27.1 pg/m³ (Yu et al., 2018). This concentration is higher than that in the Pearl River Delta, China, (average 10.8 pg/m^3) (Liu et al., 2023), a coastal area that is farther from fluorochemical industries than is the Bohai Sea.

PFOA and PFOS levels in outdoor air vary widely worldwide, and it is also difficult to compare concentrations when the results are expressed in different units. One study conducted in 2017-2019 used the same method for four regions in LMICs (Camoiras González et al., 2021). The median concentration of PFOA in Africa (148 pg/PUF disc) was similar to that in the Pacific Islands (165 pg/PUF disc), but lower than that in Asia (183 pg/PUF disc) and the Group of Latin America and the Caribbean (GRULAC) (233 pg/PUF disc). The median concentrations of PFOS in Africa (97.7 pg/PUF disc) and Asia (101 pg/PUF disc) were similar, higher levels were observed in GRULAC (192 pg/PUF disc), and the highest levels were found in the Pacific Islands (266 pg/PUF disc) (Camoiras González <u>et al., 2021</u>).

In Asia, PFOA and PFOS levels were mainly reported for samples from China, and some information was available from Pakistan, Japan, India, and the Republic of Korea. Concentrations of PFOA and PFOS in areas remote from exposure sources were similar in China, Japan, India, and the Republic of Korea, and most median concentrations were < 5 pg/m³ (Lin et al., 2020a, 2022). However, mean PFOA concentrations in highly polluted zones such as industrial areas or areas near fluorochemical industry parks varied from 23.8 pg/m³ to 1610 pg/m³, with a maximum of 9730 pg/m³ (Guo et al., 2018; Seo et al., 2019; Lin et al., 2020a; Wang et al., 2021).

Across Europe and the United Kingdom (UK), PFOA concentrations in outdoor air have been relatively low. According to sampling campaigns conducted in the UK, Norway, and other countries in Europe, more than half of the samples did not contain PFOA and PFOS at concentrations above the detection limits (27 pg/sample per day for PFOA and 3.9 pg/sample per day for PFOS), although in north-west England, the median values for PFOA and PFOS were 400 and 6.5 pg/sample per day, respectively (Chaemfa et al., 2010). Maximum concentrations of PFOA and PFOS in the particle phase measured in Geesthacht, Germany, were both < 5 pg/m³ (Dreyer et al., 2015).

Available data on PFOA and PFOS levels in outdoor air in other countries or regions including the USA were limited. For example, the production of PFOA and PFOS was phased out in the USA nearly 20 years ago, and in one study in which particulate matter with diameter < $2.5 \,\mu m$ (PM_{2.5}) samples were collected from five sites in North Carolina, USA, it was reported that most PFOA and PFOS concentrations were < 1 pg/m^3 (Zhou et al., 2021).

[The Working Group noted that these data suggest that, in the absence of an emission source, levels of PFOA and PFOS in outdoor air are low.]

(ii) Indoor air

The sources of PFOA and PFOS in indoor air include consumer products, building materials, and outdoor air (<u>Winkens et al., 2017</u>; <u>Janousek et al., 2019</u>). The results of previous studies have

suggested that PFOA and PFOS concentrations in indoor air exceed those in outdoor air (Goosey and Harrad, 2012). However, there were only a few studies in which PFOA and PFOS levels in indoor air were reported. The studies were conducted in Canada, the USA, Europe, and China, with samples collected from multiple sites, including bedrooms, homes, offices, cars, living rooms, and a laboratory and hallway (see Table 1.9). The median PFOA concentrations in indoor air collected from bedrooms in Canada (21 pg/m³) (Shoeib et al., 2011) and eastern Finland (15.2 pg/m³) (Winkens et al., 2017) were similar to those in living rooms (24 pg/m³) and offices (18 pg/m³) in the UK (Goosey and Harrad, 2012), but lower than median values in living rooms (56 pg/m^3) , cars (76 pg/m³), offices (96 pg/m³), and school classrooms (89 pg/m³) in Ireland (Harrad et al., 2019). The median PFOS concentrations in bedrooms in Canada (< 0.02 pg/m³) (Shoeib et al., 2011) and eastern Finland (1.24 pg/m³) (Winkens et al., 2017), and in living rooms in Ireland (< 0.4 pg/m³) (Harrad et al., 2019) were similar, and lower than those in living rooms (11 pg/m³) in the UK (Goosey and Harrad, 2012), and in cars (13 pg/m³), offices (8.9 pg/m³), and school classrooms (9.3 pg/m³) in Ireland (Harrad et al., 2019), and much lower than those in offices in the UK (55 pg/m³) (Goosey and Harrad, 2012). These findings suggest that the function of these spaces might influence the concentrations of PFOA and PFOS, but more data are needed to confirm these influences.] In addition, one study at the University of North Carolina, USA, found that the floor waxing process in a laboratory and hallway increased mean PFOS concentrations from < 0.22 pg/m³ before waxing to 8.88 pg/m³ during waxing (Zhou et al., 2022).

[The Working Group noted that the available data on PFOA and PFOS levels in indoor air and their determinants were sparse.]

(iii) Settled dust

PFOA and PFOS are widely detected in dust samples because of continuous releases from consumer products (Jian et al., 2017; Zhu et al., 2023). de la Torre et al. (2019) evaluated 65 samples of house dust from three European countries. The median concentrations of PFOA in these dust samples from Belgium, Italy, and Spain were similar (1.54 ng/g, 1.56 ng/g and 1.00 ng/g, respectively), and median concentrations of PFOS in dust samples from these three countries were also low (0.77 ng/g, 0.33 ng/g, and 0.03 ng/g, respectively) (Table 1.10) (de la Torre et al., 2019). A study that collected dust samples from 184 homes in North Carolina, USA, and 49 fire stations in the USA and Canada showed that the median concentration of PFOA in dust samples collected from fire stations (17.6 ng/g) was higher than that from homes (7.9 ng/g) (Table 1.10) (Hall et al., 2020). Likewise, the median concentration of PFOS in fire stations (64.5 ng/g) was much higher than from homes (4.4 ng/g). Another study measured levels in 81 dust samples from homes in Indiana, USA, and found similar median concentrations of PFOA (5.9 ng/g) and PFOS (10 ng/g) (<u>Zheng et al., 2023</u>).

(b) Water

PFOA and PFOS are generally not removed from source water during standard water treatment (Wee and Aris, 2023). They have been detected in surface water, groundwater, wastewater, and in raw and finished drinking-water. Although the global extent of water contamination by PFOA and PFOS has not been completely characterized, PFOA and PFOS have been measured in water sources on all continents (Kurwadkar et al., 2022). Studies on highly contaminated water are discussed in subsection (iv).

Reference	Location and collection	Characteristics of sampling	aracteristics Analytical sampling method (reporting		PFOA concentration (pg/m ³)		ration	Comments	
	date	(number, sites, sampler, time, flow and duration)	limits)	Mean (range)	Median (IQR)	Mean (range)	Median (IQR)		
<u>Goosey and</u> <u>Harrad.</u> (2012)	Birmingham, UK, September 2008 to March 2009	20, living room, PAS, 1.0 m ³ /day PFOA; 0.8 m ³ /day PFOS, 28–35 days/ sample	HPLC-MS/MS (NR)	52 (< 1.9-440)	24	38 (< 1.0-400)	11	Long sampling time. No. of samples was relatively large. Good reflection of PFAS levels in homes.	
		12, offices, PAS, 1.0 m³/day PFOA; 0.8 m³/day PFOS, 28–35 days/sample		58 (< 1.9–200)	18	56 (12–89)	55	Long sampling time. No. of samples was relatively large. Good reflection of PFAS levels in homes.	
<u>Harrad et al.</u> (2019)	Dublin, Galway, and Limerick, Ireland, August 2016	34, living room, PAS, 60 days/ sample 1.0 m³/day PFOA; 0.8 m³/day PFOS	HPLC-MS/MS (LOD: PFOA, 0.3 pg/m ³ ; PFOS, 0.4 pg/m ³)	72 (< 0.3–386)	56	14 (< 0.4–208)	< 0.4	Long sampling time. No. of samples was large. Good reflection of PFAS levels at selected sites.	
	to January 2017	31, cars, PAS, 60 days/sample		162 (1.2–790)	76	22 (< 0.4–152)	13		
		34, offices, PAS, 60 days/sample		153 (< 0.3–1210)	96	89 (< 0.4–1290)	8.9		
		28, school classrooms, PAS, 60 days/sample		210 (< 0.3-728)	89	188 (< 0.4–1590)	9.3		
<u>Shoeib et al.</u> (2011)	Vancouver, Canada, 2007–2008	59, bedroom, PAS, 4 wk/sample	LC-MS/MS, MDL, 0.47 pg/m ³ , 0.02 pg/m ³	113 (3.4–2570)	21	(< 0.02, < 0.02)	< 0.02	Long sampling time. No. of samples was large. Good reflection of PFAS levels in bedrooms.	
<u>Winkens et al.</u> (2017)	Kuopio, eastern Finland, 2014–2015	57, bedroom, PAS, 21 days/sample	LC-MS/MS (MDL: PFOA, 4.48 pg/m ³ ; PFOS, 0.47 pg/m ³)	21.2 (< 4.48–99.8)	15.2	1.33 (< 0.47–5.04)	1.24	Long sampling time. No. of samples was relatively large. Good reflection of PFAS levels in bedrooms.	

Table 1.9 Occurrence of PFOA and PFOS in indoor air

Table 1.9	(continued)							
Reference	Location and collection	Characteristics of sampling	Analytical method (reporting	PFOA concentration (pg/m ³)		PFOS concentration (pg/m ³)		Comments
	date	(number, limits) sites, sampler, time, flow and duration)	limits)	Mean (range)	Median (IQR)	Mean (range)	Median (IQR)	
<u>Zhou et al.</u> (2022)	University of North Carolina, USA, August to September	3, laboratory and hallway, before floor waxing, AAS, 16/min, 24 h/sample	HPLC-MS/MS (MDL: PFOA, 0.82 pg/m ³ ; PFOS, 0.25 pg/m ³)	[12.69]	NR	[< 0.22]	NR	Sampling time and size were limited.
	2019	3, laboratory and hallway, during floor waxing, AAS, 16/min, 18 h/sample		[8.83]	NR	[8.88]	NR	Sampling time and size were limited.
		3, laboratory and hallway, after floor waxing, AAS, 16/min, 24 h/sample		[8.17]	NR	[< 0.22]	NR	Sampling time and size were limited.

AAS, active air sampler; h, hour(s); HPLC, high-performance liquid chromatography; IQR, interquartile range; LC, liquid chromatography; LOD, limit of detection; LOQ, limit of quantification; MDL, method detection limit; min, minute(s); MS/MS, tandem mass spectrometry; NR, not reported; PAS, passive air sampler; PFAS, per- and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; UK, United Kingdom; United States of America; wk, week(s).

Table 1.10 Occurrence of PFOA and PFOS in dust								
Reference	Location and collection date	Characteristics of sampling	Analytical method (reporting limits)	PFOA con (ng/g	centration dust)	PFOS co (ng,	ncentration /g dust)	Comments
		(number, sites)		Mean (range)	Median (IQR)	Mean (range)	Median (IQR)	
<u>de la Torre</u> <u>et al. (2019)</u>	Belgium, September 2016 to January 2017	Homes (<i>n</i> = 22)	HPLC-MS/MS (LOQ: PFOA, 0.11 ng/g; PFOS, 0.04 ng/g)	NR (0.31–24.2)	1.54	NR (< 0.04–6.81)	0.77	Long sampling time. No. of samples was relatively large. Good reflection of PFAS levels in dust.
	Italy, September 2016 to January 2017	Homes (<i>n</i> = 22)		NR (0.21–53.0)	1.56	NR (< 0.04–11.9)	0.33	Long sampling time. No. of samples was relatively large. Good reflection of PFAS levels in dust.
	Spain, September 2016 to January 2017	Homes (<i>n</i> = 21)		NR (0.42–12.5)	1.00	NR (< 0.04–2.45)	0.03	Long sampling time. No. of samples was relatively large. Good reflection of PFAS levels in dust.
<u>Hall et al.</u> (2020)	Fire stations, USA and Canada, 2015 and 2018	Fire stations $(n = 49)$	HPLC-MS/MS (MDL: PFOA, 1.60 ng/g dust; PFOS, 1.44 ng/g dust)	NR	17.6	NR	64.5	No. of samples was relatively large. Good reflection of PFAS levels in dust.
	North Carolina, USA, 2014–2016	Homes (<i>n</i> = 184)	HPLC-MS/MS (MDL: PFOA, 0.26 ng/g dust; PFOS, 0.20 ng/g dust)	NR	7.9	NR	4.4	Long sampling time. No. of samples was relatively large. Good reflection of PFAS levels in dust.
Zheng et al. (2023)	Indiana, USA, August to December 2020	Homes (<i>n</i> = 81)	HPLC-MS/MS (MDL: PFOA, 0.01 ng/g dust; PFOS, 0.02 ng/g dust)	(< 0.01–1900)	5.9	(< 0.02–1100)	10	Long sampling time. No. of samples was relatively large. Good reflection of PFAS levels in dust.

HPLC, high-performance liquid chromatography; IQR, interquartile range; LOQ, limit of quantification; MDL, method detection limit; MS/MS, tandem mass spectrometry; NR, not reported; PFAS, per- and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; UK, United Kingdom; USA, United States of America.

(i) Surveys of surface water

Examples of PFOA and PFOS measurements in surface waters (lakes or rivers) are presented in Table S1.11 (Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: https://publications. iarc.who.int/636). Mean values of PFOA and PFOS were generally below or in the low nanograms-per-litre range in locations without any reported PFAS pollution source. One example of higher levels reported downstream of an industrial source was in Alabama, USA, where values reported were 598 µg/L for PFOA and 144 µg/L for PFOS (<u>ATSDR, 2021</u>). Reported concentrations in ocean water were generally well below the nanogram-per-litre range (see Table S1.11).

Kurwadkar and colleagues reviewed levels of PFAS substances in surface water, groundwater and wastewater (Kurwadkar et al., 2022). They included information on the Asia-Pacific region, collected under the Second Global Monitoring Report on Persistent Organic Pollutants, which showed that PFOS detection was becoming more frequent. Levels of PFOS ranged from not detected to 47 ng/L in China; from 0.02 to 230 ng/L in Japan; from 0.12 to 33 ng/L in the Republic of Korea; from 0.39 to 42 ng/L in the Philippines; and from not detected to 54 ng/L in Thailand (United Nations Environment Programme, UNEP, as cited in Kurwadkar et al., 2022). Limited data were available for most of South America and Africa.

Muir and Miaz (2021) assembled an extensive summary of PFOA and PFOS measurements and total emissions for rivers across the world. There was a high degree of variability but widespread detectable levels of PFOA, with the highest concentrations identified in Europe in the River Po, Italy (200 ng/L); in Asia, in the Hokkesantanigawa River, Japan (360 ng/L); and in China in numerous rivers (e.g. the Daling River, 233 ng/L). Estimated riverine emissions of PFOA to the sea exceeded 1000 kg/year for many rivers, with estimates for the Yangtze River reaching 10 000 to 40 000 kg/year.

For PFOS, high concentrations were reported for the Llobregat and Besos rivers in Spain (> 250 ng/L) and the Ganges in India (142 ng/L) (<u>Muir and Miaz, 2021</u>). Riverine emissions of PFOS to the sea were estimated to have exceeded 2000 kg/year for the Pearl and Xi Rivers in China, and the Saint Lawrence River in North America. [The Working Group noted that several of these measurements were taken in the early 2000s and may not represent more recent riverine discharges.]

In a meta-analysis of publications on PFAS in wastewater treatment plant effluent streams, some indications of trends over time were presented (<u>Cookson and Detwiler, 2022</u>). Multiple results in China indicated a clear upward trend during 2006–2019 for both PFOA and PFOS. In the data for the USA, a clear downward trend was evident for PFOA over the period 2004–2020, but there was no overall trend for PFOS.

In a systematic review, <u>Land et al. (2018)</u> observed declining trends in PFOA and PFOS levels in Tokyo Bay between 2004 and 2006; in marine and fresh waters on the west coast of the Republic of Korea between 2008 and 2012; and in Bohai Bay on the east coast of China between 2011 and 2013.

Across the USA and across Europe there were many sources of data on PFAS surface-water contamination, from both local government monitoring and research projects, and many of these sources have been assembled into an online searchable resource (<u>Dagorn et al., 2023;</u> <u>Environmental Working Group, 2023;</u> <u>PFAS</u> <u>Project Laboratory, 2023</u>). These maps show the widespread locations where PFOA and PFOS are detectable in the USA and across Europe but do not provide summary exposure data for PFOA and PFOS.

PFOA and PFOS have been detected in many rainwater samples collected from urban and rural areas of Europe, Asia, and North America. Levels near local emission sources can be very high, for example, PFOA concentrations measured in rainwater near to a fluoropolymer plant in China (median, [615 ng/L]; maximum, 2752 ng/L) (<u>Liu et al., 2017</u>). Dispersion has been very widespread, with detectable concentrations of [0.22 ng/L] for PFOA and [0.006 ng/L] for PFOS reported in Antarctica (<u>Casas et al., 2021</u>). Reported urban rainfall levels tended to be up to about 10 ng/L, and rural levels were generally < 1 ng/L (see Table S1.11).

PFOA and PFOS have been detected in both coastal and sea and ocean waters, with lower concentrations in ocean waters (see Table S1.11). PFOA and PFOS have been found to be substantially concentrated in sea foam and rising mist, which can be blown inland and contaminate surface water (Sha et al., 2022). Muir and Miaz conducted an extensive review of measurements from ocean and coastal waters, lakes, and rivers, and incorporated 29 500 measurements of 87 individual PFAS analytes, including PFOA and PFOS (Muir and Miaz, 2021). During 2015–2019, concentrations in seas were highest for PFOA in the Bohai and Yellow seas (median, 9.0 ng/L) and for PFOS in the Indian Ocean (median, 0.087 ng/L). The lowest concentrations were found in the Mediterranean Sea for PFOA (median, 0.001 ng/L) and in the Arctic Sea for PFOS (0.02 ng/L). Comparison of surveys conducted in 2000-2009, 2010-2014, and 2015-2019 revealed clear upward trends in concentrations of PFOA in the Bohai Sea, Yellow Sea, and East China Sea, and a steep decline in the Mediterranean. For PFOS, upward trends were evident in the Indian Ocean. [The Working Group noted, as did the authors, that deriving medians across studies with different sampling sites, design, and timing of sample collection, as well as different method detection limits (MDLs), introduces considerable uncertainty for assessing contrasts across space and time.]

PFOA and PFOS have been detected in fresh snow at levels that were very low in remote areas such as Antarctica (Xie et al., 2020) and higher in China (Shan et al., 2015), see Table S1.11 (Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: <u>https://publications.iarc.who.int/636</u>). For PFOA and PFOS concentrations measured in snow and ice core samples, see Section 1.4(c)(i).

(ii) Groundwater

Some examples of PFOA and PFOS concentrations measured in groundwater are presented in Table S1.11 (Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: https://publications. iarc.who.int/636). The occurrence of PFAS in groundwater from different areas in the world, including Australia, China, India, and islands of Malta has been described in a review (Xu et al., 2021a). PFOA was the dominant PFAS detected in three of the eight locations studied. For instance, in rural areas of eastern China, PFOA concentrations ranged from 7 to 175.2 ng/L, with a mean value of 90.8 ng/L (Chen et al., 2016). Also in China, but in the alluvial-pluvial plain of the Hutuo River, PFOA concentrations ranged from 0 to 1.76 ng/L (mean, 0.63 ng/L) in groundwater (Liu et al., 2019). PFOA was found at concentrations in the range of 0-8.03 ng/L (mean, 1.46 ng/L) in groundwater from valleys in Gozo on the Maltese Islands (Sammut et al., 2019). In the case of PFOS, higher concentrations than those of other PFAS were found only in 13 shallow monitoring bores surrounding legacy landfills in Melbourne, Australia, with a range of 1.3–4800 ng/L and mean value of 413.3 ng/L (<u>Hepburn et al., 2019</u>).

In 2019, 254 samples were collected from five aquifer systems in the eastern USA to evaluate PFAS occurrence in groundwater used as a source of drinking-water. In this study, PFOA and PFOS represent two of the three most frequently detected PFAS in public-supply wells, with 2.4% (n = 6) of the samples containing PFOA plus PFOS at concentrations of > 70 ng/L, and median concentrations detected were 4.6 ng/L and 6.7 ng/L for PFOA and PFOS, respectively (McMahon et al., 2022).

In a study developed in Sweden, a national screening for perfluorinated pollutants in drinking-water was performed. The most abundant individual PFAS in surface and ground-water supplies was PFOS, followed by PFOA (Holmström et al., 2014).

[The Working Group noted that some studies did not report on PFOA and PFOS separately.]

(iii) Drinking-water and drinking-water supplies

PFOA and PFOS have been measured in drinking-water (e.g. tap water, bottled water) in various locations (Fig. 1.4; Table S1.11, Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: https://publications.iarc.who.int/636). [The Working Group noted that mean concentrations in drinking-water from sites without any known contamination were usually below 10 ng/L (Fig. 1.4; see also Section 1.4.1(b)(iv) for concentrations measured at sites with reported contamination sources).]

Domingo and Nadal (2019) reviewed the scientific literature on PFAS exposure via drinking-water and highlighted that most information was coming from the EU, USA, and China. They reported on water sampling efforts in Europe (France, Italy, Sweden, Spain, Norway, Belgium, and the Faroe Islands), the Americas (USA, Canada, Brazil), and Asia (China, Japan, Afghanistan, India, and the Republic of Korea), Africa, and Australia. Levels of PFOA and PFOS in drinking-water and drinking-water sources ranged from non-detectable to > 500 ng/L.

There have been few formal efforts to characterize PFOA and PFOS levels in drinking-water sources on a national or international level. Because communities draw drinking-water from both surface and groundwater sources, large-scale efforts focus on water used for drinking-water supplies, regardless of whether this is surface or groundwater. In 2013-2015, the USEPA required 6000 public water systems (PWS) to test for PFOA and PFOS (and four other PFAS) in source water under the Third Unregulated Contaminant Monitoring Rule (UCMR 3) programme (US EPA, 2017b). The prevalence of PFOA at levels above the minimum reporting levels (MRLs) was low (0.09% of samples from 0.3% of PWS were above the MRL of 20 ng/L), as was that of PFOS (0.3% of samples from 0.9% of PWS exceeded the MRL of 40 ng/L) (US EPA, 2017b). Levels of PFOA reported ranged from 20 to 349 ng/L (median, [32 ng/L]); levels of PFOS ranged from 41 to 1800 ng/L (median, [66 ng/L]); the frequency of detection of these chemicals increased over the reporting period (Guelfo and Adamson, 2018). Detectable levels of PFOA and PFOS spanned three orders of magnitude, with PFOS levels being higher than those of PFOA (<u>US EPA, 2017b</u>). On the basis of these detections, it was estimated that more than 6 million people in the USA had drinking-water that exceeded 70 ng/L for the sum of PFOA and PFOS (Hu et al., 2016). Detection of PFOA and PFOS was significantly associated with nearby military firefighting training areas, AFFF-certified airports, and wastewater treatment plants. Detectable PFOA was also associated with major industrial sites that produced or used PFOA and/or PFOS (<u>Hu et al., 2016</u>).

More recently, UCMR 5, being conducted in 2023–2025, is measuring 29 PFAS with lower MRLs (4 ng/L for PFOA and PFOS) and in a larger number of PWS than UCMR 3 (<u>US EPA</u>, 2024). In initial data available up to October 2023 for 10 020 samples from 3072 PWS, PFOA was reported to be above the MRL in 6.1% of samples and 9.5% of PWS, and PFOS was detected at above the MRL in 6.4% of samples and 9.5% of PWS.



Fig. 1.4 Examples of PFOA and PFOS concentrations in drinking-water from sites without known sources of contamination

LOD, limit of detection; LOQ, limit of quantification; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; USA, United States of America.

Selected publications, see Table S1.11 for detail and references (Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: <u>https://publications.iarc.who.int/636</u>). When available, the median was plotted. When values were below the LOQ or LOD, the LOQ or LOD was plotted instead. Note the logarithmic scale.

(iv) Local major contamination of drinkingwater sources

Fluorochemical manufacture and use of firefighting foams are associated with PFOA and PFOS contamination of drinking-water around the world. Some examples are described below.

The first such contamination identified was from a facility manufacturing PTFE in Parkersburg, West Virginia, USA, which contaminated surface water and drinking-water supplies in West Virginia and Ohio, with more than 80 000 people supplied with water contaminated with PFOA to varying extents. Levels of PFOA in water supplies measured since the early 2000s ranged from 10 to 100 ng/L in the least-contaminated water district to up to 10 μ g/L in the most-contaminated water district (Shin et al., 2011a). Modelling of the water contamination indicated progressive increases in contamination from the 1950s to 2000, when emissions were curtailed (Shin et al., 2011a). In a series of measurements from 62 private wells used for drinking-water in the same area, the median PFOA concentration was 200 ng/L (range, 6–13 300 ng/L) (Hoffman et al., 2011).

In Veneto, Italy, groundwater used for drinking-water was contaminated by chemical production; PFOA was the main contaminant, together with a mixture of mainly shorter-chain PFAS. In 152 samples collected from the contaminated area in 2013, the PFOA concentration was above the LOQ in 90% of samples, with a median concentration of 319.5 ng/L (maximum, 1475 ng/L), and the PFOS concentration was above the LOQ in 78% of samples, with a median concentration of 18 ng/L (maximum, 117 ng/L) (<u>Pitter et al., 2020</u>).

Firefighting foams and their use at airports and air force bases have resulted in PFAS contamination, particularly PFOS, in drinking-water. In Ronneby, Sweden, about one third of a community of 28 000 people were supplied with contaminated drinking-water in which PFOS was measured at 8000 ng/L and PFOA at 100 ng/L before the waterworks was closed (Xu et al., 2021b).

In Australia, PFOA and PFOS were detected in groundwater near a military base in Williamtown, New South Wales, at concentrations of 1800 ng/L and 5560 ng/L, respectively (Kurwadkar et al., 2022).

(c) Soil

Soil has been highlighted as a global sink for and long-term source of PFOA and PFOS (Brusseau et al., 2020). The estimated halflives of PFOA and PFOS in soil are at least tens of years, although the true half-lives may be longer, because no significant degradation was noticeable during the experiments that have been conducted (UNEP, 2006, 2017). PFOA and PFOS can reach the soil directly, or via degradation of their precursors, from various input fertilizers; the use of PFAS-based firefighting foams; leaching from contaminated asphalt and concrete affected by the extensive use of AFFF in firefighting training centres and airfields; seepage of leachate from landfills; discharge of effluents from wastewater treatment plants; contaminated irrigation water; contaminated discharge from fluorochemical industries; and atmospheric deposition (Costello and Lee, 2020; Abou-Khalil et al., 2022; Panieri et al., 2022; Douglas et al., 2023). Brusseau et al. (2020) comprehensively reviewed the literature on PFAS in soil. Both PFOA and PFOS were ubiquitously distributed globally in soil, with or without nearby point sources. The median for maximum concentrations of PFOA and PFOS reported globally in soil near primary point sources was 8722 and 83 ng/g, respectively, whereas the median for maximum background soil (i.e. no direct input sources) concentrations worldwide was 2.7 and 2.7 ng/g, respectively (Brusseau et al., 2020). The highest reported concentration of PFOA in soil (50 000 ng/g) was measured in soil contaminated with AFFF from a US military site (Brusseau et al., 2020), and the highest PFOA concentration (460 000 ng/g) was measured at firefighting training grounds in Australia (CRCCARE, 2018). Regarding background soil levels, the highest PFOA concentration of 47.5 ng/g was measured in soil from Shanghai, China (Li et al., 2010), and the highest PFOS concentration of 162 ng/g was reported for Alnabru, Norway (NEA, 2017). Table 1.12 provides a summary of selected studies on the occurrence of PFOA and PFOS in soil and lists PFOA and PFOS concentrations from various sites with different sources of contamination, as well as background concentrations from non-contaminated sites.

sources including: application of biosolids as

A systematic review of concentrations of 12 PFAS (including PFOA and PFOS) in 1042 soil samples from 15 countries on 6 continents reported significantly higher Σ_{12} PFAS levels (dominated by PFOA and PFOS) in the northern

Table 1.12 Occurrence of PFOA and PFOS in soil

Location and collection date	No. of samples	PFOA concentrations (ng/g)	PFOS concentrations (ng/g)	Comments	Reference
Antarctica, 2010	3	< MQL	Range, 0.31–0.54 Mean, [0.45]		<u>Llorca et al. (2012a)</u>
Australia, NR	6	Range, 13.6–58.1 Mean, 34.5	Range, 2180–15 300 Mean, 7800	AFFF use at airport sites	<u>Bräunig et al.</u> (2019)
Australia, NR	3	Range, 0.33–0.39 Mean, 0.36	Range, 6.4–7.2 Mean, 6.8	No direct input sources	<u>Bräunig et al.</u> (2019)
Africa					
Uganda, 2015	18	Range, 0.25–0.91	Range, 0.6–3.0	No direct input sources	<u>Dalahmeh et al.</u> (2018)
Asia					
China, 2009	32	Range, < 0.05–34.2 Mean, 2.5	Range, 0.68–189 Mean, 22.6	Soil around a fluorochemical- manufacturing plant	<u>Wang et al. (2010)</u>
China, NR	86	Range, < 0.1–0.9 Mean, 0.2	Range, 0.02–2.4 Mean, 0.3	No direct input sources	<u>Pan et al. (2011)</u>
Nepal, 2010	14	Range, < 0.1–0.26	Range, < 0.09-0.13	No direct input sources	<u>Tan et al. (2014)</u>
Republic of Korea, 2009	13	Range, < 0.2–3.4 Mean, 2.2	Range, < 0.2–1.7 Mean, 0.8	No direct input sources.	<u>Naile et al. (2013)</u>
Europe					
Germany, 2006	1	650	8600	Soil affected by contaminated industrial waste	<u>Wilhelm et al.</u> (2008)
Norway, 2008	39	Range, < 1.0–141.5 Median, 12.8 GM, 16.0	Range, 24.5–11 923 Median, 641 GM, 516.6	Fire training sites $(n = 4)$	<u>SFT (2008)</u>
Norway, 2016	9	< 0.01	Range, < 0.02–7.06 Median < 0.02	No direct input sources	<u>Skaar et al. (2019)</u>
Sweden, 2011–2012	45	Range, < 0.1–219 Median, 1.4 GM, 2.6	Range, < 0.5–8520 Median, 39 GM, 42.5	AFFF-contaminated soil near a military airport	<u>Filipovic et al.</u> (2015a)
Sweden, NR	31	Range, < 0.02–0.57 Median, < 0.02 Mean, 0.04	Range, < 0.02–1.7 Median, 0.3 Mean, 0.43	No direct input sources	<u>Kikuchi et al.</u> (2018)
North America					
USA, 2010	6	Range, 0.29–0.54 Median, 0.33	Range, 0.93–2.1 Median, 1.4	No direct input sources	<u>Scher et al. (2018)</u>
USA, 2019	2469	Range, 0.07–50 000 Median, 1.4 GM, 2	Range, 0.09–373 000 Median, 18 GM, 22	Sites affected by AFFF use	<u>Brusseau et al.</u> (2020)

Table 1.12 (continued)

Location and collection date	No. of samples	PFOA concentrations (ng/g)	PFOS concentrations (ng/g)	Comments	Reference
South America					
Tierra del Fuego, Argentina, 2010	30	Range, < MQL–1.5 Mean, 0.3	Range, < MQL–5.4 Mean, 1.4	No direct input sources	<u>Llorca et al. (2012a)</u>

AFFF, aqueous film-forming foam; GM, geometric mean; MQL, method quantification limit; NR, not reported; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; USA, United States of America.

hemisphere, which was attributed to greater PFAS emissions, compared with the southern hemisphere (Lv et al., 2023). On a continental scale (from highest to lowest), mean concentrations of PFOA were ranked, North America > Asia > Europe > Africa > Oceania > South America, and mean PFOS concentrations were ranked, North America > Africa > Europe > Asia > Oceania > South America (Lv et al., 2023).

A meta-analysis of PFAS soil-to-groundwater concentration ratios for samples collected from 324 sites where AFFF was used across 56 military installations throughout the USA demonstrated that soil is a significant reservoir for PFAS at these contaminated sites (Hunter Anderson et al., 2019). Moreover, analysis of PFAS depth profiles in the soil indicated significant retention of PFOA and PFOS in the vadose zone over decades, serving as a significant long-term source of PFAS in groundwater (Guo et al., 2020; Lv et al., 2023). In a recent study of temporal trends in PFAS concentrations in soil samples from eastern China, it was reported that PFOA concentrations increased by 86.4% between 2011 and 2021, whereas PFOS concentrations decreased by 28.2% during the same period. The distinct difference between PFOA and PFOS in terms of temporal changes in soil concentrations during the studied decade was attributed to the fact that PFOS was added to the Stockholm Convention on Persistent Organic Pollutants (POPs) in 2009, while PFOA was added later in 2019 (Cheng et al., 2023).

(d) Food

PFOA and PFOS are introduced into foods in various ways, mainly depending on the food origin, but also on packaging (<u>Schaider et al.</u>, <u>2017</u>) and processing (<u>Choi et al.</u>, <u>2018</u>). Plantbased foods may be contaminated via atmospheric deposition or uptake from water and soil, including from use of sewage sludge as fertilizer (<u>Ghisi et al.</u>, <u>2019</u>). A study on crops grown in outdoor lysimeters demonstrated uptake of PFOA and PFOS, including in various edible parts of the crop (Felizeter et al., 2021). Uptakes vary both between and within species and may partly be explained by different plant properties (Costello and Lee, 2020). PFOA and PFOS become incorporated into animal-based foods because animals are exposed to these PFAS via water, feed, soil, and air (Death et al., 2021). [The Working Group noted that there was a lack of data on the contributions of different sources of PFOA and PFOS contamination in foods. This may be of more importance for source tracking and reduction than for exposure characterization.]

Concentrations of PFOA and PFOS have been determined in various food products, including food for infants such as formula and baby food, in a range of studies worldwide (Mikolajczyk et al., 2023). Some studies on processed food were available (e.g. Jogsten et al., 2009; Jeong et al., 2019; Genualdi et al., 2022; Vendl et al., 2022), but most of the data were based on the analysis of raw food products. In general, most data were available for fish and seafood, but there have been an increasing number of studies on other food groups during recent years (Domingo and Nadal, 2017; Jian et al., 2017; Pasecnaja et al., 2022). PFOA and PFOS concentrations detailed in selected studies and reports are presented in Table S1.13 and Table S1.14, respectively (see Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Reporting limits varied to a large extent between studies (<u>Pasecnaja et al., 2022</u>). [The Working Group noted that high reporting limits, especially when considering data generated in the early 2000s, have resulted in low detection frequencies, resulting in challenges when comparing studies.] As a result of an increasing focus on the need for more sensitive methods, lower reporting limits have been observed in more recent studies (e.g. <u>Lacina et al., 2011</u>; <u>Vestergren et al., 2012</u>; <u>Sadia et al., 2020</u>). For example, in the study by <u>Vestergren et al. (2012</u>), in which a particular effort had been made to increase sensitivity, the method limit of quantification (MLQ) ranged between 1.8 and 9.6 pg/g for PFOA, depending on food type, and between 1.5 and 8.0 pg/g for PFOS. In comparison, in the study by <u>Clarke et al. (2010)</u>, for example, the LOQ for both PFOA and PFOS was 1000 pg/g. [The Working Group noted that because levels of PFOA and PFOS are low in many food products, improvements in MLQ have an impact on detection frequencies (<u>EU, 2022</u>).]

The studies presented in supplementary Table S1.13 and Table S1.14 were published recently, present data from different regions worldwide, and include information on several food categories. As an example, a study on 266 samples collected during 2018-2019 from 26 countries located in Africa, Asia (excluding China), and Latin America included data on several food groups and had high detection rates (Fiedler et al., 2022). The mean concentrations of PFOA in vegetables, fish and other seafood, beef, chicken, milk, and eggs were 7.58, 12.4, 6.44, 4.61, 0.99, and 8.34 pg/g, respectively. The corresponding mean concentrations of PFOS in the same food groups were 2.45, 124, 37.6, 5.80, 22.1, and 45.6 pg/g, respectively. [The Working Group noted that with the low LOQ in this study, PFOA and PFOS contamination was detected more frequently.]

The mean concentrations in samples from Europe (EFSA Panel on Contaminants in the Food Chain, 2020) and China (Fan et al., 2021) were in general higher than those from Africa, Asia (excluding China), and Latin America (Fiedler et al., 2022), but detection frequencies in studies in the USA were too low to compare, except for fish and seafood (US FDA, 2022a; Young et al., 2022). For example, the mean concentrations of PFOA in eggs were 106 and 150 pg/g in samples from Europe (EFSA Panel on Contaminants in the Food Chain, 2020) and China (Fan et al., 2021), respectively, while the mean concentration in the study on samples from Africa, Asia (excluding China), and Latin America was 8.34 pg/g (<u>Fiedler et al., 2022</u>).

For PFOS, the highest mean concentrations were generally seen in fish and seafood, when compared with other food groups (see Table S1.13 and Table S1.14, Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: https://publications. iarc.who.int/636). For example, the mean concentration of PFOS in fish and shrimp from China was 2760 pg/g, whereas the mean concentration in meat and meat products was 300 pg/g (Fan et al., 2021). This was in line with data reported in review papers (Domingo and Nadal, 2017; Jian et al., 2017; Pasecnaja et al., 2022). These observations were supported by the results of a study that found PFOS to be very bioaccumulative (bioaccumulation factor, > 5000) and also biomagnifying (trophic magnification factor, > 1) in a freshwater food web in Canada (Munoz et al., 2022). For PFOA, among various countries, the mean concentrations were highest in China across all food groups, with the highest concentrations found in fish and meat (see Table S1.13 and Table S1.14, Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: https://publications. iarc.who.int/636).

In a study on more than 500 composite samples of locally caught freshwater fish collected across the USA in 2013–2015, the median concentration of total PFAS (of which PFOS constituted 74%) was more than 200 times as high as the levels found in commercially relevant fish analysed by the US FDA in 2019–2022 (Barbo et al., 2023). The median and 90th percentile concentrations of PFOS in locally caught freshwater fish fillets across the USA were 8410 pg/g and 41 400 pg/g, respectively (Barbo et al., 2023), whereas the concentrations reported by the US FDA were generally in the low hundreds of picograms per gram, or less (Barbo et al., 2023).

Elevated concentrations of PFOA and PFOS in food have been reported in areas with known PFAS contamination (e.g. Hölzer et al., 2011; Langberg et al., 2022; Lasters et al., 2022). For instance, elevated concentrations of PFOA and PFOS were measured in hen eggs from private gardens situated within a 10 km radius of a fluorochemical-production plant in Antwerp, Belgium (Lasters et al., 2022). The highest concentrations were observed for PFOS (130-241 000 pg/g), and decreasing concentrations were observed with increasing distance from the plant (Lasters et al., <u>2022</u>). This was in line with the results of a study showing increasing PFOA and PFOS concentrations in eggs of hens exposed to increasing concentrations in drinking-water (Wilson et al., 2021). In Lyon, France, PFOS concentrations in home-produced eggs near a fluorochemical-production facility were higher than in commercially produced eggs: home-produced, median, [965 pg/g] and range, 105–5240 pg/g; commercially produced, median, [113 pg/g] and range, 34–650 pg/g (<u>Préfète du Rhône, 2023</u>).

Environmental contamination may also result in elevated levels in dairy products. In a study on two dairy farms in the USA with known contamination of groundwater, samples were collected between 2018 and 2021. Milk samples from one of the two farms contained PFOS at elevated concentrations. PFOS concentrations of up to 4.22 ng/g were reported in the milk from this farm, while retail milk and control milk did not contain PFOS at detectable levels (US FDA, 2021a)

Food from wild game species may also contain elevated levels of PFOA and PFOS (<u>Death et al., 2021</u>). The European Food Safety Authority (EFSA) Panel on Contaminants in the Food Chain (<u>EFSA Panel on Contaminants</u> in the Food Chain, 2020) reported that edible offal from game animals contained mean levels of PFOA and PFOS that were more than 10 and 100 times, respectively, as high as in any other food group. In a study from China that included food samples from two provinces, Hubei (n = 121) and Zhejiang (n = 106), geographical differences were observed between regions and sampling sites (Zhang et al., 2017a). PFOS levels were higher in Hubei than in Zhejiang, particularly for food of animal origin. This was expected because PFOS was produced in Hubei province. The concentrations of PFOA were similar in the two regions.

Temporal trends for abiotic and biological environmental samples, including food items, were evaluated in a systematic review by Land et al. (2018). Both non-significant and decreasing trends were observed for PFOS, depending on region and study. Non-significant trends were predominant for PFOA. The authors concluded that, despite interventions to reduce exposure to PFOA and PFOS, no clear temporal trends have been observed globally, probably because of the high persistence of these compounds in the environment. [There were no available data from the southern hemisphere and Asia.]

(e) Consumer products

PFOA and PFOS are present in numerous consumer products; for example, textiles, outdoor clothing, cleaning products, paints, coatings, carpets, floor coverings, floor polish, leather, cosmetics, printing inks, adhesives, ski wax, and lubricants (Glüge et al., 2020). However, it is often not clear if they were added intentionally, are impurities of other components, or are degradation products (Glüge et al., 2020). Also, PFOA has been used as a processing aid when manufacturing fluoropolymers used, for example, in non-stick cookware. Residues of PFOA may thus be present in these products (Sinclair et al., 2007).

The amount of available data on PFOA and PFOS in consumer products is considerably smaller than that on these compounds in food and drinking-water, and almost all data have been published in the last decade. [The Working Group noted that results are reported in different metrics, e.g. ng/g or $\mu g/m^2$, which makes com-

parison challenging.] Below, selected studies on consumer products available on the market in different regions worldwide are presented.

PFOA and PFOS were determined in 115 randomly selected consumer products (textiles, carpets, cleaning and impregnating agents, leather, baking and sandwich papers, paper baking forms, and ski waxes) purchased in Germany in 2010 (Kotthoff et al., 2015). Detection frequencies varied between product categories. For PFOA and PFOS, respectively, the detection frequencies were 100% and 100% for outdoor textiles, 78% and 100% for nanosprays and impregnation sprays, 88% and 100% for ski wax, and 30% and 90% for carpets. PFOA was also found in 100% of the gloves, 100% of textiles for awnings, and 63% of leather products. PFOS was detected in 69% of paper-based food contact materials. PFOA and/or PFOS were detected in < 50% of the remaining products (wood glue and cleaners). For PFOA, the highest median (and maximum) concentrations were observed in ski wax and nano- and impregnation sprays, 15.5 (maximum, 2033.1) and 15.9 (maximum, 28.9) ng/g, respectively (Kotthoff et al., 2015). The highest median (and maximum) levels of PFOS were observed in outdoor textiles and leather with concentrations of 9.5 (maximum, 35.4) and 5.6 (maximum, 5.6) μ g/m², respectively.

PFOA and PFOS were determined in 25 samples of consumer products available to private consumers in Japan and purchased between 1981 and 2009 (car wash/coating products, sprays for fabrics and textiles, insecticides, rust inhibitors, and paints) (Ye et al., 2015). PFOA was found in one sample of spray for fabrics and textiles (36 ng/g) and one rust inhibitor (11 ng/g), and PFOS was observed in one sample of spray for fabrics and textiles (59 ng/g) (Ye et al., 2015). In a study in Norway, PFOA and PFOS were determined in 45 samples of furniture textile, carpet, clothing, and food contact materials (Vestergren et al., 2015). All samples were imported from China and purchased in Norway in 2012–2013. PFOA was found in 26 samples, at concentrations between 0.005 (carpet) and 0.91 μ g/m² (curtain). PFOS was detected at 1.7 μ g/m² in one carpet sample (<u>Vestergren et al., 2015</u>).

Seventeen samples of paper and cardboard food packaging materials purchased from retailers and grocery stores in Egypt in 2013 were analysed for PFAS (Shoeib et al., 2016). PFOA and PFOS were detected in 79% and 58% of the samples, at median concentrations of 2.40 ng/g and 0.29 ng/g, respectively.

In a study in the USA, PFAS were determined in 61 samples of furnishings, apparel, and bedding purchased in 2020 (Rodgers et al., 2022). PFOA and PFOS were detected in seven and one product, respectively. The maximum concentration of PFOA was 22.5 ng/g, and the concentration of PFOS was 2.1 ng/g in the one product in which it was detected (Rodgers et al., 2022). PFAS were determined in 160 textile products purchased in Albany, New York, USA, between 2016 and 2019 (Zhu and Kannan, 2020). PFOA was detected in 20% of the products, at a maximum concentration of 32.7 μ g/m², whereas PFOS was found in 3.8% of the products, at a maximum concentration of 0.167 µg/m² (Zhu and Kannan, 2020). PFOA and PFOS were determined in 32 textile samples purchased in Thailand; mean concentrations were 2.74 $\mu g/m^2$ (range, 0.31–14.14 $\mu g/m^2$) and $0.18 \,\mu\text{g/m}^2$ (range, $0.02-0.46 \,\mu\text{g/m}^2$), respectively (Supreevasunthorn et al., 2016).

PFOA was determined in sunscreens and cosmetics, primarily from Japan, for which fluorinated compounds were listed as ingredients. Among these products, 8 of 9 sunscreens and 13 of 15 cosmetics contained PFOA at concentrations above the LOQ; concentrations ranged from 3.7 to 5700 ng/g (Fujii et al., 2013). A wide range of PFAS were determined in 38 cosmetics and personal care products for which organofluorine compounds were listed as ingredients and that were available on the North American market in 2020–22 (Harris et al., 2022). PFOA and PFOS

were found at levels above the LOQ in 65.8% and 26.3% of the samples, respectively. The median (and maximum) concentrations of PFOA and PFOS were 13.6 ng/g (28 600 ng/g) and < LOQ (16.5 ng/g), respectively (Harris et al., 2022). In another study in which PFOA and PFOS concentrations were determined in cosmetics (n = 29, n)12 from the USA and 17 from Canada) available on the North American market in 2020, PFOS was detected in only two Canadian samples, at concentrations of 15.5 and 6.6 ng/g (Whitehead et al., 2021). In a study on 43 different cosmetics for which PFAS were listed as ingredients and that were available on the European market in 2020, PFOA was detected in only one foundation/beauty balm cream, at a concentration of 112 ng/g, and PFOS was not detected (Pütz et al., 2022).

[The Working Group noted that there were few data available on the same products at different time points; it was thus not feasible to evaluate potential time trends and the effects of regulations and voluntary phase-outs (see Section 1.5).]

The results of these studies demonstrated that PFOA and PFOS are commonly present in a wide range of consumer products. In several of the studies, PFOA and/or PFOS were most frequently detected in textiles and fabrics at concentrations between the LOD and 35 μ g/m². Also, in some of the studies on cosmetics, detection frequencies were high, and concentrations detected were up to micrograms per gram product.

1.4.2 Occupational exposure

Populations with occupational exposure are generally recognized as having some of the highest levels of exposure to PFOA and PFOS (<u>Christensen and Calkins, 2023</u>). In occupational settings, the exposure route of greatest importance is typically assumed to be inhalation, but dermal uptake and ingestion of dust may also contribute, depending on the workplace conditions (De Silva et al., 2021).

Occupational exposure to PFOA and PFOS may result from fluorochemical-production processes; use of PFAS as a processing aid in other manufacturing settings; use of PFOS as a mist suppressant to reduce exposure to other chemical hazards; contact with products containing PFOA or PFOS, as well as precursor compounds; and contact with contaminated environmental media or waste infrastructure (see Table S1.15, Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: https://publications.iarc.who.int/636). The information regarding determinants of exposure was limited, but indicated that processing conditions, such as high temperature, low pH, and use of PFAS in dry powder form, are linked to elevated PFAS exposure in fluorochemical-manufacturing settings (Freberg et al., 2010; Kaiser et al., 2010; Christensen and Calkins, 2023). Limited data on bulk and dust monitoring suggested that dust present in speciality textile-manufacturing settings, such as those producing flame-retardant or water-repellent materials, increases the risk of PFAS exposure (Sha et al., 2018; Christensen and Calkins, 2023). For non-manufacturing industries, such as retail and office buildings, factors including the building's age, the presence of carpeting, and the extent of ventilation are strongly linked to PFAS exposure (Langer et al., 2010; Sha et al., 2018; Christensen and Calkins, 2023).

In this section, occupational exposures are first described using biomonitoring data, followed by industrial hygiene samples, and work environment samples.

(a) Biomonitoring data

Biomonitoring has been used to assess exposure to PFOA and PFOS in different settings. Fluorochemical-production workers have some of the highest serum PFOA and PFOS concentrations reported in the literature (see Fig. 1.5 and Fig. 1.6; and Table S1.15, Annex 1, Supplementary material for Section 1, Exposure Characterization,

8 Fig. 1.5 Examples of PFOA concentrations in serum, plasma, and whole blood in occupationally exposed populations



NHANES, National Health and Nutrition Examination Survey; PFOA, perfluorooctanoic acid; PTFE, polytetrafluoroethylene; USA, United States of America. *Whole blood. Statistics include median, minimum, and maximum concentrations (ng/mL), the most recent year of sample collection is indicated. See Table S1.15 for detail and references (Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: <u>https://publications.iarc.who.int/636</u>). [The Working Group noted that these values are thought to be representative of the literature on occupational exposure. For studies in which data were reported for multiple subgroups, only selected groups are included.] Concentrations from the adult general population study NHANES are given for the years 1999–2000 and 2019–2020 for comparison (see Section 1.4.3).

Studies

Fig. 1.6 Examples of PFOS concentrations in serum, plasma, or whole blood in occupationally exposed populations



NHANES, National Health and Nutrition Examination Survey; PFOS, perfluorooctanesulfonic acid; USA, United States of America. *Whole blood. Statistics include median, minimum, and maximum concentrations (ng/mL), the most recent year of sample collection is indicated. See Table S1.15 for detail and references (Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: https://publications.iarc.who.int/636). [The Working Group noted that these values are thought to be representative of the literature on occupational exposure. For studies in which data were reported for multiple subgroups, only selected groups are included.] Concentrations from the adult general population study NHANES are given for the years 1999–2000 and 2019–2020 for comparison (see Section 1.4.3). online only, available from: https://publications. iarc.who.int/636). In samples collected in the year 2000 from 25 PFOA-production workers at a site in Italy, Costa et al. (2009) reported a geometric mean concentration of PFOA in serum of 11 700 ng/mL (range, 1540–86 300 ng/mL) (Costa et al., 2009). However, the highest reported PFOA serum concentration (114 100 ng/mL) was in a sample collected in 1995 from a male worker at a US facility that produced APFO, the ammonium salt form of PFOA (Olsen et al., 2000). In a study by Fu et al. (2016) of fluorochemical-production workers from a chemical plant in Hubei province, China, serum samples collected from 101 workers in the sulfonation department contained some of the highest concentrations of PFOS, with a mean of 14 002 ng/mL (range, 416-118 000 ng/mL). PFOA was also reported in this study; however, concentrations were higher for workers in the electrolytic department than in the sulfonation department, peaking at 32 000 ng/mL (Fu et al., 2016).

Using blood samples collected from workers at a fluoropolymer-manufacturing facility in West Virginia, USA, between 1972 and 2004, Woskie et al. (2012) constructed a job-exposure matrix (JEM) to retrospectively assess exposure spanning from 1950 to 2004. This facility used a process to manufacture certain fluoropolymers, such as PTFE, that involved the use of APFO as a surfactant in the polymerization of tetrafluoroethylene (TFE). Serum PFOA concentrations were highest in workers with tasks involving fine powder and granular PTFE (mean, 5470 ng/mL; range, 90-59 400 ng/mL), whereas workers involved in non-PTFE production (no use of APFO or PFOA) had the lowest serum concentrations (mean, 240 ng/mL; range, 7-4140 ng/mL). However even the latter experienced exposure that far exceeded that of the general population. For example, for the period 2000-2004, the geometric mean serum concentration for non-PTFE (no use of APFO or PFOA) production workers was 140 ng/mL, whereas the median serum concentration for the local community in 2005-2006 was 40 ng/mL, and geometric mean concentrations in adults aged > 20 years in the National Health and Nutrition Examination Survey (NHANES) were about 4-5 ng/mL between 1999-2000 and 2003-2004 (Woskie et al., 2012). Trends over time also differed by job category in modelled output, with exposures in fine powder/granular PTFE chemical operators peaking in 1980 (median serum concentration, > 6000 ng/mL), just before the implementation of exposure control measures. However, workers with only intermittent or background exposure to PFOA experienced higher exposure in 2000 (median serum concentration was estimated at nearly 1600 ng/mL), corresponding with peak production (Woskie et al., 2012).

Exposure in first responders (including firefighters), the second most frequently characterized worker population, is typically described as resulting from interactions with products containing PFAS, most notably AFFF, but exposure from turnout gear or the built environment has also been the subject of recent research (Peaslee et al., 2020; Young et al., 2021). Inhalation, dermal contact, and ingestion (e.g. via hand-to-mouth contact) are potential exposure routes because of contact with the product as well as with dust or air at the fire station and response scene (Mazumder et al., 2023; Rosenfeld et al., 2023). The highest serum concentrations of PFOS were reported among 149 firefighters working at AFFF training facilities in Australia; the mean PFOS concentration was 74 ng/mL (range, 3.4–391 ng/mL), compared with a mean concentration of 12 ng/mL for the general population in Australia. In this study, employment before the 2003 phase-out of PFOS-based AFFF at the facilities was positively associated with PFOS concentrations in serum samples collected in 2013. PFOA was not elevated in this population, with mean serum concentrations of 4.6 ng/mL (range, 0.3-18 ng/mL) (Rotander et al., 2015).
Results from other studies of firefighters suggest that there are differences by type of firefighter, firefighting activity, and geography. PFOS concentrations were higher in airport firefighters, (median, 10.69 ng/mL; range, 4.28–30.42 ng/mL) than in suburban firefighters (median, 4.04 ng/mL; range, 1.57-9.34 ng/mL) from the same geographical region; however, no difference was observed for PFOA (Leary et al., 2020). Burgess et al. (2023) compared serum concentrations collected from firefighters in municipal fire departments in three distinctly different regions of the USA and reported concentrations that were elevated above those reported in NHANES for branched PFOS (sm-PFOS, sum of perfluoromethylheptane sulfonate isomers) in all four departments, as well as linear PFOS (n-PFOS) and linear PFOA (n-PFOA) in two departments (Burgess et al., 2023). In contrast, PFOS levels were similar to those from NHANES for a sample of 101 male municipal firefighters in California, USA (Dobraca et al., 2015), and lower than those from NHANES in 138 volunteer municipal firefighters in New Jersey, USA (Graber et al., 2021).

In plasma from first responders from New York State and National Guard employees who responded to the collapse of the World Trade Center, New York City, USA, in the terrorist attack of 11 September 2001, PFOA concentrations were approximately twofold those of the general population. In this study, Tao et al. (2008a) used samples collected 6 months to 2 years after the collapse to assess exposure to PFAS categorically according to more and less exposure to smoke or dust. They observed higher PFOA concentrations in smoke-exposed individuals than in dust-exposed individuals, with the highest levels occurring in the group that was more highly exposed to smoke (mean, 10.21 ng/mL; range, 0.67-61 ng/mL). Background PFOA concentrations for 2001-2002 in the USA ranged from a median of 4.7 ng/mL for the full population to a mean of 6.98 ng/mL for non-Hispanic White

men (see Section 1.4.3 for exposure in the general population). PFOS was detected in all samples, with mean concentrations ranging from 22.9 to 33.9 ng/mL across the study groups; however, concentrations were not elevated above those in the general US population (median, 25.8 ng/mL) or in the general population in two US cities (in Portland, Oregon, the median was 26.0 ng/mL, and in Boston, Massachusetts, the median was 29.5 ng/mL) (Tao et al., 2008a).

There were few available biomonitoring data for other worker populations. In the Human Biomonitoring for Europe (HBM4EU) samples collected in project, 2018-2019 contained median PFOS concentrations of 4.97 ng/mL (maximum, 1513 ng/mL) in welders and 4.83 ng/mL (maximum, 789 ng/mL) in metal plating workers - an exposure that is anticipated to have resulted from the use of PFOS as a mist suppressant in chrome plating baths (Göen et al., 2024). Shi et al. (2016) reported a median serum PFOS concentration in metal plating workers in China of 40 ng/mL (range, 2.4–1323 ng/mL) (Shi et al., 2016). In a study of workers in shoe and leather-related industries (2011) from the same region of China, mean serum concentrations were 6.93 ng/mL (range, 0.17-117.77 ng/mL) and 14.18 ng/mL (range, 0.05-31.66 ng/mL) for PFOA and PFOS, respectively (Zhang et al., 2011). In a separate study of textile factory workers (2009) in another region of China, the mean blood concentration of PFOA was 5.46 ng/mL (range, 2.35-10.93 ng/mL), and that of PFOS was 5.73 ng/mL (range, 1.34-14.75 ng/mL) (Lu et al., 2014). In the same study, mean concentrations measured in barbers, who may be exposed through the use of products containing PFOA and PFOS, were 3.18 ng/mL (range, 0.78-12.18 ng/mL) and 2.56 ng/mL (range, 0.44-7.72 ng/mL) for PFOA and PFOS, respectively (Lu et al., 2014). PFOA concentrations measured in whole blood from ski-wax technicians working with teams competing in World Cup events in Europe between 2007 and 2011

ranged from 1.9 to 630 ng/mL (mean, 130 ng/mL) (Nilsson et al., 2013a). In this population, for which samples were collected at multiple time points across multiple ski seasons, PFOA concentrations increased in technicians with "low" initial concentrations, but decreased or remained at steady-state in technicians with "high" initial concentrations (Nilsson et al., 2010, 2013a). The median PFOS concentration (12.2 ng/mL) was not elevated when compared with exposure in the general population (Nilsson et al., 2010).

In studies of agricultural workers in China and Sri Lanka, as well as retail and office workers in the USA, concentrations of PFOA and PFOS have been reported that are similar to those in the corresponding general population (see Table S1.15, Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: https://publications.iarc. who.int/636) (Guruge et al., 2005; Fraser et al., 2012; Zhou et al., 2014; Wu et al., 2019; Clarity et al., 2021), with the exception of commercial fishery workers in China, for whom the potential for a large dietary exposure through consumption of employer-provided fish probably contributed to the mean serum PFOS concentration of 11 400 ng/mL (range, 82.6-31 400 ng/mL) (Zhou <u>et al., 2014</u>).

Although biomonitoring data collected over time and by occupation or industry were relatively limited, the available data indicated the potential for exposure to PFOA and PFOS in diverse occupational settings, including primary (e.g. fluorochemical production) and secondary (e.g. metal plating, textile mill) manufacturing, public safety (e.g. firefighters) and services (e.g. ski-wax technicians, barber). The exposure characterization was the most robust for fluorochemical-production workers, followed by first responders. Although the magnitude of exposure in these populations differed substantially, there was evidence for intrapopulation differences in exposure by task or activity, with higher concentrations being measured in workers with

100

tasks or activities involving known contact with materials containing PFOA or PFOS.

[The Working Group noted that blood concentrations across occupational populations differ by orders of magnitude. For example, median concentrations of PFOA and PFOS reported for fluorochemical-production workers are often 100 to 10 000 times as high as those of firefighters, depending on the study. The Working Group also noted that PFOA and PFOS have primarily been measured in worker populations with a known source of exposure (e.g. manufacturing of fluorochemicals, use of AFFF); the absence of data for the many occupations with the potential for PFOA and PFOS exposure is not evidence of the absence of exposure.]

(b) Industrial hygiene samples

Characterization of PFOA and PFOS in the work environment, for example through air samples, is limited by the availability and consistency of methods (see Section 1.3) and comparable data. Although large variability exists in the reported concentrations, measures of the work environment, including samples of workplace air, dust, surfaces, and other work-related materials, frequently contain higher concentrations of PFOA and PFOS in facilities engaged in manufacturing or use of PFAS-laden products (such as fluorochemical production, secondary manufacturing, firefighting, and ski-wax application) than in other occupational environments (such as offices, schools, retail stores, and hotels). In these studies, measurement of PFOA and PFOS is often accompanied by measurement of related or precursor compounds (see Section 1.4), such as fluorotelomer alcohols (e.g. 8:2 FTOH), fluorotelomer sulfonic acids (e.g. 8:2 FTS), sulfonamides (e.g. N-EtFOSA), and phosphoric acid diesters (e.g. 8:2 diPAP) (Christensen and Calkins, 2023). Precursor compounds such as these may transform in the environment or the body to PFOA, PFOS, or other PFAS (Kolanczyk et al., 2023). For additional information on the toxicokinetics of PFOA and PFOS, see Section 4.1.

Occupational exposure at a chemical plant in Hubei province, China, was described by Gao et al. (2015) as indicating higher concentrations of PFOS than PFOA in indoor dust, with geometric mean concentrations of 830 ng/g dust (range, LOD to 658 343 ng/g dust) and 360 ng/g dust (range, 41.3–85 139 ng/g dust), respectively. However, the reverse was observed in total suspended particles, with geometric means of 0.4 ng/m³ (range, 0.03-78 ng/m³) and 0.94 ng/m³ (range, 0.04-1123 ng/m³) for PFOS and PFOA, respectively (Gao et al., 2015). At a facility producing APFO and PFOA in the USA, Kaiser et al. (2010) reported an 8-hour time-weighted average (TWA) median air concentration of 34 μ g/m³ (range, 4–65 μ g/m³) measured near the process sumps. Concentrations were higher at lower pH and water levels. They also reported that the sublimation rate measured for PFOA (360 µg/hour) was higher than that for APFO (0.302 µg/hour) (<u>Kaiser et al., 2010</u>).

In the studies on ski-wax technicians, personal breathing zone samples collected over three ski seasons spanning 2007 to 2010 contained higher concentrations of 8:2 FTOH (a precursor to PFOA) than PFOA, with concentrations ranging from 0.834 to 997 μ g/m³ and from 0.027 to 14.9 μ g/m³ for 8:2 FTOH and PFOA, respectively. Area aerosol samples did not contain FTOH; however, PFOA concentrations were higher in the inhalable fraction than the respirable fraction, with mean concentrations of $16 \,\mu\text{g/m}^3$ (range, 2.11–52.8 $\mu\text{g/m}^3$) and 9.91 $\mu\text{g/m}^3$ (range, 0.62–26.8 µg/m³), respectively (Nilsson et al., 2013b). The metabolism of FTOH to PFOA in workers' blood was supported by the presence of FTOH degradation products, 5:3 FTCA and 7:3 FTCA, in the blood of workers in this study (Nilsson et al., 2013a). Although concentrations were lower, Freberg et al. (2010) reported a similar relationship between inhalable and respirable fractions, with PFOA concentrations ranging

from 5.1 to 35 ng/m³ and 5.6 to 38 ng/m³ in inhalable and respirable fractions, respectively. PFOA was detected at higher concentrations in powder wax (median, 2.7 µg/g product; range, 0.29-12 µg/g) than in solid block wax (median, 0.68 µg/g product; range, < LOQ to 3.8 µg/g). PFOS was not detected in any dust samples and was only detected in a few of the powder ski-wax samples (maximum, 0.149 µg/g) (Freberg et al., 2010).

Hall et al. (2020) reported higher concentrations of PFOA and PFOS in dust samples collected from 49 fire stations across the USA and Canada than in samples from 184 homes in North Carolina, with median (and maximum) concentrations in fire stations and homes of 17.6 ng/g dust (maximum, 791 ng/g dust) and 7.9 ng/g dust (2350 ng/g dust) for PFOA, respectively and 64.5 ng/g dust (74 370 ng/g dust) and 4.4 ng/g dust (2810 ng/g dust) for PFOS, respectively (Hall et al., 2020). Within fire stations, concentrations of the PFOS precursor, N-ethyl-perfluorooctane sulfonamido acetic acid (N-EtFOSAA), in dust were higher in living areas (median, 87.5 ng/g dust; range, 0.748-1800 ng/g dust) than in gear storage (median, 7.84 ng/g dust; range, < MDL to 299 ng/g dust) or the apparatus bay (median, 3.51 ng/g dust; range, < MDL to 159 ng/g dust) (Young et al., 2021). Using silicone wristbands worn while on- or off-shift, Levasseur et al. (2022) reported that PFOS concentrations while on-duty and responding to fires were 2.5 times as high as off-duty exposures; however, PFOA concentrations while on-duty and responding to fires were lower than off-duty exposures (Levasseur et al., 2022). When analysing textiles used in new firefighter personal protective equipment (PPE), Maizel et al. (2023) detected PFOA and PFOS in 7 of 20 textiles tested, with concentrations all < 2 ng/g. However, the highest concentrations were reported for precursor compounds, including 6:2 fluorotelomer methacrylate (mean, 1570 ng/g), 6:2 FTOH (mean, 613 ng/g), and 6:2 fluorotelomer sulfonic acid (mean, 393 ng/g). In

a separate study of PPE worn by firefighters, sets of used and unused turnout gear thermal liners, moisture barriers, and outer shells were analysed. PFOS levels were largely below the LOD, but PFOA was detected at higher concentrations than other PFAS, with the highest concentration measured in a used thermal liner from trousers worn in 2014 (850 ppb) (Peaslee et al., 2020).

Area air and dust collected over 17 days in 2014 from a speciality, water-repellent textile-manufacturing facility in China were analysed for PFOA, PFOS, and numerous precursor compounds. Heydebreck et al. (2016) reported higher concentrations in the gas phase than the particle phase in air samples, with generally lower concentrations in settled dust. Fluorotelomer alcohols (8:2 FTOH and 10:2 FTOH) were the dominant analytes measured in workplace air, with the highest concentrations reported for 8:2 FTOH from heat setting (91.3 μ g/m³) and drying (87.7 μ g/m³) operations in one of the workshops, two processes that occur after the durable water-repellent coating has been applied to the textile. PFOA was the ionic PFAS measured at the highest concentrations in workplace air, with highest concentrations measured during the drying operation in the same workshop (8.48 ng/m³). PFOS was generally below the method detection limit (5.33 ng/g) (Heydebreck et al., 2016).

Some studies have described exposure in occupational environments where the exposure sources are similar to sources for the general public, such as through direct contact or contact with dust from consumer products (e.g. clothing) or the built environment (e.g. carpets). In these settings, exposures vary across studies of classrooms, offices, and retail stores. In the UK, <u>Goosey</u> and Harrad (2011) reported higher concentrations of PFOS in dust collected from classrooms (mean, 980 ng/g dust; range, 22–3700 ng/g dust) than from offices (mean, 370 ng/g dust; range, 20–1000 ng/g dust). Concentrations of PFOA were variable in classrooms, (mean, 310 ng/g dust; range, 18-1700 ng/g dust), and in offices (mean, 550 ng/g dust; range, < LOD to 6000 ng/g dust) (Goosey and Harrad, 2011). PFOA has been reported to be the predominant compound in samples analysed for perfluoroalkyl carboxylic and sulfonic acids from electronic shops, offices, libraries, and internet cafés (Besis et al., 2019), and to be present at higher concentrations in office than in residential settings. However, this is not the case for PFOS (D'Hollander et al., 2010; Goosey and Harrad, 2011; Fraser et al., 2012). In office settings, Fraser et al. (2012) reported a strong positive association between FTOHs measured in office air and serum PFOA concentrations measured in office workers, with geometric mean air concentration of 8:2 FTOH of 9.92 ng/m³ (range, 0.28–70.6 ng/m³).

In the only study of dermal exposure, skin exposure to the pesticide sulfluramid (*N*-EtFOSA), a precursor of PFOS, in pesticide manufacturing workers during an 8-hour shift was measured at six different locations across the body. Exposure was greatest on the hands, with a mean of 89.7 μ g/day, followed by the left leg (73.0 μ g/day) and arms (72.1 μ g/day); however, there was substantial variability (Machado-Neto et al., 1999).

(c) Protection measures to limit exposure

Approaches to reducing occupational exposures are commonly categorized into the hierarchy of controls - an effectiveness-based hierarchy of actions. This framework categorizes elimination and substitution as the most effective risk management measures, with PPE considered the least effective because of reliance on correct and consistent use by individual workers (NIOSH, 2023). Effective control measures have been documented in research studies involving fluorochemical-production facilities and ski waxing. For both industries, the use of local exhaust ventilation near the exposure source was linked to reductions in PFAS exposure. Other measures that led to reductions included

maintaining pH levels above 7 to reduce volatilization potential and wetting PFAS-containing dry powders at fluorochemical-production facilities, and replacing powder wax with block wax at ski-waxing facilities (Christensen and Calkins, <u>2023</u>). The retrospective assessment by <u>Woskie</u> et al. (2012) demonstrated that, despite increases in production between 1980 and 2000, incorporation of exposure controls such as engineering and PPE resulted in decreasing serum concentrations in workers in the most highly exposed job category (Woskie et al., 2012). [The Working Group noted that the implementation of effective control measures may affect occupational exposures, including those relevant to epidemiological studies. In this case, samples collected after the implementation of effective controls may not be representative of exposures that occurred under prior conditions (and vice versa).]

1.4.3 Exposure of the general population

The general population is exposed via multiple sources to PFOA and PFOS and, given their widespread use, environmental contamination, and long persistence, both compounds are detectable in the blood of virtually all people tested (OECD, 2015b). Serum or plasma are the most common biological matrices used in biomonitoring campaigns. Long-chain PFAS such as PFOA and PFOS are not commonly measured in urine or breast milk samples, because of the lack of sensitivity of most available analytical tools (Worley et al., 2017) (see Section 1.3.4).

Measured levels in whole blood, serum or plasma are useful indicators of exposure since they reflect accumulated intake from all sources and, given the long half-lives of PFOA and PFOS, they are quite stable indicators of body burden (see also Section 4.1). Where people are exposed to a local substantial source of PFOA or PFOS, such as contaminated drinking-water, or occupational exposure, such exposure will be the main source (Pitter et al., 2020). For the general population not living close to a major point source, measured serum levels will reflect diverse exposure sources, including food, water, air, indoor dust, and consumer products. Also, individual serum levels will vary, reflecting not only degree of intake but individual variability in efficiency of uptake, distribution, and excretion (Section 4.1).

(a) Human exposure estimation

The general population is exposed via the diet, drinking-water, household dust, consumer products, and inhalation of contaminated air (Sunderland et al., 2019; De Silva et al., 2021). [The Working Group noted that the data presented in this section mainly draw from studies performed after 2000. Extrapolation to earlier points in time was difficult because of the sparse data available before 2000.] When drinking-water is contaminated by a specific pollution source with high emissions, drinking-water is the main exposure source. For example, in the Mid-Ohio Valley "C8" study population in the USA, the principal PFOA exposure source was drinking-water. In the most highly contaminated water district, the population's PFOA serum levels were 17-fold those in the water district with the lowest contamination, and drinking-water was the main contributor to the total body burden (<u>Steenland et al., 2009</u>).

For general populations without a recognized emission source or not living in a highly contaminated location, several studies have sought to estimate the relative importance of different exposure sources, and these are summarized in <u>Table 1.16</u> and <u>Table 1.17</u>. Some of the studies have in addition estimated the possible pathways of exposure. There was some variability in the dietary contribution but, in all cases, diet has been estimated to be the most important exposure source. Exposure can derive both from the food being contaminated from uptake during growing or grazing, and from migration from food packaging materials (see Section 1.4.1(d)).

Location, sampling	Relative contribution of exposure route (%)						Comments	Reference	
time	Oral			Inhalation Dermal		Via	-		
	Diet	Dust	Water	Food packaging			precursors		
Germany, 2005; Japan, 2004	85	6	1	3			2-8		Vestergren et al. (2008); Vestergren and Cousins (2009)
Norway, 2008	84	5	11		0.13				<u>Haug et al.</u> (2011)
USA, 2003/2004	66	9	24		< 1	< 1			<u>Lorber and</u> <u>Egeghy</u> (2011)
North America, Europe, Republic of Korea, Japan 2007/2008	47	8	12		6		27 ^a		<u>Gebbink</u> <u>et al. (2015)</u>
Republic of Korea, 2009	41		37		22		5		<u>Tian et al.</u> (2016)
China, 2013/2014	> 99		< 1						<u>Shan et al.</u> (2016)
Finland, 2005/2006, 2010/2011, 2014/2015	95	< 2.5			< 2.5			Children aged 10 years	<u>Balk et al.</u> (2019)
Ireland, 2016/2017/2018	NR NR	1 3	37 74		62 23			Adults Children	<u>Harrad et al.</u> <u>(2019)</u>
Norway, 2013/2014	92	4			3	< 1			<u>Poothong</u> <u>et al. (2020)</u>

Table 1.16 Estimated relative contribution (%) of various routes of exposure to total PFOA in the general population

NR, not reported; PFOA, perfluorooctanoic acid.; USA, United States of America.

^a Value given for the intermediate exposure scenario; estimated contribution varied according to exposure scenario from 13% to 64%.

In 2020, the European Food Safety Authority made an assessment of exposure of the European population to several PFAS, including PFOA and PFOS, on the basis of data available concerning the presence of these PFAS in different food categories and on consumption data (EFSA Panel on <u>Contaminants in the Food Chain, 2020</u>). For the lower-bound scenario, which was considered the most realistic by the panel, median dietary exposure to PFOA was estimated to range between 0.17 and 0.41 ng/kg body weight (bw) per day for different age categories (<u>Table 1.18</u>). Median dietary exposure to PFOS for the same scenario was estimated to range between 0.36 and 1.34 ng/kg bw per day. Similar average intakes were estimated in the Netherlands: 0.2 ng/kg bw per day for PFOA and 0.3 ng/kg bw per day for PFOS (Noorlander et al., 2011). In both studies and for both compounds, mean dietary exposure was highest for toddlers (defined as children aged 1–3 years).

In a study conducted in the USA in 2020 (Zheng et al., 2020), the occurrence and distribution of PFAS, including PFOA and PFAS, was determined in the childcare environment (dust and nap mats), and children's exposure

Location	Relative contribution of exposure route (%)					Comments	Reference		
	Oral			Inhalation Dermal	Via				
	Diet	Dust	Water	Food packaging			precursors		
USA, 2003/2004	72	6	22		< 1	< 1			<u>Egeghy</u> and Lorber (2011)
Norway, 2008	96	1	1		2				<u>Haug et al.</u> (2011)
North America, Europe, Republic of Korea, Japan 2007/2008	66	10	7		2		16 ^a		<u>Gebbink</u> et al. (2015)
Republic of Korea, 2009	93		4		3				<u>Tian et al.</u> (2016)
China, 2013/2014	100		< 1						<u>Shan et al.</u> (2016)
Finland, 2005/2006, 2010/2011, 2014/2015	95	< 2.5			< 2.5			Children aged 10 years	<u>Balk et al.</u> (2019)
Ireland, 2016/2017/2018	NR NR	21 55	30 35		49 10			Adults Children	<u>Harrad</u> <u>et al. (2019)</u>
Norway, 2013/2014	75				3				<u>Poothong</u> <u>et al.</u> (2020)

Table 1.17 Estimated relative contribution of various routes of exposure to total PFOS in the general population

PFOS, perfluorooctanesulfonic acid; USA, United States of America.

^a Value given for the intermediate exposure scenario; estimated contribution varied with exposure scenario from 11% to 33%.

through dust ingestion and dermal absorption was estimated. The estimated daily intake of PFOA through dust ingestion for toddlers was 0.03 ng/kg bw per day (median value). In the case of dermal absorption, the estimated daily intake was 0.002 ng/kg bw per day (median value). For PFOS, the equivalent values were 0.002 and 0.001 ng/kg bw per day, respectively (Zheng et al., 2020). A modelling exercise for US children and adults considered both direct exposure to PFOS and exposure to precursors. Median adult intake was 4.2 ng/kg bw per day, about half of which was from precursors. This estimate was validated by comparing with intake calculated from a one-compartment pharmacokinetic model with a range of values for the volume of distribution

 $(V_{\rm d})$. With the more plausible $V_{\rm d}$ values (see Section 4.1), agreement was quite close (Egephy and Lorber, 2011).

For some individuals, a considerable part of the intake could be from personal care products and cosmetics (<u>Husøy et al., 2023</u>). [The Working Group noted that recent data suggested that dermal uptake is likely to be higher than was previously assumed (see <u>Abraham and Monien</u>, <u>2022</u>, and Section 4.1).]

(b) Biomonitoring data for the general population (serum and plasma)

Repeated population surveys with the aim of measuring PFOA and PFOS concentrations in serum or plasma, or in archived blood samples,

Age group ^a	Dietary exposure (ng/kg bw per day) ^b				
	PFOA	PFOS			
Infants	0.19	0.36			
Toddlers	0.41	1.34			
Other children	0.30	1.02			
Adolescents	0.17	0.53			
Adults	0.18	0.58			
Elderly adults	0.17	0.59			

Table 1.18 Median dietary exposure to PFOA and PFOS for different age groups in the population of Europe

bw, body weight; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid.

^a Age ranges: infants, < 12 months; toddlers, \geq 12 to < 36 months; other children, \geq 36 months to < 10 years; adolescents, \geq 10 to < 18 years; adults: \geq 18 to < 65 years; elderly, \geq 65 to < 75 years; very elderly, \geq 75 years.

^b Only the lower-bound estimates are presented since these were considered to be more realistic by the European Food Safety Authority Panel. From <u>EFSA Panel on Contaminants in the Food Chain (2020)</u>.

have permitted exposure trends to be observed in several countries (Fig. 1.7). For most studies, estimated total PFOS or PFOA concentrations are presented, but more recent studies present concentrations of isomers, distinguishing linear and different branched isomers, more commonly for PFOS than PFOA (e.g. NHANES, 2023).

In Japan between 1983 and 1999, results for PFOA showed a clear trend, with geometric mean concentrations in men rising from [2.5 to 11] ng/mL, and in women from [1.8 to 8.1] ng/mL, corresponding to a mean annual increase of 0.49 and 0.42 ng/mL, respectively. For PFOS, there was no clear trend, with mean concentrations in the range of approximately [15–23] ng/mL for men and [13 to 19] ng/mL for women (Harada et al., 2007).

In China, in the region of Shenyang, the results of a study from 2006 showed mean PFOS concentrations of 142 ng/mL (range, 31.7–225 ng/mL) for men and 170 ng/mL (range, 80.4–310 ng/mL) for women (<u>Yeung et al., 2006</u>). Concentrations of PFOS in a previous study (Jin et al., 2003) were 40 ng/mL (range, 5.32–145 ng/mL) for men and 45.5 ng/mL (range, 10.6–142 ng/mL) for women. On this basis, PFOS concentrations measured in the study conducted in 2006 were 3–4 times as high as those reported in the previous study (<u>Yeung et al., 2006</u>).

In the USA, analysis of archived blood samples collected in 1974 (serum) and 1989 (plasma) from volunteer participants in a large community health study indicated an increase in PFOA concentrations from median values of 2.3 μ g/L in 1974 to 5.6 μ g/L in 1989; for PFOS, the equivalent figures were 29.5 μ g/L in 1974 and 34.7 μ g/L in 1989 (Olsen et al., 2005).

A unique study reporting PFOA and PFOS concentrations in the same 59 individuals over a long time period in Tromsø, Norway, showed clear trends (Nøst et al., 2014). Samples were collected in five rounds – in 1979, 1986, 1994, 2001, and 2007 – and for both PFOA and PFOS, average concentrations peaked in 2001. Correlations were high between each pair of subsequent rounds for both PFOA and PFOS (Spearman correlation, ρ , in the range 0.6–0.8; all P < 0.05), indicating some stability in exposure as determined by single measurements (Nøst et al., 2014).

Trends towards falling concentrations in the last 20–30 years have been shown in several countries (Fig. 1.7). For example, data from the USA derived from NHANES, a large national biomonitoring programme with repeated sampling cycles that has included PFAS in monitoring campaigns



Fig. 1.7 Median PFOA and PFOS concentrations reported in blood samples from the adult general population in several countries

PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; USA, United States of America. Data aggregated from <u>HBM4EU (2023)</u>, <u>CDC (2023)</u>, and <u>Government of Canada (2023)</u>.

since 1999, have shown this trend very clearly. For PFOA, geometric mean serum concentrations were 5.2 μ g/L in 1999–2000, close to 4 μ g/L between 2003 and 2008, then declined steadily in subsequent rounds, falling to 1.42 μ g/L in 2017–2018. Equivalent trend data were observed for PFOS, with geometric mean serum concentrations of 30.4 μ g/L in 1999, falling in each survey, down to 4.3 μ g/L in 2017–2018 (Kato et al., 2011; NHANES, 2023). A similar pattern with downward trend was reported for PFOA and PFOS concentrations measured in archived blood spots collected in 1997–2007 from infants in New York, USA (Spliethoff et al., 2008).

A decreasing trend was also observed in Australia (Toms et al., 2014; Eriksson et al., 2017), Japan (Okada et al., 2013), and in several countries in the EU (Fig. 1.7).

In Norway, a steady increasing trend for PFOA concentrations in serum was observed from 1977 (0.58 μ g/L) up to the 1990s (5.2 μ g/L in 1993), then a decline to 2.7 μ g/L in 2006. For PFOS, serum concentrations rose from 3.8 μ g/L in 1977 to 33 μ g/L during the 1990s, falling to 12 μ g/L by 2006 (Haug et al., 2009).

In Germany, data from archived plasma samples from 20 participants (10 men and 10 women) randomly chosen from the monitoring programmes in Münster between 1982 and 2010 were analysed; PFOA concentrations were found to be highest in 1986 (7.4 µg/L) and decreased from 2007 (5.2 µg/L) to 2010 (3.1 µg/L), but in other years there were no clear trends. For PFOS, the pattern was clearer, rising from 15.4 µg/L in 1982 to 28.6 µg/L in 1989, and subsequently falling steadily to 12.7 µg/L in 2005 and 3.8 µg/L in 2010 (Schröter-Kermani et al., 2013).

A similar pattern of decline since 2000 is evident in data assembled from many recent smaller studies across Europe. In the HBM4EU project, data were assembled across 12 European countries, combining 32 different surveys. The surveys were not all directly comparable because of variation in the age and sex composition, but together they provided a picture of falling serum levels over time and exposure ranges between countries at the same points in time (HBM4EU, 2023). In this European project, although early studies were sparse (with only one study including data from 2000), PFOA body burdens were comparable to US NHANES results, with PFOA concentrations in the range of 3 to 6 μ g/L up to around 2010, falling to 1 to 2 μ g/L in recent years. For PFOS, levels in the EU were somewhat lower than in the USA, being mainly between 6 and 10 μ g/L around 2010 for the European data, falling to between 1 and 3 μ g/L in recent years (CDC, 2023; HBM4EU, 2023) (Fig. 1.7).

Exposure data for teenagers in this EU project suggested that exposure levels were significantly higher in north and west Europe than in the south and east. Concentrations of PFOA and PFOS were significantly higher in boys than in girls, and significantly higher concentrations were found in teenagers from households with a higher education level. In the same EU project, the consumption of seafood and fish at least twice per week was significantly associated with a 21% (95% CI, 12–31%) increase in PFOS concentrations. The same trend was observed for PFOA but was not statistically significant (<u>Richterová et al.</u>, 2023).

PFOA and PFOS levels have been shown to vary by age and sex (Frisbee et al., 2009; Kato et al., 2011; Pitter et al., 2020; NHANES, 2023). Serum levels are consistently higher in males than females, reflecting differences in excretion (with women excreting additionally via menstruation, pregnancy and lactation), and possibly differences in intake and pharmacokinetics (see Li et al., 2022c, and Section 4.1). By age, serum levels measured in cross-sectional surveys showed some differences, with older people having higher serum levels. This may reflect variation in the routes of exposure according to age and biological changes, but the time trends of exposure would also be important, given the long halflives in people. Infants can have high levels from

maternal and lactational transfer that fall in the post-lactation period (Fromme et al., 2010).With emissions and ambient levels falling, higher levels in older people will in part reflect the fact that they were exposed at earlier time periods when intake was likely to be higher (Nøst et al., 2014). In NHANES data for 1999–2000 and 2003–2004, a modest increasing slope was evident in PFOS levels from age 12 to \geq 60 years, but no slope was evident for PFOA (Calafat et al., 2007). In the Mid-Ohio C8 population, which had a wider age range, there was a clear increasing trend from age < 10 to \geq 80 years for PFOS levels in males, but for females the trend was decreasing until age 30-39 years, then rising thereafter (Frisbee et al., 2010). For PFOA, concentrations in females are lower than in males in most age groups, but both males and females show a similar pattern, with a minimum at around age 30 years.

NHANES also provided information on ethnicity: there were some small differences between White and Black people, but PFOS levels were markedly lower for Hispanic people, with smaller differences for PFOA (<u>Calafat et al.</u>, 2007).

[The Working Group noted that although there were differences between countries, the overall pattern in general population serum or plasma samples across the world has been a rise in concentrations since the earliest measurements in the 1970s, reaching a peak in the 1990s or close to 2000. Subsequently, trends towards falling serum concentrations have been observed for both PFOA and PFOS. The most notable difference between countries was a higher level of PFOS in earlier samples from the USA compared with other countries.]

Multiple PFAS with long half-lives and slow rates of excretion, such as PFOA and PFOS, have been monitored in serum samples. Serum concentrations tend to be correlated with each other; for example, logarithmic concentrations showed a significant Pearson correlation coefficient of 0.66 between PFOA and PFOS in the NHANES data (Calafat et al., 2007). Correlation coefficients for circulating PFOA and PFOS levels were similar in several general populations in the cancer studies reviewed in Section 1.6.1, but there was large variability, with values ranging from < 0.15 to > 0.7 (see Table S1.22, Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: <u>https://publications.iarc.who.int/636</u>). There were also significant correlations with other widespread PFAS with long half-lives, notably perfluorononanoic acid (PFNA) and perfluorohexanesulfonic acid (PFHxS). [The Working Group noted that these correlations may reflect a correlation in exposure or a correlation between different PFAS in individual rates of uptake and excretion.]

(c) Biomonitoring data for populations living at contaminated sites

Several large communities have experienced high exposure to PFAS because of environmental contamination, related mainly to the use firefighting foams containing PFAS (e.g. airports, military facilities), certain industrial facilities where PFAS are produced or used and emitted to the environment, and sites related to PFAScontaining waste (Salvatore et al., 2022). This has led to higher blood concentrations of PFOA and PFOS in some of these communities compared with the general population (see Fig. 1.8).

In the USA, high serum concentrations of PFOA were measured in samples collected in 2005–2006 from 69 030 residents living near a PTFE-production facility in West Virginia, USA (the C8 Health Project); the overall geometric mean was 32.9 μ g/L, and the arithmetic mean was 82.9 μ g/L. Exposures in that community varied substantially across six water districts; the mean serum concentration of PFOA was 16 μ g/L in the two water districts with the lowest concentrations of PFOA in water, and 228 μ g/L in the water district with the highest concentrations (Frisbee et al., 2009).





NHANES, National Health and Nutrition Examination Survey; PFAS, per- and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; USA, United States of America.

The dotted line shows the respective concentration measured in the NHANES population in 2010 (<u>NHANES, 2023</u>). For the individual study populations, see <u>ATSDR (2023)</u>, <u>Pitter et al.</u> (2020), <u>Ingelido et al. (2010)</u>, <u>Hölzer et al. (2008)</u>, <u>Frisbee et al. (2009)</u>, <u>Xu et al. (2021)</u>, and <u>Herrick et al. (2017)</u>.

110

Also in the USA, as early as the 1970s, the Fairchild Air Force Base located in the City of Airway Heights in Spokane County, Washington, used AFFF containing PFAS during firefighter training. Over time, PFAS from the AFFF entered the ground, moved into the groundwater to off-site locations, and affected nearby municipal wells. In samples from 2019, geometric mean PFOA and PFOS serum concentrations of 9.72 μ g/L and 42.4 μ g/L, respectively, were reported for community residents (ATSDR, 2023).

In China, a lake adjacent to a fluorochemical-production factory was contaminated with PFAS, and serum concentrations of several PFAS were assessed in fishermen and their families who were exposed primarily via eating locally caught fish (Zhou et al., 2014). Among family members of fishery employees, extremely high PFOS levels were found, with a median concentration of linear PFOS of 2720 ng/mL, branched PFOS of 620 ng/mL, and sum of PFOS of 3540 ng/mL. PFOA concentrations were slightly elevated, with a median of 11.7 ng/mL.

At the end of 2013, drinking-water from one of the two municipal waterworks in Ronneby, Blekinge County, Sweden, was found to be contaminated by firefighting foams used at a nearby military airfield. Drinking-water containing high levels of PFOS and PFHxS, and to a lesser extent PFOA, had been distributed to approximately one third of Ronneby households (total population, approximately 30 000) since the mid-1980s (Xu et al., 2021b). Blood samples and demographic data were collected from 3297 Ronneby residents and 226 individuals from a reference group. The population geometric means for serum PFOA and PFOS concentrations were 6.8 and 135 µg/L for all Ronneby residents, i.e. 35 and 4.5 times, respectively, as high as for the reference group (Xu et al., 2021b).

In spring 2013, groundwater of part of the Veneto region in north-eastern Italy was found to be contaminated with mostly PFOA and to a smaller degree with PFOS and other PFAS from a factory that had been manufacturing a variety of PFAS since the 1960s. A population of 140 000 was potentially affected, and a population-based screening programme including measurement of serum PFAS was offered by the regional health service to residents who were exposed to PFAS via contaminated drinking-water (Ingelido et al., 2018). Among 18 122 subjects aged 14–39 years living in the Veneto region, the median concentration of PFOA was elevated, at 44 μ g/L (Pitter et al., 2020), whereas the median concentration of PFOS, 3.9 μ g/L, was close to levels reported for the Italian general population (Ingelido et al., 2010).

A study of 641 residents of Arnsberg, Germany, in 2006 reported geometric mean PFOA serum concentrations of 22.1, 23.4, and 25.3 μ g/L in children, mothers, and men, respectively, because of surface water contamination from upstream agricultural use of soil conditioner mingled with industrial waste (Hölzer et al., 2008). PFOA levels of children and adults living in Arnsberg were 4.5–8.3 times as high as those of the reference population used in the study and living in non-contaminated sites (Hölzer et al., 2008).

(d) Other biological matrices used in biomonitoring

Although serum samples are most commonly used in biomonitoring campaigns, PFOA and PFOS have also been measured in other biological matrices, such as breast milk and urine. [The Working Group noted that these biological matrices could also be used in biomonitoring, particularly in biomonitoring campaigns performed in highly contaminated sites.]

PFOA and PFOS have been detected in breast milk, which is a significant route of exposure to infants through breastfeeding. [The Working Group noted that concentrations in breast milk are much lower than in serum; however, the large volume of breast milk ingested by infants on a body-weight basis results in considerable exposure.] In a study of 109 paired maternal serum and breast milk samples in a population with high PFAS exposure in Sweden, breast milk concentrations were 0.03 ng/mL for PFOA and 0.130 ng/mL for PFOS (Blomberg et al., 2023). The transfer efficiency or ratio of breast milk to serum concentration was 2.16% for PFOA and 1.02% for PFOS (Blomberg et al., 2023). In a summary of 23 studies, all except 4 reported concentrations in breast milk that were above the LOQ in > 50% of samples; however, LOQs varied between studies (Fromme et al., 2022). Median values above the LOQ for PFOA were 7.2, 26, and 138 ng/L for three studies in Spain (Serrano et al., <u>2021; Motas Guzmàn et al., 2016; and Beser et al.,</u> 2019, respectively), and median values were 139, 121 and 35 ng/L in three studies in China (Awad et al., 2020; Liu et al., 2010, 2011). Median values for PFOA in two studies in the USA were 14 and 36 ng/L (Tao et al., 2008b; Zheng et al., 2021). In a study of a contaminated site in Germany, PFOA could be quantified in all breast milk samples, with a mean value of 199 ng/L (range, 33-854 ng/L) (Fromme et al., 2022). PFOS was observed in only 3 out of 13 samples, at levels of 33 ng/L, 35 ng/L, and 61 ng/L.

PFOA and PFOS can be detected in urine, although concentrations in urine are much lower than in serum. In a study of 104 paired samples in a population with high PFAS exposure in Sweden, the median ratio of urinary to serum level was 0.23% for PFOA and 0.07% for linear PFOS and ranged from 0.02% to 0.07% for branched PFOS. Median urinary concentrations for the three sampling rounds carried out were between 0.017 and 0.025 ng/mL for PFOA and between 0.050 and 0.075 ng/mL for PFOS (Li et al., 2022c). In general population campaigns, values in urine samples are mostly below the LOD. In the NHANES 2013-2014 round, urine levels were above the LOD (0.1 ng/mL) for < 0.1%of the population, even though serum levels were above the LOD for almost 100% of participants (<u>Calafat et al., 2019</u>).

[The Working Group noted that PFOA and PFOS have been measured in other biospecimens, such as nails, hair, and semen, but these have rarely been used to assess exposure for epidemiological studies.]

1.5 Regulations and guidelines

Regulations, guidelines, and guidance for PFOA and PFOS have been established by international, national, and local governing bodies, as well as nongovernmental organizations (e.g. standards, non-profit, and professional organizations). The aim is to reduce human exposure and environmental contamination via approaches covering production, use, and disposal; occupational exposures; food and consumer products; environmental media; and biomonitoring. Unless otherwise stated, numerical standards and guidelines for PFOA and PFOS are generally based on non-cancer effects.

Internationally, PFOS and PFOA and their salts derivatives are recognized as persistent organic pollutants and were included in the Stockholm Convention on 2009 and 2019, respectively. PFOS is listed under Annex B (Restriction) (measures must be taken to restrict production and use), whereas PFOA is listed under Annex A (Elimination) (measures must be taken to eliminate production and use) (UNEP, 2023).

Various regions and countries have also specific regulations in place to prevent the use of PFAS such as PFOA and PFOS (OECD, 2023b). More detailed information on the actions being developed and on regulations in place in each country or region can be found in the supplementary material (Annex 2, Actions and regulations for the elimination of PFAS worldwide, online only, available from: <u>https://publications.</u> <u>iarc.who.int/636</u>).

Country	PFOA (mg/m ³)		APFO (mg/m ³)		PFOS and its salts (mg/m ³)	
	8-hour	Short-term	8-hour	Short-term	8-hour	Short-term
ACGIH			0.01 ^b			
Australia			0.1			
Belgium			0.01 ^f			
Canada – Ontario			0.01			
Canada – Quebec			0.01 ^b			
Denmark			0.01 ^b	0.02 ^{b,c}		
Germany (AGS)					0.01 ^{a,b}	0.08 ^{a,b,c}
Germany (DFG)	0.005 ^{a,b}	0.04 ^{a,b,c}			0.01 ^{a,b}	0.08 ^{a,b,c}
Ireland			0.01			
Japan (JSOH)	0.005 ^d					
New Zealand			0.1			
Singapore			0.01			
Spain			0.01 ^b			
Sweden					900	1400 ^c
Switzerland	0.005°	0.04 ^e			0.01 ^e	0.08 ^e

Table 1.19 Occupational exposure thresholds for PFOA, APFO, and PFOS, by country

ACGIH, American Conference of Governmental Industrial Hygienists; AGS, Ausschuss für Gefahrstoffe (Hazardous Substances Committee); APFO, ammonium perfluorooctanoate; DFG, Deutsche Forschungsgemeinschaft (German Research Foundation); JSOH, Japan Society for Occupational Health; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid.

^a Inhalable fraction.

^b Skin.

° 15-minute average.

^d Not applicable to women of child-bearing potential.

^e Inhalable aerosol.

^f Skin, mucous membranes, and eyes.

From <u>IFA (2022)</u>.

1.5.1 Occupational limits

Occupational exposure limits for air concentrations are available for PFOA, APFO, and PFOS (Table 1.19). Germany and Switzerland have identical 8-hour TWAs and short-term exposure limits (STELs): TWA for inhalable PFOA, 0.005 mg/m^3 (STEL, 0.04 mg/m^3) and TWA for inhalable PFOS, 0.01 mg/m³ (STEL, 0.08 mg/m³). Japan also uses the value of 0.05 mg/m³ as the 8-hour TWA for PFAS; however, Sweden's 8-hour TWA for PFOS is 900 mg/m³ (and the STEL is 1400 mg/m³). APFO is assigned an 8-hour TWA of 0.01 mg/m³ in Belgium, Canada (Ontario and Quebec), Denmark, Ireland, Singapore, and Spain, (IFA, 2022). The same value is adopted by the American Conference of Governmental Industrial Hygienists (ACGIH) (ACGIH, 2023).

Australia and New Zealand set their 8-hour TWA at 0.1 mg/m³, and Denmark has a STEL of 0.02 mg/m³. Belgium, Denmark, Germany, Quebec, Spain, and the ACGIH all assign a skin notation to their guidance, indicating that dermal protection is needed to prevent skin absorption (IFA, 2022).

1.5.2 Consumer products and food

See <u>Table 1.20</u>.

Numerous countries have set recommended limits for exposure to PFAS in consumer products and food. Food Standards Australia New Zealand (FSANZ) and the National Health and Medical Research Council (NHMRC) of Australia set a tolerable daily intake (TDI) for PFOA of 160 ng/kg bw and a combined intake

Consumer product	Country or region	Reference
Food packaging	USA, European Union, Japan	<u>US FDA (2022b)</u> OECD (2023a)
Children's products	Some states in the USA	<u>ITRC (2023a)</u> ITRC (2023b)
Carpets, textiles, rugs, and fabric treatments, furniture	European Union, some states in the USA	<u>ITRC (2023a)</u> <u>ITRC (2023b)</u> <u>MNPCA (2023); Maine DEP (2023)</u>
Cookware	Some states in the USA	<u>MNPCA (2023)</u>
Cosmetics and other personal products	Some states in the USA	<u>MNPCA (2023)</u>
Firefighting foams	Canada, European Union, Australia, some states in the USA	ECHA (2023) ECCC (2017) ITRC (2023b)

Table 1.20 Examples of consumer products in which the presence or use of PFOA and PFOS is restricted

PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; USA, United States of America.

of PFOS and PFHxS of 20 ng/kg bw (Australian Government, 2017). In the USA, the Agency for Toxic Substances and Disease Registry (ATSDR) developed intermediate-duration oral minimal risk levels of 3 ng/kg per day for PFOA and 2 ng/kg per day for PFOS. The ATSDR minimal risk levels are estimates of the daily intake below which harm to human health is not anticipated to occur and are often used as screening levels for environmental media (e.g. water) (ATSDR, 2021).

In 2008, the European Food Safety Authority's Panel on Contaminants in the Food Chain (CONTAM) established TDIs of 150 ng/kg bw per day for PFOS and 1500 ng/kg bw per day for PFOA. In 2020, the same agency established a new safety threshold for PFOA, PFOS, and two other PFAS (PFHxS and PFNA), a group tolerable weekly intake (TWI) of 4.4 ng/kg bw per week.

1.5.3 Environmental guidelines

National and local jurisdictions have established regulations and guidelines on acceptable concentrations of PFOA and PFOS in drinking-water and other environmental compartments. [The Working Group noted that these guidelines are evolving on the basis of current science and regulatory processes.] Many of these regulations, particularly those pertaining to drinking-water, have been updated in recent years and are closely tracked by organizations such as the Interstate Technology and Regulatory Council (ITRC). More information on water and soil regulations is available online in tables that are maintained by the ITRC (<u>ITRC, 2023c</u>).

<u>Table 1.21</u> presents a non-exhaustive list of some regulations for different environmental compartments.

Canada additionally has guidelines for PFOS in surface water, aquatic life, fish tissue, and wildlife diet, as part of the Federal Environmental Quality Guidelines (ECCC, 2023). In the USA, several states have implemented PFAS limits for a variety of environmental media. PFOA and PFOS in drinking-water, surface water, groundwater, and sediment or soil (residential, industrial or commercial, and construction site) are regulated in various combinations in up to 20 states. A few states additionally have testing requirements or allowable concentrations for PFOA and PFOS in biosolids and wastewater. Consumption advisories or limits on concentrations of PFOS and, to a lesser extent, PFOA in fish as well as in shellfish, deer, turkey, beef, and milk exist in numerous states for different consumption patterns and

Environmental compartment	Country or region	Limit established; year	Reference
Drinking-water	New Zealand	PFOA, 560 ng/L; PFOS, 70 ng/L; 2017	Australian Government (2017)
Drinking-water	Canada	PFOA, 200 ng/L; PFOS, 600 ng/L; 2018	<u>Health Canada (2018a, b)</u>
Drinking-water	European Union	500 ng/L for total PFAS; 100 ng/L for the sum of 20 PFAS, including PFOA and PFOS; 2020	<u>EU (2020)</u>
Drinking-water	Denmark	2 ng/L for the total of PFOA, PFOS, PFNA, and PFHxS; 2021	<u>Danish Environmental</u> <u>Protection Agency (2023)</u>
Drinking-water	UK	10–100 ng/L for PFOS or PFOA; 2021	<u>DWI (2021)</u>
Drinking-water	USA	PFOA, 0.004 ng/L; PFOS, 0.02 ng/L; 2022 (Interim Health Advisory)	Office of the Federal Register (2022)
Recreational water	New Zealand	PFOA, 10 000 ng/L; PFOS, 2000 ng/L; 2017	<u>Australian Government</u> (2017)
Ambient water	Canada (British Columbia)	PFOA, 200 ng/L; PFOS, 600 ng/L; 2020	<u>BC MECCS (2020)</u>
Groundwater	European Union	4.4 ng/L (sum of 24 PFAS, including PFOA and PFOS); 2022	European Commission (2022)
Soil	Canada	PFOS, 0.01 mg/kg dry weight; 2021 PFOA soil screening values are 0.70, 1.05, and 9.94 mg/kg soil, for agricultural/residential, commercial, and industrial land use; 2019	<u>CCME (2021)</u> <u>Health Canada (2019)</u>
Ambient air	USA	PFOS, PFOA, and APFO concentrations ranging from 0.006 to 0.082 μ g/m ³ , 0.007 to 0.07 μ g/m ³ , and 0.024 to 0.05 μ g/m ³ , respectively; varies by regulation	<u>ITRC (2023a, b)</u>

Table 1.21 Examples of guidelines in place for environmental compartments

APFO, ammonium perfluorooctanoate; PFAS, per- and polyfluoroalkyl substances; PFNA, perfluorononanoic acid; PFOA, perfluorooctanoic acid; PFHxS, perfluorohexanesulfonic acid; PFOS, perfluorooctanesulfonic acid; UK, United Kingdom; USA, United States of America.

populations (<u>ECOS, 2023</u>; <u>ITRC, 2023a</u>, <u>b</u>). [The Working Group noted that this is a dynamic area and new regulations with lower proposed regulatory thresholds are being established.]

1.5.4 Guidance and biomonitoring reference values

Health-based threshold guidance values for biomonitoring are available in Germany and the USA. In 2016, the Human Biomonitoring Commission (HBM Commission) of the German Environment Agency (UBA) established concentrations below which no adverse health effects are expected to occur (HBM-I values), according to current knowledge and assessment, of 2 ng/mL for PFOA and 5 ng/mL for PFOS in blood serum or plasma (<u>Hölzer</u> et al., 2021). In 2019, concentrations above which there is an increased risk of adverse health effects (HBM-II values) were established of 5 ng/mL and 10 ng/mL for PFOA and PFOS, respectively, in blood plasma in women of childbearing age, and of 10 ng/mL and 20 ng/mL for PFOA and PFOS, respectively, in blood plasma of all other populations (Schümann et al., 2021). In its report of 2022, the US National Academies of Sciences, Engineering, and Medicine (NASEM) identified two threshold values for the sum of seven PFAS in serum or plasma, including PFOA and PFOS, to guide clinical care and exposure reduction efforts: 2 ng/mL and 20 ng/mL. NASEM recommended that clinicians provide the usual standard of care at concentrations of < 2 ng/mL; encourage exposure reduction and screen for certain medical conditions at concentrations of 2 to < 20 ng/mL; and encourage exposure reduction and screen for additional medical conditions at concentrations of \geq 20 ng/mL (NASEM, 2022).

1.6 Quality of exposure assessment in key epidemiological studies of cancer and mechanistic studies in humans

1.6.1 Quality of exposure assessment in key cancer epidemiology studies

(a) Exposure assessment methods

The exposure assessment methods employed in 12 case–control studies and 30 cohort studies, including 18 nested case–control studies, were reviewed and are described below by study design. Details on each of the studies are summarized in Table S1.22 (Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: <u>https://publications.iarc.</u> who.int/636).

These studies employed primarily one of two methods of exposure assessment for PFOA and PFOS: biological measurement of PFOA and PFOS in the blood (whole blood, serum, or plasma) or job (residential history) exposure matrices to estimate historical exposures. The biological matrix for the analysis of PFOA and PFOS was blood in two studies, plasma in five studies, and serum in the other studies.

An overview of chemical analysis methods used for detection and quantification of PFOA and PFOS in human biological samples is presented in Section 1.3.4. In the epidemiological cancer studies in which the exposure assessment was based on biomonitoring, targeted analytical methods were applied in all except four studies that used non-targeted methods, which do not permit quantification of concentrations but rely on semiquantitative determination of intensity level for identified PFAS (Chang et al., 2023; <u>Chen et al., 2023; van Gerwen et al., 2023; Zhang et al., 2023</u>).

As described in Section 1.1, several isomers exist for both PFOA and PFOS. In a few epidemiological cancer studies, isomer-specific determinations were performed that were summed for analysis (Itoh et al., 2021; Li et al., 2022a; Purdue et al., 2023; Rhee et al., 2023b; Winquist et al., 2023), but in most studies only one concentration was reported for PFOA and one for PFOS. The exception was the study by van Gerwen et al. (2023) who considered linear and branchedchain PFOS separately in their non-targeted analysis. [The Working Group noted that when one concentration value was reported for PFOA or PFOS, it was assumed that this represented the sum of branched and linear isomers, even though this was not always specified in the study.] In the studies in which non-targeted methods were used (see Section 1.3.4), compound-specific intensities, not concentrations, were reported (Chang et al., 2023; Chen et al., 2023; van Gerwen et al., 2023; Zhang et al., 2023).

(i) Case–control studies

In total, 12 relevant case–control studies were reviewed for the present monograph (Bonefeld-Jørgensen et al., 2011; Vieira et al., 2013; Hardell et al., 2014; Wielsøe et al., 2017; Lin et al., 2020b; Tsai et al., 2020; Itoh et al., 2021; Cao et al., 2022; Li et al., 2022a; Liu et al., 2022a; Velarde et al., 2022; Chen et al., 2023). In all studies except that by Vieira et al. (2013), both PFOA and PFOS were evaluated, and the exposure assessment was based on biomonitoring in the blood (serum, plasma, or whole blood).

<u>Vieira et al. (2013)</u> evaluated incident cancers in residents (according to address at time of cancer diagnosis) in six PFOA-contaminated water districts and 13 counties in Ohio and West Virginia, USA. PFOA concentrations in water, available for each of the six districts, varied by community. Water district information was available for all individuals, and logistic regression analyses compared individuals in contaminated water districts with those in neighbouring water districts. For residents of Ohio, where approximately one third of the sample population lived, residential addresses were geocoded and then PFOA serum concentrations were assigned on the basis of modelled estimates (Shin et al., 2011a, b), assuming 10 years residence at that address. Exposure was then divided into four categories. However, analysis for residents of West Virginia was limited to residence by water district.

For all other studies based on general populations, blood samples were collected from participants during the same time periods for the cases and controls. For the cases in studies by Bonefeld-Jørgensen et al. (2011) and Cao et al. (2022), the timing of the blood draw relative to when treatment started was not reported, and in the study by Lin et al. (2020b), blood samples were collected 1 week after the identification of the case by pathology. In the study by <u>Chen et al.</u> (2023), blood spot samples were collected at birth, and diagnosis occurred on average 9.3 months after birth for unilateral retinoblastoma and 22 months after birth for bilateral retinoblastoma. For the remaining studies (Hardell et al., 2014; Wielsøe et al., 2017; Tsai et al., 2020; Itoh et al., 2021; Li et al., 2022a; Velarde et al., 2022), blood samples were collected between the time of diagnosis and the start of treatment. Controls were selected from participants in ongoing cross-sectional studies (Bonefeld-Jørgensen et al., 2011; Wielsøe et al., 2017); invited on the basis of selection from population registries (Hardell et al., 2014) or breast cancer screening programmes (Cao et al., 2022; Li et al., <u>2022a</u>); in connection with medical check-ups (Itoh et al., 2021); through advertisements at the hospital and in the community (Tsai et al., 2020; Liu et al., 2022a); or invited after hospitalization due to other diagnoses or illnesses (Lin et al., 2020b). In studies by Bonefeld-Jørgensen et al. (2011), Hardell et al. (2014), Wielsøe et al. (2017), Itoh et al. (2021) and Chen et al. (2023), cases

and controls were matched on age and region of residence, whereas <u>Lin et al. (2020b)</u> matched cases and controls on age and sex. [The Working Group noted that, given the temporal trends in PFOA and PFOS blood levels, it is important that time of blood sample collection is matched or adjusted for.]

In all studies, targeted chemical analyses were performed using LC-MS/MS, except in the study by Chen et al. (2023), in which non-targeted methods were used. Because PFOA and PFOS levels were not quantified using standard targeted methods by Chen et al. (2023), direct comparisons with the levels from other studies were not possible. Li et al. (2022a) performed separate determinations for eight PFOA isomers and nine PFOS isomers, and internal standards of linear PFOA and PFOS isomers were used. In the study by <u>Hardell et al. (2014)</u>, only linear isomers of PFOA and PFOS were determined. In the remaining studies, one concentration for PFOA and one for PFOS were reported. In these studies, it was not stated whether only linear isomers were considered or whether other isomers were also included in the reported concentrations.

In addition to PFOA and PFOS, all studies except that by <u>Chen et al. (2023)</u> included at least four of the other most prominent PFAS in human blood (<u>EFSA Panel on Contaminants in the Food</u> <u>Chain, 2020</u>). <u>Bonefeld-Jørgensen et al. (2011), Li</u> <u>et al. (2022a)</u>, and <u>Wielsøe et al. (2017</u>) assessed exposure both for single PFAS and for the sum of several PFAS. In the studies by <u>Hardell et al.</u> (2014), <u>Tsai et al. (2020)</u>, <u>Lin et al. (2020b)</u>, <u>Liu</u> <u>et al. (2022a)</u>, <u>Cao et al. (2022)</u>, and <u>Chen et al.</u> (2023), only single PFAS were assessed.

Bonefeld-Jørgensen et al. (2011) and Wielsøe et al. (2017) measured other carcinogens, i.e. polychlorinated biphenyls (PCBs), β -hexachlorocyclohexane, cadmium and cotinine (as a biomarker for tobacco smoke), via biomonitoring. Some studies collected information on exposure to other carcinogens, i.e. barbecuing, hair dyeing, smoking, alcohol consumption, use of estrogen or estrogen-replacement therapy, meat consumption, via questionnaires (see Table S1.22, Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: <u>https://publications.iarc.</u><u>who.int/636</u>).

(ii) Cohort studies

This section includes cohort studies designed to study PFOA and PFOS exposure in occupational settings and contaminated communities and case-control studies nested in other general population cohort studies. Eleven cohort studies focusing on cancer incidence or mortality were reviewed by the Working Group. These included eight occupational cohort analyses (Alexander et al., 2003; Alexander and Olsen, 2007; Lundin et al., 2009; Steenland and Woskie, 2012; Consonni et al., 2013; Raleigh et al., 2014; Steenland et al., 2015; Girardi and Merler, 2019), two cohort studies in highly exposed communities (Barry et al., 2013; Li et al., 2022b), and one in the general population in the USA (Wen et al., 2022). Three of the cohort analyses were conducted in the C8 study area (a fluorochemical-production plant in Parkersburg, West Virginia, USA, and the six water districts in Ohio and West Virginia in which water was contaminated by a chemical plant that used APFO in the production of PTFE) and focused on either occupational exposure to PFOA or exposure through residential consumption of drinkingwater (Steenland and Woskie, 2012; Barry et al., 2013; Steenland et al., 2015). Additionally, four studies were conducted among fluorochemical-manufacturingworkers at a PFOS-production site in Alabama, USA (Alexander and Olsen, 2007) and at an APFO-manufacturing site in Minnesota, USA (Alexander et al., 2003; Lundin et al., 2009; Raleigh et al., 2014). The study by Li et al. (2022b) was based in a general population that was highly exposed to PFAS, but an ecological approach using water districts was followed, rather than measurement of subject-specific

PFAS exposure. The majority of these studies that evaluated specific PFAS focused on PFOA. In the general population study (<u>Wen et al., 2022</u>) and the community exposure study (<u>Li et al., 2022b</u>), exposure to both PFOA and PFOS was evaluated.

Occupational cohort studies

Six of the occupational cohort analyses focused on PFOA (<u>Lundin et al., 2009; Steenland</u> and Woskie, 2012; <u>Consonni et al., 2013; Raleigh</u> <u>et al., 2014; Steenland et al., 2015; Girardi and</u> <u>Merler, 2019</u>) and two on PFOS (<u>Alexander et al.,</u> <u>2003; Alexander and Olsen, 2007</u>).

Another cohort analysis on fluoropolymer production was a mortality analysis that did not include estimates for PFOA exposure and is not discussed further in the present monograph (Leonard et al., 2008).

All of these occupational cohort analyses relied on job history to classify potential exposure to PFOA or PFOS. Occupational exposure to PFOS was evaluated in workers in a film and chemical plant in Alabama, USA, (Alexander et al., 2003; Alexander and Olsen, 2007) using an exposure matrix developed by <u>Olsen et al. (2003)</u> that classified workers into three categories on the basis of potential exposure to POSF (a precursor of PFOS): ever high; ever low/never high; or no exposure. No measure of cumulative exposure was included. Serum samples were analysed but were not used to develop an exposure matrix. A variety of perfluorinated amides, alcohols, acrylates, and other fluorochemical polymers were produced at the plant (e.g. PFOA was used as a by-product or emulsifier until 1988) but were not included in the exposure assessment (Olsen et al., 2003).

For PFOA, several different approaches were used in the occupational cohort studies. Two studies used a JEM created using expert opinion (Lundin et al., 2009; Consonni et al., 2013); one used air sampling measurements together with a JEM (Raleigh et al., 2014); and others used biomarkers to enhance JEMs (Steenland and

Woskie, 2012; Steenland et al., 2015; Girardi and Merler, 2019). In the study on workers at the fluorochemical-production plant in Parkersburg, West Virginia (Steenland and Woskie, 2012; Steenland et al., 2015), exposure to PFOA was assessed using a JEM and then linked to serum exposure levels in samples collected between 1979 and 2004 from workers in eight work categories (Woskie et al., 2012). Cumulative exposure was estimated in ppm-years (µg/mL serumyears). Girardi and Merler (2019) used a similar approach to estimate cumulative serum PFOA exposure among workers at a factory in Veneto, Italy. Although PFOS was also produced, at lower volumes, in this factory (an average of [33 tonnes/ year] compared with [227 tonnes/year] of PFOA), exposure to PFOS was not estimated. For APFOmanufacturing workers at the Minnesota factory (Lundin et al., 2009), exposure to PFOA was estimated according to three categories: definite; probable; and no occupational exposure, based on job history. Cumulative PFOA exposure was then estimated using weights based on serum levels of workers in different areas of the manufacturing facility; lifetime exposure was estimated based on the product of the weight and the exposure days. For manufacturing workers using APFO (the ammonium salt of PFOA), air samples were collected for combinations of department/ job title/work area/equipment/task (Raleigh et al., 2014). Job histories were then linked to the air samples to create a TWA of APFO exposure $(\mu g/m^3$ -years), and then all jobs were summed to create an overall summary APFO air-exposure variable.

<u>Consonni et al. (2013)</u> evaluated mortality among workers at a plant involved in TFE synthesis and polymerization. The TFE synthesis and polymerization process uses APFO (the ammonium salt of PFOA) and, as a result, workers were commonly co-exposed to both TFE and APFO (88%, in the study by <u>Consonni et al.</u>, <u>2013</u>). A semiquantitative JEM using arbitrary units was created, and cumulative exposure was estimated.

Studies of communities with contaminated drinking-water

Cancer risk associated with the consumption of PFAS-contaminated drinking-water was evaluated in three communities: the C8 Study in Ohio and West Virginia, in the USA; Ronneby, Sweden; and the Veneto region, in Italy.

The C8 study focused on water districts where drinking-water was contaminated by PFOA, also known as "C8", from a fluorochemical-production plant (<u>Barry et al., 2013</u>). One cohort study of cancer incidence was conducted in this region. This study, which included both residents and workers, used the exposure assessment metric from the study by <u>Shin et al. (2011a, b)</u> to assign cumulative PFOA exposure to individuals on the basis of residential history, and the exposure metric from <u>Woskie et al. (2012</u>) to assign PFOA exposure related to occupational exposure. Exposure was modelled based on a continuous measure of cumulative PFOA exposure as well as categories of exposure.

The study in Ronneby, Sweden, by Li et al. (2022b) relied on residential history to assign water source into categories: ever high, never high, early high, late high, short high or long high PFAS exposure. Differences in exposure between these categories were supported by measurement of PFAS blood levels in the population, with the highest levels found in the late high group (Li et al., 2022b). Water from this region was contaminated with multiple PFAS, and exposures were particularly high for PFAS related to firefighting foam (PFOS and PFHxS). Exposure assessment in this analysis was not chemical-specific and used residence as a surrogate for exposure. [The Working Group recognized that it was not possible to distinguish PFOS from PFHxS because of the elevated levels of both compounds and the presence of somewhat elevated PFOA levels that correlated with

levels of PFOS and PFHxS, even though PFOA levels were much lower than those of PFOS and PFHxS.]

Another location in the world where there is extensive contamination of water with PFAS is the region of Veneto, Italy, where a factory produced PFOA between 1968 and 2014 (Girardi and Merler, 2019). [No publication has comprehensively described the exposure experience in this community. The Working Group reviewed several papers and reports to characterize PFOA and PFOS exposure in this community and included details in Annex 2, Actions and regulations for the elimination of PFAS worldwide, online only, available from: https://publications. iarc.who.int/636.] Drinking-water contamination was discovered in 2013, and since that time extensive environmental and human biological sampling has been conducted (Ingelido et al., 2018; Pitter et al., 2020; Giglioli et al., 2023). Initially, the highly contaminated area, also known as the "red area", was composed of 21 municipalities, with 126 000 inhabitants. In 2018, nine additional municipalities were added, some of which were only partially supplied by the contaminated waterworks; the updated red area has a size of 595 km² and a total population of approximately 140 000.

General population cohorts including nested case–control studies

Wen et al. (2022) used NHANES exposure data from 1999 to 2014 to evaluate cancer mortality in adults in a general population sample in the USA. The NHANES is a nationally representative sampling of the population, designed to assess the health and nutritional status of adults and children in the USA. This evaluation used serum measurements of PFOA and PFOS, and other PFAS; only one serum measurement was available for each individual. Deaths were identified through linkage to the National Death Index, with a median follow-up of 81 months (range, 46–112 months). Cancer mortality risk was estimated using tertiles of exposure for PFOA and PFOS, but the majority of the analysis focused on the PFAS mixture.

In total there were 18 case-control studies nested within cohorts that used biomonitoring of PFAS in their analyses (Eriksen et al., 2009; Bonefeld-Jørgensen et al., 2014; Ghisari et al., 2017; Hurley et al., 2018; Cohn et al., 2020; Mancini et al., 2020; Shearer et al., 2021; Feng et al., 2022; Frenoy et al., 2022; Goodrich et al., 2022; Chang et al., 2023; Purdue et al., 2023; Rhee et al., 2023a, b; van Gerwen et al., 2023; Winquist et al., 2023; Zhang et al., 2023; Madrigal et al., 2024). Two studies used the E3N (Etude épidémiologique auprès de femmes de la Mutuelle générale de l'Education nationale) prospective cohort of women in the national education system in France (Mancini et al., 2020; Frenoy et al., 2022). Two studies used the Danish National Birth Cohort (Bonefeld-Jørgensen et al., 2014; Ghisari et al., 2017). One study used a cohort of retired Chinese motor-company employees (Feng et al., 2022); another was nested in a cohort of US Air Force Servicemen (Purdue et al., 2023). The others included a cohort of California teachers (Hurley et al., 2018), a Child Health and Development Studies pregnancy cohort in California (Cohn et al., 2020), a population-based national maternity cohort in Finland (Madrigal et al., 2024), and the Alpha-Tocopherol, Beta-Carotene Cancer Prevention (ATBC) Study in Finland (Zhang et al., 2023), and the Mount Sinai BioMe medical record-linked biobank in the USA (van Gerwen et al., 2023). Four studies used the US-based Prostate, Lung, Colorectal, and Ovarian Cancer (PLCO) Cancer Screening Trial cohort (Shearer et al., 2021; Chang et al., 2023; Rhee et al., 2023a; Zhang et al., 2023), and two used the Californiaand Hawaii-based Multiethnic Cohort in the USA (Goodrich et al., 2022; Rhee et al., 2023b).

<u>Chang et al. (2023)</u>, <u>van Gerwen et al.</u> (2023) and <u>Zhang et al. (2023)</u> used non-targeted analysis. In these studies, the analysis was conducted on quantiles of intensity measures of the relative levels of PFOA and PFOS found in pre-diagnosis serum samples. Because PFOA and PFOS levels were not quantified using standard targeted methods, direct comparisons with the PFAS levels reported in other studies were not possible. However, the authors (<u>Changet al., 2023</u>; <u>Zhang et al., 2023</u>) reported strong correlations of between 0.76 and 0.77 between the untargeted analysis and the standard targeted analysis.

In all studies except that by Hurley et al. (2018), blood samples were collected before case ascertainment. Hurley et al. collected samples between 9 months and 8.5 years (average, 35 months) after case diagnosis. In the other studies, the time between sample collection and case ascertainment varied (where this was reported). Zhang et al. (2023) reported that the time between sample collection and cancer diagnosis was 0-18 years (median, 9 years) for the PLCO subcohort analysed. For the study by van Gerwen et al. (2023), sample collection took place 0-1 year before diagnosis (average, 0.08 years) for 65% of cases and an average of 4 years before diagnosis for the remaining 35% of cases. For the study by Goodrich et al. (2022), the median time span between collection of blood sample and diagnosis was 7.2 years (range, 0.9–16.4 years). In the study by Ghisari et al. (2017), cases were diagnosed 11–12 years after initial blood draw, while for Eriksen et al. (2009) cases were diagnosed a median of 7 years after enrolment (and blood draw) (range, 0-12 years). Shearer et al. (2021) reported a mean of 8.8 years (range, 2-18 years) between blood draw and diagnosis, and Chang et al. (2023) reported a median of 5.6 years (range, 2-18 years) between diagnosis and blood draw. In the study by <u>Purdue et al. (2023)</u>, the median time between blood collection and diagnosis was 5 years (range, 0–19.8 years). Rhee et al. (2023a) reported a median time between blood collection and diagnosis of 9 years (interquartile range, 5-13 years). Madrigal et al. (2024) reported that cases were diagnosed at least 3 years after delivery

(samples were collected during the first trimester of pregnancy).

Both Purdue et al. (2023) and Rhee et al. (2023a) had access to multiple blood samples, which allowed them to evaluate how the rank ordering of exposure might change over time. For a subset of participants in the study by <u>Purdue</u> et al. (2023), a second blood sample was collected. To explore differences related to the time of collection, Purdue et al. (2023) analysed the data separately for participants with two samples and also created a combined variable based on the classification of the median level at each time point. They reported an overall Spearman coefficient of 0.6 for both PFOA and PFOS in repeat samples and an intraclass correlation coefficient (ICC) of 0.5–0.6, with stronger correlation for repeat samples taken after < 4.7 years and weaker correlation for repeat samples taken after > 4.7 years. Rhee et al. (2023a) analysed blood from 60 controls at enrolment, 1 year after enrolment, and 5 years after enrolment to assess long-term intra-individual variability in PFAS concentration. The ICC for three measures was 0.73 for of PFOA and 0.85 for PFOS; these values suggest that measurements of PFOA and PFOS were reliable over time in this study.

Eight of the studies focused on breast cancer (Bonefeld-Jørgensen et al., 2014; Ghisari et al., 2017; Hurley et al., 2018; Cohn et al., 2020; Mancini et al., 2020; Feng et al., 2022; Frenoy et al., 2022; Chang et al., 2023). Some of the studies were in birth cohorts for which blood samples were collected during pregnancy and maternal breast cancers identified subsequently (Bonefeld-Jørgensen et al., 2014; Ghisari et al., 2017). Alternatively, Cohn et al. (2020) used maternal blood collected 1-3 days postpartum to investigate breast cancer in the daughters. Chang et al. (2023), Feng et al. (2022), Frenoy et al. (2022), Mancini et al. (2020), and Hurley et al. (2018) reported on prospective studies of adult general populations. Three studies focused on people in professions related to education

(Hurley et al., 2018; Mancini et al., 2020; Frenoy et al., 2022), and one study was in a cohort in an industrial motor company (Feng et al., 2022), although occupational exposures were not its focus.

Several nested case-control studies were part of general cancer screening or prevention trials, such as the PLCO cohort (<u>Chang et al., 2023</u>; <u>Rhee et al., 2023a</u>; <u>Zhang et al., 2023</u>), the Cancer Prevention Study II Lifelink Cohort (<u>Winquist et al., 2023</u>), or the ATBC Study (<u>Zhang et al.,</u> <u>2023</u>). One study used a hospital-based biobank in the USA (<u>van Gerwen et al., 2023</u>), and another was a population-based national maternity cohort in Finland (<u>Madrigal et al., 2024</u>).

Many studies measured multiple PFAS in their samples; however, much of the outcome analysis focused on potential associations with a limited number of individual PFAS (Eriksen et al., 2009; Ghisari et al., 2017; Hurley et al., 2018; Cohn et al., 2020; Mancini et al., 2020; Shearer et al., 2021; Goodrich et al., 2022). Some authors attempted to sum a variable number of the measured PFOA or PFOS isomers, and use these summed metrics in their analysis (Bonefeld-Jørgensen et al., 2014; Feng et al., 2022), and Frenoy et al. (2022) used principal components analysis and Bayesian kernel machine regression on all the PFAS measurements, van Gerwen et al. (2023) used untargeted analysis to examine intensities of eight detectable PFAS, including linear PFOA and branched and linear PFOS, which were examined individually in their analysis.

(b) Critical review of exposure assessment in key epidemiological studies

Blood is considered a suitable matrix for exposure assessment (Vorkamp et al., 2021), and measured blood concentrations are an objective measure of exposure. In most studies in which blood measurements were used, the analytical methods used were state-of-the-art in 2023, the LOQs for PFOA and PFOS were sufficiently low to ensure high quantification frequencies, and the measurement error in the targeted chemical analyses was low (see Section 1.3.4). In some studies, the quantification method used was non-targeted and thus semiquantitative; therefore, exact concentrations were not available. However, ranking of levels is possible. Several occupational cohort studies that estimated cumulative exposures used older, less specific or precise methods, with higher LODs (Alexander et al., 2003; Steenland and Woskie, 2012). In some studies, estimation of serum levels combined state-of-the-art measurements of community exposures with older data from occupational cohort studies (Barry et al., 2013; Steenland et al., 2015).

The measured concentrations in blood represent combined exposure through all exposure pathways (see Section 1.4.3 on biomonitoring). Since PFOA and PFOS have long elimination half-lives (see Section 4.1), and repeated measures in humans show strong ICCs (Blake et al., 2018; Rhee et al., 2023a), the measured concentrations represent exposure over a relatively long period of time. These factors limit the potential for non-differential exposure misclassification, in general. Using repeated measures data from Rhee et al. (2023a) and Purdue et al. (2023), the Working Group evaluated the potential for exposure misclassification and resulting bias if just one biological sample is used; the results of this analysis demonstrated that using a single sample represented rather well the mean of repeated samples collected a median of 4-5 years apart in two cohort studies of populations with background levels (Spearman correlations of 0.87 and 0.83 for the PLCO and US Air Force Servicemen cohorts, respectively) (see Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications.iarc.who.int/636). Repeat biomonitoring of PFOA and PFOS in the general population is described in Section 1.4.3.

It is important to be careful when comparing measured concentrations reported in the

various studies, since PFOA and PFOS isomers have been treated differently among studies. This is of particular importance for PFOS since branched isomers may comprise up to 50% of the total concentration of PFOS (Haug et al., 2009); whether or not the branched isomers are included could make a significant difference to participant exposure levels. Results from studies using untargeted methods also present limitations when comparing exposure concentrations with results from other studies.

Most studies relied on a single blood sample to classify lifetime exposure to PFOA and PFOS. In case–control studies, one blood sample was collected near the time of diagnosis. In the cohort and nested case–control studies, the time between blood collection and diagnosis ranged from 0 to 20 years, as described above. Thus, there is a possibility that measured blood levels of PFOA and PFOS do not reflect exposure at crucial windows in cancer development. However, as described above, the results of studies of repeated human serum measurements of PFOA and PFOS have shown strong correlations over time.

(i) Case-control studies

In the study by <u>Vieira et al. (2013)</u>, exposure was assigned on the basis of address at the time of cancer diagnosis; this could result in exposure misclassification if individuals changed addresses before cancer diagnosis. However, the authors stated that the median residence time at current address was 17 years, suggesting that this issue was unlikely to be a source of exposure misclassification.

All the other case-control studies used biomonitoring for exposure assessment, and thus generally had the same strengths and limitations. While blood samples provide specific measures of PFOA and PFOS exposure, biological samples are influenced by interindividual variability. For the case-control studies, the fact that blood samples were collected at or near the time of diagnosis means that these biological markers may be influenced by the disease process. If cancer were to alter the absorption, distribution, metabolism, or excretion (ADME) of PFOA and PFOS, then the measured levels in the cases could not be compared with measured levels in the controls, thus resulting in differential exposure misclassification.

A limitation of these studies is that most did not measure other carcinogens in the blood samples, and that only limited information on exposure to other carcinogens was available from the questionnaires. In the studies by Bonefeld-Jørgensen et al. (2011) and Wielsøe et al. (2017), other substances classified by IARC in Group 1, carcinogenic to humans (PCBs, β-hexachlorocyclohexane, cadmium, and cotinine as a biomarker of tobacco smoking), were measured. Bonefeld-Jørgensen et al. (2011) reported high correlations between PFAS and other persistent organic pollutants (r = 0.42 - 0.55; P < 0.05), although no information on specific compounds was reported. A strength of the exposure assessment in this study was that correlations with biomarkers of co-exposures were assessed.

In summary, for all case-control studies (except Vieira et al., 2013), blood levels were measured and used as the exposure metric. A main strength was that the measured levels represent combined exposure through all exposure pathways. Measurement error was also thought to be low in all studies in which targeted analyses were performed, whereas the untargeted methods applied in other studies might have lower precision. A major weakness of all the case-control studies was that the blood samples for the cases were collected after the participants had been diagnosed. Thus, the measured levels may not reflect exposure at crucial windows in cancer development, and if cancer alters the ADME of PFAS, there could be differential exposure misclassification.

(ii) Cohort studies

In the majority of studies with occupational exposure and in communities with high exposure, PFOA and PFOS exposure was determined by exposure reconstruction, based either on occupational or residential history. Most studies used exposure reconstruction techniques that provided cumulative exposure estimates to rankorder individuals according to PFOA and PFOS exposure. These cumulative exposure estimates allowed for exposure-response analysis, which may strengthen the argument for causality. In several studies, cumulative serum-level estimates were developed using retrospective modelling (Steenland and Woskie, 2012; Barry et al., 2013; Steenland et al., 2015; Girardi and Merler, 2019); these studies have added strength because they included both environmental and biological measurements to support their estimates. In one study, cumulative estimates of air levels of APFO were developed that enabled workers to be ranked according to exposure, because the main source of PFOA was expected to be occupational (Raleigh et al., 2014). In other studies, cumulative categorical estimates were developed based on occupational history information (Alexander et al., 2003; Alexander and Olsen, 2007; Lundin et al., 2009; Consonni et al., 2013). One study relied solely on residence to assign a categorical exposure, although serum levels were said to validate the categories (Li et al., 2022b); this study also lacked specificity for individual PFAS, limiting its utility to the evaluation of the carcinogenicity of PFOA or PFOS individually. Many of these studies focused only on PFOA (Lundin et al., 2009; Steenland and Woskie, 2012; Barry et al., 2013; Steenland et al., 2015; Girardi and Merler, 2019); none presented isomer-specific estimates of exposures. In the study by Girardi and Merler (2019), workers may have been exposed to other PFAS, including PFOS, but these exposures were not evaluated. In one occupational cohort (Consonni et al., 2013) focusing

124

on TFE workers, a very high correlation between cumulative weighted categorical exposures to TFE and cumulative weighted categorical exposures to APFO ($\rho = 0.72$) was reported in exposed workers, therefore, it was difficult to ascertain differences between these exposures. Another study focused on categories of POSF-exposed workers, resulting in estimates only of indirect exposure to its metabolite PFOS; however, serum levels of PFOS were used to validate the exposure estimates. Co-exposure to PFOA was likely but was not assessed (Alexander et al., 2003; Alexander and Olsen, 2007).

In all cohorts, exposure was ascertained before cancer diagnosis or cancer death. Because exposure was assigned before diagnosis and all individuals were evaluated in the same way, the potential for differential exposure misclassification was limited for both cohort studies and the resulting nested case–control studies.

All the nested case-control studies and one cohort analysis (Wen et al., 2022) relied on biomarker measurement of PFAS in serum or plasma samples, although van Gerwen et al. (2023) and Zhang et al. (2023) used an untargeted analysis method. As discussed for the case-control studies, blood is an appropriate matrix for biomonitoring of PFOA and PFOS. The use of non-targeted methods does not allow quantification of PFAS concentrations but does provide appropriate rank ordering of individuals. Most studies evaluated PFOA and PFOS separately. Frenoy et al. (2022) primarily used principal components analysis to characterize exposure to both PFOA and PFOS together with other PFAS and brominated flame retardants, which made individual PFOA or PFOS determinations challenging.

All studies except that by <u>Hurley et al. (2018)</u> used blood samples collected before case ascertainment, although the range of time between blood collection and case ascertainment varied widely. PFOA and PFOS have a relatively long half-life in blood, making them good measures

of long-term exposure. However, single sample exposure measurements may not reflect exposure at crucial windows in cancer disease development. All studies, except those by Purdue et al. (2023) and Rhee et al. (2023a), used a single blood sample to determine exposure status. Purdue et al. (2023) collected samples at two points in time and analysed them both separately and as a combined exposure metric; this may reduce exposure misclassification but also reduced the study power since not all participants had two samples. The results of the ICC analysis by Rhee et al. (2023a) suggested that PFOA and PFOS concentrations in blood samples remain relatively constant over time, suggesting that a single measure may correctly classify individuals. A bias analysis of these samples by the Working Group demonstrated little misclassification error when considering samples collected within an interval of 5-8 years (see Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications. iarc.who.int/636).

While all cohort and nested case-control studies accounted for potential co-exposures to some substances classified by IARC as carcinogenic to humans (Group 1), mostly by questionnaire, most studies focused solely on PFAS exposure. Many studies quantified additional PFAS in serum samples and presented risk estimates for individual and total PFAS as well. The most common co-exposures to carcinogens were smoking, alcohol consumption, and use of oral contraceptives, although information on occupation type was also collected by Eriksen et al. (2009) and Fengetal. (2022). Madrigal et al. (2024) also measured PCB congeners, organochlorine pesticides, and polybrominated diphenyl ethers (PBDEs) in serum samples. At present, little is known about the correlation between exposure to PFAS and to other substances classified by IARC as carcinogens.

1.6.2 Quality of exposure assessment in key mechanistic studies in exposed humans

(a) Exposure assessment methods

The exposure assessment methods used in the key mechanistic studies in humans are discussed below according to study design. [The Working Group did not review all mechanistic studies in exposed humans but reviewed a representative sample of studies for each type of study design.]

(i) Cross-sectional studies

The Working Group reviewed the exposure assessment methods used in 18 studies with a cross-sectional design (Knox et al., 2011; Fletcher et al., 2013; Watkins et al., 2014; Lin et al., 2016, 2020c; Lopez-Espinosa et al., 2016; Liu et al., 2018b; Pan et al., 2019; Abraham et al., 2020; Aimuzi et al., 2020; Di Nisio et al., 2020; Kvalem et al., 2020; Clarity et al., 2021; Lopez-Espinosa et al., 2021; Omoike et al., 2021; Cheng et al., 2022; Zhang et al., 2022; Wang et al., 2023). The studies were conducted in the USA and several European and several Asian countries. In all these studies, both PFOA and PFOS were evaluated, and the exposure assessment was based on biomonitoring.

In 16 of these studies, PFOA and PFOS concentrations were measured in the serum or plasma fractions of blood. These matrices are considered suitable for exposure assessment of environmental contaminants, including long-chain PFAS such as PFOA and PFOS (Calafat et al., 2019; Vorkamp et al., 2021; NASEM, 2022) and have been used as the exposure metric in most epidemiological studies of PFAS. PFOA and PFOS concentrations were measured in cord blood in one study (Liu et al., 2018b), in semen (as well as in serum) in the study by Pan et al. (2019), and in the placenta in the study by Wang et al. (2023). Relatively few studies have used semen or placenta for the assessment of exposure to PFAS.

Twelve of the 18 studies were of participants from the general population. In 6 of these 12

studies, the study populations included men who visited a fertility clinic (Pan et al., 2019), patients undergoing surgery for benign diseases or an elective reason (Cheng et al., 2022), children (Lin et al., 2016), and pregnant women (Liu et al., 2018b; Aimuzi et al., 2020; Wang et al., 2023). Another 6 of the 12 studies (Knox et al., 2011; Fletcher et al., 2013; Watkins et al., 2014; Lopez-Espinosa et al., 2016, 2021; Di Nisio et al., 2020) were of populations with elevated exposure to PFOA from contaminated drinking-water. However, exposure to PFOS in these populations was not higher than in the general population.

In all studies, PFAS were measured at the same time point as the assessment of the outcome, and in one study (Watkins et al., 2014), they were also measured 4–5 years before assessment of the outcome, but the two measures of PFAS were averaged to give a single exposure measure. In all the studies, PFOA and PFOS were analysed using LC-MS/MS.

Four studies (Fletcher et al., 2013; Di Nisio et al., 2020; Lin et al., 2020c; Cheng et al., 2022) reported only PFOA and PFOS. All the other studies also reported other PFAS. Although Knox et al. (2011) measured levels of other PFAS, they evaluated potential associations with the outcome only for PFOA and PFOS; Xie et al. (2023) reported 17 PFAS and considered the total concentration of the 17 PFAS that were evaluated.

In two studies (Omoike et al., 2021; Zhang et al., 2022), serum cotinine levels were measured as a biomarker for tobacco smoke, and in 11 studies (Knox et al., 2011; Fletcher et al., 2013; Watkins et al., 2014; Lin et al., 2016, 2020c; Pan et al., 2019; Aimuzi et al., 2020; Di Nisio et al., 2020; Lopez-Espinosa et al., 2021; Wang et al., 2023; Xie et al., 2023) information was obtained about either current or overall exposure to tobacco and/or tobacco smoke via questionnaires. In 10 studies (Knox et al., 2011; Watkins et al., 2014; Lin et al., 2016, 2020c; Pan et al., 2019; Aimuzi et al., 2020; Di Nisio et al., 2020; Di Nisio et al., 2020; Lopez-Espinosa et al., 2020; Lopez-Espinosa et al., 2020; Di Nisio et al., 2020; Lopez-Espinosa et al., 2021; Zhang et al., 2022; Xie et al., 2023), information

on alcohol consumption was obtained using a questionnaire. <u>Watkins et al. (2014)</u> and <u>Lopez-Espinosa et al. (2021)</u> also obtained information on regular use of anti-inflammatory drugs over time through a questionnaire, <u>Cheng et al. (2022)</u> obtained information on use of hypolipidaemic drugs, and <u>Knox et al. (2011)</u> excluded participants who were taking hormonal medications.

In two studies, biomonitoring data were collected for contaminants other than PFAS. Abraham et al. (2020) measured PFAS in stored blood samples that were collected in the late 1990s and had previously been 2,3,7,8-substituted analysed for polychlorinated dibenzodioxins (PCDDs) and polychlorinated dibenzofurans (PCDFs), including 2,3,7,8-tetrachlorodibenzo-p-dioxin (TCDD), which is classified as carcinogenic to humans, Group 1; IARC, 1997); non-dioxinlike-, mono-ortho-, and coplanar PCBs (classified as carcinogenic to humans, Group 1; IARC, 2015); 4,4'-dichlorodiphenyltrichloroethane (DDT) and its metabolites (classified as probably carcinogenic to humans, Group 2A; IARC, 2017); hexachlorobenzene and β-hexachlorocyclohexane (both classified as possibly carcinogenic to humans, Group 2B; IARC, 1987, 2001); lead (classified as probably carcinogenic to humans, Group 2A; IARC, 1979, 2006), cadmium (classified as carcinogenic to humans, Group 1; IARC, 1993, 2012), and mercury. <u>Clarity et al. (2021)</u>, in a study of firefighters, measured urinary levels of four brominated flame retardants and metabolites of six organophosphate flame retardants for which there was potential occupational exposure.

(ii) Prospective birth cohort studies

Exposure assessment methods were reviewed for six mechanistic studies with a prospective birth cohort design (<u>Grandjean et al., 2012;</u> <u>Goudarzi et al., 2017; Miura et al., 2018; Manzano-Salgado et al., 2019; Dalsager et al., 2021; Liu et al., 2022b</u>). These studies were conducted in Denmark, the Faroe Islands (Denmark), Spain, Japan, and the USA. All six studies included mother-child pairs with singleton births from the general population. Both PFOA and PFOS were evaluated, and the exposure assessment was based on blood biomonitoring.

In all six studies, PFOA and PFOS were analysed using LC-MS/MS in maternal blood serum or plasma collected during pregnancy. <u>Grandjean et al. (2012)</u> also measured PFAS in the children at age 5 years. Among the six studies, outcomes were measured in children at time points ranging from birth to age 12 years.

Miura et al. (2018) reported results only for PFOA and PFOS, whereas the other five studies also reported on other PFAS.

In some studies, information was collected on other exposures, including smoking, diet, and other environmental contaminants. Liu et al. (2022b) measured serum cotinine as a biomarker for maternal tobacco smoking. Grandjean et al. (2012), Manzano-Salgado et al. (2019), and <u>Dalsager et al. (2021)</u> collected information on smoking during pregnancy, and Goudarzi et al. (2017) collected information on parental smoking and environmental tobacco smoke when the children were aged 4 years. Manzano-Salgado et al. (2019) also collected information on maternal diet, including fish consumption, with a questionnaire. Miura et al. (2018) did not provide information on exposure to any other agents. Five of the studies did not obtain biomonitoring data for contaminants other than PFAS, whereas Grandjean et al. (2012) measured PCBs in the serum samples; and none of the studies evaluated or measured exposure to agents other than those mentioned above.

(iii) Longitudinal and repeated-measures studies

Exposure assessment methods were reviewed for three studies with a longitudinal or repeated measures design (<u>Kim et al., 2016</u>, <u>2020</u>; <u>Blake et al., 2018</u>). In all three studies, LC-MS/MS was used to analyse serum levels of PFAS. Kim et al. (2020) measured serum levels of PFOA, PFOS, and 12 other PFAS and assessed outcomes in children from the general population of the Republic of Korea at the same three time points (ages 2, 4, and 6 years). Information on maternal smoking during pregnancy was collected. Blake et al. (2018) measured serum levels of PFOA, PFOS, and six other PFAS in a cohort of adults who were living near a river in the USA that was contaminated with PFOA and who were identified as being at high risk of elevated exposure to PFAS, particularly PFOA. The study group was a subset of residents near a uranium processing site, but this subset was unlikely to have uranium exposure above background. PFAS levels were measured at enrolment in the study and at one or two later time points for each participant, and outcomes were assessed at the same and/or different time point(s) as the collection of samples for measurement of serum levels of PFAS. In the first serum measurement, PFOA and PFOS were detected in all samples. No information on smoking or alcohol consumption was collected. Kim et al. (2016) measured levels of PFOA, PFOS, and 13 other PFAS in the serum of older adults (aged > 60 years) from the general population of the Republic of Korea who participated in a clinical trial on the effect of vitamin C on the outcomes. Serum levels of PFAS were measured at enrolment and at two additional time points over a 10-week period. Exposure to tobacco smoke (using urinary cotinine as a surrogate) and exposure to air pollutants (PM_{10}), ozone, and nitrogen oxide) were evaluated.

(iv) Study on pathology samples

Exposure assessment was reviewed for a study on PFAS levels in glioma and non-glioma brain tissue in patients (aged 2–77 years) with glioma, in China (Xie et al., 2023). The study included paired glioma and non-glioma brain tissue for 18 patients, as well as glioma or non-glioma brain tissue that did not come from the same patients, making a total of 137 glioma and 40 non-glioma brain tissue samples. PFOA, PFOS, and 15 other PFAS were analysed using LC-MS/MS in these brain tissue samples to evaluate the potential association between PFAS levels and glioma pathological grade, as well as related biomarkers.

The MRL for PFOA and PFOS in brain tissue was 0.05 ng/g (wet weight). PFOA and PFOS were detected at concentrations above the RL in 69% and 82%, respectively, of the glioma tissue samples, and in 33% and 65%, respectively, of the non-glioma tissue samples. The areas of the brain that were sampled for the non-glioma tissue samples were not provided, and a study by <u>Di</u> <u>Nisio et al. (2022)</u> showed that PFAS levels vary widely in different parts of the brain. This study did not report on brain tissue concentrations of contaminants other than PFAS.

(b) Critical review of exposure assessment in key mechanistic studies in exposed humans

(i) Cross-sectional studies

Exposure assessment in all the cross-sectional studies was based on biomonitoring data, and the studies shared many strengths and limitations. In all of these studies, the analytical methods used were state-of-art at the time when the studies were conducted, and the LODs or LOQs for PFOA and PFOS, when provided, were sufficiently low to ensure detection or quantification of PFOA and PFOS (when present) in all or most samples.

In cross-sectional studies in general, it is not possible to determine the temporal relationship between exposure and outcome. Relying solely on measurements made at a certain point in time makes it difficult to comprehensively assess the impact of long-term exposure on health. For cross-sectional studies in general, a single measurement may not accurately reflect longterm exposure levels, because the concentration of chemicals in the human body may fluctuate with changes in the environment and lifestyle habits over time. However, measured serum or plasma concentrations of PFOA and PFOS are objective measures that integrate exposure from various sources and pathways, including contributions from metabolism of precursors to PFOA or PFOS (Section 4.1), and measurement error in the chemical analysis is low. Because PFOA and PFOS have long elimination half-lives (several years; see Section 4.1), the concentrations measured at a single time point represent past exposure over a relatively long period of time (see Section 1.4.3). For these reasons, measurement of serum or plasma PFOA and PFOS concentrations at the same time as the outcome appraisal is considered to be an acceptable method of exposure assessment for the outcomes considered in these studies, and this is also true for measurement of PFOA and PFOS in cord blood (Liu et al., 2018b). In 16 of the 18 cross-sectional studies, it was reported that PFOA and PFOS were detected at levels above the LOD or LOQ in all or almost all samples; Lin et al. (2020c) and Di Nisio et al. (2020) did not provide this information. These factors limit the potential for non-differential exposure misclassification, in general.

All studies except one collected blood samples once and assessed the outcome at the same time point (or during the same period, <u>Zhang et al.</u>, <u>2022</u>) as the serum or plasma PFAS levels. In the study by <u>Watkins et al. (2014</u>), serum levels of PFAS were measured at two time points – several years before and at the same time that the outcome was assessed – and the analysis was based on the mean of the two serum PFAS values.

A potential limitation of cross-sectional studies is that exposures to other agents that were not measured or evaluated may be correlated with PFOA and PFOS exposure and may also have an impact on the outcome (e.g. act as confounders or effect-modifiers). As one example, exposure to dioxins can result in immune system suppression (WHO, 2016). Different outcome(s) were evaluated in each study, and substances that are potential confounders would probably

differ according to the outcome. Thirteen studies assessed exposure to tobacco smoke with serum cotinine measurements or questionnaires, and 10 studies assessed exposure to alcohol with questionnaires. Two studies (<u>Watkins et al., 2014</u> and <u>Lopez-Espinosa et al., 2021</u>) obtained information on regular use of anti-inflammatory drugs; one study (<u>Cheng et al., 2022</u>) obtained information on use of hypolipidaemic drugs; and one study (<u>Knox et al., 2011</u>) excluded participants who were taking hormonal medications.

Several studies measured exposures to contaminants other than PFAS. Abraham et al. (2020) measured several other POPs and heavy metals in plasma, and Grandjean et al. (2012) measured PCBs; in both studies, these other contaminants were evaluated as potential confounders of associations with PFAS. Additionally, <u>Clarity et al. (2021)</u> measured 10 flame retardants or their metabolites in the urine. However, exposures to other agents that may have an impact on the outcomes were not evaluated in the cross-sectional studies. This consideration may be particularly applicable in the study by <u>Clarity et al. (2021)</u> on firefighters and office workers. In this study, associations between the outcome and PFOA and PFOS were stronger in firefighters, who are exposed to many other contaminants in addition to PFAS (see Table S1.23, Annex 1, Supplementary material for Section 1, Exposure Characterization, online only, available from: <u>https://publications.iarc.</u> who.int/636) compared with office workers.

In one of the studies, <u>Abraham et al. (2020)</u> evaluated potential associations between serum levels of PFAS and antibody response to vaccination in children aged 1 year, including breast-fed and formula-fed children. In this study, samples were collected between 1997 and 1999, which corresponds with the period of highest PFOA and PFOS levels in the general population (see Section 1.4.3).

(ii) Prospective birth cohort studies

In the six prospective birth cohort studies, maternal serum or plasma PFAS level measured during pregnancy was used as an indicator of prenatal PFAS exposure for the children, in whom the outcomes were assessed at birth and/or at later time points. In one study (Grandjean et al., 2012), PFAS levels were also assessed in the children at age 5 years. The analytical methods used were state-of-art, and the LODs or LOQs for PFOA and PFOS, when provided, were sufficiently low to ensure detection or quantification of PFOA and PFOS in all or most samples. Because PFOA and PFOS have long elimination half-lives (several years; see Section 4.1), the concentrations measured in serum or plasma represent maternal exposure over a relatively long period of time.

Blood serum or plasma concentrations are an objective measure of exposure; the concentrations represent the combined exposure through all exposure pathways over a period of time and include contributions from the metabolism of precursors to PFOA or PFOS (see Section 1.4(d) or Section 4.1); and the measurement error in the chemical analyses is low. These factors limit the potential for non-differential exposure misclassification, in general. Five of the six studies (Goudarzi et al., 2017; Miura et al., 2018; Manzano-Salgado et al., 2019; Dalsager et al., 2021; Liu et al., 2022b) reported low LODs or LOQs for PFOA and PFOS, and the sixth study (Grandjean et al., 2012) did not provide information on the values of the LODs or LOQs. [The Working Group noted that even though not explicitly reported, data reported on tertiles of measured concentrations suggested that detection frequencies for PFOA and PFOS were high.] In the study by Manzano-Salgado et al. (2019), PFOA and PFOS were detected at concentrations above the LOD or LOQ in all or almost all samples, whereas Grandjean et al. (2012), Miura

et al. (2018), and <u>Liu et al. (2022b)</u> did not provide this information.

Factors such as plasma volume expansion and changes in glomerular filtration rate that occur during pregnancy may result in decreased PFAS concentrations in serum or plasma, and this effect may be greater when PFAS is measured later in pregnancy (reviewed in US EPA SAB, 2022). Maternal PFAS concentration was measured in the first trimester of pregnancy in the studies by Manzano-Salgado et al. (2019) and Dalsager et al. (2021), in the second or third trimester of pregnancy by Miura et al. (2018), in the first, second, or third trimester by Liu et al. (2022b), and in the third trimester by Grandjean et al. (2012) and Goudarzi et al. (2017). [The Working Group noted that although serum PFAS concentrations may decrease during pregnancy, this is unlikely to result in substantial exposure misclassification in studies in which blood PFAS concentrations are measured at the same time point in pregnancy in all participants. There is a higher risk of exposure misclassification in studies when serum PFAS concentrations are not measured during the same time period (e.g. trimester) in all participants.]

In two of the studies (Miura et al., 2018; Liu et al., 2022b), exposure and outcome were assessed in the same cord blood samples at birth, limiting the potential for non-differential exposure misclassification related to PFAS exposures other than from maternal fetal transfer. However, potential associations between the outcome and maternal PFAS concentrations were evaluated by Manzano-Salgado et al. (2019) at ages 1.5, 4, and 7 years, and by Dalsager et al. (2021) and Goudarzi et al. (2017) at up to age 4 years. The potential association between the outcome and maternal PFAS concentrations was evaluated by Liu et al. (2022b) at age 7 or 12 years as well as at birth, and by Grandjean et al. (2012) at ages 5 and 7 years. However, the potential impact of PFAS exposures that occurred postnatally was not considered, except by Grandjean et al. (2012),

who also assessed the association between serum PFAS concentration at age 5 years with the outcome at age 7 years. [The Working Group noted that prenatal exposures are an important time window of exposure for epigenetic changes.] Health outcomes assessed in these children may be associated with postnatal PFAS exposure instead of or in addition to prenatal exposure. Breastfeeding has an impact on postnatal exposure, with the magnitude of the impact being dependent on breastfeeding duration, as well as exposure through drinking-water, diet, consumer products, and other sources. Although there may be some relationship between exposure to the mother (and associated prenatal exposure) and postnatal exposure (e.g. if the mother and child both drink the same contaminated drinking-water), maternal/prenatal and postnatal exposure are not necessarily strongly correlated. For example, Grandjean et al. (2012) reported weak correlations (Pearson coefficients of 0.19 for PFOA and 0.27 for PFOS) for maternal PFAS concentrations at week 32 of pregnancy and postnatal PFAS concentrations at age 5 years.

In these studies, exposures to other agents that were not measured in the mothers or children (see Section 1.6.2(a) above) may be correlated with PFAS exposure and may also have an impact on the outcome as confounders or effect-modifiers.

(iii) Longitudinal and repeated-measures studies

Longitudinal or repeated measures were used in three studies. The strengths of these studies include that repeated measurements provide information on the variability of biomarkers over time. Other strengths include that, in all three studies, the analytical methods used were stateof-the-art, and the LODs or LOQs for PFOA and PFOS were sufficiently low to ensure detection or quantification of PFOA and PFOS in all or almost all samples. Because PFOA and PFOS have long elimination half-lives (several years; see Section 4.1), the concentrations measured in serum represent exposure over a relatively long period of time. Blood serum concentrations are an objective measure of exposure, the concentrations represent the combined exposure through all exposure pathways over a period of time, and the measurement error in the chemical analyses is low. These factors limit the potential for non-differential exposure misclassification, in general.

(iv) Study on pathology samples

In this study with a case-control design, Xie et al. (2023) measured concentrations of PFOA, PFOS, and 17 other PFAS in samples of glioma and non-glioma brain tissue. Although the analytical method (LC-MS/MS) was stateof-the-art, the percentage of samples in which PFOA and PFOS were detected at levels above the RL was 69% and 82%, respectively, of the glioma tissue samples, and 33% and 65%, respectively, of the non-glioma tissue samples, compared with other studies in which PFAS were detected in all or almost all samples in serum or plasma, or other matrices. [The Working Group noted that the low number of samples and low detection frequencies limited the informativeness of this study.]

In this study, paired glioma and non-glioma samples were available from only 18 patients, and the remainder of the total of 137 glioma and 40 non-glioma brain tissue samples did not come from the same patients. Additionally, the areas of the brain that were sampled for the non-glioma tissue samples were not reported. The comparisons of PFAS concentrations in glioma versus non-glioma tissue samples in this study were highly uncertain because, as previously stated, the specific part(s) of the brain that were sampled and compared were not known, and PFAS levels vary widely in different parts of the brain (Di Nisio et al., 2022). Also, comparison of PFAS levels in tumour and non-tumour brain tissues from different individuals is challenging to

interpret because PFAS exposures vary widely among individuals. Finally, it is possible that PFAS accumulate more in tumour tissue than in non-tumour tissue in the brain, resulting in reverse causation.

References

- Abdallah MA, Wemken N, Drage DS, Tlustos C, Cellarius C, Cleere K, et al. (2020). Concentrations of perfluoroalkyl substances in human milk from Ireland: implications for adult and nursing infant exposure. *Chemosphere*. 246:125724. doi:10.1016/j. chemosphere.2019.125724 PMID:31887492
- Abou-Khalil C, Sarkar D, Braykaa P, Boufadel MC (2022). Mobilization of per- and polyfluoroalkyl substances (PFAS) in soils: a review. *Curr Pollut Rep.* 8(4):422–44. doi:10.1007/s40726-022-00241-8
- Abraham K, Mielke H, Fromme H, Völkel W, Menzel J, Peiser M, et al. (2020). Internal exposure to perfluoroalkyl substances (PFASs) and biological markers in 101 healthy 1-year-old children: associations between levels of perfluorooctanoic acid (PFOA) and vaccine response. *Arch Toxicol.* 94(6):2131–47. doi:10.1007/ s00204-020-02715-4 PMID:32227269
- Abraham K, Monien BH (2022). Transdermal absorption of ${}^{13}C_4$ -perfluorooctanoic acid (${}^{13}C_4$ -PFOA) from a sunscreen in a male volunteer what could be the contribution of cosmetics to the internal exposure of perfluoroalkyl substances (PFAS)? *Environ Int.* 169:107549. doi:10.1016/j.envint.2022.107549 PMID:36191486
- ACGIH (2023). 2023 TLVs® and BEIs®. Based on the documentation of the threshold limit values for chemical substances and physical agents & biological exposure indices. Cincinnati (OH), USA: American Conference of Governmental Industrial Hygienists. Available from: <u>https://hse-me.ir/wp-con-tent/uploads/2023/05/2023 ACGIH Threshold Limit Values TLVs and Biological Exposure Indices.pdf</u>.
- ACS (2019). Perfluorooctanesulfonic acid. Molecule of the week archive. Washington (DC), USA: American Chemical Society Publishing. Available from: <u>https:// www.acs.org/molecule-of-the-week/archive/p/</u> <u>perfluorooctanesulfonic-acid.html</u>, accessed April 2023.
- Ahmed E, Mohibul Kabir KM, Wang H, Xiao D, Fletcher J, Donald WA (2019). Rapid separation of isomeric perfluoroalkyl substances by high-resolution differential ion mobility mass spectrometry. *Anal Chim*

Acta. 1058:127–35. doi:<u>10.1016/j.aca.2019.01.038</u> PMID:<u>30851846</u>

- Aimuzi R, Luo K, Huang R, Huo X, Nian M, Ouyang F, et al. (2020). Perfluoroalkyl and polyfluoroalkyl substances and maternal thyroid hormones in early pregnancy. *Environ Pollut*. 264:114557. doi:10.1016/j. envpol.2020.114557 PMID:32388293
- Alexander BH, Olsen GW (2007). Bladder cancer in perfluorooctanesulfonyl fluoride manufacturing workers. Ann Epidemiol. 17(6):471–8. doi:10.1016/j. annepidem.2007.01.036 PMID:17448680
- Alexander BH, Olsen GW, Burris JM, Mandel JH, Mandel JS (2003). Mortality of employees of a perfluorooctanesulphonyl fluoride manufacturing facility. *Occup Environ Med.* 60(10):722–9. doi:<u>10.1136/oem.60.10.722</u> PMID:<u>14504359</u>
- ASTM International (2017). Standard test method for determination of polyfluorinated compounds in soil by liquid chromatography tandem mass spectrometry (LC/MS/MS). D7968-17a. West Conshohocken (PA), USA: American Society for Testing and Materials. Available from: <u>https://cdn.standards.iteh.ai/samples</u> /98013/9a15621dc5524cc986bce9c0be6e2e74/ASTM-D7968-17a.pdf, accessed October 2024.
- ATSDR (2021). Toxicological profile for perfluoroalkyls. Atlanta (GA), USA: United States Department of Health and Human Services, Public Health Service, Agency for Toxic Substances and Disease Registry. Available from: <u>http://www.atsdr.cdc.gov/ToxProfiles/</u> <u>tp200.pdf</u>, accessed January 2023.
- ATSDR (2023). Per- and polyfluoroalkyl substances (PFAS) exposure assessment. Report. Airway Heights, Spokane County, Washington. Atlanta (GA), USA: United States Department of Health and Human Services, Public Health Service, Agency for Toxic Substances and Disease Registry. Available from: <u>https://www.atsdr.cdc.gov/pfas/docs/ATSDR-PFAS-EA-Site-D-SpokaneCounty-Report-508.pdf</u>, accessed October 2024.
- Australian Government (2017). Health-based guidance values for PFAS for use in site investigations in Australia. Canberra (ACT), Australia: Australian Government. Available from: <u>https://</u> www.health.gov.au/resources/publications/healthbased-guidance-values-for-pfas-for-use-in-siteinvestigations-in-australia?utm_source=health. gov.au&utm_medium=callout-auto-custom&utm_ campaign=digital_transformation, accessed_March 2023.
- Awad R, Zhou Y, Nyberg E, Namazkar S, Yongning W, Xiao Q, et al. (2020). Emerging per- and polyfluoroalkyl substances (PFAS) in human milk from Sweden and China. *Environ Sci Process Impacts*. 22(10):2023– 30. doi:10.1039/D0EM00077A PMID:32940316

- Balk FGP, Winkens Pütz K, Ribbenstedt A, Gomis MI, Filipovic M, Cousins IT (2019). Children's exposure to perfluoroalkyl acids – a modelling approach. *Environ Sci Process Impacts*. 21(11):1875–86. doi:10.1039/ <u>C9EM00323A</u> PMID:31549993
- Barber JL, Berger U, Chaemfa C, Huber S, Jahnke A, Temme C, et al. (2007). Analysis of per- and polyfluorinated alkyl substances in air samples from northwest Europe. J Environ Monit. 9(6):530–41. doi:10.1039/ b701417a PMID:17554424
- Barbo N, Stoiber T, Naidenko OV, Andrews DQ (2023). Locally caught freshwater fish across the United States are likely a significant source of exposure to PFOS and other perfluorinated compounds. *Environ Res.* 220:115165. doi:10.1016/j.envres.2022.115165 PMID:36584847
- Barry V, Winquist A, Steenland K (2013). Perfluorooctanoic acid (PFOA) exposures and incident cancers among adults living near a chemical plant. *Environ Health Perspect*. 121(11–12):1313–8. doi:<u>10.1289/ehp.</u> <u>1306615 PMID:24007715</u>
- BC MECCS (2020). Source drinking water quality guidelines. Guideline summary. Water quality guideline series, WQG-01. Victoria (BC), Canada: British Columbia Ministry of Environment and Climate Change Strategy. Available from: <u>https:// www2.gov.bc.ca/assets/gov/environment/air-landwater/water/waterquality/water-quality-guidelines/ approved-wqgs/drinking-water-and-recreation/ source drinking water quality guidelines bcenv. pdf, accessed October 2024.</u>
- Bečanová J, Melymuk L, Vojta Š, Komprdová K, Klánová J (2016). Screening for perfluoroalkyl acids in consumer products, building materials and wastes. *Chemosphere*. 164:322–9. doi:10.1016/j.chemosphere.2016.08.112 PMID:27592321
- Benskin JP, De Silva AO, Martin JW (2010a). Isomer profiling of perfluorinated substances as a tool for source tracking: a review of early findings and future applications. In: De Voogt P, editor. Reviews of environmental contamination and toxicology. Volume 208. Perfluorinated alkylated substances. pp. 111–160. New York (NY), USA: Springer New York.
- Benskin JP, Yeung LW, Yamashita N, Taniyasu S, Lam PK, Martin JW (2010b). Perfluorinated acid isomer profiling in water and quantitative assessment of manufacturing source. *Environ Sci Technol.* 44(23):9049–54. doi:<u>10.1021/es102582x</u> PMID:<u>21058746</u>
- Beser MI, Pardo O, Beltrán J, Yusà V (2011). Determination of per- and polyfluorinated substances in airborne particulate matter by microwave-assisted extraction and liquid chromatography-tandem mass spectrometry. J Chromatogr A. 1218(30):4847–55. doi:10.1016/j. chroma.2011.04.082 PMID:21616498

- Beser MI, Pardo O, Beltrán J, Yusà V (2019). Determination of 21 perfluoroalkyl substances and organophosphorus compounds in breast milk by liquid chromatography coupled to orbitrap high-resolution mass spectrometry. *Anal Chim Acta*. 1049:123–32. doi:<u>10.1016/j.</u> <u>aca.2018.10.033</u> PMID:<u>30612643</u>
- Besis A, Botsaropoulou E, Samara C, Katsoyiannis A, Hanssen L, Huber S (2019). Perfluoroalkyl substances (PFASs) in air-conditioner filter dust of indoor microenvironments in Greece: implications for exposure. *Ecotoxicol Environ Saf.* 183:109559. doi:10.1016/j. ecoenv.2019.109559 PMID:31408822
- Blake BE, Pinney SM, Hines EP, Fenton SE, Ferguson KK (2018). Associations between longitudinal serum perfluoroalkyl substance (PFAS) levels and measures of thyroid hormone, kidney function, and body mass index in the Fernald Community Cohort. *Environ Pollut*. 242(Pt A):894–904. doi:10.1016/j.envpol.2018.07.042 PMID:30373035
- Blomberg AJ, Norén E, Haug LS, Lindh C, Sabaredzovic A, Pineda D, et al. (2023). Estimated transfer of perfluoroalkyl substances (PFAS) from maternal serum to breast milk in women highly exposed from contaminated drinking water: a study in the Ronneby Mother-Child Cohort. *Environ Health Perspect*. 131(1):17005. doi:10.1289/EHP11292 PMID:36688826
- Bonefeld-Jørgensen EC, Long M, Bossi R, Ayotte P, Asmund G, Krüger T, et al. (2011). Perfluorinated compounds are related to breast cancer risk in Greenlandic Inuit: a case control study. *Environ Health*. 10(1):88. doi:10.1186/1476-069X-10-88 PMID:21978366
- Bonefeld-Jørgensen EC, Long M, Fredslund SO, Bossi R, Olsen J (2014). Breast cancer risk after exposure to perfluorinated compounds in Danish women: a case-control study nested in the Danish National Birth Cohort. *Cancer Causes Control*. 25(11):1439–48. doi:10.1007/s10552-014-0446-7 PMID:25148915
- Bräunig J, Baduel C, Barnes CM, Mueller JF (2019). Leaching and bioavailability of selected perfluoroalkyl acids (PFAAs) from soil contaminated by firefighting activities. *Sci Total Environ*. 646:471–9. doi:<u>10.1016/j.</u> <u>scitotenv.2018.07.231</u> PMID:<u>30056234</u>
- Brusseau ML, Anderson RH, Guo B (2020). PFAS concentrations in soils: background levels versus contaminated sites. *Sci Total Environ*. 740:140017. doi:<u>10.1016/j.</u> <u>scitotenv.2020.140017</u> PMID:<u>32927568</u>
- Buck RC, Franklin J, Berger U, Conder JM, Cousins IT, de Voogt P, et al. (2011). Perfluoroalkyl and polyfluoroalkyl substances in the environment: terminology, classification, and origins. *Integr Environ Assess Manag.* 7(4):513–41. doi:10.1002/ieam.258 PMID:21793199
- Burgess JL, Fisher JM, Nematollahi A, Jung AM, Calkins MM, Graber JM, et al. (2023). Serum per- and poly-fluoroalkyl substance concentrations in four municipal US fire departments. *Am J Ind Med.* 66(5):411–23. doi:10.1002/ajim.23413 PMID:35864570

- Butt CM, Berger U, Bossi R, Tomy GT (2010). Levels and trends of poly- and perfluorinated compounds in the arctic environment. *Sci Total Environ*. 408(15):2936– 65. doi:10.1016/j.scitotenv.2010.03.015 PMID:20493516
- Cai M, Zhao Z, Yin Z, Ahrens L, Huang P, Cai M, et al. (2012). Occurrence of perfluoroalkyl compounds in surface waters from the North Pacific to the Arctic Ocean. *Environ Sci Technol.* 46(2):661–8. doi:<u>10.1021/ es2026278</u> PMID:<u>22128794</u>
- Calafat AM, Kato K, Hubbard K, Jia T, Botelho JC, Wong LY (2019). Legacy and alternative per- and poly-fluoroalkyl substances in the US general population: paired serum-urine data from the 2013–2014 National Health and Nutrition Examination Survey. *Environ Int.* 131:105048. doi:10.1016/j.envint.2019.105048 PMID:31376596
- Calafat AM, Kuklenyik Z, Reidy JA, Caudill SP, Tully JS, Needham LL (2007). Serum concentrations of 11 polyfluoroalkyl compounds in the US population: data from the national health and nutrition examination survey (NHANES). *Environ Sci Technol.* 41(7):2237–42. doi:10.1021/es062686m PMID:17438769
- Camoiras González P, Sadia M, Baabish A, Sobhanei S, Fiedler H (2021). Air monitoring with passive samplers for perfluoroalkane substances in developing countries (2017–2019). *Chemosphere*. 282:131069. doi:<u>10.1016/j.</u> <u>chemosphere.2021.131069</u> PMID:<u>34470153</u>
- Canadian Council of Ministers of the Environment (2021). Canadian soil and groundwater quality guidelines for the protection of environmental and human health: perfluorooctane sulfonate (PFOS). Canadian environmental quality guidelines, 1999. Winnipeg (MB), Canada: Canadian Council of Ministers of the Environment (CCME). Available from: <u>https:// ccme.ca/en/canadian-soil-and-groundwater-qualityguidelines-for-the-protection-of-environmentaland-human-health-perfluorooctane-sulfonate-pfos, accessed October 2024.</u>
- Cao L, Guo Y, Chen Y, Hong J, Wu J, Hangbiao J (2022). Per-/polyfluoroalkyl substance concentrations in human serum and their associations with liver cancer. *Chemosphere*. 296:134083. doi:10.1016/j. chemosphere.2022.134083 PMID:35216980
- CAS (2023). CAS Scifinder. Washington (DC), USA: American Chemical Society. Available from: <u>https://www.cas.org/products/scifinder/explore</u>, accessed November 2023.
- Casas G, Martinez-Varela A, Vila-Costa M, Jiménez B, Dachs J (2021). Rain amplification of persistent organic pollutants. *Environ Sci Technol*. 55(19):12961–72. doi:10.1021/acs.est.1c03295 PMID:34553911
- CDC (2022). Calculation of PFOS and PFOA as the sum of isomers. Atlanta (GA), USA: Centers for Disease Control and Prevention. Available from: <u>https://www.cdc.gov/environmental-exposure-report/php/about-the-data/calculation-of-pfos-and-pfoa-as-sum-of-isomers.</u>

<u>html?CDC_AAref_Val=https://www.cdc.gov/</u> <u>exposurereport/pfos_pfoa_calculation.html</u>, accessed October 2024.

- CDC (2023). National Health and Nutrition Examination Survey. Atlanta (GA), USA: National Center for Health Statistics, Centers for Disease Control and Prevention. Available from: <u>https://www.cdc.gov/nchs/nhanes/</u> <u>index.htm</u>, accessed October 2024.
- Chaemfa C, Barber JL, Huber S, Breivik K, Jones KC (2010). Screening for PFOS and PFOA in European air using passive samplers. *J Environ Monit*. 12(5):1100–9. doi:10.1039/b921628f PMID:20461269
- Chang VC, Rhee J, Berndt SI, Moore SC, Freedman ND, Jones RR, et al. (2023). Serum perfluorooctane sulfonate and perfluorooctanoate and risk of postmenopausal breast cancer according to hormone receptor status: an analysis in the Prostate, Lung, Colorectal and Ovarian Cancer Screening Trial. *Int J Cancer*. 153(4):775-782. doi:10.1002/ijc.34487 PMID:36843273
- Chen S, Jiao XC, Gai N, Li XJ, Wang XC, Lu GH, et al. (2016). Perfluorinated compounds in soil, surface water, and groundwater from rural areas in eastern China. *Environ Pollut*. 211:124–31. doi:10.1016/j. envpol.2015.12.024 PMID:26745397
- Chen Y, Paul KC, Walker DI, Jones DP, Wang X, Ritz BR, et al. (2023). Neonatal per- and polyfluoroalkyl substance exposure in relation to retinoblastoma. *Environ Res.* 240(Part 2):117435. doi:<u>10.1016/j.envres.2023.117435</u> PMID:<u>37866539</u>
- Cheng X, Wei Y, Zhang Z, Wang F, He J, Wang R, et al. (2022). Plasma PFOA and PFOS levels, DNA methylation, and blood lipid levels: a pilot study. *Environ Sci Technol.* 56(23):17039–51. doi:<u>10.1021/acs.est.2c04107</u> PMID:<u>36374530</u>
- Cheng Y, An Q, Qi H, Li R, Liu W, Gu B, et al. (2023). Temporal trends of legacy and emerging PFASs from 2011 to 2021 in agricultural soils of eastern China: impacts of the Stockholm Convention. *Environ Sci Technol.* 57(25):9277–86. doi:<u>10.1021/acs.est.2c07873</u> PMID:<u>37307015</u>
- Choi H, Bae IA, Choi JC, Park SJ, Kim M (2018). Perfluorinated compounds in food simulants after migration from fluorocarbon resin-coated frying pans, baking utensils, and non-stick baking papers on the Korean market. *Food Addit Contam Part B Surveill*. 11(4):264–72. doi:10.1080/19393210.2018.1499677 PMID:30052135
- Christensen BT, Calkins MM (2023). Occupational exposure to per- and polyfluoroalkyl substances: a scope review of the literature from 1980–2021. *J Expo Sci Environ Epidemiol*. doi:<u>10.1038/s41370-023-00536-y</u> PMID:<u>36977833</u>
- Clarity C, Trowbridge J, Gerona R, Ona K, McMaster M, Bessonneau V, et al. (2021). Associations between polyfluoroalkyl substance and organophosphate flame retardant exposures and telomere length in a cohort of

women firefighters and office workers in San Francisco. *Environ Health.* 20(1):97. doi:<u>10.1186/s12940-021-</u>00778-z PMID:<u>34454526</u>

- Clarke DB, Bailey VA, Routledge A, Lloyd AS, Hird S, Mortimer DN, et al. (2010). Dietary intake estimate for perfluoroccanesulphonic acid (PFOS) and other perfluoroccompounds (PFCs) in UK retail foods following determination using standard addition LC-MS/MS. *Food Addit Contam Part A Chem Anal Control Expo Risk Assess.* 27(4):530-45. doi:10.1080/19440040903476590 PMID:20198525
- Cohn BA, La Merrill MA, Krigbaum NY, Wang M, Park JS, Petreas M, et al. (2020). In utero exposure to polyand perfluoroalkyl substances (PFASs) and subsequent breast cancer. *Reprod Toxicol*. 92:112–9. doi:<u>10.1016/j.</u> <u>reprotox.2019.06.012</u> PMID:<u>31323350</u>
- Consonni D, Straif K, Symons JM, Tomenson JA, van Amelsvoort LG, Sleeuwenhoek A, et al. (2013). Cancer risk among tetrafluoroethylene synthesis and polymerization workers. *Am J Epidemiol*. 178(3):350–8. doi:10.1093/aje/kws588 PMID:23828249
- Cookson ES, Detwiler RL (2022). Global patterns and temporal trends of perfluoroalkyl substances in municipal wastewater: a meta-analysis. *Water Res*. 221:118784. doi:<u>10.1016/j.watres.2022.118784</u> PMID:<u>35949071</u>
- Costa G, Sartori S, Consonni D (2009). Thirty years of medical surveillance in perfluooctanoic acid production workers. *J Occup Environ Med.* 51(3):364–72. doi:10.1097/JOM.0b013e3181965d80 PMID:19225424
- Costello MCS, Lee LS (2020). Sources, fate, and plant uptake in agricultural systems of per- and polyfluoroalkyl substances. *Curr Pollut Rep.* doi:10.1007/ <u>s40726-020-00168-y</u>
- Cousins IT, DeWitt JC, Glüge J, Goldenman G, Herzke D, Lohmann R, et al. (2020). The high persistence of PFAS is sufficient for their management as a chemical class. *Environ Sci Process Impacts*. 22(12):2307–12. doi:10.1039/D0EM00355G PMID:33230514
- CRCCARE (2018). Practitioner guide to risk-based assessment, remediation and management of PFAS site contamination. Technical report No. 43. Callaghan (NSW), Australia: Cooperative Research Centre for Contamination Assessment and Remediation of the Environment. Available from: <u>https://crccare.com/wp-content/uploads/2024/02/</u> <u>CRCCARETechreport43_PFASpractitionerguide</u> web.pdf, accessed October 2024.
- D'Hollander W, Roosens L, Covaci A, Cornelis C, Reynders H, Campenhout KV, et al. (2010). Brominated flame retardants and perfluorinated compounds in indoor dust from homes and offices in Flanders, Belgium. *Chemosphere*. 81(4):478–87. doi:10.1016/j. chemosphere.2010.07.043 PMID:20709355
- Dagorn G, Aubert R, Horel S, Martinon L, Steffen T (2023). Forever pollution: explore the map of Europe's PFAS contamination. Paris, France:
Le Monde. 23 February 2023, updated 17 February 2024. Available from: <u>https://www.lemonde.</u> <u>fr/en/les-decodeurs/article/2023/02/23/forever-pollution-explore-the-map-of-europe-s-pfas-contamination 6016905 8.html</u>, accessed October 2024.

- Dalahmeh S, Tirgani S, Komakech AJ, Niwagaba CB, Ahrens L (2018). Per- and polyfluoroalkyl substances (PFASs) in water, soil and plants in wetlands and agricultural areas in Kampala, Uganda. *Sci Total Environ*. 631–632:660–7. doi:<u>10.1016/j.scitotenv.2018.03.024</u> PMID:<u>29539594</u>
- Dalsager L, Christensen N, Halekoh U, Timmermann CAG, Nielsen F, Kyhl HB, et al. (2021). Exposure to perfluoroalkyl substances during fetal life and hospitalization for infectious disease in childhood: a study among 1,503 children from the Odense Child Cohort. *Environ Int.* 149:106395. doi:<u>10.1016/j.envint.2021.106395</u> PMID:<u>33508532</u>
- Danish Environmental Protection Agency (2023). Decree on water quality and supervision of water supply facilities. Odense, Denmark: Danish Environmental Protection Agency. Available from: <u>https://www.rets information.dk/eli/lta/2023/1023</u>, accessed October 2024. [Danish]
- de la Torre A, Navarro I, Sanz P, Mártinez MLÁ (2019). Occurrence and human exposure assessment of perfluorinated substances in house dust from three European countries. *Sci Total Environ*. 685:308–14. doi:<u>10.1016/j.scitotenv.2019.05.463</u> PMID:<u>31176217</u>
- De Silva AO, Armitage JM, Bruton TA, Dassuncao C, Heiger-Bernays W, Hu XC, et al. (2021). PFAS exposure pathways for humans and wildlife: a synthesis of current knowledge and key gaps in understanding. *Environ Toxicol Chem.* 40(3):631–57. doi:10.1002/ etc.4935 PMID:33201517
- Death C, Bell C, Champness D, Milne C, Reichman S, Hagen T (2021). Per- and polyfluoroalkyl substances (PFAS) in livestock and game species: a review. *Sci Total Environ*. 774:144795. doi:<u>10.1016/j.scitotenv.</u> <u>2020.144795</u> PMID:<u>33609849</u>
- Di Nisio A, Pannella M, Vogiatzis S, Sut S, Dall'Acqua S, Rocca MS, et al. (2022). Impairment of human dopaminergic neurons at different developmental stages by perfluoro-octanoic acid (PFOA) and differential human brain areas accumulation of perfluoroalkyl chemicals. *Environ Int.* 158:106982. doi:10.1016/j. envint.2021.106982 PMID:34781208
- Di Nisio A, Rocca MS, Sabovic I, De Rocco Ponce M, Corsini C, Guidolin D, et al. (2020). Perfluorooctanoic acid alters progesterone activity in human endometrial cells and induces reproductive alterations in young women. *Chemosphere*. 242:125208. doi:10.1016/j. chemosphere.2019.125208 PMID:31896193

- Dobraca D, Israel L, McNeel S, Voss R, Wang M, Gajek R, et al. (2015). Biomonitoring in California firefighters: metals and perfluorinated chemicals. *J Occup Environ Med*.57(1):88–97. doi:<u>10.1097/JOM.000000000000307</u> PMID:25563545
- Domingo JL, Nadal M (2017). Per- and polyfluoroalkyl substances (PFASs) in food and human dietary intake: a review of the recent scientific literature. *J Agric Food Chem.* 65(3):533–43. doi:<u>10.1021/acs.jafc.6b04683</u> PMID:<u>28052194</u>
- Domingo JL, Nadal M (2019). Human exposure to per- and polyfluoroalkyl substances (PFAS) through drinking water: a review of the recent scientific literature. *Environ Res.* 177:108648. doi:<u>10.1016/j.envres.2019.108648</u> PMID:<u>31421451</u>
- Douglas GB, Vanderzalm JL, Williams M, Kirby JK, Kookana RS, Bastow TP, et al. (2023). PFAS contaminated asphalt and concrete – knowledge gaps for future research and management. *Sci Total Environ*. 887:164025. doi:10.1016/j.scitotenv.2023.164025 PMID: 37169188
- Dreyer A, Kirchgeorg T, Weinberg I, Matthias V (2015). Particle-size distribution of airborne polyand perfluorinated alkyl substances. *Chemosphere*. 129:142–9. doi:10.1016/j.chemosphere.2014.06.069 PMID:25027782
- DWI (2021). Guidance on the Water Supply (Water Quality) Regulations 2016 specific to PFOS (perfluorooctane sulphonate) and PFOA (perfluorooctanoic acid) concentrations in drinking water. London, UK: Drinking Water Inspectorate. Available from: <u>https://dwi-content.s3.eu-west-2.amazonaws.com/</u> <u>wp-content/uploads/2021/01/12110137/PFOS-PFOAguidance-2021.pdf</u>, accessed October 2024.
- ECCC (2017). Aqueous film-forming foam and the Prohibition of Certain Toxic Substances Regulations. Gatineau (QC), Canada: Environment and Climate Change Canada. Available from: <u>https://www.canada.ca/en/environment-climatechange/services/management-toxic-substances/ list-canadian-environmental-protection-act/ perfluorooctane-sulfonate/film-forming-foamprohibition-toxic-substances.html, accessed March 2023.</u>
- ECCC (2023). Per- and polyfluoroalkyl substances (PFAS) information sheet. Gatineau (QC), Canada: Environment and Climate Change Canada. Available from: <u>https://www.canada.ca/en/health-canada/services/chemical-substances/fact-sheets/chemical-sglance/per-polyfluoroalkyl-substances.html</u>, accessed September 2023.
- ECHÀ (2023). ECHA's committees: EU-wide PFAS ban in firefighting foams warranted. Helsinki, Finland: European Health Agency. Available from: <u>https://echa.europa.eu/-/</u>

<u>echa-s-committees-eu-wide-pfas-ban-in-firefighting-</u> <u>foams-warranted</u>, accessed September 2023.

- ECOS (2023). Processes & considerations for setting state PFAS standards. Washington (DC), USA: The Environmental Council of the States (ECOS). Available from: <u>https://www.ecos.org/wp-content/</u> <u>uploads/2023/03/2023-ECOS-PFAS-Standards-Paper-</u> <u>Update.pdf</u>, accessed October 2024.
- EFSA Panel on Contaminants in the Food Chain (CONTAM); Knutsen HK, Alexander J, Barregård L, Bignami M, Brüschweiler B, et al. (2018). Risk to human health related to the presence of perfluorooc-tane sulfonic acid and perfluorooctanoic acid in food. *EFSA J.* 16(12):e05194. doi:<u>10.2903/j.efsa.2018.5194</u> PMID:<u>32625773</u>
- EFSA Panel on Contaminants in the Food Chain (EFSA CONTAM Panel); Schrenk D, Bignami M, Bodin L, Chipman JK, Del Mazo J, et al. (2020). Risk to human health related to the presence of perfluoroalkyl substances in food. *EFSA J*. 18(9):e06223. <u>https://efsa.onlinelibrary.wiley.com/doi/epdf/10.2903/j.efsa.2020.6223</u> PMID:<u>32994824</u>
- Egeghy PP, Lorber M (2011). An assessment of the exposure of Americans to perfluorooctane sulfonate: a comparison of estimated intake with values inferred from NHANES data. J Expo Sci Environ Epidemiol. 21(2):150–68. doi:10.1038/jes.2009.73 PMID:20145679
- Ehresman DJ, Froehlich JW, Olsen GW, Chang SC, Butenhoff JL (2007). Comparison of human whole blood, plasma, and serum matrices for the determination of perfluorooctanesulfonate (PFOS), perfluorooctanoate (PFOA), and other fluorochemicals. *Environ Res.* 103(2):176–84. doi:10.1016/j.envres.2006.06.008 PMID:16893538
- Environmental Working Group (2023). Mapping the PFAS contamination crisis: new data show 3,186 sites in 50 states, the District of Columbia and two territories. PFAS Contamination in the US. Washington (DC), USA: Environmental Working Group. Available from: <u>https://www.ewg.org/interactive-maps/pfas</u> contamination/, accessed October 2024.
- Eriksen KT, Sørensen M, McLaughlin JK, Lipworth L, Tjønneland A, Overvad K, et al. (2009). Perfluorooctanoate and perfluorooctanesulfonate plasma levels and risk of cancer in the general Danish population. J Natl Cancer Inst. 101(8):605–9. doi:10.1093/jnci/djp041 PMID:19351918
- Eriksson U, Mueller JF, Toms LL, Hobson P, Kärrman A (2017). Temporal trends of PFSAs, PFCAs and selected precursors in Australian serum from 2002 to 2013. *Environ Pollut*. 220(Pt A):168–77. doi:10.1016/j. envpol.2016.09.036 PMID:27726977
- EU (2020). Directive (EU) 2020/2184 of the European Parliament and of the Council of 16 December 2020 on the quality of water intended for human consumption (recast). Off J Eur Union. L435/1.

https://eur-lex.europa.eu/legal-content/EN/TXT/ PDF/?uri=CELEX:32020L2184.

- EU (2022). Commission recommendation (EU) 2022/1431 of 24 August 2022 on the monitoring of perfluoroalkyl substances in food. Off J Eur Union. L221/105. <u>https://eur-lex.europa.eu/legal-content/EN/TXT/</u> PDF/?uri=CELEX:32022H1431.
- European Commission (2022). Annexes to the proposal for a directive of the European Parliament and of the Council amending Directive 2000/60/EC establishing a framework for community action in the field of water policy. Directive 2006/118/EC on the protection of groundwater against pollution and deterioration and Directive 2008/105/EC on environmental quality standards in the field of water policy. Brussels, Belgium: European Commission. Available from: <u>https:// environment.ec.europa.eu/system/files/2022-10/ Annexes%20to%20the%20proposal_0.pdf</u>, accessed October 2023.
- European Environment Agency (2021). Fluorinated polymers in a low carbon, circular and toxic-free economy. Technical report. Eionet report – ETC/ WMGE 2021/9. Copenhagen, Denmark: European Topic Centre Waste and Materials in a Green Economy, European Environment Agency. Available from: <u>https://www.eionet.europa.eu/etcs/etc-cm/products/</u> <u>etc-cme-and-etc-wmge-report-9-2021-fluorinated-</u> <u>polymers-in-a-low-carbon-circular-and-toxic-free-</u> <u>economy/@@download/file/EEA%20task%20on%20</u> <u>fluoropolymers_Full%20report_December%2014%20</u> <u>2021%20(1).pdf</u>, accessed October 2024.
- Fan X, Wang Z, Li Y, Wang H, Fan W, Dong Z (2021). Estimating the dietary exposure and risk of persistent organic pollutants in China: a national analysis. *EnvironPollut*.288:117764. doi:<u>10.1016/j.envpol.2021.117764</u> PMID:<u>34280741</u>
- Felizeter S, Jürling H, Kotthoff M, De Voogt P, McLachlan MS (2021). Uptake of perfluorinated alkyl acids by crops: results from a field study. *Environ Sci Process Impacts.* 23(8):1158–70. doi:10.1039/D1EM00166C PMID:34259284
- Feng Y, Bai Y, Lu Y, Chen M, Fu M, Guan X, et al. (2022). Plasma perfluoroalkyl substance exposure and incidence risk of breast cancer: a case–cohort study in the Dongfeng–Tongji cohort. *Environ Pollut.* 306:119345. doi:10.1016/j.envpol.2022.119345 PMID:35472559
- Fiedler H, Sadia M, Baabish A, Sobhanei S (2022). Perfluoroalkane substances in national samples from global monitoring plan projects (2017–2019). *Chemosphere*. 307(Pt 3):136038. doi:10.1016/j. <u>chemosphere.2022.136038</u> PMID:<u>35977568</u>
- Filipovic M, Woldegiorgis A, Norström K, Bibi M, Lindberg M, Österås A-H (2015a). Historical usage of aqueous film forming foam: a case study of the widespread distribution of perfluoroalkyl acids from a military airport to groundwater, lakes, soils

and fish. *Chemosphere*. 129:39–45. doi:<u>10.1016/j.</u> <u>chemosphere.2014.09.005</u> PMID:<u>25262531</u>

- Flaherty JM, Connolly PD, Decker ER, Kennedy SM, Ellefson ME, Reagen WK, et al. (2005). Quantitative determination of perfluorooctanoic acid in serum and plasma by liquid chromatography tandem mass spectrometry. J Chromatogr B Analyt Technol Biomed Life Sci. 819(2):329–38. doi:10.1016/j.jchromb.2005.03.002 PMID:15833298
- Fletcher T, Galloway TS, Melzer D, Holcroft P, Cipelli R, Pilling LC, et al. (2013). Associations between PFOA, PFOS and changes in the expression of genes involved in cholesterol metabolism in humans. *Environ Int.* 57–58:2–10. doi:<u>10.1016/j.envint.2013.03.008</u> PMID:<u>23624243</u>
- Fraser AJ, Webster TF, Watkins DJ, Nelson JW, Stapleton HM, Calafat AM, et al. (2012). Polyfluorinated compounds in serum linked to indoor air in office environments. *Environ Sci Technol.* 46(2):1209–15. doi:<u>10.1021/es2038257</u> PMID:<u>22148395</u>
- Freberg BI, Haug LS, Olsen R, Daae HL, Hersson M, Thomsen C, et al. (2010). Occupational exposure to airborne perfluorinated compounds during professional ski waxing. *Environ Sci Technol.* 44(19):7723–8. doi:<u>10.1021/es102033k</u> PMID:<u>20831156</u>
- Frenoy P, Perduca V, Cano-Sancho G, Antignac JP, Severi G, Mancini FR (2022). Application of two statistical approaches (Bayesian kernel machine regression and principal component regression) to assess breast cancer risk in association to exposure to mixtures of brominated flame retardants and per- and polyfluorinated alkylated substances in the E3N cohort. *Environ Health.* 21(1):27. doi:10.1186/s12940-022-00840-4 PMID:35216589
- Frisbee SJ, Brooks AP Jr, Maher A, Flensborg P, Arnold S, Fletcher T, et al. (2009). The C8 Health Project: design, methods, and participants. *Environ Health Perspect*. 117(12):1873–82. doi:<u>10.1289/ehp.0800379</u> PMID:<u>20049206</u>
- Frisbee SJ, Shankar A, Knox SS, Steenland K, Savitz DA, Fletcher T, et al. (2010). Perfluorooctanoic acid, perfluorooctanesulfonate, and serum lipids in children and adolescents: results from the C8 Health Project. *Arch Pediatr Adolesc Med.* 164(9):860–9. doi:10.1001/archpediatrics.2010.163 PMID:20819969
- Fromme H, Fuchs V, Albrecht M, Aschenbrenner B, Röhl C, Janitzki N, et al. (2022). Polychlorinated dioxins and dibenzofurans (PCDD/F), polybrominated dioxins and dibenzofurans (PBDD/F), polychlorinated biphenyls (PCB), polybrominated diphenyl ethers (PBDE), and per- and polyfluoroalkyl substances (PFAS) in German breast milk samples (LUPE 8). *Sci Total Environ*. 825:154066. doi:10.1016/j.scitotenv.2022.154066
 PMID:35217048

- Fromme H, Mosch C, Morovitz M, Alba-Alejandre I, Boehmer S, Kiranoglu M, et al. (2010). Pre- and postnatal exposure to perfluorinated compounds (PFCs). *Environ Sci Technol.* 44(18):7123–9. doi:10.1021/ es101184f PMID:20722423
- Fu J, Gao Y, Cui L, Wang T, Liang Y, Qu G, et al. (2016). Occurrence, temporal trends, and half-lives of perfluoroalkyl acids (PFAAs) in occupational workers in China. *Sci Rep.* 6(1):38039. doi:<u>10.1038/srep38039</u> PMID:<u>27905562</u>
- Fujii Y, Harada KH, Koizumi A (2013). Occurrence of perfluorinated carboxylic acids (PFCAs) in personal care products and compounding agents. *Chemosphere*. 93(3):538–44. doi:<u>10.1016/j.chemosphere.2013.06.049</u> PMID:<u>23932147</u>
- Gao K, Fu J, Xue Q, Li Y, Liang Y, Pan Y, et al. (2018). An integrated method for simultaneously determining 10 classes of per- and polyfluoroalkyl substances in one drop of human serum. *Anal Chim Acta*. 999:76–86. doi:10.1016/j.aca.2017.10.038 PMID:29254577
- Gao Y, Fu J, Cao H, Wang Y, Zhang A, Liang Y, et al. (2015). Differential accumulation and elimination behavior of perfluoroalkyl acid isomers in occupational workers in a manufactory in China. *Environ Sci Technol.* 49(11):6953–62. doi:10.1021/acs.est.5b00778 PMID:25927957
- Garnett J, Halsall C, Winton H, Joerss H, Mulvaney R, Ebinghaus R, et al. (2022). Increasing accumulation of perfluorocarboxylate contaminants revealed in an Antarctic firn core (1958–2017). *Environ Sci Technol.* 56(16):11246–55. doi:10.1021/acs.est.2c02592 PMID:35881889
- Gebbink WA, Berger U, Cousins IT (2015). Estimating human exposure to PFOS isomers and PFCA homologues: the relative importance of direct and indirect (precursor) exposure. *Environ Int.* 74:160–9. doi:10.1016/j.envint.2014.10.013 PMID:25454233
- Genualdi S, Beekman J, Carlos K, Fisher CM, Young W, DeJager L, et al. (2022). Analysis of per- and poly-fluoroalkyl substances (PFAS) in processed foods from FDA's Total Diet Study. *Anal Bioanal Chem.* 414(3):1189–99. doi:<u>10.1007/s00216-021-03610-2</u> PMID:<u>34435208</u>
- Ghisari M, Long M, Røge DM, Olsen J, Bonefeld-Jørgensen EC (2017). Polymorphism in xenobiotic and estrogen metabolizing genes, exposure to perfluorinated compounds and subsequent breast cancer risk: a nested case-control study in the Danish National Birth Cohort. *Environ Res.* 154:325–33. doi:<u>10.1016/j.</u> envres.2017.01.020 PMID:<u>28157646</u>
- Ghisi R, Vamerali T, Manzetti S (2019). Accumulation of perfluorinated alkyl substances (PFAS) in agricultural plants: a review. *Environ Res.* 169:326–41. doi:<u>10.1016/j.</u> <u>envres.2018.10.023</u> PMID:<u>30502744</u>
- Giglioli S, Colombo L, Azzellino A (2023). Cluster and multivariate analysis to study the diffuse contamination of emerging per- and polyfluoroalkyl substances

(PFAS) in the Veneto Region plain (northeastern Italy). *Chemosphere*. 319:137916. doi:<u>10.1016/j</u>. <u>chemosphere.2023.137916</u> PMID:<u>36706810</u>

- Gilliland FD, Mandel JS (1996). Serum perfluorooctanoic acid and hepatic enzymes, lipoproteins, and cholesterol: a study of occupationally exposed men. Am J Ind Med. 29(5):560–8. doi:10.1002/ (SICI)1097-0274(199605)29:5<560::AID-AJIM17> 3.0.CO;2-Z PMID:8732932
- Girardi P, Merler E (2019). A mortality study on male subjects exposed to polyfluoroalkyl acids with high internal dose of perfluorooctanoic acid. *Environ Res.* 179(PtA):108743. doi:<u>10.1016/j.envres.2019.108743</u> PMID:<u>31542491</u>
- Glüge J, Scheringer M, Cousins IT, DeWitt JC, Goldenman G, Herzke D, et al. (2020). An overview of the uses of per- and polyfluoroalkyl substances (PFAS). *Environ Sci Process Impacts*. 22(12):2345–73. doi:10.1039/ D0EM00291G PMID:33125022
- Göen T, Abballe A, Bousoumah R, Godderis L, Iavicoli I, Ingelido AM, et al.; HBM4EU Chromates Study Team (2024). HBM4EU chromates study PFAS exposure in electroplaters and bystanders. *Chemosphere*. 346:140613. doi:10.1016/j.chemosphere.2023.140613
 PMID:<u>37944767</u>
- Gómez V, Torres M, Karásková P, Přibylová P, Klánová J, Pozo K (2021). Occurrence of perfluoroalkyl substances (PFASs) in marine plastic litter from coastal areas of Central Chile. *Mar Pollut Bull*. 172:112818. doi:<u>10.1016/j.</u> <u>marpolbul.2021.112818</u> PMID:<u>34391011</u>
- Goodrich JA, Walker D, Lin X, Wang H, Lim T, McConnell R, et al. (2022). Exposure to perfluoroalkyl substances and risk of hepatocellular carcinoma in a multiethnic cohort. *JHEP Rep.* 4(10):100550. doi:10.1016/j. jhepr.2022.100550 PMID:36111068
- Goosey E, Harrad S (2011). Perfluoroalkyl compounds in dust from Asian, Australian, European, and North American homes and UK cars, classrooms, and offices. *Environ Int.* 37(1):86–92. doi:<u>10.1016/j. envint.2010.08.001</u> PMID:<u>20810169</u>
- Goosey E, Harrad S (2012). Perfluoroalkyl substances in UK indoor and outdoor air: spatial and seasonal variation, and implications for human exposure. *Environ Int*. 45:86–90. doi:<u>10.1016/j.envint.2012.04.007</u> PMID:<u>22580294</u>
- Goss K-U (2008). The pKa values of PFOA and other highly fluorinated carboxylic acids. *Environ Sci Technol*. 42(2):456–8. doi:10.1021/es702192c PMID:18284146
- Goudarzi H, Miyashita C, Okada E, Kashino I, Chen CJ, Ito S, et al. (2017). Prenatal exposure to perfluoroalkyl acids and prevalence of infectious diseases up to 4 years of age. *Environ Int.* 104:132–8. doi:10.1016/j. envint.2017.01.024 PMID:28392064

- Government of Canada 2023 Canadian biomonitoring dashboard. Ottawa (ON), Canada: Government of Canada. Available from: <u>https://health-infobase.</u> <u>canada.ca/biomonitoring/</u>, accessed October 2024.
- Graber JM, Black TM, Shah NN, Caban-Martinez AJ, Lu SE, Brancard T, et al. (2021). Prevalence and predictors of per- and polyfluoroalkyl substances (PFAS) serum levels among members of a suburban US volunteer fire department. *Int J Environ Res Public Health*. 18(7):3730. doi:10.3390/ijerph18073730 PMID:33918459
- Grandjean P, Andersen EW, Budtz-Jørgensen E, Nielsen F, Mølbak K, Weihe P, et al. (2012). Serum vaccine antibody concentrations in children exposed to perfluorinated compounds. *JAMA*. 307(4):391–7. doi:10.1001/ jama.2011.2034 PMID:22274686
- Guelfo JL, Adamson DT (2018). Evaluation of a national data set for insights into sources, composition, and concentrations of per- and polyfluoroalkyl substances (PFASs) in US drinking water. *Environ Pollut*. 236:505–13. doi:10.1016/j.envpol.2018.01.066 PMID:29427949
- Guo B, Zeng J, Brusseau ML (2020). A mathematical model for the release, transport, and retention of per- and polyfluoroalkyl substances (PFAS) in the vadose zone. *Water Resour Res.* 56(2):e2019WR026667. doi:10.1029/2019WR026667 PMID:33223573
- Guo M, Lyu Y, Xu T, Yao B, Song W, Li M, et al. (2018). Particle size distribution and respiratory deposition estimates of airborne perfluoroalkyl acids during the haze period in the megacity of Shanghai. *Environ Pollut.* 234:9–19. doi:10.1016/j.envpol.2017.10.128 PMID:29154207
- Guo R, Megson D, Myers AL, Helm PA, Marvin C, Crozier P, et al. (2016). Application of a comprehensive extraction technique for the determination of poly- and perfluoroalkyl substances (PFASs) in Great Lakes Region sediments. *Chemosphere*. 164:535–46. doi:10.1016/j. chemosphere.2016.08.126 PMID:27619064
- Guruge KS, Taniyasu S, Yamashita N, Wijeratna S, Mohotti KM, Seneviratne HR, et al. (2005). Perfluorinated organic compounds in human blood serum and seminal plasma: a study of urban and rural tea worker populations in Sri Lanka. *J Environ Monit*. 7(4):371–7. doi:10.1039/b412532k PMID:15798805
- Hall SM, Patton S, Petreas M, Zhang S, Phillips AL, Hoffman K, et al. (2020). Per- and polyfluoroalkyl substances in dust collected from residential homes and fire stations in North America. *Environ Sci Technol.* 54(22):14558–67. doi:10.1021/acs.est.0c04869 PMID:33143410
- Hansen KJ, Clemen LA, Ellefson ME, Johnson HO (2001).
 Compound-specific, quantitative characterization of organic fluorochemicals in biological matrices. *Environ Sci Technol.* 35(4):766–70. doi:<u>10.1021/es001489z</u> PMID:<u>11349290</u>

- Harada K, Koizumi A, Saito N, Inoue K, Yoshinaga T, Date C, et al. (2007). Historical and geographical aspects of the increasing perfluorooctanoate and perfluorooctane sulfonate contamination in human serum in Japan. *Chemosphere*. 66(2):293–301. doi:10.1016/j. chemosphere.2006.05.010 PMID:16793116
- Hardell E, Kärrman A, van Bavel B, Bao J, Carlberg M, Hardell L (2014). Case-control study on perfluorinated alkyl acids (PFAAs) and the risk of prostate cancer. *Environ Int.* 63:35–9. doi:<u>10.1016/j.envint.2013.10.005</u> PMID:<u>24246240</u>
- Harrad S, Wemken N, Drage DS, Abdallah MA, Coggins AM (2019). Perfluoroalkyl substances in drinking water, indoor air and dust from Ireland: implications for human exposure. *Environ Sci Technol*. 53(22):13449–57. doi:10.1021/acs.est.9b04604 PMID:31702898
- Harris KJ, Munoz G, Woo V, Sauvé S, Rand AA (2022). Targeted and suspect screening of per- and polyfluoroalkyl substances in cosmetics and personal care products. *Environ Sci Technol*. 56(20):14594–604. doi:10.1021/acs.est.2c02660 PMID:36178710
- Haug LS, Huber S, Schlabach M, Becher G, Thomsen C (2011). Investigation on per- and polyfluorinated compounds in paired samples of house dust and indoor air from Norwegian homes. *Environ Sci Technol.* 45(19):7991–8. doi:10.1021/es103456h PMID:21417377
- Haug LS, Thomsen C, Becher G (2009). Time trends and the influence of age and gender on serum concentrations of perfluorinated compounds in archived human samples. *Environ Sci Technol.* 43(6):2131–6. doi:<u>10.1021/</u> <u>es802827u</u> PMID:<u>19368225</u>
- HBM4EU (2023). European human biomonitoring dashboard. Copenhagen, Denmark: Human Biomonitoring for Europe, European Environment Agency. Available from: <u>https://hbm.vito.be/eu-hbm-dashboard</u>, accessed October 2024.
- Health Canada (2018a). Guidelines for Canadian drinking water quality: guideline technical document – perfluorooctane sulfonate (PFOS). Catalogue No. H144-13/9-2018E-PDF. Ottawa (ON), Canada: Water and Air Quality Bureau, Healthy Environments and Consumer Safety Branch, Health Canada. Available from: <u>https://www.canada.ca/en/health-canada/services/</u> publications/healthy-living/guidelines-canadiandrinking-water-quality-guideline-technicaldocument-perfluorooctane-sulfonate/document.html, accessed October 2024.
- Health Canada (2018b). Objective for Canadian drinking water quality - per- and polyfluoroalkyl substances. Catalogue No. H144-13/8-2018E-PDF. Ottawa (ON), Canada: Water and Air Quality Bureau, Healthy Environments and Consumer Safety Branch, Health Canada. Available from: <u>https://www.canada.ca/ en/health-canada/services/publications/healthyliving/guidelines-canadian-drinking-water-quality-</u>

technical-document-perfluorooctanoic-acid/ document.html, accessed October 2024.

- Health Canada (2019). Updates to Health Canada soil screening values for perfluoroalkylated substances (PFAS). Ottawa (ON), Canada: Health Canada. Available from: <u>https://buyandsell.gc.ca/cds/public /2020/06/19/1e18568783a9ee9bddc91262c6d7679a/attachment_007.pdf</u>, accessed October 2024.
- Hepburn E, Madden C, Szabo D, Coggan TL, Clarke B, Currell M (2019). Contamination of groundwater with per- and polyfluoroalkyl substances (PFAS) from legacy landfills in an urban re-development precinct. *Environ Pollut*. 248:101–13. doi:<u>10.1016/j.envpol.2019.02.018</u> PMID:<u>30784829</u>
- Herrick RL, Buckholz J, Biro FM, Calafat AM, Ye X, Xie C, et al. (2017). Polyfluoroalkyl substance exposure in the Mid-Ohio River Valley, 1991–2012. *Environ Pollut.* 228:50–60. doi:<u>10.1016/j.envpol.2017.04.092</u> PMID:<u>28505513</u>
- Herzke D, Olsson E, Posner S (2012). Perfluoroalkyl and polyfluoroalkyl substances (PFASs) in consumer products in Norway – a pilot study. *Chemosphere*. 88(8):980–7. doi:10.1016/j.chemosphere.2012.03.035 PMID:22483730
- Heydebreck F, Tang J, Xie Z, Ebinghaus R (2016). Emissions of per- and polyfluoroalkyl substances in a textile manufacturing plant in China and their relevance for workers' exposure. *Environ Sci Technol*. 50(19):10386– 96. doi:10.1021/acs.est.6b03213 PMID:27617679
- Higgins CP, Field JA, Criddle CS, Luthy RG (2005). Quantitative determination of perfluorochemicals in sediments and domestic sludge. *Environ Sci Technol*. 39(11):3946–56.doi:10.1021/es048245pPMID:15984769
- Hoffman K, Webster TF, Bartell SM, Weisskopf MG, Fletcher T, Vieira VM (2011). Private drinking water wells as a source of exposure to perfluorooctanoic acid (PFOA) in communities surrounding a fluoropolymer production facility. *Environ Health Perspect*. 119(1):92–7. doi:10.1289/ehp.1002503 PMID:20920951
- Holmström K, Wetterstrand S, Hedenberg G (2014). National screening of perfluorinated pollutants (PFAA) in drinking water. Report No. 2014-20. Bromma, Sweden: Svenskt Vatten Utveckling. Available from: <u>https://vav.griffel.net/filer/SVU-rapport_2014-20.pdf</u>, accessed October 2024. [Swedish]
- Hölzer J, Göen T, Just P, Reupert R, Rauchfuss K, Kraft M, et al. (2011). Perfluorinated compounds in fish and blood of anglers at Lake Möhne, Sauerland area, Germany. *Environ Sci Technol.* 45(19):8046–52. doi:10.1021/es104391z PMID:21534594
- Hölzer J, Lilienthal H, Schümann M (2021). Human biomonitoring (HBM)-I values for perfluorooctanoic acid (PFOA) and perfluorooctane sulfonic acid (PFOS) description, derivation and discussion. *Regul Toxicol Pharmacol.* 121:104862. doi:10.1016/j. yrtph.2021.104862 PMID:33444659

- Hölzer J, Midasch O, Rauchfuss K, Kraft M, Reupert R, Angerer J, et al. (2008). Biomonitoring of perfluorinated compounds in children and adults exposed to perfluorooctanoate-contaminated drinking water. Environ Health Perspect. 116(5):651–7. doi:10.1289/ ehp.11064 PMID:18470314
- Hu XC, Andrews DQ, Lindstrom AB, Bruton TA, Schaider LA, Grandjean P, et al. (2016). Detection of poly- and perfluoroalkyl substances (PFASs) in US drinking water linked to industrial sites, military fire training areas, and wastewater treatment plants. *Environ Sci Technol Lett.* 3(10):344–50. doi:<u>10.1021/acs.</u> estlett.6b00260 PMID:27752509
- Hunter Anderson R, Adamson DT, Stroo HF (2019). Partitioning of poly- and perfluoroalkyl substances from soil to groundwater within aqueous film-forming foam source zones. *J Contam Hydrol*. 220:59–65. doi:10.1016/j.jconhyd.2018.11.011 PMID:30527585
- Hurley S, Goldberg D, Wang M, Park JS, Petreas M, Bernstein L, et al. (2018). Breast cancer risk and serum levels of per- and poly-fluoroalkyl substances: a casecontrol study nested in the California Teachers Study. *Environ Health*. 17(1):83. doi:10.1186/s12940-018-0426-6 PMID:30482205
- Husøy T, Caspersen IH, Thépaut E, Knutsen H, Haug LS, Andreassen M, et al. (2023). Comparison of aggregated exposure to perfluorooctanoic acid (PFOA) from diet and personal care products with concentrations in blood using a PBPK model – results from the Norwegian biomonitoring study in EuroMix. *Environ Res.* 239(Pt 2):117341. doi:10.1016/j.envres.2023.117341 PMID:37839534
- IARC (1979). Some halogenated hydrocarbons. IARC Monogr Eval Carcinog Risk Chem Hum. 20:1–609.
 Available from: <u>https://publications.iarc.who.int/38</u> PMID:296120
- IARC (1987). Overall evaluations of carcinogenicity: an updating of IARC Monographs volumes 1 to 42. IARC Monogr Eval Carcinog Risks Hum Suppl. 7:1-440. Available from: <u>https://publications.iarc.who.int/139</u> PMID:3482203
- IARC (1993). Beryllium, cadmium, mercury, and exposures in the glass manufacturing industry. *IARC Monogr Eval Carcinog Risks Hum.* 58:1–444. Available from: https://publications.iarc.who.int/76 PMID:8022054
- IARC (1997). Polychlorinated dibenzo-para-dioxins and polychlorinated dibenzofurans. IARC Monogr Eval Carcinog Risks Hum. 69:1–666. Available from: <u>https://</u> publications.iarc.who.int/87 PMID:9379504
- IARC (2001). Some thyrotropic agents. IARC Monogr Eval Carcinog Risks Hum. 79:1–763. Available from: <u>https://</u> <u>publications.iarc.who.int/97</u>
- IARC (2006). Inorganic and organic lead compounds. *IARC Monogr Eval Carcinog Risks Hum.* 87:1–506. Available from: <u>https://publications.iarc.who.int/105</u> PMID:<u>17191367</u>

- IARC (2012). Arsenic, metals, fibres, and dusts. IARC Monogr Eval Carcinog Risks Hum. 100C:1–501. Available from: <u>https://publications.iarc.who.int/120</u> PMID:23189751
- IARC (2015). Polychlorinated biphenyls and polybrominated biphenyls. *IARC Monogr Eval Carcinog Risks Hum.* 107:1–502. Available from: <u>https://publications.</u> <u>iarc.who.int/131</u> PMID:<u>29905442</u>
- IARC (2016). Some chemicals used as solvents and in polymer manufacture. *IARC Monogr Eval Carcinog Risks Hum*, 110:1–276. Available from: <u>https://</u> <u>publications.iarc.who.int/547</u> PMID:<u>31829531</u>
- IARC (2017). DDT, lindane, and 2,4-D. IARC Monogr Eval Carcinog Risks Hum. 113:1–503. Available from: <u>https://publications.iarc.who.int/550</u> PMID:29912510
- IFA (2022). GESTIS International limit values for chemical agents [online database]. Sankt Augustin, Germany: Institut für Arbeitsschutz der Deutschen Gesetzlichen Unfallversicherung (Institute for Occupational Safety and Health of the German Social Accident Insurance). Available from: <u>https://limitvalue.ifa.dguv.de/Web</u> Form _gw2.aspx, accessed March 2023.
- Ingelido AM, Abballe A, Gemma S, Dellatte E, Iacovella N, De Angelis G, et al. (2018). Biomonitoring of perfluorinated compounds in adults exposed to contaminated drinking water in the Veneto Region, Italy. *Environ Int.* 110:149–59. doi:10.1016/j.envint.2017.10.026 PMID:29108835
- Ingelido AM, Marra V, Abballe A, Valentini S, Iacovella N, Barbieri P, et al. (2010). Perfluorooctanesulfonate and perfluorooctanoic acid exposures of the Italian general population. *Chemosphere*.80(10):1125–30. doi:<u>10.1016/j.</u> <u>chemosphere.2010.06.025</u> PMID:<u>20633921</u>
- ISO (2009). ISO water quality. Determination of perfluorooctanesulfonate (PFOS) and perfluorooctanoate (PFOA). Method for unfiltered samples using solid phase extraction and liquid chromatography/mass spectrometry, ISO 25101:2009. Geneva, Switzerland: International Organization for Standardization. Available from: <u>https://www.iso.org/standard/42742.</u> <u>html</u>.
- ISO (2019). ISO 21675, 2019. Water quality determination of perfluoroalkyl and polyfluoroalkyl substances (PFAS) in water – method using solid phase extraction and liquid chromatography-tandem mass spectrometry (LC–MS/MS). Geneva, Switzerland: International Organization for Standardization. <u>Available from:</u> <u>https://www.iso.org/standard/71338.html</u>.
- Itoh H, Harada KH, Kasuga Y, Yokoyama S, Onuma H, Nishimura H, et al. (2021). Serum perfluoroalkyl substances and breast cancer risk in Japanese women: a case-control study. *Sci Total Environ*. 800:149316. doi:<u>10.1016/j.scitotenv.2021.149316</u> PMID:<u>34392213</u>
- ITRC (2022a). PFAS technical and regulatory guidance document and fact sheets. PFAS-1. Section 2 – naming conventions and use. Washington (DC), USA:

Interstate Technology & Regulatory Council, PFAS Team. Available from: <u>https://pfas-1.itrcweb.org/</u>, accessed November 2023.

- ITRC (2022b). PFAS technical and regulatory guidance document and fact sheets. PFAS-1. Section 3 – firefighting foams. Washington (DC), USA: Interstate Technology & Regulatory Council, PFAS Team. Available from: <u>https://pfas-1.itrcweb.org/</u>, accessed November 2023.
- ITRC (2022c). PFAS per- and polyfluoroalkyl substances. 2.2 Chemistry, terminology, and acronyms. Washington (DC), USA: Interstate Technology & Regulatory Council, PFAS Team. Available from: <u>https://pfas-1.itrcweb.org/2-2-chemistry-terminology-</u> and-acronyms/, accessed November 2023.
- ITRC (2023a). PFAS technical and regulatory guidance document and fact sheets. PFAS-1. Section 8 basis of regulations. Washington (DC), USA: Interstate Technology & Regulatory Council, PFAS Team. Available from: <u>https://pfas-1.itrcweb.org/8-basis-of-regulations/#8_2</u>, accessed November 2023.
- ITRC (2023b). PFAS technical and regulatory guidance document and fact sheets. PFAS-1. Section 8 – basis of regulations, PFAS regulatory programs summary table. Washington (DC), USA: Interstate Technology & Regulatory Council, PFAS Team. Available in list of external data tables from: <u>https://pfas-1.itrcweb.org/</u>, accessed November 2023.
- ITRC (2023c). PFAS water and soil values table excel file. Washington (DC), USA: Interstate Technology & Regulatory Council, PFAS Team. Available from: <u>https://pfas-1.itrcweb.org/fact-sheets/</u>, accessed November 2023.
- Jahnke A, Huber S, Temme C, Kylin H, Berger U (2007). Development and application of a simplified sampling method for volatile polyfluorinated alkyl substances in indoor and environmental air. *J Chromatogr A*. 1164(1–2):1–9. doi:<u>10.1016/j.chroma.2007.06.068</u> PMID: <u>17659294</u>
- Janda J, Nödler K, Brauch HJ, Zwiener C, Lange FT (2019). Robust trace analysis of polar (C₂-C₈) perfluorinated carboxylic acids by liquid chromatography-tandem mass spectrometry: method development and application to surface water, groundwater and drinking water. *Environ Sci Pollut Res Int.* 26(8):7326–36. doi:10.1007/ s11356-018-1731-x PMID:29557039
- Janousek RM, Lebertz S, Knepper TP (2019) Previously unidentified sources of perfluoroalkyl and polyfluoroalkyl substances from building materials and industrial fabrics. *Environ Sci Process Impacts*. 21(11):1936–45. doi:10.1021/acs.est.2c07185 PMID:36940151
- Jeong YJ, Bang S, Kim J, Chun SH, Choi S, Kim J, et al. (2019). Comparing levels of perfluorinated compounds in processed marine products. *Food Chem Toxicol*. 126:199–210. doi:10.1016/j.fct.2019.01.028 PMID:30797874

- Jian JM, Guo Y, Zeng L, Liang-Ying L, Lu X, Wang F, et al. (2017). Global distribution of perfluorochemicals (PFCs) in potential human exposure source a review. *Environ Int.* 108:51–62. doi:<u>10.1016/j.envint.2017.07.024</u> PMID:<u>28800414</u>
- Jin H, Lin S, Dai W, Feng L, Li T, Lou J, et al. (2020). Exposure sources of perfluoroalkyl acids and influence of age and gender on concentrations of chlorinated polyfluorinated ether sulfonates in human serum from China. *Environ Int.* 138:105651. doi:10.1016/j. envint.2020.105651 PMID:32200315
- Jin YH, Liu X, Zhang X, et al. (2003). [Perfluorooctane sulfonate situation in serum of common population.] *Chin J Publ Health*. 19:1200–1. Available from: <u>https://www.zgggws.com/en/article/id/17273</u>. [Chinese]
- Jogsten IE, Perelló G, Llebaria X, Bigas E, Martí-Cid R, Kärrman A, et al. (2009). Exposure to perfluorinated compounds in Catalonia, Spain, through consumption of various raw and cooked foodstuffs, including packaged food. *Food Chem Toxicol.* 47(7):1577–83. doi:10.1016/j.fct.2009.04.004 PMID:19362113
- Johnson GR, Brusseau ML, Carroll KC, Tick GR, Duncan CM (2022). Global distributions, source-type dependencies, and concentration ranges of per- and polyfluoroalkyl substances in groundwater. *Sci Total Environ*. 841:156602. doi:<u>10.1016/j.scitotenv.2022.156602</u> PMID: <u>35690215</u>
- Jones W, Harden M (2022). Manual extraction of PFAS in drinking water in compliance with ISO 21675. Application note AN967. Uppsala, Sweden: Biotage. Available from: <u>https://lcms.cz/</u> <u>labrulez-bucket-strapi-h3hsga3/AN 967 PFAS in</u> <u>drinking water ISO 21675 377314d46c/AN967-PFAS-in-drinking-water-ISO-21675.pdf</u>, accessed October 2024.
- Kadar H, Veyrand B, Barbarossa A, Pagliuca G, Legrand A, Bosher C, et al. (2011). Development of an analytical strategy based on liquid chromatographyhigh resolution mass spectrometry for measuring perfluorinated compounds in human breast milk: application to the generation of preliminary data regarding perinatal exposure in France. *Chemosphere*. 85(3):473–80. doi:10.1016/j.chemosphere.2011.07.077 PMID:21880346
- Kaiser MA, Dawson BJ, Barton CA, Botelho MA (2010). Understanding potential exposure sources of perfluorinated carboxylic acids in the workplace. *Ann Occup Hyg.* 54(8):915–22. doi:<u>10.1093/annhyg/meq066</u> PMID: <u>20974675</u>
- Kärrman A, Ericson I, van Bavel B, Darnerud PO, Aune M, Glynn A, et al. (2007). Exposure of perfluorinated chemicals through lactation: levels of matched human milk and serum and a temporal trend, 1996–2004, in Sweden. *Environ Health Perspect*. 115(2):226–30. doi:10.1289/ehp.9491 PMID:17384769

- Kato K, Kalathil AA, Patel AM, Ye X, Calafat AM (2018). Per- and polyfluoroalkyl substances and fluorinated alternatives in urine and serum by on-line solid phase extraction-liquid chromatography-tandem mass spectrometry. *Chemosphere*. 209:338–45. doi:10.1016/j. chemosphere.2018.06.085 PMID:29935462
- Kato K, Wong LY, Jia LT, Kuklenyik Z, Calafat AM (2011). Trends in exposure to polyfluoroalkyl chemicals in the US population: 1999–2008. *Environ Sci Technol.* 45(19):8037–45. doi:10.1021/es1043613 PMID:21469664
- Kenny S (2021). Emerging issues in food waste management: persistent chemicals. Washington (DC), USA: Office of Research and Development, United States Environmental Protection Agency.
- Kikuchi J, Wiberg K, Stendahl J, Ahrens L (2018). Analysis of per- and polyfluoroalkyl substances (PFASs) in soil from Swedish background sites. Swedish University of Agricultural Sciences, Department of Aquatic Sciences and Assessment. Available from: <u>https://pub.epsilon.</u> <u>slu.se/15902/7/kikuchi j et al 190211.pdf</u>, accessed October 2024.
- Kim HY, Kim KN, Shin CH, Lim YH, Kim JI, Kim BN, et al. (2020). The relationship between perfluoroalkyl substances concentrations and thyroid function in early childhood: a prospective cohort study. *Thyroid.* 30(11):1556–65. doi:10.1089/thy.2019.0436 PMID:32368952
- Kim JH, Park HY, Jeon JD, Kho Y, Kim SK, Park MS, et al. (2016). The modifying effect of vitamin C on the association between perfluorinated compounds and insulin resistance in the Korean elderly: a double-blind, randomized, placebo-controlled crossover trial. *Eur J Nutr.* 55(3):1011–20. doi:10.1007/s00394-015-0915-0 PMID:25939797
- Knox SS, Jackson T, Javins B, Frisbee SJ, Shankar A, Ducatman AM (2011). Implications of early menopause in women exposed to perfluorocarbons. *J Clin Endocrinol Metab.* 96(6):1747–53. doi:<u>10.1210/jc.2010-2401 PMID:21411548</u>
- Kolanczyk RC, Saley MR, Serrano JA, Daley SM, Tapper MA (2023). PFAS biotransformation pathways: a species comparison study. *Toxics*. 11(1):74. doi:<u>10.3390/ toxics11010074</u> PMID:<u>36668800</u>
- Kotthoff M, Müller J, Jürling H, Schlummer M, Fiedler D (2015). Perfluoroalkyl and polyfluoroalkyl substances in consumer products. *Environ Sci Pollut Res Int*. 22(19):14546–59. doi:<u>10.1007/s11356-015-4202-7</u> PMID: 25854201
- Kourtchev I, Hellebust S, Heffernan E, Wenger J, Towers S, Diapouli E, et al. (2022). A new on-line SPE LC–HRMS method for the analysis of perfluoroalkyl and polyfluoroalkyl substances (PFAS) in PM_{2.5} and its application for screening atmospheric particulates from Dublin and Enniscorthy, Ireland. *Sci Total Environ*. 835:155496. doi:10.1016/j.scitotenv.2022.155496 PMID: 35483471

- Kubwabo C, Stewart B, Zhu J, Marro L (2005). Occurrence of perfluorosulfonates and other perfluorochemicals in dust from selected homes in the city of Ottawa, Canada. *J Environ Monit.* 7(11):1074–8. doi:10.1039/b507731c PMID:16252056
- Kuklenyik Z, Needham LL, Calafat AM (2005). Measurement of 18 perfluorinated organic acids and amides in human serum using on-line solid-phase extraction. *Anal Chem.* 77(18):6085–91. doi:10.1021/ ac0506711 PMID:16159145
- Kuklenyik Z, Reich JA, Tully JS, Needham LL, Calafat AM (2004). Automated solid-phase extraction and measurement of perfluorinated organic acids and amides in human serum and milk. *Environ Sci Technol.* 38(13):3698–704. doi:<u>10.1021/es040332u</u> PMID:<u>15296323</u>
- Kurwadkar S, Dane J, Kanel SR, Nadagouda MN, Cawdrey RW, Ambade B, et al. (2022). Per- and polyfluoroalkyl substances in water and wastewater: a critical review of their global occurrence and distribution. *Sci Total Environ.* 809:151003. doi:10.1016/j. scitotenv.2021.151003 PMID:34695467
- Kvalem HE, Nygaard UC, Lødrup Carlsen KC, Carlsen KH, Haug LS, Granum B (2020). Perfluoroalkyl substances, airways infections, allergy and asthma related health outcomes implications of gender, exposure period and study design. *Environ Int.* 134:105259. doi:10.1016/j.envint.2019.105259 PMID:31733527
- Kwok KY, Yamazaki E, Yamashita N, Taniyasu S, Murphy MB, Horii Y, et al. (2013). Transport of perfluoroalkyl substances (PFAS) from an arctic glacier to downstream locations: implications for sources. *Sci Total Environ*. 447:46–55. doi:<u>10.1016/j.scitotenv.2012.10.091</u> PMID:<u>23376515</u>
- Lacina O, Hradkova P, Pulkrabova J, Hajslova J (2011). Simple, high throughput ultra-high performance liquid chromatography/tandem mass spectrometry trace analysis of perfluorinated alkylated substances in food of animal origin: milk and fish. *J Chromatogr A*. 1218(28):4312–21. doi:<u>10.1016/j.chroma.2011.04.061</u> PMID:<u>21621213</u>
- Land M, de Wit CA, Bignert A, Cousins Ian T, Herzke D, Johansson JH, et al. (2018). What is the effect of phasing out long-chain per- and polyfluoroalkyl substances on the concentrations of perfluoroalkyl acids and their precursors in the environment? A systematic review. *Environ Evid*. 7(1):4. doi:10.1186/s13750-017-0114-y
- Langberg HA, Hale SE, Breedveld GD, Jenssen BM, Jartun M (2022). A review of PFAS fingerprints in fish from Norwegian freshwater bodies subject to different source inputs. *Environ Sci Process Impacts*. 24(2):330–42. doi:10.1039/D1EM00408E PMID:35079763
- Langer V, Dreyer A, Ebinghaus R (2010). Polyfluorinated compounds in residential and nonresidential indoor air. *Environ Sci Technol.* 44(21):8075–81. doi:<u>10.1021/es102384z</u> PMID:<u>20925396</u>

- Langlois I, Oehme M (2006). Structural identification of isomers present in technical perfluorooctane sulfonate by tandem mass spectrometry. *Rapid Commun Mass Spectrom*. 20(5):844–50. doi:<u>10.1002/rcm.2383</u> PMID:16470576
- Lankova D, Lacina O, Pulkrabova J, Hajslova J (2013). The determination of perfluoroalkyl substances, brominated flame retardants and their metabolites in human breast milk and infant formula. *Talanta*. 117:318–25. doi:10.1016/j.talanta.2013.08.040 PMID:24209347
- Lasters R, Groffen T, Eens M, Coertjens D, Gebbink WA, Hofman J, et al. (2022). Home-produced eggs: an important human exposure pathway of perfluoroalkylated substances (PFAS). *Chemosphere*. 308(Pt 1):136283. doi:10.1016/j.chemosphere.2022.136283 PMID:36075366
- Leary DB, Takazawa M, Kannan K, Khalil N (2020). Perfluoroalkyl substances and metabolic syndrome in firefighters: a pilot study. *J Occup Environ Med.* 62(1):52–7. doi:<u>10.1097/JOM.000000000001756</u> PMID:31658221
- Leeson A, Thompson T, Stroo HF, Anderson RH, Speicher J, Mills MA, et al. (2021). Identifying and managing aqueous film-forming foam-derived per- and poly-fluoroalkyl substances in the environment. *Environ Toxicol Chem.* 40(1):24–36. doi:10.1002/etc.4894 PMID:33026660
- Leonard RC, Kreckmann KH, Sakr CJ, Symons JM (2008). Retrospective cohort mortality study of workers in a polymer production plant including a reference population of regional workers. *Ann Epidemiol.* 18(1):15–22. doi:10.1016/j.annepidem.2007.06.011 PMID:17900928
- Levasseur JL, Hoffman K, Herkert NJ, Cooper E, Hay D, Stapleton HM (2022). Characterizing firefighter's exposure to over 130 SVOCs using silicone wristbands: a pilot study comparing on-duty and off-duty exposures. *Sci Total Environ*. 834:155237. doi:10.1016/j. scitotenv.2022.155237 PMID:35447169
- Li F, Zhang C, Qu Y, Chen J, Chen L, Liu Y, et al. (2010). Quantitative characterization of short- and long-chain perfluorinated acids in solid matrices in Shanghai, China. *Sci Total Environ*. 408(3):617–23. doi:<u>10.1016/j.</u> <u>scitotenv.2009.10.032</u> PMID:<u>19896166</u>
- Li H, Hammarstrand S, Midberg B, Xu Y, Li Y, Olsson DS, et al. (2022b). Cancer incidence in a Swedish cohort with high exposure to perfluoroalkyl substances in drinking water. *Environ Res.* 204(Pt C):112217. doi:10.1016/j. envres.2021.112217 PMID:34662573
- Li L, Zhai Z, Liu J, Hu J (2015). Estimating industrial and domestic environmental releases of perfluorooctanoic acid and its salts in China from 2004 to 2012. *Chemosphere*. 129:100–9. doi:10.1016/j. <u>chemosphere.2014.11.049</u> PMID:25541533
- Li X, Song F, Liu X, Shan A, Huang Y, Yang Z, et al. (2022a). Perfluoroalkyl substances (PFASs) as risk factors for breast cancer: a case-control study in Chinese

population. *Environ Health*. 21(1):83. doi:<u>10.1186/</u> <u>\$12940-022-00895-3</u> PMID:<u>36085159</u>

- Li Y, Andersson A, Xu Y, Pineda D, Nilsson CA, Lindh CH, et al. (2022c). Determinants of serum half-lives for linear and branched perfluoroalkyl substances after long-term high exposure a study in Ronneby, Sweden. *Environ Int.* 163:107198. doi:10.1016/j. envint.2022.107198 PMID:35447437
- Li Y, Fletcher T, Mucs D, Scott K, Lindh CH, Tallving P, et al. (2018). Half-lives of PFOS, PFHxS and PFOA after end of exposure to contaminated drinking water. *Occup Environ Med.* 75(1):46–51. doi:<u>10.1136/oemed-2017-104651</u> PMID:<u>29133598</u>
- Lin CY, Chen PC, Lo SC, Torng PL, Sung FC, Su TC (2016). The association of carotid intima-media thickness with serum level of perfluorinated chemicals and endothelium-platelet microparticles in adolescents and young adults. *Environ Int*. 94:292–9. doi:10.1016/j. envint.2016.06.004 PMID:27288966
- Lin CY, Lee HL, Hwang YT, Su TC (2020c). The association between total serum isomers of per- and polyfluoroalkyl substances, lipid profiles, and the DNA oxidative/ nitrative stress biomarkers in middle-aged Taiwanese adults. *Environ Res.* 182:109064. doi:10.1016/j. envres.2019.109064 PMID:31884197
- Lin H, Taniyasu S, Yamashita N, Khan MK, Masood SS, Saied S, et al. (2022). Per- and polyfluoroalkyl substances in the atmospheric total suspended particles in Karachi, Pakistan: profiles, potential sources, and daily intake estimates. *Chemosphere*. 288(Pt2):132432. doi:10.1016/j. chemosphere.2021.132432 PMID:34606903
- Lin H, Taniyasu S, Yamazaki E, Wei S, Wang X, Gai N, et al. (2020a). Per- and polyfluoroalkyl substances in the air particles of Asia: levels, seasonality, and size-dependent distribution. *Environ Sci Technol*. 54(22):14182–91. doi:<u>10.1021/acs.est.0c03387</u> PMID:<u>33156616</u>
- Lin HW, Feng HX, Chen L, Yuan XJ, Tan Z (2020b). Maternal exposure to environmental endocrine disruptors during pregnancy is associated with pediatric germ cell tumors. *Nagoya J Med Sci*. 82(2):323–33. doi:<u>10.18999%2Fnagjms.82.2.323</u> PMID:<u>32581411</u>
- Liu H, Chen Q, Lei L, Zhou W, Huang L, Zhang J, et al. (2018b). Prenatal exposure to perfluoroalkyl and polyfluoroalkyl substances affects leukocyte telomere length in female newborns. *Environ Pollut*. 235:446–52. doi:10.1016/j.envpol.2017.12.095 PMID:29310088
- Liu J, Li J, Liu Y, Chan HM, Zhao Y, Cai Z, et al. (2011). Comparison on gestation and lactation exposure of perfluorinated compounds for newborns. *Environ Int.* 37(7):1206–12. doi:<u>10.1016/j.envint.2011.05.001</u> PMID:<u>21620474</u>
- Liu J, Li J, Zhao Y, Wang Y, Zhang L, Wu Y (2010). The occurrence of perfluorinated alkyl compounds in human milk from different regions of China. *Environ Int.* 36(5):433–8. doi:<u>10.1016/j.envint.2010.03.004</u> PMID:<u>20400180</u>

- Liu LS, Guo YT, Wu QZ, Zeeshan M, Qin SJ, Zeng HX, et al. (2023). Per- and polyfluoroalkyl substances in ambient fine particulate matter in the Pearl River Delta, China: levels, distribution and health implications. *Environ Pollut.* 334:122138. doi:<u>10.1016/j.envpol.2023.122138</u> PMID:<u>37453686</u>
- Liu M, Zhang G, Meng L, Han X, Li Y, Shi Y, et al. (2022a). Associations between novel and legacy per- and polyfluoroalkyl substances in human serum and thyroid cancer: a case and healthy population in Shandong Province, east China. *Environ Sci Technol.* 56(10):6144– 51. doi:10.1021/acs.est.1c02850 PMID:34618433
- Liu W, He W, Wu J, Wu W, Xu F (2018a). Distribution, partitioning and inhalation exposure of perfluoroalkyl acids (PFAAs) in urban and rural air near Lake Chaohu, China. *Environ Pollut*. 243(Pt A):143–51. doi:<u>10.1016/j. envpol.2018.08.052</u> PMID:<u>30172120</u>
- Liu Y, Eliot MN, Papandonatos GD, Kelsey KT, Fore R, Langevin S, et al. (2022b). Gestational perfluoroalkyl substance exposure and DNA methylation at birth and 12 years of age: a longitudinal epigenome-wide association study. *Environ Health Perspect*. 130(3):37005. doi:10.1289/EHP10118 PMID:35266797
- Liu Y, Li X, Wang X, Qiao X, Hao S, Lu J, et al. (2019). Contamination profiles of perfluoroalkyl substances (PFAS) in groundwater in the alluvial–pluvial plain of Hutuo River, China. *Water*. 11(11):1–2316. doi:<u>10.3390/</u> <u>w11112316</u> PMID:<u>32021704</u>
- Liu Z, Lu Y, Shi Y, Wang P, Jones K, Sweetman AJ, et al. (2017). Crop bioaccumulation and human exposure of perfluoroalkyl acids through multi-media transport from a mega fluorochemical industrial park, China. *Environ Int.* 106:37–47. doi:10.1016/j.envint.2017.05.014 PMID:28558301
- Llorca M, Farré M, Tavano MS, Alonso B, Koremblit G, Barceló D (2012a). Fate of a broad spectrum of perfluorinated compounds in soils and biota from Tierra del Fuego and Antarctica. *Environ Pollut*. 163:158–66. doi:<u>10.1016/j.envpol.2011.10.027</u> PMID:<u>22325444</u>
- Löfstedt Gilljam J, Leonel J, Cousins IT, Benskin JP (2016). Is ongoing sulfluramid use in South America a significant source of perfluorooctanesulfonate (PFOS)? Production inventories, environmental fate, and local occurrence. *Environ Sci Technol.* 50(2):653–9. doi:10.1021/acs.est.5b04544 PMID:26653085
- Longnecker MP, Smith CS, Kissling GE, Hoppin JA, Butenhoff JL, Decker E, et al. (2008). An interlaboratory study of perfluorinated alkyl compound levels in human plasma. *Environ Res.* 107(2):152–9. doi:10.1016/j. envres.2008.01.005 PMID:18295197
- Lopez-Espinosa MJ, Carrizosa C, Luster MI, Margolick JB, Costa O, Leonardi GS, et al. (2021). Perfluoroalkyl substances and immune cell counts in adults from the Mid-Ohio Valley (USA). *Environ Int.* 156:106599. doi:10.1016/j.envint.2021.106599 PMID:33993002

- Lopez-Espinosa M-J, Mondal D, Armstrong BG, Eskenazi B, Fletcher T (2016). Perfluoroalkyl substances, sex hormones, and insulin-like growth factor-1 at 6–9 years of age: a cross-sectional analysis within the C8 Health Project. *Environ Health Perspect*. 124(8):1269– 75. doi:10.1289/ehp.1509869 PMID:26794451
- Lorber M, Egeghy PP (2011). Simple intake and pharmacokinetic modeling to characterize exposure of Americans to perfluoroctanoic acid, PFOA. *Environ Sci Technol.* 45(19):8006–14. doi:<u>10.1021/es103718h</u> PMID:<u>21517063</u>
- Lorenzo M, Campo J, Picó Y (2015). Optimization and comparison of several extraction methods for determining perfluoroalkyl substances in abiotic environmental solid matrices using liquid chromatography-mass spectrometry. *Anal Bioanal Chem.* 407(19):5767–81.doi:<u>10.1007/s00216-015-8759-2</u>PMID: <u>26025550</u>
- Lu C, Shi YL, Zhou Z, Liu NN, Meng ZF, Cai YQ (2014). Perfluorinated compounds in blood of textile workers and barbers. *Chin Chem Lett.* 25(8):1145–8. doi:10.1016/j.cclet.2014.03.023
- Lundin JI, Alexander BH, Olsen GW, Church TR (2009). Ammonium perfluorooctanoate production and occupational mortality. *Epidemiology*. 20(6):921–8. doi:<u>10.1097/EDE.0b013e3181b5f395</u> PMID:<u>19797969</u>
- Luque N, Ballesteros-Gómez A, van Leeuwen S, Rubio S (2012). A simple and rapid extraction method for sensitive determination of perfluoroalkyl substances in blood serum suitable for exposure evaluation. *J Chromatogr A*. 1235:84–91. doi:<u>10.1016/j.chroma.</u> 2012.02.055 PMID:22420956
- Lv L, Liu B, Zhang B, Yu Y, Gao L, Ding L (2023). A systematic review on distribution, sources and sorption of perfluoroalkyl acids (PFAAs) in soil and their plant uptake. *Environ Res.* 231(Pt 1):116156. doi:<u>10.1016/j.</u> <u>envres.2023.116156</u> PMID:<u>37196690</u>
- Machado-Neto JG, Queiroz ME, Carvalho D, Bassini AJ (1999). Risk of intoxication with sulfluramid in a packing plant of Mirex-S. *Bull Environ Contam Toxicol.* 62(5):515–9. doi:10.1007/s001289900906 PMID:10227829
- Macheka-Tendenguwo LR, Olowoyo JO, Mugivhisa LL, Abafe OA (2018). Per- and polyfluoroalkyl substances in human breast milk and current analytical methods. *Environ Sci Pollut Res Int*. 25(36):36064–86. doi:10.1007/ s11356-018-3483-z PMID:30382519
- Madrigal JM, Troisi R, Surcel HJ, Ohman H, Kivela J, Kiviranta H, et al. (2024). Prediagnostic serum concentrations of per- and polyfluoroalkyl substances and risk of papillary thyroid cancer in the Finnish Maternity Cohort. *Int J Cancer*. 154(6):979–991. doi:10.1002/ ijc.34776 PMID:37902275
- Maine DEP (2023). PFAS in products. Augusta (ME), USA: Maine Department of Environmental Protection. Available from: <u>https://www.maine.gov/dep/spills/</u>

topics/pfas/PFAS-products/#:~:text=Beginning%20 January%201%2C%202023%2C%20fabric,in%20 the%20State%20of%20Maine, accessed October 2024.

- Maizel A, Thompson A, Tighe M, Escobar Veras S, Rodowa A, Falkenstein-Smith R, et al. (2023). Perand polyfluoroalkyl substances in new firefighter turnout gear textiles. NIST technical note. NIST TN 2248. Gaithersburg (MD), USA: National Institute of Standards and Technology. Available from: <u>https:// nvlpubs.nist.gov/nistpubs/TechnicalNotes/NIST.</u> <u>TN.2248.pdf</u>, accessed October 2024.
- Mancini FR, Cano-Sancho G, Gambaretti J, Marchand P, Boutron-Ruault MC, Severi G, et al. (2020). Perfluorinated alkylated substances serum concentration and breast cancer risk: evidence from a nested case-control study in the French E3N cohort. *Int J Cancer.* 146(4):917–28. doi:10.1002/ijc.32357 PMID:31008526
- Manzano-Salgado CB, Granum B, Lopez-Espinosa MJ, Ballester F, Iñiguez C, Gascón M, et al. (2019). Prenatal exposure to perfluoroalkyl substances, immune-related outcomes, and lung function in children from a Spanish birth cohort study. *Int J Hyg Environ Health*. 222(6):945–54. doi:10.1016/j.ijheh.2019.06.005 PMID:31262703
- Martín J, Rodríguez-Gómez R, Zafra-Gómez A, Alonso E, Vílchez JL, Navalón A (2016). Validated method for the determination of perfluorinated compounds in placental tissue samples based on a simple extraction procedure followed by ultra-high performance liquid chromatography-tandem mass spectrometry analysis. *Talanta*. 150:169–76. doi:10.1016/j.talanta.2015.12.020 PMID:26838396
- Mazumder N, Hossain MT, Jahura FT, Girase A, Hall AS, Lu J, et al. (2023). Firefighters' exposure to per-and polyfluoroalkyl substances (PFAS) as an occupational hazard: a review. *Front Mater.* 10:1143411. doi:<u>10.3389/fmats.2023.1143411</u> PMID:<u>38074949</u>
- McMahon PB, Tokranov AK, Bexfield LM, Lindsey BD, Johnson TD, Lombard MA, et al. (2022). Perfluoroalkyl and polyfluoroalkyl substances in groundwater used as a source of drinking water in the eastern United States. *Environ Sci Technol*. 56(4):2279–88. doi:<u>10.1021/acs.</u> <u>est.1c04795</u> PMID:<u>35113548</u>
- McMurdo CJ, Ellis DA, Webster E, Butler J, Christensen RD, Reid LK (2008). Aerosol enrichment of the surfactant PFO and mediation of the water–air transport of gaseous PFOA. *Environ Sci Technol.* 42(11):3969–74. doi:10.1021/es7032026 PMID:18589953
- MDEQ (2018). General PFAS sampling guidance. Revised 16 October 2018. Lansing (MI), USA: Michigan Department of Environmental Quality. Revised 16 October 2018..
- Mikolajczyk S, Warenik-Bany M, Pajurek M (2023). Infant formula and baby food as a source of perfluoroalkyl substances for infants. *Environ Pollut*. 317:120810. doi:<u>10.1016/j.envpol.2022.120810</u> PMID:<u>36470453</u>

- Miner KR, Clifford H, Taruscio T, Potocki M, Solomon G, Ritari M, et al. (2021). Deposition of PFAS 'forever chemicals' on Mt Everest. *Sci Total Environ*. 759:144421. doi:10.1016/j.scitotenv.2020.144421 PMID:33353778
- Miralles-Marco A, Harrad S (2015). Perfluorooctane sulfonate: a review of human exposure, biomonitoring and the environmental forensics utility of its chirality and isomer distribution. *Environ Int.* 77:148–59. doi:10.1016/j.envint.2015.02.002 PMID:25728452
- Miura R, Araki A, Miyashita C, Kobayashi S, Kobayashi S, Wang SL, et al. (2018). An epigenome-wide study of cord blood DNA methylations in relation to prenatal perfluoroalkyl substance exposure: the Hokkaido study. *Environ Int.* 115:21–8. doi:<u>10.1016/j.envint.2018.03.004</u> PMID:<u>29544137</u>
- MNPCA (2023). PFAS ban. St Paul (MN), USA: Minnesota Pollution Control Agency. Available from: <u>https:// www.pca.state.mn.us/get-engaged/pfas-ban</u>, accessed October 2024.
- Monge Brenes AL, Curtzwiler G, Dixon P, Harrata K, Talbert J, Vorst K (2019). PFOA and PFOS levels in microwave paper packaging between 2005 and 2018. *Food Addit Contam Part B Surveill*. 12(3):191–8. doi:<u>10</u>. <u>1080/19393210.2019.1592238</u> PMID:<u>30945614</u>
- Motas Guzmàn M, Clementini C, Pérez-Cárceles MD, Jiménez Rejón S, Cascone A, Martellini T, et al. (2016). Perfluorinated carboxylic acids in human breast milk from Spain and estimation of infant's daily intake. *Sci Total Environ*. 544:595–600. doi:10.1016/j. scitotenv.2015.11.059 PMID:26674688
- Muir D, Miaz LT (2021). Spatial and temporal trends of perfluoroalkyl substances in global ocean and coastal waters. *Environ Sci Technol*. 55(14):9527–37. doi:<u>10.1021/</u> <u>acs.est.0c08035</u> PMID:<u>33646763</u>
- Munoz G, Mercier L, Duy SV, Liu J, Sauvé S, Houde M (2022). Bioaccumulation and trophic magnification of emerging and legacy per- and polyfluoroalkyl substances (PFAS) in a St. Lawrence River food web. *Environ Pollut.* 309:119739. doi:10.1016/j. envpol.2022.119739 PMID:35817301
- Naile JE, Khim JS, Hong S, Park J, Kwon B-O, Ryu JS, et al. (2013). Distributions and bioconcentration characteristics of perfluorinated compounds in environmental samples collected from the west coast of Korea. *Chemosphere*. 90(2):387–94. doi:10.1016/j. chemosphere.2012.07.033 PMID:22955048
- Naile JE, Seong Khim J, House JN, Jones PD, Giesy JP (2010). Standard purity and response factors of perfluorinated compounds. *Toxicol Environ Chem*. 92(7):1219–32. doi:10.1080/02772240903397556
- NASEM (2022). Guidance on PFAS exposure, testing, and clinical follow-up. Washington (DC), USA: The National Academies Press. Available from: <u>https://doi.org/10.17226/26156</u>, accessed October 2024.

- NCBI (2023a). Perfluorooctanoic acid. Compound summary for CID 9554. PubChem [online database]. Bethesda (MD), USA: National Center for Biotechnology Information. Available from: <u>https:// pubchem.ncbi.nlm.nih.gov/compound/9554</u>, accessed April 2023.
- NCBI (2023b). Perfluorooctanesulfonic acid. Compound summary for CID 74483. PubChem [online database]. Bethesda(MD), USA: National Center for Biotechnology Information. Available from: <u>https://pubchem.ncbi.</u> <u>nlm.nih.gov/compound/74483</u>, accessed April 2023.
- NCBI (2023c). Ammonium perfluorooctanoate. Compound summary for CID 2776282. PubChem [online database]. Bethesda (MD), USA: National Center for Biotechnology Information. Available from: <u>https://pubchem.ncbi.nlm.nih.gov/compound/</u><u>Ammonium-perfluorooctanoate</u>, accessed October 2024.
- NCBI (2023d). Sodium perfluorooctanoate. Compound summary for CID 23667657. PubChem [online database]. Bethesda (MD), USA: National Center for Biotechnology Information. Available from: <u>https:// pubchem.ncbi.nlm.nih.gov/compound/Sodiumperfluorooctanoate</u>, accessed October 2024.
- NCBI (2023e). Potassium perfluorooctanoate. Compound summary for CID 4253803. PubChem [online database]. Bethesda (MD), USA: National Center for Biotechnology Information. Available from: <u>https:// pubchem.ncbi.nlm.nih.gov/compound/Potassiumperfluorooctanoate</u>, accessed October 2024.
- NCBI (2023f). Ammonium 1,1,2,2,3,3,4,4,5,5,6,6,7,7,8,8,8heptadecafluorooctane-1-sulfonate. Compound summary for CID 15607692. PubChem [online database]. Bethesda (MD), USA: National Center for Biotechnology Information. Available from: <u>https:// pubchem.ncbi.nlm.nih.gov/compound/15607692</u>, accessed October 2024.
- NCBI (2023g). Lithium perfluorooctane sulfonate. Compound summary for CID 23677927. PubChem [online database]. Bethesda (MD), USA: National Center for Biotechnology Information. Available from: <u>https://pubchem.ncbi.nlm.nih.gov/compound/</u> <u>Lithium-perfluorooctane-sulfonate</u>, accessed October 2024.
- NCBI (2023h). Potassium perfluorooctanesulfonate. Compound summary for CID 23669238, PubChem [online database]. Available from: <u>https://pubchem.</u> <u>ncbi.nlm.nih.gov/compound/Potassium-perfluorooctanesulfonate</u>, accessed October 2024.
- NEA (2017). Environmental pollutants in the terrestrial and urban environment. Report M-1076. Trondheim, Norway: Norwegian Environment Agency. Available from: <u>https://www.miljodirektoratet.no/globalassets/</u> <u>publikasjoner/m1076/m1076.pdf</u>, accessed April 2023.

- NHANES (2023). Biomonitoring data tables for environmental chemicals. Atlanta (GA), USA: National Health and Nutrition Examination Survey, Centers for Disease Control and Prevention. Available from: <u>https://www. cdc.gov/exposurereport/data_tables.html?NER</u> <u>SectionItem=NHANES</u>, accessed May 2023.
- Nielsen CJ (2012). PFOA isomers, salts and precursors. Literature study and evaluation of physico-chemical properties. Oslo, Norway: Centre for Theoretical and Computational Chemistry, University of Oslo. Available from: https://www.researchgate.net/publication/256292766 PFOA Isomers Salts and Precursors Literature study and evaluation of physico-chemical properties, accessed November 2023.
- Nilsson H, Kärrman A, Rotander A, van Bavel B, Lindström G, Westberg H (2013a). Biotransformation of fluorotelomer compound to perfluorocarboxylates in humans. *Environ Int.* 51:8–12. doi:<u>10.1016/j.envint.2012.09.001</u> PMID:<u>23138016</u>
- Nilsson H, Kärrman A, Rotander A, van Bavel B, Lindström G, Westberg H (2013b). Professional ski waxers' exposure to PFAS and aerosol concentrations in gas phase and different particle size fractions. *Environ Sci Process Impacts*. 15(4):814–22. doi:10.1039/ c3em30739e PMID:23426585
- Nilsson H, Kärrman A, Westberg H, Rotander A, van Bavel B, Lindström G (2010). A time trend study of significantly elevated perfluorocarboxylate levels in humans after using fluorinated ski wax. *Environ Sci Technol.* 44(6):2150–5. doi:10.1021/es9034733 PMID:20158198
- NIOSH (2023). Hierarchy of controls. Bethesda (MD), USA: National Institute for Occupational Safety and Health, Centers for Disease Control and Prevention. Available from: <u>https://www.cdc.gov/niosh/learning/</u> <u>safetyculturehc/module-3/2.html</u>, accessed October 2024.
- Noorlander CW, van Leeuwen SP, Te Biesebeek JD, Mengelers MJ, Zeilmaker MJ (2011). Levels of perfluorinated compounds in food and dietary intake of PFOS and PFOA in the Netherlands. *J Agric Food Chem.* 59(13):7496–505. doi:<u>10.1021/jf104943p</u> PMID:<u>21591675</u>
- Nordic Council of Ministers (2017). Analysis of PFASs and TOF in products. Copenhagen, Denmark: Nordic Council of Ministers. Available from: <u>https:// norden.diva-portal.org/smash/get/diva2:1118439/</u> <u>FULLTEXT01.pdf</u>, accessed October 2024.
- Nøst TH, Vestergren R, Berg V, Nieboer E, Odland JØ, Sandanger TM (2014). Repeated measurements of perand polyfluoroalkyl substances (PFASs) from 1979 to 2007 in males from northern Norway: assessing time trends, compound correlations and relations to age/birth cohort. *Environ Int*. 67:43–53. doi:10.1016/j. envint.2014.02.011 PMID:24657493

- Nübler S, Esteban López M, Castaño A, Mol HGJ, Haji-Abbas-ZarrabiK,SchäferM,etal.(2022).Interlaboratory comparison investigations (ICIs) and external quality assurance schemes (EQUASs) for human biomonitoring of perfluoroalkyl substances (PFASs) in serum as part of the quality assurance programme under HBM4EU. *Sci Total Environ.* 847:157481. doi:<u>10.1016/j.</u> <u>scitotenv.2022.157481</u> PMID:<u>35868372</u>
- OECD (2008). PFOA and APFO. SIDS initial assessment profile. Paris, France: Organisation for Economic Cooperation and Development. Available from: <u>https:// hpvchemicals.oecd.org/ui/handler.axd?id=c4b4700f-48f3-4aca-b6a7-6606b1a4fa48</u>, accessed October 2024.
- OECD (2015a). No. 30: Working towards a global emission inventory of PFASs: focus on PFCAS – status quo and the way forward. OECD Environment, Health and Safety Publications Series on Risk Management. No 30. Paris, France: Environmental Directorate Organisation for Economic Cooperation and Development. Available from: <u>https://www.oecd-ilibrary.org/environment/</u> working-towards-a-global-emission-inventoryof-pfass-focus-on-pfcas-status-quo-and-the-wayforward f97f34b1-en, accessed November 2023.
- OECD (2015b). Risk reduction approaches for per- and polyfluoroalkyl substances (PFASs) – a cross-country analysis. Paris, France: Organisation for Economic Cooperation and Development. Available from: <u>https://www.oecd.org/en/publications/risk-reductionapproaches-for-per-and-polyfluoroalkyl-substancespfass-a-cross-country-analysis_ed861d08-en.html, accessed October 2024.</u>
- OECD (2023a). People's Republic of China. Country information. Paris, France: Organisation for Economic Cooperation and Development. Available from: <u>https://</u> <u>www.oecd.org/content/dam/oecd/en/topics/policysub-issues/risk-management-risk-reduction-andsustainable-chemistry2/pfas-country-information/ People%E2%80%99s%20Republic%20of%20China. pdf, accessed October 2024.</u>
- OECD (2023b). Portal on per and poly fluorinated chemicals – country information. Paris, France: Organisation for Economic Cooperation and Development. Available from: <u>https://www.oecd.org/en/topics/</u> <u>risk-management-risk-reduction-and-sustainablechemistry.html</u>, accessed March 2023.
- Office of the Federal Register (2012). National emission standards for hazardous air pollutant emissions: hard and decorative chromium electroplating and chromium anodizing tanks; and steel pickling HCl process facilities and hydrochloric acid regeneration plants. 40 CFR Part 63. *Federal Register.* 77(182):58220. United States Code of Federal Regulations. Washington (DC), USA: National Archives and Records Administration. Available from: https://www.govinfo.gov/content/pkg/ FR-2012-09-19/pdf/2012-20642.pdf, accessed October 2024.

- Office of the Federal Register (2022). Notices. *Federal Register*. 87(118):36848–9. Washington (DC), USA: Office of the Federal Register. Available from: <u>https://www.govinfo.gov/content/pkg/FR-2022-06-21/pdf/2022-13158.pdf</u>, accessed October 2024.
- Okada E, Kashino I, Matsuura H, Sasaki S, Miyashita C, Yamamoto J, et al. (2013). Temporal trends of perfluoroalkyl acids in plasma samples of pregnant women in Hokkaido, Japan, 2003–2011. *Environ Int*. 60:89–96. doi:<u>10.1016/j.envint.2013.07.013</u> PMID:<u>24013022</u>
- Olsen GW, Burris JM, Burlew MM, Mandel JH (2000). Plasma cholecystokinin and hepatic enzymes, cholesterol and lipoproteins in ammonium perfluorooctanoate production workers. *Drug Chem Toxicol*. 23(4):603–20. doi:10.1081/DCT-100101973 PMID:11071397
- Olsen GW, Huang HY, Helzlsouer KJ, Hansen KJ, Butenhoff JL, Mandel JH (2005). Historical comparison of perfluorooctanesulfonate, perfluorooctanoate, and other fluorochemicals in human blood. *Environ Health Perspect*. 113(5):539–45. doi:10.1289/ehp.7544 PMID:15866760
- Olsen GW, Logan PW, Hansen KJ, Simpson CA, Burris JM, Burlew MM, et al. (2003). An occupational exposure assessment of a perfluorooctanesulfonyl fluoride production site: biomonitoring. *AIHA J (Fairfax, VA)*. 64(5):651–9. doi:<u>10.1080/15428110308984859</u> PMID: <u>14521435</u>
- Omoike OE, Pack RP, Mamudu HM, Liu Y, Strasser S, Zheng S, et al. (2021). Association between per and polyfluoroalkyl substances and markers of inflammation and oxidative stress. *Environ Res.* 196:110361. doi:10.1016/j.envres.2020.110361 PMID:33131681
- Padilla-Sánchez JA, Haug LS (2016). A fast and sensitive method for the simultaneous analysis of a wide range of per- and polyfluoroalkyl substances in indoor dust using on-line solid phase extraction–ultrahigh performance liquid chromatography–time-of-flightmass spectrometry. J Chromatogr A. 1445:36–45. doi:10.1016/j.chroma.2016.03.058 PMID:27062722
- Pan Y, Cui Q, Wang J, Sheng N, Jing J, Yao B, et al. (2019). Profiles of emerging and legacy per-/polyfluoroalkyl substances in matched serum and semen samples: new implications for human semen quality. *Environ Health Perspect*. 127(12):127005. doi:10.1289/EHP4431 PMID:31841032
- Pan Y, Shi Y, Wang J, Jin X, Cai Y (2011). Pilot investigation of perfluorinated compounds in river water, sediment, soil and fish in Tianjin, China. *Bull Environ Contam Toxicol.* 87(2):152–7. doi:<u>10.1007/s00128-011-</u> <u>0313-0</u> PMID:<u>21626112</u>
- Panieri E, Baralic K, Djukic-Cosic D, Buha Djordjevic A, Saso L (2022). PFAS molecules: a major concern for the human health and the environment. *Toxics*. 10(2):10. doi:10.3390/toxics10020044 PMID:35202231

- Pasecnaja E, Bartkevics V, Zacs D (2022). Occurrence of selected per- and polyfluorinated alkyl substances (PFASs) in food available on the European market a review on levels and human exposure assessment. *Chemosphere*. 287(Pt 4):132378. doi:10.1016/j. chemosphere.2021.132378 PMID:34592212
- Paul AG, Jones KC, Sweetman AJ (2009). A first global production, emission, and environmental inventory for perfluorooctane sulfonate. *Environ Sci Technol.* 43(2):386–92. doi:10.1021/es802216n PMID:19238969
- Peaslee GF, Wilkinson JT, McGuinness SR, Tighe M, Caterisano N, Lee S, et al. (2020). Another pathway for firefighter exposure to per- and polyfluoroalkyl substances: firefighter textiles. *Environ Sci Technol Lett*. 7(8):594–9. doi:<u>10.1021/acs.estlett.0c00410</u>
- PFAS Project Laboratory, Silent Spring Institute, PFAS-REACH team (2023). PFAS sites and community resources. An interactive mapping project from the PFAS-REACH team. Newton (MA), USA. Available from: <u>https://experience.arcgis.com/experience/1</u> 2412ab41b3141598e0bb48523a7c940/?views=Key-Abbreviations, accessed October 2024.
- Pitter G, Da Re F, Canova C, Barbieri G, Zare Jeddi M, Daprà F, et al. (2020). Serum levels of perfluoroalkyl substances (PFAS) in adolescents and young adults exposed to contaminated drinking water in the Veneto Region, Italy: a cross-sectional study based on a health surveillance program. *Environ Health Perspect*. 128(2):27007. doi:10.1289/EHP5337 PMID:32068468
- Poothong S, Lundanes E, Thomsen C, Haug LS (2017). High throughput online solid phase extraction–ultra high performance liquid chromatography–tandem mass spectrometry method for polyfluoroalkyl phosphate esters, perfluoroalkyl phosphonates, and other perfluoroalkyl substances in human serum, plasma, and whole blood. *Anal Chim Acta*. 957:10–9. doi:10.1016/j. aca.2016.12.043 PMID:28107829
- Poothong S, Papadopoulou E, Lundanes E, Padilla-Sánchez JA, Thomsen C, Haug LS (2019). Dried blood spots for reliable biomonitoring of poly- and perfluoroalkyl substances (PFASs). *Sci Total Environ*. 655:1420–6. doi:<u>10.1016/j.scitotenv.2018.11.214</u> PMID:<u>30577133</u>
- Poothong S, Papadopoulou E, Padilla-Sánchez JA, Thomsen C, Haug LS (2020). Multiple pathways of human exposure to poly- and perfluoroalkyl substances (PFASs): from external exposure to human blood. *Environ Int*. 134:105244. doi:<u>10.1016/j. envint.2019.105244</u> PMID:<u>31711019</u>
- Powley CR, George SW, Ryan TW, Buck RC (2005). Matrix effect-free analytical methods for determination of perfluorinated carboxylic acids in environmental matrixes. *Anal Chem.* 77(19):6353–8. doi:10.1021/ac0508090 PMID:16194099
- Préfète du Rhône (2023). Présence de substances perfluorées au sud de Lyon. Lyon, France: Préfète du Rhône. Available from: <u>https://www.rhone.gouv.fr/</u>

Actualites/Substances-perfluorees-PFAS/Presence-desubstances-perfluorees-au-sud-de-Lyon; https://www. auvergne-rhone-alpes.developpement-durable.gouv. fr/IMG/pdf/resultats oeufs particuliers fev2023-2. pdf; https://www.auvergne-rhone-alpes.developpement-durable.gouv.fr/IMG/pdf/valeurs detaillees oeufs particuliers juin2023.pdf; https://www.auvergne-rhone-alpes.developpement-durable.gouv.fr/ IMG/pdf/resultats oeufs professionnels sept2023. pdf, accessed November 2023.

- Prevedouros K, Cousins IT, Buck RC, Korzeniowski SH (2006). Sources, fate and transport of perfluorocarboxylates. *Environ Sci Technol.* 40(1):32–44. doi:<u>10.1021/</u> <u>es0512475</u> PMID:<u>16433330</u>
- Purdue MP, Rhee J, Denic-Roberts H, McGlynn KA, Byrne C, Sampson J, et al. (2023). A nested case-control study of serum per- and polyfluoroalkyl substances and testicular germ cell tumors among US Air Force servicemen. *Environ Health Perspect*. 131(7):77007. doi:10.1289/EHP12603 PMID:37458713
- Pütz KW, Namazkar S, Plassmann M, Benskin JP (2022). Are cosmetics a significant source of PFAS in Europe? Product inventories, chemical characterization and emission estimates. *Environ Sci Process Impacts.* 24(10):1697–707. doi:10.1039/D2EM00123C PMID:35959763
- Raleigh KK, Alexander BH, Olsen GW, Ramachandran G, Morey SZ, Church TR, et al. (2014). Mortality and cancer incidence in ammonium perfluorooctanoate production workers. *Occup Environ Med.* 71(7):500–6. doi:10.1136/oemed-2014-102109 PMID:24832944
- Ramírez Carnero A, Lestido-Cardama A, Vazquez Loureiro P, Barbosa-Pereira L, Rodríguez Bernaldo de Quirós A, Sendón R (2021). Presence of perfluoroalkyl and polyfluoroalkyl substances (PFAS) in food contact materials (FCM) and its migration to food. *Foods*. 10(7):1443. doi:10.3390/foods10071443 PMID:34206351
- Rhee J, Barry KH, Huang WY, Sampson JN, Hofmann JN, Silverman DT, et al. (2023a). A prospective nested case-control study of serum concentrations of perand polyfluoroalkyl substances and aggressive prostate cancer risk. *Environ Res.* 228:115718. doi:<u>10.1016/j.</u> <u>envres.2023.115718</u> PMID:<u>36958379</u>
- Rhee J, Chang VC, Cheng I, Calafat AM, Botelho JC, Shearer JJ, et al. (2023b). Serum concentrations of perand polyfluoroalkyl substances and risk of renal cell carcinoma in the Multiethnic Cohort Study. *Environ Int.* 180:108197. PMID:<u>37741007</u>
- Richterová D, Govarts E, Fábelová L, Rausová K, Rodriguez Martin L, Gilles L, et al. (2023). PFAS levels and determinants of variability in exposure in European teenagers – results from the HBM4EU aligned studies (2014–2021). *Int J Hyg Environ Health*. 247:114057. doi:10.1016/j.ijheh.2022.114057 PMID:36327670

- Robel AE, Marshall K, Dickinson M, Lunderberg D, Butt C, Peaslee G, et al. (2017). Closing the mass balance on fluorine on papers and textiles. *Environ Sci Technol.* 51(16):9022–32. doi:<u>10.1021/acs.est.7b02080</u> PMID:28712295
- Rodgers KM, Swartz CH, Occhialini J, Bassignani P, McCurdy M, Schaider LA (2022). How well do product labels indicate the presence of PFAS in consumer items used by children and adolescents? *Environ Sci Technol.* 56(10):6294–304. doi:<u>10.1021/acs.est.1c05175</u> PMID:<u>35506608</u>
- Rosenfeld PE, Spaeth KR, Remy LL, Byers V, Muerth SA, Hallman RC, et al. (2023). Perfluoroalkyl substances exposure in firefighters: sources and implications. *Environ Res.* 220:115164. doi:10.1016/j. envres.2022.115164 PMID:36584840
- Rotander A, Toms LM, Aylward L, Kay M, Mueller JF (2015). Elevated levels of PFOS and PFHxS in fire-fighters exposed to aqueous film forming foam (AFFF). *Environ Int.* 82:28–34. doi:<u>10.1016/j.envint.2015.05.005</u> PMID:<u>26001497</u>
- Royal Society of Chemistry (2023). Perfluorooctanesulfonic acid. ChemSpider [online database]. London, UK: Royal Society of Chemistry. Available from: <u>http:// www.chemspider.com/Chemical-Structure.67068.</u> <u>html?rid=d1ac6c78-42f0-4c33-9e71-95ee96069765</u>, accessed April 2023.
- Sadia M, Yeung LWY, Fiedler H (2020). Trace level analyses of selected perfluoroalkyl acids in food: method development and data generation. *Environ Pollut*. 263(Pt A):113721. doi:10.1016/j.envpol.2019.113721 PMID:32229370
- Salvatore D, Mok K, Garrett KK, Poudrier G, Brown P, Birnbaum LS, et al. (2022). Presumptive contamination: a new approach to PFAS contamination based on likely sources. *Environ Sci Technol Lett.* 9(11):983–90. doi:10.1021/acs.estlett.2c00502 PMID:36398312
- Sammut G, Sinagra E, Sapiano M, Helmus R, de Voogt P (2019). Perfluoroalkyl substances in the Maltese environment – (II) sediments, soils and groundwater. *Sci Total Environ*. 682:180–9. doi:10.1016/j. scitotenv.2019.04.403 PMID:31121344
- Schaider LA, Balan SA, Blum A, Andrews DQ, Strynar MJ, Dickinson ME, et al. (2017). Fluorinated compounds in US fast food packaging. *Environ Sci Technol Lett.* 4(3):105–11. doi:10.1021/acs.estlett.6b00435 PMID:30148183
- Scher DP, Kelly JE, Huset CA, Barry KM, Hoffbeck RW, Yingling VL, et al. (2018). Occurrence of perfluoroalkyl substances (PFAS) in garden produce at homes with a history of PFAS-contaminated drinking water. *Chemosphere*. 196:548–55. doi:10.1016/j. chemosphere.2017.12.179 PMID:29329087
- Schröter-Kermani C, Müller J, Jürling H, Conrad A, Schulte C (2013). Retrospective monitoring of perfluorocarboxylates and perfluorosulfonates in

human plasma archived by the German Environmental Specimen Bank. *Int J Hyg Environ Health*. 216(6):633– 40. doi:<u>10.1016/j.ijheh.2012.08.004</u> PMID:<u>22999890</u>

- Schultes L, Peaslee GF, Brockman JD, Majumdar A, McGuinness SR, Wilkinson JT, et al. (2019). Total fluorine measurements in food packaging: how do current methods perform? *Environ Sci Technol Lett.* 6(2):73–8. doi:10.1021/acs.estlett.8b00700
- Schümann M, Lilienthal H, Hölzer J (2021). Human biomonitoring (HBM)-II values for perfluorooctanoic acid (PFOA) and perfluorooctane sulfonic acid (PFOS) – description, derivation and discussion. *Regul Toxicol Pharmacol.* 121:104868. doi:10.1016/j. yrtph.2021.104868 PMID:33484797
- Seacat AM, Thomford PJ, Hansen KJ, Clemen LA, Eldridge SR, Elcombe CR, et al. (2003). Sub-chronic dietary toxicity of potassium perfluorooctanesulfonate in rats. *Toxicology*. 183(1–3):117–31. doi:10.1016/S0300-483X(02)00511-5 PMID:12504346
- Seo SH, Son MH, Shin ES, Choi SD, Chang YS (2019). Matrix-specific distribution and compositional profiles of perfluoroalkyl substances (PFASs) in multimedia environments. *J Hazard Mater*. 364:19–27. doi:10.1016/j. jhazmat.2018.10.012 PMID:30336332
- Sepulvado JG, Blaine AC, Hundal LS, Higgins CP (2011). Occurrence and fate of perfluorochemicals in soil following the land application of municipal biosolids. *Environ Sci Technol.* 45(19):8106–12. doi:<u>10.1021/</u> <u>es103903d</u> PMID:<u>21446724</u>
- Serrano L, Iribarne-Durán LM, Suárez B, Artacho-Cordón F, Vela-Soria F, Peña-Caballero M, et al. (2021). Concentrations of perfluoroalkyl substances in donor breast milk in Southern Spain and their potential determinants. *Int J Hyg Environ Health*. 236:113796. doi:10.1016/j.ijheh.2021.113796 PMID:34192647
- SFT (2008). Screening of polyfluorinated organic compounds at four fire training facilities in Norway. TA-2444/2008. Oslo, Norway: Norwegian Pollution Control Authority. Available from: <u>https://kudos.dfo. no/documents/9379/files/9266.pdf</u>, accessed April 2023.
- Sha B, Dahlberg AK, Wiberg K, Ahrens L (2018). Fluorotelomer alcohols (FTOHs), brominated flame retardants (BFRs), organophosphorus flame retardants (OPFRs) and cyclic volatile methylsiloxanes (cVMSs) in indoor air from occupational and home environments. *Environ Pollut.* 241:319–30. doi:10.1016/j. envpol.2018.04.032 PMID:29843014
- Sha B, Johansson JH, Tunved P, Bohlin-Nizzetto P, Cousins IT, Salter ME (2022). Sea spray aerosol (SSA) as a source of perfluoroalkyl acids (PFAAs) to the atmosphere: field evidence from long-term air monitoring. *Environ Sci Technol.* 56(1):228–38. doi:<u>10.1021/acs.est.1c04277</u> PMID:<u>34907779</u>

- Shan G, Chen X, Zhu L (2015). Occurrence, fluxes and sources of perfluoroalkyl substances with isomer analysis in the snow of northern China. J Hazard Mater. 299:639–46. doi:10.1016/j.jhazmat.2015.07.074 PMID:26280369
- Shan G, Wang Z, Zhou L, Du P, Luo X, Wu Q, et al. (2016). Impacts of daily intakes on the isomeric profiles of perfluoroalkyl substances (PFASs) in human serum. *Environ Int*. 89–90:62–70. doi:10.1016/j. envint.2016.01.002 PMID:26826363
- Shearer JJ, Callahan CL, Calafat AM, Huang WY, Jones RR, Sabbisetti VS, et al. (2021). Serum concentrations of per- and polyfluoroalkyl substances and risk of renal cell carcinoma. *J Natl Cancer Inst.* 113(5):580–7. doi:10.1093/jnci/djaa143 PMID:32944748
- Shi Y, Vestergren R, Xu L, Zhou Z, Li C, Liang Y, et al. (2016). Human exposure and elimination kinetics of chlorinated polyfluoroalkyl ether sulfonic acids (Cl-PFESAs). *Environ Sci Technol*. 50(5):2396–404. doi:10.1021/acs.est.5b05849 PMID:26866980
- Shin HM, Vieira VM, Ryan PB, Detwiler R, Sanders B, Steenland K, et al. (2011a). Environmental fate and transport modeling for perfluorooctanoic acid emitted from the Washington Works facility in West Virginia. *Environ Sci Technol.* 45(4):1435–42. doi:10.1021/ es102769t PMID:21226527
- Shin HM, Vieira VM, Ryan PB, Steenland K, Bartell SM (2011b). Retrospective exposure estimation and predicted versus observed serum perfluorooctanoic acid concentrations for participants in the C8 Health Project. *Environ Health Perspect*. 119(12):1760–5. doi:10.1289/ehp.1103729 PMID:21813367
- Shoeib M, Harner T, M Webster G, Lee SC (2011). Indoor sources of poly- and perfluorinated compounds (PFCS) in Vancouver, Canada: implications for human exposure. *Environ Sci Technol.* 45(19):7999–8005. doi:10.1021/es103562v PMID:21332198
- Shoeib M, Vlahos P, Harner T, Peters A, Graustein M, Narayan J (2010). Survey of polyfluorinated chemicals (PFCs) in the atmosphere over the northeast Atlantic Ocean. *Atmos Environ*. 44(24):887–2893. doi:10.1016/j. atmosenv.2010.04.056
- Shoeib T, Hassan Y, Rauert C, Harner T (2016). Polyand perfluoroalkyl substances (PFASs) in indoor dust and food packaging materials in Egypt: trends in developed and developing countries. *Chemosphere*. 144:1573–81. doi:10.1016/j.chemosphere.2015.08.066 PMID:26517384
- Shoemaker J, Tettenhorst D (2020). Method 537.1. Determination of selected per- and polyfluorinated alkyl substances in drinking water by solid phase extraction and liquid chromatography/tandem mass spectrometry (LC/MS/MS). Washington (DC), USA: United States Environmental Protection Agency, Office of Research and Development, National Center for Environmental Assessment. Available from: https://

cfpub.epa.gov/si/si public record report.cfm?dirE ntryId=348508&Lab=CESER&simpleSearch=0&sho wCriteria=2&searchAll=537.1&TIMSType=&dateBe ginPublishedPresented=03%2F24%2F2018, accessed October 2024.

- Sinclair E, Kim SK, Akinleye HB, Kannan K (2007). Quantitation of gas-phase perfluoroalkyl surfactants and fluorotelomer alcohols released from nonstick cookware and microwave popcorn bags. *Environ Sci Technol.* 41(4):1180–5. doi:10.1021/es062377w PMID:17593716
- Skaar JS, Ræder EM, Lyche JL, Ahrens L, Kallenborn R (2019). Elucidation of contamination sources for poly- and perfluoroalkyl substances (PFASs) on Svalbard (Norwegian Arctic). *Environ Sci Pollut Res Int.* 26(8):7356–63. doi:10.1007/s11356-018-2162-4 PMID:29754295
- Song XL, Wang N, He FY, Cheng CL, Wang F, Wang JL, et al. (2023). [Determination of trace perfluorinated compounds in environmental water samples by dispersive solid-phase extraction-high performance liquid chromatography-tandem mass spectrometry using carbon nanotube composite materials]. Se Pu. 41(5):409–16. doi:10.3724/SP.J.1123.2022.09016 PMID:37087606 [Chinese]
- Sottani C, Minoia C (2002). Quantitative determination of perfluorooctanoic acid ammonium salt in human serum by high-performance liquid chromatography with atmospheric pressure chemical ionization tandem mass spectrometry. *Rapid Commun Mass Spectrom*. 16(7):650–4. doi:10.1002/rcm.620 PMID:11921242
- Spliethoff HM, Tao L, Shaver SM, Aldous KM, Pass KA, Kannan K, et al. (2008). Use of newborn screening program blood spots for exposure assessment: declining levels of perfluorinated compounds in New York State infants. *Environ Sci Technol*. 42(14):5361–7. doi:10.1021/es8006244 PMID:18754394
- Srivastava P, Williams M, Du J, Navarro D, Kookana R, Douglas G, et al. (2022). Method for extraction and analysis of per- and poly-fluoroalkyl substances in contaminated asphalt. *Anal Methods*. 14(17):1678–89. doi:10.1039/D2AY00221C PMID:35438700
- Steenland K, Tinker S, Frisbee S, Ducatman A, Vaccarino V (2009). Association of perfluorooctanoic acid and perfluorooctane sulfonate with serum lipids among adults living near a chemical plant. Am J Epidemiol. 170(10):1268–78. doi:10.1093/aje/kwp279 PMID:19846564
- Steenland K, Woskie S (2012). Cohort mortality study of workers exposed to perfluorooctanoic acid. *Am J Epidemiol.* 176(10):909–17. doi:<u>10.1093/aje/kws171</u> PMID:<u>23079607</u>
- Steenland K, Zhao L, Winquist A (2015). A cohort incidence study of workers exposed to perfluorooctanoic acid (PFOA). Occup Environ Med. 72(5):373–80. doi:10.1136/oemed-2014-102364 PMID:25601914

- Stoiber T, Evans S, Naidenko OV (2020). Disposal of products and materials containing per- and polyfluoroalkyl substances (PFAS): a cyclical problem. *Chemosphere*. 260:127659. doi:10.1016/j.chemosphere.2020.127659 PMID:32698118
- Sunderland EM, Hu XC, Dassuncao C, Tokranov AK, Wagner CC, Allen JG (2019). A review of the pathways of human exposure to poly- and perfluoroalkyl substances (PFASs) and present understanding of health effects. J Expo Sci Environ Epidemiol. 29(2):131– 47. doi:10.1038/s41370-018-0094-1 PMID:30470793
- Supreeyasunthorn P, Boontanon SK, Boontanon N (2016). Perfluorooctane sulfonate (PFOS) and perfluorooctanoic acid (PFOA) contamination from textiles. J Environ Sci Health Part A Tox Hazard Subst Environ Eng. 51(6):472–7. doi:10.1080/10934529.2015.1128713 PMID:26864911
- Swedish Chemicals Agency (2020). Hard chrome metal plating – use of PFOS as mist suppressant and its alternatives. Keml, H20–05692. Sundbyberg, Sweden: Swedish Chemicals Agency.
- Tan B, Wang T, Wang P, Luo W, Lu Y, Romesh KY, et al. (2014). Perfluoroalkyl substances in soils around the Nepali Koshi River: levels, distribution, and mass balance. *Environ Sci Pollut Res Int.* 21(15):9201–11. doi:10.1007/s11356-014-2835-6 PMID:24705951
- Tao L, Kannan K, Aldous KM, Mauer MP, Eadon GA (2008a). Biomonitoring of perfluorochemicals in plasma of New York State personnel responding to the World Trade Center disaster. *Environ Sci Technol.* 42(9):3472–8. doi:10.1021/es8000079 PMID:18522136
- Tao L, Kannan K, Wong CM, Arcaro KF, Butenhoff JL (2008b). Perfluorinated compounds in human milk from Massachusetts, USA. *Environ Sci Technol*. 42(8):3096–101. doi:10.1021/es702789k PMID:18497172
- Thomsen C, Haug LS, Stigum H, Frøshaug M, Broadwell SL, Becher G (2010). Changes in concentrations of perfluorinated compounds, polybrominated diphenyl ethers, and polychlorinated biphenyls in Norwegian breast-milk during twelve months of lactation. *Environ Sci Technol.* 44(24):9550–6. doi:10.1021/es1021922 PMID:21090747
- Tian Z, Kim SK, Shoeib M, Oh JE, Park JE (2016). Human exposure to per- and polyfluoroalkyl substances (PFASs) via house dust in Korea: implication to exposure pathway. *Sci Total Environ*. 553:266–75. doi:<u>10.1016/j.</u> <u>scitotenv.2016.02.087</u> PMID:<u>26933964</u>
- Toms L-M, Thompson J, Rotander A, Hobson P, Calafat AM, Kato K, et al. (2014). Decline in perfluorooctane sulfonate and perfluorooctanoate serum concentrations in an Australian population from 2002 to 2011. *Environ Int.* 71:74–80. doi:10.1016/j.envint.2014.05.019 PMID:24980755
- Tsai MS, Chang SH, Kuo WH, Kuo CH, Li SY, Wang MY, et al. (2020). A case-control study of perfluoroalkyl substances and the risk of breast cancer in Taiwanese

women. *Environ Int.* 142:105850. doi:10.1016/j. envint.2020.105850 PMID:32580117

- UNEP (2006). Risk profile on perfluorooctane sulfonate. Stockholm Convention. Nairobi, Kenya: United Nations Environment Programme. Available from: <u>https://chm.pops.int/TheConvention/POPsReviewCommittee/</u><u>Reports</u>, accessed April 2023.
- UNEP (2017). Risk profile on pentadecafluorooctanoic acid (CAS No. 335-67-1, PFOA, perfluorooctanoic acid), its salts and PFOA-related compounds. Stockholm Convention. Nairobi, Kenya: United Nations Environment Programme. Available from: <u>http://chm. pops.int/TheConvention/POPsReviewCommittee/ ReportsandDecisions/tabid/3309/Default.aspx</u>, accessed April 2023.
- UNEP (2023). PFASs listed under the Stockholm Convention. Stockholm Convention. Nairobi, Kenya: United Nations Environment Programme. Available from: <u>http://chm.pops.int/Implementation/ IndustrialPOPs/PFAS/Overview/tabid/5221/Default.</u> aspx, accessed March 2023.
- US ÉPA (2016). Fact sheet. PFOA & PFOS drinking water health advisories. EPA 800-F-16-003. Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://www. epa.gov/sites/default/files/2016-06/documents/ drinkingwaterhealthadvisories pfoa pfos updated 5.31.16.pdf, accessed October 2024.</u>
- US EPA (2017a). Technical fact sheet perfluorooctane sulfonate (PFOS) and perfluorooctanoic acid (PFOA). Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://19january2021snapshot.epa.gov/sites/</u> <u>static/files/2017-12/documents/ffrrofactsheet</u> <u>contaminants_pfos_pfoa_11-20-17_508_0.pdf</u>, accessed April 2023.
- US EPA (2017b). The Third Unregulated Contaminant Monitoring Rule (UCMR 3): data summary. Office of Water. EPA 815-S-17-001. Washington (DC), USA: United States Environmental Protection Agency. Available from: https://19january2021snapshot.epa.gov/ dwucmr/data-summary-third-unregulated-contaminant-monitoring-rule_.html, accessed October 2024.
- US EPA (2019). Method 533. Determination of per- and polyfluoroalkyl substances in drinking water by isotope dilution anion exchange solid phase extraction and liquid chromatography/tandem mass spectrometry. 815-B-19-020. Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://www.epa.gov/dwanalyticalmethods/method-533-determination-and-polyfluoroalkyl-substancesdrinking-water-isotope</u>, accessed October 2024.
- US EPA (2020). Interim guidance on the destruction and disposal of perfluoroalkyl and polyfluoroalkyl substances and materials containing perfluoroalkyl

and polyfluoroalkyl substances. Interim guidance for public comment. Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://www.epa.gov/system/files/documents/2021-11/</u> <u>epa-hq-olem-2020-0527-0002_content.pdf</u>, accessed November 2023.

- US EPA (2021). Other Test Method 45 (OTM-45) measurement of selected per- and polyfluorinated alkyl substances from stationary sources. Washington (DC), USA: United States Environmental Protection Agency. Available from: https://www.epa.gov/sites/default/files/2021-01/documents/otm 45 semivolatile_pfas 1-13-21.pdf.
- US EPA (2022). Fact sheet: 2010/2015 PFOA Stewardship Program. Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://www.epa.gov/assessing-and-managingchemicals-under-tsca/fact-sheet-20102015-pfoastewardship-program#meet</u>, accessed March 2023.
- US EPA (2023a). Perfluorooctanesulfonic acid. 1763-23-1. DTXSID3031864. CompTox Chemicals Dashboard. Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://comptox. epa.gov/dashboard/chemical/properties/DTX</u> <u>SID3031864</u>, accessed April 2023.
- US EPA (2023b). PFAS analytical methods development and sampling research. Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://www.epa.gov/water-research/</u> <u>pfas-analytical-methods-development-and-samplingresearch</u>, accessed November 2023.
- US EPA (2023c). Hazardous waste test methods. US EPA SW-846 validated method 8327. Per- and polyfluoroalkyl substances (PFAS) using external standard calibration and liquid chromatography/tandem mass spectrometry (LC/MS/MS). Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://www.epa.gov/system/files/ documents/2021-07/8327.pdf</u> and <u>https://www.epa. gov/system/files/documents/2021-07/additionalpeformance-data-from-validation-study-for-methods-3512-and-8327.pdf</u>, accessed October 2024.
- US EPA (2023d). 4th Draft Method 1633. Analysis of perand polyfluoroalkyl substances (PFAS) in aqueous, solid, biosolids, and tissue samples by LC-MS/MS. EPA 821-D-23-001. July 2023. Washington (DC), USA: Office of Water, United States Environmental Protection Agency. Available from: https://www.epa. gov/system/files/documents/2024-01/method-1633final-for-web-posting.pdf, accessed November 2023. [Superseded by Method 1633 revision A, December 2024, available from: https://www.epa.gov/system/ files/documents/2024-01/method-1633-final-for-webposting.pdf]

- US EPA (2023e). Access Chemical Data Reporting Data. Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://www.epa.</u> <u>gov/chemical-data-reporting/access-chemical-data-reporting-data</u>, accessed December 2023.
- US EPA (2024). The Fifth Unregulated Contaminant Monitoring Rule (UCMR 5). Data summary: July 2024. Washington (DC), USA: United States Environmental Protection Agency. Available from: <u>https://www.epa.gov/system/files/documents/2023-08/ucmr5-datasummary_0.pdf</u>, accessed October 2024.
- US EPA SAB (2022). Review of EPA's analyses to support EPA's national primary drinking water rulemaking for PFAS. Final report. Washington (DC), USA: United States Environmental Protection Agency Science Advisory Board. Available from: <u>https://sab.epa.</u> <u>gov/ords/sab/f?p=100:12:17203034137454</u>, accessed October 2024.
- US FDA (2021a). Analytical results for PFAS in 2018–2021 dairy farm sampling (parts per trillion). Silver Spring (MD), USA: United States Food and Drug Administration. Available from: <u>https://www.fda.gov/media/127850/download?attachment</u>, accessed November 2023.
- US FDA (2021b). Determination of 16 per and polyfluoroalkyl substances (PFAS) in processed food using liquid chromatography-tandem mass spectrometry (LC-MS/MS). US FDA method number C-010.02. Silver Spring (MD), USA: United States Food and Drug Administration. Available from: <u>https://www.fda.gov/ media/183867/download</u>, accessed October 2024.
- US FDA (2022a). Analytical results of testing food for PFAS from environmental contamination. Silver Spring (MD), USA: United States Food and Drug Administration. Available from: <u>https://www.fda.gov/ food/process-contaminants-food/analytical-resultstesting-food-pfas-environmental-contamination,</u> accessed October 2024.
- US FDA (2022b). Per- and polyfluoroalkyl substances (PFAS). Silver Spring (MD), USA: United States Food and Drug Administration. Available from: <u>https://www.fda.gov/food/environmental-contaminants-food/and-polyfluoroalkyl-substances-pfas</u>, accessed March 2023.
- US FDA (2023). Authorized uses of PFAS in food contact applications. Silver Spring (MD), USA: United States Food and Drug Administration. Available from: <u>https://www.fda.gov/food/process-contaminants-food/authorized-uses-pfas-food-contact-applications</u>, accessed February 2024.
- van der Veen I, Fiedler H, de Boer J (2023). Assessment of the per- and polyfluoroalkyl substances analysis under the Stockholm Convention – 2018/2019. *Chemosphere*. 313:137549. doi:<u>10.1016/j.chemosphere.2022.137549</u> PMID:<u>36535499</u>

- van Gerwen M, Colicino E, Guan H, Dolios G, Nadkarni GN, Vermeulen RCH, et al. (2023). Per- and polyfluoroalkyl substances (PFAS) exposure and thyroid cancer risk. *EBioMedicine*. 97:104831. doi:10.1016/j. ebiom.2023.104831 PMID:37884429
- van Leeuwen SP, Karrman A, van Bavel B, de Boer J, Lindstrom G (2006). Struggle for quality in determination of perfluorinated contaminants in environmental and human samples. *Environ Sci Technol.* 40(24):7854– 60. doi:10.1021/es061052c PMID:17256538
- Velarde MC, Chan AFO, Sajo MEJV, Zakharevich I, Melamed J, Uy GLB, et al. (2022). Elevated levels of perfluoroalkyl substances in breast cancer patients within the Greater Manila area. *Chemosphere*. 286(Pt 1):131545. doi:10.1016/j.chemosphere.2021.131545 PMID:34293563
- Vendl C, Pottier P, Taylor MD, Bräunig J, Gibson MJ, Hesselson D, et al. (2022). Thermal processing reduces PFAS concentrations in blue food – a systematic review and meta-analysis. *Environ Pollut*. 304:119081. doi:10.1016/j.envpol.2022.119081 PMID:35367104
- Vestergren R, Berger U, Glynn A, Cousins IT (2012). Dietary exposure to perfluoroalkyl acids for the Swedish population in 1999, 2005 and 2010. *Environ Int.* 49:120–7. doi:<u>10.1016/j.envint.2012.08.016</u> PMID:<u>23018201</u>
- Vestergren R, Cousins IT (2009). Tracking the pathways of human exposure to perfluorocarboxylates. *Environ Sci Technol*. 43(15):5565–75. doi:<u>10.1021/es900228k</u> PMID:<u>19731646</u>
- Vestergren R, Cousins IT, Trudel D, Wormuth M, Scheringer M (2008). Estimating the contribution of precursor compounds in consumer exposure to PFOS and PFOA. *Chemosphere*. 73(10):1617–24. doi:10.1016/j. chemosphere.2008.08.011 PMID:18834614
- Vestergren R, Herzke D, Wang T, Cousins IT (2015). Are imported consumer products an important diffuse source of PFASs to the Norwegian environment? *Environ Pollut*. 198:223–30. doi:10.1016/j. envpol.2014.12.034 PMID:25644935
- Vieira VM, Hoffman K, Shin HM, Weinberg JM, Webster TF, Fletcher T (2013). Perfluorooctanoic acid exposure and cancer outcomes in a contaminated community: a geographic analysis. *Environ Health Perspect.* 121(3):318–23. doi:<u>10.1289/ehp.1205829</u> PMID:<u>23308854</u>
- Vorkamp K, Castaño A, Antignac JP, Boada LD, Cequier E, Covaci A, et al. (2021). Biomarkers, matrices and analytical methods targeting human exposure to chemicals selected for a European human biomonitoring initiative. *Environ Int.* 146:106082. doi:10.1016/j. envint.2020.106082 PMID:33227583
- Wang H, Li W, Yang J, Wang Y, Du H, Han M, et al. (2023). Gestational exposure to perfluoroalkyl substances is associated with placental DNA methylation and birth size. *Sci Total Environ*. 858(Pt 1):159747. doi:<u>10.1016/j.</u> <u>scitotenv.2022.159747</u> PMID:<u>36309289</u>

- Wang P, Zhang M, Li Q, Lu Y (2021). Atmospheric diffusion of perfluoroalkyl acids emitted from fluorochemical industry and its associated health risks. *Environ Int.* 146:106247. doi:10.1016/j.envint.2020.106247
 PMID:33276313
- Wang Q-W, Yang G-P, Zhang Z-M, Zhang J (2018b). Optimization of sample preparation and chromatography for the determination of perfluoroalkyl acids in sediments from the Yangtze estuary and East China Sea. *Chemosphere*. 205:524–30. doi:10.1016/j. <u>chemosphere.2018.04.143</u> PMID:29709802
- Wang Y, Fu J, Wang T, Liang Y, Pan Y, Cai Y, et al. (2010). Distribution of perfluorooctane sulfonate and other perfluorochemicals in the ambient environment around a manufacturing facility in China. *Environ Sci Technol.* 44(21):8062–7. doi:<u>10.1021/es101810h</u> PMID:<u>20879709</u>
- Wang Y, Zhong Y, Li J, Zhang J, Lyu B, Zhao Y, et al. (2018a). Occurrence of perfluoroalkyl substances in matched human serum, urine, hair and nail. *J Environ Sci* (*China*). 67:191–7. doi:10.1016/j.jes.2017.08.017 PMID:29778152
- WangZ, CousinsIT, ScheringerM, BuckRC, Hungerbühler K (2014). Global emission inventories for C4–C14 perfluoroalkyl carboxylic acid (PFCA) homologues from 1951 to 2030, part I: production and emissions from quantifiable sources. *Environ Int.* 70:62–75. doi:<u>10.1016/j.envint.2014.04.013</u> PMID:<u>24932785</u>
- Washington JW, Henderson WM, Ellington JJ, Jenkins TM, Evans JJ (2008). Analysis of perfluorinated carboxylic acids in soils II: optimization of chromatog-raphy and extraction. *J Chromatogr A*. 1181(1–2):21–32. doi:10.1016/j.chroma.2007.12.042 PMID:18201708
- Watkins DJ, Wellenius GA, Butler RA, Bartell SM, Fletcher T, Kelsey KT (2014). Associations between serum perfluoroalkyl acids and LINE-1 DNA methylation. *Environ Int.* 63:71–6. doi:<u>10.1016/j.envint.2013.10.018</u> PMID:<u>24263140</u>
- Wee SY, Aris AZ (2023). Revisiting the "forever chemicals", PFOA and PFOS exposure in drinking water. *NPJ Clean Water*. 6(1):57. doi:10.1038/s41545-023-00274-6
- Wen X, Wang M, Xu X, Li T (2022). Exposure to perand polyfluoroalkyl substances and mortality in US adults: a population-based cohort study. *Environ Health Perspect*. 130(6):67007. doi:<u>10.1289/EHP10393</u> PMID:<u>35731224</u>
- Whitehead HD, Venier M, Wu Y, Eastman E, Urbanik S, Diamond ML, et al. (2021). Fluorinated compounds in North American cosmetics. *Environ Sci Technol Lett*. 8(7):538–44. doi:10.1021/acs.estlett.1c00240
- WHO (2016). Dioxins. Effects on human health. Geneva, Switzerland: World Health Organization. Available from: <u>https://www.who.int/news-room/fact-sheets/detail/dioxins-and-their-effects-on-human-health#:~:text=Short%2Dterm%20exposure%20</u>

of%20humans,endocrine%20system%20and%20 reproductive%20functions, accessed November 2023.

- Wielsøe M, Kern P, Bonefeld-Jørgensen EC (2017). Serum levels of environmental pollutants is a risk factor for breast cancer in Inuit: a case control study. *Environ Health*. 16(1):56. doi:<u>10.1186/s12940-017-0269-6</u> PMID:<u>28610584</u>
- Wilhelm M, Kraft M, Rauchfuss K, Hölzer J (2008). Assessment and management of the first German case of a contamination with perfluorinated compounds (PFC) in the region Sauerland, North Rhine– Westphalia. J Toxicol Environ Health A. 71(11–12):725– 33. doi:10.1080/15287390801985216 PMID:18569570
- Wilson TB, Stevenson G, Crough R, de Araujo J, Fernando N, Anwar A, et al. (2021). Evaluation of residues in hen eggs after exposure of laying hens to water containing per- and polyfluoroalkyl substances. *Environ Toxicol Chem.* 40(3):735–43. doi:10.1002/etc.4723 PMID:32274818
- Winkens K, Koponen J, Schuster J, Shoeib M, Vestergren R, Berger U, et al. (2017). Perfluoroalkyl acids and their precursors in indoor air sampled in children's bedrooms. *Environ Pollut.* 222:423–32. doi:10.1016/j. envpol.2016.12.010 PMID:28012670
- Winquist A, Hodge JM, Diver WR, Rodriguez JL, Troeschel AN, Daniel J, et al. (2023). Case-cohort study of the association between PFAS and selected cancers among participants in the American Cancer Society's Cancer Prevention Study II LifeLink cohort. *Environ Health Perspect*. 131(12):127007. doi:10.1289/EHP13174 PMID:38088576
- Worley RR, Moore SM, Tierney BC, Ye X, Calafat AM, Campbell S, et al. (2017). Per- and polyfluoroalkyl substances in human serum and urine samples from a residentially exposed community. *Environ Int.* 106:135–43. doi:<u>10.1016/j.envint.2017.06.007</u> PMID: <u>28645013</u>
- Woskie SR, Gore R, Steenland K (2012). Retrospective exposure assessment of perfluorooctanoic acid serum concentrations at a fluoropolymer manufacturing plant. *Ann Occup Hyg.* 56(9):1025–37. doi:10.1093/ annhyg/mes023 PMID:22539556
- Wu N, Cai D, Guo M, Li M, Li X (2019). Per- and poly-fluorinated compounds in saleswomen's urine linked to indoor dust in clothing shops. *Sci Total Environ*. 667:594–600. doi:<u>10.1016/j.scitotenv.2019.02.287</u>
 PMID:<u>30833258</u>
- Xie MY, Sun XF, Wu CC, Huang GL, Wang P, Lin ZY, et al. (2023). Glioma is associated with exposure to legacy and alternative per- and polyfluoroalkyl substances. *J Hazard Mater.* 441:129819. doi:<u>10.1016/j.</u> jhazmat.2022.129819 PMID:<u>36084455</u>
- Xie Z, Wang Z, Magand O, Thollot A, Ebinghaus R, Mi W, et al. (2020). Occurrence of legacy and emerging organic contaminants in snow at Dome C in the

Antarctic. *Sci Total Environ*. 741:140200. doi:<u>10.1016/j.</u> scitotenv.2020.140200 PMID:<u>32599399</u>

- Xu B, Liu S, Zhou JL, Zheng C, Weifeng J, Chen B, et al. (2021a). PFAS and their substitutes in groundwater: occurrence, transformation and remediation. *J Hazard Mater.* 412:125159. doi:10.1016/j.jhazmat.2021.125159 PMID:<u>33951855</u>
- Xu Y, Nielsen C, Li Y, Hammarstrand S, Andersson EM, Li H, et al. (2021b). Serum perfluoroalkyl substances in residents following long-term drinking water contamination from firefighting foam in Ronneby, Sweden. *Environ Int.* 147:106333. doi:10.1016/j. envint.2020.106333 PMID:33360412
- Ye F, Zushi Y, Masunaga S (2015). Survey of perfluoroalkyl acids (PFAAs) and their precursors present in Japanese consumer products. *Chemosphere*. 127:262–8. doi:<u>10.1016/j.chemosphere.2015.02.026</u> PMID:<u>25753850</u>
- Yeung LW, De Silva AO, Loi EI, Marvin CH, Taniyasu S, Yamashita N, et al. (2013). Perfluoroalkyl substances and extractable organic fluorine in surface sediments and cores from Lake Ontario. *Environ Int*. 59:389–97. doi:10.1016/j.envint.2013.06.026 PMID:23911339
- Yeung LW, So MK, Jiang G, Taniyasu S, Yamashita N, Song M, et al. (2006). Perfluorooctanesulfonate and related fluorochemicals in human blood samples from China. *Environ Sci Technol*. 40(3):715–20. doi:<u>10.1021/ es052067y</u> PMID:<u>16509308</u>
- Young AS, Sparer-Fine EH, Pickard HM, Sunderland EM, Peaslee GF, Allen JG (2021). Per- and polyfluoroalkyl substances (PFAS) and total fluorine in fire station dust. *J Expo Sci Environ Epidemiol*. 31(5):930–42. doi:<u>10.1038/s41370-021-00288-7</u> PMID:<u>33542478</u>
- Young W, Wiggins S, Limm W, Fisher CM, DeJager L, Genualdi S (2022). Analysis of per- and poly(fluoroalkyl) substances (PFASs) in highly consumed seafood products from US markets. J Agric Food Chem. 70(42):13545–53. doi:<u>10.1021/acs.jafc.2c04673</u> PMID:<u>36251396</u>
- Yu S, Liu W, Xu Y, Zhao Y, Wang P, Wang X, et al. (2018). Characteristics of perfluoroalkyl acids in atmospheric PM₁₀ from the coastal cities of the Bohai and Yellow Seas, northern China. *Environ Pollut.* 243 (Pt B):1894– 903. doi:10.1016/j.envpol.2018.10.002 PMID:30408878
- Zacs D, Fedorenko D, Pasecnaja E, Bartkevics V (2023). Application of nano-LC-nano-ESI-Orbitrap-MS for trace determination of four priority PFAS in food products considering recently established tolerable weekly intake (TWI) limits. *Anal Chim Acta*. 1251:341027. doi:<u>10.1016/j.aca.2023.341027</u> PMID:<u>36925299</u>
- Zhang H, Vestergren R, Wang T, Yu J, Jiang G, Herzke D (2017a). Geographical differences in dietary exposure to perfluoroalkyl acids between manufacturing and application regions in China. *Environ Sci Technol.* 51(10):5747–55. doi:10.1021/acs.est.7b00246 PMID:28387520

- Zhang L, Liu J, Hu J, Liu C, Guo W, Wang Q, et al. (2012). The inventory of sources, environmental releases and risk assessment for perfluorooctane sulfonate in China. *Environ Pollut*. 165:193–8. doi:10.1016/j. envpol.2011.09.001 PMID:21995873
- Zhang T, Fu S, Yu K, Albanes D, Moore SC, Purdue MP, et al. (2023). Nested case–control studies investigating serum perfluorooctanoate and perfluorooctane sulfonate levels and pancreatic ductal adenocarcinoma in two cohorts. *Environ Health Perspect*. 131(10):107702. doi:10.1289/EHP13208 PMID:37844029
- Zhang T, Sun H, Gerecke AC, Kannan K, Müller CE, Alder AC (2010). Comparison of two extraction methods for the analysis of per- and polyfluorinated chemicals in digested sewage sludge. *J Chromatogr A*. 1217(31):5026–34. doi:10.1016/j.chroma.2010.05.061 PMID:20580368
- Zhang W, Lin Z, Hu M, Wang X, Lian Q, Lin K, et al. (2011). Perfluorinated chemicals in blood of residents in Wenzhou, China. *Ecotoxicol Environ Saf.* 74(6):1787–93. doi:10.1016/j.ecoenv.2011.04.027 PMID:21570120
- Zhang Y, Mustieles V, Sun Y, Oulhote Y, Wang YX, Messerlian C (2022). Association between serum perand polyfluoroalkyl substances concentrations and common cold among children and adolescents in the United States. *Environ Int.* 164:107239. doi:<u>10.1016/j. envint.2022.107239</u> PMID:<u>35447424</u>
- Zheng G, Boor BE, Schreder E, Salamova A (2020). Indoor exposure to per- and polyfluoroalkyl substances (PFAS) in the childcare environment. *Environ Pollut*. 258:113714. doi:<u>10.1016/j.envpol.2019.113714</u> PMID:<u>31901805</u>
- Zheng G, Eick SM, Salamova A (2023). Elevated levels of ultrashort- and short-chain perfluoroalkyl acids in US homes and people. *Environ Sci Technol*. 57(42):15782– 93. doi:10.1021/acs.est.2c06715 PMID:<u>37818968</u>
- Zheng G, Schreder E, Dempsey JC, Uding N, Chu V, Andres G, et al. (2021). Per- and polyfluoroalkyl substances (PFAS) in breast milk: concerning trends for current-use PFAS. *Environ Sci Technol*. 55(11):7510– 20. doi:10.1021/acs.est.0c06978 PMID:33982557

- Zhou J, Baumann K, Chang N, Morrison G, Bodnar W, Zhang Z, et al. (2022). Per- and polyfluoroalkyl substances (PFASs) in airborne particulate matter (PM_{2.0}) emitted during floor waxing: a pilot study. *Atmos Environ*. 268:118845. doi:<u>10.1016/j.</u> <u>atmosenv.2021.118845</u> PMID:<u>34899026</u>
- Zhou J, Baumann K, Mead RN, Skrabal SA, Kieber RJ, Avery GB, et al. (2021). PFOS dominates PFAS composition in ambient fine particulate matter (PM_{2.5}) collected across North Carolina nearly 20 years after the end of its US production. *Environ Sci Process Impacts*. 23(4):580–7. doi:10.1039/D0EM00497A PMID:33725038
- Zhou Z, Shi Y, Vestergren R, Wang T, Liang Y, Cai Y (2014). Highly elevated serum concentrations of perfluoroalkyl substances in fishery employees from Tangxun Lake, China. *Environ Sci Technol*. 48(7):3864–74. doi:<u>10.1021/ es4057467</u> PMID:<u>24588690</u>
- Zhu H, Kannan K (2020). Total oxidizable precursor assay in the determination of perfluoroalkyl acids in textiles collected from the United States. *Environ Pollut*. 265(Pt B):114940. doi:<u>10.1016/j.envpol.2020.114940</u> PMID:<u>32540590</u>
- Zhu L, Hajeb P, Fauser P, Vorkamp K (2023). Endocrine disrupting chemicals in indoor dust: a review of temporal and spatial trends, and human exposure. *Sci Total Environ*. 874:162374. doi:10.1016/j. scitotenv.2023.162374 PMID:36828075
- Zushi Y, Tamada M, Kanai Y, Masunaga S (2010). Time trends of perfluorinated compounds from the sediment core of Tokyo Bay, Japan (1950s–2004). *Environ Pollut*. 158(3):756–63. doi:<u>10.1016/j.envpol.2009.10.012</u> PMID:<u>19883962</u>

2. CANCER IN HUMANS

Since the previous evaluation of perfluorooctanoic acid (PFOA) by the IARC Monographs programme in 2014 (Volume 110; IARC, 2016), new epidemiological studies have investigated the occurrence of cancer in relation to exposure to PFOA and to perfluorooctanesulfonic acid (PFOS). A comprehensive search was conducted to identify the studies reporting cancer outcomes (defined as incidence or mortality) that were considered in the present evaluation, including studies of cohorts with occupational and high environmental exposure to PFOA or PFOS; prospective nested case-control or case-cohort studies in populations with background levels of exposure; case-control studies evaluating exposure to PFOA and/or perfluoroalkyl and polyfluoroalkyl substance(s) (PFAS), assessed after a cancer diagnosis; and an ecological study in a population with high contrast (determined through measured serum concentrations) between exposure to PFOA and relatively low exposures to other PFAS. The search identified several cohorts that were each reported in multiple publications and included the continuation of follow-up for cancer occurrence over time; in these instances, detailed reviews were conducted only for the most recent or most informative studies in a given cohort.

The Working Group excluded one ecological study of mortality conducted in the Veneto region of Italy, an area with a high level of PFOA contamination, because that study had notable limitations (Mastrantonio et al., 2018). The crude exposure assessment used (contaminated versus uncontaminated area) was based on drinking-water measurements without biological measurements in the population. Although serum concentrations were later assessed in a younger population (aged 15-39 years) in this region (<u>Pitter et al., 2020</u>), they were not available for the older population pertinent to the outcome investigated in the study (mortality). Human biomonitoring subsequent to the publication of Mastrantonio et al. (2018) detected substantial exposure in some areas previously classified as unexposed; this would have biased estimates towards the null value. In addition, many of the risk ratios reported for men and women combined fell outside of the range of the sex-specific risk ratios reported in the study, making it difficult to interpret the findings.

In total, 36 studies were reviewed in detail: 21 cohort studies (also comprising prospective nested case-control or case-cohort studies), some describing different cancer sites in several publications; 11 case-control studies; and 4 meta-analyses. In addition, the Working Group conducted an ecological analysis of orchiectomy rates as a surrogate for testicular cancer in residents of a contaminated area of northern Italy (see Section 2.3, and Annex 3, Supplementary analyses used in reviewing evidence on cancer in

humans, available from: https://publications.iarc. who.int/636). Section 2.1 summarizes the cohort studies, nested case-control and nested casecohort studies, and two case-control studies on multiple cancer sites (reported in Vieira et al., 2013). Results for specific cancer sites are summarized in Sections 2.2 to 2.7, with findings from cohort and nested case-control or case-cohort studies described first, followed by findings obtained using other study designs. Studies of breast cancer were further sorted by design, with separate subsections for cohort-based studies and case-control studies or meta-analyses. The Working Group also conducted a meta-analysis of studies on kidney cancer, as well as a methodological simulation study to evaluate the representativeness of serum PFOA measurements from a single time point as a surrogate for longerterm measurements (over a period of 5-8 years); this is summarized in Annex 3 (Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications.iarc. who.int/636). Finally, a synthesis of the evidence relating to cancer in humans is presented in Section 2.8.

2.1 Cohort descriptions

See <u>Table 2.1</u>.

2.1.1 PFOA-production workers (Cottage Grove, Minnesota, USA)

<u>Raleigh et al. (2014)</u> updated data on cancer incidence and mortality in a previously investigated (<u>Gilliland and Mandel, 1993; Lundin et al.,</u> 2009) cohort of workers exposed to PFOA at a facility manufacturing ammonium perfluorooctanoate (APFO, the ammonium salt of PFOA) (the Cottage Grove plant) in Minneapolis, Minnesota, USA. The cohort included 4668 workers (men, 79%) who were employed for \geq 1 year between 1947 (when production of APFO was initiated) and 2002 (when production was terminated). A reference population was also followed, this being a cohort of 4359 workers (men, 88%) who were employed for \geq 1 year before 1999 at a tape and abrasive production facility (the Saint Paul plant) where there was no production of APFO and that was located in the same suburban area and managed by the same company as the APFO-manufacturing facility.

Workers at the Cottage Grove plant were exposed by inhalation of PFOA vapour and ammonium salt particulates during regular production, through cleaning of equipment, changing filters, quality control checks, and maintenance, and through bystander exposure. For all cohort members, individual exposure by inhalation to APFO (in mg/m³ of air), as a daily time-weighted average (TWA), was estimated from work history records (period, department, job title), industrial hygiene monitoring data (205 personal samples and 659 area samples collected in 1977-2000, from all processes and tasks in APFO-production areas in the chemical division of the Cottage Grove plant), information from former and current workers and from industrial hygiene professionals, and APFOproduction levels. Daily TWAs for jobs in APFO production ranged from 1×10^{-4} to 4.0×10^{-1} g/m³ $[0.1 \ \mu g/m^3$ to 400 $\mu g/m^3]$. Exposures for non-APFO production jobs in the chemical division and the non-chemical division ranged, according to expert judgement, from 1×10^{-8} to 3×10^{-5} mg/m³ [1 × 10⁻⁵ and 3×10^{-2} µg/m³] and from 1×10^{-8} to 1×10^{-6} mg/m³ $[1 \times 10^{-5}]$ and $1 \times 10^{-3} \,\mu\text{g/m}^3$], respectively. To account for ubiquitous background exposure, all workers (including the reference population) were assigned an exposure that was one order of magnitude lower than that for the workers in the Cottage Grove non-chemical division. The final cumulative exposure metric was quartiles of µg/ m³-years. Medical surveillance of 148 workers employed in the Cottage Grove chemical division in 2000 found a geometric mean serum concentration of PFOA of 815 ng/mL (2538, 979, and

Table 2.1 Description of cohort studies (including nested case-control studies) on exposure to PFOA or PFOS and cancer

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Lundin et al. (2009) MN, USA Enrolment, 1947–1997/ follow-up, 1947–2002 (mortality) Cohort	3993 employees; Cottage Grove (MN) PFOA cohort; workers employed at an APFO-production plant for ≥ 365 days before 31 December 1997; most recent follow-up for some cancer sites (see those listed here), later follow-up by <u>Raleigh et al. (2014)</u> Exposure assessment method: based on job history; jobs classified as definite, probable, and no or minimal occupational APFO exposure	Large intestine, rectum, oesophagus, stomach Thyroid CNS Lymphatic and haematopoietic Lymphosarcoma- reticulosarcoma Hodgkin lymphoma Leukaemia	See Table S2.5 ^a See Table S2.6 ^a	Exposure assessment critique:Key strengths were the industrial hygienereview of jobs; weighting of jobs based on serummeasurements; assignment of exposure weightsbased on blood monitoring (authors indicatedthat other weights were considered but felt thatthese weights allowed better differentiationbetween probable and definite exposures overtime).Key limitations were the crude exposureassessment by job classification, and lack of job-specific data on PFOA serum levels (but serumPFOA levels for work areas were collected in2000).Other strengths: Occupational cohort withrelatively high exposures; analyses presentedbased on both job classification and cumulativeexposure estimates.Other limitations: Small occupational cohortwith limited number of deaths; potentialhealthy-worker effect due to external comparisonof rates from general population; limitedinformation on covariates. Smoking data werecollected but not included in the final models.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Raleigh et al. (2014) MN, USA Enrolment, 1947–2002/ follow-up, 1947–2008 (mortality), 1988–2008 (incidence) Cohort	9027 employees (4668 exposed workers, 4359 reference workers); Cottage Grove (MN) PFOA cohort latest update (previous: Gilliland and Mandel, 1993, and Lundin et al., 2009); workers employed for ≥ 1 yr in 1947–2002 at an APFO facility (Cottage Grove; n = 4668); reference workers without any exposure to APFO employed at a tape and abrasives production facility located in the same suburban geographical area and managed by the same company (Saint Paul; $n = 4359$) Exposure assessment method: exposure matrix used production-process air measurements and expert judgement in applying production volume data and proximity to production areas to assign department and job exposures historically; exposure matrix and job history were used to calculate cumulative exposure ($\mu g/m^3$ -years)	Kidney Urinary bladder Prostate Breast Liver Pancreas All cancers combined	See <u>Table 2.2</u> See <u>Table 2.2</u> See <u>Table 2.3</u> See <u>Table 2.4</u> See Table S2.5 ^a See Table S2.5 ^a See Table S2.7 ^a	<i>Exposure assessment critique</i> : Key strengths were that the only PFAS exposure in plant was to APFO (PFOA); cumulative co- exposure to TFE was estimated to be minimal Key limitations were that development of cumulative exposure metric was based on APFO air concentrations and not internal dose; reference population in APFO unexposed same plant was assumed to have exposures of the general population, but the method of determination was unclear (1×10^{-7} to 1×10^{-9} mg/m ³). <i>Other strengths</i> : A reference population sharing similar socioeconomic characteristics as the exposed population and a long follow-up period. <i>Other limitations</i> : Lacking data on cancer- incidence before start of follow-up up to 40 yr after first exposure. No information on health behaviours (potential confounding). Small numbers of cancers of kidney, pancreas, liver, testis. No accounting for alcohol or smoking.
Alexander et al. (2003) Decatur (AL), USA Enrolment, 1961–1997/ follow-up, 1961–1998 (mortality) Cohort	2083 employees; Decatur (AL) PFOS cohort; production workers (men, 83%) who worked \geq 365 days in a plant producing speciality films and fluorochemicals, a main one being POSF; most recent follow-up of all cancers except bladder, which is described in a later study by <u>Alexander and Olsen</u> (2007) Exposure assessment method: expert judgement; workers were categorized as ever in a "high" exposure job, ever in a "low" exposure job but not a "high" exposure job, only in jobs without POSF exposure, or \geq 1 yr in a "high" exposure job	Breast Liver and bile ducts, large intestine, oesophagus, digestive organs and peritoneum Lymphatic and haematopoietic Melanoma Respiratory system Bronchus, trachea, lung All cancers combined	See <u>Table 2.4</u> See Table S2.5 ^a See Table S2.6 ^a See Table S2.6 ^a See Table S2.6 ^a See Table S2.6 ^a See Table S2.7 ^a	<i>Exposure assessment critique:</i> A key strength was the range of workplace exposure durations and levels (a large exposure contrast). Key limitations were that exposure assessment did not use any measure of cumulative exposure, but simply categorized each worker in 1 of 3 ever/never/only job classifications, which could produce exposure misclassification; many likely co-exposures to potential carcinogens or other fluorochemicals, including PFOA (however, PFOA concentrations were probably low). <i>Other limitations:</i> Occupational cohort with few cancer deaths (overall, 39; high exposure group, 18), limited to mortality, lack of data on smoking and alcohol, mostly male (83%).

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Alexander and Olsen (2007) Decatur (AL), USA Enrolment, 1961–1997/ follow-up, 1970–2002 (mortality and incidence) Cohort	1588; Decatur (AL) PFOS cohort; production workers in the <u>Alexander</u> <u>et al. (2003)</u> cohort; living cohort members completed a questionnaire (response rate, 73.9%) to identify incident bladder cancer cases; bladder cancer decedents were identified using underlying cause of death from death certificates; analyses excluded 495 living cohort members who did not return the questionnaire Based on the exposure assessment described in <u>Alexander et al. (2003)</u> , cumulative exposure was calculated weighing the exposure categories of nonexposed, low exposed and high exposed with a factor of 1, 3, and 10, respectively, for each year in that job.	Urinary bladder	See <u>Table 2.2</u>	<i>Exposure assessment critique:</i> Key strengths were the range of workplace exposure durations and levels; cumulative exposure was estimated using a weighted approach of exposure categories. Key limitations were that the crude weighted approach to calculate cumulative exposure could produce exposure misclassification; many likely co-exposures to potential carcinogens or other fluorochemicals (however, PFOA concentrations were probably low). <i>Other strengths:</i> use of incidence data with 74% participation rate in survey; attempt to validate self-reported cancer for survey respondents. <i>Other limitations:</i> occupational cohort with only 11 cases of bladder cancer, 2 in the highest category of exposure. Bladder cancer incidence identified by survey of cohort (6 cases) and death certificates (5 deaths) no cancer registry matching, only partial data on smoking, no ability to validate 5 cases of bladder cancer identified by death certificate, mostly male (83%).
Leonard et al. (2008) Parkersburg (WV), USA Enrolment, 1948– 2002/follow-up, 1948–2002 (mortality) Cohort	6027 workers; Parkersburg (WV), polymer-production PFOA cohort; most recent follow-up for some cancer sites (see those listed here), later follow-up by <u>Steenland and Woskie (2012)</u> ; workers (men, 81%) at a polymer-manufacturing facility for ≥ 1 day in 1948–2002 Exposure assessment method: no quantitative exposure assessment; workers in a polymer-production facility were identified using the company's administrative records; ~30% worked in processes using APFO; all participants had detectable levels of serum PFOA	Large intestine, rectum, oesophagus, stomach Thyroid Melanoma	See Table S2.5 ^a See <u>Table 2.4</u> See Table S2.6 ^a	Strengths: Occupational cohort with relatively high exposures. Complete cohort ascertainment and follow-up. Local reference groups increase comparability with respect to socioeconomic factors and health behaviours. <i>Limitations</i> : No assessment of exposure to specific chemicals (the company used a wide variety of chemicals including PFOA). Small numbers.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Steenland and Woskie (2012) Parkersburg (WV), USA Enrolment, 1948– 2002/follow-up, 1952–2008 (mortality) Cohort	5791 workers; Parkersburg (WV), polymer-production PFOA cohort; workers (men, 81%) at a polymer- manufacturing facility who had potential exposure to fluoropolymers and sufficiently detailed work histories Exposure assessment method: JEM was based on a total of eight job category/job group combinations; jobs were classified on the basis of PFOA exposure potential and the JEM was improved through the use of blood samples to assign serum PFOA levels over time	Kidney Urinary bladder Testis Prostate Breast Liver and gallbladder Pancreas NHL Leukaemia Lung Mesothelioma All cancers combined	See <u>Table 2.2</u> See <u>Table 2.2</u> See <u>Table 2.3</u> See <u>Table 2.3</u> See <u>Table 2.4</u> See Table S2.5 ^a See Table S2.6 ^a	<i>Exposure assessment critique:</i> Key strengths were that the JEM incorporated changes in exposure over time; because serum levels were used to construct the JEM, residential exposure to PFOA in drinking- water was included in estimates; any exposure misclassification is likely to be non-differential. A key limitation was the lack of description of other exposures. <i>Other strengths:</i> Ability to evaluate associations with PFOA in a population exposed to levels much higher than in the general population. <i>Other limitations:</i> Limited ability to evaluate mortality for some cancers due to small numbers of deaths, particularly for cancers among women (given the small number of female workers in the study) and cancers that are relatively rare and/or less likely to be fatal.
Steenland et al. (2015) Parkersburg (WV), USA Enrolment, 1948– 2002/follow-up, 1951 to interview date in 2008–2011 (incidence) Cohort	3713 employees; a subset of the Parkersburg (WV) polymer-production PFOA cohort in <u>Steenland and Woskie</u> (2012); polymer-production workers (men, 80%) who responded (self or next- of-kin) to a questionnaire about health outcomes and for whom measured or estimated occupational and residential exposure estimates were available Exposure assessment method: cumulative PFOA serum concentrations estimated on the basis of JEMs and residential history; historical PFOA serum levels were modelled via a JEM based on > 2000 serum measurements (<u>Woskie et al., 2012</u>); non-occupational exposure from drinking-water (address- based) was also estimated:	Urinary bladder Prostate Colon and rectum Melanoma	See <u>Table 2.2</u> See <u>Table 2.3</u> See Table S2.5 ^a See Table S2.6 ^a	Exposure assessment critique: Key strengths were good characterization of exposure in both occupational and residential settings; minimal potential for non-differential exposure misclassification. A key limitation was that loss to follow-up of 40% of workers could lead to differential exposure misclassification if related to PFOA exposure. Other strengths: Ability to evaluate associations between PFOA and cancer incidence in a population exposed to levels much higher than in the general population. Use of medical records to confirm self-reported cancer diagnoses likely reduced non-differential outcome misclassification.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Steenland et al. (2015) Parkersburg (WV), USA Enrolment, 1948– 2002/follow-up, 1951 to interview date in 2008–2011 (incidence) Cohort (cont.)	yearly serum estimates from the occupational exposure model were used for the years when people worked at the plant if these were higher than residential estimates, or if they were lower, the residential (community) estimates were used			Other limitations: Possibility of selection bias as the investigation included only 62% of the target population; relatively small numbers of validated cancer cases and inability to evaluate less-common malignancies. Possible under- ascertainment of cases due to medical record confirmation.
Eriksen et al. (2009) Denmark Enrolment, 1 December 1993 to 31 May 1997/follow-up, 1 December 1993 to 1 July 2006 (incidence) Case–cohort	Case-cohort within the Diet, Cancer, and Health cohort, which included men and women aged 50–65 yr without cancer at enrolment. Cases: urinary bladder, 332; prostate, 713; liver, 67; pancreas, 128; incident cases identified through cancer registry linkage Comparison cohort: 772 (680 men, 92 women); subcohort of participants randomly selected without cancer at the end of follow-up Exposure assessment method: quantitative plasma measurements; analytical method was state-of-the-art; a single sample collected at enrolment (1993–1997) was analysed for PFOA and PFOS	Urinary bladder Prostate Liver Pancreas	See <u>Table 2.2</u> See <u>Table 2.3</u> See Table S2.5 ^a See Table S2.5 ^a	<i>Exposure assessment critique:</i> Key strengths were that plasma levels measuredat baseline represent the combined exposurethrough all exposure pathways; measurementerror low.Key limitations were that single samplescollected at time of enrolment may notreflect exposure at crucial windows in cancerdevelopment; measured only PFOA and PFOSand no information on exposure to other PFAS. <i>Other strengths:</i> Large cohort with numerousincident cancers (n = 1240) followed 0–12 yrafter baseline enrolment; control of confounders;internal comparison; low loss to follow-up. <i>Other limitations:</i> Low exposure contrast in apopulation with background exposure levels.Analyses only considered PFOA and PFOSseparately.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Barry et al. (2013) Mid-Ohio Valley (OH and WV), USA Enrolment, August 2005 to August 2006/ follow-up, 1952 to 2011 (incidence) Cohort	32 254 (28 541 community members and 3713 workers); C8 Science Panel Study; included people enrolled in the C8 Health Project who lived, worked, or attended school for ≥ 1 yr between 1950 and 3 December 2004 in a district with contaminated water in the vicinity of a chemical plant using PFOA in manufacturing processes (Parkersburg, WV, polymer-production facility), as well as a subset of those from the original Parkersburg (WV) polymer- production PFOA occupational cohort who worked at the plant between 1948 and 2002 Exposure assessment method: residential and occupational exposure estimates were combined to estimate cumulative PFOA serum concentrations; historical PFOA serum levels were modelled via a JEM based on > 2000 serum measurements (Woskie et al., 2012); non-occupational exposure from drinking-water (address-based) was also estimated	Kidney Urinary bladder Testis Prostate Breast Liver, pancreas, colon and rectum, oesophagus, stomach Thyroid Brain Leukaemia, lymphoma Lung Melanoma	See <u>Table 2.2</u> See <u>Table 2.3</u> See <u>Table 2.3</u> See <u>Table 2.4</u> See Table S2.5 ^a See <u>Table 2.4</u> See Table S2.6 ^a See Table S2.6 ^a See Table S2.6 ^a See Table S2.6 ^a	Exposure assessment critique:Key strengths were that exposure assessment wasdone the same way for all participants; estimatesaccounted for both residential and occupationalexposure to PFOA.A key limitation was that serum PFOA levelswere available only in 2005–2006.Other strengths: Wide range of PFOA exposurelevels; authors presented both no lag and 10-yrlag models; availability of detailed informationon potential confounding factors; relatively highparticipation rates; and validation of cancerdiagnoses through medical chart review andstate registries.Other limitations: possibility of selection bias,particularly for cancers with a high rate offatality; and relatively few validated cases forprospective analyses (after C8 Health Projectenrolment). Potential limitation of a survivorcohort but unlikely to be biased unless thosewith higher exposure had lower post-diagnosissurvival rates than those with lower exposure(Barry et al., 2015).Other comments: 62% of the polymer productionplant cohort (Steenland and Woskie, 2012) isincluded in the study population (includingworkers who did and did not participate in theC8 Health Project).

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Consonni et al. (2013) USA, UK, Italy, Germany, Netherlands Enrolment, 1950– 2002/follow-up, 1950–2008 Cohort	5879 male workers (APFO-exposed, 4205); The pooled international TFE cohort included male workers who were ever employed or employed for a minimum of 6 or 12 mo at one or more of six TFE production sites in North America and Europe between 1950 and 2002; the principal occupational exposures were TFE and APFO (facilitates production of TFE) Exposure assessment method: a JEM provided yearly semiquantitative estimates (in arbitrary units) of TFE and APFO exposure for relevant job titles at each production site, from the start of TFE production to 2002 (Sleeuwenhoek	Kidney and other organs of the urinary tract Urinary bladder Testis Prostate Liver and intrahepatic bile duct, pancreas, colon, rectum, oesophagus, stomach Brain Lymphatic and haematopoietic, NHL, multiple myeloma, leukaemia	See <u>Table 2.2</u> See <u>Table 2.2</u> See <u>Table 2.3</u> See <u>Table 2.3</u> See Table S2.5 ^a See Table S2.6 ^a See Table S2.6 ^a	Exposure assessment critique:A key strength was the availability of job historyfor all participants.Key limitations were that only expert judgementwas used to determine exposure levels; nomeasured exposures; high correlations betweenexposure to TFE monomer (IARC Group 2A)and PFOA, which precludes evaluation of effectsof the individual compounds.Other strengths: The cohort included all TFEproduction sites worldwide during the entireperiod of production and had almost completeenrolment and follow-up.Other limitations: low statistical power to detectrisk of rare cancers.
	and Cherrie, 2012)	All cancers combined	See Table S2.0 ^a	
Ghisari et al. (2017) Denmark Enrolment, 1996– 2002/follow-up, through 2010 Nested case–control	Nested within the Danish National Birth Cohort of ~100 000 pregnant women: nulliparous women at the time of blood draw during pregnancy were followed for breast cancer. Cases: 178 cases of breast cancer in nulliparous women at the time of blood draw during pregnancy Controls: 233; frequency-matched on age Exposure assessment method: quantitative serum measurement; analytical method was state-of-the-art; a single sample collected at enrolment (1996–2002) was analysed for PFAS at ascertainment for cases and controls	Breast (premenopausal)	See <u>Table 2.4</u>	<i>Exposure assessment critique</i> : Key strengths were that serum levels represent the combined exposure through all exposure pathways; measurement error low. Key limitations were that single samples at time of enrolment may not reflect exposure at crucial windows in cancer development; focused analysis on only 4 PFAS separately even though others had 98.8–100% samples detectable (PFHpS, PFNA) and others had 50–89% detectable (PFHpA, PFDA, PFDoA, PFUnA, PFTrA) (<u>Bonefeld-Jørgensen et al., 2014</u>). Did not sum PFAS in any way.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Ghisari et al. (2017) Denmark Enrolment, 1996– 2002/follow-up, through 2010 Nested case–control (cont.)				Other strengths: Blood samples collected beforebreast cancer diagnosis; exposure duringpregnancy may be an important exposurewindow for breast cancer; large sample ofpremenopausal cases; consideration of relevantSNPs.Other limitations: Focused on premenopausalbreast cancer, did not consider postmenopausalbreast cancer; no information on tumourcharacteristics; small exposure contrast.Other comments: Earlier follow-up by Bonefeld-Jørgensen et al. (2014). A few dozen breast cancercases from that study were excluded here due toconcern about status due to a coding error.
Hurley et al. (2018) CA, USA Enrolment, 1995–1996/ follow-up, 1 January 2006 to 1 August 2014 (incidence) Nested case–control	Nested within the California Teachers Study; 133 479 female public-school teachers and other professionals were followed annually for cancer incidence Cases: 902 cases with a diagnosis of invasive breast cancer at age < 80 yr, no prior history of breast cancer, who provided a blood specimen, answered the questionnaire, and were continuous residents of CA; participation rate, 65% Controls: 858 women drawn from a probability sample of at-risk cohort members, frequency-matched on age, race/ethnicity, and residence; participation rate, 55% Exposure assessment method: quantitative serum measurements; analytical method was state-of-the- art; a single sample was collected after diagnosis of invasive breast cancer (average, 35 mo) and analysed for PFAS	Breast	See <u>Table 2.4</u>	Exposure assessment critique: Key strengths were that serum levels represent the combined exposure through all exposure pathways; measurement error low. Key limitations were that blood samples were collected on average 35 mo (range, 9 mo to 8.5 yr) after diagnosis, which may not reflect exposure at crucial windows in cancer development; if breast cancer alters ADME of PFAS, there could be possible differential exposure misclassification; did not account for mixtures of PFAS and did not use all PFAS measurement data available. Other strengths: Case ascertainment with statewide cancer registry linkage and pathology confirmation; considered several established breast cancer risk factors as confounders/ modifiers; evaluated associations by combined ER and PR status and menopausal status; large number of cancer-registry identified cases and controls.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments	
Girardi and Merler (2019) Vicenza province, Veneto Region, Italy Enrolment, 1960– 2008/follow-up, 1970–2018 (mortality) Cohort	462 PFAS workers; 1383 railroad workers (comparison cohort); workers in a perfluorocarbon-production facility in Trissino manufacturing PFOA, PFOS, other perfluorinated compounds, and other chemicals; comparison populations included the regional general population and workers in a local railroad industry who were not exposed to chemicals; for both occupational cohorts, the workers included were men employed for ≥ 6 mo Exposure assessment method: cumulative serum levels were estimated for each worker's history, 1970–2008; serum data collected in 2000–2013 were used to model historical exposures in three job categories by incorporating fixed effects for variables related to subject of measurement as well as historical data on PFOA production.	Liver and intrahepatic bile ducts (ICD-9, 155) Colon Oesophagus, stomach Lymphatic and haematopoietic, NHL Lung All cancers combined	See Table S2.5 ^a See Table S2.5 ^a See Table S2.6 ^a See Table S2.6 ^a See Table S2.7 ^a	Exposure assessment critique: Key strengths were that historical exposures were modelled using serum measurements and estimation of cumulative exposures to PFOA (ng/mL-years); the study evaluated whether workers' home drinking-water was in contaminated area (Red Zone), but unclear how this information was used; APFO exposures were accounted for in PFOA measurements as it dissociates to PFOA in the body. Key limitations were that few samples were available to model serum levels in job categories 2 and 3; other PFAS exposures in plant were not accounted for (including PFOS and perfluorobutylsulfonyl fluoride); other potential carcinogenic co-exposures within factory were not accounted for, nor were alcohol or smoking use assessed. <i>Other strengths</i> : High exposure contrast; internal comparisons with non-exposed workers. <i>Other limitations</i> : Included only men; small occupational cohort with few deaths (<i>n</i> = 107); few cancer deaths for liver and lympho- haematopoietic (7 each) (the 2 causes with positive trends with exposure); no data on some causes of death of interest (e.g. bladder, prostate).	

\geq
R
\cap
2
3
\leq
~
Õ
ഹ
Ŗ
\geq
P
<u>ل</u> ے
UN
1
<u> </u>
ŝ
51

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Mancini et al. (2020a) France Enrolment, 1990/ follow-up, through 2013 (incidence) Nested case–control	Nested within the E3N cohort of 98 995 women born in 1925–1950 and covered by the French National Education System insurance; participants were invited to complete follow-up questionnaires (including dietary) every 2–3 yr and donate blood between 1994 and 1999 Cases: 194 incident cases of post- menopausal breast cancer diagnosed among women with serum (≥ 3 aliquots) collected before diagnosis, a completed dietary questionnaire in 1993, and randomly selected from 240 eligible cases of breast cancer Controls: 194; density-sampled at time of case occurrence and matched on age within 2 yr, menopausal status at blood collection, BMI at blood collection, and year of blood collection Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; a single sample collected before diagnosis of breast cancer was analysed for total PFOA and PFOS, not for isomers of PFOA or PFOS	Breast (post- menopausal)	See <u>Table 2.4</u>	 Exposure assessment critique: Key strengths were that serum levels represent the combined exposure through all exposure pathways; measurement error low. Key limitations were that single samples before diagnosis may not reflect exposure at crucial windows in cancer development. Other strengths: Blood samples collected before diagnosis with a long follow-up period; extensive adjustment for plausible confounders; inclusion of hormone receptor subtype information; low loss to follow-up. Other limitations: Limited statistical power, particularly when exploring differences by subtype; low exposure contrast in general population sample; did not include premenopausal breast cancer cases. Other comments: Frenoy et al. (2022) conducted additional exposure-response analyses.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Shearer et al. (2021) USA Recruitment: 1993– 2001, Follow-up (from blood drawn): median 8.8 yr (incidence) Nested case–control	Nested within the PLCO cohort, which comprises ~150 000 adults aged 55–74 yr from study centres in 10 cities; about half (assigned to the screening arm) provided a blood sample at baseline and were followed for incident cancer Cases: 324; source of cancer diagnosis not reported. Controls: 324; density-sampled on calendar time and individually matched on age categories, sex, race and ethnicity, study centre, and year of blood draw Exposure assessment method: quantitative serum measurements; analytical method was state-of the-art; blood samples collected at enrolment into PLCO study; samples from cases and controls were analysed at the same time for PFAS in serum	Kidney (RCC)	See <u>Table 2.2</u>	Exposure assessment critique:Key strengths were that serum levels representthe combined exposure through all exposurepathways; measurement error low.A key limitation was that single measurement ofserum levels may not have reflected cumulativeor long-term exposure, although only minormisclassification of long-term exposure over aperiod of 5–8 yr was seen, based on a simulationstudy (see Annex 3 in the present monograph).Other strengths: Large number of kidney cancercases ($n = 324$); an average of 8 yr of follow-upfollowing baseline serum measurement of avariety of PFAS; good data on confounders;internal comparisons with control over kidneyfunction; adjustment for exposure to other PFAS.Other limitations: Low exposure contrast in apopulation with background levels.Other comments: PFAS concentrations weremissing for two (excluded) cases.
Chang et al. (2023) USA Enrolment, 1993–2001/ follow-up, through November 2013 (incidence) Nested case–control	Nested within the PLCO cohort (see <u>Shearer et al., 2021</u>) Cases: 621; all incident cases of invasive breast cancer diagnosed among postmenopausal women who were not using MHT at baseline (unless their cancers were hormone receptor- negative) Controls: 621; controls were selected using incidence-density sampling; all were postmenopausal, still alive and cancer-free at the time of case diagnosis; matching on age at baseline, date of blood draw, and baseline MHT use	Breast (post- menopausal)	See <u>Table 2.4</u>	Exposure assessment critique: Key strengths were that serum levels represent the combined exposure through all exposure pathways; exposure was assessed before outcome; any misclassification is likely to be non-differential. Key limitations were that single measurement of serum levels may not capture relevant window of exposure for cancer development, especially among the cases diagnosed close to sample collection (but the authors conducted analyses stratified by time since blood draw, which addresses this concern); also, only minor misclassification of long-term exposure over a period of 5–8 yr, based on a simulation study (Annex 3); exposure assessment relied upon relative quantification of PFOA and PFOS (but relative measures have correlated well with

targeted absolute concentration measurements).

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Chang et al. (2023) USA Enrolment, 1993–2001/ follow-up, through November 2013 (incidence) Nested case–control (cont.)	Exposure assessment method: PFOA and PFOS serum levels were measured using a non-targeted method (unclear whether both branched and linear isomers were included); in the same participants, untargeted serum levels were correlated with levels measured using a standard, targeted method (Spearman correlation coefficient was 0.76 and 0.77 for total PFOS and total PFOA, respectively)			Other strengths: Large number of breast cancer cases (n = 621); an average of 8 yr of follow-up after serum measurements on a variety of PFAS at baseline; good data on confounders, internal comparisons; adjustment for exposure to other PFAS; stratified analyses by hormone status of cancer. Other limitations: Low exposure contrast in a population with background levels; limited power to consider hormone receptor negative tumours, no premenopausal cases. Other comments: PFAS concentrations were missing for two (excluded) cases.
Rhee et al. (2023a) USA Recruitment, 1993– 2001, follow-up (from blood draw), median, 9 yr (incidence) Nested case–control	Nested within the PLCO cohort (see <u>Shearer et al., 2021</u>) Cases: 750 cases of aggressive prostate cancer (defined as stage III or IV, Gleason score \geq 8, or Gleason score 7 and death from prostate cancer), diagnosed > 300 days after blood collection Controls: 750; alive and cancer-free at time of case diagnosis, and individually matched to cases on age at baseline, race/ ethnicity, study centre, calendar and study year of blood collection, and prior freeze-thaw cycle Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; a single sample was collected at time of enrolment into PLCO; time between blood draw and diagnosis was 9 yr (median IQR, 5.13 yr)	Prostate (aggressive/ advanced)	See <u>Table 2.3</u>	<i>Exposure assessment critique:</i> Key strengths were that serum levels represent the combined exposure through all exposure pathways; measurement error low; PFAS were measured in blood samples collected before diagnosis; analysis of blood samples collected 0,1, and 5 yr after enrolment showed a high degree of reproducibility with ICCs of > 0.7 for PFOA and PFOS. <i>Other strengths</i> : large case control study with 750 cases and matched controls; data on a broad range of confounders; smoking was controlled for in the analysis; other exposures are unlikely to be correlated with PFAS in this general population sample; mutual adjustment for other PFAS under study. <i>Other limitations</i> : general population with low exposure contrast; large number of comparisons.
Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
--	---	---------------------------------	----------------------------------	--
Cohn et al. (2020) Oakland (CA), USA Enrolment, at birth between 1959–1967/ follow-up, birth to March 2013 (incidence) Nested case–control	Nested within the CHDS pregnancy cohort, which includes 19 044 live births from pregnant members of the Kaiser Foundation Health Plan who received obstetric care between 1959 and 1967 and who provided blood specimens during pregnancy and at birth; > 99% of eligible women enrolled, and 74% of cases had a blood sample and complete information on potential confounders and effect modifiers Cases: 102 incident cases of invasive or non-invasive breast cancer diagnosed by age 52 yr, with a maternal perinatal blood sample and complete information on potential confounders and effect modifiers Controls: 310; 3 per case, density- sampled on case age and matched on birth year and trimester of maternal blood draw Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; maternal blood samples were collected before offspring birth; blood samples from cases and controls were retrieved ~50 yr later for PFAS analysis	Breast	See <u>Table 2.4</u>	<i>Exposure assessment critique</i> : Key strengths were that serum levels represent the combined exposure through all exposure pathways; measurement error low; perinatal exposure may be relevant for later breast cancer; misclassification of exposure is unlikely because cases were matched to controls for year of enrolment in the study (thus, changes in concentration over time were addressed). Key limitations were that PFAS exposure was not measured directly in study participants; only one maternal blood sample (during pregnancy or after labour) was used, while PFAS levels may vary during pregnancy; no information on exposures during the individual's lifetime. <i>Other strengths</i> : long follow-up; cases were likely to have been accurately determined via the California cancer registry. <i>Other limitations</i> : Only cases diagnosed before age 52 yr were included; risk of incomplete or biased case ascertainment; small sample size and limited statistical power; no information concerning tumour hormone-receptor status; adjustment only on potential maternal confounders and no variable collected at the daughter's individual level; lack of information on migration out of state.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Li et al. (2022a) Ronneby, southern Sweden Enrolment,1985–201/ follow-up, 1985–2016 (incidence) Cohort	60 507; the Ronneby Register cohort includes all individuals who ever lived in the Ronneby municipality in 1985–2013; one third of the households received PFAS-contaminated drinking-water from a waterworks situated near a military airfield where PFAS-containing firefighting foam was used in 1985–2013 (individuals considered to have "ever- high" exposure, 15 811); subsets with long-term exposure (≥ 11 yr) in the latest part of the follow-up period (2005–2013) were considered to be more highly exposed Exposure assessment method: residential location (water source) used to categorize participants into groups of potential exposure based on time period or duration of residency, or a residence in a neighbouring reference municipality; serum levels collected in 2014–2015 for residents and the neighbouring municipality were used to validate categories	Kidney Urinary bladder Testis Prostate Breast Liver, bile duct or gall bladder, pancreas, colon, rectum, oesophagus, stomach Thyroid Brain NHL, multiple myeloma, chronic lymphocytic leukaemia, chronic myelogenous leukaemia Melanoma Trachea and lung All cancers combined	See <u>Table 2.2</u> See <u>Table 2.3</u> See <u>Table 2.3</u> See <u>Table 2.4</u> See <u>Table 2.4</u> See Table S2.5 ^a See Table S2.6 ^a	<i>Exposure assessment critique:</i> Key strengths were that categories reflect serumlevels, which include exposure through allpathways; attempted to incorporate length ofexposure period in analysis.Key limitations were the potential formisclassification of exposures, since there was noinformation on individual water consumptionpatterns or use of bottled water or filtration athome; no cumulative years of exposure used,except two categories of short and long highexposure; no accounting for potential co-exposure; lack of historical information onarea-level PFAS drinking-water contamination,particularly during earlier years of the studyperiod.Other strengths: Large study population;strong exposure contrast; unbiased inclusion;complete follow-up; long follow-up for partof the population; reference group from samemunicipality.Other limitations: Mixed exposure profilewithout possibility to single out effects dueto specific compounds; little information onpotential confounders.Other comments: PFAS exposure mainly PFOS,PFHxS, together comprising > 90% of total PFASin water, and PFOA (water and blood samples).

Table 2.1 (continued)					
Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments	
Feng et al. (2022) Shiyan, China Enrolment, September 2008 to June 2010 and April to October 2013/ follow-up, 2008–2018 (incidence) Case–cohort	Case-cohort within the Dongfeng- Tongji cohort, which included 18 387 female retirees of automotive companies, without cancer at enrolment, with sufficient blood samples Cases: 226; the total number of diagnoses of incident breast cancer included 13 diagnoses in the comparison cohort Comparison cohort: 990 (including 13 cases); women randomly selected according to age strata The 13 cases included in the comparison cohort of 990 women served as controls until they received a cancer diagnosis Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; a single serum sample collected before diagnosis of breast cancer was analysed for six PFAS (including PFOA and PFOS), but not for isomers of PFOA or PFOS	Breast	See <u>Table 2.4</u>	 Exposure assessment critique: Key strengths were that serum levels represent the combined exposure through all exposure pathways; measurement error low; availability of prediagnostic serum samples (mean, 9.6 yr before diagnosis); measurement of several PFAS compounds. A key limitation was that single samples before diagnosis may not reflect exposure at crucial windows in cancer development. Other strengths: cases identified by reviewing medical records or death certificates; information on potential confounding variables collected through face-to-face interview and physical examination; high baseline participation. Other limitations: study population limited to retired workers; no information concerning tumour hormone-receptor status; no information on the likely completeness of diagnoses; cases identified by death certificate only (number not identified) would have an unknown diagnosis date; low exposure contrast. Other comments: ~10% lost to follow-up; does not mention how age-stratification was used to select comparison cohort; 90% of cases were postmenopausal; examined individual PFAS (6) and summed categories of PFCAs and PFSAs as well as sum of all PFAS. 	

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Wen et al. (2022) USA Enrolment, 1999–2014/ follow-up, 1999–2015 (mortality) Cohort	11 747 from the NHANES cohort, a nationally representative cross-sectional survey of adults (aged ≥ 18 yr) followed for mortality through 2015 Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; a single serum sample collected before death was analysed for 12 PFAS, but not for isomers of PFOA or PFOS	All cancers combined	See Table S2.7ª	Exposure assessment critique:Key strengths were that serum levels representthe combined exposure through all exposurepathways; considered mixtures of PFAS;measurement error low.Key limitations were that single samples beforedeath may not reflect exposure at crucialwindows in cancer development; unclear timingof blood sample relative to diagnosis/treatment.Other strengths: NDI linkage, nationallyrepresentative of the USA; relatively good controlfor potential confounders; adjustment for otherPFAS.Other limitations: Short follow-up time (median,81 mo); heterogenous outcome, representativeof incidence only in the case of high fatality ofcancers; use of volunteer-based population thatmay be healthier than the general population.Other comments: Analysed PFOA and PFOSseparately as well as total PFAS, total PFASexcluding PFOA and total PFAS excluding PFOSto address mixture issues.
Goodrich et al. (2022) CA and HI, USA Enrolment, 1993–1996/ follow-up, from mid- 1990s for > 20 yr Nested case–control	Nested within the MEC cohort, which is a community sample of 215 251 men and women aged 45–75 yr enrolled during 1993–1996 in HI and CA (primarily Los Angeles county) when responding to a 26-page postal questionnaire on mainly diet, demographic, and health issues Cases: 50 incident cases of non-viral HCC Controls: 50 individuals from the MEC, matched on age, sex, race/ethnicity, and study area	Liver (HCC)	See Table S2.5ª	<i>Exposure assessment critique</i> : Key strengths were use of prediagnostic plasma PFAS measurements; plasma levels represent the combined exposure through all exposure pathways; measurement error low. A key limitation was that single samples before diagnosis (at recruitment) may not reflect exposure at relevant windows in HCC development.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Goodrich et al. (2022) CA and HI, USA Enrolment, 1993–1996/ follow-up, from mid- 1990s for > 20 yr Nested case–control (cont.)	Exposure assessment method: quantitative plasma measurements; analytical method state-of-the-art; a single plasma sample was collected at recruitment, before HCC diagnosis, and analysed for 6 PFAS including PFOA and PFOS, but not for isomers of PFOA or PFOS			Other strengths: Exposure and outcome were ascertained independently and with high accuracy, with a median 7.2 yr between blood sample and diagnosis; BMI and diabetes status considered as potential confounders. Other limitations: Insufficient information on attrition, completeness of follow-up, statistical analysis; low exposure contrast (general population sample); did not account for mixture of PFAS in analysis of exposures.
Rhee et al. (2023b) CA and HI, USA Enrolment, 1993–1996; Follow-up through 2018 Nested case–control	Nested within the MEC cohort; see <u>Goodrich et al. (2022)</u> Cases: 428; all RCC cases identified as of 2018 in the MEC study with available pre-diagnostic serum sample; incident cases identified through linkage with the SEER HI registry and the CA state cancer registry Controls: 428 controls who were MEC participants alive at the time of the matched case diagnosis and matched 1:1 to cases on sex, race/ethnicity, study centre, age and date at serum collection, time of serum collection, and fasting status Exposure assessment method: quantitative plasma measurements; analytical method was state-of-the-art and included isomers of PFOA and PFOS; single plasma sample collected before or after (21%) RCC diagnosis; all were analysed for 11 PFAS, including PFOA and PFOS; separate analysis of linear and branched isomers of PFOS and PFOA was performed but only the summed results were reported.	Kidney (RCC)	See <u>Table 2.2</u>	Exposure assessment critique: Key strengths were the availability of pre- diagnostic sample for most participants, plasma levels represent the combined exposure through all exposure pathways; measurement error low. Key limitations were that single samples may not reflect exposure at crucial windows in RCC disease development; if RCC development alters ADME of PFAS, there could be possible differential exposure misclassification for samples collected after diagnosis (21%). Other strengths: Large sample size; consideration of multiple PFAS adjustment; stratification by race/ethnicity. Other limitations: Some stratified analyses by race/ethnicity had low statistical power. Other comments: pre-diagnostic sample collected between 1994–2006; in 1994, samples were collected only among participants selected to be included in case–control studies, then between 2001–2006, samples were taken from all survivors in the MEC cohort.

\geq
R
\cap
2
3
\underline{O}
2
0
G
R
\geq
Ρ
Т
S
I
<u> </u>
μ
01

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Purdue et al. (2023) USA Enrolment, 1988–2017/ follow-up, through 2018 Nested case–control	Nested within a cohort of active-duty US Air Force servicemen with ≥ 1 serum sample stored in the Department of Defence Serum Repository between 1988 and 2017; further eligibility criteria were no prior history of cancer 1990–2018 and age ≤ 39 yr Cases: 530 overall (187 with two samples); TGCTs diagnosed in the Department of Defence Cancer Registry Controls: 530 overall (187 with two samples); one control per case density- sampled with replacement among eligible US Air Force servicemen on active duty and cancer-free as of the case diagnosis date and matched on date of birth, race/ethnicity (seven groups), year entering military service, year of baseline serum sample collection, and year of second sample collection (if applicable) Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; stored serum samples were analysed for PFAS; in analyses of men with two samples, categories based on above or below the median at each time were evaluated	Testis	See <u>Table 2.3</u>	 Exposure assessment critique: Key strengths were that serum levels represent the combined exposure through all exposure pathways; low potential for exposure misclassification; measurements of several PFAS compounds including PFOS and PFOA isomers; 2 repeated prediagnostic samples several years apart in a subset of the population; other military related PFAS exposures were considered (drinking-water). Key limitations were that for most participants only one serum measurement was available; no information on other exposures; no cumulative exposure metric and in particular inability to examine specific exposure windows in early life. Other strengths: Nested design; well characterized source population; large number of cases; serum samples obtained 0–19 yr before diagnosis; reasonable exposure contrast, analyses adjusted for other PFAS compounds, information on a range of covariates. Other limitations: Loss to follow-up (men leaving the military) and completeness of case ascertainment not quantified; large percentage (29%) excluded due to missing serum specimens; residual confounding by prenatal PFAS concentrations is an unresolved issue of potential importance; data on strong determinants of TGCT are lacking.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Madrigal et al. (2024) Finland Enrolment, 1986–2010/ follow-up, through 2016 Nested case–control	Nested in the Finnish Maternity Cohort, a national registry of women who donated serum during the first trimester of pregnancy (> 90% of pregnancies in Finland between 1983 and 2016) Cases: 400 cases were randomly selected from those diagnosed among women who donated serum for their first pregnancy and had a live, full-term birth delivered between 1987 and 2010, and who had no prior diagnosis of cancer at enrolment Controls: 400 controls individually matched on year of delivery (4–5 yr increments) and age at first birth (3 yr increments) Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; a single serum sample was collected ≥ 3 yr before diagnosis	Thyroid (papillary)	See <u>Table 2.4</u>	 Exposure assessment critique: Key strengths were that prediagnostic serum sample levels were measured, which represent the combined exposure through all exposure pathways; all samples analysed in the same manner; measurement error low; selected Group 1 carcinogens were measured in the blood samples. A key limitation was that single sample collected during pregnancy may not reflect exposure at crucial windows in cancer development, although only minor misclassification of longterm exposure over a period of 5–8 yr, based on a simulation study (Annex 3). Other strengths: Analyses controlled for 19 PFAS as well as several PCBs, organochlorine pesticides, and PBDEs; the age of the cohort during follow-up included peak years of thyroid cancer incidence. Other limitations: Low-level exposure with small exposure contrast; controls were not matched on the exact year of delivery, but on increments of 4–5 yr, which might affect comparison of PFAS levels because of temporal trends in levels of PFAS; data on pre-pregnancy body mass index, a thyroid cancer risk factor, was largely missing; no information was available on medical or environmental exposure radiation.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Zhang et al. (2023) ATBC cohort: Finland, PLCO: USA ATBC: Enrolment, 1985–1988/follow-up, through 2011; PLCO: Enrolment, 1993–2001/follow-up, through 2010 Nested case–control	Two nested case-control studies, one within the ATBC study, the other within PLCO study (for PLCO see <u>Shearer et al.,</u> <u>2021</u>) ATBC is a randomized trial in White male smokers (aged 50–69 yr at recruitment) to evaluate the chemopreventive effects of alpha- tocopherol and beta-carotene on lung cancer (<i>n</i> = 29 246) Cases: 251 from ATBC and 360 from PLCO; cases from the ATBC study were male smokers who participated in a prevention trial who developed pancreatic ductal adenocarcinoma identified in the Finnish Cancer Registry; cases from the PLCO study were men and women ascertained by annual mail-in surveys, cancer registries and/or the NDI Controls: 251 from ATBC, 360 from PLCO; in both cohorts, controls were individually matched on age and date of blood draws, and sex; there was additional matching on race in PLCO only Exposure assessment method: PFOA and PFOS levels were measured in serum using a non-targeted method; a single serum sample was collected before diagnosis	Pancreas (ductal adenocarcinoma)	See Table S2.5ª	Exposure assessment critique: Key strengths were that prediagnostic serum levels represent the combined exposure through all exposure pathways; measurement error low; long follow-up time (median time between blood draw and diagnosis was 9–12 yr). Key limitations were that non-targeted analyses prevented comparison of sample concentrations across studies; single samples may not reflect exposure at crucial windows in cancer, although only minor misclassification of long-term exposure over a period of 5–8 yr, based on a simulation study (Annex 3); exposure assessment relied upon relative quantification of PFOA and PFOS, but relative measures have correlated well with targeted absolute concentration measurements. Other strengths: Information on potential confounders collected by trained staff through questionnaires and for height and weight in the ATBC Study; excellent case ascertainment. Other limitations: Low-level exposure with small exposure contrast; included only White men who smoke. Other comments: ATBC study participants were aged 50–69 yr at baseline and PLCO participants were aged 55–74 yr at baseline; the two nested case–controls were analysed separately.

178

Table 2.1 (continued)					
Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments	
van Gerwen et al. (2023) Mount Sinai (NY), USA Enrolment, 2008–2021 Nested case–control	Nested within Bio <i>Me</i> , a medical record- linked biobank within the Institute for Personalized Medicine at the Icahn School of Medicine at Mount Sinai, including residents of New York City and the larger metropolitan area Cases: 88 adult patients diagnosed with thyroid cancer according to ICD codes 193 (9th revision) and C73 (10th revision) Controls: 88 healthy (non-cancer) participants, pair-matched on sex, age (± 5 yr), race/ethnicity, BMI, smoking status (ever/never), and calendar year of sample collection Exposure assessment method: PFOA and PFOS levels were measured in plasma using an untargeted analytical method; a single plasma sample was collected, only 35% of these were collected > 1 yr before diagnosis; all samples were analysed for 8 PFAS including PFOA and PFOS; analysed linear and branched isomers of PFOS, but not PFOA separately	Thyroid	See <u>Table 2.4</u>	<i>Exposure assessment critique</i> : Key strengths were that plasma levels represent the combined exposure through all exposure pathways; plasma samples were collected ≥ 1 yr before diagnosis for a subset, albeit small, of the cases. Key limitations were the use of untargeted analysis with semiquantitative measurements, so concentrations not known or comparable with other studies (however, participant exposures can be ranked); single samples may not reflect exposure at crucial windows in cancer development, especially since 65% samples were collected < 1 yr before diagnosis; if thyroid cancer development alters ADME of PFAS there could be possible differential exposure misclassification for those samples collected after diagnosis. <i>Other strengths</i> : Source population represents a diverse racial/ethnic and socioeconomic population; availability of histologic data for the cases; analyses adjusted for age, sex, race, and body mass index, and sample storage time, and, for some analyses, adjustment of analyses of specific PFAS compounds for other PFAS compounds. <i>Other limitations</i> : Small sample size, particularly for cases with plasma collected > 1 yr before diagnosis; short follow-up time; the possibility of detection bias, given cases were identified in a hospital and ambulatory practice setting (however, this was minimized by selection of controls from the same network, and patients and practitioners were unaware of PFAS	

measurements).

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
yan Gerwen et al. (2023) Mount Sinai (NY), USA Enrolment, 2008–2021 Nested case–control (cont.)				Other comments: Analyses were conducted for all thyroid cancer cases, cases whose plasma collection was < 1 yr before diagnosis (cross-sectional group) and cases whose plasma collection was \geq 1 yr before diagnosis (longitudinal group).
Winquist et al. (2023) 20 US states Enrolment 1998–2001; follow-up through 30 June 2015 Case–cohort	Case-cohort design within the CPS-II Lifelink Cohort (<i>n</i> =39 371); participants in the CPS-II Nutrition cohort (recruitment, 1992–1993) who were alive and agreed to a blood sample collection between 1998 and 2001 Cases: 3762 incident cases with a first cancer diagnosis of kidney, bladder, breast (females only), prostate (males only), or pancreatic cancer, leukaemia, or lymphoma, detected through self- report or NDI linkage and verified through medical records review or cancer registry Controls: 999; a sex-stratified simple random sample of 499 women and 500 men (~3% of the eligible cohort); stratification sampling was to ensure an adequate number of subcohort participants in sex-specific analyses (for breast and prostate cancers) Exposure assessment method: quantitative plasma measurements; analytical method was state-of-the-art, except no branched isomers of PFOA and PFOS were analysed; a single plasma sample was collected before diagnosis	Kidney (all combined) Kidney (RCC) Urinary bladder Prostate Breast (post- menopausal) Pancreas Haematological malignancies	See <u>Table 2.2</u> See <u>Table 2.2</u> See <u>Table 2.3</u> See <u>Table 2.4</u> See Table S2.5 ^a See Table S2.6 ^a	<i>Exposure assessment critique:</i> Key strengths were that prediagnostic plasma levels represent the combined exposure through all exposure pathways; measurement error low. A key limitation was that single samples may not reflect exposure at crucial windows in cancer development, although only minor misclassification of long-term exposure is expected over a period of 5–8 yr, based on a simulation study (Annex 3). <i>Other strengths:</i> Large number of cases. <i>Other limitations:</i> Survivor cohort with blood collected from persons mostly over aged 65, thus the study would not include persons who may have had PFOA- or PFOS-related cancer developed earlier in life, resulting in downward bias.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Vieira et al. (2013) OH and WV, USA 1996–2005 (incidence) Case–control	Cases: 25 107 index cancer cases were retrieved from cancer registries covering a community sample with relatively high exposure to PFOA because of contamination of drinking-water from the Parkersburg (WV) polymer- production plant; 18 different cancers were analysed Controls: number varied; for each cancer site evaluated, controls were cases of cancer for all other sites, with the exclusion of four cancers of a priori interest (kidney, testis, pancreas, and liver) that have been associated with PFOA in studies in experimental animals or humans Two case-control studies are described, one including both WV and OH cases (first), the other only OH cases (second)	Kidney (urinary pelvis/ UUT) Urinary bladder Testis Prostate Breast Liver Pancreas Colon and rectum Thyroid Brain Leukaemia, multiple myeloma, NHL, melanoma Lung	See <u>Table 2.2</u> See <u>Table 2.3</u> See <u>Table 2.3</u> See <u>Table 2.4</u> See <u>Table 2.5^a</u> See <u>Table 52.5^a</u> See <u>Table 52.5^a See <u>Table 2.4</u> See <u>Table 52.6^a</u> See <u>Table 52.6^a</u> See <u>Table 52.6^a</u></u>	Exposure assessment critique: Key strengths were the availability of data on measured serum levels for a large number of residents of contaminated water districts, permitting analyses by approximate levels of exposure in each water district; exposure reconstruction for OH provided detailed exposure assessment; any misclassification is likely to be non-differential; the second case- control study based in OH estimated serum level for individuals based on a model shown to be able to predict well observed levels for 30 000 residents of the six contaminated water districts at one point in time (2005/2006, Spearman correlation 0.71). Key limitations were the assignment of an ecological exposure (by water district) in the first case-control study; the use of estimated individual serum levels in the 2nd case- control study based in OH, and the somewhat arbitrary assumption in that second study that the estimated serum levels 10 yr before case diagnosis were the most relevant, as well as the assumption that cases and controls had remained in the same residence for 10 yr; no other exposures (except smoking) were considered.

Reference, location, enrolment/follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology)	Exposure category or level	Comments
Vieira et al. (2013) OH and WV, USA 1996–2005 (incidence) Case–control (cont.)	Exposure assessment method: address at diagnosis was used to assign PFOA exposure; individuals from OH (about one third of the sample) were geocoded whereas individuals from WV were assigned exposure on the basis of geographical unit; water-district PFOA levels were available for all individuals; for individuals from OH, PFOA serum values could be estimated on the basis of exposure models (Shin et al., 2011a, b) A five-level exposure variable was created			Other strengths: The large number of incident cancers from cancer registries; the reasonably large number of exposed cases in the contaminated water districts for many specific cancers (although for analyses of rarer cancers by categories of exposure small numbers were sometimes an issue); the large exposure contrast between water districts as well as within individuals in the second case-control study. <i>Limitations</i> : Fairly few potential confounders used in the analyses; the use of controls with cancer was a less marked limitation.

ADME, absorption, distribution, metabolism, and excretion; AL, Alabama; approx., approximately; APFO, ammonium perfluorooctanoate; ATBC, Alpha-Tocopherol, Beta-Carotene Cancer Prevention; BMI, body mass index; CA, California; CHDS, Offspring in the Child Health and Development Studies; CI, confidence interval; CNS, central nervous system; CPS-II, Cancer Prevention Study II; E3N, Etude épidémiologique auprès de femmes de la Mutuelle générale de l'Education nationale; HCC, hepatocellular carcinoma; HI, Hawaii; ICD, International Classification of Diseases; IQR, interquartile range; JEM, job-exposure matrix; MEC, Multiethnic Cohort; MHT, menopausal hormone therapy; MN, Minnesota; mo, month(s); NDI, National Death Index; NHANES, National Health and Nutrition Examination Survey; NHL, non-Hodgkin lymphoma; NY, New York; OH, Ohio; PCB, polychlorinated biphenyl; PBDE, polybrominated diphenyl ethers; PFCA, perfluoroalkyl carboxylic acid; PFDA, perfluorodecanoic acid; PFDA, perfluorododecanoic acid; PFHpA, perfluoroheptanoic acid; PFHpS, perfluorobeptanesulfonic acid; PFHxS, perfluorohexanesulfonic acid; PFNA, perfluoronanoic acid; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PFTA, perfluorotridecanoic acid; PFUA, perfluoroundecanoic acid; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PFTA, perfluorotridecanoic acid; PFUA, perfluoroundecanoic acid; PLCO, Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial; POSF, perfluorooctanesulfonyl fluoride; RCC, renal cell carcinoma; SEER, Surveillance, Epidemiology, and End Results; SMR, standardized mortality ratio; SNP, single-nucleotide polymorphism; TFE, tetrafluoroethylene; TGCT, testicular germ cell tumour; UK, United Kingdom; USA, United States of America; WI, Wisconsin; WV, West Virginia.

^a Tables S2.5, S2.6, and S2.7 are available in Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: https://publications.iarc.who.int/636.

282 ng/mL in workers with all work, some work, and no work, respectively, in PFOA production) versus 4.5 ng/mL in blood donors in the same area (<u>Raleigh et al., 2014</u>). The toxicokinetics of PFOA were not considered when modelling the cumulative exposure metric, which was based solely on air concentration and duration of exposure.

Vital status was ascertained by the National Death Index (NDI) until the end of 2008 for 99.3% of the population, and cases of incident cancer were identified by linkage to the Minnesota Cancer Surveillance System and the Wisconsin Cancer Reporting System; reporting of cancer had been mandatory since 1988. The case capture of the cohort was estimated to be about 85%, and higher for highly exposed workers (individual out-of-region migration data were not available).

Standardized mortality ratios (SMRs) for cancers of the prostate, pancreas, urinary bladder, female breast, kidney, and liver were estimated from 1960 through 2008 using the entire Minnesota population as the referent.

The follow-up period for cancer incidence was from 1 January 1988 until the end of 2008. In total, 665 incident cancers of the prostate, kidney, pancreas, bladder, liver, breast, testis, and thyroid were identified. [Estimates for testicular, thyroid, and liver cancer were not provided.] Relations between cancer-specific risk and time-dependent cumulative APFO exposure were estimated by Cox regression, with adjustment for sex, year of birth, and age.

[The Working Group noted several strengths associated with this primarily male cohort, including individual assessment of cumulative air exposure, with some evidence that this exposure metric was correlated with serum concentration; limited and minimal co-exposure to tetrafluoroethylene (TFE); a relatively large number of some incident cancers in the exposed cohort (prostate, bladder, breast); a reference population with socioeconomic characteristics that were similar to those of the exposed population; and a long follow-up period. Limitations were the lack of information on dermal exposure; the fact that this was a survivor cohort (in which follow-up for cancer incidence began many years after the start of follow-up) with the potential for downward healthy-worker survivor bias; the lack of data on workers who left Minnesota or Wisconsin (although the out-migration of the workers with higher exposure was similar to that of the reference workers); and the small numbers of some cancers (kidney, pancreas, liver, testes, and thyroid). Risk estimates were most probably biased towards the null, because measurement error is likely to be non-differential. Finally, a discrepancy was noted between the description of the adjustment for Cox regression models in the text (age, year of birth, and sex) and that in the tables (age and year of birth only).]

2.1.2 PFOS-production workers (Decatur, Alabama, USA)

Alexander et al. (2003) studied a cohort of 3512 production workers who had been exposed to perfluorooctanesulfonyl fluoride (POSF), which is degraded or metabolized to PFOS, at a plant in Decatur, Alabama, USA, that produced speciality films and fluorochemicals in two different facilities. All the cohort members had worked at the plant between 1961 and 1997. The study was limited to those who had worked for ≥ 1 year at the plant (n = 2083; men, 83%). Follow-up for mortality was conducted between 1961 and 1998. Exposure was estimated using serum concentration of PFOS, based on a sample of 232 employees randomly selected for serum sampling in 1998, with 80% participation (chemical plant, n = 126; film plant, n = 60). The authors noted that there was also exposure to PFOA in this subsample, at slightly lower levels; serum PFOA concentrations correlated well with serum PFOS concentrations. PFOA manufacturing at this plant started in 1998; this was the same year that serum samples were

collected, but there may have been incidental exposure previously. Chemical plant workers had high serum concentrations of PFOS (geometric mean, 0.9 ppm [900 µg/L]), whereas film plant workers had lower concentrations (geometric mean, 0.1 ppm [100 μ g/L]). On the basis of these measurements, a job-exposure matrix (JEM) was developed for all workers, jobs being classified into three exposure groups. The groups were defined as having: (i) no or minimal workplace exposure to PFOS (e.g. jobs in film plant); (ii) low potential workplace exposure to PFOS (e.g. engineers, quality control technicians, environmental health and safety workers, administrative assistants, managers); or (iii) high potential workplace exposure to PFOS (e.g. cell operators, chemical operators, maintenance workers, mill operators, waste operators, crew supervisors). Cumulative exposure (duration multiplied by intensity of exposure) was estimated after accounting for changing jobs over time, based on assigned weights of 1, 3, and 10, respectively, for jobs in the three exposure groups. The SMRs were calculated by comparing the mortality of all workers with that of residents of Alabama; in further analyses, each worker subgroup was compared with the Alabama population. Of the 2083 workers included, 47% (982) had worked at some time in jobs in which exposure to PFOS was considered high, 14% (289) worked in low-exposure areas, but never held a job in the high-exposure areas, and 812 (39%) were considered to have no or minimal workplace exposure to fluorochemicals. In the cohort, 145 deaths were identified (including 39 deaths from cancer); there were 65 deaths in the high-exposure group, 27 in the low-exposure group, and 53 in the non-exposed group. Deaths were ascertained using the NDI and the US Social Security Death Index. There were 3 deaths from bladder cancer, which was the cause of death that showed the highest excess, compared with the reference population.

<u>Alexander and Olsen (2007)</u> further followed the PFOS cohort studied by <u>Alexander et al.</u> (2003), focusing on bladder cancer. A postal questionnaire sent to all living current and former employees of the facility in 2002 (n = 1895) to identify cases of incident bladder cancer was returned by 1400 cohort members (response rate, 74%). The underlying cause of death (ascertained through the NDI) from death certificates for 188 deceased workers was used to identify bladder cancer decedents. The analysis drew on data from questionnaire respondents and from decedents (1588 cohort members); 11 cases of bladder cancer were included (6 identified in the questionnaire and 5 identified using death certificates). Cumulative exposure was analysed for the same three groups as before (none, low, high). Serum concentrations of PFOS were estimated from a subsample, as described in Alexander et al. (2003). Job categories assigned to no, low, and high exposure had geometric mean serum PFOS concentrations of 0.11–0.29 µg/mL, 0.39-0.89 µg/mL, and 1.30-1.97 µg/mL, respectively (slightly different from the values reported in Alexander et al., 2003). The incidence of bladder cancer was analysed using national rates from 1970 to 2002, adjusted for age, sex, and calendar year. Internal analyses used Poisson regression, adjusting for age and sex.

[The Working Group noted that this cohort, the same in both studies, had large exposure contrast, and that some serum measurements were available, although they were not used in the exposure assessment. Cancer incidence data were published by Alexander and Olsen (2007), who also used internal comparisons. Limitations included the small number of observed cancers. Alexander et al. (2003) limited their study to mortality, reporting a large excess for bladder cancer on the basis of only 3 cases, and analyses only against national rates. Alexander and Olsen (2007) studied only 11 cases of incident bladder cancer (with no bladder cancer excess). Another limitation was co-exposure to PFOA, which was measured at somewhat similar levels to those of PFOS in a sample of employees in 1998

(Alexander and Olsen, 2007), and correlated with PFOS serum concentrations - although PFOA was not produced until 1998 and earlier levels were probably lower. Both studies were limited by sparse data on smoking, but there were fewer concerns about confounding by smoking in the study by Alexander and Olsen (2007) because the authors conducted an internal analysis (which is less subject to confounding) as well as external comparisons. Another limitation was the failure to adjust for other PFAS in the plant, and in general for other workplace exposures in a chemical plant. The Working Group also noted that perfluorooctanesulfonyl fluoride (POSF), the predecessor of PFOS used in this plant, is itself a reactive compound and may be toxic.]

2.1.3 Polymer-production workers in Parkersburg (West Virginia, USA)

The polymer-production plant in Parkersburg, West Virginia, USA, was a facility producing several types of polymer from a wide variety of monomers and other chemicals; it began operating in 1948. Workers in the plant, especially those involved in various activities related to the production of certain polymers or copolymers, were exposed to PFOA. Steenland and Woskie (2012) conducted an updated investigation of mortality in a previously studied cohort of plant workers (Leonard et al., 2008). Briefly, this cohort included 6027 workers who were employed at the plant for ≥ 1 day between 1948 and 2002. The analyses by Steenland and Woskie (2012) focused on 5791 workers (women, 19%) with sufficiently detailed work histories for PFOA exposure estimation and non-missing values for date of birth. Workers were followed for mortality from 1952 through 2008; deaths were identified using the NDI (for 1979 or later), or from death certificate data (from the US Social Security Administration and state death certificates) for earlier years. For the PFOA exposure assessment, the investigators used serum PFOA

concentrations determined from 2125 blood samples collected from 1308 workers between 1979 and 2004 to produce regression models estimating serum PFOA levels for eight different combinations of job category and job group over time. Comparisons of modelled PFOA levels with the serum concentrations measured for this study population demonstrated high agreement (Spearman rank correlation coefficient, $\rho = 0.8$) (Woskie et al., 2012). The SMRs for mortality from cancer overall and for specific cancers - liver, pancreas, lung, breast, prostate, testis, kidney, and bladder, and mesothelioma, non-Hodgkin lymphoma (NHL), and leukaemia – were calculated for workers in the cohort (overall and by quartiles of cumulative estimated PFOA exposure) compared with workers at other plants in the Appalachian region and managed by the same company, as well as with reference rates for the general US population. A total of 1084 deaths were recorded during follow-up (with a mean length of follow-up of 30 years). In a subsequent investigation for a subset of 3713 workers, who were all included in both the occupational cohort and the C8 Science Panel study (described in Section 2.1.5) by Steenland et al. (2015), the incidence of selected cancers for which there were 20 or more cases during follow-up (melanoma and cancers of the bladder, colorectum, and prostate) was evaluated for workers aged from 20 years or from the year 1951 (whichever was later) until the period (2008–2011) when interviews were conducted with workers or their next of kin (the latter making up 6% of the interviews included in the analysis). Analyses focused on self-reported cancers or cancers reported by the next of kin that were validated by a review of medical records (355 valid cases). Cox proportional hazards regression analyses were used to estimate the risks of developing specific cancers in relation to quartiles of estimated cumulative PFOA exposure with age, as the underlying timescale, with adjustment for sex, race, education, body mass index (BMI), and time-varying smoking and alcohol consumption.

[The Working Group noted several strengths of this study, including the detailed historical exposure assessment and the ability to evaluate associations between PFOA and cancer in a population exposed to levels much higher than those in the general population. The estimated average annual serum PFOA concentration in the cohort overall (mean, 350 ng/mL) was nearly two orders of magnitude higher than the measured serum PFOA concentration (geometric mean, 3.9 ng/mL) reported in the US National Health and Nutrition Examination Survey (NHANES) in 2003–2004 (Calafat et al., 2007). The modelled PFOA exposure levels were highly correlated with serum PFOA concentrations in the more than 2000 measurements available for this study population (Spearman correlation, 0.8), indicating a valid model. Exposure to PFOS or other PFAS was not characterized in the cohort but was considered likely to be low, given the nature of the polymer-production work at the plant at the time of the study. Although some processes in the production of fluoropolymers involve TFE (classified by IARC as probably carcinogenic to humans, Group 2A; IARC, 2016), the Working Group noted that, owing to its high volatility and explosive potential, processes involving TFE were conducted in a separate area of the plant with limited access and that exposure control measures (e.g. closed systems) were used to prevent emissions of this gas. Therefore, workers' exposure to TFE in the plant was probably minimal during normal operations.

The use of a reference group of workers from other plants in the same region was also a strength of the mortality analysis, in terms of addressing the potential downward bias of the risk estimates because of the healthy-worker effect. However, exposures to other potential carcinogens were not assessed for the reference workers; if such exposures were higher or more prevalent in the

186

reference group, or both, then the resulting risk estimates for some cancers could have been biased towards the null. The Working Group noted that mortality from mesothelioma was elevated in the cohort (on the basis of 6 deaths among the PFOAexposed workers), suggesting possible exposure to asbestos in this population. However, the potential for confounding by asbestos exposure in analyses for other cancer sites of particular interest with respect to PFOA was considered likely to be low. Other limitations of the mortality analysis included the inability to evaluate some cancers because of the small numbers of deaths, in particular for cancers among women (given the small number of female workers included in the cohort) and for cancers that are less likely to be fatal (e.g. testicular cancer).]

[The Working Group did not consider the overlap between the subset of workers included in both the investigation of cancer incidence by Steenland et al. (2015) and the C8 Science Panel investigation (Barry et al., 2013) to be a major limitation, given that workers comprised a small proportion of the population in the latter study. In addition to the strengths of the PFOA exposure assessment already noted, a strength of the study by Steenland et al. (2015) was the ability to control for established cancer risk factors (e.g. BMI, smoking, and alcohol consumption) that might have confounded the associations between PFOA and specific cancers. The use of medical records to confirm self-reported cancer diagnoses probably reduced non-differential outcome misclassification, thereby potentially improving the ability to detect any true associations with PFOA. However, this approach might also have resulted in the underascertainment of cancer cases. Taken together with the relatively small sample size, this might have further limited the statistical power for analyses of specific cancers, and there were too few cases to evaluate cancers of the kidney and testis. Finally, the Working Group noted that the subset of workers in this analysis were less likely to have died than those who were not included, and also differed with respect to several demographic characteristics (e.g. those who were included were younger and more likely to be female), raising the possibility of selection bias if the exposure–response patterns differed between the workers who were included and those who were excluded. The potential effect of such selection bias (if present) on the direction and magnitude of the reported risk estimates was unclear.]

2.1.4 Diet, Cancer, and Health Cohort

Eriksen et al. (2009) studied data collected for 57 053 participants enrolled in a prospective Danish cohort between 1993 and 1997 - the Diet, Cancer, and Health Cohort. Approximately 160 000 potential participants were recruited from the general population, with data accessible from a national database, in two counties in Denmark (Aarhus and Copenhagen) (Tjønneland et al., 2007); of these, 57 053 agreed to participate. Participants were Danish citizens aged 50-65 years with no previous cancer diagnosis at enrolment. Plasma concentrations of PFOA and PFOS were measured at baseline using liquid chromatography and tandem mass spectrometry (LC-MS/MS). Virtually all samples had concentrations above the limits of detection, and 50 random samples were measured twice, with good agreement between measurements. Cancer incidence data for the cohort were available from the Danish Cancer Register. Investigators identified 1240 cases of cancer of the prostate, bladder, pancreas, or liver (in 1111 men and 129 women), diagnosed in 1993-2006, with a follow-up period of 0-12 years (median, 7 years) after baseline. Cancers of the prostate, bladder, pancreas, and liver (713, 332, 128, and 67 cases, respectively) were analysed in relation to baseline plasma concentrations of PFOA and PFOS. A subcohort of 680 men and 92 women who were randomly selected and had not had a diagnosis of cancer at the end of the follow-up period were used as

controls for the cancer cases, according to a casecohort design. Median plasma levels (5th and 95th percentiles) of PFOA and PFOS in those not later diagnosed with cancer were 6.6 (3.0– 13.0) ng/mL and 34.3 (16.2–61.8) ng/mL, respectively. Information on potential confounders (BMI, smoking, occupation, education, alcohol intake, diet) was collected using a questionnaire and differed according to cancer type. Analyses were conducted by quartile of PFOA or PFOS concentration. Linearity was first evaluated using spline models; where there was no significant deviation from linearity, a linear trend was assessed using a continuous variable for plasma PFOA or PFOS concentration.

[The strengths of this study included the use of a large cohort with numerous incident cancers identified using a reliable cancer registry (n = 1240), the measurement of plasma PFOA and PFOS concentrations at baseline, good control for confounders (e.g. age, sex, BMI, detailed smoking data, diet), internal comparisons, and little loss to follow-up. Limitations were a relatively low exposure contrast in a population with background exposure levels, the characterization of exposure on the basis of a single measurement at enrolment, and a somewhat limited period of follow-up.]

2.1.5 C8 Science Panel study

The C8 Science Panel conducted a cohort study of community residents and workers exposed to PFOA (C8 is a synonym of PFOA) from a fluoropolymer-production plant in the Mid-Ohio Valley on the border of West Virginia and Ohio, USA (<u>Barry et al., 2013</u>). Between the 1950s and the early 2000s, PFOA was released from the plant in air emissions and as liquid and solid waste, contaminating local public water supplies and private wells. A settlement from a class action lawsuit initially funded a large community health study known as the C8 Health Project, which was conducted in 2005–2006

(Frisbee et al., 2009). The C8 Science Panel investigated a cohort of adults aged \geq 20 years enrolled in the C8 Health Project (Winquist et al., 2013), as well as individuals who were employed in the plant and included in an occupational cohort, as described in Section 2.1.3; analyses were restricted to individuals who had completed at least one subsequent survey between 2008 and 2011 and who had retrospective environmental or occupational PFOA exposure estimates. These surveys solicited detailed information on demographic and health characteristics; an extensive residential history was included. Of the 32 254 individuals included in the analytical cohort, 28 541 were community cohort participants and 3713 had worked at the plant. For community participants, annual estimates of serum PFOA concentrations were calculated using an environmental fate and transport model for each year of life between 1952 and 2011, as described in Shin et al. (2011a, b). For the workers, estimates of occupational PFOA exposure were calculated as described in Section 2.1.3 and combined with environmental exposure estimates. Barry et al. (2013) conducted an investigation of cancer incidence among cohort participants, who were followed up from age 20 years onwards for an average of 33 years. For this analysis, cancers reported by participants in the surveys conducted in 2008–2011 were confirmed by consulting state cancer registries in Ohio and West Virginia or a review of medical records; a total of 2507 validated cancer diagnoses were identified and included in the statistical analyses. Proportional hazard regression models were used to estimate the risk of developing specific cancers at each year of age in relation to estimated serum PFOA concentrations (modelled per 1-unit increase on the natural log scale) both up to the time of diagnosis or censoring (unlagged analysis) and for estimated exposures 10 years in the past (lagged analyses). Exposure-response associations based on quartiles of estimated PFOA concentrations were also reported for kidney,

testicular, and thyroid cancers. All models were run with age as the timescale and were adjusted for sex, education, 5-year birth year period, and time-varying smoking and alcohol consumption. Data from this study and a nested case–control study of Prostate, Lung, Colorectal, and Ovarian Cancer (PLCO study) (<u>Shearer et al., 2021</u>) were combined for a pooled analysis of renal cell carcinoma (RCC) (<u>Steenland et al., 2022</u>).

The Working Group considered this cohort study to be highly informative, in light of several important features and strengths. With 32 254 participants, it is, to the knowledge of the Working Group, the largest cohort to evaluate cancer risk in community members and workers with high exposure to PFOA. Based on measurements of serum PFOA concentration from the C8 Health Project in 2005–2006, exposure levels were found in the investigation of **Barry et al.** (2013) to be considerably higher in the overall cohort, compared with the general US population (medians of 26.1 ng/mL and 4.0 µg/L [ng/mL], respectively) (Winquist et al., 2013). Among C8 Health Project participants (Frisbee et al., 2009), serum concentrations of perfluorohexanesulfonic acid (PFHxS) and perfluorononanoic acid (PFNA) were more modestly elevated, relative to the general US population (geometric means were 39% and 73% higher, respectively, compared with 2003-3004 US NHANES data), and serum PFOS concentrations were similar to those observed in the general population. A particular strength of the C8 Science Panel study was the detailed characterization of estimated serum PFOA concentrations from 1952 or the participant's year of birth (whichever was later) through 2011; modelled serum PFOA concentrations corresponded well (Spearman correlation, 0.71) with measured serum concentrations for the cohort in 2005–2006 (Winquist et al., 2013). This assessment of PFOA enabled analyses of exposure-response associations with cancer incidence in both unlagged and lagged analyses. The adjustment for established cancer risk

factors (e.g. smoking, alcohol consumption) that might have confounded the associations between PFOA and specific cancers was also a strength of the analyses in this cohort. A strength of the outcome ascertainment was the validation of self-reported cancer cases by linking with Ohio and West Virginia cancer registries and medical chart reviews; this approach could have reduced the potential for attenuation of risk estimates, owing to non-differential disease misclassification. Finally, there were high rates of participation in the target populations of community members in the C8 Health Project and workers in the occupational cohort (81.5% and 72.9%, respectively) (Winquist et al., 2013), reducing the likelihood of selection bias affecting the direction or magnitude of the observed associations. The Working Group also noted several limitations of this study. Direct measurements of serum PFOA concentrations were available only in 2005–2006; this might not have reflected PFOA exposure during etiologically relevant time periods when data on measured concentrations were not available. However, as noted previously, modelled estimates and measurements of serum PFOA concentrations were highly correlated, and any exposure misclassification would be non-differential and more likely to bias risk estimates towards the null. Also, given the design of the study as a survivor cohort, community members and workers who died before the cohort enumeration would not have been included, resulting in the potential underascertainment of cancers with a high rate of fatality in this population. However, given that PFOA exposure was considered to be unlikely to be related to survival time, the effect of this aspect of the study design on the resulting risk estimates was considered likely to be minimal (<u>Barry et al., 2015</u>). Despite the large sample size, the study had relatively limited statistical power to detect associations with some less common cancers, and for prospective analyses of cancer risk (i.e. for cases diagnosed after enrolment in the C8 Health Project). Finally, the Working Group

noted that the cancer cases included in this study probably overlapped with those included in the study by Vieira et al. (2013), although the case ascertainment approaches differed for the two studies. The study of Barry et al. (2013) included self-reported cases of cancer that were confirmed either by linking with West Virginia or Ohio cancer registries or by medical record abstraction, including those diagnosed in other states or before the availability of data from state registries. In contrast, the study of Vieira et al. (2013) considered cancer cases from 13 counties in West Virginia and Ohio (including both contaminated water districts and other adjacent areas without water contamination) that were identified from West Virginia and Ohio cancer registries for the years 1996–2005. However, the degree of overlap of the cases included by Barry et al. (2013) and Vieira et al. (2013) was unknown.]

2.1.6 Pooled cohort of international tetrafluoroethylene workers

This pooled cohort of international TFE workers included workers employed at one or more of six TFE synthesis and polymerization sites in North America and Europe (Gendorf, Germany; Dordrecht, the Netherlands; Spinetta Thornton-Cleveleys, Marengo, Italy; UK: Bayonne, New Jersey, USA; Parkersburg, West Virginia, USA) that, at the time of the study, comprised the entire population of workers in TFE manufacture in Europe and the USA (Consonni et al., 2013). TFE is a flammable and explosive gas and is mainly used in closed systems as a monomer in the production of fluorinated polymers, including polytetrafluoroethylene (PTFE), which is widely used in consumer products such as waterproof and breathable membranes for clothes and as coatings on carpets. APFO, the ammonium salt of PFOA, is used as a polymerization aid in PTFE production.

Excluding 778 female workers and 122 male workers with missing data, the cohort included

5879 male workers who, for ≥ 1 day (in three plants), 6 months (in one plant), or 1 year (the other plants), were employed at a TFE-manufacturing facility in 1950–2002. Enrolment of eligible workers was based on company rosters. [Completeness of enrolment of eligible workers was not reported.]

The synthesis and polymerization of TFE entail potential exposure to the TFE monomer and to PFOA, which is released from its ammonium salt (APFO) during production. Individual semiquantitative levels of work-related exposure to TFE and PFOA were estimated using expert judgement to create a plant- and job-specific exposure matrix with yearly estimates (in arbitrary units) of exposure, declining by 10% for each decade from the start of TFE production until 2002 (Sleeuwenhoek and Cherrie, 2012). Only a few measurements of TFE air concentrations at the various plants were available to assist the exposure assessment (Sleeuwenhoek and Cherrie, 2012). The number of workers who had ever been exposed to TFE was 4773 (81.2%), while 1081 (18.4%) workers had never been exposed. Among workers who had ever been exposed to TFE, 4205 were also exposed to APFO. All workers exposed to APFO were also potentially exposed to TFE, mainly through accidental leaks, from opening of autoclaves, and from decomposition of PTFE. There was a high correlation between TFE and PFOA exposure intensities (Sleeuwenhoek and Cherrie, 2012), based on arbitrary units (Spearman correlation, 0.72). At two of the plants (Gendorf, Thornton-Cleveleys), previous exposure to vinyl chloride monomer might have occurred. No information was available on occupational exposure to agents known to promote the development of leukaemia. The largest TFE-production site (Parkersburg, West Virginia, USA) accounted for the largest number of unexposed workers.

Ascertainment of vital status (complete for 98.8% of the study population) and, where appropriate, cause of death was determined by epidemiology units at the company level (UK, USA), by university epidemiology departments (Germany, the Netherlands), or by local health units (Italy), through record-linking procedures or individual follow-up. [Record-linking procedures are expected to give a higher degree of completeness.] The mortality follow-up period was 1950-2008. Causes of death were recorded from death certificates according to the International Classification of Diseases (ICD) ICD-9 or ICD-10 classification. The mean values were 55 years for attained age, 9.2 years for duration of exposure to TFE, and 23 years for time since first exposure to TFE. For selected cancers, SMRs compared with national data were provided for ever APFO-exposed (n = 4205) and by cumulative APFO exposure, divided into four levels, among ever TFE-exposed and among three categories (low, medium, high) of cumulative TFE exposure (n = 4773). There were 534 deaths among men who had ever been exposed to APFO, of which 159 deaths were caused by cancer.

[The cohort included all TFE-production and polymerization sites worldwide at the time of the study and benefited from near-complete follow-up. Limitations were mainly related to the semiquantitative exposure assessment, with only a few TFE and no PFOA measurements available, no validation of estimated exposures by measurement, a low statistical power to detect less common cancers, and a high correlation between potential exposure to TFE monomer (classified as probably carcinogenic to humans, Group 2A; IARC, 2016) and PFOA. However, exposure to TFE among workers at the Parkersburg facility was considered very unlikely for the vast majority of workers, because processes involving TFE were conducted in a separate area of the plant with limited access, and strict hygiene-control practices (e.g. closed systems) were used to prevent emissions of this highly flammable and explosive compound. Moreover, possible exposure to other occupational and non-occupational carcinogens was not accounted for.]

2.1.7 Danish National Birth Cohort

The Danish National Birth Cohort was recruited for a nationwide cohort study that included data on about 100 000 pregnancies that occurred from 1996 through 2002 (Olsen et al., 2001). Approximately 50% of Danish women who were pregnant during this period were invited to participate when consulting their general practitioner during their first pregnancy visit, usually at weeks 6-12 of gestation; of those invited, about 60% agreed to participate in the cohort. Study participants completed questionnaires on lifestyle factors and environmental exposures, including diet, body size, alcohol intake, and smoking history, during a computer-administered interview at two time points during pregnancy as well as at two time points after pregnancy (6 and 18 months after delivery). Blood samples were collected once during the first trimester and once during the second trimester of pregnancy; cord blood was also collected at delivery.

A nested case-control study was designed to evaluate serum PFAS from blood samples collected during the first trimester of pregnancy in relation to risk of premenopausal breast cancer in mothers recruited to the Danish National Birth Cohort. Linkage to a nationwide cancer registry was used to identify 250 women who had received diagnoses of premenopausal breast cancer, with follow-up until 2010. These 250 women with breast cancer were matched to 233 randomly selected controls, with frequency-matching by age and limitation to those who were nulliparous (Bonefeld-Jørgensen et al., 2014). [The Working Group interpreted this nulliparity restriction to refer to women who, at the time of their blood sample during pregnancy, had not had a previous live birth.] In a follow-up analysis focused on gene-environment interaction, 72 of the original cases included as part of the study of **Bonefeld-**Jørgensen et al. (2014) were excluded because they had been withdrawn from the Danish National

Patient Register for unknown reasons, resulting in a case group of 178, with a control group of 233 (Ghisari et al., 2017). [The Working Group noted that the removal of the 72 cases probably had no implication on the matching, since the age distribution among cases and controls appeared to be balanced after this removal (Ghisari et al., 2017).] Bonefeld-Jørgensen et al. (2014) conducted sensitivity analyses excluding these 72 cases and observed some differences in their findings. [The Working Group prioritized the results presented by Ghisari et al. (2017), given the withdrawal of some of the women from the study; the article by Bonefeld-Jørgensen et al. (2014) was, therefore, not tabulated as part of this monograph.]

The first-trimester blood samples (taken some time in weeks 6–14 of gestation) were stored and used for PFAS analysis. For cases, the average age at blood sampling was 30.6 years (range, 21-42 years), and the average at diagnosis was 41.1 years (range, 32-53 years). Serum concentrations of PFAS, including PFOA and PFOS, were assessed using LC-MS/MS. The association between PFAS and breast cancer risk was evaluated using unconditional logistic regression models, with PFAS concentrations transformed using a natural log transformation characterized as a continuous variable, and with adjustment for potential confounders (age at blood sampling, BMI before pregnancy, gravidity, oral contraceptive use, age at menarche, smoking during pregnancy, alcohol intake during pregnancy, maternal education, physical activity).

[The Working Group noted some strengths of this analysis, including a fairly large sample size of premenopausal cases of breast cancer and the measurement of PFAS serum concentration years before breast cancer diagnosis. The study included only a single PFAS concentration as a measure of exposure, assessed during the first trimester of pregnancy; in itself, this is not expected to introduce confounding because it was collected early in pregnancy (Sagiv et al., 2018). Further, the Working Group noted that while only having a single measure of PFAS may be considered a limitation, there is some evidence, from analyses of repeat sampling of PFOA, that a single sample may represent longterm averages over a 5-8 year period, with potential misclassification resulting in only minor bias towards the null (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications.iarc. who.int/636). However, a concern remained that single measurements of PFOA or PFOS, despite their long half-life, might not represent average exposure over the longer periods that may be relevant to cancer etiology. Although many key plausible confounders were adjusted for in the study, it would have been better, given temporal trends in PFAS, to match or adjust for the year of blood sampling and consider adjustment for other PFAS. Pregnancy might be an important window of susceptibility for breast cancer, but this design might also limit the comparability of these results with those obtained using measurements for serum samples collected from non-pregnant women. There was a lack of data on cancer subtypes (e.g. histology, hormone-receptor status), and results were presented for all cancer subtypes combined, which could mask cancer subtype-specific associations and bias the overall risk estimate towards the null. Other limitations include relatively small exposure contrasts in a population with background exposures, and the potential non-applicability of the findings to postmenopausal breast cancer. The reasoning behind the exclusion of the 72 women, who might, in fact, not have had breast cancer is a limitation and resulted in a smaller sample size, but the Working Group noted that although the results of analyses slightly differed with and without them, the conclusions remained similar.]

2.1.8 California Teachers Study

The California Teachers Study (CTS) is an ongoing prospective cohort study that includes women who were current and former publicschool professionals in California, USA, who were enrolled in the California State Teachers Retirement System in 1995 (Bernstein et al., 2002). A self-administered baseline questionnaire was posted to 329 684 women, with approximately 40% responding (Bernstein et al., 2002). In 1995–1996, 133479 women were enrolled in the cohort by completing the questionnaire. The baseline questionnaire covered menstrual and reproductive history, use of exogenous hormones, diet, smoking, alcohol use, height, weight, family history of cancer, and individual's medical history. The mean baseline age for the study participants was 54.1 years (standard deviation, 14.8 years); the ethnicity of the cohort was primarily non-Hispanic White (86.7%) (Bernstein et al., 2002).

Study participants were followed up annually to update details on cancer diagnoses, deaths, and residential moves. Participants were also sent more detailed follow-up questionnaires, focusing on exposures of interest. The study also uses linkages to state and national mortality files and reports from next of kin for dates and causes of death (<u>Hurley et al., 2018</u>).

A nested breast cancer case–control study (913 cases, 1270 controls) was conducted within the CTS and included blood sample collection and an interview-administered questionnaire (Hurley et al., 2018). Women were eligible if they had received a diagnosis of invasive breast cancer between 1 January 2006 and 1 August 2014, were aged < 80 years at the time of diagnosis, had no previous history of breast cancer at the time of entry into the CTS, and had resided in California continuously from the time of cohort entry to the time of diagnosis. Breast tumours were identified by linkage to the California Cancer Registry and were confirmed by pathology (99%). Thus, as long as participants remained in California, they were actively followed up for cancer diagnosis. Controls were randomly sampled with frequency-matching by 5-year age group, race or ethnicity, and the California Cancer Registry regional entry of residence. Participation rates were 55% for controls and 65% for cases.

Phlebotomist-collected blood samples were stored and assayed for serum PFAS using LC-MS/ MS. For the cases, blood samples were collected, on average, 35 months after a diagnosis of breast cancer (range, 9 months to 8.5 years). Samples collected before October 2011 were excluded, owing to concerns regarding time trends in PFAS levels and time trends in sample collection by case status, primarily affecting controls. [The Working Group noted that it was unlikely that this exclusion would introduce any bias, since it resulted largely in the exclusion of controls, rather than cases.] After exclusions, the final sample size was 902 cases and 858 controls. The associations between each PFAS detected in the serum samples and the risk of breast cancer were estimated using unconditional logistic regression (given the breaking of initial matching), adjusting for confounders (age at baseline, race or ethnicity, region of residence, blood draw date, the square of blood draw date, season of blood draw, total pack-years smoking, BMI, family history of breast cancer, age at first fullterm pregnancy, menopausal status at blood draw, and pork consumption). Concentrations of PFAS were considered both as continuous variables (log₁₀-transformed) and as categorical variables (based on tertiles of PFAS concentrations in the controls). Estimates were stratified by menopausal status at blood draw, estrogen receptor and progesterone receptor (ER/PR) status of the tumour, and other factors.

[The Working Group noted that the strengths of this study included a large number of cases identified through the cancer registry and population-based controls. Other strengths included adjustment for a large number of potential confounders and the stratified analysis by a number of important factors regarding breast cancer, including hormone-receptor status of the tumour. The primary limitation was the use of a single blood sample collected between 9 months and 8.5 years after diagnosis and presumably after at least initial treatment, both of which might affect blood PFAS concentration, and that this did not reflect the probable etiologically relevant period. This limitation rendered the study of minimal informativeness.]

2.1.9 Perfluorocarbon-production workers

Girardi and Merler (2019) studied the association between PFAS (including PFOA and PFOS) and mortality in a cohort of 462 male employees who had worked \geq 6 months before 2009 in a factory in Italy. There were 14 658 person-years and 107 deaths, with an average follow-up time of 31.7 years. The factory had produced PFOA, POSF, and other chemicals (including one other PFAS, perfluorobutylsulfonyl fluoride) since 1968. Follow-up covered the period 1970 to 2018. Information on the underlying causes of death was obtained from the regional epidemiological department and, if not available, from the local health unit register for deaths after 1990, or from the complete death certificate, as recorded by the birth and death register of the municipality where the death had occurred before 1990. Results were given for a wide variety of outcomes, including all cancers combined and six specific cancer types: oesophagus, stomach, colon, liver, lung, all lymphoma or haematopoietic cancers, and NHL.

Measurements of PFOA serum concentration in workers, available for a subsample of the cohort (n = 120), were used to develop a regression model for job-specific levels across time. These models were then used to estimate a cumulative serum PFOA concentration for each cohort member. Employees were classified: (i) by three PFAS (either PFOA or PFOS) exposure categories (office workers, never in PFAS-exposed department, ever in PFAS-exposed department); and (ii) by tertiles of estimated cumulative PFOA serum concentrations. SMRs were calculated for the exposed cohort compared with the regional population (adjusted for age, sex, calendar time). Poisson regression risk ratios were also calculated, taking workers of a nearby metalworking factory, who were working with the Italian train system, as referents. [The Working Group noted that there was some exposure to asbestos in the metalworking factory, which might have biased deaths from lung disease towards the null.] Additional analyses were conducted to calculate SMRs and risk ratios across categories of probability of PFAS exposure and tertiles of cumulative serum PFOA concentration using the regional population and the metalworking cohort as the referent, respectively.

Serum PFOA concentrations among 120 workers in the period 2000–2013 (696 measurements) showed a geometric mean of 4048 ng/mL (range, 19–91 900 ng/mL). For these same 120 workers (615 measurements), serum PFOS results showed a much lower geometric mean of 148.8 ng/mL (range, 10–3386 ng/mL). The intrasample correlation between the PFOA and PFOS concentrations was high (Spearman correlation, $\rho = 0.59$; P < 0.001).

[The Working Group noted the exceptionally higher levels of PFOA exposure than in other occupational cohorts, with a resulting high exposure contrast, as a strength. The use of a JEM and some serum measurements to build a model of cumulative PFOA exposure that had evidence of a good fit to observed data, and comparisons with non-exposed workers, which might reduce confounding, were also strengths. While there was exposure to PFOS, serum PFOS concentrations were much lower than those of PFOA, and PFOS was not considered to be a potential confounder. Limitations were: a small occupational cohort with few deaths (107 deaths, 42 from cancer), no data on confounders (although the use of a worker comparison population might

reduce confounding), a small number of deaths (7 each) from liver cancer and lymphohaematopoietic cancer (the two causes with positive trends with exposure), no data on some causes of death of major interest (e.g. cancers of the bladder, prostate, or testis). This study was of moderate informativeness, owing to the documented high exposure, but was limited by the small number of cancer outcomes.]

2.1.10 E3N cohort

E3N (Etude épidémiologique auprès de femmes de la Mutuelle générale de l'Education nationale) is a prospective population-based cohort study of 98 995 women in France that was initiated to identify risk factors for cancer and other chronic diseases in women (Clavel-Chapelon et al., 2015). In 1990, a questionnaire was sent to almost 500 000 women aged 40–65 years who were part of a national health insurance programme for workers, primarily teachers from the French national education system, inviting them to enrol in the study. Approximately 20% responded to the questionnaire, with 98 995 participants enrolling in the cohort.

At baseline, participants completed a self-administered questionnaire and consented to the study team accessing their health insurance records. Participants completed follow-up questionnaires every 2–3 years after baseline, with an average response rate of approximately 80%, and with limited loss to follow-up (< 3%). These questionnaires included information on a range of demographic and lifestyle factors and suspected risk factors for cancer. Between 1994 and 1999, approximately 25 000 participants (participation rate, 40%) donated blood samples. Sample aliquots were stored in liquid nitrogen in a biobank.

A nested prospective case–control study was conducted to evaluate serum PFOA and PFOS concentrations in relation to breast cancer risk

(Mancini et al., 2020a). Cases of breast cancer were identified by self-report, from the health insurance files, or through death certificates. Deaths could be reported by next of kin or ascertained from the health insurance files; cause of death was identified using the NDI. Pathology reports were available for most of the cases (93%), but self-reported cases without a pathology report were included in the analysis. In the cohort, 281 cases of breast cancer were identified that were diagnosed before 2013 and for which at least three aliquots of serum were available. Cases for which the dietary questionnaire had not been completed or diagnosed before the blood sampling or before completing the dietary questionnaire were excluded. The length of time between drawing of the blood sample and cancer diagnosis was not reported. From the eligible 240 cases of incident cancer remaining after exclusions, 194 cases of incident postmenopausal breast cancer were randomly selected for the study; this reduction was due to budgetary constraints. A control (also n = 194) was sampled from the cohort for each case, using a density-sampling approach based on not having a breast cancer at the time that the corresponding case was diagnosed, with matching by age (± 2 years), menopausal status and BMI at blood collection, and year of blood collection. Mean age at diagnosis was 68.8 years (range, 58.3–84.9 years). Information on tumour hormone-receptor expression was available for ER for 158 cases (77%), and for PR for 155 cases (80%). In total, 132 tumours were positive for ER (ER+) and 98 were positive for PR (PR+).

Concentrations of PFOA and PFOS were measured using LC-MS/MS. Both PFOA and PFOS, categorized in quartiles, were evaluated in relation to breast cancer risk using conditional logistic regression. [The Working Group noted that while collection of a single blood sample might be considered a limitation, there is some evidence, from repeat sampling of PFOA, that single samples may represent long-term averages over a 5-8 year period, with potential misclassification resulting in only minor bias to the null (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: <u>https://publications.iarc.who.int/636</u>). However, there remains the concern that single measurements of PFOA or PFOS, despite the long half-life of these chemicals, might not represent average exposure over longer periods that might be relevant to cancer etiology.] Mancini et al. (2020a) explored several statistical models for confounder adjustment and considered a large number of confounders, selected a priori (total serum lipids, BMI, smoking status, physical activity, education level, history of benign breast disease, family history of breast cancer, parity or age at first full-term pregnancy, breastfeeding duration, age at menarche, age at menopause, use of oral contraceptives, current use of menopausal hormone therapy, adherence to a "Western" or "Mediterranean" diet). Stratified analyses evaluated how these associations varied by ER and PR status of the tumour.

The Working Group noted that strengths included the prospective design and collection of blood specimens, the availability of blood data before diagnosis, extensive confounder control, limited loss to follow-up, and the availability of detailed diagnostic information (e.g. ER and PR status for nearly all cases of breast cancer). Limitations included relatively small exposure contrasts in a general population sample with background serum PFOA and PFOS concentrations and a lack of cases in premenopausal women. The response rate for the blood donation was low, which might affect the generalizability of the findings. The analyses by hormone-receptor subtype, while important, were limited by the small sample sizes, and there was a lack of information on the time between blood sampling and diagnosis.]

2.1.11 PLCO Cancer Screening Trial cohort

The PLCO was a large randomized controlled trial (about 150 000 adults; 76 685 men and 78 216 women), aged 55–74 years from 10 large cities in the USA), conducted in 1993–2001, and designed and sponsored by the National Cancer Institute. The goal was to determine the effects of screening on cancer-related mortality and secondary end-points in men and women aged 55-74 years, recruited between 1993 and 2001 (Prorok et al., 2000; Rhee et al., 2023a). The target populations for recruitment differed between the 10 clinical sites in the trial: recruitment methods included mass mailings using purchased mailing lists or lists of patients in local areas. Eligible women had not previously received diagnoses of cancer and had not undergone the screening tests in the 3 years preceding enrolment for testing in the trial. In the PLCO Trial, blood samples were to be collected from participants in the screening arm and stored for future etiological research (Hayes et al., 2000): blood samples were collected from 95% of screening-arm participants at baseline. Serum PFAS concentrations in the controls of the PLCO study were similar to those in the US NHANES study collected at about the same time, suggesting that the studied population was representative of the US population (Shearer <u>et al., 2021</u>).

Four separate nested case-control studies were conducted for the PLCO Trial cohort, investigating the association between PFOA or PFOS and cancers of the kidney, prostate, breast, and pancreas.

Shearer et al. (2021) conducted a nested casecontrol study within the PLCO cohort; this study involved 324 cases (216 men, 108 women) with RCC (the main subtype of kidney cancer) and 324 matched controls who were alive and free of RCC after the diagnosis dates of their corresponding matched case. Controls were matched individually on age at enrolment, sex, race and ethnicity, study centre, and year of blood draw. Exposure assessment was based on PFOA, PFOS, and other PFAS measured in serum collected 2-18 years before cancer diagnosis (mean, 8.8 years). Analyses using conditional logistic regression controlled for BMI, kidney function (estimated glomerular filtration rate, eGFR), smoking status (never, former, or current), history of hypertension, prior sample freeze-thaw cycles, and calendar year of blood draw. Analyses were conducted for eight different PFAS measured at baseline, including PFOA and PFOS. Additional analyses considered PFOA, PFOS, and PFHxS jointly. Analyses considered quartiles of serum concentrations, as well as continuous (log₂-transformed) serum levels. Geometric mean concentrations of PFOA among controls were 4.0 and 4.5 µg/L [ng/mL] for women and men, respectively; those for PFOS were 31.3 and 38.1 µg/L [ng/mL] for women and men, respectively, similar to serum concentrations for the general population at the time.

<u>Steenland et al. (2022)</u> included the cases and controls from <u>Shearer et al. (2021)</u> in a pooled analysis of PFOA and kidney cancer, combined with 103 cases and 511 matched controls from a PFOA-exposed cohort in West Virginia and Ohio previously reported by <u>Barry et al. (2013)</u>. This pooled analysis was conducted to derive a dose-response curve between serum PFOA concentration and risk of kidney cancer using two of the largest studies of kidney cancer and PFOA exposure, and to conduct a risk assessment of excess lifetime risk of kidney cancer for specific PFOA serum concentrations and rates of drinking-water consumption.

<u>Rhee et al. (2023a)</u> studied 750 cases and 750 matched controls nested within the PLCO cohort. They looked at associations between a variety of PFAS, including PFOA and PFOS, measured at baseline, and subsequent prostate cancer. There were 750 men with aggressive prostate cancer (defined as stage III or IV or Gleason score \geq 8, or Gleason score 7 and death from prostate cancer). Cases were diagnosed > 300 days after baseline

blood collection (median, 9 years). Controls were selected from among eligible participants who were alive and cancer-free at the time of the case diagnosis and were individually matched on age at baseline, race or ethnicity, calendar year of baseline blood collection, and characteristics of blood sample (e.g. whether thawed or not). Analyses were further adjusted for BMI, smoking status, family history of prostate cancer, history of diabetes, and serum concentrations of seven other PFAS. All eight PFAS were detected in more than 95% of samples; most PFAS were moderately correlated, with the highest correlation being between PFOA and PFOS ($\rho = 0.70$). Rhee et al. (2023a) also collected multiple serum samples (at baseline, and at 1- and 5-year follow-up) from a subset of controls (n = 60) and found that the variance between participants was generally markedly higher than the variance within participants; intraclass correlation coefficients across repeats for PFOA and PFOS were 0.73 and 0.85, respectively.

Chang et al. (2023) conducted a nested casecontrol study of breast cancer among women who were postmenopausal at baseline in the PLCO cohort. There were 621 cases and 621 controls, individually matched on age at baseline, date of blood draw, and menopausal hormone therapy use at baseline, who were alive and cancer-free past the follow-up time of their corresponding matched cases. Prediagnostic serum concentrations of PFOA and PFOS, measured 1 year after baseline, with a median of 5.6 years before case diagnosis, were the exposures of interest. In another study, intensity levels of PFOA and PFOS were assessed using the same untargeted metabolomics platform and were highly correlated with targeted measured serum concentrations of PFOA and PFOS (Spearman correlations, 0.77 and 0.76, respectively). Analyses were conducted for all cases combined by quartile of PFOA and PFOS intensity levels, and by ER (ER+, 435 cases) or PR (PR+, 299 cases) status, and joint ER/PR status. Models were adjusted for age at blood

sampling, established breast cancer risk factors (age at menarche, age at first live birth, number of live births, age at menopause, duration of menopausal hormone therapy use, first-degree family history of female breast cancer, personal history of benign breast disease, BMI, smoking status, vigorous physical activity), natural log-transformed intensity levels of PFOA (for the PFOS model) or PFOS (for PFOA models), and variables, whose removal resulted in a \geq 10% change in the odds ratios (study centre, race or ethnicity, and education).

Zhang et al. (2023) studied data for 360 cases of pancreatic ductal adenocarcinoma (the most common type of pancreatic cancer) and 360 matched controls in a nested case-control study in the PLCO cohort. The same study also involved another 251 cases and 251 controls from the Alpha-Tocopherol, Beta-Carotene (ATBC) Cancer Prevention Study, conducted in Finland. The PLCO case and control groups were matched on age, date of blood draw, sex, and race. Blood collected at PLCO baseline was analysed for PFOA and PFOS (median follow-up period, 9 years), using relative intensities rather than absolute concentrations (i.e. based on non-targeted analysis of PFOA and PFOS intensity levels). Previous work has shown that such rankings correlate well with measurements of absolute concentrations (Spearman correlation, 0.76) (Rhee et al., 2023c). Additionally, although untargeted analysis was used by Zhang et al. (2023), in previous PLCO studies, serum PFOA and PFOS concentrations were comparable to those measured among the general population in the NHANES survey in 1999–2000 (Rhee et al., <u>2023a</u>). Data were analysed using conditional logistic regression, with continuous variables and quintiles, and controlling for smoking, BMI, age at blood draw, and diabetes. It did not appear that each PFAS was adjusted for the other.

[The Working Group noted strengths in all four PLCO nested case-control studies. First, all the studies involved large numbers of cases

and controls; second, they all used measured serum concentrations of PFOA and PFOS before cancer diagnosis, which is used as the exposure metric. Other strengths included consideration of a broad range of confounders, and data on tumour characteristics. Another strength was the mutual control for other PFAS under study (except in <u>Zhang et al., 2023</u>), and the availability of repeat samples in the prostate cancer study to assess reproducibility. Limitations in all four studies included the use of a single measurement to characterize long-term exposure, although the Working Group noted that in Rhee et al. (2023a) repeated measures of PFAS in a subsample indicated good reproducibility within individuals. Another weakness was the relatively low contrasts in exposure, typical of a general population study, making it more difficult to identify potentially corresponding health effect contrasts. While not necessarily a weakness, it was noted that the prostate cancer study by Rhee et al. (2023a) and the breast cancer study by Chang et al. (2023) were restricted to aggressive cancers and postmenopausal cases, respectively. However, for etiological research, it is probably more important to focus on aggressive than indolent prostate cancer, and postmenopausal breast cancer represents the great majority of breast cancer cases. Finally, the Working Group noted that the PLCO cohort had a higher percentage of White participants, e.g. 88-89% in Rhee et al. (2023a) and Chang et al. (2023), compared with the current US population (75%) (United States Census Bureau, 2022).]

2.1.12 Child Health and Development Studies pregnancy cohort

The Child Health and Development Studies (CHDS) pregnancy cohort included 20754 pregnancies that resulted in 19 044 live births, including 9300 daughters. Between 1959 and 1967, pregnant women in the Oakland area of California, USA, receiving obstetric care

through the Kaiser Foundation Health Plan were invited to participate (> 99% of all eligible women enrolled). The researchers obtained access to the medical records of all pregnant women recruited in the cohort, and to those of their children, and collected blood samples from the mothers (mostly, one blood sample per trimester and postpartum). Moreover, all mothers participated in an in-person interview during pregnancy (van den Berg et al., 1988).

To collect information on residence history and update residency and name changes over time, the CHDS cohort was linked to the California Department of Motor Vehicles. Data on the history of residential location of all women recruited in the CHDS cohort were used to identify the population at risk of cancer, corresponding with geographical surveillance by California's cancer registries. Deaths and cases of cancer were identified through linkage of the CHDS with the California Vital Status Records and the California Cancer Registry, respectively. Cases were also identified by self-report in a survey of CHDS daughters conducted in 2010– 2013 (Cohn et al., 2020).

A nested breast cancer case-control study was conducted within the CHDS cohort, including 102 cases with breast cancer and 310 controls. Cases were identified through surveillance and self-report until March 2013 and were defined as CHDS daughters with cases of incident invasive or non-invasive breast cancer, which were diagnosed when they were aged < 52 years. Only cases of cancer for which a maternal perinatal blood sample was available for the analyses were selected; this led to the inclusion of 86% of all the cases of breast cancer identified. Three controls were matched to each case on birth year and trimester of maternal blood draw. Controls were selected randomly from among CHDS daughters not known to have received a diagnosis of breast cancer at the age of diagnosis of the corresponding case (Cohn et al., 2020).

After collection, serum samples had been stored at -20 °C, and concentrations of PFAS, including PFOA and PFOS, were measured using LC-MS/MS. The association between maternal PFAS serum levels and the daughter's risk of breast cancer was evaluated using age-matched conditional logistic regression, with PFOA and PFOS serum concentrations analysed as continuous variables after log, transformation and adjusting for potential maternal confounders, such as maternal age, race, overweight in early pregnancy, parity, maternal history of breast cancer, maternal serum log₂-transformed dichlorodiphenyldichloroethylene (*p*,*p*'-DDE), maternal serum log₂-transformed dichlorodiphenyltrichloroethane (*o*,*p*'-DDT), and whether the daughter was breastfed (Cohn et al., 2020). Models evaluating PFOS also considered, a priori, inclusion of a PFOS precursor, (N-EtFOSAA), and total maternal cholesterol, both log₂-transformed, and their product term (to test for interaction).

[The Working Group noted as a strength of the study that cases were likely to have been accurately determined via the California Cancer Registry.]

The Working Group also noted that blood PFOA and PFOS concentrations decrease during pregnancy, owing to expanding plasma volume, decreased albumin concentration, and increased glomerular filtration rate. Nevertheless, it has been reported for previous studies that blood PFOA and PFOS concentrations measured during different pregnancy trimesters and postpartum in mothers, as well as measurements of cord blood, are well correlated (Glynn et al., 2012; Kato et al., 2014; Pan et al., 2017; Nielsen et al., 2020). This implies that blood PFOA and PFOS concentrations measured at different times during pregnancy and postpartum can be predictive of fetal exposure during pregnancy. Conversely, Cohn et al. (2020) analysed and compared, in the same study, PFOA and PFOS concentrations measured during pregnancy (22% of the samples) and in the early postpartum period (78% of the samples). This could have introduced a potential non-differential exposure misclassification bias, because not all blood samples were collected at the same time during pregnancy.]

[The Working Group noted that the study included examination of the prenatal exposure window, which is of interest with regard to the etiology of breast cancer in general. However, it was noted that no serum PFAS levels were directly available from study participants, namely, the CHDS daughters, and exposure was notably restricted to prenatal exposure, limiting the generalizability of the results and the possibility for comparison with other studies. Another weakness noted was that no individual information concerning the CHDS daughters was available in the study (<u>Cohn et al., 2020</u>), so that the analyses did not include important confounders. In the 1960s, PFAS contamination was expected to be still low in the USA general population (ATSDR, 2020) and this was reflected by the very low serum PFOA concentrations measured in women of the CHDS cohort, which affects the informativeness and the comparability of the results of the study (Cohn et al., 2020). In contrast, serum PFOS concentrations measured in women of the CHDS cohort were unexpectedly high, considering the time period of blood sampling; no explanation was provided by the authors for these high values. Also affecting generalizability was the restriction to cases diagnosed in daughters younger than 52 years. Additionally, the Working Group noted the lack of information on what percentage of the cohort moved out of the state and could not be followed up. Finally, there was a lack of data on cancer subtypes (e.g. histology, hormone-receptor status), and results were presented based on all cancer subtypes combined, which could mask cancer subtype-specific associations and bias the overall risk estimate towards the null. Overall, on the basis of these limitations, the Working

Group considered this study to be of minimal informativeness.]

2.1.13 The Ronneby Register cohort

Li et al. (2022a) investigated cancer incidence in residents in an area with high-level environmental exposure to, primarily, PFOS and PFHxS, in Sweden. The municipality of Ronneby, on the Baltic coast in the southern part of Sweden, had about 28 000 residents in 2013, and drinking-water was supplied by two waterworks. One of these, situated 2 km from a military airfield, supplied one third of the households of Ronneby municipality (a map of the area is given in Xu et al., 2021). In December 2013, measurements of PFAS in drinking-water from this waterworks revealed sum of PFAS concentrations above 10 000 ng/L [10 ng/mL], whereas the concentration for the other waterworks was below 90 ng/L [0.09 ng/mL] – but still higher than in the drinking-water of the neighbouring municipality of Karlshamn. It was found that PFOS and PFHxS accounted for > 90% of the total PFAS, while the PFOA contamination was relatively limited and strongly overlapping with the concentrations in the reference population. The source of the contamination was the use of PFAS-containing firefighting foam at the airfield from about 1985 until the waterworks was closed, by the end of 2013. No measurements were available before 2013, but the study authors assumed that levels in the drinking-water increased during the years after 1985 and decreased after the end of the exposure to contaminated water, in late 2013 (Li et al., 2018). The Ronneby Register cohort includes 60 507 individuals (men, 53%) who ever lived in Ronneby municipality in the period 1985-2013. Individual exposure was classified by coupling registry information on yearly residential address with information on which addresses had been supplied with contaminated water by the waterworks. Exposed individuals were those who had ever lived in the contaminated district ("ever-high", n = 15 811; 26% of the Ronneby population). This group was subdivided by the calendar period and by the number of years living at an ever-high residence: "early-high" (1985–2004), "late-high" (2005–2013), "shorthigh" (1–10 years), and "long-high" (\geq 11 years). An internal referent was defined: inhabitants who had ever lived in Ronneby municipality in 1985–2013 but never at addresses receiving contaminated drinking-water. There were [44 696 (74%)] residents with never-high exposure (data derived from Li et al., 2022a, Table 4); the mean age at entry into the cohort was between 30 and 33 years, according to sex and exposure group.

The external reference groups included a regional population (the population of Blekinge County, excluding Ronneby municipality) and a national population (the whole Swedish population). The exposure classification was validated by measurements of several PFAS in the serum of 3084 people from Ronneby municipality (everhigh and never-high), sampled in 2014-2015, and in the serum of 226 people from a neighbouring municipality, sampled in 2016. The ratio of geometric mean levels of PFOS in the latehigh group, relative to reference residents, was (239 ng/mL)/(3.9 ng/mL) = 61.3, that for PFHxS was (210 ng/mL)/(0.84 ng/mL) = 250, and that for PFOA was (13 ng/mL)/(1.5 ng/mL) = 8.7. Data on cancer occurrence during the follow-up period 1985-2016 were obtained from the Swedish Cancer Register (using ICD-7 codes). In all, 5702 first-occurring cancers were identified in the Ronneby Register cohort (n = 60507), with 495 identified in people in the group with the highest exposure, the late-high group, including 374 in people who were in the long-high group. Age-, sex-, and calendar year-standardized incidence ratios (SIRs) were computed for a large number of cancer sites for residents who had never, or had ever, resided at addresses with contaminated water, compared with those residing in uncontaminated areas (external analysis). Internal comparisons based on Cox regression models were made using the calendar year as the underlying timescale and were adjusted for age and sex. Information on lifestyle and health behaviours was not available, but annual data on highest attained education (from 1990 onwards), residence, work address, and demographic data were obtained from Swedish nationwide registers.

The strengths of this cohort were the large general population sample with complete ascertainment and follow-up, owing to the use of high-quality Swedish population registers with complete population coverage, and a strong documented exposure contrast. A limitation was the mixed exposure profile, with high levels of PFOS and PFHxS, and somewhat elevated but significantly lower levels of PFOA, as well as the lack of individual serum measurements, or individual water contamination or consumption measurements, hence necessitating an ecological exposure assignment into groups by area and time of residence. Conversely, the grouplevel exposure assignments may have captured the large exposure contrasts in this population and were supported by a large number of serum measurements. The limited information on potential confounders may be a minor issue, since this is unlikely to be dependent on the water distribution system, which also fits with the sensitivity analysis adjusted for highest attained education (which has been shown to correlate with smoking in Sweden; Eek et al., 2010) showing no change of results. A lack of historical information on area-level contamination of drinking-water with PFAS, particularly during earlier years of the study period was, however, a limitation.]

2.1.14 Dongfeng-Tongji cohort

The Dongfeng-Tongji (DFTJ) cohort is an ongoing prospective study including over 41 000 retired workers recruited from an automotive company in Shiyan, China (<u>Wang et al., 2013</u>). The company is one of the three largest vehicle

manufacturers in China and was founded in 1969, so that most first-generation workers had already retired when the DFTJ cohort was initiated. Participants in the DFTJ cohort were recruited in two waves: the first from September 2008 to June 2010, which included 27 009 participants; and the second from April to October 2013, which included 14 120 participants (Feng et al., <u>2022</u>). The participation rate was approximately 87% during the first wave (Wang et al., 2013) and was not reported for the second wave; however, responders and non-responders reported similar sociodemographic characteristics (Feng et al., 2022). At inclusion, all participants answered face-to-face interviews, underwent physical examinations, and provided a blood sample. For each participant, 10 mL of peripheral venous blood was collected once at inclusion after overnight fasting. Plasma was separated from the blood sample and stored at -80 °C (Wang et al., 2013; Feng et al., 2022).

PFAS plasma levels were low in this cohort: PFOS had the highest median plasma concentrations (10.36 ng/mL), followed by PFOA (1.19 ng/mL).

Cases of incident breast cancer, diagnosed from September 2008 until the end of 2018 (median follow-up 9.6 years), were identified by reviewing participants' medical records or death certificates provided by the five hospitals owned by the automotive company and by the local Center for Disease Control and Prevention.

A case-cohort study was conducted among women included in the DFTJ cohort to investigate the association between plasma levels of six PFAS, including PFOA and PFOS, and breast cancer risk. Women with prevalent cases of cancer at baseline, those with insufficient blood specimens, and those who were lost to follow-up were excluded from the case-cohort study. Among the remaining 18 387 women, 226 were identified as incident breast cancer cases during follow-up. A subcohort of 990 women was randomly selected from the base cohort (n = 18 387) according to strata determined by age, and among these women, 13 (1.31%) developed breast cancer during follow-up (Feng et al., 2022).

For all women included in the subcohort, the plasma concentrations of six PFAS, including PFOA and PFOS, were measured using LC-MS/ MS. The association between PFAS and breast cancer risk was evaluated using Barlow-weighted Cox proportional hazard models adjusted for potential confounders selected a priori, such as age, BMI, smoking, drinking, marital status, education level, occupation type, batch to enter the cohort, parity, menopausal status, history of mastitis, use of hormone replacement therapy, and family history of cancer. PFAS concentrations were natural log-transformed and included one-by-one in separate models for six PFAS, modelled as continuous and categorical variables (in quartile groups identified based on the distribution of each PFAS in the subcohort) (Feng et al., 2022).

[The Working Group noted that this study represents a large cohort of retired Chinese workers with low-level exposure to PFOA, PFOS, and other PFAS compounds. Strengths are high baseline participation, good control for confounders obtained by interviews, limited loss to follow-up, ascertainment of diagnoses by medical records in five company-financed hospitals and death certificates. However, details were not provided on the probable completeness of diagnoses using these methods, nor the percentage of women whose diagnoses were confirmed only via death certificate, which would result in an unknown diagnosis date. Nevertheless, it cannot be predicted in which direction the risk estimates could be affected by the unknown diagnosis date. Other strengths included the availability of prediagnostic serum samples (an average of 9.6 years before diagnosis, but the range of time span from PFAS measurements to diagnosis was not provided). Blood samples were collected only once at baseline.

The Working Group noted that there was some evidence, from analyses of repeat samples of PFOA, that single samples may represent longterm averages over a 5-8-year period, with potential misclassification resulting in only minor bias to the null (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications. iarc.who.int/636). However, a concern remained that single PFOA or PFOS samples may not represent lifelong average exposure. Whether the selection of retired workers had an impact on risk estimates was uncertain, but there was no a priori reason to suggest any marked bias due to this selection. No cumulative lifelong exposure metric was available, and the very low exposure contrast for PFOA and PFOS limited the evaluation of exposure-response associations.]

2.1.15 NHANES 1999–2014 cohort

The NHANES is a continuously conducted and nationally representative cross-sectional survey. Participants are selected through a statistical process using census data to be representative of the noninstitutionalized population of the USA and are recruited via mailed letters inviting them to participate in NHANES. Between 1999 and 2014, individuals completed a household interview and a medical examination that included a blood sample collection and an assessment of anthropometric measures. Questionnaires assessed information including demographics, socioeconomic status, alcohol use and smoking history, diet, and medical history. [The Working Group noted that detailed information on the individuals included in the cohort, such as selection and participation rates, was not readily available.]

Among participants with stored blood samples, serum concentrations of PFOA and PFOS were quantified using LC-MS/MS (Wen et al., 2022).

The US National Center for Health Statistics has linked NHANES 1999-2014 cohort participants to the NDI to identify deaths and determine the underlying causes of death using probabilistic matching criteria based on identifiers such as social security numbers and date of birth. Participants were followed for cause-specific mortality until 31 December 2015. If the individual did not match to the NDI, this person was assumed to be alive as of the end of follow-up date. Mortality from all cancers combined was one of the cause-specific death categories included in the files available for public use, and no site-specific cancer mortality data were available. For 11 747 of the cohort participants aged \geq 18 years at baseline and with blood samples analysed for PFAS, 1251 deaths were observed during the median follow-up of 81 months (interquartile range, IQR, 46–112 months). Of these deaths, 19.8% (248) were from cancer. The medians and 25th and 75th percentiles of serum PFOA and PFOS concentrations at baseline were 3.27 ng/mL (2.00, 5.00) and 11.60 ng/mL (6.40, 22.40), respectively, reflecting general population levels. [The Working Group noted that this study included only a single measure, which may be considered a limitation; however, there was some evidence, from analyses of repeat samples of PFOA, that single samples may represent long-term averages over a 5-8-year period, with potential misclassification resulting in only minor bias to the null (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications. iarc.who.int/636). However, there remained the concern that single PFOA or PFOS samples may not represent average exposure over longer periods.]

Cox proportional hazards models were used to estimate the hazard ratios (HRs) and 95% confidence intervals (CIs) for the association between PFOA and PFOS categorized in tertiles and overall cancer mortality adjusting for potential confounders (Wen et al., 2022). The confounders included were sex, age, race or ethnicity, smoking status, alcohol intake, physical activities, hypertension, diabetes, healthy eating index, creatinine clearance rate, serum total cholesterol, and serum cotinine. Analyses of PFOA and PFOS were adjusted for other PFAS.

The Working Group noted some strengths in this study, including probably complete ascertainment of cancer mortality in this large population, which was selected to be a representative sample of the US population, making results generalizable in the USA, relatively good control over potential confounders, and adjustment for other PFAS in analyses of specific PFAS. Weaknesses noted were the unclear timing of the blood sampling relative to diagnosis or treatment; a short follow-up time for some of the participants, which may not reflect the relevant etiological window; and the use of a volunteer-based population that may be healthier than the general population, which may not be fully captured by sampling weights and thus may result in some selection bias. The study focused on all-cancer mortality, without data on cancer incidence or on mortality for specific cancers, making these data of limited informativeness for the Working Group's assessment of individual cancers. The observed estimates of association with cancer mortality would not be applicable to cancer incidence unless the cancer itself had a high rate of fatality.]

2.1.16 Multiethnic Cohort study

Kolonel et al. (2000) described the design and implementation of the Multiethnic Cohort (MEC) study, which was established to study diet and cancer in the USA. The MEC included a community sample of 215 251 citizens (men, 45%) aged 45–75 years who were enrolled during 1993–1996 in Hawaii and California (primarily Los Angeles County) when responding to a mailed 26-page baseline questionnaire on mainly dietary, demographic, and health issues. The sampling frame was established using drivers' license files, voters' registration files, and Health Care Financing Administration files. The recruitment procedures aimed to obtain a balanced distribution of five specific ethnic groups comprising White, African-American, Latino, Japanese-American, and native Hawaiian people; therefore, less common groups were preferentially sampled. Response proportions spanned 20% to 49%, being lowest in Latino and highest in Japanese-American people. In the final sample, Japanese-American people comprised the largest group (28% in men and 25% in women) followed by White (24-22%), Latino (24–21%), African-American (13–19%) and native Hawaiian (6-7%). The participants represented a more educated subset of the general population. In addition to quantitative information about food and dietary components based upon portion size information, the questionnaire included data on smoking, drinking, obesity, and vigorous physical activity.

Goodrich et al. (2022) performed a casecontrol study of hepatocellular carcinoma (HCC) nested within the MEC. Incident cancers occurring during the 20 years of follow-up, including non-viral HCC (not of viral etiology), were identified from the early 1990s onwards by the Surveillance, Epidemiology and End Results (SEER) programme of the National Cancer Institute, which includes California. Additional information on health conditions was obtained from Medicare claims and California hospital discharge records (Goodrich et al., 2022). [It was uncertain whether the 50 cases of incident HCC constituted all the HCC cases arising during the follow-up period, given that the authors excluded HCC of viral etiology, and no information was provided about how the 50 controls matched on sex, age, race or ethnicity, and study area were selected.] Plasma concentrations of six PFAS (including PFOA and PFOS) were measured in prediagnostic fasting blood samples collected before diagnosis (median, 7.2 years; range, 0.9–16.4 years).] The unadjusted geometric mean of PFOS was 29.2 μ g/L [ng/mL] in both cases and controls. For PFOA, concentrations were 4.21 μ g/L [ng/mL] in cases and 4.78 μ g/L [ng/mL] in controls, and for PFHxS values were 1.84 μ g/L [ng/mL] in cases versus 2.07 μ g/L [ng/mL] in controls.

The average age at blood collection was similar for cases (69.7 years) and controls (69.2 years). Men comprised 62% of the sample and 64% were residents of Hawaii. The prevalence of high BMI, high alcohol intake, and diabetes mellitus was much higher among cases than among controls.

Adjusted odds ratios for the association between plasma concentrations of each PFAS and risk of non-viral HCC were computed by conditional logistic regression, which accounted for the matching variables. Sensitivity analyses further adjusting for baseline BMI and baseline diabetes status were performed, but additional covariates (such as other PFAS) were not included in the statistical models. An additional sensitivity analysis considered ordinary logistic regression with covariates that included the matching variables. To account for possible non-linear associations, smoothing splines were inspected, and additional analyses contrasting risk above and below the 85th percentile (which corresponded to the 90th percentile in NHANES for PFOS) were carried out.

<u>Rhee et al. (2023b)</u> performed a nested case-cohort study of RCC within the MEC. The study identified 428 cases of incident RCC, which included all cases with prediagnostic serum samples available and diagnosed before 2018 using record linkages to the Hawaii Tumor Registry and the California Cancer Registry. The 428 controls, who were participants with no RCC diagnosis who were alive at the time of the case diagnosis, were identified using 1:1 matching on sex, race or ethnicity, study centre (Hawaii, California), age at serum collection (\pm 1 year), time of serum collection (\pm 3 hours), and fasting status (0 to < 6 hours, 6 to < 8 hours, 8 to < 10 hours, and \geq 10 hours). Concentrations of nine PFAS (including PFOA and PFOS) were assessed in prediagnostic serum samples collected between 1994 and 2006.

Adjusted ORs and 95% CIs were estimated for the association between PFOA and PFOS (log₂-transformed and categorized in quartiles based on the distribution in the controls) and the risk of RCC using conditional logistic regression adjusting for the matching factors as well as BMI, eGFR, smoking status, and hypertension history. Analyses were conducted with further adjustment for other measured PFAS and with stratification by matching factors and other covariates, including race and ethnicity, sex, age at blood draw, calendar year of blood draw, years from blood draw to RCC diagnosis, and eGFR status.

[Strengths of the nested case–control studies conducted within the MEC included prediagnostic measurements of several PFAS compounds in a racially and ethnically diverse population, independent ascertainment of exposure and outcome with high accuracy, a strong focus on possible mechanistic pathways related to PFAS related metabolism, and baseline information on relevant potential confounders including education, socioeconomic level, and health behaviours such as smoking. Regarding the statistical analysis of the HCC study (Goodrich et al., 2022), risk estimates were not adjusted for date of blood sample collection, smoking, alcohol consumption, and other PFAS. Moreover, small numbers in combination with low exposure contrast limited the informativeness of this study. For the RCC study (Rhee et al., 2023b), strengths included the large sample size, adjustment for eGFR and other factors, and the consideration of multiple PFAS for adjustment. Although it was a strength to consider this association across multiple racial and ethnic groups, there was limited statistical power for some of these comparisons.]

2.1.17 Cohort of US Air Force servicemen

Purdue et al. (2023) performed a case-control study nested within a cohort of US Air Force servicemen (with prospectively collected blood specimens) to examine the risk of testicular germ cell tumours (TGCTs; the most common variety) according to adult serum concentrations of selected PFAS, including PFOA and PFOS. The US Air Force was using firefighting foams containing PFOS, PFOA, and other PFAS compounds since the late 1960s until 2018, when the use of long alkyl chain PFAS compounds was discontinued. The US Department of Defense (DoD) has identified over 200 Air Force installations with known or possible release of PFOS and PFOA and, in some airbases, these compounds have been measured in groundwater and drinking-water in amounts exceeding 70 parts per trillion (ppt), the 2016 US EPA lifetime health advisory threshold. The US DoD has since 1985 stored serum samples of members of the Air Force service (and other military branches) collected for the purposes of HIV testing before induction and at periodic medical examinations, overseas assignments, and major overseas deployments. From 2004, all service members had blood samples taken every second year. Samples were stored at -30° C at a central serum repository (Department of Defense Serum Repository, DoDSR), which contains sera from more than 10 million service members (also including US Army and US Navy personnel). DoDSR records have been linked to records of the Defense Medical Surveillance System, providing individual demographic, occupational, and health data.

<u>Purdue et al. (2023)</u> identified TGCT cases by linking DoDSR records with the DoD cancer registry, which contains data on patients diagnosed with cancer at military treatment facilities in the USA. In all, 530 male servicemen with active-duty status, at least one prediagnostic serum sample, no previous history of cancer and aged < 40 years at the time of TGCT diagnosis were identified from 1988 through 2017. In cases with at least two prediagnostic samples and with the earliest sample collected ≥ 5 years before the TGCT diagnosis, a second prediagnostic sample collected as close to the 5-year prediagnostic lag-time as possible was analysed (n = 187). The median time between collection of selected samples was 4 years (range, 0.1–13.3 years). For each case, one randomly selected male control participant was identified among active-duty Air Force servicemen, with no history of cancer at the time of the case diagnosis, by matching on birth year, race or ethnicity (seven groups), year entering military service, and year of serum sampling. The first serum samples were collected 0.3-0.4 years (median values) after entry into military service and 0-20 years before the diagnosis of TGCT. Serum concentrations of nine PFAS, including linear and branched PFOA and PFOS isomers, were measured by LC-MS/ MS. The PFAS serum concentrations of men in the Air Force service were comparable to those of men in the NHANES cohort, albeit slightly higher in earlier years and slightly lower in later years. Fewer than 1% of participants (n = 5) had occupational exposure as a firefighter during military service. Unadjusted and adjusted risk of TGCT was analysed separately for PFOA (sum of linear and branched isomers), PFOS (sum of linear and branched isomers), and other PFAS by conditional logistic regression of matched pairs grouped by quartiles of serum concentrations in controls. Besides matching factors, adjusted analyses accounted for military grade, number of deployments before diagnosis, and the six other PFAS.

[The Working Group noted several strengths of this study, including the nested design, a well-characterized source population, a large, matched dataset, measurements of PFOA and PFOS isomers, two repeated prediagnostic samples collected several years apart in a subset of the population, a reasonable exposure contrast (for PFOS, the upper quartile was > 42.2 ng/mL and the lower quartile was \leq 18.3 ng/mL), and analyses accounting for effects of other PFAS compounds. Limitations were mainly the loss to follow-up of men leaving the military and missing serum samples for a large proportion of TGCT cases (217 out of 747 cases, 29%) that was not addressed in supplementary analyses. The completeness of TGCT ascertainment was not documented but may be a minor issue since an association between PFAS exposure and completeness of TGCT ascertainment seems unlikely. In most cases, only one serum measurement per person was available, and data on known strong determinants of TGCT (such as cryptorchidism) were lacking, but the association of such determinants with PFAS exposure seems unlikely. The Working Group noted that occupational exposure as a firefighter is classified as carcinogenic to humans (Group 1) (IARC, 2023), but the evidence for risk of testicular cancer was *limited*, and very few (n = 5) of the US Air Force servicemen cohort members had been exposed as a military firefighter.]

2.1.18 Finnish Maternity Cohort

The Finnish Maternity Cohort (FMC) is a national registry of women who donated serum during the first trimester of pregnancy. The registry was established in 1983 using residual serum from a national programme to screen for congenital infections (infections transmitted from mother to child during pregnancy) and is estimated to include > 90% of pregnancies among Finnish women during the period 1983-2016 (Pukkala et al., 2007; Holl et al., 2008; Lehtinen et al., 2017). Women donated serum at municipal maternity care units, usually between weeks 10 and 14 of gestation (Madrigal et al., 2024). The registry included each woman's personal identification number and data related to reproductive history, residence at the time of collection,
dates of sample collection and processing, and expected delivery date.

Madrigal et al. (2024) conducted a nested case-control study on the incidence of papillary thyroid cancer, which accounts for approximately 90% of thyroid cancers in Finland (Hakala et al., 2012), by linking the FMC to the nationwide Finnish Population Registry, Cancer Registry, and Medical Birth Register until 2016. The population registry provided information on emigration status and vital status. The cancer registry, which covers all incident cancer cases in Finland since 1953, included date of diagnosis, histology, and stage at diagnosis (Finnish Cancer Registry, <u>2023</u>). The Medical Birth Register, established in 1987, includes data on gestational age, reproductive history, smoking status, BMI before pregnancy and at the prenatal visit, and information about the delivery and infant or fetus, but not prior history of breastfeeding (Finnish Institute for Health and Welfare, 2023). Information on BMI before pregnancy was largely missing. Madrigal et al. (2024) randomly selected 400 cases of papillary thyroid cancer from among all cases (total number not reported) diagnosed among women in the FMC who provided serum in 1986–2010 during their first pregnancy and for whom this pregnancy had resulted in a fullterm live birth with delivery between 1987 and 2010. Cases were restricted to those whose age at sample collection was 18-39 years and who were diagnosed with thyroid cancer \geq 3 years after delivery. First pregnancies only were used to avoid any changes in PFAS levels related to breastfeeding during a previous pregnancy. Age at cancer diagnosis ranged from 23 to 61 years (mean, 40.9 years). Living, cancer-free controls were individually matched on year of delivery (4–5-year increments) and age at first birth (3-year increments). Serum levels of PFAS and other persistent pollutants were analysed by the Environmental Health Unit Laboratory of the Finnish Institute for Health and Welfare using LC-MS/MS (Koponen et al., 2013; Koponen and

Kiviranta, 2019). Analytes included 19 PFAS, 13 polychlorinated biphenyl congeners (PCBs), nine organochlorine pesticides, and three polybrominated diphenyl ethers. The Spearman rank correlation coefficient for levels of PFOA and PFOS was 0.61. Statistical analyses consisted of conditional logistic regression of continuous exposures (log_2 -transformed) and of categories (25th, 50th, 75th, and 90th percentiles). For each PFAS detected in > 60% of the controls, including PFOA and PFOS, analyses were conducted with no covariates; with adjustment for any PFAS, PCBs, or organochlorine pesticides correlated (Spearman correlation, 0.3–0.61); and with adjustment for smoking.

[The Working Group noted several strengths of this study, including collection of serum before thyroid cancer diagnosis; adjustment in the analysis for other PFAS, PCBs, and organochlorine pesticides correlated ($\rho = 0.3-0.61$) with the analyte of interest; follow-up of the cohort covering the peak years of thyroid cancer incidence; and the availability of data in the Medical Birth Register on several potential confounders. The Working Group noted that one would expect only minor misclassification of longterm exposure because of reliance on a single prediagnostic sample according to a simulation study (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: <u>https://publications.iarc.who.int/636</u>). PFAS concentrations might have been lower than pre-pregnancy levels because of increased plasma volume (Chapman et al., 1998) and glomerular filtration rates (Shankar et al., 2011) in the first trimester; however, a study of PFAS and birth outcomes suggests that little confounding may have occurred (Sagiv et al., 2018). The controls were not matched on the exact year of delivery but on increments of 4-5 years, which might affect comparison of PFAS levels because of temporal trends, although, given the estimated half-lives for PFOA and PFOS, such an effect was thought to be minimal. The Working Group did not

consider that the study had important surveillance bias among women diagnosed under age 40 years, when women may have frequent reproductive health-related visits; given that neither the women nor their medical providers were aware of their PFAS serum levels, these levels are not expected to affect thyroid cancer surveillance, and there were no large differences in stage at diagnosis by age at diagnosis. Analyses were not adjusted for pre-pregnancy BMI (a risk factor for thyroid cancer), which was missing for 85% of the women, nor were there data available on medical or environmental exposure to radiation. Finally, the study population had low-level exposure with a small exposure contrast.]

2.1.19 The Alpha-Tocopherol, Beta-Carotene Cancer Prevention Study

The Alpha-Tocopherol, Beta-Carotene (ATBC) Cancer Prevention Study was a randomized chemoprevention trial the primary aim of which was to evaluate the effects of supplementation with alpha-tocopherol and beta-carotene on lung cancer incidence (Alpha-Tocopherol, Beta Carotene Cancer Prevention Study Group, 1994). Secondary aims included evaluating the effects on other major cancers and overall and cause-specific mortality. The participants in the ATBC trial were White male smokers, aged 50-69 years at recruitment, who were identified in the Central Population Register as residing in south-western Finland, who responded to a questionnaire on their smoking history and willingness to participate, and who attended two clinic visits at which they completed a baseline study questionnaire and had trained nurses measure height, weight, blood pressure, heart rate, and visual acuity. The questionnaires included medical, smoking, and occupational history, and dietary habits over the past 12 months. Excluded from the study were people with a previous diagnosis of cancer other than non-melanoma skin cancer or carcinoma in situ; chronic renal insufficiency; cirrhosis of the liver; chronic alcoholism; receiving anticoagulant therapy; other medical conditions that might limit participation for 6 years; and current use of the vitamin supplements under investigation in the trial (Alpha-Tocopherol, Beta Carotene Cancer Prevention Study Group, 1994). An overnight fasting blood sample was collected at the initial clinic visit, with serum stored at –70° C. Recruitment began in 1985 and continued until 1988 when a total of 29 246 men were randomized to one of four treatment groups in a 2×2 factorial design. After late exclusions of 113 men found not to be eligible, the final study population numbered 29 133. Follow-up consisted of three annual clinic visits, with cancer cases ascertained through the Finnish Cancer Registry and deaths through the Central Population Register. The intervention continued until 30 April 1993, with publication of the trial results in 1994 (Alpha-Tocopherol, Beta Carotene Cancer Prevention Study Group, 1994). After the cessation of the trial, researchers continued to follow the cohort for 20 years, regularly updating data on mortality and cancer incidence.

Zhang et al. (2023) reported on two nested case-control studies on the incidence of pancreatic ductal adenocarcinoma, the most common type of pancreatic cancer. One study was conducted within the ATBC cohort, together with a parallel study conducted within the PLCO cohort (see description of this study above). Within the ATBC study, 251 cases of pancreatic ductal adenocarcinoma were ascertained until December 2011. A total of 251 controls were incidence-density sampled and matched to the cases on age at blood draw (\pm 5 years) and date of blood draw (within 30 days). Relative serum levels of PFOA and PFOS were measured using untargeted ultra-performance liquid chromatography-tandem mass spectrometry and/or gas chromatography-mass spectrometry. PFOS measurements were available only for 130 cases and controls. Statistical analyses consisted of conditional logistic regression to calculate odds ratios and 95% confidence intervals per standard deviation increase of \log_{10} -transfomed PFOA or PFOS levels and quintiles based on the distribution of the controls, with adjustment for age at blood draw, smoking (years smoked and cigarettes per day), diabetes, and BMI.

The Working Group noted that the strengths of the ATBC study included prediagnostic blood samples, detailed information on potential confounders collected through questionnaires and, for height and weight, by trained staff, and excellent case ascertainment. In addition, the numbers of cases in the ATBC (n = 251) and PLCO (n = 360) studies reported by <u>Zhang et al.</u> (2023) were large. The Working Group noted that the use of a single blood sample collected at baseline would be expected to result in only minor misclassification of long-term exposure, according to a simulation study (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications.iarc.who.int/636). The Working Group noted that the study limitations included the low-level exposure with a small exposure contrast. The restriction of participants to White male smokers may affect the generalizability of the results. Furthermore, the study relied upon relative quantification of PFOA and PFOS; however, in previous research, relative measures have correlated well with targeted absolute concentration measurements (Rhee <u>et al., 2023c</u>).]

2.1.20 New York Mount Sinai Hospital BioMe biobank cohort

Bio*Me* is a biobank linked to medical records within the Institute for Personalized Medicine at the Icahn School of Medicine at Mount Sinai, New York, USA. The collection of plasma samples, medical records, and questionnaire data from patients at Mount Sinai who lived in New York City and the larger metropolitan area started in 2007 and is currently ongoing (Icahn School of Medicine at Mount Sinai, 2023). Participants are enrolled from ambulatory care practices across the Mount Sinai Health System in New York City (Bar-Mashiah et al., 2022). No eligibility criteria were established, to make the cohort as inclusive as possible. As of September 2019, 52 500 patients were enrolled, and about 600 new patients are being enrolled each month (Icahn School of Medicine at Mount Sinai, 2023; van Gerwen et al., 2023). There did not appear to be follow-up of patients other than that conducted through the Mount Sinai hospital or ambulatory network.

van Gerwen et al. (2023) identified 88 cases of thyroid cancer within the BioMe biobank for whom the time between plasma collection and thyroid cancer diagnosis was ≥ 1 year (n = 31; defined as longitudinal cases) or < 1 year (n = 57; defined as cross-sectional cases). The authors did not specify how cases were identified. Of the 88 identified cases of thyroid cancer, 74 were papillary thyroid cancer, as confirmed in pathology reports. Further inclusion or exclusion criteria (e.g. previous cancer) were not specified. Controls were pair-matched to cases on sex, age, race or ethnicity, BMI, smoking status, and calendar year of blood sample collection for PFAS measurement. Eighteen individual PFAS (including PFOA and PFOS) were measured using untargeted methods with liquid chromatography-high resolution mass spectrometry. Analyses were conducted for all thyroid cancer cases, for only papillary thyroid cancer, and for overall cases, stratified by time of blood sample collection in relation to diagnosis (longitudinal cases, n = 31, or cross-sectional cases, n = 57). Median age at sample collection was 43.5 years for cases and controls. Most of the population were women (83%). The mean time between sample collection and cancer diagnosis was 1.5 years for all thyroid cancer cases combined, 4.0 years for the longitudinal cases, and 0.1 years for the cross-sectional cases.

[The Working Group noted that the strengths of the study included the availability

of histological data for the cases and analyses adjusted for age, sex, race, and BMI, and sample storage time, and, for some analyses, adjustment for other specific PFAS compounds. Also, plasma samples were collected ≥ 1 year before diagnosis for a subset, albeit small, of the cases. Limitations included the small sample size, particularly for cases for which plasma was collected > 1 year before diagnosis (longitudinal cases), with the remainder having plasma collected < 1 year before diagnosis or at diagnosis (cross-sectional cases). In addition, the study was based on the use of untargeted assay methods, which limits direct comparisons with other studies. Also, thyroid cancer might be detected among asymptomatic patients who sought medical care for unrelated reasons, which raises a concern for detection bias, given that the cases were recruited in ambulatory practice, especially with such short follow-up. However, the Working Group noted that, since the case and control participants were recruited from within the same network of Mount Sinai ambulatory care practices, a generally comparable medical screening pattern could be assumed among cases and controls; thus, detection bias was unlikely, also considering that patients and practitioners were unaware of PFAS measurements.]

2.1.21 Cancer Prevention Study II LifeLink cohort

The American Cancer Society (ACS) Cancer Prevention Study II (CPS-II) enrolled 1 185 106 participants from 50 US states and the District of Columbia who completed a questionnaire, and mortality was ascertained using the NDI. A subset of this cohort, the CPS-II Nutrition Cohort, started in 1992–1993 by including 184 194 participants aged 50–74 years from 21 states followed with biennial questionnaires for cancer incidence, further verified through medical records or cancer registry files. Between 1998 and 2001, the CPS-II LifeLink Cohort started by recruiting 39371 members from 20 states from within the CPS-II Nutrition Cohort. Those participants were required to be alive at the time of recruitment into the CPS-II LifeLink Cohort, since participation included a baseline questionnaire and a blood sample collection. Participants in the CPS-II LifeLink Cohort are followed for cancer incidence within the CPS-II Nutrition Cohort (Winquist et al., 2023).

Winquist et al. (2023) performed a study with a case-cohort design within the CPS-II LifeLink Cohort. Participants in the CPS-II LifeLink Cohort were eligible to participate in the casecohort study if they were men or postmenopausal women who were cancer-free (excluding non-melanoma skin cancer) at the time of blood collection. The median age at the time of enrolment in LifeLink was 71 years for men and 69 years for women. From these eligible participants, the case group was defined as individuals with first primary cancers of kidney (n = 158), bladder (n = 401), prostate (men only, n = 1610), female breast (n = 786), or pancreas (n = 172); or haematopoietic malignancies (n = 635) as of 30 June 2015. The median follow-up time for the members of the subcohort was 14.3 years. The comparison subcohort included 499 women and 500 men (representing 3% of the LifeLink cohort meeting the inclusion criteria). PFOA and PFOS were measured together with other PFAS using LC-MS/MS. Several covariates were available, and the analyses were adjusted for identified cancer risk factors associated with PFAS exposure. Notably, the main models were not adjusted for BMI, because BMI was considered to be on the causal pathway. Of the participants in the comparison cohort, 98% were non-Hispanic White and 79% were aged \geq 65 years at blood collection. Some participants identified as cases were included in the comparison subcohort (4 kidney cancers, 9 bladder cancers, 11 breast cancers, 58 prostate cancers, 7 pancreatic cancers, 16 haematological malignancies).

The Working Group noted as strengths the large number of cases and the collection of blood samples before diagnosis. Because of the design as a survivor cohort, and the long time period that had elapsed between enrolment in the CPS-II Nutrition Cohort and enrolment in the LifeLink cohort, it was likely that this study would not have included some persons who may have had cancer related to PFOA or PFOS, especially those who developed cancers earlier in life in a susceptible exposed population. This survivor bias would have biased the results downwards (i.e. towards the null or even towards inverse associations). Indeed, participants in this cohort were cancerfree survivors of the CPS-II Nutrition Cohort whose blood was collected in 1998-2001, when most of them were aged > 65 years. Although using a single sample to measure PFAS is a potential limitation, there is some evidence, from analyses of repeat samples of PFOA, that single samples may represent long-term averages over a 5-8-year period, with potential misclassification resulting in only minor bias to the null (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications.iarc.who.int/636). However, there remains the concern that single PFOA or PFOS samples may not represent average exposure over longer periods, which is particularly relevant here given the older age at blood draw. Thus, it is reasonable to assume that most of the CPS Nutrition Cohort would have been exposed to PFOA or PFOS well before the time of blood collection. The Working Group considered this study to be of minimal informativeness.]

2.1.22 Case–control studies in West Virginia and Ohio

<u>Vieira et al. (2013)</u> conducted two casecontrol studies of 18 different incident cancers during the years 1996–2005 among residents of 13 counties in Ohio and West Virginia, USA, which included both contaminated and non-contaminated water districts near the same polymer-production plant in Parkersburg, West Virginia, which was the source of contamination in the population studied by <u>Barry et al. (2013)</u>. This cohort was described in Section 2.1.5 above. The two case-control studies were included in the same publication (<u>Vieira et al., 2013</u>).

The source population in the study by Vieira et al. was cancer registries of Ohio and West Virginia, and the study included counties outside the contaminated water districts studied by the C8 Science Panel. The source population in Barry et al. was the population living in six contaminated water districts near the plant in Parkersburg, West Virginia, who had participated in the C8 Health Project baseline study of 69 000 residents in the water districts and had provided blood samples in which PFOA was measured.

The final data set consisted of 7869 cases from Ohio and 17 238 cases from West Virginia, from 13 counties, in whom cancer had been diagnosed at age \geq 15 years, with 18 cancer categories (i.e. bladder, brain, female breast, cervix, colon or rectum, kidney, leukaemia, liver, lung, melanoma of the skin, multiple myeloma, NHL, ovary, pancreas, prostate, testis, thyroid, and uterus).

In the first case-control study conducted by Vieira et al., cases (of the 18 cancers of interest) and controls (controls were all other cancers apart from the cancer of interest, and excluding kidney, pancreatic, testicular, and liver cancers) were compared with regard to residence in a contaminated or non-contaminated water district. This study included cases from both West Virginia and Ohio. Odds ratios were calculated for residence versus non-residence in contaminated water districts, adjusted for age, sex, diagnosis year, smoking status (current, past, and unknown, with never smoker as the referent) and insurance provider (government-insured Medicaid, uninsured, and unknown, with privately insured as the referent). Analyses were done for each of the six contaminated water

districts versus non-contaminated districts (the districts had different degrees of contamination, and serum levels for a large number of residents of each contaminated district were known), and for all six contaminated districts combined.

These same authors also conducted a separate case-control study among Ohio residents only. The Ohio registry provided more residential detail than did the West Virginia registry, enabling geocoding of exact addresses. Exposure in the case-control study in Ohio was based on estimated individual serum levels of PFOA at specific addresses at specific points in time. The individual serum PFOA levels were estimated using linked environmental and toxicokinetics models (Shin et al., 2011a, b). The environmental models estimated air and water concentrations of PFOA between 1951 and 2008, integrating emissions data from the facilities, fate, and transport characteristics of PFOA, and addresses of case and control participants, and then, using estimated water consumption and PFOA serum half-life data, annual serum levels for those drinking the contaminated water were estimated. The authors assumed that the serum levels estimated 10 years before case diagnosis (and analogously for matched controls) were the exposure of interest. Odds ratios were calculated, relative to the unexposed, for the low $(3.7-12.8 \ \mu g/L)$ [ng/mL]), medium (12.9–30.7 µg/L [ng/mL]), high (30.8–109 µg/L [ng/mL]), and very high $(110-655 \ \mu g/L \ [ng/mL])$ exposure categories. The second study used the same set of potential confounders as the first study (see above), but additionally considered race.

[The Working Group noted that there was probably some overlap between the cancer cases considered in the study by <u>Vieira et al. (2013)</u> and those in <u>Barry et al. (2013)</u>, although the extent of overlap was unknown. The Working Group noted that the strengths of this study were the good case ascertainment via cancer registries, the large number of incident cancers from cancer registries, and the reasonably large number of exposed cases of many specific cancers in people in the contaminated water districts (although small numbers were sometimes an issue for analyses of rarer cancers by category of exposure). The case-control study in Ohio benefited from being able to estimate serum levels for individuals on the basis of a model that was shown to provide a good prediction of the observed levels for 30 000 residents of the six contaminated water districts at one point in time (2005-2006) (Spearman correlation, 0.71; Winquist et al., 2013). The Working Group also noted limitations, including the assignment of an ecological exposure (by water district) in the first case-control study, as well as the use of estimated individual serum levels in the second case-control study (data from Ohio only) based on a model. In this second casecontrol study, a limitation was also the somewhat arbitrary assumption that the estimated serum levels 10 years before case diagnosis were the most relevant, as well as the assumption that the case and control participants had remained in the same residence for 10 years. Another limitation was the fairly small number of potential confounders available in the analyses. A potential limitation for both studies was the use of people with cancer as the controls, although the authors excluded those cancers thought to be potentially positively associated with PFOA. Use of cancer controls might bias estimates to the null, if any of the included cancers were in fact associated with PFOA. The use of cancer controls also might not reflect the general population with regard to potential confounders such as socioeconomic status and diet, but these potential differences in confounders were considered unlikely to have substantive effects in this population with very high exposure.]

2.2 Cancers of the urinary tract

See <u>Table 2.2</u>.

2.2.1 Kidney cancer

Three occupational cohort studies (Steenland and Woskie, 2012; Consonni et al., 2013; Raleigh et al., 2014), two population-based cohort studies (Barry et al., 2013; Li et al., 2022a), two population-based nested case-control studies (Shearer et al., 2021; Rhee et al., 2023b), one population-based case-cohort study (Winquist et al., 2023) and one population-based casecontrol study (Vieira et al., 2013) investigated the association between PFOA or PFOS exposure and mortality from and/or relative risk of kidney cancer. Some addressed PFOA exposures in settings where co-exposure to other PFAS compounds beyond background levels was unlikely, indicating that associations, if any, would primarily be due to PFOA (Steenland and Woskie, 2012; Barry et al., 2013; Vieira et al., 2013; Raleigh et al., 2014). Other studies addressing general populations with background exposure to multiple PFAS compounds provided PFASspecific estimates (Winquist et al., 2023) or estimated PFOA associations after controlling for co-exposure to other PFAS compounds (Shearer et al., 2021; Rhee et al., 2023b).

<u>Raleigh et al. (2014)</u> investigated mortality and cancer incidence in an occupational cohort including 4668 employees working for \geq 365 days from 1947 through 2002 at an APFO facility in Cottage Grove, Minnesota (in the Minneapolis metropolitan area), USA, and 4359 employees working for \geq 365 days before 1999 at a tape and abrasives production facility (reference group). Individual cumulative airborne exposure to APFO was estimated. The study updated earlier studies of the same cohort (<u>Gilliland and Mandel</u>, <u>1993</u>; <u>Lundin et al.</u>, 2009) (see Section 2.1.1). There was no indication of increased risk of kidney cancer on the basis of either mortality data (24 deaths across the exposed and reference populations) or incidence data (35 cases).

[The Working Group noted that study strengths were complete ascertainment of the cohort, very limited loss to cancer follow-up, and quantitative cumulative exposure assessment with a large exposure contrast. Co-exposure to TFE (IARC Group 2A; with *inadequate* evidence in humans, but *sufficient* evidence in experimental animals, with evidence that it is a potent carcinogen in rats and mice, <u>IARC</u>, <u>2016</u>) was addressed explicitly and found to be minimal. The small number of cases created uncertain risk estimates. Non-differential misclassification of exposure may have caused bias towards the null, and risk estimates with reference to unexposed workers should be interpreted with caution.]

Steenland and Woskie (2012) studied cause-specific mortality among 5791 fluoropolymer-production workers (men, 81%) in a polymer-production plant in Parkersburg, West Virginia, USA. The study was an extension by an additional 6 years of the cohort study by Leonard et al. (2008) and with a comprehensive quantitative exposure assessment. The cohort was described in detail earlier (Section 2.1.3). The mean and median estimated PFOA serum concentrations in workers from the Parkersburg plant were 350 ng/mL and 403 ng/mL, respectively, compared with a median of 4 ng/mL in the population of the USA. Mortality rates for exposed workers were compared with those for other workers from the same company in the region and the USA population.

The SMR (with other workers from the same company as the referent) for kidney cancer in the highest quartile of estimated cumulative serum PFOA concentration was 2.66 (95% CI, 1.15–5.24; 8 cases) with no lag, 2.82 (95% CI, 1.13–5.81; 7 cases) after a lag of 10 years, and 3.67 (95% CI, 1.48–7.57; 7 cases) after a lag of 20 years. Exposure–response analyses indicated a positive trend for kidney cancer in analyses with no lag and less consistently with a 10-year lag or a

Population size, Comments Reference, Cancer type Exposed Risk estimate Covariates Exposure category location, description, exposure (histopathology), or level cases or (95% CI) controlled enrolment/ assessment method incidence or deaths follow-up period, mortality study design Exposed to APFO (SMR, MN referent): Raleigh et al. 9027 (4668 exposed Kidney, mortality Age, sex, calendar *Exposure* assessment (2014) workers, 4359 reference period critique: Unexposed (Saint 18 1.23(0.73-1.95)MN, USA workers); Cottage See Table 2.1. Paul plant) Enrolment, Grove (MN) PFOA *Other strengths*: Exposed (Cottage 6 0.53(0.20-1.16)1947-2002/followcohort; workers See Table 2.1. Grove plant) up, 1947–2008 employed for $\geq 1 \text{ yr}$ Reference population Estimated cumulative airborne APFO exposure Kidney, mortality Age, sex, calendar sharing similar during 1947-2002 at an (mortality), 1988quartile (SMR, MN referent): period 2008 (incidence) APFO facility (Cottage socioeconomic 1st quartile 1 0.32(0.01-1.77)Cohort Grove; n = 4668); characteristics. $(< 2.6 \times 10^{-5} \, \mu g/$ reference workers Other limitations: See m³-years) without any exposure Table 2.1. Lacking 2nd quartile 2 0.74(0.09 - 2.69)to APFO, employed at data on workers $(2.6 \times 10^{-5} \text{ to})$ a tape and abrasives who left Minnesota $< 1.4 \times 10^{-4} \, \mu g/$ production facility or Wisconsin: m³-years) located in the same small numbers; no suburban geographical 3rd quartile 2 0.66(0.08 - 2.38)accounting for health area and managed by $(1.4 \times 10^{-4} \text{ to})$ behaviours. $< 7.3 \times 10^{-4} \, \mu g/$ the same company (Saint Paul; n = 4359) m³-years) Exposure assessment 4th quartile 1 0.42(0.01-2.34)method: see Table 2.1 $(\geq 7.3 \times 10^{-4} \, \mu g/$ m³-years)

Table 2.2 Epidemiological studies on exposure to PFOA and PFOS and cancers of the urinary tract

Table 2.2 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
<u>Raleigh et al.</u> (2014)		Kidney, incidence	Estimated cumulativ quartile (HR):	ve airborne	APFO exposure	Age, sex*, year of birth	Other comments: *The Working Group			
MN, USA Enrolment, 1947–2002/follow- up, 1947–2008 (mortality), 1988– 2008 (incidence) Cohort (cont.)			Unexposed (Saint Paul plant)	19	1		assumed that the models were also			
			lst quartile (< 2.9 × 10 ⁻⁵ μg/ m³-years)	4	1.07 (0.36–3.16)		adjusted for sex, as reported in the methods of <u>Raleigh</u>			
			2nd quartile (2.9 × 10 ⁻⁵ to < 1.5 × 10 ⁻⁴ μg/ m ³ -years)	4	1.07 (0.36–3.17)		<u>ct ai. (2014)</u> .			
			3rd quartile (1.5×10^{-4} to < $7.9 \times 10^{-4} \mu g/m^{3}$ -years)	4	0.98 (0.33–2.92)					
			4th quartile ($\geq 7.9 \times 10^{-4} \mu g/m^3$ -years)	4	0.73 (0.21–2.48)					
		Urinary bladder,	Exposed to APFO (SMR, MN referent):			Age, sex, calendar				
		mortality	Unexposed (Saint Paul plant)	8	0.62 (0.27–1.22)	period				
			Exposed (Cottage Grove plant)	8	0.89 (0.38–1.76)					
		Urinary bladder, mortality	Estimated cumulativ quartile (SMR, MN	ve airborne referent):	APFO exposure	Age, sex, calendar period				
			lst quartile (< 2.6 × 10⁻⁵ μg/ m³-years)	1	0.40 (0.01–2.25)					
			2nd quartile (2.6 × 10 ⁻⁵ to < 1.4 × 10 ⁻⁴ μg/ m ³ -years)	2	0.93 (0.11–3.38)					

Table 2.2 (continued)

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Raleigh et al. (2014) MN, USA Enrolment,		Urinary bladder, mortality (cont.)	3rd quartile (1.4×10^{-4} to < 7.3×10^{-4} µg/ m ³ -years)	4	1.61 (0.44-4.13)	Age, sex, calendar period	
1947–2002/follow- up, 1947–2008 (mortality), 1988– 2008 (incidence) Cohort		Urinary bladder, incidence	4th quartile ($\geq 7.3 \times 10^{-4} \mu g/m^3$ -years)	1	0.53 (0.01–2.97)		
			Estimated cumulativ quartile (HR):	ve airborne	APFO exposure	Age, sex*, year of birth	
(cont.)			Unexposed (Saint Paul Plant)	43	1		
			1st quartile (< 2.9 × 10 ⁻⁵ μg/m³)	7	0.81 (0.36–1.81)		
			2nd quartile (2.9 × 10 ⁻⁵ to < $1.5 \times 10^{-4} \mu g/m^3$)	6	0.78 (0.33-1.85)		
			3rd quartile $(1.5 \times 10^{-4} \text{ to} < 7.9 \times 10^{-4} \mu\text{g/m}^3)$	15	1.50 (0.80–2.81)		
			4th quartile ($\geq 7.9 \times 10^{-4} \mu\text{g/m}^3$)	12	1.66 (0.86–3.18)		

Table 2.2 (con	ntinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Alexander and	1588; Decatur	Urinary bladder,	PFOS exposure cate	gory (SIR, U	JS referent):	Age, sex, calendar	Exposure assessment
<u>Olsen (2007)</u>	(AL) PFOS cohort;	incidence	Never exposed	2	0.61 (0.07-2.19)	year	critique:
Decatur (AL), USA	production workers in the <u>Alexander</u>		Ever exposed (low or high)	9	1.70 (0.77–3.22)		See <u>Table 2.1</u> . Other strengths: See
Enrolment,ef al. (2003) cohort;1961–1997/follow- up, 1970–2002a questionnaire was administered to living cohort members(mortality and incidence)cohort members (response rate, 73.9%) to identify incident		Ever high	6	1.74 (0.64–3.79)		<u>Table 2.1</u> . Use of incidence data with	
		Ever low	7	2.26 (0.91-4.67)		74% participation rate in survey; use of	
		High for ≥ 1 yr	3	1.12 (0.23-3.27)			
	(response rate, 73.9%)		High or low for	6	1.31 (0.48-2.85)		cumulative exposure
	to identify incident		$\geq 1 \text{ yr}$				with internal
	cases of bladder cancer; bladder	Urinary bladder, incidence	Cumulative PFOS ex in high PFOS-expos	kposure (yea ed jobs; SIR	ars of employment R, US referent):	Age, sex, calendar year	comparisons, good exposure contrast;
	cancer decedents		0 to < 1	2	1.07 (0.12-3.85)		attempt to validate
	were identified using		1 to < 5	4	0.95 (0.25-2.43)		self-reported
	death from death		5 to < 10	3	2.72 (0.55-73.95)		cancer for survey
	certificates; analyses		≥ 10	2	1.43 (0.16-5.15)		Other limitations:
excluded 4 cohort me	excluded 495 living cohort members who	Urinary bladder, incidence	Cumulative PFOS exposure (yes in high PFOS-exposed jobs; RR		ars of employment):	Age, sex	See <u>Table 2.1</u> . Occupational cohor
	did not return the		0 to < 1	2	1		with only 11 cases of
qu	questionnaire		1 to < 5	4	0.83 (0.15-4.65)		bladder cancer, 2 in
	Exposure assessment		5 to < 10	3	1.92 (0.30-12.06)		the highest category
	method: see <u>Table 2.1</u>		≥ 10	2	1.52 (0.21–10.99)		of exposure; bladder cancer incidence identified by survey of cohort (6 cases)

PFOA and PFOS

and death certificates (5 deaths); no cancer

registry matching; no ability to validate 5 bladder cancers

identified by death certificate, mostly male (82%); only partial data on smoking.

Table 2.2 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
Steenland and	5791 workers;	Kidney, mortality	PFOA-exposed work	ers (SMR):		Age, sex, calendar	Exposure assessment			
Woskie (2012) Parkersburg (WV), USA Enrolment,	Parkersburg (WV, USA), polymer- production PFOA occupational cohort;		Other workers referent (same region and company)	12	1.28 (0.66–2.24)	period	<i>critique</i> : See <u>Table 2.1</u> . <i>Other strengths</i> : See <u>Table 2.1</u> . Ability to			
1948-2002/follow-	workers (men,		US referent	12	1.09 (0.56-1.9)		evaluate associations			
up, 1952–2008 81% (mortality) man Cohort faci pote fluo suff wor	81%) at a polymer- manufacturing	Kidney, mortality	Cumulative serum ex workers referent, san	xposure, no ne region ar	lag (SMR, other nd company):		with PFOA in a population exposed to levels much higher than in the general population. <i>Other limitations:</i> See <u>Table 2.1</u> , Restriction			
	potential exposure to fluoropolymers with sufficiently detailed work histories		1st quartile (0 to < 904 ppm-years)	1	1.07 (0.02–3.62)					
			2nd quartile (904 to < 1520 ppm-years)	3	1.37 (0.28–3.99)					
	Exposure assessment method: see <u>Table 2.1</u>		3rd quartile (1520 to < 2700 ppm-years)	0	0.00 (0.00-1.42)		to mortality rates and small numbers			
			4th quartile (≥ 2700 ppm-years)	8	2.66 (1.15–5.24)		of kidney cancer; reverse causation			
			Trend-test P-value, 0	.02			due to reduced			
		Kidney, mortality	Cumulative serum en workers referent, san	xposure, 10 ne region ar	-yr lag (SMR, other nd company):		is an unresolved			
			1st quartile (0 to < 798 ppm-years)	2	1.05 (0.13–3.79)		no excess of kidney			
			2nd quartile (798 to < 1379 ppm-years)	2	0.87 (0.11-3.15)		in <u>Steenland et al.</u> (2015), who studied			
			3rd quartile (1379 to < 2384 ppm-years)	1	0.44 (0.01–2.44)		a subset of these workers ($n = 3717$).			
			4th quartile (≥ 2384 ppm-years)	7	2.82 (1.13-5.81)					
			Trend-test P-value, 0	.02						

Table 2.2 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
<u>Steenland and</u> <u>Woskie (2012)</u>		Kidney, mortality	Cumulative serum en other workers referen	xposure, 20 nt, same reg	-yr lag (SMR, ion and company):	Age, sex, calendar period				
Parkersburg (WV), USA Enrolment, 1948–2002/follow- up, 1952–2008 (mortality) Cohort (cont.)			1st quartile 1 (0 to < 515 ppm-years)	3	1.34 (0.28–3.91)					
			2nd quartile (515 to < 1057 ppm-years)	1	0.46 (0.01–2.55)					
			3rd quartile (1057 to < 1819 ppm-years)	0	0.00 (0.00-2.03)					
			4th quartile (≥ 1819 ppm-years)	7	3.67 (1.48-7.57)					
			Trend-test <i>P</i> -value, 0	0.003						
		Urinary bladder,	PFOA-exposed work	ters (SMR):						
		mortality	Other workers referent (same region and company)	10	1.08 (0.52–1.99)					
			US referent	10	0.95 (0.46-1.75)					
		Urinary bladder, mortality	Cumulative serum en workers referent, san	xposure, no ne region ai	lag (SMR, other nd company):					
			1st quartile (0 to < 904 ppm-years)	2	1.24 (0.15-4.47)					
			2nd quartile (904 to < 1520 ppm-years)	6	2.49 (0.97-5.78)					
			3rd quartile (1520 to < 2700 ppm-years)	1	0.39 (0.01–2.17)					
			4th quartile (≥ 2700 ppm-years)	1	0.36 (0.10-2.01)					

Table 2.2 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
Steenland et al.	3713 workers; a subset	Urinary bladder,	Cumulative PFOA ex	xposure, 10	-yr lag (RR):	Age, sex, race,	Exposure assessment			
(2015) Parkersburg	of Parkersburg (WV, USA), polymer-	incidence	1st quartile (< 0.8 μg/mL-years)	NR	1	education, BMI, time-varying	<i>critique</i> : See <u>Table 2.1</u> .			
(WV), USA pro Enrolment, 1948– coh 2002/follow-up, and 1951 to interview pol	production PFOA cohort in <u>Steenland</u>		2nd quartile (0.8 to < 3.44 μg/mL-years)	NR	0.55 (0.12–2.61)	smoking, time- varying alcohol <u>Table 2</u> consumption, evaluat year of birth with PI popula	Other strengths: See <u>Table 2.1</u> . Ability to			
	and Woskie (2012); polymer-production		3rd quartile (3.44 to < 7.04 μg/mL-years)	NR	0.47 (0.10-2.21)		with PFOA in a			
(incidence) Cohort	who responded (self or next-of-kin) to a		4th quartile (≥ 7.04 μg/mL-years)	NR	0.31 (0.06–1.54)		to levels much higher than in the general			
	questionnaire about		Trend-test <i>P</i> -value, 0	0.03			population.			
	health outcomes and						Other limitations:			
	estimated occupational						bladder cancers			
	and residential						(n = 29).			
	exposure estimates;									
	29 incident cases of									
	bladder cancer									
	method: see Table 2.1									

Table 2.2 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Eriksen et al.	Case-cohort within	Urinary bladder,	Baseline plasma PFC	OA concenti	ation (IRR):	Age, sex, smoking	Exposure assessment		
Denmark Enrolment, 1 December 1993 to 31 May 1997/follow-up, 1 December 1993 to 1 July 2006 Case-cohort	Health cohort Cases: 332 cases of cancer of the urinary bladder Comparison cohort: 772 (680 men, 92 women); subcohort of participants randomly selected without cancer at the end of follow-up Exposure assessment	Urinary bladder, incidence	2nd quartile 2nd quartile 3rd quartile 4th quartile Continuous (per 1 ng/mL increase) Baseline plasma PFC 1st quartile 2nd quartile 3rd quartile	84 82 83 332 OS concentr 83 84 83	1 0.71 (0.46–1.07) 0.92 (0.61–1.39) 0.81 (0.53–1.24) 1.00 (0.95–1.05) ation (IRR): 1 0.76 (0.50–1.16) 0.93 (0.61–1.41) 0.572 (0.57)	intensity, smoking duration, years of school attendance, occupation associated with bladder cancer risk (rubber industry; textile industry; metal processing; glass industry;	See <u>Table 2.1</u> . <i>Other strengths</i> : See <u>Table 2.1</u> . Large cohort with numerous incident cancers ($n = 1240$) followed 0–12 yr after baseline enrolment; control of confounders; use of internal comparison.		
	method: see <u>Table 2.1</u>		4th quartile Continuous (per 10 ng/mL increase)	82 332	0.70 (0.46–1.07) 0.93 (0.83–1.03)	truck, bus, taxi drivers; painter, hairdresser; waiter; cook)	Other limitations: See <u>Table 2.1</u> . Low exposure contrast in a population with background exposure levels.		

PFOA and PFOS

Table 2.2 (con	tinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Barry et al. (2013)32 254 (28 541KidneMid-Ohio Valleycommunity memberscommunity members(OH and WV),and 3713 workers);USAC8 Science PanelEnrolment,Study; included peopleAugust 2005 toenrolled in the C8August 2006/Health Project whofollow-up, 1952-lived, worked, or2011 (incidence)attended school forCohort ≥ 1 yr between 1950and 3 December 2004in a contaminated-water district in thevicinity of a chemicalplant using PFOAin manufacturingprocesses (Parkersburg,WV; polymer-production facility),as well as a subset ofthose from the originalParkersburg (WV),polymer-productionUrinapFOA occupationalincidecohort who worked atthe plant between 1948and 2002Exposure assessmentmethod: see Table 2.1Urina	32 254 (28 541 community members and 3713 workers); C8 Science Panel Study; included people enrolled in the C8	Kidney, incidence	Estimated cumulativ (ng/mL), no lag (HR 1st quartile 2nd quartile 3rd quartile 4th quartile	ve PFOA ser): NR NR NR NR	1 1.23 (0.70–2.17) 1.48 (0.84–2.60)	Age, time-varying smoking, time- varying alcohol consumption, sex, education, birth year	Exposure assessment critique: See <u>Table 2.1</u> . Other strengths: See <u>Table 2.1</u> . Large cohort and strong
		4th quartile Continuous (per unit on natural log scale) Trend-test <i>P</i> -value, 0	NR 105 0.18	1.58 (0.88–2.84) 1.10 (0.98–1.24)	(5-yr calendar intervals)	exposure contrast, lagged analyses, adjustment for several covariates. Other limitations:	
	in a contaminated- water district in the	Kidney, incidence Urinary bladder, incidence	Estimated cumulativ (ng/mL), 10-yr lag (H	See <u>Table 2.1</u> . Self- reported cancer data.			
	vicinity of a chemical plant using PFOA in manufacturing processes (Parkersburg, WV; polymer- production facility), as well as a subset of those from the original Parkersburg (WV), polymer-production		1st quartile 2nd quartile 3rd quartile 4th quartile Continuous (per unit on natural log scale) Trend-test <i>P</i> -value, C	NR NR NR 105	1 0.99 (0.53–1.85) 1.69 (0.93–3.07) 1.43 (0.76–2.69) 1.09 (0.97–1.21)		Co-exposure to other PFAS in residents not evaluated.
	PFOA occupational cohort who worked at the plant between 1948 and 2002 Exposure assessment method: see <u>Table 2.1</u>		Estimated cumulativ (ng/mL), no lag (HR Continuous (per unit on natural log scale)): 105	1.00 (0.89–1.12)		
		Urinary bladder, incidence	Estimated cumulativ (ng/mL), 10-yr lag (H	ve PFOA ser HR):	rum concentration		
			Continuous (per unit on natural log scale)	105	0.98 (0.88–1.10)		

Table 2.2 (con	tinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Consonni et al. (2013)	5879 male workers	Kidney and other organs in the	Cumulative APFO e	xposure (SN	AR, national	Age, calendar	Exposure assessment
USA, UK, Italy, Germany, the	exposed); the pooled international TFE	urinary tract, mortality	Ever APFO- exposed	10	1.69 (0.81–3.11)	periou, country	See <u>Table 2.1</u> . Other strengths:
Netherlands	cohort includes male		< 16 unit-year	3	1.57 (0.32-4.59)		See <u>Table 2.1</u> . The
Enrolment,	workers who were ever		16–138 unit-year	3	1.50 (0.31-4.39)		cohort includes all
1950-2002/follow- up, 1950-2008employed o for 6 or 12 r or more of s	for 6 or 12 mo at one		139+ unit-year	4	2.00 (0.54-5.12)		worldwide during
	or more of six TFE- production sites in North America and Europe in 1950– 2002; the principal occupational exposures were TFE and APFO (facilitating production of PTFE) Exposure assessment method: see <u>Table 2.1</u>		Trend-test P-value, ().28			the entire period
Cohort		Urinary bladder,	SMR (national refere	ent):			of production and
Conort		mortality	Ever APFO- exposed	3	0.55 (0.11–1.60)		of production and benefits from almost complete enrolment and follow-up data. <i>Other limitations</i> : See <u>Table 2.1</u> . Low statistical power to detect less-common cancers; high exposure correlations between TFE monomer and PFOA which precluded evaluation of effects of the individual

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Shearer et al. (2021) USA Enrolment, 1993–2001; follow- up (from blood draw): median, 8.8 yr (incidence) Nested case– control	Nested within the PLCO cohort (see <u>Table 2.1</u>) Cases: 324; cancer source not reported Controls: 324; density- sampled on calendar time and individually matched on age categories, sex, race and ethnicity, study centre, and year of blood draw Exposure assessment method: see <u>Table 2.1</u>	Kidney (RCC), incidence	Serum PFOA (OR): $< 4.0 \ \mu g/L$ $\ge 4.0 \ to 5.5 \ \mu g/L$ $> 5.5 \ to 7.3 \ \mu g/L$ $> 7.3 \ to 27.2 \ \mu g/L$ Continuous (per unit on \log_2 scale) Trend-test <i>P</i> -value, O	47 83 69 125 324	1 1.47 (0.77–2.80) 1.24 (0.64–2.41) 2.63 (1.33–5.20) 1.71 (1.23–2.37)	Age, sex, race/ ethnicity, study centre, study year <i>Exposure asses</i> critique:of blood draw, BMI, smokingSee Table 2.1. See Table 2.1. F status, historySee Table 2.1. F specificity of th outcome. Adju for kidney fum filtration rate, thaw cycle, calendar year of of blood drawSee Table 2.1. F specificity of th outcome. Adju for kidney fum for blood draw, previous freeze- for blood draw, previous freeze- for blood draw, previous freeze- for kidney fum for kidney fum for kidney fum for kidney fum for kidney fum filtration rate, previous freeze- for kidney fum for kidney fum for kidney fum for kidney fum for kidney fum filtration rate, previous freeze- for kidney fum for kidney fum for kidney fum filtration rate, previous freeze- for kidney fum for kidney fum filtration rate, previous	<i>Exposure assessment</i> <i>critique:</i> See <u>Table 2.1</u> . <i>Other strengths:</i> See <u>Table 2.1</u> . High specificity of the outcome. Adjustment for kidney function to exclude reverse causation and for relevant potential confounders.
		Kidney (RCC), incidence	Serum PFOA (OR): $< 4.0 \ \mu g/L$ $\ge 4.0 \ to 5.5 \ \mu g/L$ $> 5.5 \ to 7.3 \ \mu g/L$ $> 7.3 \ to 27.2 \ \mu g/L$ Continuous (per unit on \log_2 scale) Trend-test <i>P</i> -value, O	47 83 69 125 324	1 1.41 (0.69–2.90) 1.12 (0.52–2.42) 2.19 (0.86–5.61) 1.68 (1.07–2.63)		Other limitations: See <u>Table 2.1</u> . Exposure assessment at a single time point likely attenuated risk estimates; no lagged analyses and no analyses of risk according to cumulative exposure; external validity is limited by a study population defined by phlebotomy and including mainly non-Hispanic Whites.

_

_

Table 2.2 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Shearer et al. (2021) USA Enrolment, 1993–2001; follow-		Kidney (RCC), incidence	Serum PFOA (OR): Time from blood draw, ≥ 8 yr: Continuous (per unit on \log_2 scale)	NR	1.66 (1.25–2.19)	Age, sex, race/ ethnicity, glomerular filtration rate, BMI, history of			
up (from blood draw): median,		Kidney (RCC), incidence	Serum PFOS (OR): $\leq 26.3 \text{ µg/L}$	60	1	hypertension, smoking status,			
8.8 yr (incidence)			> 26.3 to 38.4 µg/L	82	1.67 (0.84–3.30)	previous freeze-			
control			> 38.4 to 49.9 µg/L	61	0.92 (0.45–1.88)	calendar year of			
(cont.)			> 49.9 to 154.2 μg/L	121	2.51 (1.28-4.92)	blood draw, study year of blood draw, study centre			
			Continuous (per unit on \log_2 scale)	324	1.39 (1.04–1.86)				
			Trend-test <i>P</i> -value, (0.009					
		Kidney (RCC),	Serum PFOS (OR):			Age, sex, race/			
		incidence	\leq 26.3 µg/L	60	1	ethnicity, study			
			> 26.3 to 38.4 µg/L	82	1.24 (0.59–2.57)	of blood draw			
			> 38.4 to 49.9 µg/L	61	0.53 (0.22–1.24)	BMI, smoking			
			> 49.9 to	121	1.14 (0.45–2.88)	status, history			
			Continuous (per unit on \log_2 scale)	324	0.92 (0.60–1.42)	of hypertension, glomerular filtration rate,			
			Trend-test <i>P</i> -value, ().64		previous freeze- thaw cycle, calendar year of blood draw, PFOA serum concentration, PFHxS serum concentration			

PFOA and PFOS

Table 2.2 (con	tinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Steenland et al. (2022) USA Enrolment, 1993–2014 (PLCO); August 2005 to August 2006 (C8 Panel	Cases: PLCO, 324; C8 Panel study, 103; all cases of RCC; cases from the PLCO cohort were the same as those identified in <u>Shearer et al. (2021);</u> cases from the C8 study were identified	Kidney (RCC), incidence Kidney (RCC), incidence	Serum PFOA, 2-piec transformed) model Continuous (per ng/mL increase up to the knot (9.5 ng/mL)) Serum PFOA, 2-piec transformed) model	e linear spli (log odds): 427 e linear spli (best-fitting	ine (not 0.135 (0.071–0.198) ine (natural log- g) (log odds):	Age, sex, race/ ethnicity, study centre (PLCO), year of blood draw (PLCO), birth year (C8), BMI, hypertension	Exposure assessment method: See <u>Table</u> 2.1 for <u>Shearer et al.</u> (2021) and <u>Barry</u> <u>et al. (2013)</u> . Other strengths: See <u>Table 2.1</u> . Pooled analysis of large and informative studies
study); follow- up, 1993–2014 (PLCO), 1952– 2011 (C8 Panel study) Nested case– control	study were identified in the C8 panel cohort study (<u>Barry</u> <u>et al., 2013</u>) using the topographical code C64.9 and excluding urothelial carcinomas (e.g. morphology codes 8120, 8130), to capture mostly RCCs Controls: PLCO, 324; C8 panel study, 511; for the PLCO component, controls were the same as those identified in <u>Shearer et al. (2021</u>); for the C8 component, up to 5 controls per case were selected, matched on sex, race, year of birth (within 5 yr); controls were required to have survived past the age at which the case was diagnosed Exposure assessment method: see <u>Table 2.1</u> for <u>Shearer et al. (2021</u>)		Continuous (per unit increase up to the knot (In PFOA = 2.55))	427	0.656 (0.333-0.979)		informative studies on kidney cancer. <i>Other limitations</i> : See <u>Table 2.1</u> . Cumulative serum levels were not available in the PLCO study.

S

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Li et al. (2022a) Ronneby, southern Sweden Enrolment, up, 1985–2016 Cohort Cohort Cohort Enrolment, up, 1985–2016 Cohort Co	60 507; the Ronneby Register Cohort included all individuals	Kidney, incidence	Men, residential exp contaminated drink county excluding Ro	osure to hig ing-water (Sonneby refer	ghly PFAS- SIR, Blekinge rent):	Age, calendar year	<i>Exposure assessment</i> <i>critique:</i> See <u>Table 2.1</u> . <i>Other strengths</i> : See <u>Table 2.1</u> . Complete registration of a large cohort; no loss to follow-up; long follow-up period. <i>Other limitations</i> :
	who ever lived in		Never	46	0.67 (0.49-0.90)		
	Ronneby municipality		Ever	17	0.86 (0.50-1.38)		
	in 1985–2013; one third of the households received	Kidney, incidence	Women, residential contaminated drink county excluding Ro	exposure to ing-water (Sonneby refer	highly PFAS- SIR, Blekinge rent):	Age, calendar year	
	drinking-water from		Never	43	1.17 (0.84-1.57)		
	a waterworks situated		Ever	16	1.47 (0.84-2.39)		See Table 2.1. Mixed
	near a military airfield where PFAS-containing	Kidney, incidence	Residential exposure to highly PFAS- contaminated drinking-water (HR):			Calendar year, age, sex	PFAS exposure (mainly PFOS,
	firefighting foam was		Never	89	1		PFHxS and PFOA); no adjustment for known determinants
	used in 1985–2013		Ever	33	1.27 (0.85–1.91)		
	(15 811 individuals considered "ever-	Kidney, incidence	Time period of residential exposure to highly PFAS-contaminated drinking-water (HR):			Calendar year, age, sex	of kidney cancer such
	long-term exposure		Never	89	1		overweight: relativel
	$(\geq 11 \text{ yr})$ in the latest		Early (1985–2004)	19	1.05 (0.64–1.73)		few cases producing
	part of the follow-up		Late (2005-2013)	14	1.85 (1.00-3.40)		uncertain risk
	period (2005–2013) were considered to be	Kidney, incidence	Duration of residential exposure to highly PFAS- contaminated drinking-water (HR):			Calendar year, age, sex	estimates.
	more highly exposed		Never	89	1		
	Exposure assessment		Short (1–10 yr)	15	1.11 (0.64–1.92)		
	method: see <u>Table 2.1</u>		Long ($\geq 11 \text{ yr}$)	18	1.47 (0.87-2.49)		
		Urinary bladder, incidence	Men, residential exp contaminated drink county excluding Ro	oosure to hig ing-water (Sonneby refer	ghly PFAS- SIR, Blekinge rent):	Age, calendar year	
			Never	166	0.94 (0.80-1.09)		
			Ever	57	1.10 (0.84-1.43)		

Covariates

controlled

Comments

Table 2.2 (cor	Table 2.2 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)					

follow-up period, study design	mortality					
<u>Li et al. (2022a)</u> Ronneby, southern Sweden	Urinary bladder, incidence	Women, residential contaminated drink county excluding R	exposure king-water onneby ref	Age, calendar year		
Enrolment,		Never	35	0.69 (0.48-0.95)		
1985–2013/tollow-		Ever	17	1.13 (0.66–1.80)		
up, 1985–2016 (incidence) Cobort	Urinary bladder, incidence	Residential exposur contaminated drink	e to highly king-water	Calendar year, age, sex		
(cont.)		Never	200	1		
()		Ever	74	1.30 (0.99–1.69)		
	Urinary bladder, incidence	Time period of resid PFAS-contaminated	dential exp d drinking	osure to highly -water (HR):	Calendar year, age, sex	
		Never	200	1		
		Early (1985-2004)	46	1.20 (0.87-1.66)		
		Late (2005-2013)	28	1.50 (0.98-2.29)		
	Urinary bladder, incidence	Duration of residen contaminated drink	tial exposu king-water	re to highly PFAS- (HR):	Calendar year, age, sex	
		Never	200	1		
		Short (1–10 yr)	39	1.23 (0.87–1.73)		
		Long (≥ 11 yr)	35	1.39 (0.95-2.02)		

Table 2.2 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
<u>Rhee et al. (2023b)</u>	Nested within the MEC	Kidney (RCC),	PFOA serum concent	tration (OR	.):	Sex, race/	Exposure assessment			
CA and FI, USA Conort Enrolment, Cases: 428; all RCC 1993–1996/follow- up, through 2018 of 2018 in the MEC Nested case– study, with available control pre-diagnostic serum sample; incident cases identified through	cohort Cases: 428; all RCC	incidence	1st quartile $(< 3.27 \text{ µg/L})$	107	1	ethnicity, study centre, age at	<i>critique</i> : See <u>Table 2.1</u> .			
	cases identified as of 2018 in the MEC study, with available		(= 3.27 µg/L) 2nd quartile (> 3.27 to 4.47 µg/L)	99	1.26 (0.80–1.97)	serum collection, date of serum collection, time of	Other strengths: See <u>Table 2.1</u> . Large sample size;			
		3rd quartile (> 4.47 to 6.22 μg/L)	122	1.26 (0.78–2.05)	serum collection, fasting status, smoking status,	consideration of multiple PFAS adjustment;				
	linkage with the SEER HI registry and the CA		4th quartile (> 6.22 μg/L)	100	1.04 (0.60–1.81)	BMI, history of hypertension,	stratification by race/ ethnicity.			
HI registry and the CA state cancer registry Controls: 428; controls were MEC participants alive at the time of the matched case diagnosis and matched 1:1 to cases on sex race or		Continuous (per unit on log ₂ scale) Trend-test <i>P</i> -value, 0	428 .75	0.89 (0.67–1.18)	eGFR, PFOS (log ₂ - transformed), PFHxS (log ₂ - transformed), PFNA (log ₂ - transformed), FOSA detected	Other limitations: See <u>Table 2.1</u> . Some of the stratified analysis by race/ethnicity have limited statistical power.				
	ethnicity, study centre, age and date at serum	Kidney (RCC), incidence	White participants, I (OR):	PFOA serun	n concentration	Sex, study centre, age at serum				
	collection, time of serum collection, and		lst quartile (≤ 3.27 μg/L)	19	1	collection, date of serum collection,				
	Exposure assessment		2nd quartile (> 3.27 to 4.47 μg/L)	15	2.08 (0.62-6.98)	collection,				
	inemou. see <u>rable 2.1</u>		3rd quartile (> 4.47 to 6.22 μg/L)	24	3.63 (0.84–15.8)	smoking status, BML history of				
			4th quartile (> 6.22 μg/L)	22	2.94 (0.56–15.5)	BMI, history of hypertension, eGFR, PFOS (log ₂ - transformed), PFHxS (log ₂ - transformed), PFNA (log ₂ - transformed), FOSA detected				
			Continuous (per unit on \log_2 scale)	80	2.12 (0.87–5.18)					
			Trend-test <i>P</i> -value, 0	.48						

Table 2.2 (con	tinued)							
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
Rhee et al. (2023b) CA and HI, USA		Kidney (RCC), incidence	African-American p concentration (OR):	articipants	, PFOA serum	Sex, study centre, age at serum		
Enrolment, 1993–1996/follow-			lst quartile (≤ 3.27 μg/L)	24	1	collection, date of serum collection,		
up, through 2018 Nested case–			2nd quartile (> 3.27 to 4.47 μg/L)	15	1 (0.23-4.33)	time of serum collection,		
control (cont.)			3rd quartile (> 4.47 to 6.22 μg/L)	17	1.01 (0.24-4.23)	smoking status, smoking status, BMI, history of hypertension, eGFR, PFOS (log ₂ - transformed), PFHxS (log ₂ -		
		Kidney (RCC), incidence	4th quartile (> 6.22 μg/L)	16	1.08 (0.23-5.13)			
			Continuous (per unit on \log_2 scale)	72	1.01 (0.51–1.98)			
			Trend-test P-value, (end-test <i>P</i> -value, 0.91				
			Japanese-American concentration (OR):	participant	s, PFOA serum	PFNA (log ₂ - transformed),		
			lst quartile (≤ 3.27 μg/L)	14	1	FOSA detected		
			2nd quartile (> 3.27 to 4.47 μg/L)	25	2.62 (0.79-8.69)			
			3rd quartile (> 4.47 to 6.22 μg/L)	37	2.65 (0.77-9.15)			
			4th quartile (> 6.22 μg/L)	31	3.29 (0.84–12.88)			
			Continuous (per unit on \log_2 scale)	107	1.00 (0.47–2.13)			
			Trend-test P-value, ().22				

Table 2.2 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Rhee et al. (2023b) CA and HI, USA Enrolment, 1993–1996/follow- up, through 2018 Nested case– control (cont.)		Kidney (RCC), incidence	Native Hawaiian par concentration (OR): 1st quartile $(\leq 3.27 \ \mu g/L)$ 2nd quartile $(> 3.27 \ to 4.47 \ \mu g/L)$ 3rd quartile $(> 4.47 \ to 6.22 \ \mu g/L)$ 4th quartile $(> 6.22 \ \mu g/L)$ Continuous (per unit on \log_2 scale) Trend-test <i>P</i> -value, 0	ticipants, P 12 10 17 11 50 .04	FOA serum 1 0.3 (0.04–2.31) 0.28 (0.03–2.39) 0.08 (0.01–0.94) 0.57 (0.21–1.55)	Sex, study centre, age at serum collection, date of serum collection, time of serum collection, fasting status, smoking status, BMI, history of hypertension, eGFR, PFOS (log ₂ - transformed), PFHxS (log ₂ - transformed),			
		Kidney (RCC),	PFOA serum concentration (OR):			PFNA (log ₂ - transformed), FOSA detected Sex, race/			
		incidence	Calendar year blood drawn, before 2002: Continuous (per unit on log ₂ scale)	90	1.49 (0.77–2.87)	ethnicity, study centre, age at serum collection, date of serum collection, time of serum collection, fasting status, smoking status, BMI, history of hypertension, eGFR, PFOS (log ₂ - transformed), PFHxS (log ₂ - transformed), PFNA (log ₂ - transformed), FOSA detected			

Table 2.2 (con	tinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
<u>Rhee et al. (2023b)</u>		Kidney (RCC),	PFOA serum concen	tration (OR	k):	Sex, race/	
CA and HI, USA Enrolment, 1993–1996/follow- up, through 2018 Nested case–		incidence	Calendar year blood drawn, in 2002 or later: Continuous (per unit on log ₂ scale)	336	0.80 (0.56–1.13)	ethnicity, study centre, age at serum collection, date of serum collection, time of	
control		Kidney (RCC),	PFOS serum concen	tration (OR):	serum collection,	
(cont.)		incidence	1st quartile (< 16.65 μg/L)	118	1	smoking status,	
			2nd quartile (16.65 to < 25.05 μg/L)	105	1.05 (0.66–1.66)	hypertension, eGER_PEOS (log -	
			3rd quartile (25.05 to < 36.40 μg/L)	100	0.99 (0.58–1.68)	transformed), PFHxS (log ₂ -	
			4th quartile $(\geq 36.40 \ \mu g/L)$	105	0.93 (0.51-1.72)	transformed), PFNA (log ₂ -	
			Continuous (per unit on log, scale)	428	0.95 (0.74–1.23)	transformed), FOSA detected	
			Trend-test P-value, 0).72			
		Kidney (RCC),	PFOS serum concent	tration (OR):		
		incidence	Calendar year blood drawn, before 2002: Continuous (per unit on log ₂ scale)	90	0.77 (0.40–1.48)		
		Kidney (RCC),	PFOS serum concen	tration (OR):		
		incidence	Calendar year blood drawn, in 2002 or later: Continuous (per unit on log ₂ scale)	336	0.96 (0.73–1.28)		

-

Table 2.2 (cor	ntinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Winquist et al.	Case-cohort within the	Kidney, incidence	Serum PFOA concer	ntration (H)	R):	Sex, year of	Exposure assessment critique: See Table 2.1. Other strengths: See Table 2.1. Large number of cases. Other limitations: See Table 2.1. Survivor cohort with blood collected from persons mostly over aged 65, thus the study would not include persons who may have had PFOA- or PFOS- related cancer developed earlier in life, resulting in bias towards the null or
<u>(2023)</u> 20 states_USA	CPS-II Lifelink Cohort (see Table 2.1)		1st quartile	39	1	serum sample	
Enrolment, 1998– 2001/follow-up,	Cases: 3762 overall (kidney cancer, 158,		(< 3.900 lig/lifL) 2nd quartile (3.900 to < 5.200 ng/mL)	39	0.93 (0.56-1.56)	serum collection, race, education,	
through 30 Juneof which 109 were2015RCC, and urinary	of which 109 were RCC, and urinary bladder 401): incident		3rd quartile (5.200 to < 7.300 ng/mL)	39	0.83 (0.49–1.40)	smoking status, alcohol	
Case-conort	cases from the CPS- II Lifelink Cohort	77-1 1	4th quartile (≥ 7.300 ng/mL)	39	1.20 (0.71–2.04)	consumption	
(surviving CPS-II Nutrition cohort	(surviving CPS-II Nutrition cohort		Continuous (per unit on log ₂ scale)	156	1.08 (0.88–1.33)	Waama Gaamaa	
	participants) with a	Kidney, incidence	Women, serum PFO	A concentr	ation (HR):	Year of serum sample collection, age at serum collection, race, education, smoking status, alcohol consumption	
	kidney, urinary bladder detected through self- report or NDI linkage,		1st quartile (< 3.900 ng/mL)	17	1		
			2nd quartile (3.900 to < 5.200 ng/mL)	13	0.80 (0.34-1.87)		
	and verified through medical-record review or cancer registry		3rd quartile (5.200 to < 7.300 ng/mL)	17	1.04 (0.45–2.44)		
	Controls: 999; a sex-stratified simple		4th quartile (≥ 7.300 ng/mL)	18	1.94 (0.87–4.35)		even towards inverse associations.
	random sample of 499 women and 500 men		Continuous (per unit on log ₂ scale)	65	1.33 (0.97–1.83)		associations.
	(~3% of the eligible	Kidney, incidence	Men, serum PFOA c	oncentratio	on (HR):		
	cohort); stratification sampling was to ensure		1st quartile (< 3.900 ng/mL)	22	1		
	an adequate number of subcohort participants		2nd quartile (3.900 to < 5.200 ng/mL)	26	0.87 (0.43–1.75)		
	in sex-specific analyses (for breast and prostate cancers) Exposure assessment method: see <u>Table 2.1</u>		3rd quartile (5.200 to < 7.300 ng/mL)	22	0.65 (0.31–1.35)		
			4th quartile (≥ 7.300 ng/mL)	21	0.81 (0.39–1.68)		
			Continuous (per unit on log ₂ scale)	91	0.89 (0.66–1.20)		

Table 2.2 (continued)											
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments				
Winquist et al.		Kidney, incidence	Serum PFOS concen	tration (HF	R):	Sex, year of					
(2023) 20 states, USA Enrolment, 1998– 2001/follow-up, through 30 June 2015 Case-cohort (cont.)			1st quartile (< 13.000 ng/mL)	35	1	serum sample collection, age at					
			2nd quartile (13.000– < 18.000 ng/mL)	39	0.92 (0.54–1.57)	serum collection, race, education, smoking status, alcohol consumption					
			3rd quartile (18.000- < 26.000 ng/mL)	42	0.97 (0.58–1.63)						
			4th quartile (≥ 26.000 ng/mL)	40	1.14 (0.67–1.92)						
			Continuous (per unit on log, scale)	156	1.03 (0.84–1.26)						
		Kidney, incidence	Women, serum PFO	S concentra	Year of serum						
			1st quartile (< 13.000 ng/mL)	19	1	sample collection, age at serum					
			2nd quartile (13.000 to < 18.000 ng/mL)	10	0.37 (0.14–0.94)	collection, race, education, smoking					
			3rd quartile (18.000 to < 26.000 ng/mL)	17	0.76 (0.35–1.66)	status, alcohol consumption					
			4th quartile (≥ 26.000 ng/mL)	19	0.93 (0.40-2.15)						
			Continuous (per unit on log ₂ scale)	65	1.06 (0.70–1.59)						

Table 2.2 (con	Fable 2.2 (continued)											
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments					
Winquist et al.		Kidney, incidence	Men, serum PFOS co	oncentratio	n (HR):	Year of serum						
(2023) 20 states, USA Enrolment, 1998– 2001/follow-up, through 30 June 2015 Case–cohort (cont.)			1st quartile (< 13.000 ng/mL)	16	1	sample collection, age at serum						
		2nd quartile (13.000 to < 18.000 ng/mL)	29	1.72 (0.82–3.61)	collection, race, education, smoking							
			3rd quartile (18.000 to < 26.000 ng/mL)	25	1.39 (0.66–2.93)	consumption						
			4th quartile (≥ 26.000 ng/mL)	21	1.33 (0.62–2.85)							
			Continuous (per unit on log ₂ scale)	91	1.00 (0.79–1.28)							
		Kidney (RCC),	Serum PFOA concer	ntration (HI	R):	Sex, year of						
		incidence	1st quartile (< 3.900 ng/mL)	27	1	serum sample collection, age at serum collection, race, education, smoking status, alcohol consumption						
			2nd quartile (3.900 to < 5.000 ng/mL)	25	1.00 (0.54–1.87)							
			3rd quartile (5.000 to < 7.400 ng/mL)	28	0.74 (0.40-1.36)							
			4th quartile (≥ 7.400 ng/mL)	27	1.21 (0.65–2.27)							
			Continuous (per unit on log, scale)		1.06 (0.83–1.35)							

Table 2.2 (con	Table 2.2 (continued)											
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments					
Winquist et al.		Kidney (RCC),	Women, serum PFO	A concentra	ation (HR):	Year of serum						
(2023) 20 states, USA Enrolment, 1998–		incidence	1st quartile (< 3.900 ng/mL)	8	1	sample collection, age at serum						
Enrolment, 1998– 2001/follow-up,			2nd quartile (3.900 to < 5.000 ng/mL)	8	1.33 (0.42-4.19)	collection, race, education,						
through 30 June 2015 Case–cohort (cont.)			3rd quartile (5.000 to < 7.400 ng/mL)	13	1.66 (0.54–5.12)	status, alcohol consumption						
			4th quartile (≥ 7.400 ng/mL)	13	3.14 (1.04–9.54)							
			Continuous (per unit on log ₂ scale)	42	1.54 (1.05–2.26)							
		Kidney (RCC), incidence	Men, serum PFOA c	oncentratio								
			1st quartile (< 3.900 ng/mL)	19	1							
			2nd quartile (3.900– < 5.000 ng/mL)	17	0.79 (0.36–1.74)							
			3rd quartile (5.000– < 7.400 ng/mL)	15	0.45 (0.20-1.01)							
			4th quartile (≥ 7.400 ng/mL)	14	0.64 (0.28–1.46)							
			Continuous (per unit on log, scale)	65	0.80 (0.57–1.11)							

Table 2.2 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
Winquist et al.		Kidney (RCC),	Serum PFOS concen	tration (HR	k):	Sex, year of serum sample collection, age at serum collection, race, education, smoking status, alcohol consumption				
(2023) 20 states, USA		incidence	1st quartile (< 13.000 ng/mL)	25	1					
Enrolment, 1998– 2001/follow-up, through 30 June 2015 Case–cohort (cont.)			2nd quartile (13.000 to < 19.000 ng/mL)	29	0.82 (0.45–1.49)					
			3rd quartile (19.000 to < 26.000 ng/mL)	24	0.96 (0.51–1.80)					
			4th quartile (≥ 26.000 ng/mL)	29	1.13 (0.61–2.07)					
			Continuous (per unit on \log_2 scale)	107	1.08 (0.84–1.38)					
		Kidney (RCC),	Women, serum PFO	S concentra	tion (HR):	Year of serum				
		incidence	1st quartile (< 13.000 ng/mL)	11	1	sample collection, age at serum				
			2nd quartile (13.000 to < 19.000 ng/mL)	6	0.40 (0.12–1.35)	collection, race, education, smoking status, alcohol consumption				
			3rd quartile (19.000 to < 26.000 ng/mL)	10	0.89 (0.32–2.46)					
			4th quartile (≥ 26.000 ng/mL)	15	1.29 (0.45-3.74)					
			Continuous (per unit on log ₂ scale)	42	1.30 (0.77–2.20)					

Table 2.2 (continued)								
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
Winquist et al.		Kidney (RCC),	Men, serum PFOS c	oncentratio	n (HR):	Year of serum		
(2023) 20 states, USA Enrolment, 1998– 2001/follow-up, through 30 June 2015 Case–cohort (cont.)	incidence Urinary b incidence	incidence	1st quartile (< 13.000 ng/mL)	14	1	sample collection, age at serum collection, race, education, smoking status, alcohol consumption		
			2nd quartile (13.000 to < 19.000 ng/mL)	23	1.25 (0.57–2.74)			
			3rd quartile (19.000 to < 26.000 ng/mL)	14	1.10 (0.46-2.60)			
			4th quartile (≥ 26.000 ng/mL)	14	0.98 (0.42-2.29)			
			Continuous (per unit on log ₂ scale)	65	0.97 (0.73-1.29)			
		Urinary bladder, incidence	Serum PFOA concentration (HR):			Sex, year of		
			1st quartile (< 3.800 ng/mL)	95	1	serum sample collection, age at serum collection, race, education, smoking status, alcohol consumption		
			2nd quartile (3.800 to < 5.100 ng/mL)	97	0.84 (0.56–1.26)			
			3rd quartile (5.100 to < 6.700 ng/mL)	99	0.87 (0.58–1.30)			
			4th quartile (≥ 6.700 ng/mL)	105	0.86 (0.58–1.27)			
			Continuous (per unit on log ₂ scale)	396	0.93 (0.77–1.13)			

Table 2.2 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Winquist et al.		Urinary bladder,	Women, serum PFOA concentration (HR):			Year of serum			
(2023) 20 states, USA Enrolment, 1998– 2001/follow-up, through 30 June 2015 Case–cohort (cont.)		incidence	1st quartile (< 3.800 ng/mL)	25	1	sample collection, age at serum			
			2nd quartile (3.800 to < 5.100 ng/mL)	25	1.23 (0.60–2.52)	collection, race, education, smoking status, alcohol consumption			
			3rd quartile (5.100 to < 6.700 ng/mL)	12	0.68 (0.27–1.70)				
			4th quartile (≥ 6.700 ng/mL)	20	0.81 (0.37–1.78)				
			Continuous (per unit on \log_2 scale)	82	0.91 (0.63–1.31)				
		Urinary bladder,	Men, serum PFOA c	oncentratio	on (HR):				
		incidence	1st quartile (< 3.800 ng/mL)	70	1				
			2nd quartile (3.800 to < 5.100 ng/mL)	72	0.80 (0.49–1.32)				
			3rd quartile (5.100 to < 6.700 ng/mL)	87	0.92 (0.57–1.49)				
			4th quartile (≥ 6.700 ng/mL)	85	0.87 (0.54–1.40)				
			Continuous (per unit on log ₂ scale)	314	0.93 (0.74–1.17)				

Table 2.2 (continued)								
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
Winquist et al. (2023) 20 states, USA Enrolment, 1998– 2001/follow-up, through 30 June 2015 Case–cohort (cont.)		Urinary bladder, incidence	Serum PFOS concentration (HR):			Sex, year of		
			1st quartile (< 13.000 ng/mL)	95	1	serum sample collection, age at		
			2nd quartile (13.000 to < 18.000 ng/mL)	92	0.81 (0.54–1.21)	serum collection, race, education, smoking status, alcohol consumption		
			3rd quartile (18.000 to < 25.000 ng/mL)	106	1.07 (0.72–1.60)			
			4th quartile (≥ 25.000 ng/mL)	103	0.96 (0.64–1.44)			
			Continuous (per unit on log ₂ scale)	396	1.01 (0.86–1.20)			
		Urinary bladder, incidence	Women, serum PFOS concentration (HR):			Year of serum		
			1st quartile (< 13.000 ng/mL)	27	1	sample collection, age at serum		
			2nd quartile (13.000– < 18.000 ng/mL)	17	0.51 (0.24–1.05)	collection, race, education, smoking status, alcohol consumption		
			3rd quartile (18.000– < 25.000 ng/mL)	20	0.65 (0.33-1.30)			
			4th quartile (≥ 25.000 ng/mL)	18	0.63 (0.29–1.35)			
			Continuous (per unit on log ₂ scale)	82	0.82 (0.58–1.16)			

Table 2.2 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Winquist et al.		Urinary bladder,	Men, serum PFOS co	oncentration	n (HR):	Year of serum			
(2023) 20 states, USA Enrolment, 1998– 2001/follow-up, through 30 June 2015 Case–cohort (cont.)	incidence	incidence	1st quartile (< 13.000 ng/mL)	68	1	sample collection, age at serum			
			2nd quartile (13.000– < 18.000 ng/mL)	75	0.92 (0.57–1.49)	collection, race, education, smoking status, alcohol consumption			
			3rd quartile (18.00– < 25.000 ng/mL)	86	1.20 (0.75–1.94)				
			4th quartile (≥ 25.000 ng/mL)	85	1.10 (0.68–1.78)				
			Continuous (per unit on log ₂ scale)	314	1.06 (0.78–1.28)				

Reference, location, enrolment/ follow-up period, study designPopulation size, description, exposure assessment methodCancer type (histopathology), incidence or mortalityExposure category or levelExposed cases or deathsRisk estimate (95% CI)Covariates controlledCommentsVieira et al. (2013) OH and WV, USA Clincidence)Cases: study 1: kidney, cancer, 751, and bladder cancer, 751, and bladder cancer, 1350; study 2: (incidence)Kidney, incidenceStudy 1. Residence in a PFOA-contaminated water district (OH and WV) (OR): UnexposedAge, sex, diagnosis year, insurance provider, smoking statusExposure assessment critique: See Table 2.1. Other strengths: See Table 2.1. Well- ascertained cases based on cancer registries. Other comments: See Table 2.1. Substantial overlap of the study population addres table 2.1. Substantial overlap of the study population addres table 2.1. Substantial overlap of the study population address by a C8 Science Panel Project by Barry et al.	Table 2.2 (continued)								
Vieira et al. (2013) OH and WV, USA 1996–2005 (incidence)Cases: study 1: kidney cancer, 751, and bladder cancer, 1350; study 2: kidney cancer, 246, and bladder cancer, 395; index cancer cases were retrieved from cancer registries covering a comunity sample with relatively high exposure to PFOA because of contamination of dininking-water fromKidney, incidenceStudy 1. Residence in a PFOA-contaminated water district (OH and WV) (OR): Unexposed 657Age, sex, diagnosis year, insurance <i>Exposure assessment</i> critique: See Table 2.1. Other strengths: See Table 2.1. Well- ascertained cases based on cancer registries. Other of the study population addressed by a C8 Science Panel Project by Barry et al.Vieira et al. (2013) (DH and WV, USA (incidence)Cases: study 1: kidney, incidence kidney, incidenceStudy 1. Residence in a PFOA-contaminated water district (OH and WV) (OR): Unexposed study 2. Individual-level annual PFOA serum exposure, assuming 10-yr residency and latency (OH only) (OR): UnexposedAge, race, sex, diagnosis year, insurance <i>Exposure assessment</i> critique: See Table 2.1. Other strengths: See Table 2.1. Well- ascertained cases based on cancer registries. Other comments: See Table 2.1. Substantial overlap of the study population addressed by a C8 Science Panel Project by Barry et al.	Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
the Parkersburg (WV),High222.0 (1.3-3.2)(2013).polymer-production(30.8-109 μg/L)(30.8-109 μg/L)2.0 (1.0-3.9)plant; 18 differentVery high92.0 (1.0-3.9)cancers were analysed(110-655 μg/L)(110-655 μg/L)(bladder, brain, femalerectum, kidney,100-655 μg/L)breast, cervix, colon/rectum, kidney,100-655 μg/L)leukaemia, liver, lung,melanoma of the skin,100-655 μg/L)multiple myeloma,NHL, ovary, pancreas,prostate, testis, thyroid,and uterus)00	Vieira et al. (2013) OH and WV, USA 1996–2005 (incidence) Case–control	Cases: study 1: kidney cancer, 751, and bladder cancer, 1350; study 2: kidney cancer, 246, and bladder cancer, 395; index cancer cases were retrieved from cancer registries covering a community sample with relatively high exposure to PFOA because of contamination of drinking-water from the Parkersburg (WV), polymer-production plant; 18 different cancers were analysed (bladder, brain, female breast, cervix, colon/ rectum, kidney, leukaemia, liver, lung, melanoma of the skin, multiple myeloma, NHL, ovary, pancreas, prostate, testis, thyroid, and uterus)	Kidney, incidence	Study 1. Residence in district (OH and WV Unexposed Any exposed water district Study 2. Individual- exposure, assuming (OH only) (OR): Unexposed Low (3.7–12.8 µg/L) Medium (12.9–30.7 µg/L) High (30.8–109 µg/L) Very high (110–655 µg/L)	n a PFOA-cd 7) (OR): 657 94 level annua 10-yr reside 187 11 17 22 9	ontaminated water 1 1.1 (0.9–1.4) 1 PFOA serum ency and latency 1 0.8 (0.4–1.5) 1.2 (0.7–2.0) 2.0 (1.3–3.2) 2.0 (1.0–3.9)	Age, sex, diagnosis year, insurance provider, smoking status Age, race, sex, diagnosis year, insurance provider, smoking status	Exposure assessment critique: See Table 2.1. Other strengths: See Table 2.1. Well- ascertained cases based on cancer registries. Other comments: See Table 2.1. Substantial overlap of the study population addressed by a C8 Science Panel Project by <u>Barry et al.</u> (2013).	
	(indea)								
---	--	--	--	----------------------------------	---	------------------------------	----------		
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Vieira et al. (2013) OH and WV, USA	Controls: study 1: 23 548 (for kidney),	Urinary bladder, incidence	Study 1. Residence in district (OH and WV	n a PFOA-co /) (OR):	ontaminated water	Age, sex, diagnosis year,			
1996-2005	22 198 (for bladder)		Unexposed	1213	1	insurance			
(incidence) Case-control	other cancers; study 2: 7339 (for kidney), 6944		Any exposed water district	137	0.8 (0.7–1.0)	provider, smoking status			
(cont.) (for bladder); for each cancer site evaluated, controls were cases of	Urinary bladder, incidence	Study 2. Individual-l exposure, assuming (OH only) (OR):	evel annual 10-yr reside	l PFOA serum ency and latency	Age, race, sex, diagnosis year, insurance				
	(excluding sites in the		Unexposed	326	1	provider, smoking			
	kidney, testis, pancreas, and liver, which have		Low (3.7–12.8 μg/L)	23	0.9 (0.6–1.4)	status			
	been associated with PFOA in studies in		Medium (12.9–30.7 μg/L)	21	0.9 (0.6–1.4)				
	experimental animals or humans)		High (30.8–109 μg/L)	21	1.2 (0.8–2.0)				
	Exposure assessment method: See <u>Table 2.1</u>		Very high (110–655 μg/L)	4	0.6 (0.2–1.5)				

Table 2.2 (continued)

AL, Alabama; APFO, ammonium perfluorooctanoate; approx., approximately; BMI, body mass index; CA, California; CI, confidence interval; CPS-II, Cancer Prevention Study II; eGFR, estimated glomerular filtration rate; FOSA, perfluorooctane sulfonamide; HI, Hawaii; HR, hazard ratio; IRR, incidence rate ratio; MEC, Multiethnic Cohort; MN, Minnesota; NDI, National Death Index; NR, not reported; OH, Ohio; OR, odds ratio; PFAS, per- and polyfluoroalkyl substances; PFHxS, perfluorohexanesulfonic acid; PFNA, perfluorononanoic acid; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PLCO, Prostate, Lung, Colorectal and Ovarian Cancer Screening Trial; ppm, parts per million; PTFE, polytetrafluoroethylene; RCC, renal cell carcinoma; RR, rate ratio; SEER, Surveillance, Epidemiology, and End Results; SIR, standardized incidence ratio; SMR, standardized mortality ratio; TFE, tetrafluoroethylene; UK, United Kingdom; US, United States; USA, United States of America; WV, West Virginia; yr, year(s). 20-year lag. Numbers were small (1–7 cases in each quartile of exposure with a 10-year lag), but the test for trend across quartiles was significant.

[The Working Group noted that the detailed historical exposure assessment using blood samples to model serum PFOA levels across time was an improvement in the exposure assessment. Results were limited by restriction to mortality rates and by small numbers of fatal kidney cancer. Possible co-exposure to TFE (IARC Group 2A) could not be excluded but seemed of minor importance. Earlier indications that PFOA may be associated with kidney disease and reduced glomerular filtration and thereby introduce reverse causation in studies on PFAS and kidney cancer have not been corroborated in later studies (Dhingra et al., 2016, 2017).]

Barry et al. (2013) evaluated the risk of kidney cancer, among other cancers, in 32 254 adult community residents in the Mid-Ohio Valley, USA, exposed to drinking-water contaminated with PFOA as a result of chemical plant emissions and in workers at a local chemical plant producing PTFE (C8 Health Project). The cohort was described in detail earlier (Section 2.1.5). Briefly, information on cancer occurrence was obtained by interview in 2005–2011 for the period from 1952 onwards. Cumulative exposure to PFOA in community residents was assessed using serum measurements in 2005–2006, historical regional and occupational data from several sources, and PFOA toxicokinetics. The estimated annualized serum PFOA concentrations matched well with measured levels (Spearman correlation, 0.71, comparing predicted levels with 2005-2006 measured levels, Winquist et al., 2013). Cumulative PFOA serum estimates in workers were estimated using a chemical plant-specific IEM.

Estimated cumulative serum PFOA concentration was associated with risk of kidney cancer; the hazard ratio for a one-unit increase in natural log-transformed serum PFOA was 1.10 (95% CI, 0.98–1.24; P = 0.10; 105 cases). Quartile analysis

also indicated positive trends with increasing exposure. The adjusted hazard ratio for the fourth quartile versus the first was 1.58 (95% CI, 0.88-2.84; linear trend test, P = 0.18). Risk estimates based upon 10-year lagged analyses were slightly attenuated in the fourth quartile but not in the third quartile.

[The Working Group noted that this study presented improvements over other cohort studies because of its large study population; the large exposure contrast including both high-level occupational PFOA exposure, environmental PFOA exposure, and PFOA background exposure; the comprehensive exposure modelling using biological measurements in combination with environmental data, also taking PFOA toxicokinetics and variation across time into account; and the statistical analyses adjusting for a number of covariates and including lagged analyses. Although the study almost entirely included residents alive in 2005, the participation in the C8 Health Project was high, and the cohort was largely representative of the target population (Winquist et al., 2013). Moreover, a simulation study did not indicate that failure to include residents who died from kidney cancer before enrolment would bias risk estimates towards null (Barry et al., 2015). Nevertheless, capture of a larger part of the at-risk population would have added additional value to this study.]

<u>Consonni et al. (2013)</u> investigated causespecific mortality rates in an international occupational cohort of 5879 male TFE workers, of whom 4205 were exposed to APFO. An individual semiquantitative estimate of cumulative TWA exposure to APFO was assigned from a study-specific JEM. The cohort was described in detail earlier (Section 2.1.6). Using national data as the referent, the risk of cancer of the kidney and urinary organs other than bladder (ICD-9 code 189) was elevated (SMR, 1.69; 95% CI, 0.81–3.11) but with no indication of an exposure–response relation (<u>Consonni et al., 2013</u>). [The Working Group noted that the informativeness of this study was limited because of the small numbers of exposed men with cancer (n = 10), the semiquantitative exposure assessment with few measurements available, and the high correlation between TFE and PFOA exposure. However, exposure to TFE at the Parkersburg facility – the largest facility of the study – was considered to be very low because of strict hygiene controls for this flammable and explosive compound.]

Shearer et al. (2021) conducted a general population-based case-control study, nested within the PLCO cohort, addressing the risk of RCC according to prediagnostic serum concentrations of eight PFAS compounds, including PFOA and PFOS. The PLCO cohort was described in detail earlier (Section 2.1.11). In brief, 324 cases of RCC and 324 individually matched controls with baseline serum samples were enrolled in 1993–2001 from the screening arm of a multicentre randomized cancer screening trial in USA that included approximately 150 000 citizens (approximately half, 74 000, randomly assigned to the screening arm of the trial, provided blood samples at the baseline screening examination; Hayes et al., 2000). The adjusted risk of RCC was increased in individuals with higher PFOA serum concentration. The adjusted odds ratio in the highest exposure quartile (> $7.3-27.2 \mu g/L$ [ng/mL]) versus the lowest (< 4.0 μ g/L [ng/mL]) was 2.63 (95% CI, 1.33-5.20) and, using a continuous exposure metric, the approximate risk related to a doubling of the serum concentration was 1.71 (95% CI, 1.23-2.37). Several potential confounders were controlled either by matching or by including variables in the models. The estimates of relative risk for PFOA changed little when PFOS and PFHxS were included in multivariable analysis and did not vary by kidney function (P for heterogeneity, 0.97), duration of time since blood sampling (P for heterogeneity, 0.32), and prior freeze-thaw cycles of the specimen (P for heterogeneity, 0.63). There was no indication of risk modification by sex (*P* for heterogeneity, 0.87) or age (*P* for heterogeneity, 0.66); although estimates did not differ significantly, associations seemed stronger among those with normal BMI (*P* for heterogeneity, 0.74), those without a history of hypertension (*P* for heterogeneity, 0.31), and among former and current smokers (*P* for heterogeneity, 0.24).

The adjusted risk of RCC was also increased in individuals with higher PFOS serum concentrations with a significant exposure–response trend, but the risk was attenuated when adjusted for PFOA and PFHxS serum concentrations.

The Working Group noted that this general population study was distinguished from other case-control studies by having PFOA analyses adjusted for other PFAS compounds, by benefiting from blood samples collected on average 8.8 years before diagnosis, and by adjustment for several potential confounders, which added strongly to the reliability of the results. Although using a single sample to measure PFAS was a potential limitation, there is some evidence, from analyses of repeat samples of PFOA, that single samples may represent long-term averages over a 5-8-year period, with potential misclassification resulting in only minor bias to the null (see Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: <u>https://publications.iarc.who.int/636</u>). Exposure assessment at a single time point precludes analyses of risk according to cumulative exposure in specified exposure windows; however, some insight was obtained by analyses stratified by years from blood collection to diagnosis. Overall, this study within a general population added substantially to the evaluation of risk of kidney cancer after exposure to PFOA and PFOS.]

Steenland et al. (2022) conducted a pooled analysis of two studies on PFOA and RCC, described above (<u>Barry et al., 2013; Shearer et al.,</u> 2021). Both studies were based upon quantitative assessment of PFOA serum concentrations and enabled exposure–response modelling for the purposes of assessment of lifetime excess risk and setting of limit values. The pooled analysis included 427 cases and 835 controls. The best-fitting dose–response model for the pooled data was a two-piece linear spline model with natural log-transformed serum PFOA and a knot at 2.55 (serum PFOA concentration, approximately 12.5 ng/mL). The log odds of RCC increased up to the knot and was flat thereafter. [The Working Group noted that the focus of this paper was on risk assessment and calculation of limit values and on quantitative exposure–response modelling that is important for causal inference.]

Li et al. (2022a) investigated the risk of kidney cancer in a community sample with a high level of exposure in Sweden, with follow-up from 1985 (when PFAS contamination of waterworks started) until the end of 2016 (Section 2.1.13). Exposure was categorized according to period and duration of living in a contaminated area. PFOA constituted only a minor part of the PFAS contamination, which was dominated by PFOS and PFHxS. SIRs, calculated separately for men and women, were adjusted for age and calendar year using regional reference rates. The SIR for kidney cancer for participants who had ever resided in an area with high PFAS contamination was elevated in women, but not in men. Internal comparisons of exposed and unexposed residents, adjusted for age, sex, and calendar year, revealed an increased risk among those who had ever been exposed (HR, 1.27; 95% CI, 0.85-1.91), with slightly higher risk in residents with longer and more recent exposure. [Major strengths included the complete registration of the cohort, no loss to follow-up, and a long follow-up period. Major limitations were the ecological exposure assessment based upon residence without individual estimates related to PFOS exposure, and the relatively few cases of kidney cancer, producing uncertain risk estimates.]

Rhee et al. (2023b) conducted a nested casecontrol study of prediagnostic serum concentrations of nine PFAS among 428 cases of RCC and 428 individually matched controls within the MEC (see Section 2.1.16). The MEC included more than 215 000 men and women aged 45-75 years at baseline (1993-1996) and represents a very racially, ethnically, and socioeconomically diverse population. Cohort members were living in Hawaii and California (primarily Los Angeles County), USA. Cases were ascertained via cancer registries in California and Hawaii. The controls were individually matched to cases on race, ethnicity, sex, age at serum sample, date of serum sampling, study centre (Hawaii or California), fasting status at time of sample, and time of day of sampling. The controls were not diagnosed with RCC at the time when their matched case was diagnosed. Eleven PFAS were measured, including PFOA, PFOS, PFHxS, and PFNA. Exposure levels were similar to those measured in the general population in NHANES. Analyses were conducted by conditional logistic regression maintaining the matched pairs, were mutually adjusted for all PFAS, and controlled for the matching factors, as well as for smoking status, eGFR, history of hypertension, and BMI. Analyses were carried out modelling the exposure both as categorical (using quartiles) or as continuous (using log₂) serum levels. PFOA and PFOS were correlated (Spearman correlation, $\rho = 0.61$), and PFNA was correlated with both PFOA and PFOS ($\rho = 0.57$ and $\rho = 0.48$, respectively). The legacy PFAS (PFOA, PFOS, PFHxS, and PFNA) were detected in \geq 97% of study participants. PFOA was not associated with renal cancer in the overall study group, with the OR for quartile 4 versus quartile 1 being 1.04 (95% CI, 0.60-1.81) and the OR for continuous log₂ PFOA being 0.89 (95% CI, 0.67–1.18). However, a positive association was observed for White participants, with an OR per log, PFOA concentration of 2.12 (95% CI, 0.87–5.18) and higher ORs for upper quartiles (ranging from 2.1 to 3.6 for quartiles 2 to 4 versus

quartile 1). There was also a suggestive association for those sampled before 2002 (OR per \log_2 PFOA concentration, 1.49; 95% CI, 0.77–2.87).

[The Working Group considered this nested case-control study to be informative, given the large sample size, the adjustment for multiple PFAS, the multiple racial or ethnic groups studied, the good cancer ascertainment via registries, and the availability of serum levels before diagnosis. The limitations were mainly the small sample sizes for different racial or ethnic groups.]

Winquist et al. (2023) conducted a casecohort study within the prospective CPS-II LifeLink Cohort of the ACS, with measurements of PFOA, PFOS, and several other PFAS in prediagnostic serum samples collected during 1998-2001 (Section 2.1.21). Overall, there was no increased risk of kidney cancer or RCC with increasing serum PFOA. In women, serum PFOA concentration was positively associated with RCC (HR per doubling of serum PFOA, 1.54; 95% CI, 1.05-2.26), whereas no association was observed in men. [The Working Group noted several strengths, including the casecohort design, the large sample size, the good cancer ascertainment via registries and examination by histological subtype, and availability of prediagnostic serum samples. The limitations were mainly the low exposure levels and narrow exposure contrast. There may be a survivor bias downwards for kidney cancer, for which a relatively high proportion of cases are diagnosed before age 65 years, because of the gap between exposure and enrolment.]

<u>Vieira et al. (2013)</u> conducted a case-control study in Ohio and West Virginia, USA, to investigate the risk of 18 cancers, including kidney cancer, in a community sample with relatively high exposure to PFOA due to contamination of drinking-water from the polymer-production plant in Parkersburg, West Virginia, USA (Section 2.1.22). Incident cancers diagnosed in 1996–2005 were identified from cancer registries. The control population was people with other cancers, except cancers of the kidney, testis, liver, or pancreas. Logistic regression was used to estimate odds ratios, which were adjusted for age, sex, diagnosis year, smoking status, and insurance provider. For the Ohio subset with individual-level serum estimates of exposure, the adjusted odds ratios (AOR) for kidney cancer were higher in the highest exposure categories, with indications of a dose-response relation: AOR, 0.8 (95% CI, 0.4–1.5; 11 cases), 1.2 (95% CI, 0.7–2.0; 17 cases), 2.0 (95% CI, 1.3-3.2; 22 cases), and 2.0 (95% CI, 1.0-3.9; 9 cases) versus unexposed for low, medium, high, and very high exposure categories, respectively. Estimates of PFOA annual serum levels 10 years before diagnosis were 3.7-12.8 µg/L [ng/mL], 12.9–30.7 µg/L [ng/mL], 30.8– 109 µg/L [ng/mL], and 110–655 µg/L [ng/mL] for these four categories.

For the combined populations of Ohio and West Virginia without individual-level exposure estimates, the odds ratio for kidney cancer was 1.1 (95% CI, 0.9–1.4; 94 exposed cases) for participants exposed to contaminated water districts relative to unexposed participants.

[The Working Group noted that some of the cancer cases were overlapping cases from the study by <u>Barry et al. (2013</u>). Strengths were the large study population with a strong exposure contrast and estimates of individual-level exposure for a subset of the population. Misclassification of exposure was likely to be non-differential, resulting in attenuated risk estimates (if truly deviating from unity). Limitations of the main analysis applying individual-level exposure estimates were mainly related to modelled exposure data.]

The Working Group conducted a random-effects meta-analysis to estimate the rate ratio (RR) per unit (linear) of serum PFOA, by following the same methodology outlined in <u>Bartell and Vieira (2021)</u>. Details of the methodology are outlined in Annex 3 (Supplementary analyses used in reviewing evidence on cancer in humans, available from: <u>https://publications.iarc.who.</u>

int/636). The studies by Raleigh et al. (2014) and Consonni et al. (2013) were not included because of the lack of serum measurements. Categorical rate ratios based on contrasting the upper category (usually quartiles) with the referent were used, along with the assumed midpoints of the upper category and referent, to regress the log of the rate ratios on the midpoints to obtain a single linear continuous coefficient that estimates the change in log rate ratio per 10 ng/mL of (linear) PFOA. When including the studies by Steenland and Woskie (2012), Barry et al. (2013), Vieira et al. (2013), Shearer et al. (2021), Rhee et al. (2023b), and Winquist et al. (2023), the meta-analysis described above gave a result for an increase in the meta-rate ratio (meta-RR) per increase of 10 ng/mL of PFOA as 1.15 (95% CI, 0.97–1.37; $I^2 = 0.91$). In a sensitivity analysis excluding the studies by Steenland and Woskie (2012) and Vieira et al. (2013), given the concern regarding overlap with Barry et al. (2013), this sensitivity analysis gave a result for an increase in the meta-rate ratio per increase of 10 ng/mL PFOA as 1.21 (95% CI, 0.94–1.57; *I*² = 0.95). [The Working Group noted that a general limitation of the meta-analysis was the assumption of a linear exposure-response relation, although it has been observed that in studies with continuous exposure coefficients, (Barry et al., 2013; Shearer et al., 2021; Rhee et al., 2023b; Winquist et al., 2023) a logarithmic transformation of PFOA levels seems to fit the data better than do the untransformed PFOA levels. Other main limitations of the meta-analysis were: (i) the estimate of the linear coefficient using assumed midpoints of only two categories (uppermost and lowest); (ii) the use of average duration of exposure to transform cumulative exposure in the studies by Barry et al., Steenland and Woskie, and Vieira et al. to an assumed average exposure; and (iii) the assumption in the studies by Rhee et al., Shearer et al., and Winquist et al. that a single PFOA measurement is a good estimate of long-term lifetime average exposure (beyond a 5-8-year

duration discussed in Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: <u>https://publications.</u> <u>iarc.who.int/636</u>). Given these limitations, as well as the high heterogeneity across studies with different strengths and weaknesses, the Working Group chose to not rely primarily on the metaanalysis of exposure–response relation to determine the hazard identification for kidney cancer in humans.]

2.2.2 Bladder cancer

See <u>Table 2.2</u>.

Raleigh et al. (2014) studied bladder cancer mortality and incidence among 4668 APFOexposed workers at an APFO-production facility in Minnesota, USA (see Section 2.1.1 for more details). APFO is the ammonium salt of PFOA; the two substances are usually considered chemically equivalent in aqueous biological media such as the human body (Vierke et al., 2012). Raleigh et al. also studied 4359 unexposed workers at a different plant. Workers at both plants were employed for ≥ 1 year at their respective plants. For bladder cancer mortality, using the Minnesota population as the referent, the SMR for unexposed workers was 0.62 (95% CI, 0.27-1.22; 8 deaths). The SMRs for exposed workers, divided into quartiles of estimated APFO air exposure, were 0.40, 0.93, 1.61, and 0.53, based on 1, 2, 4, and 1 death, respectively. Bladder cancer incidence was ascertained using the Minnesota state cancer registry. Bladder cancer incidence hazard ratios, by quartile of estimated cumulative APFO air exposure, using unexposed workers as the referent, were 0.81 (95% CI, 0.36–1.81; 7 cases), 0.78 (95% CI, 0.33–1.85; 6 cases), 1.50 (95% CI, 0.80-2.81; 15 cases), and 1.66 (95% CI, 0.86-3.18; 12 cases).

<u>Alexander et al. (2003)</u> studied a cohort of 2083 production workers (145 deaths) who were exposed to PFOS at a plant in Decatur, Alabama, USA, that produced speciality films and

fluorochemicals, and who had worked ≥ 1 year at the plant between 1961 and 1997. Based on serum measurements for a sample of workers, a JEM was developed for all workers whereby jobs were classified into three exposure groups. Alexander and Olsen (2007) further followed this PFOS cohort, focusing on bladder cancer incidence; cases were identified using both questionnaire (6 cases) and death certificates (5 cases). Groups with no, low, and high exposure were estimated to have serum PFOS levels of [110-290 ng/mL], [390-890 ng/mL], and [1300-1970 ng/mL], respectively. Among those with any PFOS exposure, the bladder cancer SIR was 1.70 (95% CI, 0.77-3.22; 9 cases) compared with US cancer rates. Using a US population as the referent, the SIRs according to increasing cumulative exposure were 1.07 (95% CI, 0.12-3.85), 0.95 (95% CI, 0.25-2.43), 2.72 (95% CI, 0.55-73.95) and 1.43 (95% CI, 0.16-5.15). Comparing the three groups with the highest cumulative exposure with the group with the lowest cumulative exposure (internal referent, two cases), relative risks by increasing exposure were 0.83 (95% CI, 0.15-4.65; 4 cases), 1.92 (95% CI, 0.30-12.06; 3 cases), and 1.52 (95% CI, 0.21-10.99; 2 cases). A further study of medical care for some of these employees was conducted by Olsen et al. (2004), but this study was limited to certain categories of workers eligible for employer-provided care during the period 1993-1998. [The Working Group considered this study to provide only minimal information for estimating cancer incidence in this cohort.]

Steenland and Woskie (2012) studied cancer mortality among 5791 workers exposed to PFOA at a polymer-production plant in Parkersburg, West Virginia, USA (see Section 2.1.3 for more details). Compared with other non-exposed workers at other plants in the same company, the authors found an SMR for bladder cancer of 1.08 (95% CI, 0.52–1.99; 10 deaths). By quartile of estimated cumulative exposure, SMRs were 1.24 (95% CI, 0.15–4.47; 2 deaths), 2.49 (95% CI, 0.97–5.78; 6 deaths), 0.39 (95% CI, 0.01–2.17; 1 death), and 0.36 (95% CI, 0.10–2.01; 1 death). <u>Steenland et al. (2015)</u> followed a subset (n = 3713) of the PFOA-exposed workers in <u>Steenland and</u> <u>Woskie (2012)</u> for bladder cancer incidence. Bladder cancers were found via interview and confirmed via medical records, or via matching to local cancer registries. These authors found, when analysing estimated cumulative serum exposure by quartiles with a 10-year lag and using the lowest quartile as the referent in an internal comparison, RRs of 0.55 (95% CI, 0.12–2.61), 0.47 (95% CI, 0.10–2.21), and 0.31 (95% CI, 0.06–1.54), respectively, based on 29 cases of incident bladder cancer.

Eriksen et al. (2009) conducted a case-cohort study (713, 332, 128, and 67 patients with prostate, bladder, pancreatic, and liver cancer, respectively, and 772 cancer-free participants selected randomly from the full cohort) in a general-population national cohort of 57 053 people in Denmark. Analysis of bladder cancer incidence was done using baseline-measured plasma level of both PFOA and PFOS (Section 2.1.4 for more details). All participants had no previous diagnosis of cancer at the beginning of follow-up. Follow-up for cancer patients ranged from 0 to 12 years (median, 7 years). Analyses of IRRs by quartile of PFOA measured at baseline, using quartile 1 (84 cases) as the referent, were 0.71 (95% CI, 0.46–1.07), 0.92 (95% CI, 0.61–1.39), and 0.81 (95% CI, 0.53-1.24), respectively, based on 82, 83, and 83 cases, respectively. Corresponding RRs for PFOS measured at baseline, using quartile 1 (83 cases) as the referent, were 0.76 (95% CI, 0.50–1.16), 0.93 (95% CI, 0.61–1.41), and 0.70 (95% CI, 0.46-1.07), based on 84, 83, and 82 cases, respectively.

Barry et al. (2013) analysed bladder cancer incidence in a cohort of 32 254 participants with both low and high exposure to drinking-water containing PFOA (with high exposure being similar to the high levels in occupational cohorts), who were living near the plant in Parkersburg, West Virginia, USA (see Section 2.1.5 for more details). The median PFOA concentration measured in all cohort members in 2005–2006 was 26.1 µg/L [ng/mL], and the mean was 86.6 µg/L [ng/mL] (the US general population concentration was about $4 \mu g/L [ng/mL]$ at the time) (Winquist et al., 2013). Approximately 12% of participants in this study had worked in the Parkersburg plant that was the source of the PFOA contamination. Cancer incidence was determined via interview, with confirmation from medical records or from linkage with Ohio and West Virginia cancer registries. Hazard ratios were estimated per unit of increase in natural log-transformed cumulative serum level (a continuous variable), with serum levels over time estimated by a model with good correlation (Spearman correlation, 0.71) to observed serum levels that were available in 2005-2006 for all cohort members (Winquist et al., 2013). Hazard ratios were 1.00 (95% CI, 0.89–1.12) with no lag and 0.98 (95% CI, 0.88–1.10) with a 10-year lag (0 exposure assigned during most recent 10 years), based on 105 cases of incident bladder cancer. [The Working Group noted that among the non-occupational studies with bladder cancer outcomes, the larger studies with the best-characterized individual exposure were those of Eriksen et al. (2009) and Barry et al. (2013). The former was a study of a general population with low background levels of exposure, whereas the latter included both low-exposure participants and participants with very high exposures similar to occupational levels. Hence, the exposure contrasts were much smaller in the former than the latter, but the results of these two larger studies were nonetheless concordant in finding no association with bladder cancer for either PFOA or PFOS (Barry et al. did not study PFOS).]

<u>Consonni et al. (2013)</u> conducted an international cohort study of mortality in male workers at six TFE-production sites who were concomitantly exposed to APFO (or equivalently PFOA, as APFO breaks down to PFOA when soluble). The Spearman correlation between APFO and TFE in this study was 0.72 (see Section 2.1.6 for more details). Restricting the cohort to workers who had ever had exposure to APFO, in the supplemental data, the authors reported a bladder cancer SMR (versus national rates) of 0.55 (95% CI, 0.11-1.60; 3 deaths). [The Working Group noted that the small numbers of cases of incident bladder cancer and of deaths from bladder cancer in each of the three occupational cohorts reported in five papers (Alexander et al., 2003; Alexander and Olsen, 2007; Steenland and Woskie, 2012; Consonni et al., 2013; Steenland et al., 2015) limited the ability to evaluate associations between PFOA and PFOS and bladder cancer. Analyses of bladder cancer incidence in other studies noted below had better statistical precision. However, exposure contrasts between high and no or low exposure were often much reduced in the non-occupational studies.]

Li et al. (2022a) studied bladder cancer incidence in Ronneby, Sweden, among 60 507 residents among whom one third of households had been exposed to relatively high levels of both PFOS and PFOA in drinking-water contaminated by nearby military firefighting operations. [The Working Group noted that, although the authors were unable to estimate separate effects of the two exposures, the PFOS level was more than tenfold that of PFOA, on the basis of a subset of the participants with measured levels of these compounds in serum (see Section 2.1.13 for more details).] For men who had never resided in a high-exposure area, the SIR for bladder cancer incidence (the area surrounding Ronneby was used as the referent) was 0.94 (95% CI, 0.80-1.09; 166 cases), whereas for women it was 0.69 (95% CI, 0.48-0.95; 35 cases). For men ever living in a high-exposure area, the SIR was 1.10 (95% CI, 0.84-1.43; 57 cases), and for women the corresponding estimate was 1.13 (95% CI, 0.66–1.80; 17 cases). When Ronneby residents with everhigh exposure were compared with those with

never-high exposure, the bladder cancer hazard ratio was 1.30 (95% CI, 0.99–1.69). When Ronneby residents with ever-high exposure were further subdivided into "early-high" (lower exposure) and "late-high" (higher exposure), hazard ratios compared with the "never-high" exposure group were 1.20 (95% CI, 0.87–1.66) and 1.50 (95% CI, 0.98–2.29), respectively. [The Working Group noted that the study by Li et al. also had large exposure contrasts but was somewhat weakened by the fact that exposure was assigned ecologically depending on whether or not the participants lived in the Ronneby area.]

Winquist et al. (2023) studied 39 371 surviving participants in the CPS-II Nutrition Cohort (enrolled in 1991–1992) who resided in urban or suburban areas of 20 states in the USA and who had been recruited for participation in the CPS-II LifeLink Cohort. CPS-II LifeLink participants completed a LifeLink cohort baseline questionnaire and provided a blood sample in 1998-2001 (median age: 70 years overall, 71 years for men, 69 years for women) (Section 2.1.21). Using a casecohort approach, 396 cases of incident bladder cancer were identified and verified among those without previous cancer and compared with a randomly sampled subcohort of 500 men and 499 women. PFOA, PFOS, and several other PFAS compounds were measured in the collected blood samples. In the subsample, PFOS was present at the highest concentrations (median, 18.0 ng/mL), followed by PFOA (median, 5.2 ng/mL); levels were similar to those reported in NHANES. Cases were compared with the subcohort at risk at time of case occurrence via Cox regression. Hazard ratios for bladder cancer incidence were adjusted for sex, year of serum sample collection, age at serum collection; race and education from the 1982 baseline survey; smoking status and alcohol consumption from the 1997 survey (but not adjusted for other PFAS). Overall, for the sexes combined, there were no clear associations between PFOA or PFOS and bladder cancer. The hazard ratios for PFOA quartiles 2 to 4 versus

quartile 1 in relation to bladder cancer were 0.84 (95% CI, 0.56–1.26), 0.87 (95% CI, 0.58–1.30), and 0.86 (95% CI, 0.58–1.27), and there was no continuous (using \log_2 of serum levels) trend (HR, 0.93; 95% CI, 0.77–1.13; P = 0.478). For PFOS, quartile analyses showed hazard ratios of 0.81 (95% CI, 0.54–1.21), 1.07 (95% CI, 0.72–1.60), and 0.96 (95% CI, 0.64–1.44), and there was no evidence for an association, with a continuous hazard ratio of 1.01 (95% CI, 0.86–1.20; P = 0.890). Sex-specific analyses also showed no clear association for either PFOA or PFOS.

[The study by <u>Winquist et al. (2023)</u> had low exposure contrasts, with a single baseline sample, a moderate number of cases, and good case ascertainment. A weakness was that this was a survivor cohort, with median age at enrolment of approximately 70 years and follow-up starting at time of blood draw approximately 8–9 years after enrolment, which would preclude the identification of bladder cancer cases during this period (eligibility for follow-up after serum sample excluded any prior cancer, fatal or not), resulting in a potential downward bias and minimal informativeness.]

Vieira et al. (2013) conducted two casecontrol studies of incident bladder cancer among residents of 13 counties in Ohio and West Virginia, USA, including both contaminated and non-contaminated water districts near the same plant in Parkersburg, West Virginia, which was the source of contamination in the population studied by <u>Barry et al. (2013)</u> (see Section 2.1.22). In the first case–control study, cases and controls (all other cancer cases excluding kidney, pancreatic, testicular, and liver cancers), obtained from both Ohio and West Virginia cancer registries, were compared with regard to residence in a contaminated or non-contaminated water district. The bladder cancer OR for exposed residents in a contaminated water district was 0.8 (95% CI, 0.7–1.0; 137 exposed cases) versus residents in non-contaminated water districts. These same authors also conducted a separate case-control study among Ohio residents; the cases were people with bladder cancer, and the controls were people with other cancers in the Ohio counties, again excluding kidney, pancreatic, testicular, and liver cancers. Exposure in the second study was based on estimated individual annual serum levels of PFOA at specific addresses at specific points in time, 10 years before the diagnosis dates for cases and controls. Relative to the unexposed, ORs for the participants with low, medium, high, and very high-exposure 10 years before diagnosis were 0.9 (95% CI, 0.6-1.4; 23 cases), 0.9 (95% CI, 0.6-1.4; 21 cases), 1.2 (95% CI, 0.8–2.0; 21 cases), and 0.6 (95% CI, 0.2–1.5; 4 cases), respectively, in the second case-control study.

[The Working Group noted that the study by <u>Vieira et al. (2013)</u> included participants with the same large exposure contrasts as in <u>Barry et al. (2013)</u>, but it was also somewhat weakened by small numbers in high-exposure groups and assignment of either group-level exposure or broadly estimated individual exposure 10 years before diagnosis.]

2.3 Cancers of the male genital tract

See <u>Table 2.3</u>.

2.3.1 Testicular cancer

Of the studies listed in <u>Table 2.3</u>, associations between PFOA and/or PFOS exposure and testicular cancer were evaluated in two cohort studies (<u>Barry et al., 2013</u>; <u>Li et al., 2022a</u>), one prospective nested case–control study (<u>Purdue et al., 2023</u>), and one cancer registry-based case– control study (<u>Vieira et al., 2013</u>) that probably had some overlap with the <u>Barry et al. (2013</u>) cohort study. Two occupational cohort mortality studies (<u>Steenland and Woskie, 2012</u>; <u>Consonni et al., 2013</u>) reported SMRs based on 1 death from testicular cancer; a third occupational cohort (<u>Raleigh et al., 2014</u>) identified 5 cases of incident testicular cancer among PFOA-exposed workers but did not report estimates of association with testicular cancer. [Given the small numbers of deaths attributable to testicular cancer and lack of risk estimates in the latter studies, the Working Group focused on the investigations by <u>Barry et al.</u> (2013), Li et al. (2022a), <u>Purdue et al. (2023)</u>, and <u>Vieira et al. (2013)</u> in the following summary.] In addition, the Working Group conducted an analysis of data from studies carried out in the Veneto region of Italy (an area in which drinking-water is contaminated with PFAS).

Barry et al. (2013) evaluated the risk of testicular cancer in a study of 32 254 community residents and workers exposed to PFOA from a fluoropolymer-production plant in the Mid-Ohio Valley, USA (see the description of the C8 Science Panel study in Section 2.1.5). In analyses that included 17 validated incident testicular cancer cases, the authors observed evidence of an exposure-response association with estimated cumulative PFOA serum concentrations (unlagged analysis: adjusted HR for 1-unit increase in natural log-transformed levels, 1.34; 95% CI, 1.00-1.79). The corresponding hazard ratio in analyses comparing those in the highest and lowest PFOA exposure quartiles was 3.17 (95% CI, 0.75–13.45); in the categorical analysis, P = 0.04 for the exposure-response trend based on the within-category midpoints. The patterns of associations were similar, albeit slightly attenuated, in analyses with exposures lagged by 10 years (continuous: HR 1.28; 95% CI, 0.95-1.73; categorical, quartile 4 versus quartile 1: HR, 2.36; 95% CI, 0.41–13.65; P for trend, 0.02). Of the 17 cases of testicular cancer with complete covariate data that were included in these analyses, 15 were reported among community members and 2 among workers; in analyses excluding those employed at the plant, stronger associations were observed (continuous with no lag: HR, 1.73; 95% CI, 1.24–2.40; continuous with 10-year lag: HR, 1.53; 95% CI, 1.09-2.15).

Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Raleigh et al.	9027 men (3716	Prostate,	Exposed to APFO (SI	MR, MN re	ferent):	Age, calendar	Exposure assessment
(2014) MN, USA	exposed, 3834 reference); Cottage	mortality	Unexposed (Saint Paul plant)	48	1.03 (0.76–1.37)	period	<i>critique</i> : See <u>Table 2.1</u> .
Enrolment, 1947–2002/	Grove (MN), PFOA cohort; workers		Exposed (Cottage Grove plant)	24	0.83 (0.53–1.23)		<i>Other strengths</i> : See <u>Table 2.1</u> .
follow-up, 1947– employed for ≥ 1 yr 2008 (mortality), during 1947–2002 at an	employed for ≥ 1 yr during 1947–2002 at an	Prostate, mortality	Estimated cumulativ quartile (SMR, MN r	e airborne . eferent):	APFO exposure		Other limitations: See <u>Table 2.1</u> . Risk estimates
(incidence) Cohort	APFO facility (Cottage Grove; <i>n</i> = 4668); reference workers without any exposure	APFO facility (Cottage Grove; <i>n</i> = 4668); reference workers	1st quartile (< 2.6 × 10 ⁻⁵ μg/ m ³ -year)	5	0.66 (0.21–1.54)		cancer due to the small number of incident cases
to APFO employed a a tape and abrasives production facility	to APFO employed at a tape and abrasives production facility located in the same		2nd quartile (2.6 × 10 ⁻⁵ to < $1.4 \times 10^{-4} \mu g/m^{3}$ -year)	8	1.15 (0.50–2.27)		(n = 5).
	suburban geographical area and managed by the same company (Saint Paul; $n = 4359$)		3rd quartile (1.4×10^{-4} to < $7.3 \times 10^{-4} \mu g/m^{3}$ -year)	3	0.37 (0.08–1.07)		
	Exposure assessment method: see <u>Table 2.1</u>		4th quartile ($\geq 7.3 \times 10^{-4} \mu\text{g}/\text{m}^3$ -year)	8	1.29 (0.56–2.54)		
		Prostate, mortality	Estimated cumulative airborne APFO exposure quartile (HR):			Age, year of birth	
			Unexposed (Saint Paul plant)	NR	1		
			1st quartile (< 2.9 × 10 ⁻⁵ μg/ m³-year)	NR	0.34 (0.25–1.60)		
			2nd quartile (2.9 × 10 ⁻⁵ to < 1.5 × 10 ⁻⁴ μg/ m ³ -year)	NR	1.12 (0.53–2.37)		

Table 2.3 Epidemiological studies on exposure to PFOA and PFOS and cancers of the male genital tract

Table 2.3 (co	ntinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
<u>Raleigh et al.</u> (<u>2014)</u> MN, USA Enrolment,		Prostate, mortality (cont.)	3rd quartile (1.5×10^{-4} to < $7.9 \times 10^{-4} \mu g/m^{3}$ -year)	NR	0.36 (0.11–1.17)	Age, year of birth	
1947–2002/ follow-up, 1947– 2008 (mortality),			4th quartile ($\geq 7.9 \times 10^{-4} \mu g/m^3$ -year)	NR	1.32 (0.61–2.84)		
1988–2008 (incidence)		Prostate, incidence	Estimated cumulativ quartile (HR):	e airborne .	APFO exposure		
Cohort (cont.)			Unexposed (Saint Paul plant)	253	1		
			lst quartile (< 2.9 × 10 ⁻⁵ μg/ m ³ -year)	42	0.80 (0.57–1.11)		
			2nd quartile (2.9×10^{-5} to < $1.5 \times 10^{-4} \mu g/m^{3}$ -year)	42	0.85 (0.61–1.19)		
			3rd quartile (1.5×10^{-4} to < $7.9 \times 10^{-4} \mu g/m^{3}$ -year)	49	0.89 (0.66–1.21)		
			4th quartile ($\geq 7.9 \times 10^{-4} \mu g/m^3$ -year)	55	1.11 (0.82–1.49)		

254

Table 2.3 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
Steenland and	5791; Parkersburg	Testis, mortality	PFOA-exposed work	ers (SMR):		Age, calendar	Exposure assessment			
Woskie (2012)(WV), polymer- productionParkersburgproduction(WV), USAoccupational PFOAEnrolment,cohort; workers (men, 1948-2002/1948-2002/81%) at a polymerfollowum 1052morrufa during		Other workers referent (same region and company)	1	1.80 (0.05–10.03)	period	<i>critique</i> : See <u>Table 2.1</u> . <i>Other strengths</i> : See <u>Table 2.1</u> . Ability to				
		US referent	1	0.74 (0.02-4.12)		evaluate associations in				
1010 mortality	facility who had	Prostate, mortality	PFOA-exposed work	ers (SMR):			a high PFOA-exposed			
2008 (mortality) facility wh Cohort potential e fluoropoly sufficiently	potential exposure to fluoropolymers with sufficiently detailed work histories		Other workers referent (same region and company)	21	0.76 (0.47–1.16)		<i>Other limitations</i> : See <u>Table 2.1</u> . Limited statistical power to assess mortality from testicular and prostate cancers.			
	Exposure assessment		US referent	21	0.72 (0.45–1.10)					
	method: see Table 2.1	Prostate, mortality	Cumulative serum en workers referent, san							
			1st quartile (0 to < 904 ppm-years)	6	1.07 (0.39–2.34)					
			2nd quartile (904 to < 1520 ppm-years)	6	0.82 (0.30-1.78)					
			3rd quartile (1520 to < 2700 ppm-years)	5	0.65 (0.21–1.51)					
			4th quartile (≥ 2700 ppm-years)	4	0.57 (0.16-1.46)					

Table 2.3 (col	ntinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Steenland et al.	3713 (2955 male); A	Prostate,	Cumulative PFOA ex	posure, no	lag (RR):	Age, race,	Exposure assessment
(<u>2015)</u> Parkersburg	subset of Parkersburg (WV), polymer-	incidence	1st quartile (< 3.03 μg/mL-years)	NR	1	education, BMI, time-	<i>critique</i> : See <u>Table 2.1</u> .
(WV), USA Enrolment,	USA production PFOA nent, cohort in <u>Steenland</u> 2002/ and Woskie (2012);	OA <u>nland</u>	2nd quartile (3.03 to < 6.16 μg/mL-years)	NR	1.81 (0.69–4.78)	varying smoking, time-	<i>Other strengths</i> : See <u>Table 2.1</u> . Ability to
1948-2002/and Woskie (2012);follow-up,polymer-production1951-interviewworkers (mean, 80%)date in 2008-who responded (self2011 (incidence)or next-of-kin) to a		3rd quartile (6.16 to < 11.42 μg/mL-years)	NR	2.45 (0.96-6.25)	varying alcohol consumption, year of birth	between PFOA and prostate cancer incidence in a high PFOA-exposed population. <i>Other limitations</i> : See	
		4th quartile (≥ 11.42 μg/ mL-years)	NR	1.88 (0.72-4.88)			
Conort	health outcomes and		Trend-test P-value, 0	.11			selection bias given that
	who had measured or	Prostate,	Cumulative PFOA ex	xposure, 10-	-yr lag (RR):		the investigation included
	estimated occupational and residential	incidence	1st quartile (< 0.8 μg/mL-years)	NR	1		only 62% of the target population; inability to
	exposure estimates; 129 incident cases of		2nd quartile (0.8 to < 3.44 μg/mL-years)	NR	1.92 (0.56–6.58)		evaluate risk of testicular cancer and other relatively
	Exposure assessment		3rd quartile (3.44 to < 7.04 μg/mL-years)	NR	1.89 (0.57–6.34)		less common malignancies.
	inemoti. See <u>Table 2.1</u>		4th quartile (≥ 7.04 μg/mL-years)	NR	2.15 (0.64–7.26)		
			Trend-test P-value, 0	.10			

Table 2.3 (continued)

Table 2.3 (co	ntinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Eriksen et al.	Case-cohort within	Prostate,	Baseline plasma PFC	DA concentr	ation (IRR):	Age, years	Exposure assessment
(2009)	the Diet, Cancer and	incidence	1st quartile	179	1	of school	critique:
Denmark	Health cohort (see		2nd quartile	178	1.09 (0.78–1.53)	attendance,	See <u>Table 2.1</u> .
Enrolment, December 1993	<u>Table 2.1</u>) Cases: 713 incident		3rd quartile	178	0.94 (0.67–1.32)	BMI, dietary	Table 2.1. Large number of
to May 1997/	cases of prostate cancer		4th quartile	178	1.18 (0.84–1.65)	and vegetable	prostate cancer cases and
follow-up, 1	Comparison cohort:		Continuous (per	713	1.03 (0.99–1.07)	intake	non-cases.
December 1993	772 (680 men, 92		1 ng/mL increase)				Other limitations: See
to 1 July 2006	women); subcohort of	Prostate,	Baseline plasma PFC	OS concentr	ation (IRR):		Table 2.1. Results not
Case-cohort	participants randomly	incidence	1st quartile	179	1		reported for different time
	selected without cancer		2nd quartile	178	1.35 (0.97–1.87)		intervals between serum
at t	at the end of follow-up		3rd quartile	180	1.31 (0.94–1.82)		of prostate cancer: lack of
	method: see Table 2.1		4th quartile	176	1.38 (0.99–1.93)		mutually adjusted analyses
			Continuous (per 10 ng/mL increase)	713	1.05 (0.97–1.14)		of PFOS and PFOA.

PFOA and PFOS

Table 2.3 (co	ntinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Barry et al.	32 254 (28 541	Testis, incidence	Estimated cumulativ	ve PFOA ser	um	Age, time-	Exposure assessment
<u>(2013)</u>	community members		concentration (ng/m	L), no lag (l	HR):	varying	critique:
Mid-Ohio Valley	and 3713 workers);		1st quartile	NR	1	smoking, time-	See <u>Table 2.1</u> .
(OH and WV)	C8 Science Panel		2nd quartile	NR	1.04 (0.26-4.22)	varying alcohol	Other strengths: See
August 2005 to	people enrolled		3rd quartile	NR	1.91 (0.47–7.75)	education.	participation from those
August 2006/	in the C8 Health		4th quartile	NR	3.17 (0.75–13.45)	birth year	in the C8 Health Project;
follow-up, 1952 to 2011 (incidence)	Project who lived, worked, or attended school for ≥ 1 yr		Continuous (per unit on natural log scale)	17	1.34 (1.00–1.79)	(5-yr calendar intervals)	validation of diagnosed cancers. <i>Limitations</i> : See Table 2.1
Cohort	between 1950 and		Trend-test P-value, 0	Limited statistical power			
	3 December 2004 in a contaminated-water	Testis, incidence	Estimated cumulativ concentration (ng/m	re PFOA ser L), 10-yr lag	rum g (HR):		to assess risk of testicular cancer.
	district in the vicinity		1st quartile	NR	1		
	of a chemical plant		2nd quartile	NR	0.87 (0.15-4.88)		
	(Parkersburg (WV),		3rd quartile	NR	1.08 (0.20-5.90)		
	using PFOA in		4th quartile	NR	2.36 (0.41-13.65)		
	manufacturing, as well as a subset of those from the original		Continuous (per unit on natural log scale)	17	1.28 (0.95–1.73)		
	Parkersburg (WV),		Trend-test P-value, 0	.02			
polym occupa who w	polymer production occupational cohort who worked at the	Testis, incidence	Community resident PFOA serum concen (HR):				
	plant between 1948		1st quartile	NR	1		
	and 2002		2nd quartile	NR	0.80(0.16-3.97)		
	Exposure assessment		3rd quartile	NR	3.07 (0.61–15.36)		
	inctitou. see <u>rable 2.1</u>		4th quartile	NR	5.80 (0.97-34.58)		
			Continuous (per unit on natural log scale)	15	1.73 (1.24–2.40)		
			riend-test P-value, 0	.03			

Table 2.3 (co	ntinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
<u>Barry et al.</u> (2013) Mid-Ohio Valley		Testis, incidence	Community resident PFOA serum concent (HR):	ts: estimate tration (ng,	d cumulative /mL), 10-yr lag	Age, time- varying smoking, time-	
(OH and WV)			1st quartile	NR	1	varying alcohol	
Enrolment,			2nd quartile	NR	0.98 (0.13-7.14)	consumption,	
August 2005 to			3rd quartile	NR	1.54 (0.19–12.21)	education, birth year (5-yr calendar	
follow-up,			4th quartile	NR	4.66 (0.52-41.63)		
1952 to 2011 (incidence)			Continuous (per unit on natural log scale)	15	1.53 (1.09–2.15)	intervals)	
(cont.)			Trend-test <i>P</i> -value, 0	0.02			
		Prostate, incidence	Estimated cumulativ	ve PFOA ser 1L), no lag (1	rum HR):		
			Continuous (per unit on natural log scale)	446	0.99 (0.93–1.04)		
		Prostate, incidence	Estimated cumulativ concentration (ng/m	ve PFOA ser 1L), 10-yr lag	rum g (HR):		
			Continuous (per unit on natural log scale)	446	0.99 (0.94–1.05)		

Table 2.3 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Consonni et al.	5879 male workers	Testis, mortality	SMR (national refere	ent):		Age, calendar	Exposure assessment		
<u>(2013)</u> USA, UK,	(APFO-exposed, 4205); the pooled		Ever APFO- exposed	1	1.35 (0.03–7.49)	period, country	<i>critique</i> : See <u>Table 2.1</u> .		
Italy, Germany, int	international TFE	Prostate,	SMR (national referent):			Age, calendar	Other strengths: See		
Netherlands Enrolment, 1950–2002/ follow-up, 1950–2008 Cohort	cohort includes male workers who were ever employed or employed for 6 or 12 mo at one or more of six TFE-production sites in North America and Europe from 1950 to 2002; the principal occupational exposures were TFE and APFO (aiding production of PTFE) Exposure assessment method: see <u>Table 2.1</u>	mortality	Ever APFO- exposed	3	0.24 (0.05-0.70)	period, country	<u>Table 2.1</u> . <u>Other limitations:</u> See <u>Table 2.1</u> . Limited statistical power to assess mortality from testicular and prostate cancers.		

Table 2.3 (co	ntinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Rhee et al. (2023a) USA Enrolment, 1993–2001; follow-up (from blood draw), median, 9 yr (incidence) Nested case– control	Nested within the PLCO cohort (see Table 2.1 of Shearer et al., 2021) Cases: 750; aggressive prostate cancer (defined as stage III or IV, Gleason score \geq 8, or Gleason score 7 and death from prostate cancer) diagnosed \geq 300 days after blood collection Controls: 750; alive and cancer-free at time of case diagnosis, and individually matched to cases on age at baseline, race/ethnicity, study centre, calendar and study year of blood collection, and prior freeze-thaw cycle Exposure assessment method: see Table 2.1	Prostate (aggressive/ advanced), incidence Prostate (aggressive/ advanced), incidence	Serum PFOA concer $< 2.90 \ \mu g/L$ $2.90 \ to < 3.80 \ \mu g/L$ $3.80 \ to < 4.67 \ \mu g/L$ $4.67 \ to < 6.50 \ \mu g/L$ $\geq 6.50 \ \mu g/L$ Continuous (per unit increase on $\log_2 \text{ scale}$) Trend-test <i>P</i> -value, C Serum PFOA concer $< 2.90 \ \mu g/L$ $2.90 \ to < 3.80 \ \mu g/L$ $3.80 \ to < 4.67 \ \mu g/L$ $4.67 \ to < 6.50 \ \mu g/L$ Continuous (per unit increase on $\log_2 \text{ scale}$) Trend-test <i>P</i> -value, C Trend-test <i>P</i> -value, C	ntration (OF 194 155 130 149 122 750 0.005 ntration (OF 194 155 130 149 122 750 0.02	<pre> {): 1 0.75 (0.55-1.03) 0.65 (0.47-0.91) 0.69 (0.49-0.97) 0.57 (0.39-0.82) 0.82 (0.71-0.96) {): 1 0.75 (0.53-1.07) 0.72 (0.49-1.07) 0.67 (0.44-1.03) 0.54 (0.32-0.91) 0.79 (0.63-0.99) </pre>	Age, race/ ethnicity, study centre, calendar year of blood collection, study year of blood collection, prior freeze- thaw Age, race/ ethnicity, study centre, calendar year of blood collection, study year of blood collection, study year of blood collection, prior freeze- thaw, BMI, smoking status, family history of prostate cancer, history of diabetes, PFOS, PFHxS, PFNA, N-EtFOSAA, FOSA, N-MeFOSAA, PFHpS	Exposure assessment critique: See Table 2.1. Other strengths: See Table 2.1. Other limitations: See Table 2.1.

Table 2.3 (continued)									
Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
	Prostate (aggressive/ advanced),	No. of years from blo for a 1-unit increase scale):	ood draw to in serum Pl	diagnosis (OR FOA on log ₂	Age, race/ ethnicity, study centre,				
	incidence	< 1 to 3 yr	115	0.67 (0.51-0.87)	calendar				
		> 3 yr	635	0.90 (0.79–1.03)	year of blood collection				
	Prostate	Serum PFOS concen	tration (OR	.):	Age, race/				
	(aggressive/	< 19.10 µg/L	170	1	ethnicity,				
	advanced), incidence	19.10 to < 25.50 μg/L	145	0.86 (0.62–1.18)	study centre, calendar				
		25.50 to < 33.50 μg/L	168	0.99 (0.72–1.37)	year of blood collection,				
		33.50 to < 47.12 μg/L	136	0.80 (0.58–1.12)	of blood				
		$\geq 47.12 \mu g/L$	131	0.74 (0.51-1.06)	prior freeze-				
		Continuous (per unit increase on \log_2 scale)	750	0.93 (0.83–1.05)	thaw				
	ntinued) Population size, description Exposure assessment method	Population size, description Exposure assessment method Cancer type (histopathology), incidence or mortality Prostate (aggressive/ advanced), incidence Prostate (aggressive/ advanced), incidence Prostate (aggressive/ advanced), incidence Prostate (aggressive/ advanced), incidence	Population size, description methodCancer type (histopathology), incidence or mortalityExposure category or levelProstate (aggressive/ advanced), incidenceNo. of years from bld for a 1-unit increase acale): (1 to 3 yr > 3 yrProstate (aggressive/ advanced), incidenceNo. of years from bld for a 1-unit increase scale): (1 to 3 yr > 3 yrProstate (aggressive/ advanced), incidenceSerum PFOS concent (25.50 μg/L 25.50 to < 33.50 μg/L 3.50 to < 47.12 μg/L ≥ 47.12 μg/L Continuous (per unit increase on log2 scale) Trend-test P-value, 0	Population size, description Exposure assessment methodCancer type (histopathology), incidence or mortalityExposure category or levelExposed cases or deathsProstate (aggressive/ advanced), incidenceNo. of years from blood draw to for a 1-unit increase in serum Pl advanced), incidenceNo. of years from blood draw to for a 1-unit increase in serum Pl advanced), scale): incidenceProstate (aggressive/ advanced), incidenceNo. of years from blood draw to for a 1-unit increase in serum Pl scale): incidenceProstate (aggressive/ advanced), incidenceSerum PFOS concentration (OR (aggressive/ advanced), 19.10 to 145 25.50 to 145 25.50 to 145 25.50 to 168 25.50 to 136 47.12 µg/L 247.12 µg/L 247.12 µg/L 131 Continuous (per 	Population size, description methodCancer type (histopathology), incidence or mortalityExposure category or levelExposed cases or deathsRisk estimate (95% CI)Prostate (aggressive/ advanced), incidenceNo. of years from blod draw to diagnosis (OR for a 1-unit increase in serum PFOA on log2 scale): incidenceProstate (1 to 3 yr1150.67 (0.51-0.87) 0.90 (0.79-1.03)Prostate (aggressive/ advanced), incidenceSerum PFOS concentration (OR): (25.50 µg/L)100 μg/L1701 1 0.86 (0.62-1.18) (33.50 μg/L)Prostate (advanced), incidenceSerum PFOS concentration (OR): (25.50 µg/L)1680.99 (0.72-1.37) (33.50 µg/L)33.50 to < 136 0.80 (0.58-1.12) (47.12 µg/L)1310.74 (0.51-1.06) (Continuous (per or) unit increase on log2, scale)Trend-test P-value, 0.08Trend-test P-value, 0.08193	Population size, description methodCancer type (histopathology), incidence or mortalityExposure category or levelExposed cases or deathsRisk estimate (95% CI)Covariates controlledProstate (aggressive/ advanced), incidenceNo. of years from blood draw to diagnosis (OR scale): incidenceAge, race/ ethnicity, study centre, of a 1-unit increase in serum PFOA on log_ scale): incidenceAge, race/ ethnicity, study centre, of a 1-unit increase in serum PFOA on log_ scale): incidenceAge, race/ ethnicity, study centre, of a 1-unit increase in serum PFOA on log_ scale): incidenceAge, race/ ethnicity, ethnicity, scale): incidenceAge, race/ ethnicity, study centre, calendar incidenceAge, race/ ethnicity,<			

T

Reference, location, enrolment/ descriptionPopulation size, descriptionCancer type (histopathology) incidence or mortalityExposure category levelExposure category cases or dataRisk estimate (95% C1)Covariate controlledCommentsRhee et al. (2023a) USA Enrolment, follow up (from lobod drawy, median, 9 yr (incidence)Prostate (4023a) (2023b)Prostate (4023b)Serum PFOS concentation (OR (2023b)Age, race/ ethnicity, tudy centre, calendar (25.50 lo (23.50 log/L)17011993-2001: follow up (from lobod drawy, median, 9 yr (incidence)Prostate (25.50 log/L)1360.88 (0.53-1.46)ofbiod collection, study year ofbiod collection, study year ofbiod collection, study, smithy history of prostate cancer, history of diabetes, pFOA, PFIAS, PFNA, N-EEFOSAA, PFNA, N-EEFOSAA, PFNA, N-EEFOSAA, PFNA, N-EEFOSAA, PFNA, N-EEFOSAA, PFNA, N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA N-EEFOSAA, PFNA 	Table 2.3 (continued)										
Bhe ct al. (2023a)Prostate (aggressive) advanced) incidenceSerum PFOS concentration (OR): $v 10$ Age, race/ ethnicity, study centre, calendar year of blood collection, study year median, 9 yr (incidence)Prostate (25.50 to (25.50 to) (25.50 to) (2	Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Rhee et al.		Prostate	Serum PFOS concent	tration (OR	.):	Age, race/				
USA Enrolment, 1993-2001; follow-up (from blood draw), median, 9 yr (incidence)advanced), $< 33.50 \ ug/L$ 168 $< 1.07 \ (0.69-1.66)$ $< 33.50 \ ug/L$ study centre, calendar (alcient, study year of blood collection, study year of blood collection, of blood collection, protateNested case- control (cont.) $< 47.12 \ ug/L$ 131 $0.88 \ (0.53-1.46)$ prof freeze- collection, prof freeze- smokingcontrol (cont.) $< 47.12 \ ug/L$ 131 $0.84 \ (0.45-1.58)$ prof freeze- smokingcontrol (cont.)Continuous (per $0g_s$ scale) 750 $0.99 \ (0.79-1.23)$ thaw, BMI, smokingcontrol (cont.)Trend-test P-value, 0.34 $N^{-1}NC^{-1}$	<u>(2023a)</u>		(aggressive/	< 19.10 µg/L	170	1	ethnicity,				
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	USA Enrolment,		advanced), incidence	19.10 to < 25.50 µg/L	145	0.93 (0.64–1.37)	study centre, calendar				
blood draw), median, 9 yr (incidence) Nested case- control (continuous (per 2 47.12 µg/L 131 0.84 (0.45-1.58) prior freeze- control (continuous (per 750 0.99 (0.79-1.23) thaw, BMI, collection, prostate cancer, history of prostate cancer, history of diabetes, prostate cancer, history of diabe	1993–2001; follow-up (from			25.50 to	168	1.07 (0.69–1.66)	year of blood collection,				
$\begin{array}{ccc} (\operatorname{incidence}) & \geq 47.12 \ \mu g/L & 131 & 0.84 \ (0.45-1.58) & \operatorname{prior freeze-} \\ \operatorname{control} & \operatorname{Continuous (per} & 750 & 0.99 \ (0.79-1.23) & \operatorname{thaw, BMI,} \\ (\operatorname{cont.}) & \operatorname{unit increase on} & & \operatorname{smoking} \\ \operatorname{log_2, scale} & & \operatorname{trend-test} P-\operatorname{value, 0.34} & & \operatorname{history of} \\ \operatorname{prostate} & & \operatorname{cancer, history} \\ \operatorname{of diabetes,} & \operatorname{pFOA, PFHxS,} \\ \operatorname{pFOA, PFHxS,} \\ \operatorname{pFNA,} \\ \operatorname{N-EtFOSAA,} \\ \operatorname{FOSA,} \\ \operatorname{N-MeFOSAA,} \\ \operatorname{PFHpS} \\ \end{array} \\ \begin{array}{c} \operatorname{Prostate} & & \operatorname{No. of years from blood draw to diagnosis (OR \\ (aggressive/ \\ advanced), \\ \operatorname{scale}: & & \operatorname{scale}: \\ \operatorname{incidence} & < 1 \ to 3 \ yr & 115 & 0.85 \ (0.70-1.04) \\ < 3 \ to 5 \ yr & 89 & 0.94 \ (0.74-1.18) \\ \operatorname{year of blood} \\ \operatorname{collection} \\ \end{array} \\ \begin{array}{c} collection, \\ prostate \\ cancer, history \\ response \\ respons \\ respons \\ response \\ respon$	blood draw), median, 9 yr			33.50 to < 47.12 µg/L	136	0.88 (0.53-1.46)	study year of blood				
Net restPrint increaseProvide the form of th	(incidence)			$> 47.12 \mu g/L$	131	0 84 (0 45-1 58)	collection,				
(cont.) unit increase on log, scale) Trend-test <i>P</i> -value, 0.34 Trend-test <i>P</i> -value, 0.34 Trend-test <i>P</i> -value, 0.34 N-istory of prostate (aggressive/ (aggressive/ advanced), scale): incidence (1 to 3 yr (1 to 3 yr (2 to 5 yr (2 to 9 yr (2 to 1) (2 to	control			Continuous (per	750	0.99(0.79-1.23)	thaw, BMI,				
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	(cont.)			unit increase on		,	smoking				
$\begin{array}{llllllllllllllllllllllllllllllllllll$				\log_2 scale)			status, family				
ProstateNo. of years from blood draw to diagnosis (OR (aggressive/ advanced), incidenceAge, race/ ethnicity, scale): <td></td> <td></td> <td></td> <td>Trend-test <i>P</i>-value, 0</td> <td>.34</td> <td></td> <td>history of</td> <td></td>				Trend-test <i>P</i> -value, 0	.34		history of				
ProstateNo. of years from blood draw to diagnosis (OR (aggressive/ advanced), incidenceAge, race/ ethnicity, study centre, study centr							cancer, history				
Prostate (aggressive/ advanced),No. of years from blood draw to diagnosis (OR for a 1-unit increase in serum PFOS on log2 active)Age, race/ ethnicity, study centre, study centre, s							of diabetes,				
$\begin{array}{c} \mbox{PFNA,} & N-EtFOSAA, \\ \mbox{FOSA,} & N-MeFOSAA, \\ \mbox{PFHpS} \end{array}$							PFOA, PFHxS,				
ProstateNo. of years from blood draw to diagnosis (OR (aggressive/ advanced), incidenceAge, race/ ethnicity, study centre, incidence1to 3 yr1150.85 (0.70-1.04) 0.94 (0.74-1.18) year of blood collection25 to 9 yr1550.98 (0.82-1.19) 0.95 (0.84-1.09)							PFNA,				
ProstateNo. of years from blood draw to diagnosis (OR (aggressive/ advanced), incidenceAge, race/ ethnicity, study centre, calendar100 - MeFOSAA, PFHpS200 - Method Prostate201 - Method (aggressive/ advanced), incidence201 - Method (aggressive/ advanced), incidence201 - Method (aggressive/ advanced), incidence201 - Method (aggressive/ advanced), incidence201 - Method (aggressive/ advanced), incidence201 - Method (advanced), <b< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td>FOSA.</td><td></td></b<>							FOSA.				
Prostate (aggressive/ advanced), incidenceNo. of years from blood draw to diagnosis (OR for a 1-unit increase in serum PFOS on log2 scale):Age, race/ ethnicity, study centre, calendar2for a 1-unit increase in serum PFOS on log2 scale):ethnicity, study centre, calendar3to 3 yr1150.85 (0.70–1.04) year of blood collection5to 9 yr1550.98 (0.82–1.19) o.99 yr9yr3910.95 (0.84–1.09)							<i>N</i> -MeFOSAA,				
$\begin{array}{c c c c c c c c c c c c c c c c c c c $			Prostate	No. of years from blo	od draw to	diagnosis (OR	Age, race/				
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$			(aggressive/ advanced).	for a 1-unit increase scale):	in serum Pl	FOS on \log_2	ethnicity, study centre				
> 3 to 5 yr 89 0.94 (0.74-1.18) year of blood collection > 5 to 9 yr 155 0.98 (0.82-1.19) collection > 9 yr 391 0.95 (0.84-1.09)			incidence	< 1 to 3 yr	115	0.85 (0.70-1.04)	calendar				
> 5 to 9 yr 155 0.98 (0.82-1.19) collection > 9 yr 391 0.95 (0.84-1.09)				> 3 to 5 yr	89	0.94 (0.74–1.18)	year of blood				
> 9 yr 391 0.95 (0.84–1.09)				> 5 to 9 yr	155	0.98 (0.82-1.19)	collection				
				> 9 yr	391	0.95 (0.84-1.09)					

Table 2.3 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
Li et al. (2022a) Ronneby, southern Sweden	60 507 (including 31 938 men); the Ronneby Register	Testis, incidence	Residential exposure contaminated drink county excluding Ro	Residential exposure to highly PFAS- contaminated drinking-water (SIR, Blekinge county excluding Ronneby referent):			<i>Exposure assessment</i> <i>critique</i> : See Table 2.1.			
Enrolment,	Cohort included all		Never	30	0.85 (0.57-1.21)		<i>Other strengths</i> : See <u>Table 2.1</u> . Complete			
1985-2013/	individuals who ever		Ever	14	1.28 (0.70-2.15)					
follow-up, 1985– 2016 (incidence)	lived in Ronneby municipality in 1985–	Testis, incidence	Residential exposure to highly PFAS- contaminated drinking-water (HR):				ascertainment of community members in the cohort and follow up			
Collort	households received		Never	31	1		through register-based			
PFAS-contaminated		Ever	14	1.38 (0.73-2.61)		linkages; high contrast in				
	drinking-water from a waterworks	Testis, incidence	Time period of residential exposure to highly PFAS-contaminated drinking-water (HR):			PFAS exposur cohort.	PFAS exposures within the cohort.			
	situated near a		Never	31	1		<i>Other limitations</i> : See <u>Table 2.1</u> . Limited ability to assess potential effects			
	military airfield where		Early (1985–2004)	9	1.35 (0.64-2.84)					
	PFAS-containing		Late (2005–2013)	5	1.46 (0.55-3.83)					
	firefighting foam was used in 1985–2013	Testis, incidence	Duration of residential exposure to highly PFAS- contaminated drinking-water (HR):				of PFOS and PFOA individually; limited			
	with exposure		Never	31	1		risk of testicular cancer.			
	considered "ever-		Short (1–10 yr)	9	1.32 (0.63-2.79)					
	high"); subsets with		Long ($\geq 11 \text{ yr}$)	5	1.51 (0.56-4.03)					
lon (≥ par per	long-term exposure	Prostate,	Residential exposure	e to highly I	PFAS-					
	$(\geq 11 \text{ yr})$ in the latest part of the follow-up	incidence	contaminated drink county excluding Ro	ing-water (Sonneby refer	SIR, Blekinge rent):					
	period (2005–2013)		Never	712	1.14 (1.05–1.22)					
	highly exposed		Ever	181	0.96 (0.82-1.11)					
	Exposure assessment	seu Resement Prostate,	Residential exposure	e to highly I	PFAS-					
	method: see <u>Table 2.1</u>	incidence	contaminated drink	ing-water (1	HR):					
			Never	712	1					
			Ever	181	0.83 (0.71-0.98)					

Table 2.3 (co	ntinued)							
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
<u>Li et al. (2022a)</u> Bonnoby		Prostate,	Time period of resid	ential expo	sure to highly	Calendar year,		
southern Sweden		Incluence	Novor	712		age		
Enrolment,			$\frac{1000}{1000}$	/12	1 0.88 (0.72, 1.08)			
1985-2013/			Earry $(1963-2004)$	114 67	0.86(0.72 - 1.08)			
follow-up, 1985– 2016 (incidence)		Prostate, incidence	Duration of resident contaminated drink	ial exposur ing-water (l	e to highly PFAS- HR):			
(cont.)			Never	712	1			
			Short (1–10 yr)	95	0.96 (0.78-1.20)			
			Long (≥ 11 yr)	86	0.72 (0.58-0.91)			
<u>Purdue et al.</u> (2023)	Nested within a cohort of active-duty US Air	Nested within a cohort Testis, inciden of active-duty US Air	Testis, incidence	Serum PFOA (first/only sample) concentration (OR):			Date of birth, <i>Exposure assessment</i> , race/ethnicity, <i>critique</i> :	
USA	Force servicemen (see		\leq 4.45 ng/mL	161	1	year entered military service, sample availability, military grade.	See <u>Table 2.1</u> . Other strengths: See <u>Table 2.1</u> . Other limitations: See	
Enrolment,	<u>Table 2.1</u>)		4.46-5.87 ng/mL	115	0.7 (0.4–1.0)			
1988-2017/ follow-up	with two samples):		5.88–7.85 ng/mL	121	0.7 (0.5–1.0)			
through 2018	TGCT diagnosed		> 7.85 ng/mL	133	0.8 (0.5–1.2)		Table 2.1.	
Nested case- control	in the Department of Defence Cancer		Trend-test <i>P</i> -value, 0	0.46		number of deployments	1000 2.1	
	Registry	egistry Testis, incidence	Serum PFOA (first/only sample) concentration (OR):			Date of birth, race/ethnicity,		
			$\leq 4.45 \text{ ng/mL}$	161	1	year entered		
			4.46-5.87 ng/mL	115	0.7 (0.4-1.0)	military		
			5.88–7.85 ng/mL	121	0.7 (0.4–1.1)	availability		
			> 7.85 ng/mL	133	0.8 (0.5–1.4)	military grade.		
				Trend-test <i>P</i> -value, 0).86		number of deployments, other PFAS (PFOS, PFHxS, PFNA, PFDA, PFUnDA, <i>N</i> -MeFOSAA)	

Table 2.3 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
(2023)(187 with two sam one control per cash density-sampled with replacement amon eligible US Air For through 2018Nested case- control (cont.)date and matched date and matched date of birth, race/ ethnicity (seven military service, you of baseline serum sample collection, and year of second sample collection applicable)Exposure assessme method: see Table	(187 with two samples); one control per case, density-sampled with replacement among eligible US Air Force servicemen on active duty and cancer-free as of the case diagnosis date and matched by date of birth, race/	Testis, incidence	(OR): ≤ 4.25 ng/mL 4.26-5.65 ng/mL 5.66-7.55 ng/mL > 7.55 ng/mL Trend-test <i>P</i> -value, 0 Serum PFOA (second (OR):	55 52 39 41 .35 d sample) co	1 1.0 (0.6–1.8) 0.7 (0.4–1.4) 0.7 (0.4–1.5)	race/ethnicity, year entered military service, sample availability, military grade, number of deployments Date of birth, race/ethnicity,				
	ethnicity (seven groups), year entering military service, year of baseline serum sample collection, and year of second sample collection (if applicable) Exposure assessment method: see <u>Table 2.1</u>	3	<pre>≤ 4.25 ng/mL 4.26-5.65 ng/mL 5.66-7.55 ng/mL > 7.55 ng/mL Trend-test <i>P</i>-value, 0</pre>	55 52 39 41 .22	1 1.0 (0.5–2.0) 0.6 (0.3–1.4) 0.6 (0.2–1.6)	year entered military service, sample availability, military grade, number of deployments, other PFAS (PFOS, PFHxS, PFNA, PFDA, PFUnDA, <i>N</i> -MeFOSAA) Date of birth, race/ethnicity, year entered military service, sample availability, military grade, number of deployments				
		Testis, incidence	Serum PFOS (first/or (OR): ≤ 18.3 ng/mL 18.4–29.3 ng/mL 29.4–42.2 ng/mL > 42.2 ng/mL Trend-test <i>P</i> -value, 0	131 116 153 130 .64	1 1.0 (0.6–1.5) 1.4 (0.8–2.3) 1.2 (0.7–2.0)					

T

266

Table 2.3 (continued)											
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments				
Purdue et al. (2023) USA Enrolment, 1988–2017/ follow-up, through 2018 Nested case– control (cont.)	Testis, incid	Testis, incidence	Serum PFOS (first/or (OR): ≤ 18.3 ng/mL 18.4–29.3 ng/mL 29.4–42.2 ng/mL > 42.2 ng/mL Trend-test <i>P</i> -value, 0	11y sample) 131 116 153 130 .15	concentration 1 1.2 (0.7–1.9) 1.9 (1.0–3.4) 1.8 (0.9–3.6)	Date of birth, race/ethnicity, year entered military service, sample availability, military grade, number of deployments, other PFAS (PFOA, PFHxS, PFNA, PFDA, PFUnDA,					
		Testis, incidence	Serum PFOS (second (OR): ≤ 13.2 ng/mL 13.3–21.2 ng/mL 21.3–33.5 ng/mL > 33.5 ng/mL Trend-test <i>P</i> -value, 0	42 38 50 57 .02	1 1.1 (0.6–1.9) 1.9 (0.9–4.1) 2.6 (1.1–6.4)	Date of birth, race/ethnicity, year entered military service, sample availability, military grade, number of deployments					

Table 2.3 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
<u>Purdue et al.</u> (2023)		Testis, incidence	Serum PFOS (second (OR):	l sample) co	oncentration	Date of birth, race/ethnicity,				
USA			$\leq 13.2 \text{ ng/mL}$	42	1	year entered				
Enrolment,			13.3–21.2 ng/mL	38	1.5 (0.7-3.3)	military				
1988-2017/			21.3-33.5 ng/mL	50	2.8 (1.1-7.0)	service, sample				
tollow-up,			> 33.5 ng/mL	57	4.6 (1.4–15.1)	availability,				
Nested case-			Trend-test P-value, 0	.009		number of				
control (cont.)		Testis (seminoma),	Serum PFOS (first/only sample) concentration (OR):			deployments, other PFAS				
		incidence	Below median	NR	1	(PFOA, PFHxS,				
			Above median	NR	1.8 (1.0-3.3)	PFNA, PFDA,				
	Testis (semi	Testis (seminoma),	Serum PFOS (second sample) concentration (OR):			N-MeFOSAA)				
		incidence	Below median	NR	1					
			Above median	NR	2.8(1.2-6.3)					

Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Winquist et al.	Case-cohort study	Prostate,	Serum PFOA concer	tration (HI	R):	Year of	Exposure assessment
(2023) 20 US states	within the CPS-II Lifelink Cohort (see	incidence	1st quartile (< 4.000 ng/mL)	398	1	serum sample collection,	critique: See <u>Table 2.1</u> . Other strengths: See <u>Table 2.1</u> . Other limitations: See <u>Table 2.1</u> .
Enrolment 1998– 2001; follow-up	<u>Table 2.1</u>) Cases: 3762 overall		2nd quartile (4.000 to < 5.300 ng/mL)	391	0.82 (0.60–1.11)	age at serum collection, race, education, smoking status, alcohol consumption	
through 30 June 2015	(1610 prostate cancers); incident cases from the		3rd quartile (5.300 to < 6.900 ng/mL)	405	0.93 (0.68–1.27)		
Case-cohort	CPS-II Lifelink Cohort (surviving CPS-II Nutrition cohort participants) with first cancer diagnosis of prostate (men only) detected through self- report or NDI linkage and verified through medical records review or cancer registry; all participants with incident cancers		4th quartile (≥ 6.900 ng/mL)	405	0.83 (0.61–1.14)		
			Continuous (per unit on log ₂ scale)	1599	0.93 (0.79–1.08)		
		Prostate,	Serum PFOS concen				
		etected through self- incidence port or NDI linkage ad verified through tedical records review r cancer registry; l participants with acident cancers omparison cohort: 29; a sex-stratified mple random sample f 499 women and 20 men (~3% of the eligible cohort); ratification sampling as to ensure an dequate number of abcohort participants	1st quartile (< 14.000 ng/mL)	389	1		
			2nd quartile (14.000 to < 19.000 ng/mL)	392	0.94 (0.70-1.26)		
			3rd quartile (19.000 to < 26.000 ng/mL)	410	1.11 (0.81–1.50)		
	Comparison cohort: 999; a sex-stratified		4th quartile (≥ 26.000 ng/mL)	408	1.08 (0.80–1.46)		
	of 499 women and 500 men (~3% of the eligible cohort); stratification sampling		Continuous (per unit on log ₂ scale)	1599	1.00 (0.88–1.14)		
	adequate number of subcohort participants						
	in sex-specific analyses (for breast and prostate cancers)						
	Exposure assessment method: see <u>Table 2.1</u>						

Table 2.3 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
<u>Vieira et al.</u>	Cases: study 1: 134	Testis, incidence	Study 1: residence in	a PFOA-co	ontaminated	Age, diagnosis	Exposure assessment		
(<u>2013)</u>	cancers of the testis,		water district (OH ar	nd WV) (O	R):	year, insurance	critique:		
UH and WV,	36/8 cancers of the		Unexposed	116	1	provider,	See <u>Table 2.1</u> Other strengths: See <u>Table 2.1</u> . Other limitations: See <u>Table 2.1</u> . Other comments: Substantial overlap with <u>Barry et al. (2013)</u> .		
1996–2005 (incidence)	cancers of the testis,		Any exposed water district	18	1.0 (0.6–1.8)	sinoking status			
Case-control	prostate index cancer		Little Hocking	8	5.1 (1.6–15.6)				
Gube control	cases were retrieved		Lubeck	2	0.9 (0.2-4.5)				
	from cancer registries		Tuppers Plains	2	0.4 (0.1–2.0)				
covering a communi sample with relativel high exposure to PFOA because of contamination of	covering a community		Belpre	1	0.6 (0.1–5.0)				
	sample with relatively		Pomeroy	0	NC				
	PEOA because of		Mason	5	0.5 (0.2–1.5)				
	contamination of	Testis, incidence	Analysis 2: individual-level annual PFOA serum			Age, race, diagnosis year, insurança			
	drinking-water from		exposure, assuming 10-yr residency and latency						
	the Parkersburg (WV),		Unexposed	50	1	provider, smoking status			
	polymer-production		I_{OW} (3.7–12.8 µg/I)	1	0.2(0.0-1.6)				
	plant Controls: NR: for each		Low (3.7–12.8 μg/L) Medium	3	0.2(0.0-1.0) 0.6(0.2-2.2)	Ũ			
	cancer site evaluated.		(12.9–30.7 µg/L)	5	0.0 (0.2 2.2)				
	controls were cases		High (30.8–109 µg/L)	1	0.3 (0.0-2.7)				
	of cancer at all other		Very high	6	2.8 (0.8-9.2)				
	sites among men,		(110–655 μg/L)						
with the ex of four can priori inter testis, panc and liver) th been associ PFOA in str	with the exclusion of four cancers of a	Prostate, incidence	Analysis 1: Residence in a PFOA-contaminated water district (OH and WV) (OR):			Age, diagnosis year, insurance			
	testis pancreas		Unexposed	3244	1	provider,			
	and liver) that have		Any exposed water district	434	0.9 (0.8–1.1)	smoking status			
	PFOA in studies in		Little Hocking	36	1.4 (0.9–2.3)				
	experimental animals		Lubeck	78	1.2 (0.9–1.6)				
	or humans		Tuppers Plains	56	0.8 (0.6-1.1)				
	Exposure assessment		Belpre	56	0.8 (0.6–1.1)				
	method: see <u>Table 2.1</u>		Pomeroy	12	1.3 (0.6–2.6)				
			Mason	196	0.9 (0.7–1.0)				

Table 2.3 (co	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Vieira et al. (2013) OH and WV.	<u>ieira et al.</u> 013) H and WW	Prostate, incidence	Analysis 2: individua exposure, assuming (OH only) (OR):	al-level ann 10-yr reside	ual PFOA serum ency and latency	Age, race, diagnosis year, insurance	
USA			Unexposed	941	1	provider,	
1996-2005			Low (3.7–12.8 µg/L)	71	1.1(0.8-1.5)	smoking status	
(incidence) Case-control			Medium (12.9–30.7 μg/L)	65	0.8 (0.6–1.0)		
(cont.)			High (30.8–109 μg/L)	47	0.8 (0.5–1.1)		
			Very high (110–655 μg/L)	31	1.5 (0.9–2.5)		
Hardell et al. (2014)Cases: 201; patient. with prostate cance admitted for treatr	Cases: 201; patients with prostate cancer admitted for treatment	Prostate, incidence	Serum PFOA concer ≤1.9 ng/mL(median for controls)	ntration (OI 93	R): 1	Age, BMI, year of sampling	Exposure assessment critique: Key strengths were that whole blood levels represent
Sweden	at the University		> 1.9 ng/mL	108	1.1 (0.7–1.7)		
2007–2011	Hospital in Orebro	Prostate, incidence	Serum PFOS concentration (OR):				the combined exposure
Case-control	Controls: 186; cancer- free controls from		≤8.3ng/mL(median for controls)	92	1		through all exposure pathways; measurement error low. Key limitations were that if prostate cancer alters ADME of PFAS there could be possible differential
	Örebro County who		> 8.3 ng/mL	109	1.0 (0.6-1.5)		
	were identified from	Prostate (Gleason	Serum PFOA concer	ntration (OI	R):		
the Swedish pop registry and ma	the Swedish population registry and matched	score, 7–10), incidence	≤1.9 ng/mL(median for controls)	56	1		
	to cases on age		> 1.9 ng/mL	67	1.2 (0.7–1.8)		exposure misclassification,
	Exposure assessment	Prostate (Gleason	Serum PFOS concen	tration (OR	t):		as blood collection of cases
method: quan measurement	measurements; analytical method	score, 7–10), incidence	≤8.3ng/mL(median for controls)	53	1		was at or after diagnosis; single samples at time of case hospitalization may not reflect exposure at
	was state-of-the-art:		> 8.3 ng/mL	70	1.1 (0.7–1.9)		
	a single blood sample	Prostate (PSA	Serum PFOA concer	ntration (OI	R):		crucial windows in cancer
	was collected during the same time period	level ≥ 11), incidence	≤1.9 ng/mL(median for controls)	39	1		development.
	for cases and matched controls; blood was collected before		> 1.9 ng/mL	52	1.3 (0.8–2.1)		

Table 2.3 (co	ntinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description Exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Hardell et al.	treatment, during	Prostate (PSA	Serum PFOS concent	tration (OR	k):	Age, BMI, year	Other strengths: Availability
<u>(2014)</u> Örebro County,	hospitalization to receive treatment or at	level ≥ 11), incidence	≤8.3ng/mL(median for controls)	47	1	of sampling	of information on disease aggressiveness (Gleason score and PSA) and family history of prostate cancer. <i>Other limitations</i> : Lack of adjustment for other PFAS; relatively small sample size.
Sweden	general practitioners	itioners	> 8.3 ng/mL	44	0.8 (0.4–1.3)		
Case-control (cont.)		Prostate, incidence	Heredity and serum	PFOA conc	centration (OR):		
			No heredity, ≤ 1.9 ng/mL	77	1		
			Heredity, ≤ 1.9 ng/mL	16	1.1 (0.5–2.6)		
			No heredity, > 1.9 ng/mL	84	1.0 (0.6–1.5)		
			Heredity, > 1.9 ng/mL	24	2.6 (1.2-6.0)		
		Prostate,	Heredity and serum	PFOS conc	entration (OR):	Age, BMI, year	
		incidence	No heredity, ≤ 8.3 ng/mL	72	1	of sampling	
			Heredity, ≤ 8.3 ng/mL	20	1.2 (0.6–2.5)		
			No heredity, > 8.3 ng/mL	89	0.9 (0.5–1.4)		
			Heredity, > 8.3 ng/mL	20	2.7 (1.04–6.8)		

ADME, absorption, distribution, metabolism, and excretion; APFO, ammonium perfluorooctanoate; approx., approximately; BMI, body mass index; CI, confidence interval; CPS-II, Cancer Prevention Study II; N-EtFOSAA, 2-(N-ethyl-perfluorooctane sulfonamido) acetic acid; FOSA, perfluorooctane sulfonamide; HR, hazard ratio; IRR, incidence rate ratio; N-MeFOSAA, 2-(N-methyl-perfluorooctane sulfonamido) acetic acid; MN, Minnesota; mo, month(s); NC, not calculated; NDI, National Death Index; NR, not reported; OH, Ohio; OR, odds ratio; PFAS, per- and polyfluoroalkyl substances; PFDA, perfluorodecanoic acid; PFHpS, perfluoroundecanoic acid; PFHxS, perfluorohexanesulfonic acid; PFNA, perfluorononanoic acid; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PFUnDA, perfluoroundecanoic acid; PLCO, Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial; ppm, part per million; PSA, prostate-specific antigen; PTFE, polytetrafluoroethylene; RR, rate ratio; SIR, standardized incidence ratio; SMR, standardized mortality ratio; TGCT, testicular germ cell tumour; TFE, tetrafluoroethylene; UK, United Kingdom; US, United States; USA, United States of America; WV, West Virginia; yr, year(s). [The Working Group considered this study to be highly informative for the relation between PFOA and testicular cancer. Its primary strengths included the detailed characterization of estimated serum PFOA levels over time, the high response rate among participants in the C8 Health Project, and the validation of diagnosed testicular cancers through state registries and medical chart review. However, the small number of confirmed cases of testicular cancer was a limitation.]

Li et al. (2022a) investigated the incidence of testicular cancer in the Ronneby community cohort in Sweden, which had high exposures to PFAS (primarily PFOS and PFHxS, but also PFOA to some extent) from contaminated drinking-water (Section 2.1.13). Based on 14 observed cases of incident testicular cancer among residents in areas with contaminated drinking-water, elevated but imprecise risks were observed relative to regional rates (SIR, 1.28; 95% CI, 0.70-2.15) and relative to those in the Ronneby municipality among people who had never resided in contaminated water districts (HR, 1.38; 95% CI, 0.73-2.61). In analyses of people residing in districts with contaminated water during the later years when levels of PFAS contamination were higher (versus those never residing in districts with contaminated water), the hazard ratio was 1.46 (95% CI, 0.55-3.83). For those with a longer duration of residence in contaminated districts, the hazard ratio was 1.51 (95% CI, 0.56-4.03). The risk estimates for exposure in later years and for longer duration of residence were both based on analyses with 5 exposed cases.

[The strengths of this study included the complete ascertainment of community members in the cohort and follow-up through register-based linkages, and the high contrast in PFAS exposures within the cohort. However, the PFAS exposure profile and exposure assessment approach in this investigation – which characterized the potential for PFAS exposure overall rather than for specific PFAS and did not account for individual-level factors such as water consumption or use of bottled water or water filtration – limited the ability to isolate potential effects of PFOS and PFOA individually and also probably resulted in non-differential exposure misclassification, which typically might be expected to attenuate the reported risk estimates. The Working Group also noted that the findings from this study may not be directly comparable to those for PFOA in <u>Barry et al. (2013)</u>, given that PFAS exposure in this study was dominated by PFOS and PFHxS. PFOA serum levels (in samples collected 1–2 years after cessation of exposure) were lower in this population than in the C8 Science Panel study. Finally, with only 14 exposed cases of testicular cancer, the study had limited statistical power to evaluate associations with this malignancy.]

Purdue et al. (2023) conducted a nested case-control study of prediagnostic serum PFAS concentrations and risk of TGCT among US Air Force servicemen, using sera collected between 1988 and 2017 and stored in the DoD Serum Repository (Section 2.1.17). The study included 530 cases of TGCT among servicemen aged < 40 years and on active duty at diagnosis and 530 individually matched controls; of these, 187 cases and 187 matched controls also had measured PFAS concentrations from a second prediagnostic serum sample collected a median of 4 years after the first sample. In analyses conditioned on matching factors and adjusted for military grade and number of deployments, serum PFOA concentrations were not associated with TGCT risk on the basis of measurements in the first or only samples in the study population overall (fourth versus first quartile, OR, 0.8; 95% CI, 0.5–1.2; P for trend, 0.46) or the second samples from 187 case-control sets (OR, 0.7; 95% CI, 0.4–1.5; *P* for trend, 0.35); the results were similar after additionally adjusting for other PFAS. For PFOS, although no association with TGCT risk was seen in analyses of the

first or only samples (fourth versus first quartile, OR, 1.2; 95% CI, 0.7–2.0; P for trend, 0.64), the authors observed an exposure-response association with PFOS concentrations in the second samples (OR, 2.6; 95% CI, 1.1-6.4; P for trend, 0.02). After adjustment for other PFAS, the corresponding risk estimates for the fourth versus first quartiles of PFOS concentrations in the first/only and second serum samples were 1.8 (95% CI, 0.9–3.6; P for trend, 0.15) and 4.6 (95% CI, 1.4-15.1; *P* for trend, 0.009), respectively. Associations with seminomas (which are typically diagnosed at older ages than are nonseminomas) were observed for PFOS (e.g. above versus below median PFOS concentrations with adjustment for other PFAS: first or only sample, OR, 1.8; 95% CI, 1.0–3.3; second sample, OR, 2.8; 95% CI, 1.2-6.3).

[The Working Group considered this study to be highly informative because of a number of strengths, including its large sample size, the measurements of PFOA and PFOS in prediagnostic samples, the availability of repeated samples during potential etiologically relevant time periods from a subset of cases and controls, the ability to adjust for other PFAS, and the identification of cases in an age range during which most TGCTs are diagnosed. With respect to the timing of the repeated sample collections in this study, the Working Group noted that the first or only samples were often collected shortly after entering military service (a median of 0.3 and 0.4 years after enlistment for cases and controls, respectively) and probably reflected exposure patterns before military service. In contrast, the second samples (when available or selected) were typically collected after several years of service and may be more representative of PFAS levels during active duty, although PFOA and PFOS levels were still generally similar to those for comparably aged men in the US population overall. Participants with second samples also tended to be older and were more likely to be diagnosed with seminoma (which is consistent with

the typical age distributions of TGCT subtypes). As such, it is possible that the association with PFOS observed in the analyses of second samples may reflect patterns of risk related to exposure during military service, during different etiological time windows, and/or for seminomas in particular. A limitation of this study was the lack of information provided on associations with PFOA for histological subtypes of TGCT, which precluded an assessment of potential differences in the relation between PFOA and seminoma and nonseminoma tumours in this population.]

A single non-nested case-control study evaluated testicular cancer risk in relation to exposure to PFOA. A case-control study by Vieira et al. (2013) using data from cancer registries in Ohio and West Virginia evaluated various malignancies, including testicular cancer, among Mid-Ohio Valley residents with exposure to PFOA-contaminated drinking-water (Section 2.1.22). In analyses based on ecological exposure assignment for residence in areas of West Virginia and Ohio with contaminated water (all participants in the study), the investigators found no association with testicular cancer overall (adjusted OR, 1.0; 95% CI, 0.6-1.8; 18 cases in districts with contaminated water). An elevated OR was observed in the water district with the highest levels of PFOA exposure (Little Hocking: adjusted OR, 5.1; 95% CI, 1.6-15.6; 8 exposed cases). In analyses based on estimated serum PFOA concentrations among Ohio participants (not available for West Virginia participants), an elevated risk of testicular cancer was observed among those in the highest category of exposure compared with unexposed individuals, although the confidence interval was wide and included the null value (adjusted OR, 2.8; 95% CI, 0.8–9.2; 6 exposed cases).

[The Working Group noted that the cancer cases included in <u>Vieira et al. (2013)</u> overlapped with those in the study by <u>Barry et al. (2013)</u>, although the degree of overlap was unknown. It was also noted that in the analyses of more

detailed estimates of serum PFOA concentrations (only available for Ohio participants), the testicular cancer cases from Little Hocking may have been overrepresented in the highest exposure category (as this district had the highest levels of PFOA contamination). The degree of overlap was not reported, but the risk estimates from the two analyses may not be independent.]

In the Veneto region of Italy, an area with water contaminated with PFAS (overwhelmingly PFOA) from a local manufacturing plant, residents were invited to participate in a surveillance programme (participation rate, 63.5%) to address public concern about their exposure. Some of the participants lived in areas of the region with less-contaminated water. Among adults aged 14-39 years at recruitment, more than 18 000 people (9230 men) participating in this programme provided serum (Pitter et al., 2020). The median serum PFOA concentration was 44.4 ng/mL. An epidemiological investigation evaluated the frequency of orchiectomies in this region between 1997 and 2014 (Sistema Epidemiologico Regionale, 2016). Orchiectomy was used as a proxy for a diagnosis of testicular cancer (sensitivity and positive predictive values of 91.7% (95% CI, 88.0-95.4%) and 92.8% (95% CI, 89.3-96.2%), respectively, in this region). Orchiectomies were ascertained using information in hospital discharge records, which included address of residence and the main medical procedures from hospital stays and were completed for the purpose of reimbursement from the Italian national health system. SIRs for orchiectomy were estimated for the 21 municipalities separately, comparing the observed orchiectomies (n = 70 overall) versus expected numbers based on regional rates in 5-year age groups (Sistema Epidemiologico Regionale, 2016). The Working Group combined the serum PFOA data and orchiectomy rates by municipality (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications.iarc.who.int/636); as shown

in Fig. 2.1, a strong correlation (Spearman correlation, 0.57; P = 0.006) was observed between serum PFOA concentrations and rates of orchiectomy (standardized on age by 5-year age groups from 15–54 years to the overall regional rate) by municipality. The Working Group also conducted a Poisson regression of observed orchiectomies on median PFOA levels across 21 municipalities, using the log of expected events as an offset, and correcting for dispersion. The RR for each unit (ng/mL) increase of PFOA was 1.018 (95% CI, 1.006–1.031; P = 0.003).

[The Working Group considered the findings from these data from a region with high PFOA exposure to be informative because of the large number of serum measurements in the population, high PFOA levels, and good ascertainment of orchiectomy, which was shown to be an excellent surrogate for diagnosis of testicular cancer in this region. The ecological design and small numbers of orchiectomies by municipality and resulting imprecise SIR values were limitations.]

A review by <u>Bartell and Vieira (2021)</u> included a meta-analysis of associations between PFOA and testicular cancer from the Vieira et al. (2013) study and those reported in <u>Barry et al. (2013)</u>; they found a 3% increase in the risk of testicular cancer for each 10 ng/mL increase in estimated serum PFOA concentration (random-effects meta-RR, 1.03; 95% CI, 1.02-1.04). Results from a fixed-effects meta-analysis were similar. [The Working Group considered the informativeness of this meta-analysis to be reduced because of the unknown degree of overlap between the studies by Vieira et al. (2013) and Barry et al. (2013), and because the study by <u>Purdue et al. (2023)</u> was not available that time. If there were substantial overlap between the studies by Vieira et al. (2013) and <u>Barry et al. (2013)</u>, then the resulting meta-RRs could be overestimated. Another metaanalysis by Seyvedsalehi and Boffetta (2023) was not considered because it also did not include the Purdue et al. (2023) study and did not contribute any other information.]



Fig. 2.1 Serum PFOA concentrations and orchiectomy rates by municipality, in Veneto, Italy

PFOA, perfluorooctanoic acid; SIR, standardized incidence ratio.

^a Age-standardized to the regional Veneto population.

Pearson correlation coefficient, 0.58; P = 0.006.

Note that SIRs were plotted because they followed approximately normal distribution.

2.3.2 Prostate cancer

As summarized in Table 2.3 and below, there have been six investigations of prostate cancer incidence, mortality, or both in cohorts with occupational (Steenland and Woskie, 2012; Consonni et al., 2013; Raleigh et al., 2014; Steenland et al., 2015) or high environmental (Barry et al., 2013; Li et al., 2022a) exposure to PFOA and/or PFOS, and three investigations of serum PFOA and PFOS concentrations and risk of prostate cancer nested within general population cohorts (Eriksen et al., 2009; Rhee et al., 2023a; Winquist et al., 2023). Prostate cancer was also evaluated in two case-control studies (Vieira et al., 2013; Hardell et al., 2014). As described above and in Section 2.1.22, the study by Vieira et al. (2013) included individuals in the Mid-Ohio Valley with high exposure to PFOA, and the population overlapped with that of the cohort study by Barry et al. (2013). Hardell et al. (2014) conducted a population-based casecontrol study of serum PFAS concentrations and prostate cancer in a population with background levels of exposure in Sweden. Beyond these studies, an occupational cohort of 652 PFOSexposed employees at a fluorochemical-production facility evaluated prostate cancer (5 exposed cases) identified from health claims data (Olsen et al., 2004). [The Working Group considered the study by Olsen et al. (2004) to be uninformative, given the small number of exposed cases and the focus on prevalent (rather than incident) cancers as the outcome; as such, it was not included in <u>Table 2.3</u> or in the summary below.]

Raleigh et al. (2014) examined prostate cancer incidence and mortality among 3716 male workers at an APFO-production facility in Cottage Grove, Minnesota, USA (Section 2.1.1). They found no evidence of excess prostate cancer mortality on the basis of 24 deaths among PFOA-exposed workers (relative to Minnesota state rates: SMR, 0.83; 95% CI, 0.53–1.23), and no associations were observed in exposure– response analyses with unexposed workers as the reference group. Exposure–response analyses of prostate cancer incidence (on the basis of 188 cases among PFOA-exposed workers) were similarly null.

Steenland and Woskie (2012) evaluated prostate cancer mortality among 5791 workers exposed to PFOA at a fluoropolymer-production plant in Parkersburg, West Virginia, USA, in the Mid-Ohio Valley (Section 2.1.3). In analyses based on 21 deaths among PFOA-exposed workers, the authors found no evidence of excess prostate cancer mortality relative either to workers from other plants within the same company in the same region (SMR, 0.76; 95% CI, 0.47-1.16) or to the general US population (SMR, 0.72; 95% CI, 0.45-1.10), and no associations were observed in exposure-response analyses of estimated cumulative PFOA exposure. In a subsequent analysis of prostate cancer incidence that included 129 cases among a subset of 2955 male workers from this cohort, Steenland et al. (2015) observed elevated but imprecise estimates of prostate cancer rates among those with higher estimated cumulative PFOA exposure. Relative to those in the lowest quartile, the rate ratios in the second, third, and fourth quartiles were 1.81 (95% CI, 0.69-4.78), 2.45 (95% CI, 0.96–6.25), and 1.88 (95% CI, 0.72-4.88), respectively, and the *P* for trend was 0.11. Similar patterns were observed in analyses lagged by 10 years.

[The Working Group noted that the small numbers of deaths from prostate cancer in <u>Raleigh et al. (2014)</u> and <u>Steenland and Woskie</u> (2012) limited the ability to evaluate associations with prostate cancer mortality in both studies. Analyses of prostate cancer incidence by <u>Steenland et al. (2015)</u> and <u>Raleigh et al. (2014)</u> – which included 129 and 188 cases, respectively – had somewhat better statistical power.]

Eriksen et al. (2009) conducted a prospective case-cohort study nested within a cohort of older Danish adults with background levels of exposure to PFOA and PFOS. Samples were collected at cohort enrolment between 1993 and 1997, and prediagnostic plasma concentrations of PFOA and PFOS were measured for 713 cases of prostate cancer and 680 non-cases (Section 2.1.4). For PFOS, the authors observed about 30-40% increased risks of prostate cancer in the three upper quartiles compared with the lowest quartile (e.g. fourth versus first quartile, incidence rate ratio, IRR, 1.38; 95% CI, 0.99–1.93); in regression analyses in which plasma PFOS concentration was included as a continuous variable, the IRR corresponding to a 10 ng/mL increase in plasma PFOS levels was 1.05 (95% CI, 0.97–1.14). For PFOA, an exposure-response pattern was not apparent in analyses based on quartiles of measured levels; a modest increase in risk was observed when PFOA was modelled continuously (per 1 ng/mL increase, IRR, 1.03; 95% CI, 0.99–1.07).

The Working Group noted that the main strengths of this study were the large number of cases of prostate cancer and non-cases, a well-defined national cohort with complete ascertainment of incident cancer cases, data on a wide range of potential confounding factors, and a reasonable exposure contrast in a population with plasma PFOA and PFOS concentrations consistent with background levels of exposure. The measurements of PFOA and PFOS were conducted using samples collected a median of 7 years before cancer diagnosis; however, the authors did not report results for different time intervals between serum collection and diagnosis of prostate cancer to assess potential etiologically relevant periods of exposure. Also, the investigators observed a strong correlation between plasma PFOA and PFOS concentrations (Spearman correlation, 0.70) but did not evaluate prostate cancer risk in analyses with mutual adjustment for both chemicals, limiting the ability to assess the associations with each of these exposures independently.]

Barry et al. (2013) evaluated the risk of incident prostate cancer among community members and workers in the C8 Science Panel study (Section 2.1.5). A total of 446 validated prostate cancer cases were included in the analyses; the investigators found no evidence of an exposure–response association with estimated cumulative PFOA serum concentrations (unlagged analysis: adjusted HR corresponding to a unit increase in natural log-transformed levels, 0.99; 95% CI, 0.93–1.04). The corresponding risk estimate from an analysis with a 10-year lag period was similar (HR, 0.99; 95% CI, 0.94–1.05).

[The Working Group noted that the strengths of the study by <u>Barry et al. (2013)</u> included the detailed enumeration of the cohort, ascertainment or confirmation of cancer diagnoses, and relatively large numbers of incident prostate cancer cases.]

Consonni et al. (2013) evaluated prostate cancer mortality in a pooled international cohort of 4773 male workers who had ever been exposed to TFE (Section 2.1.6). Among those who had ever been exposed to APFO (n = 4205), the investigators observed reduced mortality from prostate cancer in analyses based on 3 observed deaths (using national reference rates: SMR, 0.24; 95% CI, 0.05–0.70).

[The Working Group noted that the small numbers of deaths from prostate cancer in this occupational cohort limited the ability to evaluate associations with prostate cancer mortality.]

Rhee et al. (2023a) conducted a nested case-control study of aggressive prostate cancer (750 cases, 750 matched controls) in relation to prediagnostic serum PFAS concentrations (including PFOA and PFOS) within the PLCO Cancer Screening Trial cohort (Section 2.1.11). Aggressive prostate cancer was defined as having stage III or IV disease, Gleason score \geq 8, or Gleason score 7 and death from prostate cancer. The study included cases with serum samples collected > 300 days before prostate cancer diagnosis (a median of 9 years from blood collection to diagnosis). Controls were selected from among participants who were alive and cancerfree as of the case diagnosis date and were individually matched to cases on age at baseline,

race or ethnicity, study centre, calendar year and study year of blood collection, and previous freeze-thaw cycles. For a subset of 60 controls, the investigators measured PFAS concentrations in sera collected at three time points up to 6 years apart. In overall logistic regression analyses of PFOA conditioned on matching factors, the investigators observed an inverse association with aggressive prostate cancer (per doubling in serum PFOA concentration, OR for a 1-unit increase in PFOA serum concentration on the log₂ scale, 0.82; 95% CI, 0.71–0.96). This association remained apparent in analyses adjusted for prostate cancer risk factors (BMI, smoking status, family history of prostate cancer, history of diabetes) and other PFAS – PFOS, PFHxS, PFNA, N-EtFOSAA, perfluorooctane sulfonamide (FOSA), 2-N-methyl-perfluorooctane sulfonamido acetate (N-MeFOSAA), and perfluoroheptanesulfonic acid (PFHpS). However, in analyses restricted to cases diagnosed > 3 years after blood collection, the association was less apparent (OR_{log2}, 0.90; 95% CI, 0.79-1.03). For PFOS, a modest inverse association was observed in analyses conditioned on matching factors but not adjusted for other covariates (OR_{log2} , 0.93; 95% CI, 0.83-1.05), whereas no association was observed after adjustment for prostate cancer risk factors and other PFAS (OR_{log2} , 0.99; 95% CI, 0.79-1.23). Analyses of serial samples collected up to 6 years apart from a subset of controls demonstrated good within-subject agreement in measurements of PFOA and PFOS over time, with overall intraclass correlation coefficients of 0.73 (95% CI, 0.62-0.81) and 0.85 (95% CI, 0.78–0.90) for PFOA and PFOS, respectively (<u>Rhee et al., 2023a</u>).

[The Working Group identified several strengths of this study that contributed to its informativeness, including its large sample size, measurements of serum PFOA and PFOS concentrations in prediagnostic samples, and the ability to adjust for measured concentrations of other PFAS and other potential confounding factors.]
In the Ronneby community cohort (Section 2.1.13), Li et al. (2022a) identified 181 cases of prostate cancer among residents in areas with PFAS-contaminated drinking-water. The investigators found that prostate cancer incidence rates among men who resided in exposed areas were similar to regional rates (SIR, 0.96; 95% CI, 0.82–1.11) and that risk for these men was lower than that for men in the Ronneby municipality who had never resided in contaminated water districts (HR, 0.83; 95% CI, 0.71-0.98). Inverse associations with prostate cancer risk were also observed among those residing in contaminated districts during the later years (HR, 0.76; 95% CI, 0.59-0.98; 67 exposed cases) and those with longer duration of residence in contaminated districts (HR, 0.72; 95% CI, 0.58-0.91; 86 exposed cases).

[The Working Group noted that the strengths of the study (Li et al., 2022a) included the detailed enumeration of the cohort, ascertainment and confirmation of cancer diagnoses, and relatively large numbers of incident prostate cancer cases. However, a limitation of the exposure assessment in the study by Li et al. (2022a) was the inability of their analysis to distinguish between the potential effects of PFOS and PFHxS.]

Winquist et al. (2023) evaluated prostate cancer in their case-cohort investigation in the ACS CPS-II LifeLink Cohort (see Section 2.1.21). In analyses with 1599 selected prostate cancer cases, they found no associations with prostate cancer risk for either PFOA or PFOS; when log₂-transformed levels were modelled continuously, the observed hazard ratios were 0.93 (95% CI, 0.79–1.08) and 1.00 (95% CI, 0.88–1.14) for PFOA and PFOS, respectively. Analyses based on quartiles of PFOA and PFOS were similarly null. [The Working Group noted several strengths of this study, including its large sample size, measurements of serum PFOA and PFOS concentrations in prediagnostic samples, and the ability to adjust for other potential confounding factors. Limitations included the relatively low exposure contrast in the study population.]

In addition to the cohort-based studies summarized above, the Working Group also reviewed two case–control studies of PFOA and/ or PFOS exposure and prostate cancer (<u>Vieira</u> et al., 2013; <u>Hardell et al., 2014</u>).

Vieira et al. (2013) evaluated prostate cancer in their case-control studies on multiple cancer types among Mid-Ohio Valley residents exposed to PFOA from contaminated drinking-water. In analyses with 434 PFOA-exposed cases in West Virginia and Ohio, they found no association with prostate cancer (adjusted OR, 0.9; 95% CI, 0.8–1.1); a modest increased risk was observed in the water district with the highest levels of PFOA exposure (adjusted OR, 1.4; 95% CI, 0.9-2.3; 36 exposed cases). Among Ohio participants, an elevated but imprecise OR was also observed among those in the highest category of PFOA exposure compared with unexposed individuals (adjusted OR, 1.5; 95% CI, 0.9-2.5; 31 exposed cases).

[The Working Group noted that the degree of overlap in the cancer cases included in the studies by <u>Vieira et al. (2013)</u> and <u>Barry et al.</u> (2013) was unknown, and as such the results of the two studies cannot necessarily be interpreted independently.]

Hardell et al. (2014) conducted a case–control study of serum PFAS concentrations (including PFOA and PFOS) and prostate cancer in Örebro County, Sweden. The study included 201 cases and 186 population-based controls, with blood samples collected in the period 2007–2011. Samples were collected from cases after diagnosis of prostate cancer but before initiating treatment. The investigators found no associations with PFOA or PFOS in overall analyses and among those with markers indicative of more advanced disease (Gleason score 7–10, or prostate-specific antigen, PSA \geq 11). However, for participants who reported a family history of disease (prostate cancer in a first-degree relative) and had serum concentrations above the median (24 cases), increased risks were observed for both PFOA (OR, 2.6; 95% CI, 1.2–6.0) and PFOS (OR, 2.7; 95% CI, 1.04–6.8) relative to participants with no family history and serum concentrations at or below the median for the respective chemicals.

[The Working Group noted several limitations of this study. Samples were collected from cases after diagnosis of prostate cancer, and it was possible that the measurements of serum PFOA and PFOS concentrations may have been influenced by disease status, which could have resulted in differential exposure misclassification between cases and controls, possibly biasing risk estimates either towards or away from the null value. It was also possible that PFAS concentrations at the time of diagnosis may not have reflected exposure levels during an etiologically relevant time period. However, the sample collections occurred before cases were treated, so any potential treatment-related effects on PFAS levels were not a concern in this study. Other limitations included the non-participation of some selected cases and matched controls (response rates were 79% and 54%, respectively), the lack of adjustment for other PFAS in the statistical analyses, and the relatively small sample size, particularly for analyses stratified by family history of prostate cancer.]

2.4 Cancers of the breast and thyroid gland

2.4.1 Cancer of the breast

See <u>Table 2.4</u>.

(a) Cohort and nested case-control studies

There were 12 cohort or nested case-control studies that contributed evidence on PFOA and PFOS exposure and the risk of breast cancer in women. Three of these studies were occupational cohorts (Alexander et al., 2003; Steenland and Woskie, 2012; Raleigh et al., 2014). [The Working

Group noted that these occupational cohorts included few women and thus had extremely limited power with which to consider associations with breast cancer, resulting in limited inference from those studies.] Two studies (Barry et al., 2013; Li et al., 2022a) considered how environmental exposure to high levels of PFAS in the contaminated environment from nearby industrial or occupational sources was related to breast cancer risk using modelled exposure assessment. The remaining studies included two nested casecontrol studies focused on PFAS measurements during pregnancy in relation to subsequent risk of breast cancer in the individual (Ghisari et al., 2017) and the offspring (Cohn et al., 2020), as well as blood measurements in nested substudies within larger population-based cohorts (Hurley et al., 2018; Mancini et al., 2020a; Feng et al., 2022; Chang et al., 2023; Winquist et al., 2023).

Raleigh et al. (2014) evaluated cancer incidence and mortality in an occupational cohort that included 4668 workers (of whom 952 were women) who had worked for ≥ 1 year and were exposed to APFO at a factory in Cottage Grove, Minneapolis, USA, between 1947 and 2002, and a comparison group of 4359 employees (of whom 526 were women) who were unexposed workers at a factory in Saint Paul (see Section 2.1.1). Individual inhalation exposure was estimated using a JEM, and information on cancer incidence and mortality was obtained via linkages to registries, with follow-up until 2008. Women represented only 21% of the workers at the Cottage Grove facility and 12% at the Saint Paul facility. There were 26 deaths (11 exposed, 15 unexposed) from breast cancer (25 among women) and 62 cases (34 exposed, 28 unexposed) of incident breast cancer. There was little evidence to suggest that increased APFO exposure was associated with a higher SMR for breast cancer relative to population mortality rates or with an increased hazard ratio for incident breast cancer cases, although this was based on few cases. [The Working Group noted that very

				-				
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
Raleigh et al.	9027 (952 exposed	Breast, mortality	Exposed to APFO (S	MR, MN re	ferent):	Age, sex, calendar	Exposure assessment	
<u>(2014)</u> MN, USA	women, 526 reference women); Cottage		Unexposed (Saint Paul plant)	15	1.39 (0.78–2.29)	period	<i>critique</i> : See <u>Table 2.1</u> .	
Enrolment, 1947–2002/	nt, Grove (MN) PFOA 2/ cohort; workers	Grove (MN) PFOA cohort; workers		Exposed (Cottage Grove plant)	11	0.82 (0.41–1.47)		Other strengths: See Table 2.1. Unlikely
follow-up, 1947–2008	employed for ≥ 1 yr in 1947–2002 at an	Breast, mortality	Estimated cumulative airborne APFO exposure quartile (SMR, MN referent):				Reference population	
(incidence)	Grove; <i>n</i> = 4668); reference workers		1st quartile (< 2.6 × 10 ⁻⁵ μg/ m ³ -vears)	5	0.80 (0.26–1.86)		sharing similar socioeconomic characteristics.	
Conort	to APFO employed at a tape and abrasives production facility		2nd quartile (2.6×10^{-5} to < 1.4×10^{-4} µg/ m ³ -vears)	3	0.88 (0.18–2.56)		Table 2.1. Lacking data on workers that left Minnesota or Wisconsin Small	
	suburban geographical area and managed by the same company (Saint Paul; $n = 4359$)		3rd quartile (1.4×10^{-4} to < $7.3 \times 10^{-4} \mu g/m^{3}$ -years)	2	0.73 (0.09–2.62)		numbers especially for women (12% of the cohort).	
	Exposure assessment method: see <u>Table 2.1</u>		4th quartile ($\geq 7.3 \times 10^{-4} \mu g/m^3$ -years)	1	1.02 (0.03-5.69)			

Table 2.4 Epidemiological studies on exposure to PFOA or PFOS and cancers of the breast and thyroid gland

$\overline{>}$
R
\cap
\leq
0
ž
5
õ
R
$\widetilde{>}$
Ρ
Ţ
^v
I
ŭĩ

Table 2.4 (continued)

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Raleigh et al. (2014) MN, USA Enrolment, 1947–2002/ follow-up, 1947–2008 (mortality), 1988–2008 (incidence) Cohort (cont.)		Breast, incidence	Estimated cumulativ quartile (HR): Unexposed (Saint Paul plant) 1st quartile (< $2.9 \times 10^{-5} \mu g/m^3$ -years) 2nd quartile ($2.9 \times 10^{-5} to$ < $1.5 \times 10^{-4} \mu g/m^3$ -years) 3rd quartile ($1.5 \times 10^{-4} to$ < $7.9 \times 10^{-4} \mu g/m^3$ -years) 4th cumutile	ve airborne 28 8 8 14	APFO exposure 1 0.36 (0.16–0.79) 0.65 (0.29–1.42) 1.47 (0.77–2.80)	Age, [sex], year of birth	
			4th quartile ($\geq 7.9 \times 10^{-4} \mu\text{g}/\text{m}^3$ -years)	4	0.85 (0.29–2.46)		

Table 2.4 (co	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Lundin et al. (2009) MN, USA Enrolment, 1947–1997/ follow-up, 1947–2002 (mortality) Cohort	3993 employees; Cottage Grove (MN) PFOA cohort; workers employed at a PFOA- production plant for ≥ 365 days before 31 December 1997 Exposure assessment method: see <u>Table 2.1</u>	Thyroid, mortality	Employed in APFO- referent): Never Ever probable/ never definite Ever definite	exposed job 1 0	0 (SMR, MN 2.16 (0.05–12.00) 0 (0.00–8.45) 0 (0.00–42.96)	Age, sex, calendar period	Exposure assessment critique: See <u>Table 2.1</u> . Other strengths: Occupational cohort with relatively high exposures. Other limitations: Small cohort with limited number of deaths; potential healthy-worker effect due to external comparison of rates with general population; limited information on
Alexander et al. (2003) Decatur (AL), USA Enrolment, 1961–1997/ follow-up, 1961–1998 (mortality) Cohort	2083 (241 exposed and 112 unexposed women); Decatur (AL) PFOS cohort; production workers (men, 83%) who worked ≥ 365 days in a plant producing speciality films and fluorochemicals, one of the main ones being perfluorooctane- sulfonyl (POSF). Exposure assessment method: See <u>Table 2.1</u>	Breast, mortality	PFOS exposure grou All jobs Only non-exposed Ever low, never high Ever high	up (SMR, AI 2 2 0 0	2 referent): 1.57 (0.19–5.66) 5.11 (0.62–18.45) 0 0	Sex, age, calendar period	<i>Exposure assessment</i> <i>critique</i> : See <u>Table 2.1</u> . <i>Other strengths</i> : See <u>Table 2.1</u> <i>Other limitations</i> : Occupational cohort with few breast cancer deaths (n = 2); outcome assessment limited to mortality; mostly men (83%).

Table 2.4 (c	ontinued)							
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
Steenland and	5791 (women,	Breast, mortality	PFOA-exposed work	kers (SMR):		Age, sex, calendar	Exposure assessment	
Parkersburg (WV), USA Enrolment,	19%); Parkersburg (WV), polymer- production PFOA cohort; workers (men,	(WV), polymer- production PFOA cohort; workers (men,		Other workers referent (same region and company)	4	0.65 (0.13–1.90)	period	See <u>Table 2.1</u> . Other strengths: See <u>Table 2.1</u> .
1948-2002/	81%) at a polymer-	Breast, mortality	US referent	4	0.79 (0.21–2.02)		Other limitations:	
1952–2008 Cohort	manufacturing facility who had potential exposure to fluoropolymers and had sufficiently detailed work histories		Cumulative serum exposure, no lag (SMR, other workers referent, same region and company), women only:				Occupational cohort with few breast cancer deaths $(n = 4)$;	
			1st quartile (0 to < 904 ppm-years)	2	1.49 (0.18–5.39)		outcome assessment limited to mortality; mostly men (81%)	
	Exposure assessment method: see <u>Table 2.1</u>		2nd quartile (904 to < 1520 ppm-years)	0	0.00 (0.00-3.56)			
			3rd quartile (1520 to < 2700 ppm-years)	1	0.87 (0.02-4.83)			
			4th quartile (≥ 2700 ppm-years)	0	0.00 (0.00-3.42)			

Reference, location, euroInent/ follow-up period, study designPopulation size, dasessment method mortalityCancer type (histopathology), incidence or mortalityExposure category reledExposure dategory cases or deathsRisk estimate (95% CI)Covariates controlledCommentsIconard etal. (2008)6027 workers; Parkersburg (WV), polymer-production occupational PFOA cohort; workers (men, 81%) at a polymer, facility who had mortalityThyroid, mortalityWorkers in the Parkersburg (WV), polymer- production plant (SMR); Referent UV and facility who had potential exposure to fluow-up, detailed work histories; most recent relion and coarer sites Exposure assessment region and comparabilityThyroid, mortalityWorkers in the Parkersburg (WV), polymer- production plant (SMR); Referent UV 3 [2.856 (0.589–8.347)] (6.286 (1.297–18.369)]Sex, age, calendar portical exposure comparability with relatively high or some cancer sites Exposure assessment method: recordsThyroid, mortalityWorkers (men, social cohort worker sign and comparability with respect to sociace-connic factors and health behaviours. Other immeths; Other immeths; 	Table 2.4 (co	ontinued)						
Leonard et al. (2008)6027 workers: Parkersburg (WV), polymer-production polymer-production polymer-production cohort: workers (men, dialacturing follow-up, manufacturing follow-up, tofluoropolymers and had sufficiently detailed work histories; most recent follow-up follow-up follow-up follow-up follow-up follow-up, manufacturing tofluoropolymers and had sufficiently detailed work histories; most recent follow-up follow-up f	Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
PFOA.	Leonard et al. (2008) Parkersburg (WV), USA Enrolment, 1948–2002/ follow-up, 1948–2002 (mortality) Cohort	6027 workers; Parkersburg (WV), polymer-production occupational PFOA cohort; workers (men, 81%) at a polymer- manufacturing facility who had potential exposure to fluoropolymers and had sufficiently detailed work histories; most recent follow-up for some cancer sites Exposure assessment method: records	Thyroid, mortality	Workers in the Park production plant (SP Referent US population Referent WV population Referent other workers (same region and company)	ersburg (W MR): 3 3 3	V), polymer- [3.120 (0.644–9.119)] [2.856 (0.589–8.347)] [6.286 (1.297–18.369)]	Sex, age, calendar period	Exposure assessment critique: See Table 2.1. Other strengths: Occupational cohort with relatively high exposures; complete cohort ascertainment and follow-up; use of local reference groups increased comparability with respect to socioeconomic factors and health behaviours. Other limitations: Small numbers. Other comments: The Parkersburg (WV, USA), facility manufactured a broad range of commercial products including fluoropolymers, nylon filaments, and acrylic polymers; all study participants, regardless of work area, had detectable levels of serum PFOA.

Table 2.4 (co	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Barry et al. (2013) Mid-Ohio Valley (OH and WV) Enrolment.	cy et al.32 254 (women, 17 360; community, 16 602; ohioBreast, i.Ohioand occupational, ey (OH and 0 758); C8 ScienceBreast, i)Panel Study included olment, ustBreast, i.0Panel Study included olment, ustBreast, i.5-AugustProject who lived, school for ≥ 1 yr idence)Breast, i.6/follow-up, ortworked, or attended school for ≥ 1 yr idenceThyroid, incidence.0between 1950 and contaminated-water district in the vicinity of a chemical plant (Parkersburg (WV), polymer production) using PFOA in manufacturing, as well as a subset of those from the original Parkersburg (WV),Image: Comparison of the compar	Breast, incidence	Estimated cumulativ (ng/mL), no lag (HR Continuous (per unit on natural log scale)	ve PFOA ser): 559	rum concentration 0.94 (0.89–1.00)	Age, time-varying smoking, time- varying alcohol <i>Exposure asse</i> critique: See Table 2.1. Other strength Wide range of PFOA exposu levels; availab detailed infor on potential confounding relatively high participation validation of of diagnoses thr medical chart Other limitati Mostly retros with relatively validated case prospective and	Exposure assessment critique: See <u>Table 2.1</u> . Other strengths: Wide range of PFOA exposure
August 2005-August 2006/follow-up, 1952 to 2011 (incidence) Cohort		Thyroid,	Estimated cumulativ (ng/mL), 10-yr lag (I Continuous (per unit on natural log scale) Estimated cumulativ (ng/mL), no lag (HR	ve PFOA ser HR): 559 ve PFOA ser):	um concentration 0.93 (0.88–0.99) rum concentration		levels; availability of detailed information on potential confounding factors; relatively high participation rates;
		contaminated-water district in the vicinity of a chemical plant (Parkersburg (WV), polymer production) using PFOA in manufacturing, as well as a subset of those from the original Parkersburg (WV),	(ng/nic), no lag (like 1st quartile 2nd quartile 3rd quartile 4th quartile Continuous (per unit on natural log scale) Trend-test <i>P</i> -value, (Excluding perconti). NR NR NR 86).25 me before e	1 1.54 (0.77-3.12) 1.48 (0.74-2.93) 1.73 (0.85-3.54) 1.10 (0.95-1.26)		validation of cancer diagnoses through medical chart review. <i>Other limitations</i> : Mostly retrospective with relatively few validated cases for prospective analyses.
	polymer-production occupational cohort who worked at the plant between 1948 and 2002 Exposure assessment method: see Table 2.1	incidence	known to have lived water districts: estin concentration (ng/m Continuous (per unit on natural log scale)	or worked nated cumu L), no lag (NR	in the contaminated- lative PFOA serum HR): 1.06 (0.92–1.23)		

Table 2.4 (c	Table 2.4 (continued)								
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Barry et al. (2013) Mid-Ohio Valley (OH and	Thyroid,	Estimated cumulativ	ve PFOA ser	um concentration	Age, time-varying				
	incidence	(ng/mL), 10-yr lag (l	HR):		smoking, time-				
		Ist quartile	NR		consumption sex				
WV)			2nd quartile	NK	2.06 (0.93-4.56)	education, birth			
Enrolment,			Ath quartile	NR	2.02(0.90-4.52) 1.51(0.67-3.39)	year (5-yr calendar			
August 2005-August 2006/follow-up, 1952 to 2011	st -up,		Continuous (per unit on natural log scale)	86	1.04 (0.89–1.20)	intervals)			
(incidence)			Trend-test P-value, ().57					
Cohort (cont.)		Thyroid, incidence	Excluding person-ti- known to have lived water districts: estin concentration (ng/m	me before e or worked i nated cumu nL), 10-yr la					
			Continuous (per unit on natural log scale)	NR	1.02 (0.87–1.19)				
		Thyroid, incidence	Community residen serum concentration	ts: estimate n (ng/mL), n	d cumulative PFOA o lag (HR):				
			1st quartile	NR	1				
			2nd quartile	NR	1.54 (0.73-3.26)				
			3rd quartile	NR	1.71 (0.81–3.59)				
			4th quartile	NR	1.40 (0.66–2.97)				
			Continuous (per unit on natural log scale)	78	1.04 (0.89–1.23)				
			Trend-test <i>P</i> -value, ().46					

Table 2.4 (c	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Barry et al.		Thyroid,	Community residen	ts: estimate	d cumulative PFOA	Age, time-varying	
<u>(2013)</u> Mid Ohio		incidence	serum concentration	n (ng/mL), I	0-yr lag (HR):	smoking, time-	
Valley (OH and			Ist quartile	NR		consumption sex	
WV)			2nd quartile	NK	2.09(0.91 - 4.82)	education, birth	
Enrolment,			3rd quartile	NK	1.92(0.82 - 4.50)	year (5-yr calendar	
August			4th quartile	IN K 79	1.42(0.60-5.57) 1.00(0.84, 1.20)	intervals)	
2005-August 2006/follow-up, 1952 to 2011			unit on natural log scale)	78	1.00 (0.84–1.20)		
(incidence)			Trend-test P-value, ().56			
Cohort		Thyroid,	Workers: estimated	cumulative	PFOA serum		
(cont.)		incidence	concentration (ng/m	nL), no lag (HR):		
			1st quartile	NR	1		
			2nd quartile	NR	4.64 (0.42–50.8)		
			3rd quartile	NR	9.70 (0.67–141.2)		
			4th quartile	NR	14.72 (0.85–253.9)		
			Continuous (per unit on natural log scale)	8	1.93 (1.00–3.71)		
			Trend-test <i>P</i> -value, (0.04			
		Thyroid, incidence	Workers: estimated concentration (ng/m	cumulative nL), 10-yr la	PFOA serum g (HR):		
			1st quartile	NR	1		
			2nd quartile	NR	1.65 (0.09-31.5)		
			3rd quartile	NR	4.52 (0.10-198.4)		
			4th quartile	NR	5.85 (0.13-257.1)		
			Continuous (per unit on natural log scale)	8	1.12 (0.61–2.05)		
			Trend-test <i>P</i> -value, (0.01			

Table 2.4 (c	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Ghisari et al. (2017) Denmark Enrolment, 1996–2002/ follow-up, through 2010 Nested case– control	Nested within the Danish National Birth Cohort (see <u>Table 2.1</u>) Cases: 178; nulliparous women at the time of blood draw during pregnancy followed for breast cancer, selected from ~100 000 pregnant women Controls: 233; nulliparous women at the time of blood draw during pregnancy frequency-matched on age Exposure assessment method: see <u>Table 2.1</u>	Breast (pre- menopausal), incidence Breast (pre- menopausal), incidence Breast (pre- menopausal), incidence Breast (pre- menopausal), incidence	Serum PFOA (ng/mi Continuous (per unit on natural log scale) Serum PFOS (ng/mI Continuous (per unit on natural log scale) CYP19 (C > T) geno transformed PFOA, CC CT TT CYP19 (C > T) geno transformed PFOS, m CC CT	L) (RR): 158 L) (RR): 158 type (RR pe ng/mL): 35 59 34 type (RR pe ng/mL): 35 59 34 type (RR pe ng/mL): 35 59 34	1.17 (0.63–2.17) 1.15 (0.64–2.08) er unit natural log 7.24 (1.00–52) 0.79 (0.26–2.38) 0.55 (0.14–2.24) er unit natural log 6.42 (1.08–38.3) 1.16 (0.44–3.10) 0.45 (0.10–1.97)	Age at blood draw, BMI before pregnancy, total gravidities, oral contraceptive use, age at menarche, smoking status during pregnancy, alcohol intake during pregnancy, physical activity, education	Exposure assessment critique: See <u>Table 2.1</u> . Other strengths: See <u>Table 2.1</u> . Other limitations See <u>Table 2.1</u> .

Table 2.4 (c	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Hurley et al.	Nested within the	Breast, incidence	Serum PFOA (ng/m	L) (OR):		Age at baseline,	Exposure assessment
<u>(2018)</u>	California Teachers		1st tertile	331	1	race/ethnicity,	<i>critique</i> : See <u>Table 2.1</u> . <i>Other strengths</i> : See <u>Table 2.1</u> . ^a This is the standard covariate set used
CA, USA	Study (See <u>Table 2.1</u>)		2nd tertile	298	0.901 (0.705-1.152)	region of residence,	
Enrolment,	Cases: 902; California		3rd tertile	273	0.925 (0.715–1.197)	date of blood draw,	
follow-up, 1 January	public-school teachers and other professionals		Continuous (per unit log ₁₀ scale)	902	0.733 (0.496–1.081)	(date of blood draw) ² , season of blood draw, total	
2006 to	with a diagnosis		Trend-test <i>P</i> -value, 0.54			smoking pack-years,	for all analyses.
1 August 2014	of invasive breast	Breast, incidence	Serum PFOS (ng/m)	L) (OR):		BMI, family history	Additional covariates
(incidence)	cancer, age < 80 yr		1st tertile	318	1	of breast cancer,	are indicated as
Nested case-	at diagnosis with no		2nd tertile	297	0.883 (0.691–1.129)	age at first full-term pregnancy, pork consumption ^a , and menopausal status at	required.
control	cancer who provided		3rd tertile	287	0.898 (0.695-1.161)		
	a blood specimen		Continuous (per unit log ₁₀ scale)	902	0.934 (0.683–1.277)		
	and answered a		Trend-test P-value,	0.41		blood draw	
	were continuous	Breast, post-	Serum PFOA (ng/mL) (OR):			Standard covariates ^a	
	residents of CA;	menopausal	1st tertile	306	1		
	participation rate, 65%	at blood draw,	2nd tertile	287	0.889 (0.689-1.147)		
	Controls: 858;	incidence	3rd tertile	266	0.912 (0.699-1.189)		
	from probability		Continuous (per unit log ₁₀ scale)	859	0.715 (0.476-1.073)		
	cohort members.		Trend-test P-value,	0.49			
	frequency-matched	Breast (post-	Serum PFOS (ng/m)	L) (OR):		Standard covariates ^a	
	on age, race/ethnicity, and residence;	menopausal at	1st tertile	293	1		
		blood draw),	2nd tertile	284	0.843 (0.653-1.088)		
	participation rate, 55%	incidence	3rd tertile	282	0.860 (0.661-1.118)		
	method: see <u>Table 2.1</u>		Continuous (per unit log ₁₀ scale)	859	0.885 (0.641–1.223)		
			Trend-test P-value,	0.26			

nments

Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
<u>Hurley et al.</u>		Breast (ER- and	Serum PFOA (ng/m	L) (OR):		Age at baseline,				
<u>(2018)</u>		PR–), incidence	1st tertile	43	1	race/ethnicity,				
CA, USA	SA		2nd tertile	35	0.846 (0.510-1.403)	region of residence,				
Enrolment,			3rd tertile	29	0.792 (0.460-1.365)	date of blood draw,				
follow-up,			Continuous (per unit on log ₁₀ scale)	107	0.528 (0.239–1.165)	draw) ² , season of				
2006 to			Trend-test P-value, ().39		activity				
1 August 2014		Breast (ER- and	Serum PFOS (ng/ml	L) (OR):		1				
(incidence)		PR–), incidence	1st tertile	47	1					
Nested case-			2nd tertile	32	0.628 (0.378-1.041)					
control			3rd tertile	28	0.615 (0.357-1.059)					
(cont.)			Continuous (per unit on log ₁₀ scale)	107	0.573 (0.323-1.016)					
			Trend-test P-value, (0.06						

Table 2.4 (c	Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments				
<u>Mancini et al.</u> (2020a) France	Nested within E3N cohort (see <u>Table 2.1</u>) Cases: 194; incident	Breast (post- menopausal), incidence	Serum PFOA (OR): 1st quartile (1.3-4.8 ng/mL)	85	1	Age at blood draw, BMI at blood draw, menopausal status	<i>Exposure assessment</i> <i>critique</i> : See <u>Table 2.1</u> .				
Enrolment, 1990/follow-up,	nt, postmenopausal ow-up, breast cancers among	opausal ncers among	2nd quartile (4.8–6.8 ng/mL)	118	1.69 (0.89–3.21)	at blood draw, year of blood draw, total	Other strengths: Adjustment for				
through 2013 (incidence)	women with serum (≥ 3 aliquots) collected		3rd quartile (6.8–8.8 ng/mL)	91	0.88 (0.43-1.80)	serum lipids, BMI, smoking status, physical activity	breast cancer risk factors; pathology				
Nested case- before diagnosis, a control completed dietary questionnaire in 1993		4th quartile (8.8–21.4 ng/mL)	94	0.92 (0.43-1.98)	(MET-h/week), education level,	cases. Other limitations:					
	and randomly selected from 240 eligible	Breast (post-	Trend-test <i>P</i> -value, (Serum PFOS (OR):).43		history of benign breast disease,	Limited statistical power, especially to				
	breast cancers Controls: 194; density-	menopausal), incidence	1st quartile (5.8–13.6 ng/mL)	80	1	family history of breast cancer,	explore differences by subtype.				
	sampled at time of case occurrence		2nd quartile (13.6–17.3 ng/mL)	109	1.94 (1.00–3.78)	full-term pregnancy,					
	age within 2 yr, menopausal status at		3rd quartile (17.3–22.5 ng/mL)	99	2.03 (1.02-4.04)	duration, age at menarche, age at					
	blood collection, BMI at blood collection,		4th quartile (22.5–85.3 ng/mL)	100	1.72 (0.88–3.36)	menopause, use of oral contraceptives,					
	and year of blood collection		Trend-test <i>P</i> -value, ().25		current use of MHT, adherence					
	Exposure assessment method: see <u>Table 2.1</u>					to Western diet, adherence to Mediterranean diet					

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Mancini et al.		Breast (post-	Serum PFOA (OR):			Age at blood draw,			
(2020a) France		menopausal, ER+), incidence	1st quartile (1.3–4.8 ng/mL)	NR	1	BMI at blood draw, menopausal status			
Enrolment, 1990/follow-up,			2nd quartile (4.8–6.8 ng/mL)	NR	1.72 (0.88–3.36)	at blood draw, year of blood draw, total			
through 2013 (incidence)			3rd quartile (6.8–8.8 ng/mL)	NR	1.34 (0.66–2.73)	serum lipids, BMI, smoking status,			
Nested case- control (cont.)			4th quartile (8.8–21.4 ng/mL)	NR	1.42 (0.68–2.95)	(MET-h/week),			
		Proact (post	Trend-test <i>P</i> -value, ().64		history of benign			
		menopausal, ER+), incidence	1st quartile (5.8–13.6 ng/mL)	NR	1	family history of breast cancer, parity/age at first full-term pregnancy, total breastfeeding duration, age at			
			2nd quartile (13.6–17.3 ng/mL)	NR	1.85 (0.90–3.82)				
			3rd quartile (17.3–22.5 ng/mL)	NR	2.22 (1.05-4.69)				
		4th quartile (22.5–85.3 ng/mL)	NR	2.33 (1.11-4.90)	menarche, age at menopause, use of oral contraceptives				
		Breast (post-	Trend-test <i>P</i> -value, (Serum PFOA (OR):	0.04		current use of MHT, adherence			
		menopausal, ER–), incidence	1st quartile (1.3–4.8 ng/mL)	NR	1	to Western diet, adherence to			
			2nd quartile (4.8–6.8 ng/mL)	NR	7.73 (1.46–41.08)	Mediterranean diet			
			3rd quartile (6.8–8.8 ng/mL)	NR	3.18 (0.55–18.47)				
			4th quartile (8.8–21.4 ng/mL)	NR	3.98 (0.67–23.52)				
			Trend-test P-value, ().59					

Table 2.4 (co	Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
Mancini et al.		Breast (post-	Serum PFOS (OR):			Age at blood draw,				
(2020a) France	menopausal, ER–), incidence	1st quartile (5.8–13.6 ng/mL)	NR	1	BMI at blood draw, menopausal status					
Enrolment, 1990/follow-up,	nrolment, 990/follow-up, 19912		2nd quartile (13.6–17.3 ng/mL)	NR	15.40 (1.84–129.19)	at blood draw, year of blood draw, total				
through 2013 (incidence) Nested case– control (cont.)		3rd quartile (17.3–22.5 ng/mL)	NR	4.74 (0.45-49.62)	serum lipids, BMI, smoking status,					
		4th quartile (22.5–85.3 ng/mL)	NR	7.07 (0.73-68.03)	physical activity (MET-h/week),					
			Trend-test P-value, ().72		history of benign				
		Breast (post-	Serum PFOA (OR):			breast disease, family history of breast cancer, parity/age at first full-term pregnancy, total breastfeeding duration, age at menarche, age at menopause, use of oral contracentives				
		menopausal), PR+), incidence	1st quartile (1.3–4.8 ng/mL)	NR	1					
			2nd quartile (4.8–6.8 ng/mL)	NR	1.40 (0.67–2.93)					
			3rd quartile (6.8–8.8 ng/mL)	NR	1.28 (0.59–2.77)					
			4th quartile (8.8–21.4 ng/mL)	NR	1.54 (0.70-3.69)					
			Trend-test P-value, ().37		current use of				
		Breast (post-	Serum PFOS (OR):			MHT, adherence				
		menopausal, PR+), incidence	1st quartile (5.8–13.6 ng/mL)	NR	1	to Western diet, adherence to Mediterranean diet				
			2nd quartile (13.6–17.3 ng/mL)	NR	1.84 (0.82–4.14)					
			3rd quartile (17.3–22.5 ng/mL)	NR	2.47 (1.07–5.65)					
			4th quartile (22.5–85.3 ng/mL)	NR	2.76 (1.21-6.30)					
			Trend-test P-value, (0.02						

Table 2.4 (co	able 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments				
<u>Mancini et al.</u>		Breast (post-	Serum PFOA (OR):			Age at blood draw,					
<u>(2020a)</u> France		menopausal, PR–), incidence	1st quartile (1.3–4.8 ng/mL)	NR	1	BMI at blood draw, menopausal status					
Enrolment, 1990/follow-up,			2nd quartile (4.8–6.8 ng/mL)	NR	3.44 (1.30-9.10)	at blood draw, year of blood draw, total					
through 2013 (incidence) Nested case- control			3rd quartile (6.8–8.8 ng/mL)	NR	1.80 (0.62–5.19)	serum lipids, BMI, smoking status, physical activity (MET-h/week), education level,					
			4th quartile (8.8–21.4 ng/mL)	NR	1.69 (0.56–3.12)						
(cont.)	Breast	-	Trend-test <i>P</i> -value, 0.90			history of benign					
		Breast (post-	Serum PFOS (OR)	ND		breast disease, family history of breast cancer, parity/age at first full-term pregnancy, total breastfeeding duration, age at menarche, age at menopause, use of					
		PR–), incidence	1st quartile (5.8–13.6 ng/mL)	NK	1						
			2nd quartile (13.6–17.3 ng/mL)	NR	3.47 (1.29–9.15)						
			3rd quartile (17.3–22.5 ng/mL)	NR	1.82 (0.61–5.45)						
			4th quartile (22.5–85.3 ng/mL)	NR	1.71 (0.57–5.10)						
			Trend-test <i>P</i> -value, ().93		oral contraceptives, current use of MHT, adherence to Western diet, adherence to Mediterranean diet					

Table 2.4 (c	Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments				
designChang et al. (2023)Nested within PLCO cohortUSACases: 621; all incident invasive breast cancer1993–2001/ follow-up, throughto and including through November 2013 women who were postmenopausal and not using hormone therapy at baseline (unless they were hormone receptor- negative cases) Controls: 621; controls were selected using incidence density sampling, all were postmenopausal, still alive and cancer- free at the time of case diagnosis with matching by age at baseline MHT use Exposure assessment method: see Table 2.1	Nested within PLCO cohort Cases: 621; all incident invasive breast cancer cases diagnosed up to and including November 2013 among women who were postmenopausal and not using hormone therapy at baseline (unless they were hormone receptor- negative cases) Controls: 621; controls were selected using incidence density sampling, all were postmenopausal, still alive and cancer- free at the time of case diagnosis with matching by age at harding data charace	Breast (post- menopausal), incidence Breast (post-	Serum PFOA (OR): 1st quartile 2nd quartile 3rd quartile 4th quartile Trend-test <i>P</i> -value, (147 148 162 164 0.83	1 0.91 (0.64–1.30) 1.07 (0.73–1.55) 1.01 (0.66–1.55)	Age at baseline, date of blood draw, MHT use at baseline, age at blood draw, study centre, race/ ethnicity, education, age at menarche, age at first live birth and number of births, age at menopause, duration of MHT use, first degree family history of female breast cancer, personal history of benign breast disease, BMI, smoking status, vigorous physical activity ^a , PFOS (natural log transformed) Standard covariates ^a	Exposure assessment critique: See Table 2.1. Other strengths: See Table 2.1. Limitations: See Table 2.1. a This is the standard covariate set used for all analyses. Additional covariates are indicated as required.				
	baseline, date of blood draw and baseline MHT use Exposure assessment method: see <u>Table 2.1</u>	d menopausal), incidence	1st quartile 2nd quartile 3rd quartile 4th quartile Trend-test P-value (145 158 167 151 58	1 1.21 (0.84–1.74) 1.39 (0.96–1.99) 1.17 (0.77–1.79)	and PFOA (natural log transformed)					
	Breast (post- menopausal, ER+/PR+), incidence	Serum PFOA (OR): 1st quartile 2nd quartile 3rd quartile 4th quartile Trend-test <i>P</i> -value, 0	NR NR NR NR 0.41	1 1.14 (0.66–1.97) 0.99 (0.55–1.80) 0.81 (0.40–1.62)	Standard covariates ^a and PFOS (natural log transformed)						

Table 2.4 (co	Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments				
<u>Chang et al.</u>		Breast (post-	Serum PFOS (OR):			Standard covariates ^a					
<u>(2023)</u>		menopausal,	1st quartile	NR	1	and PFOA (natural					
USA		ER+/PR+),	2nd quartile	NR	1.46 (0.84–2.54)	log transformed)					
Enrolment,		incidence	3rd quartile	NR	2.19 (1.21-3.98)						
follow-up.			4th quartile	NR	1.89 (0.97-3.69)						
through			Trend-test P-value, (0.08							
November 2013		Breast (post- menopausal, ER-/PR-), incidence	Serum PFOA (OR)			Standard covariates ^a					
Nested case– control (cont.)			1st quartile	NR	1	and PFOS (natural					
			2nd quartile	NR	0.90 (0.38-2.10)	log transformed)					
		incidence	3rd quartile	NR	2.23 (0.90-5.54)						
			4th quartile	NR	1.62 (0.62-4.23)						
			Trend-test P-value, (0.21							
		Breast (post-	Serum PFOS (OR):		Standard covariates ^a						
		menopausal,	1st quartile	NR	1	and PFOA (natural					
		ER-/PR-),	2nd quartile	NR	1.01 (0.38-2.63)	log transformed)					
		incidence	3rd quartile	NR	1.12 (0.48-2.62)						
			4th quartile	NR	0.60 (0.19-1.83)						
			Trend-test P-value, (0.34							
		Breast (post-	Serum PFOA (OR):			Standard covariates ^a					
		menopausal,	1st quartile	NR	1	and PFOS (natural					
		ER+), incidence	2nd quartile	NR	1.07 (0.68-1.66)	log transformed)					
			3rd quartile	NR	1.01 (0.64-1.61)						
			4th quartile	NR	1.03 (0.61-1.75)						
			Trend-test P-value, ().96							
		Breast (post-	Serum PFOA (OR):			Standard covariates ^a					
		menopausal,	1st quartile	NR	1	and PFOS (natural					
		ER–), incidence	2nd quartile	NR	0.84 (0.36-1.95)	log transformed)					
			3rd quartile	NR	2.08 (0.85-5.07)						
			4th quartile	NR	1.63 (0.63-4.20)						
			Trend-test P-value, ().19							

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
<u>Chang et al.</u>		Breast (post-	Serum PFOS (OR):			Standard covariates ^a	
2023)		menopausal,	1st quartile	NR	1	and PFOA (natural	
SA		ER+), incidence	2nd quartile	NR	1.26 (0.81-1.95)	log transformed)	
nrolment,			3rd quartile	NR	1.59 (1.01-2.50)		
/95-2001/ llow-up			4th quartile	NR	1.29 (0.77-2.15)		
rough			Trend-test P-value, ().39			
lovember 2013		Breast (post-	Serum PFOS (OR):			Standard covariates ^a	
ested case-		menopausal,	1st quartile	NR	1	and PFOA (natural	
ontrol		ER–), incidence	2nd quartile	NR	0.98 (0.39-2.47)	log transformed)	
ont.))		3rd quartile	NR	1.13 (0.49–2.62)		
			4th quartile	NR	0.52 (0.18-1.55)		
			Trend-test P-value, (0.20			
		Breast (post-	Serum PFOA (OR):			Standard covariates ^a	
		menopausal,	1st quartile	NR	1	and PFOS (natural log transformed)	
		PR+), incidence	2nd quartile	NR	1.14 (0.66–1.96)		
			3rd quartile	NR	1.02 (0.57–1.83)		
			4th quartile	NR	0.77 (0.39–1.52)		
			Trend-test P-value, ().31			
		Breast (post-	Serum PFOA (OR):			Standard covariates ^a	
		menopausal,	1st quartile	NR	1	and PFOS (natural	
		PR–), incidence	2nd quartile	NR	0.90 (0.47-1.70)	log transformed)	
			3rd quartile	NR	2.05 (1.06-3.94)		
			4th quartile	NR	1.48 (0.75–2.93)		
			Trend-test <i>P</i> -value, ().15			
		Breast (post-	Serum PFOS (OR):			Standard covariates ^a	
		menopausal,	1st quartile	NR	1	and PFOA (natural	
	PR+), incid	PR+), incidence	2nd quartile	NR	1.55 (0.90–2.67)	log transformed)	
			3rd quartile	NR	2.34 (1.29-4.23)		
			4th quartile	NR	1.79(0.92 - 3.48)		

Table 2.4 (co	Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments				
Chang et al. (2023) USA Enrolment, 1993–2001/ follow-up, through November 2013 Nested case– control (cont.)		Breast (post- menopausal, PR–), incidence	Serum PFOS (OR): 1st quartile 2nd quartile 3rd quartile 4th quartile Trend-test <i>P</i> -value, 0	NR NR NR NR).15	1 1.00 (0.52–1.92) 0.91 (0.50–1.64) 0.61 (0.29–1.31)	Standard covariates ^a and PFOA (natural log transformed)					
Cohn et al. (2020) Oakland (CA), USA Enrolment, at birth between 1959 and 1967/ follow-up, birth to March 2013 (incidence) Nested case- control	Nested within the CHDS cohort (see <u>Table 2.1</u>) Cases: 102; offspring in the Child Health and Development Studies pregnancy cohort who had incident invasive or non-invasive breast cancer diagnosed by age 52 yr and who had a maternal perinatal blood sample and complete information on potential confounders and effect modifiers Controls: 310; 3 per case, density-sampled on case age and matched on birth year and trimester of maternal blood draw Exposure assessment method: see <u>Table 2.1</u>	Breast, incidence	Log ₂ maternal serum Continuous (for 4th quartile median vs 1st quartile median (difference of 3.15 ng/mL))	n PFOS (OR 102): 0.3 (0.1–0.9)	Age, birth year, trimester of maternal blood draw, maternal age at pregnancy, maternal history of breast cancer, African-American, primipara, maternal overweight at first prenatal visit, maternal serum $\log_2(p,p'-DDE)$, maternal serum $\log_2(o,p'-DDE)$, maternal serum $\log_2(o,p'-DDT)$, daughter breastfed, $\log_2(N-$ EtFOSAA), $\log_2(totalcholesterol), \log_2(N-EtFOSAA) × \log_2(total cholesterol)$	Exposure assessment critique: See <u>Table 2.1</u> . Other strengths: See <u>Table 2.1</u> . Other limitations: See <u>Table 2.1</u> .				

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Li et al. (2022a) 6 Ronneby, w	et al. (2022a) 60 507 (28 569 ponneby, women: 20 933 never puthern high, 7636 ever high exposure): the high exposure): the	Breast, incidence	Women, residential contaminated drink	exposure to	highly PFAS- SIR, Blekinge county	Age, calendar year	<i>Exposure assessment</i> <i>critique:</i> See <u>Table 2.1</u> .
southern			excluding Ronneby	referent):			
Sweden high exposure); the Enrolment, Ronneby Register 1985–2016/ Cohort included all follow-up, individuals who ever 1985–2016 lived in Ronneby (incidence) municipality 1985– Cohort 2013; one third of the households received PFAS-contaminated		Never	525	0.80(0.73 - 0.87)		Large study	
	Breast, incidence	Women, residential	exposure to	0.75 (0.64–0.88) highly PFAS- HR):		population, strong exposure contrast;	
		Never	525	1		unbiased inclusion;	
		Ever	156	0.95 (0.79-1.13)		up: long follow-up	
	households received PFAS-contaminated	Breast, incidence	Women, time period of residential exposure to highly PFAS-contaminated drinking-water (HR):				for part of the population; reference
	drinking-water		Never	525	1		group from same municipality. <i>Other limitations</i> :
	from a waterworks		Early (1985–2004)	102	0.94 (0.76-1.16)		
	situated near a military airfield where		Late (2005-2013)	54	0.96 (0.72-1.29)		
	PFAS-containing firefighting foam was	Breast, incidence	Women, duration of residential exposure to highly PFAS-contaminated drinking-water (HR):				profile without
	used in 1985–2013		Never	525	1		out effects due to
	(15 811 individuals		Short (1–10 yr)	89	1.01 (0.80–1.26)		specific compounds
	considered "ever-		Long (≥ 11 yr)	67	0.87 (0.67–1.13)		limited information
high"); subsets with long-term exposure (≥ 11 yr) in the latest	Thyroid, incidence	Men, residential exposure to highly PFAS- contaminated drinking-water (SIR, Blekinge county excluding Ronneby referent):			Calendar year, age, sex	on potential confounders.	
	part of the follow-up		Never	14	1.33 (0.73-2.23)		
period (2005–2013) were considered more highly exposed Exposure assessment method.see Teble 2.1		Ever	3	0.89 (0.18-2.61)			
	highly exposed Exposure assessment method: see Table 2.1	Thyroid, incidence	Women, residential contaminated drink excluding Ronneby	exposure to ing-water (referent):	highly PFAS- SIR, Blekinge county		
	<u>Interiou. see <u>Intole 2.1</u></u>		Never	32	1.38 (0.94–1.95)		
		Ever	16	2.08 (1.19-3.38)			

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
<u>Li et al. (2022a)</u>		Thyroid,	Residential exposur	e to highly I	PFAS-contaminated			
Konneby,		incidence	drinking-water (HK	.):	1			
Sweden			Never	46	1			
Enrolment, 1985–2016/		Thyroid, incidence	Time period of residential exposure to highly PFAS- contaminated drinking-water (HR):					
1985_2016			Never	46	1			
(incidence)			Early (1985-2004)	11	1.20 (0.62-2.33)			
Cohort			Late (2005–2013)	8	1.69 (0.77-3.73)			
(cont.)		Thyroid, incidence	Duration of resident contaminated drink	tial exposur ing-water (1	e to highly PFAS- HR):			
			Never	46	1			
			Short (1–10 yr)	12	1.35 (0.71-2.56)			
			Long ($\geq 11 \text{ yr}$)	7	1.38 (0.60-3.18)			

Table 2.4 (co	Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments				
Feng et al.	Nested within the	Breast, incidence	Plasma PFOA conce	ntration (H	R):	Calendar time,	Exposure assessment				
(2022)Dongfeng-TongjiShiyan, Chinacohort (see Table 2.1)Enrolment,Cases: 226; incidentSeptemberbreast cancer drawn2008 to Junefrom 18 387 female2010 and Aprilretirees of an autoto Octoberfacility who provided2013/follow-up,a specimen; total of2008 to 2018226 breast cancer(incidence)diagnoses includedCase-cohort13 diagnoses among	Dongfeng-Tongji cohort (see <u>Table 2.1</u>)		1st quartile (< 0.84 ng/mL)	53	1	age, BMI, smoking, drinking, marital	<i>critique</i> : See <u>Table 2.1</u> .				
		2nd quartile (0.84–1.18 ng/mL)	48	0.88 (0.56–1.39)	status, education level, occupation,	Other strengths: See Table 2.1.					
		3rd quartile (1.19–1.79 ng/mL)	58	1.28 (0.80-2.04)	batch to enter cohort, parity,	<i>Other limitations</i> : See <u>Table 2.1</u> .					
		4th quartile (≥ 1.80 ng/mL)	67	1.69 (1.05–2.70)	menopausal status, history of mastitis,						
	diagnoses included 13 diagnoses among women in the		Continuous (per unit on natural log scale)	226	1.35 (1.03–1.78)	history of cancer					
	subcohort	Breast, incidence	Plasma PFOS concer	ntration (HI	R):						
	Comparison cohort: 990 (including 13		1st quartile (< 6.39 ng/mL)	53	1						
	cases); subcohort of women randomly		2nd quartile (6.39–10.35 ng/mL)	48	0.75 (0.47–1.19)						
	selected according to age strata.		3rd quartile (10.36–15.66 ng/mL)	67	1.05 (0.66–1.67)						
	among the 990 in the		4th quartile (≥ 15.67 ng/mL)	58	0.87 (0.54–1.39)						
	served as controls until time of cancer diagnosis		Continuous (per unit on natural log scale)	226	0.88 (0.66–1.16)						
	Exposure assessment	Breast (post-	Plasma PFOA conce	ntration (H	R):						
method:	method: see <u>Table 2.1</u>	menopausal),	Low (< 1.19 ng/mL)	90	1						
		incidence	High (≥ 1.19 ng/mL)	115	1.53 (1.06–2.20)						
			Continuous (per unit on natural log scale)	205	1.34 (1.01–1.77)						

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Feng et al. (2022) Shiyan, China Enrolment, September 2008 to June 2010 and April to October 2013/follow-up, 2008 to 2018 (incidence) Case-cohort (cont.)		Breast (post- menopausal), incidence	Plasma PFOS concer Low (< 10.36 ng/mL) High (≥ 10.36 ng/mL) Continuous (per unit on natural log scale)	ntration (H) 84 121 205	R): 1 1.13 (0.80–1.58) 0.91 (0.71–1.17)	Calendar time, age, BMI, smoking, drinking, marital status, education level, occupation, batch to enter cohort, parity, history of mastitis, age at menopause, use of HRT, family history of cancer			
Madrigal et al. (2024) Finland Enrolment, 1986–2010/ follow-up, through 2016 Nested case– control	Nested within the Finnish Maternity Cohort (see <u>Table 2.1</u>) Cases: 400; National registry of nulliparous women who donated serum during the first trimester of pregnancy. 400 cases were randomly selected from cases diagnosed among women who donated serum for their first pregnancy and had a live, full- term birth delivered between 1987–2010, and who had no prior diagnosis of cancer at enrolment	Thyroid (papillary), incidence	PFOA serum concer $\leq 2.82 \text{ ng/mL}$ > 2.82 to 3.77 ng/mL > 3.77 to 4.85 ng/mL > 4.85 to 6.75 ng/mL > 6.75 ng/mL Continuous (per unit on \log_2 scale) Trend-test <i>P</i> -value, 0	ntration (OF 94 105 98 78 25 400).31	<pre>R): 1 1.10 (0.73-1.64) 0.99 (0.65-1.50) 1.30 (0.80-2.11) 0.54 (0.27-1.08) 0.90 (0.68-1.19)</pre>	Calendar year of delivery, age at first birth, PFOS, <i>N</i> -EtFOSAA, PFHpS detected	Exposure assessment critique: See Table 2.1. Other strengths: See Table 2.1; data available in the Medical Birth Registry included many host factors and potential confounders. Other limitations: See Table 2.1; data on host factors and potential confounders were collected during the pregnancy only.		

Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
Madrigal et al.	Controls: 400;	Thyroid	PFOA serum concer	ntration (OF	k):	Calendar year of				
(2024)individually matchedFinlandon year of deliveryEnrolment,(4–5-yr increments)1986–2010/and age at first birthfollow-up,(3-yr increments)through 2016Exposure assessmentNested case-method: see Table 2.1control(cont.)	individually matched	(papillary),	\leq 2.82 ng/mL	94	1	delivery, age at first				
	incidence	> 2.82 to 3.77 ng/mL	105	1.13 (0.76–1.69)	birth, smoking status at the time of pregnancy					
		> 3.77 to 4.85 ng/mL	98	1.05 (0.70–1.57)						
		> 4.85 to 6.75 ng/mL	78	1.40 (0.89–2.21)						
		> 6.75 ng/mL	25	0.63 (0.34-1.14)						
()			Continuous (per unit on log, scale)	400	0.95 (0.75-1.20)					
			Trend-test P-value, ().48						
		Thyroid	Age < 40 yr, PFOA serum concentration (OR):			Calendar year of				
		(papillary), incidence	Continuous (per unit on log, scale)	185	1.37 (0.92–2.03)	delivery, age at first birth				
		Thyroid (papillary), incidence Thyroid (papillary), incidence	Age < 40 yr, PFOA serum concentration (OR):			Calendar year of				
			Continuous (per unit on log ₂ scale)	185	1.20 (0.71–2.01)	delivery, age at first birth, PFOS, <i>N</i> -EtFOSAA, PFHpS				
			Age \geq 40 yr, PFOA s	erum conce	ntration (OR):	Calendar year of				
			Continuous (per unit on log ₂ scale)	215	0.77 (0.57–1.04)	delivery, age at first birth				
		Thyroid	Age \geq 40 yr, PFOA s	erum conce	ntration (OR):	Calendar year of				
		(papillary), incidence	Continuous (per unit on log ₂ scale)	215	0.70 (0.45–1.08)	delivery, age at first birth, PFOS, <i>N</i> -EtFOSAA, PFHpS detected, <i>N</i> -EtFOSAA detected				

Table 2.4 (c	Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments				
Madrigal et al.		Thyroid	PFOS serum concen	tration (OR	.):	Calendar year of					
<u>(2024)</u>		(papillary),	≤ 11.49 ng/mL	98	1	delivery, age at					
Enrolment, 1986–2010/ follow-up, through 2016		incidence	> 11.49 to 15.76 ng/mL	94	0.98 (0.61–1.57)	first birth, PFOA, N-EtFOSAA, PFHpS					
			> 15.76 to 22.63 ng/mL	119	1.28 (0.76–2.18)	detected, total PCBs, hexachlorobenzene,					
Nested case-			> 22.63 to 27.95 ng/mL	54	0.95 (0.50-1.82)	metabolites					
(cont.)			> 27.95 ng/mL	35	0.86 (0.38-1.95)	metabolites					
. ,			Continuous (per unit on log, scale)	400	1.14 (0.81–1.59)						
			Trend-test P-value, (0.74							
		Thyroid	PFOS serum concen	tration (OR	.):	Calendar year of					
		(papillary),	≤ 11.49 ng/mL	98	1	delivery, age at first					
		incidence	> 11.49 to 15.76 ng/mL	94	0.98 (0.62–1.54)	birth, smoking status at the time of					
			> 15.76 to 22.63 ng/mL	119	1.23 (0.75–2.00)	pregnancy					
			> 22.63 to 27.95 ng/mL	54	0.92 (0.52–1.62)						
			> 27.95 ng/mL	35	0.87 (0.45-1.65)						
			Continuous (per unit on log, scale)	400	1.04 (0.81–1.33)						
			Trend-test P-value, (0.61							
		Thyroid	Age < 40 yr, PFOS so	erum conce	ntration (OR):	Calendar year of					
		(papillary), incidence	Continuous (per unit on log ₂ scale)	185	1.34 (0.92–1.96)	delivery, age at first birth					

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Madrigal et al. (2024) Finland Enrolment, 1986–2010/ follow-up, through 2016 Nested case– control		Thyroid (papillary), incidence	Age < 40 yr, PFOS se Continuous (per unit on log ₂ scale)	erum conce 185	ntration (OR): 1.14 (0.68–1.93)	Calendar year of delivery, age at first birth, PFOA, <i>N</i> -EtFOSAA, total PCBs, hexachlorobenzene, β-HCH, chlordane metabolites, DDT metabolites			
(cont.)		Thyroid (papillary), incidence	Age \geq 40 yr, PFOS se Continuous (per unit on log, scale)	erum conce 215	ntration (OR): 0.86 (0.61–1.20)	Calendar year of delivery, age at first birth			
		Thyroid (papillary), incidence	Age \geq 40 yr, PFOS so Continuous (per unit on \log_2 scale)	erum conce 215	ntration (OR): 1.01 (0.60–1.71)	Calendar year of delivery, age at first birth, PFOA, <i>N</i> -EtFOSAA, total PCBs, hexachlorobenzene, β-HCH, chlordane metabolites, DDT metabolites			

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
van Gerwen	Nested within Bio <i>Me</i>	Thyroid,	Plasma PFOA conce	ntration (O	R):	Age, BMI, sex, race,	Exposure assessment		
et al. (2023) cohort (see <u>Table 2.1</u>). Mount Sinai, Cases: 88 adult	incidence and prevalence	Continuous (per unit on log ₂ scale)	88	0.99 (0.63–1.56)	storage time of plasma sample	<i>critique</i> : See <u>Table 2.1</u> .			
New York, USA Enrolment	patients diagnosed with thyroid cancer		Continuous (per IQR increase)	88	0.99 (0.53–1.83)		Other strengths: See <u>Table 2.1</u> .		
2008–2021 Nested case	using ICD codes 193 (9th Payisian) and C73	Thyroid,	Plasma PFOA conce	ntration (O	R):	Age, BMI, sex,	Other limitations: See		
Nested case- (9th Revision) and C73 control (10th Revision) within Bio <i>Me</i> , a medical record-linked biobank within the Institute for Personalized Medicine at the Icahn School of Medicine at Mount	incidence and prevalence	Continuous (per unit on log ₂ scale)	88	0.74 (0.31–1.72)	race, storage time of plasma sample,	Other comments: Analyses were			
		Continuous (per IQR increase)	88	0.66 (0.20-2.08)	other PFAS	repeated for the time between plasma			
	Thyroid, incidence	Longitudinal study j after sample collecti (OR):	(diagnosed ≥ 1 yr PFOA concentration	Age, BMI, sex, race, storage, time of plasma sample	sample collection and thyroid cancer diagnosis: < 1 yr				
	Sinai Controls: 88 healthy		Continuous (per unit on log ₂ scale)	31	1.52 (0.77–2.98)		(cross-sectional group) and ≥ 1 yr (longitudinal group).		
	(non-cancer) participants, pair- matched on sex, age	Thyroid, prevalence	Cross-sectional stud after sample collecti (OR):	ly populatio on), plasma	n (diagnosed < 1 yr PFOA concentration				
	(± 5 yr), race/ethnicity, BMI, smoking		Continuous (per unit on log ₂ scale)	57	0.84 (0.49–1.40)				
	ever smoked ≥ 100	Thyroid, incidence and	Plasma sb-PFOS (branched PFOS) concentration (OR):			Age, BMI, sex, race, storage time of			
	entire life", yes/no), and calendar year of	prevalence	Continuous (per unit on log ₂ scale)	88	1.32 (0.99–1.81)	plasma sample			
	sample collection Exposure assessment		Continuous (per IQR increase)	88	1.73 (0.97–3.24)				
method: see <u>Table 2</u>	method: see <u>Table 2.1</u>	Thyroid, incidence and	Plasma sb-PFOS (br (OR):	anched PFC	OS) concentration	Age, BMI, sex, race, storage time of plasma sample, other PFAS			
		prevalence	Continuous (per unit on log ₂ scale)	88	1.21 (0.43-3.55)				
			(Continuous (per IOR increase)	88	1.47 (0.18–12.26)			

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
<u>van Gerwen</u> <u>et al. (2023)</u> Mount Sinai,		Thyroid, incidence	Longitudinal study p after sample collection PFOS) concentration	population (on), plasma n (OR):	diagnosed ≥ 1 yr sb-PFOS (branched	Age, BMI, sex, race, storage time of plasma sample			
New York, USA Enrolment	New York, USA Enrolment 2008–2021		Continuous (per unit on log ₂ scale)	31	3.09 (1.73-6.13)	1 1			
2008–2021 Nested case– control (cont.)	Thyroid, prevalence	Cross-sectional study population (diagnosed < 1 yr after sample collection), plasma sb-PFOS (branched PFOS) concentration (OR):							
		Continuous (per unit on log ₂ scale)	57	1.13 (0.83–1.56)					
		Thyroid,	Plasma n-PFOS (line	ear PFOS) co	oncentration (OR):				
		incidence and prevalence	Continuous (per unit on log ₂ scale)	88	1.56 (1.17–2.15)				
			Continuous (per IQR increase)	88	2.32 (1.34-4.26)				
		Thyroid,	Plasma n-PFOS (line	ear PFOS) co	oncentration (OR):	Age, BMI, sex,			
		incidence and prevalence	Continuous (per unit on log ₂ scale)	88	2.80 (1.32-6.45)	race, storage time of plasma sample,			
			Continuous (per IQR increase)	88	7.09 (1.69–34.54)	other PFAS			
		Thyroid, incidence	Longitudinal study population (diagnosed \geq 1 yr after sample collection), plasma <i>n</i> -PFOS (linear PFOS) concentration (OR):			Age, BMI, sex, race, storage time of plasma sample			
		Continuous (per unit on \log_2 scale)	31	2.67 (1.59-4.88)	. 1				
		Thyroid, prevalence	Cross-sectional stud after sample collection PFOS) concentration	Cross-sectional study population (diagnosed < 1 yr after sample collection), plasma <i>n</i> -PFOS (linear PFOS) concentration (OR):					
			Continuous (per unit on log ₂ scale)	57	1.45 (1.07–2.01)				

Table 2.4 (co	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
van Gerwen		Thyroid	Plasma PFOA concer	ntration (Ol	R):		
<u>et al. (2023)</u> Mount Sinai,		(papillary), incidence and	Continuous (per unit on log ₂ scale)	74	1.03 (0.63–1.68)		
New York, USA Enrolment 2008–2021 Nested case–		prevalence	Continuous (per IQR increase)	74	1.03 (0.55–1.96)		
		Thyroid (papillary),	Plasma sb-PFOS (branched PFOS) concentration (OR):				
(cont.)		incidence and prevalence	Continuous (per unit on log ₂ scale)	74	1.30 (0.95–1.83)		
			Continuous (per IQR increase)	74	1.61 (0.91–2.97)		
		Thyroid	Plasma n-PFOS (line	ear PFOS) co	oncentration (OR):		
		(papillary), incidence and	Continuous (per unit on log ₂ scale)	74	1.56 (1.13–2.21)		
		prevalence	Continuous (per IOR increase)	74	2.22 (1.24-4.20)		

Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
Winquist et al.	Case-cohort within	Breast (post-	Women, serum PFO	A concentr	ation (HR):	Year of serum	Exposure assessment			
(2023)the CPS-II Lifelink20 states, USACohort	menopausal), incidence	1st quartile (< 3.700 ng/mL)	193	1	sample collection, age at serum	<i>critique</i> : See <u>Table 2.1</u> .				
Enrolment, 1998–200/	Enrolment, Cases: 3762 overall 1998–200/ (786 female breast);		2nd quartile (3.700 to < 5.000 ng/mL)	196	0.80 (0.56-1.15)	collection, race, education, smoking	Strengths: See <u>Table 2.1</u> .			
follow-up,incidence casesthroughfrom the CPS-II30 June 2015Lifelink CohortCase-cohort(surviving CPS-IINutrition cabort		3rd quartile (5.000 to < 6.900 ng/mL)	189	0.75 (0.52–1.09)	status, alcohol consumption	Table 2.1.				
	(surviving CPS-II Nutrition cohort		4th quartile (≥ 6.900 ng/mL)	202	0.82 (0.57–1.17)					
	participants) with first cancer diagnosis		Continuous (per unit on log ₂ scale)	780	0.96 (0.82–1.12)					
	of kidney, bladder,	Breast (post-	Women, serum PFO	Women, serum PFOS concentration (HR):						
	breast (females only), prostate (males	menopausal), incidence	1st quartile (< 12.000 ng/mL)	160	1					
only), or pancreatic cancer, leukaemia, or lymphoma, detected through self-report or NDI linkage and verified through medical records review or cancer registry; all participants with		2nd quartile (12.000 to < 17.000 ng/mL)	195	0.66 (0.45-0.97)						
		3rd quartile (17.000 to < 24.000 ng/mL)	211	0.84 (0.57–1.23)						
		4th quartile (≥ 24.000 ng/mL)	214	0.70 (0.48–1.01)						
	incident cancers		Continuous (per unit on log ₂ scale)	780	0.87 (0.75–1.01)					

Table 2.4 (c	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Winquist et al. (2023) 20 states, USA Enrolment, 1998–200/ follow-up, through 30 June 2015 Case–cohort (cont.)	Comparison cohort: 999; a sex-stratified simple random sample of 499 women and 500 men (~3% of the eligible cohort); stratification sampling was to ensure an adequate number of subcohort participants in sex-specific analyses (for breast and prostate cancers) Exposure assessment method: see Table 2.1						
Vieira et al. (2013) OH and WV, USA 1996–2005 (incidence) Case–control	Cases: study 1: 4057 female breast, 343 thyroid cancer; study 2: 1260 female breast, 94 thyroid; cancer cases were retrieved from cancer registries covering a community sample with relatively high exposure to PFOA because of contamination of drinking-water from the Parkersburg (WV), polymer-production plant	Breast, incidence	Analysis 1. Residence district (OH and WY Females: Unexposed Any exposed water district Analysis 2. Individu exposure, assuming only) (OR) Females: Unexposed Low (3.7–12.8 µg/L) Medium (12.9–30.7 µg/L) High (30.8–109 µg/L) Very high	te in a PFOA V) (OR) 3621 436 al-level ann 10-yr reside 1037 72 77 45 29	L-contaminated water 1 1.0 (0.9–1.1) ual PFOA serum ency and latency (OH 1 0.9 (0.7–1.2) 1.1 (0.8–1.5) 0.7 (0.5–1.0) 1.4 (0.9–2.3)	Age, diagnosis year, insurance provider, smoking status Age, race, diagnosis year, insurance provider smoking status	Exposure assessment critique: See <u>Table 2.1</u> . Other strengths: ascertainment of cases from cancer registries; large exposure contrast. Other limitations: use of other types of cancer as controls; lack of adjustment for several potential confounding variables; lack of information concerning tumour hormone-receptor status.

Table 2.4 (continued)										
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments			
<u>Vieira et al.</u> Controls: NR; for each (2013)	Thyroid,	Analysis 1: residence	e in a PFOA	-contaminated water	Age, sex, diagnosis					
(2013) OH and WV	cancer site evaluated,	incidence	district (OH and W	V) (OR):	1	year, insurance				
USA of cancer at all other 1996–2005 sites among women, (incidence) with the exclusion		Unexposed	303	I 1.1 (0.7.1.5)	status					
		Any exposed water district	40	1.1 (0.7–1.5)	status					
(incidence)	se-control of four cancers of a		Little Hocking	3	0.8 (0.3-2.7)					
(cont) of four cancers of a		Lubeck	7	1.2 (0.6–2.6)						
(cont.)	testis, pancreas,		Tuppers Plains	2	0.3 (0.1–1.4)					
	and liver) that have		Belpre	5	0.9 (0.4-2.2)					
	been associated with		Pomeroy	0	NC					
	PFOA in studies in		Mason	23	1.4 (0.9–2.2)					
	experimental animals	Other (specify),	Analysis 2: individu	al-level ann						
	Exposure assessment	thyroid, incidence	exposure, assuming only) (OR):	10-yr reside						
	method: see <u>Table 2.1</u>		Unexposed	79	1					
			Low	5	0.9 (0.4-2.3)					
			(3.7–12.8 µg/L)							
			Medium (12.9–30.7 μg/L)	5	0.9 (0.4–2.3)					
			High (30.8–109 μg/L)	3	0.7 (0.2–2.1)					
			Very high (110–655 μg/L)	2	0.8 (0.2–3.5)					

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Wielsøe et al. (2017)Cases: 77 cases of breast cancer;Greenlandrecruited at diagnosisEnrolment, 2000–2003/at Dronning Ingridsfollow-up, 2011–2014(where all breast cancer cases in Greenland are registered) during two	Breast, incidence	Serum PFOA concer	ntration (OF	R):	Age, BMI, cotinine	Exposure assessment			
		1st tertile	14	1	levels, parity,	critique:			
		2nd tertile	26	1.86 (0.80-4.31)	breastfeeding	Key strengths were that serum levels represent the combined exposure through all			
		3rd tertile	37	2.64 (1.17-5.97)					
		Continuous (per unit increase)	77	1.26 (1.01–1.58)					
	Breast, incidence	Serum PFOS concentration (OR):				exposure pathways;			
		1st tertile	8	1		measurement error			
	time periods: 2000–		2nd tertile	25	3.13 (1.20-8.15)		low. Key limitations		
	2003 and 2011–2014;		3rd tertile	44	5.50 (2.19–13.84)				
	women of Greenland Inuit descent		Continuous (per unit increase)	77	1.02 (1.01–1.03)		cancer alters ADME of PFAS there could be possible differential exposure misclassification, as blood was collected after diagnosis (also see <u>Bonefeld-</u> <u>Jorgensen et al.</u> , <u>2011</u>); single samples at time of case hospitalization may not reflect exposure at crucial windows in cancer development.		
Table 2.4 (c	ontinued)								
---	--	---	-------------------------------	-------------------------------	---------------------------	--------------------------	---		
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Wielsøe et al. (2017) Greenland Enrolment, 2000–2003/ follow-up, 2011–2014 Case–control (cont.)	Controls: 81 controls for the participants recruited during 2000–2003 were selected from two cross-sectional studies on healthy persons with POPs serum measurements in the same period; the controls recruited during 2011–2014 were patients with nonmalignant diagnoses at the Dronning Ingrids hospital; controls were frequency-matched on age and geographical living area to cases; all controls were in people of Greenland Inuit descent Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; a single blood sample was collected at the hospital before treatment						<i>Other strengths</i> : cases confirmed by a positive histological sample. <i>Other limitations</i> : exclusion of cases and controls from the final analyses not clearly explained; some of the controls were hospital patients with nonmalignant abnormalities in the uterus, ovaries and breasts; small sample size and limited statistical power; cross- sectional design; no information about the delay between diagnosis and the collection of blood or if treatment occurred before blood collection; unexplained elevation in median PFOS level for cases recruited in early time period.		

Table 2.4 (c	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Tsai et al.Cases: 120 patie(2020)aged 25-80 yr aTaiwan, Chinadiagnosis, recru2014-2016NTUHCase-controlControls: 119 ccaged 25-80 yr awithout any hisof malignancy;recruited throuadvertisementsposters and flyeat NTUH and ifthe communitycontrols receivea small financiacompensation(~US\$ 6.30) aftecompleting theExposure assessmethod: quantiserum measureanalytical methwas collected diapprox. the samperiod for casesmatched controlblood was colleat the hospital btreatment	Cases: 120 patients aged 25–80 yr at diagnosis, recruited at NTUH Controls: 119 controls aged 25–80 yr and without any history	Breast, incidence Breast, incidence	Serum PFOA concer Continuous (per unit increase on natural log scale) Serum PFOS concer Continuous (per unit increase on	ntration (OF 120 tration (OR 120	 (0.89 (0.59–1.34) (0.59–1.34) (0.64–1.79) 	Age, history of pregnancy, oral contraceptionExposure critique: critique: key stren were that BMI, menopause, education levelBMI, menopause, education levellevels rep combined through a pathware	<i>Exposure assessment</i> <i>critique</i> : Key strengths were that plasma levels represent the combined exposure through all exposure
	advertisements on posters and flyers at NTUH and in the community;	Intensitient of sectionBreast, incidencecruited through vertisements on sters and flyersBreast, incidenceNTUH and in e community; ntrols received mall financial mpensation US\$ 6.30) after mpleting the study prosure assessment ethod: quantitative rum measurements; alytical method ss state-of-the-art; ingle blood sample ts collected during prox. the same time riod for cases and atched controls; bod was collected the hospital before eatmentBreast, incidenceBreast (ER-), incidenceBreast (ER-), incidence	natural log scale) Serum PFOA concer natural log scale): Age ≤ 50 yr Age > 50 yr	ntration (OF 60 60	R per unit increase on 1.14 (0.66–1.96) 0.78 (0.40–1.51)	History of pregnancy, oral contraception use, abortion,	blood samples were collected after diagnosis, a strength was that they were collected
	controls received a small financial compensation (~US\$ 6.30) after completing the study Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; a single blood sample was collected during approx. the same time period for cases and matched controls; blood was collected at the hospital before treatment		Serum PFOS concern natural log scale): Age ≤ 50 yr	tration (OR 60	2.34 (1.02–5.38)	BMI, menopause, education level	 before treatment; measurement error low. Key limitations were that if breast cancer alters ADME of PFAS there could be possible differential exposure misclassification as plasma collected after diagnosis; single samples at time of case hospitalization may not reflect exposure at crucial windows in cancer development.
			Serum PFOA concernatural log scale): Age \leq 50 yr Age $>$ 50 yr	ntration (OF 11 12	0.42 (0.17–1.06) 1.08 (0.33–3.59)		

Table 2.4 (c	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Tsai et al. (2020) Taiwan, China 2014–2016 Case–control (cont.)		Breast (ER –), incidence Breast (ER+), incidence Breast (ER+), incidence	Serum PFOS concent natural log scale): Age \leq 50 yr Serum PFOA concent natural log scale): Age \leq 50 yr Age $>$ 50 yr Serum PFOS concent natural log scale): Age \leq 50 yr Age \geq 50 yr Age $>$ 50 yr	ntration (OR 11 12 ntration (OF 49 48 ntration (OR 49 48	per unit increase on 0.23 (0.05–1.15) 0.66 (0.20–2.22) 2 per unit increase on 1.41 (0.77–2.56) 0.70 (0.35–1.42) per unit increase on 3.25 (1.29–8.23) 0.53 (0.24–1.18)	History of pregnancy, oral contraception use, abortion, BMI, menopause, education level	Other strengths: cases confirmed by positive histological samples; controls included participants without any history of malignancy; models adjusted for important confounding variables; available information on tumour hormone- receptor status. Other limitations: small sample size and limited statistical power; cross sectional design; strategy for recruiting controls could have induced a control selection bias if people positively responding to advertisement had a healthier lifestyle and a higher medical awareness compared

Table 2.4 (c	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Itoh et al.	Cases: 401 women	Breast, incidence	Serum PFOA concer	ntration (OI	R):	Age and residential	Exposure assessment
<u>(2021)</u> Japan	aged 20–74 yr with new invasive breast		1st quartile (0.72–3.98 ng/mL)	167	1	area (urban or rural), BMI, height, menopausal status and age at menopause, age at first childbirth,	<i>critique</i> : Key strengths
Enrolment, May 2001 to	cancer, admitted to any of the four		2nd quartile (4.00–5.57 ng/mL)	100	0.45 (0.25-0.80)		were that serum levels represent
2005	in the study; of 412		3rd quartile (5.57–7.62 ng/mL)	82	0.39 (0.20-0.73)		exposure through all
Case-control	405 (98%) agreed to		4th quartile (7.64–62.98 ng/mL)	52	0.21 (0.10-0.45)	of breast cancer, smoking status,	Measurement error low.
	Controls: 401 selected among individuals attending two of the hospitals for medical check-ups during the study period; they were confirmed to not have cancer and were matched with cases by age (within 3 yr) and residential area (urban or rural); two of the control participants refused to provide blood specimens and two refused to allow their samples to be used in the present analysis		Trend-test <i>P</i> -value, (0.0001		physical activity, age at menarche, number of births, breastfeeding duration, alcohol intake, isoflavone intake, and education level	Key limitations were that if breast cancer alters ADME of PFAS there could be possible differential exposure misclassification; no information available concerning the delay between diagnosis and blood sample used for PFAS measurements and if cases had received cancer treatment before blood sample; single samples at time of case hospitalization may not reflect exposure at crucial windows in cancer development; minimal information on potential carcinogenic co- exposures.

Table 2.4 (co	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
<u>Itoh et al.</u> (2021)	Exposure assessment method: quantitative	Breast, incidence	Serum PFOA concer	ntration (OF	ξ): 1	Age and residential area (urban or	<i>Other strengths</i> : cases were histologically
Japan	serum measurements;		(0.72–3.98 ng/mL)	107	1	rural), BMI, height,	confirmed invasive
May 2001 to	Enrolment, analytical method May 2001 to state-of-the-art; a		2nd quartile (4.00–5.57 ng/mL)	100	0.37 (0.19–0.73)	status and age at	response rate reduced
September single serum sa 2005 was collected d Case-control hospitalization (cont.) cases of invasiv cancer and ma non-cancer con in the hospital medical check-	was collected during	pie ing or ned ols r	3rd quartile (5.57–7.62 ng/mL)	82	0.39 (0.18-0.84)	at first childbirth,	selection bias; large
	hospitalization for cases of invasive cancer and matched non-cancer controls in the hospital for medical check-up		4th quartile (7.64–62.98 ng/mL) Trend-test <i>P</i> -value, 0	52	at first childbirth, family history of breast cancer, smoking status, physical activity, age at menarche, number of births, breastfeeding duration, alcohol intake, isoflavone intake, and education level, serum total concentrations of PCBs (lipid- adjusted), fish and shellfish intake, vegetable intake, and calendar year of blood sampling	sample size; detailed information on diet, available information on tumour hormone- receptor status; analysis examined impact of individual isomers and combinations of isomers, including the sum of 6 PFOS isomers and the sum of 2 PFOA isomers as well as combinations of PFSAs and PFCAs.	

Table 2.4 (c	continued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Itoh et al.		Breast, incidence	Serum PFOS concen	tration (OR	R):	Age and residential	Other limitations:
<u>(2021)</u> Japan			1st quartile (1.13–10.25 ng/mL)	183	1	area (urban or rural), BMI, height,	potential selection bias for controls;
Enrolment, May 2001 to September 2005		2nd quartile (10.29–14.27 ng/mL)	85	0.41 (0.22-0.77)	menopausal lack of i status and age at and adju	lack of information and adjustment	
			3rd quartile (14.27–19.24 ng/mL)	86	0.37 (0.19-0.71)	menopause, age at first childbirth,	for socioeconomic status; cross
Case–control (cont.)			4th quartile (19.28–377.33 ng/mL)	47	0.14 (0.07–0.31)	family history of breast cancer,	sectional design; no adjustment for
	Τr	Trend-test <i>P</i> -value, <	< 0.0001		shioking status, physical activity, age at menarche, number of births, breastfeeding duration, alcohol intake, isoflavone intake, and education level	medical check- up examinees as controls may have caused selection bis due to a higher medical awareness and different socioeconomic status compared to the source population for cases.	

Table 2.4 (c	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Itoh et al.		Breast, incidence	Serum PFOS concen	tration (OR	2)	Age and residential	
<u>(2021)</u> Japan			1st quartile (1 13–10 25 ng/mL)	183	1	area (urban or rural), BMI, height,	
Enrolment, May 2001 to			2nd quartile (10.29–14.27 ng/mL)	85	0.38 (0.18-0.82)	menopausal status and age at	
September 2005			3rd quartile (14.27–19.24 ng/mL)	86	0.31 (0.14–0.69)	menopause, age at first childbirth,	
Case–control (cont.)			4th quartile (19.28–377.33 ng/mL)	47	0.15 (0.06-0.39)	of breast cancer,	
			Trend-test <i>P</i> -value, ().0001		physical activity, age at menarche, number of births, breastfeeding duration, alcohol intake, isoflavone intake, and education level, serum total concentrations of PCBs (lipid- adjusted), fish and shellfish intake, vegetable intake, and calendar year of blood sampling	

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Itoh et al.		Breast (pre-	Serum PFOA concer	ntration (OI	<u></u> ;	Age and residential	
(<u>2021)</u> Japan		menopausal), incidence	Low (0.7-4.5 ng/mL)	NR	1	area (urban or rural), BMI,	
Enrolment, May 2001 to			$\begin{array}{c} \text{(0.7 II.5 hg/mL)} \\ \text{Middle} \\ (4.5-6.7 hg/mL) \end{array}$	NR	0.72 (0.38-1.37)	height, age at first childbirth,	
September 2005			(4.5–0.7 $\text{ ig/inL})$ High	NR	0.66 (0.28–1.54)	family history of breast cancer,	
Case-control (cont.)			Trend-test <i>P</i> -value, ().26		smoking status, physical activity,	
. ,		Breast (pre-	Serum PFOS concen	tration (OR	k):	age at menarche,	
		menopausal), incidence	Low (1.1–11.5 ng/mL)	NR	1	number of births, breastfeeding	
			Middle (11.5–17.0 ng/mL)	NR	0.52 (0.27–1.01)	duration, alcohol intake, isoflavone	
			High (17.1–377.33 ng/mL)	NR	0.28 (0.09-0.85)	intake, education level	
			Trend-test P-value (007			
		Breast (post-	Serum PEOA concer	ntration (OF	Age and residential		
		menopausal),	Low $(0.7, 4.5 \text{ ng/mL})$	NR	1	area (urban or rural). BMI.	
			(0.7-4.5 mg/mL) Middle	NR	0.61 (0.34–1.07)	height, age at menopause, age	
			(4.5–6.7 ng/mL) High NR (6.73–62.98 ng/mL)	NR	0.41 (0.23-0.75)	at first childbirth, family history	
			Trend-test <i>P</i> -value, 0	0.005		smoking status, physical activity,	
						age at menarche, number of births,	
						breastfeeding duration, alcohol	
						intake, isoflavone intake, education	

menopause

Ta

322

ption size, C ption, exposure (1 nent method in n B B n	(histopathology), incidence or mortality Breast (post-	Exposure category or level	Exposed cases or deaths	(95% CI)	controlled	Comments
B n ii	Breast (post-	6 DEG 6	cases or deaths	(95% CI)	controlled	Comments
n iı	1)	Serum PFOS concen	tration (OR):	Age and residential	
	menopausal), incidence	Low (1.1–11.5 ng/mL)	NR	1	area (urban or rural), BMI,	
		Middle (11.5–17.0 ng/mL)	NR	0.60 (0.33–1.09)	height, age at menopause, age	
		High (17.1–377.33 ng/mL)	NR	0.35 (0.19–0.66)	family history	
se–control nt.)		Trend-test <i>P</i> -value, 0	0.001		smoking status, physical activity, age at menarche, number of births, breastfeeding duration, alcohol intake, isoflavone intake, education level, years after menopause	
В	Breast (ER- and	Serum PFOA concer	ntration (OF	R):	Age and residential	
Р	PK–), incidence	Low (0.7–4.5 ng/mL)	NR	1	area (urban or rural), BMI, height,	
		Middle (4.5–6.7 ng/mL)	NR	0.78 (0.40–1.49)	menopausal status and age at	
		High (6.73–62.98 ng/mL) Trend-test <i>P</i> -value, 0	NR 0.23	0.62 (0.30-1.32)	at first childbirth, family history of breast cancer, smoking status, physical activity, age at menarche, number of births, breastfeeding duration, alcohol intake isoflayone	
	j	PR–), incidence	PR–), incidence Low (0.7–4.5 ng/mL) Middle (4.5–6.7 ng/mL) High (6.73–62.98 ng/mL) Trend-test <i>P</i> -value, 0	PR–), incidence Low NR (0.7–4.5 ng/mL) Middle NR (4.5–6.7 ng/mL) High NR (6.73–62.98 ng/mL) Trend-test <i>P</i> -value, 0.23	PR–), incidence Low NR 1 (0.7-4.5 ng/mL) Middle NR 0.78 (0.40–1.49) (4.5–6.7 ng/mL) High NR 0.62 (0.30–1.32) (6.73–62.98 ng/mL) Trend-test <i>P</i> -value, 0.23	PR-), incidence Low NR 1 (0.7-4.5 ng/mL) Middle NR 0.78 (0.40-1.49) (4.5-6.7 ng/mL) High NR 0.62 (0.30-1.32) (6.73-62.98 ng/mL) Trend-test <i>P</i> -value, 0.23 A first childbirth, family history of breast cancer, smoking status, physical activity, age at menarche, number of births, breastfeeding duration, alcohol intake, isoflavone intake, education level

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comment
Itoh et al.		Breast (ER+ and	Serum PFOA concer	ntration (OF	R):	Age and residential	
<u>(2021)</u> Japan		PR–), incidence	Low (0.7–4.5 ng/mL)	NR	1	area (urban or rural), BMI, height,	
Enrolment, May 2001 to			Middle (4.5–6.7 ng/mL)	NR	0.86 (0.44–1.68)	menopausal status and age at	
2005			High (6.73–62.98 ng/mL)	NR	0.27 (0.11–0.69)	at first childbirth,	
(cont.)			Trend-test <i>P</i> -value, (0.007		of breast cancer.	
()		Breast (ER+ and	Serum PFOA concer	ntration (OI	R):	smoking status,	
		PR+), incidence	Low (0.7–4.5 ng/mL)	NR	1	physical activity, age at menarche, number of births, breastfeeding duration, alcohol intake, isoflavone	
			Middle (4.5–6.7 ng/mL)	NR	0.63 (0.39–1.01)		
			High (6.73–62.98 ng/mL)	NR	0.57 (0.33-0.97)		
			Trend-test P-value, (0.035		intake, education	
		Breast (ER – and	Serum PFOS concen	tration (OR	.):	level	
		PR–), incidence	Low (1.10–11.5 ng/mL)	NR	1		
			Middle (11.5–17.0 ng/mL)	NR	0.61 (0.31–1.20)		
			High (17.1–377.33 ng/mL)	NR	0.44 (0.20-0.96)		
			Trend-test P-value, (0.037			
		Breast (ER+ and	Serum PFOS concen	tration (OR	k):		
		PR–), incidence	Low	NR	1		
			(1.10–11.5 ng/mL)				
			Middle (11.5–17.0 ng/mL)	NR	1.07 (0.52–2.20)		
			High (17.1–377.33 ng/mL)	NR	0.33 (0.13-0.83)		
			Trend-test P-value, (0.016			

324

Table 2.4 (c	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Itoh et al.		Breast (ER+ and	Serum PFOS concer	tration (OR	L):	Age and residential	
<u>(2021)</u> Japan		PR+), incidence	Low (1.10–11.5 ng/mL)	NR	1	area (urban or rural), BMI, height,	
Enrolment, May 2001 to			Middle (11.5–17.0 ng/mL)	NR	0.56 (0.34-0.90)	menopausal status and age at	
September 2005			High (17.1–377.33 ng/mL)	NR	0.33 (0.18-0.59)	menopause, age at first childbirth,	
(cont.)			Trend-test <i>P</i> -value, (0.0001		of breast cancer,	
						physical activity,	
						age at menarche, number of births,	
						breastfeeding duration alcohol	
						intake, isoflavone	
						intake, education	

level

Table 2.4 (co	ontinued)						
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Liu et al. (2022)	Cases: 134 cases were	Thyroid,	Serum PFOA (OR):			Age, sex, diabetes	Exposure assessment
Jinan City, Shandong	diagnosed with thyroid cancer by pathological	incidence	1st quartile (< 7.9 ng/mL)	69	1	status	<i>critique</i> : Key strengths
Province, east examin China Shando 2016–2017 Qianfo Case–control particip group s thyroid 2 week Contro	examination at the Shandong Provincial		2nd quartile (7.9 to < 10.9 ng/mL)	23	0.24 (0.12-0.50)		were that serum levels represent
	Quantoshan Hospital; participants in the case group stopped taking thyroid medication for 2 weeks Controls: 185 controls were randomly selected from patients undergoing routine medical visits at the hospital with normal		3rd quartile (10.9 to < 16.1 ng/mL)	21	0.24 (0.11-0.49)		exposure pathways:
			4th quartile (≥ 16.1 ng/mL)	21	0.20 (0.09-0.44)		measurement error low.
			Trend-test <i>P</i> -value,	< 0.001			Key limitations
		Thyroid, incidence	Serum PFOS (OR):			Age, sex, diabetes	were that if thyroid
			1st quartile (< 4.7 ng/mL)	49	1	status	of PFAS there
			2nd quartile (4.7 to < 7.5 ng/mL)	48	0.81 (0.42–1.53)		differential exposure
	examination and no		3rd quartile (7.5 to < 10.8 ng/mL)	17	0.26 (0.12-0.57)		as serum collected
	history of thyroid disease or taking iodine or thyroid hormone drugs during the blood collection, and frequency- matched to the case group on age (± 5 yr) and sex		4th quartile (≥ 10.8 ng/mL)	20	0.28 (0.12-0.66)		periods for cases;
			Trend-test <i>P</i> -value, ().001			single samples at time of case hospitalization may not reflect exposure at crucial windows i cancer development

326

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Liu et al. (2022) Jinan City, Shandong Province, east China 2016–2017 Case–control (cont.)	Exposure assessment method: quantitative serum measurements; analytical method was state-of-the-art; a single blood sample was collected during the same time period for cases and matched controls; blood was collected between treatment periods; control samples collected during routine visits to the hospital						Other strengths: use of novel statistical methods to evaluate the impact of PFAS on thyroid function and thyroid hormones using a WQS model. Other limitations: limited exposure contrast; small sample size; limited confounding adjustment; potential for reverse causation.		
Velarde et al. (2022) Philippines 2018 Case–control	Cases: 75 cases recruited through the Philippine General Hospital Breast Cancer Center, including Filipino women aged 18–60 yr, with no comorbidity Controls: 75 women aged 18–59 yr, without prior diagnosis of cancer and without family history of breast, ovarian, and endometrial cancer in first-degree relatives; controls were recruited through posters, social media advertisements, and by word of mouth	Breast, incidence	Serum PFOA concer 1st quartile (0.56–1.47 ng/mL) 2nd quartile (1.50–1.77 ng/mL) 3rd quartile (1.77–2.30 ng/mL) 4th quartile (2.31–8.46 ng/mL) Trend-test <i>P</i> -value, 0	ntration (OF 18 14 21 13 0.380	<pre>R): 1 0.64 (0.21-1.90) 1.05 (0.38-2.93) 0.44 (0.14-1.36)</pre>	Age, region of residence	Exposure assessment critique: Key strengths were that serum levels represent the combined exposure through all exposure pathways; measurement error low. Key limitations were that if breast cancer alters ADME of PFAS there could be possible differential exposure misclassification as serum collected at case identification		

Table 2.4 (continued)								
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
Velarde et al.	Exposure assessment	Breast, incidence	Serum PFOS concer	tration (OF	ξ):	Age, region of	(however, cases	
(2022) Philippines	method: quantitative serum measurements;		1st quartile (0.17–2.15 ng/mL)	9	1	residence	had not received neoadjuvant	
2018 Case–control	analytical method was state-of-the-art; a		2nd quartile (2.20–3.02 ng/mL)	11	1.36 (0.42–4.52)		chemotherapy before blood sample	
(cont.)	single serum sample was collected from cases and non-cancer community controls; measured 12 PEAS		3rd quartile (3.05–3.82 ng/mL)	11	1.25 (0.38–4.17)		used for PFAS measurements); single samples at time of case and control identification	
			4th quartile (3.90–23.03 ng/mL)	35	2.38 (0.81–7.31)			
	but did not measure isomers of PFOA or PFOS		Trend-test <i>P</i> -value, (0.400			may not reflect exposure at crucial windows in cancer development; no info on other carcinogens (e.g. alcohol and smoking). <i>Other strengths</i> : histologically confirmed malignant breast cancer. <i>Other limitations</i> : lack of adjustment for important confounders; small sample size and limited statistical power; no information concerning hormone-receptor status; cross sectional design; strategy for recruiting controls could have induced a control selection bias if people positively	

Table 2.4 (continued)								
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments	
Velarde et al. (2022) Philippines 2018 Case-control (cont.)							responding to advertisement had healthier lifestyle and a higher medical awareness compared with the source population for cases; analysis by each PFAS separately did not account for isomers of PFOA or PFOS; measured a variety of other exposures but analysed separately from PFAS relative to outcome. <i>Other comments</i> : All participants had no prior use of hormonal contraceptives or HRT within 1 mo from the last day of use of an oral agent, or within 6 mo from the last day of use of an intramuscular agent.	

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
<u>Li et al. (2022b)</u>	Cases: 373 cases	Breast, incidence	Plasma PFOA conce	ntration (O	R):	Age, BMI, smoking	Exposure assessment		
China 2012–2016	recruited at diagnosis from the Tianjin		1st quartile (< 2.4 ng/mL)	96	1	history, age at menarche, age of	<i>critique</i> : Key strengths		
Case-control	Medical University Cancer Institute and		2nd quartile (2.24–3.35 ng/mL)	67	0.66 (0.41–1.08)	menopause, parity, breastfeeding	were that plasma levels represent the combined exposure through all exposure pathways; blood samples of cases were collected within a week after breast cancer diagnosis and before treatment; measurement error		
	Hospital Controls: 657 controls were randomly selected from the participants in the Chinese National Breast Cancer Screening Program; cohort from a time period similar to that		3rd quartile (3.35–5.11 ng/mL)	83	1.19 (0.75–1.90)	duration, use of estrogen or estrogen replacement therapy, family history of breast cancer, education, monthly household income per capita, red meat consumption.			
			4th quartile (≥ 5.11 ng/mL)	127	2.83 (1.79-4.49)				
			Continuous (per standard deviation on natural log scale)	373	1.57 (1.31–1.89)				
			Trend-test <i>P</i> -value, 0.000			pickled, fried,	low.		
	of the cases	Breast, incidence nent itive nents; d art; mple ood re	Plasma PFOS concer	ntration (OI	R):	smoked, and barbecued food consumption.	Key limitations were that if breast cancer alters ADME of PFAS there could be possible differential exposure misclassification, as plasma was collected after diagnosis in cases; single samples at time of case hospitalization may not reflect exposure at crucial windows in		
	Exposure assessment method: quantitative plasma measurements; analytical method was state-of-the-art; a single blood sample was collected; blood was collected at the hospital before treatment		1st quartile (< 7.45 ng/mL)	119	1				
			2nd quartile (7.45–12.18 ng/mL)	85	0.61 (0.40-0.95)				
			3rd quartile (12.18–17.72 ng/mL)	83	0.58 (0.37–0.91)				
			4th quartile (≥ 17.72 ng/mL)	86	0.64 (0.41–1.00)				
			Continuous (per standard deviation on natural log scale)	373	0.81 (0.68–0.96)				
			Irend-test P-value, (0.002			cancer development.		

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Li et al. (2022b) China 2012–2016 Case–control (cont.)		Breast, incidence	Plasma PFOA conce deviation increase of ER– ER+ PR– PR+	ntration (O n natural lo 96 218 131 183	R for one standard g scale): 1.08 (0.82–1.41) 1.47 (1.19–1.80) 1.03 (0.81–1.30) 1.36 (1.09–1.69)	Age, BMI, smoking history, age at menarche, age of menopause, parity, breastfeeding duration, use of estrogen or estrogen replacement therapy, family history of breast cancer, education, monthly household income per capita, red meat consumption, pickled, fried, smoked, and barbecued food consumption.	Other strengths: histologically confirmed malignant breast cancer; adjustment for important confounding variables; detailed information on diet; available information on the ER/PR status of breast cancer; large number of case and controls allowed for stratified analyse: with good statistical power. Other limitations: cross-sectional		

Table 2.4 (continued)									
Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments		
Li et al. (2023) Shijiazhuang, Hebei Province, China January to May 2022 Case–control	Cases: 150 recent hospital-based diagnoses of thyroid cancer, histologically confirmed by the hospital pathology unit, among adults aged 20–78 yr residing in Shijiazhuang for 10 yr or longer Controls: 150 healthy individuals, aged 26– 83 yr, receiving routine physical examinations and residing in Shijiazhuang for 10 yr or longer and without thyroid nodules or thyroid disease; controls were individually matched to cases on sex and age $(\pm 5 \text{ yr})$ Exposure assessment method: plasma measurements of all participants	Thyroid, incidence	Plasma PFOA conce 1st tertile 2nd tertile 3rd tertile Continuous (per unit on natural log scale) Trend-test <i>P</i> -value, (ntration (O NR NR 150 0.006	R): 1 0.14 (0.05–0.39) 0.32 (0.15–0.69) 0.78 (0.52–1.17)	Age, sex, BMI, smoking status, drinking status, education, household income	Exposure assessmentcritique:Key strengthswere that plasmalevels representthe combinedexposure through allexposure pathways;measurement errorlow.Key limitationswere that if thyroidcancer alters ADMEof PFAS therecould be possibledifferential exposuremisclassification;single samplesat time of casehospitalization maynot reflect exposureat crucial windows incancer development.Other limitations:Selection bias due toselection of controlsamong participantswho were undergoingroutine physicalexamination;potential for reversecausation.		

Table 2.4 (continued)

Reference, location, enrolment/ follow-up period, study design	Population size, description, exposure assessment method	Cancer type (histopathology), incidence or mortality	Exposure category or level	Exposed cases or deaths	Risk estimate (95% CI)	Covariates controlled	Comments
Li et al. (2023) Shijiazhuang, Hebei Province, China January to May 2022 Case-control (cont.)		Thyroid, incidence	Plasma PFOS concer 1st tertile 2nd tertile 3rd tertile Continuous (per unit on natural log scale) Trend-test <i>P</i> -value, 0	ntration (OR NR NR NR 150 0.655	1 0.68 (0.33-1.41) 1.21 (0.60-2.45) 1.02 (0.77-1.36)	Age, sex, BMI, smoking status, drinking status, education, household income	

ADME, absorption, distribution, metabolism, and excretion; AL, Alabama; APFO, ammonium perfluorooctanoate; approx., approximately; BMI, body mass index; CA, California; CI, confidence interval; CPS-II, Cancer Prevention Study II; CYP, cytochrome P450; DDE, dichlorodiphenyldichloroethylene; DDT, dichlorodiphenyltrichloroethane; E3N, Etude épidémiologique auprès de femmes de la Mutuelle générale de l'Education nationale; ER, estrogen receptor; *N*-EtFOSAA, 2-(*N*-ethyl-perfluorooctane sulfonamido) acetic acid; β -HCH, β -hexachlorocyclohexane; HR, hazard ratio; HRT, hormone replacement therapy; ICD, International Classification of Diseases; IQR, interquartile range; MET-h, metabolic equivalent of task per hour; MHT, menopausal hormone therapy; MN, Minnesota; mo, month(s); NC, not calculated; NHL, non-Hodgkin lymphoma; NR, not reported; NTUH, National Taiwan University Hospital; OH, Ohio; OR, odds ratio; PCB, polychlorinated biphenyl; PFAS, per- and polyfluoroalkyl substances; PFCA, perfluoroalkyl carboxylic acid; PFHPS, perfluoroheptanesulfonic acid; PFNA, perfluorononanoic acid; PFOA, perfluorooctanesulfonyl fluoride; ppm, parts per million; PR, progesterone receptor; RR, relative risk; SIR, standardized mortality ratio; TFE, tetrafluoroethylene; US, United States; USA, United States of America; vs, versus; WQS, weighted quantile sum; WV, West Virginia; yr, year(s).

few women were included in this occupationally exposed cohort and there were few cases of breast cancer, limiting the ability to draw conclusions. The low number of cases may have been further affected by residential migration out of Minnesota and Wisconsin. Linkage to a cancer registry was a strength here when considering breast cancer, since it is not a cancer with a high rate of fatality and relying on NDI linkage alone would underestimate cases.]

Alexander et al. (2003) investigated a cohort of 2083 (353 women) PFOS-exposed production workers (Section 2.1.2). Mortality follow-up was conducted using linkage to the NDI until 1998. PFOS exposure was estimated based on a JEM that was validated from a subset of workers from whom blood samples had been collected. Only 17% of the cohort were women. There were two breast cancer-specific deaths identified (both among workers who only held non-exposed jobs); the resulting SMR was very imprecise. [The Working Group noted that there were very few women included in this occupationally exposed cohort and only 2 cases of breast cancer, both among workers holding non-exposed jobs, limiting the ability to draw conclusions. Further, there was little information available for confounding adjustment. The study relied on NDI linkage to identify cases, which would have underestimated the number of breast cancer cases, given the favourable survival after diagnosis, resulting in non-differential outcome misclassification that probably caused bias towards the null.]

Steenland and Woskie (2012) conducted a study of PFOA-exposed workers at the polymer-production plant in Parkersburg, West Virginia, USA (see Section 2.1.3). There were 5791 workers (19% women) who were employed for \geq 1 day between 1948 and 2002 and for whom there were sufficient work history details to estimate PFOA exposure using a JEM informed by a subset with measured PFAS levels. A total of 4 deaths related to breast cancer were observed during follow-up from 1952 to 2008; mortality was not elevated overall, nor did it increase with quartile of estimated PFOA exposure. [The Working Group noted that although this study was in a highly exposed cohort, it was not well powered for breast cancer evaluation because of the small proportion of women, the few breast cancer-related deaths identified, and the lack of incidence data.]

Barry et al. (2013) conducted an investigation of community residents and workers who were exposed to PFOA from a polymer-production plant in the West Virginia and Ohio region, USA (Section 2.1.5). The study included 32 254 community residents and workers (17 360 women) who had a measurement of serum PFOA between 2005-2006, had participated in at least one survey between 2008 and 2011, and for whom either environmental or occupational modelled cumulative PFOA estimates were available. There was a modest inverse association between estimated cumulative PFOA exposure levels (natural log-transformed) and validated breast cancer (559 cases) with a hazard ratio of 0.94 (95% CI, 0.89-1.00), and results remained similar with a 10-year lag. [The Working Group noted that this study was informative because of its large size and consideration of cancer risk in highly exposed community members and in people exposed occupationally. It also considered confounders including education and alcohol intake, although it did not report information on other established breast cancer risk factors such as reproductive history. However, these results were based on estimated PFOA serum levels using data from 2005-2006, which may not include the most etiologically relevant time window. Finally, this study did not include information on breast cancer characteristics, including hormone receptor-related tumour subtypes, and presented results for breast cancer overall, which could mask any subtype-specific associations.]

Ghisari et al. (2017) evaluated the association between serum PFAS concentrations and breast cancer risk in a nested case-control study of pregnant nulliparous women in Denmark (see Section 2.1.7). PFAS, including PFOA and PFOS, were measured in blood samples collected during the first trimester of pregnancy (1996– 2002), and breast cancer cases in the mothers were ascertained using linkage to a nationwide cancer registry, with follow-up until 2010. The study included 158 cases of breast cancer and 215 randomly selected controls. After adjusting for confounders, no association was observed between serum PFOA or PFOS concentrations and breast cancer incidence. However, when considering interactions with cytochrome P450 (CYP) family member 19 (CYP19, aromatase), which acts on the aromatization of androgens to estrogens, increases in levels of both PFOA and PFOS were associated with a notably higher incidence of breast cancer among women who had the CC genotype (relative risk for a 1-unit increase in natural log-transformed PFOA, 7.24; 95% CI, 1.00-52; and relative risk for a 1-unit increase in natural log-transformed PFOS, 6.42; 95% CI, 1.08–38.3), with significant P values for interaction (for PFOA, P = 0.047; for PFOS, P = 0.055). [The Working Group noted that this study had a number of important strengths, including serum PFAS levels that were measured at baseline and adjustment for several relevant breast cancer risk factors. Pregnancy may be an important window of susceptibility during which exposures may be particularly relevant for subsequent risk of breast (Terry et al., 2019). The study was also somewhat underpowered to investigate interactions with genotype and had very few years of follow-up after pregnancy, therefore focusing on premenopausal breast cancer. This study did not include information on diagnoses of postmenopausal breast cancer or other characteristics, including hormone-receptor tumour subtypes, and presented results for breast cancer

overall, which could mask any subtype-specific associations.]

Hurley et al. (2018) analysed serum PFAS levels in relation to breast cancer risk in a nested case-control study within the prospective CTS cohort in the USA (see Section 2.1.8). Breast cancer cases were identified by linkage to cancer registries and were analysed in relation to blood samples collected on average 35 months after a cancer diagnosis and any treatment (range, 9 months to 8.5 years). Average serum PFOA and PFOS levels in this cohort (median in controls, PFOA, 2.48 ng/mL, and PFOS, 6.95 ng/mL) were generally lower than those measured in previous studies (e.g. Ghisari et al., 2017), with the exception of PFOA in the study by <u>Wielsøe et al. (2017)</u>. Among the 902 cases and 858 controls with serum PFAS concentrations, there was little evidence for an association between breast cancer and either PFOA or PFOS. There was also no association observed when the analyses were limited to either premenopausal or postmenopausal breast cancers or when considering the combined ER/PR status of the tumour. [The Working Group noted that this study included several established breast cancer risk factors and was able to consider stratification by menopausal status and hormone-receptor status, which are important factors to consider. However, the collection of blood samples on average 35 months after a case diagnosis was a major limitation as it was unclear whether these measurements reflect the relevant etiological window for breast cancer or whether they may have been influenced by breast cancer or any treatment.]

Mancini et al. (2020a) investigated the association between serum PFAS measures and breast cancer risk in a nested case–control study in the E3N cohort of women in France (Section 2.1.10). Blood samples were collected in the period 1994–1999, and women were followed for breast cancer until 2013. There were 194 cases of postmenopausal breast cancer and 194 matched controls. For PFOA, the association for all breast cancers was elevated in the second quartile but not in the third and fourth quartiles. When stratifying by ER and PR status, this increase in risk for the second quartile was driven by ERor PR- tumours (e.g. quartile 2 versus quartile 1 for ER-, OR, 7.73; 95% CI, 1.46-41.08), although estimates were imprecise. A non-monotonic association was also observed for increasing serum levels of PFOS, with higher ORs in the second and third quartile, and associations that were elevated but with wide confidence intervals for the fourth quartile. However, a monotonic trend with increasing PFOS levels was observed for ER+ and, separately, PR+ tumours (e.g. quartile 4 versus quartile 1 for ER+, OR, 2.33; 95% CI, 1.11–4.90). [The Working Group noted that this study was particularly informative since serum samples were collected prospectively, there was a long follow-up period, and the authors were able to evaluate how associations varied by ER or PR status of the tumour, although the confidence intervals were wide.]

Chang et al. (2023) conducted a nested casecontrol study within the PLCO Cancer Screening Trial (Section 2.1.11). This study included 621 cases of invasive postmenopausal breast cancer diagnosed until November 2013 and 621 controls in postmenopausal women who were selected with matching on age at baseline, date of blood draw, and baseline use of hormone replacement therapy. There was no association between PFOA, by quartiles of exposure, and overall breast cancer risk. The ORs for serum PFOS, categorized in quartiles, were elevated but mainly with wide confidence intervals in relation to overall breast cancer risk. However, for PFOS, associations were evident for hormone receptor-positive breast cancer (ER+/PR+ quartile 3 versus quartile 1, OR, 2.19; 95% CI, 1.21-3.98 and quartile 4 versus quartile 1, OR, 1.89; 95% CI, 0.97-3.69). For PFOA, there was a non-monotonic positive exposure-response relation observed for ER-/ PR- tumours, with wide confidence intervals (ER-/PR-: quartile 3 versus quartile 1, OR, 2.23; 95% CI, 0.90–5.54; and quartile 4 versus quartile 1, OR, 1.62; 95% CI, 0.62–4.23). [The Working Group noted that this study was very informative because it was the largest prospective study evaluating prediagnostic serum PFAS levels in relation to breast cancer risk. The findings from this report were strengthened by the evaluation of differences in joint ER/PR status of the tumour. However, the assessment of PFAS levels using untargeted measurement methods limited direct comparisons with other studies. Finally, these results were generalizable only to postmenopausal women, because premenopausal breast cancer cases were not included.]

In a case-control study nested in the CHDS cohort in California, USA (see Section 2.1.12), Cohn et al. (2020) estimated the relation between maternal serum PFAS levels during pregnancy and the daughter's risk of breast cancer by age 52 years. There were 102 cases identified using validated self-report and registry linkage and they were matched to 310 controls. No association was observed for maternal PFOA exposure in utero in relation to breast cancer risk in the daughters, although the specific results were not reported. Maternal PFOS exposure in utero was inversely associated with breast cancer risk in daughters in a model that included terms for log,-transformed N-ethyl-perfluorooctane sulfonamido acetic acid (N-EtFOSAA), which is a precursor of PFOS, log₂-transformed total cholesterol, and their interaction. The OR for the fourth quartile median versus the first quartile median (an increase of 3.15 ng/mL) in log₂-transformed maternal PFOS was 0.3 (95% CI, 0.1-0.9). [The Working Group noted that although this study was unique in its focus on maternal serum PFAS levels in relation to daughter's breast cancer risk, it did not incorporate other measures of PFAS during childhood, adolescence, or adulthood, which may also be relevant. Additionally, the case counts were small, especially with stratification, which meant that interpretation of these findings was challenging and that the findings were not easily comparable to those of other studies. This study mainly focused on premenopausal breast cancer diagnoses or did not include other breast cancer characteristics, including hormone-receptor tumour subtypes, and presented results for breast cancer overall, which could mask any subtype-specific associations.]

Li et al. (2022a) followed more than 60 000 individuals (more than 28 000 of whom were women) who lived in Ronneby municipality in Sweden between 1985 and 2013; approximately one third of the participants were exposed to water contaminated with PFAS, primarily with PFOS and PFHxS and, to a lesser extent, PFOA (Section 2.1.13). Exposure assessment was based on annual residential addresses and information on drinking-water supply, and cases were identified on the basis of linkage to the cancer registry until 2016. With 681 cases of female breast cancer identified, there was no evidence of an excess risk of breast cancer: SIRs were below the null and were similar for women both with "never-high" or "ever-high" exposure living at an address supplied with PFAS-contaminated water compared with an external reference group. In the internal cohort comparison analysis, there was no difference in the hazard ratios for breast cancer across categories based on estimated duration or timing of exposure. [The Working Group noted that although this study included a large general population sample with a strong exposure contrast and a near-complete registry-based case identification, there was limited control for confounding, particularly for established breast cancer risk factors such as education and reproductive history, and the mixed exposure to multiple PFAS did not allow for the identification of associations with individual compounds. This study did not incorporate information on other breast cancer characteristics, including hormone-receptor status, and it presented results for breast cancer overall, which could mask any subtype-specific associations.]

Feng et al. (2022) evaluated the association between plasma PFAS and breast cancer risk in an ongoing prospective study in Shiyan, China, of retired workers from an automotive company (Section 2.1.14). Incident breast cancer cases were identified by medical record review or death certificates. The nested case-cohort sample included a random subcohort of 990 participants and all non-subcohort participants identified with incident breast cancer (n = 213). The random subcohort included 13 cases of breast cancer, thus there was a total of 226 incident breast cancer cases. Plasma PFAS levels were quantified for the entire case-cohort sample. Increasing levels of serum PFOA were associated with a higher risk of breast cancer (natural log-transformed PFOA levels, HR, 1.35; 95% CI, 1.03–1.78; quartile 4 versus quartile 1, HR, 1.69; 95% CI, 1.05–2.70), but no increase in risk was observed for PFOS. The association for PFOA was similar when the analysis was restricted to postmenopausal women. [The Working Group noted that this study provided compelling evidence, using prospective sample collection and a case-cohort design with adjustment for many potential confounders. However, it was unable to explore how this association may vary by hormone-receptor tumour subtype and, by presenting results for breast cancer overall, could mask any subtype-specific associations.]

Winquist et al. (2023) evaluated the concentrations of several PFAS compounds in serum samples in relation to breast cancer incidence as part of a nested case-cohort study in the ACS prospective CPS-II LifeLink Cohort. Between 1998 and 2001, participants were selected if they had no previous cancer diagnosis and donated blood samples at a median age of 70 years (69 years for women) (Section 2.1.21). Cancer cases were identified by self-report and using NDI linkage. There were 786 cases of postmenopausal breast cancer and 499 women in the subcohort, and the median follow-up time was 14 years. Higher serum PFOA and PFOS concentrations were not related to higher incidence of breast cancer (PFOA, quartile 4 versus quartile 1, HR, 0.82; 95% CI, 0.57–1.17; PFOS, quartile 4 versus quartile 1, HR, 0.70; 95% CI, 0.48–1.01). [The Working Group noted that this study included prospective sample collection and a case–cohort design with adjustment for many potential confounders. However, the blood sample used to measure PFAS was collected at a median age of 69 years, which is after the peak age at diagnosis for breast cancer. Further, it did not consider variability in the associations by hormone-receptor tumour subtype and by presenting results for breast cancer overall could mask any subtype-specific associations.]

(b) Case-control studies and meta-analyses

The eight case-control studies contributing evidence on PFOA and PFOS exposure and risk of breast cancer in women are described below. [The Working Group noted that nearly all the case-control studies listed below had a design in which exposure was measured after disease diagnosis, thus reverse causation bias cannot be excluded. Indeed, the disease could potentially affect PFOA and PFOS internal levels as a consequence of physiological changes associated with tumour development, such as altered albumin levels or altered glomerular filtration rate. Despite this concern, the Working Group considered it unlikely that such alterations would be observed in patients with breast cancer at diagnosis and that there was too little information available concerning the toxicokinetics of PFOA and PFOS in patients with cancer to reach conclusions on the presence of reverse causation bias in case-control studies. In all the case-control studies, except for that by Vieira et al. (2013) in which PFOA serum levels were inferred from geocoded addresses, environmental exposure, and toxicokinetic models, exposure classification was based on PFOA and PFOS measurements in blood samples collected only once when entering the study. The Working Group also questioned at the time of diagnosis reflect exposure during the most relevant windows of exposure with regard to breast cancer risk. Indeed, for cancer (and in particular for breast cancer) the relevant time windows of exposure are several years before diagnosis, so that the levels of exposure at the time of diagnosis may not be pertinent to the disease. Nevertheless, since there is some evidence that single samples may represent longterm average levels of exposure to PFOA over a 5-8-year period (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications.iarc. who.int/636), and since there was limited information concerning the most relevant window of exposure to PFOA and PFOS with regard to breast cancer, the Working Group could not come to a conclusion on the informativeness of these studies.]

whether blood PFOA and PFOS levels measured

Vieira et al. (2013) investigated the relation between exposure to PFOA and breast cancer risk among residents living near the polymer-production plant in Parkersburg, West Virginia, USA, in two case-control studies (Vieira et al., 2013) (Section 2.1.22). In the case-control study in West Virginia and Ohio, 4057 cases of female breast cancer diagnosed from 1996 through 2005 in five Ohio counties and eight West Virginia counties were included in the study, whereas controls comprised all other cancers registered during the same study period (excluding kidney, pancreatic, testicular, and liver cancers). Of the 13 counties included in this study, 6 areas were classified as contaminated public water districts, and living within a contaminated water district was the exposure of interest for the main analyses. In the other case-control study, additional analyses were conducted only for Ohio counties for which it was possible to geolocalize the street addresses for all cancer cases and then to estimate serum PFOA concentration as an exposure metric assuming 10 years residence and latency. All analyses were restricted to women and

adjusted for age, race (White or non-White, only for Ohio), smoking status, and health insurance provider. No evidence of associations with breast cancer risk was observed in the two studies. [The Working Group noted that other types of cancer were used as controls and that the presence of exposure misclassification bias cannot be excluded because the exposure was estimated on the basis of the address of residence at diagnosis, although participants could have moved between water districts. Potential exposure misclassification was probably non-differential, so the bias would be expected to be towards the null. Finally, the Working Group noted that several important confounding variables, mainly related to reproductive history (e.g. age at menarche, number of pregnancies, age at menopause) and exogenous hormone exposure (e.g. use of contraceptive pill, use of hormone replacement therapy), were not included in the analyses.]

Bonefeld-Jørgensen et al. (2011) conducted a case-control study among women of Inuit descent in Greenland. Between 2000 and 2003, 31 women with breast cancer were recruited at the Dronning Ingrids Hospital in Nuuk, where all breast cancer cases of Greenland are registered. Women acting as controls (n = 115) were selected by frequency-matching on age and district of residence from a cross-sectional study conducted in 2000 (Côté et al., 2006) and from the Arctic Monitoring and Assessment Programme study conducted between 1999 and 2005 (Deutch et al., 2007). This study was extended by Wielsøe et al. (2017), who enrolled 66 cases and 62 controls during 2011–2014. Although cases were always recruited at the Dronning Ingrids Hospital in Nuuk, controls enrolled during 2011–2014 were selected by frequency-matching on age and district of residence among patients admitted to the hospital in the department of orthopaedic surgery, or to the department of gynaecology and obstetrics because of the diagnosis of non-malignant abnormities in the uterus, ovaries, or breast. Controls recruited from 2000 to 2003

were then reduced to 1 control per case, so that the final study population included 77 cases of breast cancer and 81 controls. [The Working Group noted that it was unclear how the controls for the period 2000-2003 were selected from the two surveys. Moreover, the authors did not explain on which criteria the final study population was selected, so that selection bias could not be excluded. The Working Group also noted that controls enrolled between 2011 and 2014 were hospital patients attending the orthopaedic surgery department or with non-malignant abnormalities in the uterus, ovary, or breast. If there were an association between PFOA or PFOS exposure and the health conditions affecting the patients recruited as controls, selecting controls from among women admitted at the hospital could have introduced a bias.] Blood samples were collected at diagnosis for the cases and when enrolled in the study for the controls, and PFOA and PFOS serum levels were measured for both cases and controls. After adjusting for age, BMI, serum cotinine levels, number of pregnancies, and breastfeeding, the authors reported a positive association between serum levels of PFOA (OR per unit increase of PFOA, 1.26; 95% CI, 1.01-1.58) and of PFOS (OR per unit increase of PFOS, 1.02; 95% CI, 1.01-1.03) and breast cancer risk. [The Working Group noted that no information was available concerning the delay between diagnosis and the collection of blood samples used for PFAS measurements, thus it could not be excluded that patients with breast cancer changed their behaviours after diagnosis and that this change could have an impact on circulating levels of PFAS. Moreover, the authors did not specify whether the women enrolled as cases had received cancer treatment before blood samples were collected. This could potentially affect PFOA and PFOS internal levels because of physiological changes associated with the treatment. The lack of evidence on the impact of cancer treatment on PFOA and PFOS internal levels did not permit the Working Group to

reach a conclusion on the possible presence of bias. Finally, the Working Group noted that the median serum level of PFOS (45.60 ng/mL) for breast cancer cases among women recruited between 2000 and 2003 was more than double that measured in controls (18.06 ng/mL) selected in the same time period but also those measured in cases (19.35 ng/mL) and controls (18.20 ng/mL) recruited between 2011 and 2014. The authors did not provide an explanation or interpretation of this important variation in PFOS levels that would be expected to have had an impact on the results.] Additional analyses to explore interactions between gene polymorphisms and PFOA and PFOS serum levels with regard to breast cancer risk were conducted using the same study population as Bonefeld-Jorgensen et al. (2011), including 31 cases and 115 controls (Ghisari et al., 2014). [The Working Group noted that the interaction between genotype and exposure was not formally tested by Ghisari et al. (2014), so that the results were considered to be not informative. Moreover, the limited number of cases included had a strong impact on the statistical power of the analyses, preventing correct interpretation of the results.]

Between 2014 and 2015, 120 cases of histologically confirmed breast cancer were consecutively recruited from women attending the National Taiwan University Hospital, China (Tsai et al., 2020). A total of 119 women without any history of malignancy were recruited as controls between 2014 and 2016 through advertisements of posters and flyers at the hospital and in the community. All participants answered a questionnaire and donated a blood sample at enrolment. For the cases, blood samples were collected before receiving any treatment for breast cancer. Plasma PFOA and PFOS levels were measured for both cases and controls. Adjusted ORs of 0.89 (95% CI, 0.59-1.34) and 1.07 (95% CI, 0.64-1.79) were calculated for a natural log 1-unit increase in PFOA and PFOS, respectively. When the analyses were stratified on the basis of age of participants (> 50 years versus \leq 50 years), an adjusted OR of 2.34 (95% CI, 1.02-5.38) for PFOS exposure was observed for women aged \leq 50 years, whereas an adjusted OR of 0.62 (95% CI, 0.29-1.29) was observed for women aged > 50 years. When also considering the tumour ER status, PFOS exposure was significantly associated only with risk of ER+ breast cancer in women aged \leq 50 years (OR per unit increase in natural log-transformed PFOS levels, 3.25 (95% CI, 1.29–8.23). The other results were generally not positive. [The Working Group noted that the small number of cases and controls included in the study could have limited the statistical power of the analyses, especially when stratifying on the basis of age and tumour hormone-receptor status. Moreover, the Working Group noticed that the recruitment strategy for the controls could have induced a selection bias, because people positively responding to advertisement through posters and flyers could have had healthier lifestyles and a higher medical awareness compared with the source population for cases.]

A multicentric hospital-based case-control study conducted in Japan between 2001 and 2005 included 401 cases of histologically confirmed invasive breast cancer (Itoh et al., 2021). Controls were selected from among individuals attending hospital medical check-ups during the study period who had not been diagnosed with cancer. The controls were matched individually to cases on age and residential area (urban or rural). At recruitment, all participants completed a self-administered questionnaire, and a blood sample was collected. Among participants serving as cases, blood samples were collected before any cancer treatment. Multivariable analysis showed a precise inverse association between risk of breast cancer and serum concentrations of PFOA (OR for fourth quartile versus first quartile, 0.21 (95% CI, 0.10-0.45) and PFOS (OR for fourth quartile versus first quartile, 0.14; 95% CI, 0.07–0.31). Results from models that additionally adjusted for vegetable intake, fish and shellfish

intake, calendar year of blood sampling, and quartiles of serum lipid-adjusted total concentration of PCBs remained virtually unchanged. The association between PFOA or PFOS and risk of breast cancer did not differ accordingly to menopausal status or hormone-receptor status. [The Working Group noted that the use of medical check-up examinees as controls may have caused selection bias because of their higher medical awareness and possibly different socioeconomic status compared with the source population for cases. Moreover, educational and socioeconomic status were not included as adjustment variables in the main analyses.]

Velarde et al. (2022) recruited 75 cases of histologically confirmed breast cancer in women aged 18-60 years in the Philippines, with no comorbidity, who visited the Philippine General Hospital between January and December 2018. Patients who underwent neoadjuvant chemotherapy were excluded from the study. Controls were randomly recruited through posters, social media advertisements, and by word of mouth. The control group included 75 women within the age range of 18-59 years, without a previous diagnosis of cancer and without a family history of breast, ovarian, or endometrial cancer in first-degree relatives. This study did not observe any associations between serum PFOA and PFOS levels and breast cancer risk. [The Working Group noted that the study did not adjust for important confounding variables, such as anthropometric characteristics, reproductive history, and hormone exposure. Indeed, the final model included only age and region of residence as covariables. Moreover, the small number of included cases and controls limited the statistical power of the analyses and thus the informativeness of the results. Finally, the Working Group noted that the recruitment strategy for the controls may have caused selection bias, because people positively responding to advertisements through posters and social media could have a higher medical awareness or possibly a different

socioeconomic status compared with the source population for cases.]

Li et al. (2022b) conducted a case-control study that included 373 cases of breast cancer and 657 controls, all participants having available blood samples. Cases were recruited at the Tianjin Medical University Cancer Institute and Hospital, China, between January 2012 and December 2016. Diagnosis of malignant breast cancer was confirmed histologically, and a blood sample was collected within 1 week after diagnosis and before receiving any treatment. Controls were randomly selected among women participating in the Chinese National Breast Cancer Screening Program (CNBCSP) cohort. The CNBCSP was launched in 2012 and included women without a history of cancer who lived in four cities (Shijiazhuang, Tangshan, Xingtai, and Handan) in Hebei Province for ≥ 3 years, and were aged 40-74 years (Wu et al., 2023). Both case and control participants answered a questionnaire (including dietary information) at recruitment. Li et al. (2022b) found that plasma concentrations of PFOA were positively associated with breast cancer risk. The authors estimated an adjusted OR for an increase of 1 standard deviation (SD) in natural log-transformed PFOA plasma levels of 1.57 (95% CI, 1.31–1.89). PFOA was more strongly associated with the ER+ (OR, 1.47; 95% CI, 1.19–1.80) and PR+ (OR, 1.36; 95% CI, 1.09-1.69) breast cancer than with receptor-negative tumours. An inverse association was observed between PFOS plasma levels and breast cancer risk, with an OR for one SD increase in natural log-transformed PFOS plasma levels of 0.81 (95% CI, 0.68-0.96). [The Working Group considered as strengths of this study that all cases were histologically confirmed malignant breast cancer, and blood samples were collected within 1 week after diagnosis and before any cancer treatment. Moreover, the Working Group noted that the large number of cases and controls permitted stratified analyses while still ensuring a good statistical power.

The Working Group noted as a limitation the fact that the controls were selected from women participating in the breast cancer screening programme, who may have had a higher medical awareness and possibly different socioeconomic status compared with the source population for cases.]

Three meta-analyses have been conducted on the association between exposure to PFOA and PFOS and breast cancer risk. The first meta-analysis (Jiang et al., 2022) included eight studies: seven case-control studies, among which three were case-control studies nested in prospective cohort studies, and one cross-sectional study. The exposure assessment for all included studies was based on PFOA and PFOS blood measurements. The overall results showed that PFOA was positively associated with breast cancer risk, and the pooled OR was 1.32 (95% CI, 1.19–1.46), whereas PFOS was not associated with breast cancer risk (pooled OR, 1.01; 95% CI, 0.87-1.17). [The Working Group noted that the present meta-analyses included studies with different designs (case-control studies, nested casecontrol studies, cross-sectional study) which could have caused heterogeneity and instability of the pooled OR. Moreover, the Working Group noted that the results were mainly driven by the only cross-sectional study included in the meta-analyses (Omoike et al., 2021) and that this study was identified as the main source of heterogeneity by the authors. Finally, the Working Group noted that the authors counted studies multiple times when performing comparisons between exposure categories.]

In the second meta-analysis, <u>Cong et al.</u> (2023) included eleven studies: nine case-control studies, of which three were nested in prospective cohort studies, one cohort study, and one casecohort study. PFOA and PFOS blood levels were used as the main exposure variable in all studies except for one study in which individual cumulative PFOA serum concentration estimates were calculated retrospectively from 1952 through 2011. The results of the meta-analyses found little evidence of a positive association between PFOA and PFOS and breast cancer risk (pooled OR, 1.07; 95% CI, 0.84-1.38; and pooled OR, 1.01; 95% CI, 0.95-1.08, respectively). The authors observed significant heterogeneity among the included studies for both PFOA ($I^2 = 85.9\%$; P < 0.001) and PFOS ($I^2 = 65.7\%$; P = 0.003). When omitting one study at a time from the pooled analyses, a weakly positive OR was observed for PFOS in relation to breast cancer when excluding Itoh et al. (2021) (pooled OR, 1.02; 95% CI, 1.01-1.03, $I^2 = 2.6\%$; P = 0.41). Results remained unchanged for PFOA. [The Working Group noted that studies having different designs (case-control studies, nested case-control studies, case-cohort study and cohort study) and applying different methods to estimate the exposure were included in this meta-analysis and that this could explain the high observed heterogeneity. Moreover, the results of the meta-analyses seemed to be strongly influenced by the only study that highlighted an inverse association between PFOA and PFOS and breast cancer risk.]

The third meta-analysis on the association between exposure to PFOA and PFOS and breast cancer risk was conducted by Chang et al. and included 11 case-control studies, 5 of which were nested in prospective cohort studies (Changet al., 2024). For all studies included in the meta-analyses PFOA and PFOS levels were measured in blood samples (serum or plasma). The results of the meta-analyses were not consistent with an association between PFOA and PFOS blood levels and the risk of breast cancer overall, but they noted substantial heterogeneity across studies. Indeed, the authors estimated a rate ratio for a natural log-unit increase of PFOA of 0.95 (95% CI, 0.77–1.18; $I^2 = 67\%$; P for heterogeneity, < 0.01) and for a natural log-unit increase of PFOS of 0.98 (95% CI, 0.87–1.11; $I^2 = 54\%$; P for heterogeneity, 0.02). In subanalyses, when limiting to studies with prospectively collected blood samples, there was a positive association

with PFOA (RR, 1.16; 95% CI, 0.96–1.40). [The Working Group noted that this meta-analysis incorporated important subgroup analyses, including by timing of sample collection and tumour subtype. However, there was substantial heterogeneity across the published studies, limiting the informativeness of the results.]

2.4.2 Cancer of the thyroid gland

See <u>Table 2.4</u>.

The Working Group identified six cohort studies and three case-control studies investigating the risk of thyroid cancer associated with PFOA or PFOS exposure. Among the cohort studies, two were occupational cohorts (Leonard et al., 2008; Lundin et al., 2009), one was a combination of general population members and workers (Barry et al., 2013), one was composed of residents in area with highly contaminated drinking-water (Ronneby Register cohort; Li et al., 2022a), one nested case-control study was within the FMC (Madrigal et al., 2024), and one nested case-control study was within the BioMe biobank (van Gerwen et al., 2023). One of the case-control studies was population-based (Vieira et al., 2013) and two were hospital-based (Liu et al., 2022; Li et al., 2023).

(a) Cohort studies

Lundin et al. (2009) conducted a mortality study in a cohort of 3993 employees of an APFO-manufacturing facility in Cottage Grove, Minnesota, USA (see Section 2.1.1). The cohort was followed until 31 December 2002, and 807 decedents were identified. Using rates for the state of Minnesota as the referent, SMRs were calculated for different jobs classified by exposure to APFO (the ammonium salt of PFOA). There was only 1 observed death from thyroid cancer, which was assigned to the "never" exposure group. The SMR for the "never" exposure group was 2.16 (95% CI, 0.05–12.00). [The Working Group noted that the important limitations of the study included the small occupational cohort with only 1 death from thyroid cancer and crude exposure assessment by job classification, which made this study uninformative for the evaluation of an association with thyroid cancer.]

Leonard et al. (2008) conducted a retrospective cohort mortality study for the PFOA cohort in a polymer-production plant in Parkersburg, West Virginia, USA, which included 6027 participants who had worked at the facility between 1948 and 2002 (Section 2.1.3). SMRs were calculated by comparing the observed number of deaths to expected numbers derived from mortality rates for three reference populations (the US population, the West Virginia state population, and an eight-state regional employee population from the same company). There were only 3 observed deaths for thyroid cancer. The SMRs for the cohort from the Parkersburg plant were [3.120] (95% CI, [0.644–9.119]), [2.856] (95% CI, [0.589-8.347]), and [6.286] (95% CI, [1.297–18.369]), respectively, for the three reference populations (the US population, the West Virginia population, and the workers in the same company and region). [The Working Group noted that the major limitation of the study was the limited statistical power to evaluate mortality rates for thyroid cancer because of the small numbers of observed deaths, which made this study uninformative for the evaluation of an association with thyroid cancer.]

Barry et al. (2013) examined PFOA exposures and incident cancers among community residents and workers who were exposed to PFOA from a chemical plant, using the C8 Health Project cohort combined with the worker cohort from the polymer-production plant in Parkersburg, West Virginia, USA (Section 2.1.5). There were 32 254 participants in the entire cohort, with 28 541 participants classified as the community group and 3713 as the occupational group. There were 98 cases of primary thyroid cancer reported. The analysis included 86 cases of validated primary thyroid cancer with complete covariate information. In the total cohort, the hazard ratios for a 1-unit increase in natural log-transformed estimated cumulative PFOA exposure in relation to thyroid cancer were 1.10 (95% CI, 0.95-1.26) for unlagged exposures and 1.04 (95% CI, 0.89-1.20) for exposures lagged by 10 years. When stratified by community residents and workers, the hazard ratios for cumulative PFOA exposure in relation to thyroid cancer were 1.04 (95% CI, 0.89-1.23) and 1.93 (95% CI, 1.00–3.71), respectively, for unlagged exposures, and 1.00 (95% CI, 0.84-1.20) and 1.12 (95% CI, 0.61–2.05), respectively, for exposures lagged by 10 years. In sensitivity analyses, when excluding years before each participant began living or working in the contaminated water districts in the survival models, results were similar to the reported results above. When calculating hazard ratios by PFOA quartile in the total cohort, there was no indication of an exposure-response relation between PFOA exposure and thyroid cancer. However, an exposure-response relation was indicated when calculating hazard ratios by PFOA quartile among the occupational group (8 cases) but not the community group (78 cases). [The Working Group noted as strengths the large cohort, strong exposure contrast, assessment of individual cumulative PFOA exposure, and lagged analyses. Limitations included self-reported cancer cases, the low sample size for thyroid cancer, and lack of evaluation of residents' co-exposure to other PFAS.]

Li et al. (2022a) studied cancer incidence in the Ronneby Register cohort, which included a community of residents in Sweden with high-level environmental exposure to PFAS, dominated by PFOS and PFHxS, in drinking-water (Section 2.1.13). SIRs were calculated by comparing with a regional external reference population (the population of Blekinge County excluding Ronneby municipality) and the national reference population (the whole population of Sweden). By the end of the follow-up on 31 December 2016, there were 17 cases of incident thyroid cancer in men and 48 cases in women. External comparisons and internal comparisons were both performed within the Ronneby Register cohort. To facilitate comparison, Ronneby residents were assigned to mutually exclusive groups: "never-high" and "ever-high" based on the source of drinking-water at their residence. When compared with the regional external reference population, women in the ever-high group had nominally higher estimates (defined as > 25% difference) for cancers of the thyroid than did women in the never-high group, with an SIR of 2.08 (95% CI, 1.19-3.38) in the ever-high group, and 1.38 (95% CI, 0.94-1.95) in the never-high group. However, that relation was not observed among men. In internal comparisons, the never-high group was used as the referent, and the thyroid cancer hazard ratio for different groups was calculated. The authors observed modestly increased point estimates but with wide confidence intervals, which showed limited indications of an exposure-response relation between PFAS exposure and the incidence of thyroid cancer for time period of high exposure ("early-high" in 2004 or earlier versus "latehigh" in 2005 or later) or for duration of time in a high-exposure area ("short-high" for ≤ 10 years versus "long-high" for ≥ 11 years) compared with "never-high" group. [The Working Group noted as strengths the large study population, strong exposure contrast, and unbiased inclusion. The group-based exposure assessment can be assumed to provide unbiased risk estimates but less exposure contrast and broader confidence intervals than would be expected with individual-level estimates. However, even at the group level, there was a large exposure contrast, which was one of the strengths of this study. The main limitations were the crude exposure assessment, not including individual water intake or sources of exposure other than drinking-water, the mixed exposure profile, and the limited information on potential confounders.]

Madrigal et al. (2024) conducted a nested case-control study of papillary thyroid cancer in the FMC, restricting eligibility to women for whom serum samples were collected in their first pregnancy and whose pregnancy resulted in a full-term live birth, with delivery dates from 1987 to 2010 (see Section 2.1.18). Thyroid cancer cases and controls were identified by the nationwide Finnish Cancer Registry and the population registry until 2016. All cases were randomly selected women with primary papillary thyroid cancer diagnosed \geq 3 years after the delivery date, without a history of other cancers. Controls were individually matched to cases on year of delivery (increments of 4-5 years) and age at first birth (increments of 3 years). A total of 800 participants (400 cases of thyroid cancer and 400 controls) were included in the nested casecontrol analysis. No clear pattern was observed in the association between papillary thyroid cancer risk and serum concentrations of PFOA (OR per log₂, 0.90; 95% CI, 0.68–1.19) or PFOS (OR per log₂, 1.14; 95% CI, 0.81–1.95). When stratified by age at diagnosis (< 40 years, \geq 40 years), the associations per each doubling in concentrations of PFOA and PFOS were elevated but imprecise (OR per log₂, 1.20; 95% CI, 0.71–2.01; and OR per log₂, 1.14; 95% CI, 0.68–1.93; respectively) among women diagnosed before age 40 years. However, among women diagnosed at age ≥ 40 years, the associations were inverse or were close to 1.0 (OR per log₂, 0.70; 95% CI, 0.45–1.08; and OR per log₂, 1.01; 95% CI, 0.60-1.71; for PFOA and PFOS, respectively). [The Working Group noted that use of a single prediagnostic sample would result in only minor misclassification of long-term exposure over a period of 5–8 years, on the basis of a simulation study (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications. iarc.who.int/636). The lack of information on pre-pregnancy BMI (a risk factor for thyroid cancer) and the population's low-level exposure

with small exposure contrast were noted as limitations.]

van Gerwen et al. (2023) conducted a case-control study nested within the BioMe biobank of Mount Sinai hospital in New York (see Section 2.1.20). Among 88 cases (57 with < 1 year between sample collection and thyroid cancer diagnosis) and 88 controls, plasma PFOA concentration was not associated with thyroid cancer risk, whereas plasma concentrations of branched PFOS and linear-PFOS were associated with increased thyroid cancer risk (ORs for increment of log₂-plasma concentration of branched PFOS and linear-PFOS were 1.32; 95% CI, 0.99–1.81; and 1.56; 95% CI, 0.99–1.81; respectively). In the sensitivity analysis restricted to 31 cases with > 1 year between sample collection and incident thyroid cancer diagnosis (median time, 3.7 years), log₂-plasma concentrations of branched PFOS and linear-PFOS were also associated with increased risk of thyroid cancer (OR, 3.09; 95% CI, 1.73-6.13; and OR, 2.67; 95% CI, 1.59–4.88; respectively). [The Working Group noted the limited follow-up time in this study and that use of a single prediagnostic sample would result in only minor misclassification of long-term exposure over a period of 5-8 years, on the basis of a simulation study carried out by the Working Group (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications. iarc.who.int/636).]

(b) Case–control studies

One population-based case-control study was conducted by <u>Vieira et al. (2013)</u> among residents of 13 counties in Ohio and West Virginia surrounding the Parkersburg polymer-production facility (see Section 2.1.22). The final data set included 343 cases of thyroid cancer. Controls were defined as all other cancers in the study data set, except for cancers of the kidney, pancreas, testis, liver, and thyroid. All cancer diagnoses were classified as exposed (living within a contaminated water district) or unexposed (not living in a contaminated water district) using geocoding. The AORs varied among the water districts exposed to contaminated water, with the AOR of the overall exposure risk being 1.1 (95% CI, 0.7–1.5). Furthermore, for each case in Ohio, annual PFOA serum levels were calculated by linking environmental, exposure, and toxicokinetics models. The AORs were calculated using individual-level exposure categorized on the basis of the distribution of annual PFOA serum concentrations among the exposed study population. Using the unexposed group as the reference category, the AORs for very high, high, medium, and low individual-level exposures were 0.8 (95% CI, 0.2-3.5), 0.7 (95% CI, 0.2-2.1), 0.9 (95% CI, 0.4-2.3), and 0.9 (95% CI, 0.4-2.3), respectively. [The Working Group noted that the strengths of the study included its focus on a population with high PFOA exposure, the strong contrast in exposure levels, and the estimation of individual-level exposure for a subset of the people. Limitations included the use of other cancers as the referent, the lack of geocoded residence information among participants from West Virginia, and the risk of exposure misclassification (reliance on the address at the time of diagnosis rather than a complete residential history in analyses among Ohio participants).]

Liu et al. (2022) conducted a hospital-based case-control study in the Shandong Provincial Qianfoshan Hospital in Jinan City, Shandong Province, China, from 2016 to 2017. A total of 319 participants (134 cases of thyroid cancer and 185 controls) were included in the case-control analysis. The control group was randomly selected from patients undergoing routine medical visits at the hospital, with normal thyroid B-ultrasound examination, without a history of thyroid disease, and without taking iodine or thyroid hormone drugs during the blood collection. Serum samples of the participants were used to assess exposure to individual PFAS compounds. Serum samples for the case group were collected after the patients had stopped taking thyroid medication for 2 weeks under the guidance of their doctors. Serum samples for the control group were collected when they underwent routine medical visits at the hospital. The associations between serum levels of PFAS (including PFOA and PFOS) and thyroid cancer were examined using logistic regression models. Concentrations of PFAS compounds were categorized into quartiles according to the distribution in the control group. Compared with the first quartile of PFOA concentration, the ORs for the second, third, and last quartiles were 0.24 (95% CI, 0.12-0.50), 0.24 (95% CI, 0.11-0.49), and 0.20 (95% CI, 0.09-0.44), respectively, with a *P* for trend of < 0.001. Compared with the first quartile of PFOS concentration, the ORs for the second, third, and last quartiles of PFOS concentration were 0.81 (95% CI, 0.42-1.53), 0.26 (95% CI, 0.12-0.57), and 0.28 (95% CI, 0.12-0.66), respectively, with a P for trend of 0.001. [The Working Group noted that the limitations of the study included the sampling of serum after diagnosis and treatment, limited exposure contrast, small sample size, and the likelihood of potential reverse causation.]

Li et al. (2023) conducted a hospital-based case-control study in the Fourth Hospital of Hebei Medical University in Shijiazhuang, Hebei Province, from January to May 2022. All cases were newly arising thyroid cancer cases in the hospital, confirmed histologically by the hospital pathology unit, among patients who had resided in Shijiazhuang for \geq 10 years. Controls were healthy individuals attending routine physical examinations in the health examination centre who had resided in Shijiazhuang for ≥ 10 years without thyroid cancer or other malignancies and were individually matched to cases on age $(\pm 5 \text{ years})$ and sex. A total of 300 participants (150 cases of thyroid cancer and 150 healthy controls) were included in the case-control analysis. Plasma samples were collected before the start of thyroid cancer therapy for the cases

and during the physical examination for the controls. The associations between plasma levels of PFAS compounds (including PFOA and PFOS) and thyroid cancer were examined using conditional logistic regression and restricted cubic spline models. Plasma PFAS concentrations were analysed as continuous variables and categorized variables (classified into tertiles according to the distribution among controls). The results showed no consistent indication of a positive exposureresponse relation between plasma PFOA or PFOS and thyroid cancer, with the ORs associated with a 1-unit increase in natural log-transformed levels being 0.78 (95% CI, 0.52-1.17) and 1.02 (95% CI, 0.77-1.36), respectively. Further, compared with the first tertile of PFOA concentration, the OR for the highest tertile of PFOA concentration was 0.32 (95% CI, 0.15–0.69), indicating an inverse association between PFOA and thyroid cancer risk (P for trend, 0.006). However, the restricted cubic spline model did not show this inverse dose-response relation. [The Working Group noted that the study relied on postdiagnostic serum samples, which might have been affected by reverse causation. The Working Group noted that only minor misclassification of long-term exposure because of reliance on a single prediagnostic sample would be expected, according to a simulation study (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications. iarc.who.int/636).]

2.5 Cancers of the digestive tract

2.5.1 Liver cancer

See Table S2.5 (Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: <u>https://publications.iarc.</u><u>who.int/636</u>).

There were 11 epidemiological studies with information on liver cancer. Most were cohort studies, but three were case-control studies (one nested within a cohort), and one was a case-cohort study. Six studies were conducted in the USA (Alexander et al., 2003; Steenland and Woskie, 2012; Barry et al., 2013; Vieira et al., 2013; Raleigh et al., 2014; Goodrich et al., 2022), and one each in Denmark (Eriksen et al., 2009), China (Cao et al., 2022), Italy (Girardi and Merler, 2019), and Sweden (Li et al., 2022a). One included cohorts from multiple countries (Consonni et al., 2013). Five were occupational cohort mortality studies (Alexander et al., 2003; Steenland and Woskie, 2012; Consonni et al., 2013; Raleigh et al., 2014; Girardi and Merler, 2019); for four of these PFOA was the exposure of interest, whereas for one (Alexander et al., 2003) the exposure of interest was PFOS. The three community cohort studies (Eriksen et al., 2009; Barry et al., 2014; Li et al., 2022a) had incident cancer data, as did all three of the case-control studies (Vieira et al., 2013; Cao et al., 2022; Goodrich et al., 2022). Eriksen et al. (2009) reported results for PFOA and PFOS, Barry et al. (2014) reported on PFOA, and Li et al. (2022a) was not able to identify a specific PFAS of interest, but levels of PFOS were the highest in the studied population. Among the case-control studies, Vieira et al. (2013) focused on PFOA, whereas Cao et al. (2022) and Goodrich et al. (2022) reported results for both PFOA and PFOS. One additional study (Olsen et al., 2004) had some cross-sectional data on liver cancer among active and some inactive employees, but it was not considered informative regarding liver cancer incidence or mortality and therefore is not discussed further.

(a) Cohort, case–cohort, and nested case– control studies

<u>Raleigh et al. (2014)</u> studied liver cancer mortality among 4668 PFOA-exposed workers and 4359 unexposed workers at a different plant, all working for \geq 1 year (Section 2.1.1 for more details). Using Minnesota rates as the referent, the SMR for exposed workers was 0.81 (95% CI, 0.35–1.59; 8 deaths from liver cancer). When estimated PFOA air exposure was divided into four quartiles, the SMRs for exposed workers versus the Minnesota population were 1.40, 0.86, 0.75, and 0.00, based on only 4, 2, 2, and 0 deaths from liver cancer, respectively. When exposed workers were compared with non-exposed workers, combined quartiles 1 and 2 of cumulative PFOA exposure showed a hazard ratio of 2.09 (95% CI, 0.69–6.31), whereas combined quartiles 3 and 4 had a hazard ratio of 0.67 (95% CI, 0.14–3.27).

<u>Alexander et al. (2003)</u> studied mortality in a cohort of 2083 production workers (145 deaths) who were exposed to PFOS at a plant in Decatur, Alabama, USA, that produced speciality films and fluorochemicals, and who had worked \geq 1 year at the plant between 1961 and 1997 (Section 2.1.2). On the basis of only 2 deaths from biliary and liver cancer, these authors estimated an SMR for the entire cohort (using an Alabama referent) of 1.61 (95% CI, 0.20–5.82).

Steenland and Woskie (2012) studied mortality from liver and gall bladder cancer among 5791 workers exposed to PFOA at a polymer-production plant in Parkersburg, West Virginia, USA (Section 2.1.3). Compared with non-exposed workers at other plants within the same company, the authors found an SMR of 1.07 (95% CI, 0.51–1.96) based on 10 deaths from liver and gall bladder cancer. By quartile of estimated cumulative exposure, SMRs were 2.39 (95% CI, 0.65–6.13; 4 deaths), 0 (95% CI, 0–1.81; 0 deaths;), 2.01 (95% CI, 0.65–4.68; 5 deaths), and 0.32 (95% CI, 0.01–1.76; 1 death).

[The Working Group noted that, for all these occupational cohort mortality studies (Alexander et al., 2003; Steenland and Woskie, 2012; Raleigh et al., 2014), the numbers of deaths from liver cancer were too small to be informative.]

Eriksen et al. (2009) conducted a case-cohort study (67 patients with liver cancer and 782 cancer-free participants selected randomly from the full cohort) in a general population national cohort of 57 053 people in Denmark. Analyses of liver cancer incidence were done using baseline-measured plasma levels of both PFOA and PFOS (Section 2.1.4). All participants had no previous diagnoses of cancer at the beginning of follow-up. Follow-up for cancer patients ranged from 0 to 12 years (median, 7 years). Analyses of IRRs for liver cancer by quartile of PFOA, using quartile 1 as referent, were 1.00 (95% CI, 0.44-2.23), 0.49 (95% CI, 0.22-1.09), and 0.60 (95% CI, 0.26-1.37), respectively, based on 17, 17, and 16 cases, respectively. Corresponding IRRs for PFOS were 0.62 (95% CI, 0.29-1.33), 0.72 (95% CI, 0.33–1.56), and 0.59 (95% CI, 0.27–1.27). [The Working Group noted that the number of cases was larger in the case-cohort study by Eriksen et al. (2009) (67 cases) compared with other studies reporting on liver cancer after PFOA or PFOS exposure; however, this study was limited to some degree by low exposure contrasts for both PFOA and PFOS.]

Barry et al. (2013) analysed liver cancer incidence in a cohort of 32 254 participants with both low and high exposure to PFOA from drinking-water (with high exposure being similar to the high levels in occupational cohorts), who were living near the Parkersburg polymer-production plant in West Virginia, USA (Section 2.1.5). The median PFOA level measured in all cohort members in 2005-2006 was 26 µg/L [ng/mL], and the mean was 87 µg/L [ng/mL] (whereas in the USA the general population levels were about $4 \mu g/L [ng/mL]$ at the time). Approximately 12% of participants in this study had worked in the Parkersburg plant that was the source of the PFOA contamination. Cancer incidence was determined via interview with confirmation from medical records, or by matching to Ohio and West Virginia cancer registries. Liver cancer hazard ratios per unit natural log-transformed cumulative serum level were 0.73 (95% CI, 0.43-1.23) and 0.74 (95% CI, 0.43-1.26), based on 9 cases, for unlagged and 10-year lagged estimates, respectively. [The Working Group noted

that the exposure–response analysis was based on a continuous variable, with serum levels over time estimated by a model with good correlation ($\rho = 0.71$) to observed serum levels, which were available in 2005 or 2006 for all cohort members. However, the number of deaths from or number of cases of incident liver cancer in <u>Barry et al. (2014)</u> (8 cases) was too small to draw conclusions.]

Consonni et al. (2013) conducted an international cohort mortality study of male workers at six TFE-production sites, who were concomitantly exposed to APFO (or equivalently PFOA) (Spearman correlation, 0.72). [The Working Group noted that the high correlation between TFE and PFOA exposure precluded evaluation of the effects of the individual compounds (Section 2.1.6). At two plants there was also possible exposure to vinyl chloride, a liver carcinogen, but no details were given.] Restricting the cohort to workers who ever had exposure to APFO, the authors reported an SMR for liver and bile duct cancer (versus national rates) of 1.43 (95% CI, 0.57–2.94), based on 7 deaths from liver cancer. The authors reported a trend with increasing cumulative APFO exposure that was estimated on the basis of a JEM, with liver cancer SMRs for low-, medium-, and high-exposure groups of 0.70 (95% CI, 0.02-3.87; 1 death), 1.25 (95% CI, 0.15–4.52; 2 deaths), and 2.14 (95% CI, 0.58–5.49; 4 deaths) (*P* for trend, 0.24).

Girardi and Merler (2019) studied mortality among industrial workers who were exposed to very high levels of PFOA and, to a somewhat lesser extent, PFOS (Section 2.1.9). In a subsample of 120 workers for whom 696 serum samples were available, the geometric mean concentration of PFOA was 4048 ng/mL (a geometric mean of 8862 ng/mL was found in the subgroup of PFOA operators), whereas for PFOS it was 148.8 ng/mL. SMRs and risk ratios for liver cancer mortality in exposed workers (7 deaths) versus the general population (SMR, 2.32; 95% CI, 1.11–4.87) and versus non-exposed workers at another plant (risk ratio, 6.69; 95% CI, 1.71–26.2) were elevated. Relative to non-exposed workers (3 deaths), liver cancer mortality increased by estimated cumulative serum levels of PFOA by tertile, with risk ratios of 3.07 (95% CI, 0.31-30.0; 1 death), 8.39 (95% CI, 1.40-50.3; 2 deaths), and 9.28 (95% CI, 2.07–41.5; 4 deaths), by tertile of increasing serum level. Death from liver cirrhosis was also markedly elevated, based on 6 deaths (SMR, 1.71; 95% CI, 0.77-3.81; and risk ratio, 3.87; 95% CI, 1.18–12.7), compared with the unexposed workers cohort. [The Working Group noted that the authors suggested that the excess of cirrhosis could be due to high exposure to PFOA, as PFOA is a liver toxin. The Working Group noted the excess of cirrhosis could be a sign of confounding by alcohol, which is also associated with liver cancer.]

[The Working Group also noted that the number of liver cancer deaths in the occupational cohort mortality study by <u>Consonni et al. (2013)</u> was too small to be informative. The study by <u>Girardi and Merler (2019)</u> also had only a small number of cases, making it less informative, but was notable for the strong exposure-response relation, with very high PFOA exposure, and good exposure estimation.]

Li et al. (2022a) studied liver cancer incidence in Ronneby, Sweden, in 60 507 residents. One third of households were exposed to relatively high levels of both PFOS and, to a lesser extent, PFOA from drinking-water contaminated by nearby military firefighting operations (based on a subset with serum levels, PFOS being the most elevated). The authors were unable to separate exposures to different PFAS, particularly PFOS and PFHxS (Section 2.1.13). In a sample of 3084 Ronneby residents and 226 non-Ronneby residents, the geometric means for the high-exposure group in Ronneby (n = 2052) were 199, 176, and 11 ng/mL for PFOS, PFHxS, and PFOA, respectively. For men who never resided in a high-exposure area, the SIR for liver cancer (area surrounding Ronneby used as a reference) was 1.12 (95% CI, 0.72-1.66; 24 cases), and for women

it was 0.98 (95% CI, 0.45–1.86; 9 cases). For men ever living in a high-exposure area, the SIR was 1.52 (95% CI, 0.70–2.89; 9 cases) and for women the corresponding estimate was 1.52 (95% CI, 0.41-3.88; 4 cases).

[The cohort study in Ronneby, Sweden, by Li et al. (2022a) had a larger number of cases (n = 25) compared with other studies reporting on liver cancer after PFOA or PFOS exposure, but a limitation was the ecological assignment of exposure on the basis of residence, and some uncertainty regarding the role of PFOS versus that of PFHxS, another PFAS that was present at high levels in the drinking-water.]

Goodrich et al. (2022) conducted a nested case-control study of PFOA and PFOS (baseline measurements), and HCC not of viral origin, in a large multiethnic cohort, with 50 cases and 50 controls (Section 2.1.16). Geometric mean concentrations of plasma PFOA and PFOS did not differ between cases and controls, and the use of continuous measures of PFOA and PFOS did not show any statistically significant positive associations with liver cancer. When restricting the definition of exposure to above the 85th percentile for PFOS (54.9 ng/mL, which corresponded to the 90th percentile in NHANES) and PFOA (8.6 ng/mL), exposure was associated markedly with liver cancer for PFOS (OR for PFOS, 4.50; 95% CI, 1.20-16.00), but not for PFOA (OR for PFOA, 1.20; 95% CI, 0.52-2.80), after adjusting for the matching variables of age, sex, race or ethnicity, and study site. However further adjustment for BMI lowered the OR for high PFOS (> 54.9 ng/mL) relative to low PFOS to 2.90 (95% CI, 0.78-10.00).

(b) Case-control studies

Vieira et al. (2013) conducted two case-control studies of incident liver cancer among residents of 13 counties in Ohio and West Virginia, USA, which included both contaminated and non-contaminated water districts near the same Parkersburg polymer-production plant in West Virginia that was the source of contamination in the population studied by Barry et al. (2013) (see Section 2.1.22). In the first case-control study, cases and controls (all other cancer cases excluding kidney, pancreatic, testicular, and liver cancers) obtained from both Ohio and West Virginia cancer registries were compared with regard to residence in a contaminated or non-contaminated water district. The liver cancer OR for exposed water district residents was 1.1 (95% CI, 0.7-1.6; 23 exposed cases) versus residents in non-contaminated water districts. These authors also conducted a separate case-control study among Ohio residents; cases were participants with liver cancer and controls were participants with other cancers in the Ohio counties, again excluding kidney, pancreatic, testicular, and liver cancers. Exposure in the second study was based on estimated individual serum levels of PFOA at specific addresses at specific points in time. The methods for estimating individual serum PFOA levels from linked environmental, exposure, and toxicokinetics models are described in detail elsewhere (Shin et al., 2011a). The environmental models integrated facility emissions data; fate, and transport characteristics of PFOA; addresses of cases and controls; and hydrogeological properties of the study area to estimate PFOA air and water concentrations from 1951 through 2008. Exposure was the estimated individual serum level 10 years before the diagnosis dates for cases and controls. Relative to non-contaminated water districts (50 cases), the ORs for those with low, medium, and high exposure 10 years before diagnosis were 1.1 (95% CI, 0.4-3.1; 4 cases), 0.9 (95% CI, 0.3-2.5; 4 cases), and 1.0 (95% CI, 0.3–3.1; 3 cases), respectively (there were no cases among those with very high exposure).

<u>Cao et al. (2022)</u> studied 203 cases of incident liver cancer compared with 203 hospital-based controls in a hospital in China during 2019– 2021. [The Working Group noted that the study did not mention how controls were matched.] Serum PFOA and PFOS were measured in study
participants. [The Working Group noted that the timing of collection with respect to cancer diagnosis was not reported.] In cases and controls combined, the mean serum PFOS and PFOA concentrations were 9.8 ng/mL and 8.3 ng/mL, respectively. Log-transformed PFOS (a continuous variable) was associated with liver cancer (OR, 2.609; 95% CI, 1.179–4.029), after adjustment for covariates, but log-transformed PFOA was not (OR, 1.036; 95% CI, 1.002–1.070). [The Working Group considered that the base of log transformation of PFOS was 10; however, it was not specified in the manuscript.]

[The Working Group noted that the three case-control studies had large numbers of cases but suffered from other limitations. The study by Vieira et al. (2013), which included two different case-control studies, had a fairly large number of cases in the first study (179 cases, 23 exposed) but was limited by an ecological assignment of exposure. The second case-control study in Vieira et al. (2013) had fewer cases (61 cases, 11 exposed), with better assessment of estimated individual exposure levels, but was again limited by small numbers of exposed cases. The two other case-control studies by Goodrich et al. (2022) and Cao et al. (2022) showed some positive associations but had their own limitations. In the nested case-control study by Goodrich et al. (2022), positive findings for PFOS were observed only when baseline plasma concentrations were dichotomized between the top 15% and the bottom 85% and were diminished after control for BMI. In the other case-control study by Cao et al. (2022), serum levels were available only at time of diagnosis, making etiological inference difficult.]

(c) Meta-analysis

<u>Seyyedsalehi and Boffetta (2023)</u> published a meta-analysis of liver cancer in relation to PFAS that included all the studies cited in the present monograph as well as two ecological studies judged to be not informative, and one other that was a predecessor of the study by <u>Steenland and</u> <u>Woskie (2012)</u> cited here [for those reasons, the Working Group did not considered this metaanalysis to be informative].

2.5.2 Pancreatic cancer

See Table S2.5 (Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: <u>https://publications.iarc.</u><u>who.int/636</u>).

A total of nine studies investigating the association between PFAS (mainly PFOA) and cancer of the pancreas are presented below according to three different types of population: three studies among workers in chemical plants producing or using PFOA (<u>Steenland and Woskie, 2012;</u> <u>Consonni et al., 2013; Raleigh et al., 2014</u>), three from communities surrounding a plant from which there had been environmental release of PFOA and contamination of public and private water supplies (<u>Barry et al., 2013; Vieira et al.,</u> <u>2013; Li et al., 2022a</u>), and three from studies in the general population with background exposures (<u>Eriksen et al., 2009; Winquist et al., 2023</u>).

Raleigh et al. (2014) investigated mortality and cancer incidence among APFO-production workers (n = 4668) compared with tape and abrasives production workers (n = 4359) in two manufacturing facilities owned by the same company in Minnesota, USA, between 1947 and 2002 (see Section 2.1.1). Hazard ratios for mortality using the unexposed workers as the referent were calculated for quartile-based categories of PFOA exposure created using a task-based JEM; the hazard ratios were 0.32 (95% CI, 0.08–1.35), 0.89 (95% CI, 0.34–2.31), 0.82 (95% CI, 0.32–2.12), and 1.23 (95% CI, 0.50-3.00) on the basis of 18 deaths from pancreatic cancer observed in all the exposed categories. Hazard ratios for incidence in exposed workers compared with unexposed workers were 0.13 (95% CI, 0.02-1.03; 1 case) for quartiles 1 and 2 combined and 1.36 (95% CI,

0.59–3.11; 9 cases) for the upper two quartiles combined (<u>Raleigh et al., 2014</u>).

Steenland and Woskie (2012) studied mortality among 5791 workers employed for ≥ 1 day at a polymer-production plant in Parkersburg, West Virginia, USA, between 1948 and 2002 (see Section 2.1.3). For exposed workers compared with other non-exposed workers at other plants within the same company and region, SMRs calculated for pancreatic cancer by quartile of cumulative serum PFOA level (estimated using a JEM) were 1.18 (95% CI, 0.32–3.03; 4 cases), 1.02 (95% CI, 0.28–2.61; 4 cases), 1.09 (95% CI, 0.35–2.54; 5 cases), and 0.92 (95% CI, 0.30–2.16; 5 cases) from the lowest to the highest quartile categories, respectively.

[Both of these occupational cohort studies (Steenland and Woskie, 2012; Raleigh et al., 2014) had the advantage of the ability to evaluate associations with PFOA in a population exposed to levels much higher than those in the general population; however, a major limitation for both was the small number of observed cases.]

Eriksen et al. (2009) conducted a case-cohort study within a prospective cohort of men and women from the general population in Denmark. Eligible participants were aged 50-65 years at enrolment. The investigators measured PFOA and PFOS concentrations in plasma samples collected before cancer diagnosis (Section 2.1.4). IRRs were calculated on the basis of 128 cases of pancreatic cancer and 772 subcohort participants and were adjusted for age, sex, smoking status, smoking intensity, smoking duration, dietary fat intake, and fruit and vegetable intake. IRRs for pancreatic cancer were 1.55 (95% CI, 0.85-2.80) and 0.91 (95% CI, 0.51-1.65) in the highest quartiles of plasma PFOA and PFOS concentration, respectively, compared with the lowest quartile. The IRR per increase in PFOA of 1 ng/mL was 1.03 (95% CI, 0.98-1.10) and that per increase in PFOS of 10 ng/mL was 0.99 (95% CI, 0.86–1.14). The strengths of this study included a relatively large number of cases and adjustment for

potential confounders such as smoking. Because the PFOA and PFOS measurements were from samples collected before diagnosis, concentrations were less likely to be influenced by the presence of cancer. However, a single measurement at enrolment may not reflect exposure at crucial windows in cancer development. Since the study was carried out among the general population with background exposure levels, exposure contrasts might be too small to detect an association.]

Within communities surrounding a plant from which there had been environmental release of PFOA and contamination of public and private water supplies, the C8 Science Panel (Section 2.1.5) conducted a cohort study of a total of 32 254 community residents and workers exposed to PFOA from a fluoropolymerproduction plant in the Mid-Ohio Valley on the border of West Virginia and Ohio, USA (Barry et al., 2013). For community participants, annual estimates of cumulative serum PFOA concentrations were estimated from 1952 to 2011 using a model by Shin et al. (2011a, b). For the workers, estimates of occupational PFOA exposure were calculated as described in Woskie et al. (2012) and were combined with estimates of environmental exposure. Self-reported cancer according to the surveys administered in 2008-2011 was verified through medical records and cancer registry review. The hazard ratio per unit of natural log-transformed estimated cumulative PFOA serum concentration was 1.00 (95% CI, 0.78-1.29; 24 cases) after adjustment for time-varying smoking, time-varying alcohol consumption, sex, education, birth year (5-year calendar intervals), and age. [The strengths of this study included its use of individual-level exposure modelling using lifetime residential history, the validation of the exposure modelling, the wide range of PFOA exposure levels, and control for potential confounders such as smoking. The main limitation was the small number of cases of incident pancreatic cancer.

In addition, community members and workers who died before enrolment would not have been included, owing to the design of the study as a survivor cohort. This might lead to the potential underascertainment of cancers with a high fatality rate in this population. However, given that PFOA exposure was considered unlikely to be related to survival time, the impact of this aspect of the study design on the resulting risk estimates was likely to be minimal (<u>Barry et al.</u>, <u>2015</u>).]

Consonni et al. (2013) conducted a mortality study in the pooled international TFE cohort that included 5879 male workers who were ever employed or employed for a minimum of 6 or 12 months at one or more of six TFE-production sites in North America and Europe between 1950 and 2002 (see Section 2.1.6). Causes of death were ascertained from 1950 through 2008. Semiquantitative levels of work-related exposure to TFE and APFO were assessed by a plant- and job-specific exposure matrix with yearly estimates (in arbitrary units) of exposure from the start of TFE production until the end of 2002. Among the subset of workers who had ever been exposed to APFO (n = 4205), the SMR using national rates as a referent was 1.05 (95% CI, 0.51-1.94; 10 deaths). In addition, the SMRs for groups with low, medium, and high cumulative APFO exposure were 0 (0 deaths), 1.30 (95% CI, 0.35-3.33, 4 deaths), and 1.84 (95% CI, 0.67-4.00, 6 deaths), respectively (P for trend, 0.34). [Consonni et al. (2013) studied work-related exposure to TFE and/or APFO, and high correlations were observed between exposure to TFE monomer (IARC Group 2A; IARC, 2016) and PFOA, which precludes evaluation of effects of the individual compounds. This study observed fewer than 20 cases, and the small number was a major limitation.]

The Ronneby Register cohort (see Section 2.1.13 for details) comprised 60 507 individuals who had ever lived in the Ronneby municipality during the period when drinking-water was contaminated with a mixture of PFAS, mainly PFOS, PFHxS, and PFOA (1985–2013) (Li et al., 2022a). Cancer incidence data were obtained through linkage to the Swedish Cancer Register (1985–2016). SIRs for incident pancreatic cancer among residents who had ever lived in a highly exposed area were 0.46 (95% CI, 0.17–1.01; 6 cases) for men and 0.81 (95% CI, 0.39-1.50; 10 cases) for women, using the regional external population as the referent. Groups of residents who had ever lived in the contaminated area were subdivided by the number of years living at an ever-high area, and calendar year-, age-, and sex-adjusted hazard ratios compared the ever-high group to the neverhigh group comprising residents who had never lived in the contaminated area. Hazard ratios for this internal comparison were below unity. The strengths of this study included the large general population sample with complete ascertainment and follow-up, owing to high-quality Swedish population registers with complete population coverage, and a strong documented exposure contrast. The limitations of this study were the mixed exposure profile without the possibility to single out effects caused by specific compounds, the small number of cases, and the lack of information on important confounders such as smoking. Additionally, SIRs from the external comparisons might be viewed as ecological comparisons.]

Zhang et al. (2023) conducted two independent nested case-control studies within the ATBC Cancer Prevention Study and the PLCO Cancer Screening Trial (Sections 2.1.11 and 2.1.19). Prediagnostic serum samples were measured for relative levels of PFOA and PFOS among 251 matched pairs from ATBC comprising male smokers aged 50–69 years at baseline (1985–1988) in Finland who were followed until December 2011, and 360 matched pairs from PLCO comprising men and women, mostly non-smokers, aged 55–74 years at baseline (1993–2001) in the USA who were followed until 15 May 2010. ORs for pancreatic ductal adenocarcinoma were adjusted for age and date at blood draw, smoking, diabetes, and BMI. ORs were 2.37 (95% CI, 1.24-4.51) and 1.82 (95% CI, 0.82–4.03) in the highest quintiles of serum PFOA and PFOS concentrations, respectively, compared with the lowest quintile in ATBC. The ORs per SD increase were 1.27 (95% CI, 1.04-1.56) and 1.13 (95% CI, 0.88-1.45) for PFOA and PFOS, respectively. For PLCO, the ORs per SD increase were below unity for both PFOA and PFOS. ORs for only men who had ever smoked or were still in the habit of smoking were lower than those for all participants. [The strengths of this study included prediagnostic serum samples, the relatively large number of cases, and adjustment for potential confounders such as smoking. The limitations of this study included low-level exposure with a small exposure contrast. The Working Group noted that there was unexplained inconsistency between the results for the ATBC and the PLCO for male smokers only.]

Winquist et al. (2023) conducted a casecohort study within the ACS CPS-II LifeLink Cohort (Section 2.1.21). Prediagnostic serum samples were collected during 1998-2001, and participants (median age, 70 years) were followed for cancer incidence until June 2015. Serum concentrations of PFAS were measured for 172 cases of pancreatic cancer and 999 subcohort participants, and hazard ratios were calculated with adjustment for age and year at blood draw, education, race or ethnicity, smoking, and alcohol use. Hazard ratios (95% CI) for pancreatic cancer per concentration doubling were 0.94 (95% CI, 0.74-1.21) and 0.87 (95% CI, 0.70-1.10) for PFOA and PFOS, respectively. In sex-specific analyses, hazard ratios per PFOA doubling were 0.71 (95% CI, 0.52-0.96) and 1.14 (95% CI, 0.78-1.67) for men and women, respectively, although similar hazard ratios for both sexes were observed for PFOS. [The strengths of this study included prediagnostic serum samples, the relatively large number of cases, and adjustment for potential confounders such as smoking. The study limitations included low-level exposure with a small exposure contrast. In addition, because of its design as a survivor cohort, this study would not have included some people who may have had PFOA- or PFOS-related cancer, especially those who developed cancers earlier in life in a susceptible exposed population. This survivor bias would have biased the results downwards (i.e. towards the null or even towards inverse associations).]

Vieira et al. (2013) conducted two case-control studies of 18 different incident cancers during the years 1996–2005 among residents of 13 counties in Ohio and West Virginia, USA, including both contaminated and non-contaminated water districts near the same polymer-production plant in Parkersburg, West Virginia, USA, that was the source of contamination in the population studied by <u>Barry et al. (2013)</u> (see Section 2.1.22). In the first case-control study, cases and controls (all other cancer cases excluding cancers of the kidney, liver, pancreas, and testis) were compared with regard to residence in a contaminated or non-contaminated water district. The OR for pancreatic cancer was 1.0 (95% CI, 0.8-1.3; 58 exposed cases) after adjustment for age, sex, diagnosis year, insurance provider, and smoking status. In the second case-control study, restricted to the Ohio data because of availability of geocoded street addresses, serum PFOA concentrations were estimated by environmental, exposure, and toxicokinetics models designed by Shin et al. (2011a, b). The ORs for pancreatic cancer in the low, medium, high, and very high exposure categories compared with the unexposed, calculated after adjustment for age, race, sex, diagnosis year, insurance provider, and smoking status, were 1.3 (95% CI, 0.7-2.3; 12 exposed cases), 0.9 (95% CI, 0.5-1.7; 10 exposed cases), 1.1 (95% CI, 0.6-2.3; 9 exposed cases), and 0.6 (95% CI, 0.1-2.5; 2 exposed cases), respectively. [The Working Group noted that the studies by Barry et al. (2013) and Vieira et al. (2013) were overlapping rather than independent studies in

that the same geographical areas and some of the same cases were included in both analyses, although the extent of overlap was unknown. The strengths of this study were the large number of incident cancers from cancer registries and the reasonably large number of exposed cases in the contaminated water districts. The second casecontrol study based in Ohio benefited from being able to estimate serum levels for individuals on the basis of a validated model. The limitations were the assignment of an ecological exposure (by water district) in the first case-control study and the somewhat arbitrary assumption in the second case-control study that the estimated serum levels 10 years before case diagnosis were the most relevant, as well as the assumption that cases and controls had remained in the same residence for 10 years. Additionally, the control group included cases of other cancers (bladder, brain, female breast, cervix, leukaemia, lung, melanoma, multiple myeloma, NHL, ovary, prostate, thyroid, and uterus), which may not be representative of the source population because of differences in lifestyle and socioeconomic status among cancer cases. In particular, it might bias estimates towards the null, if any of the included cancers were in fact associated with PFOA. Otherwise, the Working Group considered these potential differences in confounders to be unlikely to have substantive effects in this population with a very high exposure.]

2.5.3 Colorectal cancer and other cancers of the digestive tract (other than liver and pancreas)

See Table S2.5 (Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: <u>https://publications.iarc.</u><u>who.int/636</u>).

Five occupational cohort studies (<u>Alexander</u> et al., 2003; <u>Leonard et al., 2008; <u>Lundin et al.,</u> 2009; <u>Consonni et al., 2013; Steenland et al.,</u> 2015; <u>Girardi and Merler, 2019</u>) and three studies</u>

from communities surrounding a plant from which there had been environmental release of PFOA and contamination of public and private water supplies (<u>Barry et al., 2013</u>; <u>Vieira et al.,</u> <u>2013</u>; <u>Li et al., 2022a</u>) investigated the association between PFOA or PFOS (or both) and cancers of the colorectum and other digestive organs (oesophagus and stomach).

Olsen et al. (2004) studied workers at two manufacturing plants between 1993 and 1998 in Decatur, Alabama, USA. "Episode of care" was identified using health claim data between 1993 and 1998 and was compared between 652 workers at a fluorochemical-production plant (exposed group) and 659 workers at a film plant (non-exposed group). [Episode of care is not a definitive measure of risk because it could include cases of incident cancer, prevalent cancer, and tentatively diagnosed cancer. Mortality in the same company was reported in a study by <u>Alexander et al. (2003)</u> included in the present monograph. Therefore, the study by <u>Olsen et al. (2004)</u> was judged to be uninformative.]

Innes et al. (2014) conducted a cross-sectional study to examine the association between serum concentrations of PFOA and PFOS and self-reported colorectal cancer diagnosis, verified by chart review for 47 359 participants in a comprehensive health survey between 2005 and 2006 by the C8 Health Study Project. [Since the participants in this study overlapped with those in a cohort study by <u>Barry et al. (2013)</u>, and since prevalent cases were used as the case group and the serum concentrations of these participants were influenced by the presence and/or treatment of cancer, the study was judged to be uninformative.]

Among the occupational cohort studies was a study by <u>Lundin et al. (2009)</u> of mortality among of 3993 workers at an APFO-production plant in Cottage Grove, Minnesota, USA, between 1947 and 1997, with follow-up until 2002 (see Section 2.1.1). Using rates for the state of Minnesota as a referent, SMRs were calculated

according to classification of jobs by exposure to APFO. For colon cancer, SMRs for "never", "ever probable/never definite", and "ever definite" exposure groups were 1.30 (95% CI, 0.75-2.12; 16 deaths), 0.88 (95% CI, 0.42-1.62; 10 deaths), and 1.07 (95% CI, 0.13-3.86; 2 deaths), respectively. For rectal cancer, SMRs for "never" and "ever probable/never definite" exposure groups were 0.40 (95% CI, 0.01-2.22; 1 death) and 1.28 (95% CI, 0.26-3.76; 3 deaths), respectively (0 deaths in the "ever definite" category). For oesophageal cancer, SMRs for "never", "ever probable/never definite", and "ever definite" exposure groups were 0.59 (95% CI, 0.07-2.13; 2 deaths), 0.31 (95% CI, 0.01-1.70; 1 death), and 1.54 (95% CI, 0.04-8.57; 1 death), respectively. For stomach cancer, SMRs for "never" and "ever probable/ never definite" exposure groups were 0.74 (95% CI, 0.15-2.15; 3 deaths) and 1.06 (95% CI, 0.29-2.71; 4 deaths), respectively (0 deaths in the "ever definite" category).

Alexander et al. (2003) studied the mortality of a cohort of 2083 production workers who were exposed to PFOS at a plant in Decatur, Alabama, USA, that produced speciality films and fluorochemicals, and who had worked for \geq 1 year at the plant between 1961 and 1997 (Section 2.1.2). Using rates for the state of Alabama as referent, SMRs for all cohort members were 0.30 (95% CI, 0.01–1.66; 1 death) for colon cancer and 1.76 (95% CI, 0.21–6.35; 2 deaths) for oesophageal cancer. In addition, the SMR for cohort members ever employed in a low-exposure job, but never a high-exposure job, was 1.43 (95% CI, 0.04–7.94; 1 death) for colon cancer.

Leonard et al. (2008) investigated mortality among 6027 workers exposed to PFOA at a polymer-production plant in Parkersburg, West Virginia, USA (see Section 2.1.3). Eligible workers were employed at the plant for \geq 1 day between 1948 and 2002 and were followed for mortality from 1948 to 2002. SMRs were computed in comparison to the US population, the West Virginia state population, and an

356

eight-state regional employee population from the same company on the basis of 17 deaths from colon cancer, 5 from rectal cancer, 4 from oesophageal cancer, and 3 from stomach cancer. SMRs estimated using three different reference populations were less than unity except for that for rectal cancer using the reference population of workers from the other regional facilities within the same company (SMR, [1.321]; 95% CI, [0.429-3.082]). Steenland and Woskie (2012) reported an extension of this study by an additional 6 years of follow-up and comprehensive quantitative exposure assessment. Steenland et al. (2015) conducted an incidence study of a subset of the PFOA-exposed workers (n = 3713) in Steenland and Woskie (2012). Rate ratios for quartiles of cumulative serum PFOA level estimated by JEM were calculated by adjusting for age, year of birth, sex, race, education, BMI, and time-varying smoking and alcohol consumption. Compared with those in the lowest quartile, the rate ratios in the second, third, and highest quartiles were 0.58 (95% CI, 0.18-1.87), 1.43 (95% CI, 0.49-4.19), and 1.20 (95% CI, 0.39-3.62), respectively, on the basis of 41 cases of incident colorectal cancer (P for trend, 0.68).

The C8 Health Study (Section 2.1.5) included a total of 32 254 community residents and workers exposed to PFOA from a polymer-production plant in the Mid-Ohio Valley, USA (Barry et al., 2013). Cumulative serum PFOA concentrations were estimated for community residents and workers, taking into account community and occupational exposure, and cancer diagnosis was assessed through self-reported questionnaire and validation through medical-record review and cancer registry data (Barry et al., 2013). Hazard ratios per unit natural log-transformed estimated cumulative PFOA serum concentration were 0.99 (95% CI, 0.92-1.07; 264 cases) for incident colorectal cancer, 0.96 (95% CI, 0.70–1.32; 15 cases) for incident oesophageal cancer, and 0.72 (95% CI, 0.45-1.14; 12 cases) for incident stomach cancer, after adjustment for time-varying smoking, time-varying alcohol consumption, sex, education, birth year (5-year calendar intervals), and age.

In the pooled international TFE cohort study, follow-up was conducted for mortality (1950-2008) of 5879 male workers who were ever employed or employed for a minimum of 6 or 12 months at one of six TFE-production sites in North America and Europe between 1950 and 2002 (Section 2.1.6). Among the subset of workers who had ever been exposed to APFO (n = 4205), the SMR using national rates as the referent was 1.44 (95% CI, 0.72-2.57) for cancer of the oesophagus, whereas SMRs for cancers of the stomach, colon, and rectum were below or around unity (<u>Consonni et al., 2013</u>). In addition, SMRs for oesophageal cancer for groups of low, medium, and high cumulative APFO exposure were 1.62 (95% CI, 0.44-4.14; 4 deaths), 1.54 (95% CI, 0.42-3.93; 4 deaths), and 1.16 (95% CI, 0.24–3.39; 3 deaths) (*P* for trend, 0.60).

Girardi and Merler (2019) reported on mortality among 462 male employees who had worked for \geq 6 months before 2009 at a factory manufacturing PFOA, PFOS, and other chemicals in Trissino, Veneto, Italy (Section 2.1.9). They were followed for mortality from 1970 to 2018. SMRs were calculated in comparison with the regional mortality rates, and mortality risk ratios were estimated by a Poisson regression model using rates from non-exposed workers in other plants. For colon cancer, the SMR was 1.72 (95% CI, 0.72–4.14) and the mortality risk ratio was 2.84 (95% CI, 0.74-10.9), based on 5 deaths; for oesophageal cancer, the SMR was 2.31 (95% CI, 0.68–6.50) and the mortality risk ratio was 3.62 (95% CI, 0.59-22.3), based on 3 deaths; and for stomach cancer, the SMR was 1.30 (95% CI, 0.42-4.02) and the mortality risk ratio was 2.43(95% CI, 0.54–10.9), based on 3 cases.

[The strengths of the incidence study by <u>Steenland et al. (2015)</u> included use of estimated average annual serum PFOA concentrations and adjustment for several potential confounders such as smoking, alcohol drinking, and BMI. In contrast, the five occupational cohort studies of cancer mortality (Alexander et al., 2003; Leonard et al., 2008; Lundin et al., 2009; Consonni et al., 2013; Girardi and Merler, 2019) included small numbers of deaths (fewer than 17) and lack of adjustment for important confounders such as smoking, alcohol drinking, and BMI. The strengths of the study by **Barry** et al. (2013) included a relatively large number of cases of colorectal cancer and control for potential confounders such as smoking, but the small number of cases of oesophageal and stomach cancer was a limitation of this study. In addition, Consonni et al. (2013) studied work-related exposure to TFE and APFO and noted high correlations between exposure to TFE monomer (IARC Group 2A, IARC, 2016) and PFOA, which precluded evaluation of the effects of the individual compounds.]

The Ronneby Register cohort (Section 2.1.13) comprised 60 507 individuals who had ever lived in the Ronneby municipality during a period when drinking-water was contaminated with a mixture of PFAS, mainly PFOS, PFHxS, and PFOA (1985–2013), and incidence data were linked to Swedish Cancer Register (1985–2016) (Li et al., 2022a). Using the regional external population as the referent, SIRs for rectal cancer among residents who had ever lived in the contaminated area were 1.25 (95% CI, 0.89–1.69; 41 cases) for men and 1.33 (95% CI, 0.91–1.88; 32 cases) for women. SIRs for stomach cancer were 1.10 (95% CI, 0.70-1.64; 24 cases) for men and 1.03 (95% CI, 0.55-1.76; 13 cases) for women. For colon and oesophageal cancer, SIRs were below or around unity for both men and women. The group of residents who had ever lived in the contaminated area was subdivided by the number of years living at an ever-high area, and calendar year-, age-, and sex-adjusted hazard ratios were calculated, comparing the ever-high group with the never-high group of residents who had never lived in the contaminated area. Hazard ratios for the ever-high group were 1.25 (95% CI, 0.95–1.64) for rectal cancer and 1.14 (95% CI, 0.79-1.66) for stomach cancer. In addition, hazard ratios for short-high (1–10 years) and long-high (\geq 11 years) exposure were 1.16 (95% CI, 0.80-1.69) and 1.34 (95% CI, 0.94-1.90) for rectal cancer, respectively, and 0.86 (95% CI, 0.51-1.46) and 1.56 (95% CI, 0.95-2.55) for stomach cancer, respectively. [The strengths of this study included the large general population sample with complete ascertainment and follow-up, owing to the high-quality Swedish population registers with complete population coverage, and the strong documented exposure contrast. The study limitations were the mixed exposure profile without the possibility to single out effects caused by specific compounds, the small numbers of cases, and the lack of information on important confounders such as smoking, alcohol drinking, and BMI. Additionally, SIRs from the external comparisons might be viewed as ecological comparisons.]

Vieira et al. (2013) conducted two case-control studies among residents of 13 counties in Ohio and West Virginia, USA (Section 2.1.22). In the first case-control study, after adjustment for age, sex, diagnosis year, smoking status, and insurance provider, the odds ratio for colorectal cancer was 0.9 (95% CI, 0.8-1.0; 383 exposed cases). In the second case-control study, restricted to the Ohio data, and after adjustment for age, race, sex, diagnosis year, smoking status, and insurance provider, ORs for colorectal cancer for the categories with low, medium, high, and very high exposure compared with the unexposed were 1.0 (95% CI, 0.8-1.3; 72 exposed cases), 0.9 (95% CI, 0.7-1.2; 64 exposed cases), 1.3 (95% CI, 1.0-1.7; 63 exposed cases), and 0.6 (95% CI, 0.3-1.0; 13 exposed cases), respectively. The strengths of the study by <u>Vieira et al. (2013)</u> also included the large number of incident colon cancers from cancer registries and the reasonably large number of exposed cases in the contaminated water districts.]

2.6 Cancers of the brain and lymphatic and haematopoietic tissue

2.6.1 Cancers of the eye and brain, and other cancers of the nervous system

See Table S2.6 (Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: <u>https://publications.iarc.</u> <u>who.int/636</u>).

The Working Group identified four cohort studies and two case-control studies investigating the risk of brain cancer associated with PFOA or PFOS exposure. Two of the cohort studies included occupational cohorts (Lundin et al., 2009; Consonni et al., 2013), one of the cohort studies included the C8 Health Project cohort (Barry et al., 2013), and one of the cohort studies included the Ronneby Register cohort (Li et al., 2022a). The case-control studies were population-based (Vieira et al., 2013). In addition, the Working Group reviewed one case-control study on retinoblastoma (Chen et al., 2024).

(a) Cohort studies

A cohort study was conducted on mortality among 3993 employees of an APFO-manufacturing facility located in Cottage Grove, Minnesota, USA (Lundin et al., 2009) (see Section 2.1.1). During the follow-up until 2002, 807 decedents were identified. Using the rates for the state of Minnesota as the referent, SMRs were calculated for different jobs classified by exposure to APFO (the ammonium salt of PFOA). Only 7 deaths were observed for cancer of the central nervous system, with 5 deaths assigned to the "ever probable/never definite" exposure group and 2 deaths assigned to the "never" exposure group. The SMRs for the "ever probable/never definite" exposure group and the "never" exposure group were 1.16 (95% CI, 0.37-2.70) and 0.44 (95% CI, 0.05–1.59), respectively. [The Working Group noted that the significant limitations of the study included the small occupational cohort with a limited number of deaths and the crude exposure assessment by job classification, which made this study uninformative for cancers of the central nervous system.]

Barry et al. (2013) focused on PFOA exposure and incident cancers among community residents and workers exposed to PFOA from a chemical plant, using the C8 Health Project cohort in combination with the cohort of workers from the polymer-production plant in Parkersburg, West Virginia, USA (see Section 2.1.5). The study population comprised 28 541 community members and 3713 workers, with 32 254 participants in the entire cohort. Cancer cases were captured by self-report by the participant and confirmed by medical chart review or state cancer registry matching in Ohio and West Virginia. The number of reported cases of primary brain cancer was 33. The analysis included 17 cases of validated primary brain cancer for whom there was complete covariate information. The authors calculated cumulative PFOA serum concentration estimates for each community participant on the basis of regional historical data. For participants who had ever worked in the polymer-production plant in Parkersburg, a JEM was applied to estimate occupational exposure levels and combined with estimated serum levels from residential exposure to contaminated drinking-water. A proportional hazards regression model was applied in a stratified analysis adjusting for age, time-varying smoking, time-varying alcohol consumption, sex, education, and birth year. Risk estimates based upon models in which exposure was unlagged or lagged 10 years were similar. The hazard ratios for a 1-unit increase in natural log-transformed cumulative exposure in relation to brain cancer were 1.13 (95% CI, 0.84-1.51) for unlagged exposure and 1.06 (95% CI, 0.79-1.41) for exposure lagged by 10 years in the whole cohort. For community residents, increased exposure to PFOA was associated with

a slightly increased risk of brain cancer (HR, 1.14; 95% CI, 0.78–1.65; 13 cases), whereas for the workers, there was no clear evidence of a trend in risk of brain cancer (HR, 0.82; 95% CI, 0.26–2.59; 4 cases). [The Working Group noted as strengths the large cohort, strong exposure contrast, assessment of individual cumulative PFOA exposure, and lagged analyses. Limitations included the self-reported cancer cases, no evaluation of co-exposure to other PFAS in residents, and wide confidence intervals in the estimate for occupational workers.]

The association between occupational exposure to PFOA and mortality from brain cancer was investigated in the pooled international TFE worker cohort, in which data were pooled from workers from one or more of six TFE-production sites in North America and Europe (Consonni et al., 2013) (see Section 2.1.6). The epidemiology departments or the local health unit performed ascertainment of vital status and cause of death through record linkage or individual follow-up procedures. Exposure assessment was performed by a personal semiquantitative estimate using a JEM. There were 4 cases of brain cancer among 4205 men who had ever been exposed to PFOA, and the SMR for brain cancer associated with exposure to PFOA was 0.64 (95% CI, 0.17–1.63), using national rates as the referent. [The Working Group noted as strengths the inclusion of all TFE-production sites worldwide, and the complete enrolment and follow-up data. Limitations included the high correlations between TFE and PFOA exposure and the small number of cases of brain cancer observed, which limited the informativeness of this study.]

Li et al. (2022a) investigated cancer incidence in the Ronneby Register cohort, a community of residents with high-level environmental exposure to a mixture of PFAS, in Sweden. By the end of the follow-up (31 December 2016), the study had identified 150 cases of incident brain cancer (80 men and 70 women) (see Section 2.1.13). All information on brain cancer diagnosis was obtained from the nationwide Swedish Cancer Register. To facilitate comparison, Ronneby residents were assigned to mutually exclusive groups, "never-high" and "ever-high", based on whether they were exposed to PFAS-contaminated water at their residence. When comparing the study population to the general population of Blekinge County excluding Ronneby, the incidence of brain cancer was increased in the "ever-high" group among men (SIR, 1.29; 95% CI, 0.83–1.93) but decreased in the "never-high" group among women (SIR, 0.73; 95% CI, 0.55-0.96). In internal comparisons, the "ever-high" group was further subdivided by the time period of high exposure ("early-high" in 2004 or earlier, "late-high" in 2005 or later) and duration of time in a high-exposure area ("short-high" for ≤ 10 years, and "long-high" for \geq 11 years). Hazard ratios for early-high and late-high were 1.20 (95% CI, 0.78-1.84) and 1.31 (95% CI, 0.76-2.26), respectively, and those for short-high and long-high were 1.06 (95% CI, 0.66-1.69) and 1.50 (95% CI, 0.92-2.44), respectively. [The Working Group noted as strengths the large study population, strong exposure contrast, and unbiased inclusion. The main limitations included the small number of cases, the crude exposure assessment (not including individual water intake or other sources of exposure than drinking-water), the mixed exposure profile, and the limited information on potential confounders such as smoking habits, BMI, and occupational exposure.]

(b) Case-control study

A case-control study was conducted using cancer registry data for residents of counties in Ohio and West Virginia surrounding the polymer-production plant in Parkersburg, West Virginia, USA, from which PFOA had been emitted into drinking-water sources (Vieira et al., 2013) (see Section 2.1.22). The study included incident cancer cases drawn from registry data from 1996 through 2005. Controls comprised all other cancers in the study data set, except cancers of

360

the kidney, pancreas, testis, and liver. There were 506 cases of brain cancer in the final data set, of which 150 came from Ohio. All people with a cancer diagnosis were classified as exposed (living within a contaminated water district) or unexposed (not living in contaminated water districts) using geocoding. The AORs varied among the water districts exposed to contaminated water, with the AOR for the overall exposure risk being 1.0 (95% CI, 0.8–1.3). In a second case-control study, the authors restricted the analysis to Ohio participants for whom annual PFOA serum concentrations could be estimated on the basis of an existing PFOA exposure prediction model. Individual-level annual exposure was categorized as very high, high, medium, low, and unexposed. Using the unexposed group as the reference category, the AORs for high, medium, and low individual-level exposures were 0.6 (95% CI, 0.2-1.6), 1.8 (95% CI, 1.1-3.2), and 1.5 (95% CI, 0.8-2.7), respectively. No cases of brain cancer occurred in the group with very high exposure. Findings were similar in various sensitivity analyses (e.g. using cumulative PFOA serum exposure instead of annual exposure; using exposure level for exposure estimates that did not account for latency; including cases of kidney, liver, pancreatic, and testicular cancer in the control group). [The Working Group noted that the strengths of the study included its focus on a population with high PFOA exposure, the strong contrast in exposure levels, and the estimation of individual-level exposure for a subset of the population. Limitations included the use of other cancers as the reference group and the potential for exposure misclassification (reliance on the address at the time of diagnosis rather than a complete residential history in analyses among Ohio participants). However, both these limitations would be expected to have resulted in a bias towards the null.]

A population-based case-control study was conducted that included 501 children aged < 5 years with a diagnosis of retinoblastoma

between 1983 and 2013 identified and randomly selected from the California Cancer Registry (Chen et al., 2024). Controls (n = 899) were selected from California birth rolls and frequency-matched to cases on year of birth. For cases and controls, neonatal dry blood samples were available, collected from the newborn heel-stick test, which is done 12-48 hours after birth for neonatal genetic screening. This sampling is a routine procedure, for > 99% of all neonates in California, and samples are stored by the California Newborn Screening Program for genetic disease. The blood spot was used for quantification of PFOA, PFOS, and PFNA. Outliers for PFAS measurement identified through a principal component analysis were excluded (n = 10), leaving a total of 497 cases and 893 controls for the analysis. Children with a PFOA concentration above the mean, compared with those with a concentration below the mean, had a higher risk of retinoblastoma (OR, 1.16; 95% CI, 0.90-1.50), particularly so for those born from US-born mothers (OR, 1.41; 95% CI, 1.00-2.02). Children with a neonatal heel-stick PFOS concentration of above the mean, compared with those with a concentration of below the mean, had a 29% higher risk of retinoblastoma (OR, 1.29; 95% CI, 1.00–1.67), with risk being elevated in both US-born and Mexico-born mothers. When restricting to unilateral retinoblastoma cases, the OR for a PFOS concentration of above the mean versus below the mean was 1.42 (95% CI, 1.03–1.97), whereas for bilateral retinoblastoma cases the OR was 1.14 (95% CI, 0.82-1.62). [The Working Group noted the limited sample size for the stratified analysis by mother's birthplace. The population-based design and the use of prediagnostic samples collected for medical reasons unrelated to the case status was a strength of this study, since it minimized selection bias and provided a measurement of PFOA and PFOS exposure unrelated to diagnosis or treatment. Such measurements are probably representative of fetal exposure; however, uncertainty remained

concerning the capture of the relevant window of exposure for cancer development. In addition, PFAS were measured by a semiquantitative non-targeted method, which limited comparability across studies.]

2.6.2 Cancers of lymphatic and haematopoietic tissue and other cancers

See Table S2.6 (Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: <u>https://publications.iarc.</u><u>who.int/636</u>).

Five occupational cohort studies have investigated mortality for cancers of lymphatic and haematopoietic tissue, melanoma, lung, or mesothelioma (Alexander et al., 2003; Leonard et al., 2008; Lundin et al., 2009; with later follow-up of mortality for selected cancers by Steenland and Woskie, 2012; Consonni et al., 2013; Steenland et al., 2015, Girardi and Merler, 2019). Two cohort studies (Barry et al., 2013; Li et al., 2022a) and one case-control study (Vieira et al., 2013, partly overlapping with **Barry et al.**, 2013) addressing highly exposed community residents have investigated the incidence of cancers of lymphatic and haematopoietic tissues and melanoma according to PFOA and/or PFOS exposure. A large US case-cohort study of the general population with low background exposure examined a range of cancers of lymphatic and haematopoietic tissue (Winquist et al., 2023). A small case-control study with cross-sectional sampling of exposure data by Lin et al. (2020) examined associations between germ cell tumours in preschool children and maternal serum concentrations of PFOA and PFOS. [The Working Group noted the unclear methods used for control selection and some very high PFOA measurements that were not discussed. This study was considered uninformative for the evaluation of human cancer hazard and was not considered further.]

Lundin et al. (2009) conducted a mortality study among 3993 workers at an APFOproduction plant in Cottage Grove, Minnesota, USA, between 1947 and 2002 (see Section 2.1.1). Using the rates for the state of Minnesota as the referent, SMRs were calculated according to classification of jobs by exposure to APFO (the ammonium salt of PFOA) in three categories: never, ever probable/never definite, and ever definite. Of 29 deaths from cancers of lymphatic and haematopoietic tissue, only one was among those definitely exposed to APFO (SMR, 0.37; 95% CI, 0.01-2.08) and 14 were probably, but never definitely exposed (SMR, 0.96; 95% CI, 0.53-1.61). None of the 3 deaths from lymphosarcoma-reticulosarcoma or 13 deaths from other lymphatic and haematopoietic cancers were among the definitely exposed, but 2 deaths (SMR, 1.80; 95% CI, 0.22-6.51) and 5 deaths (SMR, 0.71; 95% CI, 0.23-1.66), respectively, were among the probably, but never definitely, exposed for these cancer types. The single death from Hodgkin lymphoma was observed in the group of workers who had never been exposed. For leukaemia, 7 of 12 deaths were among the probably but never definitely exposed (SMR, 1.27; 95% CI, 0.51–2.61) with only 1 death among the definitely exposed (SMR, 0.96; 95% CI, 0.02-5.34). [The Working Group noted that this study was focusing on mortality, and its informativeness with respect to specific, relatively rare, cancers was limited by the small numbers and the crude exposure assessment that did not allow for analysis of cumulative exposure or lagged analyses.]

Alexander et al. (2003) studied mortality from cancers of lymphatic and haematopoietic tissues combined and melanoma between 1961 and 1998 among 2083 workers enrolled from 1961 through 1997 at the PFOS facility in Decatur, Alabama, USA (see Section 2.1.2 for a full description). The median duration of follow-up was 25.9 years, and a total of 4 deaths from lymphatic and haematopoietic cancer, 3 deaths from melanoma, and 15 deaths from respiratory system cancers were identified. Workers were classified as highly exposed to PFOS (and PFOA) according to a company-specific JEM based upon a survey of PFOS serum measurements and included a subset of workers in the chemical division, whereas workers in the film-producing division were unexposed at work. The geometric mean serum PFOS concentration for chemical division employees was 0.9 ppm [900 ng/mL] and for film division employees it was 0.1 ppm [100 ng/mL]. Using the Alabama state population as the referent, SMRs for lymphatic and haematopoietic cancers were not increased in the entire cohort including both chemical and film divisions, or in ever potentially highly exposed employees (Table S2.6, Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: https://publications.iarc.who. int/636). The SMR for melanoma was increased in both the entire cohort and exposed employees, but estimates were based on few (\leq 3) exposed cases. Mortality was not elevated from respiratory system cancers (all trachea, bronchus, and lung) overall or in any exposure category.

[The Working Group noted that this study had complete data for a highly exposed occupational cohort with long follow-up, but numbers of less-frequent cancers, such as cancers of lymphatic and haematopoietic tissue and melanoma, were low and did not allow estimation of an association with PFOS with reasonable precision. The exposure assessment was rather crude, without assessment of cumulative exposure, and co-exposures to potential carcinogens and other fluorochemicals were likely. Therefore, the Working Group considered that this study provided limited information for the evaluation of cancers of lymphatic and haematopoietic tissue, melanoma, or respiratory system cancers.]

Leonard et al. (2008) studied mortality from several specific cancers of lymphatic and haematopoietic tissue and from melanoma among 6027 workers (men, 81%) who had ever worked at the polymer-production facility in

Parkersburg, West Virginia, USA, between 1948 and 2002. With follow-up until 2002, mortality was not elevated for melanoma (3 deaths) for workers versus any of the three reference groups considered. A later follow-up until 2008 was published by Steenland and Woskie (2012) (see Section 2.1.3 for a full description). Only the latest follow-up data for NHL, leukaemia, lung cancer, and mesothelioma are reported here (Steenland and Woskie, 2012). The latest follow-up for melanoma is also reported in Table S2.6, as in Steenland et al. (2015). The mean follow-up was 30 years, and 14, 14, 84, and 6 deaths from NHL, leukaemia, lung cancer, and mesothelioma, respectively, were observed. SMRs were computed using the US population and an eight-state regional employee population from the same company (other workers in the same company and region) as referents.

Compared with other workers in the same company and region, increases were not observed for NHL (SMR, 1.05; 95% CI, 0.57–1.76), leukaemia (SMR, 1.05; 95% CI, 0.57–1.76), or lung cancer (SMR, 0.78; 95% CI, 0.62–1.64) in the PFOA-exposed cohort. SMRs according to individual cumulative PFOA exposure estimates without a lag did not indicate dose–response associations for these causes of death. Mortality from mesothelioma was elevated in the cohort (SMR, 2.85; 95% CI, 1.05–6.20), especially in the fourth quartile of cumulative PFOA exposure (5 deaths).

[The Working Group noted that this was the largest of the three US occupational PFAS cohorts (partly because there was no restriction with respect to duration of employment) and was characterized by a high degree of completeness of case ascertainment and cohort follow-up. A major strength of the updated follow-up was estimation of individual cumulative serum PFOA levels. The magnitude of occupational exposure to suspected or known human carcinogens such as asbestos was not quantified, but some co-exposure could not be ruled out.]

Barry et al. (2013) evaluated the risk of cancers of lymphatic and haematopoietic tissue and of melanoma in 28 541 community residents in the Mid-Ohio Valley, USA, who were exposed to PFOA in drinking-water as a result of emissions from the polymer-production plant in Parkersburg, West Virginia, and in 3713 employees working at this plant (a total of 32 254 individuals; men, 46%) (see Section 2.1.5). The average duration of follow-up after age 20 years was 33 years and during this period 66, 136, 241, and 108 cases of incident leukaemia, lymphoma, melanoma, and lung cancer, respectively, were identified. Adjusted hazard ratios for selected cancers of lymphatic and haematopoietic tissue and melanoma were computed by proportional hazard regression by estimated cumulative PFOA exposure (continuous variable). The risk of leukaemia, lymphoma (type not specified), melanoma, and lung cancer did not increase with increasing estimated cumulative exposure (Table S2.6, Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: <u>https://publications.iarc.who.</u> int/636). Risk estimates based upon models where exposure was unlagged, lagged 10 years, or lagged 20 years were similar (data for a 20-year lag were not reported in the manuscript). Results based on all self-reported cancer cases were similar to those for based on validated cases only (the results of the analysis using validated cases only were not reported in the manuscript). [The Working Group noted that these findings in this large community cohort with individual assessment of cumulative exposure did not consistently indicate that PFOA is associated with increased risk of these cancers at exposure levels encountered in a community with mainly high environmental exposure.]

<u>Consonni et al. (2013)</u> investigated cause-specific mortality rates in an international occupational cohort of 5879 male TFE workers of whom 4205 were also exposed to APFO (see Section 2.1.6). An individual semiquantitative estimate of cumulative TWA airborne exposure was assigned from a study-specific JEM. In total, 49 deaths from lung cancer and 19 deaths from lymphatic and haematopoietic cancer were identified during follow-up from 1950 to 2008 in workers who had ever been exposed to APFO.

Using national rates as the referent, SMRs for lymphatic and haematopoietic tissue cancers combined, for NHL, and for multiple myeloma were not elevated in male workers who had ever been exposed to APFO. Mortality from leukaemia was increased (SMR, 1.61; 95% CI, 0.88–2.88), but without indications of increasing risk with increasing cumulative exposure. Lung cancer mortality was lower in workers than in the reference population (SMR, 0.73; 95% CI, 0.54–0.97).

[The Working Group noted that this cohort included all TFE-production sites worldwide during the entire period of production and benefited from almost complete enrolment and follow-up data. The informativeness of this study was limited. Internal analyses were not performed. Analyses stratified by level of cumulative exposure only included few cases (e.g. 3 or 4 cases of leukaemia in tertiles of cumulative exposure).]

Girardi and Merler (2019) reported mortality from cancers of lymphatic and haematopoietic tissue and lung in 1970-2018 among 462 male employees enrolled from 1960 through 2008 at a factory manufacturing PFOA, PFOS, and other chemicals in Trissino, Veneto, Italy (see Section 2.1.9 for details). A cohort of railroad workers from the geographical region constituted the reference group. For the factory-worker and railroad-worker cohorts, the mean duration of employment was 12.5 and 9.7 years, respectively, and the mean length of follow-up was 31.7 and 34.3 years, respectively. Loss to follow-up was < 3%. The geometric mean for PFOA was 4048 ng/mL, highest among PFOA operators (geometric mean, 8826 ng/mL; range, 335-86 300 ng/mL).

Using both regional and reference factory data as referents, mortality from cancers of lymphatic and haematopoietic tissue (7 deaths) was increased in the entire factory-worker cohort and increased with increasing estimated cumulative level of PFOA exposure, as indicated by tertile analysis (Table S2.6, Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: https:// publications.iarc.who.int/636). Mortality from lung cancer (6 deaths) was not elevated in the factory-worker cohort compared with either reference group.

[The Working Group noted that this study had the advantage of complete data for an occupational cohort with high-level exposure, long follow-up, biological monitoring data, and estimates of cumulative exposure to PFOA. Subsets of employees seemed to have the highest recorded levels of PFOA among the available PFAS cohorts so far. Major limitations were that few samples were available to model some job categories; the small size of the factory-worker cohort with only 7 deaths from cancers of lymphatic and haematopoietic tissue among 462 employees followed for about 32 years, resulting in imprecise confidence intervals; inability to distinguish the effects of different PFAS compounds and other potential carcinogenic co-exposures; and limited confounding control. Factory workers were exposed to several chemicals in addition to PFOA and PFOS.]

Li et al. (2022a) reported sex-stratified risk estimates for the most common specific cancers of lymphatic and haematopoietic tissue, melanoma, and lung cancer among 60 507 residents (15 811 highly exposed; men, 52%) of Ronneby municipality, Sweden, reporting on follow-up from 1985. PFOA constituted only a minor proportion of the PFAS (mainly PFOS and PFHxS) that contaminated the drinking-water (for details, see Section 2.1.13).

The SIR, adjusted for sex, age and calendar year, for NHL was not increased in male or female

Ronneby residents who had ever been exposed to highly contaminated drinking-water, but the internal analysis within the Ronneby municipality revealed an elevated hazard ratio for residents with high-level exposure for > 10 years and for residents with high-level exposure during the latest period, where contamination levels were assumed to be higher, relative to residents who had never been exposed to highly contaminated water. The latter risk estimates were uncertain, with broad confidence intervals including unity.

The SIRs for multiple myeloma and chronic lymphocytic leukaemia were not increased in residents who had ever been exposed (men or women), and no consistent increase in risk was seen according to time or duration of exposure in residents ever living in a highly contaminated district. The SIR for chronic myeloid leukaemia was increased in Ronneby residents with low-level (only men) or high-level exposure, but numbers were low and precluded more detailed analysis.

The SIR for melanoma was increased in Ronneby residents with low-level (only men) or high-level exposure, and internal analysis indicated higher risk among residents with highlevel exposure for > 10 years and especially for residents with high-level exposure during the latest period (HR, 1.54; 95% CI, 1.09–2.19).

[The Working Group noted that major strengths included complete registration of the cohort, no loss to follow-up, and a long follow-up period. Major limitations were the crude ecological exposure assessment, without individual estimates related to PFOS exposure.]

Winquist et al. (2023) conducted a casecohort study within the ACS prospective CPS-II LifeLink Cohort, with measurement of PFOA, PFOS, and several other PFAS in prediagnostic serum samples collected during 1998–2001. Overall, there was no increase in the incidence of lymphatic and haematopoietic cancers associated with serum PFOA or PFOS serum concentrations. [The Working Group noted several strengths, including the case-cohort design, large sample size, good cancer ascertainment via registries and examination by histological subtype, and prediagnostic serum samples. Limitations were mainly the low exposure levels, narrow exposure contrast, and probable attenuation of risk estimates because of delayed blood sampling relative to time of enrolment.]

<u>Vieira et al. (2013)</u> conducted a case–control study to investigate the risk of 18 cancers in a community sample with relatively high exposure to PFOA because of contamination of drinking-water by the polymer-production plant in Parkersburg, West Virginia, USA. Using all other cancers except kidney, testicular, liver, and pancreatic as controls, odds ratios were estimated for exposed versus unexposed and for subsets of exposed across districts (see Section 2.1.18).

The odds of NHL were elevated among exposed residents in contaminated water districts (152 cases) relative to the unexposed, but the excess was limited to the very-high and medium exposure categories. Leukaemia (72 exposed cases), multiple myeloma (36 exposed cases), and melanoma (168 exposed cases) were not associated with exposure in contaminated water districts, and the odds of these cancers did not increase with increasing exposure category.

Lung cancer (632 exposed cases) was associated with exposure to contaminated water (OR, 1.2; 95% CI, 1.1–1.3), but the elevation was observed only in the high-exposure category and not in the very-high exposure category.

[The Working Group noted that this was a relatively large study population with a strong exposure contrast and with estimates of individual-level exposure for a subset of the population. Limitations included the use of other cancers as controls, which may cause bias towards the null if PFAS exposure is a risk factor for the cancers in the control group, or the opposite if PFAS exposure is associated with risk factors – for instance, smoking and alcohol consumption – that are probably more prevalent in the cancer controls than in the background population. Individual-level exposure misclassification was most likely to be independent of the cancer outcome, with probable bias towards the null as a result.]

2.7 Cancer of all sites combined

See Table S2.7 (Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: <u>https://publications.iarc.</u><u>who.int/636</u>).

There were seven cohorts that contributed evidence on PFOA and/or PFOS exposure and the risk of cancer overall. Five of these were occupational cohorts that used JEMs to estimate exposure and were focused primarily on men and on cancer mortality. In contrast, Li et al. (2022a) examined overall cancer incidence in residents with high environmental exposure, and Wen et al. (2022) evaluated serum PFOA and PFOS levels in relation to mortality using NHANES, which was more representative of exposure levels in the general US population. All these studies used data linkages to ascertain cancer outcomes, most commonly using information from death certificates. [The Working Group noted that although this approach had the benefit of often providing very complete outcome data, the focus on cancer mortality did not provide much information on the relation between PFAS exposure and specific cancers that have a longer survival time after diagnosis. Additionally, the studies in this section considered all cancer diagnoses as a single outcome, a heterogeneous category that may mask important associations with individual cancer outcomes.]

Raleigh et al. (2014) evaluated overall cancer mortality in an occupational cohort that included 4668 workers exposed to PFOA at an APFO factory in Cottage Grove, Minnesota, USA, between 1947 and 2002 and a comparison group of 4359 employees who were unexposed workers at a tape and abrasive production facility in Saint Paul, Minnesota (see Section 2.1.1, PFOA-production workers). Individual inhalation exposure was estimated using a JEM created from expert evaluation and industrial hygiene data. Mortality information was obtained from the NDI. There were 332 cancer deaths identified among the exposed workers. Overall cancer mortality for individuals working at the exposed plant (Cottage Grove, SMR, 0.87; 95% CI, 0.78–0.97) was lower than that for workers at the unexposed location (Saint Paul, SMR, 1.04; 95% CI, 0.95-1.13). Higher APFO exposure was not associated with a higher SMR for overall cancer mortality (quartile 4, SMR, 0.92; 95% CI, 0.71-1.16). [The Working Group noted that although this study had individual cumulative air exposure assessment with some evidence of this exposure metric being correlated with serum level, unlikely co-exposure to TFE, and a relatively higher number of overall cancer mortality cases compared with the individual cancer types, the heterogeneous nature of the outcome limited the inference from these findings. The study also lacked data on workers who left Minnesota or Wisconsin or on potential confounding factors such as smoking, which, if associated with occupational exposure, may have led to residual confounding.]

Alexander et al. (2003) studied a population of 3512 PFOS-exposed production workers at a plant in Alabama, USA (see Section 2.1.2, PFOSproduction workers). 2083 participants were identified who had worked for ≥ 1 year between 1961 and 1997. Mortality follow-up was conducted using linkage to the NDI. The individual's PFOS exposure was estimated on the basis of job history and information from a subset (n = 232)for whom blood samples had been collected in 1998 and PFOS levels measured. Based on this subset, all workers were categorized according to their possible exposure (no workplace exposure, low potential exposure, or high potential exposure). Of the 2083 workers who met the criterion of working \geq 1 year at the plant, 39 cancer deaths (total deaths, 145) were observed. The SMR for all

cancer deaths was 0.72 (95% CI, 0.51-0.98). It was similar when limiting to employees who were ever employed in a high-exposure job (SMR, 0.84; 95% CI, 0.50–1.32; 18 deaths), as well as for those who were ever employed in a low-exposure job but never in a high-exposure job (SMR, 0.52; 95% CI, 0.19-1.14), or those who worked in a high-exposure job for ≥ 1 year (SMR, 0.84; 95% CI, 0.46–1.41). [The Working Group noted that this study used a JEM informed by a subset of workers with blood measurements and had a strong exposure contrast but few deaths and no cancer incidence data. This study evaluated mortality using an NDI linkage, which would have underestimated associations with specific cancer types that have more favourable survival after diagnosis. The heterogeneous nature of the outcome limited the inference from these findings. Furthermore, the study was conducted predominantly in men and had limited control for confounding by factors such as smoking, which, if associated with occupational exposure, may have led to residual confounding.]

Steenland and Woskie (2012) conducted a mortality study among a cohort of PFOAexposed workers at the polymer-production plant in Parkersburg, West Virginia, USA (see Section 2.1.3). There were 5791 workers who were employed for \geq 1 day between 1948 and 2002 and who had sufficient work histories to allow for estimation of PFOA exposure. PFOA exposure was estimated using information on work history and from a subset with serum PFOA measurements. This cohort was highly exposed, with estimated serum PFOA concentrations that were two orders of magnitude higher than those in the general population. SMRs were calculated comparing workers in the cohort to workers at other factories in the same company in a similar geographical region and to the general US population. A total of 1084 deaths were observed during follow-up from 1952 to 2008; of these, 304 were determined to be cancer-related, ascertained via linkage to the NDI or from death certificate data. Relative to

workers at other factories within the same region and company, mortality for all cancer types in participants in the Parkersburg polymer-production plant cohort was not elevated overall or when considering quartiles of estimated exposure to PFOA (e.g. quartile 4, SMR, 0.94; 95% CI, 0.76–1.16). The consideration of either a 10-year lag (e.g. quartile 4, SMR, 0.92; 95% CI, 0.73–1.15) or 20-year lag (data not reported) did not alter conclusions. [The Working Group noted that this study included a highly exposed cohort with long follow-up period and used a comparison group that included other workers, which may have attenuated any healthy-worker effect. Another strength of the study was the detailed exposure assessment using an enhanced JEM with serum exposure levels based on measurements from workers. However, the study did not evaluate the incidence of all cancers combined. Using cancer mortality data from linkages may underestimate associations with specific types of incident cancer, particularly those with more favourable survival after diagnosis. The heterogeneous nature of the outcome was a main limitation. There was limited control for confounding, which could have led to residual confounding if lifestyle factors such as smoking were related to occupational status in the cohort.]

Consonni et al. (2013) evaluated cancer mortality in a cohort of workers who were employees at six TFE-production sites in North America and Europe and were exposed to APFO (the ammonium salt of PFOA) as part of the manufacturing process between 1950 and 2002 (see Section 2.1.6). Job-specific exposure matrices based on the potential for exposure were used to estimate semiguantitative exposure to both TFE and PFOA. Vital status was obtained until 2008 using a variety of methods and linkages across the various geographical locations where the factories were located. Among 4205 workers who had ever been exposed to APFO, there were a total of 534 deaths, including 159 deaths from cancer. Overall, there was no association

between all-cancer mortality and cumulative estimated APFO exposure, when comparing with a national referent (e.g. highest cumulative exposure, SMR, 0.78; 95% CI, 0.59-1.02). Co-exposure to high levels of TFE and high levels of APFO was also not associated with an elevated SMR (0.81; 95% CI, 0.60–1.06). [The Working Group noted that although this study was a comprehensive population of international TFE workers at the time it was conducted, it used a semiguantitative exposure assessment with no validation of estimated exposures. The study was limited to men and did not include information on potential confounding factors such as smoking status, which, if associated with occupational exposure, could have led to residual confounding. It was also difficult to discern whether any observed effects in this study would be caused by TFE (IARC Group 2A, <u>IARC, 2016</u>), if present, or by PFOA, given the high correlation between the exposures. Using cancer mortality data from linkages may underestimate associations with specific types of incident cancer, particularly those with more favourable survival after diagnosis. The heterogeneous nature of the outcome limited the inference from these findings.]

Girardi and Merler (2019) investigated mortality in a cohort of 462 PFAS-exposed workers at a factory in Trissino, Veneto, Italy, and compared mortality rates with those for regional general populations and 1383 railroad workers who were not exposed to PFAS compounds (see Section 2.1.9). PFAS exposure was estimated using a JEM, which was informed in part by serum PFOA concentrations. Exposure was categorized into tertiles of estimated PFOA and was also evaluated on the basis of categories of exposure (ever at PFAS department, never at PFAS department, and in offices). Vital status was obtained from death certificates, for deaths between 1970 and 2018. This was a highly exposed occupational cohort (n = 120 with measured PFOA; geometric mean, 4048 ng/mL), with 107 deaths observed, 42 of which were from cancer. There was no

excess mortality observed when compared with regional rates for comparison overall (all cancers, SMR, 1.00; 95% CI, 0.74-1.36) although the overall cancer mortality risk was elevated when compared to that for the railworkers (risk ratio, 1.32; 95% CI, 0.91–1.91). There was little evidence of association with categorical estimates of PFAS exposure, with imprecise increases in the SMR for the highest estimated tertile of PFOA (SMR, 1.22; 95% CI, 0.79-1.87) and among those ever working in a PFAS department (SMR, 1.46; 95% CI, 0.85–2.51), when using regional rates as the referent. However, the estimate for the highest tertile of PFOA was more pronounced when the railroad workers were used as the referent (risk ratio, 1.65; 95% CI, 1.02-2.65), and an increase was also evident for those ever working in a PFAS department (risk ratio, 1.97; 95% CI, 1.10-3.54). [The Working Group noted that this was a highly exposed cohort, for which serum levels were used in conjunction with a JEM to inform the exposure classification. The population was limited to men and although the study did not include any information on confounders such as smoking, it used both national rates and an unexposed worker population to reduce the impact of the healthy-worker effect, which may also limit residual confounding. However, despite a long follow-up period, there were few cancer-related deaths, and the use of death certificates to ascertain cancer mortality data may underestimate possible associations with specific types of incident cancer, particularly those with more favourable survival after diagnosis. The heterogeneous nature of the outcome limited the inference from these findings.]

Li et al. (2022a) examined overall cancer incidence in more than 60 000 individuals who lived in Ronneby municipality in Sweden in 1985– 2013 (see Section 2.1.13). This study population included approximately one third who were exposed to water contaminated with a mixture of PFAS compounds, mainly PFOS, PFHxS, and, to a lesser extent, PFOA. The exposure assessment

was based on annual residential addresses and information on drinking-water supply, and cases were identified using cancer registry linkage until 2016. There were 5702 cases of cancer identified. There was no evidence that exposure to highly contaminated PFAS drinking-water was associated with excess incidence, as SIRs were around or below the null. SIRs were also similar for both the "never-high" and "ever-high" exposure groups, defined on the basis of living at an address supplied with PFAS-contaminated water, compared with an external reference group. In the internal cohort comparison analysis, there was little difference in the hazard ratios for overall cancer incidence according to estimated exposure duration or timing of exposure, although there may have been a slight increase for high exposures between 2005 and 2013 (late period) (HR, 1.09; 95% CI, 0.99-1.20) but not for high exposures between 1985 and 2004 (early period). Li et al. (2022a) also conducted sensitivity analyses further adjusting for highest education attained. Potential confounding by smoking was partly accounted for, since duration of education and smoking are highly correlated (<u>Eek et al., 2010</u>). [The Working Group noted that this study had a large general population sample with a high environmental level of PFAS exposure and near-complete registry-based case identification. Other strengths included the use of both an external reference group and internal comparisons. However, the exposure assessment was limited by not having any individual-level measurements of exposure and by including areas that were contaminated by multiple PFAS, which limited inferences regarding associations with individual compounds. The minimal information on individual-level confounders, except for education (which was included in sensitivity analyses), may not be as important in this context, given that exposure was determined on the basis of the water distribution system.]

Wen et al. (2022) evaluated the association between mortality and serum measurements

of PFOA and PFOS using data from NHANES, which is a continuously conducted and nationally representative cross-sectional survey designed to represent the non-institutionalized US population (see Section 2.1.15). Blood samples were collected in 1999-2014, and participants were followed up for mortality using linkage to the NDI until the end of 2015. Of the 1251 deaths that occurred during the study follow-up period, 248 were from cancer. Increasing serum PFOA levels were not related to higher incidence of cancer-related mortality (PFOA tertile 3 versus tertile 1, HR, 1.06; 95% CI, 0.68–1.71). In contrast, increasing PFOS level was related to higher adjusted hazard ratio for overall cancer-related mortality in a dose-dependent manner (PFOS tertile 2 versus tertile 1, HR, 1.26; 95% CI, 0.75-2.06; PFOS tertile 3 versus tertile 1, HR, 1.75; 95% CI, 1.10–2.83), adjusting for the other measured PFAS in addition to sex, age, race or ethnicity, education, smoking status, physical activity, hypertension, healthy eating index, creatinine clearance rate, serum total cholesterol, and serum cotinine. [The Working Group noted that the strengths of this investigation were the use of a nationally representative population with serum measurements of PFOS and PFOA, with probable complete ascertainment of cancer mortality. Despite relatively good control for potential confounders, including other PFAS, the Working Group noted that the analysis did not adjust for calendar time, which is important given temporal trends in PFAS concentrations. Other limitations included the short follow-up time for some of the participants, which may not reflect the relevant etiological window, and, for some individuals, the blood sample may have been collected after cancer diagnosis and treatment, since participants with cancer at baseline were not excluded. There was also a relatively small number of cancer-related deaths, and the focus on overall cancer mortality, a heterogeneous outcome, may mask associations for individual cancer types.]

2.8 Evidence synthesis for cancer in humans

This section provides a synthesis of the epidemiological evidence on cancer in humans exposed to PFOA or PFOS. The synthesis is based upon a total of 36 epidemiological studies available to the Working Group.

The first epidemiological study addressing risk of cancer associated with exposure to PFAS was an occupational mortality study in a cohort of workers manufacturing APFO (the ammonium salt of PFOA), in the Cottage Grove plant in Minneapolis, Minnesota, USA (<u>Gilliland and</u> <u>Mandel, 1993</u>). This study was published more than five decades after large-scale manufacture of PFOA was initiated.

2.8.1 Studies evaluated

The epidemiological evidence on the carcinogenicity of PFOA and PFOS in humans is available from studies with three different exposure settings. First, occupational exposure of workers in chemical plants manufacturing or using PFOA or PFOS; second, high environmental exposure in communities contaminated by emissions from chemical plants or other specific sources, such as the use of aqueous firefighting foam; and last, background exposure of the general population. Studies within these three settings typically have different epidemiological designs with different strengths and limitations; thus, comparing findings for a particular cancer site across the various exposure settings may assist causal inference. The studies that the Working Group considered the most informative and hence to which the most weight was given when balancing the evidence for carcinogenicity in humans were large cohort and nested case-control studies from all three major exposure settings.

(a) Occupational cohort studies

Chemical plants manufacturing or using PFOA or PFOS were established at three major sites in USA and some facilities in Europe from the late 1940s onwards, and follow-up studies of worker cohorts from these sites contributed substantially to the evidence on human carcinogenicity of PFOA and, to a lesser degree, of PFOS. The three US cohorts were the PFOAmanufacturing facility in Cottage Grove, Minnesota (Gilliland and Mandel, 1993; Lundin et al., 2009; Raleigh et al., 2014), the polymer-production plant in Parkersburg, West Virginia (Leonard et al., 2008; Steenland and Woskie, 2012; Steenland et al., 2015), and the fluorochemical-production facility in Decatur, Alabama (Alexander et al., 2003; Alexander and Olsen, 2007). A number of plants in Europe and the Parkersburg polymer-production facility were included in the pooled international TFE cohort (Consonni et al., 2013), and a small PFASmanufacturing plant in the Veneto region, Italy, contributed data on workers with extremely high PFOA serum concentrations (Girardi and Merler, 2019).

The occupational cohort studies were distinguished by PFOA or PFOS serum concentrations that were up to two orders of magnitude higher in workers than in the background population, with exposure contrasts facilitating the evaluation of exposure-response relations. The occupational cohorts generally had several decades of follow-up since first exposure, and exposure profiles were dominated by only a few PFAS compounds, depending on manufacturing processes. Thus, occupational exposures to either PFOA and PFOS were usually confined largely to one or the other, without being mixed. However, co-exposure to other PFAS and TFE was possible in some studies. Some occupational cohorts had information on blood concentrations of PFOA and/or PFOS, which were used to develop company-specific JEMs estimating serum levels across time and jobs. This provided reliable information, given that blood levels represent an internal dose resulting from all exposure routes and may be superior to exposure metrics based on air concentrations that account for exposure only by the inhalation route.

A major limitation of occupational cohort studies is their relatively small sample sizes for specific cancers, inability to address cancers of the breast and female genital tract (since most chemical-plant workers are men), and the use of mortality rather than incidence by most of the studies. Mortality studies provide weaker data on etiology and generally smaller sample sizes than do incidence studies, especially for cancers with a low fatality rate. Another concern with occupational studies is that some are based on cross-sectional samples excluding workers at risk before cohort enrolment, although this will affect cancers with a low fatality rate to a lesser extent and may create only a rather weak downward bias for cancers with a high rate of fatality. Moreover, mortality studies often lack histological data, which might lead to attenuation of risk estimates for all cases combined if the risk is associated only with certain histological subtypes. Despite these caveats, the Working Group considered three occupational studies to be particularly informative, i.e. those by Alexander and Olsen (2007), Raleigh et al. (2014), and Steenland et al. (2015). All three used incidence data and provided risk estimates for a range of cancers according to cumulative quantitative exposure metrics, although the study by Alexander and Olsen (2007) still suffered from small numbers.

(b) Studies of high environmental exposure

These studies can be particularly informative because, like occupational studies, they may enable detection of health effects that may be less marked in general population studies with low exposures. Only three high-level environmental exposure studies were available: <u>Barry</u> et al. (2013), Vieira et al. (2013), and Li et al. (2022a). Exposure levels and contrasts in these settings were between the low background levels of the general population and the very high occupational levels. They benefited from large study populations, resulting in more precise risk estimates and high comparability of exposed and unexposed people recruited from the same geographical regions.

The C8 Science Panel Project included a group of workers (11.5% of the cohort) with occupational exposure to primarily PFOA at a polymer-production plant in Parkersburg, West Virginia, USA, and residents in the Mid-Ohio Valley USA, which was contaminated by emissions from this facility from the late 1940s until about 2005 (Barry et al., 2013). This Mid-Ohio Valley cohort had an important strength in its modelling of individual cumulative serum PFOA concentrations from birth onwards based on a large number of parameters, including plant emission data, measured drinking-water levels, residential histories, individual consumption of tap water, and toxicokinetic data for PFOA in humans. The estimated PFOA serum concentrations correlated well with a large number of PFOA measurements made in 2005 and 2006. The serum concentrations of PFOA, PFHxS, and PFNA were elevated by about 500%, 75% and 40%, respectively, compared with US background levels, whereas the PFOS serum concentration was not increased (Frisbee et al., 2009).

The case–control study of West Virginia and Ohio residents (Vieira et al., 2013) somewhat overlapped the C8 Science Panel study. Since this non-nested case–control study was based upon cancer registry records from a longer period and from larger geographical areas, the total number of cancer cases was higher than those of the analyses by <u>Barry et al. (2013)</u>. This provided more accurate risk estimates, which is particularly important when considering rare cancer types. However, a limitation was the reference group that comprised people with other cancers (except pancreatic, kidney, testicular, and liver cancer), which may attenuate risk estimates if these other cancers are also associated with PFAS. Also, exposure misclassification was of some concern, because the residential address used to assign exposure based on the same model used in <u>Barry et al. (2013)</u> was known only at the time of diagnosis (for details, see Section 2.1.22). However, since the error could mostly be of Berkson type, exposure misclassification might not cause substantial attenuation of risk estimates (<u>Armstrong, 1998</u>).

The third study, in the Ronneby Register cohort, included more than 60 000 community residents living in an area where parts of the population received drinking-water contaminated with PFAS from a nearby airfield (Li et al., 2022a). In contrast to the Mid-Ohio Valley cohort, the Ronneby Register cohort used a crude assignment of exposure based on earlier and current residential addresses, but it was supported by large exposure contrasts in PFAS concentrations in drinking-water across the various water supplies of the municipality. Another limitation of the Ronneby Register cohort, which is in general an issue for most non-occupational PFAS studies, was overlapping exposures to various PFAS. Whereas Ronneby had very high levels of PFOS and PFHxS, levels of PFOA greatly overlapped those of the unexposed Swedish population in the region.

(c) Studies in the general population with background exposure

Studies of subsets of the general population were often case-control studies nested within large cohorts or trials created for other purposes. With this design it is possible to cost-effectively sample large series of cases of a specific cancer; to take advantage of individual data on social, lifestyle, and health issues of particular relevance for a specific cancer; to use frozen blood samples to obtain prediagnostic measurements of contaminants; and to limit potential bias and

372

confounding by matching on relevant characteristics. The main limitation pertaining to population-based studies is low exposure levels, low exposure contrasts, and background exposure to numerous other PFAS. Several chlorinated persistent organic pollutants are also widespread and have even longer biological half-lives than do PFAS, but the two classes of chemicals do not share physicochemical characteristics and in general serum concentrations are not correlated. Positive findings that are not corroborated in studies of high-exposure contrast (e.g. occupational or high environmental exposures) may seem contradictory, although for many carcinogens it has been shown that risk increases greatly with increasing levels at low exposure and then tails off or reaches a plateau at higher exposures (Stayner et al., 2003; Lanphear, 2017; Steenland et al., 2022). Suggested biological explanations include saturation of metabolic pathways, enhanced detoxification, and greater DNA repair efficiency at higher exposure levels (Stayner et al., 2003). Increasing exposure measurement error with increasing level of exposure can also result in the exposure-response relation reaching a plateau (Stayner et al., 2003). Healthy-worker survivor bias may also be a factor reducing the apparent risk in occupational cohort studies.

Despite limitations, several case-control studies nested within large cohorts were considered informative for this evaluation. They included studies based upon the Danish Diet, Cancer, and Health Cohort, addressing associations of PFOA and PFOS with cancers of the urinary bladder, prostate, liver, and pancreas in men (Eriksen et al., 2009); four studies based on the intervention arms in the PLCO Trial, addressing cancers of the kidney (Shearer et al., 2021), breast (Chang et al., 2023), prostate (Rhee et al., 2023a), and pancreas (Zhang et al., 2023); the US Air Force servicemen cohort, addressing testicular cancer (Purdue et al., 2023); two studies based on the US MEC, addressing HCC (Goodrich et al., 2022) and kidney cancer (Rhee

et al., 2023b); a study based in the ATBC Study in Finland, on pancreatic cancer (Zhang et al., 2023); a study of women in the FMC, addressing thyroid cancer (Madrigal et al., 2024); a casecohort study on the association between PFAS and cancers of the kidney, pancreas, breast, prostate, and lymphatic and haematopoietic tissue among participants in the ACS CPS-II LifeLink Cohort (Winquist et al., 2023); and a small nested case-control study evaluating thyroid cancer in New York, USA (van Gerwen et al., 2023). Finally, four nested case-control studies with prediagnostic PFAS measurements, which addressed risk of breast cancer in population samples with a low level of exposure, i.e. a study of women in the French education system (E3N; Mancini et al., 2020a); the Danish National Birth Cohort (Ghisari et al., 2017); the US Child Health and Development Cohort (<u>Cohn et al., 2020</u>); and the Dongfeng-Tongji cohort of female retirees from a large motor company in China (Feng et al., 2022).

A number of hospital-based and non-nested case-control studies were considered less informative, because the control groups did not clearly represent the same population from which the cases were chosen, resulting in potential unpredictable bias. Moreover, the exposure assessment in these studies was based on postdiagnostic measurements of PFAS in blood samples, which are expected to provide less-reliable information on exposure during the relevant time windows than do prediagnostic baseline samples. Risk estimates may be biased if prodromal disease states, the fully developed disease, or the treatment affect serum concentrations of PFAS (this is labelled reverse causation), but little is known on this issue and the direction of bias, if any, is unpredictable. For these reasons, such casecontrol studies and one nested case-control study (Hurley et al., 2018) were given less weight when balancing the epidemiological evidence for causal associations for cancers of the breast (Wielsøe et al., 2017; Tsai et al., 2020; Itoh et al., 2021; Li et al., 2022b; Velarde et al., 2022), thyroid

(Liu et al., 2022; Li et al., 2023), prostate (Hardell et al., 2014) and liver (Cao et al., 2022).

2.8.2 Exposure assessment quality considerations

Information on individual cumulative exposure to PFOA and/or PFOS that enabled analyses of exposure–response relations including lagged analyses was considered of critical importance for the evaluation of epidemiological studies on the carcinogenicity of PFOA and PFOS. A systematic description and appraisal of exposure assessment in all available epidemiological studies is provided in Section 1.6.1.

Among nine occupational cohort studies in which exposure assessment primarily relied on job history, three studies focusing on PFOA and one study on PFOS used quantitative estimates of cumulative exposure (Alexander and Olsen, 2007; Steenland and Woskie, 2012; Raleigh et al., 2014; Steenland et al., 2015). These estimates were derived from company-specific JEMs informed by industrial hygiene and/or biological measurements and accounted for temporal shifts in exposure levels. In particular, the approach used in the study by Steenland and Woskie (2012) (and Steenland et al., 2015), which incorporated industrial hygiene and biological measurements into modelled serum concentrations, was considered superior to the others.

The inevitable misclassification of exposure related to group-based exposure assignment in these studies may not necessarily cause attenuation of risk estimates towards the null. Depending on the degree of Berkson-type measurement error, it may primarily result in unbiased but less precise risk estimates (Armstrong, 1998). However, errors involved in group mean exposure measurement used in JEMs also could cause bias towards or away from the null. Exposure assessment in the small cohort of PFAS-manufacturing workers in Italy was also modelled via cumulative PFOA serum concentrations based upon a JEM informed by measurements, but the effects of considerable co-exposure to other PFAS compounds were not accounted for by the analyses (Girardi and Merler, 2019). Other occupational cohort studies applying crude or semiquantitative assignments of exposure levels and without quantitative estimates of individual cumulative exposure were considered at higher risk of exposure misclassification for lifetime exposure and therefore provided less-reliable risk estimates (Alexander et al., 2003; Leonard et al., 2008; Lundin et al., 2009; Consonni et al., 2013).

Among the three studies addressing risk of cancer in residents living in areas contaminated by local PFOS and/or PFOA emissions, one study of the population in the Mid-Ohio Valley, West Virginia, USA, was considered particularly informative because of the modelled annual serum PFOA concentrations from birth onwards (Barry et al., 2013), and another partly overlapping study assessed exposure 10 years before diagnosis, making the exposure assessment slightly less informative (Vieira et al., 2013). The Ronneby Register cohort study in Sweden had well-documented, strong contrasts among residents with respect to PFOS and PFHxS serum concentrations, but the exposure assessment was entirely based upon timing and duration of residence at contaminated and uncontaminated addresses and did not allow estimation of the effects of the individual compounds (Li et al., 2022a).

All studies addressing risk of cancer in general population samples used the concentration of PFAS in at least one blood sample as a proxy for cumulative exposure. This approach is supported by the long biological half-lives of PFOA and PFOS in humans (see Section 4.1 for details) and some indications of high stability of blood concentrations across several years within individuals who provided repeated samples (Blake et al., 2018; Purdue et al., 2023; Rhee et al., 2023a). Furthermore, simulation studies based on available data with repeated measurements up to 8 years apart indicated that bias towards the null because of non-differential misclassification would be modest (Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: <u>https://publications.</u> <u>iarc.who.int/636</u>).

2.8.3 Co-exposures to other agents of relevance to cancer hazard identification

Mutually independent information on the carcinogenicity of PFOA and PFOS in humans in the available epidemiological studies was obtained by two main approaches.

First, some occupational and environmental settings were associated with exposure to specific PFAS compounds at levels many times as high as background levels, whereby co-exposure to other PFAS compounds above background levels was unlikely, given the characteristics of the production processes and sources of exposure. This applied to the occupational cohorts of workers at the APFO-producing plant in Cottage Grove, Minnesota, USA (Raleigh et al., 2014); the studies of fluoropolymer-production workers in Parkersburg, West Virginia, USA (Steenland and Woskie, 2012; Steenland et al., 2015); and the C8 Science Panel cohort of workers and residents of contaminated areas of the Mid-Ohio Valley, USA (Barry et al., 2013; Vieira et al., 2013). In all these studies, PFOA serum concentrations were substantially elevated above background levels, whereas PFOS serum concentrations were not. Serum concentrations of PFHxS and PFNA were also somewhat above background levels in Mid-Ohio Valley residents, but the correlation with PFOA was modest, indicating that exposure via a source other than the plant in Parkersburg, West Virginia, was likely (Frisbee et al., 2009). Moreover, co-exposure to TFE (classified in IARC Group 2A; IARC, 2016) may have occurred at some European workplaces (Consonni et al., 2013) but was considered unlikely at the plant in

Parkersburg because use was strictly controlled under normal operations (Steenland and Woskie, 2012). For PFOS, there were no occupational or environmental settings without some co-exposure to other PFAS compounds or carcinogens. The occupational cohort of fluorochemical-production workers in Decatur, Alabama was characterized by high exposure to PFOS, but exposure to several other fluorochemicals including PFOA was possible or likely (Alexander et al., 2003; for details, see Table S1.22, Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: https://publications.iarc. who.int/636). Likewise, the Ronneby Register cohort was characterized by PFOS exposures an order of magnitude above background levels, but levels of PFHxS were also substantially higher than background levels whereas PFOA levels were not (Li et al., 2022a).

Second, in all studies of general population samples, PFAS exposure was mixed. Most studies estimated exposure by measurement of PFAS compounds in one or more blood samples, and estimates were typically provided for both PFOA and PFOS. In some studies, mutual adjustment was performed for the effects of other PFAS compounds (Cohn et al., 2020; Shearer et al., 2021; Wen et al., 2022; Chang et al., 2023; Purdue et al., 2023; Rhee et al., 2023a, b; Madrigal et al., 2024), which helped to identify individual effects. The correlation coefficients of PFOA and PFOS in the above studies ranged from 0.50 (Wen et al., <u>2022</u>) to 0.70 (<u>Rhee et al., 2023a</u>), and therefore the possibility of unstable statistical models or overadjustment was unlikely. The same concern applied to correlations between PFOA, PFOS, and other common legacy PFAS (Shearer et al., 2021; Rhee et al., 2023b).

2.8.4 Bias and confounding

Exposure- and outcome-dependent selection into studies was not considered to be an important source of bias in the most informative studies available for this evaluation. Most occupational studies were based upon rosters kept by major companies (Alexander and Olsen, 2007; Raleigh et al., 2014; Steenland et al., 2015), and the large, nested case-control studies mostly used existing independent databases or public registries to define the study populations (see Table 2.1). Selection bias in terms of a healthy-worker effect (sometimes viewed as a confounder) and healthy-worker survivor bias is of concern when considering occupational studies. Both would be expected to lead to downward bias. The former bias can be mitigated by using internal rather than external comparisons, and the latter is of less concern if there is little evidence that high exposure is associated with leaving employment in a highly exposed job or altogether. Furthermore, compared with studies on other chronic diseases, studies on cancer may be less susceptible to both healthy-worker effects and healthy-worker survivor effects.

The C8 Science Panel cohort of workers and residents in the PFOA-contaminated Mid-Ohio Valley area in Ohio and West Virginia was considered particularly informative for this evaluation because of its size, validated estimates of cumulative internal exposure, large exposure contrast, and extensive covariate information. It was mainly based upon a cross-sectional population sample of residents alive at the time of interview and with most at-risk years occurring before baseline interviews. Selection bias was unlikely because the participation rate was about 80%, and the data did not indicate preferential participation of residents from contaminated areas who had a history of cancer (Barry et al., 2013). Moreover, simulation analyses demonstrated that lacking information about fatalities occurring in the population before enrolment would not affect risk estimates (Barry et al., 2015), unless survival after diagnosis was associated with exposure level, judged a priori to be unlikely.

In most studies, case identification and ascertainment were based upon population-wide cancer registries, death certificates, or death registries (or a combination of these), and in one study the additional data from personal recall of cancer were verified by medical records (<u>Barry</u> <u>et al., 2013</u>). The approaches for case identification in general were not expected to introduce major outcome misclassification.

Only a few informative studies distinguished subtypes of specific cancers, mainly for breast cancer (Hurley et al., 2018; Mancini et al., 2020a; Chang et al., 2023). Examples of environmental exposures causing risk of some but not of other specific cancer subtypes are few (e.g. wood dust causes adenocarcinoma but rarely causes squamous cell carcinoma of the sinonasal cavity; IARC, 2012). Considering that PFAS may modulate endocrine regulation and signalling (for details, see Section 4.2.8), there is a rationale for examination of receptor-defined subtypes of, particularly, breast cancer. However, since the effects of PFAS may depend on endogenous hormone levels and may be inhibitory in some situations but stimulatory in others, it is difficult to put forward a priori hypotheses, which complicates the interpretation of epidemiological findings. There is no mechanistic evidence indicating that the main subtypes of testicular cancer (seminoma, and non-seminoma) have different etiologies in young men in whom these tumours develop from carcinoma in situ in cells of developmental origin (Rajpert-De Meyts, 2006). It is disputed whether subtypes of testicular cancer for which incidence peaks later in adulthood have different etiologies (Coupland et al., 1999; Stang, 2009). At present, studies addressing specific cancer subtypes are mainly explorative and foremost of importance as starting points for forthcoming studies.

Demographic characteristics such as race, ethnicity, sex, age, residence area, socioeconomic status and calendar period are strong determinants of cancer and are also associated with PFAS exposure in the general population (Steenland et al., 2009; Eriksen et al., 2011; Buekers et al., 2018; Momenimovahed and Salehiniya, 2019; Rhee et al., 2023b). With few exceptions, these factors were controlled by design and/or analysis in the highly informative studies. Some studies, mainly nested case-control studies (for details, see Table 2.1), also accounted for the effects of smoking, alcoholic beverage consumption, and BMI, based on data collected by personal interview (Barry et al., 2013); however, such information was not available in the occupational cohort studies, the case-control studies by Vieira et al. (2013), and the Ronneby Register cohort study (Li et al., 2022a). In occupational studies, internal analyses and comparisons of exposed with unexposed workers in the same types of jobs from nearby plants during the same calendar period mitigated confounding due to differences in social and lifestyle factors, whereas confounding in studies of heterogeneous populations not accounting for these factors may result in bias in an unpredictable direction. Nested case-cohort and case-control studies designed to address one or more specific cancers often included information on a range of determinants of these specific cancers, such as hepatitis for primary liver cancer (Goodrich et al., 2022); hypertension and possible reduced glomerular filtration for kidney cancer (Shearer et al., 2021); reproductive factors for breast cancer (e.g. Ghisari et al., 2017; Cohn et al., 2020; Mancini et al., 2020a; Chang et al., 2023); or specific occupational exposures for bladder cancer (Eriksen et al., 2009).

As the main potentially confounding factors, sex, age, time, geography, socioeconomic status, and possibly race or ethnicity were measured and analysed with high accuracy, residual confounding by these factors was considered unlikely. Cooking practices such as frying and consumption of a number of food items (such as eggs, potatoes, red meat, snacks, and vegetables) have been associated with serum concentrations of PFOA and PFOS in a number of studies. In a general population sample, these factors explained 14% and 24% of the variation in concentrations of PFOA and PFOS, respectively (Eriksen et al., 2011). Various foodstuffs have also been associated with some cancers, and therefore confounding by diet (with unpredictable magnitude and direction) cannot be ruled out in the general population studies, whereas confounding by diet was very unlikely in the occupational studies and the studies of communities with high-level exposure, because the dietary intake of PFAS was marginal compared with the main source of exposure.

2.8.5 Specific cancer sites and exposure to PFOA

(a) Kidney cancer

Three partly overlapping studies of workers and residents in West Virginia and Ohio, USA, have consistently shown increased risk of kidney cancer in relation to occupational and/or highlevel environmental exposure (Steenland and Woskie, 2012; Barry et al., 2013; Vieira et al., 2013). The occupational mortality study reported an SMR for fluoropolymer workers in the highest exposed quartile of estimated cumulative PFOA serum concentrations compared with unexposed workers of 2.66 (95% CI, 1.15-5.24; 8 deaths) with indications of an exposure-response relation (Steenland and Woskie, 2012). The highly informative cohort study of workers and residents found an increasing risk of incident kidney cancer with increasing cumulative PFOA serum levels, albeit with borderline statistical significance (Barry et al., 2013). The adjusted hazard ratio for the fourth quartile of cumulative PFOA serum concentrations versus the first was 1.58 (95% CI, 0.88-2.84; 105 cases; linear trend test, P = 0.18; using the log continuous PFOA serum concentration, P = 0.10). Findings were consistent with results of the third partly overlapping study from this geographical area, a register-based case-control study in Ohio that reported an

adjusted odds ratio for incident kidney cancer in exposed people in the highest PFOA serum concentration quartile (110–655 μ g/L [ng/mL]), versus the unexposed, of 2.0 (95% CI, 1.0–3.9; total, 246 cases) (Vieira et al., 2013). In this study, there was some concern about the appropriateness of the control group, which comprised people with all other cancers excluding those of the testis, liver, and pancreas.

The results of two (less informative) occupational cohort studies did not corroborate or refute the above findings. The cohort study of APFO workers at the Cottage Grove facility in Minneapolis, Minnesota, USA did not find indications of an increased incidence of kidney cancer in exposed workers. The hazard ratio for the fourth quartile versus the unexposed workers (Saint Paul plant) was 0.73 (95% CI, 0.21-2.48; 16 exposed cases) and there was no indication of increasing risk across increasing quartiles of exposure (see Table 2.2). However, the wide confidence intervals were not incompatible with the effects observed in the earlier studies (Raleigh et al., 2014). The exposure metric was based upon air measurements of PFOA, which may be less reliable than biological measurements if exposure occurs through pathways other than inhalation or if there is large variation in pulmonary absorption of PFOA due to, for instance, differential use of respiratory protection equipment or high pulmonary ventilation in some physically demanding jobs. The international study of mortality in TFE-production workers (Consonni et al., 2013) was not informative because of the semiquantitative exposure assessment and the small number of cases (n = 10).

Unlike the above five studies of highly exposed populations, two nested case-control studies and a case-cohort study using a single prediagnostic PFOA serum concentration addressed risk associated with the much lower background exposure of the general US population (Shearer et al., 2021; Rhee et al., 2023b; Winquist et al., 2023). The study based upon the PLCO Trial cohort reported an adjusted odds ratio for RCC (constituting about 80-90% of all kidney cancers) in the highest exposure quartile (> $7.3-27.2 \mu g/L [ng/mL]$) versus the lowest (< 4.0 µg/L [ng/mL]) of 2.63 (95% CI, 1.33–5.20) (Shearer et al., 2021). Adjusted for other PFAS compounds, the OR was 2.19 (95% CI, 0.86-5.61). This relative risk for RCC observed in the general population was similar to that for kidney cancer observed among people with an exposure more than one order magnitude higher. If these associations are causal, this indicates a non-linear exposure-response relation with a steep increase in risk at very low exposure levels, which tails off or even reaches a plateau with higher exposure (Steenland et al., 2022). The other nested case-control study of an ethnically diverse US population with background exposure levels did not find an association between prediagnostic PFOA serum concentrations and risk of incident RCC overall (OR for a 1-unit increase in PFOA serum concentration on the \log_2 scale, 0.89; 95% CI, 0.67-1.18), but – consistent with the earlier findings - the risk was elevated in White participants (23% of the study population), albeit with wide confidence intervals (adjusted OR for a 1-unit increment in PFOA serum concentration on the log₂ scale, 2.12; 95% CI, 0.87–5.18; <u>Rhee</u> et al., 2023b). Finally, the case-cohort study conducted within the CPS-II LifeLink Cohort (in which 98% of participants were White) found no increased risk for all kidney cancers (HR for continuous log₂-plasma PFOA concentrations was 1.08; 95% CI, 0.88-1.33, 156 cases; Winquist et al., 2023). For RCC, the hazard ratio was 1.06 (95% CI, 0.83-1.35). Among women (38% of the case-cohort group), for all kidney cancers there was an increased hazard ratio of 1.33 (95% CI, 0.97-1.83; 65 cases) and for RCC it was 1.54 (95% CI, 1.05-2.26; 42 cases). Of note was that this was a "survivor cohort", in which the median age when follow-up started was 70 years, about 8 years after enrolment began in the CPS-II. At age 40-60 years, the rate of RCC is twice as high in men as in women, which could have contributed to a differential survivor effect by sex (Mancini et al., 2020b; NCI, 2023).

The Working Group concluded that increased risks of kidney cancer overall or RCC in relation to PFOA exposure were reported by two independent and highly informative studies (Barry et al., 2013; Shearer et al., 2021). These studies included large study populations and long follow-up, across which there were large exposure contrasts spanning background, high environmental, and extremely high occupational exposure. There was comprehensive individual-level assessment of cumulative exposure in one of these studies (Barry et al., 2013). Exposureresponse relations were observed overall in these two independent populations. The findings were not corroborated overall by those of two other less-informative occupational studies (Raleigh et al., 2014; Consonni et al., 2013), and only among subgroups in two other general population studies (Rhee et al., 2023b; Winquist et al., 2023). In the random-effects meta-analysis conducted by the Working Group, which was based on six studies (three of which were from the Mid-Ohio Valley, as well as Shearer et al., 2021, Rhee et al., 2023b, and Winquist et al., 2023) a meta-rate ratio per 10 ng/mL of 1.16 (95% CI, 0.98-1.38; $I^2 = 0.91$) was estimated for PFOA. The limitations of the meta-analysis were estimation of the linear exposure-response relation from two categorical data points, assumptions about the duration of exposure in three studies, assumptions about the midpoint in the high-exposure category in three studies, and lack of independence of three of the studies.

Taken together, the body of epidemiological evidence indicated that a causal association between PFOA and RCC is credible, but the evidence was not considered sufficiently consistent to rule out chance and bias with confidence. The studies did not allow for an evaluation of kidney cancers of non-RCC histology subtype.

(b) Testicular cancer

The cohort study of the highly exposed Mid-Ohio Valley population with PFOA exposure substantially above background levels (Barry et al., 2013) and the nested case-control study of US Air Force servicemen with exposure levels in the range of the general US population (Purdue et al., 2023) were considered the most informative for the evaluation of testicular cancer. Barry et al. (2013) reported an adjusted hazard ratio for incident testicular cancer of 1.34 (95% CI, 1.00–1.79) for a 1-unit increase in natural log-transformed serum concentrations in unlagged analyses. This observation was not corroborated by <u>Purdue</u> et al. (2023), who reported an OR for testicular germ cell tumour (TGCT; the most common type of testicular cancer) of 0.8 (95% CI, 0.5–1.4), comparing the highest exposed quartile with the lowest, based on 530 cases and matched controls. There was no indication of an exposure-response relation (P for trend, 0.86), and similar results were observed in an analysis of the second sample collected in a subset of the population. The range of measured serum PFOA levels in 2005-2006 was 0.25-4752 ng/mL (median, 24.2 ng/mL) for residents in the study by Barry et al. (2013) compared with a geometric mean of 5.8 ng/mL for controls in the study by Purdue et al. (2023). Therefore, the higher exposure contrast in the former study may explain the discrepant findings. Moreover, the study of Air Force servicemen did not control for residential area, which may cause bias in an unpredictable direction. Findings in the study by Vieira et al. (2013) were compatible with an increased risk of testicular cancer, but the cancer cases included somewhat overlapped the cases in the study by Barry et al. (2013), and the study offered no improvements in design or analysis.

One additional population was exposed to high levels of PFOA (and to a much lesser extent other PFAS) resulting from industrial contamination in the Veneto region of Italy, and serum concentration data (more than 18 000 measurements in 2016 among those aged 14-39 years) were reported by municipality (n = 21) by <u>Pitter et al. (2020)</u>. Orchiectomies by the same groupings of municipality in the Veneto region between 1997 and 2014 were reported separately (Sistema Epidemiologico Regionale, 2016). Orchiectomy was found to have high sensitivity and positive predictive value for testicular cancer in this region (Sistema Epidemiologico Regionale, 2016). The Working Group combined the serum and orchiectomy rate data and observed a strong positive correlation (Spearman correlation, 0.57; P = 0.006; 21 cases) between serum PFOA concentrations and rates of orchiectomy (standardized on age by 5-year age groups from ages 15 to 54 years to the overall regional rate) by municipality.

Few other studies addressed the of the association between PFOA exposure and testicular cancer. Results of two occupational mortality studies were also compatible with an increased risk but were based on very few cases (< 3), not permitting detailed analysis (Steenland and Woskie, 2012; Consonni et al., 2013). Moreover, these studies of mortality were considered less informative, owing to the high survival rate for testicular cancer, since mortality-based risk estimates reflect a mix of etiological and prognostic factors.

The Working Group concluded that there were indications in two independent populations for an increased risk of testicular cancer associated with PFOA serum concentrations in residents with a high level of exposure. In the third informative study, a null association was seen, but exposure levels were at background in this population. Overall, the Working Group concluded that a positive association between PFOA and testicular cancer is credible. However, chance and/or bias could not be ruled out as explanations for these findings, given the small number of cases in the few available studies and that one of the positive studies was of ecological design.

(c) Bladder cancer

Two occupational cohort studies (Raleigh et al., 2014; Steenland et al., 2015), the cohort study of the Mid-Ohio Valley population with high exposure (Barry et al., 2013), and the partly overlapping registry-based case-control study (Vieira et al., 2013) provided data on the incidence of bladder cancer in relation to individual-level estimates of cumulative PFOA exposure. None of these studies that included large study populations with a strong exposure contrast indicated an increased risk of bladder cancer in relation to PFOA exposure, and the results were consistent with those of the Danish Diet Cancer and Health Cohort study (Eriksen et al., 2009) and a large US case-cohort study (Winquist et al., 2023). The former study only addressed low-level background exposure and did not adjust for co-exposure to PFOS, but PFOS was not associated with increased risk. Finally, the international occupational mortality study did not observe an increased risk of fatal bladder cancer (Consonni et al., 2013), and some indication of increased risk of fatal bladder cancer in an occupational cohort study (Steenland and Woskie, 2012) was not corroborated by the subsequent incidence study that had an improved exposure assessment (Steenland et al., 2015).

The Working Group concluded that the epidemiological evidence in aggregate did not indicate a positive association between PFOA at environmental or occupational exposure levels and urinary bladder cancer, but noted that the occupational cohort studies in particular include few exposed cases, limiting informativeness, and that exposure misclassification may have biased associations towards the null.

(d) Prostate cancer

Altogether, six studies on the risk of prostate cancer and PFOA exposure were fairly consistent in reporting null or inverse associations regardless of study design, type of population (background exposure, high environmental or occupational exposure), method of exposure assessment (estimates of external exposure using various approaches or measurements of serum concentration) or outcome (incident cases or mortality) (see Section 2.3.2 for details). This collection of studies included highly informative studies with a large exposure contrast and lifelong estimates of cumulative PFOA serum concentrations (Barry et al., 2013); high comparability of exposed and non-exposed (Rhee et al., 2023a); extensive control for potential confounding, also including education, BMI and diet (Eriksen et al., 2009); a large study size (Winquist et al., 2023); and examination of more aggressive (i.e. fatal) prostate cancer (Steenland and Woskie, 2012; Raleigh et al., 2014), although cases were few in the latter. One study of incident prostate cancer in an occupational cohort found a higher risk in the second to fourth quartiles based on estimated cumulative PFOA exposure, compared with the lowest quartile, but without a consistent exposure-response trend (Steenland et al., 2015).

An inherent issue in all studies of incident prostate cancer was detection bias in populations undergoing different levels of medical surveillance, because of the common occurrence of latent disease that may be detected by blood assays, and which may cause bias in an unpredictable direction.

The Working Group concluded that the results of several studies addressing diverse populations in different countries and with different designs fairly consistently did not indicate an association between exposure to PFOA and prostate cancer, but considering exposure misclassification that most probably caused bias towards the null, issues related to outcome ascertainment, and latency periods of < 30 years in several studies, the epidemiological evidence did not preclude that such associations may exist.

(e) Breast cancer

The available evidence included two occupational cohort studies with high-level exposure to PFOA but with few cases of incident (Raleigh et al., 2014) or fatal (Steenland and Woskie, 2012) breast cancer, one large cohort study addressing high-level environmental exposure of community residents on the basis of modelled estimates of individual cumulative lifelong serum PFOA concentrations (Barry et al., 2013), and a case-control study of Mid-Ohio Valley residents with a high level of exposure (Vieira et al., 2013, partly overlapping Barry et al., 2013). These studies found no associations between PFOA exposure and breast cancer overall but did not separately evaluate pre- and postmenopausal cancer or subtypes of breast cancer. Moreover, three large cohort or nested case-control studies addressing background exposure of the general population did not report increased risk of incident breast cancer with increasing prediagnostic PFOA serum concentrations, either overall or for pre- or postmenopausal breast cancer, when these were analysed separately (Ghisari et al., 2017; Cohn et al., 2020; Chang et al., 2023). These studies extensively controlled for confounders; however, most could not address risk for specific subtypes. A large case-cohort study of female retirees from a motor vehicle company in China (Dongfeng-Tongji cohort; see Section 2.1.14 for details) reported higher risk of incident breast cancer with higher levels of prediagnostic PFOA plasma concentrations (Feng et al., 2022). The hazard ratio for the highest quartile of PFOA serum concentration versus the lowest was 1.69 (95% CI, 1.05–2.70), with a positive trend. It was not clear how this selected sample of retirees compared with the general population of Chinese women and how selection might influence risk estimates. Moreover, a large cohort study in France of primarily teachers (E3N cohort; see Section 2.1.10 for details) found an increased risk with wide confidence intervals for

postmenopausal cancer in the second quartile of prediagnostic PFOA serum concentrations (OR, 1.69; 95% CI, 0.89–3.21) but not in higher quartiles (<u>Mancini et al., 2020a</u>). There was no indication of an exposure–response relation, and risk estimates were not adjusted for PFOS, which also was related to risk of breast cancer in this study.

The association between PFOA and breast cancer was also studied in a large, nested case-control study of US teachers (CTS; see Section 2.1.8 for details, Hurley et al., 2018) and five non-nested case-control studies that all evaluated associations with background exposure of the general population (Wielsøe et al., 2017; Tsai et al., 2020; Itoh et al., 2021; Velarde et al., 2022; Li et al., 2022b). In spite of methodological strengths, such as extensive control for potential confounding, including a range of known determinants related to reproduction, lifestyle, and other environmental contaminants in some of the studies, all were based on measurement of PFOA in postdiagnostic blood samples, and only three studies specified whether efforts were made to collect samples before treatment (Tsai et al., 2020; Velarde et al., 2022; Li et al., 2022b). Moreover, several studies were limited by small sample sizes and control groups with questionable representativeness of the population from which cases were recruited. Findings with respect to breast cancer overall in relation to PFOA exposure were diverse in these six studies - some studies provided indications of increased risk (Wielsøe et al., 2017; Li et al., 2022b), others did not (Hurley et al., 2018; Tsai et al., 2020; Itoh et al., 2021; Velarde et al., 2022).

Breast cancer is a heterogeneous tumour type, and subtypes defined by different ER or PR characteristics may have different etiologies (<u>Yager</u> <u>and Davidson, 2006</u>). Therefore, it might assist causal inference to distinguish risk according to tumour subtype – not least because there is evidence that PFAS compounds may interfere with receptor-mediated hormonal signalling (see Section 4.2.8 for details). Five studies of incident breast cancer in general population samples provided risk estimates according to breast cancer receptor characteristics, but with inconsistent results. The large, nested case-control studies of the E3N cohort (Mancini et al., 2020a) and the PLCO cohort (Chang et al., 2023) both used prediagnostic serum samples and found some indications of increased risk related to all receptor subtypes (Mancini et al., 2020a; ER+, ER-, PR+, PR-) or to some but not others (Chang et al., 2023; ER-, PR-, ER-/PR-) but without an exposure-response relation, with limited statistical power and without adjustment for the effects of other PFAS (except Chang et al., 2023). Among three case-control studies that used cross-sectional sampling of blood specimens, Li et al. (2022b) reported increased risk with an exposure-response pattern for ER+ and PR+, but not for ER- and PR-. Itoh et al. (2021) reported null or reduced risk in all examined receptor type combinations (ER+/PR+; ER+/PR-; and ER–/PR–), and <u>Tsai et al. (2020)</u> found (with one exception) null or reduced risk in ER+ and ERsubtypes. Although some of these studies were distinguished by extensive adjustment for potential confounders, including both known determinants for breast cancer and, in some cases, other persistent organic compounds (Itoh et al., 2021), they had other methodological drawbacks (for details, see Section 2.4 and Table 2.4). A general issue pertaining to many studies examining receptor subtypes was low statistical power, which complicates causal inference. Similarly, the only study with analyses stratified by polymorphisms in selected xenobiotic and metabolizing genes had limited informativeness because of insufficient statistical power (Ghisari et al., 2017). Finally, a systematic review and metaanalysis included 18 papers of which 11 were eligible for meta-analysis (Chang et al., 2024). The summary rate ratio per 1-unit increase in natural log-transformed serum or plasma PFOA concentration was 0.95 (95% CI, 0.77-1.18) in analyses including all risk estimates. Excluding

382

studies that assessed exposure after diagnosis of breast cancer revealed a summary rate ratio of 1.16 (95% CI, 0.96–1.40). There was considerable heterogeneity across studies.

The Working Group concluded that the most informative epidemiological studies showed a slightly elevated but uncertain association with PFOA. Overall, the two most informative studies (Mancini et al., 2020a; Chang et al., 2023) were null overall but were the only prospective studies that examined postmenopausal breast cancer cases by ER/PR receptor status. Both found non-linear positive associations with ER- and PR- postmenopausal breast cancer. The statistical power was low in studies examining associations with specific tumour subtypes or stratified by levels of endogenous hormone levels (pre- or postmenopausal cancer), limiting the ability to identify causal associations. Moreover, there were few data on risk at exposure levels above background. Overall, despite some evidence of associations for certain subgroups, the available epidemiological evidence was not considered sufficiently consistent to permit a conclusion to be made about the presence of a causal association between exposure to PFOA and breast cancer.

(f) Thyroid cancer

The study of residents with high environmental exposure in the Mid-Ohio Valley, USA, included 86 cases of validated incident cancer and was considered the most informative of a total of seven studies providing data on risk of incident cancer of the thyroid gland (<u>Barry et al., 2013</u>). This study found indications of increased risk of incident thyroid cancer overall but no exposure– response relation, attenuated 10-year lagged risk estimates, and wide confidence intervals (HR per unit cumulative serum PFOA concentration, natural log scale, no lag, 1.10; 95% CI, 0.95–1.26). The corresponding hazard ratio in the subset of workers with substantially higher exposure at the polymer-production plant in Parkersburg,

West Virginia, USA, was 1.93 (95% CI, 1.00–3.71) but with strong attenuation in 10-year lagged analyses based on 8 cases (Barry et al., 2013). The partly overlapping case-control studies in West Virginia and Ohio based on 343 cases did not observe an increased risk of incident thyroid cancer in relation to PFOA exposure (Vieira et al., 2013). A case-control study nested within a cohort of nulliparous pregnant women in Finland found no associations overall but weak associations in women aged > 40 years at diagnosis (OR, 1.20; 95% CI, 0.71-2.01) (Madrigal et al., 2024), and a small nested case-control study likewise reported null results (van Gerwen et al., 2023). Two occupational mortality studies of predominantly male workers were considered uninformative because of low numbers and because mortality is a less appropriate outcome measure because of the high survival rate for thyroid cancer (Leonard et al., 2008; Lundin et al., 2009). Finally, two case-control studies in China addressing background exposure levels reported strong inverse associations between postdiagnostic PFOA serum concentrations and risk of thyroid cancer (Liu et al., 2022; Li et al., 2023).

The Working Group concluded that there was no consistent epidemiological evidence for increased risk of thyroid cancer in relation to occupational or environmental exposure to PFOA across the available studies, which generally had small numbers of cases.

(g) Liver cancer

Of the nine studies addressing the association between PFOA and liver cancer, the cohort study of residents with high environmental exposure in Mid-Ohio Valley, USA, (<u>Barry et al.</u>, 2013) and the Diet, Cancer and Health Cohort (<u>Eriksen et al.</u>, 2009) were considered particularly informative. These cohorts provided incidence data, had large and well-characterized study populations with high completeness of verified cases, validated lifelong estimates of cumulative internal exposure or prediagnostic serum sampling, and meticulous control for confounding, including adjustment for tobacco smoking, alcoholic beverage consumption, and - in one study - occupation as a waiter or cook, which have been associated with risk of liver cancer. These studies found no associations between levels of PFOA exposure and risk of liver cancer. This finding was also consistent with results of the Ohio and West Virginia cancer registry-based case-control studies (Vieira et al., 2013; for details, see Section 2.1.22), which was also in a highly exposed population (overlapping with <u>Barry et al., 2013</u>). Six other studies of liver cancer reported similar essentially null results but were considered less informative. Four of these were occupational cohort studies with very high exposure contrast but with too few cases for causal inference (Steenland and Woskie 2012; Consonni et al., 2013; Raleigh et al., 2014; Girardi and Merler, 2019). A nested case-control study of the MEC was distinguished by its study of non-viral HCC and found no associations with prediagnostic PFOA plasma concentrations (Goodrich et al., 2022). Finally, a hospital-based case-control study in China was less informative because of various methodological limitations (Cao et al., 2022; for details, see Section 2.5.1). The Working Group concluded that most findings for liver cancer were null, and that most studies, including the one positive high-exposure study (Girardi and Merler, 2019), had few cases.

(h) Pancreatic cancer

Two occupational cohort studies of mortality (Steenland and Woskie, 2012; Raleigh et al., 2014) and incident cancer (Raleigh et al., 2014), a cohort study of a community with high environmental exposure (Barry et al., 2013), and the large nested case-cohort or case-control studies addressing background exposures in the general population in Denmark (Eriksen et al., 2009), in the US PLCO cohort (Zhang et al., 2023), and the US CPS-II LifeLink Cohort

(Winquist et al., 2023) found no indications of increased risk of pancreatic cancer in relation to PFOA exposure. In contrast, a case-control study of male smokers nested within a cancer prevention study in Finland reported an overall increased risk of pancreatic cancer (OR per SD increase in PFOA on the log₁₀ scale, 1.27; 95% CI, 1.04-1.56) (Zhang et al., 2023). The reasons for these discrepant findings compared with earlier studies were unknown. Men within the PLCO cohort who had ever smoked did not have an increased risk of pancreatic cancer (Zhang et al., 2023; see Table S2.5; Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: https://publications.iarc. who.int/636). Finally, in the international occupational mortality study of TFE synthesis and polymerization workers, the SMR for workers with the highest cumulative exposure estimate versus the national reference rate was 1.84 (95% CI, 0.67-4.00; 10 exposed cases) (Consonni et al., 2013).

The Working Group concluded that the epidemiological evidence on risk of pancreatic cancer at high levels of occupational exposure to PFOA concerned very few exposed cases and that findings in studies addressing high environmental and background levels were generally null.

(i) Colorectal cancer and cancers of the digestive tract other than liver and pancreas

The most informative occupational cohort study that investigated risk of colorectal cancer found increased incidence in the third and fourth quartiles versus the first quartile of estimated cumulative PFOA serum concentrations but with wide confidence limits and without an exposure-response relation (Steenland et al., 2015). The results of four occupational mortality studies were conflicting, but all reported on very few exposed cases (Leonard et al., 2008; Lundin et al., 2009; Consonni et al., 2013; Girardi and Merler, 2019). The case-control study in West Virginia and Ohio, USA, reported an increased incidence of colorectal cancer in participants with high, but not very high, estimated PFOA serum levels (OR, 1.3; 95% CI, 1.0-1.7; Vieira et al., 2013), but these findings were not corroborated by those of the partly overlapping study of Mid-Ohio Valley residents (Barry et al., 2013). Risk of oesophagus and stomach cancer was addressed by the Mid-Ohio Valley study (Barry et al., 2013), with essentially null findings, and by four occupational mortality studies (Leonard et al., 2008; Lundin et al., 2009; Consonni et al., 2013; Girardi and Merler, 2019). The findings of these studies were conflicting but all - except Consonni et al. (2013) - included fewer than 5 exposed cases (see Section 2.5.3 and Table S2.5 for details; Annex 4, Supplementary material for Section 2, Cancer in Humans, online only, available from: <u>https://publications.iarc.who.int/636</u>).

The Working Group concluded that there was no clear or consistent epidemiological evidence for an increased risk of cancer of the colorectum, oesophagus, or stomach in relation to PFOA exposure. There were no studies in low-exposure general populations. In occupational studies or studies on high environmental exposure, there were very few exposed cases, resulting in highly uncertain risk estimates.

(j) Cancers of lymphatic and haematopoietic tissue

The study of residents with high environmental exposure in the Mid-Ohio Valley, USA, found no indications of increased risk of incident leukaemia or non-specified lymphoma (<u>Barry</u> <u>et al., 2013</u>). These findings were fairly consistent with those from a US case-cohort study (<u>Winquist et al., 2023</u>) and with the mortality study of the fluoropolymer worker cohort in the Parkersburg polymer-production plant, which did not find indications of increased risk of fatal NHL or leukaemia in exposed workers (<u>Steenland</u> and <u>Woskie, 2012</u>). In contrast, the case-control studies in West Virginia and Ohio, USA, reported

an increased incidence of NHL among exposed groups and provided some indications of an exposure-response relation. The OR for residents with the highest estimated PFOA serum concentration, assuming 10-year residency and latency, versus unexposed residents was 1.8 (95% CI, 1.0–3.4; <u>Vieira et al., 2013</u>). The reason for this discrepancy was unknown, but the cohort study with modelled and validated lifelong cumulative exposure assessment was considered to be the most informative. Other occupational mortality studies added little to the evidence because of few exposed cases, resulting in very imprecise risk estimates, reporting risk for other subgroups of cancers of lymphatic and haematopoietic tissue, crude exposure assessment, and other methodological issues (Gilliland and Mandel, 1993; Lundin et al., 2009; Consonni et al., 2013; Girardi and Merler, 2019; for details, see Section 2.6.3).

The Working Group concluded that the studies addressing effects at high occupational exposure levels included very few exposed cases of cancers of lymphatic and haematopoietic tissue and that the strongest evidence on risk related to high environmental levels did not indicate an increased risk. Overall, the epidemiological evidence was insufficient to permit causal inference and to exclude the possibility that causal associations between PFOA exposure and cancer of lymphatic and haematopoietic tissue may exist.

(k) Other cancer types

One study of community residents with high exposure found weak indications of increased risk of brain cancer (Vieira et al., 2013), but findings were not corroborated by those of the most informative study of highly exposed Mid-Ohio Valley residents (Barry et al., 2013). Two occupational mortality studies did not find an increased risk, but there were 5 or fewer deaths in exposed people (Lundin et al., 2009; Consonni et al., 2013). No increased risk for melanoma associated with PFOA exposure was reported in the study of residents with high environmental exposure in the Mid-Ohio Valley, USA (Barry et al., 2013; Vieira et al., 2013). Two occupational cohort mortality studies included very few exposed cases of melanoma, resulting in very imprecise risk estimates (Leonard et al., 2008; Steenland et al., 2015). A positive association between PFOA exposure and mesothelioma was reported in one study of a polymer-production plant in Parkersburg, West Virginia, USA (Steenland and Woskie, 2012), but this finding was not replicated in other studies and was likely to be caused by exposure to asbestos at the plant. Some indication of an increased risk of lung cancer in relation to exposure to PFOA was reported in one study (Vieira et al., 2013), but five other studies, including some that were highly informative, did not find an increased risk (Alexander et al., 2003; Steenland and Woskie, 2012; Barry et al., 2013; Consonni et al., 2013; Girardi and Merler, 2019). The available epidemiological evidence base for evaluation of cancer at these organ sites was sparse and generally null. Finally, a case-control study found indications of an increased risk of retinoblastoma (adjusted OR per IQR increase in blood PFOA, 1.03; 95% CI, 0.97–1.09) in a population with background exposure levels (Chen et al., 2024).

(I) All sites combined

Three occupational cohort studies did not find indications of increased mortality from all types of cancers combined in analyses based upon internal comparisons or comparisons with non-exposed workers, which reduce the likelihood that risk estimates were attenuated because of primary healthy-worker selection or healthy-worker survivor bias (<u>Steenland and</u> <u>Woskie, 2012; Consonni et al., 2013; Raleigh et al., 2014</u>). These results are in line with those of the NHANES 1999–2014 cohort addressing general population background exposure, in which cancer mortality was not associated with PFOA serum levels (<u>Wen et al., 2022</u>). In contrast to these five studies that all had high statistical power, one small study in Italy of perfluorocarbon-production workers who had the highest serum levels of PFOA ever published provided some, but inconsistent, indications of increased mortality from all cancers combined (<u>Girardi</u> and <u>Merler, 2019</u>).

The Working Group noted that null associations observed in studies of overall risk of cancer are of minimal value when it comes to causal inference of cancer etiology, because specific compounds such as PFOA cannot be expected to contribute to the occurrence of all cancer types, and therefore associations with any specific cancer type may be masked by null associations with other cancer types.

2.8.6 Specific cancer sites and exposure to PFOS

(a) Kidney cancer

The Swedish Ronneby Register cohort study of residents with high environmental exposure found some evidence for an increased risk of incident kidney cancer; when comparing with unexposed residents, the hazard ratio for residents with longer exposure was 1.47 (95% CI, 0.87-2.49) and for residents with more recent exposure was 1.85 (95% CI, 1.00-3.40) (Li et al., 2022a). However, the study did not distinguish the effects of PFOS (the predominant PFAS in drinking-water in this population) from those of other PFAS compounds that were also present at exposure levels above background (Li et al., 2022a). Of the two large nested case-control studies of general populations with substantial lower background exposure and with prediagnostic blood samples, one study observed an association in analyses not adjusted for other PFAS; however, neither study observed associations after adjustment for other PFAS (Shearer et al., 2021; Rhee et al., 2023b). A third large nested population-based case-control study also found no association overall, in men or women

(<u>Winquist et al., 2023</u>). The only occupational cohort study that primarily addressed PFOS exposure (<u>Alexander and Olsen, 2007</u>) did not report kidney cancer data.

The Working Group considered that the epidemiological evidence on the association between PFOS and kidney cancer was too sparse to permit evaluation.

(b) Testicular cancer

The study of US Air Force servicemen exposed to levels comparable to those in the general population was the only available study with a sufficient number of cases that addressed the risk of TGCT (the most common type of testicular cancer) related to PFOS exposure and adjusted for other PFAS (Purdue et al., 2023). In an analysis that included the entire study population (using first or only samples), the adjusted OR for the highest versus the lowest exposed quartile was 1.8 (95% CI, 0.9-3.6; P for trend, 0.15), whereas the OR for the subset of the population with repeated blood samples (second blood sample only; about one third of the participants) was 4.6 (95% CI, 1.4–15.1; *P* for trend, 0.009). These estimates were adjusted for other PFAS. The reasons for these discrepant results - if not due to chance - were unknown. The men with repeated samples had accumulated more exposed years, but PFOS measurement levels were similar between the two samples, and both were similar to background levels (e.g. as measured in NHANES). Both measurements were obtained by analysis of samples collected on average about 5 years before diagnosis. The Ronneby Register cohort study reported a hazard ratio of 1.51 (95% CI, 0.56-4.03; 45 incident cases) among residents with the longest exposure compared with residents in the same municipality who were not exposed to contaminated drinking-water (Li et al., 2022a). Although PFOS was the main contaminant of drinking-water among Ronneby residents, it was not possible to distinguish the effects of PFOS
from those of other PFAS compounds that were also present at levels above background.

The Working Group concluded that in the two available studies, an imprecise or inconsistent positive association was observed between PFOS exposure and cancer of the testis. Overall, the evidence did not permit the evaluation of a causal association between PFOS and testicular cancer because there were too few informative studies, unexplained inconsistencies between findings, or potential confounding by other PFAS compounds (i.e. PFHxS).

(c) Bladder cancer

An occupational cohort study of PFOSexposed workers at a chemical plant in Alabama, USA, found indications of an increased incidence of bladder cancer in workers with the highest cumulative PFOS exposure in internal comparisons (Alexander and Olsen, 2007), but these included only a small number of cases, and co-exposure to other PFAS was likely. Incomplete registration and ascertainment of diagnoses may have caused non-differential misclassification of the outcome and bias towards the null. The Ronneby Register cohort study reported moderately elevated risks of bladder cancer among residents with later and longer exposure to PFOS-contaminated drinking-water compared with unexposed residents (HR, 1.50; 95% CI, 0.98-2.28; and HR, 1.39; 95% CI, 0.95-2.02; respectively) (Li et al., 2022a). The group-based exposure assessment not accounting for individual variation limited the options to explore risk at the full range of exposure. The Danish Diet, Cancer, and Health Cohort study (Eriksen et al., 2009) and a US case-cohort study (Winguist et al., 2023) did not find increased risk of bladder cancer. Eriksen et al. (2009) applied a more stringent design including detailed adjustment for smoking and several occupations that have been associated with bladder cancer. Although the average exposure was lower than in Ronneby, exposure levels were overlapping. The Working

Group concluded that there were findings indicating an increased risk of bladder cancer in two studies of workers and residents with high environmental exposure, but not in two studies of populations with lower (background) levels of PFOS exposure. The Working Group concluded that there were too few studies available to permit a conclusion to be drawn about the association between PFOS and bladder cancer.

(d) Prostate cancer

Two highly informative cohort studies of background exposure levels in the general Danish and US population investigated the association between PFOS and risk of incident prostate cancer (Eriksen et al., 2009; Rhee et al., 2023a). Eriksen et al. reported moderately increased risks of prostate cancer in the three upper quartiles compared with the lowest quartile, but exposure-response analyses did not provide solid evidence for a linear trend (IRR per 10 ng/mL increase in PFOS concentration, 1.05; 95% CI, 0.97-1.14). The US population-based study reported an OR per unit increase on log, scale of 0.99 (95% CI, 0.79-1.23) (Rhee et al., 2023a). Other environmentally exposed population studies did not provide substantial additional information (Hardell et al., 2014; Li et al., 2022a; Winquist et al., 2023). In particular, there were no data available from occupational cohorts with much higher exposure levels.

The Working Group concluded that two large cohort studies of the general population found no consistent evidence for increased risk of prostate cancer in relation to PFOS exposure, but there were no available data at higher exposure levels in an occupational setting.

(e) Breast cancer

Almost all the available studies on associations between PFOS and breast cancer were based upon general population samples, which limited causal inference because of the narrow ranges of exposure. Findings in the most informative nested case-control and case-cohort studies based upon large samples, incident data, prediagnostic serum PFOS measurements, and good confounder control were partly conflicting. The study conducted in the French E3N Cohort (Mancini et al., 2020a) provided some indications of an increased risk of overall breast cancer at higher PFOS exposure levels. This association appeared to be stronger and linear when restricted to hormone receptor-positive breast cancers. Similarly, in the US PLCO cohort (Chang et al., 2023), PFOS appeared to be positively associated only with hormone receptor-positive breast cancers. These results were not confirmed in the case-cohort study conducted in the Dongfeng-Tongji cohort in China (Feng et al., 2022), or in the CPS-II LifeLink Cohort (Winquist et al., 2023); however, these two studies did not explore the association by hormone-receptor tumour subtype. The only informative available study of populations with higher environmental exposure, the Swedish Ronneby Register cohort study, did not find associations between exposure to PFAS (including PFOS) and overall breast cancer risk, but did not investigate associations with specific tumour subtypes (Li et al., 2022a). Findings in several case-control studies were less informative because of methodological issues relating to postdiagnostic or post-treatment PFOS measurements, and potential confounding (for details, see Section 2.4 and Table 2.4).

In summary, there was little evidence of an association between PFOS exposure and breast cancer overall. However, the two most informative studies (Mancini et al., 2020a; Chang et al., 2023), which were the only prospective studies to examine the association by hormone-receptor tumour subtype, found an imprecise but increased risk of hormone receptor-positive breast cancers associated with higher levels of PFOS. This finding was somewhat contradicted by the null findings among postmenopausal women in the Dongfeng-Tongji cohort (Feng et al., 2022) and the CPS-II cohort (Winquist

et al., 2023), which did not stratify by receptor status (most postmenopausal breast cancers are hormone receptor-positive). Given the inconsistencies across studies, the Working Group considered that the available evidence on risk of breast cancer associated with PFOS exposure was inconclusive.

(f) Thyroid cancer

The only studies available to the Working Group were the Swedish Ronneby Register cohort study of residents with high environmental exposure (Li et al., 2022a); a case-control study nested within the FMC including nulliparous women from the general population (Madrigal et al., 2024); a case-control nested within the BioMe cohort (van Gerwen et al., 2023); and two case-control studies on risk related to background exposure of the general population (Liu et al., 2022; Li et al., 2023). The Ronneby study reported an increased risk of thyroid cancer (type unspecified) in exposed women (SIR based on regional reference rates, 2.08; 95% CI, 1.19-3.38; 16 exposed cases) but not in men (3 exposed cases). The FMC study did find indications of an increased risk of papillary thyroid cancer among women diagnosed at age < 40 years; however, when adjusted for exposure to other PFAS, the association was greatly attenuated the OR for serum PFOS increment by log, in women aged < 40 years at diagnosis was 1.14 (95% CI, 0.68-1.93; 185 cases). Although PFOS was present at by far the highest concentrations in contaminated drinking-water in the Ronneby municipality, the effects of other PFAS could not be accounted for in this study (Li et al., 2022a), the study by van Gerwen et al. (2023) had very small numbers in the longitudinal subsample, and the two case-control studies were less informative because of postdiagnostic measurements of exposure, potential for bias related to selection of reference groups, or few cases (Liu et al., 2022; Li et al., 2023).

The Working Group noted that there were inconsistent indications of positive associations between PFOS exposure and thyroid cancer in women, but that, overall, studies were too few to permit an evaluation of causal associations.

(g) Liver cancer

Of the five studies addressing the association between PFOS and liver cancer, the nested case-cohort study of the Danish Diet, Cancer, and Health Cohort (Eriksen et al., 2009) was considered particularly informative because of the large well-characterized study population, high completeness of verified cases, prediagnostic serum sampling, and meticulous control for confounding. No association between levels of PFOS exposure and risk of liver cancer was identified. This study was limited by low exposure levels and rather narrow exposure contrast, but its findings were consistent with those of the Ronneby Register cohort study that addressed much higher environmental exposure (Li et al., 2022a). Also, the Ronneby study did not find an increased risk of liver cancer but included < 10 exposed cases. An occupational mortality study was not informative because it included even fewer cases (Alexander et al., 2003). In contrast, the nested case-control study in the MEC, which was distinguished by its study of non-viral HCC, found some indications of an increased, BMI-adjusted, risk for higher plasma PFOS concentrations (> 54.9 ng/mL, corresponding to the NHANES 90th percentile) compared with lower concentrations (OR, 2.90; 95% CI, 0.78-10.00), but without a clear exposureresponse relation (Goodrich et al., 2022) and a risk estimate based upon a post-hoc grouping of exposure. Finally, a Chinese hospital-based case-control study was less informative because of various methodological limitations (Cao et al., 2022; for details see Section 2.5.1).

The Working Group concluded that the most informative studies found no associations between PFOS exposure and risk of liver cancer and, overall, the available epidemiological studies were too few to permit an evaluation of causal associations.

(h) Other cancer types

One cohort study of residents with high environmental exposure (Li et al., 2022a) and two large studies of background exposure (Eriksen et al., 2009; Zhang et al., 2023) found no indications of increased risk of pancreatic cancer. An occupational mortality study of PFOSproduction workers in Decatur, Alabama, USA, found no increased risk of fatal cancer of the digestive organs and peritoneum combined, but estimates were based on only 5 deaths (Alexander et al., 2003). The Ronneby Register cohort found no significant increase in the incidence of cancer of the colon, rectum, oesophagus, or stomach in residents with high environmental exposure (Li et al., 2022a). The only available study on associations between PFOS and brain cancer was the Ronneby Register cohort study, which found weak indications of an increased risk of incident brain cancer among highly exposed residents (Li et al., 2022a), but risk estimates were imprecise and the effects of PFOS - which was present at highest concentrations in contaminated drinking-water - and other PFAS present at lower concentrations could not be distinguished. Two occupational cohort studies of fatal cancer of lymphatic and haematopoietic tissue in workers with exposure to PFOS were not informative because of the crude exposure assessments and very few cases (Alexander et al., 2003; Girardi and Merler, 2019). The Ronneby Register cohort study had greater statistical power, but lower exposure levels did not reveal any increased risk of several specific types of cancer of lymphatic and haematopoietic tissue, including NHL, multiple myeloma, and chronic lymphocytic leukaemia, whereas residents who had ever been exposed had a higher risk of chronic myeloid leukaemia but with too few cases to allow more detailed analysis (Li et al., 2022a). One occupational

cohort study addressed risk of fatal melanoma, but the few cases resulted in very imprecise estimates (Alexander et al., 2003). The Ronneby Register cohort study had higher statistical power and found some evidence for an increased risk of melanoma in the subset of residents with the latest ever-high exposure to PFAS-contaminated drinking-water (HR, 1.54; 95% CI, 1.09-2.19) but not in those with the longest ever-high exposure (HR, 1.14; 95% CI, 0.80-1.64), compared with those with no exposure. The findings were not corroborated or refuted by other studies. Finally, a case-control study did not find an increased risk of retinoblastoma (adjusted OR per IQR increase in blood PFOS, 1.02; 95% CI, 0.95-1.09) in a population with background exposure levels (Chen et al., 2024).

The Working Group noted that there were too few studies available for the evaluation of associations between PFOS exposure and melanoma, retinoblastoma, and cancers of the pancreas, colon, rectum, oesophagus, stomach, brain, or lymphatic and haematopoietic tissue.

(i) All sites combined

Overall cancer mortality was not increased among PFOS-exposed workers at the PFOSproduction facility in Decatur, Alabama, USA (Alexander et al., 2003), and this was consistent with results from the Ronneby Register cohort study in which combined cancer incidence was not elevated in a population of residents with above-background exposure to PFOS in particular, but also PFHxS and PFOA to a lesser degree (Lietal., 2022a). On the other hand, overall cancer mortality was increased among members of the NHANES 1999-2014 cohort who had the highest serum PFOS values compared with the lowest tertile (Wen et al., 2022), and a small study in Italy with low statistical power reported some inconsistent indications of increased all-cancer mortality in workers with very high exposure to PFOS as well as other PFAS (Girardi and Merler, 2019).

The Working Group noted that the results of the few available studies were conflicting and that the number of informative epidemiological studies was too few to permit an evaluation of the evidence on PFOS and PFOA exposure and risk of all cancers combined.

References

- Alexander BH, Olsen GW (2007). Bladder cancer in perfluorooctanesulfonyl fluoride manufacturing workers. *Ann Epidemiol*. 17(6):471–8. doi:<u>10.1016/j.</u> <u>annepidem.2007.01.036</u> PMID:<u>17448680</u>
- Alexander BH, Olsen GW, Burris JM, Mandel JH, Mandel JS (2003). Mortality of employees of a perfluorooctanesulphonyl fluoride manufacturing facility. *Occup Environ Med.* 60(10):722–9. doi:<u>10.1136/oem.60.10.722</u> PMID:<u>14504359</u>
- Alpha-Tocopherol, Beta Carotene Cancer Prevention Study Group (1994). The effect of vitamin E and beta carotene on the incidence of lung cancer and other cancers in male smokers. *N Engl J Med.* 330(15):1029– 35. doi:10.1056/NEJM199404143301501 PMID:8127329
- Armstrong BG (1998). Effect of measurement error on epidemiological studies of environmental and occupational exposures. *Occup Environ Med.* 55(10):651–6. doi:<u>10.1136/oem.55.10.651</u> PMID:<u>9930084</u>
- ATSDR (2020). PFAS. Stories from the field. Chamblee (GA), USA: Agency for Toxic Substances and Disease Registry. Available from: <u>https://www.atsdr.cdc.gov/</u>2019ATSDRAnnualReport/stories/pfas.html, accessed August 2024.
- Bar-Mashiah A, Soper ER, Cullina S, Belbin GM, Kenny EE, Lucas AL, et al. (2022). CDH1 pathogenic variants and cancer risk in an unselected patient population. *Fam Cancer.* 21(2):235–9. doi:<u>10.1007/s10689-021-00257-x</u> PMID:<u>33886068</u>
- Barry V, Darrow LA, Klein M, Winquist A, Steenland K (2014). Early life perfluorooctanoic acid (PFOA) exposure and overweight and obesity risk in adult-hood in a community with elevated exposure. *Environ Res.* 132:62–9. doi:<u>10.1016/j.envres.2014.03.025</u> PMID:<u>24742729</u>
- Barry V, Klein M, Winquist A, Darrow LA, Steenland K (2015). Disease fatality and bias in survival cohorts. *Environ Res.* 140:275-81. doi:10.1016/j. envres.2015.03.039 PMID:25880887
- Barry V, Winquist A, Steenland K (2013). Perfluorooctanoic acid (PFOA) exposures and incident cancers among adults living near a chemical plant. *Environ Health Perspect.* 121(11–12):1313–8. doi:<u>10.1289/ehp.1306615</u> PMID:<u>24007715</u>

- Bartell SM, Vieira VM (2021). Critical review on PFOA, kidney cancer, and testicular cancer. J Air Waste Manag Assoc. 71(6):663–79. doi:<u>10.1080/10962247.202</u> <u>1.1909668</u> PMID:<u>33780327</u>
- Bernstein L, Allen M, Anton-Culver H, Deapen D, Horn-Ross PL, Peel D, et al. (2002). High breast cancer incidence rates among California teachers: results from the California Teachers Study (United States). Cancer Causes Control. 13(7):625–35. doi:10.1023/A:1019552 126105 PMID:12296510
- Blake BE, Pinney SM, Hines EP, Fenton SE, Ferguson KK (2018). Associations between longitudinal serum perfluoroalkyl substance (PFAS) levels and measures of thyroid hormone, kidney function, and body mass index in the Fernald Community Cohort. *Environ Pollut.* 242(Pt A):894–904. doi:10.1016/j. envpol.2018.07.042 PMID:30373035
- Bonefeld-Jørgensen EC, Long M, Bossi R, Ayotte P, Asmund G, Krüger T, et al. (2011). Perfluorinated compounds are related to breast cancer risk in Greenlandic Inuit: a case control study. *Environ Health*. 10(1):88. doi:10.1186/1476-069X-10-88 PMID:21978366
- Bonefeld-Jørgensen EC, Long M, Fredslund SO, Bossi R, Olsen J (2014). Breast cancer risk after exposure to perfluorinated compounds in Danish women: a case-control study nested in the Danish National Birth Cohort. *Cancer Causes Control*. 25(11):1439–48. doi:10.1007/s10552-014-0446-7 PMID:25148915
- Buekers J, Colles A, Cornelis C, Morrens B, Govarts E, Schoeters G (2018). Socio-economic status and health: evaluation of human biomonitored chemical exposure to per- and polyfluorinated substances across status. *Int J Environ Res Public Health*. 15(12):2818. doi:10.3390/ ijerph15122818 PMID:30544905
- Calafat AM, Wong LY, Kuklenyik Z, Reidy JA, Needham LL (2007). Polyfluoroalkyl chemicals in the US population: data from the National Health and Nutrition Examination Survey (NHANES) 2003–2004 and comparisons with NHANES 1999–2000. *Environ Health Perspect*. 115(11):1596–602. doi:10.1289/ehp.10598 PMID:18007991
- Cao L, Guo Y, Chen Y, Hong J, Wu J, Hangbiao J (2022). Per-/polyfluoroalkyl substance concentrations in human serum and their associations with liver cancer. *Chemosphere*. 296:134083. doi:<u>10.1016/j.</u> <u>chemosphere.2022.134083</u> PMID:<u>35216980</u>
- Chang CJ, Ish JL, Chang VC, Daniel M, Jones RR, White AJ (2024). Exposure to per- and polyfluoroalkyl substances and breast cancer risk: a systematic review and meta-analysis of epidemiologic studies. *Am J Epidemiol.* 193(8):1182–96. doi:<u>10.1093/aje/kwae010</u> PMID:<u>38400646</u>
- Chang VC, Rhee J, Berndt SI, Moore SC, Freedman ND, Jones RR, et al. (2023). Serum perfluorooctane sulfonate and perfluorooctanoate and risk of postmenopausal breast cancer according to hormone receptor status: an

analysis in the Prostate, Lung, Colorectal and Ovarian Cancer Screening Trial. *Int J Cancer*. 153(4):775–82. doi:<u>10.1002/ijc.34487</u> PMID:<u>36843273</u>

- Chapman AB, Abraham WT, Zamudio S, Coffin C, Merouani A, Young D, et al. (1998). Temporal relationships between hormonal and hemodynamic changes in early human pregnancy. *Kidney Int*. 54(6):2056–63. doi:10.1046/j.1523-1755.1998.00217.x PMID:9853271
- Chen Y, Paul KC, Walker DI, Jones DP, Wang X, Ritz BR, et al. (2024). Neonatal per- and polyfluoroalkyl substance exposure in relation to retinoblastoma. *Environ Res.* 240(Pt 2):117435. doi:10.1016/j.envres.2023.117435 PMID:37866539
- Clavel-Chapelon F; E3N Study Group (2015). Cohort profile: the French E3N cohort study. *Int J Epidemiol*. 44(3):801–9. doi:<u>10.1093/ije/dyu184</u> PMID:<u>25212479</u>
- Cohn BA, La Merrill MA, Krigbaum NY, Wang M, Park JS, Petreas M, et al. (2020). In utero exposure to polyand perfluoroalkyl substances (PFASs) and subsequent breast cancer. *Reprod Toxicol.* 92:112–9. doi:<u>10.1016/j.</u> <u>reprotox.2019.06.012</u> PMID:<u>31323350</u>
- Cong X, Liu Q, Li W, Wang L, Feng Y, Liu C, et al. (2023). Systematic review and meta-analysis of breast cancer risks in relation to 2,3,7,8-tetrachlorodibenzo-*p*-dioxin and per- and polyfluoroalkyl substances. *Environ Sci Pollut Res Int.* 30(37):86540–55. doi:10.1007/s11356-023-28592-9 PMID:37420152
- Consonni D, Straif K, Symons JM, Tomenson JA, van Amelsvoort LG, Sleeuwenhoek A, et al. (2013). Cancer risk among tetrafluoroethylene synthesis and polymerization workers. *Am J Epidemiol*. 178(3):350–8. doi:10.1093/aje/kws588 PMID:23828249
- Côté S, Ayotte P, Dodin S, Blanchet C, Mulvad G, Petersen HS, et al. (2006). Plasma organochlorine concentrations and bone ultrasound measurements: a cross-sectional study in peri-and postmenopausal Inuit women from Greenland. *Environ Health*. 5(1):33. doi:10.1186/1476-069X-5-33 PMID:17184534
- Coupland CAC, Chilvers CE, Davey G, Pike MC, Oliver RT, Forman D; United Kingdom Testicular Cancer Study Group (1999). Risk factors for testicular germ cell tumours by histological tumour type. *Br J Cancer.* 80(11):1859–63. doi:10.1038/sj.bjc.6690611 PMID:10468310
- Deutch B, Pedersen HS, Asmund G, Hansen JC (2007). Contaminants, diet, plasma fatty acids and smoking in Greenland 1999-2005. *Sci Total Environ*. 372(2–3):486– 96. doi:<u>10.1016/j.scitotenv.2006.10.043</u> PMID:<u>17157358</u>
- Dhingra R, Lally C, Darrow LA, Klein M, Winquist A, Steenland K (2016). Perfluorooctanoic acid and chronic kidney disease: longitudinal analysis of a Mid-Ohio Valley community. *Environ Res.* 145:85–92. doi:10.1016/j.envres.2015.11.018 PMID:26656498
- Dhingra R, Winquist A, Darrow LA, Klein M, Steenland K (2017). A study of reverse causation: examining the associations of perfluorooctanoic acid serum levels with

two outcomes. *Environ Health Perspect*. 125(3):416–21. doi:<u>10.1289/EHP273</u> PMID:<u>27529882</u>

- Eek F, Ostergren PO, Diderichsen F, Rasmussen NK, Andersen I, Moussa K, et al. (2010). Differences in socioeconomic and gender inequalities in tobacco smoking in Denmark and Sweden; a cross sectional comparison of the equity effect of different public health policies. *BMC Public Health*. 10(1):9. doi:<u>10.1186/1471-2458-10-9</u> PMID:<u>20064219</u>
- Eriksen KT, Sørensen M, McLaughlin JK, Lipworth L, Tjønneland A, Overvad K, et al. (2009). Perfluorooctanoate and perfluorooctanesulfonate plasma levels and risk of cancer in the general Danish population. *J Natl Cancer Inst.* 101(8):605–9. doi:<u>10.1093/jnci/djp041</u> PMID:<u>19351918</u>
- Eriksen KT, Sørensen M, McLaughlin JK, Tjønneland A, Overvad K, Raaschou-Nielsen O (2011). Determinants of plasma PFOA and PFOS levels among 652 Danish men. *Environ Sci Technol*. 45(19):8137–43. doi:<u>10.1021/</u> <u>es100626h</u> PMID:<u>20939531</u>
- Feng Y, Bai Y, Lu Y, Chen M, Fu M, Guan X, et al. (2022). Plasma perfluoroalkyl substance exposure and incidence risk of breast cancer: a case–cohort study in the Dongfeng-Tongji cohort. *Environ Pollut*. 306:119345. doi:10.1016/j.envpol.2022.119345 PMID:35472559
- Finnish Cancer Registry (2023). Helsinki, Finland: Finnish Cancer Registry. Available from: <u>https://cancerregistry.</u> <u>fi/</u>, accessed October 2023.
- Finnish Institute for Health and Welfare (2023). Register Description. Medical Birth Register. Helsinki, Finland: Finnish Institute for Health and Welfare. Available from: <u>https://thl.fi/en/web/thlfi-en/statistics-and-data/ data-and-services/register-descriptions/newborns</u>, accessed September 2023.
- Frenoy P, Perduca V, Cano-Sancho G, Antignac JP, Severi G, Mancini FR (2022). Application of two statistical approaches (Bayesian Kernel Machine Regression and Principal Component Regression) to assess breast cancer risk in association to exposure to mixtures of brominated flame retardants and per- and polyfluorinated alkylated substances in the E3N cohort. *Environ Health.* 21(1):27. doi:10.1186/s12940-022-00840-4 PMID:35216589
- Frisbee SJ, Brooks AP Jr, Maher A, Flensborg P, Arnold S, Fletcher T, et al. (2009). The C8 Health Project: design, methods, and participants. *Environ Health Perspect*. 117(12):1873–82. doi:<u>10.1289/ehp.0800379</u> PMID:<u>20049206</u>
- Ghisari M, Eiberg H, Long M, Bonefeld-Jørgensen EC (2014). Polymorphisms in phase I and phase II genes and breast cancer risk and relations to persistent organic pollutant exposure: a case-control study in Inuit women. *Environ Health*. 13(1):19. doi:10.1186/1476-069X-13-19 PMID:24629213

- Ghisari M, Long M, Røge DM, Olsen J, Bonefeld-Jørgensen EC (2017). Polymorphism in xenobiotic and estrogen metabolizing genes, exposure to perfluorinated compounds and subsequent breast cancer risk: a nested case-control study in the Danish National Birth Cohort. *Environ Res.* 154:325-33. doi:10.1016/j. envres.2017.01.020 PMID:28157646
- Gilliland FD, Mandel JS (1993). Mortality among employees of a perfluorooctanoic acid production plant. *J Occup Med*. 35(9):950–4. doi:<u>10.1097/00043764-199309000-00020</u> PMID:<u>8229349</u>
- Girardi P, Merler E (2019). A mortality study on male subjects exposed to polyfluoroalkyl acids with high internal dose of perfluorooctanoic acid. *Environ Res.* 179(Pt A):108743. doi:<u>10.1016/j.envres.2019.108743</u> PMID:<u>31542491</u>
- Glynn A, Berger U, Bignert A, Ullah S, Aune M, Lignell S, et al. (2012). Perfluorinated alkyl acids in blood serum from primiparous women in Sweden: serial sampling during pregnancy and nursing, and temporal trends 1996–2010. *Environ Sci Technol*. 46(16):9071–9. doi:10.1021/es301168c PMID:22770559
- Goodrich JA, Walker D, Lin X, Wang H, Lim T, McConnell R, et al. (2022). Exposure to perfluoroalkyl substances and risk of hepatocellular carcinoma in a multiethnic cohort. *JHEP Rep Innov Hepatol.* 4(10):100550. doi:10.1016/j.jhepr.2022.100550 PMID:36111068
- Hakala T, Kellokumpu-Lehtinen P, Kholová I, Holli K, Huhtala H, Sand J (2012). Rising incidence of small size papillary thyroid cancers with no change in disease-specific survival in Finnish thyroid cancer patients. *Scand J Surg.* 101(4):301–6. doi:10.1177/145749691210100415 PMID:23238509
- Hardell E, Kärrman A, van Bavel B, Bao J, Carlberg M, Hardell L (2014). Case-control study on perfluorinated alkyl acids (PFAAs) and the risk of prostate cancer. *Environ Int.* 63:35–9. doi:<u>10.1016/j.envint.2013.10.005</u> PMID:<u>24246240</u>
- Hayes RB, Reding D, Kopp W, Subar AF, Bhat N, Rothman N, et al.; Prostate, Lung, Colorectal and Ovarian Cancer Screening Trial Project Team (2000). Etiologic and early marker studies in the Prostate, Lung, Colorectal and Ovarian (PLCO) cancer screening trial. *Control Clin Trials.* 21(6 Suppl):349S–55S. doi:10.1016/S0197-2456(00)00101-X PMID:11189687
- Holl K, Lundin E, Kaasila M, Grankvist K, Afanasyeva Y, Hallmans G, et al. (2008). Effect of long-term storage on hormone measurements in samples from pregnant women: the experience of the Finnish Maternity Cohort. *Acta Oncol.* 47(3):406–12. doi:10.1080/02841860701592400 PMID:17891670
- Hurley S, Goldberg D, Wang M, Park JS, Petreas M, Bernstein L, et al. (2018). Breast cancer risk and serum levels of per- and poly-fluoroalkyl substances: a casecontrol study nested in the California Teachers Study.

Environ Health. 17(1):83. doi:<u>10.1186/s12940-018-0426-6</u> PMID:<u>30482205</u>

- IARC (2012). Arsenic, metals, fibres, and dusts. IARC Monogr Eval Carcinog Risks Hum. 100C:1–501. Available from: <u>https://publications.iarc.who.int/120</u> PMID:23189751
- IARC (2016). Some chemicals used as solvents and in polymer manufacture. IARC Monogr Eval Carcinog Risks Hum. 110:1–276. Available from: <u>https:// publications.iarc.who.int/547</u> PMID:31829531
- IARC (2023). Occupational exposure as a firefighter. *IARC* Monogr Identif Carcinog Hazard Hum. 132:1–728.
 Available from: <u>https://publications.iarc.who.int/615</u>. PMID:<u>37963216</u>
- Icahn School of Medicine at Mount Sinai (2023). BioMe* BioBank Program. New York (NY), USA: Icahn School of Medicine at Mount Sinai, Charles Bronfman Institute for Personalized Medicine. Available from: <u>https://icahn.mssm.edu/research/ipm/programs/</u> <u>biome-biobank</u>, accessed October 2023.
- Innes KE, Wimsatt JH, Frisbee S, Ducatman AM (2014). Inverse association of colorectal cancer prevalence to serum levels of perfluorooctane sulfonate (PFOS) and perfluorooctanoate (PFOA) in a large Appalachian population. *BMC Cancer.* 14(1):45. doi:10.1186/1471-2407-14-45 PMID:24468211
- Itoh H, Harada KH, Kasuga Y, Yokoyama S, Onuma H, Nishimura H, et al. (2021). Serum perfluoroalkyl substances and breast cancer risk in Japanese women: a case-control study. *Sci Total Environ.* 800:149316. doi:10.1016/j.scitotenv.2021.149316 PMID:34392213
- Jiang H, Liu H, Liu G, Yu J, Liu N, Jin Y, et al. (2022). Associations between polyfluoroalkyl substances exposure and breast cancer: a meta-analysis. *Toxics*. 10(6):318. doi:10.3390/toxics10060318 PMID:35736926
- Kato K, Wong LY, Chen A, Dunbar C, Webster GM, Lanphear BP, et al. (2014). Changes in serum concentrations of maternal poly- and perfluoroalkyl substances over the course of pregnancy and predictors of exposure in a multiethnic cohort of Cincinnati, Ohio pregnant women during 2003–2006. *Environ Sci Technol*. 48(16):9600–8. doi:10.1021/es501811k PMID:25026485
- Kolonel LN, Henderson BE, Hankin JH, Nomura AM, Wilkens LR, Pike MC, et al. (2000). A multiethnic cohort in Hawaii and Los Angeles: baseline characteristics. *Am J Epidemiol*. 151(4):346–57. doi:<u>10.1093/ oxfordjournals.aje.a010213</u> PMID:<u>10695593</u>
- Koponen J, Kiviranta H (2019). Quantitation of perfluoroalkyl acids, parabens and cotinine from single low volume serum sample by validated analytical method. *Int J Environ Anal Chem.* 99(13):1268–85. doi:10.1080/ 03067319.2019.1618850
- Koponen J, Rantakokko P, Airaksinen R, Kiviranta H (2013). Determination of selected perfluorinated alkyl acids and persistent organic pollutants from a small volume human serum sample relevant for

epidemiological studies. *J Chromatogr A*. 1309:48–55. doi:<u>10.1016/j.chroma.2013.07.064</u> PMID:<u>23972455</u>

- Lanphear BP (2017). Low-level toxicity of chemicals: no acceptable levels? *PLoS Biol.* 15(12):e2003066. doi:10.1371/journal.pbio.2003066 PMID:29257830
- Lehtinen M, Surcel HM, Natunen K, Pukkala E, Dillner J (2017). Cancer Registry follow-up for 17 million personyears of a nationwide maternity cohort. *Cancer Med*. 6(12):3060–4. doi:10.1002/cam4.1222 PMID:29071810
- Leonard RC, Kreckmann KH, Sakr CJ, Symons JM (2008). Retrospective cohort mortality study of workers in a polymer production plant including a reference population of regional workers. *Ann Epidemiol*. 18(1):15–22. doi:10.1016/j.annepidem.2007.06.011 PMID:17900928
- Li H, Hammarstrand S, Midberg B, Xu Y, Li Y, Olsson DS, et al. (2022a). Cancer incidence in a Swedish cohort with high exposure to perfluoroalkyl substances in drinking water. *Environ Res.* 204(Pt C):112217. doi:10.1016/j. envres.2021.112217 PMID:34662573
- Li H, Yang M, Yang J, Seery S, Ma C, Liu Y, et al. (2023). Per- and polyfluoroalkyl substances and the associated thyroid cancer risk: a case-control study in China. *Chemosphere*. 337:139411. doi:10.1016/j. <u>chemosphere.2023.139411</u> PMID:37419160
- Li X, Song F, Liu X, Shan A, Huang Y, Yang Z, et al. (2022b). Perfluoroalkyl substances (PFASs) as risk factors for breast cancer: a case-control study in Chinese population. *Environ Health*. 21(1):83. doi:<u>10.1186/s12940-022-00895-3</u> PMID:<u>36085159</u>
- Li Y, Fletcher T, Mucs D, Scott K, Lindh CH, Tallving P, et al. (2018). Half-lives of PFOS, PFHxS and PFOA after end of exposure to contaminated drinking water. *Occup Environ Med.* 75(1):46–51. doi:<u>10.1136/oemed-2017-104651</u> PMID:<u>29133598</u>
- Lin HW, Feng HX, Chen L, Yuan XJ, Tan Z (2020). Maternal exposure to environmental endocrine disruptors during pregnancy is associated with pediatric germ cell tumors. *Nagoya J Med Sci.* 82(2):323–33. PMID:<u>32581411</u>
- Liu M, Zhang G, Meng L, Han X, Li Y, Shi Y, et al. (2022). Associations between novel and legacy per- and polyfluoroalkyl substances in human serum and thyroid cancer: a case and healthy population in Shandong province, East China. *Environ Sci Technol*. 56(10):6144– 51. doi:10.1021/acs.est.1c02850 PMID:34618433
- Lundin JI, Alexander BH, Olsen GW, Church TR (2009). Ammonium perfluorooctanoate production and occupational mortality. *Epidemiology*. 20(6):921–8. doi:10.1097/EDE.0b013e3181b5f395 PMID:19797969
- Madrigal JM, Troisi R, Surcel HM, Öhman H, Kivelä J, Kiviranta H, et al. (2024). Prediagnostic serum concentrations of per- and polyfluoroalkyl substances and risk of papillary thyroid cancer in the Finnish Maternity Cohort. *Int J Cancer*. 154(6):979–91. doi:10.1002/ ijc.34776 PMID:37902275

- Mancini FR, Cano-Sancho G, Gambaretti J, Marchand P, Boutron-Ruault MC, Severi G, et al. (2020a). Perfluorinated alkylated substances serum concentration and breast cancer risk: Evidence from a nested case-control study in the French E3N cohort. *Int J Cancer.* 146(4):917–28. doi:10.1002/ijc.32357 PMID:31008526
- Mancini M, Righetto M, Baggio G (2020b). Genderrelated approach to kidney cancer management: moving forward. *Int J Mol Sci.* 21(9):3378. doi:<u>10.3390/</u> <u>ijms21093378</u> PMID:<u>32397685</u>
- Mastrantonio M, Bai E, Uccelli R, Cordiano V, Screpanti A, Crosignani P (2018). Drinking water contamination from perfluoroalkyl substances (PFAS): an ecological mortality study in the Veneto Region, Italy. *Eur J Public Health*. 28(1):180–5. doi:<u>10.1093/eurpub/ckx066</u> PMID:<u>28541558</u>
- Momenimovahed Z, Salehiniya H (2019). Epidemiological characteristics of and risk factors for breast cancer in the world. *Breast Cancer (Dove Med Press)*. 11:151–64. doi:<u>10.2147/BCTT.S176070</u> PMID:<u>31040712</u>
- NCI (2023). Cancer Statistics Explorer Network. Bethesda (MD), USA: Surveillance, Epidemiology, and End Results Program, National Cancer Institute. Available from: <u>https://seer.cancer.gov/statistics-network/</u>, accessed August 2024.
- Nielsen C, Andersson Hall U, Lindh C, Ekström U, Xu Y, Li Y, et al. (2020). Pregnancy-induced changes in serum concentrations of perfluoroalkyl substances and the influence of kidney function. *Environ Health*. 19(1):80. doi:10.1186/s12940-020-00626-6 PMID:32641055
- Olsen GW, Burlew MM, Marshall JC, Burris JM, Mandel JH (2004). Analysis of episodes of care in a perfluorooctanesulfonyl fluoride production facility. *J Occup Environ Med.* 46(8):837-46. doi:10.1097/01. jom.0000135546.70469.87 PMID:15300136
- Olsen J, Melbye M, Olsen SF, Sørensen TI, Aaby P, Andersen AM, et al. (2001). The Danish National Birth Cohort—its background, structure and aim. *Scand J Public Health*. 29(4):300–7. doi:<u>10.1177/1403494801029</u> <u>0040201</u> PMID:<u>11775787</u>
- Omoike OE, Pack RP, Mamudu HM, Liu Y, Wang L (2021). A cross-sectional study of the association between perfluorinated chemical exposure and cancers related to deregulation of estrogen receptors. *Environ Res.* 196:110329. doi:10.1016/j.envres.2020.110329 PMID:33068574
- Pan Y, Zhu Y, Zheng T, Cui Q, Buka SL, Zhang B, et al. (2017). Novel chlorinated polyfluorinated ether sulfonates and legacy per-/polyfluoroalkyl substances: placental transfer and relationship with serum albumin and glomerular filtration rate. *Environ Sci Technol.* 51(1):634–44. doi:10.1021/acs.est.6b04590 PMID:27931097

- Pitter G, Da Re F, Canova C, Barbieri G, Zare Jeddi M, Daprà F, et al. (2020). Serum levels of perfluoroalkyl substances (PFAS) in adolescents and young adults exposed to contaminated drinking water in the Veneto region, Italy: a cross-sectional study based on a health surveillance program. *Environ Health Perspect*. 128(2):27007. doi:10.1289/EHP5337 PMID:32068468
- Prorok PC, Andriole GL, Bresalier RS, Buys SS, Chia D, Crawford ED, et al. (2000). Design of the prostate, lung, colorectal and ovarian (PLCO) cancer screening trial. *Control Clin Trials*. 21(6 Suppl):273S–309S. doi:<u>10.1016/</u> <u>S0197-2456(00)00098-2</u> PMID:<u>11189684</u>
- Pukkala E, Andersen A, Berglund G, Gislefoss R, Gudnason V, Hallmans G, et al. (2007). Nordic biological specimen banks as basis for studies of cancer causes and control-more than 2 million sample donors, 25 million person years and 100,000 prospective cancers. *Acta Oncol.* 46(3):286–307. doi:10.1080/02841860701203545 PMID:17450464
- Purdue MP, Rhee J, Denic-Roberts H, McGlynn KA, Byrne C, Sampson J, et al. (2023). A nested case-control study of serum per- and polyfluoroalkyl substances and testicular germ cell tumors among US Air Force servicemen. *Environ Health Perspect*. 131(7):77007. doi:10.1289/EHP12603 PMID:37458713
- Rajpert-De Meyts E (2006). Developmental model for the pathogenesis of testicular carcinoma in situ: genetic and environmental aspects. *Hum Reprod Update*. 12(3):303–23. doi:<u>10.1093/humupd/dmk006</u> PMID:<u>16540528</u>
- Raleigh KK, Alexander BH, Olsen GW, Ramachandran G, Morey SZ, Church TR, et al. (2014). Mortality and cancer incidence in ammonium perfluorooctanoate production workers. *Occup Environ Med.* 71(7):500–6. doi:10.1136/oemed-2014-102109 PMID:24832944
- Rhee J, Barry KH, Huang WY, Sampson JN, Hofmann JN, Silverman DT, et al. (2023a). A prospective nested case-control study of serum concentrations of perand polyfluoroalkyl substances and aggressive prostate cancer risk. *Environ Res.* 228:115718. doi:<u>10.1016/j.</u> envres.2023.115718 PMID:<u>36958379</u>
- Rhee J, Chang VC, Cheng I, Calafat AM, Botelho JC, Shearer JJ, et al. (2023b). Serum concentrations of perand polyfluoroalkyl substances and risk of renal cell carcinoma in the Multiethnic Cohort Study. *Environ Int.* 180:108197. doi:10.1016/j.envint.2023.108197 PMID:37741007
- Rhee J, Loftfield E, Albanes D, Layne TM, Stolzenberg-Solomon R, Liao LM, et al. (2023c). A metabolomic investigation of serum perfluorooctane sulfonate and perfluorooctanoate. *Environ Int.* 180:108198. doi:10.1016/j.envint.2023.108198 PMID:37716341

- Sagiv SK, Rifas-Shiman SL, Fleisch AF, Webster TF, Calafat AM, Ye X, et al. (2018). Early-pregnancy plasma concentrations of perfluoroalkyl substances and birth outcomes in Project Viva: confounded by pregnancy hemodynamics? *Am J Epidemiol*. 187(4):793–802. doi:10.1093/aje/kwx332 PMID:29155920
- Saugo M, Ioverno E, Olivieri A, Bertola F, Pasinato A, Ducatman A (2024). PFOA and testis cancer in the Veneto Region (Italy). *Environ Health*. 23(1):33. doi:10.1186/s12940-024-01064-4 PMID:38549149
- Seyyedsalehi MS, Boffetta P (2023). Per- and poly-fluoroalkyl substances (PFAS) exposure and risk of kidney, liver, and testicular cancers: a systematic review and meta-analysis. *Med Lav.* 114(5):e2023040. PMID:<u>37878255</u>
- Shankar A, Xiao J, Ducatman A (2011). Perfluoroalkyl chemicals and chronic kidney disease in US adults. *Am J Epidemiol.* 174(8):893–900. doi:<u>10.1093/aje/kwr171</u> PMID:<u>21873601</u>
- Shearer JJ, Callahan CL, Calafat AM, Huang WY, Jones RR, Sabbisetti VS, et al. (2021). Serum concentrations of per- and polyfluoroalkyl substances and risk of renal cell carcinoma. *J Natl Cancer Inst.* 113(5):580–7. doi:10.1093/jnci/djaa143 PMID:32944748
- Shin HM, Vieira VM, Ryan PB, Detwiler R, Sanders B, Steenland K, et al. (2011a). Environmental fate and transport modeling for perfluorooctanoic acid emitted from the Washington Works Facility in West Virginia. *Environ Sci Technol.* 45(4):1435–42. doi:10.1021/ es102769t PMID:21226527
- Shin HM, Vieira VM, Ryan PB, Steenland K, Bartell SM (2011b). Retrospective exposure estimation and predicted versus observed serum perfluorooctanoic acid concentrations for participants in the C8 Health Project. *Environ Health Perspect*. 119(12):1760–5. doi:10.1289/ehp.1103729 PMID:21813367
- Sistema Epidemiologico Regionale (2016). Ricognizione epidemiologica iniziale sulle orchiectomie per tumore del testicolo rilevate nell'area interessata dalla contaminazione idropotabile da PFAS. Venice, Italy: Sistema Epidemiologico Regionale. Available from: <u>https://sian.aulss9.veneto.it/index.cfm?method=mys.</u> <u>apridoc&iddoc=809</u>, accessed August 2024. [Italian] [Summarized in English by Saugo et al. (2024)]
- Sleeuwenhoek A, Cherrie JW (2012). Exposure assessment of tetrafluoroethylene and ammonium perfluorooctanoate 1951-2002. *J Environ Monit*. 14(3):775–81. doi:<u>10.1039/c2em10930a</u> PMID:<u>22322341</u>
- Stang A (2009). Etiology of testicular germ cell tumors: lumping or splitting? A plea against lumping. Eur J Epidemiol. 24(2):65–7. doi:<u>10.1007/s10654-008-9311-x</u> PMID:<u>19109758</u>
- Stayner L, Steenland K, Dosemeci M, Hertz-Picciotto I (2003). Attenuation of exposure-response curves in occupational cohort studies at high exposure levels.

Scand J Work Environ Health. 29(4):317–24. doi:<u>10.5271/</u> <u>sjweh.737</u> PMID:<u>12934726</u>

- Steenland K, Hofmann JN, Silverman DT, Bartell SM (2022). Risk assessment for PFOA and kidney cancer based on a pooled analysis of two studies. *Environ Int.* 167:107425. doi:10.1016/j.envint.2022.107425 PMID:35905598
- Steenland K, Jin C, MacNeil J, Lally C, Ducatman A, Vieira V, et al. (2009). Predictors of PFOA levels in a community surrounding a chemical plant. *Environ Health Perspect*. 117(7):1083–8. doi:10.1289/ ehp.0800294 PMID:19654917
- Steenland K, Woskie S (2012). Cohort mortality study of workers exposed to perfluorooctanoic acid. Am J Epidemiol. 176(10):909–17. doi:10.1093/aje/kws171 PMID:23079607
- Steenland K, Zhao L, Winquist A (2015). A cohort incidence study of workers exposed to perfluorooctanoic acid (PFOA). Occup Environ Med. 72(5):373–80. doi:10.1136/oemed-2014-102364 PMID:25601914
- Terry MB, Michels KB, Brody JG, Byrne C, Chen S, Jerry DJ, et al.; Breast Cancer and the Environment Research Program (BCERP) (2019). Environmental exposures during windows of susceptibility for breast cancer: a framework for prevention research. *Breast Cancer Res.* 21(1):96. doi:10.1186/s13058-019-1168-2 PMID:31429809
- The ATBC Cancer Prevention Study Group (1994). The Alpha-Tocopherol, Beta-Carotene Lung Cancer Prevention Study: design, methods, participant characteristics, and compliance. *Ann Epidemiol*. 4(1):1–10. doi:10.1016/1047-2797(94)90036-1 PMID:8205268
- Tjønneland A, Olsen A, Boll K, Stripp C, Christensen J, Engholm G, et al. (2007). Study design, exposure variables, and socioeconomic determinants of participation in Diet, Cancer and Health: a population-based prospective cohort study of 57 053 men and women in Denmark. *Scand J Public Health*. 35(4):432–41. doi:10.1080/14034940601047986 PMID:17786808
- Tsai MS, Chang SH, Kuo WH, Kuo CH, Li SY, Wang MY, et al. (2020). A case-control study of perfluoroalkyl substances and the risk of breast cancer in Taiwanese women. *Environ Int.* 142:105850. doi:10.1016/j. envint.2020.105850 PMID:32580117
- United States Census Bureau (2022). Population estimates. QuickFacts [online database]. Hillcrest Heights (MD), USA: United States Census Bureau. Available from: <u>https://www.census.gov/quickfacts/fact/table/</u> <u>US/PST045222</u>, accessed November 2023.
- van den Berg BJ, Christianson RE, Oechsli FW (1988). The California Child Health and Development Studies of the School of Public Health, University of California at Berkeley. *Paediatr Perinat Epidemiol.* 2(3):265–82. doi:10.1111/j.1365-3016.1988.tb00218.x PMID:3070486

- van Gerwen M, Colicino E, Guan H, Dolios G, Nadkarni GN, Vermeulen RCH, et al. (2023). Per- and polyfluoroalkyl substances (PFAS) exposure and thyroid cancer risk. *EBioMedicine*. 97:104831. doi:10.1016/j. ebiom.2023.104831 PMID:37884429
- Velarde MC, Chan AFO, Sajo MEJV, Zakharevich I, Melamed J, Uy GLB, et al. (2022). Elevated levels of perfluoroalkyl substances in breast cancer patients within the Greater Manila area. *Chemosphere*. 286(Pt 1):131545. doi:10.1016/j.chemosphere.2021.131545 PMID:34293563
- Vieira VM, Hoffman K, Shin HM, Weinberg JM, Webster TF, Fletcher T (2013). Perfluorooctanoic acid exposure and cancer outcomes in a contaminated community: a geographic analysis. *Environ Health Perspect.* 121(3):318–23. doi:10.1289/ehp.1205829 PMID:23308854
- Vierke L, Staude C, Biegel-Engler A, Drost W, Schulte C (2012). Perfluorooctanoic acid (PFOA) – main concerns and regulatory developments in Europe from an environmental point of view. *Environ Sci Eur.* 24(1):16. doi:10.1186/2190-4715-24-16
- Wang F, Zhu J, Yao P, Li X, He M, Liu Y, et al. (2013). Cohort profile: the Dongfeng–Tongji cohort study of retired workers. *Int J Epidemiol.* 42(3):731–40. doi:<u>10.1093/ije/ dys053</u> PMID:<u>22531126</u>
- Wen X, Wang M, Xu X, Li T (2022). Exposure to perand polyfluoroalkyl substances and mortality in US adults: a population-based cohort study. *Environ Health Perspect*. 130(6):67007. doi:10.1289/EHP10393 PMID:35731224
- Wielsøe M, Kern P, Bonefeld-Jørgensen EC (2017). Serum levels of environmental pollutants is a risk factor for breast cancer in Inuit: a case-control study. *Environ Health*. 16(1):56. doi:<u>10.1186/s12940-017-0269-6</u> PMID:<u>28610584</u>
- Winquist A, Hodge JM, Diver WR, Rodriguez JL, Troeschel AN, Daniel J, et al. (2023). Case-cohort study of the association between PFAS and selected cancers among participants in the American Cancer Society's Cancer Prevention Study II LifeLink cohort. *Environ Health Perspect*. 131(12):127007. doi:10.1289/EHP13174 PMID:38088576

- Winquist A, Lally C, Shin HM, Steenland K (2013). Design, methods, and population for a study of PFOA health effects among highly exposed mid-Ohio valley community residents and workers. *Environ Health Perspect.* 121(8):893–9. doi:10.1289/ehp.1206450 PMID:23735518
- Woskie SR, Gore R, Steenland K (2012). Retrospective exposure assessment of perfluorooctanoic acid serum concentrations at a fluoropolymer manufacturing plant. *Ann Occup Hyg.* 56(9):1025–37. doi:10.1093/annhyg/mes023 PMID:22539556
- Wu S, Liang D, Shi J, Li D, Liu Y, Hao Y, et al. (2023). Evaluation of a population-based breast cancer screening in north China. J Cancer Res Clin Oncol. 149(12):10119–30. doi:<u>10.1007/s00432-023-04905-w</u> PMID:<u>37266660</u>
- Xu Y, Nielsen C, Li Y, Hammarstrand S, Andersson EM, Li H, et al. (2021). Serum perfluoroalkyl substances in residents following long-term drinking water contamination from firefighting foam in Ronneby, Sweden. *Environ Int.* 147:106333. doi:10.1016/j. envint.2020.106333 PMID:33360412
- Yager JD, Davidson NE (2006). Estrogen carcinogenesis in breast cancer. N Engl J Med. 354(3):270–82. doi:<u>10.1056/</u> <u>NEJMra050776</u> PMID:<u>16421368</u>
- Zhang T, Fu S, Yu K, Albanes D, Moore SC, Purdue MP, et al. (2023). Nested case-control studies investigating serum perfluorooctanoate and perfluorooctane sulfonate levels and pancreatic ductal adenocarcinoma in two cohorts. *Environ Health Perspect*. 131(10):107702. doi:10.1289/EHP13208 PMID:37844029

3. CANCER IN EXPERIMENTAL ANIMALS

3.1 Perfluorooctanoic acid (PFOA)

Perfluorooctanoic acid (PFOA) was previously evaluated by the *IARC Monographs* programme in 2014 and the evaluation was published in Volume 110 (<u>IARC, 2016</u>). In its evaluation at that time, the Working Group concluded that there was *limited evidence* in experimental animals for the carcinogenicity of PFOA. Since the previous evaluation of PFOA by the *IARC Monographs* Programme, there have been new studies investigating the occurrence of cancer in experimental animals in relation to exposure to PFOA.

3.1.1 Mouse

See <u>Table 3.1</u>.

(a) Oral administration (drinking-water)

A cancer promotion study using the KC mouse model was conducted by <u>Kamendulis</u> et al. (2022). A mouse model was developed by selectively introducing a *Kras^{G12D}* mutation in pancreatic ductal cells using a Cre-lox technology, i.e. *PDX-1-Cre;LSL-Kras^{G12D}* transgenic mouse model (KC model) (<u>Hingorani et al., 2003</u>). This KC mouse model spontaneously develops pancreatic intraepithelial neoplasia (PanIN), mimicking human lesions as progression through four stages. At 9 months, 80% of mutant mice have PanIN lesions (considered to be a cancer precursor lesion), and eventually

develop invasive and metastatic adenocarcinoma (<u>Hingorani et al., 2003</u>).

Groups of male and female LSL-Kras^{G12D}; Pdx-1 Cre (KC) transgenic mice (age, 8 weeks) were treated with PFOA (specifically, the ammonium salt; purity, 96%) at a concentration of 5 ppm in drinking-water for 4 or 7 months. The numbers of mice [number of each sex not reported] were 10 and 11 at 6 months, and 9 and 9 at 9 months, for the control and PFOA groups, respectively. [The Working Group noted that the administered dose of PFOA in milligrams per kilogram body weight (bw) was not reported.] Controls received tap water. PFOA exposure did not significantly alter body weight at either time point in treated mice compared with controls. No information on survival or food consumption was reported. At the end of the feeding period, the mice were killed (at age 6 or 9 months). PanIN grade, inflammation score and stroma evaluation were performed by pathologists blinded to the experimental groups on haematoxylin-and-eosin-stained slides using light microscopy to evaluate each section.

Administration of PFOA in drinking-water did not cause a significant increase in the incidence of any type of neoplasm in either males or females.

In the same study, there was a significant increase in both the PanIN lesion area (58%) and the number of PanIN lesions per mm² of pancreas

Table 3.1 Studies of carcinogenicity in mice exposed to PFOA					
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence or multiplicity	Significance	Comments	
Initiation-promotion (tested as promoter) Mouse (transgenic), <i>LSL-Kras^{G12D};Pdx-1 Cre</i> (KC) (M, F) (combined) 2 mo 6 mo <u>Kamendulis et al.</u> (2022)	Oral administration (drinking-water) PFOA (ammonium salt), 96% Tap water 0, 5 ppm, for 4 mo 10, 11 NR, NR	No significant increase in t treated animals	umour incidence in	 Principal strengths: end-points studied at two time points (age 6 and 9 mo, see below); PFOA measured in serum and pancreatic tissue at both time points; PanIN grade, inflammation score and stroma evaluation performed by pathologists blinded to the treatment. Principal limitations: intake of drinking-water was not measured, thus the PFOA dose was not known; data were combined for males and females; only one dose used; no survival data; short duration of exposure; limited number of animals per group. Other comments: the mean PanIN grade did not significantly differ between control and PFOA-treated mice at 6 mo; the composite histopathology severity score derived by incorporating PanIN lesion stage, inflammation and stromal density, was significantly increased at 6 mo; the lesion number per area was significantly increased. 	
Initiation-promotion (tested as promoter) Mouse (transgenic), <i>LSL-Kras^{G12D};Pdx-1 Cre</i> (KC) (M, F) (combined) 2 mo 9 mo <u>Kamendulis et al.</u> (2022)	Oral administration (drinking-water) PFOA (ammonium salt), 96% Tap water 0, 5 ppm for 7 mo 9, 9 NR, NR	No significant increase in t treated mice	umour incidence in	Principal strengths: end-points studied at two time points (age 6 and 9 mo); PFOA measured in serum and pancreatic tissue at both time points; PanIN grade, inflammation score, and stroma evaluation performed by pathologists blinded to the treatment. <i>Principal limitations</i> : intake of drinking-water was not measured, thus the PFOA dose was not known; data were combined for males and females; only one dose used; no survival data; short duration of exposure; small number of animals per group. <i>Other comments</i> : the mean PanIN grade did not significantly differ between control and PFOA- treated mice at 9 mo; the composite histopathology severity score, derived by incorporating PanIN lesion stage, inflammation, and stromal density, was not significantly increased at 9 mo; the lesion number per area was not significantly increased.	

Table 3.1 (contin	Table 3.1 (continued)					
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence or multiplicity	Significance	Comments		
Full carcinogenicity Mouse, C57BL/6J- <i>Apc</i> ^{Min/+} (M) Day 1 of gestation 11 wk Ngo et al. (2014)	Gavage PFOA (ammonium salt), ≥ 98% Water 0, 0.01, 0.1, 3.0 mg/kg bw per day 15, 3, 19, 0 15, 3, 19, 0	Small intestine (duodenum, jeju Tumour incidence: 15/15, 3/3, 19/19, 0/0 Tumour multiplicity: 146.7 ± 72.4, 128.0 ± 127.1, 82.2 ± 38.3, NR <i>Colon</i> Tumour incidence: 12/15 (80%), 3/3 (100%), 17/19 (89%), 0/0 Tumour multiplicity: 2.5 ± 2.2, 4.0 ± 3.5, 2.4 ± 2.2, NR	num or ileum) NS NS NS NS	 Principal strengths: males and females studied; multiple doses used; analysed background levels of PFOA in feed and drinking-water; analysed internal doses of PFOA; tested stability of PFOA; blocks of PFOA administration were compared statistically. Principal limitations: no histopathological examination of the small intestinal tumours was performed; small number of mice per group. Other comments: study of transplacental exposure; increase in the incidence and multiplicity of spontaneous tumours was studied in this mouse model; small intestinal tumours were observed in all Min/+ mice in all experimental groups, including the vehicle group, demonstrating 100% incidence in this end-point, as is usual in this mouse model. 		
Full carcinogenicity Mouse, C57BL/6J- <i>Apc</i> ^{Min/+} (F) Day 1 of gestation 11 wk <u>Ngo et al. (2014)</u>	Gavage PFOA (ammonium salt), ≥ 98% Water 0, 0.01, 0.1, 3.0 mg/kg bw per day 23, 15, 26, 2 23, 15, 26, 2	Small intestine (duodenum, jeju Tumour incidence: 23/23, 15/15, 26/26, 2/2 Tumour multiplicity: 151.0 \pm 102.3, 102.9 \pm 40.7, 119.5 \pm 73.0, 84.0 \pm 36.8 <i>Colon</i> Tumour incidence: 9/23 (39%), 8/15 (53%), 17/27 (63%), 1/2 (50%) Tumour multiplicity: 0.6 \pm 1.0, 0.8 \pm 1.1, 1.1 \pm 1.1, 1.0 \pm 1.4	num or ileum) NS NS NS NS	 Principal strengths: males and females studied; multiple doses used; analysed background levels of PFOA in feed and drinking-water, analysed internal doses of PFOA, tested stability of PFOA, blocks of PFOA administration were compared statistically. Principal limitations: no histopathological examination of the small intestinal tumours was performed; small number of mice per group; and short duration. Other comments: study of transplacental exposure; increase in incidence and multiplicity of spontaneous tumours was studied in this mouse model; small intestinal tumours were observed in all Min/+ mice in all experimental groups including the vehicle group, demonstrating 100% incidence in this end-point, as is usual in this mouse model. 		

Table 21 (contin ٦)

Table 3.1 (continued)					
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence or multiplicity	Significance	Comments	
Full carcinogenicity Mouse, CD-1 (F) Exposure on days 1–17 of gestation Killed at 18 mo Filgo et al. (2015)	Gavage PFOA (ammonium salt), > 98% (pure, linear product) Deionized water Control, 0.01, 0.1, 0.3, 1, 5 mg/kg bw, once per day 29, 29, 37, 26, 31, 21 29, 29, 37, 26, 31, 21	Liver Hepatocellular adenoma 0/29, 0/29, 1/37 (2.7%), 4/26* (15.38), 0/31, 1/21 (4.8%) Hepatocellular adenoma, multip 0/29, 1/29 (3.4%), 0/37, 0/26, 0/31, 0/21 Hepatocellular carcinoma 0/29, 0/29, 0/37, 1/26 (3.8%), 0/31, 1/21 (4.8%) Haemangiosarcoma 0/29, 0/29, 0/37, 1/26 (3.8%), 0/31, 2/21 (9.5%) Histiocytic sarcoma 0/29, 0/29, 1/37 (2.7%), 0/26, 1/31 (3.2%), 1/21 (4.8%) Lymphoma 1/29 (3.4%), 0/29, 0/37, 1/26 (3.8%), 1/31 (3.2%), 1/21 (4.8%)	* <i>P</i> < 0.05, Fisher exact test ole NS NS <i>P</i> < 0.01, Cochran- Armitage trend test NS	Principal strengths: three mouse strains used; multiple doses used. Principal limitations: only females were studied; number of tumours per animal was not reported; no statistical comparison between blocks of mice was reported; there was no statement that the dams were randomized to the treatment groups. Other comments: for mice killed before 18 mo when tumours were counted, only the percentage of mice born, which is unknown, was stated; thus, the numbers of mice reported at the start and surviving are both the numbers surviving at 18 mo and included in the study.	
Full carcinogenicity Mouse, 129/Sv wildtype (F) Exposure on days 1–17 of gestation Killed at age 18 mo Filgo et al. (2015)	Gavage PFOA (ammonium salt), > 98% (pure, linear product) Deionized water Control, 0.1, 0.3, 0.6, 1 mg/kg bw, once per day 10, 10, 8, 6, 8 10, 10, 8, 6, 8	Liver Adenoma 0/10, 0/10, 0/8, 0/6, 0/8 Haemangiosarcoma 0/10, 0/10, 0/8, 0/6, 0/8 Histiocytic sarcoma 0/10, 1/10 (10%), 0/8, 0/6, 0/8 Ito cell tumour 0/10, 0/10, 0/8, 0/6, 0/8	NS NS NS	 Principal strengths: three mouse strains used; multiple doses used. Principal limitations: only females were studied; number of tumours per mouse was not reported; no statistical comparison between blocks of mice was reported; there was no statement that the dams were randomized to the treatment groups. Other comments: for mice killed before 18 mo when tumours were counted, only the percentage of mice born, which is unknown, was reported. Thus, the numbers of mice given at start and surviving are both the numbers surviving at 18 mo and included in the study. 	

400

Table 3.1 (continu	Table 3.1 (continued)					
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence or multiplicity	Significance	Comments		
Full carcinogenicity Mouse, 129/Sv PPARa knockout (F) Exposure on days 1–17	Gavage PFOA (ammonium salt), > 98% (pure, linear product) Deionized water	<i>Liver</i> Hepatocellular adenoma 0/6, 1/10 (10%), 1/10 (10%), 1/9 (11%), 2/9 (22%)	NS	<i>Principal strengths</i> : three mouse strains used; multiple doses used. <i>Principal limitations</i> : only females were studied; number of tumours per mouse was not reported;		
of gestation Killed at age 18 mo <u>Filgo et al. (2015)</u>	Control, 0.1, 0.3, 1, 3 mg/kg bw, once per day 6, 10, 10, 9, 9 6, 10, 10, 9, 9	Haemangiosarcoma 1/6 (16.7%), 0/10, 0/10, 0/9, 0/9 Histiocytic sarcoma	NS	no statistical comparison between blocks of mice was reported; there was no statement that the dams were randomized to the treatment groups. <i>Other comments</i> : for mice killed before 18 mo when		
		0/6, 0/10, 0/10, 0/9, 0/9 Ito cell tumour	NS	tumours were counted, only the percentage of mice born, which is unknown, was reported. Thus, the		
		0/6, 0/10, 1/10 (10%), 0/9, 0/9	NS	numbers of mice given at start and surviving are both the numbers surviving at 18 mo and included in the study.		

bw, body weight; F, female; M, male; mo, month(s); NR, not reported; NS, not significant; PanIN, pancreatic intraepithelial neoplasia; PFOA, perfluorooctanoic acid; PPARa, peroxisome proliferator-activated receptor alpha; ppm, parts per million; vs, versus; wk, week(s).

(twofold) in the group receiving PFOA versus the controls at 6 months, but not at 9 months. [The Working Group noted that PanIN progresses to pancreatic ductal adenocarcinoma through stages characterized by morphological changes and nuclear atypia (see <u>Hezel et al., 2006</u>).] [The Working Group noted that this model effectively examines the shortening of latency by the treatment. The Working Group also noted that this study included two time points and measured PFOA concentrations in the serum and tissue at both time points; however, there was a lack of information on the number of animals per sex (males and females were combined); a limited number of animals (for most end-points, the number of mice per group was 9-11); a lack of information on survival and on randomization to treatment groups; one dose only was used; and there was a lack of information on the exact dose of PFOA ingested, since the intake of drinking-water was not measured.]

(b) Transplacental exposure

Ngo et al. (2014) examined the tumorigenic effects of gestational exposure to PFOA (specifically, the ammonium salt) in C57BL/6J-Apc^{Min/+} mice, a mouse model that develops spontaneous intestinal tumours because of a heterozygous Min mutation in the tumour suppressor gene adenomatous polyposis coli (Apc). These mice are sensitive to chemicals that mutate or delete (parts of) the remaining wildtype Apc allele, and it is a model both for the inherited disorder, familial adenomatous polyposis, and for sporadic colorectal cancer. The wildtype C57BL/6J-Apc+/+ dams were mated to heterozygous C57BL/6J-Apc^{Min/+} males. Pregnant wildtype females were treated with PFOA (purity, \geq 98%) by gavage at a dose of 0, 0.01, 0.1, or 3.0 mg/kg bw per day on days 1–17 of gestation. Insufficient rates of pregnancy and littering and low F₁ survival were observed in the first experimental block - block 1, 0 (vehicle, distilled water), 0.1, and 3.0 mg/kg per day; 104 exposed dams (age, 7-8 weeks) - thus, a second block was added for which the PFOA exposure was lower – block 2, 0, 0.01, and 0.1 mg/kg per day; 100 exposed dams (age, 9-10 weeks). The PFOA solutions were made separately for the two experimental blocks, and the gavage volumes for all doses were below 1 mL/100 g bw. The PFOA solutions were tested by chemical analysis and found to be stable during the experiment. Furthermore, the tap water (used as drinking-water for the mice) and both the breeding and maintenance diets, as well as the distilled water (used as the vehicle for PFOA), were analysed and showed very low background PFOA levels (picograms per litre and picograms per gram in water and feed, respectively). Internal exposure was quantified (1 or 2 mice per time point) in the dams on day 18 of gestation, postnatal day 23 (block 1) or postnatal days 26-28 (block 2), and F₁ pups on postnatal days 25-27 (depending on the block). The limit of quantification (LOQ) for PFOA was 0.05 ng/mL serum. Although minimal data were generated, they confirmed that the internal exposure within dams and pups increased with dose and decreased with time post-dosing (day 18 of gestation versus postnatal day 23 in dams). Serum concentrations of PFOA were significantly increased in mice exposed to PFOA, with mice in the control groups for both ages exhibiting an average PFOA concentration of 0.003 µg/mL, whereas serum concentrations in PFOA-treated KC mice aged 6 or 9 months were 41.96 \pm 16.45 and 26.35 \pm 17.53 µg/mL, respectively. PFOA concentrations in pancreatic tissue were also elevated (in the range of nanograms per milligram protein) in mice treated with PFOA. For $Min/+F_1$ male offspring, the numbers of mice in each dose group were 15, 3, 19, and 0, in the groups exposed to PFOA at a dose of 0 (vehicle, water), 0.01, 0.1, or 3.0 mg/kg bw, respectively. The numbers of $Min/+ F_1$ female offspring obtained in each dose group (both blocks together) were 23, 15, 26, and 2 in the groups exposed in utero to PFOA at dose of 0 (vehicle, water), 0.01, 0.1, or 3.0 mg/kg bw, respectively. For the dams weighed

on days 1–18 of gestation, there were no differences in body weight as area under the curve (AUC, arbitrary units) between the experimental groups, in either experimental block 1 or 2, and there was no difference in body weight between the two experimental blocks. For the pups aged 3–18 days, including both *Min/+* and wildtype (+/+) mice of both sexes, there were some significant differences in body weight between the treatment groups but in varying directions; thus, there were no consistent differences in pup body weight AUC for the pups between experimental blocks 1 and 2. Considering the individual pups in both experimental blocks, PFOA at doses of 3.0 and 0.1 mg/kg bw per day decreased the pup body weight compared with that of pups treated with the vehicle only (water). The offspring were weaned when aged 21 days and housed as a litter per cage, males and females separately. They were genotyped for the heterozygous Min/+ mutation using DNA collected from ear punches. The $Apc^{+/+}$ mice were not expected to develop intestinal tumours at age 11 weeks and were used for studies on non-cancer end-points. All C57BL/6J-Apc^{Min/+} offspring mice were killed at age 11 weeks, before the onset of serious anaemia caused by their spontaneous tumours (based on experience with this model), and were used to study the effects of PFOA on intestinal tumorigenesis. The number, diameter, and localization of tumours in the small intestine and colon were measured by transillumination in an inverse light microscope. The reviewer scored lesions at 20× magnification and was blinded to mouse treatment. The diameter of the tumours was scored using an eyepiece graticule. Statistical analysis of incidence was performed on both an individual and litter basis; furthermore, the two experimental blocks were combined in the analysis if no statistical differences in incidence were found between them.

Neoplastic lesions (tumours of the small intestine) were observed in all *Min/+* mice in all experimental groups including the vehicle group,

demonstrating 100% incidence of this end-point, as is usual in this mouse model. PFOA, at any dose, did not cause a significant increase in the number of small intestine tumours, compared with the vehicle (control). In male Min/+ mice only, treated with PFOA at 0.01 mg/kg bw, the small intestine tumours were larger in size than those in mice that were treated with the vehicle. Most of the small intestine tumours were localized in the distal two thirds, i.e. in the middle and distal parts, of the small intestine, irrespective of treatment or sex, as seen in previous experiments with Min/+ mice (see Andreassen et al., 2002). [The Working Group noted that there was no clear linear dose-response relation in the number and size of small intestine tumours and that these results were found both when the data were analysed with individual mice or with the litter as the statistical unit.]

The incidence of colon tumours at the individual level showed no significant differences between experimental blocks 1 and 2. The only significant difference between the treatment groups was that the group treated with PFOA at 0.1 mg/kg bw had a higher incidence of colon tumours than did the group treated with the vehicle, for males and females together, in experimental block 1 (P = 0.039, Fisher exact test, two-tailed probability). However, when this result was tested with the litter as the statistical unit, it did not reach statistical significance. There were no statistically significant differences in the number or diameter of colon tumours between experimental blocks 1 and 2 on the individual level, and therefore the data from both experimental blocks were evaluated together. There were no significant differences in the number of tumours of the colon in mice from any of the groups treated with PFOA compared with that in mice in the vehicle group. The experimental design, such as duration of the study with termination at 11 weeks, was based on previous experience with this model. [The Working Group noted that, usually, when Min/+ mice are killed

at age 11 weeks, most intestinal tumours identified are adenomas (see <u>Moser et al., 1990</u>).] [The Working Group noted that this study used both sexes, multiple doses, tested PFOA stability, and analysed the internal dose and the background levels of PFOA in feed and drinking-water. However, no histological examination of tumours of the small intestine or colon was performed.]

Filgo et al. (2015) studied liver toxicity in CD-1 and 129/Sv strains of mice treated with PFOA (specifically, the ammonium salt; purity, > 98%) administered orally (by gavage) after gestational exposure. Both wildtype and peroxisome proliferator-activated receptor alpha (PPARa)-knockout transgenic 129/Sv mice were used. Two blocks of time-pregnant CD-1 mice (12, 12, 14, 13, 12, and 6 dams) were treated with distilled water (vehicle control), or PFOA at a dose of 0.01, 0.1, 0.3, 1, or 5 mg/kg bw, respectively, resulting in a final number of 29, 29 37, 26, 31, and 21 female offspring per group, respectively, surviving to age 18 months. Some mice died before 18 months (28%, 17%, 16%, 28%, 24%, and 22% of the numbers at the beginning of the experiment from the control group and groups treated with PFOA at 0.01, 0.1, 0.3, 1, and 5 mg/kg bw, respectively), because of sudden unknown causes (found on check; 28% of early deaths) or severe dermatitis (common to CD-1 mice; 32%) and other health problems (40%) that required pre-emptive euthanasia. [The Working Group noted that the percentages of early death in this study were reported to be in line with survival rates reported in several other studies of control CD-1 mice aged 18 months, with an average death rate before 18 months of 21.7% (see Giknis and Clifford, 2010).] For the 129/Sv mice, three blocks of animals were used, each separated by 2-3 weeks. 129/Sv wildtype mice were dosed with vehicle, or PFOA at 0.1, 0.3, 0.6, or 1 mg/kg bw, resulting in a final number of 10, 10, 8, 6, and 8 female offspring surviving to age 18 months (to be consistent with the CD-1 mice) and included in the necropsy (from 7, 7, 5, 3, and 5 pregnant dams,

404

respectively). PPARa-knockout mice were dosed with vehicle, or PFOA at 0.1, 0.3, 1, or 3 mg/kg bw, resulting in a final number of 6, 10, 10, 9, and 9 offspring (from 5, 9, 8, 7, and 9 pregnant dams), respectively. Different dose ranges were used for the three strains because of differences in strain sensitivities to PFOA. The highest dose used per strain was selected to minimize developmental toxicities and litter loss (see Abbott et al., 2007). The lower doses were selected such that resulting adolescent mice would have PFOA blood serum concentrations comparable to those reported for highly exposed humans (see Macon et al., 2011). PFOA was administered to all mice by oral gavage on days 1-17 of gestation. To determine the dose amounts, the dams were weighed daily before dosing. At birth, the pups were individually weighed and sexed. Pups within a treatment group were pooled and randomly redistributed among the dams of their respective treatment groups, and litters were equalized to 10 male and female pups. [The Working Group noted that litter effects could not be evaluated because of the cross-fostering that only occurred in the CD-1 mice.] Among the CD-1 mice, small litters (fewer than 4 pups) were excluded from the study. Pups were weaned at age 21 days, and only female offspring were retained in this study and housed 3-5 mice per cage. At 18 months, all mice underwent full necropsy, and livers were collected from all surviving mice in the exposure groups. [The Working Group noted that the mice that died for various reasons before 18 months were not included in this study because of inconsistencies in age and quality of tissues that could be retrieved.] Liver sections underwent a pathology peer review by a team of board-certified veterinary pathologists (pathology working group) to determine the incidence of neoplastic and non-neoplastic lesions, and "INHAND" (International Harmonization of Nomenclature and Diagnostic Criteria for Lesions in Rats and Mice) liver nomenclature was used when evaluating liver lesions (Thoolen et al., 2010). [The Working Group noted that this study was not designed as a liver carcinogenesis study but was the consequence of finding liver tumours when investigating unscheduled deaths of PPARaknockout mice in which no tumours were expected to be found. In addition, mice that died before termination of the study were not examined and, therefore, potential tumours in the liver were not included that could have affected the statistical analysis.]

Neoplastic lesions were present in female CD-1 mice treated with PFOA. The incidence of hepatocellular adenoma - 0/29, 0/29, 1/37 (2.7%), 4/26 (15.4%), 0/31, and 1/21 (4.8%) for the groups at 0 (control), 0.01, 0.1, 0.3, 1, and 5 mg/kg bw, respectively – was significantly increased (P < 0.05, Fisher exact test) at 0.3 mg/kg bw, and exceeded the upper bound of the range for historical controls - 3/897 (0.3%); range, 0-2%- reported by <u>Giknis and Clifford (2010)</u>. There was one mouse with hepatocellular adenoma (multiple) in the group at the lowest PFOA dose. [The Working Group noted that there was a significant increase in the incidence of adenoma at the intermediate dose (0.3 mg/kg bw) compared with controls and that the upper bound of the range for historical controls was exceeded, but that there was no significant trend in the incidence of adenoma. Therefore, the Working Group was uncertain about the causal association between these tumours and PFOA exposure.] Hepatocellular carcinomas occurred in one mouse per group at 0.3 and 5.0 mg/kg bw, and histiocytic sarcomas developed in one mouse per group at 0.1, 1.0, and 5.0 mg/kg bw, but the incidence did not reach statistical significance for either tumour type. There was a significant positive trend (P < 0.01, Cochran–Armitage trend test) in the incidence of liver haemangiosarcoma - 0/29, 0/29, 0/37, 1/26 (3.8%), 0/31, 2/21 (9.5%) in the groups at 0 (control group), 0.01, 0.1, 0.3, 1, and 5 mg/kg bw, respectively – with the incidence exceeding the upper bound of the range observed in historical controls - 3/897 (0.3%),

range, 0–2% – reported by <u>Giknis and Clifford</u> (2010). In the control group, the only tumour found was a single malignant lymphoma. [The Working Group noted that lymphoma was a background lesion in historical controls for CD-1 females – 112/900 (12.4%); range, 0–6% (see <u>Giknis and Clifford, 2010</u>).] The overall incidence of malignant lymphoma was 3/144 (2.1%) after PFOA exposure versus 1/29 (3.4%) in the controls.

In the vehicle-treated 129/Sv wildtype mice, no tumours were found. The only tumour found in PFOA-treated 129/Sv wildtype mice was a histiocytic sarcoma in the group at 0.1 mg/kg bw. Hepatocellular adenomas developed in five PFOA-treated PPARa-knockout mice – one mouse in each of the three groups at the lower doses and two mice in the group at the highest dose (3 mg/kg bw) – leading to an overall incidence of 5/38 (13.2%) in PFOA-treated mice. An Itocelltumour developed in one PPA Ra-knockout mouse treated with PFOA at 0.3 mg/kg bw. The Working Group noted that the power to detect an effect was low for this study because of the low number of animals and that the knockout control group contained only six mice.]

Regarding non-neoplastic lesions, basophilic oreosinophilic foci were found in one CD-1 mouse in each PFOA-treated group at 0.01 mg/kg bw, 0.1 mg/kg bw (basophilic foci), and 0.3 mg/kg bw (eosinophilic foci). A significant positive trend in the incidence of oval cell hyperplasia, Ito cell hypertrophy, and centrilobular hepatocyte hypertrophy was observed in CD-1 mice after PFOA exposure, with the incidence being significantly increased for Ito cell hypertrophy and centrilobular hepatocyte hypertrophy in the group at 5 mg/kg bw. Chronic inflammation was common in CD-1 mice, and there was a dose-related increase in severity scores in PFOAexposed livers; mean severity in the two groups at the highest dose was significantly higher than in the controls. In PPARa-knockout mice, clear cell focus developed in one mouse in the group

treated with PFOA at 0.1 mg/kg bw, and eosinophilic foci developed in one mouse in each group at 0 (control) and 0.3 mg/kg bw. In 129/Sv wildtype mice, eosinophilic foci developed in one mouse in at each group at 0.3 and 0.6 mg/kg bw.

Non-neoplastic changes were also numerous in the 129/Sv strain after PFOA exposure. Significant positive trends were observed in the incidence of both bile duct hyperplasia and bile duct inclusion bodies (hyaline droplets) in 129/Sv PPARa-knockout mice, but there was no increase in the incidence of either bile duct hyperplasia (although there was a decreasing trend in severity with dose) or hyaline droplet accumulation (although there was a decreasing trend in incidence) in 129/Sv wildtype mice. The incidence of Ito cell hypertrophy decreased with increasing PFOA dose in PPARa-knockout mice. There was a significant positive trend in the incidence of haematopoietic cell proliferation with increasing PFOA dose in PPARa-knockout mice, but not in the 129/Sv wildtype mice. There was a significant increase in the incidence of centrilobular hepatocyte hypertrophy with increasing PFOA dose in the PPARa-knockout mice. Although the incidence of centrilobular hepatocyte hypertrophy in 129/Sv wildtype mice did not significantly change with PFOA dose, the severity increased significantly with PFOA dose. A similar increase in mean severity was noted in PPARa-knockout mice, but that effect did not reach statistical significance.

[The Working Group noted that only females were studied. For all three mouse strains, when tumours were counted for mice killed before 18 months, only percentages (%) of mice born were stated. The initial numbers of F_1 females for each study were not provided. Thus, in <u>Table 3.1</u> the numbers of mice given at start and surviving are both the numbers surviving at 18 months and included in the study. The number of tumours per mouse was not reported, thus, multiplicity was not known. The numbers of mice in the wildtype and knockout studies were low. No

statistical comparisons between experimental blocks of mice treated with PFOA were reported. There was no statement that the dams were randomized to the treatment groups. It was not reported whether histopathology was done without knowledge of treatment.]

3.1.2 Rat

See Table 3.2.

(a) Oral administration (feed)

In a study by <u>Biegel et al. (2001)</u> that was designed to compare the carcinogenic effects of Wyeth-14,6431 (designated as WY group), a rodent peroxisome proliferator and carcinogen, with those of PFOA (specifically, ammonium salt; designated as the C8 group), an initial group of 156 male Sprague-Dawley rats [Crl:CD BR (CD)] (age, 6 weeks) were treated with feed containing PFOA (purity, 98-100%) at a dose of 300 ppm for 24 months. Two control groups (156 rats in each) were either fed ad libitum (designated as the control group) or received the same amount of feed as the PFOA-treated group (pair-fed control group, designated as the CP-C8 group), respectively. The average daily dose of PFOA was 13.6 mg/kg bw per day in the C8 group. There were initially 156 rats per group, and 6 rats per group were randomly selected and killed at eight interim time points (approximately 1, 3, 6, 9, 12, 15, 18, and 21 months) for histology evaluation (48 rats) and measurements of cell proliferation and peroxisome proliferation (48 rats), leaving 60 rats per group for the 2-year observation for carcinogenesis (Biegel et al., 2001). Hormone measurement was performed at the eight interim time points using 10 rats per group that were randomly selected and not killed. At study termination, survival rates were approximately 15%, 35%, 48%, and 16% for the control, CP-C8, C8, and WY groups, respectively. [The Working Group noted that, although not clearly documented, rats for hormone measurement were

Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence	Significance	Comments
Full carcinogenicity Rat, Crl:CD BR (M) 49 d [7 wk] 24 mo Biegel et al. (2001)	Oral administration (feed) PFOA (ammonium salt), 98–100% Feed 0 (controls fed ad libitum), 0 (pair-fed controls; CP-C8), 300 (C8) ppm (approximately 0, 0, and 13.6 mg/kg bw) 156, 156, 156 80, 79, 76	<i>Liver</i> Hepatocellular adenoma 2/80 (3%), 1/79 (1%), 10/76 (13%)* Hepatocellular carcinora 0/80, 2/79 (3%), 0 /76 Hepatocellular adenoma 2/80 (3%), 3/79 (4%), 10/76 (13%)* <i>Testis</i> Leydig cell adenoma 0/80, 2/78 (3%), 8/76 (11%)* <i>Pancreas</i> Acinar cell adenoma 0/80, 1/79 (1%), 7/76 (9%)* Acinar cell carcinoma 0/80, 0/79, 1/76 (1%) Acinar cell adenoma or 0/80, 1/79 (1%), 8/76 (11%)*	a * $P < 0.05$, Dunnett test; $[P = 0.0038$, Fisher exact test] na NS a or carcinoma (combined) * $P < 0.05$, Dunnett test; $[P = 0.0336$, Fisher exact test] * $P < 0.05$, Dunnett test; $[P = 0.0448$, Fisher exact test] * $P < 0.05$, Dunnett test; $[P = 0.0279$, Fisher exact test] NS carcinoma (combined) * $P < 0.05$, Dunnett test; $[P = 0.0145$, Fisher exact test]	Principal strengths: long-term study; adequatenumber of rats per group; study covered mostof the lifespan.Principal limitations: only one dosegroup; one sex used; age of rats whenassessed for lesions (when killed) was notclearly documented; no results or data fortrend test(s) were reported, despite largedifferences in survival rates among groups;scheduled and unscheduled deaths werenot distinguishable and were shown as thedenominator of the rat numbers in Table 2 ofthis paper.Other comments: only the liver, testes,epididymides, pancreas, and organs withgross lesions were examined microscopically;of 165 rats per group, 48 rats were designatedfor interim kill for measurement of cellproliferation, and another 48 rats forperoxisome proliferation (6 rats × 8 timepoints); hormone analysis was performedwithout killing; 60 rats were likely to bedesignated for pathological evaluation forthe 2-yr time period; peer-review of thedata on pancreatic lesions by a panel ofpathologists (Caverly Rae et al., 2014) usingthe same diagnostic criteria as those appliedin the study by Biegel et al. (2001) generatedthe following incidence data (which werestatistically significant from those for pair-fed controls, *P < 0.05): pancreatic acinar

Table 3.2 Studies of carcinogenicity in rats exposed to PFOA

Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence	Significance	Comments
Full carcinogenicity Rat, Sprague- Dawley Crl: COBS (SD) BR (M) 39–41 d [6 wk] 104 wk <u>Butenhoff et al.</u> (2012a)	Oral administration (feed) PFOA (ammonium salt), 97.2% Feed 0, 30, 300 ppm (actual doses, 0, 1.3, and 14.2 mg/kg bw per day) 50, 50, 50 49, 50, 50	Testis and epididymis Leydig cell adenoma 0/49, 2/50 (4%), 7/50 (14%)* Liver Hepatocellular carcinom 3/49 (6%), 1/50 (2%), 5/50 (10%) Adrenal medulla Pheochromocytoma (bu 2/49 (4%), 4/50 (8%), 4/50 (8%) Pheochromocytoma (mu 0/49, 1/50 (2%), 0/50 Pituitary gland Adenoma 17/48 (35%), 17/47 (36%), 13/46 (28%) Thyroid gland, C-cell Adenoma 0/43, 2/47 (4%), 4/47 (9%) Carcinoma 2/43 (5%), 0/47, 0/47	$[P = 0.010, Cochran-Armitage trend test]*P \le 0.05, Fisher exact test (two-tailed)[*P = 0.0067, Fisher exact test (one-tailed)]maNSenign)NSalignant)NSNSNS$	<i>Principal strengths</i> : adequate number of animals per group, males and females used, adequate duration; well-conducted study. <i>Other comments</i> : 65 rats in the control and higher-dose groups, 50 rats in the lower-dose group (15 rats from the control and higher-dose groups were killed at 1 yr); no neoplasms at 1-yr interim kill; peer review of the pancreatic lesion data by a panel of pathologists (Caverly Rae et al., 2014) using the same diagnostic criteria as those applied in the study by Biegel et al. (2001) generated the following incidence data: pancreatic acinar cell hyperplasia, 3/46, 1/46, 10/47* [* <i>P</i> = 0.0382, Fisher exact test (one-tailed)].

Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence	Significance	Comments
Full carcinogenicity Rat, Sprague- Dawley Crl: COBS (SD) BR (F) 39–41 d [6 wk] 104 wk <u>Butenhoff et al.</u> (2012a)	Oral administration (feed) Ammonium salt, 97.2% Feed 0, 30, 300 ppm Actual doses: 0, 1.6, 16.1 mg/kg bw per day 50, 50, 50 46, 45, 44	Mammary gland Fibroadenoma 10/46 (22%), 19/45 (42%)*, 21/44 (48%)** Adenocarcinoma 7/46 (15%), 14/45 (31%), 5/44 (11%) Lymphangiosarcoma 0/46, 0/45, 1/44 (2%) Adrenal medulla Pheochromocytoma (m 0/50, 0/50, 1/49 (2%) Liver Hepatocellular carcinoma 0/50, 0/50, 1/50 (2%) Pituitary gland Adenoma 33/46 (72%), 39/47 (83%), 36/50 (72%) Adenocarcinoma 9/50 (18%),	<pre>[P = 0.024, Cochran-Armitage trend test] [*P = 0.0302, Fisher exact test (one- tailed)] [**P = 0.0086, Fisher exact test (one- tailed)] NS NS nalignant) NS MS NS</pre>	 Principal strengths: adequate number of animals per group; males and females used; adequate duration. Other comments: 65 rats in the control and high-dose groups, 50 rats in the low-dose group (15 rats from the control and high-dose groups were killed at 1 yr); no neoplasms at 1-yr interim kill. Peer review of the mammary gland data by a panel of pathologists (Hardisty et al., 2010) using contemporary diagnostic criteria generated the following incidence data (with no statistical significance): adenocarcinoma of the mammary gland, 9/50 (18%), 16/50 (32%), 5/50 (10%); adenoma of the mammary gland: 1/50 (2%), 0/50, 0/50; fibroadenoma of the mammary gland: 16/50 (32%), 20/50 (40%); fibroadenoma (multiple) of the mammary gland: 2/50 (4%), 6/50 (12%), 3/50 (6%) (not adjusted for survival).
		16/50 (32%), 5/50 (10%)		

Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence	Significance	Comments
Full carcinogenicity Rat, Hsd:Sprague- Dawley (M) Perinatal then PND 21–23 (study 2) 2 yr NTP (2020)	Oral administration (feed) PFOA, 98.8% NIH-07 (perinatal phase) and NTP-2000 (post- weaning phase) 0/0, 0/20, 0/40, 0/80, 300/0, 300/20, 300/40, 300/80 ppm Feed 50, 50, 50, 50, 50, 50, 50, 50, 50 36, 42, 35, 37, 35, 38, 38, 39	<i>Liver</i> Hepatocellular adenoma 0/50, 0/50, 7/50 (14%)*, 11/50 (22%)**, 0/50, 1/50 (2%), 5/50 (10%), 10/50 (20%)*** Hepatocellular carcinor 0/50, 0/50, 0/50, 0/50, 0/50, 0/50, 0/50, 0/50, (8%) Hepatocellular adenoma 0/50, 0/50, 7/50 (14%)*, 11/50 (22%)**, 0/50, 1/50 (2%), 5/50 (10)***, 12/50 (24%)****	a (includes multiple) P < 0.001, Cochran–Armitage trend test * $P = 0.050$, Rao–Scott adjusted poly-3 test; [$P = 0.0062$, Fisher exact test] ** $P = 0.010$, Rao–Scott adjusted poly-3 test; [$P = 0.0003$, Fisher exact test] *** $P = 0.006$, Rao–Scott adjusted poly-3 test; [$P = 0.0006$, Fisher exact test] ma Positive trend for F ₁ males with F ₀ exposure only a or carcinoma (combined) P < 0.001, Cochran–Armitage trend test * $P = 0.050$, Rao–Scott adjusted poly-3 test; [$P = 0.0062$, Fisher exact test] ** $P = 0.010$, Rao–Scott adjusted poly-3 test; [$P = 0.0062$, Fisher exact test] *** $P = 0.003$, Rao–Scott adjusted poly-3 test; [$P = 0.0003$, Rao–Scott adjusted poly-3 test; [$P = 0.0003$, Fisher exact test]	Principal strengths: adequate number of animals used; randomly allocated in groups; adequate duration; males and females used; multiple doses used; well-conducted GLP study. Other comments: historical controls: hepatocellular adenoma, all routes, 2/340 ($0.67\% \pm 1.03\%$); range, $0-2\%$; hepatocellular carcinoma, $0/340$; hepatocellular adenoma or carcinoma (combined), 2/340 ($0.67\% \pm 1.03\%$); range, $0-2\%$; pancreatic acinar cell adenoma, all routes, 45/340 ($12.33\% \pm 10.07\%$); range, $0-28\%$; pancreatic acinar cell adenocarcinoma, 2/340 ($0.52\% \pm 0.85\%$); range, $0-2\%$; pancreatic acinar cell adenoma or adenocarcinoma (combined), 45/340 ($12.33\% \pm 10.07\%$); range, 0-28%.

Table 3.2 (continued)						
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence	Significance	Comments		
Full		Pancreas				
carcinogenicity		Acinar cell adenoma (i	ncludes multiple)			
Rat, Hsd:Sprague- Dawley (M) Perinatal then PND 21-23 (study 2) 2 yr <u>NTP (2020)</u> (cont.)		3/50 (6%), 28/50 (56%)**, 26/50 (52%)**, 32/50 (64%)**, 7/50 (14%), 18/50 (36%)*, 30/50 (60%)**, 30/50 (60%)**	P < 0.001, Cochran–Armitage trend test * $P = 0.016$, Rao–Scott adjusted poly-3 test; [$P = 0.0099$, Fisher exact test] ** $P < 0.001$, Rao–Scott adjusted poly- 3 test; [$P < 0.0001$, Fisher exact test]			
		Acinar cell adenocarci	noma			
		0/50, 3/50 (6), 1/50 (2%), 3/50 (6%), 0/50, 2/50 (4%), 1/50 (2%), 3/50 (6%)	NS			
		Acinar cell adenoma o	r adenocarcinoma (combined)			
		3/50 (6%), 29/50 (58%)**, 26/50 (52%)**, 32/50 (64%)**, 7/50 (14%), 20/50 (40%)*, 30/50 (60%)**,	P < 0.001, Cochran–Armitage trend test * $P = 0.006$, Rao–Scott adjusted poly-3 test; [$P < 0.0001$, Fisher exact test] ** $P < 0.001$, Rao–Scott adjusted poly- 3 test; [$P < 0.0001$, Fisher exact test]			

Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence	Significance	Comments
Full carcinogenicity Rat, Hsd:Sprague- Dawley (F) Perinatal then PND 21–23 (study 1) 2 yr NTP (2020)	Oral administration (feed) PFOA, 98.8% NIH-07 (perinatal phase) and NTP-2000 (post- weaning phase) 0/0, 0/300, 0/1000, 150/300, 300/1000 ppm Feed 50, 50, 49, 50, 50 23, 28, 23, 32, 23	<i>Liver</i> Hepatocellular adenoma 2/50 (4%), 0/50, 1/49 (2%), 0/50, 3/50 (6%) Hepatocellular carcinom 1/50 (2%), 1/50 (2%), 3/49 (6%), 0/50, 4/50 (8%) Hepatocellular adenoma 3/50 (6%), 1/50 (2%), 4/49 (8%), 0/50, 6/50 (12%) <i>Pancreas</i> Acinar cell adenoma 0/50, 0/50, 1/49 (2%), 0/50, 3/50 (6%) Acinar cell adenocarcino 0/50, 0/50, 1/49 (2%), 0/50, 2/50 (4%) Acinar cell adenoma or a 0/50, 0/50, 2/49 (4%), 0/50, 5/50 (10%) <i>Uterus</i> Adenoma (extended eval 1/50 (2%), 0/49, 0/48, 0/50, 0/48 Acinar cell adenocarcino 1/50 (2%), 5/49 (10%), 7/48 (14%)*, 3/50 (6%), 5/48 (10%)	NS NS NS NS or carcinoma (combined) NS NS NS MS adenocarcinoma (combined) (P = 0.018, Rao-Scott adjusted trend poly-3 test) huation) NS oma (extended evaluation) * $P = 0.005, Rao-Scott adjusted poly-3 test; [P = 0.0258, Fisher exact test]$	Principal strengths: adequate number of animals used, randomly allocated in groups; adequate duration; males and females used; multiple doses used; well-conducted GLP study. Other comments: historical controls: hepatocellular adenoma, all routes, 14/340 ($3.63\% \pm 2.59\%$); range, $0-8\%$; hepatocellular carcinoma: 1/340 ($0.33\% \pm 0.82\%$); range, 0-2%; hepatocellular adenoma or carcinoma (combined), 15/340 ($3.96\% \pm 2.77\%$); range, 0-8%; pancreatic acinar cell adenoma, all routes, 0/340; pancreatic acinar cell adenocarcinoma, 0/340; pancreatic acinar cell adenoma or adenocarcinoma (combined), 0/340; adenoma of the uterus (standard evaluation), all routes, 1/150 ($0.67\% \pm 1.15\%$); range, $0-2\%$; adenocarcinoma of the uterus (standard evaluation), 11/150 ($7.33\% \pm 4.62\%$); range, $2-10\%$; adenoma or adenocarcinoma (combined) of the uterus, 12/150 ($8\% \pm 3.46\%$); range (standard evaluation), 4-10%.

Table 3.2	(continued)
-----------	-------------

Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence	Significance	Comments
Full carcinogenicity Rat, Hsd:Sprague- Dawley (F) Perinatal then PND 21–23 (study 1) 2 yr NTP (2020) (cont.)		Acinar cell adenoma or (extended evaluation) 2/50 (4%), 5/49 (10%), 7/48 (15%), 3/50 (6%), 5/48 (10%) Acinar cell adenoma (st combined) 1/50 (2%), 1/49 (2%), 0/49, 0/50, 0/50 Acinar cell adenocarcin combined) 1/50 (2%), 5/49 (10%), 7/49 (14%)*, 3/50 (6%), 5/50 (10%) Acinar cell adenoma or or extended evaluation 2/50 (4%), 5/49 (10%), 7/49 (14%), 3/50 (6%), 5/50 (10%)	adenocarcinoma (combined) NS andard or extended evaluation NS oma (standard or extended evaluation *P = 0.050, Rao–Scott adjusted poly-3 test; [$P = 0.0277$, Fisher exact test] adenocarcinoma (combined) (standard combined) NS	

bw, body weight; d, day(s); F, female; GLP, Good Laboratory Practice; M, male; mo, month(s); NIH, National Institutes of Health; NS, not significant; NTP, National Toxicology Program; PFOA, perfluorooctanoic acid; PND, postnatal day(s); ppm, parts per million; vs, versus; wk, week(s); yr, year(s).

not killed and were returned to the group for further treatment and observation. Accordingly, the Working Group estimated that the final number of rats for pathological evaluation at the 24-month time point was around 60 animals per group. Survival data were provided in graphic form only (the actual numbers were not reported) and therefore numbers were estimated by the Working Group from the graph presented in the original publication.] All surviving rats underwent complete necropsy with histopathological evaluation. The liver, testes, epididymis, pancreas, and organs with gross lesions were examined microscopically.

At 24 months, exposure to PFOA (C8 group) significantly increased the incidence of hepatocellular adenoma - 10/76 (13%) versus 1/79 (1%), (P < 0.05, Dunnett test; [P = 0.0038, Fisher exact]test]) - and of hepatocellular adenoma or carcinoma (combined) - 10/76 (13%) versus 3/79 (4%), (P < 0.05, Dunnett test; [P = 0.0336, Fisher exact]test]) – compared with the pair-fed control group CP-C8. There was a significant increase (P < 0.05, Dunnett test; [P = 0.0448, Fisher exact test]) in the incidence of testicular Leydig cell adenoma in the C8 group compared with the CP-C8 group - 8/76 (11%) versus 2/78 (3%). There were significant increases (P < 0.05, Dunnett test; [P = 0.0279, Fisher exact test]) in the incidence of pancreatic acinar cell adenoma in the C8 group compared with the CP-C8 group - 7/76 (9%) versus 1/79 (1%) – and in the incidence of pancreatic acinar cell adenoma or carcinoma (combined) in the C8 group (*P* < 0.05, Dunnett test; [*P* = 0.0145, Fisher exact test]) compared with the CP-C8 group -8/76 (11%) versus 1/79 (1%). [The Working Group noted that the numbers of rats that were submitted for pathological diagnosis were indicated only in the footnote of Table 2 of the publication and are the sum of scheduled and unscheduled deaths.]

Regarding non-neoplastic lesions in the C8 group compared with the CP-C8 group, there was a significant increase in the incidence of pancreatic acinar cell hyperplasia.

[The Working Group noted that this was a long-term study using a large group size and that the duration was most of the lifespan, and that the stability of the test article was evaluated and the concentration in the diet measured. However, the study was limited by the use of a single dose; the use of one sex only; because the age of each animal at death was not clearly stated; and because no results or data for trend test(s) were reported, despite large differences in survival rates among groups.]

Butenhoff et al. (2012a) published a report of a well-conducted study carried out between 1981 and 1983. The original reports of this study (US EPA, 1987) were published in a peer-reviewed publication by **Butenhoff et al.** (2012a). In this study of chronic toxicity that complied with Good Laboratory Practice (GLP), groups of male and female Sprague-Dawley rats [Crl:COBS CD(SD)BR] (age, 39-41 days) were treated with feed containing PFOA (specifically, the ammonium salt; purity, 97.2%) at a concentration of 0 (control), 30, or 300 ppm, corresponding to an average daily dose of approximately 0, 1.3, and 14.2 mg/kg bw in males, and 0, 1.6, and 16.1 mg/kg bw in females. The control group and group at the higher dose contained 65 males and 65 females, and the group at the lower dose contained 50 males and 50 females. An interim kill at 1 year involved 15 males and 15 females from both the control group and the group at 300 ppm, and the remaining rats, 50 of each sex, were killed at 104 weeks (US EPA, 1987); also reported by Butenhoff et al., 2012a). At study termination at 104 weeks, survival was 70%, 72%, and 88% for male rats, and 50%, 48%, and 58% for female rats, for the groups at 0 (control), 30, and 300 ppm, respectively; the increased survival rate observed in males at the higher dose was statistically significant, compared with the males in the control group. For male rats at the higher dose, body-weight gains were decreased by more than 10% throughout the 66 weeks of the study, compared with controls. The largest decrease was approximately 21% by week 6. In males at the lower dose, a slight but not significant (5%) decrease in body weight was observed at week 6, with little additional decrease subsequently. In female rats treated with PFOA, mean body weights did not change during the first 18 months of the study. After 18 months, there was a gradual decrease in mean body weight in females at 300 ppm; the maximum decrease was 11% lower than that of the controls at 22 months. Mean feed consumption relative to body weight was increased in all the PFOA-treated males throughout the study. In females, there was a trend towards lowered food consumption in both PFOA-treated groups.

In males, there was a significant positive trend [P = 0.010, Cochran–Armitage test] in the incidence of testicular Leydig cell adenoma, and the incidence – 0/49, 2/50 (4%), and 7/50 (14%) for the groups at 0 (control), 30, and 300 ppm, respectively – was significantly increased at 300 ppm ($P \le 0.05$, Fisher exact test, two-tailed; [P = 0.0067, Fisher exact test, one-tailed]).

In females, there was a significant positive trend [P = 0.024, Cochran–Armitage test] in the incidence of fibroadenoma of the mammary gland, and the incidence – 10/46 (22%), 19/45 (42%), and 21/44 (48%) for the groups at 0 (control), 30, and 300 ppm, respectively – was significantly increased at 30 and 300 ppm ([P = 0.0302, P = 0.0086, Fisher exact test, respectively]).

In 2010, a pathology working group was convened to review the original slides of mammary glands from the study by <u>US EPA</u> (1987), to provide a consensus diagnosis for neoplasms of the mammary gland using contemporary diagnostic criteria (<u>Hardisty et al., 2010</u>). The pathology working group concluded that some lesions originally diagnosed as lobular hyperplasia had features consistent with fibroadenoma of the mammary gland (mainly in slides from the control group), and that, consequently, PFOA did not induce neoplasms of the mammary gland. Both the initial data on mammary pathology (<u>US EPA, 1987</u>) and the reviewed data (<u>Hardisty et al., 2010</u>) were reported by <u>Butenhoff</u> et al. (2012a).

There was an increase in the incidence of hepatocellular hypertrophy in males and females at the higher dose, and an increase in the incidence of liver cystoid degeneration, vascular mineralization of the testis, and portal mononuclear cell infiltrate in males at the higher dose (Butenhoff et al., 2012a). Increases in the incidence of pancreatic acinar cell hyperplasia in male rats - 0/46 (0%), 2/46 (4%), and 2/49 (4%) in the groups at 0 (control), 30, and 300 ppm, respectively – were not statistically significant. Pancreatic acinar hyperplasia was not reported in female rats in this study. [The Working Group noted that this study used an adequate number of rats per group, both sexes, and an adequate duration of exposure. Discrepancy between the original study pathology and the review pathology (Hardisty et al., 2010) regarding the diagnosis of mammary lesions was noted. The Working Group also noted that increases in the incidence of Leydig cell adenoma were the only positive finding when using contemporary diagnostic criteria. In addition, the Working Group noted that faster elimination occurs in female rats than in males, as outlined in Section 4.1 of the present monograph, which may explain why minimal effects were observed in females.]

In a review of the pancreatic lesions observed in male rats in US EPA (1987), also reported by Butenhoff et al. (2012a), using the same diagnostic criteria as those applied in the study by Biegel et al. (2001), a significant positive trend (P < 0.05, Cochran–Armitage trend test) in the incidence of pancreatic acinar cell hyperplasia was observed in males, and the incidence was significantly increased [P = 0.0382, Fisher exact test] at the higher dose – 3/46 (7%), 1/46 (2%), and 10/47 (21%) (Caverly Rae et al., 2014). There were no statistically significant or test-related increases in the incidence of acinar cell adenoma or in acinar cell carcinoma separately with either PFOA dose, but there was a significant positive trend (P < 0.05, Cochran–Armitage trend test) in the incidence of all three lesions combined (hyperplasia, adenoma, and carcinoma) - 3/46(7%), 1/46 (2%), and 11/47 (23%). [The Working Group noted that only one neoplasm was observed, which was a carcinoma in the group at the higher dose. Both pancreatic acinar cell hyperplasia and pancreatic acinar adenoma were considered to be proliferative lesions, and this review supported the conclusion that the pancreas is a target of PFOA in male rats. In addition, the Working Group noted that hyperplasia, adenoma, and carcinoma were combined under the assumption that they are sequential pathological lesions.]

In a well-conducted study of chronic toxicity that complied with GLP and in which early-life exposure to PFOA on carcinogenicity outcomes was investigated, PFOA (purity, 98.8%) was administered to groups of 36 Hsd:Sprague-Dawley pregnant rats from day 6 of gestation through lactation, and subsequently to their pups for 2 years (NTP, 2020, revised in 2023). The control group comprised 103 pregnant females. F_0 females received feed containing PFOA at a concentration of 0, 150, or 300 ppm and were housed individually during gestation and together with their respective litters during lactation. The pups (F_1) were weaned on postnatal days 21–23. All F_1 exposure groups comprised 50 males and 50 females and were treated with feed containing PFOA at a concentration of 150 or 300 ppm for males and 300 or 1000 ppm for females. The initial dose setting, i.e. 150 ppm and 300 ppm during the mating and preweaning period (F_0) combined with 300 ppm and 1000 ppm (females) or 150 ppm and 300 ppm (males) for 2-year dietary exposure to the offspring (F_1) was tolerated only by female offspring (designated as study 1 for females only). [The Working Group noted that elimination is faster in female rats than in males (as outlined in Section 4.1 in the present monograph), which

was used to explain the higher post-weaning doses used in females.] Therefore, a second study was started that was focused entirely on males, and post-weaning concentrations were lowered (designated as study 2 for males only). A single perinatal exposure concentration was used, i.e. 300 ppm for the F_0 rats, and 20, 40 and 80 ppm for the F_1 rats. Total and live litter sizes and survival of the F_1 rats during lactation were not affected by exposure.

The treatment groups are indicated by the given doses in parts per million for the F_0 (gestation/lactation) and F_1 (post-weaning) as F_0/F_1 , such as 0/1000.

At termination of study 2 (2 years, males only), group mean body weights for the groups at 0/20, 0/40, 0/80, 300/0, 300/20, and 300/40 ppm (males) were within 10% of those for the respective control groups (0/0 ppm or 300/0 ppm). The terminal mean body weight of the group at 300/80 ppm was 13% less than that of the control group at 0/0 ppm. Post-weaning consumption of PFOA in males was 1.1/1.0, 2.2/2.1 and 4.6/4.6 mg/kg per day for the groups at 20, 40, and 80 ppm, with or without perinatal exposure. At termination of study 1 (2 years, females only), group mean body weights for the groups at 0/1000 and 300/1000 ppm were lower (19% and 27%, respectively) than those in the 0/0 ppm control group (females). Group mean feed consumption in females over the course of the study averaged 93%, 99%, 96%, and 88% of that in the 0/0 ppm control group for the groups at 0/300, 150/300, 0/1000, and 300/1000 ppm, respectively. After weaning, PFOA consumption for females in the groups at 0/300 and 150/300 ppm averaged 18.2 and 18.4 mg/kg per day, respectively. PFOA consumption averaged 63.4 and 63.5 mg/kg per day for the groups at 0/1000 and 300/1000 ppm, respectively (NTP, 2020).

In F_1 male rats (2 years, study 2), there was a significant positive trend in the incidence of hepatocellular adenoma (includes multiple) (P < 0.001, Cochran–Armitage trend test) in

both the F_0 exposed and the F_0 unexposed groups, with the incidence being significantly increased (P = 0.050, Rao–Scott adjusted poly-3 test, [P = 0.0062, Fisher exact test]; P = 0.010,Rao–Scott adjusted poly-3 test, [P = 0.0003,Fisher exact test]; P = 0.006, Rao–Scott adjusted poly-3 test, [P = 0.0006, Fisher exact test]) at 0/40, 0/80, and 300/80 ppm, respectively) in both the F_0 exposed and the F_0 unexposed groups, i.e. 0/50, 0/50, 7/50 (14%), 11/50 (22%) at 0/0, 0/20, 0/40, 0/80 ppm, and 0/50, 1/50 (2%), 5/50 (10%), 10/50 (20%) at 300/0, 300/20, 300/40, and 300/80 ppm. In addition, the incidence in all treated groups, except the group at 300/20 ppm, exceeded the upper bound of the range observed in historical controls from this laboratory - $2/340 (0.067\% \pm 1.03\%)$; range, 0–2%. There was a significant positive trend in the incidence of hepatocellular carcinoma (P = 0.049, Cochran– Armitage trend test) in male rats with perinatal exposure. No carcinomas were observed in the male rats with only post-weaning exposure. There was a significant positive trend in the incidence of hepatocellular adenoma or carcinoma (combined) (P < 0.001, Cochran–Armitage trend test) in both the F_0 exposed and the F_0 unexposed groups, with the incidence being significantly increased (P = 0.050, Rao–Scott adjusted poly-3 test, [P = 0.0062, Fisher exact test]; P = 0.010,Rao–Scott adjusted poly-3 test, [P = 0.0003,Fisher exact test]; P = 0.003, Rao–Scott adjusted poly-3 test, [P = 0.0001, Fisher exact test] at 0/40, 0/80, and 300/80 ppm, respectively, in both the F_0 exposed and the F_0 unexposed groups, i.e. 0/50, 0/50, 7/50 (14%), and 11/50 (22%) for the groups at 0/0, 0/20, 0/40, and 0/80 ppm, and 0/50, 1/50 (2%), 5/50 (10%), and 12/50 (24%) for the groups at 300/0, 300/20, 300/40, 300/80 ppm. In addition, the incidence in all treated groups, except in the group at 300/20 ppm, exceeded the upper bound of the range observed in historical controls from this laboratory $-2/340 (0.067\% \pm 1.03\%)$; range, 0-2%.

There was a significant positive trend in the incidence of acinar cell adenoma of the pancreas (includes multiple) (P < 0.001, Cochran–Armitage trend test) in both the F_0 exposed and the F₀ unexposed groups, with the incidence being significantly increased (P < 0.0001, Rao-Scott adjusted poly-3 test, [P > 0.0001, Fisher exact test] at 0/20, 0/40, 0/80, and 300/40 ppm; and P = 0.016, Rao-Scott adjusted poly-3 test, [P = 0.0002, Fisher exact test] at 300/20 ppm) in both the F_0 exposed and the F_0 unexposed groups - 3/50 (6%), 28/50 (56%), 26/50 (52%), and 32/50 (64%) for the groups at 0/0, 0/20, 0/40, and 0/80 ppm, and 7/50 (14%), 18/50 (36%), 30/50 (60%), and 30/50 (60%) for the groups at 300/0, 300/20, 300/40, and 300/80 ppm, respectively). In addition, the incidence in all treated groups exceeded the upper bound of the range observed in historical controls from this laboratory – all routes, 45/340 (12.33% \pm 10.07%); range, 0–28%. The incidence of pancreatic acinar cell adenocarcinoma was not statistically significant in any of the treated groups versus controls and exceeded the upper bound of the range observed in historical controls from this laboratory - all routes, $2/340 (0.52\% \pm 0.85\%)$; range, 0-2% – for the groups at 0/20, 0/80, 300/20, and 300/80 ppm. There was a significant positive trend in the incidence of acinar cell adenoma or adenocarcinoma (combined) of the pancreas (*P* < 0.001, Cochran– Armitage trend test) in both the F_0 exposed and the F_0 unexposed groups, with the incidence being significantly increased (P < 0.0001, Rao– Scott adjusted poly-3 test, [P < 0.0001, Fisherexact test] for all treated groups) in both the F_0 exposed and the F_0 unexposed groups – 3/50 (6%), 29/50 (58%), 26/50 (52%), and 32/50 (64%) for the groups at 0/0, 0/20, 0/40, and 0/80 ppm, and 7/50 (14%), 20/50 (40%), 30/50 (60%), and 30/50 (60%) for the groups at 300/0, 300/20, 300/40, and 300/80 ppm, respectively). In addition, the incidence in all treated groups exceeded the upper bound of the range observed in historical

controls from this laboratory – all routes, 45/340 (12.33% ± 10.07%); range, 0–28%.

The effect of perinatal exposure (F_0) over the effect of postnatal exposure (F_1) was not apparent for hepatocellular adenoma and acinar cell adenoma of the pancreas. There was a suggestive but not statistically significant effect of perinatal exposure on the incidence of hepatocellular carcinoma in male rats – 0/50, 0/50, 0/50, and 0/50 versus 0/50, 0/50, 0/50, 4/50 (P = 0.049 by Rao–Scott adjusted poly-3 test). [The Working Group noted that hepatocellular carcinoma is a rare neoplasm (0/340 in historical controls).]

In female rats (2 years, study 1), there was a significant positive trend (P = 0.018, Rao–Scott adjusted poly-3 test) in the incidence of pancreatic acinar cell adenoma or adenocarcinoma (combined) with perinatal exposure -0/50, 0/50,and 5/50 (10%) for the groups at 0/0, 150/300, and 300/1000 ppm, respectively). There was a significant increase in the incidence of adenocarcinoma of the uterus (standard or extended evaluation combined) (P = 0.050, Rao-Scott adjusted poly-3 test, [P = 0.0227, Fisher exact test] for the group at 0/1000 ppm) in F₀ exposed groups -1/50 (2%), 5/49 (10%), 7/49 (14%), 3/50 (6%), and 5/48 (10%) for the groups at 0/0, 0/300, 0/1000, 150/300, and 300/1000 ppm, respectively). [The Working Group noted that the new and extended evaluation used a combination of two sectioning methods. Because of this change in methods, the historical controls were of limited utility for the results obtained by the new method. The Working Group also noted that the new data reflected the 2023 revision.]

In 2023, a revision was made due to the identification of an error in the combining process for the uterine adenocarcinomas: "One animal with a squamous cell carcinoma in the 0/1000 ppm group was inadvertently combined in the adenocarcinoma analysis of the extended evaluation. The number of animals examined during the standard, extended and standard or extended (combined) evaluations was also corrected in the 0/300, 0/1000, and 300/1000 ppm groups" (NTP, 2020; revised in 2023). [The Working Group noted that the squamous cell carcinoma is of the same origin as the endometrial epithelium and can be combined with the adenoma and carcinoma. The Working Group also noted that the significant difference in the incidence of adenocarcinoma of the uterus in the group with the highest exposure without perinatal exposure was still statistically significant.]

Regarding non-neoplastic lesions, exposure to PFOA resulted in increases in the incidence of non-neoplastic lesions in the liver (hepatocyte cytoplasmic alteration; hepatocyte hypertrophy; hepatocyte single cell death; necrosis; and pigment in males and females) in males; hepatocyte cytoplasmic alteration; hepatocyte hypertrophy; hepatocyte single cell death; necrosis; pigment; bile duct hyperplasia; hepatocyte increased mitoses, in females; pancreatic acinus hyperplasia in male rats; and follicular cell hypertrophy of the thyroid gland of female rats. [The Working Group noted that pancreatic acinus adenoma and adenocarcinoma are rare lesions in females of this rat species and that pancreatic acinus hyperplasia was also considered to be rare; although these effects were of low incidence, they were consistent with the increased incidence of pancreatic acinar cell lesions reported in male rats.]

[The Working Group noted that this was a well-conducted study that complied with GLP and that used an adequate number of rats per group, both sexes, and an adequate duration of exposure. The Working Group also noted that internal exposure was measured in male and female rats, and that the stability of the test article was tested.]

(b) Initiation-promotion studies

See <u>Table 3.3</u>.

In an initiation-selection-promotion study, adult male Wistar rats [age not reported] were initiated with diethylnitrosamine (DEN)

	•	•		
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence or multiplicity	Significance	Comments
Initiation-promotion (tested as promoter) Rat, Wistar (ICO:WI IOPS AF:Han) (M) Age NR ("adult") 7 mo after initiation <u>Abdellatif et al. (1990)</u>	Oral administration (feed) PFOA, NR (analytical grade) Feed Basal feed (control), PB (positive control), PFOA (% feed) NDEA (single i.p. injection), 2-AAF (feed), carbon tetrachloride (gavage), PFOA and PB (feed) – Initiation: NDEA, 200 mg/kg bw (all three groups) – Selection: 2 wk after initiation, 0.03% 2-AAF for 2 wk; carbon tetrachloride, 2 mL/kg bw in 1:1 corn oil (after 1 wk of 2-AAF) – Promotion: 0% (control), 0.05% PB or 0.15% PFOA for 7 mo 7, 8, 12 7, 8, 12	Liver Total tumours Tumour incidence: 0/7, 6/8 (75%)**, 4/12 (33%)* Tumour multiplicity: 0, 3.4, 1.2	* <i>P</i> < 0.05, Student <i>t</i> -test ** <i>P</i> < 0.02, Student <i>t</i> -test	Principal strengths: the only report on the promoting activity of PFOA identified in rats via an initiation-selection-promotion protocol; PFOA concentrations measured in serum; end-points were measure at two time points. Principal limitations: only a limited number of rats were used in the experiment; histopathological examination of the liver only; only one sex used; average daily dose of PFOA was not reported.
Initiation-promotion (tested as promoter) Rat, Wistar (ICO:WI IOPS AF:Han) (M) Age NR ("adult") 12 mo <u>Abdellatif et al. (1991)</u>	Oral administration (feed) PFOA, NR (analytical grade) Feed Basal feed (control), PB (positive control), PFOA (0.005%), PFOA (0.02%) % in feed NDEA (single i.p. injection), PB and PFOA (feed) – Initiation: NDEA, 200 mg/kg bw (all three groups) – Promotion: basal feed (control), 0.05% PB, or 0.005% or 0.02% PFOA for 12 mo 10, 10, 10, 10 7, 7, 7, 9	<i>Liver</i> Hepatocellular carci Tumour incidence: 0/7, 2/7 (28%), 1/7 (14.3%), 5/9 (55.5%)*	noma *P < 0.05, Scheffé test NR	Principal strengths: the only report on promoting activity of PFOA identified in rats via an initiation- selection-promotion protocol; PFOA concentrations measured in serum; end-points were measured at two time points. Principal limitations: only a limited number of rats were included in the experiment; average daily dose of PFOA was not reported; histopathological examination of the liver only; only one sex used.

Table 3.3 Initiation-promotion studies in rats and fish exposed to PFOA

Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence Significance or multiplicity	Comments
Initiation–promotion (tested as promoter) Rat, Sprague-Dawley (F) 3 wk 7 mo <u>Su et al. (2022)</u>	Oral administration (gavage) PFOA, 95% Sesame oil 0, 0.01 mg/kg bw PFOA, zeranol, or PFOA + zeranol for 3 wk; at the age of 50 d, all rats received a single dose of DMBA 37, 37 37, 37	Mammary glandPapillary adenocarcinoma or cribricarcinomaTumour incidence:NS $35/37$ (94.6%)Tumourmultiplicity: 3.5 ± 2.2 , 3.7 ± 2.2 Total tumours:121, 129	Principal limitations: dose of the initiation agentiformDMBA was not optimal: extremely high tumour incidence in control (DMBA-treated group) did not allow investigation of potential enhancement of tumour incidence in the DMBA/PFOA-treated group; only one sex used; histological examination of the mammary gland only; only one dose. Other comments: the mixture of invasive papillary adenocarcinoma type 2 (prevalent) and invasive cribriform carcinoma was the most frequent mixed type for the PFOA group; histologically identified mammary tumours were also investigated by RNA- seq and qRT-PCR analyses; immunohistochemical analysis of selected receptors and effects on the endocrine system.
Initiation-promotion (tested as promoter) Rainbow trout (<i>Oncorhynchus mykiss</i>), Mount Shasta strain (M, F) (combined) 10 wk post-hatch 6 mo <u>Tilton et al. (2008)</u>	Oral administration (feed) PFOA, NR Feed 0, 200, 1800 ppm Initiation: aqueous exposure to AFB ₁ or to vehicle (ethanol) for 30 min; after 3 mo, fed experimental diets containing lower or higher dose of PFOA 140, 140, 140 NR, NR, NR	Liver Mixed tumour, malignant $36\%, 34\%, 71\%^*$ * $P < 0.05$, log regression ar (compared w AFB ₁ /control Hepatocellular adenoma 3%, 0%, 8% NR Hepatocellular carcinoma 10%, 11%, 46% NR	Principal strengths: adequate number of fish; two doses used. Principal limitations: data for males and females combined; no survival data; histopathological examination of the liver only. Other comments: no tumours were observed in non-initiated fish treated with feed containing PFOA; liver tumour enhancement after AFB ₁ /PFOA treatment might be related to induced estrogen-like signalling; the historical incidence of spontaneous liver tumours in trout (age 9 mo) fed control feed was 0.1%.

Table 3.3 (continued)							
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Tumour incidence or multiplicity	Significance	Comments			
Initiation-promotion (tested as promoter) Rainbow trout (<i>Oncorhynchus mykiss</i>), Mount Shasta strain (M, F) (combined) 10 wk post-spawn 12 mo <u>Benninghoff et al.</u> (2012)	Oral administration (feed) PFOA, "analytical grade" aqueous exposure Sham/control, Sham/PFOA, AFB ₁ / control, AFB ₁ /PFOA ppm Initiation: aqueous exposure to AFB ₁ (10 ppb [0.01 ppm]) for 30 min; promotion: after 1 mo, fed experimental diets containing PFOA (2000 ppm) 5 d/wk for 6 mo 250, 250, 250, 250 NR, NR, NR, NR	<i>Liver</i> Total tumours 0, 0, 13%, 62%* Hepatocellular aden 0, 0, 26%, 10% Hepatocellular carci 0, 0, 0, 27% Mixed carcinoma 0, 0, 47%, 54%	*P < 0.01, logistic regression analysis (compared with AFB ₁ /control) oma NS noma NS	Principal strengths: the experiment wassupplemented with hepatic gene expression analysis,adequate number of fish per group.Principal limitations: a short-term exposure wasused in the global gene-expression experiment;males and females combined; only one dose used; nosurvival data.Other comments: no liver tumours were observedin non-initiated fish treated with PFOA; increasedmultiplicity and size of liver tumours, but statisticalanalysis not provided.			
Initiation-promotion (tested as promoter) Rainbow trout, Mount Shasta strain (M, F) (combined) 10 wk post-spawn 12 mo <u>Benninghoff et al.</u> (2012)	Oral administration (feed) PFOA, "analytical grade" Aqueous exposure Sham/control, sham/PFOA, MNNG/control, MNNG/PFOA Initiation: aqueous exposure to MNNG (35 ppm) for 30 min; promotion: after 1 mo, fed experimental diets containing PFOA (2000 ppm) 5 d/wk for 6 mo 250, 250, 250, 250 NR, NR, NR, NR	Liver Total tumours 0, 0, 51%, 81%* Hepatocellular aden 0, 0, 25%, 26% Hepatocellular carci 0, 0, 28%, 11% Hepatocellular carci 0, 0, 39%, 55% Stomach, kidney, swi No significant increa	*P < 0.0001, logistic regression analysis (compared with MNNG/control) oma NS noma NS noma [mixed] NS <i>im bladder</i> ase in the incidence of	<i>Principal strengths</i> : the use of MNNG as the initiation agent allowed investigation of whether tumorigenesis in other organs (kidney and swim bladder) was affected by PFOA treatment; adequate number of fish per group. <i>Principal limitations</i> : the MNNG dose was too high for estimation of effects of PFOA on stomach carcinogenesis (stomach tumour incidence in control fish was 99%); males and females combined; only one dose used; no survival data. <i>Other comments</i> : significant increase in liver tumour multiplicity (<i>P</i> < 0.005, Kruskal–Wallis test with Dunnett post hoc test for multiple comparisons).			

2-AAF, 2-acetylaminofluorene; AFB, aflatoxin B,; bw, body weight; d, day(s); DMBA, 7,12-dimethylbenz[*a*]anthracene; F, female; i.p., intraperitoneal; M, male; min, minute(s); MNNG, *N*-methyl-*N*'-nitro-*N*-nitrosoguanidine; mo, month(s); NDEA, *N*-nitrosodiethylamine; NR, not reported; NS, not significant; PB, phenobarbital; PFOA, perfluorooctanoic acid; ppb, parts per billion; ppm, parts per million; RNA-seq, RNA sequencing; qRT-PCR, quantitative reverse transcription-polymerase chain reaction; vs, versus; wk, week(s); yr, year(s).

(200 mg/kg bw, intraperitoneally), followed 2 weeks later by a selection procedure - feed containing 0.03% 2-acetylaminofluorene (2-AAF) for 2 weeks and in the middle of this treatment, after week 1, a treatment with a single dose of carbon tetrachloride at 2 mL/kg bw by gavage, 1:1 in corn oil (Abdellatif et al., 1990, also reported by Abdellatif et al., 1991 and Nilsson et al., 1991). One week after the selection procedure, the rats were divided into three groups receiving a basal diet (control), or a diet supplemented with either 0.05% phenobarbital (positive control) or 0.015% PFOA (analytical grade) for 23 weeks. [The average daily doses of 2-AAF, phenobarbital, and PFOA were not reported.] There were 7 rats in the control group, 8 rats in the phenobarbital-treated group, and 12 rats in the PFOA-treated group. Body weight was slightly but non-significantly decreased in the PFOA-treated group compared with the control group. [No data on survival at study termination were reported.] Liver samples were collected, and histological and histochemical evaluations were performed.

There was a significant increase (P < 0.05, Student *t*-test) in the incidence of total liver tumours in the phenobarbital-treated and PFOAtreated groups compared with the control group: 0/7, 6/8 (75%), and 4/12 (33%) for the control group, and groups receiving phenobarbital and PFOA, respectively. In the PFOA-treated group, 25% of the liver tumours were hepatocellular carcinomas (type I) and 8% were other tumours. In the phenobarbital-treated group, 63% were hepatocellular carcinomas (type I), and 12% were hepatocellular carcinomas (type IV). The tumour multiplicity was 3.4 and 1.2 for the phenobarbital- and PFOA-treated rats, respectively (Abdellatif et al., 1990).

Regarding pre-neoplastic lesions, some eosinophilic, basophilic, or mixed cell type foci and a few nodules were detected (<u>Abdellatif et al.</u>, <u>1991</u>). According to <u>Nilsson et al.</u> (1991), there were 8 nodules in the PFOA-treated group.

In the initiation-promotion study performed by the same research team (Abdellatif et al., 1991), groups of 15 adult male Wistar rats [age not reported] received a single intraperitoneal dose of DEN at 200 mg/kg bw for initiation. Control groups did not receive initiation treatment with DEN. After 2 weeks, the rats were fed basal feed (control), or feed containing 0.05% phenobarbital (positive control), or 0.005% or 0.02% PFOA (analytical grade) until termination at 12 months. The average daily doses of phenobarbital and PFOA were not reported. From each group, 5 rats were killed at the interim time of 3 months, and 10 rats were killed 12 months after the start of the experiment. Survival in the initiated groups was 7/10 (total in the control, phenobarbital-treated, and 0.005% PFOA-treated groups) and 9/10 in 0.02% PFOA-treated rats. Liver samples were collected, and histological and histochemical evaluations were performed.

There was a significant increase (P < 0.05, Sheffe test) in the incidence of hepatocellular carcinoma in the group at the higher dose of PFOA (DEN-initiated) - 0/7, 1/7 (14.3%), and 5/9 (55.5%), for the control group and the groups at 0.005% PFOA and 0.02% PFOA, respectively - at 12 months. For the positive control, phenobarbital, the result was 2/7 (28.6%) (Abdellatif et al., <u>1991</u>). All the malignant tumours were well-differentiated type I hepatocellular carcinoma in rats treated with 0.005% PFOA. In the rats treated with 0.02% PFOA, four out of nine rats had moderately differentiated type II hepatocellular carcinoma, and one rat had a poorly differentiated type III hepatocellular carcinoma. No tumours were identified in rats that died at an early stage of the experiment, all within the first 8 months of the study, with the cause of death reported to be pneumonia in all cases. Tumour multiplicity was not reported for any treatment groups.

Regarding pre-neoplastic lesions, the rats with or without malignant tumours had foci and nodules containing a mainly eosinophilic, but
also basophilic, clear cell population or a mixed cell pattern (Abdellatif et al., 1991). No foci, nodules or malignant tumours were observed in non-initiated control rats either after 3 or 12 months, or in the initiated rats killed after 3 months. [The Working Group noted that PFOA concentrations were measured in the serum, and end-points were measured at two time points. However, both studies (Abdellatif et al., 1990, 1991) used only one sex; the purity of PFOA was not reported for either of these two protocols, only that it was of the purest available analytical grade; the average daily doses of PFOA, phenobarbital, and 2-AAF, and survival at study termination were not reported; and the histopathological examination was limited to the liver.]

In a study by <u>Su et al. (2022</u>), the effect of pubertal exposure to an environmentally relevant dose of PFOA was investigated, using a model of 7,12-dimethylbenz[a]anthracene (DMBA)induced tumorigenesis in the rat mammary gland. The aim of the study was to investigate whether exposure to PFOA during puberty might alter susceptibility to breast cancer. Female Sprague-Dawley rats (age, 21 days) were randomized into 36 or 37 rats per group and exposed via gavage to sesame oil (controls), or to PFOA (purity, 95%) at a dose of 0.01 mg/kg bw, or to a combination of PFOA and zeranol (a metabolite of the mycotoxin zearalenone) (0.01 mg/kg bw), 5 days per week from age 21 to 42 days. At age 50 days, all rats were challenged with a single dose of DMBA (30 mg/kg bw) via gavage. There was no significant difference in body weight between treated and control groups. Survival was not significantly affected by PFOA treatment. The rats were monitored for the development of mammary gland tumours for 7 months.

There were no significant differences in tumour incidence or the number of tumours per rat in the groups treated with PFOA or with PFOA plus zeranol compared with the DMBA control group. Overall, tumour latency, based on tumour-free survival, was not significantly affected with PFOA alone.

Regarding pre-neoplastic lesions, none were reported. [The Working Group noted that this study used only one sex and one dose, and histological examination was performed on the mammary glands only. The Working Group also noted that tumour incidence in both the control group (DMBA-treated) and in groups treated with PFOA and DMBA was extremely high – 35/37 (94.6%) – therefore, it may have been very difficult to detect any promotion effects.]

In an initiation–promotion study in rainbow trout (Oncorhynchus mykiss), approximately 1000 fry were initiated at 10 weeks post-hatch with aqueous exposure to aflatoxin B_1 (AFB₁) at 0.01 ppm for 30 minutes. The non-initiated controls were sham-exposed trout, exposed to vehicle alone (0.01% ethanol). After 3 months, initiated trout were randomly divided among experimental treatment groups (140 animals per group) and fed a semi-purified casein-based diet containing PFOA [purity not reported] at a concentration of 200 or 1800 ppm (equivalent to doses of 5 and 50 mg/kg per day, respectively) for 5 days per week. [The Working Group noted that the concentration of PFOA in the water tank was not reported.] At 9 months post-initiation, juvenile fish were killed and sampled for liver tumour histological identification and examination using haematoxylin and eosin (Tilton et al., 2008).

No tumours were observed in non-initiated fish fed with control or PFOA diets, indicating lack of carcinogenic potential by themselves in this model. There was a significant increase (P < 0.05, logistic regression analysis) in the incidence of total liver tumours (cholangiocellular carcinoma, hepatocellular adenoma, hepatocellular carcinoma, and mixed adenoma and mixed carcinoma) – 36%, 34%, and 71% for the AFB₁/0 (control group), and the groups treated with AFB₁/200 ppm PFOA, and AFB₁/1800 ppm PFOA, respectively. [The Working Group noted that mixed adenoma and carcinoma comprised both cholangiocellular and hepatocellular cell types that are considered to be originated from a common progenitor cell of bile duct cells and liver cells.] Specifically, there was a significant increase in overall tumour incidence (71%) in the group treated with AFB₁/1800 ppm PFOA compared with the control group (P < 0.05, logistic regression analysis). There was also an increase in the incidence of hepatocellular carcinoma -10%, 11%, and 46% for the groups treated with AFB₁/0 (control), AFB₁/200, and AFB₁/1800 ppm, respectively - and in the incidence expressed as a percentage of hepatocellular adenoma - 3%, 0%, and 8% for the groups treated with $AFB_1/0$ (control), AFB₁/200, and AFB₁/1800 ppm, respectively - although no statistical testing for the individual tumour types was reported. In addition, the multiplicity of the induced tumours per animal was also increased.

Regarding non-neoplastic lesions, PFOA exposure produced hepatomegaly and basophilic foci (<u>Tilton et al., 2008</u>). [The Working Group noted that this study used an adequate number of animals per group and tested two doses of PFOA. However, data were combined for males and females, the purity of PFOA was not reported, no information on survival was provided, and histopathological examination was performed on the liver only.]

In another initiation-promotion study in Mount Shasta rainbow trout (*Oncorhynchus mykiss*), PFOA was evaluated by initiating about 3500 fry (age, 10 weeks) with AFB₁ at 10 ppb [0.01 ppm] or about 1000 fry (age, 10 weeks) with *N*-methyl-*N'*-nitro-*N*-nitrosoguanidine (MNNG) at 35 ppm for 30 minutes (<u>Benninghoff</u> et al., 2012). Since MNNG is a multiorgan carcinogen, MNNG initiation was used to determine whether the tumour-promoting effects of dietary PFOA were specific to hepatocarcinogenesis or dependent on the initiating carcinogen. Non-initiated sham controls in the two experiments were treated with the vehicle (0.01% ethanol or 0.01% dimethyl sulfoxide, DMSO, respectively). After initiation, the fry were given untreated feed (a semi-purified casein-based diet) for 1 month. Then, within each initiation cohort, trout were randomly divided into the treatment groups. In the first (AFB₁) cohort, fish were fed experimental diets containing PFOA at 2000 ppm (equivalent to 50 mg/kg per day, analytical grade), ad libitum, 5 days per week for 6 months. Untreated feed was used for controls. There were four exposure groups, each containing 250 fish: sham/control, sham/+PFOA, AFB₁/control, and AFB₁/+PFOA. The MNNG-initiated trout were exposed to the vehicle or PFOA at 2000 ppm. There were four exposure groups, each containing 250 fish: sham/-PFOA, sham/+PFOA, MNNG/-PFOA, and MNNG/+PFOA. The diet was prepared on a monthly basis and kept frozen at -20°C before use. The fish were terminated at 12.5 months post-spawn and examination of tumours was performed. Neoplasms were classified according to the criteria described in Hendricks et al. (1984).

A total of cholangiocellular adenomas or cholangiocellular carcinomas, hepatocellular adenomas or hepatocellular carcinomas, and mixed adenomas, and mixed carcinomas that consist of both cholangiocellular and hepatocellular cell types were counted as liver tumours. Initiation with AFB₁ at 10 ppb resulted in a moderate increase in liver tumour incidence (13%) compared with the control group. PFOA exposure significantly enhanced the incidence of liver tumours (62%) (P < 0.01, logistic regression analysis), and increased liver tumour multiplicity and size (both P < 0.05, Kruskal–Wallis test with Dunnett post hoc test for multiple comparisons). There was a significant increase (P < 0.0001, logistic regression analysis) in liver tumour incidence in the MNNG/PFOA group (86%) compared with the MNNG/control group (51%). Tumour multiplicity and size were also significantly increased by PFOA treatment (both P < 0.001, Kruskal–Wallis test with Dunnett post hoc test for multiple comparisons). After MNNG initiation, kidney and stomach carcinogenesis was not significantly affected by PFOA exposure. Mixed carcinoma followed by hepatocellular adenoma and hepatocellular carcinoma were the most prevalent liver tumour types in both experiments, with the prevalence being lower than that seen in groups treated with AFB₁ alone and MNNG alone (individual tumour types were not statistically analysed). [The Working Group noted that this study used an adequate number of animals per group; however, data were combined for males and females, only one dose of PFOA was tested, and the purity for PFOA was reported only as "analytical grade". It was noted that the results of these studies indicated that PFOA can act as a promoter in this fish model. Furthermore, the data reflect chemical-specific responses in the liver with both AFB₁ (liver-specific) and MNNG (multiorgan) initiators.]

3.2 Perfluorooctanesulfonic acid (PFOS)

See <u>Table 3.4</u>.

3.2.1 Mouse

Transplacental exposure

The tumorigenic effects of gestational exposure to PFOS were evaluated in C57BL/6J- $Apc^{Min/+}$ mice, a mouse model that develops intestinal tumours because of a mutation in the tumour suppressor gene, adenomatous polyposis cell (Apc) (Ngo et al., 2014). The wildtype ($Apc^{+/+}$) females were mated to heterozygotic males ($Apc^{Min/+}$). Wildtype dams were then treated by gavage with PFOS (purity, \geq 98%) at a dose of 0 (water vehicle), 0.01, 0.1, or 3.0 mg/kg per day on days 1–17 of gestation. Insufficient rates of pregnancy and littering rates and low F₁ survival were observed in the first experimental block – block 1, 0 (water vehicle), PFOS at 0.1 and 3.0 mg/kg bw per day; 104 mice (age, 7–8 weeks) – thus a second block was added that had a lower PFOS exposure (block 2, 0 (water vehicle), PFOS at 0.01 and 0.1 mg/kg bw per day; 100 dams (age, 9–10 weeks). The numbers of mice (Min/+) obtained in each dose group (both blocks combined) were 15, 10, 12, and 7 for males and 23, 6, 13, and 5 for females exposed in utero to vehicle (water), 0.01, 0.1, and 3.0 mg/kg bw of PFOS, respectively. Because of difficulty in ascertaining pregnancy status, exposure varied on days 14 to 17 of gestation. F_1 offspring were genotyped via polymerase chain reaction (PCR) using DNA collected from ear punches. Offspring were genotyped as wildtype $(Apc^{+/+})$ and heterozygous $(Apc^{Min/+})$, and only the *Min*/+ mice were used for the carcinogenesis study. Efforts were made to verify dosing level by measuring PFOS stability in the dosing solution. Furthermore, PFOS concentrations in tap water and feed used for the study were quantified, and levels of PFOS in the serum of mice in the vehicle control group were below the LOQ. Internal exposure was quantified in the serum (2) mice per time point) in dams on day 18 of gestation, and postnatal day 23 (block 1) or postnatal days 26–28 (block 2), and F_1 pups on postnatal days 25-27 (depending on block). The LOQ in serum was 0.05 ng/mL. Although minimal data were generated, they confirmed that there was internal exposure within dams and pups and that it increased with dose and decreased with time after dosing (day 18 of gestation versus postnatal day 23 in dams). Evaluation of intestinal tumorigenesis occurred at age 11 weeks in the F_1 *Min*/+ mice. The number, diameter, and localization of tumours in the small intestine and colon were measured by transillumination in an inverse light microscope. The reviewer scored lesions at 20× magnification and was blinded to treatment. Statistical analysis of incidence was conducted on both an individual and litter basis; furthermore, blocks were combined in the analysis if no consistent differences were found between blocks 1 and 2. None of the PFOS doses

Table 3.4 Studies of carcinogenicity in experimental animals exposed to PFOS					
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Results	Significance	Comments	
Full carcinogenicity Mouse, C57BL/6J- <i>Apc</i> ^{Min/+} (M) Day 1 of gestation 11 wk Ngo et al. (2014)	Oral administration (gavage) PFOS, ≥ 98% Water 0, 0.01, 0.1, 3.0 mg/kg bw per day 15, 10, 12, 7 15, 10, 12, 7	Small intestine (duodenu No significant increase in 15/15, 10/10, 12/12, 7/7 <i>Colon</i> No significant increase in 15/15, 3/3, 19/19, 0	m, jejunum or ileum) n tumour incidence NS n tumour incidence NS	 Principal strengths: males and females studied; multiple doses; analysed background levels of PFOS in feed and drinking-water; analysed internal doses of PFOS; tested stability of PFOS; blocks of PFOS administration were compared statistically. Principal limitations: small number of mice per group; short duration of study; histopathological examination not conducted; batch number of PFOS was not stated; varied exposure (14–17 d). Other comments: study of transplacental exposure; increase in the incidence of spontaneous tumours was studied in this mouse model; tumours of the small intestine were observed in all experimental groups of <i>Min</i>/+ mice, demonstrating 100% incidence for this end-point, as is usual in this mouse model; multiplicity was not increased by PFOS exposure. 	

Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Results	Significance	Comments
Full carcinogenicity Mouse, C57BL/6J- <i>Apc</i> ^{Min/+} (F) Day 1 of gestation 11 wk <u>Ngo et al. (2014)</u>	Oral administration (gavage) PFOS, ≥ 98% Water 0, 0.01, 0.1, 3.0 mg/kg bw per day 23, 6, 13, 5 23, 6, 13, 5	Small intestine (duodenum, No significant increase in t 23/23, 6/6, 13/13, 5/5 <i>Colon</i> No significant increase in t	, jejunum or ileum) umour incidence NS umour incidence	 Principal strengths: males and females studied; multiple doses; analysed background concentrations of PFOS in feed and drinking- water; analysed internal doses of PFOS; tested stability of PFOS; blocks of PFOS administration were compared statistically. Principal limitations: small and unbalanced number of mice per group; short duration of study; histopathological examination not conducted; batch number of PFOS was not reported. Other comments: study of transplacental exposure; increase in incidence of spontaneous tumours was studied in this mouse model; small intestinal tumours were observed in all experimental groups of Min/+ mice, demonstrating 100% incidence for this end-point, as is usual in this mouse model; multiplicity was not increased by PFOS exposure.
Full carcinogenicity Rat, Crl:CD (SD)IGS BR (M) Approx. 41 d 104 wk <u>Butenhoff et al. (2012b)</u>	Oral administration (feed) PFOS (potassium salt), 86.9% Acetone 0, 0.5, 2, 5, 20 ppm 7 d/wk 50, 50, 50, 50, 51 11, 11, 17, 25, 23	<i>Liver</i> Hepatocellular adenoma 0/60, 3/50, 3/50, 1/50, 7/60* <i>Thyroid gland</i> Follicular cell adenoma 3/60, 5/49, 4/50, 4/49, 4/59 Follicular cell carcinoma 3/60, 1/49, 1/50, 2/49, 1/59 Follicular cell adenoma or 6/60, 6/49, 5/50, 5/49, 5/59	P = 0.0276, Cochran– Armitage trend test *P = 0.046, Dunnett <i>t</i> -test NS NS carcinoma (combined)	<i>Principal strengths</i> : adequate number of rats used; randomly allocated in groups; males and females used; adequate duration. <i>Other comments</i> : "N at Start" removed rats from 4, 14, and 52 wk interim necropsy; note: Laboratory Report and Butenhoff include 52-wk interim animals in " <i>n</i> " for tumour incidence in the 0 and 20 ppm groups (e.g. 20 ppm <i>n</i> = 7/60 including 52 interim weeks vs <i>n</i> = 7/50 using only ≥ 53 wk); survival data in laboratory report were calculated using <i>n</i> = 50.

Table 3.4 (continued)

Table 3.4 (continued)

Route Agent tested, purity	Results	Significance	Comments
Dose(s) No. of animals at start No. of surviving animals			
	Pancreas Islet cell adenoma 4/60, 3/49, 4/50, 4/50, 4/60 Islet cell carcinoma 1/60, 2/49, 2/50, 5/50, 5/60	NS [P = 0.02, Cochran-Armitage trend test (not survival adjusted)] $[P = 0.13, poly-3 trend test(survival adjusted, withinterim animals)][P = 0.117, poly-3 trend test(survival adjusted, withoutinterim animals)]$	
	Islet cell adenoma or carcin	noma (combined)	
	5/60, 5/49, 6/50, 8/50, 9/60	NS	
Oral administration (feed) PFOS (potassium salt), 86.9% Acetone 0, 0.5, 2, 5, 20 ppm 7 d/wk 50, 50, 50, 50, 50 25, 15, 10, 17, 26	<i>Liver</i> Hepatocellular adenoma 0/60, 1/50, 1/49, 1/50, 5/60* Hepatocellular carcinoma 0/60, 0/50, 0/49, 0/50, 1/60 Hepatocellular adenoma on 0/60, 1/50, 1/49, 1/50, 6/60* <i>Thyroid gland</i> Follicular cell adenoma 0/60, 0/50, 0/49, 2/50,	P = 0.0153, Cochran- Armitage trend test * $P = 0.0386$, Dunnett <i>t</i> -test NS carcinoma (combined) P = 0.0057, Cochran- Armitage trend test * $P = 0.0204$, Dunnett <i>t</i> -test NS	Principal strengths: adequate number of rats used; randomly allocated in groups; males and females used; adequate duration. Other comments: inclusion of animals killed at the 52-wk interim time point in statistical analysis; experiment terminated at 103 wk for the group at 2 ppm; "N at Start" removed rats from 4, 14, and 52 wk interim necropsy; note: US EPA (2002) and Butenhoff included 52-wk interim animals in " <i>n</i> " for tumour incidence in the groups at 0 and 20 ppm (e.g. 20 ppm, n = 7/60 including 52 interim weeks vs $n = 7/50using only ≥ 53 wk); survival data in thislaboratory report were calculated using n = 50.$
	Oral administration (feed) PFOS (potassium salt), 86.9% Acetone 0, 0.5, 2, 5, 20 ppm 7 d/wk 50, 50, 50, 50, 50 25, 15, 10, 17, 26	No. of animals at start No. of surviving animals $\begin{array}{c} Pancreas\\ Islet cell adenoma\\ 4/60, 3/49, 4/50, 4/50, 4/60\\ Islet cell carcinoma\\ 1/60, 2/49, 2/50, 5/50, 5/60\\ \hline\end{array}$	No. of animals at start No. of surviving animals $\begin{array}{ c c c c c } \hline Pancreas\\ Islet cell adenoma\\ 4/60, 3/49, 4/50, 4/50, NS\\ 4/60\\ Islet cell carcinoma\\ 1/60, 2/49, 2/50, 5/50, [P = 0.02, Cochran- 5/60 Armitage trend test (not survival adjusted)] [P = 0.13, poly-3 trend test (survival adjusted, with interim animals)] [P = 0.117, poly-3 trend test (survival adjusted, without interim animals)] Islet cell adenoma or carcinoma (combined) 5/60, 5/49, 6/50, 8/50, 9/60 NS Oral administration (feed) PFOS (potassium salt), 86.9% Acetone 0, 0.5, 2, 5, 20 ppm 7 d/wk 50, 50, 50, 50, 50, 50 25, 15, 10, 17, 26 Hepatocellular carcinoma 0/60, 0/50, 0/49, 0/50, NS 1/60 Hepatocellular adenoma or carcinoma (combined) 0/60, 0/50, 0/49, 0/50, NS 1/60 Hepatocellular adenoma or carcinoma (combined) 0/60, 0/50, 0/49, 0/50, NS 1/60 Hepatocellular adenoma or carcinoma (combined) 0/60, 0/50, 0/49, 1/50, P = 0.0057, Cochran- 6/60* Armitage trend test *P = 0.0204, Dunnett t-test Thyroid gland Follicular cell adenoma 0/60, 0/50, 0/49, 2/50, NS$

428

Table 3.4 (continued)						
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Results	Significance	Comments		
Full carcinogenicity		Follicular cell carcinoma				
Rat, Crl:CD (SD)IGS (F) Approx. 41 d		0/60, 0/50, 0/49, 1/50, 0/60	NS			
104 wk		Follicular cell adenoma or	carcinoma (combined)			
(cont.)		0/60, 0/50, 0/49, 3/50*, 1/60	* <i>P</i> = 0.047, Dunnett <i>t</i> -test			
		Mammary gland				
		Fibroadenoma				
		20/60, 27/50*, 19/48, 24/50, 11/60	* <i>P</i> = 0.0337, Dunnett <i>t</i> -test			
		Adenoma				
		7/60, 6/50, 5/48, 7/50, 4/60	NS			
		Carcinoma				
		11/60, 12/50, 15/48, 11/50, 14/60	NS			
		Fibroadenoma or adenoma	a (combined)			
		23/60, 30/50*, 22/48, 26/50, 15/60	* <i>P</i> = 0.318, Dunnett <i>t</i> -test			
		Fibroadenoma, adenoma, o	or carcinoma (combined)			
		29/60, 36/50*, 31/48**, 29/50, 24/60	P = 0.0482, Cochran- Armitage trend test *P = 0.0474, Dunnett <i>t</i> -test **P = 0.0066, Dunnett <i>t</i> test			

Table 3.4 (continued)

Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Results	Significance	Comments
Full carcinogenicity Rat, Crl:CD (SD)IGS BR rats (M) Approx. 41 d 52 wk <u>Butenhoff et al. (2012b)</u>	Oral administration (feed) PFOS (potassium salt), 86.9% Acetone 0, 20 (recovery) ppm 0, 20 ppm, 7 d/wk for 52 wk followed by basal feed with acetone control for 52 wk 50, 40 11, 11	Liver Hepatocellular adenoma 0/60, 0/40 Thyroid gland Follicular cell adenoma 3/60, 9/39* Follicular cell carcinoma 3/60, 1/39 Follicular cell adenoma or 6/60, 10/39 Pancreas Islet cell adenoma 4/60, 11/40 Islet cell carcinoma 1/60, 3/40 Islet cell adenoma or carcin 5/60, 4/40	*P = 0.0280, Dunnett <i>t</i> -test NS carcinoma (combined) NS NS NS NS 10ma (combined)	Principal strengths: adequate number of rats used; randomly allocated in groups; males and females used; adequate duration. Other comments: "N at Start" removed rats from the interim necropsy at 4, 14, and 52 wk; note: US EPA (2002) and Butenhoff included 52-wk interim rats in "n" for tumour incidence in the groups at 0 and 20 ppm (e.g. 20 ppm n = 7/60 including 52 interim weeks vs $n = 7/50using only ≥ 53 wk); survival data in thislaboratory report were calculated using n = 50.$
Full carcinogenicity Rat, Crl:CD (SD)IGS (F) Approx. 41 d 52 wk <u>Butenhoff et al. (2012b)</u>	Oral administration (feed) PFOS (potassium salt), 86.9% Acetone 0, 20 (recovery) ppm 0, 20 ppm, 7 d/wk for 52 wk followed by basal feed with acetone control for 52 wk 50, 40 25, 19	<i>Liver</i> Hepatocellular adenoma 0/60, 2/40 Hepatocellular carcinoma 0/60, 0/40 Hepatocellular adenoma or 0/60, 2/40 <i>Thyroid gland</i> Follicular cell adenoma 0/60, 1/40 Follicular cell carcinoma 0/60, 0/40 Follicular cell adenoma or 0/60, 1/40	NS NS r carcinoma (combined) NS NS NS carcinoma (combined)	Principal strengths: adequate number of rats used, randomly allocated in groups, males and females used, adequate duration; well- conducted GLP study. Other comments: inclusion of 52-wk interim rats in statistical analysis; experiment terminated at 103 wk for the group at 2 ppm; "N at Start" removed rats from interim necropsy at 4, 14, and 52 wk; note: <u>US EPA</u> (2002) and Butenhoff included 52-wk interim rats in "n" for tumour incidence in the groups at 0 and 20 ppm (e.g. 20 ppm, $n = 7/60$ including 52 interim weeks vs $n = 7/50$ using only ≥ 53 wk); survival data in this laboratory report were calculated using $n = 50$.

Table 3.4 (continued)					
Study design Species, strain (sex) Age at start Duration Reference	Route Agent tested, purity Vehicle Dose(s) No. of animals at start No. of surviving animals	Results	Significance	Comments	
Full carcinogenicity Rat, Crl:CD (SD)IGS (F) Approx. 41 d 52 wk <u>Butenhoff et al. (2012b)</u> (cont.)		Mammary gland Fibroadenoma 20/60, 15/40 Adenoma 7/60, 4/40 Carcinoma 11/60, 4/40 Fibroadenoma or adenoma 23/60, 16/40 Fibroadenoma, adenoma, o 29/60, 17/40	NS NS (combined) or carcinoma (combined) NS		
Initiation-promotion (tested as promoter) Rainbow trout, (<i>Oncorhynchus mykiss</i>) (M, F) (combined) 15 wks (at initiation) 6 mo <u>Benninghoff et al. (2012)</u>	Oral administration (feed) PFOS, unspecified 0.01% EtOH Sham/control, Sham/ PFOS, AFB ₁ /control, AFB ₁ / PFOS ppm 5d/wk 250, 250, 250, 250 NR, NR, NR, NR	<i>Liver</i> Total tumours 0, 0, 1%, 13%*	* <i>P</i> < 0.01, logistic regression analysis (compared with AFB ₁ / control)	Principal strengths: adequate number of animals used; randomly allocated in groups. Principal limitations: males and females combined; only one dose; purity not reported. Other comments: survival and incidence number not reported, just the percentage incidence in each group; appeared that multiplicity increased somewhat, but numbers not provided; liver tumour diameter not increased.	
Initiation-promotion (tested as promoter) Zebrafish (<i>Danio rerio</i>), <i>Kras</i> ^{v12} transgenic (M, F) (combined) 90 d post fertilization 10 d Zhu et al. (2021)	Aqueous exposure PFOS (potassium salt), > 98% 0.1% DMSO (v/v) DMSO, DOX, PFOS, DOX + PFOS µg/L 7 d/wk 24, 24, 24, 24 NR, NR, NR	<i>Liver</i> Hepatocellular adenoma 0/6, 3/6, 0/6, 1/6 Hepatocellular carcinoma 0/6, 2/6, 0/6, 5/6*	[NS] [* <i>P</i> = 0.0076, Fisher exact test]	<i>Principal strengths</i> : liver histology conducted. <i>Principal limitations</i> : neoplasm incidences not reported; short duration; small number of animals per group; data combined for males and females; limited histopathological description. <i>Other comments</i> : incidence derived from a graph; histology conducted on 6 fish per treatment.	

AFB₁, aflatoxin B₁; bw, body weight; d, day(s); DMSO, dimethyl sulfoxide; DOX, doxycycline; EtOH, ethanol; F, female; M, male; mo, month(s); NR, not reported; NS, not significant; PFOS, perfluorooctanesulfonic acid; ppm, part per million; v/v, volume per volume; wk, week(s).

PFOA and PFOS

affected body weight, compared with dams and F_1 offspring in the vehicle control group.

There was a 100% incidence of tumours of the small intestine in heterozygous Min/+ mice, as is usual in this mouse model. Exposure to PFOS did not cause a significant increase in the number or multiplicity of small intestine tumours among male and female heterozygous mice. Tumour diameters were not significantly increased with increasing PFOS exposure within male mice, but there was an increase in tumour size in females in the groups treated with PFOS at 0.01 and 3.0 mg/kg bw compared with the control group (mice at the intermediate dose of 0.1 mg/kg bw were unaffected). Fewer tumours were observed in the colon than in the small intestine, as usual in this model, and no statistical differences were observed between PFOS-exposed groups and the control group with regard to incidence, number per mouse (multiplicity), or size of colon tumours. The localization of tumours along the small intestine and colon was not affected by treatment with PFOS. [The Working Group noted that this study used both sexes and multiple doses; PFOS stability was tested and the internal dose was analysed, as were background levels of PFOS in feed and drinking-water. However, no histological examination was performed.]

3.2.2 Rat

Oral administration (feed)

There was only one study on the carcinogenicity of PFOS in rodents. For this well-conducted study that complied with GLP, the data were available in the original laboratory report (<u>US EPA, 2002</u>) and as a manuscript published at a later date (<u>Butenhoff et al., 2012b</u>). In this study, PFOS (T-6295; purity, 86.9%) was administered in the feed at a concentration of 0 (control), 0.5, 2, 5, or 20 ppm to initial groups of 70, 60, 60, 60, 70, and 40 (20 ppm recovery group) Sprague-Dawley (Crl:CD(SD)IGS BR) male and female rats, respectively. Of these rats, 50 rats per group received control feed or PFOS for the full 2-year exposure (groups at 0-20 ppm). A recovery group of 40 males and 40 females was also included; rats in this group were treated with feed containing PFOS at 20 ppm for 52 weeks, and then allowed a recovery of 52 weeks. Controls received the control feed (basal feed with acetone as vehicle). Interim necropsies were carried out at 4, 14, and 53 weeks, when clinical chemistry, PFOS concentrations, and liver end-points (palmitoyl-CoA oxidase activity for PPARa activity, cell proliferation) were evaluated. PFOS consumption was 0.024, 0.098, 0.242, and 0.984 mg/kg bw per day for males and 0.029, 0.120, 0.299, and 1.251 mg/kg bw per day for females over the 104-week period, for the groups at 0.5, 2, 5, and 20 ppm, respectively. PFOS consumption for the recovery groups at 20 ppm was 1.144 and 1.385 mg/kg bw per day for males and females, respectively. Body weight was lower in the recovery groups of males and females at 20 ppm compared with that in the control group at the end of the dosing period, but body weights in the recovery group post-exposure tended back towards control values. At 104 weeks, survival of rats selected for the 2-year evaluation was: 11/50, 11/50, 17/50, 25/50, 23/51, and 11/40 in males and 25/50, 15/50, 10/50 (at 102 weeks), 17/50, 26/50, and 19/40 in females, for the groups at 0 (control), 0.5, 2, 5, 20 ppm, and 20 ppm (recovery), respectively. There was some indication of longer survival among male rats at 5 and 20 ppm compared with controls, possibly because of lower survival in the controls. [The Working Group noted that survival was lower in the male control groups compared with the treated groups, and the Peto statistical test was performed, taking differences in the survival into account.] One group of treated females (2 ppm) had decreased survival compared with controls (US EPA, 2002; also reported by Butenhoff et al., 2012b).

Rats from the 53-week evaluation of the control group and the group at 20 ppm (highest dose) and from the 104-week terminal necropsy were included to determine tumour incidence within the groups. For the evaluation of the incidence of non-neoplastic lesions, rats were included from the 14-, 53-, and 104-week necropsies.

In males, there was a significant positive trend (P = 0.0276, Cochran–Armitage trend test) in the incidence of hepatocellular adenoma, and the incidence - 0/60, 3/50, 3/50, 1/50, and 7/60 for the groups at 0 (control), 0.5, 2, 5, and 20 ppm, respectively – was significantly increased at 20 ppm (*P* = 0.0456, Dunnett *t*-test), compared with controls. Incidence in the recovery group at the highest dose (20 ppm) was similar to that in controls (0/60 versus 0/40). There was no significant increase in the incidence of thyroid follicular cell adenoma in continuously exposed male rats - 3/60, 5/49, 4/50, 4/49, and 4/59 for the groups at 0 (control), 0.5, 2, 5, and 20 ppm, respectively – however, there was a significant increase (P = 0.0280, Dunnett *t*-test) in the incidence of thyroid follicular cell adenoma in the recovery group at 20 ppm (9/39) compared with the control group (3/60).

In females, there was a significant positive trend (P = 0.0153, Cochran–Armitage trend test) in the incidence of hepatocellular adenoma -0/60, 1/50, 1/49, 1/50, and 5/60, for the groups at 0 (control), 0.5, 2, 5, and 20 ppm, respectively with the incidence being significantly increased (P = 0.0386, Dunnett t-test) in the group at 20 ppm compared with controls. There was a significant positive trend (P = 0.0057, Cochran–Armitage trend test) in the incidence of hepatocellular adenoma or carcinoma (combined), with the incidence - 0/60, 1/50, 1/49, 1/50, and 6/60 for the groups at 0 (control), 0.5, 2, 5, and 20 ppm, respectively – being significantly increased (P = 0.0204, Dunnett t-test) in the groups at 20 ppm compared with controls. In the group at 20 ppm, the incidence of hepatocellular carcinoma was 2% (1/60). For females, there was no difference between the recovery group (20 ppm) and the controls. There was a significant increase (P = 0.047, Dunnett t-test) in the incidence of thyroid follicular cell

adenoma or carcinoma (combined) in females in the group at 5 ppm - 0/60, 0/50, 0/49, 3/50, and 1/60 for the groups at 0 (control), 0.5, 2, 5, and 20 ppm, respectively. The incidence of mammary gland fibroadenoma - 20/60, 27/50, 19/48, 24/50, and 11/60 for the groups at 0 (control), 0.5, 2, 5, and 20 ppm, respectively - was significantly increased in the group at 0.5 ppm compared with controls (P = 0.0337, Dunnett *t*-test). The incidence of mammary gland fibroadenoma or adenoma (combined) was increased in the group at 0.5 ppm (P = 0.0318, Dunnett *t*-test). Not reported in Butenhoff et al. (2012b), but present in the laboratory report (US EPA, 2002), the incidence of mammary gland fibroadenoma, adenoma, or carcinoma (combined) - 29/60, 36/50, 31/48, 29/50, and 24/60 for the groups at 0 (control), 0.5, 2, 5, and 20 ppm, respectively – was significantly increased in the groups at 0.5 and 2.0 ppm (P = 0.0474 and P = 0.0066, Dunnett *t*-test, respectively). The increases in the incidence of combined mammary neoplasms within these groups were mostly because of increases in the incidence of fibroadenoma. There was no increase in the incidence of tumours in females in the recovery group (20 ppm) compared with the controls. [The Working Group noted that there was a tumour response for cancer of the thyroid gland in male (recovery group at 20 ppm) and female (continuous exposure at 5 ppm) rats compared with controls. However, the response was not dose-dependent in females, and no increase was observed in males at the highest dose and continuous exposure. Furthermore, mammary gland tumours were only observed at the lowest exposures and the association with PFOS was uncertain. The positive findings from this study were liver tumours in males and females.]

Regarding non-neoplastic lesions, these were observed in the liver in both males and females, primarily in the group at 20 ppm, and included hepatocellular hypertrophy, eosinophilic granular cytoplasm, hepatocellular pigmentation, individual hepatocyte necrosis, hepatocellular vacuolation, and cystic degeneration (males only). In females, there was an increase in the incidence of lymphohistiocytic infiltrate and pigmented macrophage infiltrate within the liver. Increased liver weight (absolute and relative) in males and hepatocyte hypertrophy in both males and females in the group at 20 ppm were observed at weeks 14 and 53. However, for both sexes, no significant increases in the incidence of liver cell proliferation were observed at weeks 4 and 14 (proliferating cell nuclear antigen, PCNA) or at week 53 (bromodeoxyuridine).

[The Working Group noted that this study used an adequate number of animals per group, both sexes, and an adequate duration of exposure. The reason for the inclusion of interim animals (control group and 20 ppm) from week 52 with animals from week 104 in the report by US EPA (2002) and Butenhoff et al. (2012b) was unclear, as the exposure was significantly different. A review of the histopathology results for males and females from week 52 showed that many of the rats had no neoplasms (26/39) or only had pituitary adenomas (11/39). The Working Group also noted the nearly significant positive trend in the incidence of pancreatic islet carcinoma, but there were no significant changes by pairwise comparison. The Working Group was uncertain of this finding and the association with PFOS exposure because hyperplasia, adenomas, and the combination of adenoma or carcinoma were also not significant in males, and no pancreatic islet cell effect in female rats was observed. The Working Group also noted that the pancreatic islet cell tumours were not reported in **Butenhoff** et al. (2012b) but were reported in US EPA (2002). The Working Group conducted survival-adjusted statistical analyses on the data for pancreatic islet cell carcinoma, because it was noted that survival in controls was low, using the poly-3 test method. There was no significant difference in the trend test results in analyses including the 53-week interim animals (P = 0.130) or excluding

the 53-week interim animals (P = 0.117), and, similarly to in the report by <u>US EPA (2002)</u>, no significant pairwise comparisons with incidence in the controls were found.]

3.2.3 Fish

Initiation-promotion studies

Benninghoff et al. (2012) used Mount Shasta rainbow trout (Oncorhynchus mykiss) to evaluate PFOS promoter activity by initiating 1000 fry (age, 15 weeks) for 30 minutes with either aflatoxin B_1 (AFB₁) at a concentration of 10 ppb or a sham control of 0.01% ethanol. These initiated fry were then treated with a diet containing PFOS (purity not reported; analytical grade) at a concentration of 100 ppm (equivalent to 2.5 mg/kg bw per day). In all, there were four exposure groups, each containing 250 fish: sham/control, sham/PFOS, AFB₁/control, and AFB₁/PFOS. The dietary concentration of PFOS was selected on the basis of a pilot study in which the concentration used, 2000 ppm, resulted in high mortality. The PFOS diet was provided for 6 months; during this time, the diet was prepared on a monthly basis and kept frozen at -20 °C before use. Body weight was decreased after AFB₁/PFOS exposure compared with sham/control (P < 0.05, logistic regression analysis), whereas body weight in sham/control and sham/PFOS exposure groups was not significantly different. Liver weights (absolute and relative) were increased after treatment with PFOS. with and without initiation. The fish were killed at 12.5 months post-spawn, and examination of tumours was performed.

After initiation with AFB_1 at 10 ppb and in the absence of subsequent PFOS exposure, there was a 1% induction of liver tumours. In the sham controls with and without PFOS exposure, there was no induction of liver tumours. After initiation followed by PFOS exposure, there was a significant increase in the incidence of liver tumours, compared with controls (P < 0.01, logistic regression analysis; 13% compared with 1%). Tumour multiplicity and tumour size after PFOS exposure were not different from those for the controls. Mixed carcinoma was the most prevalent tumour type after AFB₁/PFOS exposure and AFB₁/control. [The Working Group noted that an adequate number of fish per group was used in this study; however, data were combined for males and females, only one dose of PFOS was tested, and PFOS purity was not reported.]

In a study by Zhu et al. (2021), male $Kras^{V12}$ transgenic zebrafish (Danio rerio) (age, 90 days post-fertilization), in which hepatocellular carcinomas can be initiated via doxycycline (DOX)induced expression of the Kras G12V oncogene in the liver (see <u>Chew et al., 2014</u>), were used to determine whether PFOS alone or DOX + PFOS could initiate or promote hepatocellular carcinoma, respectively. Adult transgenic zebrafish were immersed in PFOS (purity, > 98%) with or without DOX, in the dark for 10 days, to avoid photodegradation of the DOX. Four exposure groups were evaluated: 0.01% DMSO (control), DOX (20 mg/L), PFOS (500 μ g/L), and DOX + PFOS (20 mg/L plus 500 μ g/L, respectively). The selection of PFOS exposures was made on the basis of a short-term (4 day) study in zebrafish larvae exposed to PFOS at concentrations of 50, 100, 200, 500, and 1000 µg/L. After the 10-day exposure, the fish were killed, and livers were evaluated. Three exposure replicates were conducted, with 8 fish per replicate. From these 24 fish, livers from 6 fish were used for histological analysis and livers from 3 fish were used for transcriptomics analysis. [The Working Group noted that PFOS concentrations were not verified either in the aqueous exposure or in the internal dose.]

The hepatosomatic index (liver weight relative to body weight) in adult zebrafish was increased after treatment with DOX or DOX + PFOS, compared with DMSO (both P < 0.05), but not with PFOS alone, and was increased after treatment with DOX + PFOS versus DOX (P < 0.05). Liver size (mm²), and fluorescence intensity (resulting from expression of a liver-specific enhanced green fluorescent protein) were increased in the group exposed to DOX + PFOS compared with the group exposed to DOX alone (both P < 0.05). The incidence of hepatocellular carcinoma was higher in the group exposed to DOX + PFOS (5/6) than in the group exposed to DOX alone (2/6), and no hepatocellular carcinomas were observed in the groups exposed to PFOS alone (0/6) or DMSO (0/6). [The Working Group noted that a small number of fish were used (n = 6) for histopathological evaluation of liver tumours. Statistical analysis by the Working Group showed no significant difference between the DOX group (2/6; 40%) and the DOX + PFOS group (5/6; 83%) [P = 0.1212, Fisher exact test]. The duration of the study appeared to be based on a 3-month study in which the establishing of this transgenic line was reported, and in which increasing mortality was observed shortly after induction or initiation in a 3-month study.]

3.3 Evidence synthesis for cancer in experimental animals

3.3.1 PFOA

The carcinogenicity of PFOA has been assessed in two well-conducted GLP studies, one in male and female Sprague-Dawley rats treated by oral administration (in the feed) in a combination of F_0 (in utero and lactation) and F_1 (postweaning) exposure (<u>NTP, 2020</u>) and the other in male and female Sprague-Dawley rats treated by oral administration (in the feed) (US EPA, 1987); also reported by Butenhoff et al., 2012a) [histological re-analysis by Hardisty et al. (2010) (mammary gland) and Caverly Rae et al., 2014 (pancreas)]. The carcinogenicity of PFOA was also evaluated in studies that did not comply with GLP. Specifically, these were studies of oral administration (feed) in male Sprague-Dawley rats (Biegel et al., 2001); oral administration (gavage) in male and female C57BL/6J-Apc^{Min/+}

mice (Ngo et al., 2014); and studies in female CD-1 mice, female 129/Sv wildtype, and 129/Sv PPAR α -knockout mice (Filgo et al., 2015). In addition, there were six initiation–promotion studies of oral administration (feed) in male Wistar rats (Abdellatif et al., 1990, 1991; also reported by Nilsson et al., 1991), in male and female rainbow trout (Tilton et al., 2008; Benninghoff et al., 2012); of oral administration (drinking-water) in male and female *LSL-Kras^{G12D};Pdx-1 Cre* (KC) transgenic mice (Kamendulis et al., 2022); and of oral administration (gavage) in female Sprague-Dawley rats (Su et al., 2022).

In the dietary study that complied with GLP in F₁ male and female Sprague-Dawley rats (<u>NTP, 2020</u>), a significant positive trend in the incidence of hepatocellular adenoma (includes multiple) was observed in males, and the incidence was significantly increased in both the F_0 exposed and F_0 unexposed groups. There was a positive trend in the incidence of hepatocellular carcinoma only in F_1 males with F_0 exposure. There was a significant positive trend in the incidence of hepatocellular adenoma or carcinoma (combined) in F_1 males in both the F_0 exposed and F₀ unexposed groups, and the incidence was significantly increased in both groups. There was a significant positive trend in the incidence of acinar cell adenoma of the pancreas (includes multiple) in F_1 males in both the F_0 exposed and F_0 unexposed groups, and the incidence was significantly increased in both groups. There was a significant positive trend in the incidence of acinar cell adenoma or adenocarcinoma (combined) of the pancreas in F_1 males in both the F₀ exposed and F₀ unexposed groups, and the incidence was significantly increased in both groups. In female rats, there was a significant increase in the incidence of adenocarcinoma of the uterus in the group with F_1 exposure at the highest dose without F_{0} exposure. In female rats, there was a significant positive trend in the incidence of pancreatic acinar cell adenoma or adenocarcinoma (combined) in F_1 rats with F_0 exposure. [The Working Group noted that a low incidence of pancreatic acinar cell adenoma or carcinoma (combined) and of pancreatic acinus hyperplasia was observed in females; these rare lesions in female rats were considered to be associated with PFOA exposure.]

In another dietary study that complied with GLP in male and female Sprague-Dawley rats (Butenhoff et al., 2012a), there was a significant positive trend in the incidence of testicular Leydig cell adenoma, and the incidence was significantly increased at the highest dose in males. In females, there was a significant positive trend in the incidence of fibroadenoma of the mammary gland, and the incidence was significantly increased at both doses. A pathology working group was convened by the study sponsor(s) to review the original slides of the mammary glands from the study by <u>US EPA (1987)</u>, a study that was also reported by **Butenhoff et al.** (2012a), and concluded that PFOA did not induce neoplasms of the mammary gland in those studies (Hardisty et al., 2010; Butenhoff et al., 2012a). [The Working Group agreed with the conclusion of <u>Hardisty</u> et al. (2010) that PFOA did not induce neoplasms of the mammary gland.]

In the single-dose dietary study in male Sprague-Dawley rats (<u>Biegel et al., 2001</u>), there was a significant increase in the incidence of hepatocellular adenoma, and of hepatocellular adenoma or carcinoma (combined). There was a significant increase in the incidence of testicular Leydig cell adenoma. There were significant increases in the incidence of pancreatic acinar cell adenoma, and in the incidence of pancreatic acinar cell adenoma or carcinoma (combined).

In a study of oral administration (gavage) in female CD-1 mice, there was a significant increase in the incidence of hepatocellular adenoma at the intermediate dose only, and a significant positive trend in the incidence of liver haemangiosarcoma (Filgo et al., 2015). No increase in the incidence of hepatic neoplasms was observed in treated female 129/Sv wildtype and PPARα-knockout mice. [The Working Group considered the liver haemangiosarcomas to be possibly associated with PFOA exposure; however, the Working Group acknowledged that 16–28% of unscheduled deaths in all groups were not examined. The Working Group was uncertain regarding the biological significance of the hepatocellular adenoma results.]

In studies of oral administration (feed) and initiation-promotion in rats (Abdellatif et al., <u>1990, 1991; also reported by Nilsson et al., 1991)</u> and fish (Tilton et al., 2008; Benninghoff et al., 2012), the promoting activity of PFOA was investigated. There was a significant increase in the incidence of total tumours of the liver at the intermediate and higher doses (Abdellatif et al., 1990) and a significant increase in the incidence of hepatocellular carcinoma at the highest dose (Abdellatif et al., 1991) in male Wistar rats. There was a significant increase in the incidence of total tumours of the liver (malignant and benign) (Tilton et al., 2008) and of total tumours of the liver Benninghoff et al. (2012) in male and female rainbow trout. [The Working Group noted that the proportion of malignant tumours (as a percentage of the total liver tumours) was higher than that of benign tumours. The Working Group also noted that PFOA acted as a cancer promoter in these studies.]

In a study of oral administration (gavage) in male and female C57BL/6J- $Apc^{Min/+}$ mice (Ngo et al., 2014), and a promotion study of oral administration (drinking-water) in male and female LSL- $Kras^{G12D}$;Pdx-1 Cre (KC) transgenic mice (Kamendulis et al., 2022) there was no significant increase in the incidence of tumours. In a promotion study of oral administration (gavage) in female Sprague-Dawley rats, no significant increase in tumour incidence was found (Su et al., 2022). [The Working Group noted that this negative result may have been due to the high initiating dose.] [The Working Group noted that an effect in the liver and pancreas in male rats was observed consistently throughout the studies, while an effect in the pancreas in female rats was only observed in the NTP study. This effect in the female pancreas is possibly a result of the higher exposure to PFOA in females than in males, compensating for the faster elimination in females than in males.]

3.3.2 PFOS

The carcinogenicity of PFOS has been assessed in one well-conducted study that complied with GLP in male and female Sprague-Dawley rats treated by oral administration (feed) (<u>US EPA, 2002</u>; also reported by <u>Butenhoff et al.,</u> 2012b). The carcinogenicity of PFOS was also evaluated in three studies that did not comply with GLP. One study was of oral administration (gavage) in male and female C57BL/6J-*Apc*^{Min/+} mice (<u>Ngo et al., 2014</u>). Two initiation–promotion studies were of oral administration (feed) in male and female rainbow trout (*Oncorhynchus mykiss*) (<u>Benninghoff et al., 2012</u>) and of aqueous exposure in male and female *Kras*^{V12} transgenic zebrafish (<u>Zhu et al., 2021</u>).

In a dietary study in male and female Sprague-Dawley rats (Butenhoff et al., 2012b), there was a significant positive trend in the incidence of hepatocellular adenoma in males, and the incidence was significantly increased at the highest dose. In females, there was a significant positive trend in the incidence of hepatocellular adenoma, and hepatocellular adenoma or carcinoma (combined), with the incidence being significantly increased at the highest dose. There was a significant increase in the incidence of thyroid follicular cell adenoma or carcinoma (combined) at the higher intermediate dose in females. There was a significant increase in the incidence of fibroadenoma of the mammary gland and of fibroadenoma or adenoma (combined) at the lowest dose. There was significant positive trend in the incidence of fibroadenoma, adenoma or carcinoma (combined) of the mammary gland, with the incidence being significantly increased

at the two lower doses. In the recovery group of males, there was a significant increase in the incidence of thyroid follicular cell adenoma. [The Working Group noted that the liver was a target organ for PFOS in both male and female rats. The Working Group also noted that the association between PFOS exposure and the incidence of thyroid follicular cell tumours and mammary gland tumours was uncertain.]

In an initiation-promotion study of oral administration (in the feed) of PFOS, there was a significant increase in the incidence of total liver tumours in male and female rainbow trout (*Oncorhynchus mykiss*) at the highest dose (Benninghoff et al., 2012). [The Working Group noted that this study provided evidence that PFOS can be a cancer promoter in a rainbow trout model.]

In a study of oral administration (gavage) in male and female C57BL/6J- $Apc^{Min/+}$ mice (Ngo et al., 2014) and of aqueous exposure in male and female $Kras^{V12}$ transgenic zebrafish (Zhu et al., 2021), there was no significant increase in the incidence of tumours.

References

- Abbott BD, Wolf CJ, Schmid JE, Das KP, Zehr RD, Helfant L, et al. (2007). Perfluorooctanoic acid induced developmental toxicity in the mouse is dependent on expression of peroxisome proliferator activated receptor-alpha. *Toxicol Sci.* 98(2):571–81. doi:<u>10.1093/toxsci/ kfm110</u> PMID:<u>17488742</u>
- Abdellatif AG, Préat V, Taper HS, Roberfroid M (1991). The modulation of rat liver carcinogenesis by perfluorooctanoic acid, a peroxisome proliferator. *Toxicol Appl Pharmacol.* 111(3):530–7. doi:<u>10.1016/0041-</u> <u>008X(91)90257-F</u> PMID:<u>1684073</u>
- Abdellatif AG, Préat V, Vamecq J, Nilsson R, Roberfroid M (1990). Peroxisome proliferation and modulation of rat liver carcinogenesis by 2,4-dichlorophenoxyacetic acid, 2,4,5-trichlorophenoxyacetic acid, perfluorooctanoic acid and nafenopin. *Carcinogenesis*. 11(11):1899–902. doi:10.1093/carcin/11.11.1899 PMID:2225320
- Andreassen A, Møllersen L, Vikse R, Steffensen I-L, Mikalsen A, Paulsen JE, et al. (2002). One dose of 2-amino-1-methyl-6-phenylimidazo[4,5-*b*]pyridine

(PhIP) or 2-amino-3-methylimidazo[4,5-f]quinoline (IQ) induces tumours in Min/+ mice by truncation mutations or LOH in the Apc gene. *Mutat Res.* 517(1-2):157-66. doi:10.1016/S1383-5718(02)00065-7 PMID:12034317

- Benninghoff AD, Orner GA, Buchner CH, Hendricks JD, Duffy AM, Williams DE (2012). Promotion of hepatocarcinogenesis by perfluoroalkyl acids in rainbow trout. *Toxicol Sci.* 125(1):69–78. doi:<u>10.1093/toxsci/ kfr267</u> PMID:<u>21984479</u>
- Biegel LB, Hurtt ME, Frame SR, O'Connor JC, Cook JC (2001). Mechanisms of extrahepatic tumor induction by peroxisome proliferators in male CD rats. *Toxicol Sci.* 60(1):44–55. doi:10.1093/toxsci/60.1.44 PMID:11222872
- Butenhoff JL, Chang S-C, Olsen GW, Thomford PJ (2012b). Chronic dietary toxicity and carcinogenicity study with potassium perfluorooctanesulfonate in Sprague-Dawley rats. *Toxicology*. 293(1–3):1–15. doi:<u>10.1016/j.</u> tox.2012.01.003 PMID:<u>22266392</u>
- Butenhoff JL, Kennedy GL Jr, Chang S-C, Olsen GW (2012a). Chronic dietary toxicity and carcinogenicity study with ammonium perfluorooctanoate in Sprague-Dawley rats. *Toxicology*. 298(1–3):1–13. doi:10.1016/j. tox.2012.04.001 PMID:22531602
- Caverly Rae JM, Frame SR, Kennedy GL, Butenhoff JL, Chang S-C (2014). Pathology review of proliferative lesions of the exocrine pancreas in two chronic feeding studies in rats with ammonium perfluorooctanoate. *Toxicol Rep.* 1:85–91. doi:<u>10.1016/j.toxrep.2014.04.005</u> PMID:<u>28962229</u>
- Chew TW, Liu XJ, Liu L, Spitsbergen JM, Gong Z, Low BC (2014). Crosstalk of Ras and Rho: activation of RhoA abates Kras-induced liver tumorigenesis in transgenic zebrafish models. *Oncogene*. 33(21):2717–27. doi:10.1038/onc.2013.240 PMID:23812423
- Filgo AJ, Quist EM, Hoenerhoff MJ, Brix AE, Kissling GE, Fenton SE (2015). Perfluorooctanoic acid (PFOA)induced liver lesions in two strains of mice following developmental exposures: PPARα is not required. *Toxicol Pathol.* 43(4):558–68. doi:<u>10.1177/0192623314558463</u> PMID:25398757
- Giknis MLA, Clifford CB (2010). Spontaneous neoplastic lesions in the Crl:CD-1(ICR) mouse in control groups from 18 month to 2 year studies – March 2010. Charles River Laboratories. Available from: <u>https://www.criver. com/sites/default/files/resources/doc_a/Spontaneous</u> <u>NeoplasticLesionsintheCrlCD-1ICRMouseinContro</u> <u>lGroupsfrom18Monthto2YearStudies%E2%80%94M</u> <u>arch2010.pdf</u>.
- Hardisty JF, Willson GA, Brown WR, McConnell EE, Frame SR, Gaylor DW, et al. (2010). Pathology Working Group review and evaluation of proliferative lesions of mammary gland tissues in female rats fed ammonium perfluorooctanoate (APFO) in

the diet for 2 years. *Drug Chem Toxicol*. 33(2):131–7. doi:<u>10.3109/01480541003667610</u> PMID:<u>20307141</u>

- Hendricks JD, Meyers TR, Shelton DW (1984). Histological progression of hepatic neoplasia in rainbow trout (*Salmo gairdneri*). *Natl Cancer Inst Monogr*. 65:321–36. PMID:<u>6087143</u>
- Hezel AF, Kimmelman AC, Stanger BZ, Bardeesy N, Depinho RA (2006). Genetics and biology of pancreatic ductal adenocarcinoma. *Genes Dev.* 20(10):1218–49. doi:10.1101/gad.1415606 PMID:16702400
- Hingorani SR, Petricoin EF 3rd, Maitra A, Rajapakse V, King C, Jacobetz MA, et al. (2003). Preinvasive and invasive ductal pancreatic cancer and its early detection in the mouse. *Cancer Cell*. 4(6):437–50. doi:10.1016/ S1535-6108(03)00309-X PMID:14706336
- IARC (2016). Some chemicals used as solvents and in polymer manufacture. *IARC Monogr Eval Carcinog Risks Hum.* 110:1–276. Available from: <u>https://</u> <u>publications.iarc.who.int/547</u> PMID:<u>31829531</u>
- Kamendulis LM, Hocevar JM, Stephens M, Sandusky GE, Hocevar BA (2022). Exposure to perfluorooctanoic acid leads to promotion of pancreatic cancer. *Carcinogenesis.* 43(5):469–78. doi:<u>10.1093/carcin/ bgac005</u> PMID:<u>35022659</u>
- Macon MB, Villanueva LR, Tatum-Gibbs K, Zehr RD, Strynar MJ, Stanko JP, et al. (2011). Prenatal perfluorooctanoic acid exposure in CD-1 mice: low-dose developmental effects and internal dosimetry. *Toxicol Sci.* 122(1):134–45. doi:<u>10.1093/toxsci/kfr076</u> PMID:<u>21482639</u>
- Moser AR, Pitot HC, Dove WF (1990). A dominant mutation that predisposes to multiple intestinal neoplasia in the mouse. *Science*. 247(4940):322–4. doi:<u>10.1126/</u> <u>science.2296722</u> PMID:<u>2296722</u>
- Ngo HT, Hetland RB, Sabaredzovic A, Haug LS, Steffensen I-L (2014). In utero exposure to perfluorooctanoate (PFOA) or perfluorooctane sulfonate (PFOS) did not increase body weight or intestinal tumorigenesis in multiple intestinal neoplasia (*Min*/+) mice. *Environ Res.* 132:251–63. doi:<u>10.1016/j.envres.2014.03.033</u> PMID:<u>24834819</u>
- Nilsson R, Beije B, Préat V, Erixon K, Ramel C (1991). On the mechanism of the hepatocarcinogenicity of peroxisome proliferators. *Chem Biol Interact*. 78(2):235–50. doi:<u>10.1016/0009-2797(91)90017-2</u> PMID:<u>2040027</u>

- NTP (2020). NTP Technical Report on the toxicology and carcinogenesis studies of perfluorooctanoic acid (CASRN 335-67-1) administered in feed to Sprague Dawley (Hsd:Sprague Dawley[®] SD[®]) rats (revised). *Natl Toxicol Program Tech Rep Ser.* (598) Revised February 2023.
- Su Y, Santucci-Pereira J, Dang NM, Kanefsky J, Rahulkannan V, Hillegass M, et al. (2022). Effects of pubertal exposure to butyl benzyl phthalate, perfluorooctanoic acid, and zeranol on mammary gland development and tumorigenesis in rats. *Int J Mol Sci.* 23(3):1398. doi:10.3390/ijms23031398 PMID:35163327
- Thoolen B, Maronpot RR, Harada T, Nyska A, Rousseaux C, Nolte T, et al. (2010). Proliferative and nonproliferative lesions of the rat and mouse hepatobiliary system. *Toxicol Pathol.* 38(7 Suppl):5S-81S. doi:10.1177/0192623310386499 PMID:21191096
- Tilton SC, Orner GA, Benninghoff AD, Carpenter HM, Hendricks JD, Pereira CB, et al. (2008). Genomic profiling reveals an alternate mechanism for hepatic tumor promotion by perfluorooctanoic acid in rainbow trout. *Environ Health Perspect*. 116(8):1047– 55. doi:10.1289/ehp.11190 PMID:18709148
- US EPA (1987). Two-year oral (diet) toxicity/oncogenicity study of fluorochemical FC-143 in rats. Prepared by Sibinski LJ. Riker Experiment No. 0281CR0012, Riker Laboratories Inc/3M Company. Washington (DC), USA: United States Environmental Toxicology Program. Available from: <u>https://hero.epa.gov/hero/ index.cfm/reference/details/reference_id/5432403</u>.
- US EPA (2002). 104-Week dietary chronic toxicity and carcinogenicity study with perfluorooctane sulfonic acid potassium salt (PFOS; T-6295) in rats. Technical report. Prepared by Thomford PJ. Prepared by Thomford PJ for Covance Laboratories Inc. Study No. 6329-183. Washington (DC), USA: United States Environmental Toxicology Program. Available from: https://hero.epa.gov/hero/index.cfm/reference/details/reference_id/5029075.
- Zhu Y, Yang D, Duan X, Zhang Y, Chen D, Gong Z, et al. (2021). Perfluorooctane sulfonate promotes doxycycline-induced liver tumor progression in male *Kras*^{v12} transgenic zebrafish. *Environ Res.* 196:110962. doi:10.1016/j.envres.2021.110962 PMID:33675800

4. MECHANISTIC EVIDENCE

4.1 Absorption, distribution, metabolism, and excretion

Perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS) are characterized by long half-lives in humans (years) and much shorter half-lives in experimental animals (days or weeks). Numerous toxicokinetic (TK) studies in laboratory animals were available, and several physiologically based pharmacokinetic (PBPK) models of PFOA and PFOS have been developed. However, human TK data are still scarce in comparison with those from experimental animals, and the mechanistic bases for the observed TK differences among species remain to be fully elucidated.

4.1.1 Humans

- (a) Exposed humans
- (i) Absorption

Exposure to PFOA and PFOS in non-occupational settings occurs mainly via the oral route. After a single oral dose of ammonium perfluorooctanoate (APFO, 50–1200 mg, as a gelatin capsule) was administered during a clinical trial of patients with cancer, the serum concentrations reached a plateau after approximately 600 hours for the highest dose, but they were still increasing after 864 hours, the latest time point at which they were assessed, for lower doses (<u>Convertino</u> et al., 2018). [The Working Group noted that these doses were much higher than estimated human environmental exposures and that there were no healthy controls in the study. The Working Group also noted that the salt and protonated forms used in this study were comparable to those used in other studies described in this section.] No direct estimates of the oral absorption efficiency of PFOA or PFOS were available for humans. The oral absorption efficiency is assumed to be close to 100%, according to TK models that simulate the human serum concentrations of PFOA, on the basis of comparisons with data obtained from experimental animals. PFOA and PFOS are excreted in the bile (Fujii et al., 2015) and are reabsorbed from the gastrointestinal tract into the enterohepatic circulation; a high absorption efficiency would be required to be consistent with the observed long half-lives of these compounds (Harada et al., 2007).

The inhalation and dermal routes can significantly contribute to PFOA and PFOS exposure in occupational settings, such as in firefighters (Christensen and Calkins, 2023; Mazumder et al., 2023). In a transdermal absorption study of a single volunteer, whole-skin application of [¹³C]PFOA mixed into sunscreen resulted in the maximum serum concentration 22 days post-application, and the absorption efficiency was estimated to be 1.6% of the administered dose (Abraham and Monien, 2022). No data regarding the absorption of PFOA or PFOS by inhalation were available to the Working Group.

(ii) Distribution

Few data were available regarding the distribution of PFOA and PFOS, and most of the existing data were obtained from postmortem samples. The liver, blood, and lungs appear to be important sites of accumulation for both compounds, and there are generally lower levels of accumulation in the kidneys, bone, muscle, brain, and fat (Olsen et al., 2003a; Maestri et al., 2006; Pérez et al., 2013; Yeung et al., 2013a; Fàbrega et al., 2014; Mamsen et al., 2019; Di Nisio et al., 2022). In one study, PFOA concentrations in the liver were found to be similar to those in the blood, whereas PFOS concentrations in the liver were higher by approximately 2.7-fold (<u>Maestri et al., 2006</u>). In contrast, accumulation in the liver of laboratory animals appeared to be more marked, as discussed below.

In two other studies, PFOA and PFOS concentrations in human follicular fluid were found to be comparable to those in the serum (Petro et al., 2014; McCoy et al., 2017). PFOA and PFOS have been detected in the thyroid at concentrations just below those in the serum (Pirali et al., 2009). PFOA and PFOS have also been found in semen and cerebrospinal fluid, but at much lower levels than in serum (Raymer et al., 2012; Wang et al., 2018a; Di Nisio et al., 2019). In addition, PFOA and PFOS have been detected in the amniotic fluid, placenta, cord blood, and embryonic tissues at lower concentrations than those in the maternal serum (Zhang et al., 2013a; Mamsen et al., 2019).

The elimination of PFOA and PFOS from human serum has been demonstrated to follow first-order kinetics, and therefore the overall distribution can be characterized using the volume of distribution (V_d) (Harada et al., 2005; Olsen et al., 2007; Russell et al., 2015). Based on a Bayesian TK analysis of data from 13 studies in which PFOA and PFOS serum levels, PFOA and PFOS concentrations in drinking-water, and background PFOA and PFOS exposures were considered, the $V_{\rm d}$ for a one-compartment TK model was estimated to be 0.43 L/kg and 0.32 L/kg for PFOA and PFOS, respectively (Chiu et al., <u>2022</u>). [The Working Group noted that these values are higher than some of the previously proposed PFOA and PFOS V_d estimates, but they appeared to be based on the largest dataset, analysed using the best available methods.] The previously widely used PFOA V_{d} estimate of 0.17 L/kg was derived from a single dataset using a similar approach (Thompson et al., 2010). This value was used as the initial estimate for the PFOA V_{d} in the Bayesian analysis performed by Chiu et al. (2022).

On the basis of their structure and physicochemical properties, PFOA and PFOS are unlikely to cross cellular membranes directly. Various membrane transporters are thought to mediate their transmembrane transport, as described in more detail in the section regarding in vitro studies below.

Blood is an important compartment for the accumulation of PFOA and PFOS. PFOA and PFOS in the blood primarily distribute to the serum, but a significant fraction also partitions to blood cells, indicating that the use of a default factor of two to account for the volume of plasma in whole blood (a common practice when no prior knowledge about chemical distribution between blood fractions is available) may not be an appropriate method for the blood-toserum concentration conversion (Jin et al., 2016; Poothong et al., 2017; Liu et al., 2023a). Indeed, in two studies in which the mass fraction in plasma $(F_{\rm p})$ was measured, it was 0.7–0.8 for PFOA and 0.8-0.85 for PFOS (Jin et al., 2016; Liu et al., <u>2023a</u>). [The Working Group noted that F_p would equal 0.5 if the compound distributed to plasma.] In fractionated plasma collected from healthy volunteers (n = 4), PFOA and PFOS were bound to albumin, with little affinity for lipoproteins (Forsthuber et al., 2020).

An analysis of biomonitoring data using PBPK modelling showed that serum concentrations of PFOA and PFOS were lower in women of reproductive age than in age-matched men. This was thought to be the result of menstrual blood loss and placental and lactational transfer associated with pregnancy and childbirth (Wong et al., 2014; Gomis et al., 2017).

PFOA and PFOS cross the placenta and can also be transferred to infants via breast milk. In a review of the available studies, it was calculated that the placental transfer (median and range), defined as the ratio of PFAS fetal (cord blood) concentration to the maternal serum concentration, was 0.79 (0.60-1.5) for PFOA and 0.37 (0.29–0.56) for PFOS (<u>Pizzurro et al., 2019</u>). PFOS concentrations in maternal serum during the first trimester were significantly higher than those during the second and third trimesters, but there were no significant differences in PFOA concentration between trimesters (Mamsen et al., 2019). In another study, PFOA and PFOS concentrations both significantly decreased by approximately 40% between 16 weeks of gestation and time of delivery (Kato et al., 2014). [The Working Group noted that decreases in PFOA and PFOS levels in pregnancy were probably because of blood volume expansion and increase in the glomerular filtration rate (GFR) and, to a minor extent, transfer to the fetus. Differences among studies could be caused by differences in PFOA levels and experimental variation.]

Lactational transfer (median and range of the values obtained from the available studies), defined as the ratio of the breast milk to the maternal serum concentration, was calculated to be 0.04 (0.03–0.12) for PFOA and 0.01 (0.01–0.03) for PFOS (Pizzurro et al., 2019). Lactational PFOA transfer probably accounts for the higher (2.7–4.6-fold) PFOA serum concentrations in infants aged 2–6 months than in mothers. However, PFOS concentrations were similar in infants and mothers (Fromme et al., 2010; Gyllenhammar et al., 2018). By age 18 months, the mean PFOA concentration in the infants' serum was similar to that in the mothers' serum (Højsager et al., 2022), probably because of the cessation of breastfeeding and the growth-dependent dilution of PFOA according to body burden. There was very high inter-individual variability in the time-course profiles for PFOA and PFOS concentrations in infants (Mogensen et al., 2015a). Serum PFOA and PFOS concentrations in children were similar at age 2, 4, and 6 years (Kim et al., 2020). On the basis of the available data, Goeden et al. (2019) developed a TK model that predicts serum PFOA concentrations in people of all ages, including the very young.

The available evidence suggested that PFOA and PFOS undergo enterohepatic circulation. Although these compounds were not detected (< 0.5 ng/g) in the faeces of a single participant (serum concentrations 6.8 and 26.0 ng/g for PFOA and PFOS, respectively), the addition of cholestyramine, which inhibits the reabsorption of bile acids and thereby increases their excretion, to the diet for 1 week (4 g, three times per day) resulted in detectable faecal PFOA (0.96 ng/g) and PFOS (9.06 ng/g) (Genuis et al., 2010). Harada et al. (2007) measured the PFOA and PFOS concentrations in bile (4 participants) and, using the total and urinary clearances of these compounds, calculated their biliary reabsorption rates (the proportions reabsorbed) to be 0.89 and 0.97, respectively. The biliary reabsorption rate of PFOA was found to be 0.98 in another study (n = 5) in which similar methodology was used (Fujii et al., 2015).

PFOA and PFOS are present in the environment as a mixture of linear and branched isomers. PFOA is primarily present as a linear isomer in various exposure matrices, whereas the PFOS isomer composition is more variable (Shan et al., 2016). In human blood and serum, the fraction of the linear PFOA isomer (*n*-PFOA) is 96–100% (Zhang et al., 2014a, 2017a), which is consistent with the high proportion of *n*-PFOA in food and drinking-water and its expected slower elimination compared with branched isomers (Zhang et al., 2013b; Zhou et al., 2014). In contrast, the proportion of *n*-PFOS in the blood or serum (35–80% of total blood PFOS) is lower than would be expected in most exposure scenarios (Zhang et al., 2014a, 2017a; Gebbink et al., 2015; Salihović et al., 2015; Nilsson et al., 2022). This apparent relative enrichment of branched isomers could be the result of the metabolism of unknown PFOS precursor(s) to form branched isomers (Gebbink et al., 2015; Shan et al., 2016).

The proportion of *n*-PFOA in cord serum has been reported to be 98–99% (Beesoon et al., 2011; Zhang et al., 2017a), and that of *n*-PFOS to be 52–75% of the total, which was lower than that in the maternal serum in studies in which such data were available (Hamm et al., 2010; Beesoon et al., 2011; Zhang and Qin, 2014; Zhang et al., 2017a).

(iii) Metabolism

No in vivo data on the metabolism of PFOA and PFOS were available to the Working Group. On the basis of their chemical structure and data demonstrating a lack of metabolism in experimental animals, PFOA and PFOS are not expected to be metabolized in humans.

The metabolism of precursor compounds, including fluorotelomer alcohols, perfluoroalkyl sulfonamides, and amidoalcohols, to form PFOA and PFOS was estimated to contribute to 2–8% and 2–5% of the internal doses of PFOA and PFOS, respectively, in an intermediate-level exposure scenario, and to 28–55% and 60–80%, respectively, in a high-level exposure scenario (Vestergren et al., 2008). By using exposure and biomonitoring data in a TK model, Gomis et al. (2016) estimated that the metabolism of 8:2 fluorotelomer alcohol contributed to 45% of the serum concentration of PFOA in ski-waxers who have a high level of occupational exposure.

(iv) Excretion

PFOA and PFOS are characterized by very long half-lives in humans, with estimates of 2–5 years reported in most studies (Harada et al., 2005; Olsen et al., 2007; Spliethoff et al., 2008; Costa et al., 2009; Bartell et al., 2010; Brede et al., 2010; D'eon and Mabury, 2011; Seals et al., 2011; Glynn et al., 2012; Olsen et al., 2012; Yeung et al., 2013b; Zhang et al., 2013b; Russell et al., 2015; Fu et al., 2016; Gomis et al., 2016, 2017; Worley et al., 2017a; Li et al., 2018a; Xu et al., 2020a). [The Working Group noted that these were observational population kinetic studies, in which participants may have been subject to unspecified background exposures, in addition to known contamination sources.] Recently, Chiu et al. (2022) applied a Bayesian TK analysis to data from 13 studies of the association between exposure and serum concentration; they reported serum half-life $(T_{1/2})$ estimates for the population geometric mean of 3.14 years for PFOA and 3.36 years for PFOS.

[The Working Group noted that several studies have investigated the serum half-lives of PFOS isomers (Xu et al., 2020a; Li et al., 2022a; Nilsson et al., 2022), but the results were inconsistent among studies.]

<u>Xu et al. (2020a)</u> reported a longer $T_{1/2}$ for *n*-PFOS than for the sum of branched-chain forms of PFOS, but the other two studies conducted by the same research group reported shorter $T_{1/2}$ estimates for *n*-PFOS than for the branched isomers (Li et al., 2022a; Nilsson et al., 2022).

In humans, PFOA and PFOS are primarily eliminated in the urine and faeces. In women of reproductive age, blood loss during menstruation, fetal transfer during pregnancy, and lactational transfer during breastfeeding are also thought to reduce the body burden of PFOA and PFOS (Mondal et al., 2014; Wong et al., 2014; Gomis et al., 2017; Pizzurro et al., 2019). Multiple studies reported renal clearance of PFOA and PFOS, with estimates of 0.03–0.8 mL/ (kg-day) and 0.01–0.03 mL/(kg-day), respectively, calculated in a 24-hour period and assuming an average human body weight of 50–70 kg (<u>Harada et al., 2005; Zhang et al., 2013b; Zhou et al., 2014; Fujii et al., 2015; Gao et al., 2015a; Zhang et al., 2015a; Fu et al., 2016</u>).

Linear PFOA and PFOS isomers appeared to be eliminated more slowly than branched isomers in the urine (<u>Zhang et al., 2013b</u>; <u>Gao et al., 2015a</u>). In only one study, PFOA and PFOS levels were reported to be above the limit of detection in the faeces, and only after addition of the bile acid sequestrant cholestyramine to the diet (<u>Genuis et al., 2010</u>).

A comparison of the urinary and total elimination rates by <u>Fujii et al. (2015)</u> showed that the faecal and urinary PFOA clearances contribute approximately 54% (0.052 mL/(kg-day) and 46% (0.044 mL/(kg-day), respectively, to the overall (serum) clearance, calculated in a 24-hour period and assuming a human body weight of 50 kg.

Using the same approach, <u>Harada et al. (2007)</u> found that urinary PFOA clearance constitutes approximately 20% of the total serum clearance – 0.03 mL/(kg-day) to 0.15 mL/(kg-day) – and urinary PFOS clearance constitutes approximately 14% – 0.015 mL/(kg-day) to 0.106 mL/(kg-day).

[The Working Group noted that the urinary clearance appears to be of less importance as an excretion route for PFOA and PFOS in humans than the estimated faecal clearance. The limits of detection for PFOA and PFOS in faeces appear to be higher than for other physiological matrices, which may explain the scarcity of published data.]

The body burden of PFOA and PFOS, indicated by serum concentrations, increases in response to exposure to these chemicals. Their excretion increases with increasing serum concentration until, if the intake is constant, an equilibrium is reached. The relation between intake and the serum concentration at steady state can be estimated, assuming first-order kinetics. One-compartment TK models are widely used in the literature to back-calculate PFOA and PFOS exposure (ng/kg per day) from serum levels (ng/mL). The outcome in these models depends on selected $V_{\rm d}$ and $T_{1/2}$ values. Several two-compartment human TK models have also been developed for PFOA and PFOS, to investigate gestational and lactational transfer or to fit TK data from a clinical trial with PFOA (Verner et al., 2016; Convertino et al., 2018; Goeden et al., 2019). Several multicompartment PBPK human models have been developed for PFOA and PFOS that included renal reabsorption, to account for the long retention times in humans (Loccisano et al., 2011, 2013; Fabrega et al., 2014; Worley et al., 2017b; Chou and Lin, 2019). Of note, these models do not include either the faecal elimination route (Loccisano et al., 2011, 2013; Fàbrega et al., 2014) or the complete enterohepatic loop (Worley et al., 2017b; Chou and Lin, 2019).

[The Working Group noted that not fully accounting for faecal elimination would probably cause an overestimation of the effect of renal elimination relative to total elimination, with possible implications for PBPK studies of confounders dependent on renal elimination. The Working Group noted that a recently published PBPK model for PFOA (Husøy et al., 2023) appeared to include the faecal elimination route; however, this model was not evaluated in the present monograph.]

(b) Human cells in vitro

Approximately 24% of the applied dose of PFOA was found to penetrate human skin in vitro, and the skin permeability coefficient for ionized PFOA (a physiologically relevant form) was 6.0×10^{-5} cm/hour (Franko et al., 2012).

Cell-free and cell culture in vitro experimental systems have been used to characterize the interactions of PFOA and PFOS with membrane transporters, serum albumin, liver fatty acidbinding protein (L-FABP), thyroid receptor, and transthyretin. Interactions with membrane transporters probably mediate the absorption and reabsorption of PFOA and PFOS from the gastrointestinal tract, their reabsorption in the kidney, and their ability to cross the placenta. High-affinity binding to serum albumin and L-FABP may underly the accumulation of PFOA and PFOS in serum and the liver.

Most available data regarding the interactions of PFOA and PFOS with transporters concern organic anion transporting polypeptides (OATPs), a family of proteins that are expressed in the gut, kidney, and placenta.

In uptake studies of cells that were stably transfected with human membrane transporters, PFOA was found to be transported by (human) OAT1 ($K_m = 48 \mu$ M), OAT3 ($K_m = 49.1 \mu$ M), OAT4 ($K_m = 310 \mu$ M), URAT1 ($K_m = 64.1 \mu$ M) and Na⁺/taurocholate cotransporting polypeptide (NTCP) ($K_m = 1.8 \text{ mM}$) (Nakagawa et al., 2008, 2009; Yang et al., 2010; Kummu et al., 2015), but not by OAT2, OATP1A2, or ABCG2 (Nakagawa et al., 2008; Yang et al., 2010; Kummu et al., 2015). In uptake studies of cells that were transiently transfected with human membrane transporters, PFOS was transported by (human) OSTa/ β (no reported K_m) and NTCP ($K_m = 130 \mu$ M), but not by ABCG2 (Zhao et al., 2015).

Kummu et al. (2015) proposed that OAT4 is involved in the transport of PFOA across the placenta, on the basis of a positive correlation between protein expression and the transport of PFOA by isolated placenta cells. Kimura et al. (2017) showed a decrease in the uptake of PFOA across the apical membrane of human intestinal Caco-2 cells when they were co-incubated with PFOA and inhibitors of OATPs or with the OATP substrate sulfobromophthalein. Furthermore, PFOA inhibited the uptake of sulfobromophthalein at this location (Kimura et al., 2020). HEK293 cells stably transfected with NTCP exhibited PFOA uptake ($K_{\rm m} = 1.8$ mM) and the inhibition of taurocholate uptake by PFOA ($K_{\rm i} = 7.5-28 \ \mu$ M) (<u>Ruggiero et al., 2021</u>).

PFOA and PFOS bind to several human proteins in cell-free systems, including to serum albumin, with dissociation constants $K_{\rm d}$ = 3.7 \times 10^{-6}-4 \times 10^{-4} M for PFOA, and $4.5-20 \times 10^{-5}$ M for PFOS, and to L-FABP with $K_{\rm d} = 2.4 - 50.4 \times 10^{-6} \,\mathrm{M}$ for PFOA and $18.5 \times 10^{-6} \,\mathrm{M}$ for PFOS (Han et al., 2003; Messina et al., 2005a; Chen and Guo, 2009; Li et al., 2009; Wu et al., 2009; Luo et al., 2012; Zhang et al., 2013c; Sheng et al., 2016; Maso et al., 2021). [Where appropriate, the Working Group converted association or absorption constants to dissociation constants for the purposes of comparison; see Supplementary Table S4.1, in Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: https://publications. iarc.who.int/636.] PFOA was able to displace the endogenous ligand oleic acid from human serum albumin indicating the potential to interfere in fatty acid transport in the blood (D'Eon et al., 2010).

The linear isomers *n*-PFOA and *n*-PFOS bind much more tightly to human serum albumin than do the branched isomers, with dissociation constants that are several orders of magnitude lower (Beesoon and Martin, 2015). As described in more detail in Section 4.2.8, PFOA and PFOS were able to displace thyroid hormones (T3, T4) from the human thyroid receptor and the thyroid hormone transporter protein transthyretin (Weiss et al., 2009; Ren et al., 2015, 2016).

(c) Considerations regarding exposure metrics

Cross-sectional measurements of serum concentrations of PFOA and PFOS are often used to study associations in epidemiological investigations, because of cost constraints, and with the assumption that such metrics would represent quantitative indicators of exposure over an appropriate time frame for the chronic

effects studied (see Section 1.6.2, and Annex 3, Supplementary analyses used in reviewing evidence on cancer in humans, available from: https://publications.iarc.who.int/636). In support of this approach, consecutive measurements of the serum concentrations of PFOA and PFOS, measured in the same non-occupationally exposed individuals in 1979, 1986, 1994, 2001, and 2007, were closely correlated (Spearman ρ coefficient, > 0.6), even though the mean levels increased 3.66-fold, peaking in 2001, and then decreased in 2007 (Nøst et al., 2014). In another investigation of the applicability of the use of a single serum PFOA measurement in human studies, retrospective PFOA exposure was reconstructed for participants in the C8 cohort, using geocoding, documented air and water concentrations, and a pharmacokinetic model with an approximate Bayesian computation (Zhu et al., 2022). These data were combined with the measured PFOA serum concentrations to produce a better longitudinal estimate of PFOA exposure. Associations between PFOA and pre-eclampsia based on reconstructed historical PFOA exposure estimates in this population were found to be similar to those reported, based on one-time PFOA measurements (Savitz et al., 2012), suggesting that the rank order of exposure estimates of study participants was important for statistically significant associations, and not the exact exposure amount (Zhu et al., 2022).

On the basis of current understanding of PFOA and PFOS kinetics, various physiological factors can affect their serum concentrations and lead to possible reverse causation effects in epidemiological analyses of associations. This type of bias has been termed pharmacokinetic bias, and several examples regarding PFOA and PFOS have been described in a recent review (Andersen et al., 2021). These examples are detailed below.

Dhingra et al. (2017) examined associations of menopause and estimated glomerular filtration rate (eGFR) with the measured or modelled serum concentrations of PFOA in a highly exposed cohort. Serum PFOA concentrations were from single sample collections, and the modelled PFOA concentrations were based on the retrospective exposure to PFOA and a one-compartment TK model (Shin et al., 2011). The modelled PFOA concentrations were generated independently of the measured concentrations and had a Spearman correlation of $\rho = 0.71$ (Winquist et al., 2013; Dhingra et al., 2017). Menopause and eGFR were found to be significantly associated with the measured, but not modelled, serum PFOA concentration (Winquist et al., 2013; Dhingra et al., 2017). Specifically, there was an inverse relation (negative trend) of eGFR with the quintile of measured serum PFOA concentration, but neither the modelled serum PFOA concentration nor the modelled cumulative exposure showed an association with eGFR (Dhingra et al., 2017). In the same study, a significant increasing trend for reported menopause was identified alongside an increase in the measured serum PFOA categories, but not in the modelled metrics (Dhingra et al., 2017). The Working Group noted that these findings can be interpreted as an instance of reverse causation, in which changes in eGFR, which also occur as a consequence of menopause, would directly affect PFOA excretion and serum PFOA concentration.]

The same reverse causation mechanism, in which changes in GFR would affect concentrations of PFOA and/or PFOS and the outcome of interest, has been investigated for potential associations with respect to low birth weight (Verner et al., 2015). A human pregnancy PBPK model of PFOA/PFOS was updated and parametrized by Loccisano et al. (2013) to incorporate the relation between GFR and birth weight, based on the results of previous studies of substances other than perfluoroalkyl and polyfluoroalkyl substances (PFAS). A PBPK model-driven metaanalysis was performed, investigating associations of serum PFOA and PFOS concentrations with birth weight, using data obtained from seven

studies. The use of the modified PBPK model was intended to "remove" the confounding effect of low birth weight on GFR, and to simulate direct effects of increases in serum PFOA and/or PFOS concentrations on birth weight. The resulting levels were approximately 50% of those reported on the basis of measured PFOA and PFOS concentrations. The authors concluded that a substantial proportion of the reported associations of prenatal PFOA and PFOS concentrations with birth weight was attributable to confounding by GFR (Verner et al., 2015). [The Working Group noted that the TK adjustment for GFR confounding in the described examples relied on models that assumed urinary excretion to be the only excretion route. However, faecal elimination in humans also appeared to be equally, if not more, important. Thus, such adjustments may overestimate the effects of GFR and urinary elimination on serum PFOA/PFOS levels.]

Hypotheses regarding reverse causation have been also investigated with respect to associations of serum PFOA and PFOS concentrations with serum cholesterol (EFSA, 2018; Steenland et al., 2020). Two possible mechanisms were proposed. Firstly, PFOA and PFOS might preferentially distribute to cholesterol-containing lipoprotein particles in the blood, resulting in greater accumulation in the presence of higher cholesterol concentrations, therefore explaining the reverse causation. The European Food Safety Authority (EFSA) Panel on Contaminants in the Food Chain provided a detailed analysis of this hypothetical mechanism and concluded that it was not supported by the available mechanistic data (EFSA, 2018). Secondly, the concentrations of serum PFOA and PFOS might be influenced by inter-individual variability in bile acid transporters, for which they are substrates, in enterohepatic resorption and would correlate with bile acid concentrations by virtue of sharing the same retention mechanism. However, an increase in bile acid concentrations would inhibit the metabolism of cholesterol to form bile acids, resulting in increases in cholesterol concentrations and an indirect relationship of the PFOA and PFOS concentrations with those of cholesterol. This mechanism was also briefly discussed in the updated EFSA report of 2020, in light of the available findings (EFSA, 2020).

One of the proposed reasons against this mechanism was a study that found significant associations for serum cholesterol based on estimated PFOA/PFOS external exposures, and not just serum levels (Li et al., 2020a). Additionally, in a study that reported associations between 20 measured PFAS (including PFOA and PFOS) and 19 bile acids in samples from healthy individuals (n = 20), the overall trend was that the concentrations of the majority of bile acids were negatively associated with those of PFAS (Salihović et al., 2020a), also indicating a lack of support for the shared retention mechanism. [The Working Group noted that the existing evidence does not support the proposed mechanisms of reverse causality between the measured PFAS and serum cholesterol.]

The effects of advanced cancers on the TKs of PFOA and PFOS remain unclear. PFOA has been previously tested in a phase 1 therapeutic trial of 43 patients with cancers at various stages, and the TK data obtained were made publicly available in a patent application (Elcombe et al., 2013). [The Working Group noted that these data have not been curated or peer-reviewed.] In this trial, up to 1200 mg of PFOA was administered to the patients once per week for 6 weeks, and their serum PFOA concentrations were monitored. The experimental set-up and the results have been described and discussed in separate publications (Convertino et al., 2018; Dourson et al., 2019), in which the authors noted that secondary to this very high level of exposure, which exceeded the typical level of environmental exposure by several orders of magnitude, the serum PFOA concentration appeared to reach steady state within weeks, which would correspond to a much shorter half-life estimate than the 2–5 years that has generally been recorded in the literature (see above). [The Working Group noted that it was not clear whether the apparently faster elimination was because of higher PFOA doses or altered physiological parameters in the study participants, all of whom were patients with cancer. The study did not appear to include healthy controls.] [The Working Group noted that there was no information available on the kinetic effects of cancer or cancer treatment on PFOA or PFOS levels in patients with cancer.]

4.1.2 Experimental systems

(a) Experimental systems in vivo

The TK mechanisms associated with PFOA and PFOS have been extensively investigated in the mouse and rat, and a brief review of the findings obtained using these two species follows. In addition, information obtained from primates is included, where available.

The oral absorption efficiencies of PFOA and PFOS have been estimated in animal studies by comparing the area under the curve (AUC) of the serum/plasma concentration, with the dose-adjusted AUC obtained using data collected after an intravenous (i.v.) dose (an oral/i.v. ratio of 1 indicates 100% absorption efficiency), or alternatively by estimating the amount of the administered compound that remained in the faeces during the 24 hours after administration. [The Working Group noted that the second method could underestimate absorption efficiency, because some excretion via the faecal route would be expected within the observation period.] Using the first method, a comparison of AUCs yielded an oral absorption efficiency for PFOA of close to 100% in rats (Kim et al., 2016a; Dzierlenga et al., 2020) and mice (Fujii et al., 2015), and also for PFOS in rats (Kim et al., 2016a; Huang et al., 2019a, 2021). There were no studies available measuring PFOS in mice. With the second method (subtraction of the excretion via faecal

route from 100%), the PFOA oral absorption efficiency was \geq 93% in rats (<u>Cui et al., 2010</u>), and the PFOS oral absorption efficiency in mice was \geq 98% (<u>Jandacek et al., 2010</u>).

PFOA exposure via inhalation increased the plasma PFOA concentrations of male and female rats in a dose-dependent manner (<u>Hinderliter et al., 2006</u>), and these authors concluded that a route-to-route extrapolation would be possible using the plasma PFOA concentration as an index of the internal dose metric. [No other TK studies of inhalation were available to the Working Group.]

In rats, the dermal application of APFO increased blood organofluorine concentration (a surrogate for PFOA concentration) in a dose-dependent manner, indicating that PFOA is absorbed through the skin (Kennedy, 1985). Similarly, the topical application of PFOA increased serum PFOA concentration of mice in a dose-dependent manner (Franko et al., 2012).

The Working Group identified more than 50 studies on PFOA and PFOS distribution in rats and mice, including studies of single and multiple dosing; studies of oral administration by gavage, in the drinking-water or in the diet; i.v. and intraperitoneal (i.p.) administration; and studies in pregnant animals (e.g. <u>Ylinen et al., 1990;</u> Vanden Heuvel et al., 1991; Austin et al., 2003; Luebker et al., 2005; Hundley et al., 2006; Kudo et al., 2007; Benskin et al., 2009; Cui et al., 2009; Bogdanska et al., 2011; Fujii et al., 2015; Iwabuchi et al., 2017; Dzierlenga et al., 2020). The data on organ distributions identified for PFOA and PFOS were generally consistent across the studies, with liver being the primary site of accumulation of both compounds, followed by the kidney and the blood/serum. Lower levels of accumulation were detected in the brain, fat, muscle, thyroid, testes, thymus, and skin than in the serum. PFOS was found to preferentially accumulate in the lungs in both mice and rats, but PFOA was not. The data available for monkeys were limited to the liver and serum, and showed preferential

accumulation of PFOS in the liver (Seacat et al., 2002), but much lower levels of PFOA in the liver than in the serum (Griffith and Long, 1980; Butenhoff et al., 2004). Most animal studies were conducted using low to intermediate milligram per kilogram doses (either single or repeated administration). Several of the studies included doses in the microgram per kilogram range (e.g. Seacat et al., 2002; Bogdanska et al., 2011; and Li et al., 2017a), and these generated organ distribution data that were comparable to the results obtained using higher doses.

PFOA and PFOS are efficiently transferred to fetuses during pregnancy and to pups via lactation (Lau et al., 2003; Chang et al., 2009; Fenton et al., 2009; Macon et al., 2011; Lai et al., 2017a). Similar to humans, PFOA and PFOS are involved in enterohepatic resorption, because cholestyr-amine treatment in the diet for 14 days in rats substantially increased the levels of [¹⁴C]PFOA and [¹⁴C]PFOS in the faeces after a single i.v. dose of PFOA at 13.3 or PFOS at 3.4 mg/kg (Johnson et al., 1984). PFOA and PFOS isomers showed minor differences in their organ distributions in male and female rats (Benskin et al., 2009; De Silva et al., 2009).

In the literature, there were several estimates of V_d in rats, mice, and monkeys. For comparison purposes, the following examples are from studies in which a one-compartment TK model was assumed or in which a non-parametric analysis was applied.

In rats treated with a single oral dose of PFOA (1 mg/kg) or PFOS (2 mg/kg), the PFOA V_d estimates were approximately 106 and 154 mL/kg for males and females, respectively, and the PFOS V_d estimates were approximately 280 and 289 mL/kg for males and females, respectively (Kim et al., 2016a).

In mice treated with a single i.v. dose of PFOA (0.13 mg/kg), V_d estimates were approximately 180 mL/kg in males and 150 mL/kg in females (Fujii et al., 2015). In mice treated with a single oral dose of PFOS (1 or 20 mg/kg), the

 $V_{\rm d}$ estimates were approximately 263–290 mL/kg in males and 258–261 mL/kg in females (Chang et al., 2012).

In monkeys treated with a single i.v. dose of PFOA (10 mg/kg), the V_d estimates were 181 and 198 mL/kg for males and females, respectively (Butenhoff et al., 2004). In monkeys treated with a single oral dose of PFOS (9 or 14 mg/kg), the V_d estimates were 127–135 mL/kg and 127–141 mL/kg in males and females, respectively (Chang et al., 2017).

[The Working Group noted that the magnitude of the V_d estimates for rats, mice, and monkeys (150–250 mL/kg) is consistent with the current understanding of the PFOA and PFOS distribution mechanism, with strong binding to proteins in the blood and accumulation in several organs.]

Two detailed studies of PFOA metabolism in rats were available. Vanden Heuvel et al. (1992) treated male and female rats with a single i.p. dose of $[1-^{14}C]$ PFOA (9.4 µmol/kg or 3.9 mg/kg) and investigated the tissue distribution and tissue-specific concentrations at various time points up to 28 days. They did not find PFOA metabolites or conjugates in many tissues (liver, kidney, heart, muscle, fat, or testes) or in the plasma, urine, bile, or faeces, based on the similarity of the high-performance liquid chromatography (HPLC) chromatograms of the sampled extracts to the chromatogram of the parent compound. In addition, they found no increases in the serum or urine concentrations of fluoride, which would have indicated PFOA defluorination (Vanden Heuvel et al., 1991). In a study by <u>Goecke et al. (1992)</u>, a single i.p. dose of PFOA (50 mg/kg) was administered to male rats, and 19F-nuclear magnetic resonance spectra were acquired from blood, urine, bile, and liver homogenates 3 days after dosing. Only resonances referable to the parent compound were identified in all the analyses performed, which implies a lack of covalent modification of the PFOA molecule, i.e. absence of metabolites or

conjugates (<u>Goecke et al., 1992</u>). Other studies in rats confirmed the lack of phase I or phase II metabolism of PFOA (<u>Ophaug and Singer, 1980</u>; <u>Ylinen et al., 1989</u>; <u>Kuslikis et al., 1992</u>).

[The Working Group noted that, on the basis of the structural similarities between PFOA and PFOS and similarities in their TK mechanisms among species, PFOA and PFOS are not expected to be metabolized in mammals.]

There are species- and sex-specific differences with respect to the elimination of PFOA and PFOS. Estimates of the serum $T_{1/2}$ for PFOA in male rats after oral administration ranged from 6.4 days to 13.4 days on the basis of five studies that included a total of 12 dose groups (Kudo et al., 2002; US EPA, 2003; Benskin et al., 2009; Iwabuchi et al., 2017; Dzierlenga et al., 2020). Only one study with a single dose group produced a lower estimate of oral PFOA serum $T_{1/2}$ of 1.64 days (<u>Kim et al., 2016a</u>). However, the serum $T_{1/2}$ estimates for PFOA for female rats ranged from 2.75 to 13.9 hours, based on the results of three studies comprising a total of 10 dose groups (US EPA, 2003; Kim et al., 2016a; Dzierlenga et al., 2020). It was hypothesized that the sex-specific differences in rats with respect to the elimination of PFOA are the consequence of differential expression of specific transporter proteins in the kidneys, resulting in more effective reabsorption in males (Weaver et al., 2010; <u>Pizzurro et al., 2019</u>).

After a single oral dose by gavage, the serum $T_{1/2}$ for PFOA was 21.7 days in male mice and 15.6 days in female mice (Lou et al., 2009). The only available estimates for primates were obtained in monkeys (*Simia cynomolgus* or *Macaca fascicularis*). Butenhoff et al. treated monkeys with PFOA as an i.v. dose of 10 mg/kg, yielding a serum $T_{1/2}$ estimate for PFOA of 13.6–35.3 days in males and 26.8–41.7 days in females (Butenhoff et al., 2004).

In rats, the serum $T_{1/2}$ estimates for PFOS after gavage were 8.23–41.19 days for males and 23.5–71.13 days for females in three studies of a

total of 5 dosage groups (<u>Chang et al., 2012; Kim</u> et al., 2016a; <u>Huang et al., 2019a</u>, <u>2021</u>).

Longer $T_{1/2}$ estimates were obtained for female rats than for males in one of these studies (<u>Chang</u> <u>et al., 2012</u>), but no sex-specific differences were identified in the other two.

In mice, the serum $T_{1/2}$ estimates for PFOS after oral administration were 36–43 days for males and 30–38 days for females, with no significant sex-specific differences (Chang et al., 2012). In monkeys, the serum $T_{1/2}$ estimates for PFOS after oral administration were 117–200 days for males and 102–200 days for females, also with no significant sex-specific differences (Seacat et al., 2002; Chang et al., 2017).

Linear *n*-PFOA and *n*-PFOS isomers appear to be eliminated more slowly than the branched isomers. When mixtures of PFOA or PFOS isomers were administered to male rats by oral gavage, isomer-specific blood elimination $T_{1/2}$ was determined (Benskin et al., 2009). In this study, *n*-PFOA was found to have the longest blood elimination $T_{1/2}$, 13.4 days, compared with 1.28–9.10 days for the branched isomers; the $T_{1/2}$ for *n*-PFOS (33.7 days) was at the top end of the range obtained (the geometric mean for all the PFOS isomers was 23.8 days) (Benskin et al., 2009).

In rats and mice, PFOA and PFOS are eliminated in the urine and faeces. Vanden Heuvel et al. (1991) measured [14C]PFOA elimination in male and female Sprague-Dawley rats. They found that, in females, 91% of the i.p. dose $(9.4 \,\mu\text{mol/kg or } 3.9 \,\text{mg/kg})$ was eliminated in the urine during the first 24 hours, with negligible amounts being present in the faeces, whereas in males, over the 28-day collection period, 36.4% and 35.1% of the administered dose was recovered in the urine and faeces, respectively (Vanden Heuvel et al., 1991). However, Cui et al. (2010) found greater excretion (approximately two-fold) of PFOA in the urine than in the faeces during the daily administration of PFOA by gavage at 5 or 20 mg/kg to male Sprague-Dawley rats. In this

study in male Sprague-Dawley rats, the rate of PFOS excretion in urine exceeded the excretion rate in faeces, at doses of either 5 or 20 mg/kg (Cui et al., 2010). Another study in male and female Sprague-Dawley rats also showed that urinary excretion was the primary means of eliminating PFOS after a single oral dose of 2 or 15 mg/kg of potassium perfluorooctanesulfonate (Chang et al., 2012). Furthermore, the urinary PFOA clearance, in male or female FVB/NJc1 mice, was approximately twice as high as the faecal clearance after either i.v. or oral gavage administration (0.313 μ mol/kg and 3.13 μ mol/kg, respectively, equal to 0.13 and 1.3 mg/kg) (Fujii et al., 2015).

Similarly, the clearance rate of PFOS in CD-1 mice after a single oral dose (1 or 20 mg/kg of the potassium salt, potassium perfluorooctane-sulfonate) exceeded the faecal clearance rate (<u>Chang et al., 2012</u>). [The Working Group noted that urinary excretion appears to be the primary route of elimination for PFOA and PFOS in rats and mice.]

In monkeys, larger amounts of PFOA were also excreted in the urine than in the faeces at steady state (<u>Butenhoff et al., 2004</u>). [The Working Group noted that urinary excretion is more relevant than faecal excretion in laboratory animals, while in humans the opposite is true, at least for PFOS.]

Several TK and PBPK models were available regarding PFOA and PFOS in rats, mice, and monkeys (reviewed in <u>Bernstein et al., 2021)</u>. [The Working Group noted that most of these models included the assumption that urinary elimination is the primary route, on the basis of the experimental evidence. Accordingly, the kidney reabsorption loop has been extensively studied using PBPK models. However, the enterohepatic reabsorption component in rodents, despite the evidence for its presence (Johnson et al., 1984), has not received much attention, and it has not been fully incorporated into the existing rodent PBPK models.] Organic anion transporters are thought to be involved in the renal elimination and reabsorption of PFOA also in non-human mammals. The expression levels of OATP1a1, OAT2, and OAT3 were found to be sex-dependent and responsive to changes in steroid hormone concentrations in rats (Kudo et al., 2002; Cheng et al., 2006). In rats, multiple regression analysis suggested that the clearance of PFOA was at least in part dependent on the renal expression of OAT2 and OAT3 (Kudo et al., 2002).

The renal clearance of PFOA was increased by mannitol infusion (which increases urine flow) but was reduced by a low-phosphate diet in both male and female rats (<u>Katakura et al., 2007</u>).

The renal clearance of PFOA did not change in the rat knockout model for multidrug resistance protein (MRP2) compared with wildtype rats (<u>Katakura et al., 2007</u>), indicating that MRP2 is not involved in PFOA transport.

(b) Experimental systems in vitro

Approximately 39% of the applied concentration of PFOA was found to penetrate mouse skin in vitro (Franko et al., 2012). [The Working Group noted that the PFOA penetration of mouse skin samples in vitro was higher than that for human skin samples in this study, which was 24% of the dose administered.]

The transport of PFOA by rat transporters has been reported for OAT1 ($K_m = 43.2-51 \mu$ M), OAT3 ($K_m = 65.7-80.2 \mu$ M), and OATP1a1 ($K_m = 126.5-162 \mu$ M), using transiently transfected oocytes of *Xenopus laevis* and human embryonic kidney HEK293 cells in vitro (Katakura et al., 2007; Nakagawa et al., 2009; Weaver et al., 2010). No transport was observed in vitro for rat OAT2 or URAT1 (Nakagawa et al., 2009; Weaver et al., 2010).

<u>Woodcroft et al. (2010)</u> investigated the binding of PFOA to bacterially expressed rat L-FABP and, depending on the method used (direct displacement or isothermal titration calorimetry), predicted the existence of two

Substance	Humanª Population GM (95% CI)	Monkey Range	Mouse Range or mean	Rat Range
PFOA	3.14 yr (2.69–3.73 yr)	M, 13.6–35.3 d F, 26.8–41.7 d	M, 21.7 d F, 15.6 d	M, 6.4–13.4 d F, 2.75–13.9 h
PFOS	3.36 yr (2.52–4.42 yr)	M, 117–200 d F, 102–200 d	M, 36–43 d F, 30–38 d	M, 8.23–41.19 d F, 23.5–71.13 d

Table 4.2 Serum half-life $(T_{1/2})$ estimates for key species

CI, confidence interval; d, day(s); F, female; GM, geometric mean; h, hour(s); M, male; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; yr, year(s).

^a Estimates for humans are for both men and women.

Data were obtained from Kudo et al. (2002); Seacat et al. (2002); US EPA (2003); Butenhoff et al. (2004); Benskin et al. (2009); Lou et al. (2009); Kim et al. (2016a); Chang et al. (2017); Jwabuchi et al. (2017); Huang et al. (2019a, 2021); Dzierlenga et al. (2020); Chiu et al. (2022).

or three binding sites, with $K_{\rm m}$ estimates of $3.1-52.6 \times 10^{-6}$ M. Furthermore, PFOA and PFOS displaced a fluorescent fatty acid analogue from its binding site on rat L-FABP in vitro, indicating their potential to interfere with fatty-acid binding in vivo (Luebker et al., 2002). PFOA bound to rat and human serum albumin with similar affinities in vitro ($K_{\rm m} = 4 \times 10^{-4}$ M) (Han et al., 2003).

(c) Interspecies differences

There are marked interspecies differences in the $T_{1/2}$ of PFOA and PFOS (summarized in Table 4.2). The rat demonstrates sex-specific differences with respect to PFOA but not PFOS, and the female rat has the shortest PFOA $T_{1/2}$ overall. However, no sex-specific differences have been identified in other species with respect to either PFOA or PFOS. The mechanisms responsible for the long $T_{1/2}$ in humans have not been elucidated. It has been hypothesized that tighter binding to endogenous proteins, such as albumin in the serum or L-FABP in the liver, or more efficient reabsorption by OATs in the kidney may be responsible. However, in vitro studies of these molecular targets did not report differences in the affinities of the rat and human proteins for PFOA (<u>Nakagawa et al., 2008</u>). Of note, owing to a shorter $T_{1/2}$ for PFOA, implying more rapid excretion, female rats appear to represent a more complex combination of laboratory species and sex, such that extrapolation of data on PFOA to

humans is more difficult than from male rats or mice. [The Working Group noted that longer human PFOA and PFOS $T_{1/2}$ appear to correlate with relatively greater faecal excretion, whereas the shorter half-lives in monkeys, mice, and rats appear to favour the urinary excretion of PFOA and PFOS.]

Most previous research has focused on OATs and the involvement of bile acid transporters in the intestinal reabsorption of PFOA and PFOS; therefore data regarding possible interspecies differences in this mechanism are scarce. The human kidney reabsorbs bile acids during filtration (<u>Stiehl, 1974</u>), suggesting that that bile acid transporters may be present, and these may also play a role in the renal reabsorption of PFOA and PFOS.

Although PFOA and PFOS accumulate in the liver in humans, the liver/serum and liver/blood concentration ratios for each agent are much lower than those reported in animal studies. For example, in postmortem samples collected from adults, <u>Maestri et al. (2006)</u> found liver/blood ratios of 1 for PFOA and 2.7 for PFOS (<u>Maestri et al., 2006</u>). However, a study of human fetal postmortem tissues showed that liver levels of PFOA and PFOS were lower than those in maternal serum (<u>Mamsen et al., 2019</u>). Furthermore, in repeated oral dosing studies, PFOA and PFOS were found to accumulate in the rat and mouse liver and yield much higher concentration ratios for PFOA of

3.3–5.6 and for PFOS of 4.8; <u>Cui et al., 2009</u>). Moreover, PFOA and PFOS have been shown to accumulate in the livers of fetal mice and rats (<u>Luebker et al., 2005</u>; <u>Chang et al., 2009</u>; <u>Macon</u> <u>et al., 2011</u>; <u>Ishida et al., 2017</u>; <u>Lai et al., 2017</u>a).

The kidney is another organ that shows interspecies differences with respect to PFOA accumulation. In postmortem kidney samples obtained from adult humans, PFOA and PFOS concentrations were slightly higher than those in the blood, with kidney/blood ratios of 1.2 for both compounds (Maestri et al., 2006). In rats, the kidney/blood ratio was approximately 0.7 for PFOA and approximately 1.0 for PFOS in 3-month oral multiple dose studies (Gao et al., 2015b; Iwabuchi et al., 2017), and similar values were obtained using other study designs. In studies conducted in mice exposed to PFOS, the kidney/blood ratio was estimated to be 0.5-0.9 (Bogdanska et al., 2011; Chang et al., 2012), whereas studies in mice exposed to PFOA yielded an estimated kidney/blood ratio of 0.15-0.2 (Lou et al., 2009; Fujii et al., 2015). Thus, the kidney/ blood ratio appears to be lower by 6-8-fold in mice than in humans, indicating that these agents accumulate to a lesser extent in this organ in mice. [The Working Group acknowledged that limited data were available and noted that differences in the site-specific preferential accumulation of PFOA and PFOS may contribute to species differences in adverse effects.]

Synopsis

[In summary, the Working Group noted that there were few available data regarding the absorption and distribution of PFOA and PFOS in humans. PFOA and PFOS are absorbed after oral, inhalation, or dermal exposure. On the basis of their structures, PFOA and PFOS are not likely to cross cellular membranes directly, and various membrane transporters are thought to mediate transmembrane transport. The liver, blood, and lungs are important sites of accumulation for both agents. The distribution of PFOA and PFOS appears to be driven by binding to specific proteins, such as albumin in the blood and L-FABP in the liver. The available evidence suggests that enterohepatic circulation of PFOA and PFOS and bile acid transporters in the gut could be responsible for this effect. PFOA and PFOS cross the placenta and are transferred to infants via breast milk. Neither agent is metabolized. PFOA and PFOS are excreted in the faeces, which is an important route in humans, and the urine. In women of reproductive age, blood loss during menstruation, fetal transfer during pregnancy, and transfer during breastfeeding also contribute to the loss of PFOA and PFOS. The half-lives of PFOA and PFOS in humans are approximately 3.14 and 3.36 years, respectively, whereas those in monkeys, mice, and rats are in the order of hours to months. Rats and mice also have PFOA and PFOS distribution patterns that are distinct to those in humans. The mechanisms underlying the interspecies TK differences in PFOA and PFOS are not well understood.]

4.2 Evidence relevant to key characteristics of carcinogens

This section reviews the mechanistic data for the 10 key characteristics (KCs) of carcinogens (Smith et al., 2016) encompassed by the agents PFOA and PFOS. The studies in exposed humans used to support the mechanistic evidence were evaluated for the quality of the study design, exposure assessment, and assay accuracy and precision, and were found to reflect suitable methods for human environmental epidemiological studies. The evaluated human studies also accounted for important confounding and modifying variables. The determination of the Working Group may also have been buttressed by mechanistic evidence from human primary cells and tissues or from experimental systems.

4.2.1 Is electrophilic or can be metabolically activated to an electrophile

(a) Humans

No studies in exposed humans or human cells in vitro were available to the Working Group.

(b) Experimental systems

The fluoro-carbon chain of PFOA and PFOS forms hydrophobic bonds with DNA, which may cause strand unwinding and steric hindrance for covalent binding (Lu et al., 2012, 2016a; Qin et al., 2022a). Several studies have demonstrated that PFOA binds to proteins in plasma (i.e. albumin), liver (especially L-FABP), kidney (i.e. alpha2u-globulin) and testes of rats (Vanden Heuvel et al., 1992; Luebker et al., 2002; Han et al., 2003, 2004, 2005). In addition, it has been shown that PFOA binds to cysteine residues in murine acetyl-coenzyme A (CoA) carboxylase A and B (Shao et al., 2018). PFOA has been shown to bind to rat and human albumin with similar strengths (Han et al., 2003; Messina et al., 2005b). Studies on bovine serum albumin have indicated that PFOA mainly binds by Van der Waals forces and

hydrogen bonds at the Sudlow site 1 on albumin (Yang et al., 2023). PFOA binds to haemoglobin (Perera et al., 2023). [The Working Group noted that while the binding affinity to haemoglobin in this study was comparable to the binding affinity to albumin, as found in multiple studies, this finding probably had little biological relevance, because PFOA preferentially distributed to plasma proteins in vivo (see Section 4.1).] It has been shown that PFOS binds more strongly than PFOA to rat liver L-FABP (Luebker et al., 2002).

Synopsis

[The Working Group noted that PFOA and PFOS do not appear to have electrophilic properties or to be metabolized to electrophilic compounds (see also Section 4.1.1). PFOA and PFOS form non-covalent bonds with DNA. In addition, studies have shown that PFOA can interact with proteins, possibly through noncovalent hydrogen bonds and Van der Waals forces.]

4.2.2 Is genotoxic

See <u>Tables 4.3</u> to <u>4.10</u>.

(a) Humans

(i) Exposed humans

Genotoxic effects in humans were reported in the Flemish Environment and Health Study, a cross-sectional environmental study of adolescents from the Flanders region, Belgium, exposed to various hazardous compounds, including heavy metals, polycyclic aromatic hydrocarbons (PAHs), benzene, phthalates, and organophosphate pesticides (n = 606, in the entire cohort) (Franken et al., 2017). PFOA concentrations in serum were measured in a subpopulation from Menen (Flanders, Belgium), that was the location of a shredding factory (geometric mean, 2.55 ng/mL; 95% CI, 2.44–2.65 ng/mL; *n* = 197). The study participants were recruited between May 2010 and February 2011. [The Working Group noted that the PFOA concentrations in

serum were consistent with background levels of PFOA, indicating that the participants from Menen were not highly exposed.] The study showed a positive association between serum levels of PFOA and DNA strand breaks in leukocytes, measured using the alkaline comet assay (9% interquartile range, IQR; 95% CI, 1.5-17%, adjusted for sex, age, smoking status and maximum temperature, 7 days before sample collection). However, the positive association was not statistically significant after controlling for multiple comparisons using the method of Benjamini and Hochberg, which controlled the false discovery rate (FDR) at P = 0.05 (Franken et al., 2017). [The Working Group noted that PFOA serum concentrations were not correlated (Pearson correlation test) with marker levels of exposure to heavy metals, PAHs, benzene, phthalate, and organophosphate pesticides.]

[The Working Group noted that the study by Franken et al. (2017) had reliable measurements of exposure biomarkers and DNA damage end-points, and the positive association between PFOA and DNA strand breaks did not appear to be confounded by heavy metals, PAHs, benzene, phthalate, or organophosphate pesticides, and it was not statistically significant after controlling for multiple comparisons. The DNA damage was not mediated by oxidative stress, because no association between PFOA exposure and damaged DNA was observed using the formamidopyrimidine-DNA glycosylase (Fpg)-modified comet assay in leukocytes (see also Section 4.2.5).]

DNA fragmentation in spermatozoa has been assessed in two studies that aimed to evaluate semen quality and male fertility. One study recruited male partners of pregnant women at their first antenatal care visit to hospitals in Greenland (n = 199), Poland (n = 197), and Ukraine (n = 208) (Specht et al., 2012). The exposure levels differed between the populations, as reflected in the serum concentrations of PFOA (mean concentrations, 4.8, 5.1, and 1.8 ng/mL) and PFOS (51.9, 18.6, and 8.1 ng/mL in the populations from Greenland, Poland, and Ukraine, respectively). There was no association of the PFOA and PFOS serum concentrations with DNA fragmentation index in spermatozoa of this population of fertile men (Specht et al., 2012).

Another study recruited male partners of couples at their first visit to a reproductive medical clinic in Nanjing, China (Pan et al., 2019). The population was described as heterogeneous because it included both men with fertility issues and fertile men who were partners to infertile women (n = 664). The study included PFOA and PFOS measurements in serum and semen. There were similar serum concentrations of PFOA (median, 8.6 ng/mL; lower and upper cut-off values, 6.8 ng/mL and 11.0 ng/mL) and PFOS (median, 8.4 ng/mL; lower and upper cut-off values, 5.6 ng/mL and 13.1 ng/mL). Likewise, semen concentrations were similar for PFOA (median, 0.23 ng/mL; lower and upper cut-off values, 0.15 ng/mL and 0.36 ng/mL) and PFOS (median, 0.10 ng/mL; lower and upper cut-off values, 0.06 ng/mL and 0.18 ng/mL). There were strong correlations between serum and semen concentrations of PFOA (r = 0.70; P < 0.001) and PFOS (r = 0.8; P < 0.001). Serum concentrations of PFOA and PFOS or the increase per 1-unit increase in the log-transformed concentration were not associated with the DNA fragmentation index (PFOA: $\beta = 0.046$; 95% CI, -0.052 to 0.144; PFOS: $\beta = 0.040$; 95% CI, -0.037 to 0.116), adjusted for age, body mass index (BMI), BMI², smoking, alcohol intake, and the duration of abstinence from both, whereas there were positive associations between the DNA fragmentation index and concentrations of PFOA $(\beta = 0.136; 95\% \text{ CI}, 0.064-0.209)$ and PFOS $(\beta = 0.087; 95\% \text{ CI}, 0.033 - 0.142)$ in seminal fluid (Pan et al., 2019). [The Working Group considered that the two studies were less relevant than the findings reported in the study by Franken et al. (2017) described above, because DNA fragmentation was measured using the sperm

End-point (assay)	Cells	Result ^a	Concentration (LEC or HIC)	Comments	Reference
DNA strand breaks (comet assay)	HepG2	+	1.9 μΜ	Positive control included (H ₂ O ₂)	<u>Ojo et al.</u> (2022a)
DNA strand breaks (comet assay)	HepG2	+	10 µM	Positive control included (H_2O_2)	<u>Wielsøe et al.</u> (2015)
DNA strand breaks (comet assay)	TK6	+	125 μg/mL [0.3 μM]	No positive control included	<u>Yahia et al.</u> (2016)
DNA strand breaks (comet assay)	HepG2	+	50 µM	No positive control included	<u>Yao and</u> <u>Zhong (2005)</u>
DNA strand breaks (comet assay)	Sperm (primary cells)	-	1000 µM	Positive control included (H_2O_2)	<u>Emerce and</u> <u>Çetin (2018)</u>
DNA strand breaks (comet assay)	HepG2	-	$400\mu M$	Positive control included (Ro19-8022 + light)	<u>Eriksen et al.</u> (2010)
DNA strand breaks (comet assay)	HepG2	-	400 µM	Positive control included (H ₂ O ₂)	<u>Florentin et al.</u> (2011)
DNA double- strand breaks (γH2AX assay)	НаСаТ	(+)	50 μΜ	Single concentration; no positive control included	<u>Peropadre</u> et al. (2018)
Chromosomal aberrations	Primary lymphocytes (from males)	-	2010 μg/mL	Increased frequency of endoreduplication in the cells observed at an exposure level that also caused a significant decrease in mitotic index; positive controls included (mitomycin C and cyclophosphamide for assays without or with metabolic activation, respectively)	<u>US EPA</u> (1996a)
Chromosomal aberrations	Primary lymphocytes (from males)	_	100 μg/mL	Positive controls included (mitomycin C and cyclophosphamide for assays without or with metabolic activation, respectively)	<u>Butenhoff</u> et al. (2014)
Micronuclei (CBMN assay)	HepG2	-	400 μΜ	Positive control included (mitomycin C)	<u>Florentin et al.</u> (2011)
Micronuclei (CBMN assay)	HepG2	+	100 µM	No positive control included	<u>Yao and</u> <u>Zhong (2005)</u>

Table 4.3 End-	points relevant to	aenotoxicity	in human cell	s in vitro ex	posed to PFOA
I GIALC ILA ELLA	pointe reretailte	genotomenty		5 111 1111 0 67	

CBMN, cytokinesis-block micronucleus assay; H_2O_2 , hydrogen peroxide; HIC, highest ineffective concentration; LEC, lowest effective concentration; PFOA, perfluorooctanoic acid; γ H2AX, γ -H2A histone family member X.

^a –, negative; +, positive; (+), positive in a study of limited quality.

chromatin structure assay, which can be considered an indirect indicator of DNA damage.]

(ii) Human cells in vitro

See <u>Table 4.3</u> and <u>Table 4.4</u>.

No studies conducted in human tissues were available. Four studies aiming to assess genotoxic effects induced by PFOA or PFOS treatment were conducted in human primary cells. The amount of DNA strand breaks (represented by DNA % tail as measured by comet assay) was not altered in human sperm cells by exposure to PFOA or PFOS (<u>Emerce and Cetin, 2018</u>). Exposure to PFOA or PFOS did not affect the frequency of chromosomal aberrations in human lymphocytes (<u>US EPA, 1996a</u>, <u>1999b</u>; <u>Butenhoff et al.</u>, <u>2014</u>).

Six studies have assessed the formation of DNA strand breaks using the comet assay in human hepatoma (HepG2) and in human lymphoblastoid permanent TK6 cells; four of these studies have shown high levels of

End-point (assay)	Cells	Result ^a	Concentration (LEC or HIC)	Comments	Reference
DNA strand breaks (comet assay)	HepG2	+	0.8 μΜ	Positive control included (H ₂ O ₂)	<u>Ojo et al. (2022a)</u>
DNA strand breaks (comet assay)	HepG2	+	0.2 μΜ	Positive control included (H_2O_2)	<u>Wielsøe et al.</u> (2015)
DNA strand breaks (comet assay)	Sperm	-	1000 µM	Positive control included (H_2O_2)	<u>Emerce and</u> <u>Çetin (2018)</u>
DNA strand breaks (comet assay)	HepG2	-	$400 \ \mu M$	Positive assay control included (THP-1 cells treated with Ro19-8022 + light)	<u>Eriksen et al.</u> (2010)
DNA strand breaks (comet assay)	HepG2	-	300 µM	Positive controls included (H ₂ O ₂ and benzo[<i>a</i>]pyrene)	<u>Florentin et al.</u> (2011)
Chromosomal aberrations	Primary lymphocytes (from males)	_	349 μg/mL	Negative with and without metabolic activation; positive controls included (mitomycin C and cyclophosphamide); one experiment with technical replicates	<u>US EPA (1999b)</u>
Micronuclei (CBMN assay)	HepG2	-	300 µM	Positive control included (mitomycin C)	<u>Florentin et al.</u> (2011)

Table 4.4 End-	points relevant to (aenotoxicity i	n human cells in	vitro exp	oosed to PFOS
TUNIC TIT LINA		genotoxicity i	II IIMIIIMII CCIIJ III		

 $CBMN, cytokinesis-block\ micronucleus\ assay;\ H_2O_2,\ hydrogen\ peroxide;\ HIC,\ highest\ ineffective\ concentration;\ LEC,\ lowest\ effective\ concentration;\ DEC,\ perfluorooctanesulfonic\ acid.$

^a -, negative; +, positive.

genotoxicity (Yao and Zhong, 2005; Wielsøe et al., 2015; Yahia et al., 2016; Ojo et al., 2022a). The other two studies showed unaltered levels of DNA strand breaks in human HepG2 cells after exposure to both PFOA and PFOS when compared with a positive control group (Eriksen et al., 2010; Florentin et al., 2011). No change in the frequencies of chromosome aberrations and micronuclei in the HepG2 cells was observed after exposure to PFOA and PFOS in one of these studies (Florentin et al., 2011). Yao and Zhong instead reported an increase in the frequency of micronuclei in HepG2 cells after PFOA exposure (Yao and Zhong, 2005).

Exposure to PFOA induced an increase in the frequency of phosphorylated H2A histone family member X (γ H2AX) foci in human HaCaT keratinocytes, which is an indicator of either DNA double-strand breaks or stalled replication forks (Peropadre et al., 2018).

(b) Experimental systems

(i) Non-human mammalian systems

See <u>Table 4.5</u> (PFOA in vivo), <u>Table 4.6</u> (PFOS in vivo), <u>Table 4.7</u> (PFOA in vitro), <u>Table 4.8</u> (PFOS in vitro).

DNA strand breaks (measured using the comet assay) were not induced in the liver and testes of mice exposed to PFOA in drinking-water (Crebelli et al., 2019). However, in the oocytes of mice that were exposed orally to PFOA there was an increased frequency of yH2AX foci, which is an indicator of either DNA double-strand breaks or stalled replication forks (Zhang et al., 2022a). No increases in the frequency of micronuclei in blood reticulocytes or spleen lymphocytes were observed in mice (Crebelli et al., 2019). In addition, oral exposure to PFOA did not alter the frequency of micronucleated bone marrow cells in mice (US EPA, 1995b, 1996d; Butenhoff et al., 2014) (see Table 4.5). [The Working Group considered that the negative results from the micronucleus assays were reliable, because the
End-point (assay)	Species, strain (sex)	Tissue(s)	Result ^a	Dose (LED, HID)	Route, duration, dosing regimen	Comments	Reference
DNA strand breaks (comet assay)	Mouse, C57BL/6 (M)	Liver, testis	_	5 mg/kg	Oral, 5 wk, drinking-water	Positive control included (MMS)	<u>Crebelli et al.</u> (2019)
DNA double strand breaks (γH2AX assay)	Mouse, ICR (F)	Oocytes	(+)	5 mg/kg per day	Oral, 28 days, drinking-water	Method for the detection of γ H2AX foci was not described; no positive control group	<u>Zhang et al.</u> (2022a)
Micronuclei	Mouse, C57BL/6 (M)	Blood (reticulocytes) and spleen (lymphocytes)	-	5 mg/kg	Oral, 5 wk, drinking-water	Positive control included (MMS)	<u>Crebelli et al.</u> (2019)
Micronuclei	Mouse, ICR (F and M)	Bone marrow	-	1000 mg/kg	Oral, 24–72 h, single dose	Positive control included (cyclophosphamide)	<u>Butenhoff</u> <u>et al. (2014)</u>
Micronuclei	Mouse, Crl:CD-1 (ICR) BR (F and M)	Bone marrow	-	1990 mg/kg	Oral, single dose, killed 24, 48 or 72 h post-exposure	Positive control included (cyclophosphamide)	<u>US EPA</u> (1996d)
Micronuclei	Mouse, Crl:CD-1 (ICR) BR (F and M)	Bone marrow	_	5000 mg/kg	Oral, single dose, killed 24, 48 or 72 h post-exposure	Positive control included (cyclophosphamide)	<u>US EPA</u> (1995b)

Table 4.5 End-points relevant to genotoxicity in non-human mammals in vivo exposed to PFOA

F, female; h, hour(s); γ H2AX, γ -H2A histone family member X; HIC, highest ineffective concentration; HPLC-ECD, high-performance liquid chromatography-electrochemical detection; LEC, lowest effective concentration; M, male; MMS, methyl methanesulfonate; PFOA, perfluorooctanoic acid; wk, week(s).

^a –, negative; +, positive; (+), positive in a study of limited quality.

Table 4.6 End-points relevant to genotoxicity in non-human mammals in vivo exposed to PFOS

End-point (assay)	Species, strain (sex)	Tissue	Resultª	Dose (LED, HID)	Route, duration, dosing regimen	Comments	Reference
DNA strand breaks (comet assay)	Rat, Wistar (M)	Bone marrow	+	0.6 mg/kg	Oral, 30 d, gavage at 48-h intervals	Positive control group included (mitomycin C)	<u>Çelik et al.</u> (2013)
DNA strand breaks (comet assay)	Rat, Wistar (M)	Leukocytes (whole blood)	+	0.6 mg/kg	Oral, 30 d, gavage at 48-h intervals	Positive control group included (mitomycin C)	<u>Eke and Çelik</u> (2016)
DNA strand breaks (comet assay)	Rat, Wistar (M)	Liver	+	0.6 mg/kg	Oral, 30 d, gavage at 48-h intervals	Positive control group included (mitomycin C)	<u>Eke et al. (2017)</u>
Mutations (<i>Spi</i> - assay)	Mice, <i>gpt</i> delta transgenic (M)	Liver	-	10 mg/kg per day	Oral, 28 d, gavage at 24-h intervals	No positive control group	<u>Wang et al.</u> (2015a)
Micronuclei	Mice, <i>gpt</i> delta transgenic (M)	Bone marrow	-	10 mg/kg per day	Oral, 28 d, gavage at 24-h intervals	No positive control group	<u>Wang et al.</u> (2015a)
Micronuclei	Rat, Sprague- Dawley (F and M)	Erythrocytes (blood)	_	5 mg/kg per day	Oral, 28 d, gavage once daily	A slightly increased percentage of micronucleated cells in female rats was related to PFOS-induced bone-marrow toxicity	<u>NTP (2019)</u>
Micronuclei	Rat, Wistar (M)	Bone marrow	+	1.25 mg/kg	Oral, 30 d, gavage at 48-h intervals	Positive control group included (mitomycin C)	<u>Çelik et al.</u> (2013)
Micronuclei	Rat, Wistar (M)	Leukocytes (whole blood)	+	0.6 mg/kg	Oral, 30 d, gavage at 48-h intervals	Positive control group included (mitomycin C)	<u>Eke and Çelik</u> (2016)
Micronuclei	Rat, Wistar (M)	Liver	+	0.6 mg/kg	Oral, 30 d, gavage at 48-h intervals	Positive control group included (mitomycin C)	<u>Eke et al. (2017)</u>
Micronuclei	Mouse, Crl:CD-1 (ICR) BR (F and M)	Bone marrow	-	950 mg/kg	Oral, single dose, killed 24, 48 or 72 h post-exposure	Positive control group included (cyclophosphamide)	<u>US EPA (1996e)</u>

d, day(s); F, female; h, hour(s); HIC, highest ineffective concentration; LEC, lowest effective concentration; M, male; PFOS, perfluorooctanesulfonic acid. ^a –, negative; +, positive.

Table 4.7 End-points relevant to	genotoxicity in non-huma	n mammalian cells in vitro	exposed to PFOA
----------------------------------	--------------------------	----------------------------	-----------------

End-point (assay)	Cells	Result ^a	Concentration (LEC or HIC)	Comments	Reference
DNA strand breaks (comet assay)	Oocytes (porcine)	(+)	40 μΜ	Increased comet tail length, although changes in tail intensity and tail moment were not statistically significant; it was uncertain whether the statistical analysis was based on all comets ($n = 250$) or independent experiments; no positive control included	<u>Mario et al.</u> (2022)
DNA strand breaks (comet assay)	SHE cells	-	300 µM	Positive control included (H ₂ O ₂)	<u>Jacquet et al.</u> (2012a)
DNA strand breaks (comet assay)	Rat testicular cells	-	300 µM	Positive control included (1,2-dibromo-3-chloropropane)	<u>Lindeman</u> <u>et al. (2012)</u>
DNA double- strand breaks (γH2AX assay)	Mouse oocytes	(+)	300 µM	No positive control included in the experiments; toxicity at the highest concentration (approximately 30% cell death and decreased oocyte maturation at lowest concentration)	<u>Guo et al.</u> (2021a)
DNA double- strand breaks (γH2AX assay)	Mouse oocytes	(+)	200 µM	Single concentration; statistical analysis was based on all oocytes $(n = 50-58)$, rather than independent experiments; no positive control group included	<u>Zhou et al.</u> (2022)
HGPRT locus mutations	CHO-K1	-	39 μg/mL [90 μM]	Positive control included (ethylmethanesulfoxide and dimethylbenzanthracene for the test conditions without and with metabolic activation, respectively)	<u>US EPA</u> (2002)
CD59 locus mutations	Human– hamster hybrid (A _L) cells	+	200 μΜ	Mutagenic effect observed after extended period of exposure (16 days, but not after 1, 4 or 8 days of exposure); no positive control group included	<u>Zhao et al.</u> (2011a)
Micronuclei (assay not specified)	CHL V79 fibroblasts	_	10 µM	Single concentration; positive controls included (ethylmethylsulfonate and cyclophosphamide)	<u>Buhrke et al.</u> (2013)
Chromosomal aberrations	CHO cells	-	996 μg/mL	Positive control included (mitomycin C and cyclophosphamide for the test conditions without and with metabolic activation, respectively)	<u>Butenhoff</u> <u>et al. (2014)</u>
Chromosomal aberrations	CHO cells	-	2250 μg/mL	Negative without metabolic activation; increased frequency of endoreduplication in cells and polyploidy in cells with metabolic activation; positive controls included (mitomycin C and cyclophosphamide)	<u>US EPA</u> (1996b)
Chromosomal aberrations	CHO cells	+ (with metabolic activation)	3740 μg/mL	Induction of chromosomal aberrations in cells with metabolic activation; negative without metabolic activation; positive controls included (mitomycin C and cyclophosphamide)	<u>US EPA</u> (1996c)

CHL, Chinese hamster lung; CHO, Chinese hamster ovary; γ H2AX, γ -H2A histone family member X; H_2O_2 , hydrogen peroxide; HIC, highest ineffective concentration; HGPRT, hypoxanthine-guanine phosphoribosyltransferase; LEC, lowest effective concentration; PFOA, perfluorooctanoic acid; SHE, Syrian hamster embryo.

^a –, negative; +, positive; (+), positive in a study of limited quality.

Table 4.8 End-points relevant to genotoxicity in non-human mammalian cells in vitro exposed to PFOS

End-point (assay)	Cells	Result ^a	Concentration (LEC or HIC)	Comments	Reference
DNA double-strand breaks (γH2AX assay)	Porcine ovary	(+)	100 μΜ	Single concentration; no positive control included; the statistical analysis was based on all the oocytes ($n = 27-30$), rather than independent experiments	<u>Chen et al. (2021)</u>
DNA strand breaks (comet assay)	Sperm (boar)	(+)	461 µM	Single concentration; no positive control included	<u>Oseguera-López</u> <u>et al. (2020)</u>
DNA double-strand breaks (γH2AX assay; western blot)	MEF	(+)	20 µM	Genotoxic response occurred at concentrations with < 50% viability; no positive control included	<u>Wang et al. (2015a)</u>
DNA strand breaks (comet assay)	SHE cells	-	93 µM	Positive control included (H_2O_2)	<u>Jacquet et al.</u> (2012b)
Unscheduled DNA synthesis	Rat hepatocytes	-	25 μg/mL	Positive control included (2-acetylaminofluorene)	<u>US EPA (1999c)</u>
Mutations (<i>Spi</i> - mutation assay; <i>redBA/</i> <i>gam</i> locus)	MEF	+	10 μΜ	No positive control included	<u>Wang et al. (2015a)</u>
Mutations (CD59- mutants)	Human–hamster (A _L) hybrid cells	-	200 µM	No positive control included	<u>Wang et al. (2013)</u>
Micronuclei	CHL fibroblasts (V79)	-	12.5 μg/mL	Single concentration with metabolic activation (S9 mix); positive control included (cyclophosphamide)	<u>Jernbro et al.</u> (2007)
Aneuploidy	Mouse oocytes	(+)	600 µM	Single concentration; aneuploidy caused by dysfunction of spindle assembly and chromosome alignment in mitosis; no positive control included	<u>Wei et al. (2021)</u>

CHL, Chinese hamster lung; H₂O₂, hydrogen peroxide; γH2AX, γ-H2A histone family member X; HIC, highest ineffective concentration; LEC, lowest effective concentration;

MEF, mouse embryonic fibroblasts; PFOS, perfluorooctane sulfonic acid; S9, 9000 $\times g$ supernatant; SHE, Syrian hamster embryo.

^a –, negative; +, positive; (+), positive, in a study of limited quality.

studies included positive controls for genotoxic effects (<u>Butenhoff et al., 2014</u>; <u>Crebelli et al.,</u> 2019).]

In three consecutive studies performed in rats by the same research group, oral exposure to PFOS was shown to induce an increase in DNA strand breaks, measured using the comet assay, and in the frequency of micronuclei in bone marrow cells (<u>Celik et al., 2013</u>), leukocytes (whole blood) (<u>Eke and Celik, 2016</u>), and liver (<u>Eke et al., 2017</u>).

One study assessed genotoxic effects in gpt transgenic mice, which were exposed by oral gavage to PFOS at 1.5, 4, or 10 mg/kg once daily for 28 days (Wang et al., 2015a). Compared with the control group, there were higher frequencies of mutations in the liver (2.2 and 6.8 λ mutants/106 plaques) and of micronucleated polychromatic erythrocytes in the bone marrow (3.0% and 2.9%, equal to 0.52- and 0.43-fold) at the two highest doses, although the increase was not statistically significant (Wang et al., 2015a). The Working Group noted that the authors considered the fold increases as evidence that PFOS is mutagenic in vivo. However, the greater mutagenic effect in the highest dose group may have been because of an outlier, and genotoxic effects occurred only at doses that caused increases in the serum levels of alkaline phosphatase and alanine aminotransferase, suggesting genotoxicity may have been because of tissue toxicity.]

The study presented in a report by the National Toxicology Program (NTP) showed no effect on micronucleated cells in the peripheral blood of rats exposed orally (NTP, 2019). Lastly, one study showed unaltered frequency of micronuclei in bone marrow cells after a single oral administration of PFOS in mice (US EPA, 1996e) (see Table 4.6).

Two in vitro studies showed that PFOA did not induce the formation of DNA strand breaks, assessed using the comet assay, in rodent cells (Jacquet et al., 2012a; Lindeman et al., 2012). In contrast, more recently, <u>Mario et al. (2022)</u> reported that PFOA exposure generated DNA strand breaks, assessed using the comet assay in porcine oocytes. [The Working Group noted that the study was of limited relevance because the statistical analysis seemed to be based on individual comets, rather than the mean results from independent experiments, and the study did not include a positive control group.]

In two different studies in mouse oocytes, PFOA treatment induced an increase in γ H2AX foci in comparison with the negative control (Guo et al., 2021a; Zhou et al., 2022). [The Working Group noted that the statistical analysis in the study of Zhou et al. (2022) also seemed to be based on individual comets, rather than the mean results from independent experiments, and the two studies did not include a positive control group.]

US EPA (2002) reported no increase in mutation frequency in Chinese hamster ovary CHO-K1 cells, using the hypoxanthine-guanine phosphoribosyl transferase assay. An increase in mutation frequency was observed at the CD59 locus in human-hamster hybrid (A_1) cells exposed to PFOA for 16 days, whereas a shorter exposure to PFOA for 1, 4, or 8 days was not associated with increases in mutation (Zhao et al., 2011a). No increases of chromosome aberrations or in the frequency of micronuclei were observed in PFOA-exposed rodent cells (US EPA, 1996b; Buhrke et al., 2013; Butenhoff et al., 2014). An increase of chromosome aberrations was observed in another study in CHO cells in the presence of metabolic activation (US EPA, 1996c) (see Table 4.7).

In three in vitro studies, exposure to PFOS induced DNA strand breaks (assessed using the comet assay) and γ H2AX foci in porcine ovary cells, sperm of boar, and mouse embryonic fibroblasts (Wang et al., 2015a; Oseguera-López et al., 2020; Chen et al., 2021), whereas, in a fourth study, exposure to the potassium salt of PFOS did

not alter the level of DNA strand breaks in Syrian hamster embryo cells (Jacquet et al., 2012b).

PFOS did not induce unscheduled DNA synthesis in primary cultures of rat hepatocytes (US EPA, 1999c). One study showed an increase in mutation frequency (*spi* mutation assay) in mouse embryonic fibroblasts (Wang et al., 2015a). In human-hamster (A_I) hybrid cells, exposure to PFOS did not alter the frequency of CD59- mutants (Wang et al., 2013). [The Working Group noted that a positive control group was not included in the study.]

No change in the frequency of micronuclei was reported in Chinese hamster lung fibroblasts (V79) (Jernbro et al., 2007). Lastly, Wei and colleagues observed aneuploidy in oocytes from mice after exposure to the potassium salt of PFOS, which was speculated to be caused by dysfunctions of spindle assembly and chromosome alignment in mitosis (Wei et al., 2021) [The Working Group noted that aneuploidy encompasses both gain and loss of chromosomes. However, the study had some limitations: the statistical analysis of aneuploidy appeared to be based on the number of oocytes (group sizes of 20 and 21 oocytes), rather than results from independent experiments, and the study did not include a positive control.]

(ii) Prokaryotes and other speciesSee <u>Table 4.9</u> (PFOA) and <u>Table 4.10</u> (PFOS).

Prokaryotes

PFOA has not been found to be mutagenic in various *Escherichia coli* and *Salmonella typhimurium* tester strains (Griffith and Long, 1980; US EPA, 1995a, 1996f; Oda et al., 2007; Fernández Freire et al., 2008; Buhrke et al., 2013; Butenhoff et al., 2014). PFOS has given negative results for mutagenicity in various *E. coli* and *S. typhimurium* tester strains (Simmon and Marx, 1978; US EPA, 1979, 1999a; Oda et al., 2007; NTP, 2019). [The Working Group noted that PFOA and PFOS might not readily enter cells in the absence of appropriate transporters, which could be the case for the prokaryotic systems and for *Saccharomyces cerevisiae*.]

Lower eukaryotes

Exposure of *Paramecium caudatum* (unicellular freshwater protozoa) to PFOA was associated with increased levels of DNA strand breaks (<u>Kawamoto et al., 2010</u>). PFOA exposure did not alter the frequency of mitotic recombination in *S. cerevisiae* (<u>Butenhoff et al., 2014</u>).

Exposure to PFOS did not affect levels of DNA strand breaks in *P. caudatum* (Kawamoto et al., 2010) and did not induce mitotic recombination in *S. cerevisiae* (Simmon and Marx, 1978; US EPA, 1979).

Other species

Studies on PFOA have shown increased levels of DNA strand breaks in earthworms (*Eisenia fetida*) (Zheng et al., 2016; Wang et al., 2021a), green mussels (*Perna viridis*) (Liu et al., 2014a, b), planarians (*Dugesia japonica*) (Zhang et al., 2020a), and daphnia (*Daphnia carinata*) (Logeshwaran et al., 2021), whereas exposure to PFOA in the common carp (*Cyprinus carpio*) was not associated with changes in levels of DNA strand breaks (Kim et al., 2010). Increased micronuclei frequency was observed in haemolymph cells of PFOA-exposed *Perna viridis* (Liu et al., 2014b).

DNA strand breaks, measured using the comet assay, were increased after exposure to PFOS in *Cyprinus carpio* (Kim et al., 2010), zebrafish (*Dario rerio*) (Du et al., 2016), *Eisenia fetida* (Xu et al., 2013a; Zheng et al., 2016), *Perna viridis* (Liu et al., 2014a, b), *Dugesia japonica* (Shao et al., 2019), *Daphnia carinata* (Logeshwaran et al., 2021), and *Allium cepa* (Sivaram et al., 2021). One study showed unaltered levels of DNA strand breaks in brain and liver cells from

End-point (assay)	Species (cell type)	Result ^a	Concentration (LEC or HIC)	Comments	Reference
Prokarvotes			(120 01 1110)		
Mutations	<i>Salmonella typhimurium</i> (TA98, TA100, TA1535, TA1537, and TA1538)	-	5 μmol/plate	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls	<u>Buhrke et al.</u> (2013)
Mutations	Salmonella typhimurium (TA98, TA100, TA1535, and TA1537) and Escherichia coli WP2 uvrA	-	1000 μg/plate	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls	<u>Butenhoff et al.</u> (2014)
Mutations	<i>Salmonella typhimurium</i> (TA98, TA100, TA102, and TA104)	-	500 µM	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls	<u>Fernández Freire</u> <u>et al. (2008)</u>
Mutations	<i>Salmonella typhimurium</i> (TA98, TA100, TA1535, TA1537, and TA1538)	-	500 μg/plate	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls	<u>Griffith and Long</u> (1980)
Mutations	<i>Salmonella typhimurium</i> (TA98, TA100, TA1535, and TA1537)	-	5 mg/plate	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls	<u>US EPA (1995a)</u>
Mutations	<i>Salmonella typhimurium</i> (TA98, TA100, TA1535, and TA1537)	-	5 mg/plate	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls	<u>US EPA (1996f)</u>
Mutations	Salmonella typhimurium TA1535/ pSK1002 (umu test)	-	1000 μΜ	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls	<u>Oda et al. (2007)</u>
Lower eukaryotes					
Mitotic recombination	Saccharomyces cerevisiae D4	-	500 µM	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls	<u>Butenhoff et al.</u> (2014)
DNA strand breaks (comet assay)	Paramecium caudatum	(+)	100 μΜ	Positive control groups included (2-aminoanthracene and MNNG)	<u>Kawamoto et al.</u> (2010)
Other species					
DNA strand breaks (comet assay)	<i>Eisenia fetida</i> (in the coelomocytes cells)	+	10 mg/kg soil	No positive control group included	<u>Wang et al. (2021a)</u>
DNA strand breaks (comet assay)	<i>Eisenia fetida</i> (in the coelomocytes cells)	+	600 mg/kg soil	No positive control included; there was uncertainty about whether the statistical analysis was based on all comets, independent experiments, or individual specimens	<u>Zheng et al. (2016)</u>
DNA strand breaks (comet assay)	Perna viridis (haemolymph)	+	1000 μg/mL	No positive control included	<u>Liu et al. (2014a)</u>

Table 4.9 End-points relevant to genotoxicity in non-mammalian systems exposed to PFOA

Table 4.9 (continued)

End-point (assay)	Species (cell type)	Result ^a	Concentration (LEC or HIC)	Comments	Reference
DNA strand breaks (comet assay)	Perna viridis (haemolymph)	+	1000 μg/mL	No positive control included	<u>Liu et al. (2014b)</u>
DNA strand breaks (comet assay)	Dugesia japonica	(+)	15 μg/mL	No positive control included; there was uncertainty about whether the statistical analysis was based on all comets, independent experiments, or individual specimens	<u>Zhang et al.</u> (<u>2020a)</u>
DNA strand breaks (comet assay)	Daphnia carinata	+	10 μg/mL	Positive control included (H_2O_2)	<u>Logeshwaran et al.</u> (2021)
DNA strand breaks (comet assay)	<i>Cyprinus carpio</i> (blood cells)	_	50 μg/mL	No positive control included	<u>Kim et al. (2010)</u>
Micronuclei	Perna viridis (haemolymph)	+	1000 μg/mL	No positive control included	<u>Liu et al. (2014b)</u>

H₂O₂, hydrogen peroxide; HIC, highest ineffective concentration; LEC, lowest effective concentration; MNNG, *N*-methyl-*N'*-nitro-*N*-nitrosoguanidine; PFOA, perfluorooctanoic acid. *+, positive; -, negative; (+), positive in a study of limited quality.

End-point (assay)	Species (cell type)	Result ^a	Concentration (LEC or HIC)	Comments	Reference
Prokaryotes					
Mutations	Salmonella typhimurium (TA98 and TA100) and Escherichia coli WP2 uvrA/ pKM101	_	10 mg/plate	Negative with and without metabolic activation; genotoxic effect by exposure to positive controls	<u>NTP (2019)</u>
Mutations	Salmonella typhimurium TA1535/pSK1002 (umu test)	-	1000 μΜ	Negative with and without metabolic activation; genotoxic effect by exposure to positive controls	<u>Oda et al. (2007)</u>
Mutations	Salmonella typhimurium (TA98, TA100, TA1535, TA1537, and TA1538)	_	5 mg/plate	Negative with and without metabolic activation; genotoxic effect by exposure to positive controls; report contains results from two compounds (T-2247 and T-2248) that were not identified by CAS No. ^b	<u>Simmon and</u> <u>Marx (1978)</u>
Mutations	<i>Salmonella typhimurium</i> (TA98, TA100, TA1535, TA1537, and TA1538)	-	2 mg/plate	Negative with and without metabolic activation; genotoxic effect by exposure to positive controls; results from one independent experiment, except TA100 ($n = 2$)	<u>US EPA (1979)</u>
Mutations	Salmonella typhimurium (TA98, TA100, TA1535, and TA1537) and Escherichia coli WP2 uvrA	_	5 mg/plate	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls	<u>US EPA (1999a)</u>
Lower eukaryotes					
Mitotic recombination	Saccharomyces cerevisiae D3	_	5% (w/v or v/v)	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls; report contains results from two compounds (T-2247 and T-2248) that are not identified by CAS No. ^b	<u>Simmon and</u> <u>Marx (1978)</u>
Mitotic recombination	Saccharomyces cerevisiae D4	-	0.5 mg/plate	Negative with and without metabolic activation; genotoxic effect of exposure to positive controls; result from one experiment	<u>US EPA (1979)</u>
DNA strand breaks (comet assay)	Paramecium caudatum	-	100 μΜ	Positive control groups included (2-aminoanthracene and MNNG)	<u>Kawamoto et al.</u> (2010)
Other species					
DNA strand breaks (comet assay)	<i>Cyprinus carpio</i> (blood cells)	+	5 μg/mL	No positive control included	<u>Kim et al. (2010)</u>
DNA strand breaks (comet assay)	<i>Danio rerio</i> (blood cells)	+	0.4 μg/mL	No chemical positive control included (exposure to ZnO nanoparticles was associated with increased number of DNA strand breaks)	<u>Du et al. (2016)</u>

Table 4.10 End-points relevant to genotoxicity in non-mammalian systems exposed to PFOS

Table 4.10 (continued)

End-point (assay)	Species (cell type)	Result ^a	Concentration (LEC or HIC)	Comments	Reference
DNA strand breaks (comet assay)	<i>Eisenia fetida</i> (in the coelomocytes cells)	(+)	0.25 µg/cm ²	No positive control included; there was uncertainty about whether the statistical analysis was based on all comets, independent experiments, or individual specimens	<u>Xu et al. (2013a)</u>
DNA strand breaks (comet assay)	<i>Eisenia fetida</i> (in the coelomocytes cells)	(+)	470 mg/kg soil	No positive control included; there was uncertainty about whether the statistical analysis was based on all comets, independent experiments, or individual specimens	<u>Zheng et al.</u> (2016)
DNA strand breaks (comet assay)	<i>Perna viridis</i> (haemolymph)	+	10 μg/mL	No positive control included	<u>Liu et al. (2014a)</u>
DNA strand breaks (comet assay)	<i>Perna viridis</i> (haemolymph)	+	10 μg/mL	No positive control included	<u>Liu et al. (2014b)</u>
DNA strand breaks (comet assay)	Dugesia japonica	+	5 μg/mL	No positive control included	<u>Shao et al. (2019)</u>
DNA strand breaks (comet assay)	Daphnia carinata	+	1 μg/mL	Positive control included (H ₂ O ₂)	<u>Logeshwaran</u> <u>et al. (2021)</u>
DNA strand breaks (comet assay)	Allium cepa	+	25 μg/mL	Positive control included (benzo[<i>a</i>]pyrene)	<u>Sivaram et al.</u> (2021)
DNA strand breaks (DNA precipitation assay)	<i>Larus michahellis</i> (brain and liver cells of embryos)	_	200 ng/g egg weight	No positive control included	<u>Parolini et al.</u> (2016)
Mutations	λ transgenic medaka (liver)	+	6.7 ng/mL	Mutation spectrum encompassed mainly +1 frameshift mutations	<u>Chen et al. (2016)</u>
Chromosomal aberrations	Allium cepa	+	25 μg/mL	Positive control included (benzo[<i>a</i>]pyrene)	<u>Sivaram et al.</u> (2021)
Micronuclei	<i>Danio rerio</i> (blood cells)	+	0.8 µg/mL	No chemical positive control included (exposure to ZnO nanoparticles was associated with increased number of DNA strand breaks)	<u>Du et al. (2016)</u>
Micronuclei	<i>Perna viridis</i> (haemolymph)	+	100 μg/mL	No positive control included	<u>Liu et al. (2014b)</u>

CAS No., Chemical Abstracts Service Registry Number; H₂O₂, hydrogen peroxide; HIC, highest ineffective concentration; LEC, lowest effective concentration; MNNG, *N*-methyl-*N*'-nitro-*N*-nitrosoguanidine; PFOS, perfluorooctanesulfonic acid; w/v, weight per volume; w/w, weight per weight; ZnO, zinc oxide.

^a +, positive; –, negative; (+), positive in a study of limited quality.

^b The compounds have been identified (<u>OECD, 2002</u>) as a 50% (by weight) solution of the diethanolammonium salt of perfluorooctanesulfonate in water (T-2247 CoC) and 22.5% of a reaction product of ethyl and methyl methacrylates and 22.5% of the pyridinium chloride salt of an *N*-methylperfluorooctanesulfonamidoethanol-based glutaryl amide (T-2248 CoC).

yellow-legged gull (*Larus michahellis*) embryos after exposure to PFOS (<u>Parolini et al., 2016</u>).

Increases in the frequencies of micronuclei and chromosome aberrations versus the control group were observed in haemolymph cells of PFOS-exposed *Perna viridis* (Liu et al., 2014b), in *Dario rerio* (Du et al., 2016), and in cells of *Allium cepa* (Sivaram et al., 2021). A higher frequency of +1 frameshift mutations was observed in the liver cells of λ transgenic medaka fish (Chen et al., 2016).

Synopsis

[The Working Group noted that there was a paucity of data in exposed humans regarding genotoxicity, especially investigating associations between PFOA exposure and DNA damage end-points such as mutations and chromosome aberrations. One study in exposed humans showed a positive association between the serum level of PFOA and DNA strand breaks in the leukocytes of adolescents in Menen, Belgium. This positive association was not statistically significant when corrected for multiple comparisons. Two studies on PFOS have shown inconsistent indices of DNA fragmentation in semen samples from exposed humans.

For both PFOA and PFOS, in vitro studies in human primary cells have shown a lack of genotoxicity. Studies in experimental systems in human cell lines and in non-human mammalian systems in vivo and in vitro have shown mixed results for various types of end-points, such as DNA strand breaks (measured with the comet assay or by counting γ H2AX foci), micronuclei, and chromosome aberrations. PFOA and PFOS did not exert mutagenic effects in prokaryotes.]

4.2.3 Alters DNA repair or causes genomic instability

(a) Humans

No studies in exposed humans or in human cells in vitro were available to the Working Group.

(b) Experimental systems

Estefanía González-Alvarez et al. (2022) reported that oral treatment with PFOA at the dose of 2.5 mg/kg body weight (bw) for 15 days altered the ovarian contents of proteins that are involved in DNA damage sensing and repair in lean (4 increases and 12 decreases in protein contents) and obese (12 increases and 6 decreases in protein contents) female mice.

In vitro exposure of rhesus monkey trophoblast stem cells to PFOA at 100 nM for 4 weeks produced only subtle effects on gene expression related to DNA damage checkpoint signalling (<u>Midic et al., 2018</u>).

PFOS has been shown to reduce the activity of polymerase α from calf thymus: half-maximal inhibitory concentration (IC₅₀), 24.5 μ M, and of the recombinant rat polymerase β , IC₅₀ = 46.4 μ M (Nakamura et al., 2007).

Synopsis

[The Working Group noted that there was a paucity of data on whether PFOA and PFOS altered DNA repair and caused genomic instability.]

4.2.4 Induces epigenetic alterations

See <u>Tables 4.11</u> to <u>4.14</u>.

DNA (CpG, 5'-C-phosphate-G-3'-dinucleotide) methylation, global DNA methylation, altered expression of microRNAs (miRNAs), and histone modifications are all forms of epigenetic change that have been associated with carcinogenesis (Sharma et al., 2010). The following sections detail studies that have investigated the

Table 4.11	able 4.11 End-points relevant to epigenetic alterations in humans exposed to PFOA or PFOS							
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response	Covariates adjusted for	Comments	Reference	
Epigenome- wide DNA methylation	Cord blood and peripheral leukocytes	Ohio, USA Prospective birth cohort with cross- sectional analysis	266 mother–child pairs (median: PFOS, 14 ng/mL; PFOA, 5.5 ng/mL) at birth	435 CpGs (PFAS); 2 CpGs PFOS + 12 CpGs PFOA + and –	Cell type composition, child age, child sex, annual household income, maternal race/ethnicity, and maternal smoking during pregnancy	Cohort replication included; comparison of methylation over time highlighting the persistence of epigenetic modifications	<u>Liu et al.</u> (2022a)	
Epigenome- wide DNA methylation	Cord blood leukocytes	Taiwan, China Prospective birth cohort	Sapporo cohort of the Hokkaido (Japan) study (190 mother-child pairs from the general population; discovery cohort) (median: PFOS, 5.2 ng/mL; PFOA, 1.4 ng/mL) Taiwan, China Maternal and Infant Cohort Study (37 mother-child pairs from the general population; replication cohort (PFOS, 12.2 ng/mL; PFOA, 1.8 ng/mL)	4 CpGs for PFOS –; 3 CpGs for PFOA + and –	Maternal age, parity, maternal educational level, maternal blood sampling period, maternal pre-pregnancy BMI, maternal smoking during pregnancy, gestational age, infant sex, and cord blood cell estimates	Strengths of this study included the use of a replication cohort	<u>Miura et al.</u> (2018)	
Epigenome- wide DNA methylation	Dried blood spots	New York, USA Cross-sectional study	597 neonates (median: PFOS, 1.74 ng/mL; PFOA, 1.12 ng/mL)	2 sex-specific associations for CpGs PFOS + and -; 1 CpG PFOA –	Sample plate and estimated cell count; infant sex, plurality, and epigenetically- derived ancestry (4 principal components); maternal age, race/ethnicity, education level, marital status, pre- pregnancy BMI, smoking during pregnancy, and history of pregnancy loss		<u>Robinson</u> <u>et al. (2021)</u>	

Table 4.11	(continue	d)					
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Epigenome- wide DNA methylation	Cord blood leukocytes	Colorado, USA Cross-sectional study	583 mother–child pairs (median: PFOS, 2.4 ng/mL; PFOA 1.1 ng/mL)	1 DMR – for PFOA	Infant sex, gestational age (days), maternal age (years), education level (completed high school), smoking during pregnancy, race/ethnicity, BMI, previous pregnancies, cell type		Starling et al. (2020)
Epigenome- wide DNA methylation	Blood leukocytes	Shiyan Renmin Hospital Hubei Province, China Cross-sectional study	98 male and female patients (median: PFOS, 2.29 ng/mL; PFOA, 0.85 ng/mL)	87 CpGs and 11 DMRs for PFOS 63 CpGs for PFOA	Age, BMI, sex		<u>Cheng et al.</u> (2022)
Epigenome- wide DNA methylation	Placenta	USA Cross-sectional study	260 pregnant women (median in maternal plasma: PFOS, 4.74 ng/mL; PFOA, 2.2 ng/mL)	PFOS: 3 CpG sites (2 sites + and 1 site –) in placenta	Self-reported maternal race/ ethnicity (non-Hispanic White, non-Hispanic Black, Hispanic, Asian), age (in years), offspring sex (male/ female), pre-pregnancy BMI (kg/m ²), total plasma lipid concentration (ng/mL, except PFAS), log-transformed plasma cotinine concentration (ng/mL); methylation sample plate ($n = 5$); the first three methylation PCs and the first 10 genotype PCs were used to account for population structure	Included gene expression	<u>Ouidir et al.</u> (2020)
Epigenome- wide DNA methylation	Peripheral blood leukocytes	Ohio, USA Cross-sectional study	44 total (7.5 ng/mL); PFOA high (<i>n</i> = 22; 15 ng/mL) and low (<i>n</i> = 22; 2.4 ng/mL); median values	1 CpG -	Cell type, child sex, and income	Small sample size; nothing passed FDR	<u>Kingsley</u> et al. (2017)

_
\mathcal{L}
\sim
()
\geq
~
-
\cap
\cup
_
~
-
()
\sim
\sim
U 1
_
\mathcal{L}
~
1
_
0
_
_
0,
(L)
72
01

Table 4.11	(continue	ed)					
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Epigenome- wide DNA methylation	Blood leukocytes	Six European cohorts (France, Spain, Norway, Greece, UK, Lithuania) Cross-sectional study	1173 (PFOA, 1.51 μg/L; PFOS, 2.14 μg/L)	PFOS: 12 CpGs majority –; PFOA +/–	Cell type		<u>Cadiou et al.</u> (2020)
Epigenome- wide DNA methylation	Blood leukocytes	Dordrecht, Netherlands Cross-sectional study	34 (PFOS: 40 ng/g) median	PFOS: 29 CpGs, 38 DMRs	Age and leukocyte counts	Small sample size; did not differentiate between dioxins, PCBs, and PFOS; men only	<u>van den</u> <u>Dungen et al.</u> (2017a)
Epigenome- wide DNA methylation/ Epigenetic age	Whole blood leukocytes	Ronneby, Sweden Cross-sectional	63 participants (PFOS: controls, n = 32: 2.8 ng/mL, high exposure group, n = 31: 295 ng/mL) (PFOA: controls, n = 32: 1.4 ng/mL, high exposure group, n = 31: 19 ng/mL); medians	12 DMPs, PFAS +	Neutrophil fraction	Study did not differentiate between PFOS and PFOA; study design was unclear (case– control?)	<u>Xu et al.</u> (2022)
Global (Alu elements, LINE-1) DNA methylation	Cord blood leukocytes	Taipei, Taiwan, China Cross-sectional	363 participants (PFOS, 6.07 ng/mL; PFOA, 2.05 ng/mL)	Alu methylation: PFOS –; PFOA +/–; LINE-1 methylation +/–	Parental education level, maternal BMI, maternal age, delivery method (vaginal delivery or caesarean section), parity, infant sex, gestational age, and cotinine level		<u>Liu et al.</u> (2018a)
Global DNA methylation of LINE-1	Peripheral blood leukocytes	Ohio, USA Cross-sectional	685 participants (PFOS, 14.1 ng/mL; PFOA, 57.9 ng/mL)	LINE-1 methylation PFOS +; no association with PFOA	Age, sex, BMI, smoking status (ever/never), and current alcohol consumer (yes/no)		<u>Watkins</u> et al. (2014)
Global DNA methylation of LINE-1 and Alu methylation	Umbilical cord blood	Maryland, USA Cross-sectional	30 participants (PFOS, 5.8 ng/mL; PFOA, 1.8 ng/mL)	PFOS +/-; PFOA was marginally associated (P = 0.06) with a decrease in global DNA methylation	Maternal age or gestational age	Small sample size	<u>Guerrero-</u> <u>Preston et al.</u> (2010)

Table 4.11	11 (continued)								
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response	Covariates adjusted for	Comments	Reference		
Global DNA methylation	Sperm	Greenland, Denmark; Poland; Ukraine Cross-sectional	262 participants (PFOS, 27.2 ng/mL; PFOA, 4.0 ng/mL)	DNA global methylation PFOS +/-; PFOA +/-	Age and smoking status	No results were significant when the entire sample was considered, but when broken down by sub- study, there were significant associations	<u>Leter et al.</u> (2014)		
Global DNA methylation	Blood leukocytes	Taiwan, China Cross-sectional	1425 participants (mean PFOS, 4.95 ng/mL)	5mdC/dG +	Model 1: adjusted for age, sex, smoking, alcohol consumption, BMI, and household income; Model 2: adjusted for Model 1 parameters plus HTN, DM, or hyperlipidaemia	The overall epigenetic sampling was performed in a smaller sample	<u>Lin et al.</u> (2022)		
Targeted DNA methylation	Cord blood leukocytes	Sapporo, Japan Cohort	177 mother-child pairs (PFOS, 5.2 ng/mL; PFOA, 1.3 ng/mL)	IGF2 methylation; PFOA –	Maternal age, maternal education, infant sex, maternal smoking during pregnancy, and blood sampling period		<u>Kobayashi</u> <u>et al. (2017)</u>		
miRNA expression	Blood leukocytes	Ronneby, Sweden Cross-sectional	53 pregnant women (median: PFOS: low-exposure group, 3 ng/mL; high- exposure group, 230 ng/mL) (median: PFOA: low-exposure group: 2 ng/mL; high- exposure group, 8 ng/mL)	PFOS: ↓ miR-101-3p, ↓ miR-144-3p, ↓ miR-19a-3p	None	No covariates included; only women tested	<u>Xu et al.</u> (2020b)		
Targeted DNA methylation	Cord blood leukocytes	Taiwan, China Cross-sectional	486 participants (PFOS, 6.09 ng/mL; PFOA, 2.04 ng/mL)	<i>Mest</i> promoter methylation PFOS –	Maternal age, infant sex, parental educational level, cotinine level in cord blood, maternal alcohol consumption, pre-pregnancy BMI, parity, type of delivery, and gestational age	Sex-specific findings (women higher than men)	<u>Ku et al.</u> (2022)		

Table 4.11	(continue	d)					
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Targeted DNA methylation (<i>IGF2</i> , <i>NR3C1</i>), LINE-1 DNA methylation	Placenta	Hebei Province, China Cross-sectional	180 participants; median: PFOS, 1.21 ng/g; PFOA, 1.33 ng/g	LINE-1 methylation: PFOS –; PFOA –	Age, pre-pregnancy BMI, gestational weeks, GDM, parity, newborn sex, mean intakes of carbohydrate, protein, and fat during pregnancy, and physical activity		<u>Wang et al.</u> (2023a)
Epigenome- wide DNA methylation, targeted DNA methylation, epigenetic age	Blood leukocytes	Arizona, California, and Massachusetts, USA Firefighters Cross-sectional	197 firefighters (<i>n</i> -PFOS, 4.02 ng/mL; <i>sm</i> -PFOS, 2.06 ng/mL; PFOA, 1.79 ng/mL)	Epigenetic age: positive association with 2 epigenetic clocks for <i>sm</i> -PFOS; positive association with 6 epigenetic clocks for <i>n</i> -PFOA Targeted DNA methylation: <i>sb</i> -PFOA positively associated with CpG sites in <i>PPARG</i> and <i>CD36</i> ; <i>n</i> -PFOA inversely associated with methylation at one CpG site in <i>ACOT2</i> Epigenome-wide DNA methylation: positively associated with a CpG site within <i>CAPN12</i> ; <i>n</i> -PFOS was inversely associated with methylation near the transcription start site	Age, sex, race/ethnicity, cell type estimates and PCs representing technical variation from the Infinium MethylationEPIC array, and ethnicity in site-specific analysis and EWAS		<u>Goodrich</u> et al. (2021)

Table 4.11	(continue	d)
------------	-----------	----

End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Targeted miRNA expression	Blood leukocytes	Fluorochemical plant, Jiangsu Province, China Cross-sectional	55 workers: (PFOS, 33 ng/mL; PFOA, 1272 ng/mL), 132 nearby residents: PFOS, 30.92 ng/mL; PFOA, 249 ng/mL)	PFOA positively associated with the expression of miR-26b and miR-199-3p	Age, BMI, smoking, alcohol consumption status		<u>Wang et al.</u> (2012a)
Targeted miRNA expression	Blood leukocytes	China Cross-sectional	80 participants with MetS and 64 controls (PFOS, 3.3 ng/mL; PFOA, 2.1 ng/mL)	<i>n</i> -PFOA negatively associated with miR-140-5p	Age	All men	<u>Yang et al.</u> (2020)

BMI, body mass index; CpG, cytosine–guanosine dinucleotide; dG, 2'-deoxyguanosine; DM, diabetes mellitus; DMP, differentially methylated position; DMR, differentially methylated region; EWAS, epigenome-wide association analysis study; FDR, false discovery rate; GDM, gestational diabetes mellitus; HTN, hypertension; LINE-1, long interspersed nuclear element-1; 5mdC, 5-methyl deoxycytosine; MetS, metabolic syndrome; miRNA, microRNA; *n*-, linear isomer; PC, principal component; PCB, polychlorinated biphenyl; PFAS, per- and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; *sb*-, sum of branched isomers; *sm*-PFOS, sum of perfluoromethylheptane sulfonate isomers; UK, United Kingdom; USA, United States of America.

a +, increased methylation; -, decreased methylation, +/-, increase or decrease in methylation; \uparrow , increased gene expression; \downarrow , decreased gene expression; \downarrow , decreased gene expression; $\uparrow\downarrow$ increased and decreased gene expression.

association between PFOA or PFOS exposure and these epigenetic modifications.

- (a) Humans
- (i) Exposed humans

Exposure to PFOA

Several epigenome-wide association analysis studies (EWASs) have examined the relation between maternal exposure to PFOA during pregnancy and DNA methylation in neonatal cord blood.

Noteworthy among these studies was that by Liu et al. (2022a), which focused on a prospective birth cohort of mother-child pairs from the Health Outcomes and Measures of the Environment (HOME) Study (2003–2006; Cincinnati, Ohio, USA). Pregnant women were enrolled in the study at around week 16 of gestation, and children were followed-up at age 4 weeks and at age 1, 2, 3, 4, 5, 8, and 12 years. Overall, 291 participants with data on PFOA exposure and DNA methylation were considered in the study (266 at baseline and 160 at follow-up). For this study, the median PFOA concentration in maternal serum was 5.5 ng/mL and the 25th and 75th percentile values were 3.9 and 7.9 ng/mL, respectively. After adjustment for potential confounders and multiple comparisons, the authors identified that the maternal concentration of PFOA was associated with differential methylation of 12 CpGs measured in cord blood. The CpGs displayed both hyper- and hypomethylation. Notably, several of these DNA methylation changes persisted up to age 12 years. The associations were consistent at birth and at age 12 years, having the same direction and comparable effect sizes (Liu et al., 2022a). In addition, several CpGs were annotated to genes that have been linked to cancer of the breast, prostate, pancreas, and/ or brain, such as MAGI1, KRT18, SRPRB, TNR, and SERPINA5 (Liu et al., 2022a). [The Working Group noted that a strength of this study was the comparison of differential DNA methylation in cord blood at birth and blood collected during adolescence in the same participants, highlighting the stability of the findings. This study also benefited from the inclusion of a replication cohort. The Working Group also noted that this study analysed the association between maternal exposure to other PFAS and CpG methylation and found that there were specific changes associated with PFOA that persisted over time, further supporting the chemical specificity of the finding. As a potential limitation, it is possible that exposure to PFAS during the postnatal period could have influenced CpG methylation in later life. However, the probability of the chance identification of common CpG methylation sites between the early (e.g. at birth) and adolescent time points is low. Finally, the Working Group noted that a potential limitation of this study was that serum PFAS were not measured during the same trimester of pregnancy in all participants. However, this would lead to non-differential exposure misclassification and thus bias towards the null.]

Miura et al. (2018) also performed an EWAS in 190 mother-child pairs from the prospective Sapporo cohort of the Hokkaido Study (discovery cohort) and from 37 mother-child pairs from the Taiwan Maternal and Infant Cohort Study (replication cohort), examining the relation between PFOA in maternal serum and cord blood DNA methylation. The median PFOA concentration was measured to be 1.4 ng/mL and exposure was associated with both hypo- and hypermethylation of CpGs (Miura et al., 2018). [The Working Group noted that a strength of this study was the inclusion of a replication cohort. As above, the Working Group noted a potential limitation of this study was that serum PFAS concentrations were not measured during the same trimester of pregnancy in all participants. As noted, this would lead to non-differential exposure misclassification that would bias towards the null.]

Robinson et al. (2021) examined associations between PFOA levels and DNA methylation in a

cohort study in 597 neonates in New York, USA, in which both were assessed in dried blood spots from newborns. The median PFOA concentration measured was 1.12 ng/mL. Overall, log-transformed values of PFOA were not related to site-specific DNA methylation. When comparing the participants in the top decile of PFOA concentration with the other participants, exposure was associated with decreased DNA methylation of one CpG methylation site annotated to genes *SCRT2* and *SRXN1* (Robinson et al., 2021).

Starling et al. (2020) performed an analysis of data from a prospective cohort study of motherinfant pairs in Colorado, USA (n = 583). The median PFOA concentration was 1.12 ng/mL, and the range was 0.1–15.4 ng/mL. The maternal serum level of PFOA was associated with decreased DNA methylation of a CpG site annotated to *TJAP* in the cord blood of infants (Starling et al., 2020). Additionally, PFOA was associated with altered methylation of 15 differentially methylated regions (DMRs).

In the study by <u>Cheng et al. (2022)</u>, plasma PFOA was assessed in 98 patients and the median concentration was 0.85 ng/mL. There were 63 CpG sites and eight DMRs associated with the measured plasma PFOA levels. Among the identified CpGs were those that were annotated to the genes *AFF3*, *CREB5*, *NRG2*, and *USF2*, and one of the DMRs was annotated to *IRF6* (<u>Cheng et al.</u>, 2022).

[The Working Group noted that, taken together, the EWAS studies relating maternal PFOA to cord blood-based DNA methylation identified statistically significant associations. Although the specific genes that were identified across studies were not the same, in numerous cases the CpG sites were located within cancer-associated genes.]

In addition to the EWAS described above, several studies have also investigated the relations between PFOA and gene-specific/targeted CpG methylation. For example, <u>Kobayashi et al.</u>

(2017) examined prenatal exposure to PFOA in a cohort in Japan (177 participants). The mean level was 1.3 ng/mL and the concentrations ranged from below the detection limit to 5.3 ng/mL. Exposure to PFOA was associated with lower methylation of the imprinted gene *IGF2* in cord blood (Kobayashi et al., 2017).

Similarly, Goodrich et al. (2021) investigated the PFOA-associated DNA methylation in blood leukocytes in firefighters from three states in the USA, namely Arizona, California, and Massachusetts. The mean level of linear *n*-PFOA was 1.79 ng/mL, and the 25th and 75th percentile values were 1.40 and 2.20 ng/mL. [The Working Group noted that there were lower concentrations and detection frequencies for branched *sb*-PFOA in this cohort, with a detection frequency of 31%.] The results indicated that exposure to *n*-PFOA was associated with lower methylation at one CpG site annotated to ACOT2. The results also indicated that the sum of the branched isomers of perfluorooctanoate (sb-PFOA) was associated with greater methylation at two CpG sites annotated to PPARG and CD36. When assessed via an epigenome-wide approach, sb-PFOA was associated with greater methylation at a CpG site that was annotated to CAPN12 (Goodrich et al., 2021).

The relation between PFOA and epigenetic ageing (i.e. a measure of biological ageing) has been assessed. Indicators of epigenetic age, referred to as epigenetic clocks, have been developed with the use of CpG methylation data (Li et al., 2022b). Epigenetic age is a predictor of age and/or mortality. Recent research in diverse cancer types has highlighted the crucial role of epigenetic ageing in the initiation of tumours and its potential utility in predicting cancer risk (Yu et al., 2020). Goodrich et al. (2021) highlighted a positive association between exposure to *n*-PFOA and epigenetic ageing, assessed using various epigenetic clocks. However, there were no associations identified between sb-PFOA and epigenetic ageing.

The relationship between exposure to PFOA and the expression levels of miRNAs has been assessed in several studies.

Wang et al. (2012a) performed a cross-sectional analysis of a cohort of 55 workers in a fluorochemical plant and 132 nearby residents (controls) in a suburban area of Changshu City, Jiangsu Province, China. The geometric mean levels of PFOA were 1272.31 ng/mL in the workers and 249.93 ng/mL in the residents. To explore the effect of PFOA on circulating miRNAs, serum samples from 10 workers and 10 residents were used for miRNA microarray analysis. The high serum level of PFOA (high PFOA group) was positively associated with the increased expression of miR-26b and miR-199–3p in blood leukocytes (Wang et al., 2012a).

Yang et al. (2020) performed a cross-sectional analysis in a cohort of male participants with (n = 80) and without (n = 64) metabolic syndrome from China in whom the mean level of PFOA was 2.1 ng/mL. The serum concentration of *n*-PFOA was found to be negatively associated ($\beta = -0.772$; 95% CI, -0.244 to -0.300; P < 0.01; q < 0.05) with the expression of miR-140-5p in blood leukocytes (Yang et al., 2020).

Two studies assessed PFOA in relation to the global methylation of long interspersed nuclear element 1 (LINE-1) and/or small dimeric elements ALU methylation. Specifically, <u>Guerrero-Preston et al. (2010)</u> performed a cross-sectional analysis of a cohort of newborns in Maryland, USA (30 participants). The mean PFOA concentration, measured in umbilical cord blood just after birth, was 1.8 ng/mL. PFOA was marginally associated (P = 0.06) with a low level of global DNA methylation in the umbilical cord blood (<u>Guerrero-Preston et al., 2010</u>).

Similarly, <u>Watkins et al. (2014)</u> examined the association between PFOA and LINE-1 in a subset of adults enrolled in the C8 Health Project in Ohio, USA (685 participants). The mean serum concentration of PFOA was 57.9 ng/mL. No association was observed between PFOA and LINE-1 (<u>Watkins et al., 2014</u>) (see <u>Table 4.11</u>).

Exposure to PFOS

EWAS have been used to investigate the relationship between exposure to PFOS during pregnancy and CpG methylation in neonatal cord blood. A prospective birth cohort study collected data from 266 mother-child pairs from the general population in Ohio, USA. The median maternal serum PFOS concentration was measured as 14 ng/mL and the 25th and 75th percentile values were 9.9 and 17.8 ng/mL. After adjusting for potential confounders and multiple comparisons, the study found that prenatal exposure to PFOS was associated with persistent hypermethylation of two CpGs, in both cord blood and peripheral blood later in life (at age 12 years) (Liu et al., 2022a). One of these CpGs was annotated to HPSE2, which has been linked to breast cancer (Zhang et al., 2021a).

[The Working Group noted two strengths of this study: (i) the comparison of DNA methylation at different time points in the same participants; and (ii) the inclusion of a replication cohort.]

Miura et al. investigated the relation between maternal serum PFOS and cord blood DNA methylation in mother-child pairs recruited from the Sapporo cohort of the Hokkaido Study (discovery cohort) and mother-child pairs from the Taiwan Maternal and Infant Cohort Study (replication cohort). A total of 190 mother-child pairs, with exposure and methylation data, were analysed. The median PFOS concentration was 5.2 ng/mL, and the exposure was associated with the hypomethylation of four CpGs, as well as one DMR (Miura et al., 2018). [The Working Group considered the inclusion of a replication cohort to be a strength of this study. However, the Working Group noted a potential limitation of this study, which was that the serum PFOS concentrations were not measured during the same trimester of pregnancy in all participants. This might lead to

non-differential exposure misclassification and thus bias towards the null.]

In a separate study, <u>Cheng et al. (2022)</u> examined the relation between plasma PFOS and DNA methylation in leukocytes sampled from both male and female patients (n = 98) from the Shiyan Renmin Hospital of Hubei Province in China. The patients were being treated for benign diseases or for cosmetic needs. The median plasma concentration was 2.29 ng/mL. A total of 87 CpG sites and 11 DMRs displayed associations with plasma PFOS concentrations (<u>Cheng</u> <u>et al., 2022</u>).

Ouidir et al. (2020) assessed the association between PFOS exposure (plasma median concentration, 4.74 ng/mL) in pregnant women (n = 260) enrolled in the Eunice Kennedy Shriver National Institute of Child Health and Human Development (NICHD) Fetal Growth Studies of a singleton cohort and epigenome-wide DNA methylation in the placenta. The authors identified three CpG sites with both hyper- and hypomethylation (Ouidir et al., 2020).

Robinson et al. (2021) investigated the association between PFOS exposure and DNA methylation in the Upstate KIDS cohort study, New York, USA, comprising 597 newborns, for whom the median PFOS concentration, measured in dried blood spots, was 1.74 ng/mL (25th to 75th quartile, 1.11 to 2.54 ng/mL). The results showed that PFOS was associated with lower DNA methylation, measured in dried blood spots, at one CpG site in boys, and higher DNA methylation at a different site in girls. In addition, the associations were observed only at the highest concentrations of PFOS, above the 90th percentile (Robinson et al., 2021).

Other studies have examined the relation between exposure to PFOS and epigenetic ageing. As reported above, DNA methylation can be used to estimate epigenetic age (Li et al., 2022b). Epigenetic ageing has been associated with tumour initiation and cancer risk prediction (Yu et al., 2020). Goodrich et al. (2021) conducted a cross-sectional analysis of 197 firefighters from Arizona, California, and Massachusetts, USA. They observed that the mean serum concentration of n-PFOS was 4.02 ng/mL and the 25th and 75th percentile values were 3.00 and 5.80 ng/mL, respectively. The total concentration of perfluoromethylheptane sulfonate isomers (*sm*-PFOS) was 2.06 ng/mL and the 25th and 75th percentile values were 1.40 and 3.10 ng/mL, respectively. The results showed that for n-PFOS there was no association with epigenetic ageing, whereas for *sm*-PFOS there were associations with epigenetic ageing, specifically for two of the clocks (Goodrich et al., 2021).

Besides the EWAS approach, several studies have investigated the relationship between PFOS and gene-specific or targeted DNA methylation. For example, Ku et al. (2022) analysed 486 mother-infant pairs from the Taiwan Birth Panel Study cohort, China. The mean PFOS concentration was 6.09 ng/mL and the highest measured concentration was 67.92 ng/mL. The researchers identified that, in the multivariable model after adjustments, prenatal exposure to PFOS was associated with decreased methylation in the promoter region of MEST in the cord blood of infants (Ku et al., 2022). [The Working Group noted that *MEST* is an imprinted gene that encodes a protein belonging to the α/β hydrolase superfamily and has been found to be linked to adipocyte differentiation (Kamei et al., 2007).]

The associations of exposure to PFOS with global methylation, LINE-1, and/or ALU methylation have also been assessed. Wang et al. (2023a) performed a cross-sectional analysis of 180 pregnant women enrolled in a cohort study from Tangshan City, northern China, to examine the relation between PFOS and DNA methylation in the placenta. The median PFOS concentration in the placenta was 1.39 ng/g and ranged from 0.19 to 3.70 ng/g. The level of PFOS in the placenta was inversely associated with the overall methylation of LINE-1 (Wang et al., 2023a).

Similarly, Liu et al. (2018a) analysed 363 mother–infant pairs from the Taiwan Birth Panel birth cohort study in Taiwan, China. Maternal and cord blood samples were collected at birth. The mean PFOS concentration in the cord blood was 6.07 ± 1.93 ng/mL (geometric mean \pm standard deviation, SD). Prenatal PFOS was associated with decreases in cord blood-derived ALU methylation (Liu et al., 2018a).

<u>Guerrero-Preston et al. (2010)</u> conducted a cross-sectional analysis of cohort of newborns (n = 30) in Maryland, USA. The mean PFOS concentration in the cord blood was 5.8 ng/mL. No association between PFOS and global DNA methylation was observed.

Another form of epigenetic modification is the altered expression of miRNAs. Several studies have examined the relation between PFOS and miRNA expression. For example, Xu et al. (2020b) performed a cross-sectional analysis of 53 women from the Ronneby area, Sweden, in whom the PFOS levels were up to 315 ng/mL because of contamination of drinking-water. The results highlighted that PFOS is associated with decreases in the expression of three miRNAs (miR-101-3p, miR-144-3p, and miR-19a-3p) in blood leukocytes (Xu et al., <u>2020b</u>). These miRNAs are predicted to target genes annotated to cancer or endocrine dysfunction, such as DNMT3A, EGFR, HMGCR, NR1H3, PPARA, PTGS2, and TGFA.

[The Working Group noted several factors as strengths of the studies, including: (1) an examination of the persistence of the epigenetic mark over time; (2) the inclusion of gene expression measures; and (3) the use of replication cohorts. The Working Group noted that numerous studies deployed designs where chemical exposure was analysed in relation to epigenetic end-points in a cross-sectional manner. A potential limitation of this design is that a chemical measurement and/or epigenetic modification assessed at a single time point may not capture variation over time. In support of the above, it has been reported that PFAS levels can vary over the course of gestation: blood levels are higher in the first trimester of pregnancy than in later trimesters. Most of the association analyses presented here examined the relationship between PFAS exposure assessed later in pregnancy and the epigenetic end-point. This time frame of exposure assessment would thus be expected to lead to bias towards the null. The Working Group also noted that most studies assessed the epigenetic marks in blood leukocytes, and the relevance of the epigenetic marks to tumorigenesis has not been completely established. Despite the limitations of measurements in leukocytes, the Working Group noted features of the studies that have relevance to cancer, including PFOA and PFOS-associated DNA methylation of cancer-associated genes, altered expression of miRNAs that are known to be involved in carcinogenesis, and cancer-associated features, such as global hypomethylation.]

Other studies with inconclusive results or limitations in the quality of design were considered less informative. These studies investigated the associations of PFOA (<u>Kingsley et al., 2017</u>) or PFOS (<u>van den Dungen et al., 2017a; Cadiou et al., 2020; Xu et al., 2022</u>) with genome-wide methylation, global DNA methylation, or target DNA methylation (<u>Leter et al., 2014; Lin et al., 2022</u>).

(ii) Human cells in vitro

See <u>Table 4.12</u>.

The effects of in vitro exposure to PFOA and PFOS on DNA methylation have been assessed in a variety of human cells, including breast (MCF-7, MCF-10A), brain (SH-SY5Y), liver (HepG2), lung (A549), and placenta (HTR-8/SVneo) cell lines (Tian et al., 2012; Bastos Sales et al., 2013; Guo et al., 2017; van den Dungen et al., 2017b; Jabeen et al., 2020; Zhao et al., 2022).

End-point	Tissue, cell line	Result ^a	Concentrations or dosing regimen	Comments	Reference
DNA methylation	Brain, SH-SY5Y cells	PFOA –	PFOA: 0.4 or 4 μg/L in medium for 4 d		<u>Zhao et al. (2022)</u>
DNMT activity (DNMT1 and DNMT3A); global DNA methylation; <i>DNMT</i> expression	Breast, MCF7; liver, HepG2 cells	PFOA: global methylation −, ↓ <i>DNMT1</i> expression, ↑ <i>DNMT3A</i> in MCF-7	PFOA: 0, 20, 100, 200, or 400 μM for 48 h		<u>Liu and Irudayaraj</u> (2020)
Global DNA methylation	Brain, neuroblastoma cell line (SK-N-AS)	PFOS +/-; PFOA +/-	PFOS: 10 μM; PFOA 10 μM		<u>Bastos Sales et al.</u> (2013)
Global DNA methylation	Breast, MCF-10A cells	PFOS + and PFOA +	PFOS (10 μM) and PFOA (100 μM) for 72 h		<u>Pierozan et al.</u> (2020)
Global DNA methylation	Liver, L02 cells	PFOA +/-	PFOA: 5, 10, 25, 50, or 100 mg/L		<u>Tian et al. (2012)</u>
Global DNA methylation	Liver, HepG2	PFOA –	PFOA: 0–400 μM for 24 h		<u>Wen et al. (2020)</u>
Global methylation	Breast, MCF-10A	PFOS + and PFOA +	PFOS: 10 μM; PFOA: 100 μM		<u>Pierozan et al.</u> (2020)
mRNA expression of DNMTs and BDNF, miRNA-16, miRNA-22, and miRNA-30a- 5p	Brain, SK-N-SH cells	↑ miRNA-16, ↑ miRNA-22, and ↑ microRNA-30a-5p, ↓ <i>DNMT1</i> mRNA, ↓ DNMT1 protein, ↑ <i>DNMT3A</i> mRNA, ↑ DNMT3A protein, ↑ <i>DNMT3B</i> mRNA, ↑ DNMT3B protein, ↓ BDNF protein, ↓ <i>BDNF</i> mRNA	PFOS: (0–150 μM)		<u>Guo et al. (2017)</u>
Targeted DNA methylation: DMRs, DMPs, 84 adipogenic genes	Human mesenchymal stem cells from bone marrow	DNA methylation –, 2 DMRs (<i>AXIN1</i> , <i>DKK1</i>) –, 45 DMPs (majority –)	PFOS: 0-30 μM, from day -1 to day 10; measurements on day 10	Treatment during cell differentiation	<u>van den Dungen</u> <u>et al. (2017b)</u>
Targeted gene expression (DNA methylation machinery (DNMTs, TETs)	Lung, A549 cells	Expression: PFOS: ↓ TET1 mRNA, ↑ TET2 mRNA, ↑ TET3 mRNA, ↑↓ DNMT1 mRNA, ↓ DNMT3B mRNA, ↓ DNMT3A mRNA, ↑↓ CCNE1 mRNA, ↑↓ CCNA2 mRNA, ↓ CCNB1 mRNA PFOA: ↓ TET1 mRNA, ↑ TET3 mRNA, ↓ DNMT1 mRNA, ↓ DNMT3B mRNA, ↑ DNMT3A mRNA, ↑ CCNE1 mRNA, ↓ CCNA2 mRNA, ↑↓ CCNB1 mRNA	PFOS: 0-400 μM; PFOA: 0-400 μM	DNA methylation was not assessed	<u>Jabeen et al. (2020)</u>
Targeted miRNA expression (miR-155 expression)	Liver, HepG2 cells	↑ miR-155	PFOS: 0–50 μM for 24 h		<u>Wan et al. (2016)</u>

Table 4.12 End-points relevant to epigenetic alterations in human cells in vitro exposed to PFOA or PFOS

Table 4.12 (continued)

End-point	Tissue, cell line	Result ^a	Concentrations or dosing Comments regimen	Reference
Targeted miRNA expression and gene expression (MEG3)	HTR-8/SVneo cells	PFOS: ↓ miR-770; ↓ <i>MEG3</i> , ↑ <i>PTX3</i>	PFOS: 0-10 μM	<u>Li et al. (2022b)</u>
Targeted miRNA expression: miR-19a and miR-19b expression; targeted DNA methylation (H19 methylation)	HTR-8/SVneo cells	\downarrow miR-19a; \downarrow miR-19b; \uparrow <i>H19</i> expression	PFOS: 0, 0.1, 1, or 10 μM for 24 or 48 h	<u>Li et al. (2020b)</u>
Targeted miRNA expression (miRNA-22)	Brain, SH-SY5Y cells	\downarrow miR-16 expression; \uparrow miR-22 expression, \downarrow <i>BDNF</i> , \downarrow <i>CREB</i> (100 μ M), \uparrow <i>TrkB</i>	PFOS: 0–100 μM for 48 h	<u>Li et al. (2015)</u>
Targeted gene expression (DNMTs, SIRTs), global DNA methylation, targeted miRNA expression (miR-29b)	HTR-8/SVneo cells	PFOS: \downarrow DNMT1, \downarrow DNMT3A, \downarrow DNMT3B, \downarrow SIRT1, \downarrow SIRT3; global DNA methylation; expression: \uparrow miR-29b	1, 10, and 50 μM	<u>Sonkar et al. (2019)</u>

BDNF, brain-derived neurotrophic factor; d, day(s); DMP, differentially methylated position; DMR, differentially methylated region; DNMT, DNA methyltransferase; miRNA, microRNA; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; SIRT, sirtuin; TET (enzymes), ten eleven translocation, alias for tet methylcytosine dioxygenases.

^a +, increased methylation; -, decreased methylation, +/-, increase or decrease in methylation; ↑, increased gene expression; ↓, decreased gene expression; ↑↓, increased and decreased gene expression.

Exposure to PFOA

<u>Pierozan et al. (2020)</u> examined the effects of PFOA at a concentration of 100 μ M for 72 hours and found increased global DNA methylation in breast MCF-10A cells. In another study in which PFOA was tested at a range of 0 to 400 μ M, exposure was associated with a dose-dependent decrease in global DNA methylation in HepG2 cells (Wen et al., 2020).

Liu and Irudayaraj (2020) exposed breast MCF7 and liver HepG2 cells to PFOA at concentrations ranging from 0–400 μ M. PFOA was associated with reduced global DNA methylation and altered expression of *DNMT1* in both cell types. *DNMT3A* displayed increased expression in MCF7 cells. An inconclusive trend in the expression level of *DNMT3B* was observed in both cell types (Liu and Irudayaraj, 2020).

Exposure to PFOS

<u>Pierozan et al. (2020)</u> examined the effects of PFOS exposure on MCF-10A cells using a concentration of 10 μ M. PFOS was associated with increases in global DNA methylation in breast MCF-10A cells (<u>Pierozan et al., 2020</u>).

In relation to miRNAs, Wan et al. studied PFOS at concentrations ranging from $0-50 \mu M$. They found that PFOS was associated with increased expression of miR-155 in HepG2 cells (Wan et al., 2016).

Li et al. (2020b) examined the effects of PFOS in HTR-8/SVneo cells at concentrations ranging from 0–10 μ M. They found that PFOS altered the expression of several miRNAs in HTR-8 cells, including the reduction of the expression of miR-19a and miR-19b (Li et al., 2020b). PFOS was subsequently shown to reduce the expression of miR-770 in HTR-8 cells (Li et al., 2022c). The effects of PFOS were tested in SH-SY5Y at concentrations ranging from 0–100 μ M; PFOS caused decreased expression of miR-16 and increased expression of miR-22 (Li et al., 2015). PFOS treatment for 24 or 48 hours, even at 10 μ M, decreased gene and protein expression of the DNA methyltransferases, and significantly increased reactive oxygen species (ROS) production in the first-trimester human HTR-8/SVneo trophoblast cell line. In addition, PFOS reduced global DNA methylation and increased protein lysine acetylation (Sonkar et al., 2019).

- (b) Experimental systems
- (i) Non-human mammalian systems in vivo See <u>Table 4.13</u>.

Exposure to PFOA

In relation to histone modifications, <u>Li et al.</u> (2019a) exposed female pregnant Kunming mice to PFOA doses of 1, 2.5, 5, 10, or 20 mg/kg bw per day from pregnancy day 0, i.e. gestational day (GD)1, to GD17, and found that PFOA was associated in a dose-dependent manner with decreased histone acetylation in the liver: the histone acetyl-transferase activity of the female offspring was reduced significantly up to the dose of 5 mg/kg, and the histone deacetylase activity was increased significantly up to the highest dose of 10 mg/kg. The expression of both acetyl-histone H3 and acetyl-histone H4 proteins was reduced significantly (<u>Li et al., 2019a</u>).

Rashid et al. (2020a) exposed CD-1 mice (age 30 days) to PFOA at 1, 5, 10, or 20 mg/kg per day for 10 days and found that PFOA exposure was associated with 879 DMRs and increased *DNMT1* expression in the kidney (Rashid et al., 2020a).

Ahmad et al. (2021) exposed CD-1 mice to PFOA at 5 or 20 mg/kg per day for 10 days and observed decreases in the methylation of the gene encoding transmembrane serine protease *Tmprss2* (a prognostic marker for lung adenocarcinoma) in lung tissue where also PFOA accumulated (Ahmad et al., 2021). The gene expression of *Dnmts* and *Tets* was also decreased (Ahmad et al., 2021).

End-point	Species, strain	Tissue	Result ^a	Concentrations or dosing regimen	Route, duration	Comments	Reference
Global DNA methylation level, targeted gene expression (histone demethylases <i>Kdm1a</i> and <i>Kdm4c</i>)	Mouse, CD-1	Kidney	Global DNA methylation level –, gene expression of histone demethylases Kdm1a and Kdm4c ↑	PFOS, 5, 10, 20 mg/kg per day	Oral, 14 d		<u>Wen et al.</u> (2022)
Global DNA methylation, LINE-1 methylation, <i>GSTP</i> promoter region methylation	Rat, Sprague- Dawley	Liver of offspring, on PND21	Global DNA methylation –, LINE-1 methylation –, <i>GSTP</i> promoter region methylation changes +	PFOS, 0.1, 0.6, or 2.0 mg/kg bw per day	Oral gavage to dams from GD2 to GD21	Prenatal treatment	<u>Wan et al.</u> (2010)
Histone acetylation	Mouse, Kunming	Liver of female offspring on PND21	PFOA: ↓ histone acetylation, ↓ HAT activity, ↑ HDAC activity	PFOA, 1, 2.5, 5, or 10 mg/kg/bw (0.2 mL per day) to dams	Gavage, solution in deionized water, dams, GD1 to GD17	Prenatal treatment; liver specimens collected from female offspring killed on PND21	<u>Li et al.</u> (2019a)
Histone acetylation	Rat, Wistar	Testis	PFOS: H3K9me2 +, H3K9ac +, H3K18ac +, H3K9me3 –	PFOS, 0.015 and 0.15 mg/kg per day	Oral gavage, 60 d		<u>Alam et al.</u> (2021)
Histone acetylation	Mouse, ICR	Ovary	PFOS: histone H3K14 acetylation of <i>StAR</i> promoter –	PFOS, 0.1 mg/kg per day	Drinking-water, 4 mo		<u>Feng et al.</u> (2015)
Targeted DNA methylation, targeted gene expression	Mouse, CD-1	Kidney	879 differentially methylated regions; ↑ <i>Dnmt1</i> expression, ↓ <i>Dnmt3a</i> expression, ↓ <i>Tet1</i> expression, <i>Tet2</i> expression +/-, ↑ <i>Tet3</i> expression; global DNA methylation +/-; ↑ Hdac1, Hdac3, Hdac4, Hdac 2-10 +/-; ↓ <i>RASAL1</i> mRNA expression, ↑ <i>Acta2</i> mRNA	PFOA, 1, 5, 10, or 20 mg/kg per day	Oral, 10 d		<u>Rashid</u> <u>et al.</u> (2020a)

expression, *Lrnf2* and *Dlg2* mRNA expression +/-, ↑ *Tgfb* mRNA expression

Table 4.13 End-points relevant to epigenetic alterations in non-human mammals in vivo exposed to PFOA or PFOS

End-point	Species, strain	Tissue	Result ^a	Concentrations or dosing regimen	Route, duration	Comments	Reference		
Targeted gene and miRNA expression (angiogenesis-related mRNA, miRNA, and lncRNA)	Pregnant mouse, CD-1	Placenta of dams	↓ IncRNA Xist expression	PFOS, 0.5, 2.5, 12.5 mg/kg bw per day	Oral gavage, GD1 to GD17	Analysis performed on GD18	<u>Chen et al.</u> (2018)		
Targeted gene expression (<i>Dlk1–Dio3</i> imprinted cluster)	Mice, Kunming	Testes of offspring	↓ <i>Dlk1–Dio3</i> on PND21	PFOA, 1, 2.5, or 5 mg/kg per day	Gavage during gestation, GD1 to GD17	Prenatal treatment	<u>Song et al.</u> (2018)		
Targeted gene expression (Dnmts and Tets); targeted CpG methylation	Mouse, CD-1	Lung tissue	PFOA: ↓ Dnmts and ↓ Tets expression: – CpG <i>Tmprss2</i>	PFOA, 5 or 20 mg/kg per day	Oral gavage, 10 d		<u>Ahmad</u> et al. (2021)		
Targeted gene expression (epigenetic machinery)	Mouse, CD-1	Small intestine, colon	Small intestine: mRNA: $\downarrow Dnmt1, \downarrow Dnmt3a (\uparrow\downarrow),$ $\downarrow Dnmt3b; \uparrow Tet1, \uparrow Tet2,$ $\downarrow Tet3, \downarrow Cldn2, \downarrow Cldn8,$ $\downarrow Cldn12, \uparrow Cldn4, Cldn3$ (\uparrow and \downarrow), $\uparrow Cldn15, Cldn7$ ($\uparrow\downarrow$), $Tjp1 (\uparrow\downarrow), Tjp2 (\downarrow\uparrow), Ocln$ +/- Colon: mRNA: $\downarrow Dnmt3b,$ $\downarrow Dnmt3a, Dnmt1 +/-; \downarrow Tet1,$ $Tet2 +/-, Tet3 +/-, \uparrow Cldn2,$ $\uparrow Cldn3, \uparrow Cldn8, \downarrow Cldn7,$ Cldn4 +/-, Cldn12 +/-, $Cldn15 +/-, \downarrow Tjp1, \downarrow Ocln,$ Tjp2 +/-	PFOA, 1, 5, 10, or 20 mg/kg per day	Oral gavage, 10 d		<u>Rashid</u> <u>et al.</u> (2020b)		
Targeted gene expression (H19)	Pregnant mouse, CD-1	Placenta of dams	\uparrow <i>H19</i> expression, H19 methylation –	PFOS, 0, 0.5, 2.5, or 12.5 mg/kg per day	Oral, GD1 to GD17	Analysis performed on GD18	<u>Li et al.</u> (2020b)		
Targeted miRNA and gene expression (miR- 770 expression; <i>Meg3</i> expression)	Pregnant mouse, CD-1	Placentas of dams	PFOS: ↓ miR-770 expression, ↓ <i>Meg3</i> expression, <i>Meg3</i> methylation +, ↑ <i>Ptx3</i> expression	PFOS, 0.5, 2.5, or 12.5 mg/kg per day	Gavage, GD0 to GD17	Analysis performed on GD18	<u>Li et al.</u> (2022c)		
Targeted miRNA expression	Rat, Wistar	Brain	↓ miR-466b, ↓ miR-672, ↓ miR-297	PFOS, 0–3.2 mg/kg per day	Feed, GD1 to PND7		<u>Wang et al.</u> (2012b)		

Table 4.13 (continued)

Table 4.13 (continued)

End-point	Species, strain	Tissue	Result ^a	Concentrations or dosing regimen	Route, duration Comments	Reference
Targeted miRNA expression	Chicken, Plymouth Rock	Heart	PFOA:↑miR-490-5p	PFOA, 0–2 mg/kg per day	Cell injection, ED0 to ED21	<u>Guo et al.</u> (2022a)
Targeted miRNA expression	Mouse, BALB/c	Testes	9 ↓miRNAs, 8 ↑ miRNAs, including miR-133b-3p	PFOA, 5 mg/kg per day	Oral gavage, 28 d	<u>Lu et al.</u> (2017)
Targeted miRNA expression	Mice, BALB/c	Serum	miR-28-5p, miR-32-5p, miR-122-5p, miR-192-5p, and miR-26b-5p (all ↑)	PFOA, 1.25 or 5 mg/kg per day	Oral gavage, 28 d	<u>Yan et al.</u> (2014)
Targeted MiRNA expression	Mice, ICR	Sertoli cells and Leydig cells	↑ miR-9-3p, ↑↓ miR-1954, ↑↓ miR-710	PFOS, (0.5–10 mg/kg per day)	Oral gavage, 4 wk	<u>Huang</u> <u>et al.</u> (2022a)
Targeted miRNA expression (387)	Rat, Wistar	Liver	Significantly altered miRNAs included \uparrow miR- 19b, miR-21*, miR-17-3p, miR-125a-3p, miR-16, miR- 26a, miR-1, miR-200c, and miR-451. PND1: 35 miRNAs \uparrow , 11 \downarrow miRNAs; PND7: 8 \uparrow miRNAs, 1 \downarrow miRNA; 4 miRNAs on both PND1 and PND7	PFOS, 3.2 mg/kg per day	Feed (dam), GD1 to PND7 Pups, until PND7	<u>Wang et al.</u> (2015b)
Targeted miRNA expression (miR-34a)	Mice, BALB/c	Liver	miR-34a ↑	PFOA, 5 mg/kg per day	Gavage, 28 d	<u>Cui et al.</u> (2019)
Targeted gene expression, histone modifications	Rat, Wistar	Testes	PFOA: <i>Lhr</i> \uparrow , <i>Star</i> \uparrow , <i>Hsd3b</i> \downarrow , <i>Hsd17b</i> \downarrow , <i>Arom</i> \downarrow , <i>Cyp11a1</i> +/-, <i>Cyp17a1</i> +/-; histone modification: \downarrow H3K9me1, H3K9me2, H3K9me3, H3K9ac, H3K18me1, H3K18ac, H3K23me1, and H3K23ac	PFOA, 0.015 or 0.15 mg/kg per day	Gavage, 60 d	<u>Han et al.</u> (2022)

bw, body weight; d, day(s); CpG, cytosine-guanosine dinucleotide; DMP, differentially methylated position; ED, embryonic day; GD, gestational day; HAT, histone acetyltransferase; HDAC, histone deacetylase; HIC, highest ineffective concentration; lncRNA, long non-coding RNA; LEC, lowest effective concentration; LINE-1, long interspersed nuclear element-1; mo, month(s); miRNA, microRNA; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PND, postnatal day; wk, week(s).

a +, increased methylation; -, decreased methylation; +/-, increase or decrease in methylation; †, increased gene expression; ↓, decreased gene expression; ↓, increased and decreased gene expression.

Targeted gene expression analyses have shown that PFOA exposure was associated with an increase in *DNMT* expression in the mouse kidney (<u>Rashid et al., 2020a</u>). PFOA was associated with decreases in the expression of *Dnmt1*, *Dnmt3a*, and *Dnmt3b* in the intestines of CD-1 mice exposed to PFOA at 1, 5, 10, or 20 mg/kg per day (<u>Rashid et al., 2020b</u>). Targeted gene expression analysis has shown that *Dlk1 and Dio3* have decreased expression in the testes of Kunming offspring mice exposed prenatally to PFOA at doses of 1, 2.5 or 5 mg/kg during gestation (<u>Song et al., 2018</u>).

In relation to miRNAs, in male BALB/c mice exposed to PFOA at 0.08, 0.31, 1.25, 5 or 20 mg/kg per day for 28 days, increases in the expression of miR-28-5p, miR-32-5p, miR-122-5p, miR-192-5p, and miR-26b-5p were identified in mouse serum at the doses of 1.25 and 5 mg/kg (<u>Yan et al., 2014</u>).

Developmental exposure of fertile hatchling chicken eggs (incubated to hatch) at PFOA doses of 0.5, 1, or 2 mg/kg per egg weight was also associated with increased expression of miR-490-5p in heart tissues compared with the vehicle control group (Guo et al., 2022a).

Exposure to PFOS

Wen et al. (2022) exposed CD-1 mice to PFOS at doses of 5, 10, or 20 mg/kg per day for 14 days and found decreases in global DNA methylation levels in the kidney.

Wan et al. (2010) exposed Sprague-Dawley rats to PFOS at doses of 0.1, 0.6, or 2.0 mg/kg per day from GD2 to GD21. PFOS was found to accumulate in the kidney in a dose-dependent manner and was associated with increased expression of the kidney injury markers *Acta2* and *Bcl2l1*. In addition, PFOS was found to be associated with decreased global DNA methylation and decreased LINE-1 methylation in the livers of the offspring (Wan et al., 2010). These authors also found that PFOS was associated with increased methylation the *Gstp* promoter region in the livers (Wan et al., 2010). [The Working Group noted the relevance of this finding, because it represents DNA methylation in a region of the *Gstp* gene, which is a member of the glutathione *S*-transferase (GST) gene family, involved in carcinogenesis.]

In relation to targeted gene expression, <u>Wen</u> et al. (2022) exposed CD-1 mice to PFOS at doses ranging from 5–20 mg/kg per day for 14 days. PFOS was associated with increased expression of the histone demethylases *Kdm1a* and *Kdm4c* in the kidney (<u>Wen et al., 2022</u>).

In relation to miRNAs, <u>Wang et al. (2012b)</u> observed decreased expression of miR-466b, miR-672, and miR-297 in the brains of neonatal albino Wistar rats on postnatal day (PND) 1 and PND7 that were born from mothers fed with PFOS at a dose of 3.2 mg/kg per day from GD1 to PND7, compared with neonatal brain tissue derived from mothers treated with vehicle (<u>Wang et al., 2012b</u>).

Li et al. (2022c) observed an inverse association between PFOS, at the highest dose, and miR-770 expression in the placenta sampled on GD18 from CD-1 pregnant mice treated with PFOS at doses ranging from 0.5 to 12.5 mg/kg per day by gavage from GD0 to GD17. Similarly, the expression of *MEG3*, a cancer suppressor gene, was significantly decreased in the placenta, and there was hypermethylation in a CpG site in its promoter region (Li et al., 2022c).

Wang et al. (2015b) fed pregnant albino Wistar rats with chow containing PFOS at a dose of 3.2 mg/kg per day from PND1 to PND7 and reported that 35 miRNAs were highly expressed on PND1, eight miRNAs were highly expressed on PND 7, and four miRNAs (miR-125a-3p, miR-23a*, miR-25*, and miR-494) were significantly expressed on both PND1 and PND7.

Related to histone modifications, PFOS was found to be associated with increased H3K9me2, H3K9ac, and H3K18ac, and decreased H3K9me3, in the testes of male Wistar rats treated with PFOS at a dose of 0.015 or 0.15 mg/kg per day for 60 days (<u>Alam et al., 2021</u>). PFOS was also associated with decreases in histone acetylation of the StAR promoter in the ovaries of ICR mice exposed to PFOS at a dose of 0.1 mg/kg per day for 4 months (Feng et al., 2015).

PFOS was associated with decreases, observed on GD18, in the expression of lncRNA Xist in the placentas of CD-1 mouse dams exposed to PFOS at doses of 0.5, 2.5, or 12.5 mg/kg from GD1 to GD17 (Chen et al., 2018) (see Table 4.13).

(ii) Non-human mammalian systems in vitro

See <u>Table 4.14</u>.

In vitro testing has been carried out in cells that represent the mouse brain (mHypoE-N46), mouse liver, embryonic stem cells, and macrophages; as well as in rat mitochondria and kidneys. In addition, zebrafish embryos and bovine tissues have been assessed.

Exposure to PFOA

<u>Kim et al. (2021)</u> examined the effects of PFOA exposure at concentrations of $0.25-250 \mu mol/L$ on the embryonic hypothalamic cell line N46 (mHypoE-N46). PFOA was found to increase global DNA methylation.

Mouse fibroblast preadipocytes (3T3-L1) exposed to PFOA at concentrations of $0.01-100 \mu g/mL$ showed decreases in global DNA methylation compared with control (<u>Ma et al., 2018</u>).

No apparent DNA methylation was observed in a study by Starkov and Wallace performed in mitochondria isolated from the livers of Sprague-Dawley rats and exposed to 100 μ M PFOA (Starkov and Wallace, 2002), or in the kidneys of Balb/c mice treated with PFOS at 0.1 or 1 mg/kg intraperitoneally every other day for 3 months, and rat renal tubular epithelial NRK-52E cells treated with PFOS (0–500 nM) for 6, 12 or 24 hours (Chou et al., 2017). However, Chou et al. (2017) observed increases in the upregulation of Sirt1, and in the deacetylation of peroxisome proliferator-activated receptor gamma (PPAR γ) that mediated epithelial-mesenchymal transition-associated renal fibrosis.

Exposure to PFOS

PFOS exposure was associated with increased expression of miR-9-3p in Sertoli cells and Leydig cells of male ICR mice treated orally with PFOS at doses ranging between 0.5 and 10 mg/kg bw per day for 4 weeks. PFOS exposure was not associated with changes in miR-1954 or miR-710 in Sertoli cells (Huang et al., 2022a).

Blanc et al. (2019) exposed zebrafish embryos and liver (ZF-L) cells to PFOS at a concentration equivalent to its EC_{10} (93 µM) for 96 hours and showed an increase in global DNA methylation of 13% versus control. Hallberg et al. exposed bovine oocytes complexes to PFOS at concentrations of 2 ng/g or 53 ng/g for 22 hours during their maturation and found both increases and decreases in target DNA methylation on day 8 after fertilization (Hallberg et al., 2021). The authors identified that the most altered pathways were those involved in cell death and survival, with the p53 pathway the most altered (see Table 4.15).

Synopsis

[The Working Group noted that numerous studies investigating epigenetic alterations in humans exposed to PFOA and PFOS were in cohorts with background exposures representative of the general population. Many of these studies were prospective birth cohort studies that evaluated the relation between maternal/ in utero exposure and epigenetic alterations in the neonate. These studies are of great importance, because they investigate the potential for developmental reprogramming that may influence cancer susceptibility. Numerous studies in exposed humans have identified associations between exposure to PFOA or PFOS and altered DNA methylation in cancer-associated genes. For PFOA and PFOS, the specific gene targets identified in these studies differed. The Working

Table 4.14 End-points relevant to epigenetic alterations in non-human mammalian systems in vitro exposed to PFOA or PFOS

End-point	Species	Tissue, cell line	Result ^a	Concentrations or dosing regimen	Comments	Reference
Global DNA methylation	Mouse	Brain, embryonic hypothalamic cell line N46 (mHypoE-N46)	Global DNA methylation +	PFOA, ranging study of 0–250 μ mol/L for 24 h or 48 h; cells were then exposed to PFOA at EC ₅₀ = 27.5 μ mol/L, for 24 h		<u>Kim et al.</u> (2021)
Global DNA methylation; DNA methyltransferase gene expression	Mouse	Fibroblasts, preadipocytes (3T3-L1)	Global DNA methylation –; DNA methyltransferase genes +	PFOA, 0.01–100 μg/mL for 4–8 days		<u>Ma et al.</u> (2018)
Permeability of mitochondrial membranes	Rat	Liver, mitochondria	+	PFOA, 100 μM; PFOS: 10 μM; concomitant measurements	No apparent DNA methylation or miRNA	<u>Starkov and</u> <u>Wallace</u> (2002)
Targeted DNA methylation	Cow	Egg, bovine cumulus oocyte	PFOS +/-; gene-dependent	PFOS, 2 or 53 ng/g for 22 h		<u>Hallberg</u> <u>et al. (2021)</u>
Targeted gene expression	Rat	Kidney, renal tubular epithelial cells (NRK-52E)	+ expression of EMT and renal injury biomarkers (e.g. <i>N</i> -cadherin, vimentin, <i>Snai1</i> , <i>Kim1</i> , and <i>Lcn2</i>); - expression of <i>Tjp1</i> ; + expression of <i>Sirt1</i>	PFOA, 0–500 nM for 24 h	No apparent DNA methylation or miRNA	<u>Chou et al.</u> (2017)
Targeted gene expression (Sirt1–7)	Mouse	Macrophages (RAW 264.7 cells)	SIRT1–7 mRNA: PFOS +/–; PFOA +/–	PFOS or PFOA, 0, 0.5, 5, or 50 μM for 24 h		<u>Park et al.</u> (2019)
Targeted gene expression, histone modifications; global DNA methylation	Zebrafish	Embryo, liver, ZFL cells	Global DNA methylation: PFOS +	PFOS at $EC_{10} = 93 \ \mu M$, for 48 h		<u>Blanc et al.</u> (2019)
Targeted miRNA expression (miR-145 and miR-490-3p) and mRNA and protein expression	Mouse	Embryonic stem cells	PFOS: miR-145 expression +, miR- 490-3p expression +, <i>Sox2</i> mRNA -, Sox2 protein -, <i>Nanog</i> mRNA -, Nanog protein -, <i>Oct4</i> mRNA +/-, Oct4 protein +/-, <i>Chrm2</i> expression +	PFOS, 0.2, 2, 20, or 200 μM for 24 h		<u>Xu et al.</u> (<u>2013b)</u>

Table 4.14 (continued)

End-point	Species	Tissue, cell line	Resultª	Concentrations or dosing regimen	Comments	Reference
Targeted miRNA expression (miR-134, miR-145, miR-490-3p)	Mouse	Embryonic stem cells, D3	miR-134 ↓, miR-145 ↓, miR-490-3p ↓	PFOS, 0.2, 2, 20, or 200 μ M; medium changed on days 0, 2, and 4; measurement on day 6		<u>Xu et al.</u> (2015)

EMT, epithelial-mesenchymal transition; h, hour(s); mRNA, messenger RNA; miRNA, microRNA; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid. ^a +, increased methylation; -, decreased methylation; +/-, increase or decrease in methylation; ↑, increased gene expression; ↓, decreased gene expression; ↑↓, increased and decreased gene expression.

End-point Assay	Biosample type	Location, setting Study design	Exposure level No. of participants	Response	Covariates adjusted for	Comments	Reference
MDA, by TBARS 8-OHdG, by HPLC-MS/MS	Urine	Seoul, Republic of Korea Placebo- controlled crossover trial	PFOS, 10.04 ng/mL; PFOA, 4.61 ng/mL; 141 participants, age > 60 years	↑ MDA and 8-OHdG in a dose-dependent response with PFOS, but not PFOA	Age, sex, BMI, cotinine level, PM_{10} , outdoor temperature, dew point, treatment arm, and treatment	Vitamin C did not significantly reduce MDA or 8-OHdG in the PFOA or PFOS group	<u>Kim et al.</u> (2016b)
8-oxodG, 15-F2t-isoP, by ELISA	Urine, plasma	Ceske Budejovice, Prague, and Ostrava, Czechia Cross-sectional study	PFOA means: 1.05 (Ceske), 0.96 (Prague), 0.98 (Ostrava) ng/ mL; PFOS means: 3.49 (Ceske), 3.23 (Prague), 3.35 (Ostrava) ng/mL in plasma 126 healthy non- smoking policemen; mean age, 38–40 years	↑ 8-OHdG and ↓ 15-F2t-isoP. Statistically significant ($P < 0.05$) association only between PFOS and 8-OHdG	Sampling period and locality, non-smoking policemen	Plasma PFOS and PFOA concentrations did not differ between the three different areas	<u>Ambroz et al.</u> (2022)
8-OHdG 8-Nitrosoguanine by HPLC-ESI-MS/MS	Urine	Taipei, Taiwan, China Cross-sectional study	<i>n</i> -PFOA, 3.77 ng/mL; branched PFOA, 0.08 ng/mL; <i>n</i> -PFOS, 12.92 ng/mL; branched PFOS, 0.44 ng/mL 597 participants (519 men and 78 women, aged 22–63 years; mean, 45.8 years)	↑ in a dose- dependent manner across the four quartiles of linear PFOS, but not PFOA Positive association with linear PFOS using 3 models	Model 1 was adjusted for age and sex; Model 2 was adjusted for the Model 1 parameters plus smoking status, alcohol intake, education level, BMI, hypertension and diabetes mellitus; Model 3 was adjusted for the Model 2 parameters plus LDL-C and urinary creatinine as covariates (multiple linear regression)	Recruited participants were controls and patients with acute coronary heart disease from another study; therefore, few women were enrolled in this study Both branched and linear PFOS and PFOA were measured in serum	<u>Lin et al.</u> (2020b)

Table 4.15 End-points relevant to oxidative stress in humans exposed to PFOA or PFOS

Table 4.15	(continued)
	continueu)

End-point Assay	Biosample type	Location, setting Study design	Exposure level No. of participants	Response	Covariates adjusted for	Comments	Reference
8-OHdG, by LC-MS/MS	Urine	Taiwan, China Cross-sectional study	Geometric mean and 95% CI, 3.21 (3.00–3.46) ng/mL for PFOA, 6.44 (6.05–6.89) ng/mL for PFOS 848 participants (331 men and 517 women, aged 12–30 years	No association with PFOA or PFOS using Models 1 and 2; Model 1 was adjusted for age and sex; Model 2 was adjusted for age, sex, and other risk factors, such as smoking status, BMI, systolic blood pressure, LDL-C, HDL-C, and hs-CRP	Age, sex, systolic blood pressure, LDL-C, HDL-C, insulin resistance, serum hs- CRP, history of medication, income, smoking, alcohol consumption, BMI, hypertension, diabetes mellitus	Sufficient sample size; serum PFC and microplastics were analysed	<u>Lin et al.</u> (2016)
8-OHdG, by ELISA	Urine	Flanders, Belgium	PFOA, 2.55 μg/L; 596 adolescents (324 males and 282 females, aged 14–15 years)	Serum PFOA concentration (n = 197) weakly associated (not statistically significant) with 8-OHdG $(n = 195)$, and associated with increased DNA damage (alkaline comet assay, $n = 598$)	Sex, age, BMI, smoking habits, alcohol consumption, education level, season of sample collection	Study limitation: use of ELISA method; the presence of contaminants, including lead, chromium, cadmium, arsenic, methyl mercury, and PAH complicated the interpretation	<u>Franken</u> <u>et al. (2017)</u>
Bilirubin (an antioxidant) Albumin (extracellular antioxidant) NHANES 2007–2008 Laboratory Data Overview (<u>National</u> <u>Center for Health</u> <u>Statistics, 2023</u>)	Serum	USA (NHANES 2005–2012) Cross-sectional study	Serum 95th percentile of PFOA, 8.90 (max. 104.0) ng/mL; PFOS, 49.40 (max. 281.0) ng/mL 6652 participants (3246 men, 3406 women, aged 49.48 ± 18.07 years)	Positive associations with PFOA and PFOS; dose- dependent	Age, sex, education, ethnicity, income level, cotinine, and BMI	Large sample size. However, the classical and most informative oxidative end- points were not measured. These two end-points are not considered to be specific to oxidative stress	<u>Omoike et al.</u> (2021)

Table 4.15 (continued)							
End-point Assay	Biosample type	Location, setting Study design	Exposure level No. of participants	Response	Covariates adjusted for	Comments	Reference
α-klotho antioxidant, by ELISA	Serum	USA (NHANES 2007–2016) Cross-sectional study	3981 participants (1940 men, 2041 women, aged 40–79 years)	Positive associations of PFOA and PFOS with α-klotho in participants with GF- 3B/4 stage, without albuminuria. Inverse associations between PFOS and α-klotho, with healthy kidney	Age, sex, ethnicity, obesity status, hypertension, diabetes, smoking, anaemia, alcohol consumption, glomerular filtration stage, and albuminuria	Large sample size. However, the classical oxidative biomarkers were not measured	Jain and Ducatman (2022)
Metabolome biomarkers of oxidative/ nitrosative stress: hydroxybutyric acid, pyroglutamic acid, oxoglutaric acid, D-glucurono- 6,3-lactone, deoxyarabinohexonic acid, tetrahydrobiopterin, a-carboxyethyl hydroxychromanol, and arachidonic acid, by LC/orbitrap-MS	Serum	China	Median concentration: PFOA, 7.56 nM; PFOS, 12.78 nM 181 male participants, aged 22–48 years	PFOA and PFOS were associated with direct or indirect biomarkers of oxidative/nitrosative stress	Age, BMI, smoking, alcohol consumption	Small sample of men in a single region with incomplete demographic data; a metabolomic approach, involving the measurement of biomarkers directly or indirectly involved in the oxidative/nitrosative pathways using a state-of-the- art instrument. The metabolism biomarkers were not specific for oxidative/nitrosative stress	<u>Wang et al.</u> (<u>2017)</u>

Table 4.15 (continued)								
End-point Assay	Biosample type	Location, setting Study design	Exposure level No. of participants	Response ^a	Covariates adjusted for	Comments	Reference	
ROS, by OxiSelect ROS/RNS assay kit	Umbilical cord plasma	Shanghai, China Prospective study	PFOA: girls, 0.70–29.97 ng/mL; boys, < LOD to 25.99 ng/mL PFOS: girls, 0.39–18.68 ng/mL; boys,0.62–65.61 ng/mL Newborns (299 boys, 282 girls)	Positive association with PFOS in female newborns; however, in male newborns, there were no relations with PFOA or PFOS	Maternal and paternal age, maternal education, maternal pre- pregnancy BMI, mode of delivery, gestational age at birth, infant sex, infant birth weight, and antepartum obstetric risk	It was impossible to distinguish the separate effects of PFOA and PFOS	<u>Liu et al.</u> (2018b)	
8-isoprostane- PGF2α, PGF2α, 2,3-dinor-8-iso- PGF2α, and 2,3-dinor-5,6- dihydro-8-iso- PGF2α, by LC-MS/MS	Urine	Illinois, USA Prospective birth cohorts	Geometric means: PFOA, 0.75 ng/mL; PFOS, 2.03 ng/mL 428 pregnant (15- week) mothers, aged 18–40 years	Association with PFOS but not PFOA	Sociodemographic backgrounds and geographical locations; clinical characteristic and trimesters	Relatively small sample size	<u>Taibl et al.</u> (2022)	
HO-1, by ELISA kit	Plasma	Shanxi, China Case–control study	PFOS median, 1.79 ng/mL; PFOA median, 0.79 ng/mL 144 spontaneous preterm births, and 375 full-term deliveries as controls	No association with HO-1 observed	Demographic characteristics	A nested case- control study, minimizing selection and recall bias; however, subgroups were small	<u>Liu et al.</u> (<u>2020a)</u>	

BMI, body mass index; h, hour(s); CI, confidence interval; ELISA, enzyme-linked immunosorbent assay; ESI, electrospray ionization; GF-3B/4 stage, glomerular function stage of kidney disease; HDL-C, high-density lipoprotein-cholesterol; HO-1, haem oxygenase 1; HPLC, high-performance liquid chromatography; hs-CRP, high-sensitivity C-reactive protein; LC, liquid chromatography; LC-MS/MS, liquid chromatography-tandem mass spectrometry; LDL-C, low-density lipoprotein-cholesterol; LOD, limit of detection; LTL, leukocyte telomere length; MDA, malonaldehyde; NHANES, National Health and Nutrition Examination Survey; *n*-, linear isomer; 8-NO2Gua, 8-nitrosoguanine; 8-OHdG, 8-hydroxydeoxyguanosine; 8-oxodG, 8-oxo-2'-deoxyguanosine; PAH, polycyclic aromatic hydrocarbon; PFC, perfluorochemicals; PFOA, perfluorooctanoic acid; PFDA, perfluorononanoic acid; PFOS, perfluoroctanesulfonic acid; PG, prostaglandin; PM₁₀, particulate matter of < 10 µm in diameter; RNS, reactive nitrogen species; ROS, reactive oxygen species; SD, standard deviation; TBARS, thiobarbituric acid-reactive substance; USA, United States of America; yr, year(s). ^a ↓, decrease; ↑, increase.
Group analysed the strengths and limitations of individual studies and considered the relevance of the KC-associated end-points across systems. The Working Group noted that the significant findings in humans are unlikely to be attributable to chance. The Working Group noted that the data in experimental systems corroborate the evidence observed in exposed humans. Specifically, in vivo and in vitro studies using rodents and human- and rodent-derived cells suggest that PFOA and PFOS alter DNA methylation, LINE methylation, histone modifications, and miRNA expression.]

4.2.5 Induces oxidative stress

(a) Humans

See <u>Table 4.15</u>.

(i) Exposed humans

Eleven studies on PFOA and PFOS (or both) relevant to exposed humans were identified in which oxidative stress-related end-points were measured.

Kim et al. (2016b) conducted a doubleblind, randomized, placebo-controlled crossover trial to study the effect of vitamin C on PFOA and PFOS-induced insulin resistance in 141 healthy elderly participants aged > 60 years from the Korean Elderly Environmental Panel (KEEP) study in Seoul, Republic of Korea. Serum PFOA and PFOS concentrations were $4.61 \pm 1.86 \text{ ng/mL}$ (11.1 $\pm 4.5 \mu$ M) and 10.04 ± 4.12 ng/mL (20.1 \pm 8.2 μ M), respectively. PFOS, but not PFOA, was positively associated with increased urinary end-points of oxidative stress malondialdehyde (MDA, P = 0.02) and 8-hydroxy-2'-deoxyguanosine (8-OHdG, P = 0.001) levels. Vitamin C treatment (1000 mg/day for 4 weeks) did not significantly alter MDA or 8-OHdG levels induced by PFOA or PFOS. The measurement of both end-points included the use of radiolabelled internal standards. MDA was measured using the thiobarbituric

acid-reactive substance (TBARS) assay. The urinary 8-OHdG concentration was measured using HPLC-tandem mass spectrometry (MS/MS). [The Working Group noted that the TBARS method was not specific for malonaldehyde but, nevertheless, the MDA results complemented those for 8-OHdG.]

In a recent cross-sectional study (Ambroz et al., 2022), 126 healthy, non-smoking adult policemen from three areas of Czechia were sampled for their plasma PFOA, PFOS, and 15- F_{2t} -isoprostane (IsoP), and urinary 8-oxo-2'deoxyguanosine (8-oxodG). 15- F_{2t} -IsoP, and urinary 8-oxodG were measured using enzymelinked immunosorbent assay (ELISA) kits. The PFOA and PFOS concentrations correlated with elevated urinary 8-oxodG levels and reduced plasma 15- F_{2t} -IsoP, but only the association between PFOS and 8-OHdG was statistically significant.

A cross-sectional study (Lin et al., 2020b) was conducted in 597 adult (519 men and 78 women) from a middle-aged cohort (mean age, 45.8 years) from the National Taiwan University Hospital, China, to assess the associations of serum isomers of PFOA and PFOS (branched and linear) with urinary 8-OHdG and 8-nitroguanine (8-NO₂Gua). The levels of urinary 8-OHdG and 8-NO₂Gua were measured using HPLCelectrospray ionization (ESI)-MS/MS with satisfactory accuracy. Branched PFOA and PFOS were 2.1% and 3.2% of the total PFOA and PFOS concentrations, respectively. The geometric means of urinary 8-OHdG and of 8-NO,Gua significantly increased across the four quartiles (< 8.39 and > 22.3 ng/mL for *n*-PFOS; from 6.82 to 8.65 μ g/mL for 8-OHdG, *P* for trend = 0.016; and from 0.78 to 1.21 µg/mL for 8-NO₂Gua; P for trend, 0.041) in multiple linear regression analysis, after controlling for potential confounders; however, this was not true for PFOA. The results indicate that *n*-PFOS in serum was significantly associated with urine oxidative/nitrative stress end-points in a relatively large cohort in Taiwan.

Lin et al. (2016) investigated the relation between serum PFOA and PFOS and urinary 8-OHdG in 848 participants aged 12-30 years (331 men and 517 women) in a cross-sectional study in Taiwan, China. Urinary 8-OHdG concentrations were measured using liquid chromatography (LC)-MS/MS, with the inclusion of a suitable internal standard (15N₅8-OHdG). The geometric means of serum PFOA and PFOS were 3.21 ng/mL (95% CI, 3.00-3.46 ng/mL) and 6.44 ng/mL (95% CI, 6.05-6.89 ng/mL) (i.e. 7.8, 7.2-8.4 µM and 12.9, 12.1-13.8 µM), respectively. There were no associations of serum PFOA and PFOS concentrations with 8-OHdG levels in the urine. [The Working Group noted that the cohort was younger, and the exposure level was lower than those in the most 8-oxodG studies in humans exposed to PFOS.]

In a cross-sectional study of a cohort of approximately 600 adolescents (both males and females aged 14-15 years) from the Flanders region of Belgium, urinary 8-OHdG as a measure of DNA damage was assessed by ELISA in 596 adolescents (see also Section 4.2.4) (Franken et al., 2017). In addition, damage to DNA was assessed with alkaline comet assay, and specifically, oxidative damage to DNA was also assessed with the Fpg-modified comet assay in a subpopulation of the cohort. Increased serum PFOA levels were associated with a 9% increase in DNA damage, as measured by the alkaline comet assay (95% CI, 1.5–17.0%; *n* = 196); however, in 195 participants the PFOA level was only weakly and not significantly associated with increased 8-OHdG.

In another cross-sectional study, <u>Omoike et al.</u> (2021) analysed data from the National Health and Nutrition Examination Survey (NHANES) cohort (2005–2012; n = 6652, 3246 men and 3406 women; age, 49.40 ± 18.07 years) that included the measurement of serum antioxidants bilirubin and albumin (extracellular antioxidants) as indicators of oxidative stress. The 95th percentile values for PFOA and PFOS were 8.90 (maximum, 104) ng/mL (i.e. 21.5, max. 251.2 μ M) and 42.70

(max. 281) ng/mL (i.e. 85.4, max, 561.9 μ M), respectively. PFOA and PFOS were found to be positively associated with bilirubin and albumin levels. [The Working Group noted that the actual serum levels of bilirubin and albumin were not reported in this study. Also, although the study included a substantial number of participants, the most informative oxidative stress-related end-points were not assessed; those measured were not specific for oxidative stress.]

Another NHANES cross-sectional study (Jain and Ducatman, 2022) analysed data for 3981 US adults (aged 40-79 years), 3461 with and 530 without albuminuria. The study authors hypothesized that PFAS may adversely affect the antioxidant response of the normal kidney. Serum PFAS, including PFOA and PFOS, and α-klotho, an anti-ageing protein that plays a key role in the production of antioxidant enzymes in the kidney, were measured. A positive association between PFOA or PFOS and a-klotho was observed in participants without albuminuria and kidney function in glomerular function (GF) stage 3B/4 $(15 \le eGFR < 45 \text{ mL/min}/1.73 \text{ m}^2)$. In stage GF-1 (eGFR \ge 90 mL/min/1.73 m²), an inverse association between PFOS and a-klotho was observed in individuals without albuminuria (Jain and Ducatman, 2022).

Wang et al. (2017) studied serum PFAS, including PFOA and PFOS, and serum metabolome markers related to oxidative/nitrosative stress in 181 Chinese men (aged 22-48 years; mean \pm SD, 33.2 \pm 6.4 years). Metabolism end-points included hydroxybutyric acid, pyroglutamic acid, oxoglutaric acid, D-glucurono-6,3-lactone, deoxyarabinohexonic acid, tetrahydrobiopterin, α-carboxyethyl hydroxychromanol, and arachidonic acid, and were measured using LC-Orbitrap-MS. The metabolism markers directly or indirectly correlated with lipid oxidation. Associations between PFOA or PFOS and oxidative/nitrosative stress-related end-points were observed. The authors suggested that low environmental levels of PFAS, including PFOA

and PFOS, may increase the early risk of metabolic diseases, including diabetes and cardiovascular diseases. [The Working Group noted that this study included a small male cohort from a single region with incomplete demographics, which complicates the interpretation of the observations. It was noted that the metabolic markers measured in the study might not all be specific for oxidative/nitrosative stress. However, they are involved and/or result from an oxidative stress process.]

Liu et al. (2018b) studied umbilical cord plasma PFAS, including PFOA and PFOS, and ROS and fetal leukocyte telomere length (LTL) in 299 newborn boys and 282 newborn girls from a prospective cohort in Shanghai, China. The ROS/ reactive nitrogen species (RNS) ratios in the cord plasma were measured using a commercial kit. Shorter LTL and high levels of ROS were observed in female newborns, and these were associated with PFOS. No association with PFOA or PFOS was observed in male newborns. [The Working Group noted that in this study the umbilical cord plasma PFOA level was higher than that for PFOS, in contrast to what is normally observed in adults. It was impossible to distinguish the effect of individual PFOA and PFOS exposure, because they are strongly positively correlated, as acknowledged by the authors.]

Taibl et al. (2022) tested the hypothesis that excess ROS might be a contributor to preterm birth (at less than 15 weeks of gestation) in 428 pregnant mothers (aged 18–40 years), not pregnant with multiple fetuses, from the Illinois Kids Development Study (IKIDS) and Chemicals in Our Bodies (CIOB) prospective birth cohorts between 2014 and 2019 in the USA. Serum PFAS levels (second trimester) and urinary levels of end-points of oxidative stress (second and third trimesters), including prostaglandin-F2α (PGF2α), 8-isoprostane-prostaglandin F2α (8iso-PGF2α), 2,3-dinor-8-iso-PGF2α, and 2,3dinor-5,6-dihydro-8-iso-PGF2α were measured using LC-MS/MS. The 95th percentiles and geometric means \pm SD of PFOA and PFOS were 2.17 (0.75 \pm 2.14) ng/mL (5.2, 1.7 \pm 5.2 μ M) and 7.01 (2.03 \pm 2.47) ng/mL (14, 4.1 \pm 4.9 μ M), respectively. PFOS was found to have a modest positive association with increases in the oxidative end-points measured.

Liu et al. (2020a) conducted a nested casecontrol study in Shanxi, China, that included 144 women who underwent spontaneous preterm birth and 375 who underwent full-term delivery as controls. Among the 17 PFAS measured in the maternal plasma, the median values of PFOA and PFOS were 0.79 ng/mL (1.9 μ M) and 1.79 ng/mL $(3.6 \,\mu\text{M})$, respectively. No significant differences in haem oxygenase 1 (HO-1) in the maternal plasma were observed between the spontaneous preterm birth group and the controls, with no association between HO-1 level and PFOA or PFOS. [The Working Group noted that the nested case-control study was designed to minimize selection and recall bias. However, the small sample size was considered a limitation.]

Of the eleven human studies evaluated, six were in Asian countries including China (including Taiwan) and the Republic of Korea; three studies were in the USA; one in Belgium; and one in Czechia. Among the studies, eight demonstrated a positive association between PFOS exposure and oxidative stress, and four showed that PFOA might cause oxidative stress in humans. Five studies included the measurement of the 8-OHdG or 8-oxodG form of the oxidative product, a relevant end-point of oxidative DNA damage. Three of the five studies reported a positive association for PFOS, and only one a positive association for PFOA, with 8-OHdG.

Among the studies that reported a positive association for PFOS, only two measured 8-oxodG with a highly specific analytical method. The evidence for PFOA is weak compared with that for PFOS. It was also noted that the serum/ plasma PFOS concentrations were generally higher than the PFOA concentrations, except in the umbilical cord plasma, as reported by <u>Liu</u> et al. (2018b).

[The Working Group noted that four prospective studies reported associations of the urinary excretion of 8-oxodG with the risks of specific cancers, although they did not evaluate any PFAS with this effect. Two of the studies showed associations with lung cancer (Loft et al., 2006, 2012) and the other two studies showed associations with breast cancer (Loft et al., 2013; Broedbaek et al., 2015). It was noted that the study by Broedback et al. was in patients with type 2 diabetes. In these studies few cancer cases were included; however, the studies were considered relevant because the authors reported on effects on this particular oxidative damage-related end-point.]

[The Working Group also noted there was a recent systematic review article (Chen et al., <u>2023a</u>) that analysed the associations of exposure to persistent organic pollutants (POPs) in humans, including PFAS (PFOA and PFOS were included), with oxidative stress end-points. The review included typical oxidative stress end-points, such as 8-OHdG, ROS, MDA, reduced glutathione (GSH), catalase (CAT), and superoxide dismutase (SOD), and some clinical diagnostic markers (gamma-glutamyl transferase, GGT), uric acid, and bilirubin). The metaanalysis of <u>Chen et al. (2023a)</u> acknowledged that the association of PFOA or PFOS with oxidative stress in exposed humans would require further studies.]

(ii) Human cells in vitro See <u>Table 4.16</u>.

Human primary cells

There have been fewer studies using human primary cells in vitro compared with immortalized cells to investigate oxidative stress induced by PFOA or PFAS. Alterations of oxidative stress end-points were observed in cells treated with non-cytotoxic concentrations ranging from submicromolar to under 100 μ M in most studies. Most studies set P < 0.05 for significant differences.

Han et al. (2020) exposed human epidermal keratinocytes and human dermal fibroblasts within a human full-thickness skin model (EpiDermFT (EFT-400), MatTek Corporation, Ashland, MA, USA) to PFOA at 250 μ M or 2500 μ M for 6 days. MDA lipid peroxidation marker levels, measured using TBARS, were significantly increased at both concentrations in the treated cells, but no significant changes in 8-OHdG were observed, as measured using an ELISA kit (Han et al., 2020).

For experiments using human primary cells, Orbach et al. (2018) used primary liver cells to assemble a multicellular organotypic culture model in 96-well plates (µOCMs) and collagen sandwich (CS) culture. Commercially available primary hepatocytes, derived from two adult males, were exposed to ½ or 1 median lethal concentration (LC₅₀) (250 or 500 μ M) of PFOA for 24 hours. GSH was measured to evaluate PFOA-induced oxidative stress, and PFOA was found to reduce GSH to < 10% of control (Orbach et al., 2018). [The Working Group noted that GSH was measured using the commercial HTS GSH-Glo assay kit and that GSH reduction because of binding or oxidation did not occur because cytochrome P450 (CYP) enzymes do not oxidize PFOA. It was also noted that the doses of $\frac{1}{2}$ LC₅₀ and 1 LC₅₀ were considered high.]

In a study from Pan et al. (2018), human erythrocytes were isolated from blood samples collected from six healthy, non-smoking adults from China. Cells were exposed to three PFAS compounds, including PFOA, at 0, 5, 10, 50, or 100 μ M for 3 hours, then MDA, GSH, glutathione peroxidase (GPx), SOD, and CAT were measured using the TBARS, 2, 3-naphthalenedicarboxaldehyde, 5, 5-dithiobis (2-nitrobenzoic acid), SOD assay kit, and ammonium molybdate methods, respectively. PFOA at 100 μ M induced a significant increase in MDA level, GSH levels were reduced by 10 and 100 μ M, and the CAT and

End-points	Assay	Tissue, cell line	Result ^a	Concentrations or dosing regimen	Comments	Reference
Primary cells						
8-oxodG, MDA	ELISA for 8-oxodG, TBARS for MDA	EpiDermFT skin model (human epidermal keratinocytes and human dermal fibroblasts cells)	No change in 8-oxodG ↑ MDA	PFOA: 250 or 2500 μM, 6 days of treatment every other day	No positive control; levels of 8-oxodG in unexposed controls (0.3 ng/µg DNA, corresponding to 1740 lesions/10 ⁶ dG); significant increase in MDA	<u>Han et al.</u> (2020)
GSH	HTS GSH-Glo glutathione assay	Primary human hepatic cells	↓ GSH at 250 µM	PFOA: 250 or 500 μM (1/2 LC ₅₀ and LC ₅₀), 24 h	PFOA reduced GSH to < 10%; CYP enzymes did not oxidize PFOA, GSH depletion due to binding or oxidation did not occur; doses of $\frac{1}{2}$ LC ₅₀ and 1 LC ₅₀ were considered to be high	<u>Orbach</u> <u>et al. (2018)</u>
MDA, GSH, GPx, SOD, CAT	TBA, 2, 3-NDA, DTNB, SOD by assay kits with ammonium molybdate	Erythrocytes	↑ MDA, ↓ GSH, ↓ GPx, ↓ CAT, no change in SOD	PFOA: 10 or 100 μM, 3 h for GSH; 5, 10, 50, or 100 μM for MDA, GPx, CAT, SOD	MDA: $(100 \ \mu\text{M})$ significant increase; GSH (10 and 100 μ M), GPx (100 μ M), and CAT (100 μ M): significant decrease; SOD: no change. The results suggest PFOA induces oxidative stress in erythrocytes	<u>Pan et al.</u> (<u>2018)</u>
ROS	Muse Oxidative Stress Kit, followed by flow cytometry	Embryonic stem cell system – primary spermatocytes, secondary spermatocytes, and spermatids	No increase in ROS	PFOA: 11, 25, or 100 μ M; PFOS: 24, 48, or 126 μ M on days 1–10 of the differentiation process	No significant effect on ROS generation PFOS (all concentrations) and PFOA (11 and 25 µM) exposure resulted in significantly lower ROS levels	<u>Steves et al.</u> (2018)
ROS	ROS kit	Sperm	↑ ROS	PFOA 0.25, 2.5, or 25 μg/mL (0.6, 6, or 60 μM) for 0.5 or 4 h	$60 \ \mu M$ significantly induced ROS production after 4 h	<u>Yuan et al.</u> (2020)
ROS, MDA, GSH, GSSG	DCFH-DA, TBARS, <i>o</i> -phthalaldehyde fluorescence detection	Lymphocytes	↑ ROS, ↑ MDA, ↓ GSH, ↑ GSSG	PFOS, 75, 150, or 300 μM for 2, 4, 6, 8, 10, or 12 h	20 healthy adults: aged 18–30 yr, sex not reported ↑ ROS in a time- and dose– dependent manner ↑ MDA after 6 h at 150 and 300 µM; ↓ GSH and ↑ GSSG after 4, 8, 10, and 12 h, but not after 6 h Biomarker data supportive of oxidative stress	<u>Zarei et al.</u> (<u>2018)</u>

Table 4.16 End-points relevant to oxidative stress in human cells in vitro exposed to PFOA or PFOS

500

Table 4.16 (continued)

End-points	Assay	Tissue, cell line	Resultª	Concentrations or dosing regimen	Comments	Reference
ROS	Electron spin resonance (ESR) spectroscopy, DHE-fluorescence– confocal microscopy, DHE-fluorescence microplate reader	Microvascular endothelial (HMVEC) cells	↑ ROS	PFOS, 10, 20, 50, or 100 μM for 1 h; 2 μM for 1, 2, 3, or 5 h (time study)	\uparrow ROS at 50 or 100 μM by DHE method; \uparrow ROS at 2 μM at all time points (1–5 h)	Qian et al. (2010)
ROS, GPx, HO-1	DCFH-DA, western blotting	Proximal tubular epithelial (HK-2) cells	↑ ROS, ↓ GPx4, ↓ HO-1	PFOS, 200 μM for 12 h	Significant ↑ ROS, ↓ GPx4, and ↓ HO-1 provided evidence of oxidative stress Only a single concentration and time point	<u>Wang et al.</u> (2022b)
ROS	DCFH-DA	Corneal epithelial (HCEpiC) cells	↑ ROS	PFOA 100, 200, or 400 ppm (mg/kg) for 6 h	The solubility of the PFOA in the culture medium and hence its bioavailability to cells was unknown	<u>Tien et al.</u> (2020)
ROS	DCF-DA	Insulin-producing EndoC-βH1 cells	No change in ROS	PFOA 1 nM or 1 μM for 24 h	No change in ROS	<u>Dos Santos</u> <u>et al. (2022)</u>
ROS	DCFH-DA	Placental trophoblast (HTR-8/SVneo) cells	↑ ROS	PFOA 0, 100, 200, 400, 600, 800, or 1000 μM for 24 h	Increased ROS in a dose-response manner	<u>Du et al.</u> (2022)
ROS	DCFH-DA	Placental trophoblast (HTR-8/SVneo) cells	↑ ROS	PFOS, 1, 10, or 50 μM for 24 h or 48 h	Significant ↑ ROS at 10 and 50 µM in a time- and dose-dependent manner	<u>Sonkar et al.</u> (2019)
ROS	DCFH-DA	Umbilical vein endothelial (HUVEC) cells	↑ ROS	PFOS, 100 mg/L (200 μM) for 1, 5, 12, 24, or 48 h	ROS increased in a time-dependent manner	<u>Liao et al.</u> (2012)
ROS	DCFH-DA	Umbilical vein endothelial (HUVEC) cells	↑ ROS	PFOS, 100 mg/L (200 μM) for 24 h or 40 h	PFOS increased ROS production after 24 or 40 h of exposure. In cells co-treated with anti-oxidant <i>Flos</i> <i>lonicerae</i> extract or chlorogenic acid for 40 h, ROS production was reduced to levels comparable to those of cells exposed to PFOS for 24 h. ROS levels were not reduced by co-treatment of cells exposed to PFOS for 24 h only	<u>Liao et al.</u> (2013)

	ontinueu)			-		
End-points	Assay	Tissue, cell line	Result ^a	Concentrations or dosing regimen	Comments	Reference
ROS	ROS sensor (CellROX)	Mesenchymal stem (hMSC) cells	↑ ROS	PFOS, 0.1 or 10 μM at 12, 24, 36 and 48 h; 7 and 14 days	↑ ROS at 10 μM at all time points, ↑ ROS at 7 days at 0.1 μM	<u>Gao et al.</u> (2020)
Cell lines						
Oxidatively damaged DNA	Fpg-modified comet assay	HepG2 cells	No change	PFOA or PFOS, 100 or 400 μM	Positive control (Ro19-8022 + light) increased DNA damage	<u>Eriksen</u> <u>et al. (2010)</u>
8-oxodG	Immunostaining	HaCaT cells	↑ ROS	PFOA, 50 μM for 24 h + 8 days recovery	No positive control; increased 8-oxodG at 8 days recovery	<u>Peropadre</u> <u>et al. (2018)</u>
8-oxodG	HPLC-MS/MS	TK6 cells	↑ ROS	PFOA, 125 or 250 μg/mL for 2 h	No positive control	<u>Yahia et al.</u> (2016)
ROS, 8-OHdG	DCFH-DA for ROS, immunocytochemical detection for 8-OHdG	HepG2 cells	↑ ROS, ↑ 8-OHdG	PFOA, 100, 200, or 400 μM for 3 h	ROS production increased in a dose-response manner; no positive control for ROS; 8-OHdG increased in a dose-response manner with H_2O_2 used as a positive control	<u>Yao and</u> <u>Zhong</u> (2005)
ROS (H ₂ O ₂ and superoxide anions)	Flow cytometry: DCFH- DA for H_2O_2 ; DHE for superoxide anion	HepG2 cells	↑ ROS	PFOA, 200 or 400 μM for 3 h	Minimum 3 replicates per treatment; ROS measured after 1.5, 3, 5, and 24 h and was found to peak at 3 h	<u>Panaretakis</u> et al. (2001)
ROS	DCFH-DA by flow cytometry; GSH, SOD, CAT by ELISA	Hep2G cells	↑ ROS, ↑ GSH, ↑ CAT, no change in SOD	PFOA, 10, 25, or 50 μM for 24 h	PFOA at 10, 25, or 50 μ M significantly increased ROS (\leq 5.3-fold at 50 μ M); GSH (1.7-fold), and CAT (1.4-fold) only at 10 μ M; no change in SOD	<u>Abudayyak</u> et al. (2021b
ROS	DCFH-DA	HepG2 cells	↑ ROS, ↓ GSH, ↓ GPx, ↑ SOD, ↑ CAT, GST	PFOA and PFOS 50, 100, 150, or 200 μM for 5, 10, or 15 h (ROS) and 48 h (GSH, GPx, SOD, CAT)	Significant increase in ROS at \geq 100 µM after 5, 10, or 15 h; GSH and GPx increased at \geq 100 µM; SOD and CAT increased at \geq 150 µM; GST increased at 200 µM	<u>Hu and Hu</u> (2009)
ROS	DCFH-DA	HepG2 cells	↑ ROS	PFOA or PFOS at 0.4, 4, 40, 200, 400, 1000, or 2000 μM; every 15 min for 3 h	Significant increases at all time points, but no dose-dependency	<u>Eriksen</u> et al. (2010)
ROS, TAC	Carboxy-DCFH-DA, antioxidant assay kit	HepG2 cells	↑ ROS, ↓ TAC	PFOA or PFOS: ROS: 0.2, 2, or 20 μM for 24 h; TAC: 0.02, 0.2, 2, 20, or 200 μM for 24 h	PFOA or PFOS significantly induced ROS; PFOA significantly reduced TAC; PFOS reduced TAC, but not significantly	<u>Wielsøe</u> et al. (2015)

. . . . AAC (continued)

Table 4.16 (continued)

End-points	Assay	Tissue, cell line	Result ^a	Concentrations or dosing regimen	Comments	Reference
ROS	DCFH-DA	HepG2 cells	↑ ROS	PFOA or PFOS: 0, 5, 25, 50, 100, 200, 400, or 800 μM for 3 or 24 h	PFOA or PFOS induced ROS in a dose-dependent manner	<u>Amstutz</u> et al. (2022)
ROS, NO	DCF, and Greiss reagent	HepG2 cells	↑ ROS, ↑ NO	PFOA: 100 or 200 μM for 24 h	PFOA increased ROS and NO levels and <i>NOS2A</i> mRNA expression	<u>Yarahalli</u> Jayaram et al. (2020)
ROS	CellRox green reagent	HepaRG and HepG2 cells	↑ ROS	PFOA: 10, 100, or 1000 nM for 72 h	Generated PFOA-resistant cells; conducted acute (1–3 days) and chronic (30–60 days) exposure experiments for the steatosis and fibrosis study	<u>Qi et al.</u> (2023)
ROS, NOx	DCFH-DA or fluorescent MAK145 for ROS; 2, 3-diaminonaphthalene (fluorometric assay kit for NOx)	HepG2 cells and keratinocyte (HaCaT) cell line	↑ ROS, ↑ NOx in both cells	PFOA: 10 μM for 24 h	The DCFH-DA assay yielded higher intensity compared with the MAK14 assay	<u>Magnifico</u> et al. (2022)
Nrf2-ARE	Luciferase assay	ARE reporter HepG2 cell line	↑ Nrf2– ARE	PFOA or PFOS: 1, 2, 3, 4, 5, or 6 μM for 24 h	PFOA and PFOS induced Nrf2– ARE activation; the effective concentration of induction ratio 1.5 (EC_{IRL5}) was 1.38 μ M for PFOA and 1.17 μ M for PFOS; supportive of oxidative stress being involved	<u>Ojo et al.</u> (2022b)
ROS	DCFH-DA	HepG2 cells	↑ ROS	PFOS: 10, 20, 30, 40, or 50 μM for 24 h; PFOA: 50 μM for 1, 3, 6, 12, or 24 h	Significant \uparrow ROS at PFOS concentrations $\ge 30 \ \mu\text{M}$ after 24 h of treatment; when treated with $50 \ \mu\text{M}$, \uparrow ROS gradually, peaking after 12 h, then lower after 24 h	<u>Wan et al.</u> (2016)
MDA, ROS, superoxide anion, SOD, CAT, GSH	TBARS for MDA, assay kits for other biomarkers	HepG2 cells	↑ ROS, no significant changes in other biomarkers	PFOS: 50, 100, or 200 μM for 24 or 72 h	Significant ↑ ROS with PFOA ≥ 100 µM after 24 h, but not 72 h; no significant changes in the other biomarkers measured; cells pretreated with NAC showed a reduction in PFOA-induced ROS production	<u>Yan et al.</u> (2015a)
ROS	DCFH-DA	HepG2 cells	No change in ROS	PFOA: 5, 10, 50, 100, 200, or 400 μM; PFOS: 5, 10, 50, 100, 200, or 300 μM; 1 or 24 h	No significant increase at any time point Avoided additional stress by not trypsinizing cells before adding DCFH-DA	<u>Florentin</u> et al. (2011)

Table 4.16 (c	Table 4.16 (continued)										
End-points	Assay	Tissue, cell line	Result ^a	Concentrations or dosing regimen	Comments	Reference					
ROS, GSH	DCFH-DA for ROS, GSH-Glo assay kit	HepG2 cells	No change in ROS, ↓ GSH	PFOA: 0.2, 2, or 20 μM for 24 h PFOS: 0.2, 2, or 20 μM for 24 h	No increase in ROS but a decrease in GSH as the PFOA or PFOS concentration increased; 20 µM PFOA or PFOS induced a significant decrease in GSH; oxidative stress implicated, but unclear; the avoidance of additional physical stress by not trypsinizing cells before adding DCFH-DA may have been a factor	<u>Ojo et al.</u> (2021)					
ROS, GSH/ GSSG ratio, MDA	DCFH-DA for ROS, HPLC/fluorescence detector for GSH and GSSG, assay kit for MDA	HepG2 cells	No change in ROS, MDA, GSH/GSSG	PFOA or PFOS at 100 μM for 3 h	No change in ROS, MDA, or GSH/GSSG; only a single level of exposure; no detailed data were provided	<u>Shan et al.</u> (2013)					
ROS, MDA, GSH, SOD, CAT	ROS by DCFH-DA/flow cytometry; MDA, GSH, SOD, CAT by ELISA	Pancreatic epithelioid carcinoma (PANC-1) cell line	No changes in ROS, ↑ GSH, ↑ CAT, ↑ SOD, ↑ MDA	PFOA at 10, 50, or 100 μM for 24 h	No increase in ROS levels; GSH levels increased only at 10 µM; MDA and SOD increased to similar levels regardless of the dose; the involvement of oxidative stress was unclear; CAT increased at all doses in a non-dose-dependent manner	<u>Abudayyak</u> <u>et al. (2021a</u>					
ROS	ROS-Glo H ₂ O ₂ assay; DCFH-DA assay	Differentiated neuroblastoma (SH- SY5Y) cells	No changes in ROS	PFOA, 1, 10, 100, 150, 200, or 250 μM for 4, 24, 48, or 72 h	DCFDA detected ROS increases at 100, 200, and 250 μ M after 4 h and at 250 μ M after 24 h, but no change after 48 h at any concentration	<u>Souders</u> <u>et al. (2021)</u>					
ROS, MDA, GSH, SOD	DCFH-DA followed by fluorescence photography, TBARS, GSH assay kit, SOD assay kit	Lung carcinoma (A459) cells	↑ ROS, ↑ MDA, ↑ SOD, ↓ GSH	PFOS, 25, 50, 100, or 200 μM for 24 h	PFOS at 50, 100, or 200 μM significantly increased ROS, MDA, and SOD, and reduced GSH; ROS increase was measured using fluorescence photography, but not quantified	<u>Mao et al.</u> (2013)					
mtROS	Immunofluorescence microscopy	Ovarian granulosa-like tumour (KGN) cells	↑ mtROS	PFOA, 250, 500, or 750 μM for 24 h	Significant increase in mtROS at all doses	<u>Zhang et al.</u> (2023a)					

Table 1 16 (a **പ**)

Table 4.16 (continued)

End-points	Assay	Tissue, cell line	Result ^a	Concentrations or dosing regimen	Comments	Reference
ROS	DCFH-DA	Embryonic kidney (HEK293) cells	↑ ROS	PFOS, 10–60 μM for 24 h	Significant increase in ROS production; CBD (2–80 µM) partially restored PFOS-induced ROS; however, the concentrations of PFOS and CBD that had their effects were unclear	<u>Du et al.</u> (2023)

ARE, antioxidant responsive element; CAT, catalase; CBD, cannabidiol; CYP, cytochrome P450; DCFH-DA, 2',7'-dichlorodihydrofluorescein diacetate; DCF, dichlorofluorescein; dG, 2'-deoxyguanosine; DHE, dihydroethidium; DTNB, 5,5'-dithiobis(2-nitrobenzoic acid); ELISA, enzyme-linked immunosorbent assay; Fpg, formamidopyrimidine DNA glycosylase; GPx, glutathione peroxidase; GSH, glutathione; GSSG, oxidized glutathione; GST, glutathione S-transferase; h, hour(s); H₂O₂, hydrogen peroxide; HK-2, human kidney proximal tubular epithelial cells; HMVEC, human microvascular endothelial cells; HO-1, haem oxygenase 1; HPLC, high-performance liquid chromatography; HUVEC, human umbilical vein endothelial cells; LC₅₀, median lethal concentration; LC-MS/MS, liquid chromatography-tandem mass spectrometry; LOD, limit of detection; MDA, malondialdehyde; min, minute(s); NAC, *N*-acetylcysteine; 2,3-NDA, 2,3-naphthalenedicarboxaldehyde; NOx, nitrogen oxides; NRf2, NF-E2-related factor 2; 8-OHdG, 8-hydroxydeoxyguanosine; 8-oxodG, 8-oxo-2'-deoxyguanosine; PANC-1, human pancreatic cells; TAC, total antioxidant content; TBA, thiobarbituric acid; TBARS, thiobarbituric acid-reactive substances; yr, year(s).

^a ↑, increase; ↓, decrease.

GPx levels were reduced by 100 μ M. No change was observed for SOD (<u>Pan et al., 2018</u>). [The Working Group noted that the authors concluded that PFOA induces oxidative stress in human erythrocytes.]

Steves et al. (2018) used a human stem cellbased model of spermatogenesis to assess the effects of PFOA and PFOS individually or in mixtures. Cells were treated with PFOS (24, 48, or 126 μ M) or PFOA (11, 25, or 100 μ M), starting on day 1, for the entire 10-day cell differentiation process. ROS production was measured using the Muse Oxidative Stress Kit, followed by flow cytometry. No significant changes in ROS production in the cells exposed to PFOA at 100 μ M were observed, whereas PFOS at 126 μ M significantly reduced ROS compared with the control group (0.25% dimethyl sulfoxide only). Both PFOA and PFOS significantly reduced ROS production at the two lowest concentrations. The authors concluded that ROS production induced by PFOA or PFOS is unlikely to influence the viability of spermatogenic cells in vitro.

In another study, <u>Yuan et al. (2020)</u> treated human sperm with PFOA at 0, 0.25, 2.5, or $25 \mu g/mL$ (0, 0.6, 6, or 60 μ M) for 30 minutes or 4 hours. ROS was measured using a ROS assay kit. The highest exposure concentration (60 μ M) after 4 hours induced a significant increase in ROS generation. Sperm have high levels of polyunsaturated fatty acids and low antioxidant enzyme levels. Hence, they are especially vulnerable to oxidative stress, and high levels of ROS may disrupt sperm function.

Human lymphocytes for a PFOS exposure study were isolated from blood samples of 20 healthy adults aged 18–30 years (sex not reported) (Zarei et al., 2018). The cells were treated with PFOS at 75, 150, or 300 μ M for 2, 4, 6, 8, 10, or 12 hours. ROS production was measured using 2',7'-dichlorodihydrofluorescein diacetate (DCFH-DA), MDA using TBARS, and GSH and oxidized glutathione (GSSG) using the *o*-phthalaldehyde fluorescence method. PFOS significantly (P < 0.05) induced ROS production in a time– and dose–responsive manner: the levels of ROS (measured by DCFH-DA) increased at concentrations of 150 and 300 µM PFOS at 2 hours and at all concentrations and later time points (4–12 hours), in comparison with the control. MDA levels were also increased after 6 hours of treatment with 150 or 300 µM PFOS. A reduction in GSH and increase in GSSG were observed at 4, 8, 10, and 12, but not at 6 hours, at all concentrations. Butylated hydroxytoluene (50 µM), an antioxidant, was found to inhibit PFOS-induced oxidative stress.

In another study (Qian et al., 2010), human microvascular endothelial cells (HMVEC) were exposed to PFOS at 10, 20, 50, or 100 μ M for 1 hour, or 2 μ M for 1, 2, 3, or 5 hours. Dihydroethidium-fluorescence-confocal microscopy and dihydroethidium-fluorescence microplate reader methods were used to measure ROS. PFOS increased ROS production at all concentrations (10–100 μ M for 1 hour) and time points (2 μ M for 1–5 hours). The low concentrations of PFOS applied to HMVEC cells matched occupational and environmental levels.

Wang et al. (2022a) treated human L02 liver cells with 0, 8, 64, or 256 μ M PFOA for 24 hours. PFOA significantly increased ROS production at 64 and 256 μ M. The authors concluded that PFOA may induce endoplasmic reticulum stress (ERS) and oxidative stress. [The Working Group noted that the L02 cells used in the study might be contaminated with HeLa cells (SIB, 2024). The HeLa cell line is derived from cervical cancer cells. Therefore, contamination of L02 cells with HeLa cells would complicate the interpretation of the results of this study.]

In another study (Wang et al., 2022b), human proximal tubular epithelial cells (HK-2) were exposed to PFOS at 200 μ M for 12 hours. PFOS significantly increased the level of ROS production and reduced expression of the antioxidants GPx4 and HO-1. In this study, the authors proposed that PFOS may cause ferroptosis because of an intracellular increase in iron, which would lead to an increase in ROS production and a reduction in GSH concentration.

Tien et al. (2020) identified consistently high concentrations of PFOA in the particulate matter (PM_{25-1}) fraction of indoor dust to which people can be exposed and assessed the effects of PFOA on human corneal epithelial (HCEpiC), endothelial (HCEC), and retinal pigment epithelial (RPE) cells. HCEpiC cells were treated with PFOA at concentrations of 100, 200, or 400 ppm for 6 hours. PFOA at 200 and 400 ppm increased ROS production in HCEpiC cells, and a significant increase was observed for 400 ppm exposure. The authors concluded that PFOA in dust might induce ROS production in the retina and that this may have a risk implication for age-related macular degeneration (Tien et al., 2020). [The Working Group noted that the response could have been because of the dust particles as well as the PFOA in the dust.]

Elumalai et al. (2023) reported that PFOSinduced oxidative stress contributed to apoptosis in human pancreatic 1.1b4 β cells. Human 1.1b4 cells were treated with 100 μ M PFOS for 48 or 36 hours. PFOS was found to increase mitochondrial ROS generation with significantly decreased GSH/GSSG ratios through NOX2-gp91Phox activation and the inhibition of cyclic adenosine monophosphate-protein kinase A (cAMP-PKA).

However, in another study, <u>Dos Santos</u> <u>et al. (2022)</u> treated the human insulin-producing EndoC- β -cell line (a human pancreatic β -cell model) with relatively low concentrations of PFOA (1 nM or 1 μ M). No significant changes in ROS production, as measured as 2',7'-dichlorofluorescein diacetate (DCF-DA) fluorescence, were observed.

Du et al. (2022) treated human placental trophoblast cells immortalized with SV40T antigen (HTR-8/SVneo cells) with PFOA at 0, 100, 200, 400, 600, 800, or 1000 μ M for 24 hours. PFOA was found to increase ROS production in a dose-dependent manner. The study also showed

that PFOA induced ERS, which triggered both the proliferation and apoptosis of trophoblasts via ROS generation or activation of the unfolded protein response (UPR) signalling pathway.

Sonkar et al. (2019) also studied PFOSinduced ROS production in HTR-8/SVneo cells. The cells were exposed to PFOS (1, 10, or 50 μ M) for 24 or 48 hours. ROS production in the cells was significantly induced at concentrations \geq 10 μ M at both 24 and 48 hours.

Liao et al. (2012) exposed human umbilical vein endothelial cells (HUVECs) to PFOS (100 mg/L; 200 μ M), and ROS production was measured after 1, 5, 12, 24, and 40 hours using the DCFH-DA fluorescence assay. A significant increase in ROS production was observed in a time-dependent manner.

In a follow-up study, Liao et al. (2013) reported that PFOS (at 100 mg/L, i.e. 200 μ M) significantly increased ROS production, as measured by DCFH-DA, in HUVECs. The study showed that PFOS treatment alone significantly increased ROS production. Co-treatment with PFOS and either *Flos lonicerae* extract containing chlorogenic acid (CGA) or CGA for 40 hours reduced ROS production to levels comparable to those in cells exposed to PFOS only for 24 hours. However, the increased ROS levels of cells exposed to PFOS for just 24 hours were not affected by co-treatment with either *Flos lonicerae* or CGA.

<u>Gao et al. (2020)</u> studied the effects of PFOS on human mesenchymal stem cells (hMSCs) by treating the cells with 0.1 or 10 μ M PFOS for 12, 24, 36, or 48 hours. Treatment with 10 μ M PFOS was found to significantly increase ROS production at all time points. The lower concentration of PFOS only increased ROS production significantly in cells treated for 7 days. The results suggest an accumulation effect of low-dose PFOS in hMSC cells.

[The Working Group noted that primary hepatic cells were the most-used human cells for the study of PFOA and PFOS induction of oxidative stress in vitro, with studies also using primary cells derived from other organ systems. PFOA and PFOS exposure concentrations typically ranged from low doses (submicromolar) to medium doses (up to $100-200 \mu$ M) and high doses (above 500 μ M). Except for a few studies reporting inverse associations between the exposure and response, both PFOA and PFOS were shown to have varying potencies to induce ROS production in primary liver cells and cells derived from other organ systems.]

<u>Human cell lines</u>

Among immortalized cell lines, HepG2 cells are the most widely used. This cell line was used for exposure experiments involving exposure to a wide range of concentrations of PFOA or PFOS for different periods of time.

Two studies examining the effect of PFOA and/or PFOS on the induction of oxidative DNA damage found no effects. The first study used the Fpg-modified comet assay, which included a Ro19-8022 + light as a positive control for the generation of 8-oxodG, to examine the effect of PFOA and PFOS treatment on HepG2 DNA. PFOA and PFOS treatment did not generate oxidative damage to DNA (Eriksen et al., 2010). The authors observed an increase in ROS production. Specifically, HepG2 cells were treated with PFOA or PFOS at concentrations ranging from 0.4 µM to 2 mM (0.4, 4, 40, 200, 400, 1000, or 2000 µM), and ROS production was measured every 15 minutes up to 3 hours. PFOA and PFOS induced a moderate, non-significant increase in ROS production in the cells; however, the response was not concentration-dependent (Eriksen et al., 2010). The second study (Han et al., 2020) exposed a human skin equivalent to PFOA for 6 days. A slight but non-significant increase in 8-oxodG, measured by ELISA, was observed at 0.25 mM.

In three other studies, PFOA exposure was shown to increase the levels of 8-oxodG in human hepatoma cells (HepG2), in lymphoblastoid cells (TK6), and in epidermal keratinocytes (HaCaT) [although the Working Group considered the reliability of the measurements to be questionable] (Yao and Zhong, 2005; Yahia et al., 2016; Peropadre et al., 2018). The first study reported relatively high background levels of 8-oxodG, measured by HPLC-MS/MS (i.e. 8.3 lesions/10⁶ dG) (Yahia et al., 2016), which is almost one order of magnitude higher than the generally accepted background level of 8-oxodG in mammalian cells (1 lesion/10⁶ dG) and exceeds the level at which methodological artefacts become a serious concern (5 lesions/10⁶ dG) (ESCODD, 2002). [The Working Group considered that the results from three studies with antibody-based detection of 8-oxodG in cells were inconclusive because of a lack of positive controls to verify the specificity of the assay (Yao and Zhong, 2005; Peropadre et al., 2018). The detection of 8-oxodG by an antibody-based system was considered questionable, because of non-specific binding to other biomolecules in cells or biological matrices (<u>Chao et al., 2021</u>).]

Studies demonstrating the induction of ROS production are summarized below.

PFOA (200 or 400 μ M) was found to significantly increase ROS production in HepG2 cells after a 3-hour exposure (<u>Panaretakis et al., 2001</u>). ROS were measured using flow cytometry as hydrogen peroxide (H₂O₂) and the superoxide anion O₂⁻ after reaction with DCFH-DA and dihydroethidine, respectively.

HepG2 cells were exposed to PFOA at concentrations of 10, 25, or 50 μ M for 24 hours (Abudayyak et al., 2021b). In this study, ROS production was increased by all concentrations (\leq 5.3-fold at 50 μ M), GSH was increased (1.7-fold at 10 μ M), and CAT was increased (1.4-fold at 10 μ M), but changes in SOD activity were not observed at any concentration. Fluorescein isothiocyanate-labelled DCFH-DA was used to measure ROS, whereas ELISA kits were used for the measurement of other end-points.

Hu and Hu (2009) measured the following end-points of oxidative stress – ROS, GSH, GPx, SOD, and CAT – in HepG2 cells exposed to PFOA or PFOS (50, 100, 150, or 200 μ M) for 5, 10, or 15 hours. PFOS significantly increased ROS production in a time- and dose-dependent manner (at 100, 150, and 200 μ M). PFOA induced ROS production in a similar manner, although no data were shown. At 48 hours, significant changes in antioxidant activity were observed: both PFOA and PFOS significantly reduced GSH and GPx at 100, 150, and 200 μ M and increased SOD and CAT activities at the two highest concentrations, 150 and 200 μ M. PFOA and PFOS induced an increase in GST only at 200 μ M.

Wielsøe et al. (2015) studied the effects of exposure to seven different PFAS, including PFOA and PFOS, on oxidative stress end-points, such as ROS production and the total antioxidant capacity (TAC) in HepG2 cells. The cells were exposed to concentrations of 0.02, 0.2, 2, 20, or 200 μ M for 24 hours. ROS production was significantly induced by 0.2, 2, and 20 μ M of PFOA and PFOS. TAC was reduced at all the tested concentrations of PFOA and PFOS; however, the reduction only reached significance with PFOA exposure.

Amstutz et al. (2022) exposed HepG2 cells to 5, 25, 50, 100, 200, 400, or 800 μ M of PFOA, PFOS, or several other PFAS for 3 or 24 hours. PFOA and PFOS induced concentration-dependent increases in ROS generation. <u>Yarahalli Jayaram</u> et al. (2020) exposed HepG2 cells to PFOA at 100 or 250 μ M for 24 hours. PFOA exposure at 250 μ M significantly increased ROS production and NO levels, measured using dichlorofluorescein (DCF) and Greiss reagent, respectively. The mRNA expression of *NOS2A* also increased upon exposure to PFOA, providing further evidence for oxidative effects of PFOA.

Qi et al. (2023) studied the pro-steatotic and fibrotic effects of PFOA in HepG2 and HepaRG cells. Cells were exposed to PFOA at 10, 100, or 1000 nM for 72 hours, which led to increased ROS production. Additionally, PFOA was found to increase steatosis and fibrosis in both cell lines, as indicated by the upregulation of specific genes involved in UPR signalling and non-alcoholic fatty liver disease (NAFLD).

HepG2 cells and HaCaT cells were treated with 10 μ M PFOA for 24 hours. Afterwards, ROS levels were measured using two different fluorescent probes, MAK145 (red fluorescence) and DCFH-DA, with the DCFH-DA probe being more sensitive than the MAK145 probe (Magnifico et al., 2022). In addition, levels of nitrogen oxides (NO_x) were measured using the fluorescent probe 2,3-diaminonaphthalene from a nitrate/nitrite fluorometric assay kit. Fluorescence intensity, proportional to the total NO production, was measured using a fluorometer. The results showed that PFOA significantly increased ROS production and nitrosative stress in both cell lines.

One study by <u>Ojo et al. (2022b)</u> investigated the effects of several PFAS, including PFOA and PFOS, on the Nrf2-antioxidant responsive element (ARE) pathway, individually or in combination, using ARE reporter-HepG2 cells. The induction of Nrf2-ARE is indicative of an event of cellular oxidative stress. The cells were exposed to PFOA or PFOS at 1, 2, 3, 4, or 5 μ M for 24 hours. The results showed a significant induction of oxidative stress. The 1.5-fold induction ratios (EC_{IR} = 1.5) for PFOA and PFOS were determined to be 1.38 μ M and 1.17 μ M, respectively.

Wan et al. (2016) treated HepG2 cells with PFOS at concentrations of 0, 10, 20, 30, 40, or 50 μ M for 24 hours or 50 μ M for 1, 3, 6, 12, or 24 hours. ROS levels increased significantly at PFOS concentrations \geq 30 μ M. When treated with 50 μ M, ROS levels gradually increased over time, peaking at 12 hours, and then decreased to 24 hours. GSH levels significantly decreased at PFOS concentrations \geq 20 μ M and at all time points after 6 hours during the time-course experiment. Yan et al. (2015a) exposed HepG2 cells to PFOA at concentrations of 0, 50, 100, or 200 μ M for 24 or 72 hours. ROS production was significantly increased in cells treated with PFOA concentrations $\geq 100 \ \mu$ M at 24 hours but was decreased at 72 hours. ROS production was decreased in cells pre-treated with *N*-acetylcysteine (NAC).

Several studies have reported no effects of PFOA or PFOS exposure on the production of ROS in HepG2 cells. For example, Florentin et al. (2011) assessed the induction of ROS production after exposure to PFOA or PFOS in HepG2 cells. The cells were exposed to PFOA at 0, 5, 10, 50, 100, 200, or 400 μ M, or 0, 5, 10, 50, 100, 200, or 300 μ M PFOS for 1 or 24 hours. No increase in oxidative stress was observed. No effect on ROS production was also reported by Ojo et al. (2021) after they exposed HepG2 cells to 0.2, 2, or 20 μ M PFOA or PFOS for 24 hours. It was noted that the doses used in this study were lower and more relevant to environmental exposure levels.

In another study, <u>Shan et al. (2013)</u> exposed HepG2 cells to a single concentration (100 μ M) of PFOA or PFOS for 3 hours. No significant differences in ROS production, GSH, GSSG, or MDA were observed between the PFAS test groups and the negative control. [The Working Group noticed that the study focused on the combined effects of PFOA or PFOS with pentachlorophenol. No tabulated or graphical data were provided for PFOA and PFOS alone, only descriptive text. Nevertheless, this was another study available in the literature that used relatively low concentrations of PFOA and PFOS and with no effects on the cell oxidative status.]

The concentrations of PFAS and the treatment conditions of the cells before the addition of the ROS-detection probe were considered of relevance (Ojo et al., 2021). It was noted that the cell trypsinization stage included in the ROS measurement method could have induced additional stress in the cells, as noted by <u>Florentin</u> et al. (2011). Exposure concentrations may also affect ROS production. It was reported that low concentrations may protect against ROS generation, and high concentrations may induce ROS production (<u>Steves et al., 2018</u>).

<u>Yao and Zhong (2005)</u> reported that 8-OHdG content, measured using immunocytochemical staining, was significantly increased in a dose-dependent manner when HepG2 cells were exposed to PFOA (100, 200, or 400 μ M) for 3 hours. Hydrogen peroxide was used as a positive control. ROS, measured using DCFH-DA, was also increased in a dose-dependent manner. No positive control was included in the ROS assay. As mentioned above, <u>Eriksen et al. (2010)</u> also did not observe a significant increase in ROS.

Fewer studies have examined the effects of PFOA and PFOS on the induction of oxidative stress in human cells other than HepG2. <u>Abudayyak et al. (2021a)</u> exposed the human pancreatic epithelioid carcinoma cell line PANC-1 to PFOA (0, 10, 50, or 100 μ M) for 24 hours. ROS levels were not increased by any tested concentration. However, MDA, SOD, and CAT were significantly increased upon PFOA treatment, although not in a dose-dependent manner. It was noted that GSH was increased in cells treated with 10 μ M PFOA, but not higher concentrations.

Souders et al. (2021) exposed differentiated human SH-SY5Y neuroblastoma cells to PFOA for a metabolic profiling study. Cells were treated with 1, 10, 100, 150, 200, or 250 μ M PFOA, and ROS production was measured at 4, 24, and 48 hours using the ROS-Glo H₂O₂ and DCFH-DA assays. The ROS-Glo assay did not detect changes in ROS production, except a reduction after 4 hours at 250 μ M, whereas the DCFH-DA assay detected a significant increase in ROS levels at 4 hours for 100, 200, and 250 μ M and at 24 hours for 250 μ M, but no increase at 48 hours.

In another study (Mao et al., 2013), human lung carcinoma A459 cells were treated with PFOS (25, 50, 100, or 200 μ M) for 24 hours. ROS were measured using DCFH-DA, MDA

using TBARS, and the GSH and SOD levels were measured using assay kits. PFOS significantly increased ROS, MDA, and SOD, and reduced GSH at all concentrations $\geq 50 \ \mu$ M. ROS was measured qualitatively using fluorescence photography, and an increase in fluorescence levels was observed with increasing PFOS concentrations. PFOS-induced ROS generation was also inhibited by pre-treatment with the thiol antioxidant NAC.

More recently, Zhang et al. (2023a) exposed human ovarian granulosa-like tumour cells (KGN) to PFOA at concentrations ranging from 250 to 750 μ M for 24 hours. Mitochondrial ROS (mt-ROS), determined by the superoxide indicator fluorescence ratio of red MitoSOX to MitoTracker Green was significantly increased in a dose-dependent manner across all tested concentrations.

Du et al. (2023) observed PFOS-induced ROS production (DCFH-DA method) in human embryonic kidney cells (HEK293) exposed for 24 hours to a medium containing PFOS at 40 μ M and/or cannabidiol (CBD) at 20 μ M. Co-exposure with CBD reduced the level of PFOS-induced ROS production.

(b) Experimental systems

(i) Non-human mammalian systems in vivo See <u>Table 4.17</u>.

Some studies have shown mixed results regarding 8-oxodG formation in the kidneys, liver, and testes of mice and rats after oral exposure to PFOA (Takagi et al., 1991; Abdellatif et al., 2003–2004; Zou et al., 2015; Ma et al., 2023). [The Working Group noted that some of the studies did not include positive controls. In addition, there was uncertainty about the reliability of the results, because of high background levels of 8-oxodG in rat tissues (Takagi et al., 1991; Abdellatif et al., 2003–2004) and the

measurement of 8-oxodG by a non-specific antibody-based method in the mouse tissues (Zou et al., 2015; Ma et al., 2023).]

Abdellatif et al. (2003-2004) studied the effects of dietary PFOA exposure on peroxisomal enzymes and 8-OHdG in male Wistar rats (minimum 15 rats per group), following a biphasic initiation procedure involving dosing with diethylnitrosamine (DEN) or a triphasic initiation involving dosing with DEN, 2-acetylaminofluorene, and carbon tetrachloride (CCl₄) (Abdellatif et al., 2003-2004). For the rats that underwent biphasic initiation, basal rodent diet or diet containing 0.005% or 0.02% PFOA (daily dose in mg/kg per day was not estimated) was ingested for 14 or 25 weeks. Those that underwent triphasic initiation ingested a diet without PFOA or a diet containing 0.015% PFOA for 25 weeks. CAT, measured spectroscopically, was significantly increased in the rat livers after 14 or 25 weeks compared with the basal diet group. To study the effect of PFOA on 8-OHdG induction, male Wistar rats underwent the initiation treatment and were fed either a basal diet or a basal diet containing 0.02% PFOA for 5 or 9 weeks. No significant increase in 8-OHdG was observed in the liver after 5 or 9 weeks compared with the basal diet group. [The Working Group noted that the study authors suggested that PFOA was a liver cancer promoter that may not require significant DNA damage. It was also noted that the study did not include a positive control group and that the background 8-OHdG was relatively high in the livers from the basal diet group.]

Ma et al. (2023) investigated the potential of rutin to ameliorate the oxidative effects of PFOA in mice (10 mice per group). Male ICR mice were orally dosed with PFOA (20 mg/kg bw) daily for 28 days in the presence or absence of rutin. The levels of 8-OHdG, GPx, SOD, and MDA were measured. 8-OHdG was significantly increased in the serum and testes; and MDA, GPx, and SOD were significantly increased in the testes of

End-point, assay	Species, strain, sex, cell line	Tissue	Resultª	Dose	Route, duration, dosing regimen	Comments	Reference
8-OHdG, CAT, by HPLC-ECD, spectroscopic method	Rat, Wistar, male	Liver	No change in 8-OHdG, significant increase in CAT	$\begin{array}{c} \mbox{PFOA 0.02\% in} \\ \mbox{feed for 8-OHdG,} \\ \mbox{0.005\% or 0.02\%} \\ \mbox{with DEN biphasic} \\ \mbox{initiation, or 0.015\%} \\ \mbox{PFOA after DEN,} \\ \mbox{2-acetylaminofluorene} \\ \mbox{and CCl}_4 \mbox{triphasic} \\ \mbox{initiation} \end{array}$	Oral, 5 or 9 wk for 8-OHdG, 14 or 25 wk for CAT	No positive control; tissues from unexposed rats have high background levels of 8-oxodG (i.e. 0.126 ng/µg DNA, corresponding to 730 lesions/10 ⁶ dG) ^a	<u>Abdellatif</u> <u>et al. (2003–</u> <u>2004)</u>
8-OHdG, MDA, GPx, SOD 8-OHdG by ELISA kit; MDA, SOD, and GPx by microplate reader	Mouse, ICR, male	Serum and testicular tissue	Serum and testis: ↑ 8-OHdG Testis: ↑ MDA, ↑ GPx, ↑ SOD	PFOA (20 mg/kg per day)	Daily oral gavage for 28 days	Rutin ameliorates PFOA-induced oxidative stress; no positive control group; results on 8-oxodG reported as pg/mg protein	<u>Ma et al.</u> (2023)
8-OHdG, by HPLC-ECD	Rat, Fischer 344, male	Kidney, liver	↑ 8-OHdG (liver) ≥ 3 days No changes (kidney)	PFOA, 100 mg/kg	Peritoneum single injection, dosing at 1, 3, 5, or 8 days post-injection	No positive control group; tissues from unexposed rats had high background levels of 8-oxodG (> 30 lesions/10 ⁶ dG)	<u>Takagi et al.</u> (1991)
8-OHdG, by HPLC-ECD	Rat, Fischer-344, male	Kidney, liver	↑ 8-OHdG (liver) No changes (kidney)	PFOA, 0.02% in feed	Oral, 2 wk, feed	No positive control group; tissues from unexposed rats had high background levels of 8-oxodG (i.e. 17 lesions/10 ⁶ dG)	<u>Takagi et al.</u> (1991)

Table 4.17 End-points relevant to oxidative stress in in non-human mammalian systems in vivo exposed to PFOA or PFOS

	able 4.17 (continued)									
End-point, assay	Species, strain, sex, cell line	Tissue	Resultª	Dose	Route, duration, dosing regimen	Comments	Reference			
8-OHdG, MDA, H ₂ O ₂ , SOD, CAT by ELISA for 8-OHdG, commercial kits for other biomarkers	Mouse, Kunming, male	Liver	↑ 8-OHdG, ↑ H ₂ O ₂ , ↑ MDA, ↓ SOD, ↓ CAT	PFOA, 10 mg/kg per day	Oral, daily for 14 days	No positive control group; results for 8-oxodG were reported as pg/mg protein; commercial kits may not be specific for the particular end-points Quercetin, a flavonoid, given at 75 mg/kg per day for 14 days limited the oxidative effects of PFOA	<u>Zou et al.</u> (2015)			
8-OHdG, by HPLC (UV and electrochemical detectors)	Mouse, wildtype mice (129S4/ SvlmJ) and Ppara-null mice (129S4/SvJae- Pparatm1Gonz/J), male	Liver	↑ 8-OHdG (only in Pparα-null mice)	PFOA, 12.5, 25, or 50 μmol/kg per day	Oral, daily for 4 wk	No change in 8-OHdG in the wildtype mice; the ablation of Ppara exacerbated PFOA- induced oxidative stress	<u>Minata</u> et al. (2010)			
8-OHdG, MDA, 8-OHdG by ELISA, MDA by HPLC	Rat, Sprague- Dawley, male	Urine	↑ 8-OHdG, ↑ MDA	PFOA, 10, 33, or 100 mg/kg per day	Oral gavage on days 4, 5, and 6	Urine 8-OHdG and MDA tested from day 1 to day 10; both biomarkers increased from day 3 or 4, peaked on day 5–7, then returned to baseline by day 8 or 10	<u>Rigden</u> et al. (2015)			
ROS, MDA, GSSG, CAT, SOD, and GSH, by DCFH-DA for ROS, assay kits for other biomarkers	Rat, Sprague- Dawley, male	Liver	↑ 8ROS, ↑ MDA, ↑ GSSG, ↓ SOD, ↓ CAT, ↓ GSH, ↓ GSH/GSSH	PFOS, 1 or 10 mg/kg per day	Oral, daily for 28 days	Significant dose- dependent changes in biomarkers; the observed induction of oxidative stress was also supported by increases in serum NO and liver	<u>Han et al.</u> (2018a)			

Nos2 mRNA expression

IARC MONOGRAPHS - 135

Table 4.17 (continued)

Table 4.17 (continued)										
End-point, assay	Species, strain, sex, cell line	Tissue	Result ^a	Dose	Route, duration, dosing regimen	Comments	Reference			
MDA, GSH, SOD, and CAT, by TBARS assay, GSH ELISA, SOD kit, CAT kit	Mouse, Balb/c, male	Liver	↓ MDA, ↑ GSH, ↓ SOD, ↑ CAT	PFOA, 0, 0.08, 0.31, 1.25, 5, or 20 mg/kg per day	Gavage, daily for 28 days	Multiple groups and daily dosing; PFOA disturbed the antioxidant defence system in the liver, but did not significantly induce oxidative stress	<u>Yan et al.</u> (2015a)			
8-iso-pg-PGF2a; <i>Sod1</i> , <i>Sod2</i> , <i>Gpx2</i> , <i>Cat</i> , <i>Nqo1</i> expression, by LC/MS for 8-iso- pg-PGF2a, qRT- PCR for the other biomarkers	Mouse, C57BL/6, male	Liver, pancreas	Pancreas: \uparrow all biomarkers, except for <i>Cat</i> Liver: \uparrow 8-iso-pg-PGF2 α \uparrow <i>Sod1</i> , \uparrow <i>Sod2</i> , \uparrow <i>Cat</i> , and \uparrow <i>Nqo1</i> dosed with 2.5 or 5 mg/kg; no change in <i>Gpx2</i> expression	PFOA, 0, 0.5, 2.5, or 5 mg/kg per day	Gavage, daily for 7 days	LC-MS is a robust method for 8-iso-pg- PGF2a measurement; qRT-PCR is considered to be reliable	<u>Kamendulis</u> <u>et al. (2014)</u>			
MDA, SOD, CAT, TrxR, by commercial kits	Mouse (conditional <i>Kras</i> ^{G12D} mouse model [<i>LSL-Kras</i> ^{G12D} and Pdx-1 Cre mice])	Pancreas	↑ MDA, ↑ SOD, ↑ CAT, ↑ TrxR	PFOA, 5 ppm in drinking-water	Drinking- water, 4 mo or 7 mo exposure; mice were of age 6 or 9 mo at the end of the dosing period	MDA protein level increased only at age 6 mo; SOD enzyme activity increased at both 6 and 9 mo, whereas CAT and TrxR enzyme activities were increased only at age 9 mo	<u>Kamendulis</u> et al. (2022)			

Table 4.17 (continued)End-point, assaySpecies, strain, Tis

End-point, assay	Species, strain, sex, cell line	Tissue	Resultª	Dose	Route, duration, dosing regimen	Comments	Reference
GSH, CAT, GPx, GR, GST, by spectrometry	Rat, Wistar, male and female	Liver	↑ CAT, ↑ GPx, No change in GSH, ↑ GR, ↓ GST (male), GST – no effect (female), ↑ peroxisomal β-oxidation, ↑ microsomal NADP-dependent lipid peroxidation (male, age 26 weeks)	PFOA, 0.01% of diet	Oral, 2 or 26 wk	Increases in peroxisomal β -oxidation in male and female rats after 2 or 26 wk of PFOA exposure, plus an increase in microsomal NADP-dependent lipid peroxidation, and changes in other biomarkers supported the induction of oxidative stress in male and female rats by PFOA	<u>Kawashima</u> <u>et al. (1994)</u>
Oxidized lipid products (15- F_{2t} -IsoP, 5- F_{2t} -IsoP, 8- F_{3t} -IsoP, 4(RS)-4- F_{4t} -NeuroP, 17(RS)- F_{2t} -dihomo- Iso-P, Iso-F, NeuroF, 17(RS)-SC- Δ^{15} - 11-dihomo-IsoF, 7 β -hydroxycholesterol, 27-hydroxycholesterol, 27-hydroxycholesterol, 9(S)-HETE, 11(S)- HETE, 9(S)-HETE, 20-HETE, 5(S)-HETE, 8(S)-HETE, 12(S)- HETE, 15(S)-HETE, SOD, CAT, by LC-MS/MS, and assay kits	Mouse, CD-1, pregnant and adult females and fetuses	Liver, brain, kidney	No changes in SOD or CAT, significant increases in various oxidized lipid products in fetal liver, brain, and kidney	PFOS, 3 mg/kg per day	Oral, pregnant mice from GD1 to GD17, adult female mice daily for 14 days	Comprehensive evaluation of oxidized lipid products derived from the lipid peroxidation of polyunsaturated fatty acid; both non-enzymatic and enzymatic oxidation (CYP and LOX) products were measured. Changes in oxidative stress biomarkers were more significant in the fetuses	<u>Lee et al.</u> (2015)

Table 4.17 (continued)

End-point, assay	Species, strain, sex, cell line	Tissue	Resultª	Dose	Route, duration, dosing regimen	Comments	Reference
MDA, H ₂ O ₂ , SOD, CAT, by TBARS, assay kits	Mouse, Kunming, male	Testis	↑ MDA, ↓ SOD, ↓ CAT	PFOA, 0, 2.5, 5, or 10 mg/kg per day	Oral, daily for 14 days	Oxidative stress was further supported by a reduction in Nrf2 expression	<u>Liu et al.</u> (2015a)
MDA, H2O2, SOD, CAT, by assay kits	Mouse, Kunming, male	Liver	↑ MDA, ↑ H_2O_2 , ↓ SOD, ↓ CAT	PFOA, 10 mg/kg per day	Oral, daily for 14 days	Oxidative stress was further supported by a reduction in Nrf2 expression	<u>Liu et al.</u> (2015b)
MDA, RONs, NO, GPx, GST, SOD, CAT, GSH, By TBARS, DCFH- DA, Griess reaction, 1-chloro-2,4- dinitrobenzene, autoxidation of epinephrine, H ₂ O ₂ , <i>o</i> -phthaldialdehyde	Rat, Wistar, male	Liver, kidney	↑ RONs, ↑ NO, ↓ GPx, ↓ GST, ↓ SOD, ↓ CAT, ↓ GSH, ↑ MDA	PFOA, 5 mg/kg per day	Oral, daily for 28 days	Evidence of oxidative stress is further supported by increases in xanthine oxidase and myeloperoxidase. NAC alleviated the level of oxidative stress	<u>Owumi</u> <u>et al.</u> (2021a)
MDA, RONs, NO, GPx, GST, SOD, CAT, GSH, by TBARS, DCFH- DA, Griess reaction, 1-chloro-2,4- dinitrobenzene, autoxidation of epinephrine, H ₂ O ₂ , <i>o</i> -phthaldialdehyde	Rat, Wistar, male	Testis, epididymis	 ↑ MDA, ↑ RONs (testes only), ↑ NO, ↓ GPx (epididymis only), ↓ GST (testes only), ↓ SOD, ↓ CAT, ↓ GSH 	PFOA, 5 mg/kg per day	Oral, daily for 28 days	Evidence of oxidative stress is further supported by increases in xanthine oxidase and myeloperoxidase; NAC alleviated the level of oxidative stress	<u>Owumi</u> <u>et al.</u> (2021b)
MDA, H ₂ O ₂ , by assay kits	Mouse, Kunming, male	Liver	$\uparrow MDA, \\ \uparrow H_2O_2$	PFOA, 0, 2.5, 5, or 10 mg/kg per day	Oral, daily for 14 days	Significant dose- dependent increases in biomarkers	<u>Yang et al.</u> (2014)

•							
End-point, assay	Species, strain, sex, cell line	Tissue	Resultª	Dose	Route, duration, dosing regimen	Comments	Reference
MDA, GSH, GSSG, by assay kits	Mouse, C57BL/6, male	Liver	↑ MDA, ↓ GSH/GSSG	PFOS, 0, 0.003%, 0.006%, or 0.012% of diet	Oral, daily for 14 days 0.003% for 6 wk	Increase in MDA and decrease in GSH/GSSG ratio mice consuming either a normal diet or a marginal methionine/ choline-deficient diet containing PFOS; oxidative stress biomarker data were only available for the 0.003% PFOS groups; choline reduced oxidative stress	Zhang et al. (2016a)
ROS, by CM-DCF-DA	Mouse, ICR, female	Oocyte	↑ ROS	PFOA, 0, 1, or 5 mg/kg per day	Oral, daily for 28 days	ROS significantly increased in a dose- dependent manner; increase in γH_2AX , a DNA damage marker provided supportive evidence for oxidative stress	<u>Zhang et al.</u> (<u>2022a)</u>
MDA, SOD, GPx, ROS, by TBARS assay for MDA, nitroblue tetrazolium- illumination method for SOD, benzoic acid method for GPx, DCFH-DA for ROS	Mouse, Balb/c, male	Epididymis, sperm cells	↑ MDA, No change in SOD, ↓ GPx, ↓ SOD/MDA	PFOA, 0, 1.25, 5, or 20 mg/kg per day	Gavage, daily for 28 days Sperm cells were exposed to PFOA at 0, 100 or 400 μ M for 1, 2, 4, or 8 h	No available method for ROS measurement in epididymis; therefore, sperm cells were isolated from mice for in-vitro PFOA exposure	<u>Lu et al.</u> (<u>2016b)</u>
MDA, SOD, GPx, by TBARS for MDA, xanthine oxidase assay for SOD, DTNB assay	Mouse, Kunming, pregnant female	Liver	↑ MDA, ↓ SOD, ↓ GPx	PFOA, 0, 1, 5, 10, 20, or 40 mg/kg per day	Gavage, daily from GD1 to GD7, killed on GD9	Dose-dependent effects observed for all biomarkers tested in the liver; uterine cell	<u>Zhang et al.</u> (2021b)

516

Table 4.17 (continued)

for GPx

IARC MONOGRAPHS - 135

apoptosis was observed

Table 4.17 (continued)								
End-point, assay	Species, strain, sex, cell line	Tissue	Result ^a	Dose	Route, duration, dosing regimen	Comments	Reference	
MDA, H ₂ O ₂ , SOD, CAT, by commercial kits	Mouse, Kunming, pregnant female	Ovary	↑ MDA, ↑ H_2O_2 , ↓ SOD, ↓ CAT	PFOA, 0, 2.5, 5, or 10 mg/kg per day	Gavage, daily from GD1 to GD7 or GD13	Significant increases in MDA and H_2O_2 , and decreases in SOD and CAT on GD13 (dose- dependent); significant increase in MDA on GD7 (not dose- dependent); mixed and non-significant results for H_2O_2 , SOD, and CAT on GD7	<u>Chen et al.</u> (2017)	
mtROS, by immunofluorescent microscopy (MitoSOX Red/MitoTracker Green)	Mice, unspecified strain, female	Granulosa cells/ oocytes	↑ mtROS	PFOA, 4 mg/kg per day for 30 days	Oral, in drinking-water		<u>Zhang et al.</u> (<u>2023a)</u>	
MDA, SOD, GPx, by ELISA kits	Rat, Sprague- Dawley, male	Testis	↑ MDA, ↓ SOD, ↓ GPx	PFOA, 0.01 g/kg per day	Oral gavage daily for 30 days	Both low and high doses of lipoic acid protected rats against reproductive damage by reducing oxidative stress biomarkers induced by PFOA	<u>Zhang et al.</u> (2023c)	
MDA, SOD, GPx, by commercial kits	Rats, strain not specified (albino), male	Jejunum	↑ MDA, ↓ SOD, ↓ GPx,	Potassium salt of PFOS, 5 mg/kg per day	Oral gavage daily for 28 days	Co-exposure with lemongrass essential oil (100 mg/kg per day) restored the levels of the biomarkers in rats dosed with 5 mg/kg per day	<u>Shalaby</u> et al. (2023)	
MDA, SOD, GPx, NOX4, by commercial kits and immunofluorescence staining	Mouse, C57BL/6, male	Serum (MDA, SOD, GPx); kidney (NOX4)	↑ MDA, ↓ SOD ↓ GPx	PFOS, 5 mg/kg per day	Oral gavage, 28 days	Cannabidiol partially restored the levels of markers	<u>Du et al.</u> (2023)	

IARC MONOGRAPHS - 135

Table 4.17 (continued)

End-point, assay	Species, strain, sex, cell line	Tissue	Resultª	Dose	Route, duration, dosing regimen	Comments	Reference
MDA, GPx, ROS, by assay kits (A003-1, A005-1) for MDA and GPx, DCFH-DA for ROS (in vitro)	Rat, Sprague- Dawley, male	Kidney	↑ MDA, ↓ GPx	PFOS, 1 or 10 mg/kg	Intraperitoneal injections every other day for 15 days	↑ MDA and ROS (significant at either dose), ↓ G-Px (significant only at 10 mg/kg); approx. 12% apoptosis at 10 mg/kg dose	<u>Tang et al.</u> (2022)
HO-1, SOD, Nrf2, by gene expression (mRNA) assays	Mouse, C57BL/6, male	White adipose cells	↑ Nrf2 expression, HO-1 and SOD: non-significant increases	PFOS, 100 μg/kg per day	Oral gavage for 36 days	Gene expression of oxidative stress biomarkers was measured	<u>Xu et al.</u> (2016)
MDA, H ₂ O ₂ , SOD, GPx, by TBARS, hydrogen peroxide kit, ELISA kits	Mouse, DBA/1J, male	Paw tissue	↑ MDA, ↓ GPx, ↓ SOD, H ₂ O ₂ non- significant increase	PFOS, 10 mg/kg	Rheumatoid arthritis was induced on days 1 and 21; daily oral gavage of PFOS on days 21–35	Paw tissue was extracted for analyses; PFOS induced oxidative stress in normal mice; the effect was enhanced in mice with rheumatoid arthritis	<u>D'Amico</u> <u>et al. (2022)</u>
MDA, CAT, GSH, by commercial kits	Mouse, C57BL/6, male	Ileum and colon	↑ MDA (colon and ileum), no changes in CAT, ↑ GSH versus PFBS	PFOS, 500 μg/L in drinking-water	PFOS in drinking-water for 28 days	TNF-α and IL-1β expression was also significantly increased in ileum but not in colon	<u>Chen et al.</u> (<u>2023b)</u>
MDA, H ₂ O ₂ , SOD, by assay kits	Mouse, Kunming, male	Liver	↑ MDA, ↑ H ₂ O ₂ , ↓ SOD	PFOS, 10 mg/kg per day	Daily oral gavage for 21 days	Grape seed proanthocyanidin extract was shown to be protective against PFOS- induced oxidative stress	<u>Huang et al.</u> (2020)

Table 4.17 (continued)								
End-point, assay	Species, strain, sex, cell line	Tissue	Resultª	Dose	Route, duration, dosing regimen	Comments	Reference	
MDA, SOD, CAT, GPx, by assay kits	Mouse, C57BL/6J male	Liver	↑ MDA, ↓ SOD, ↓ CAT, ↓ GPx	PFOA, 5 mg/kg per day	Daily oral gavage for 6 wk, followed by 0.5 mL saline, with or without gastrodin, daily for 8 wk	Gastrodin, isolated from the root of <i>Gastrodia</i> <i>elata</i> Blume, partially protected against the oxidative stress caused by PFOS	<u>Ma et al.</u> (2021)	
MDA, GSH, SOD, MPO, by assay kits	Rat, Sprague- Dawley, pregnant female	Lung	↑ MDA, ↑ MPO ↓ SOD, ↓ GSH	PFOS, 0, 0.1, or 2 mg/kg per day	Daily from GD0 to GD21	2 male and 2 female pups were randomly selected for the oxidative biomarker assays; significant oxidative stress was observed in the pups' lungs	<u>Chen et al.</u> (2012)	
Peroxidation (MDA), by TBARS	Mouse, C57BL/6, male	Liver	No change in MDA	PFOA, 0, 0.1, 1, or 5 mg/kg	Oral, via drinking-water containing 0.55, 5.5, or 28 mg/L PFOA	No significant increase in liver MDA in PFOA- treated mice; liver MDA significantly increased in CCl ₄ -treated mice (positive control)	<u>Crebelli</u> <u>et al. (2019)</u>	
mtROS, by DCFH-DA	Mouse, NMRI, pregnant female	Liver, brain, heart of fetus and placenta	↑ ROS in liver, brain, and heart of fetus but not placenta	PFOA, 0, 1, 10, or 20 mg/kg	Intraperitoneal injections, GD5–GD9	Significant increase in mtROS production with PFOA at 10 and 20 mg/kg (brain), and 20 mg/kg (liver, heart)	<u>Salimi et al.</u> (2019)	

Þ
R
\cap
2
3
$\underline{\circ}$
Z
0
ā
R
$\mathbf{\Sigma}$
Ð
Ť
S
1
_
ω
Ŭ

Table 4.17 (continued)

End-point, assay	Species, strain, sex, cell line	Tissue	Result ^a	Dose	Route, duration, dosing regimen	Comments	Reference
MDA, SOD, T-AOC, by TBARS, xanthine oxidase, Fe ³⁺ reduction methods	Mouse, Kunming, male and female pups	Liver, brain	No change in MDA, ↓ SOD, ↓ T-AOC	PFOA, 50 mg/kg	0 or 50 mg/kg by subcutaneous injection on PND7, 14, 21, 28, and 35	No changes in MDA levels in liver or brain at any time point; ↓ SOD in male pup brain on PND7 and PND21 and in female pup liver on PND14; ↓ T-AOC in male pup brain on PND21, in female pup liver on PND21, and in male pup liver on PND7 and PND14; ↑ T-AOC in male pup liver on PND21	<u>Liu et al.</u> (2009)
GSH, GR, by biochemical assay kits	Mouse, Balb/c, male	Liver	↑ GSH, ↑ GR	PFOS, effect at 5 mg/kg per day only	0.2, 1, or 5 mg/kg per day; oral gavage, 28 days		<u>Wang et al.</u> (2022c)
MDA, GSH, GPx, CAT, Cu-Zn-SOD, by TBARS for MDA, DTNB for GSH, assay kits for other biomarkers	Mouse, Balb/c, male	Liver, brain	Liver (no change in MDA, ↑ GSH, ↑ CAT, ↓ Cu-Zn-SOD, ↓ GPx). Brain (↑ MDA only for 30 mg/kg, ↓ GPx, no changes in other biomarkers)	PFOA, 15 or 30 mg/kg per day	Oral, 10 days One group received PFOA 30 mg/kg for another 10 days	Depletion of the antioxidant system suggested potential oxidative stress in the liver and brain	<u>Endirlik</u> <u>et al. (2022)</u>

Table 4.17 (continued)									
End-point, assay	Species, strain, sex, cell line	Tissue	Resultª	Dose	Route, duration, dosing regimen	Comments	Reference		
Hepatic peroxisomal β-oxidation and CAT, by spectrophotometry at 340 nm for β-oxidation and 240 nm for CAT activity	Rat, Fischer 344, male	Liver	↑β-oxidation ↓ CAT	PFOA, 150 mg/kg	150 mg/kg bw PFOA in corn oil given to rats of age 4, 10, 20, 50, and 100 wk	Rats were killed at pre- set time points between days 0 and 28 after exposure for oxidative stress determination; β -oxidation significantly increased in all age groups; CAT significantly reduced only in the rats aged 100 wk, probably due to senescence	<u>Badr and</u> <u>Birnbaum</u> (2004)		
β -oxidation, by $[1-{}^{14}C]$ palmitoyl- CoA as the substrate	Rat, Crl:CD BR, male	Liver	↑β-oxidation	PFOA (ammonium salt): 0.2, 2, 20, or 40 mg/kg per day	0.2, 2, 20, or 40 mg/kg per day PFOA, daily oral gavage for 14 days	No other oxidative stress biomarkers were measured	<u>Liu et al.</u> (1996)		
MDA, H ₂ O ₂ , GSH, SOD, by assay kits	Mouse, strain not specified, male	Liver	↑ MDA, ↑ H ₂ O ₂ , ↓ SOD, ↓ GSH	PFOS, 10 mg/kg per day	10 mg/kg per day PFOS, oral gavage daily for 3 wk	The study also showed the protective effects of naringin against PFOS- induced oxidative stress	<u>Lv et al.</u> (2018)		
mtROS, by fluorescent microscopy	Mouse, strain not specified, female	Cumulus– oocyte complexes	↑ mtROS at 250, 500, and 750 μM	PFOA, 50–2000 μM	PFOA for 24 h	Significant dose- dependent increases in mtROS	<u>Zhang et al.</u> (2023b)		

. . . . •

bw, body weight; CAT, catalase; CCl., carbon tetrachloride; CoA, coenzyme A; CYP, cytochrome P450; DCF, dichlorofluorescein; DCFH-DA, 2',7'-dichlorodihydrofluorescein diacetate; DEN, diethylnitrosamine, dG, 2'-deoxyguanosine; ECD, electrochemical detection; ELISA, enzyme-linked immunosorbent assay; ESR, electron spin resonance; GD, gestational day; GPx, glutathione peroxidase; GR, glutathione reductase; GSH, glutathione; GSSG, oxidized glutathione; GST, glutathione S-transferase; h, hour(s); γH₂AX, γ-H2A histone family member X; HETE, hydroxyeicosatetraenoic acid; H,O,, hydrogen peroxide; HPLC, high-performance liquid chromatography; IsoP, isoprostane; LC-MS/MS, liquid chromatographytandem mass spectrometry; LOX, lipoxygenase; MDA, malondialdehyde; mo, month(s); MPO, myeloperoxidase; mRNA, messenger RNA; mtROS, mitochondrial reactive oxygen species; NAC, N-acetylcysteine; NeuroP, neuroprostane; NO, nitrogen oxide; NR, not reported; NRf2, NF-E2-related factor 2; 8-OHdG, 8-hydroxydeoxyguanosine; 8-oxodG, 8-oxo-2'deoxyguanosine; PFBS, perfluorobutanesulfonate; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PGF, prostaglandin; PND, postnatal day; ppm, parts per million; qRT-PCR, quantitative reverse transcription-polymerase chain reaction ; RONS, reactive oxygen and nitrogen species; SOD, superoxide dismutase; T-AOC, total antioxidant capacity; TBARS, thiobarbituric acid-reactive substances; wk, week(s).

^a +, positive; \uparrow , increase; \downarrow , decrease.

the PFOA-treated mice. Rutin was able to reduce the oxidative stress induced by PFOA.

Takagi et al. (1991) measured 8-OHdG in the DNA of liver and kidney from rats dosed with PFOA. In the first experiment, male Fischer 344 rats were exposed to a single i.p. dose of PFOA (100 mg/kg) in corn oil (5 rats per group) and killed 1, 3, 5, or 8 days after dosing. In a second experiment (a feeding trial), rats were fed a powdered diet ad libitum containing 0.02% PFOA for 2 weeks (5 rats per group; the dose as mg/kg per day was not estimated). After necropsy, liver and kidney DNA was isolated. Significant increases in 8-OHdG levels were found in the livers of rats injected with a single dose of PFOA on days 3, 5, and 8. No effect of PFOA on 8-OHdG level was observed in the kidney. In the feeding trial, 8-OHdG was significantly increased in the liver but not in the kidney. [The Working Group noted that relatively high 8-OHdG levels were reported in both liver and kidney DNA of the control group rats. In addition, HPLC with electrochemical detection, as performed in this study, was considered to be a better technique than assay kits for 8-OHdG measurement.]

Zou et al. (2015) orally dosed male mice (n = 8) with PFOA at 10 mg/kg per day for 14 days. Oxidative end-points including ROS, 8-OHdG, H_2O_2 , MDA, SOD, and CAT were measured in the liver using commercial kits. The levels of 8-OHdG, H_2O_2 , and MDA were significantly increased in PFOA-treated mice compared with control mice, whereas SOD and CAT levels were significantly decreased. When the mice were co-treated with PFOA (10 mg/kg per day) and the flavonoid quercetin (75 mg/kg per day) for 14 days, quercetin was found to be able to reverse the PFOA-induced effects on these end-points.

In a study by <u>Minata et al. (2010)</u>, 39 wildtype (129S4/SvlmJ) and 40 PPAR α -null (129S4/SvJae-Pparatm1Gonz/J) male mice were randomly divided into four groups and orally dosed with PFOA (0, 12.5, 25, or 50 μ mol/kg per day) for 4 weeks. The level of 8-OHdG in the liver was

measured by HPLC using ultraviolet and electrochemical detectors. No changes in 8-OHdG levels were observed in wildtype mice, whereas the 8-OHdG levels were increased in a dose-dependent manner in the PPAR α -null mice, with a significant increase in the group at 50 μ mol/kg.

Rigden et al. (2015) dosed male Sprague-Dawley rats with PFOA and measured urinary 8-OHdG and MDA. Groups of five rats were dosed by gavage with PFOA (0, 10, 33, or 100 mg/kg bw once per day on days 4, 5, and 6). Urine samples were collected from day 1 to day 10. The two highest doses of PFOA induced significant increases in urinary MDA. The urinary 8-OHdG levels were significantly increased at a PFOA dose of 100 mg/kg. The levels of 8-OHdG started rising from day 3 and peaked from day 4 to day 7, depending on the dose, then decreased to baseline on days 8-10. The MDA levels started to increase from day 4, peaked on day 5, then decreased to baseline on day 8. The rises and peaks in 8-OHdG and MDA levels seemed to occur at about the same time after PFOA dosing.

[The Working Group noted that, altogether, the results support oxidative effects of PFOA, although the commercial kits may not be specific for the particular end-points measured, and the studies did not include a positive control group.]

Groups of six Sprague-Dawley male rats were dosed with PFOS at 0, 1, or 10 mg/kg per day for 28 days (<u>Han et al., 2018a</u>). The oxidative stress end-points measured in the liver included ROS production (measured by DCFH-DA), MDA, GSSG, CAT, SOD, and GSH (measured with assay kits). The MDA and GSSG levels were found to be significantly increased in a dose–response manner, whereas GSH, SOD, CAT, and GSH/ GSSG were significantly reduced. An increase in serum NO, coupled with an increase in mRNA expression of inducible nitric oxide synthase (iNOS) in the liver, provided additional evidence for oxidative stress induction.

Yan et al. (2015a) dosed male Balb/c mice (age 6-8 weeks) with PFOA by oral gavage (0, 0.08, 0.31, 1.25, 5, or 20 mg/kg per day) daily for 28 days (the second gavage was used for the administration of additional chemicals, but the mice that yielded the results described here only received water in this second gavage). The MDA, GSH, SOD, and CAT levels were measured in the liver after necropsy. The levels of MDA in the liver were found to be significantly decreased at all doses compared with the control. The liver GSH protein levels decreased, but not significantly, at the three lowest doses but significantly increased when the mice were dosed with 5 or 20 mg/kg. The authors also reported increased CAT ($\geq 3 \text{ mg/kg}$) and decreased SOD (\geq 5 mg/kg) enzyme activities in the livers of the mice. However, the dosing regimen was not provided in the manuscript. The study also analysed the mRNA expression of oxidative stress-responsive genes, including Cat, Sod1, Sod2, and Sesn1, in the livers of mice after exposure to PFOA for 28 days. Cat mRNA expression was unchanged after PFOA exposure. Sesn1 mRNA expression was significantly increased at the highest dose of 20 mg/kg, Sod1 mRNA was significantly increased at doses of 1.25 and 5 mg/kg, and Sod2 mRNA was significantly increased at 5 mg/kg. However, the mRNA expression changes for these three genes were considered moderate. This study suggested that the oxidative stress defence system was affected by repeated doses of PFOA, although oxidative stress, as indicated by the MDA concentration, did not increase in the mouse liver.

<u>Kamendulis et al. (2014)</u> dosed male C57BL/6 mice with PFOA at 0.5, 2.5, or 5 mg/kg for 7 days. The lipid peroxidation product 8-iso-PGF2α (by LC-MS) and the expression of the antioxidation response genes *Sod1*, *Sod2*, *GPx2*, *Cat*, and *Nq01* (by quantitative reverse transcription-polymerase chain reaction, qRT-PCR) were measured in the liver and pancreas. Cerulein was used as a positive control. The levels of 8-iso-PGF2α were significantly increased in the liver and pancreas upon treatment with PFOA. In the pancreas, all end-points, except for CAT, were significantly and positively correlated with PFOA exposure. In the liver, Sod1, Sod2, Cat, and Ngo1 expression was significantly increased by PFOA doses of 2.5 and 5 mg/kg. PFOA treatment did not have an effect on *Gpx2* expression in the liver. It was noted that cerulein treatment did not significantly induce any biomarkers, except for 8-iso-PGF2a, compared with the negative control. In another study (Kamendulis et al., 2022), PFOA exposure was thought to promote pancreatic cancer in a conditional Kras^{G12D} mouse model (LSL-Kras^{G12D} and *Pdx-1 Cre* mice). Drinking-water containing PFOA at 5 ppm was administered to mice (age 2 months) for 4 or 7 months. The mice, which were aged 6 or 9 months at the end of the dosing period, were examined for levels of MDA, and the antioxidant enzymes SOD, CAT, and thioredoxin reductase in the pancreas were measured using commercial kits. The MDA protein levels were significantly increased in mice only at age 9 months, whereas SOD enzyme activity was significantly increased at age 6 and 9 months. The CAT and thioredoxin reductase enzyme activities were only significantly increased in mice aged 6 months.

Male and female Wistar rats were fed a rodent diet containing PFOA at 0.01% w/w for 2 or 26 weeks (Kawashima et al., 1994). Microsomal NADPH-dependent lipid peroxidation, GSH, GPx (with H_2O_2 or cumene hydroperoxide), glutathione reductase (GSR), GST, CAT (H₂O₂ assay), and peroxisomal β -oxidation were measured in the liver. Peroxisomal β -oxidation was significantly increased in both male and female rats after 2 and 26 weeks. CAT levels were also increased compared with the controls at that time. In the livers of male rats, no changes in GSH were observed after 2 or 26 weeks, whereas microsomal NADPH-dependent lipid peroxidation, GPx, and GSR were increased, but were only significantly higher in male rats after 26 weeks of exposure. The GST activities

towards 1-chloro-2,4-dinitrobenzene and 1,2-dichloro-4-nitrobenzene were significantly reduced in male rats after both 2 and 26 weeks of PFOA exposure, whereas in female rats, the GST activities towards 1-chloro-2,4-dinitrobenzene and 1,2-dichloro-4-nitrobenzene were unchanged.

Lee et al. (2015) studied the effects of prenatal PFOS exposure on the fetal development of CD-1 mice. Pregnant mice were orally dosed with PFOS at 3 mg/kg per day from GD1 until GD17. In another group, non-pregnant female adult mice were dosed with PFOS at 3 mg/kg per day for 14 days. Oxidative stress biomarkers were measured in the mothers, fetuses, and non-pregnant mice. Oxidized lipid products derived from the lipid peroxidation of polyunsaturated fatty acids were measured by LC-MS/MS, and SOD and CAT were measured using assay kits. CAT was not measured in the fetuses because of insufficient sample sizes. No significant changes in SOD or CAT levels were observed in the livers or placentas of the pregnant mice. Levels of non-enzymatically oxidized lipid products, including isoprostanes (IsoP) 15-F_{2t}-IsoP, 5-F_{2t}-IsoP, and 8- F_{3t} -IsoP, and neuroprostane (NeuroP) 4(RS)-4- F_{4t} -NeuroP, 7 β -hydroxycholesterol, and 7-ketocholesterol, were significantly increased in fetal livers compared with levels in livers of control mice. Effects on enzymatically oxidized lipid products of the lipoxygenase (LOX) and CYP450 pathways were also examined. Upon PFOS treatment, the level of 15(S)-hydroxyeicosatetraenoic acid (HETE) decreased in fetal livers, whereas the level of 5(S)-HETE increased in maternal livers. As for products of the CYP450 pathway, increased levels of 27-hydroxycholesterol were only observed in fetal livers. In a second experiment, the effects of PFOS exposure on adult female brains and kidneys were compared with those on fetuses. With respect to the non-enzymatic pathway, arachidonic acid, adrenic acid, and neurofuran levels were significantly elevated in PFOS-dosed adult brains compared with non-exposed adults. However, only 8-F_{3t}-IsoP levels were significantly increased in the brains of PFOS-exposed fetuses compared with controls. With respect to the enzymatic pathway, oxidized lipid products were significantly increased in fetal brains, but not in adults. These included 9(S)-HETE (CYP-mediated), 5(S)-HETE, and 8(S)-HETE (LOX-mediated). These results indicate that oxidative stress due to PFOS exposure is more severe in fetuses than in their dams. In the kidneys of PFOS-dosed fetuses, significant increases in all the measured non-enzymatically oxidized lipid products, including 15-F_{2t}-IsoP, $5-F_{2t}$ -IsoP, $8-F_{3t}$ -IsoP, $4(RS)-4-F_{4t}$ -NeuroP, 17(RS)- F_{2t} -dihomo-IsoP, isofuran (IsoF), neurofuran, and 17(RS)-SC- Δ^{15} -11-dihomo-IsoF, were observed; whereas in PFOS-dosed adult female mice, only 15-F_{2t}-IsoP was increased, and a decrease in levels was observed for 5-F_{2t}-IsoP, 4(RS)-4-F_{4t}-NeuroP, 17(RS)-F_{2t}-dihomo-IsoP, and 17(RS)-SC- Δ^{15} -11dihomo-IsoF. The results indicated that oxidative stress induced by PFOS is more severe in fetuses than in their dams or adult female mice.

In a study by Liu et al. (2015a), adult male Kunming mice were dosed with PFOA at 2.5, 5, or 10 mg/kg per day for 14 days. The testicular levels of MDA, measured by TBARS, were significantly increased in mice exposed to PFOA at 5 or 10 mg/kg. Additionally, a significant increase in H_2O_2 , measured using the reaction with molybdenum acid, was observed. The induction of oxidative stress was further indicated by the reduction in *Nrf2* gene and protein expression, as well as reduced SOD and CAT activities (assay kits) in all the PFOA-treated mice.

Liu et al. (2015b) exposed male Kunming mice to PFOA orally at 10 mg/kg per day for 14 days. A second group of PFOA-dosed mice was cotreated orally with grape seed proanthocyanidin extract at 150 mg/kg per day. MDA, H_2O_2 , SOD, and CAT were measured in liver homogenates using assay kits (Jiancheng Institute of Biotechnology, Nanjing, China). The hepatic levels of MDA and H_2O_2 were significantly increased in PFOA-treated mice, whereas the SOD and CAT activities were reduced. In addition, *Nrf2* mRNA expression and protein levels were also decreased. The co-treatment with grape seed proanthocyanidin extract was able to restore the oxidative status of the livers of PFOAdosed mice.

Another study investigated both the hepatic and renal effects of PFOA in male Wistar rats (Owumi et al., 2021a). Groups of 10 rats were dosed orally with PFOA at 0 or 5 mg/kg per day for 28 days. Oxidative biomarkers were measured in the liver and kidney. The levels of reactive oxygen and nitrogen species (RONS) (measured with DCFH-DA), NO (Griess reaction), xanthine oxidase, and myeloperoxidase were significantly increased in both the liver and kidney upon PFOA exposure. The levels of GPx (measured using H_2O_2 and Ellman reagent), GSH (H_2O_2), GST (1-chloro-2,4-dinitrobenzene), SOD (autoxidation of epinephrine), and CAT (H_2O_2) were significantly decreased in the liver and kidneys of PFOA-exposed rats. Co-treatment with NAC was found to alleviate the oxidative stress of the PFOA-dosed rats. The effects of oxidative stress on testicular function in male Wistar rats were reported in another study (Owumi et al., 2021b). Groups of 10 rats were dosed orally with PFOA at 0 or 5 mg/kg per day for 28 days. The testes and epididymides were removed for the same suite of biomarker measurements to that described above (Owumi et al., 2021a). The levels of MDA, NO, xanthine oxidase, and myeloperoxidase were significantly increased in both the testes and epididymides of PFOA-exposed rats, whereas RONS levels were significantly increased only in the testes. The antioxidants SOD, CAT, and GSH were all significantly decreased upon PFOA treatment in both the testes and epididymides, the GPx levels were significantly decreased in the epididymides, and the GST levels were significantly decreased in the testes. Co-treatment with NAC alleviated the oxidative stress observed in the PFOA-dosed rats.

Yang et al. (2014) demonstrated oxidative stress in male Kunming mice dosed with PFOA at 2.5, 5, or 10 mg/kg per day for 14 days. The biomarkers MDA and H_2O_2 were measured using assay kits from the Jiancheng Institute of Biotechnology, Nanjing, and were found to be significantly increased in liver tissue after 2.5, 5, and 10 mg/kg, and 5 and 10 mg/kg doses, respectively.

Zhang et al. (2016a) fed male C57BL/6 mice a normal diet or a marginal methionine/cholinedeficient (mMCD) diet, both containing 0.003%, 0.006%, or 0.012% PFOS, for 23 days or 21 days (second replicate). A potential modulatory mechanism affecting hepatic steatosis and oxidative stress was investigated. Both PFOS-containing diets caused weight loss in the mice, and this weight loss was more severe with the mMCD diet. In a choline-supplementation experiment, the levels of MDA, GSH, and GSSG (assay kits) were measured in the livers of mice exposed to 0.003% PFOS in their diet for 6 weeks. The levels of MDA significantly increased, whereas the GSH/GSSG levels significantly decreased in mice fed either the 0.003% PFOS-normal diet or the 0.003% PFOS-mMCD diet. Choline supplementation alleviated the oxidative stress effects of PFOS. No biomarker data were available for the other dosage groups.

Zhang et al. (2022a) studied the effects of PFOA on oocyte maturation in ICR mice dosed with PFOA at 1 or 5 mg/kg per day for 28 days. ROS production measured by chloromethyl derivative of DCFH-DA was significantly increased in oocytes in a dose-dependent manner. PFOA-mediated induction of oxidative stress was further supported by an increase in the DNA damage marker yH2AX.

Another study investigated the effects of PFOA on the mouse epididymis (Lu et al., 2016b). Male BALB/c mice (age 6–8 weeks) were dosed by oral gavage with PFOA at 0, 1.25, 5, or 20 mg/kg per day for 28 days. At necropsy, the epididymides were extracted, and SOD, MDA, and

GPx levels were measured using the nitroblue tetrazolium-illumination, TBARS, and benzoic acid methods, respectively. Because there was no available method for measuring ROS in the epididymis, the assessment of ROS production was performed in sperm cells isolated from the epididymides and treated in vitro with PFOA at 0, 100, or 400 µM for 1, 2, 4, or 8 hours. Rosup was used as a positive control. ROS production in the sperm cells was measured spectroscopically using the DCF-DA method. The study reported significantly increased ROS production in sperm cells after 1 and 2 hours of treatment with 100 µM PFOA and at all time points after treatment with 400 µM PFOA. In the epididymides of the mice, MDA levels were significantly elevated when the mice were treated with PFOA at concentrations of 1.25 or 5 mg/kg per day, but not 20 mg/kg per day. No significant changes in SOD levels were observed upon PFOA exposure. However, the SOD/MDA ratio significantly decreased in the epididymides of mice exposed to PFOA at 1.25 or 5 mg/kg per day. Additionally, GPx levels were significantly decreased when mice were treated with PFOA at 5 or 20 mg/kg per day. In this study, the authors also demonstrated that PFOA disturbed the profile of polyunsaturated fatty acids in the epididymides of PFOA-treated mice. This observation, coupled with the increased oxidative stress in the epididymis, may explain the negative effects of PFOA on the reproductive function of male mice.

Zhang et al. (2021b) studied the mechanisms of PFOA toxicity in the uterus and liver of Kunming mice during early pregnancy. PFOA (1, 5, 10, 20, or 40 mg/kg per day) was administered to pregnant mice by gavage from GD1 to GD7, and they were killed on GD9. Their liver MDA levels significantly increased with increasing PFOA concentration, and the SOD and GPx levels significantly decreased in a dose-dependent manner. [The Working Group noted that the findings of this study suggest that oxidative damage may be involved when PFOA induces liver toxicity and uterine cell apoptosis, leading to possible embryo loss or damage.]

<u>Chen et al. (2017)</u> also studied the effects of PFOA on pregnant mice. Adult female Kunming mice were dosed by oral gavage with PFOA at 2.5, 5, or 10 mg/kg per day from GD1 to GD7 or GD13. Significant dose-dependent elevations in MDA and H_2O_2 and significant dose-dependent decreases in CAT and SOD levels were observed in the ovaries of PFOA-exposed mice on GD13. On GD7, MDA levels also significantly increased in the ovaries of mice at all the tested PFOA concentrations; however, no dose-dependent effects were observed. The levels of H_2O_2 , SOD, and CAT were not significantly changed on GD7. Biomarkers were measured using commercial kits.

In a recent study (Zhang et al., 2023a), female mice (strain not reported) were treated with PFOA at 4 mg/kg per day for 30 days. The levels of mt-ROS were significantly increased in granulosa cells upon treatment with PFOA, as assessed using the fluorescence ratio of MitoSOX Red/ MitoTracker Green.

In another study on the reproductive effects of PFOA, male Sprague-Dawley rats were dosed by oral gavage with PFOA at 0.01 g/kg bw for 30 days (Zhang et al., 2023c). Levels of MDA, SOD, and GPx in the testes were measured by ELISA kits. Levels of MDA were significantly increased, whereas levels of SOD and GPx were significantly decreased upon PFOA exposure. In addition, in PFOA-treated rats that were subsequently treated with lipoic acid at a dose of 0.1 g/kg per day for 42 days, levels of the markers were restored to those in the control rats. Lipoic acid at the low daily dose of 0.05 g/kg was less effective than the higher dose. Shalaby et al. (2023) reported that when rats were dosed with the potassium salt of PFOS by oral gavage at 5 mg/kg per day for 28 days, jejunal MDA levels were significantly increased, whereas jejunal SOD and GPx levels were significantly decreased. When rats were co-treated with lemongrass

essential oil at 100 mg/kg, 30 minutes before the PFOS dose (5 mg/kg per day) every day for 28 days, the lemongrass essential oil restored the levels of oxidative stress biomarkers.

Du et al. (2023) reported that CBD partially alleviated PFOS-induced apoptosis via the NOX4/ROS/JNK pathway. In this study, one group of male C57BL/6 mice was dosed by oral gavage with PFOS alone at 5 mg/kg per day for 4 weeks. Another group of PFOS-dosed mice was co-treated with CBD at 5 mg/kg by oral gavage daily, 4 hours after the PFOS treatment. The levels of MDA were significantly increased, whereas the SOD and GPx levels were significantly decreased in the serum of PFOS-dosed mice, as was the expression of NOX4, assessed using immunofluorescence staining, in kidney tissue. Co-treatment with CBD partially restored the levels of these markers. In addition, the authors observed alterations of MDA, SOD and GPx (oxidative stress end-points) in mouse kidney HEK293 cells in vitro.

Intraperitoneal injections of PFOS (1 or 10 mg/kg bw) were given to male Sprague-Dawley rats every other day for 15 days (Tang et al., 2022). The kidney MDA levels significantly increased in all PFOS-dosed rats compared with the control group. A significant decrease in GPx level was only observed in the group dosed with 10 mg/kg bw.

Xu et al. (2016) proposed that PFOS induces adipogenesis and glucose uptake in association with an activation of the Nrf2 signalling pathway. Male C57BL/6 mice (age 21 weeks) were given PFOS (100 μ g/kg bw) by oral gavage for 36 days. The expression of SOD and HO-1 did not increase, but *Nrf2* mRNA expression was significantly increased upon PFOS exposure, implying an upregulation of the Nrf2 signalling pathway and adipogenic gene expression in epididymal white adipose tissue. [The Working Group considered that the activation of the Nrf2 pathway suggested an oxidative stress-mediated effect of PFOS on adipogenesis.] <u>D'Amico et al. (2022)</u> studied the oxidative effects of PFOS exposure on healthy DBA/1J mice and mice with rheumatoid arthritis (induced by type II chicken collagen and complete Freund's adjuvant administered intradermally on days 0 and 21). PFOS (10 mg/kg bw) and other endocrine disruptors were given by oral gavage from day 21 to 35. Paw tissue was analysed for oxidative stress biomarkers, including MDA, H_2O_2 , SOD, and GPx. PFOS exposure significantly induced oxidative stress in healthy mice, and this effect was more severe in mice with rheumatoid arthritis.

The induction of metabolic disturbances by PFOS in different regions of the mouse gut was studied by <u>Chen et al. (2023a</u>). Male C57BL/6 mice (age 6 weeks) were exposed to PFOS at 500 µg/L in drinking-water for 28 days. CAT, GSH, and MDA levels were measured using commercial kits (Jiancheng Bioengineering Institute, Nanjing, China). The MDA levels significantly increased in the ileum and colon of mice exposed to PFOS, and this effect was more marked in the ileum than in the colon. However, no significant changes in CAT or GSH levels were observed in the ileum and colon of PFOSexposed mice compared with control mice. The levels of tumour necrosis factor alpha (TNF- α) and interleukin-1-beta (IL-1 β) were also significantly increased in the ileum but not in the colon of mice treated with PFOS.

In a study by <u>Huang et al. (2020)</u> in mice, the authors investigated the protective effects of grape seed proanthocyanidin extract against PFOS-induced oxidative stress. Male Kunming mice were dosed with PFOS (10 mg/kg bw) via gastric gavage daily for 21 days. Levels of MDA, H_2O_2 , and SOD were measured in the liver. PFOS-induced oxidative stress was indicated by an increase in MDA and H_2O_2 levels and a decrease in SOD level. The extract was shown to be protective against PFOS-induced oxidative stress. In another study (Ma et al., 2021), gastrodin, a water-soluble organic compound isolated from the root of *Gastrodia elata* Blume, was shown to reduce the oxidative stress induced by PFOA. C57BL/6J mice were dosed with PFOA (5 mg/kg bw) by oral gavage daily for 6 weeks, which was followed by treatment with gastrodin (20 mg/kg bw) or vehicle (saline) daily for 8 weeks. After PFOA administration, the liver MDA levels significantly increased, whereas the CAT, SOD, and GPx levels significantly decreased. The administration of gastrodin significantly reduced PFOA-induced oxidative stress.

<u>Ma et al. (2023)</u> also investigated the potential for rutin to reduce the PFOA-induced oxidative effects in mice. As reported above, male ICR mice were dosed with PFOA (20 mg/kg bw) daily for 28 days in the presence or absence of rutin. The serum contents of MDA, GPx, and SOD were significantly increased after exposure to PFOA. Rutin was able to reduce the oxidative stress induced by PFOA.

<u>Chen et al. (2012)</u> studied the effects of prenatal PFOS exposure on oxidative stress in the lungs of rat offspring. Pregnant female rats were dosed with PFOS (0.1 or 2 mg/kg bw) between GD1 and GD21 daily by gavage. Fresh lung tissue was collected from two male and two female pups from each group for biomarker analysis on PND0 and PND21. Pups from the dams that received the higher dose (2 mg/kg bw) showed significantly increased levels of MDA and significantly decreased GSH level and SOD activity compared with the controls at both time points. The MDA level showed a small but significant increase in the neonatal lungs of pups from dams exposed to PFOS at 0.1 mg/kg.

<u>Crebelli et al. (2019)</u> conducted a subacute experiment by supplying male C56BL/6 mice (age 6–8 weeks) with drinking-water containing PFOA (28, 5.5, or 0.55 mg/L, corresponding to 5, 1, or 0.1 mg/kg bw per day) for 5 weeks. Liver MDA levels were measured using the TBARS method. The study reported no significant changes in liver lipid peroxidation in any of the PFOA-treated groups. However, CCl_4 -treated mice (positive control) showed a significant increase in MDA level.

Salimi et al. (2019) studied the potential of PFOA to induce abortion and developmental toxicity in mice. Pregnant female NMRI mice (age 8–9 weeks) were dosed with PFOA at 1, 10, or 20 mg/kg between GD5 and GD9. mt-ROS production, measured using DCFH-DA, was significantly increased in the liver, brain, and heart of the fetuses, but not in the placenta.

Liu et al. (2009) studied the oxidative effects of PFOS in young mouse pups. The pups of Kunming mice were weaned on PND21, then dosed with PFOS at 50 mg/kg bw once by subcutaneous injection on PND7, PND14, PND21, PND28, or PND35. The levels of MDA, SOD, and TAC were determined in liver and brain 24 hours after exposure. No significant changes in MDA levels in the liver or brain were observed in pups of either sex. In the brain, significant decreases in SOD level were only observed in male pups after injections on PND7 and PND21. In the liver, a significantly decreased SOD level was only observed in female pups after injection on PND14. A significant decrease in TAC was only observed in male pup brains after injection on PND21. In the liver, a significant decrease in TAC was observed in female pups after injection on PND21 and in male pups after injection on PND7 and PND14. However, TAC was significantly increased in male pups after injection on PND21.

Wang et al. (2022c) orally dosed male Balb/c mice with PFOS at 0.2, 1, or 5 mg/kg bw daily for 28 days. GSH levels were significantly increased in the livers of mice exposed to PFOS at 5 mg/kg; however, a non-significant decrease was observed in the two lower-dosed groups. The level of GSR in the liver was increased in all the PFOS-exposed mice, but this only reached significance in the group at the highest dose.

Endirlik et al. (2022) examined the effects of PFOA exposure on the livers and brains of male Balb/c mice. The mice were orally dosed with PFOA at 15 or 30 mg/kg for 10 days. Levels of MDA and GSH were measured using the TBARS method and the 5,5'-dithiobis (2-nitrobenzoic acid) method, respectively. GPx, CAT, and Cu-Zn-SOD levels were measured using assay kits. In the liver, no changes in MDA level, but significantly increased GSH and CAT levels and significantly decreased Cu-Zn-SOD and GPx levels, were observed. In the brain, the MDA level significantly increased in mice exposed to PFOA at 30 mg/kg, whereas the GPx level significantly decreased, and no significant changes in the levels of the other biomarkers were observed. It was suggested that PFOA exposure results in a depletion of antioxidative systems and exhibits neurotoxic effects by inducing oxidative stress.

Badr and Birnbaum (2004) studied the oxidative effects of PFOA in male Fischer 344 rats ranging in age from juvenile (age 4 weeks) to post-puberty (age 10 weeks), mature adulthood (age 20 weeks), middle age (age 50 weeks), and senescence (age 100 weeks). The rats were dosed by gavage with PFOA (150 mg/kg bw) or vehicle. The rats were killed between days 0 and 28 postdosing for the measurement of hepatic peroxisomal β -oxidation and CAT activity in the liver. Hepatic peroxisomal β -oxidation was increased 3–5-fold in all PFOA-dosed groups, with an increased recovery time in older rats. Liver CAT activity was also significantly lower in senescent livers compared with those from all other groups.

Liu et al. (1996) treated adult male Crl:CD BR (CD) rats with APFO (C8) (0.2, 2, 20, or 40 mg/kg per day) for 14 days. β -Oxidation in the liver was measured using [1-¹⁴C]palmitoyl-CoA as the substrate, and potassium cyanide was used to inhibit mitochondrial β -oxidation. Hepatic β -oxidation was significantly increased upon PFOA exposure (≥ 2 mg/kg per day) in a dose-dependent manner.

Lv et al. (2018) showed the antioxidative properties of naringin (NAR, 4',5,7-trihydroxyflavonone-7-rhamnoglucoside), a naturally occurring flavonoid glycoside isolated from citrus fruits, with respect to PFOS-induced oxidative stress. Male mice were exposed to PFOS by oral gavage (10 mg/kg bw) daily for 3 weeks. Levels of MDA, H_2O_2 , SOD, and GSH were measured in the livers of the mice in the presence or absence of NAR (100 mg/kg bw). PFOS exposure significantly increased the MDA and H_2O_2 levels and reduced the SOD and GSH levels. Mice co-treated with NAR were protected from PFOS-induced oxidative stress.

[The Working Group noted that the studies described above reported the effects of acute, subacute, and subchronic repeated-dose treatment (up to 28 days) of PFOA or PFOS in vivo. The agents were mostly administered by oral gavage; the i.p. route, administration via drinking-water, or ad libitum feeding were used only in a few studies. Rodent studies dominated, and the dosing regime was typically 5–10 mg/kg per day. It was noted that such doses were much higher than exposures in humans. Despite this, PFOA and PFOS were shown to induce oxidative stress as shown also in studies in exposed humans and in human cells in vitro.]

(ii) Non-human mammalian systems in vitro

A wide range of experiments in vitro in non-human mammalian cells (primary and immortalized cells derived from different organ systems) have been used to explore the oxidative effects of PFOA and PFOS. The studies are briefly described below.

Lindeman et al. (2012) reported no significant increase in oxidatively damaged DNA (Fpg-modified comet assay) in freshly isolated rat testicular cells treated with 100 or 300 μ M PFOA for 24 hours. 1,2-Dibromo-3-chloropropane was used as a positive control.

Liver cells of animal origin have been widely used to explore the oxidative stress-inducing potential of PFOA and PFOS. Xu et al. (2019) isolated primary hepatocytes from young adult male C57BL/6J mice (age, 6 weeks). The liver cells were exposed to PFOA or PFOS (0, 10, 100, 500, or 1000 µM) for 24 hours. PFOA and PFOS significantly increased the ROS (PFOA at \geq 500 μ M; PFOS at 1000 µM), SOD (PFOA and PFOS at \geq 500 µM), and GSH (PFOA at \geq 500 µM; PFOS at 100 μ M and 1000 μ M) levels and significantly reduced CAT activity at all concentrations. The Working Group noted that PFOA or PFOS induced cell death at a concentration of $1000 \,\mu M$ and that the increases in oxidative end-points are not relevant at this concentration.] Goines and Dick (2022) also reported that PFOS increased ROS production and reduced the GSH level in HepG2 cells that were treated with 100 µM PFOS for 16 hours.

PFOS was shown to induce ROS production in kidney cells. <u>Wen et al. (2021)</u> exposed rat proximal renal tubular cells (NRK-52E) to PFOS (100 μ M) and measured ROS production after 1, 3, 6, and 24 hours of exposure. The exposed cells showed significantly increased ROS production after 1 hour, and the ROS levels remained significantly elevated at all the tested time points.

PFOA and PFOS were shown to induce ROS production in other organ systems. For example, <u>Reistad et al. (2013)</u> examined primary cultures of neurons isolated from rat cerebellum (age, 6-8 days; sex not reported). The cultured neurons were exposed to different concentrations (0, 6, 12, 25, 50, or 100 µM) of PFOA or PFOS for 24 hours. PFOA and PFOS significantly increased ROS production in a dose-dependent manner. The estimated half-maximal effective concentration (EC₅₀) values (\pm SD) were 25 \pm 11 μ M and 27 \pm 9.0 μ M for PFOA and PFOS, respectively. Increased ROS production was observed at concentrations close to the serum PFOA and PFOS concentrations found in occupationally exposed workers.

<u>López-Arellano et al. (2019)</u> studied the effects of PFOA on mouse oocytes in vitro. Ovary

explants were dissected from 17-day post-coitus mouse CD-1 embryos and cultured for 7 days. Oocytes were then treated with PFOA at $1/4 LC_{50}$ $(28.2 \ \mu\text{M})$ or LC₅₀ (112.8 μM) for 24 hours. ROS production was significantly increased in the treated oocytes compared with the controls. A similar study was conducted in porcine oocytes and cumulus cells (Mario et al., 2022). Cumulus cells are a group of granulosa cells that surround the oocytes and play a role in the process of oocyte development and maturation. The cells were treated with PFOA – $1/8 \text{ LC}_{50}$ (20 μ M) or $1/4 LC_{50} (40 \,\mu\text{M})$ – during in vitro maturation for 44 hours. Exposure to 40 µM PFOA significantly increased ROS production in the cumulus cells. An earlier study (Chen et al., 2021) also showed that PFOS exposure during in vitro maturation (100 μ M, 44–48 hours) significantly increased ROS levels in porcine oocytes.

Basini et al. (2022) studied the effects of PFOA on porcine ovarian granulosa cells that were exposed to PFOA (2, 20, or 200 ng/mL equal to 4.83, 48.3, or 483 μ M) for 48 hours. RONS, including the O^{2–}, H₂O₂, and NO levels, were assessed, and in contrast to the above findings, these were found to be significantly decreased after treatment with all the tested PFOA concentrations. Non-enzymatic scavenging activity was assessed by the ferric-reducing ability of plasma and was found not to be affected by PFOA exposure at the tested concentrations.

PFOS was suggested to have an adverse effect on cerebellar granule cells by inducing apoptosis via a ROS-mediated alteration in protein kinase C (PKC) signalling. PKC is a pivotal messenger molecule that is involved in neuronal function and development. PFOS (3 or 30 μ M) was found to increase ROS production in cerebellar granule cells from Sprague-Dawley rats aged 7 days. NAC (10 mM) pre-treatment 24 hours before PFOS exposure was shown to inhibit ROS production in the cells (Lee et al., 2012). In a study aiming to investigate the potential for PFOS to cause neurotoxicity (Li et al., 2017b),
PFOS (250 μ M, 24 hours) significantly increased ROS production in PC12 cells, a cell line derived from a rat pheochromocytoma. When cells were pre-treated with taurine, an antioxidant (80 mM, 30 minutes), PFOS-induced ROS generation was inhibited.

Qian et al. (2010) studied ROS in mouse RAW 264.7 macrophages exposed to PFOS (100 μ M, 5 minutes) using electron spin resonance spectroscopic measurement. PFOS significantly induced ROS (by 2.4-fold) in the cells. Electron spin resonance is considered to be a more robust technique for the detection of ROS because of its specificity and sensitivity. The study also showed that the production of ROS plays a role in PFOS-induced actin filament remodelling and the increase in endothelial permeability.

Elumalai et al. (2023) reported that PFOSinduced oxidative stress contributed to the apoptosis of rat insulinoma INS-1 cells. When cells were treated with 50 or 100 μ M PFOS for 36 hours, mitochondrial ROS generation increased through NOX2–gp91Phox activation and the inhibition of cAMP-PKA.

Dos Santos et al. (2022) treated the rat β -cell line INS-1E with low concentrations of PFOA (1 nM or 1 μ M) for 24 hours. No significant changes in ROS production, measured by fluorescence of DCF-DA, were observed.

Effects of PFOS (75–600 μ M, 24 hours) and PFOA (400–1000 μ M, 24 hours) on Nrf2^{-/-} knockout and wildtype C57BL/6 mouse astrocytes have also been studied (Alharthy and Hardej, 2021). The study showed that exposure to 600 μ M PFOS or 800 μ M PFOA significantly increased ROS levels, lipid peroxidation, and apoptosis in both the wildtype and Nrf2^{-/-} astrocytes. However, the effects of PFOA and PFOS were significantly larger in the Nrf2^{-/-} astrocytes than in wildtype cells. The GSH/GSSG ratio was significantly decreased in both wildtype and Nrf2^{-/-} astrocytes treated with PFOA or PFOS. However, the decrease was significantly larger in the Nrf2^{-/-} astrocytes treated with PFOA or PFOS.

Pre-treatment with butylated hydroxytoluene before PFOA and PFOS exposure significantly reduced ROS production in wildtype astrocytes but not in Nrf2^{-/-} astrocytes. In summary, the study showed that Nrf2^{-/-} astrocytes were more susceptible to PFOA and PFOS toxicity.

[The Working Group noted that the literature in experimental systems in vitro regarding the potential effects of PFOA and PFOS to induce oxidative stress was quite extensive. Oxidative stress is typically measured by generation of ROS, and this is sometimes coupled with the measurement of antioxidant enzyme activities, including those of SOD, CAT, GSH, and/or GPx. The Working Group identified more studies reporting evidence of oxidative stress induction than those reporting marginal evidence or no evidence.]

Oxidative stress was induced by PFOS in embryonic stem cell-derived cardiomyocytes from 129 mice (<u>Cheng et al., 2013</u>), in rat hepatocytes (<u>Khansari et al., 2017</u>), in rat HAPI microglial cells (<u>Wang et al., 2015c</u>), and in a co-culture of Sertoli cells and gonocytes from male Sprague-Dawley rat pups (<u>Zhang et al., 2013d</u>).

PFOA was shown to induce oxidative stress in MC3T3 osteoblast cells (Choi et al., 2016), in mouse spermatogonial GC-1 cells (Lin et al., 2020a), and in rat brain and liver mitochondria (Mashayekhi et al., 2015). Both PFOA and PFOS induced oxidative stress in rat mesangial cells (Gong et al., 2019).

Studies showing marginal or no effects included a study by <u>Berntsen et al. (2017)</u>, in which rat cerebellar granule neurons were exposed to PFOA or PFOS at 75, 150, 300, or 600 μ M for 3 hours, and H₂O₂ was used as a positive control. No significant increase in ROS production upon exposure to PFOA or PFOS was observed after 3 hours. PFOA and PFOS exposure for 3 hours resulted in a significant decrease in lipid peroxidation at many of the tested concentrations.

PFOA- or PFOS-induced oxidative stress may also affect neuronal development. <u>Slotkin et al.</u>

(2008) used PC12 cells as an in vitro model to study the effects of PFOA and PFOS exposure on undifferentiated (24 hours) and differentiating (4 days) cells. PFOA (at 10 and 250 μ M) and PFOS (at 50, 100, and 250 μ M) significantly increased MDA levels in the differentiating cells; 50 μ M chlorpyrifos was used as a positive control.

Other evidence that PFOA and PFOS induce oxidative stress in non-human cell lines was provided by several studies. Wang et al. (2015a) used *gpt delta* transgenic mouse embryonic fibroblast cells immortalized by the SV40 T antigen, <u>Oseguera-López et al. (2020)</u> performed a spermatozoa capacitation study using sperm from boars, and <u>Wei et al. (2021)</u> used mouse oocytes, and all three studies showed that PFOS induced oxidative stress in the respective cell types. A study by <u>Suh et al. (2017)</u> provided further evidence that PFOA induces oxidative stress in rat pancreatic β -cell-derived RIN-m5F cells.

In a rabbit renal proximal tubule model (Schnellmann, 1990), PFOS at 10, 20, 50, or 100 μ M did not increase MDA levels. However, GSH levels were significantly decreased in cells exposed to 100 μ M PFOS for 15 minutes.

Synopsis

[The Working Group noted that there is some evidence that PFOS induces oxidative stress in exposed humans, whereas the evidence for PFOA is mixed. Ten studies examined the effects of PFOS, of which six generated positive results, two marginal, and two negative results. Eleven studies in exposed humans examined the effects of PFOA; of these, only three studies showed a positive association of PFOA with oxidative stress, with one marginal and seven negative results.

In human primary cells in vitro, there is evidence that PFOA and PFOS induce oxidative stress. Similarly, PFOA and PFOS were found to induce ROS production in most in vitro studies using immortalized human cell lines, including HepG2 cells and other human cells, confirming the observations in both in vitro and in vivo studies in humans.

In experimental systems in vivo and in vitro, there is evidence that PFOA and PFOS induce oxidative stress.

The PFOA and PFOS concentrations used in exposure experiments in various non-human mammalian cells were similar to those used in human cell studies. The findings suggested that PFOA and PFOS can induce oxidative stress in vitro at concentrations similar to those used in human cells, in the micromolar concentration range. Some studies reported increased levels of 8-OHdG or 8-oxodG in cells or biological fluids, as measured by LC-MS/MS, which provided evidence for oxidative stress and potential oxidative damage to DNA. The generation of oxidative stress by PFOA and PFOS exposure is further supported by increased ROS production and decreased antioxidant defence end-points. Co-treatment with antioxidants has been shown to reduce oxidative stress, which provides additional evidence that oxidative stress may be involved when experimental systems are exposed to PFOA or PFOS.]

4.2.6 Induces chronic inflammation

(a) Humans

(i) Exposed humans

See <u>Table 4.18</u>.

Changes in inflammatory markers, including C-reactive protein (CRP) cytokines such as the interleukins (ILs) IL-1 β , IL-2, IL-4, IL-5, IL-6, IL-8, and IL-10, TNF- α , interferon-gamma (IFN- γ), monocyte chemoattractant protein-1 (MCP-1), neutrophil count; lymphocyte count; leptin; and adiponectin; as well as parameters related to eczema and rhino-conjunctivitis, atopic dermatitis, and lung function; were examined as part of the KC "induces chronic inflammation".

In total, the Working Group had access to 18 studies that investigated associations between

End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Inflammatory	markers in ge	eneral population	exposure				
35 inflammatory proteins	Serum/ plasma	Europe HELIX project; in 1101 mother-child pairs, PFAS measured in blood collected during pregnancy (prenatal exposure) and in children of age 8 years (postnatal exposure) (range, 6–12 years). 35 inflammatory proteins measured in the children's plasma Cross-sectional design.	Maternal PFOS: median, 6.2 ng/mL Child PFOS: median, 1.5 ng/mL n = 1101 Maternal PFOA: median, 2.2 ng/mL Child PFOA: median, 1.9 ng/mL n = 1101	↓ MIG, MIP1-β (prenatal PFOS) No association with adiponectin, BAFF, C-peptide, CRP, EGF, eotaxin, FGF-2, G-CSF, HGF, IFN-α, IFN-γ, IL-10, IL-12, IL-13, IL-15, IL-17, IL-1RA, IL-2, IL-2R, IL-4, IL-5, IL-6, IP10/CXCL10, leptin, MCP-1/CCL2, MIP- 1α/CCL3, PAI-1, or TNF-α ↑ IL-1β (prenatal PFOA) ↓ IL-8, HPG (postnatal PFOA) No association with adiponectin, BAFF, C-peptide, CRP, EGF, eotaxin, FGF-2, G-CSF, HGF, IFN-α, IFN-γ, IL-10, IL-12, IL-13, IL-15, IL-17, IL-1RA, IL-2, IL-2R, IL-4, IL-5, IL-6, IP10/CXCL10, leptin, MCP-1/CCL2, MIP- 1α/CCL3, PAI-1, or	Maternal age and education, pre-pregnancy BMI, parity, cohort, child ethnicity, age at examination, and sex	Potential divergence of PFAS exposure profiles in childhood from gestational exposure profiles because of factors such as transplacental transfer, breastfeeding, and dietary exposure. Sample size relatively large. Possibility of loss to follow-up and selection bias in longitudinal birth cohorts, children of mean age 8 years. Both mixture-approach statistical analysis and linear regression analysis were performed. All significance levels were set to 0.05 in this study, but it is unclear whether the linear analysis involved multiple correction.	Papadopoulou et al. (2021)

Table 4.18 End-points relevant to chronic inflammation in humans exposed to PFOA or PFOS

Table 4.18	(continued	d)					
End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
HbA1c, adiponectin, leptin, CRP, SHBG, and IL-6	Plasma	USA Prospective prebirth cohort with 3 years of follow-up. Measurements in mothers during and after pregnancy	PFOS: median, 24.8 ng/mL, n = 450-454, depending on the marker, during pregnancy PFOA: median, 5.6 ng/mL, n = 450-454, depending on the marker during pregnancy	No association with HbA1c, adiponectin, leptin, CRP, SHBG, or IL-6 No association with HbA1c, adiponectin, leptin, CRP, SHBG, or IL-6	Age, pre-pregnancy BMI, marital status, race/ ethnicity, education, income, smoking, parity, and breastfeeding history	Loss to follow-up between pregnancy and 3 years postpartum, requiring correction for potential selection bias. Limited power to detect subtle effects, potentially resulting in false negatives; small effect sizes with wide confidence intervals; and no adjustment for multiple testing.	<u>Mitro et al.</u> (2020)
IFN-γ, IL-6, IL-10, TNF-α, CRP	Serum	Atlanta, USA African American Maternal–Child Cohort Longitudinal assessment	PFOS: median, 2.19 ng/mL; <i>n</i> = 425 PFOA: median, 0.72 ng/mL; <i>n</i> = 425	↑ TNF-α and IFN-γ at both 8–14 wk and 24–30 wk of gestation ↑ IL-6 and CRP at 24–30 wk of gestation No associations with IFN-γ, IL-6, IL-10, TNF-α, or CRP at either 8–14 wk or 24–30 wk of gestation	Models were adjusted for gestational age at sample collection, maternal age, education, prenatal BMI, and parity	PFOS and PFOA were detected in > 98% of samples at 8–14 wk of gestation; inflammatory markers measured at up to two time points (8–14 wk and 24–30 wk of gestation).	<u>Tan et al.</u> (2023)

Table 4.18	Table 4.18 (continued)									
End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference			
87 inflammatory proteins	Plasma	Sweden Prospective adult population- based cohort Cross-sectional design	PFOS: median, 13.4 ng/mL; <i>n</i> = 965 PFOA: median, 3.3 ng/mL; <i>n</i> = 965	No association with CRP or other inflammatory markers ↓ CRP	Sex, sample storage time in freezer, smoking, exercise habits, education, energy and alcohol intake, BMI, glomerular filtration rate, glucocorticoid and COX- inhibitor treatment	Sample size was moderate, multiple covariate adjustments and correction for multiple testing were applied. Older individuals, sensitivity analysis included. CRP assessed using ELISA, Older Swedish participants, limiting the generalizability of the findings to other age and ethnic groups.	Salihović et al. (2020b)			
CRP, IL- 6, IL-1β, adiponectin, and leptin	Serum	Taiwan, China Young adults (age 12–30years) Cross-sectional design	PFOS: median, 8.93 ng/mL; <i>n</i> = 287 PFOA: median, 2.39 ng/mL; <i>n</i> = 287	No associations No association with CRP, IL-6, IL-1β, adiponectin, or leptin	Adjustment for age, sex, additional lifestyle factors, and measured parameters; a significant association was only identified when it remained consistent across all three models, avoiding model- dependent results	Sample size was relatively small.	<u>Lin et al.</u> (2011)			
IL-6, IL-10, and TNF-α	Serum of women with overweight or obesity during pregnancy or afterwards	San Francisco Bay Area, USA Participation in the MAMAS Cross-sectional design	PFOS: median, 2.83 ng/mL; <i>n</i> = 103 PFOA: median, 1.4 ng/mL; <i>n</i> = 103	↑ IL-6 No association with IL-10 or TNF-α ↑ IL-6 No association with IL-10 or TNF-α	Age, race/ethnicity, time-varying BMI, parity, education, smoking status, number of gestational weeks at baseline, and visit	Sample size was small, which may have limited the ability to detect associations ($n = 103$), lack of data on breastfeeding	<u>Zota et al.</u> (2018)			
CRP, absolute neutrophil count, and lymphocyte count	Serum	USA. NHANES 2005–2006. Adults aged ≥ 20 years. Cross-sectional design	PFOS: median, 11.4 ng/mL; <i>n</i> = 6652	↓ Neutrophil count ↑ Lymphocyte count No association with CRP	Age, sex, race/ethnicity, education, poverty income ratio, BMI, and serum cotinine	Sample size was relatively large, potential bias from participants taking anti-inflammatory or immunomodulatory medication.	<u>Omoike et al.</u> (2021)			

Table 4.18	(continue	d)					
End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
			PFOA: median, 3.2 ng/mL; <i>n</i> = 6652	↓ Neutrophil count ↑ Lymphocyte count No association with CRP			<u>Omoike et al.</u> (2021) (cont.)
MCP-1, HO- 1, and IL-8	Maternal plasma or serum	China Prospective, nested case– control study Cross-sectional assessment	PFOS: median, 1.79 ng/mL <i>n</i> = 519 (144 cases and 375 controls) PFOA: median, 0.79 ng/mL; <i>n</i> = 519 (144 cases and 375 controls)	↑ MCP-1 No association with IL-8 or HO-1 ↓ IL-8 No association with MCP-1 or HO-1	Crude models were adjusted for gestational age. Full models included covariates that were significantly associated with the levels of the biomarkers or PFOS and PFOA	Case-control study, sample size was small (144 cases and 375 controls). Participants were recruited only during the first and second trimesters (4–22 wk of gestation).	<u>Liu et al.</u> (2020a)
Leptin, adiponectin	Umbilical cord serum	China SMBCS Prospective birth cohort Cross-sectional assessment	PFOS: median, 1.94 μg/L; <i>n</i> = 1111 PFOA: median, 3.76 μg/L; <i>n</i> = 1111	↑ Adiponectin ↑ Leptin (women) ↑ Adiponectin ↑ Leptin (women)	Maternal factors (age, occupation type, education level, family annual income, pre-pregnancy BMI, gestational weight gain, weekly physical activity, smoking during pregnancy, parity); infant factors (sex and gestational age); and an interaction term for PFAS	Sample size was relatively large, biomarkers not specific for chronic inflammation.	<u>Ding et al.</u> (2023)

and sex

Table 4.18	(continued	d)					
End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Inflammatory	markers in co	mmunities with e	levated exposure				
IL-1β, IL-2, IL-4, IL-5, IL-6, IL-8, IL-10, IFN-γ, GM-CSF, TNF-α	Serum	USA Population- based cohort study, adults with median age 61 years Cross-sectional design	PFOS: median, 8.2 ng/mL; <i>n</i> = 212 PFOA: median, 3.3 ng/mL; <i>n</i> = 212	No association with IL-1 β , IL-2, IL-4, IL- 5, IL-6, IL-8, IL-10, IFN- γ , GM-CSF, or TNF- α \downarrow IL-1 β No association with IL-2, IL-4, IL-5, IL-6, IL-8, IL-10, IFN- γ , GM-CSF, or TNF- α	Age, BMI, sex, smoking history, and race/ ethnicity	Sample size was small ($n = 212$), may not be generalizable to non-AFFF-exposed populations; the biomarkers were not all measured in the entire sample, which further reduced the power. Limitations of the study included low detection frequency of cytokines and non-fasting samples, suggesting that cytokine concentrations may have been influenced by recent food intake.	<u>Barton et al.</u> (2022)
TNFα, IL-6, IL-8, IFN-γ, adiponectin, and leptin	Serum	Mid-Ohio Valley, USA The C8 Health Study of adults Cross-sectional design	PFOS: mean, 26.9 ng/mL; <i>n</i> = 200 PFOA: mean, 94.6 ng/mL; <i>n</i> = 200	$ \begin{tabular}{lllllllllllllllllllllllllllllllllll$	Age, alcohol consumption, BMI, sex, and eGFR	Serum samples were randomly selected from a larger cohort; pilot study with no multiple comparisons.	<u>Bassler et al.</u> (2019)
End-points of a	lisease outcor	ne					
Eczema and rhino- conjunctivitis	Maternal plasma	Hokkaido, Japan Prospective birth cohort with 7 years of follow-up Longitudinal design	PFOS: median, 5.1 ng/mL; <i>n</i> = 2689 PFOA: median, 1.9 ng/mL; <i>n</i> = 2689	↓ Eczema No association with rhino-conjunctivitis ↓ Eczema No association with rhino-conjunctivitis	Sex parity, maternal age at delivery, maternal smoking during pregnancy, BMI pre- pregnancy, and annual household income during pregnancy	Large sample size, loss to follow-up, self-reported questionnaires, with the possibility of recall bias affecting outcomes, lack of inflammatory biomarker measurement.	<u>Ait Bamai</u> <u>et al. (2020)</u>

Table 4.18	18 (continued)									
End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference			
Lung function, asthma, atopic dermatitis, rhinitis, LRTIs	Serum at age 10 years	Norway ECA prospective birth cohort with 10–16 years of follow up. Longitudinal design	 PFOS: median, 5.2 ng/mL; <i>n</i> = 378 PFOA: median, 1.6 ng/mL; <i>n</i> = 378 	 ↑ LRTIs (entire sample and boys aged 10–16 years) No association with reduced lung function, asthma, atopic dermatitis, or rhinitis ↑ Rhinitis (PFOA) ↑ LRTIs (entire sample and girls aged 10–16 years) No association with lung function, asthma, or atopic dermatitis 	BMI at age 16 years, puberty status, maternal education, and physical activity level at age 16 years (frequency of activities leading to breaking sweat and shortness of breath); after 10 years of follow up: physical activity and maternal education	Sample size was small, potential selection bias related to socioeconomic factors and loss to follow- up.	<u>Kvalem et al.</u> (2020)			

10010 4.10	(continue)	u)					
End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Childhood atopic dermatitis	Fetal umbilical cord blood	Shanghai, China Prospective birth cohort with 2 years of follow-up	PFOS: median, 2.5 ng/mL; <i>n</i> = 687 PFOA: median, 7.0 ng/mL; <i>n</i> = 687	No association with atopic dermatitis ↑ Atopic dermatitis in girls	Infant sex, parity (nulliparous and parous), birth weight, gestational age at delivery, mode of delivery, maternal pre-pregnancy BMI, maternal age, maternal education, maternal ethnicity, paternal age, paternal education, parental history of allergic disorders, paternal smoking during pregnancy, family income, and breastfeeding during the first 6 months; as only a few women consumed alcohol (1.98%) or smoked (0.5%), maternal alcohol consumption and smoking were not included in the fully adjusted model	Diagnosis of atopic dermatitis confirmed by two dermatologists. Adjustment for a large number of potential confounders. Loss to follow-up resulted in a substantial proportion (23%) of participants being excluded from the analysis. Limited sample size for some types of exposure.	<u>Chen et al.</u> (2018a)

Table 4.18 (continued)

Table 4.18	(continue	d)					
End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Lung function, asthma, eczema	Maternal plasma	Spain. Prospective birth cohort with 7 years of follow-up	PFOS: median, 5.8 ng/mL; <i>n</i> = 1243 PFOA: median, 2.3 ng/mL; <i>n</i> = 1243	Trends toward associations with lower risks of asthma and eczema; no association with reduced lung function Trend toward a positive association with reduced lung function (FVC and FEV1) at age 4 years, no association at age 7 years; no association with asthma or eczema	Maternal age at delivery, parity, previous breastfeeding, pre- pregnancy BMI, region of residence, and country of birth	Large sample size. Lack of correction for multiple comparisons and small effect estimates for the relationships of PFAS with lung function necessitate careful interpretation. The potential for chance findings because of multiple comparisons was acknowledged. Lack of information about <i>P</i> -value criteria and statistical significance, reliance on maternal PFAS concentrations as proxies for fetal exposure, and self-reported questionnaires for outcome assessment introduced potential for recall and misclassification	<u>Manzano-</u> <u>Salgado et al.</u> (2019)
Childhood atopic eczema, food allergy, allergic rhinitis, and asthma	Maternal plasma	Norway MoBA study, prospective population- based pregnancy sub-cohort with 7 years of follow-up	PFOS: median, 12.9 ng/mL; <i>n</i> = 1943	↓ Common cold, ear infection, urinary tract infection ↑ Bronchitis/ pneumonia No association with atopic eczema, food allergy, allergic rhinitis, asthma, throat infection, pseudocroup, or diarrhoea/gastric flu	Maternal age, maternal BMI, maternal education, parity, and smoking during the pregnancy; correction for multiple testing using false discovery rate	7 years of follow-up, a relatively large number of participants, loss to follow-up and reliance on questionnaire-based outcomes. All samples had PFOS and PFOA concentrations above the LOQ.	<u>Impinen et al.</u> (2019)

Table 4.18	(continue	d)					
End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Childhood atopic eczema, food allergy, allergic rhinitis, and asthma (cont.)			PFOA: median, 2.5 ng/mL; <i>n</i> = 1943	↓ Common cold, urinary tract infection ↑ Bronchitis/ pneumonia, pseudocroup No association with atopic eczema, food allergy, allergic rhinitis, asthma, throat infection, ear infection, or diarrhoea/gastric flu			Impinen et al. (2019) (cont.)
Reduced lung function at birth, asthma, allergic rhinitis, atopic dermatitis	Fetal umbilical cord blood	Norway ECA prospective birth cohort with 2–10 years of follow-up	PFOS: median, 5.2 ng/mL; <i>n</i> = 641 PFOA: median, 1.6 ng/mL PFOA: median, 1.6 ng/mL; <i>n</i> = 641	No association with reduced lung function at birth, asthma, allergic rhinitis, or atopic dermatitis No association with reduced lung function at birth, asthma, allergic rhinitis, or atopic dermatitis	Confounders examined: sex; birth weight; birth month; breastfeeding at 6 and 12 months; maternal smoking during pregnancy; household smoking at birth, at preschool age and at school age; parental asthma; atopic dermatitis and allergic rhinitis; parental education; and household income Regression models were adjusted for sex only and correction for multiple testing was performed using Bonferroni	High follow-up rate after 10 years and representativeness of the study sample. Potential recall bias, especially in questionnaires and interviews, lack of data on important confounding factors, such as parity and previous breastfeeding.	Impinen et al. (2018)

Table 4.18	(continue)	d)					
End-point	Biosample type	Location, setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Lung function	Umbilical cord blood plasma and serum from children	Taiwan, China Prospective birth cohort with 7 years of follow-up	Cord blood PFOS: median, 6.4 ng/mL Children's serum PFOS: median, 5.9 ng/mL; n = 165 Cord blood PFOA: median, 2.4 ng/mL Children's serum PFOA: median, 2.7 ng/mL; n = 165	Positive association with reduced lung function in children Positive association with reduced lung function in children	Sex, child height, child BMI, birth weight, maternal education, eating habits, prenatal smoking history, history of environmental tobacco smoke exposure, maternal cotinine concentration, gestational age, family income, use of pesticide or incense at home	Sample size was small, lack of multiple testing correction, differences in demographic characteristics of the participants at follow- up compared with the original cohort might have introduced selection bias, reliance on cord blood and blood for the assessment of associations with PFAS exposure, which may not have fully captured the changing concentrations of these compounds over time, particularly during early life and childbood	<u>Kung et al.</u> (2021)

AFFF, aqueous film-forming foam; BMI, body mass index; COX, cyclooxygenase; CRP, C-reactive protein; ECA, Environment and Childhood Asthma study; eGFR, estimated glomerular filtration rate; ELISA, enzyme-linked immunosorbent assay; FEV1, forced expiratory volume in 1 second: FVC, forced vital capacity; G-CSF, granulocyte colony-stimulating factor; GM-CSF, granulocyte–macrophage colony-stimulating factor; HbA1c, glycated haemoglobin; HELIX, Human Early Life Exposome; HO-1, haem oxygenase 1; IFN-γ, interferon-gamma; IL, interleukin; LOQ, limit of quantification; LRTIs, lower respiratory tract infections; MAMAS, Maternal Adiposity, Metabolism, and Stress Study; MCP-1, monocyte chemoattractant protein-1; MoBA, Norwegian Mother, Father and Child Cohort Study; NHANES, National Health and Nutrition Examination Survey; PFAS, perfluoroalkyl and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; SHGB, sex hormone-binding globulin; SMBCS, Sheyang Mini Birth Cohort Study; TNF, tumour necrosis factor; USA, United States of America; wk, week(s).

^a ↑, increase; ↓, decrease.

human exposure to PFOA and PFOS and end-points relevant to chronic inflammation. Of these, 11 studies examined the relations between PFOS and/or PFOA concentration and circulating markers of inflammation. Furthermore, seven of the studies explored the associations between the concentrations of PFOS and/or PFOA in maternal or umbilical cord blood and health outcomes (atopic dermatitis, eczema, rhinitis, asthma, and lung function) in children.

Measurement of inflammatory end-points

Altogether, 11 studies addressed associations of PFOA and PFOS concentrations with inflammatory markers, including CRP, MCP-1, IL-8, IL-6, neutrophil count, lymphocyte count, IL-10, TNF- α , leptin, and adiponectin. Of these, nine assessed associations either in cross-sectional (n = 7) or prospective (n = 2) settings, involving both children and adults from the general population (study size range, n = 103-6652). In these settings, the primary focus was on examining associations with inflammatory proteins, considering background exposure as the predominant source. Of the 11 studies, two cross-sectional studies examined potential associations in individuals exposed to elevated levels of PFAS: <u>Bassler</u> et al. (2019) focused on participants with elevated exposure due to PFOA-contaminated drinking-water in the C8 Health Study; and Barton et al. (2022) examined a population exposed to drinking-water contaminated with aqueous film-forming foam (AFFF).

Studies of exposure in the general population

The nine studies have been organized in descending order of informativeness, taking into account factors such as study size, prospective versus cross-sectional design, and the specificity of the assessed markers of inflammation for chronic inflammation. Prospective cohort studies are more informative because they enable longitudinal assessments.

In a study of the effects of prenatal (during pregnancy) and childhood (in children aged 8 years) exposure to PFOA and PFOS on cardiometabolic factors and inflammatory status in children, the authors assessed 1101 mother-child pairs from the Human Early Life Exposome project (the children had an average age of 8 years) (Papadopoulou et al., 2021). The covariates included maternal age and education, pre-pregnancy BMI, parity, cohort, child ethnicity, age, and sex. The study reported that prenatal PFOA concentration was positively correlated with that of the pro-inflammatory cytokine IL-1 β . Postnatal PFOA concentration was negatively correlated with those of IL-8 and HGF. Prenatal PFOS concentration was negatively associated with those of MIG/CXCL9 and MIP-1 β /CCL4. No association was observed between PFOA and/or PFOS concentrations and several end-points, including adiponectin, BAFF, C-peptide, CRP, EGF, eotaxin, FGF-2, G-CSF, IFN-α, IFN-γ, IL-10, IL-12, IL-13, IL-15, IL-17, IL-1RA, IL-2, IL-2R, IL-4, IL-5, IL-6, IP10/ CXCL10, leptin, MCP-1/CCL2, MIP-1a/CCL3, PAI-1, or TNF-α (Papadopoulou et al., 2021). [The Working Group noted that the strengths of this study included the assessment of both prenatal and postnatal exposure to PFOA and PFOS and multiple markers of relevance for chronic inflammation in a large sample of participants from six European cohorts.]

Mitro et al. (2020) conducted a study involving 450 pregnant women recruited between 1999 and 2002 in Massachusetts, USA, to investigate the potential links between PFOA and PFOS plasma concentrations and the markers CRP, IL-6, HbA1c, adiponectin, leptin, and sex hormonebinding globulin (SHBG) during pregnancy and in plasma samples collected 3 years postpartum. The covariates included age, pre-pregnancy BMI, marital status, race/ethnicity, education, income, smoking, parity, and breastfeeding history. No significant associations between PFOA or PFOS exposure and CRP, IL-6, HbA1c, adiponectin, leptin, or SHBG were reported (<u>Mitro et al.</u>, 2020).

In the Atlanta African American Maternal-Child Cohort (n = 425), associations between serum concentrations of a PFAS mixture (including PFOA and PFOS) and serum concentrations of IFN-y, IL-6, IL-10, TNF-a, and CRP at two different time points (8-14 weeks and 24-30 weeks of gestation) were examined. Exposure to PFOS was considered to be the main driver of the effect of the mixture, according to the four different statistical methods used for analysis. When examining associations of the PFOA and PFOS concentrations with various markers, distinctions were observed between the two. PFOS concentration displayed positive associations with IFN-y and TNF-a during both 8-14 weeks and 24-30 weeks of gestation and were positively associated with IL-6 and CRP only during the 24–30-week period. In contrast, there were no significant associations between PFOA concentration and CRP, IL-6, IFN-y, IL-10, and TNF-α during both gestation periods (Tan et al., 2023). [The Working Group noted that the findings from this longitudinal assessment indicate disparities in the associations of PFOA and PFOS concentrations with circulating markers of inflammation, suggesting differences between the two agents' induction of circulating markers of chronic inflammation.]

In a cross-sectional study of 965 elderly individuals (age, 70 years; 50% women) from Sweden, associations of PFOA and PFOS with 86 inflammatory protein markers, including multiple cytokines, such as IL-6 and TNF- α , measured using a proximity extension assay, were assessed (Salihović et al., 2020b). Examination of the associations of the PFOA and PFOS concentrations with CRP, measured using a different assay, was included as a sensitivity analysis. The concentration of PFOA was consistently inversely associated with circulating CRP after multiple adjustments (see <u>Table 4.18</u>) and correction for multiple testing (Salihović et al., 2020b). [The Working Group observed that the cross-sectional findings suggest that PFOA concentrations were mainly associated with decreased levels of circulating markers of chronic inflammation, including CRP.]

In a cross-sectional study involving 287 participants (age, 12–30 years) from Taiwan, China, no significant associations of PFOA or PFOS exposure with CRP, IL-6, IL-1 β , adiponectin, or leptin were reported (Lin et al., 2011).

In a cross-sectional study of pregnant women with overweight or obesity from the San Francisco Bay area, USA, the associations of prenatal exposure to PFOA and PFOS with levels of inflammatory markers, such as IL-6, IL-10 and TNF- α , in pregnant and postpartum women were assessed (Zota et al., 2018). The study reported a positive association between PFOS and IL-6 (P < 0.05). Additionally, the PFOA concentration showed a positive association with the pro-inflammatory cytokine IL-6 (P < 0.05). These findings suggest that exposure to PFOA and PFOS is cross-sectionally associated with elevated levels of IL-6 in pregnant and postpartum women.

Using cross-sectional data (n = 6652) from the NHANES 2005-2012, PFOA and PFOS concentrations were found to be significantly associated with decreased neutrophil count and increased lymphocyte count in linear models. However, when stratified by quintiles, the associations of PFOA concentration with the neutrophil and lymphocyte counts diminished. No significant associations between PFOA or PFOS and CRP were found (Omoike et al., 2021). The study could not account for potential bias introduced by participants taking anti-inflammatory or immune-modulatory medication. [The Working Group noted that the percentage changes in markers of chronic inflammation were of small magnitude and that the finding of decreased neutrophil count versus increased leukocyte count makes the interpretation of the findings challenging.]

In a nested case-control study comprising 144 women experiencing spontaneous preterm birth and 375 control participants (total, n = 519), the associations of serum PFOA and PFOS concentrations (both as continuous variables and after division into quartiles) with inflammatory markers, specifically MCP-1 and IL-8, were examined during weeks 4-22 of gestation. The study reported that in the full cohort, the PFOS concentration was positively associated with that of MCP-1, whereas the concentration of PFOA was inversely associated with that of IL-8, after adjustment for major confounders, including maternal age at delivery, parity, pre-pregnancy BMI, occupation, spontaneous abortion history, medication use, folic acid use, fasting status, pre-pregnancy passive smoking status, the gestational week of sampling (≤ 12 weeks or 12–27 weeks), and the sex of the child (Liu et al., 2020a). [The Working Group observed that the assessed markers may lack specificity for chronic inflammation.]

In 1111 mother-infant pairs from the Sheyang Mini Birth Cohort Study (SMBCS), associations of prenatal exposure to PFOA and PFOS with the leptin and adiponectin concentrations in umbilical cord serum were assessed, and sex-stratification was performed in each model to assess potential sexually dimorphic effects. The PFOA concentration was positively associated with leptin concentration in women only (P < 0.05) and positively associated with adiponectin concentration (P < 0.05) both in the entire sample and in men. The PFOS concentration was found to be positively associated with that of adiponectin (P < 0.05), both in the entire sample and in men, and positively associated with the leptin concentration in women (Ding et al., 2023). [The Working Group noted that the evaluated markers are not specific for chronic inflammation and that use of the findings of this study to draw conclusions regarding chronic inflammation may require additional context and complementary markers to be assessed, in

order to provide a comprehensive assessment of the inflammatory state.]

Studies in communities with elevated exposure

Two studies investigated associations between elevated exposure to PFOA and PFOS in contaminated drinking-water and inflammation end-points (see <u>Table 4.18</u>).

Barton et al. (2022) evaluated associations of PFAS exposure via AFFF-contaminated drinking-water with IL-1β, IL-2, IL-4, IL-5, IL-6, IL-8, IL-10, IFN-y, granulocyte-macrophage colony-stimulating factor (GM-CSF), and TNF-a in the PFAS Assessment of Water and Resident Exposure (PFAS-AWARE) study (n = 212). All the models were adjusted for age, race/ethnicity, smoking history, sex, and BMI. The authors reported an inverse association between serum PFOA concentration and the probability of detecting IL-1 β . No significant associations were observed with IL-2, IL-4, IL-5, IL-6, IL-8, IL-10, IFN-γ, GM-CSF, or TNF-α. The authors recognized certain constraints, such as the infrequent detection of specific cytokines, including IL-1 β , and the possibility that cytokine levels could have been affected by recent food consumption, because of non-fasted samples (Barton et al., 2022).

Bassler et al. (2019) included 200 adult C8 Health Study participants with an elevated exposure to PFOA (PFOA mean concentration, 94.6 ng/mL). Associations of PFOA and PFOS concentrations with the inflammatory markers TNF- α , IL-6, IL-8, IFN- γ , adiponectin, and leptin were assessed. The PFOA concentration was positively associated with that of IFN- γ and inversely associated with that of TNF- α . The PFOS concentration was positively associated with that of adiponectin and inversely associated with those of TNF- α and IL-8. No significant associations were observed between the PFOS concentration and those of IL-6, IFN- γ , and leptin (Bassler et al., 2019). [The Working Group noted that in the context of elevated community PFOA exposure, both elevated and decreased concentrations of chronic inflammatory markers were reported.]

Measurement of end-points of disease outcome

Studies of exposure in the general population

In a prospective cohort study including 2689 children in Japan in which maternal enrolment occurred during the first trimester of pregnancy, with follow-up questionnaires administered when the children reached age 7 years, higher maternal PFOS concentrations were associated with a decreased risk of eczema. No significant association was reported with rhino-conjunctivitis. Similarly, PFOA concentration was also inversely associated with eczema, and no significant association was found for rhino-conjunctivitis (<u>Ait Bamai et al., 2020</u>).

The Environment and Childhood Asthma (ECA) Study was a prospective birth cohort including 378 participants with PFOA and PFOS measurements made at age 10 years and follow-up data collected at age 10 years (cross-sectional data) and age 16 years (longitudinal data). The cross-sectional data collected at age 10 years showed associations of the PFOA and PFOS concentrations with asthma. The longitudinal data showed no associations of the PFOA or PFOS concentration with atopic dermatitis. No associations were observed between PFOA or PFOS and lung function. Notably, PFOA concentration was positively associated with rhinitis in all participants at age 16 years. For airway infections, the longitudinal data showed that PFOA and PFOS concentrations were positively associated with lower respiratory tract infections (LRTIs) between ages 10 and 16 years (Kvalem et al., 2020). [The Working Group noted that the study's longitudinal approach minimized selection bias and the chance of misclassifying health outcomes. The asthma diagnosis was

based on medication use and clinical diagnosis, further enhancing its accuracy and reducing misclassifications.]

The association of prenatal exposure to PFOA and PFOS with childhood atopic dermatitis was investigated in a prospective birth cohort study including 687 children who had completed a 2-year follow-up. Only in female children (n = 328), a log-unit increase in PFOA concentration was found to be associated with an increase of 2.1-fold in atopic dermatitis risk (adjusted odds ratio, 2.07; 95% CI, 1.13–3.80) after adjusting for potential confounders (Chen et al., 2018a).

Four studies reported no significant associations of PFOS and/or PFOA concentrations with atopic dermatitis/eczema, food allergy, rhinitis, asthma, and/or lung function (Impinen et al., 2018, 2019; Manzano-Salgado et al., 2019; Kung et al., 2021). More specifically, in a prospective cohort study conducted in Spain involving 1243 mother-child pairs, the associations of the prenatal concentrations of PFOA and PFOS with asthma, eczema, and lung function were investigated (see Section 1.6.2 for details). Although the findings were not significant, the study showed that the PFOA and PFOS concentrations were associated with asthma and eczema during childhood (Manzano-Salgado et al., 2019). [The Working Group noted that the interpretation of the findings was hampered since the study presented trends in associations rather than statistically significant comparisons, requiring cautious interpretation.]

In a prospective cohort study conducted in Norway, the associations of prenatal exposure to PFOA and PFOS with the development of asthma, allergies, and common infectious diseases in early childhood was assessed. The study involved a prospective cohort of 1943 mother–child pairs and followed participants up to age 7 years. Health outcomes were assessed at age 7 years, including asthma, allergies, and common infections. Questionnaires were completed at ages 3 and 7 years. PFOS was inversely associated with

the common cold, ear infection, and urinary tract infection between the age of 0-3 years (n = 1207). In addition, PFOS was positively associated with bronchitis/pneumonia, whereas no significant associations were observed with atopic eczema, food allergy, allergic rhinitis, asthma, throat infections, pseudocroup, or diarrhoea/gastric flu at ages 0-3 years and 6-7 years. The PFOA concentration was inversely associated with the common cold and urinary tract infection at age 0-3 years. PFOA was also positively associated with bronchitis/pneumonia and pseudocroup. No associations with atopic eczema, food allergy, allergic rhinitis, asthma, throat infections, ear infections, or diarrhoea/gastric flu were identified at ages 0-3 years and 6-7 years (Impinen et al., 2019).

In a study of the 0-2- and 2-10-year intervals obtained from the 2- and 10-year follow-up investigations of a prospective general population birth cohort, the ECA study, performed in Norway, associations of PFOA and PFOS with physician-diagnosed asthma, wheeze, obstructive airway disease severity, reduced lung function, atopic dermatitis, rhinitis, rhino-conjunctivitis, allergic sensitization, common cold episodes, and LRTIs were assessed. Adjustment for maternal age, maternal BMI, maternal education, parity, and smoking during pregnancy; and correction for multiple testing using the FDR were performed. No associations between PFOA or PFOS and lung function at birth, asthma, allergic rhinitis, or atopic dermatitis were observed (Impinen et al., 2018).

Kung et al. (2021) conducted a study examining the potential link between prenatal or childhood exposure to PFOA and PFOS and the development of lung function during childhood. This investigation included 165 children from Taiwan, China. In the entire cohort, the PFOA and PFOS levels in cord blood were inversely associated with lung function. In particular, intrauterine PFOS exposure was inversely correlated with mean FEV1 (forced expiratory volume in the first second), FVC (forced vital capacity), and PEF (peak expiratory flow) in childhood, especially in subgroups with lower birth weight and allergic rhinitis (Kung et al., 2021).

(ii) Human cells in vitro See <u>Table 4.19</u>.

Human primary cells

Overall, five studies were available: four studies evaluated the effects of PFOA and PFOS exposure on cytokines (IL-6, IL-8, TNF- α , IL-4, IL-10, and IFN- γ) in peripheral blood leukocytes (Brieger et al., 2011; Corsini et al., 2011, 2012; Maddalon et al., 2023a) and the fifth focused on the IL-6 and IL-1 β responses to PFOS exposure in human primary decidual stromal cells during early pregnancy (Yang et al., 2016).

Brieger et al. (2011) reported a significant association between PFOS exposure and decreased TNF- α release by human peripheral blood mononuclear cells in response to lipopolysaccharide (LPS)-stimulation at both the 4-hour and 24-hour time points. Notably, there was no significant correlation of IL-6 with either PFOA or TNF- α .

Corsini et al. (2011) measured the release of pro-inflammatory cytokines (IL-6, IL-8, TNF-α, IL-4, IL-10, and IFN-y) in LPS-stimulated human peripheral blood leukocytes and in the human promyelocytic cell line THP-1. The study showed that both PFOA and PFOS reduced the release of TNF-a in peripheral blood leukocytes and in THP-1 cells, but only PFOS reduced the release of IL-6 in peripheral blood leukocytes. PFOA reduced the release of IL-8 at 100 μ g/mL, and PFOS reduced IL-8 release at 1–100 µg/mL in a dose-dependent manner in THP-1 cells. Both PFOA and PFOS reduced the release of T-cell derived cytokines (IL-4 and IL-10), whereas IFN-y release was reduced only by PFOS in peripheral blood leukocytes (Corsini et al., 2011).

End-point	Assay	Tissue, cell line	Result ^a	Concentration	Comments	Reference
I	,			(LEC or HIC) or range		
Primary cells						
IL-6 and TNF- α	ELISA	PBMCs	↓ TNF-α after 4 and 24 h No change in IL-6	PFOS, 100 μg/mL (LEC) PFOS, 1–100 μg/mL		<u>Brieger et al.</u> (2011)
			No change in IL-6 or TNF-α	PFOA, 1–100 μg/mL		
IL-6, IL-8, TNF-α, IL-4, IL-10, IFN-γ	ELISA	Peripheral blood leukocytes	↓ IL-6, TNF-α, IL-4, IL-10, IFN-γ	PFOS, 0.1 µg/mL (LEC)	Relevance to human exposure	<u>Corsini et al.</u> (2011)
		,	No change in IL-8	PFOS, 0.1–10 μg/mL	and concentration	
			↓ TNF-α (LEC, 1 μg/mL), IL-4, IL-10 (LEC, 10 μg/mL)	PFOA, 10 μg/mL (LEC)	effects	
			No change in IL-6, IFN-γ, or IL-8	PFOA, 0.1–10 μg/mL		
IL-6, TNF-α, IL-10, IFN-γ	ELISA	Peripheral blood leukocytes	↓ IL-6, IL-10, IFN-γ after PFOS treatment in blood leukocytes obtained from a female donor	PFOS, 10 μg/mL (LEC)	Sex of the donor considered	<u>Corsini et al.</u> (2012)
			↓ TNF-α after PFOS treatment of female and male donor or after PFOA in female donor	PFOS, 0.1 μg/mL (LEC) PFOA, 1 μg/mL (LEC)		
IL-8 and TNF-α	ELISA	PBMCs from male and female donor	↓ IL-8 and TNF-α (male donor only)	PFOS, 0.2 μM	Sex of the donor considered	<u>Maddalon</u> <u>et al. (2023a)</u>
IL-6 and IL-1β	qPCR	Primary human decidual stromal cells of early pregnancy	No change	PFOS, 0.1 μM		<u>Yang et al.</u> (2016)
Cell lines						
IL-8, TNF-α	ELISA	Promyelocytic cell line THP-1	↓ IL-8, TNF-α ↓ TNF-α ↓ IL-8	PFOS, 1 μg/mL (LEC) PFOA, 10 μg/mL (LEC) PFOA, 100 μg/mL (LEC)	Relevance to human exposure and concentration effects	<u>Corsini et al.</u> (2011)
TNF-a	ELISA	Promyelocytic cell line THP-1	↓ TNF-α	PFOS, 0.1 μg/mL PFOA, 10 μg/mL (LECs)		<u>Corsini et al.</u> (2012)

Table 4.19 End-points relevant to chronic inflammation in human cells in vitro exposed to PFOA or PFOS

Table 4.19 (continued)										
End-point	Assay	Tissue, cell line	Result ^a	Concentration (LEC or HIC) or range	Comments	Reference				
IL-4, GATA-3, IFN-γ (Th1 and Th2 responses)	ELISA, western blot, qPCR	Human Jurkat cells	↓ IFN-γ, ↑ IL-4 ↓ IFN-γ, ↑ IL-4	PFOS, 1 μM (LEC) PFOA, 1 μM (LEC)		<u>Yang et al.</u> (2021)				
IL-6, IL-8	qPCR, flow cytometry	HCEpiC, HCEC, and RPE cells	↑ IL-6, IL-8	PFOA, 400 ppm for 16 h	Only PFOA was detected in all the indoor carpet samples, and had the highest (37 458 ng/g) concentrations of all the perfluorinated compounds	<u>Tien et al.</u> (2020)				
IL-6, TNF-α, and IL-10; mRNA expression of <i>IL6</i> , <i>TNFA</i> , and <i>IL10</i>	ELISA and mRNA by PCR	Human placental trophoblast (HTR-8/ Svneo) cells	↓ IL-6, ↑ TNF-α ↑ IL-10 ↓ mRNA expression of <i>IL10</i> at low concentrations ↑ <i>IL6</i> mRNA expression at high concentrations	PFOS, 0.01 mg/L (LEC) PFOS, 0.1 mg/L (LEC) PFOS, 0.01 mg/L (LEC) PFOS, 0.1 mg/L (LEC)		<u>Zhu et al.</u> (2020)				
IL-6	ELISA	CCD-18Co myofibroblasts	↓ IL-1β-induced IL-6 production ↓ IL-1β-induced IL-6 production	PFOS, 0.6 μM (300 ng/mL) every 24 h to 96 h PFOA, 0.36 μM (0.15 ng/mL) every 24 h to 96 h	PFOS showed greater cytotoxicity than PFOA in CCD-18Co myofibroblasts	<u>Giménez-</u> <u>Bastida et al.</u> (2015)				
CXCL-8, CXCL-10, IL-1α, IL-1β, IL-6	ELISA	Bronchial epithelial cells (HBEC3-TK)	\uparrow IL-1α, \uparrow IL-1β at a PFOS concentration of 10 μM with, or ≥ 30 μM without, immune stimulation via TLR3 No change in IL- 6, \downarrow CXCL-10 and CXCL-8 (with TLR3 stimulation)	PFOS concentrations ranged from 0.13 to 60 µM and exposure lasted for 48 h		<u>Sørli et al.</u> (2020)				

Table 4.19 (continued)

End-point	Assay	Tissue, cell line	Result ^a	Concentration (LEC or HIC) or range	Comments	Reference
			No significant changes in cytokine release were observed at non-cytotoxic PFOA concentrations in unstimulated cells; ↑ IL-1β in TLR3- stimulated cells	PFOA concentrations ranged from 0.13 to 10 μ M and exposure lasted for 48 h		<u>Sørli et al.</u> (2020) (cont.)
mRNA expression of IL6, IL8, TNFA, NFKB1, MAPK8	qPCR	Human pancreatic cell line (PANC-1)	No changes in <i>IL8</i> , <i>TNFA</i> , <i>NFKB1</i> , <i>MAPK8</i> . ↓ IL-6 at 100 μM	PFOA, 10, 50, or 100 μM	$LD_{_{50}}$ was 195.74 μM	<u>Abudayyak</u> <u>et al. (2021a)</u>
IL-6, IL-8	ELISA	Human hepatocarcinoma cell line (HepG2)	\uparrow IL-6, \downarrow IL-8 (at both 25 and 50 $\mu M)$	PFOA, 10, 25, or 50 μM		<u>Abudayyak</u> <u>et al. (2021b)</u>
Tachykinin	LC-MS/MS	Astroglial SVG p12 cells	↑ tachykinin at 80 μM	PFOA, 10 or 80 μM for 48 h	No positive control	<u>Osemwegie</u> <u>et al. (2021)</u>
Mast cell-mediated inflammatory response, eicosanoids		Human mast cells (HMC- 1)	↑ eicosanoids (PFOS)	PFOS or PFOA, 30 μM		<u>Park et al.</u> (2021)
IL6, TNFA	qPCR	BV2 microglial cells	↑ <i>IL6</i> at all time points ↑ <i>TNFA</i> after 1 and 3 h ↑ <i>IL6</i> at 1 or 10 μ M, after 6 h ↑ <i>TNFA</i> at 10 μ M after 6 h	PFOS, 1 μM for 1, 3, or 6 h PFOS: 0.1, 1.0, or 10 μM for 6 h		<u>Zhu et al.</u> (2015)
IL-6, and TNF- α	ELISA		↑ IL-6 and TNF-α at all time points ↑ IL-6 ↑ TNF-α	PFOS: 1 μM for 6, 12, or 24 h 0.1 μM (LEC) for 24 h 1 μM (LEC) for 24 h		

550

Table 4.19 (continued)										
End-point	Assay	Tissue, cell line	Result ^a	Concentration (LEC or HIC) or range	Comments	Reference				
IL6, IL1B, and TNFA	qPCR	Human placental cell lines (HTR-8/SVneo and JEG-3 cells)	↑ <i>IL6</i> , <i>IL1B</i> , and <i>TNFA</i>	PFOS, 0–100 μM 50 μM (LEC) in HTR-8/SVneo 20 μM (LEC) in JEG-3 cells	Primarily focused on the immediate effects of PFOS exposure; the study did not include an extensive exploration of concentration- response relationships for the observed outcomes	<u>Li et al. (2021a)</u>				
IL-8, TNF-α	ELISA	Human monocytic cell line THP-1	↓ IL-8 at 20 μM ↓TNF-α at all time points	PFOS, 0.2, 2, or 20 μM 0.2 μM (LEC)	LPS 10 ng/mL was added for 24 h	<u>Masi et al.</u> (2022)				
TNFA, IL1B, IL6, IL8, COX2	RT-PCR and qPCR	Human mast cells (HMC-1)	↑ <i>TNFA</i> and <i>IL8</i> ↑ <i>IL1B</i> and <i>IL6</i> ↑ <i>COX2</i> ↑ histamine ≥ 25 μM	PFOA, 50–400 μM for 12 h 200 μM (LEC) 50 μM (LEC) 100 μM (LEC) PFOA, 25–100 μM for 24 h 25 μM (LEC)		<u>Singh et al.</u> (2012)				

CXCL, CXC motif chemokine ligand; ELISA, enzyme-linked immunosorbent assay; h, hour(s); HCEC, human corneal endothelial cells; HCEpiC, human corneal epithelial cells; HIC, highest ineffective concentration; HMC, human mast cells; IFN, interferon; IL, interleukin; LC-MS/MS, liquid chromatography-tandem mass spectrometry; LD₅₀, median lethal dose; LEC, lowest effective concentration; LPS, lipopolysaccharide; mRNA, messenger RNA; PCR, polymerase chain reaction; PBMC, peripheral blood mononuclear cell; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; ppm, parts per million; qPCR, quantitative polymerase chain reaction; RPE, retinal pigment epithelial; RT, reverse transcription; Th, T-helper; TLR, toll-like receptor; TNF, tumour necrosis factor.

^a ↑, increase; ↓, decrease.

In another study, similar results were obtained using human leukocytes and the human promyelocytic cell line THP-1, and the authors further showed that PFOA and PFOS affect nuclear factor kappa B (NF- κ B) activation, thereby suppressing cytokine secretion by immune cells (<u>Corsini</u> <u>et al., 2012</u>).

More recently, <u>Maddalon et al. (2023a)</u> investigated male and female peripheral blood mononuclear cells that were treated with PFOS at 0.2 μ M. The authors observed that PFOS reduced both IL-8 and TNF- α release for male donors only, and in case of TNF- α , a statistically significant difference between male and female donors was observed. [The Working Group noted that the results of these studies suggested that PFOA and PFOS exposure was associated with an attenuated inflammatory response in peripheral blood leukocytes.]

<u>Yang et al. (2016)</u> investigated the impact of PFOS exposure on decidual stromal cell function. PFOS alone had no effects on the pro-in-flammatory cytokines IL-6 or IL-1 β (<u>Yang et al.</u>, 2016).

Human cell lines

One study conducted in vitro experiments using Jurkat cells to investigate the mechanisms of PFOA- and PFOS-mediated T-helper (Th) Th1 and Th2 responses. The results showed that both PFOA and PFOS enhanced the Th2 response (IL-4) in Jurkat cells at $1.0-50 \,\mu$ M via STAT6 activation, but inhibited Th1-related gene expression (IFN- γ). The effects of PFOS on IL-4 and IFN- γ were reversed in Jurkat cells that did not express STAT6, suggesting that PFOA and PFOS exposure aggravate inflammation (Yang et al., 2021).

<u>Tien et al. (2020)</u> investigated whether PFOA in indoor particulate matter induced inflammation in corneal and retinal cells. Exposure to PFOA was found to promote the secretion of IL-6 and IL-8. The authors suggested that exposure to PFOA in particulate matter may increase the risk of age-related macular degeneration by inducing oxidative stress and inflammation in the retina. Only PFOA was detected in all the indoor dust samples and it was present at the highest concentrations, of the substances investigated (<u>Tien</u> <u>et al., 2020</u>). [The Working Group noted that the observed response might be attributed to both dust particles and the presence of PFOA in the dust.]

Zhu et al. (2020) investigated the effects of PFOS on the secretion of IL-6, TNF-a, and IL-10, as well as the mRNA expression of IL6, TNFA, and IL10, in human placental trophoblast (HTR-8/Svneo) cells. The cells were exposed to PFOS at 0, 0.01, 0.1, or 1.0 mg/L for 24 hours, and the survival rates, secretion levels, and mRNA expression were measured. PFOS exposure reduced the survival rate of the cells at higher concentrations and induced changes in the secretion of inflammatory cytokines. Specifically, PFOS reduced IL-6 secretion at lower concentrations but increased TNF- α secretion and *IL6* mRNA expression at higher concentrations. In addition, PFOS increased IL-10 secretion at higher concentrations but reduced IL10 mRNA expression at lower concentrations (Zhu et al., 2020).

Giménez-Bastida et al. (2015) selected the CCD-18Co myofibroblast cell model to investigate the ability of PFOA and PFOS to modulate inflammatory mechanisms. IL-1 β , an essential pro-inflammatory cytokine, was added to the cells to induce an inflammatory response. The study showed that PFOA and PFOS individually possess the ability to impair cell viability, induce cell proliferation, and reduce cell inflammation by reducing IL-1 β -induced IL-6 production by subepithelial myofibroblasts of the colonic CCD-18Co line at non-cytotoxic concentrations (Giménez-Bastida et al., 2015) (see also Section 4.2.10).

Another study assessed the effects of PFOA and PFOS on the respiratory system using bronchial epithelial cells (HBEC3-TK) (<u>Sørli</u> <u>et al., 2020</u>). PFOA and PFOS inhibited lung

surfactant function at 0.5 mM in an acellular test and induced a pro-inflammatory response at micromolar concentrations. PFOA and PFOS were not associated with reduced cell viability at 10 µM. PFOS increased the release of the pro-inflammatory cytokine IL-1a, and also increased that of IL-1 β at \geq 30 μ M without immunostimulation by the toll-like receptor 3 (TLR3) and at 10 µM with TLR3 stimulation. PFOS reduced the release of the chemokine CXCL8 at 3.3 µM and reduced that of CXCL10 at 10 µM when cells were immunostimulated. In contrast, PFOA at 10 µM increased only the release of IL-1 β in HBEC3-TK that were immunostimulated by TLR3. The authors acknowledged that the concentrations of PFOA and PFOS tested in the assay (0.5 mM) were considerably higher than what would be deposited in the alveoli of a person inhaling contaminated air, which may affect the study's relevance to human exposure (Sørli et al., 2020).

The role of inflammation in PFOA-induced pancreatic toxicity was investigated by measuring the mRNA expression levels of *IL6*, *IL8*, *TNFA*, *NFKB1*, and *MAPK8* in a human pancreatic cell line (PANC-1). PFOA was reported to induce dose-dependent cytotoxicity (IC₅₀, 195.74 μ M), with apoptosis being the main cell-death pathway induced. PFOA did not significantly increase the mRNA expression of the analysed immune response-related biomarkers (Abudayyak et al., 2021a).

Abudayyak et al. (2021b) also investigated the effects of PFOA on human hepatocarcinoma (HepG2) cells and found that PFOA significantly increased IL-6 levels (\leq 1.8-fold; $P \leq$ 0.05) but reduced IL-8 levels at 25 µmol/L (40% decrease) and 50 µmol/L (35% decrease) (Abudayyak et al., 2021b).

<u>Osemwegie et al. (2021)</u> investigated the effects of the in vitro exposure to PFOA of the astroglial SVG p12 cell line. The study measured the gene expression of the inflammatory marker tachykinin, which was increased by 80 μ M PFOA treatment for 48 hours (<u>Osemwegie et al., 2021</u>).

The effects of PFOA and PFOS on human mast cells have also been examined. The results showed that PFOS had the greatest impact, increasing degranulation and the production of inflammatory eicosanoids in mast cells (Park et al., 2021).

PFOS had effects on BV2 microglia, and its potential contribution to inflammation in the central nervous system was assessed (Zhu et al., 2015). The results showed that PFOS increased BV2 cell activation after 12 hours of treatment at 0.1–10 μ M. Treatment with PFOS at 1 μ M for 1–6 hours significantly increased *IL6* mRNA expression and increased *TNFA* mRNA expression at 1 μ M after treatments of 1 and 3 hours, but not longer. In addition, cells treated with a higher concentration of PFOS (10 μ M) increased the expression of both *IL6* and *TNFA* after 6 hours of treatment (Zhu et al., 2015).

A study by Li and colleagues aimed to examine the potential impact of PFOS on the development of the human placental cell lines HTR-8/ SVneo and JEG-3 cells, mediated through the PPAR γ pathway. The mRNA expression of *IL6*, *IL1B*, and *TNFA* were significantly increased by PFOS exposure in HTR-8/SVneo cells at 50 and 100 μ M. In JEG-3 cells, PFOS increased the mRNA expression of *IL6* when present at 10 μ M to 30 μ M, and of *IL1B* and *TNFA* at 20 and 30 μ M (Li et al., 2021a).

Masi et al. (2022) investigated the effect of exposure to PFOS on the expression of receptor for activated C kinase 1 (RACK1) receptor and immune cell activation. The results showed that PFOS exposure downregulated RACK1 expression when treated with low doses (0.2–20 μ M) for 18 or 24 hours, and reduced LPS-induced IL-8 and TNF- α production at 20 μ M and 0.2 μ M, respectively.

In another study, the role of PFOA in human mast cell (HMC-1)-mediated allergic inflammation and the underlying mechanisms was investigated (Singh et al., 2012). The HMC-1 cells were treated with PFOA (25–400 μ M) for 12 or

24 hours. PFOA exposure was found to increase histamine release after 24 hours at concentrations of $\geq 25 \,\mu$ M and to increase the gene expression of *TNFA*, *IL1B*, *IL6*, *IL8*, and *COX2* in the mast cells after 12 hours (Singh et al., 2012).

(b) Experimental systems

(i) Non-human mammalian systems in vivo

Pro-inflammatory effects

See <u>Table 4.20</u>.

Quist et al. (2015) examined adverse hepatic changes at both the microscopic and ultrastructural levels in CD-1 mice exposed prenatally to low doses of PFOA that were within the higher bound of the reference interval for human exposure. Pregnant CD-1 mice were exposed orally by gavage to PFOA at doses ranging from 0.01 to 1 mg/kg on GD1 to GD17, and histopathological changes in the livers of the offspring were observed on PND21 and PND91. Prenatal exposure to low doses of PFOA induced histopathological changes in the liver of offspring on PND21 and PND91, including chronic active periportal inflammation, which primarily involved lymphocytes and macrophages, with fewer plasma cells and occasional neutrophils. These effects were dose-dependent (Quist et al., 2015).

In a 2-year feeding study in rats exposed to PFOA (<u>NTP</u>, 2020; see also Section 3.1.2), the incidence of chronic active inflammation in the glandular stomach and of focal inflammation in the liver of male rats was increased in postweaning exposure groups at 150 or 300 ppm at age 16 weeks, compared with controls. The study in male rats was stopped at the interim time point, and a second study was started at lower exposures (up to 80 ppm). Glandular stomach chronic inflammation in female rats was not increased at the 16-week time point, but it was at the 2-year time point (<u>NTP</u>, 2020).

Filgo et al. (2015) (see also Section 3.1.1) investigated the severity of chronic liver inflammation in three mouse strains – CD-1, 129/SV wildtype (WT), and 129/SV knockout (KO) - that were exposed to PFOA from GD1 to GD17. Chronic inflammation increased in severity only in CD-1 mice and, notably, severity was increased at the two highest doses, but incidence was not increased. Indeed, no increase in the incidence of chronic inflammation was observed in any of the three strains. [The Working Group noted that the study investigated only female mice, which are recognized as being less responsive to PFOA than are males. Among the three strains, the evidence of chronic inflammation in two 129/SV strains (WT and KO) was inconclusive; only CD-1 exhibited a positive but not significant response.]

In a study by <u>Kamendulis et al. (2022)</u>, mice (age 8 weeks) were exposed to drinking-water containing PFOA at 5 ppm for 4 or 7 months (see Sections 3.1.2, 4.2.5). The chronic inflammation score was investigated at age 6 and 9 months, and was based on histopathological staining of sections of the pancreas. The study reported an increased inflammation score at 9 months compared with controls. [The Working Group noted that the evidence supporting a connection between PFOA and chronic pancreatic inflammation in the experimental model was inconclusive.]

Son et al. (2009) investigated the effect of PFOA exposure on immune organs (the spleen and thymus) of male ICR mice. The ICR mice were exposed to drinking-water containing PFOA at various doses for 21 days. The study showed that PFOA increased the expression of TNF- α , IL-1 β , and IL-6 in the spleen, which also showed enlargement, with hyperplasia of the white pulp in the groups at 250 ppm. No changes were observed in the thymus (Son et al., 2009).

<u>Guo et al. (2021b)</u> also investigated the potential of PFOA to induce splenic atrophy in male mice. Mice were exposed to different

Table 4.20 End-points relevant to chronic inflammation effects in non-human mammalian systems in vivo exposed to PFOA or PFOS

End-point, assay	Species, strain, sex	Tissue	Resultª	Dose (LED or HID) or range	Route, duration, dosing regimen	Comments	Reference
Pro-inflammatory effe	cts						
Chronic active periportal inflammation, by histology	Pregnant mouse, CD-1, female	Livers of offspring	↑ in offspring at PND91 and PND21, dose-related	PFOA, 0.01, 0.1, 0.3, or 1 mg/kg	Oral gavage from GD1 to GD17, daily	PFOA ammonium salt	<u>Quist et al.</u> (2015)
Chronic inflammation (active or focal), by histology	Rat, Sprague- Dawley, both sexes	Glandular stomach Liver	+ active inflammation in males after 16 weeks + focal inflammation in	PFOA, 0/150, 0/300 ppm (only postweaning exposure) in study 1 PFOA,	Feed, 16-week interim time point of 2-year feeding study (study 1) Feed, 2-year study	PFOA	<u>NTP (2020)</u>
			males after 2 years	0/40, 0/80, 300/80 ppm in study 2	(study 2)		
		Forestomach	+ active inflammation in females after 2 years	PFOA, 0/1000 ppm (only postweaning) or 300/1000 ppm (both exposures) in study 1			
Chronic inflammation, by histology	Pregnant mouse, CD-1, 129/SV WT, and 129/SV KO, female	Livers of offspring	+ only in CD-1 mice at 1 and 5 mg/kg bw	PFOA, 0.01–5 mg/kg per day	Oral gavage daily from GD1 to GD17	PFOA ammonium salt; dose- related increases in severity scores in PFOA- exposed livers	<u>Filgo et. al.</u> (2015)
Chronic inflammation, by histology	Mouse, Pdx-1 (KC), both sexes	Pancreas	(†) at 9-month time point	PFOA, 5 ppm	Drinking-water, from age 8 weeks until 6 or 9 months (time points of analysis)	Inconclusive evidence	<u>Kamendulis</u> <u>et al. (2022)</u>

IARC MONOGRAPHS - 135

Table 4.20 (continued)

	•						
End-point, assay	Species, strain, sex	Tissue	Resultª	Dose (LED or HID) or range	Route, duration, dosing regimen	Comments	Reference
Tnf, Il1b, Il6, by RT-PCR	Mouse, ICR, male	Spleen Thymus	↑ <i>Tnf, Il6</i> at 250 ppm ↑ <i>Il1b</i> at 50 and 250 ppm No changes	PFOA, 0, 2, 10, 50, or 250 ppm	Oral (drinking- water) for 21 days	PFOA ammonium salt; spleen hyperplasia	<u>Son et al.</u> (2009)
TNF-α, IL-1β, IL- 6, IRF5, IL12p70, macrophages, by ELISA	Mouse, BALB/c, male	Spleen and serum	$ \uparrow TNF-\alpha, IL-1\beta, IL-6, and IRF5 at 2 and 10 mg/kg per day (spleen and serum) \uparrow IL-1\beta, IL-6, and IL12p70 at \geq 0.4 mg/kg per day (serum) \uparrow TNF-\alpha at \geq 2 mg/kg per day (serum) \uparrow IL12p70 at 10 mg/kg per day (serum) \uparrow macrophage ratio at all doses with dose dependence $	PFOA, 0, 0.4, 2, or 10 mg/kg per day	Oral (gavage) for 28 days, daily		<u>Guo et al.</u> (<u>2021b)</u>
IL-6, TNF-α, IFN-γ, IL-4, by ELISA	Mouse, C57BL/6, male	Liver	↓ TNF-α, IFN-γ, IL-4, IL-6	PFOA, 0.002% w/w, 3 ± 0.7 mg/kg per day	Oral (dietary exposure), 10 days		<u>Qazi et al.</u> (2013)
IL-5, IL-13, IL-4, IL-12, IL-2, IFN-γ, IL- 17α, IL-1α, IL-6, and TNF-α, by Luminex	Mouse, B6C3F ₁ , female	Serum	↓ IL-5, IL-13, IL-12 at high dose ↓ IL-17α, IL-1α at both doses ↑ TNF-α at high dose No significant effects on IL-4, IL-2, IFN-γ, or IL-6	PFOA, 1.88 (low) or 7.5 (high) mg/kg per day	Oral (in drinking- water), for 4 weeks. 5 days prior the end of the exposure, one single intraperitoneal injection (300 mg/mouse in 0.5 mL) of keyhole limpet hemocyanin	Effects of PFOA alone not investigated	<u>De Guise</u> <u>and Levin</u> (2021)
IL-6, COX-2, CRP, by ELISA Inflammatory cells, by histology	Mouse, Kunming, male	Liver	 ↑ IL-6, COX-2, and CRP at 10 mg/kg per day ↓ IL-6, COX-2, CRP at 2.5 and 5 mg/kg per day ↑ infiltration of inflammatory cells (and hypertrophy), dose- dependent 	PFOA, 2.5, 5, or 10 mg/kg per day	Oral for 14 consecutive days		<u>Yang et al.</u> (<u>2014)</u>

556

End-point, assay	Species, strain, sex	Tissue	Result ^a	Dose (LED or HID) or range	Route, duration, dosing regimen	Comments	Reference
Liver inflammation, steatosis, IL-1β, by histology, immunohistochemistry, ELISA	Mouse, C57BL/6J, male	Liver and serum	↑ steatosis after both diets ↑ liver inflammation in HFD-fed mice ↑ IL-1β (serum) after both diets	PFOS, 5 mg/kg per day	Oral for 4 weeks; HFD or chow diet ± PFOS		<u>Qin et al.</u> (2022b)
<i>Tnf, Il1b, Cox2, Il10,</i> by qPCR	Mouse, C57BL/6J, male	Colon, serum, cerebral cortex, hippocampus	↑ <i>Tnf</i> and <i>Cox2</i> at 3 mg/kg ↑ <i>Il1</i> at 1 and 3 mg/kg ↓ <i>Il10</i> at 1 and 3 mg/kg in colon ↓ <i>Il10</i> and ↑ <i>Tnf</i> in serum ↑ <i>Tnf</i> at 1 mg/kg and <i>Cox2</i> at 1 and 3 mg/kg in cortex and at 3 mg/kg in hippocampus ↓ <i>Il10</i> in hippocampus at 3 mg/kg	PFOA, 0.5, 1, or 3 mg/kg per day	Oral for 5 weeks		<u>Shi et al.</u> (<u>2020)</u>
TNF-α, IL-1β, IL-6, macrophages (CD11b ⁺ cells), by ELISA, RT-PCR (cytokines); flow cytometry (CD11b)	Mouse, C57BL/6, male	Peritoneal cavity, spleen	↑ TNF-α and IL-6 at ≥ 25 mg/kg, ↑ IL-1β at ≥ 5 mg/kg ↑ % macrophages in peritoneal cavity at ≥ 1 mg/kg ↑ TNF-α and IL-6 at ≥ 125 mg/kg, ↑ IL-1β at ≥ 50 mg/kg ↑ macrophages at ≥ 50 mg/kg in spleen	PFOS, cumulative doses of 0.5, 1, 5, 25, or 50 mg/kg; max 2.08 mg/kg per day	Oral for 60 days	PFOS concentrations were measured in the serum PFOS was the potassium salt	<u>Dong et al.</u> (2012a)
TNF-α and IL-6, by ELISA	Mouse, B6C3F ₁ , female	Serum, peritoneal lavage fluid	Serum cytokines: ↓ TNF-α at 1 mg/kg (↑ at 300 mg/kg), ↑ IL-6 at 1 mg/kg (↓ at 3 mg/kg) Peritoneal macrophage cytokines: ↑ TNF-α at 300 mg/kg, no change in IL-6 Peritoneal lavage fluid: ↓ TNF-α and IL-6	PFOS, 1, 3, or 300 mg/kg total administered dose, corresponding to 0, 0.0331, 0.0993, or 9.93 mg/kg per day	Oral for 28 days	PFOS potassium salt	<u>Mollenhauer</u> <u>et al. (2011)</u>

Table 4.20 (continued)

	,						
End-point, assay	Species, strain, sex	Tissue	Resultª	Dose (LED or HID) or range	Route, duration, dosing regimen	Comments	Reference
Il1b, Tnf, Nfkb1, Creb1, AP-1(mRNA expression), by qPCR	Rat, Sprague- Dawley, pregnant female	Hippocampus, cortex	Hippocampus of offspring: \uparrow <i>Il1b</i> , \uparrow <i>Tnf</i> on PND0 and PN21 \uparrow AP-1 at all doses on PND0 and at 0.6 and 2 mg/kg per day on PND21 \uparrow <i>Nfkb1</i> at 0.6 and 2 mg/kg per day on PND0 and at 2 mg/kg on PND21 \uparrow <i>Creb1</i> at 0.6 and 2 mg/kg per day on PND0 and PND21 Cortex: \uparrow Il1b at 0.6 mg/kg per day on PND0 and PND21, at 2 mg/kg per day on PND21 \uparrow <i>Tnf</i> , AP-1, <i>Nfkb1</i> at 2 mg/kg per day on PND0 and PND21 \uparrow <i>CREB</i> at 0.6 and 2 mg/kg per day on PND0 and PND21	PFOS: 0.1, 0.6, or 2.0 mg/kg per day	Oral, from GD2 to GD21	PFOS potassium salt	Zeng et al. (2011)
TNF-α and TGFβ3, by ELISA, western blot	Mouse, BALB/c, male	Testicular homogenates	↑ TNF-α, TGFβ3	PFOA, 0–20 mg/kg per day	28 consecutive days		<u>Lu et al.</u> (2016c)
IL-6, IFN-γ, TNF-α, HGF, by ELISA	Mouse, C57BL/6J, male	Serum	↑ IL-6, IFN-γ, TNF-α, HGF	PFOA, 1, 5, 10, or 20 mg/kg per day	Oral gavage up to 28 days		<u>Soltani et al.</u> (2023)
IL-1 β , IL-6, TNF- α , by ELISA	Rat, Sprague- Dawley, female	Serum	↑ IL-6, TNF-α No change in IL-1β	PFOS, 0.3 mg/kg per day	Oral daily gavage of pregnant (GD1 to birth) and lactating (PND1 to PND21) mothers		<u>Liu et al.</u> (2023b)
<i>Tnf, Il6, Nfkb1,</i> and <i>Mapk8</i> gene expression, by qRT-PCR	Rat, Wistar, male	Liver	↑ <i>Tnf, Il6, Nfkb1,</i> and <i>Mapk8</i> gene expression	PFOA, 10 mg/kg per day	Oral daily gavage for 4 weeks		<u>Naderi et al.</u> (2023)

Table 4.20 (continued)									
End-point, assay	Species, strain, sex	Tissue	Result ^a	Dose (LED or HID) or range	Route, duration, dosing regimen	Comments	Reference		
Crp, TNF-α, IL-6, and <i>Nfkb1</i> gene expression, by RT-PCR, histopathology, ELISA	Rat, albino, male	Jejenum	↑ CRP, TNF-α, IL-6, <i>Nfkb1</i>	PFOS, 5 mg/kg per day	Oral daily gavage for 28 days		<u>Shalaby et al.</u> (2023)		
CRP, IL-6, COX-2, inflammatory cell infiltration, oedema, by ELISA, histopathology	Mouse, Kunming, male	Liver	↑ CRP, IL-6, COX- 2, inflammatory cell infiltration, oedema	PFOA, 10 mg/kg per day	Intragastrically administered for 14 days		<u>Zou et al.</u> (2015)		
Anti-inflammatory effe	cts								
Oedema, dorsal-to-ventral paw thickness measured with micro-callipers	Rat, Sprague- Dawley, male	Paw	↓ oedema	PFOA, 20, 50, or 100 mg/kg	Injection into paw 30 min before induction of inflammation	Carrageenan- induced oedema; dose-response relation	<u>Taylor et al.</u> (<u>2002)</u>		
Leukocyte activation, PGE2 synthesis, by ELISA	Rat, Sprague- Dawley, female	Serum	\downarrow leukocytes and PGE2	PFOA, 150 mg/kg	SC	Acute pancreatitis induced by	<u>Griesbacher</u> et al. (2008)		
Inflammatory oedema formation, by histology		Pancreas	No change			cerulein			

AP, activator protein; bw, body weight; COX, cyclooxygenase; CRP, C-reactive protein; ELISA, enzyme-linked immunosorbent assay; GD, gestational day; HFD, high-fat diet; HID, highest effective dose; IFN, interferon; IL, interleukin; IRF, interferon regulatory factor; KO, knock-out; LED, lowest effective dose; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PG, prostaglandin; PND, postnatal day; ppm, parts per million; qRT-PCR, quantitative reverse transcription-polymerase chain reaction; SC, subcutaneous; TGF, transforming growth factor; TNF, tumour necrosis factor; WT, wildtype; w/w, weight per weight.

 a $\uparrow,$ increase; $\downarrow,$ decrease; ($\uparrow),$ decrease in a study with limitations; +, occurrence.

concentrations of PFOA (0, 0.4, 2, or 10 mg/kg per day) for 28 days. PFOA induced splenic atrophy and increased levels of TNF- α , IL-1 β , IL-6, and IRF5 in the spleen at the two highest doses (2 and 10 mg/kg). In the serum, levels of the cytokines IL-1 β , IL-6, and IL12p70 were increased by 0.4 mg/kg at higher doses, whereas TNF- α was significantly increased at doses \geq 2 mg/kg. In addition, dose-dependent increases in activated macrophages were found in the spleen (Guo et al., 2021b).

The effects of exposure to PFOA on immune-mediated liver damage in mice was assessed by Qazi et al. (2013). PFOA exposure at 3 ± 0.7 mg/kg per day for 10 days was associated with decreased hepatic levels of IL-6, TNF- α , IFN- γ , and IL-4, and PFOA caused marked hypertrophy of hepatocytes.

In the study conducted by <u>De Guise and</u> <u>Levin (2021)</u>, an investigation of levels of pro-inflammatory markers (IL-5, IL-13, IL-4, IL-12, IL-2, IFN- γ , IL-17 α , IL-1 α , IL-6, and TNF- α) was performed in mice exposed to low (1.88 mg/kg per day) or high (7.5 mg/kg per day) doses of PFOA in drinking-water for 4 weeks. The results showed a significant reduction in the serum concentrations of the Th2 cytokines IL-5 and IL-13, the Th1 cytokine IL-12, a non-significant dose-dependent increase in IL-2, and an increase in IFN- γ at the highest PFOA exposure. Both doses of PFOA reduced IL-17 α and IL-1 α , whereas TNF- α was increased at the highest dose of PFOA (<u>De Guise and Levin, 2021</u>).

The hepatic effects of PFOA exposure were also assessed in a study of mice orally dosed with different concentrations of PFOA for 14 days (<u>Yang et al., 2014</u>). The results showed that only high doses of PFOA (10 mg/kg per day) were associated with higher levels of IL-6, COX-2, and CRP (<u>Yang et al., 2014</u>). In addition, hepatic hypertrophy was induced, as indicated by a significant increase in relative liver weight; in addition, a significant increase in inflammatory cells was observed on histological examination.

Qin et al. (2022b) investigated the effects of chronic PFOS exposure (5 mg/kg per day) on liver disease progression by exploring the initiation of NLRP3 inflammasome activation and its potential role in liver inflammation. The study assessed several end-points, including the activation of liver inflammation and steatosis, as well as the release of the pro-inflammatory cytokine IL-1 β under a high-fat diet (HFD) or chow diet. The study reported significant findings, including the activation of the NLRP3 inflammasome, an increase in IL-1ß secretion, and hepatocyte steatosis in both chow- and HFD-fed animals exposed to PFOS, but exacerbation of liver inflammation around the portal vein and steatosis was also observed in HFD-fed mice exposed to PFOS (Qin et al., 2022b).

Shi et al. (2020) investigated the potential effects of PFOA exposure on the gut and brain in male C57BL/6J mice. Mice were exposed to different concentrations of PFOA (0, 0.5, 1, or 3 mg/kg bw per day) for 35 days. PFOA exposure resulted in increased TNF- α , IL-1 β , and COX-2 and decreased IL-10 levels in the colon. Different results were observed in serum and the brain (see Table 4.20).

Dong et al. (2012a) investigated the effect of 60-day oral exposure to the potassium salt of PFOS on the inflammatory response in peritoneal macrophages, splenic macrophages, and serum of adult male C57BL/6 mice. The mice were dosed daily by oral gavage with PFOS at various doses, ranging from 0.0083 to 2.0833 mg/kg per day, to achieve total administered doses (TADs) of 0.5 to 125 mg/kg over the treatment period. The study reported that PFOS exposure significantly increased the percentage of peritoneal macrophages (CD11b+ cells) at concentrations of $\geq 1 \text{ mg/kg TAD}$, and PFOS exposure elevated the ex vivo production of IL-1 β in the peritoneal cavity at concentrations of \geq 5 mg/kg TAD. Furthermore, PFOS exposure markedly enhanced the ex vivo production of TNF- α , IL-1 β , and IL-6 by peritoneal and splenic

macrophages when stimulated with LPS. In addition, the serum levels of TNF- α , IL-1 β , and IL-6 were also significantly elevated in response to LPS stimulation (Dong et al., 2012a).

Mollenhauer et al. (2011) investigated the effects of PFOS exposure on the inflammatory markers TNF-a and IL-6 in both serum and peritoneal lavage fluid in adult female B6C3F₁ mice. The mice were orally exposed to PFOS at 0, 1, 3, or 300 mg/kg TAD, corresponding to daily doses of 0, 0.0331, 0.0993, or 9.93 mg/kg per day, for 28 days. The study reported that the serum TNF-α levels were significantly decreased by PFOS at 1 mg/kg TAD compared with controls, whereas the serum IL-6 levels were increased. The IL-6 concentrations in peritoneal lavage fluid decreased with increasing PFOS dose. Additionally, the number of splenocytes expressing intracellular IL-6 was significantly decreased in the 3 mg/kg treatment group compared with controls (Mollenhauer <u>et al., 2011</u>).

The pro-inflammatory effects of PFOS on glial activation in the hippocampus and cortex were investigated in a study of rat offspring. The dams received different doses of PFOS via gavage (0.1, 0.6, or 2.0 mg/kg bw) from GD2 to GD21. The study reported increased levels of IL-1 β and TNF- α , and elevated mRNA levels of *Jun*, *Nfkb1*, and *Creb1* in the hippocampus at lower doses than in the cortex at PND0 or PND21 (Zeng et al., 2011).

Lu et al. (2016c) investigated the effect of PFOA exposure (0–20 mg/kg per day) on the cytokines TNF- α and TGF β 3 in BALB/c male mice. The authors observed that PFOA increased the TNF- α and TGF β 3-based inflammatory response in testes.

Soltani et al. (2023) investigated the effects of PFOA in C57BL/6 J male mice exposed at 1, 5, 10, or 20 mg/kg. The study reported that PFOA increased the circulating levels of cytokines such as IL-6, IFN- γ , TNF- α , and HGF.

Liu et al. (2023b) investigated the effects of PFOS exposure during pregnancy and lactation

in rats. The study reported that PFOS increased the levels of circulating TNF- α and IL-6 on PND14 in the pups.

Naderi et al. (2023) investigated the effects of PFOA exposure on the expression of *Tnfa*, *Il6*, *Nfkb1*, and JNK genes in Wistar male rats exposed to 10 mg/kg. The study reported that PFOA increased the expression of *Il6*, *Tnfa*, *Nfkb1*, and JNK genes in the liver.

Shalaby et al. (2023) investigated *Crp*, *Tnfa*, *Il6*, and *Nfkb1* gene expression in response to PFOS exposure at a dose of 5 mg/kg per day. The study showed increased expression of *Crp*, *Tnfa*, *Il6*, and *Nfkb1* genes.

Zou et al. (2015) investigated the potential of quercetin to offset the adverse effects of PFOA in mice treated intragastrically with PFOA at a dose of 10 mg/kg per day alone or in combination with quercetin at a dose of 75 mg/kg per day for 14 consecutive days. The study showed that PFOA changes the liver morphology by disrupting liver architecture and inducing marked oedema, vacuolar degeneration, hepatocellular necrosis, and inflammatory cell infiltration. In addition, PFOA significantly increased levels of CRP, IL-6, and COX-2, but supplementation with quercetin reduced these levels (Zou et al., 2015).

Overall, when assessing data from experimental animal models, the aggregated findings from the evaluated studies displayed a mixture of outcomes. In a broader context, there is a notable tendency towards a positive correlation of PFAS exposure with chronic inflammation, as indicated by the histological data, and this connection is particularly robust for PFOA, in contrast to PFOS.

Anti-inflammatory effects

See <u>Table 4.20</u>.

Taylor et al. (2002) investigated the effects of PFOA on the inflammation and hyperalgesia induced by intraplantar injection of carrageenan in male Sprague-Dawley rats, with the aim of better understanding the contribution

End-point	Assay	Tissue, cell line	Result ^a	Concentrations, range	Reference
TNF-α, IL-1β, IL-6, and IL-8	ELISA, western blot	Rat RBL-2H3 mast cells	↑ IL-1 β , IL-6, and IL-8 at 100 μ M ↑ TNF- α , IL-1 β , IL-6, and IL-8 ≥ 10 μ M in IgE- stimulated cells	PFOA, 10, 50, 100, or 500 μM	<u>Lee et al.</u> (2017)
TNF-α, IL-6	ELISA	Rat Kupffer cells and primary hepatocytes	↑ TNF-α, IL-6 in Kupffer cells at all time points No change in hepatocytes	PFOS, 100 μM for 0 or 48 h	<u>Han et al.</u> (2018b)
TNF-α	ELISA	Rat C6 glioma cells	↑ TNF-α	PFOS, 0, 0.1, 1, 5, 10, 20, 50, or 100 nM for 12 h or 20 nM for 0, 1, 3, 6, 12, 24, or 48 h	<u>Chen et al.</u> (2018b)
TNF-a	ELISA	Rat HAPI microglial cells	\uparrow TNF- α > 1 nM and following 3 h of exposure	PFOS, 0–200 nM for 6 h or 20 nM for 0, 1, 3, 6, 8, 12 or 24 h. 0.1% DMSO as the control	<u>Yang et al.</u> (2015)

Table 4.21 End-points relevant to chronic inflammation in non-human mammalian cells in vitro exposed to PFOA or PFOS

DMSO, dimethyl sulfoxide; ELISA, enzyme-linked immunosorbent assay; h, hour(s); IgE, immunoglobulin E; IL, interleukin; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; TNF, tumour necrosis factor.

^a ↑, increase.

of PPARs to the inflammatory process. The oedema (swelling) and thermal hypersensitivity were assessed by measuring the dorsal-to-ventral paw thickness. The results showed that PFOA inhibited carrageenan-induced oedema and thermal hypersensitivity. The authors concluded that PFOA exhibits anti-inflammatory and/or anti-hyperalgesic effects in vivo by interfering with the initiation of inflammation (Taylor et al., 2002). In another study, the potential anti-inflammatory effects of PFOA in cerulein-induced acute pancreatitis in female rats was investigated (Griesbacher et al., 2008). The study showed that PFOA significantly reduced leukocyte accumulation and prostanoid synthesis, and thereby displayed a potential for anti-inflammatory effects in cerulein-induced acute pancreatitis (Griesbacher et al., 2008).

(ii) Non-human mammalian systems in vitro See Table 4.21.

The immunological effects of PFOA and PFOS exposure on macrophages via sirtuin regulation

was assessed using the murine macrophage cell line RAW 264.7 (Park et al., 2019). The results showed that both PFOA and PFOS had effects on sirtuins. More specifically, PFOA reduced *Sirt7* and *Sirt4* expression at 0.5 μ M, but increased *Sirt1*, *Sirt3*, *Sirt4*, *Sirt5*, and *Sirt6* gene expression at 5 μ M. In contrast, PFOS increased *Sirt2*, *Sirt3*, *Sirt5*, and *Sirt6* gene expression (Park et al., 2019).

Lee et al. (2017) investigated the correlation between PFOA exposure and allergic inflammation by examining its effect on RBL-2H3 mast cell degranulation and allergic symptoms. The results showed that PFOA activated NF- κ B, leading to enhanced expression of TNF- α , IL-1 β , IL-6, and IL-8 in IgE-stimulated mast cells (Lee et al., 2017).

Han et al. (2018b) investigated the effects of PFOS exposure on the inflammatory responses of Kupffer cells (liver macrophages). Rat Kupffer cells and primary hepatocytes were exposed to 100 μ M PFOS for various time periods (0–48 hours). The study demonstrated that PFOS

induced the production of TNF- α and IL-6 in Kupffer cells (<u>Han et al., 2018b</u>).

<u>Chen et al. (2018b)</u> investigated the effects of PFOS exposure on the TNF- α inflammatory response in rat C6 glioma cells. The C6 cells were exposed to different concentrations (0–100 nM) of PFOS for 12 h or 20 nM PFOS for different periods of time (0, 1, 3, 6, 12, 24, or 48 hours). The study reported that PFOS increased the TNF- α inflammatory response (<u>Chen et al., 2018b</u>).

Yang et al. (2015) investigated the effect of PFOS exposure on the TNF- α inflammatory response in rat HAPI microglia. The cells were treated with PFOS at different concentrations for 6 hours (0–200 nM) or at 20 nM for 0, 1, 3, 6, 8, 12, or 24 hours. The study reported that PFOS increased the TNF- α inflammatory response at \geq 5 nM for 6 hours and at 20 nM after 3 hours or longer (Yang et al., 2015). Overall, exposure to PFOA and PFOS consistently increased chronic inflammatory responses, as demonstrated by increased IL-6 and TNF- α , in various cell types in vitro.

Synopsis

[Overall, the Working Group noted that the findings of studies of human populations, both cross-sectional and longitudinal, that examined inflammatory markers (cytokines and proteins), neutrophil and leukocyte counts, and inflammatory disease outcomes (eczema, atopic dermatitis, asthma, and allergic rhinitis) in relation to PFOA and PFOS exposure yielded mixed results. Moreover, the associations varied by the biomarker investigated, sex, and exposure source, highlighting the complexity of these relationships. The evidence is not informative enough to conclude that PFOA and PFOS induce chronic inflammation in exposed humans.

In human primary cell studies, a trend emerged for decreases in the levels of chronic inflammatory markers in response to PFOA and PFOS treatment, although the outcomes varied depending on the specific model and assay used. In experimental systems in vivo, the findings were also mixed, but in a broader context, there was a discernible inclination towards a positive association with chronic inflammation, which was more pronounced for PFOA than for PFOS. While the evidence from the in vitro studies is clearer, the results from human and experimental systems in vivo collectively support a nuanced relationship of PFOA and PFOS exposure with chronic inflammation.]

4.2.7 Is immunosuppressive

See <u>Table 4.22</u>.

- (a) Humans
- (i) Exposed humans

The evaluation of the evidence on whether PFOA and PFOS exhibit the key characteristic of carcinogens "is immunosuppressive" was prompted by observations of clinical end-points related to immunosuppression (e.g. by infections), the effects on related immune end-points (vaccination, immune cell subpopulation monitoring, performance of functional assays on immune cells collected from exposed humans), or the measurement of biomarkers in fluids from exposed humans (cytokines or other mediators of inflammation). The most informative and relevant studies are reported in <u>Table 4.22</u>.

The Working Group first described the clinical outcomes observed in humans, mostly children exposed to the two agents. The strengths of the studies were: serum levels representing combined exposure from all sources and routes, including contributions from metabolites of precursors, the monitoring of diverse types of infections (respiratory tract and digestive tract), and/or responses to diverse vaccine antigens (tetanus, diphtheria, influenza, COVID-19, etc.), and the combination of longitudinal and cross-sectional studies including a large number of individuals. Confounding factors, such as smoking, BMI, maternal education, sex of the

10016 4.22	able 4.22 End-points relevant to minimunosuppression in numans exposed to FTOA of FTO5										
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference				
Infectious diseases	Maternal plasma	Hokkaido, Japan Study on the environment and children's health to age 4 years Cross-sectional study	PFOS, 4.92 ng/mL; PFOA, 2.01 ng/mL (median) 1558 mother–child pairs	+ (increased OR of total infectious diseases for PFOS) Q4 versus Q1, OR, 1.61; 95% CI, 1.18–2.21; <i>P</i> for trend, 0.008	Maternal age, number of older siblings, maternal smoking during pregnancy, maternal education, infant sex, breastfeeding period	Cross-sectional study with a large number of mother-child pairs recruited. Limitations were: infectious disease incidences were based on maternal reports that were not corroborated by medical records. No studies on the validity of self-reported physician- diagnosed infections. These could have resulted in some level of outcome misclassification.	<u>Goudarzi</u> <u>et al.</u> (2017a)				
Infectious diseases	Maternal plasma	Hokkaido, Japan Cohort for the study of the environment and children's health Prospective birth cohort with 7 years of follow-up	PFOS, 5.12 ng/mL; PFOA, 1.94 ng/mL (median) 2689 mother-child pairs; finally analysed 2206 children	+ PFOA, increased risk of pneumonia: OR, 1.17 (95% CI, 1.01–1.37); $P = 0.043$ for the children as a whole ($n = 2689$) RSV infection: OR, 1.58 (95% CI, 1.13–2.22); P = 0.008, P for trend = 0.038, for children with no siblings ($n = 379$) PFOS was inversely associated with increased risks of pneumonia and RSV infection	Sex, parity, maternal age at delivery, maternal smoking during pregnancy, BMI pre-pregnancy, annual household income during pregnancy	Longitudinal birth cohort design with a large sample size. Limitations were: loss to follow-up, reliance on questionnaire-based outcomes, possibility of recall bias affecting, health outcomes recorded using parental self- reported questionnaires.	<u>Ait-Bamai</u> <u>et al.</u> (2020)				

Table 4.22 End-points relevant to immunosuppression in humans exposed to PFOA or PFOS

	(,						
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Respiratory tract infection	Fetal umbilical cord blood	Oslo, Norway Environment and Childhood Asthma study Prospective birth cohort with 2–10 years of follow-up	PFOS, 5.2 ng/mL; PFOA, 1.6 ng/mL (median) <i>n</i> = 641	+ positive associations of the cord blood concentrations of PFOS and PFOA with airway infections; significant association of lower respiratory tract infection in infants aged 0–10 years with PFOS (β = 0.50; 95% CI, 0.42–0.57) and PFOA (β = 0.28; 95% CI, 0.22–0.35)	Confounders examined were sex; birth weight; birth month; breastfeeding at age 6 and 12 months; maternal smoking during pregnancy; household smoking at birth, at preschool age; parental asthma, atopic dermatitis and allergic rhinitis; parental education; and household income	Strengths of the study: high follow-up rate at 10 years; highly representative study sample; prospective longitudinal design, extensive characterization of the children from birth to age 2 years, as well as at 10 years; use of questionnaires, structured interviews, and clinical examinations of the children. Limitations: potential recall bias, especially in questionnaires and interviews, lack of data on important confounding factors, such as parity and previous breastfeeding.	<u>Impinen</u> <u>et al.</u> (2018)

Table 4.22 (continued)

_
\rightarrow
_
∇
\sim
()
_
<
_
\cap
\simeq
~
()
ž
)
_
2
\geq
_
-0
_
S
· ·
•
ŝ
(F

Table 4.22	(continued)						
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Infectious diseases	Maternal plasma	Norway, MoBA nationwide cohort. Prospective population- based pregnancy sub-cohort with 7 years of follow-up	PFOS, 12.9 ng/mL, PFOA, 2.5 ng/mL (medians) n = 1943 n = 1270 (questionnaire after 3 years) n = 972 (questionnaire after 7 years)	+ positive associations between bronchitis and/or pneumonia with PFOS (RR, 1.20;95% CI, 1.07–1.34; P = 0.001) and PFOA (RR, 1.27, 95% CI, 1.12–1.43; $P < 0.001$) (age 0–3 years) Positive association between viral pseudocroup and PFOA (RR, 1.22; 95% CI, 1.07–1.38; P = 0.002) (age 0–3 years) Positive association between gastric flu/ diarrhoea and PFOA (RR, 1.48; 95% CI, 1.31–1.67; $P < 0.001$) (age 6–7 years)	Maternal age, maternal BMI, maternal education, parity, and smoking during the pregnancy Correction for multiple testing using false discovery rate	Longitudinal prospective cohort design with 7 years of follow-up, sample size was relatively large. All samples had PFOS and PFOA concentrations above the LOQ. Loss to follow-up at 7 years and reliance on questionnaire-based outcomes. Possible that a single illness episode could be reported with more than one of the symptoms of infection, making the number of episodes more uncertain.	<u>Impinen</u> <u>et al.</u> (2019)
Table 4.22	(continued)						
---	-------------------------------	---	--	--	---	---	---
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Infectious diseases	Maternal plasma	Odense, Denmark Odense Child Cohort; data collected 2010–2012 Prospective cohort	PFOS, 7.52 ng/mL; PFOA, 1.68 ng/mL (medians) 1503 mother-child pairs	+ PFOS doubling in mothers associated (adjusted HR, 1.23) with a 23% increase in the risk of hospitalization because of any infection + PFOA and PFOS doubling increased the risk of LRTI by 27% (adjusted HR, 1.27; 95% CI, 1.01–1.59) and 54% (adjusted HR, 1.54; 95% CI, 1.11–2.15), respectively	Maternal educational, maternal, parity, BMI, smoking during pregnancy	Prospective cohort study in 1503 mother-child pairs. Estimation of the number of severe infections may be imprecise. Analyses of some specific types of infections based on a small number of events (gastrointestinal tract, 40 events).	<u>Dalsager</u> <u>et al.</u> (2021)
Response to vaccination (tetanus and diphtheria)	Maternal and infant plasma	Faroe Islands, Denmark National Hospital Faroe Island Birth cohort Prospective study	Maternal PFOS GM, 27.3 ng/mL; age 5 years children PFOS GM, 16.7 ng/mL Maternal PFOA GM, 3.20 ng/mL; age 5 years children PFOS GM, 4.06 ng/mL 656 recruited 1997–2000; <i>n</i> = 587, follow-up to 2008	+ Prenatal exposure to PFOS and PFOA negatively associated with anti- diphtheria antibody concentrations, at age 5 and 7 years for PFOS and 7 years for PFOA - (negative) association between anti-tetanus titre and PFOA at age 7 years	Pairwise; possible effect of PCB exposure, birth weight, maternal smoking during pregnancy, duration of breastfeeding	Prospective cohort study addressing the response to vaccination as the end- point. Average sample size. Closed population studied (fishing community in the Faroe Islands).	<u>Grandjean</u> <u>et al.</u> (2012)

Table 4.22 (continued)

Table 4.22	continued)							
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response	Covariates adjusted for	Comments	Reference	
Response to vaccination (tetanus, diphtheria)	Infant blood collected at birth and at ages 18 months and 5 years (clinical examinations)	Faroe Islands, Denmark Birth Cohort 5, born in 2007–2009 Prospective birth cohort	Age 18 months, PFOS GM, 7.1 ng/mL; age 5 years, PFOS GM, 4.7 ng/mL Age 18 months, PFOA GM, 2.8 ng/mL; age 5 years, PFOA GM, 2.2 ng/mL Total, $n = 490$; age 18 months, $n = 275$; and age 5 years, n = 349	+ (at age 5 years, 152 (44%) children had antibody concentrations lower than the protective level of 0.1 IU/ mL for diphtheria and 126 (36%) for tetanus)	Breastfeeding duration, maternal serum PCB concentrations, caesarean section; others not mentioned	Clear vaccination schema, prospective study with follow-up.	Grandjean et al. (2017)	
Response to vaccination (<i>Haemophilus</i> <i>influenzae</i> , diphtheria, tetanus) and IFN-γ production by PBMCs	Infant blood collected at age 1 year	Berlin, Germany Cross-sectional study	Data expressed as means Formula-fed children, PFOS, 6.8 ± 3.4 ng/mL; PFOA, 3.8 ± 1.1 ng/mL Breastfed children, PFOS, 15.2 ± 6.9 ng/mL; PFOA, 16.8 ± 6.6 ng/mL 101 healthy children age 1 year (21 formula-fed, 80 breastfed for at least 4 months)	+ PFOA associations of adjusted PFOA level with antibody levels for Hib, tetanus IgG1, and diphtheria, $P < 0.05$) No significant associations of PFOS with Hib, or tetanus or diphtheria antibodies + association of PFOA level with a lower production of IFN- γ by PBMCs stimulated with tetanus and diphtheria toxoid found in 55 children) No effect of PFOA or PFOS on infections during the first year of life	Children (age 341–369 days) of German parents, either breastfed for a maximum of 2 weeks ("formula-fed" children) or breastfed (exclusively breastfed for ≥ 4 months) Healthy infants with no acute or chronic diseases, including atopic eczema, no medication, vaccinated	Strengths: children breastfed for a long time with relatively high levels of internal exposure to persistent organic pollutants. High stability of the associations identified using different methods of evaluation, broad spectrum of other contaminants also measured as possible confounders. Limitation: small sample size: 101 children (51 boys, 50 girls) examined; 21 formula-fed, and 80 breastfed for at least 4 months.	<u>Abraham</u> <u>et al.</u> (2020)	

_

Table 4.22	(continued)						
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Response to vaccination (rubella)	Blood from participants	US NHANES for years 1999–2000 and 2003–2004 Cross-sectional study	Mean PFOA concentration 6 ng/mL in men, 4.3 ng/mL in women, 4.8 ng/mL in youth Mean PFOS concentrations were 28.1 ng/mL, 22.1 ng/mL, and 25.1 ng/mL, respectively; 581 women, 621 men, and 1012 young participants (aged 12–18 years)	+ associations of both PFOA (<i>P</i> = 0.0016) and PFOS (<i>P</i> = 0.0295) quartiles with rubella titres after adjusting for covariates	Age, race/ethnicity, educational level (high school), BMI, parity, live births	Limitations: cross- sectional study, temporal nature of the associations between PFAS and humoral immune response could not be determined. No young children in the cohort. No information on vaccination status or recent rubella infection. No information on age when vaccination was performed, other infections, or variation in vaccination rates and schedules.	Pilkerton et al. (2018)

Table 4.22 (continued)

Table 4.22	(continued))					
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response	Covariates adjusted for	Comments	Reference
Response to vaccination (mumps, rubella, and measles antibodies)	Blood from participants	US NHANES 1999–2000 and 2003–2004 Cross-sectional study	GM PFOS, 20.8 ng/mL; PFOA, 4.13 ng/mL; 1191 children	 + (doubling of PFOS associated with a 7.4% decrease in mumps antibodies; doubling of PFOS and PFOA associated with 5.9% and 6.6% decreases in mumps antibodies if analysis restricted to seropositive children) + (doubling of PFOS was associated with a 13.3% decrease in rubella antibodies; for PFOA decrease not significant) No associations between PFOS or PFOA with measles antibody level 	No information on vaccination status in NHANES; however, in early 1990s, most US states have required a measles/mumps/ rubella booster before starting school; parity	Limitations: possible exposure misclassification, although less risk than in a typical cross- sectional study, because these compounds have long serum half-lives. Lack of information on receipt and/or timing of measles/mumps/rubella vaccination.	<u>Stein et al.</u> (2016)

10016 4.22	(continueu)						
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Response to vaccination (rubella, measles, and mumps antibodies)	Blood from participants	US NHANES 2003–2004 and 2009–2010 cycles Cross-sectional study	GM PFOA in entire sample, 3.33 ng/mL; 3.42 ng/mL in those with lower folate; 3.15 ng/mL in those with higher folate PFOS in entire sample, 12.44 ng/mL; 12.88 ng/mL in those with lower folate; 11.57 ng/mL for those with higher folate 819 adolescents aged 12–19 years	+ inverse associations of serum PFOS (P = 0.22) and PFOA (P = 0.03) with rubella antibodies, and between PFOA (P = 0.01) and mumps antibodies, only in adolescents with blood folate concentrations < 66th percentile (lower folate group) No associations in adolescents with higher RBC folate levels (higher folate group)	Demographic characteristics, including age (continuous), sex (dichotomous), race, household income, using self-reported questionnaires; income-to-poverty ratio, weight, height, dietary information	NHANES study collected high quality data, good sample size. Seropositivity for rubella and measles was used as a proxy for rubella/measles/mumps vaccination, because of lack of information on vaccination or boosting for the study sample.	Zhang <u>et al.</u> (2023d)

Table 4.22 (continued)

Table 4.22	(continued)						
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Response to vaccination (influenza virus, A/H3N2 strain)	Blood from participants	Mid-Ohio Valley region of Ohio and West Virginia, USA Cross-sectional study (pre-post measurements)	PFOA: first quartile: 0.25-13.7 ng/mL; second quartile: 13.8-31.5 ng/mL; third quartile: 31.6-90 ng/mL; fourth quartile: 90.4-2140 ng/mL. PFOS: first quartile: 0.1-5.8 ng/mL; second quartile: 5.9-9.2 ng/mL; third quartile: $9.3-14.5 \text{ ng/mL}$; fourth quartile: 14.7-42.3 ng/mL n = 411	+ PFOA (serum concentrations between 13.7 and 90 ng/mL) associated with a reduced response to A/H3N2 influenza virus + PFOA associated with an increased risk of not attaining the antibody threshold considered to offer long-term protection) - (no evidence that PFOS serum concentration was associated with reduced vaccine responses; no associations of self- reported colds or influenza with PFAA concentrations)	Age, medical comorbidities, and medications	Serum samples collected pre-vaccination and 21 ± 3 days post- vaccination in 2010. Number of participants was only 411. Analysis revealed a strong effect of previous influenza vaccination on the immune response, based on self-reported information.	Looker et al. (2014)

Table 4.22	(continued)						
End-point	Biosample type	Location Setting, study design	Exposure level No. of study participants	Response ^a	Covariates adjusted for	Comments	Reference
Response to vaccination against COVID-19	Blood from participants	Ronneby, Sweden Adults (age 20–60 years) from the Biomarker Cohort and a background- exposure group from Karlshamn Observational study	Ronneby group, median PFOS serum level, 47 ng/mL (adults, <i>n</i> = 309) Background group, median exposure PFOS serum level, 4 ng/mL (<i>n</i> = 47) 2 doses of mRNA vaccine Spikevax	– (PFOA and PFOS levels were not associated with antibody levels after 5 weeks and 6 months)	Sex, age, smoking Exclusion of individuals with previous SARS-CoV- 2-infection Others mentioned in the questionnaire, but analysis not provided	PFAS levels were measured before vaccination and 5 weeks (n = 350) and 6 months (n = 329) after the second vaccine dose. Prospective design with measurement of antibody responses at defined time points after vaccination. Number of participants used to assess background exposure was small. Possibility that there was an effect of PFAS within the background exposure levels cannot be excluded.	Andersson et al. (2023)
Response to vaccination against COVID-19	Blood from participants	Factories in Alabama and Wisconsin, USA Employees and retirees from two facilities (Spring 2021) Observational study	415 participants (757 observations, repeated measures analyses)	+ fully adjusted IgG concentration was -3.45% (95% CI, -7.03 to 0.26%) per 14.5 ng/mL IQR increase in PFOS + For PFOA, -1.95% (95% CI, -4.35 to 0.51) per 3.59 ng/mL IQR increase in PFOA	Age, sex, race/ ethnicity, site, BMI, smoking, immunocompromising conditions Antigenic stimulus group and time since last antigenic stimulus used as precision variables	SARS-CoV-2 neutralizing antibodies were measured to assess the polyclonal response to SARS-CoV-2. Wide distribution of serum concentrations of PFOS and PFOA (increased statistical power to detect an association). Potential longer-term associations with PFAS not assessed.	Porter et al. (2022)

β, adjusted Poisson regression estimate; BMI, body mass index; CI, confidence interval; COVID-19, disease caused by SARS-CoV-2-infection; GM, geometric mean; Hib, *Haemophilus influenzae* type b; HR, hazard ratio; IFN-γ, interferon-gamma; IgG, immunoglobulin G; IQR, interquartile range; IU, international unit; LOQ, limit of quantification; LRTI, lower respiratory tract infection; MoBA, Norwegian Mother, Father and Child Cohort Study; NHANES, National Health and Nutrition Examination Survey; OR, odds ratio;
 PBMCs, peripheral blood mononuclear cells; PCB, polychlorinated biphenyl; PFAA, perfluoroalkyl acids; PFAS, perfluoroalkyl and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; Q, quartile; RR, Poisson regression rate ratio; RSV, respiratory syncytial virus; SARS-CoV-2, severe acute respiratory syndrome coronavirus 2; SD, standard deviation; US, United States; USA, United States of America.

^a +, positive; -, negative; +/-, equivocal (variable response in several experiments within an adequate study).

infant, breastfeeding period, and others were considered in the analyses. The weaknesses of the studies were mainly restricted to the less detailed self-reported questionnaires used in some of the studies.

Increased risk of infections

The Hokkaido Study on the relation between environment and children's health recruited 20 787 pregnant women from February 2003 to March 2012. Using this cohort, Goudarzi et al. (2017a) examined the relation between prenatal exposure to various perfluoroalkyl acids (PFAA), including PFOS, and the prevalence of infectious diseases in children aged up to 4 years (Goudarzi et al., 2017a). A total of 1558 mother-child pairs were included in this data analysis. PFAA concentrations were measured in maternal plasma samples collected at 28-32 weeks of gestation. Medical history of common infectious diseases, based on physicians' diagnoses, was extracted from the questionnaires completed by the mothers. The median serum concentrations were 4.92 ng/mL for PFOS and 2.01 ng/mL for PFOA, and 67.1% of the children (n = 1046) were diagnosed with at least one of the diseases defined as infectious. After adjusting for appropriate confounders (maternal age, number of older siblings, maternal smoking during pregnancy, maternal education, infant sex, and breastfeeding period), the highest quartile of PFOS concentration was associated with an increased OR for infectious diseases as a whole in all the children. No associations between infectious diseases and the other PFAAs examined, including PFOA, were found (Goudarzi et al., 2017a).

[The Working Group noted that prenatal exposure to PFOS seemed to be associated with infectious disease occurrence in early life. As in other studies, the classification of infectious diseases was based on maternal reports that were not corroborated by medical records. No studies on the validity of self-reported physician-diagnosed infections were conducted.]

In a subsequent study conducted on the same Hokkaido cohort, Ait Bamai et al. (2020) investigated the effects of prenatal exposure to PFAS on the prevalence of infectious diseases in children aged up to 7 years (Ait Bamai et al., 2020). Of the children diagnosed with infectious diseases between ages 2 and 4 years, the analysis was limited to 2206 children (out of a total 2689) for whom data on PFAS measurements and questionnaires, administered to mothers during the first trimester of pregnancy and when children were 7 years old, were available. Median concentrations in the blood were 1.94 ng/mL for PFOA and 5.12 ng/mL for PFOS. Prenatal exposure to PFOA was associated with increased risk of infections in all the children (n = 2689) and with increased risk of respiratory syncytial virus (RSV) infection among children without any siblings (n = 379), and PFOS was inversely associated with increased risks of pneumonia and RSV infection.

In the study by Impinen et al. (2018), the objective was to determine whether prenatal exposure to PFAS was associated with asthma or other allergic diseases, or respiratory tract infections, in childhood. PFAS were measured in cord blood from 641 infants in the ECA prospective birth cohort study cohort in Oslo, Norway. The results showed that reported airway infections were significantly associated with cord blood concentrations of PFOA and PFOS (median concentrations of PFOA and PFOS were 1.6 and 5.2 ng/mL, respectively) and, specifically, that LRTIs in infants aged 0-10 years were associated with PFOA $(\beta = 0.28; 95\% \text{ CI}, 0.22-0.35)$ and PFOS $(\beta = 0.50;$ 95% CI, 0.42-0.57). Associations were also found for perfluorooctanesulfonamide, perfluorononanoic acid (PFNA), and perfluoroundecanoic acid. [The Working Group considered that, although it is not known how cord blood concentration is linked to overall future exposure, it represents a good marker of gestational exposure. These observations suggested that PFOA and PFOS were associated with an increased number of respiratory tract infections in the first 10 years of life, suggesting immunosuppressive effects of these two molecules. The common cold, asthma, allergic rhinitis, atopic dermatitis, and allergic sensitization were not significantly associated with PFOA or PFOS.]

The potential association between PFAS plasma level and the risk of infectious diseases was also investigated over a 7-year follow-up in mothers and children recruited in Norway between 1999 and 2008 in the Norwegian Mother and Child Cohort Study (MoBa) (Impinen et al., 2019). Blood samples were obtained from the mothers during mid-pregnancy (n = 1943). The median blood concentrations were 12.9 ng/mL for PFOS and 2.5 ng/mL for PFOA. From age 0 to 3 years, positive associations were found between bronchitis and/or pneumonia with both PFOA and PFOS, and between pseudocroup and PFOA only. A positive association between PFOA and gastric flu/diarrhoea from age 6 to 7 years was also found.

Dalsager et al. (2021) investigated the association between the maternal serum concentrations of PFAS during pregnancy and the rate of hospitalization of the children due to common infectious diseases between birth and age 4 years (1503 mother-child pairs were included). Serum samples collected from first-trimester pregnant women from the Odense Child Cohort (OCC) in 2010-2012 were analysed for concentrations of PFOS, PFOA, and other PFAS. Hazard ratios (HRs) expressing the relative change in the instantaneous risk of hospitalization associated with a doubling in maternal PFAS concentration were used. A doubling in maternal PFOS concentration was associated with a 23% increase in the risk of hospitalization due to any infection (HR, 1.23; 95% CI, 1.05–1.44). Every doubling of PFOA or PFOS increased the risk of LRTI by 27% (HR, 1.27; 95% CI, 1.01-1.59) or 54% (HR, 1.54; 95% CI, 1.11–2.15), respectively.

[The Working Group considered that these studies thoroughly addressed the effects of exposure during pregnancy or early life on infectious diseases in children. An association was found between PFOS and the overall risk of infectious disease, and of PFOA and PFOS exposure with the risk of LRTIs, including RSV.]

The objective of the study conducted by Granum et al. (2013) was to investigate the effect of prenatal exposure to PFAS on the responses to paediatric vaccines and immune-related health outcomes in children up to age 3 years (Granum et al., 2013). In the prospective BraMat birth cohort (n = 99), a subcohort of MoBa, pregnant women from Oslo and Akershus were recruited during 2007-2008. Maternal PFAS levels in the blood were measured at delivery (median concentrations for PFOA and PFOS were 1.1 and 5.5 ng/mL, respectively), and the children had their levels measured at age 3 years (not reported by the authors). There was a positive association between the maternal concentration of PFOA and the number of episodes of common cold in the children, and between PFOA and the number of episodes of gastroenteritis.

<u>Zhang et al. (2022b)</u> studied data from the NHANES, including for 517 children aged 3-11 years (2013-2014 cycle) and 2732 adolescents aged 12-19 years (2003-2016 cycles) (Zhang et al., 2022b). PFAS concentrations were measured in serum samples from random subsamples (one third of the total population in that age group of the cycle) for children (age 3-11 years) and adolescents (age 12–19 years). The common cold was self-reported by the participant or parent on the basis of having had a head cold or chest cold during the last month. In children, serum PFOA concentrations were related to the common cold, although the confidence intervals were wide (OR, 1.32; 95% CI, 0.67-2.62). No association was seen for PFOS and the common cold among the children. For adolescents, serum PFOS concentration showed a positive relationship with the study outcome (OR, 1.13; 95% CI, 0.96-1.32), although

with limited precision. No positive relation was found for PFOA. The Bayesian kernel machine regression model was used to study the effect of the mixture. The results showed a clear trend for common cold estimates to increase across quantiles of the total PFAS concentration in the children, whereas no clear pattern was found for the adolescents. [The Working Group considered the above two studies to be of low informativeness, because they were based on a small number of people and because the self-reported questionnaire completed by the participant or parent only described having a head cold or chest cold during the last month.]

The aim of the prospective cohort study conducted by Okada et al. (2012) was to evaluate possible associations between maternal PFOA and PFOS levels in pregnant women and infectious diseases in their infants during the first 18 months of life. Pregnant women were recruited between 2002 and 2005 from the Sapporo region of Japan. Maternal serum PFOA and PFOS levels in blood taken from a peripheral vein after the second trimester, or in certain cases immediately after delivery, were measured and related to the IgE levels in cord blood, reflecting the prenatal exposure of the fetus (n = 231), and to infectious diseases (n = 343) during the following 18 months. The median PFOA and PFOS concentrations were 1.3 ng/mL and 5.2 ng/mL, respectively. The development of infectious diseases at age 18 months was determined from questionnaires to mothers. The results showed no association between maternal PFOA and PFOS levels and infant infectious diseases at age 18 months. [The Working Group noted that this study had some limitations, with a relatively small sample size that was probably insufficient for the identification of significant relationships of PFOA and PFOS exposure with infectious diseases.]

Wang et al. (2022d) investigated the associations of prenatal exposure to PFAS with acute infectious diseases, including the common cold, bronchitis/pneumonia, and diarrhoea, during early childhood. A prospective cohort in Shandong, China, of 235 mother-infant pairs recruited between September 2010 and 2013 from the Laizhou Wan (Bay) birth cohort (LWBC) was studied. Ten selected PFAS were measured in maternal serum, including PFOA and PFOS. The geometric mean concentrations of PFOA and PFOS were and 44.88 ng/mL and 4.39 ng/mL, respectively. Questionnaires completed by parents were used to collect detailed information on acute infectious diseases after 1 year of follow-up. The questionnaire data were confirmed by the medical records. The OR for diarrhoea was 4.99 (95% CI, 1.86-13.39) per log-unit increase in PFOA. When stratified by breastfeeding duration (at least 4 months, or not), the relation between PFAS exposure and diarrhoea was more pronounced among the breastfed infants. However, there were no associations between prenatal PFAS exposure and the common cold or bronchitis/pneumonia (Wang et al., 2022d). [The Working Group noted that the sample size of the study was small, and the results should be interpreted with caution.]

Effects on responses to vaccination

In the following paragraphs, the potential effects of PFOA and PFOS on the response to vaccination are evaluated. The immune response to vaccination is normally dependent on the adaptive immune system, which involves antigen presenting cells and T and B lymphocytes. The response to common vaccines represents a relevant end-point to address the consequences for the immune system of the exposure to chemicals. Indeed, the concentrations of antigen-specific antibodies in serum represent a useful and clinically relevant indicator of immune function in humans. The studies are reported in descending order of informativeness.

<u>Grandjean et al. (2012)</u> studied the relationship between PFAS exposure and vaccination with tetanus and diphtheria toxoids in infants using the serum antibody concentrations at ages

5 and 7 years as a marker. ELISA was used for tetanus antibody and Vero cell-based neutralization assay was used for diphtheria antibody. The study was a prospective study of a birth cohort from the National Hospital in the Faroe Islands, Denmark, with a total of 656 consecutive single-child births recruited during 1997–2000, 587 of which had follow-up until 2008. Maternal serum PFAS concentrations were measured during the third trimester. Maternal geometric mean serum concentrations were 3.20 and 27.3 ng/mL for PFOA and PFOS, respectively. Geometric mean serum concentrations at age 5 years were 4.06 and 16.7 ng/mL for PFOA and PFOS, respectively. Prenatal exposure to both PFOA and PFOS, as indicated by the maternal serum concentrations, was negatively associated with the anti-diphtheria antibody concentrations. The strongest negative correlations were for the association between maternal PFOS and pre-booster diphtheria antibody concentration at age 5 years, with a two-fold higher exposure being associated with a difference of –39% (95%) CI, -55% to -17%) in the diphtheria antibody concentration. For the antibody concentrations at age 7 years, an increase of two-fold in PFOA exposure associated with differences of -36% (95% CI, -52% to -14%) and -25% (95% CI, -43% to -2%) for tetanus and diphtheria, respectively. PFOS exposure was associated with a difference in diphtheria antibody of -28% (95% CI, -46% to -3%) and a non-significant difference of -23.8%(95% CI, -44.3% to 4.2%) for tetanus. These results showed that elevated exposures to PFAS were associated with reduced humoral immune responses to routine childhood immunizations in children aged 5 and 7 years. [The Working Group noted the following strengths: the serum levels represented the combined exposure to all exposure sources and routes, including metabolites of precursors, over a relatively long period of time, due to the long half-lives of PFOA and PFOS. In addition, the study considered co-exposure to smoking and polychlorinated biphenyls (PCBs).]

In a subsequent study, Grandjean et al. (2017) attempted to link the PFAS-associated decreases in antibody concentrations to past exposure during infancy using a new prospective birth cohort in the Faroe Islands (Cohort 5, born in 2007–2009). A total of 381 children participated in the examinations at age 5 years, and 370 of these had also participated at age 18 months. To increase the statistical power of the study, joint statistical analyses were conducted with a previous birth cohort in the Faroe Islands (Grandjean et al., 2012), in which the PFAS exposure was higher. The geometric mean PFOS concentration at age 5 years was 16.7 ng/mL in the 2012 study versus 4.7 ng/mL in the 2017 study. The geometric mean PFOA concentration at age 5 years was 4.06 ng/mL in the 2012 study versus 2.2 ng/mL in the 2017 study. The serum concentrations of antibodies against the tetanus and diphtheria vaccines were determined at age 5 years. This study confirmed the inverse associations of prenatal exposure with the antibodies against both the tetanus and diphtheria toxoids. A doubling of the exposure to PFOA led to a decrease of about 30% in the antibody concentration for tetanus at age 5 years, whereas the association with diphtheria was weaker. Using the same cohorts of people recruited in the Faroe Islands, Mogensen et al. (2015b) estimated the combined (analysis of multiple exposures) change in antibody concentrations for tetanus and diphtheria. In this study, it was not possible to attribute causality to any single PFAS compound. Hence, the three 7-year concentrations, i.e. the concentrations of PFOA, PFOS, and perfluorohexanesulfonic acid (PFHxS), were combined, and the study showed that an increase of two-fold in PFAS was associated with a decrease of 54.4% (95% CI, 22.0–73.3%) in the antibody concentration after vaccination (Mogensen et al., 2015b).

A cross-sectional study in 101 healthy children (aged 1 year) from Berlin, Germany (21 formula-fed, and 80 breastfed for at least 4 months), which included measurements of internal levels of POPs and a broad panel of biological parameters, was performed at the end of the 1990s (Abraham et al., 2020). The plasma levels (mean ± SD) of PFOA and PFOS were 3.8 ± 1.1 and $6.8 \pm 3.4 \,\mu$ g/L in the 21 formula-fed children, and 16.8 ± 6.6 and $15.2 \pm 6.9 \,\mu\text{g/L}$ in the 80 children who were exclusively breastfed for at least 4 months, respectively. This study showed significant correlations of the level of PFOA, but not of PFOS, with the adjusted levels of vaccine antibodies against Haemophilus influenza type b (Hib; r = 0.32), tetanus (r = 0.25), and diphtheria (r = 0.23). These results showed an association between exposure to PFOA and lower levels of antibodies. Moreover, significant associations of PFOA level with lower production of IFN-γ after the stimulation of peripheral blood lymphocytes with tetanus (r = -0.33; P = 0.01) or diphtheria (r = -0.24; P = 0.08) toxoid were found in 55 children, suggesting alterations in the cell-mediated immune system. However, no relevant associations were observed between PFOA or PFOS and lymphocyte proliferation after specific or nonspecific stimulation with vaccine components (Abraham et al., 2020). [The Working Group noted that the strengths of the study were that studies on children in their first year are especially relevant because infants have a relatively high exposure if breastfed and may have a higher susceptibility, because their immune system is developing; and the use of a sensitive analytical method to measure PFOA, PFOS, and PFHxS in almost all samples. The plasma concentrations represented the combined exposure from all exposure sources and routes, including metabolites of precursors, and although measured at a single time point, represent exposure over a relatively long period of time, because of the long half-lives of PFOA and PFOS. Also, co-exposures to other substances, including several carcinogens, were considered in the analysis.]

To examine whether serum PFOA and PFOS were associated with a reduced immunity-response to rubella immunization, and whether interactions with sex or ethnicity warranted data stratification, Pilkerton et al. (2018) analysed a nationally representative sample of individuals aged 12 years in NHANES, USA, for the years 1999-2000 and 2003-2004. In total, 581 adult women, 621 adult men (age, 19-49 years), and 1012 young participants (age, 12-18 years) were included. The average serum PFOA concentrations were 6 ± 0.3 ng/mL in men, 4.3 ng/mL in women, and 4.8 ng/mL in the young participants. The average serum PFOS concentrations were 28.1 ng/mL in men, 22.1 ng/mL in women, and 25.1 ng/mL in young participants. Whole-group linear regression analyses of the young participants showed no significant associations of the rubella titre with either PFOA or PFOS, after adjusting for covariates (sex, age, race/ethnicity, educational level, and BMI), nor were there interactions of these PFAS with sex or ethnicity. In adults, there were significant associations of both PFOA (P = 0.0016) and PFOS (P = 0.0295) quartiles with the rubella titre after adjusting for covariates. In adults, when the effect size analyses were stratified by sex, a significant negative association was observed only for PFOA in men; the association for PFOS was borderline.

Stein et al. (2016), in a cross-sectional study, examined the relations between PFAS serum concentration and measles, mumps, and rubella antibody concentrations in the NHANES cohorts of 1999–2000 and 2003–2004 (n = 1191). The study included 1191 children. The geometric means for serum concentrations were 20.8 ng/mL (95% CI, 19.1–22.7 ng/mL) for PFOS and 4.13 ng/mL for PFOA (95% CI, 3.76–4.53 ng/mL). A doubling of PFOS was associated with a 7.4% (95% CI, –12.8% to –1.7%) decrease in mumps antibodies. When restricted to seropositive children, a 6.6% (95% CI, –11.7% to –1.5%) decrease in mumps antibodies was associated with a doubling of PFOA and a 5.9% decrease with a doubling of PFOS. The largest decrease in antibody concentration (13%) was observed for the association of rubella antibodies with a doubling of PFOS, although only in the seropositive subsets.

Crawford et al. (2023) with financial support by the industry, conducted a systematic review, using PROSPERO to determine the closeness of the association between a doubling in serum PFAS concentration and the difference in log [antibody concentration] after vaccination (Crawford et al., 2023). The literature retrieved from PubMed and Web of Science searches was evaluated, and five PFAS, including PFOA and PFOS, were assessed across 14 reports deemed eligible and published between 2012 and 2022. The evaluation of the responses to diphtheria, rubella, and tetanus infections, including a meta-analysis, were supportive of an association with PFAS, with stronger associations identified for PFOA, PFOS, and PFHxS than for PFNA or perfluorodecanoic acid.

In a study conducted in 819 adolescents aged 12-19 years in the NHANES 2003-2004 and 2009–2010 cycles who had detectable serum levels of rubella and measles antibody, Zhang et al. (2023d) found inverse associations of the serum concentrations of PFOA (P = 0.03), and to a lesser extent PFOS (P = 0.22), with rubella antibodies. An inverse correlation was also found between the serum concentrations of PFOA (P = 0.01) and mumps antibodies. However, these associations were present only in adolescents with lower folate levels (measured as a ratio between whole-blood and serum folate levels) and not in adolescents with higher levels. [The Working Group noted that the study did not investigate the potential mechanism or the effect of folate.]

An investigation of the relation between the antibody response after vaccination with an inactivated trivalent influenza vaccine and the circulating levels of PFOA and PFOS was performed in the USA (Looker et al., 2014). The study participants comprised 411 adults who lived for at least 1 year between 1950 and 2004 in one of six water districts in the Mid-Ohio Valley region of the USA, where the inhabitants had elevated PFOA exposure from contaminated drinking-water. To assess the response to influenza vaccination, haemagglutination-inhibition tests were conducted on serum samples collected pre-vaccination and 21 ± 3 days post-vaccination in 2010. Serum samples were also analysed for PFOA and PFOS concentrations. The concentrations of PFOA were as follows: first quartile: 0.25–13.7 ng/mL; second quartile: 13.8–31.5 ng/mL; third quartile: 31.6–90 ng/mL; and fourth quartile: 90.4–2140 ng/mL. The concentrations of PFOS were as follows: first quartile: 0.1-5.8 ng/mL; second quartile: 5.9-9.2 ng/mL; third quartile: 9.3-14.5 ng/mL; and fourth quartile: 14.7-42.3 ng/mL. Questionnaires were completed regarding the occurrence and frequency of recent (during the last 12 months) respiratory infections. The results showed that PFOA at serum concentrations between 13.7 and 90 ng/mL was associated with a reduced response to influenza virus A/H3N2. The authors also found an increased risk of not attaining the antibody threshold considered to offer long-term protection. However, there was no evidence that the PFOS serum concentration was associated with the vaccine response. The authors also found no evidence of an association between self-reported colds or influenza and PFOA or PFOS concentration, suggesting that the extent of suppression of the vaccine response and the associations with disease outcomes are difficult to assess, particularly in a small study sample (Looker et al., 2014).

Andersson et al. (2023) performed an observational study with prospective design of 309 adults from the Ronneby Biomarker Cohort in Ronneby, Sweden, who had a median PFOS serum concentration of 47 ng/mL, and a group from another town with background exposure (n = 47) who had a median PFOS serum concentration of 4 ng/mL. These groups received two doses of the Spikevax mRNA vaccine against

COVID-19 (SARS-CoV-2). The serum levels of seven PFAS were measured before vaccination. IgG against the SARS-CoV-2 spike antigen was measured before vaccination and 5 weeks (n = 350) and 6 months (n = 329) after the second vaccine dose. PFAS exposure, regardless of how it was estimated, was not negatively associated with antibody levels after COVID-19 vaccination.

In a study by Porter et al. (2022), a total of 415 participants were included in repeated measures analyses (757 observations) of antibodies against SARS-CoV-2. The participants were current and retired workers from two manufacturing facilities with historical use of PFAS in Alabama and Wisconsin, USA; the study was sponsored by the company. The log-transformed concentrations of anti-spike IgG and neutralizing antibodies were modelled in relation to the concentrations of PFAS at enrolment, after adjusting for antigenic stimulus. The median concentration of PFOS was 121.50 ng/mL and that of PFOA was 31.7 ng/mL. The fully adjusted IgG concentration was 3.45% lower (95% CI, -7.03% to 0.26%) for each 14.5 ng/mL (interquartile range) increase in PFOS. For PFOA, the results were comparable to those for PFOS: the concentration was 1.95% lower (95% CI, -4.35% to 0.51%) for each 3.59 ng/mL (IQR) increase in PFOA.

One of the objectives of the study conducted by Granum et al. (2013) was to investigate the effect of prenatal exposure to PFAS on the responses to paediatric vaccines. To this end, the antibody levels specific for four vaccines in the Norwegian Childhood Vaccination Program were measured: measles, rubella, tetanus, and Hib (Granum et al., 2013). Pregnant women in the prospective BraMat birth cohort (n = 99), a subcohort of MoBa, were recruited during 2007-2008 in Oslo and Arkershus, Norway. Blood samples were collected from the mothers at the time of delivery and from the children at age 3 years. In multivariate models, increased concentrations of all four PFAS, including PFOA (median, 1.1 ng/mL; interquartile range, 0.2–2.7 ng/mL)

and PFOS (median, 5.5 ng/mL; interquartile range, 1.4–11.0 ng/mL), in maternal blood (50 samples tested) were significantly associated with reduced levels of anti-rubella antibodies in the children at age 3 years (PFOA bivariate $\beta = -0.40$; 95% CI, -0.64 to -0.17; P = 0.001; and PFOS bivariate $\beta = -0.08$; 95% CI, 0.14 to -0.02; P = 0.007). However, no significant associations were found between the concentrations of PFAS, including PFOA and PFOS, and antibody levels developed in response to the other vaccines, including tetanus toxoid (Granum et al., 2013). [The Working Group noted that a major limitation of the study was the very small study sample.]

In a randomized controlled trial conducted in rural regions close to Bissau, Guinea-Bissau, where exposure to PFAS is generally low compared with that in all other parts of the world (237 infants; median serum PFOS concentration, 0.77 ng/mL; median serum PFOA concentration, 0.68 ng/mL), a doubling of serum PFOS concentration in children vaccinated at age 4–7 months was associated with a 21% lower concentration of measles antibodies (95% CI, –37% to –2%) measured at age 9 months for each 1 ng/mL increment in blood concentration (<u>Timmermann</u> et al., 2020).

Kaur et al. (2023) investigated the relation between maternal plasma PFAS concentration (n = 72) and SARS-CoV-2 anti-spike IgG antibody protein levels in a New York City, USA-based pregnancy cohort, the Gen C cohort, established in April 2020. Of all the congeners tested, the nine congeners that were measurable in the maternal plasma negatively correlated with the SARS-CoV-2 anti-spike IgG antibody level (P < 0.05) in multivariable analyses. There was a significant association for PFOA ($\beta = -0.62$; 95% CI, -1.11 to -0.12; P = 0.017), but not for PFOS ($\beta = -0.33$; 95% CI, -0.85 to 0.20; P = 0.209). The mean maternal serum levels were 1.10 ng/mL for PFOA and 1.84 ng/mL for PFOS.

In Greenlandic children who participated in the INUENDO cohort and the IVAAQ cohort,

Timmermann et al. (2022) examined associations between exposure to PFAS, PCBs, and mercury in all forms and blood concentrations of diphtheria and tetanus vaccine antibodies after vaccination. The study included cross-sectional data for children aged 7-12 years, collected in Greenland during 2012–2015. A total of 338 children were eligible for the study and had blood samples available, and 175 of these had available vaccination records. In the children, the median concentration of PFOS was 8.68 ng/mL and that of PFOA was 2.28 ng/mL. The authors found that higher childhood exposure to environmental chemicals was associated with lower antibody concentrations after vaccination and with higher odds of not having a sufficient antibody concentration to be protected against diphtheria. The ORs for not being protected against diphtheria were 1.14 (95% CI, 1.04–1.26) for PFOS and 1.41 (95% CI, 0.91–2.19) for PFOA (Timmermann et al., 2022). [The Working Group noted that the concentrations of the specific diphtheria antibodies were quite low, probably due to the time interval since the most recent vaccination or booster.]

End-points of immunity

Associations between neonatal serum concentrations of PFOA and PFOS and immunoglobulin (Ig) isotype profiles in dried blood spots from newborns were assessed in a prospective cohort of infants (3175 infants in the Upstate KIDS prospective birth cohort study, 2008–2010) (Jones et al., 2022). After correction for multiple comparisons, higher PFOA levels in the neonates' blood samples were associated with higher IgA, IgM, and IgG2 levels, and lower levels of IgE, in single-pollutant models, with P values ranging from 0.002 to < 0.00001. However, PFOS showed no significant association with the Ig isotype profile. When PFOA and PFOS were examined as a mixture, with estimates adjusted for infant sex, maternal BMI, race, parity, age, and infertility treatment, small (but significant)

to marginal joint effects of the mixture on the outcome isotypes, consistent with single-pollutant PFAS models, were found (Jones et al., 2022). [The Working Group considered that the clinical relevance of these observations remains to be established.]

Zhu et al. (2016) assessed the polarization of T-helper lymphocytes in 231 asthmatic children and 225 non-asthmatic children (controls) from the Genetic and Biomarkers study for Childhood Asthma (GBCA) in northern Taiwan, China. The levels of Th1 (IFN-y and IL-2) and Th2 (IL-4 and IL-5) cytokines were measured using ELISA. The Th2/Th1 cytokine concentration ratio and the IgE levels were in accordance with the asthmatic status of the patients. The results showed that asthmatic children had significantly higher serum PFAA concentrations, but not higher concentrations of specific PFAS, compared with the healthy controls. In male asthmatic children, significant positive associations were found between the IL-4/IFN-y ratio and PFOS and PFOA (correlation coefficients, Spearman $\rho = 0.210; P = 0.008;$ and Spearman $\rho = 0.290;$ P < 0.001, respectively), and between the IL-5/ IFN-y ratio and PFOS and PFOA (Spearman $\rho = 0.198$; P = 0.017; and Spearman $\rho = 0.189$; P = 0.017, respectively). In female asthmatic children, significant associations were found only of the IL-4/IFN-y ratio with PFOS (Spearman $\rho = 0.291$; P = 0.013), and of the IL-5/IFN- γ ratio with PFOS (Spearman $\rho = 0.245$; P = 0.037). [The Working Group considered that the limitation of these analyses should be noted, with only single serum samples being collected for cytokine measurement.

Lopez-Espinosa et al. (2021) studied the associations of PFHxS, PFOA, PFOS, and PFNA with leukocyte counts (neutrophils, monocytes, eosinophils, lymphocytes, and basophils). Serum PFAS concentrations were measured, and leukocyte types were counted in 42782 (between 2005 and 2006) and 526 (during 2010) adults living in an area with PFOA contamination of drinking-water in the Mid-Ohio Valley, USA. The circulating numbers of immune cells (CD3+ T-cells, CD3+ CD4+ T-helper cells, CD3+ CD8+ T-cytotoxic cells, CD3+CD4+CD8+double-positive T-cells, CD16+ CD56+ natural killer (NK) cells, and CD19+ B cells) were measured only in the 2010 adults. Data on the study participants were collected through questionnaires, and the covariates sex, race/ethnicity, age, education, BMI (available for the 2005–2006 survey), tobacco consumption, alcohol intake, a present or past diagnosis of an immune disease and/or cancer, and the use of anti-inflammatory medication (available for the 2010 survey) and other medication were considered. A weak consistent association was shown for both PFOA and PFOS with the total lymphocyte count, but no clear associations with the various lymphocyte subtypes. No clear associations were found between the total leukocyte count and PFAS level, with statistical significance only being reached for PFHxS and PFOS in the 2005–2006 survey (Lopez-Espinosa et al., 2021). [The Working Group noted that the small differences in the overall lymphocyte count observed in the study cannot be considered clinically meaningful.]

(ii) Human cells in vitro

Mechanisms of PFOA- and PFOS-induced immunosuppression were evaluated in vitro using human primary immune cells (lymphocytes and monocytes) and human immune cell lines. In a recent study, Kasten-Jolly and Lawrence (2022) showed a significant reduction in the proliferation of human T-cells (from peripheral blood mononuclear cells (PBMCs) isolated from healthy staff of the New York State Department of Health) in response to phytohaemagglutinin (PHA; polyclonal activation) or influenza antigen (memory T-cell activation) in the presence of 100 µM PFOA or PFOS (cells were exposed to increasing concentrations of the two chemicals at 1, 10, and 100 µM). Moreover, both PFOA and PFOS at the concentration of $100 \,\mu M$ significantly (P < 0.05) suppressed the release of IFN- γ , a cytokine known to be important for the immune response to viruses. The production of other cytokines and chemokines by human PBMCs was also affected by PFOA (MIP3 α / CCL20, ITAC, IL-10, IL1- β , IL-6, and IL-7) and PFOS (same as for PFOA, plus GM-CSF, fractalkine, IL-17, IL-13, IL-21, and TNF- α). PFOS was more potent than PFOA. However, for both molecules, IL-2 production was significantly augmented, despite an inhibition of proliferation. These effects were significant with 100 μ M of PFOA or PFOS, except for CCL20, the production of which was increased by 10 μ M of either chemical.

The cytokines released by T lymphocytes were evaluated using a whole-blood assay and peripheral blood from healthy donors after stimulation with the polyclonal activator PHA. Both PFOA and PFOS reduced PHA-induced IL-4 and IL-10 release, whereas IFN-y release was affected only by PFOS (Corsini et al., 2011). In all the tests performed, PFOS was a more potent inhibitor of cytokine production than PFOA. At the concentration of 0.1 µg/mL (equivalent to 241 nM PFOA and 200 nM PFOS), only PFOS was able to reduce the secretion of TNF-a, IL-4, IL-6, IL-10, and IFN-y. The same assay was used to assess the release of pro-inflammatory cytokines after LPS stimulation. Both PFOA and PFOS induced a dose-related decrease in TNF-a production, IL-8 release was unaffected, and the release of IL-6 was reduced by PFOS, but not by PFOA (Corsini et al., 2011). Similar findings were reported in the human promyelocytic cell line THP-1, with PFOS causing significant decreases in the LPS-induced release of TNF-a and IL-8, starting from a concentration of 1 µg/mL. In these models, the suppression of cytokine production by PFOA was dependent upon PPARa activation, whereas the effects of PFOS were independent of PPARα activation. On the other hand, Midgett et al. (2015) demonstrated that PFOA did not have an impact on IL-2 production, but PFOS

at a concentration of 100 µg/mL suppressed IL-2 production in both the Jurkat T-cell line and human primary CD4+ T cells isolated from leukocytes of healthy donors (n = 11).

Using human primary lymphocytes (6 donors) as a model, together with integrative omics analyses, including of the transcriptome and lipidome, and bioinformatics analysis, Li et al. (2020c) observed that PFOS moderately altered the production of interleukins (IL-1, IL-4, IL-6, and IL-8) in human lymphocytes and dysregulated clusters of genes and lipids that play important roles in immune function, such as in lymphocyte differentiation, the inflammatory response, and the immune response. The same group performed a metabolomic analysis on human lymphocytes and highlighted alterations in organic acids, including taurine and NG-dimethyl-l-arginine; organooxygen molecules, such as d-ribulose 5-phosphate; and lipid compounds, such as carnitines; in lymphocytes treated with PFOA (Li et al., 2020d).

<u>Corsini et al. (2012)</u>, using a human wholeblood assay and LPS stimulation, confirmed the inhibitory effect of PFOS, and to a lesser extent PFOA, on TNF- α production. The concentrations used were 0.1, 1, and 10 µg/mL. These results were confirmed using the human promyelocytic cell line THP-1.

In another study that used the monocytic THP-1 cell line under LPS stimulation, <u>Masi</u> et al. (2022) also observed that PFOS (0.2–20 μ M) induced a dose-dependent downregulation of RACK-1 receptor promoter transcriptional activity, mRNA expression, and protein levels, which was mirrored by reductions in IL-8 and TNF- α production and CD86 expression.

Houck et al. (2023) tested 147 PFAS using a phenotypic screening platform of human primary cell co-culture systems, the BioMAP Diversity PLUS panel. This platform models tissues and the disease biology of organs (vasculature, immune system, skin, and lung). The platform includes 12 assays that use 12 different human primary cell systems and includes 148 mechanistic end-points, of which several are modulators and effectors of vascular and immune biology. The authors included four known immunosuppressants (azathioprine, methotrexate, dexamethasone, and cyclosporine A), covering different mechanisms of immunosuppressive action, in the test set, along with the 147 PFAS (Houck et al., 2023). PFOA and PFOS reduced the level of IL-10 in the co-culture system, and their mechanisms of action were different from those of other known immunosuppressants.

Human NK-cell activity has recently been shown to be slightly inhibited after 24 hours of exposure of human primary PBMCs from male donors to 0.2μ M PFOS (Maddalon et al., 2023b).

[The Working Group noted that PFOS and, to a lesser extent, PFOA reduced the production of some cytokines, with the possible involvement of PPARa and lipid metabolism, and also altered lymphocyte proliferation and NK-cell activity. The latter effect was only in cells isolated from male donors.]

- (b) Experimental systems
- (i) PFOA in non-human mammalian systems in vivo

Effects on immune end-points

PFOA was evaluated using standard assays of immune function (<u>Yang et al., 2002a</u>; <u>De Witt et al., 2008</u>; <u>McDonough et al., 2020</u>; <u>De Guise and Levin, 2021</u>).

In the study by <u>De Witt et al. (2008)</u>, sheep red blood cell (SRBC) IgM antibody titres were suppressed in C57BL/6J and C57BL/6N mice in a dose-dependent manner at a PFOA exposure of \geq 3.75 mg/kg per day (ammonium salt; purity, \geq 98%) for 10–15 days given via the drinking-water. However, the IgG titres were elevated at PFOA exposure levels of 3.75 and 7.5 mg/kg per day (<u>De Witt et al., 2008</u>). Using a diet supplemented with PFOA at 0.02% (w/w), Yang et al. (2002a) showed a clear decrease in the production of antibodies of both IgM and IgG isotypes against horse erythrocytes injected intravenously ($5-10 \times 10^7$ cells in 200 mL of Earle balanced solution) into C57BL/6 mice.

A significant decrease in the IgM antibody response to the T-cell-dependent antigen keyhole limpet haemocyanin (KLH) was observed with PFOA at a dose of 1.88 mg/kg per day given to $B6C3F_1$ mice in their drinking-water for 4 weeks (De Guise and Levin, 2021).

Altered T-lymphocyte populations were observed in male ICR mice exposed to 0, 2, 10, 50, or 250 ppm (0, 0.49 ± 0.04 , 2.64 ± 0.15 , 12.63 \pm 1.15, or 47.21 \pm 3.57 mg/kg per day) PFOA (free acid; purity, 98%) in drinking-water for 21 days. In the spleen, all the doses of PFOA reduced CD8+ lymphocytes, and CD4+ lymphocytes were increased by PFOA at 50 and 250 ppm. Exposure to PFOA at 250 ppm increased CD8+ lymphocytes in the thymus. However, PFOA induced a significant decrease in body weight from the initial value at the two highest doses (Son et al., 2009) (see also Section 4.2.7(c) and Section 4.2.6(b)(i)).

Iwai and Yamashita (2006) treated male Crj:CD(SD)IGS rats orally for 14 days with PFOA (ammonium salt, 10% aqueous solution). The doses of PFOA were 0, 0.5, 5, and 50 mg/kg per day. Blood parameters (haematocrit, erythrocyte count, and haemoglobin) were reduced by PFOA at 50 mg/kg, but the reticulocyte count was not affected. No influence on lymphocyte subsets in blood (T-cells and NK cells) was observed at any of the tested doses. [The Working Group noted that the circulating levels of PFOA were not measured in this study.]

Loveless et al. (2008) conducted a study with the objective of comparing the effects of PFOA (ammonium salt, 19.5% aqueous solution) in male CD rats and CD-1 mice. Both species were dosed by oral gavage with *n*-PFOA at 0.3, 1, 10, or 30 mg/kg per day for 29 days. In the rats, the production of anti-IgM SRBC antibodies was not affected; however, systemic toxicity (reduced body-weight gain and increased corticosterone levels) was observed at the 10 and 30 mg/kg doses. Systemic toxicity was also observed in mice at the 10 and 30 mg/kg doses. In mice, reduced IgM antibody production after 10 mg/kg was observed (20% suppression), along with reduced spleen and thymus weights and cell numbers, as well as microscopic depletion/atrophy of lymphoid tissue (thymus and spleen). [The Working Group noted that non-specific alteration of the immune response to SRBCs (a T-cell-dependent antigen) because of systemic toxicity of PFOA in mice cannot be excluded.]

The proliferation in response to both T-cell (concanavalin A, ConA) and B-cell (LPS) activators of ex vivo spleen cells derived from male C57BL/6 mice fed for 10 days with a diet containing 0.02% (w/w) PFOA was attenuated by the PFOA treatment (Yang et al., 2002a).

Because of concerns regarding skin exposure, the immunotoxicity of PFOA (free acid; purity, 96%) after dermal exposure (0.5-2% w/v, or 12.5-50 mg/kg per dose administered topically on the dorsal portion of the ear) has been also evaluated in mice. The IgM antibody response to SRBCs was significantly reduced in the spleen after 4 days of dermal exposure in B6C3F₁ mice (<u>Shane et al., 2020</u>). In addition, PFOA exposure caused a significant decrease in thymus and spleen weights after 4 and 14 days of exposure. Immune-cell phenotyping identified a reduction in the frequency of splenic B-cells (<u>Shane et al.,</u> 2020).

Effects on the developing immune system in mice

Experiments have also been conducted in mice to study the effects of PFOA and PFOS on the developing immune system (Keil et al., 2008; Hu et al., 2010, 2012; Zhong et al., 2016; Torres et al., 2021). The mean SRBC-specific IgM antibody titres in the female offspring (PND48) of C57BL/6 mouse dams exposed to

PFOA (ammonium salt; purity, \geq 98%) from GD6 to GD17 were not affected by doses of up to 1 mg/kg bw given in the drinking-water (Hu et al., 2010). The adult offspring of dams exposed to PFOA at 2 mg/kg during gestation and lactation showed a reduction in the percentage of splenic CD4+CD25+Foxp3+ T-cells and in IL-10 production, suggesting alterations to regulatory T cells (Hu et al., 2012).

Effects in models of infection in mice

One study showed that PFOA (purity, 96%) can accumulate in the lungs, making this organ more susceptible to viral infections (Ahmad et al., 2021). Ahmad et al. (2021) observed in CD-1 mice orally exposed to 5 or 20 mg/kg per day PFOA for 10 days that it accumulated in the lungs, with mean concentrations of 14.14 ± 2.95 and $36.41 \pm 15.09 \,\mu\text{g/g}$, respectively, measured in the lung tissue. Changes in mRNA expression of the DNA methylation regulator genes encoding DNA methyltransferases (Dnmts) and tet methylcytosine dioxygenases (Tets) were observed, along with the genes encoding the membrane proteins angiotensin converting enzyme 2 (Ace2) and transmembrane serine protease 2 (*Tmprss2*), which are involved in SARS-CoV-2 virus infection (see also Section 4.2.4).

(ii) PFOS in non-human mammalian systems in vivo

Effects on immune end-points

Experimental data derived mainly from in vivo experiments in rodents (rats or mice) exposed for 7–60 days using the oral route (gavage or food intake). In vivo antibody production was evaluated after immunization with a specific antigen, and ex vivo experiments measured lymphocyte proliferation, cytokine production, and NK cell activity. Doses of PFOS above 20 mg/kg (total administered dose, TAD) were found to be toxic (reduced body weight) and were considered not to be relevant in general.

In vivo daily exposure of B6C3F₁ mice for 28 days to PFOS (potassium salt; stated purity, >98%) at 0, 0.166, 1.66, 3.31, 16.6, 33.1, or 166 µg/kg bw per day, given by gavage, significantly reduced the T-cell-dependent production of antibodies against SRBCs in cells isolated from the spleen, assessed using a plaque-forming cell assay. The lowest observed effect level was 0.05 mg/kg TAD over 28 days in male mice and 0.5 mg/kg TAD in female mice, corresponding to 1.66 and 3.31 µg/kg per day, respectively (Peden-Adams et al., 2008). The measured PFOS serum concentrations at these doses were 91.5 \pm 22.2 ng/mL and 666 \pm 108 ng/mL (mean \pm SD), respectively. NK-cell activity was significantly increased in male mice at the doses of 0.5, 1, and 5 mg/kg TAD, but not in females. The serum trinitrophenyl (TNP)-specific IgM titres were also reduced by PFOS after TNP-LPS (TNP conjugated with LPS) challenge, suggesting that the humoral immune effects may be attributed to B-cells rather than T cells, because both T-dependent (SRBC) and T-independent (TI) (TNP-LPS) antigen administration resulted in suppressed IgM production (Peden-Adams et al., 2008).

Dong et al. (2009) also observed a significant reduction in anti-SRBC IgM antibodies in male C57BL/6 mice daily exposed for 60 days by gavage to PFOS (potassium salt; purity, > 98%) at 0, 8.33, 83.33, 416.67, 833.33, or 2083.33 µg/kg bw per day, with a no observed adverse effect level (NOAEL) of 8.33 µg/kg per day (0.5 mg/kg TAD) and a lowest observed adverse effect level of 83.33 µg/kg per day (5 mg/kg TAD). The measured PFOS serum concentration for 0.5 mg/kg TAD was 0.674 \pm 0.166 mg/L.

In a follow-up study, <u>Dong et al. (2011)</u> used a similar protocol but slightly different doses. The NOAEL for a decrease in IgM was 16.7 μ g/kg per day and the lowest observed adverse effect level was 83.3 μ g/kg per day (TAD, 5 mg/kg per day).

Antibodies to SRBC were also found to decrease after 7 days' exposure to PFOS (potassium salt; purity, > 98%) by gavage at 5 mg/kg per day, and the PFOS level was $110.46 \pm 6.18 \text{ mg/L}$ (Zheng et al., 2009).

Delayed-type hypersensitivity (DTH), a marker of cell-mediated immunity, in adult male C57BL/6 mice treated with PFOS orally for 60 days was not altered by doses up to 50 mg/kg, corresponding to PFOS at 833.3 μ g/kg bw per day and a serum level of 51.71 ± 3.81 mg/L (Dong et al., 2011).

In a study by <u>McDonough et al. (2020)</u>, a formulation containing several PFAS (a commercial brand of electrochemically fluorinated AFFF) was administered for 10 days via gavage to female and male C57BL/6 mice (PFOS + PFOA, at 0, 1.88, 3.75, 7.5, or 10 mg/kg bw). The results showed that antigen-specific antibody production was suppressed, on average, by 13% in male mice and by 12.4% in female mice across all the doses (<u>McDonough et al., 2020</u>). [The Working Group considered that these results obtained in vivo in mice might suggest that T-cell-dependent antibody production was the main target of PFOS.]

Splenic T-cell immunophenotypes (expressing CD4 and CD8 markers or not) were minimally altered in B6C3F₁ female mice (age 7–8 weeks), but all the T-cell subpopulations were slightly but significantly modulated in male mice by PFOS at 0.1-0.5 mg/kg TAD for 28 days by the oral route. The PFOS doses corresponded to 131 ± 15.2 ng/mL for male mice and 123 ± 18.7 ng/mL for female mice (Peden-Adams et al., 2008).

Also, the absolute numbers of splenic cells expressing CD19/CD21 or major histocompatibility complex (MHC) II markers in adult female $B6C3F_1$ mice treated orally with PFOS for 28 days were not altered at doses up to 5 mg/kg TAD (Fair et al., 2011). The splenic and thymic cellularity of adult male C57BL/6 mice was significantly reduced at oral doses of PFOS from 25 mg/kg for 60 days. At this same dose, splenic and thymic CD4+ lymphocytes were altered in adult male C57BL/6 mice. The NOAEL was 5 mg/kg TAD. Splenic CD8+ and B-cell numbers were also affected at 50 mg/kg and above (Dong et al., 2009). Splenic and thymic cellularity, including CD4+, CD8+, and B-cells, were also decreased in adult male C57BL/6 mice after oral exposure to PFOS for 7 days at doses from 20 mg/kg (Zheng et al., 2009). [The Working Group considered that PFOS mostly altered lymphocyte function, but also induced a decrease in the number of cells at doses of 25 mg/kg and above.]

Other immune parameters (lymphoproliferation, NK cell activity, and cytokines) have been measured ex vivo in mice using splenocytes. NK cells are an important component of the immune surveillance for cancer and provoke the lysis of cells not expressing MHC class I molecules, such as cancer and virally infected cells.

NK cell activity, measured with a chromium assay and expressed in lytic units, was increased in male $B6C3F_1$ mice at 0.5, 1, and 5 mg/kg TAD compared with controls, but was not altered in females after 28 days' exposure to PFOS by gavage (<u>Peden-Adams et al., 2008</u>). An augmentation of splenic NK cell activity, determined by lactate dehydrogenase release, was also observed in adult male C57BL/6 mice after oral exposure to PFOS at 5 mg/kg TAD for 60 days (<u>Dong et al.,</u> 2009).

T-lymphocyte and B-lymphocyte proliferation, measured after polyclonal activation using mitogens (ConA or LPS), was not affected in B6C3F₁ mice exposed by gavage to PFOS for 28 days at doses up to 5 mg/kg TAD, and in male C57BL/6 mice after oral exposure to PFOS for 60 days at doses up to 25 mg/kg TAD (Peden-Adams et al., 2008; Dong et al., 2009).

T-cell proliferation in response to ConA stimulation in PFOA-exposed Balb/c mice was inhibited after PFOS (potassium salt; purity, > 98%) exposure, and the T-cell receptor signalling, calcium signalling, and p38 MAPK signalling pathways, as measured by microarray data analysis, and calcium ion influx were augmented in mouse splenocytes (Lv et al., 2015). However, splenic T-lymphocyte proliferation was slightly decreased in male C57BL/6 mice after oral exposure to PFOS for 7 days at 5 mg/kg per day (Zheng et al., 2009).

Ex vivo IL-6 production by B-cells was significantly increased by in vitro stimulation with either anti-CD40 or LPS in female $B6C3F_1$ mice exposed orally for 28 days to PFOS (potassium salt; stated purity, > 98%) at 1 mg/kg TAD (Fair et al., 2011). IL-4 secretion was increased by exposure to PFOS at \geq 5 mg/kg TAD administered to male C57BL/6 mice daily via gavage for 60 days (Dong et al., 2011) and also after oral exposure to PFOS for 7 days at 5 mg/kg per day (Zheng et al., 2011).

<u>Pierpont et al. (2023)</u> measured immune parameters after chronic exposure of C57BL/6 mice to PFOS (potassium salt; purity, 88.9%). Mice were exposed to PFOS at 0.15, 1.5, 15, or 50 µg/kg for 28 days, after which B cells, T cells, and granulocytes from the bone marrow, liver, spleen, lymph nodes, and thymus were assessed. No effects of PFOS on the major T- or B-cell populations, macrophages, dendritic cells, basophils, mast cells, eosinophils, neutrophils, serum antibodies, orselected serum cytokines were observed. All the experimental groups had elevated serum PFOS levels, with the highest levels obtained at 50 µg/kg per day (females, 2792.0 ± 295.3 ng/mL, and males, 2159.2 ± 164.6 ng/mL).

Interestingly, when male $B6C3F_1$ mice were exposed to a PFOS tetraethylammonium salt at a TAD of 7 mg/kg bw (equivalent to a TAD for PFOS anion of 5.55 mg/kg) in the diet for 28 days, a serum concentration of 11 µg/mL was yielded. However, no effects were found on the cellular compositions of the thymus and spleen, the number of splenic cells secreting IgM antibodies against SRBCs or TNP-LPS (Qazi et al., 2010). [The Working Group noted that only a single dose administered via the diet was used in the study.]

<u>Lefebvre et al. (2008)</u> also addressed the effects of PFOS on immune system function (potassium salt; stated purity, \geq 98%) using

exposure via the diet, but in Sprague-Dawley rats. The rats were exposed for 28 days to PFOS at 2–100 mg/kg diet (corresponding to approximately 0.14–7.58 mg/kg per day). The body-weight reductions induced were significant in male and female rats exposed to PFOS at 50 or 100 mg/kg diet. There were no effects on the DTH response to KLH, KLH-specific IgG in the serum, or splenic T- and B-cell proliferation in response to a mitogen ex vivo (Lefebvre et al., 2008).

Effects on the developing immune system in mice

In the B6C3F₁ pups of dams exposed orally to PFOS (potassium salt; purity, 91%) during GD1– GD17, NK cell function and IgM production were significantly decreased at age 8 weeks. The NOAEL was 0.1 mg/kg per day (Keil et al., 2008). C57BL/6 mouse pups were evaluated for developmental immunotoxicity after maternal oral exposure to PFOS (potassium salt; purity, > 98%) of 0.1, 1.0, or 5.0 mg/kg per day during GD1– GD17 (Zhong et al., 2016). The results showed alterations in splenic and thymic cellularity, in T- and B-cell proliferation measured ex vivo after 5 mg/kg exposure, in splenic NK-cell activity at age 8 weeks, and in the antibody response at age 4 but not 8 weeks.

Effects in models of infection in mice

Experimental models of infection are often used to address the consequences of immunosuppression on host resistance.

A 21-day exposure to PFOS (potassium salt) at 25 µg/kg per day, corresponding to PFOS at 670 \pm 47 ng/mL in the serum, resulted in a significant increase in emaciation and mortality in response to influenza A virus in B6C3F₁ mice (Guruge et al., 2009).

In another study in C57BL/6 mice exposed to PFOS (potassium salt; purity, 88.9%) at 1.5 μ g/kg per day for 4 weeks or to 3 μ g/kg per day for 14 days, no effects were found on influenza virus clearance or antibody or T-cell-specific antiviral responses, indicating that for these doses

and durations, PFOS did not suppress the antigen-specific immune response. The PFOS serum levels were 99.6 \pm 4.4 and 116 ng/mL \pm 2.8 ng/mL for 1.5 and 3 µg/kg per day, respectively (Torres et al., 2021).

In a mouse model of *Citrobacter rodentium* infection, persistent treatment with PFOS at 2 mg/kg (potassium salt; purity, > 98%) increased the bacterial count, and this was accompanied by increases in inflammatory cytokine concentrations and dysbiosis, suggesting an alteration in intestinal immunity (Suo et al., 2017).

(c) Mechanism of action of PFOA and PFOS on the immune system

Taylor et al. (2023) confirmed in C57BL6 mice treated for 15 consecutive days the significant decrease of IgM antibodies directed against SRBCs. They also observed changes in the numbers of B-cell subsets as well as mitochondrial markers after PFOA (free acid) exposure, indicating that the B-cell developmental trajectory could be altered through effects on B-cell differentiation or proliferation, leading to suppression of the T-cell-dependent antibody response (TDAR).

PFOA and PFOS have been described to have pro-inflammatory effects, mediated through cytokine production (as also reported in Section 4.2.6), which could indirectly modulate the immune response. Son et al. (2009) found that levels of TNF- α , IL-1 β , and IL-6 were augmented in the spleen in male mice exposed to PFOA (ammonium salt; purity, 98%) in drinking-water for 21 days. However, most in vitro studies conducted with human or rodent immune cells showed a decrease in pro-inflammatory cytokine production. PFOA and PFOS reduced the number of macrophages (CD11b+ cells) in the bone marrow, but not in the spleen or peritoneal cavity, in C57BL/6 mice receiving 0.02% (w/w) PFOS (tetraethylammonium salt; purity, 98%) or PFOA (free acid; purity, 96%) in the diet for 10 days (Qazi et al., 2009). This

high-dose, short-term exposure also augmented the inflammatory responses to LPS. Dietary treatment of male C57BL/6 mice with 0.002% (w/w) PFOA or 0.005% (w/w) PFOS for 10 days attenuated the hepatic levels of TNF- α , IFN- γ , and IL-4 (Qazi et al., 2010).

[The Working Group noted that the mechanisms of action of PFOA and PFOS on the immune system are still a matter of debate.]

A role of PPARa in the PFOA-induced reduction in thymus weight and cellularity has been evoked in mice (Yangetal., 2000, 2001). In PPARanull mice, reductions in spleen weight and in the number of splenocytes after PFOA treatment were absent, and the decrease in the number of thymocytes was significantly less marked (Yang et al., 2002b). The response of splenocytes isolated from the spleens of PFOA-treated PPARa-null mice to appropriate T- or B-cell activators in vitro was not altered compared with wildtype mice (Yang et al., 2002b). However, De Witt et al. (2016) found that exposure to PFOA at 30 mg/kg suppressed the TDAR in both WT and PPARa KO C57BL/6 mice, suggesting that the suppression of TDAR might be independent of PPARa involvement.

To determine whether the immunotoxicity of PFOA (free acid; purity, > 96%) is associated with lipid metabolism, male BALB/c mice were fed either a regular diet or an HFD, and exposed to PFOA at doses of 0, 5, 10, or 20 mg/kg per day for 14 days (<u>Wang et al., 2014a</u>). The results suggested that an excess of dietary lipids did not prevent the PFOA-induced immune suppression caused by peroxisome proliferators. Moreover, immunomodulation by PFOA was via the PPAR pathway and involved the induction of mitochondrial damage and the lymphocyte apoptosis pathway.

Ten-day treatment of male 129/Sv PPAR α -null mice with different dietary doses (0.001%–1% w/w) of PFOS showed that the thymic changes were partially dependent on PPAR α (Qazi et al., 2009). It has been suggested that PFOS (free acid;

purity, > 98%) may indirectly affect the immune organs by interfering with lipid metabolism, leading to co-senescence of the thymus and spleen (Wang et al., 2011).

Corsini et al. (2011) conducted in vitro experiments showing that PFOA and PFOS (free acids) suppressed LPS-induced TNF-a production in human primary cultures and THP-1 cells, whereas IL-8 was suppressed only in THP-1 cells, and IL-6 release was reduced only by PFOS. Both PFOA and PFOS reduced PHA-induced IL-4 and IL-10 release from T-cells, whereas IFN-y release was affected only by PFOS. In all instances, PFOS was more potent than PFOA. Using siRNA, a role for PPARa in PFOA-induced immunotoxicity could be shown, whereas an inhibitory effect on LPS-induced I-kB degradation was identified that could explain the immunomodulatory effect of PFOS, suggesting different mechanisms of action (Corsini et al., 2011).

Maddalon et al. (2023b) showed that PFOS (free acid) reduced RACK-1 expression in a recent report. RACK-1 is a kinase involved in immune function and cytokine expression.

Recently, <u>Zhang et al. (2023e)</u> used a systemic evidence map (SEM) approach and found 1155 studies showing that PFAS were involved in either immune effects or chronic inflammation, of which 321 qualified for inclusion in their data set. The SEM showed decreased B-cell activation and altered levels of T-cell subtypes and immunoglobulins, confirming PFAS-induced immunosuppression (<u>Zhang et al., 2023e</u>).

Cell death is often observed after immunosuppressant treatment, explaining the observations of lymphoid organ atrophy. Adult male C57BL/6 mice treated with PFOS (potassium salt; purity, > 98%) at 0, 1, 5, or 10 mg/kg per day by gavage daily for 7 days had more apoptotic cells than in control mice. The PFOS-induced production of ROS and alteration in mitochondrial membrane potential could lead to the apoptosis of splenocytes and thymocytes (Zhang et al., 2013d). In adult C57BL/6 mice dosed daily by oral gavage with PFOS (potassium salt; purity, > 98%) at 0, 0.0167, 0.0833, or 0.8333 mg/kg per day, yielding target PFOS TADs of 0, 1, 5, or 50 mg/kg, respectively, for 60 days, PFOS induced p53-dependent apoptosis through Bcl-xl downregulation, without changing Bcl-2 or Bax expression (Dong et al., 2012b). The release of cytochrome c and activation of caspase-3 confirmed the involvement of the mitochondria.

Synopsis

[The Working Group noted that in exposed humans, PFOA and PFOS have been found to be associated with the augmentation of LRTIs and diarrhoea and reduced vaccination efficacy in children in several environmental studies conducted in different countries. An association between PFOA and a reduced response to influenza vaccination has recently been identified in adults, and also for SARS-CoV-2 vaccination in a maternal cohort. The limitations of environmental studies are mainly associated with the use of self-reported data in questionnaires and the existence of confounding factors. Importantly, these observations were supported by evidence that PFOA and PFOS affect the response of human primary immune cells (proliferation and cytokine production). Moreover, evidence of immunosuppression after exposure to PFOA and PFOS, which mainly affected the TDAR (T-celldependent antibody response), has also been reported in several animal experiments.]

4.2.8 Modulates receptor-mediated effects

See Tables S4.23–S4.26 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>), and <u>Table 4.27</u>.

With regard to the modulation of receptor-dependent pathways in humans, most epidemiological data have come from studies on the receptor ligands (e.g. changes in the serum concentrations of ligands that activate receptor-dependent pathways) or hormones that regulate the expression of receptor ligands (e.g. thyroid-stimulating hormone TSH and luteinizing hormone LH). The epidemiological studies considered here can be found in Table S4.23. Human in vitro studies have investigated the activation/antagonism of multiple nuclear receptors, with a focus on the activation of PPARa. However, rodent studies have made it clear that PFOA and PFOS target more than just PPARa (Rosen et al., 2010; Attema et al., 2022; Su et al., 2022a). Table S4.24 summarizes the available studies that investigated the ability of PFOA and PFOS to bind and/or activate nuclear receptors in human systems in vitro. Evidence was gathered from transcriptional and biological assays performed in human primary hepatocytes and human liver cell models, as well as binding and reporter assays.

(a) Humans

(i) Thyroid hormone pathway

Exposed humans

See Table S4.23 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Child-adult exposure

Several studies have investigated the associations of PFOA and PFOS with markers of thyroid function and disease in children and adults. [The Working Group selected the most informative studies, based on a prospective study design, a high level of exposure (including occupational and contaminated community exposure), and large study size.]

Two prospective cohort studies of female and male adults (USA) and children (Republic of Korea) have investigated the associations of serum PFOA and PFOS levels with TSH, free triiodothyronine (FT3), total triiodothyronine (TT3), free thyroxine (FT4), and/or total thyroxine (TT4) (<u>Blake et al., 2018</u>; <u>Kim et al., 2020</u>).

In the study conducted in adults, serum collected at the initial enrolment examination and at subsequent follow-up examinations during the years 1991–2008 was analysed for PFOA, PFOS, TSH, and TT4 levels. An IQR increase in serum PFOS, but not PFOA, was associated with a 9.75% (95% CI, 1.72%–18.4%) increase in TSH, but neither PFOA nor PFOS was associated with TT4 (Blake et al., 2018). [The Working Group noted, however, that due to a change in protocol, fewer repeated TT4 measurements than TSH measurements were available. This was also a highly exposed community.]

In children, TSH levels were measured at age 2, 4, and 6 years. The relation of serum PFOA and PFOS concentrations with TSH levels at the three time points was assessed by repeated-measures analysis using linear mixed models. Serum levels of FT4 and T3 were measured once (at age 6 years). PFOA was associated with a low TSH level only in male participants, and no association was detected for PFOS (Kim et al., 2020). The PFOA level at age 6 years was not associated with T3, but it was associated with high FT4 at age 6 years, primarily in male participants (Kim et al., 2020). The PFOS level at age 6 years was associated with high levels of T3 at the same age, primarily in male participants, but was not associated with FT4 (Kim et al., 2020). Age 6 years was the one time point at which a significantly higher PFOS concentration was reported in male than female participants (Kim et al., 2020).

There have been several cross-sectional studies conducted in highly exposed populations. A cross-sectional study of highly exposed adult male workers in a fluorochemical plant, conducted by the industry, showed that their serum PFOA levels were associated with a decrease in FT4 within the normal reference range, without associations with TSH, T3, or TT4 (Olsen and Zobel, 2007). A cross-sectional study

of highly exposed adult female and male workers exposed to fluorochemicals from two manufacturing facilities showed that the serum PFOS levels in men from all the locations combined were associated with an increase in T3, but no associations were seen with TSH, FT4, or TT4. No associations of the serum PFOS levels with TSH, FT3, FT4, or TT4 were found in women in another industry-conducted study (Olsen et al., <u>2003b</u>). Large cross-sectional studies (n > 10000) have reported few associations of serum PFOA or PFOS levels with TSH or thyroid hormones. A cross-sectional study of males and females aged 14-39 years in the Veneto region of Italy found no associations of serum PFOA or PFOS levels with TSH (Gallo et al., 2022). [The Working Group noted that this was an analysis of a highly exposed community.]

In a large study of children and adolescents from a community contaminated by PFOA from a fluorochemical-production plant (the C8 study cohort in the Mid-Ohio Valley, USA), serum PFOS, but not PFOA, was associated with a small (1.1%) IQR increase in TT4 (Lopez-Espinosa et al., 2012). Neither the PFOA nor the PFOS serum level was associated with TSH (Lopez-Espinosa et al., 2012).

Similarly, moderately sized cross-sectional studies (n = 1000-2000) have reported associations between serum PFOA or PFOS levels and TSH and thyroid hormones. In a study of cross-sectional data from male and female adolescents, adults, and older adults in the USA (NHANES 2007–2008), serum PFOA level was associated with increased TSH and TT3 levels, but not with free iodothyronine (FT3), FT4, or TT4 (Jain, 2013). No associations of serum PFOS level with TSH, FT3, TT3, FT4, or TT4 were found (Jain, 2013). However, when low iodine and high thyroid peroxidase antibody (TPOAb) levels were taken into account in a subset of the same study sample, serum PFOA and PFOS levels were associated with increases in TSH, FT3, and TT3, and PFOS was negatively associated with FT4.

The authors observed that in the T0I0 group, with normal TPOAb and iodine concentrations, serum PFOA was associated with increased FT3 levels (1.2%; 95% CI, 0.1–2.4%) (Webster et al., 2016).

A study of NHANES data from the 2007–2008 and 2009–2010 cycles stratified by age and sex reported that whereas PFOA was associated with an increase in FT3 in both sexes, it was associated with an increase in TT3 only in women (Wen et al., 2013). PFOA was not found to be associated with TSH or FT4/TT4, and the PFOS level was not found to be associated with TSH or thyroid hormones in this study (Wen et al., 2013).

A similar analysis of NHANES data from the 2011–2012 participants showed that the PFOA level was not associated with TSH, FT3, TT3, FT4, or TT4, and that the PFOS level was only associated with an increase in FT4 (van Gerwen et al., 2020). However, when analysed by sex and age, the PFOA level in the same cohort was associated with reduced TSH (in females aged 12–19 years), increased TT3 and FT3 (in women aged 60-80 years), and increased FT4 (in women aged 20-39 years). PFOA was not associated with TSH or thyroid hormones in male participants in any age group (Lewis et al., 2015). The PFOS levels in the 2011–2012 NHANES cohort were reported to be associated with increased TSH (in male participants aged 12-19 years) and increased FT4 (in women aged 20–39 years). PFOS was not associated with TT3, FT3, or TT4 in male or female participants, nor with FT4 in male participants (<u>Lewis et al., 2015</u>).

Lastly, in a cross-sectional study of older (aged 63.5 ± 13.6 years, mean \pm SD) women and men, the PFOA serum level was associated with lower TSH concentrations and higher FT4 concentrations, but was not associated with FT3 (Li et al., 2022d). The PFOS serum level in this study was associated with lower TSH and FT3, but higher FT4 (Li et al., 2022c). In contrast, in a study incorporating cross-sectional data from adolescents, adults, and older adults, no associations of PFOA

or PFOS with TSH or FT4 levels were observed (Ji et al., 2012).

Several smaller studies (n = 31-633) have been conducted to investigate associations of the PFOA and PFOS serum levels with TSH and thyroid hormones in children and adults. [The Working Group considered these studies to be moderately informative, because of the smaller number of participants and/or the lower PFOA or PFOS serum levels.]

No association between the PFOA serum level and TSH was found in the majority of the small studies (<u>Bloom et al., 2010; Ji et al., 2012; Raymer</u> et al., 2012; Lin et al., 2013; <u>Shrestha et al., 2015;</u> <u>Crawford et al., 2017; Li et al., 2017c; Kang et al.,</u> 2018; <u>Khalil et al., 2018; Caron-Beaudoin et al.,</u> 2019; Li et al., 2021b, women only).

Two studies reported that the PFOA level in serum was associated with increased TSH level (Byrne et al., 2018; Liu et al., 2022b). One study reported that the PFOA level in adolescent male participants was associated with decreased TSH level (Li et al., 2021b).

No association between the PFOA serum level and TT3 was found in the majority of the small studies (Raymer et al., 2012; Shrestha et al., 2015; Byrne et al., 2018; Liu et al., 2022b). One study reported that the PFOA serum level in adult women was associated with increased TT3 level (Crawford et al., 2017). In the four small studies of adults that examined the relation between PFOA serum level and FT3 level, no association was found (Li et al., 2017c, 2021b; Byrne et al., 2018; Liu et al., 2022b).

In the six small studies in adults that have examined the relation between the PFOA level and TT4 level, no association was found (Ji et al., 2012; Raymer et al., 2012; Shrestha et al., 2015; Crawford et al., 2017; Byrne et al., 2018; Liu et al., 2022b).

No association between the PFOA serum level and FT4 was found in the majority of the small studies (<u>Bloom et al., 2010; Lin et al., 2013;</u> <u>Shrestha et al., 2015; Crawford et al., 2017; Li</u> et al., 2017c, 2021b; Byrne et al., 2018; Kang et al., 2018; Khalil et al., 2018; Caron-Beaudoin et al., 2019; Liu et al., 2022b).

No association between the PFOS serum level and TSH was found in the majority of the small studies (Bloom et al., 2010; Ji et al., 2012; Raymer et al., 2012; Lin et al., 2013; Shrestha et al., 2015; Crawford et al., 2017; Li et al., 2021b; Byrne et al., 2018; Kang et al., 2018; Khalil et al., 2018; Caron-Beaudoin et al., 2019; Liu et al., 2022b).

One study reported that the PFOS level was associated with increased TSH level (<u>Li et al., 2017c</u>).

Another study reported that the PFOS levels in adult men and women were associated with reduced TSH level (Dallaire et al., 2009). No association between the PFOS serum level and TT3 was found in the majority of the small studies (Shrestha et al., 2015; Crawford et al., 2017; Byrne et al., 2018; Liu et al., 2022b). One study reported that the PFOS levels in adult men and women were associated with reduced TT3 level (Dallaire et al., 2009). One study reported that the PFOS level in adult males was associated with increased TT3 level (Raymer et al., 2012). In the majority of the small studies in adults, the PFOS serum level was not associated with the FT3 level (Byrne et al., 2018; Li et al., 2021b; Liu et al., 2022b). One study reported that the PFOS levels in adult women and men were associated with reduced FT3 (Li et al., 2017c). In the majority of small studies in adults, the PFOS serum level was not associated with the TT4 level (Ji et al., 2012; Crawford et al., 2017; Byrne et al., 2018; Liu et al., 2022b). One study reported that the PFOS level in adult men was not associated with TT4 (Raymer et al., 2012), and another reported that the PFOS levels in adult women and men were associated with increased TT4 (Shrestha et al., 2015). No association between the PFOS serum level and FT4 was found in the majority of small studies (Bloom et al., 2010; Lin et al., 2013; Crawford et al., 2017; Byrne et al., 2018; Kang et al., 2018; Khalil et al., 2018; Caron-Beaudoin et al., 2019;

Li et al., 2021b). Three studies reported that the PFOS levels in adult women and men were associated with increased FT4 (Dallaire et al., 2009; Shrestha et al., 2015; Liu et al., 2022b). One study reported that the PFOS levels in adult men and women were associated with reduced FT4 (Li et al., 2017c).

[The Working Group noted that a recent meta-analysis of 12 epidemiological studies showed a significant association between blood levels of PFOS and increased FT4, although the effect size was small (pooled z value, 0.05; 95% CI, 0.03–0.08) (Kim et al., 2018). No associations were found of PFOS with TT4, TT3, or TSH; no associations were found between PFOA and TSH or thyroid hormones (Kim et al., 2018).]

Maternal exposure/maternal hormone status

A large number of studies have been conducted on serum PFOA and PFOS concentrations during pregnancy and associations with thyroid hormone outcomes in mothers and newborns. No prospective or large $(n > 10\ 000)$ cross-sectional studies have been conducted; however, moderately sized cross-sectional studies (n = 1000-2000) have been conducted. The Working Group thus considered these to be the most informative.] A cross-sectional study in Sweden tested the association of the maternal PFOA and PFOS levels (blood collected during the first and second trimester, median, 10 weeks) with TSH, FT3, TT3, FT4, and TT4. Neither PFOA nor PFOS were associated with TSH, and PFOA was only associated with higher FT4. PFOS was associated with a lower TT3 and a non-linear (inverted U) increase in TT4 (Derakhshan et al., 2022).

Similarly, a cross-sectional study conducted in Shanghai, China, which tested the associations of the maternal PFOA and PFOS levels (blood collected during the first and early second trimesters) with TSH, FT3, and FT4, reported that PFOA was associated with an increase in FT4 (<u>Aimuzi et al., 2020</u>).

A cross-sectional study conducted in Odense, Denmark, which examined the associations of the maternal PFOA and PFOS (blood collected during the first and second trimesters) with TSH and FT4, found that both PFOA and PFOS were associated with an increase in FT4 (Jensen et al., 2022).

In contrast, a cross-sectional study conducted in Demark that tested the associations of the maternal PFOA and PFOS levels with TSH and FT4 during the first and second trimesters reported no associations (Inoue et al., 2019). [The Working Group noted that haemodilution occurs during the early third trimester (\geq 30 weeks of gestation) (Assali and Brinkman, 1972). Moreover, because in all these studies blood was collected no later than the late second trimester (< 27 weeks of gestation), the Working Group considered that haemodilution could not be a factor that would have had an impact on the outcomes.]

Several smaller studies (n = 152-919) have been conducted to investigate associations of the maternal PFOA and PFOS serum levels with TSH and thyroid hormones. [The Working Group considered these studies to be moderately informative, because of the smaller number of participants and/or the lower PFOA or PFOS serum levels.]

In seven of the eight small studies of mothers, no association between the PFOA serum level and TSH was found (<u>Berg et al., 2015; Kato et al.,</u> 2016; <u>Preston et al., 2018; Itoh et al., 2019; Reardon et al., 2019; Xiao et al., 2020; Sarzo et al., 2021</u>), but one showed an increase in TSH in women with high TPOAb titres (<u>Webster et al., 2014</u>).

No association was found between maternal PFOA level and TT3 in one study (<u>Berg et al.</u>, 2015), but in another study a decrease in TT3 was reported (<u>Sarzo et al.</u>, 2021).

In the four small studies of mothers, no association between the PFOA serum level and FT3

was found (Berg et al., 2015; Itoh et al., 2019; Reardon et al., 2019; Xiao et al., 2020). In the four small studies of adults that examined the relation between PFOA level and TT4 level, no associations were found (Berg et al., 2015; Preston et al., 2018; Xiao et al., 2020). In the seven small studies that examined the relation between PFOA level and FT4, no associations were found (Webster et al., 2014; Berg et al., 2015; Kato et al., 2016; Itoh et al., 2019; Reardon et al., 2019; Xiao et al., 2020; Sarzo et al., 2021).

Mixed results have been reported on the association between the maternal PFOS serum level and TSH. Four studies have reported no association (Preston et al., 2018; Itoh et al., 2019; Sarzo et al., 2021; Xiao et al., 2020). Three studies have reported that the maternal PFOS level was associated with an increase in TSH (Berg et al., 2015; Reardon et al., 2019; Xiao et al., 2020), although one of these only reported this association in people with high TPOAb titres (Webster et al., 2014). One study reported that the maternal PFOS level was associated with low TSH only in TPOAb-positive participants (Preston et al., 2018). One study reported that the maternal PFOS level was associated with reduced TSH (Kato et al., 2016). The two studies that investigated the relation between the maternal PFOS serum level and TT3 reported no association (Berg et al., 2015; Sarzo et al., 2021). In the four small studies that examined the relation between the maternal PFOS serum level and FT3 level, no associations were found (Berg et al., 2015; Itoh et al., 2019; Reardon et al., 2019; Xiao et al., 2020). No association between the PFOS serum level and TT4 was found in the three studies that examined this relation (Berg et al., 2015; Preston et al., 2018; Xiao et al., 2020). No association between the PFOS serum level and FT4 was found in the seven small studies that tested this relation (Webster et al., 2014; Berg et al., 2015; Kato et al., 2016; Itoh et al., 2019; Reardon et al., 2019; Xiao et al., 2020; Sarzo et al., 2021).

Maternal exposure/neonate hormone status

Studies of the associations of prenatal PFOA or PFOS serum levels (measured in the neonatal cord serum) with neonatal TSH and thyroid hormones have also been conducted.

No prospective, large $(n > 10\ 000)$ or moderately sized (n > 1000) cross-sectional studies have been conducted; however, several studies with 300-700 participants have been conducted. [The Working Group selected the largest available studies and/or those with the highest PFOA or PFOS levels as being the most informative.] Three cross-sectional studies measured both PFOA or PFOS and hormone levels in neonatal cord blood, two of which were conducted in China and one in the Republic of Korea. Guo et al. (2021c) assessed the associations of PFOA or PFOS with TSH, TT3, FT3, TT4, and FT4 in cord sample samples. [The Working Group noted that the cord blood PFOA or PFOS levels may be a more reliable measure of exposure over the full course of gestation, rather than a single maternal sample that may be affected by haemodilution.] Associations were found only with thyroxine.

The neonatal cord blood PFOA level was associated with increased TT4, and the PFOS level was associated with increased TT4 and FT4 (Guo et al., 2021c). Aimuzi et al. (2019) also assessed the associations of PFOA or PFOS with TSH, FT3, and FT4 in cord blood samples. PFOA was associated with an increase in FT4 and a decrease in FT3; however, this only occurred in male neonates (Aimuzi et al., 2019). PFOS was associated with reduced TSH and increased FT3 across all the neonates and increased FT3 in the male neonates only (Aimuzi et al., 2019). In contrast, no association was reported of the neonatal cord serum PFOA or PFOS level with TSH, TT3, or TT4 in a small study (n = 43) (Kim et al., 2011).

A cross-sectional study conducted in Japan tested the association of the maternal PFOA and PFOSlevels during the first trimester of pregnancy

with neonatal cord blood thyroid hormone levels (TSH, FT3, and FT4). Only one association was found: maternal PFOS was directly associated with an increase in TSH in male neonates (Itoh et al., 2019). A second study in Japan, which tested the association of the maternal PFOA and PFOS levels during the second or third trimester with neonatal TSH and FT4 in blood collected by heel puncture, also reported that maternal PFOS was associated with an increase in the cord blood TSH concentration in all the newborns (Kato et al., 2016). A cross-sectional study conducted in China tested the relationship between the maternal PFOA or PFOS levels during the second trimester and the neonatal cord blood thyroid hormone levels (TSH, TT3, FT3, TT4, and FT4). The maternal PFOA and PFOS levels were both associated with increased neonatal cord blood TT3 and FT3, but no associations were found for TSH or thyroxine (Liang et al., 2020). A cross-sectional study conducted in the USA that evaluated the relation between maternal PFOA and PFOS levels during the first and second trimesters with T4 levels in newborn heel-puncture blood found no associations (Preston et al., 2018).

The remaining studies on the relations between maternal PFOA or PFOS levels and neonatal TSH and thyroid hormones were very small (n < 200), and therefore the Working Group considered them to be of only moderate importance. In the one study that analysed associations of the maternal PFOA and PFOS levels during the third trimester with neonatal cord blood hormones (TSH, TT3, FT3, TT4, and FT4), TSH was the only hormone that was associated with the maternal PFOA and PFOS levels, with both PFOA and PFOS being associated with an increase in TSH (Xiao et al., 2020). In the one study that analysed maternal PFOA and PFOS levels during the first and second trimesters and neonatal cord blood hormones (TSH, TT3, FT3, TT4, and FT4), the only association that was found was that maternal PFOA and PFOS levels were both associated with lower FT4 levels in

neonates, but only when the neonates were born to mothers with high TPOAb levels (<u>Lebeaux</u> <u>et al., 2020</u>). The cord blood PFOA level, but not that of PFOS, was positively associated with newborn heel puncture TT4 in female, but not in male, neonates (<u>de Cock et al., 2014</u>). Lastly, the neonatal dried blood spot PFOA and PFOS concentrations were positively associated with T4 and negatively associated with TSH (<u>Rosen</u> <u>Vollmar et al., 2023</u>).

Human cell lines and reporter assays

Few studies have investigated the thyroid hormone receptor (TR) pathway in human in vitro models. Human thyroid peroxidase activity in FTC-238 human follicular carcinoma cells (48-hour exposure) was consistently inhibited by PFOS (≥ 0.1 nM), but by only a high concentration of PFOA (1 μ M) (Song et al., 2012). PFOA (IC₅₀ = 1.8–3 μ M, in cell-free transthyretin preincubation) and PFOS (IC₅₀ = 0.6 μ M) displaced T4 from transthyretin (TTR) in a TTR-TR β CALUX assay performed in U2OS human osteosarcoma cells (Behnisch et al., 2021; Sprengel et al., 2021). PFOS ($\geq 0.1 \ \mu$ M) antagonized TR β activation by T3 in a one-hybrid reporter assay performed in CV-1 cells (Du et al., 2013) (see Table S4.23).

Synopsis

[The Working Group noted that the evidence for associations of PFOA and PFOS with TSH or thyroid hormones in children and adults was generally weak (see summary findings for KC8 in <u>Table 4.27</u>). In the most informative studies (prospective study design, high level of exposure (including occupational and contaminated community exposure), or large study size), the results were mixed, with some pattern to the reported associations of PFOA or PFOS levels with the TSH or thyroid hormone levels being identified. The strongest associations were between the PFOS serum level and increased FT4; however, the collective evidence (for TSH, TT3, FT3, and TT4) did not support an association with the PFOA or PFOS serum level. Importantly, changes in thyroid hormones were rarely accompanied by clinically relevant changes in TSH level in adults or children. Also, there was little evidence about the effects of PFOA and PFOS on the human thyroid hormone receptor (TR) pathway in human cells in vitro.]

Steroid receptors – androgen and estrogen pathways

Exposed humans

See Table S4.23 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Child-adult exposure

Several studies have investigated the associations of PFOA and PFOS with LH, testosterone, and estrogens in children and adults. LH is responsible for stimulating testosterone production in the gonads, and testosterone is converted to estrogens through the action of aromatase. Therefore, these hormones are analysed together. [The Working Group selected the most informative studies based on a prospective study design, a high level of exposure (including occupational and contaminated community exposure), and large study size.] A single prospective cohort study of pre-adolescent girls (n = 704) in the USA found no associations between serum PFOA and estradiol pre- or post-thelarche but did show negative associations with estrone and testosterone 6 months before thelarche (Pinney et al., 2023). The same study showed no associations of serum PFOS with estradiol, estrone, or testosterone at any time point pre- or post-thelarche (Pinney et al., 2023). A cross-sectional study (n = 1041) of men aged 18–21 years in Denmark found no associations of serum PFOA or PFOS with LH, testosterone, or estradiol (Petersen et al., 2022). Similarly, a cross-sectional study (n = 920) of men aged 28–25 years in Shanghai,

China, found no associations of serum PFOA with LH, testosterone, or estradiol (Luo et al., 2021). PFOS was also found to not be associated with LH or testosterone, but PFOS was reported to be negatively associated with estradiol (Luo et al., 2021). Two cross-sectional studies have been conducted of the C8 cohort, composed of participants who lived in contaminated communities in the Mid-Ohio Valley, USA. In a study of children aged 6–9 years (n = 2292), serum PFOA was positively associated with lower testosterone only in male children, whereas serum PFOS was associated with lower testosterone in both girls and boys (Lopez-Espinosa et al., 2016). In the same study, PFOA was not associated with estradiol in male or female children, but PFOS was associated with lower estradiol in boys (Lopez-Espinosa et al., 2016). In a large cross-sectional analysis of adult women in the C8 cohort (n = 29.957), serum PFOA was not found to be associated with estradiol, but serum PFOS was found to be associated with lower estradiol in women aged > 42 years (Knox et al., 2011).

In a cross-sectional study of young men (age 24–26 years; n = 263) in the Faroe Islands, Denmark, the PFOA level was not found to be associated with LH, free or total testosterone, or estradiol; however, the PFOS level was reported to be associated with an increase in LH (Petersen et al., 2018).

In a cross-sectional analysis of NHANES data (USA) regarding people aged > 12 years (n = 1682) from the 2011–2012 cohort, no association was found in any group between the PFOA or PFOS level and the total serum testosterone (Lewis et al., 2015). However, in an analysis of the NHANES 2015–2016 cohort, including people aged > 12 years (n = 1886), PFOS, but not PFOA, was found to be associated with higher testosterone levels in male participants only (Xie et al., 2021). PFOA was associated with a linear decrease in estradiol only in women aged 20–49 years, whereas PFOS concentrations in the second quartile only were found to be associated

with higher estradiol in Q2 in women aged 20-49 years (Xie et al., 2021).

In another cross-sectional study of women in the USA, in which the participants aged 47–52 years had a median serum PFOS concentration of 25 ng/mL, neither the PFOA nor the PFOS serum level was found to be associated with testosterone or estradiol (Harlow et al., 2021).

In a cross-sectional study of male and female adolescents aged 13–15 years in Taiwan, China, in whom the median serum PFOS concentration was ≥ 28 ng/mL, neither PFOA nor PFOS was associated with testosterone or estradiol in the female participants (Zhou et al., 2016). In the male participants, PFOA was associated with higher estradiol, and PFOS was associated with lower testosterone (Zhou et al., 2016).

In a further analysis of these participants within a larger study, the authors reported that specifically in adolescents with asthma, both PFOA and PFOS were associated with lower testosterone and higher estradiol (Zhou et al., 2017).

Several small cross-sectional studies (n = 59-651) have also been conducted. [The Working Group considered these studies to be moderately informative, because of the smaller number of participants and/or the lower PFOA or PFOS serum levels.] The majority of studies found no association of serum PFOA or PFOS with LH (Joensen et al., 2009; Joensen et al., <u>2013; Tsai et al., 2015; Cui et al., 2020</u>). One study reported that PFOA was associated with an increase in LH in young men, whereas PFOS showed no association with LH (Raymer et al., <u>2012</u>). Mixed results have been reported for the relation of serum PFOA with testosterone (free or total), including no association (Joensen et al., <u>2009</u>, <u>2013</u>; <u>Tsai et al., 2015</u>), a positive association (Raymer et al., 2012; Heffernan et al., 2018; Wang et al., 2021b), and a negative association (<u>Cui et al., 2020</u>).

Mixed results also have been reported for the relation of serum PFOS with testosterone (free or total), including no association (Joensen et al., 2009; Raymer et al., 2012; Heffernan et al., 2018), a positive association (Wang et al., 2021b), and a negative association (Joensen et al., 2013; Tsai et al., 2015; Cui et al., 2020). None of these studies reported an association between serum PFOA or PFOS with estradiol (Joensen et al., 2009, 2013; Raymer et al., 2012; Barrett et al., 2015; Tsai et al., 2015; Heffernan et al., 2018; Cui et al., 2020; Wang et al., 2021b).

Two cross-sectional studies in Italy investigated the associations of PFOA and PFOS serum levels with the expression of androgen and estrogen receptors in leukocytes.

In a study of fertile and infertile women (age 18–40 years; n = 111), PFOA was not found to be associated with sex hormone receptor mRNA expression, whereas PFOS was reported to be associated with higher androgen receptor (AR) expression (Caserta et al., 2013). In a study of fertile and infertile men (age 27–40 years; n = 153), only PFOA was associated with lower expression of both estrogen receptors and androgen receptors (La Rocca et al., 2015).

[The Working Group noted that a recent meta-analysis of 11 studies examined the associations of PFOA and PFOS with reproductive hormones (estradiol and total testosterone) (Li et al., 2024). PFOS was associated with reduced serum testosterone in men, although the effect size was notably small, and no association was found in women (Li et al., 2024). However, earlier systematic analyses did not report significant associations between PFOS and low testosterone (Bach et al., 2016; Petersen et al., 2020). No associations of PFOA or PFOS with estradiol were reported by Li et al. (2024).]

Prenatal exposure

Several studies of prenatal exposure to PFOA and PFOS have investigated their associations with intermediates in the sex hormone synthesis pathway, as well as with testosterone and estradiol. No prospective cohort or large (n > 10000) cross-sectional studies were available to the Working Group.

In a cross-sectional study of amniotic fluid in Denmark (n = 645; male fetuses only), PFOS was found to be associated with increased progesterone, 17-hydroxyprogesterone, androstene-dione, and testosterone (PFOA was not analysed) (Toft et al., 2016).

In a cross-sectional analysis of the association between maternal serum PFOA and PFOS concentrations during the third trimester and the male and female neonatal cord blood hormones (Shandong, China; n = 349), neither maternal PFOA nor PFOS were reported to be associated with neonatal testosterone (Yao et al., 2021). Maternal PFOA, but not PFOS, concentration during pregnancy was found to be associated with higher levels of neonatal estradiol (Yao et al., 2021).

A second cross-sectional analysis of the association between maternal serum PFOA and PFOS concentrations during the third trimester and male and female neonatal cord blood hormones (Sapporo, Japan; n = 224) similarly reported that neither maternal PFOA nor PFOS concentrations during pregnancy were associated with neonatal testosterone (Kobayashi et al., 2021). In the study by Kobayashi and colleagues, maternal concentration of PFOS, but not PFOA, during pregnancy was reported to be associated with increased neonatal levels of estradiol (Kobayashi et al., 2021).

Two cross-sectional studies in Hubei, China, investigated the associations of neonatal cord blood PFOA and PFOS with neonatal estrogens (estrone, estradiol, and estriol) in male and females. In the larger of the two studies (n = 942), both neonatal PFOA and PFOS were positively associated with neonatal cord blood estrone and estradiol (Liu et al., 2021). However, only PFOA was associated with higher estriol (Liu et al., 2021). In the second study (n = 424), neonate PFOA was

positively associated with only estrone, whereas PFOS was associated with higher estrone and estriol (<u>Wang et al., 2019</u>).

In a cross-sectional study of neonatal cord blood PFOA and PFOS and sex hormones in Shandong Province, China (n = 351), PFOA was reported to be positively associated with cord blood estradiol, but not total testosterone, and PFOS was positively associated with cord blood total testosterone, but not estradiol (Yao et al., 2019).

There have been a few other moderately informative, smaller (n = 72-373) cross-sectional studies of associations between maternal PFOA and PFOS levels during pregnancy and steroid hormones in neonates, children, and young adults. The maternal PFOA concentration after the second trimester was negatively associated with the testosterone precursor dehydroepiandrosterone (DHEA) in neonatal cord blood, and maternal PFOS was positively associated with neonatal DHEA (Goudarzi et al., 2017b). Neither maternal PFOA nor PFOS concentrations during pregnancy were associated with neonatal androstenedione (Goudarzi et al., 2017b). Two cross-sectional studies have tested the associations of the maternal serum PFOA and PFOS concentrations during pregnancy with anatomical biomarkers of prenatal hormone exposure: the ratio of the lengths of the second and fourth digits (2D:4D; negatively associated with testosterone exposure and positively associated with estrogen exposure) and anogenital distance (positively associated with testosterone exposure). The maternal serum PFOA concentration during the second and third trimesters, but not that of PFOS, was associated with an increased mean 2D:4D digit ratio in male, but not female, neonates with different ESR1 genotypes (Nishimura et al., 2022).

No associations between the maternal serum PFOA or PFOS concentration during pregnancy and the anogenital distance of male or female neonates were observed (<u>Arbuckle et al., 2020</u>). No associations of the maternal serum PFOA or

PFOS concentration during the first and second trimesters with the infant (age 4 months) serum LH, DHEA, androstenedione, 17-hydroxyprogesterone, or testosterone were found (Jensen et al., 2020). The maternal PFOA and PFOS concentrations during the first and second trimesters were positively associated with testosterone in the adolescent daughters (Maisonet et al., 2015). However, the maternal PFOA and PFOS concentrations during the third trimester were not found to be associated with LH, total testosterone, DHEAS, or estradiol in the young adult daughters (Kristensen et al., 2013). The maternal PFOA and PFOS concentrations during the third trimester were not associated with testosterone or estradiol in the young adult sons, although PFOA was associated with higher LH (Vested et al., 2013).

In vitro effects on estrogen receptors a and β

See Table S4.24 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>)

Human primary cells

Only three studies have investigated the effects of PFOA on the estrogen receptor (ER) (ER α and ER β) pathway in human primary cells. In human primary hepatocytes exposed for 24 hours to PFOA ($\geq 25 \mu$ M), the expression of the ER α gene (*ESR1*) was significantly downregulated (Buhrke et al., 2015).

In HUVECs, PFOS (100 μ M, 48 hours) significantly increased the expression of the ERa gene (*ESR1*) (Liao et al., 2012). In human primary placental cytotrophoblasts, PFOS ($\geq 0.001 \mu$ M, 24 hours) reduced CYP19 protein expression and 17 β -estradiol secretion (Zhang et al., 2015b).

Human cell lines

In MCF7 human breast cancer cells, PFOA (100 μ M, 24–48 hours) did not affect the mRNA expression of the ER target genes *TFF1*, *EGR3*,

ESR1, GREB1, or PGR (Behr et al., 2018; Li et al., 2020e); however, PFOA significantly suppressed the estradiol-dependent expression of EGR3 and *TFF1* in one of these studies (Li et al., 2020e). In contrast, PFOS (50 µM, 24-48 hours) upregulated TFF1 but not EGR3 and suppressed the estradioldependent expression of these target genes (Li et al., 2020e). In the other study; however, PFOS (100 μ M, 24 hours) had no effect on *TFF1*, ESR1, GREB1, PGR, or CTSD expression (Behr et al., 2018). In T47D human breast cancer cells, whereas neither PFOA (0.001 μ M, 24 hours) nor PFOS (0.001 μ M, 24 hours) increased the expression of known ER target genes (*PR* and *pS2*), they did increase the estradiol-dependent induction of *pS2* (Sonthithai et al., 2016). In addition, PFOS induced the expression of ER β in HepG2 cells (<u>Xu et al., 2017</u>).

PFOA and PFOS have also been investigated for their ability to modulate estradiol production and secretion. In H295R human adrenocortical carcinoma cells, PFOA (48 hours) was reported to have no effect on 17β -estradiol secretion at \leq 100 µM (<u>Kraugerud et al., 2011</u>; <u>Wang et al.</u>, <u>2015d; Behr et al., 2018</u>), but to reduce 17β -estradiol secretion at 1.6 µM (Rosenmai et al., 2013), and to increase its secretion at 50 μ M (Rosenmai et al., 2013). In H295R cells, Kang et al. also reported increases in estradiol at 10 µM and 100 µM PFOA (Kang et al., 2016). PFOA (600 μ M, 48 hours) was shown to increase aromatase activity in H295R cells (Kraugerud et al., 2011) and to reduce aromatase activity $(IC_{50} = 80 \ \mu M)$ in JEG-3 human placental carcinoma cells (Gorrochategui et al., 2014). In H295R cells, PFOS was shown to have no effect on 17β-estradiol secretion at \leq 100 µM (<u>Behr et al.</u>, 2018), but to increase this at 0.03, 200, or 600 μ M (Kraugerud et al., 2011; Du et al., 2013; van den Dungen et al., 2015; Kang et al., 2016).

Human binding and reporter assays

PFOA and PFOS have been shown to bind human ERa at high concentrations in competitive binding assays using the human ERa ligand-binding domain with similar binding affinities (Qiu et al., 2020). The majority of ERa reporter studies employing endogenous human ERa and estrogen response element-driven reporters performed in breast and ovarian carcinoma cell lines (BG1, ovarian; MCF7 and T47D, breast) showed that PFOA and/or PFOS did not transactivate ERa (Yao et al., 2014; Kang et al., 2016; Sonthithai et al., 2016; Evans et al., 2022). However, other studies using this approach showed that PFOA (EC $_{50}$ = 65 $\mu M)$ and PFOS (EC₅₀ = 29 μ M) stimulated ERa transactivation (Kjeldsen and Bonefeld-Jørgensen, 2013) or that PFOS (EC₂₀ = 12 μ M), but not PFOA, stimulated ERa transactivation (Li et al., 2020e), but with significantly lower efficacy than did estradiol. In addition, PFOA (10 nM to 1 μ M) and PFOS (1 nM to 1 μ M) increased estradiol-stimulated ERa transactivation in an endogenous ERa-driven reporter assay performed in T47D cells (Sonthithai et al., 2016).

In ER reporter studies employing the forced expression of full-length human ERa in HEK293T human kidney cells, PFOA ($\geq 0.1 \, \mu M$) and PFOS ($\geq 0.001 \ \mu$ M) stimulated the transactivation of ERa (Benninghoff et al., 2011; Houck et al., 2021). ERa reporter studies using a one-hybrid approach have also been conducted. In one study of human HepG2 cells, PFOA (half-maximal activity concentration (AC₅₀ = 8 μ M) and PFOS (AC₅₀ = 4.4 μ M) stimulated ER α -mediated transactivation (Houck et al., 2021). In one study of human kidney HEK293T cells, neither PFOA $(\leq 100 \,\mu\text{M})$ nor PFOS $(\leq 100 \,\mu\text{M})$ transactivated ERa or ER β (Behr et al., 2018). However, in experimental designs in which PFOA and PFOS were applied together with estradiol, PFOA (100 µM) was shown to enhance estradiol-driven ER β activation, and PFOS (100 μ M) was shown to enhance estradiol-driven ER α and ER β activation in a one-hybrid assay in HEK293T cells (<u>Behr et al., 2018</u>).

In vitro effects on the androgen receptor

See Table S4.24 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Human primary cells

No studies available to the Working Group investigated the effects of PFOA or PFOS on androgen receptor (AR) pathways in human primary cells.

Human cell lines

In LNCaP human prostate adenocarcinoma cells (24-hour exposure), neither PFOA $(\leq 100 \ \mu\text{M})$ nor PFOS $(\leq 100 \ \mu\text{M})$ induced the mRNA expression of AR target genes (e.g. AR and PSA) (Behr et al., 2018). In H295R human adrenocortical carcinoma cells, PFOA has been reported not to have an effect on testosterone secretion (PFOA \leq 100 μ M, 48 hours) (<u>Rosenmai</u> et al., 2013; Wang et al., 2015d; Behr et al., 2018) or to increase testosterone secretion (PFOA \geq 0.6 µM) (<u>Kraugerud et al., 2011</u>). The effect of PFOS has also been tested using the H295R steroidogenesis assay. PFOS was reported to have no significant effect on testosterone secretion in H295R cells (PFOS \leq 200 μ M) (<u>Behr et al., 2018</u>), to reduce testosterone production (PFOS $\geq 1 \,\mu$ M) (Du et al., 2013), and to increase testosterone secretion (PFOS, 0.6-600 µM) (Kraugerud et al., 2011; van den Dungen et al., 2015).

Human binding and reporter assays

PFOA (\leq 300 µM) and PFOS (\leq 300 µM) did not transactivate endogenous human AR in reporter assays performed in breast (MDA-kB2) or prostate (22Rv1/MMTV) cancer cells (Kang et al., 2016; Behr et al., 2018) or HepG2 hepatoma cells (Houck et al., 2021). PFOS (\leq 3 µM) did not

transactivate endogenous human AR in reporter assays performed in breast (MDA-kB2) cancer cells (<u>Du et al., 2013</u>). PFOA (\leq 100 µM) did not transactivate full-length overexpressed human AR in a reporter assay performed in CHO cells (Kjeldsen and Bonefeld-Jørgensen, 2013; <u>Rosenmai et al., 2013</u>). PFOS (\leq 100 µM) also did not transactivate full-length overexpressed human AR in a reporter assay performed in CHO cells (Kjeldsen and Bonefeld-Jørgensen, <u>2013</u>). PFOS (> 50 μ M), but not PFOA, enhanced testosterone-induced AR reporter transactivation in MDA-kB2 cells (Behr et al., 2018), but PFOA (IC₅₀ = 11 μ M) and PFOS (IC₅₀ = 5 μ M) inhibited AR reporter transactivation in CHO cells (Kjeldsen and Bonefeld-Jørgensen, 2013). No studies that used binding assays were identified.

Synopsis

[Overall, the Working Group noted that in the studies in exposed humans reviewed above there was no evidence that PFOA or PFOS were associated with LH in children and adults, and there was little evidence that PFOA was associated with testosterone. Whereas the results of a recent meta-analysis (Li et al., 2024) showed that a small but significant decrease in serum total testosterone was associated with PFOS in men, the most informative and moderately informative studies reviewed here reported no association between PFOS and testosterone.

There was also little evidence provided by either the most informative or the moderately informative studies that PFOA is associated with estradiol level (either higher or lower). There were several studies that supported an association of PFOS with lower estradiol level; however, this association was not recapitulated in the moderately informative studies. There does appear to be the potential for sex-specific differences, with men being more likely to show associations of PFOA and PFOS with testosterone and estradiol than women. Human studies of in utero exposure to PFOA and PFOS did not show associations with sex hormones or sex hormone-dependent end-points in neonates or later in life. More studies reported no association than reported significant associations. However, most of the associations reported were between PFOA or PFOS and higher concentrations or effects of sex hormones.

In addition, PFOA and PFOS seemed not to modulate testosterone production or act as AR ligands, as measured with AR reporter assays.

Overall, inconsistent findings were reported with regard to PFOA and PFOS modulating the ER pathway-mediated effects in human primary cells, human cell lines, and human reporter assays. In systems in which ERs are expressed endogenously, high (i.e. non-human-relevant) PFOA and PFOS concentrations were required to modulate ER-dependent gene expression and transactivation, as well as estradiol production (see summary findings for KC8 in <u>Table 4.27</u>).]

(iii) Steroid receptors – progesterone pathway

See Table S4.23 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Exposed humans

One cross-sectional study conducted in Norway (n = 178) assessed the association between serum PFOA and PFOS and salivary progesterone in women and found that in nulliparous, but not parous, adult women, PFOS, but not PFOA, was negatively associated with salivary progesterone (<u>Barrett et al., 2015</u>).

One cross-sectional study conducted in Denmark (n = 545) assessed the association between PFOS and progesterone in amniotic fluid during the second trimester and reported that PFOS was associated with higher progesterone (Toft et al., 2016).

Two cross-sectional studies ($n \le 224$ and n = 189, respectively) have assessed the associations of prenatal exposure to PFOA and PFOS with neonatal progesterone levels. Maternal concentration of PFOS, but not PFOA, during the third trimester was negatively associated with cord blood progesterone in the first of these (Kobayashi et al., 2021). In the second study, the maternal pregnancy (from the first to third trimester) PFOA level was negatively associated with the cord blood progesterone in male and female neonates, whereas the maternal PFOS level was not associated with progesterone in neonates of either sex (Itoh et al., 2016).

A single cross-sectional study conducted in China (n = 374, women and men) investigated the relations between PFOA or PFOS and progesterone in neonatal cord blood and reported that neonatal levels of PFOA and PFOS were not associated with neonatal levels of progesterone or 17-hydroxyprogesterone (Liu et al., 2020b).

A single cross-sectional study conducted in Belgium (n = 170, women and men) investigated the associations between PFOA or PFOS in neonatal cord blood and the expression of progesterone-receptor target genes in leukocytes and found that the blood levels of PFOA and PFOS were associated with higher levels of progesterone receptor-mediated gene expression (Remy et al., 2016).

Synopsis

[The Working Group noted that, in the few studies described above, little evidence was available on the potential association of in utero exposure to PFOA and PFOS with the progesterone pathway. PFOS appeared to be more strongly associated with progesterone and progester-one-induced pathways than PFOA, but both positive and negative associations were reported (see summary findings for KC8 in <u>Table 4.27</u>).]

602

(iv) Steroid receptors – glucocorticoid pathway

See Table S4.23 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Exposed humans

A single cross-sectional study conducted in Sweden (n = 1048) assessed the association between maternal PFOA or PFOS exposure during the first trimester and the urinary cortisol and cortisone concentrations during the third trimester (Dreyer et al., 2020). The maternal serum PFOA was found not to be associated with urinary cortisol or cortisone. The maternal serum PFOS was associated with lower urinary cortisone but not cortisol (Drever et al., 2020). One cross-sectional study conducted in Denmark (n = 545) assessed the association between PFOS and cortisol in amniotic fluid and reported that PFOS was associated with higher cortisol (Toft et al., 2016). Two cross-sectional studies (n < 400) that assessed the associations of prenatal exposure to PFOA and PFOS with neonatal corticosteroid levels have been conducted. Maternal concentration of PFOS, but not PFOA, after the second trimester was negatively associated with cord blood cortisol, cortisone, and the cortisol/ cortisone ratio (Goudarzi et al., 2017b). Neonatal cord blood PFOA and PFOS were found to be positively associated with 11-deoxycortisol, but not cortisol or cortisone (Liu et al., 2020b).

<u>In vitro – human primary cells, cell lines, and</u> <u>reporter assays</u>

See Table S4.24 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

A single study has investigated the effect of PFOA or PFOS on the glucocorticoid receptor (GR) pathway in human primary cells. PFOS ($0.001-1 \mu$ M, 24 hours) reduced the upregulation
of 11 β -hydroxysteroid dehydrogenase (HSD)1, the enzyme responsible for the generation of cortisol, an endogenous GR ligand, in human primary decidual stromal cells (<u>Yang et al., 2016</u>).

Using microsomes from human and rat kidneys, PFOA and PFOS were demonstrated to inhibit the catalytic activity of 11 β -HSD2 (the enzyme that deactivates cortisol), with PFOS having a lower IC₅₀ than PFOA (0.05 versus 24.4 μ M, respectively) (Zhao et al., 2011b, 2023). In the same studies, rat enzyme activities were measured, and the PFOS IC₅₀ was 0.29 μ M, whereas that of PFOA was 3.8 μ M.

In H295R cells, PFOA ($\leq 600 \mu$ M, 48 hours) had no effect on cortisol secretion (<u>Kraugerud</u> et al., 2011; <u>Rosenmai et al., 2013</u>; <u>Wang et al.,</u> 2015d). In H295R cells, PFOS was shown to have no effect on cortisol secretion (PFOS $\leq 600 \mu$ M) (<u>Kraugerud et al., 2011</u>), nor to increase cortisol secretion (200 μ M) (<u>van den Dungen et al., 2015</u>). In a full-length GR reporter assay performed in T47D cells, PFOS (~30–60 μ M, 24 hours) enhanced cortisol-driven GR reporter activation to a modest extent (<u>Wilson et al., 2016</u>).

Synopsis

[The Working Group noted that there was a small number of available studies in which alteration of glucocorticoids and exposure to PFOA or PFOS during pregnancy were investigated, and few of these studies provided evidence that was supportive of a positive association. PFOS appeared to be more strongly associated than PFOA with corticosteroid levels, and the associations were generally negative (see summary findings for KC8 in Table 4.27).

A modest set of data suggested that PFOS suppressed cortisol production and deactivation but could enhance GR activation by cortisol, whereas PFOA had little effect on the GR pathway.]

(v) Other nuclear receptors and types of receptors

See Table S4.23 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Exposed humans

Several studies have investigated the association of PFOA and PFOS with vitamin D receptor-related biomarkers.

Two cross-sectional studies conducted in the USA, one of which was in adults (n = 7040)and one in children (n = 78), investigated the associations of serum PFOA and PFOS levels with total serum 25-hydroxyvitamin D (Khalil et al., 2018; Etzel et al., 2019). In adults, the PFOS level, but not that of PFOA, was associated with a lower serum vitamin D level (Etzel et al., 2019). In children, neither PFOA nor PFOS was associated with the serum vitamin D level (Khalil et al., 2018). A cross-sectional study of pregnant women in the USA (n = 442) investigated the associations of serum PFOA and PFOS levels during the first trimester with free and total serum 25-hydroxyvitamin D levels during the first and second trimesters and reported that serum PFOA level was not associated with the vitamin D level (Chang et al., 2021). In the same study, the serum PFOS level was associated with higher total serum 25-hydroxyvitamin D during both the first and second trimesters. The serum PFOS level was only associated with higher free serum 25-hydroxyvitamin D during the second trimester, and only in pregnancies with male fetuses (Chang et al., 2021).

In a cross-sectional study in China (n = 992) that investigated the associations of neonatal cord blood PFOA and PFOS levels with the total serum 25-hydroxyvitamin D, it was reported that the serum PFOS level, but not that of PFOA, was associated with higher neonatal levels of vitamin D (Liu et al., 2023c).

[The Working Group noted that PFOS appeared to be more associated with serum vitamin D level than PFOA; however, the direction of the association was negative in one study and positive in two studies.]

Three studies have investigated the associations of serum PFOA and PFOS in adult women and men with the expression of nuclear receptor and *AHR* mRNA expression in leukocytes.

In a cross-sectional study of adult women conducted in Italy (n = 154), serum PFOA was negatively associated with *PXR* and *AHR* expression in peripheral blood cells but was not associated with *PPARG* expression (Caserta et al., 2013). In the same study, the serum PFOS concentration positively correlated with *PXR* expression in peripheral blood cells (Caserta et al., 2013).

In a cross-sectional study of adult men conducted in Italy (n = 153), serum level of PFOA was negatively associated with the expression of *PXR* and *AHR* in peripheral blood cells but not with the expression of *PPARG* (La Rocca et al., 2015). In the same study, PFOS was not associated with the expression of *PXR*, *AHR*, or *PPARG* (La Rocca et al., 2015).

In a cross-sectional study of adult women and men conducted in the USA (n = 290), serum PFOA level was associated with lower expression of *LXRB*, but not *LXRA*, in peripheral blood cells, and the serum PFOS level was associated with lower expression of *LXRA*, but not *LXRB* (Fletcher et al., 2013). In the same study, no associations of PFOA or PFOS with *PPARA*, *PPARD*, or *PPARG* were reported (Fletcher et al., 2013).

[The Working Group noted that it is difficult to interpret the significance of associations between serum PFOA and PFOS levels and the expression of nuclear receptors in leukocytes, because of the small number of studies and disparate results.]

<u>PPARa – in vitro</u>

See Table S4.24 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

The majority of human in vitro studies have focused on the activation of PPARa. However, rodent studies have made it clear that PFOA and PFOS target more than just PPARa (<u>Rosen</u> <u>et al., 2010; Attema et al., 2022; Su et al., 2022a</u>). Table S4.24 summarizes the available studies that investigated the ability of PFOA and PFOS to bind and/or activate nuclear receptors in human systems in vitro. Evidence was gathered from transcriptional and biological assays performed in human primary hepatocytes and human liver cell models, as well as binding assays and reporter assays.

Human primary cells

Multiple studies in human primary liver cells have tested whether PFOA or PFOS activate human PPARa. In human primary hepatocytes treated with PFOA or PFOS for 24–28 hours, human *PPARA* mRNA expression was induced by PFOA ($\geq 5 \mu$ M) and PFOS ($\geq 25 \mu$ M) (Bjork and Wallace, 2009; Bjork et al., 2011; Rosen et al., 2013; Buhrke et al., 2015; Marques et al., 2022). Ingenuity Pathway Analysis of microarray mRNA expression data induced by PFOA identified PPARa as a predicted upstream regulator of effects on the cell cycle (Buhrke et al., 2015).

Similarly, in human primary liver spheroids composed of hepatocytes and Kupffer cells that were treated with PFOA or PFOS for 1–14 days and analysed by targeted RNA sequencing (RNASeq), the fatty acid β -oxidation pathway was significantly upregulated by PFOA (20 μ M) at all time points and by PFOS (20 μ M) after 1 and 4 days (<u>Rowan-Carroll et al., 2021</u>). PPARa was predicted to be a strongly activated upstream regulator of the gene expression changes induced by PFOA and PFOS (<u>Rowan-Carroll et al., 2021</u>). When comparing full transcriptomes, the benchmark concentrations for PFOA and PFOS were similar after 14 days of exposure (5–10 μ M); however, PFOS was found to regulate the transcription of more genes, in general, than PFOA (Rowan-Carroll et al., 2021).

In studies that tested the effects of PFOA or PFOS in both rodent and human primary hepatocytes, PPARa was less potently activated in human versus mouse primary hepatocytes: 5.1 times less potently for PFOA and 1.9 times less potently for PFOS (Rosen et al., 2013). Differences in efficacy have also been noted, with PFOA and PFOS more efficaciously activating PPARa in rat than human primary hepatocytes (although the authors noted that there were differences in the baseline expression of genes in rat and human hepatocytes that affected the apparent induction of expression) (Bjork et al., 2011).

Human cell lines

Similarly, studies of human liver cell lines have shown that PFOA and PFOS increase the mRNA expression of human PPARa target genes. In HepaRG human liver cells, 24-hour exposure to PFOA (100 µM) or PFOS (100 µM) upregulated the mRNA expression of known targets of PPARa, including those associated with fatty acid β -oxidation (Louisse et al., 2020, 2023; Murase et al., 2023). In addition, PFOS, but not PFOA, except at the highest concentration of 200 µM, was shown to increase lipid accumulation in HepRG cells (Louisse et al., 2020). In HepG2 human liver cancer cells, PFOA ($\geq 25 \,\mu$ M) had similar effects on the mRNA expression of PPARa target genes to the PPARa agonists WY14 643 and GW7647, although with lower potency (<u>Behr et al., 2020b</u>). In contrast, in HepG2/C3a human hepatoma cells, PFOA ($\leq 200 \ \mu$ M) and PFOS (25 μ M) did not induce the mRNA expression of the PPARa target genes ACOX1, ACOT, and CYP4A1 (Bjork and Wallace, 2009).

Human binding and reporter assays

The results obtained from human primary hepatocytes and other human cell models have been corroborated by ample evidence that PFOA and PFOS activate human PPARa derived from binding and reporter assays. In competitive binding assays performed using the human PPARa ligand-binding domain, PFOA and PFOS were shown to displace well-known PPARα ligands (Li et al., 2018b; Ishibashi et al., 2019). PFOA has been shown to activate PPARamediated transcription in reporter assays using full-length human PPARa (Maloney and Waxman, 1999; Nielsen et al., 2022; Sakai et al., <u>2022</u>) or chimaeras of the human PPARa ligandbinding domain with the GAL4 DNA-binding domain (Vanden Heuvel et al., 2006; Takacs and <u>Abbott, 2007; Wolf et al., 2008a, 2012; Corsini</u> et al., 2012; Buhrke et al., 2013; Rosenmai et al., 2016, 2018; Behr et al., 2020b; Houck et al., 2021; Evans et al., 2022). PFOS also has been shown to activate PPARa reporter assays (full-length: Shipley et al., 2004; Nielsen et al., 2022; Gal-4: Wolf et al., 2008a; Behr et al., 2020b; Houck et al., 2021; Evans et al., 2022). However, some studies also showed that PFOS failed to activate PPARa reporter assays (Gal-4: Takacs and Abbott, 2007; Corsini et al., 2012; Rosenmai et al., 2018). In the reporter assays using full-length hPPARa, Cos1, Cos7, or MDA-MB-231 cells were transfected with an expression vector for the full-length hPPARa protein and a reporter construct in which reporter gene expression was driven by PPARa binding to PPAR response elements and treated with multiple concentrations of PFOA or PFOS, with reporter activity being assessed after 24 hours of exposure. In the reporter assays that used a one-hybrid approach, 3T3-L1, COS-1, THP1, HEK293, HEK293T, or HepG2 cells were transfected with an expression vector for a chimaera of the human PPARa ligand-binding domain and the GAL4 DNA-binding domain and a reporter construct in which reporter gene

expression was driven by GAL4 binding to upstream activating sequences (also known as a one-hybrid approach), and treated with multiple concentrations of PFOA or PFOS, with reporter activity being assessed after 6-24 hours of exposure. It is important to note that whereas PFOA is a full agonist of human PPARa, PFOS is a partial agonist in human PPARa reporter assays (Vanden Heuvel et al., 2006; Rosenmai et al., 2016; Behr et al., 2020b; Nielsen et al., 2022). Human PPARa is also less activated than mouse PPARa in reporter assays, with effective concentrations being 1-2.7 times higher for PFOA to activate human than mouse PPARa and 2.1-2.8 times higher for PFOS to activate human than mouse PPARα (Maloney and Waxman, 1999; Shipley et al., 2004; Vanden Heuvel et al., 2006; Takacs and Abbott, 2007; Wolf et al., 2008a).

Synopsis

[The Working Group noted that, on the basis of the studies reviewed above that were conducted in human primary cells, and various human cell lines, using binding and reporter assays, there was evidence that PFOA and PFOS transcriptionally activated PPARa. However, human PPARa was less effectively activated by PFOA or PFOS than was rodent PPARa. Also, PFOA activated human PPARa more effectively than did PFOS (see summary findings for KC8 in Table 4.27).]

CAR/PXR – in vitro

See Table S4.24 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Human primary cells

Several studies in human primary hepatocytes have tested whether PFOA or PFOS activate human constitutive androstane receptor (CAR)and pregnane X receptor (PXR)-mediated gene transcription. These receptors are discussed together because they regulate overlapping transcriptional programmes, which can make it challenging to definitively identify receptor-specific target genes (<u>Tojima et al., 2012</u>). Here, we have deferred to the study authors with respect to the identification of receptor-target gene pairs.

In human primary hepatocytes exposed for 24–48 hours, PFOS (25 μ M) induced the expression of the CAR target genes *CYP2B6* and *CYP2C19*, but PFOA (25 μ M) only induced CAR target gene expression after 48 hours (<u>Bjork et al.</u>, 2011; Marques et al., 2022).

In transcriptomic analyses of human primary hepatocytes, PFOA (25 μ M) was shown to partially activate the PXR-dependent pathway, but not the CAR pathway (<u>Buhrke et al., 2015</u>). The PXR target gene *CYP3A4* has also been shown to be upregulated by PFOA and PFOS (100 μ M, 48 hours) in human primary hepatocytes (<u>Rosen et al., 2013</u>); however, when tested at 25 μ M for 24 hours, only PFOS significantly increased *CYP3A4* mRNA expression (<u>Bjork et al., 2011</u>).

Human cell lines

In human liver cell models, PFOA activated CAR-dependent gene expression (CYP2B6) in HepaRG human liver cells (30 µM, 48 hours; <u>Abe et al., 2017</u>) and in HepG2 cells (250 μ M, 24 hours; Behr et al., 2020b). In contrast, very low-level exposure to PFOA (0.001 µM) or PFOS $(0.001 \,\mu\text{M})$ for 24–48 hours was shown to reduce CAR-mediated gene expression (CYP2C19) (Franco et al., 2020). In HepaRG human liver cells exposed for 48 hours, PFOA (\geq 50 µM) and PFOS ($\geq 1 \mu M$) induced mRNA expression of the PXR target gene CYP3A4 (Behr et al., 2020a). Very low-level exposure to PFOA (0.001 μ M) or PFOS (0.001 µM) for 24-48 hours also reduced PXR-mediated gene expression (CYP3A4) (Franco et al., 2020).

Hormone pathway	Consistent a evide	nd coherent ence	Suggestive evidence		Paucity of data	
-	PFOA	PFOS	PFOA	PFOS	PFOA	PFOS
Exposed humans						
Thyroid pathway			X	Х		
Estrogen pathway				X	Х	
Androgen pathway		Х	X			
Progesterone pathway			Х	X		
Glucocorticoid pathway				X	X	
AHR pathway					Х	Х
Primary human cells						
Thyroid pathway					Х	Х
Estrogen pathway			Х			Х
Androgen pathway					Х	Х
Progesterone pathway					Х	Х
Glucocorticoid pathway					Х	Х
AHR					Х	Х
PPARa	Х	Х				
CAR/PXR	Х	Х				
PPARγ			Х	Х		
HNF4a					Х	Х
Human cell lines, binding assays	and reporter a	ssays				
Thyroid pathway	1	,			Х	Х
Estrogen pathway			Х	Х		
Androgen pathway					Х	Х
Progesterone pathway					Х	Х
Glucocorticoid pathway					Х	Х
AHR					Х	Х
PPARa	х	Х				
CAR/PXR	Х	Х				
PPARy			Х	х		
HNF4α					Х	Х
Experimental systems (rodents)						
Thyroid pathway		Х			Х	
Estrogen pathway		х	Х			
Androgen pathway	Х	Х				
Progesterone pathway					х	х
Glucocorticoid pathway						Х
AHR					х	х
PPARα	X	Х				
CAR/PXR	X	Х				
PPARy	X	X				
HNF4a					X	х

Table 4.27 Modulation of receptor-mediated effects: pathway characterization for PFOA and PFOS

AHR, aryl hydrocarbon receptor; CAR, constitutive and rostane receptor; HNF4 α , hepatocyte nuclear factor 4 alpha, PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PPAR, peroxisome proliferator-activated receptor; PXR, pregnane X receptor.

Human binding and reporter assays

Reporter assays also have been employed to investigate the abilities of PFOA and PFOS to stimulate the transcriptional activities of CAR and PXR.

It has been shown that PFOS activates CAR indirectly, by modulating its phosphorylation, rather than binding to the ligand-binding domain (Abe et al., 2017); therefore, traditional reporter assays have not detected PFOA- or PFOS-induced transactivation of CAR (e.g. Behr et al., 2020b; Houck et al., 2021; Murase et al., 2023).

In reporter studies employing full-length human PXR, PFOA (0.1–300 µM, 24 hours) was shown to stimulate PXR transactivation when the host cell was a human hepatocyte model (Zhang et al., 2017b; Houck et al., 2021). In reporter assays using a one-hybrid approach, PFOA (0.1-300 µM, 24 hours) stimulated PXR-mediated transcription in only one of the two available studies (Behr et al., 2020a; Houck et al., 2021). In reporter studies employing full-length human PXR, PFOS (0.14–300 µM, 24 hours) was shown to stimulate PXR transactivation when the host cell was a human hepatocyte model (Zhang et al., 2017b; Houck et al., 2021). In reporter assays using a one-hybrid approach, PFOS (0.14-300 µM, 24 hours) stimulated PXR-mediated transcription in only one of the two available studies (Behr et al., 2020a; Houck et al., 2021).

Synopsis

[Overall, the Working Group noted that there was evidence in human primary cells and other human cell in vitro models that PFOA and PFOS stimulated CAR and PXR transcriptional activity. Reporter assays were not able to detect effects of PFOA or PFOS on CAR activity, because activation probably occurred through the modulation of phosphorylation, rather than through ligand binding. The results of PXR reporter assays suggested the transactivation of PXR by PFOA and PFOS, but the effects of PFOA and PFOS appeared to be dependent upon the type of reporter assay and host cell type (see summary findings for KC8 in <u>Table 4.27</u>).]

<u> PPARγ – in vitro</u>

See Table S4.24 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Human primary cells

Only four studies in human primary cells have examined whether PFOA and PFOS activate human PPARy-related pathways. In HUVECs, PFOS (100 µM) exposure for either 24 or 48 hours increased the mRNA expression of PPARG (Liao et al., 2012). In human primary hepatocytes exposed for 48 hours, PFOA (25 µM) and PFOS $(25 \,\mu\text{M})$ induced the mRNA expression of CD36 and PPARG, PFOA alone induced the expression of SCD, and neither induced the expression of FASN or GPAM (Marques et al., 2022). In transcriptomic analyses of human primary hepatocytes, PFOA ($\geq 1 \mu$ M) strongly upregulated the PPARγ pathway (<u>Buhrke et al., 2015</u>). In primary hMSCs stimulated to undergo adipogenic differentiation (7–14 days), PFOA ($\geq 0.1 \mu$ M) and PFOS $(\geq 0.1 \ \mu M)$ stimulated the expression of *FABP4* and PPARG and lipid accumulation (an indicator of adipocyte differentiation, which requires PPARy; <u>Qin et al., 2022c</u>). The authors reported similar findings in mammalian cells (Qin et al., 2022b). In another similar study (7-21 days of differentiation), PFOS ($\geq 0.1 \mu M$) induced FABP4 and PPARG expression Gao et al. (2020). In human primary subcutaneous preadipocytes stimulated to undergo differentiation (10 days), PFOA ($\geq 6 \,\mu$ M) induced the expression of *FABP4*, PLIN1, and PPARG and increased lipid accumulation (Li et al., 2019b). In human primary bone marrow mesenchymal stromal cells differentiating for 21 days, PFOA ($\geq 0.1 \ \mu$ M) and PFOS (1 μM) increased lipid accumulation (Bérubé

et al., 2023). Similarly, in human primary visceral preadipocytes differentiated for 11 days, PFOS ($\geq 5 \mu$ M) increased lipid accumulation (Xu et al., 2016). [The Working Group noted that none of the differentiation studies tested the ability of PFOA or PFOS to stimulate adipocyte differentiation in the absence of a hormone cocktail, which would be a stronger indicator of PPARy agonism.]

Human cell lines

In the human liver cell HepaRG model, exposure to PFOA (100 μ M) for 24 hours induced the expression of the classic PPARy target gene *FABP4* (Attema et al., 2022).

Human binding and reporter assays

PFOA and PFOS have been shown to bind to human PPARy at high concentrations in competitive binding assays using the human PPARy ligand-binding domain, with PFOS having higher binding affinity than PFOA (Zhang et al., 2014b; Li et al., 2018b, 2019b). In a human PPARy reporter assay in which fulllength human PPARy was expressed in HepG2 cells, PFOA and PFOS had similar potencies (~10 μ M) for the stimulation of PPARy transactivation (Zhang et al., 2014b). Less consistent results have been obtained using human PPARy one-hybrid reporter assays. PFOA (at concentrations of $\geq 25 \,\mu\text{M}$) and PFOS have been reported not to activate human PPARy hosted in human kidney HEK293T cells (<u>Behr et al., 2020b</u>); or to activate human PPARy in HEK293 cells with low efficacy (Li et al., 2019b); or to activate human PPARy in INDIGO cells with very low potency (Evans et al., 2022); or to activate human PPARy in HEK293 cells (Buhrke et al., 2013), human epithelial HeLa cells (Garoche et al., 2021), and HepG2 cells (Houck et al., 2021).

Synopsis

[The Working Group noted that there was evidence in human primary cells and other human cell in vitro models that PFOA and PFOS stimulated PPARy transcriptional activity. Caveats remain, however, in that it is unclear if increases in PPARy-mediated gene expression result from the direct activation of PPARy or from PFOA and PFOS increasing the expression of *PPARG*. Additionally, the results of reporter assay studies were inconsistent with regard to the stimulation of PPARy-transactivation by PFOA and PFOS, even when the host cell line was the same (see summary findings for KC8 in <u>Table</u> <u>4.27</u>).]

<u>HNF4a – in vitro</u>

See Table S4.24 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Human primary cells, cell lines, and reporter assays

Hepatocyte nuclear factor (HNF)4 α is a critical receptor for liver development and function (Hayhurst et al., 2001) and is an orphan nuclear receptor (Bogan et al., 2000). [Fatty acids are thought to become stably associated with the ligand-binding domain (Wisely et al., 2002); therefore, PFOA and PFOS are unlikely to interact with the HNF4 α ligand-binding domain and are more likely to have indirect effects on HNF4 α activity.] Importantly, HNF4 α is essential for the activation of CAR and PXR, by acting as a coactivator (Tirona et al., 2003).

PFOA ($\geq 25 \ \mu$ M, 24 hours) has been shown to downregulate HNF4 α , as evidenced by a large decrease in *HNF1A* mRNA expression in human primary hepatocytes (Buhrke et al., 2015). Similarly, in HepG2 cells, PFOA (25 μ M, 48 hours) reduced the protein expression of HNF4 α and a transcriptional target (HNF1 α) (Scharmach et al., 2012). Lastly, PFOA ($\geq 1 \mu M$, 24 hours) reduced endogenous HNF4 α transcriptional activity in HepG2 cells and the activity of overexpressed full-length HNF4 α in HEK293 cells in reporter assays (Scharmach et al., 2012).

[The Working Group noted that PFOA was shown to suppress the activity of a critical liver transcription factor, HNF4 α . However, all the evidence came from a single research group. The possibility that PFOS may interfere with the HNF4 α pathway has not been investigated.]

Synopsis

[The Working Group noted that, overall, in exposed humans, the data showed some association of PFOA with modulation of the thyroid, androgen, and progesterone pathways. However, the data did not support an association between PFOA and the estrogen pathway, because the results were largely negative. There was a paucity of data regarding an association of PFOA with the glucocorticoid pathway.

In exposed humans, there was some evidence for PFOS being associated with modulation of the androgen pathway, with PFOS potentially being associated with lower testosterone levels in men, although the effect size was small. For the thyroid hormone, estrogen, progesterone, and glucocorticoid pathways, the data were inconsistent, and there was insufficient explanation for the differences, meaning that the data were only indicative of an association with PFOS.

There was a paucity of data to permit the assessment of any association of either PFOA or PFOS with the aryl hydrocarbon receptor (AHR) pathway.

In human primary cells, there was evidence that both PFOA and PFOS modulated PPARa activity (receptor activation), which was corroborated by data from human cell lines, binding, and reporter assays. There was evidence that both PFOA and PFOS modulated CAR/PXR activity (receptor activation), which was corroborated by data from human cell lines. There was some evidence that PFOA and PFOS modulated the estrogen pathway (estradiol secretion and estrogen-receptor activation) in human primary cells, but the results were inconsistent across studies in other human cell in vitro models, with more evidence of estrogen-receptor activation coming from reporter assays in which estrogen receptor was overexpressed. There was some evidence that PFOA and PFOS modulated the PPAR γ pathway (adipocyte differentiation, lipid accumulation, and PPAR γ activation) in human primary cells, but the results were inconsistent across studies in other human cell in vitro models.

There was scarce evidence of associations of PFOA or PFOS with androgen-receptor activation in human primary cells, with results in other human cells in vitro (testosterone secretion and androgen-receptor activation) having largely obtained negative results. There was a paucity of data regarding any association of PFOA or PFOS with the thyroid pathway (thyroid peroxidase activity, TTR binding/T4 displacement). There was a paucity of data with which to assess the modulation of HNF4a activity (receptor activation) by PFOA and no studies examined the modulation of HNF4a by PFOS. There was a paucity of data (no studies in human primary cells) from studies examining the modulation of the progesterone receptor, the glucocorticoid receptor (and negative results regarding cortisol secretion and receptor activation in experimental human models), or the AHR by PFOA or PFOS (see summary findings for KC8 in Table 4.27).]

(b) Experimental systems

See Table S4.26 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>) and <u>Table 4.27</u>.

Both PFOA and PFOS have been shown to modulate receptor-mediated effects in experimental systems. Because the data were robust for many endocrine receptor outcomes, this review primarily focuses on in vivo studies in experimental systems. The identified studies were peer-reviewed, used well-developed methods, and the quality of the study design was deemed to be adequate. Some studies used transgenic models in which the expression of one or more endocrine receptor classes was experimentally altered.

(i) Modulation of PPARs

Non-human mammalian systems in vivo

See Table S4.25 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

One of the most extensively studied effects relates to the ability of these chemicals to activate PPARs. Numerous in vivo mammalian studies that evaluated effects on PPARs were available (Table S4.24). This class of nuclear receptors has three main subtypes: PPARa, PPAR β , and PPAR γ (Escher and Wahli, 2000). PPAR α is highly expressed in the liver and influences the expression of genes involved in lipid metabolism, including peroxisomal β -oxidation, fatty acid uptake, and triglyceride synthesis (Rakhshandehroo et al., 2007). PFOA and PFOS are structural analogues of fatty acids, the endogenous ligands for PPAR α .

In rodents, peroxisome proliferation and other metabolic effects occur in the liver after PPARα activation by either PFOA or PFOS. PFOA or PFOS exposure results in hepatomegaly in rodents (Costello et al., 2022). Studies of the exposure of PPARα-KO 129/Sv mice to PFOA or PFOS have shown that hepatomegaly was independent of PPARα activation (Filgo et al., 2015; Su et al., 2022a). PFOA or PFOS exposure in rodents also alters the activity of certain members of the cytochrome P450 (CYP) system. PPARα regulates the expression of acyl-CoA oxidase (ACOX) and CYP4A. CYP4A is involved in the oxidation of fatty acids and lipid metabolism. The increased expression of certain CYP genes in rodents exposed to either PFOA or PFOS has been used as an indicator of the activation of the receptor subclasses. PPAR α target genes include acyl-CoA oxidase (*Acox1 or Pco*), cytochrome P450 4a10 (*Cyp4a10*), and acetyl-CoA acetyltransferase 1 (*Acat1*) (Mandard et al., 2004). It has been estimated that PPAR α mediates more than 75% of the changes in hepatic gene expression induced by PFOA and PFOS in mice (Rosen et al., 2008a, 2017).

Hepatic palmitoyl-CoA oxidase (PCO; alias Acox1) activity has been used as a marker of PPARα activation and peroxisome proliferation (<u>Klaunig et al., 2003</u>). [The Working Group noted that in the literature, acyl-CoA oxidase (*Acox1*) is also termed PCO (<u>NCBI, 2023</u>).]

One study in male cynomolgus monkeys (S. cynomolgus or M. fascicularis) examined the effects of 6 months of oral exposure to PFOA on PPARa (Butenhoff et al., 2002). [The Working Group noted that the increase in PPARa activation caused by PFOA, assessed by PCO activity in this study, is qualitatively similar to that seen in male rats after PFOA exposure.] Male and female rats exposed to PFOS at 20 ppm in the diet for 14 weeks developed hepatocellular hypertrophy and vacuolation, as well as increased relative liver weight, without a significant increase in hepatic PCO activity (Seacat et al., 2003). Increased relative liver weight, hepatocellular hypertrophy, and a modest (0.2-fold) increase in hepatic PCO activity occurred in rats after 28 days of exposure to PFOS at 20 ppm (Elcombe et al., 2012b).

PPARα-null mice have been studied after exposure to PFOA or PFOS. <u>Yang et al. (2002a)</u> reported that liver weight increased in both wildtype and PPARα-null mice treated with PFOA (0.02% in the diet) for 7 days, whereas the PCO activity was only increased in the wildtype mice. <u>Wolf et al. (2008b)</u> also reported that the oral PFOA exposure of wildtype and PPARα-null mice increased liver weight and hepatic cell proliferation and caused hepatocyte hypertrophy. Increased numbers of peroxisomes were present in the livers of treated wildtype mice, whereas the livers of PPAR α -null mice had numerous vacuoles without peroxisomes (Wolf et al., 2008b). [The Working Group noted that, collectively, these findings suggest that the effects of PFOS on the rat liver do not appear to primarily occur through a PPAR α -dependent mode of action.]

Loveless et al. (2006) evaluated the effects of linear, branched, and a mixture of linear/ branched isomers of PFOA on the relative liver weight and hepatic PCO activity of male CD rats and CD-1 mice. The liver weight of the mice increased with increasing PFOA dose and serum level, but the PCO activity was lower at the highest dose and serum level than at lower doses and serum levels. In the rats, the branched isomers of PFOA were more potent at increasing relative liver weight than the linear isomer but were less potent at increasing PCO activity. Perkins et al. (2004) reported that the subchronic dietary exposure of rats to PFOA was associated with a reversible increase in liver weight, increased hepatic PCO activity, and minimal-tomild hepatocyte hypertrophy.

Short-term exposure to PFOS was associated with increased liver weights in wildtype and PPAR α -null mice (Qazi et al., 2009). In addition, altered expression of genes related to lipid metabolism, inflammation, and xenobiotic metabolism were observed in PPAR α -knockout mice exposed to PFOS, suggesting that additional pathways are activated by this chemical (Rosen et al., 2010).

Differences have been reported in the function of mouse and human PPARa (Gonzalez and Shah, 2008). The activation of mouse PPARa results in hepatocyte proliferation and dysregulation of cell-cycle genes, which does not occur in humans (Morimura et al., 2006). Transgenic mouse models that express the human PPARa gene (hPPARa) have also been used to explore the role of PPARa in PFOA and PFOS hepatotoxicity. For example, female and male hPPARa mice exposed to PFOA showed increased liver mass and histologically evident lipid accumulation (Schlezinger et al., 2020, 2021). Other studies using humanized transgenic mice have shown that hPPARa may be less responsive to either PFOA or PFOS when compared with PPARa in wildtype mice (Nakamura et al., 2009; Albrecht et al., 2013; Su et al., 2022a). [The Working Group noted that differences in the response in humanized mice are not necessarily indicative of a different response of human PPARa to PFOA, because species differences in the binding of PFOA to recognition sites on mouse DNA may exist. The Working Group also noted that the relevance to humans of the activation of PPARa with the altered hepatocyte growth and survival and clonal expansion of preneoplastic foci cells, leading to PPARa-dependent rodent liver tumour responses, has been questioned (Corton et al., 2018).]

The effects of PFOA and PFOS on PPARy have also been examined in rodents. This pathway regulates adipocyte differentiation and lipid metabolism (<u>Casals-Casas and Desvergne</u>, 2011). Four studies in mice (<u>Abbott et al.</u>, 2012; <u>Nakagawa et al.</u>, 2012; <u>Yan et al.</u>, 2015b; <u>Schlezinger et al.</u>, 2020) have shown increased liver *Pparg* mRNA expression after PFOA exposure. One study (<u>Wan Ibrahim et al.</u>, 2013) showed increased brain *Pparg* mRNA expression in neonatal mice after the PFOS exposure of pregnant dams.

Two studies in mice (<u>Yan et al., 2015b; Li et al.,</u> 2019c) yielded mixed results regarding the liver expression of PPAR β/δ after PFOA exposure.

Non-human mammalian systems in vitro

See Table S4.26 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

Cell-based in vitro studies involving either reporter assays or the expression of PPAR target genes in hepatic cells have shown that these

chemicals activated mouse, rat, and human PPARa, as well as PPARy (Vanden Heuvel et al., 2006; Takacs and Abbott, 2007; Wolf et al., 2008b; Bjork and Wallace, 2009; Bjork et al., 2011; Houck et al., 2021). Both PFOA and PFOS exposure resulted in the transactivation of PPARy constructs derived from humans, mice, zebrafish, and Xenopus in reporter assays (Garoche et al., 2021). Both PFOA and PFOS have been shown to modulate PPARy signalling in various in vitro cell systems (Watkins et al., 2015; Liu et al., 2019; Li et al., 2021a; Modaresi et al., 2022; Qin et al., 2022c). Studies performed by Takacs and Abbott (2007) using COS-1 cells transfected with mouse or human PPARa, PPAR β/δ , or PPAR γ reporter plasmids and exposed to either PFOA or PFOS showed that PFOA causes significant increases in both mouse and human PPARa reporter activity, whereas PFOS induced activation of the mouse PPARa reporter alone. PFOA and PFOS exposure also increased the activity of the mouse PPAR β /δ, but not the human PPAR β /δ reporter construct. Neither PFOA nor PFOS activated the mouse or human PPARy reporter (Takacs and <u>Abbott, 2007</u>).

(ii) Modulation of CAR/PXR

Non-human mammalian systems in vivo

See Table S4.25 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

In vivo studies using CAR-null mice (Abe et al., 2017) or mice expressing hPPARa (Schlezinger et al., 2020) have shown that PFOA exposure modulates the CAR and PXR pathways. These studies reported effects on the expression of CAR target genes, which include *Cyp2b10*, *Cyp2c29*, *Cyp2c55*, *Akr1b7*, and *Ugt2b34* (Dusek et al., 2019). The PXR target genes examined in some studies included *Oatp1a4*, *CYP3A4*, *CYP2B6*, *MDR1*, and *MRP2* (Smutny et al., 2022). Two studies performed in male C57BL/6 mice

showed increased hepatic expression of CAR after PFOA exposure (<u>Cheng and Klaassen, 2008;</u> Li et al., 2019c).

Increased liver expression of the CAR target gene *Cyp2b1* has been reported to occur in male Sprague-Dawley rats after either a 7- or 28-day (Elcombe et al., 2012a, b) exposure to PFOS at 100 ppm. Increased liver activity of the PXR target gene *Cyp3a1was* also identified after 28 days (Elcombe et al., 2012b). Three other rodent studies (Bijland et al., 2011; Dong et al., 2016; Lai et al., 2017a) also reported PFOS effects on the PXR pathway.

Non-human mammalian systems in vitro

In vitro studies using transcriptome profiling in rodent and human hepatocytes showed that PFOA and PFOS may regulate PXR target gene expression and, to lesser extent, that of CAR (Bjork et al., 2011; Buhrke et al., 2015; Abe et al., 2017; Houck et al., 2021).

(iii) Modulation of thyroid-receptor function

Non-human mammalian systems in vivo

See Table S4.25 (Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.iarc.who.int/636</u>).

One of the most extensively studied endocrine responses seen after PFOA or PFOS exposure relates to the effects of these chemicals on thyroid hormone function. In rats, PFOS exposure reduced serum T4 and T3 levels without increasing TSH secretion (<u>Chang et al., 2008;</u> <u>Davidsen et al., 2022</u>). In mice, PFOS had less consistent effects on thyroid hormone levels (Table S4.25). One study in cynomolgus monkeys reported no effects of PFOS on the serum TSH, FT4, or TT3 concentrations (<u>Chang et al., 2017</u>). Monkeys exposed to PFOS had low serum TT4 concentrations, which was not deemed to be clinically significant. [However, the Working Group noted that the exposure level in this study was limited and that long delays between exposure and the measurement of thyroid hormone concentrations occurred.]

The exposure of rodents to PFOS during pregnancy was often not associated with altered maternal or pup thyroid hormone levels (Fuentes et al., 2006; Chang et al., 2009), although one study reported reduced maternal serum TT3 and TT4 concentrations after exposure during gestation (Conley et al., 2022), and this finding was replicated in pups (Lau et al., 2003).

Adult exposure to PFOS was generally associated with reduced serum thyroid hormone concentrations, whereas the TSH level was often unaffected (see Table S4.26, Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: https://publications.iarc.who.int/636).

Putative mechanisms for the PFOS-induced disruption of thyroid hormone function include inhibition of the sodium/iodide symporter, altered synthesis of thyroglobulin, effects on thyroperoxidase, and the displacement of thyroid hormones from TTR (Weiss et al., 2009; Yu et al., 2009; Dong et al., 2016; Ren et al., 2016; Coperchini et al., 2021a; Davidsen et al., 2022).

In vivo studies examining the effects of PFOA on thyroid hormone function have been more limited in number (Table S4.26, Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: https://publications.iarc.who.int/636), with mixed effects reported in exposed non-human primates, mice, and rats (Butenhoff et al., 2002, 2012a; Blake et al., 2020).

Non-human mammalian systems in vitro

Several studies have shown that PFOA and PFOS affect the binding of thyroid hormones to the thyroid hormone transport protein TTR (Weiss et al., 2009; Ren et al., 2016; Behnisch et al., 2021). Both PFOA and PFOS have been shown to modulate thyroid hormone signalling

in rat cell systems (<u>Croce et al., 2019; Selano et al.,</u> 2019; <u>De Toni et al., 2022</u>).

(iv) Modulation of estrogen-receptor function

Non-human mammalian systems in vivo

Data concerning the direct effects of PFOA or PFOS on the estrogen receptor are limited to those obtained in single studies with negative or mixed findings (Yao et al., 2014; Xu et al., 2017). The estradiol concentrations in mice and rats after either PFOA or PFOS exposure can be variable, with some studies reporting elevated levels, and others showing a reduction in serum estrogen concentration (Table S4.26, Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: https://publications.iarc.who.int/636). Male mice exposed to PFOS did not demonstrate altered serum estradiol concentrations (Wang et al., 2014b; Qu et al., 2016; Qiu et al., 2021; Huang et al., 2022a), whereas female ICR mice exposed to PFOS had reduced estradiol concentrations (Wang et al., 2018b). Reductions in serum estradiol concentrations occurred at exposure levels that did not affect male mice (Qiu et al., 2021; Huang et al., 2022a). However, male rats exposed to PFOS had reduced serum estradiol concentrations in two studies (López-Doval et al., 2015; Salgado et al., 2015).

One study conducted in male cynomolgus monkeys (S. cynomolgus or M. fascicularis) showed no effect on serum estradiol concentration after a 26-week exposure to PFOA (<u>Butenhoff</u> et al., 2002). Several rodent studies also reported no effect of PFOA on serum estradiol concentration (<u>Perkins et al., 2004; Zhao et al., 2010;</u> <u>Tucker et al., 2015; Su et al., 2022b; Yang et al.,</u> <u>2022</u>). Other studies reported increased serum estradiol concentrations in male rats upon PFOA exposure (<u>Biegel et al., 2001; Han et al., 2022</u>) and in female rats upon PFOS exposure (<u>Qiu et al.,</u> <u>2020</u>). Zhang et al. (2020b) reported lower serum estradiol concentrations in female ICR mice exposed to PFOA.

Short-term studies using zebrafish have shown that PFOS acts as an estrogen-receptor agonist (<u>Du et al., 2013</u>).

Non-human mammalian systems in vitro

In vitro studies using ER reporter assays performed in trout or human HEK293T cells and transcription factor activation bioassays performed in human HepG2 cells have shown that PFOA and PFOS regulate ERa target gene expression (<u>Benninghoff et al., 2011; Buhrke et al., 2015; Qiu et al., 2020; Houck et al., 2021</u>).

(v) Modulation of androgen-receptor function

Non-human mammalian systems in vivo

Several studies performed in mice have shown that PFOS exposure results in reduced serum or testicular testosterone concentrations (Table S4.25, Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: https://publications. iarc.who.int/636). However, other studies in mice showed no effects of PFOS exposure (Wang et al., 2014b; Li et al., 2022e). Most studies performed in rats have also shown reduced serum testosterone concentration after PFOS exposure, with fewer demonstrating no effect or increased testosterone concentration (Table S4.25). One study conducted in cynomolgus monkeys showed no effect of a 26-week exposure to PFOA (see also above) (Butenhoff et al., 2002). PFOS exposure in rodents has also been shown to reduce epididymal sperm count and alter the expression levels of several key steroidogenic enzymes in the testes (Wan et al., 2011). Effects of PFOS on testosterone synthesis in Leydig cells have been linked to the CREB/transcription coactivator 2 (CRTC2)/steroidogenic acute regulatory protein (StAR) signalling pathway (<u>Qiu et al., 2021</u>).

Two studies performed in Sprague-Dawley rats showed increased liver testosterone 6β -hydroxylase activity after PFOS administration (Elcombe et al., 2012a, b).

PFOA also showed effects on the androgen receptor pathway. Reduced serum or plasma testosterone concentrations were reported in male rats and mice after PFOA exposure (Li et al., 2011; Owumi et al., 2021b). Reduced serum testosterone concentrations were identified in the male offspring of Kunming mice exposed during gestation (Song et al., 2018). Increased serum testosterone concentrations were identified in female CD-1 mice (Yang et al., 2022) and male CD rats (Biegel et al., 2001) after PFOA administration. However, one study reported no effect of PFOA on the circulating testosterone concentration of exposed male CD rats (Perkins et al., 2004).

(vi) Other receptor systems

Non-human mammalian systems in vivo

Effects of PFOA and PFOS on prolactin and insulin function have been reported in studies that evaluated the function of multiple hormones (Table S4.25, Annex 5, Supplementary material for Sections 4.1 and 4.2, Mechanistic Evidence, online only, available from: <u>https://publications.</u> iarc.who.int/636). Hines et al. (2009) reported that exposure to PFOA at 0.01 mg/kg bw significantly increased serum insulin in mice. Wan et al. (2014) reported that perinatal exposure to PFOS resulted in elevated levels of glucose and insulin in F₁ pups and adult CD-1 mice. Combined gestational and lactational exposure to PFOS elevated fasting serum insulin concentrations and impaired glucose tolerance in rat offspring (<u>Lv et al., 2013</u>).

Neither PFOS nor PFOA altered AHR transcriptional activity in transfected mouse Hepa1.1 2cR cells (Long et al., 2013). [The Working Group noted that a positive control, 2,3,7,8-tetrachlorodibenzo-*p*-dioxin, was included with this AHR-luciferase reporter gene bioassay.] Other studies examined whether PFOA or PFOS altered the expression of *Cyp1a1*, which encodes an enzyme and is regulated by AHR. One study performed in juvenile Atlantic salmon (Salmo salar) fed either PFOA or PFOS at 0.2 mg/kg for < 1 week showed no relation between PFOA or PFOS treatment and the hepatic expression of CYP1A1 and AHR isoforms; however, a temporary but significant increase in Cyp1a1 expression was identified in PFOA-fed salmon after 5 days and PFOS-fed salmon after 2 days (Mortensen et al., 2011). The exposure of rare minnow (Gobiocypris rarus) to PFOA increased the Ahr mRNA level in the gills and was associated with the downregulation of *Cyp1a* mRNA (Liu et al., 2008). One study examined the effects of PFOS on intestinal immunity and infection in mice with *Citrobacter rodentium* infection (Suo et al., 2017). PFOS prevented the replication of C. rodentium by promoting IL-22 production by group 3 innate lymphoid cells through the activation of AHR. The large intestinal lamina propria lymphocyte mRNA expression of Cyp1a1 was also elevated after PFOS treatment (Suo et al., 2017).

In the liver, HNF4a regulates multiple genes and lipid metabolism and is involved in hepatocellular differentiation (Yeh et al., 2019). In mice, the conditional hepatocyte-specific deletion of HNF4a results in hepatomegaly and hepatic steatosis. This phenotype is similar to that induced in rodents exposed to PFOA or PFOS (Bonzo et al., 2012). However, only one study has examined the role of HNF4a after PFOA or PFOS exposure in experimental systems (Yan et al., 2015b). This showed reduced hepatic mRNA expression of Hnf4a in mice exposed to PFOA at 1.25 or 5 mg/kg per day. The high PFOA dose used in this study was not associated with altered Hnf4a expression. Finally, an in vitro study using proteomic profiling in human hepatocytes showed that PFOA alters the expression of genes regulated by HNF4α (Scharmach et al., 2012).

Synopsis

[The Working Group noted that, overall, the studies described above in experimental systems in vivo reported evidence that both PFOA and PFOS could modulate PPARα activity. In addition, there was evidence that PFOA modulated PPARγ activity. In one rodent study PFOS was shown to modulate PPARγ activity.

There was a paucity of available data for the effects of either PFOA or PFOS on PPAR β/δ . There was evidence that both PFOA and PFOS modulated CAR/PXR function.

PFOS modulated thyroid function in exposed animals. However, there were limited data suggesting that thyroid effects were induced by PFOA exposure. There was some evidence that PFOA and PFOS altered serum estradiol concentrations in rodents; however, both non-significant and significant findings were reported.

Evidence that PFOS altered the testosterone concentrations of rodents was available from multiple studies, but there was more limited evidence for the effects of PFOA on androgen-receptor function and testosterone concentration. Finally, there was a paucity of information on the effects of both PFOA and PFOS on other receptor systems in experimental systems (see summary findings for KC8 in Table 4.27).]

4.2.9 Causes immortalization

- (a) Humans
- (i) Exposed humans

See <u>Table 4.28</u>.

The shortening or lengthening of telomeres has been associated with some types of cancers and may contribute to carcinogenesis. Telomere shortening may increase genetic instability, whereas telomere lengthening may promote deleterious cell survival and proliferation and serve as a marker of immortalization.

	-			-			
End-point, assay	Biosample type	Location, setting, study design	Exposure levelª No. of study participants	Response (significance)	Covariates adjusted for	Comments	Reference
Telomere length, by Sigma TeloTAGGG Telomere Length assay	Peripheral blood leukocytes and serum	Firefighters from San Francisco Fire Department and office workers, City and County of San Francisco, USA	163 participants (84 firefighters and 79 office workers) Women PFOA, 1.16 ± 1.76 ng/mL PFOS, 4.18 ± 2.08 ng/mL	Linear regression models (minimally adjusted model, Model 1, and a fully adjusted model, Model 2, used to assess the associations between continuous log- transformed PFAS and TL In Model 1, a doubling of PFOA concentration was associated with a 273 (95% CI, 54–493) bp increase in TL; in Model 2, a doubling in PFOA was associated with a 240 (95% CI, 25–455) bp increase in TL A doubling in PFOS concentration was associated with a 183 (95% CI, 15–352) bp increase in TL in Model 1, and a 172 (95% CI, 5–340) bp increase in TL in Model 2	Age, occupation, the number of times dairy products were eaten per week, and the number of times eggs were eaten per week. Covariates assessed include demographic variables such as race/ethnicity and education; health variables such as BMI, stress, and sleep metrics; and food frequency variables.	Potential confounders were selected a priori, on the basis of results from previous literature and prior analyses performed on these data. No tables provided for the above statistics.	<u>Clarity</u> <u>et al. (2021)</u>

Table 4.28 End-points relevant to immortalization in humans exposed to PFOA or PFOS

~
_
\mathcal{L}
\sim
()
\geq
~
-
\cap
\cup
_
~
-
()
\sim
\sim
U 1
_
\mathcal{L}
_
0
_
_
10
0,
ω
12
01

Table 4.28	(continued)						
End-point, assay	Biosample type	Location, setting, study design	Exposure level ^a No. of study participants	Response (significance)	Covariates adjusted for	Comments	Reference
Telomere length, by qPCR	Maternal whole blood and delivery cord blood	Chemicals in Our Bodies cohort	Mothers (<i>n</i> = 125) and newborns (<i>n</i> = 326) PFOA, 0.8 ± 2.0 ng/mL PFOS, 2.0 ± 2.1 ng/mL	Quantile g-computation PFAS mixture alone was associated with a modest increase in maternal TL (mean change in maternal TL per quartile increase, 0.04; 95% CI, -0.01 to 0.09) In the analysis restricted to maternal-fetal paired samples, an increase in the PFAS mixture of one quartile was positively associated with umbilical cord blood TL (mean change in umbilical cord blood TL per quartile increase, 0.11 (95% CI, 0.01–0.22) Individual PFOA (β = 0.01; 95% CI, $-0.03-0.06$), and PFOS (β = -0.01 ; 95% CI, -0.05 to 0.03)	Maternal education, race/ethnicity, maternal age, gestational age at delivery, pre- pregnancy BMI, parity, and infant sex		<u>Eick et al.</u> (2021)
Telomere length, by qPCR	Leukocytes, serum	US NHANES 1999–2000 Cross- sectional data set	773 participants; 389 men and 384 women PFOA mean (10th–90th percentile), 5.63 (2.24–8.76) ng/mL PFOS, 33.97 (13.54–57.68) ng/ mL	Fully adjusted multiple variable linear regression Each increment of one SD in the \log_{10} PFOS level was associated with a 21-bp increase in the TL ($P = 0.033$), after adjustment for potential confounders No associations of the TL with the concentrations of PFOA or other PFAS	Age, sex, race, BMI, educational level, leukocyte count, C-reactive protein, and PIR status		<u>Huang et al.</u> (2019b)

Table 4.28	(continued)						
End-point, assay	Biosample type	Location, setting, study design	Exposure level ^a No. of study participants	Response (significance)	Covariates adjusted for	Comments	Reference
Telomere length, by qPCR	Cord blood leukocytes, cord serum	Wuhan Maternal and Child Health Hospital in Wuhan, China	916 cord blood samples PFOA, 1.81 ± 2.33 ng/mL PFOS, 5.77 ± 10.32 ng/mL	Weighted quantiles of exposure and linear regression Negative association in the adjusted model: % change in neonatal TL, -5.19% (95% CI, -9.44 to $-0.73%$; $P = 0.040$) for each unit increase in WQS index of PFAS mixture The weights of PFOS and PFOA were 13.16% and 10.75%, respectively	Maternal sociodemographics, age, education, alcohol drinking habits, smoking status, pre- pregnancy BMI, parity, birth weight, gestational age, neonatal sex, and neonatal birth season Alcohol consumption and active smoking were not adjusted for, because no participants drank or smoked during pregnancy		<u>Li et al.</u> (2023a)
Telomere length, by qPCR	Buffy coat from cord blood, cord plasma	Shanghai Allergy Cohort, Shanghai, China; 2012–2013	581 participants PFOA, 7.65 ± 3.88 ng/mL PFOS, 2.93 ± 3.28 ng/mL	Generalized linear models adjusted for covariates LTL was significantly shorter in the female newborns whose PFOS, but not PFOA, concentrations were in the highest quartile, compared with those in the lowest quartile, after adjusting for potential confounders (0.926 ± 0.053 vs 0.945 ± 0.054 , respectively (mean \pm SD); $P = 0.023$)	Maternal and paternal ages, maternal education, maternal pre- pregnancy BMI, mode of delivery, gestational age at birth, infant sex, birth weight, and antepartum obstetric risk		<u>Liu et al.</u> <u>(2018b)</u>

Table 4.28	(continued)						
End-point, assay	Biosample type	Location, setting, study design	Exposure level ^a No. of study participants	Response (significance)	Covariates adjusted for	Comments	Reference
Telomere length, by qPCR	Buffy coat containing leukocytes, serum	Flemish Environment and Health Study, Flanders, Belgium	175 participants PFOA geometric mean (25th–75th percentile): 2.78 (2.13–3.65) μg/L PFOS: 7.52 (5.31–10.9) μg/L	sPLS regression model followed by OLS regression Serum PFOA level was inversely associated with LTL; sPLS model (raw coefficient, -0.017 ; 95% CI, -0.032 to -0.002 ; $P = 0.03$; and OLS model (relative change, -3.64% ; 95% CI, -6.60 to $-0.60%$; $P = 0.02$	In linear regression models the a priori covariates selected from among socioeconomic status, lifestyle, smoking habits, and ethnicity were age, sex, BMI, smoking habits, household education, and ethnicity		<u>Vriens et al.</u> (2019)
Telomere length, by qPCR	Leukocytes from cord blood, maternal serum	Birth cohort study in Guangxi, China, 2015–2018	PFOA, geometric mean, 2.379; 25th– 95th percentile, 1.716–5.599 ng/mL PFOS geometric mean, 0.983; 25th– 95th percentile, 0.662–3.736 ng/mL 499 mother– umbilical cord blood pairs	Multivariable linear regression: each ln- transformed unit concentration increase in PFOA was associated with 20.41% (95% CI, -30.44 to -8.93%) shorter LTL in spring-born infants but not in those born in other seasons	Cigarette smoking, passive exposure to tobacco, alcohol consumption, pre-pregnancy BMI, maternal age, occupation, parity, pregnancy complications, gestational age, date of birth (birth season), infant sex, birth weight		<u>Pan et al.</u> (2022)

BMI, body mass index; bp, base pair; CI, confidence interval; LTL, leukocyte telomere length; NHANES, National Health and Nutrition Examination Survey; OLS, ordinary leastsquares; PFAS, per- and polyfluoroalkyl substances; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid: PIR, poverty income ratio; qPCR, quantitative polymerase chain reaction; SD, standard deviation; sPLS, sparse partial least-squares; TeloTAGGG, commercial telomere length assay; TL, telomere length; vs, versus; WQS, weighted quantile sum. ^a PFOS and PFOA concentrations are expressed as mean ± SD or geometric mean ± geometric SD, unless otherwise stated in the table. Several cross-sectional studies examined potential links of PFOA and PFOS exposure in humans with changes in telomere length (TL).

Clarity et al. (2021) conducted a study on blood collected from female firefighters from the San Francisco Fire Department and office workers in the City and County of San Francisco, USA. The minimally adjusted model (Model 1) and fully adjusted model (Model 2) assessed the association between the continuous log-transformed PFOA and PFOS serum concentrations and leukocyte telomere length (LTL). Statistically significant positive associations of LTL with PFOA ($\beta = 0.57$; 95% CI, 0.12–1.02) and PFOS $(\beta = 0.44; 95\% \text{ CI}, 0.05-0.83)$ were observed in both cohorts, and especially in firefighters in Model 1, after adjustment for age. In Model 2, after adjustment for age, dairy consumption, and egg consumption, only PFOA was found to be significantly associated with LTL in the firefighters. [The Working Group noted that firefighters were occupationally exposed to many different chemicals, including benzene, PAHs, formaldehyde, dioxins, and polybrominated diphenyl ethers, but the effect estimate differences did not account for unmeasured chemical co-exposure in the firefighters that may have affected TL.]

A strong positive association between the blood PFOS concentration and TL was shown mainly in the female participants from among 773 adults in the NHANES cycle of 1999–2000 (Huang et al., 2019b).

In another study, prenatal exposure to PFAS mixtures that included PFOA and PFOS showed a modest association with LTL in umbilical cord blood (mean change in TL per quartile increase, 0.11; 95% CI, 0.01–0.22), and only when maternal-fetal unit pairs were considered (Eick et al., 2021). No significant association was observed of individual PFOA and PFOS blood levels with TL in maternal whole blood (Spearman correlations of 0.09 and 0.05, respectively) or umbilical cord

blood (Spearman correlations of 0.22 and -0.06, respectively) (<u>Eick et al., 2021</u>).

In contrast to these findings, <u>Li et al. (2023a)</u> showed a modest negative association between PFOS concentration and umbilical cord blood TL in male newborns only, and <u>Liu et al. (2018b)</u> identified shorter TL in the umbilical cord blood of female newborns exposed to higher levels of PFOS. In both studies there was no significant association between the level of PFOA and TL in umbilical cord blood.

No significant associations of umbilical cord blood TL with the maternal serum levels of PFOA or PFOS were observed in a birth cohort study of 499 mother–newborn pairs conducted in Guangxi, China, between June 2015 and May 2018 (Pan et al., 2022).

In addition, serum PFOA level was inversely associated with TL in 175 adults aged 50–65 years in the cross-sectional Flemish Environment and Health study (Vriens et al., 2019). [The Working Group noted that TL displays large inter-individual variation at birth and throughout the human lifespan and may depend on the differentiation and activation status of leukocytes. All the available studies were cross-sectional, and the lack of prospective studies investigating the effects of PFOA and PFOS exposure on TL in humans was a notable research gap.]

(ii) Human cells in vitro

The inhibition of gap junctional intercellular communication (GJIC) may lead to a loss of intercellular communication and play a role in mitogenic activation, promoting tumour formation and cellular proliferation, migration, and invasion. Some carcinogens may inhibit GJIC and promote cancer development.

Saejia et al. (2019) showed that treatment with PFOA did not affect cell migration, but enhanced cell invasion, adhesion, and the activity of matrix metalloproteinase-2 through the activation of the NF- κ B signalling pathway in human follicular thyroid carcinoma (FTC133) cells. Treatment

of non-tumorigenic human breast epithelial MCF-10A cells with 10 μ M PFOS for 72 hours induced cell-cycle progression, cell migration, and invasion, which may lead to breast cancer initiation and development (<u>Pierozan and Karlsson, 2018</u>).

(b) Experimental systems

Non-human mammalian cells in vitro

It has been shown that PFOA and PFOS inhibit GJIC in rat liver epithelial cells (<u>Upham et al.</u>, <u>1998</u>). The activation of extracellular receptor kinase and phosphatidylcholine-specific phospholipase C in an oxidation-dependent manner was suggested to be a mechanism of GJIC dysregulation (<u>Upham et al.</u>, <u>2009</u>).

The morphological transformation of Syrian hamster embryo cells in vitro assay is recommended by the Organisation for Economic Co-operation and Development for the detection of genotoxic and non-genotoxic carcinogens. Seven days of exposure to non-cytotoxic concentrations of PFOS alone or PFOA in combination with benzo[a]pyrene (B[a]P, 0.4 μ M) pre-treatment induced cell transformation (measured as colony forming unit frequency) in a non-genotoxic manner (Jacquet et al., 2012a, b). PFOS and its acid form induced cell transformation only at 0.37 and 3.7 µM concentrations, in a non-dose-dependent manner without an initiator, whereas PFOA at concentrations from 3.7×10^{-4} to $37.2 \,\mu$ M induced cell transformation in B[a]P-sensitized cells (Jacquet et al., 2012a, b). [The Working Group noted that immortalization-specific studies of primary or cultured human cells were not available.]

Synopsis

[The Working Group noted that the evidence that PFOA or PFOS induces cell immortalization was sparse for all test systems.]

- 4.2.10 Alters cell proliferation, cell death, or nutrient supply
- (a) Humans
- (i) Cell proliferation

Exposed humans

Xie et al. (2023) identified positive correlations of PFOA and PFOS concentrations with the expression of molecular markers of glioma, specifically Ki-67 and p53. A total of 137 glioma tissue and 40 non-glioma tissue samples were collected. The study showed that PFOA and PFOS is commonly present in cancerous and noncancerous brain tissue, but higher concentrations of PFOA and PFOS were present in glioma samples from the brain than in non-glioma samples, albeit without statistical significance. The data revealed a positive correlation between PFOA and PFOS concentrations and tumour grade. Positive correlations of PFOA and PFOS with the expression of the glioma molecular markers Ki-67 or p53 were also observed. Significant correlations were observed of tumour grade with Ki-67 and p53 expression ($r^2 = 0.33$ and $r^2 = 0.10$, respectively). Ki-67 expression significantly correlated with the level of PFOA (univariate linear regression; n = 97; $r^2 = 0.24$; P < 0.05). A multiple regression model indicated that approximately 30% of Ki-67 expression could be explained by variation in the PFOS and PFOS concentrations, and predictor importance analysis suggested that Ki-67 expression was mainly driven by the PFOA concentration. No significant univariate regression associations were observed between the concentrations of individual PFAS and p53 (Xie et al., 2023). [The Working Group noted several limitations to this study. PFOA and PFOS were measured in the glioma tissue; however, it was not specified which part(s) of the brain was sampled and compared. The PFAS concentrations varied substantially in different brain areas (Di Nisio et al., 2022). Paired glioma and non-glioma

samples were available for 18 patients, and no statistically significant differences in the concentrations of PFAS were observed between the 18 pairs of glioma and non-glioma tissue samples, possibly because of the limited sample numbers. The 137 glioma and 40 non-glioma brain tissue samples included in the study were from patients from whom only one of the two tissue types was collected. It was not informative to compare PFAS levels in tumour and non-tumour brain tissue from different participants, because PFOA and PFOS exposure varied substantially among the participants. In addition, it was noted that the expression of Ki-67 was absent in 40 out of 137 glioma samples and that the relevance of cell proliferation measurements in tumour tissues as an end-point for this KC is questionable.]

Bassler et al. (2019) explored the mechanisms of the associations of PFOA and PFOS with the development of NAFLD. Two hundred adult samples from a cross-sectional study of participants from districts with PFOA contamination of drinking-water were analysed. The serum concentrations of biomarkers of hepatocyte death/apoptosis were altered. A univariate analysis showed that serum cytokeratin 18 M30 (a marker of hepatocyte apoptosis) and cytokeratin C18 M65 (a marker of hepatocyte total cell death, necrosis and apoptosis) were both positively associated with serum PFOA and PFOS. In a multivariate analysis, M30 was positively associated with PFOA, and there was a similar trend for PFOS [The Working Group noted that although caspase-cleaved cytokeratin 18 fragments (CK-18 forms both M30 and M65 antigens) are markers of hepatocyte death, they have also been considered to be candidate markers for the detection of non-alcoholic steatohepatitis and fibrosis (Feldstein et al., 2009).] [The Working Group also noted that this study observed associations of PFOA and PFOS with markers of inflammation. See Section 4.2.6.]

Human primary cells in vitro

Human primary normal prostate epithelial cells were cultivated to form a 3D spheroid model and exposed to PFOA or PFOS at 10 nM for 3–4 weeks. PFOA and PFOS significantly increased the total number of spheroids and their size, indicating elevated stem cell self-renewal and progenitor cell proliferation. Transcriptome analyses showed an upregulation of genes encoding signalling pathway intermediates involved in cell proliferation (G2-M checkpoint, mitotic spindle, E2F targets) and oncogenesis (kRAS and MYC signalling, TNF- α via NF- κ B, IL-6/JAK/STAT3, TGFβ, and inflammatory signalling). Metabolomic analysis of PFOA and PFOS-exposed prostaspheres (spheroids) revealed upregulation of glycolytic pathways, including those involved in the Warburg effect, in response to PFOA or PFOS exposure (Hu et al., 2022).

In addition, transcriptomic analysis of human primary hepatocytes treated with non-cytotoxic doses (10 nM-10 µM PFOA or PFOS, potassium salt) for 48 or 96 hours revealed that PFOA predominantly changed the expression of genes involved in lipid metabolism and hepatic steatosis, whereas PFOS predominantly induced changes in the expression of genes involved in carcinogenesis and cell death signalling. PFOA and PFOS caused a decrease in CLDN1 mRNA. Claudin-1 protein is involved in maintaining cellular adhesion and the formation of cell junctions. PFOA and PFOS also caused an induction of the AKR1B10 gene, which is associated with the progression of hepatocellular carcinoma (Matkowskyj et al., 2014).

The effect of PFOS on human primary oesophageal epithelial cell proliferation was assayed using a cell counting kit (CCK)-8. PFOS at 10 nM had no significant effect on proliferation but increased the migration and invasion of the oesophageal squamous cell carcinoma cell lines KYSE150, KYSE140, and KYSE70 by regulating the transcription and protein stability of ZEB1 (Liu et al., 2022c).

Human cell lines in vitro

Exposure to PFOS ($10 \mu M$) or PFOA ($100 \mu M$) for 72 hours induced breast epithelial cell (MCF-10A cell line, expected to be ER α - and β -negative) proliferation and the alteration of regulatory cell-cycle proteins (cyclin D1, CDK6, p21, p53, p27, ERK 1/2, and p38), and this persisted after multiple cell divisions. Interestingly, PFOA and PFOS increased proliferation and caused a persistent increase in cyclin D1 levels, but through two different mechanisms. PFOS activated the ERK pathway and PFOA acted by inactivating p38 and reducing cyclin D1 degradation. Both compounds promoted cell migration and invasion (Pierozan et al., 2020). [The Working Group noted that this study also observed associations of PFOA and PFOS with alterations in epigenetic end-points (see Section 4.2.4).]

Gimenez-Batista et al. investigated the effects of several concentrations of PFOA (100, 72.5, 7.25, 3.6, and 0.36 μ M) or PFOS (100, 60, 6, and 0.6 μ M) on the growth of a human colon myofibroblast (CCD-18Co) cell line using an MTT (3-(4,5-dimethyl-2-thiazolyl)-2,5-diphenyl-2*H*tetrazolium bromide) assay, in the presence or absence of the pro-inflammatory cytokine IL-1b. After 96 hours of treatment, CCD-18Co myofibroblast proliferation was significantly induced in the presence or absence of IL-1b (<u>Giménez-Bastida et al., 2015</u>).

Exposure of a human non-small cell lung carcinoma cell line (A549) to one of several concentrations of PFOA or PFOS (10, 200, or 400 μ M) for 48 hours caused alterations to the cell cycle and apoptosis. Lower doses of these compounds caused the cell balance to shift towards cell proliferation, whereas exposure to higher concentrations shifted the balance towards apoptosis, as evaluated by MTT assay and confirmed by measurement of the mRNA expression of genes involved in the cell cycle and

proliferation. Increases in cellular proliferation were noted at 100 and 200 μ M PFOA and PFOS exposure levels; however, significant reductions in cell viability at \geq 600 μ M PFOA and 400 μ M PFOS were also observed (Jabeen et al., 2020).

Spheroids produced from COV434 and KGN human ovary granulosa tumour cell lines were exposed to PFOA (0.02, 0.2, 2, 20, or 200 ng/mL, or 2 mg/mL) or PFOS (0.08, 0.8, 8, 80, or 800 ng/mL, or 8 mg/mL). The proliferation of the spheroid cultures was estimated by the measurement of ATP using the CellTiter-Glo 3D cell viability assay. PFOA and PFOS increased COV434 and KGN cell proliferation in a dose-dependent manner, compared with untreated control cells (Gogola et al., 2019).

In other studies, lower viability or induction of apoptosis was observed after PFOA or PFOS exposure in primary cultures of normal human thyroid cells (NHT) (<u>Coperchini et al., 2021b</u>), in a human hepatoma cell line (HepG2) (<u>Hu and Hu, 2009</u>), and in monolayers (2D) and spheroids (3D) of neuronal cells (N2a) (<u>Choi et al., 2013</u>).

Two cancer cell lines, prostate (DU145) and breast (MCF7), were treated with very low doses of PFOA (10⁻⁶ to 10⁻¹² M), for 48 hours. PFOA increased the proliferation of DU145 and MCF7 cells, as monitored by real-time imaging. PFOA acted through distinct signalling pathways in these two cell lines (AKT/mTORC1 and plexin D1-dependent pathways in MCF7 and DU145 cells, respectively) (Charazac et al., 2022).

PFOA at 50 or 100 μ M after 72 hours significantly promoted viability, migration, and invasion of RD cells (a human embryonal rhabdomyosarcoma cell line), and significantly inhibited apoptosis. Higher concentrations (500 and 1000 μ M) inhibited cell viability. Treatment with a PI3K inhibitor antagonized the effects of PFOA on migration, invasion, and apoptosis. The authors concluded that PFOA promoted RD cell migration and invasion and inhibited apoptosis through the PI3K/AKT signalling pathway (Zhang et al., 2019). Trophoblast cells HTR-8/SVneo were exposed to 100–1000 μ M PFOA for 72 hours. Low doses increased trophoblast viability, whereas concentrations of > 400 μ M reduced cell viability. The activation of ROS-dependent ERK signalling at low doses triggered trophoblast proliferation, whereas the activation of UPR signalling triggered trophoblast apoptosis under conditions of severe ERS (Du et al., 2022).

Immortalized human granulosa HGrC1cells (a cell type of ovarian follicles) were exposed to PFOA at 1–100 μ M for 24–96 hours. Low doses (up to 10 μ M) caused increased viability and proliferation, whereas the highest dose (100 μ M) caused reductions in these parameters. YAP1-mediated proliferation was confirmed by adding a YAP1 inhibitor, which prevented the effects of PFOA. Thus, PFOA was shown to induce proliferation, migration, and invasion of HGrC1 cells, potentially via upregulation of the Hippo pathway effector YAP1 and of cell-cycle regulators such as cyclin D1 (Clark et al., 2022).

PFOA (\geq 100 nM) treatment also stimulated A2780 ovarian cancer cell invasion and migration, and increased the expression of the matrix metalloproteinases MMP-2 and 9 (Li et al., 2018c).

Human normal HL-7702 hepatocyte line was exposed to PFOA at 0-800 µM for 48 or 96 hours. Low doses (50–100 µM) increased cell proliferation by promoting a shift from the G1 to S phase, whereas at higher doses (200–400 μ M), the cell numbers were comparable with those of the control, mainly because of cell-cycle arrest in the G0/G1 phase. Notably, no apoptosis was detected, even at 400 µM. Corroborating these data, proteomic analysis detected 111 significantly expressed proteins, of which 46 were related to cell proliferation and apoptosis. The induction of cyclin D1, CDK6, cyclin E2, cyclin A2, and CDK2 in the low-dose PFOA groups was associated with larger numbers of cells in S phase, as well as fewer cells in the G0/G1 phase, indicating that PFOA shifted cells from G1 to S phase (Zhang et al., 2016b).

The viability of L-02 cells increased slightly after 24 hours of treatment with PFOA at 8 μ M, but decreased at higher PFOA concentrations (16–512 μ M) after 24 hours (<u>Wang et al., 2022a</u>).

In an experiment performed in HepG2 cells, which were treated with 200–500 μ M PFOA for 12–48 hours, it was observed that 50 μ M PFOA increased the number of cells in the G2/M phase, while reducing the number in S phase; 100 and 150 μ M PFOA increased the G0/G1 cell percentages and reduced the S/G2/M cell percentages; and \geq 400 μ M reduced the proportion of G0/G1 cells (Shabalina et al., 1999).

In contrast, cell proliferation was not altered in other cell types, for example in human choriocarcinoma (Jeg-3) and in endometrial adenocarcinoma (RL95-2) cells treated with 0.01–100 μ M PFOA for 24 hours (<u>Tsang et al., 2013</u>), in ovarian cancer cells (A2780 cells) treated with 0–200 nM PFOA for different periods of time, and in human ovarian GC cells (KGN) treated with 0.03–300 μ M PFOA for 12–48 hours (<u>Zhou</u> <u>et al., 2020</u>).

To examine the effects of PFOA during human endometrial carcinogenesis, Ishikawa cells were treated with 50 nM PFOA for 48 hours. This treatment did not affect proliferation, but it promoted both the migration and invasion of these endometrial cancer cells (Ma et al., 2016).

<u>Pierozan and Karlsson (2018)</u> showed that PFOS-induced cell proliferation and cell death are dependent on the exposure time and concentration. The breast cell line MCF-10A was treated with 0–1 mM PFOS for 24, 48, or 72 hours. Significantly lower cell viability was observed at \geq 250 µM PFOS at all time points. In contrast, exposure to 10 µM PFOS for 48 hours, or 1 or 10 µM PFOS for 72 hours, increased MTT production and the number of cells, reduced the percentage of cells in the G0/G1 phase, and increased the percentage of cells in S phase at all time points. There were also decreases in the mean fluorescence intensities associated with p27, p21, and p53 staining and an increase in the mean fluorescence intensity associated with CKD4 staining in PFOS-treated cells, compared with the controls. In addition, treatment with 10 μ M PFOS for 72 hours also stimulated MCF-10A cell migration and invasion.

However, PFOS at concentrations ranging from approximately 1 nM to 100 μ M did not induce significant effects on cell viability and/or cell death in hMSCs (Gao et al., 2020), human primary placental cytotrophoblasts isolated from placenta at full-term pregnancy (Zhang et al., 2015b), human embryo liver cells (L-02) (Zeng et al., 2021; Dong et al., 2022), human choriocarcinoma cell lines HTR-8/SVneo and JEG-3, embryonic stem cell-derived cardiomyocytes (Cheng et al., 2013; Li et al., 2021a), and renal proximal tubular epithelial cells NRK-52E (Wen et al., 2016).

Notably, ferroptosis, a process of regulatory cell death that is induced by excessive lipid peroxidation, was investigated in two studies. <u>Cui et al. (2022)</u> investigated the role of ferroptosis in HUVECs exposed to 180 μ M PFOS for 12–48 hours. The results showed that the viability of the HUVECs was significantly reduced by the PFOS treatment after 12 hours. PFOS increased the expression of the ferroptosis-related protein ACSL4 and reduced the expression of GPX4, HO-1, and FTH1. The results were confirmed using a ferroptosis inhibitor (<u>Cui et al., 2022</u>).

In the human proximal tubular epithelial cell line HK-2, treatment with PFOS at 50–250 μ M reduced cell viability after 12 hours in a dose-dependent manner and induced ferroptosis and apoptosis (<u>Wang et al., 2022b</u>).

(ii) Cell death inhibition

Human hepatocarcinoma HepG2 cells exposed to 200 μ M PFOA for \leq 72 hours showed autophagosome accumulation. [The Working Group noted that the autophagosome accumulation may have been the result of autophagy activation or have been caused by the inhibition of autophagy at the degradation step.] The proteomic

analysis performed in this study suggested the inhibition of autophagy. The same study observed autophagosome accumulation in mouse livers (<u>Yan et al., 2017</u>). [The Working Group noted that these results are helpful in understanding the potential mechanisms of the hepatotoxicity induced by PFOA. In addition, this study showed that the PFOA-dependent reduction in HepG2 cell viability may not be directly attributable to a dysfunction of autophagy.]

(iii) Angiogenesis

In HTR8/SVneo cells, an embryonic trophoblast cell type that is able to form vessel-like vascular networks in 3D matrices, exposure to PFOA at 100 μ g/mL affected morphological parameters of the pseudo-vascular network in a dose-dependent manner, including the number of pseudo-vascular junctions and the total lengths of non-branching segments. In contrast to PFOA, PFOS did not have significant effects on angiogenesis (Poteser et al., 2020).

(b) Experimental systems

(i) Non-human mammalian systems in vivo

In the chronic carcinogenicity study by <u>Butenhoff et al. (2012a)</u> (see Section 3.1.2), a significantly increased incidence of tubular hyperplasia of the ovaries of rats was observed in groups treated with PFOA, compared with controls – 0/48 (0%), 7/50 (14%), and 15/47 (33%) at 0 (control), 30, and 300 ppm, respectively. There were also increases in the incidence of hepatocellular hypertrophy in male and female rats. [The Working Group noted that although the differences were not statistically significant, the incidence of pancreas acinar hyperplasia in male rats was 0/46 (0%), 2/46 (4%), and 2/49 (4%) for the groups at 0 (control), 30, and 300 ppm, respectively).]

Dietary administration of PFOA at 300 ppm for 2 years to CD rats resulted in increased Leydig cell proliferation (46% versus 14% for the control group) and a higher incidence of adenoma (11% versus 0% for the control groups) in the testes. Also, PFOA showed a tendency to increase pancreatic acinar cell proliferation at 15, 18, and 21 months, when compared with the control groups. The incidence of acinar cell hyperplasia was 39% versus 18% or 10% for the ad libitum or pair-fed control groups, respectively; and that of adenoma was 9% versus 0% or 1% for the ad libitum or pair-fed control groups, respectively. No cell proliferation was observed in the livers (Biegel et al., 2001).

In a review of the pancreatic lesions in male rats in the studies by both <u>Biegel et al. (2001)</u> and Butenhoff et al. (2012a), using the same diagnostic criteria as those applied in the study by Biegel et al. (2001), a significant positive trend (P < 0.05, Cochran-Armitage trend test) in the incidence of pancreatic acinar cell hyperplasia was observed, with the incidence being significantly increased [P = 0.0382, Fisher exact test] at the highest dose (3/46 (7%), 1/46 (2%), and 10/47 (21%) for 0, 30, and 300 ppm PFOA, respectively) (Caverly-Rae et al., 2014). In addition, histopathological examination of the pancreas revealed focal ductal hyperplasia in C57Bl/6 mice treated with PFOA at 2.5 or 5.0 mg/kg orally for 7 days (Kamendulis et al., 2014).

In a 2-year carcinogenicity study with and without perinatal exposure (NTP, 2020) (see Section 3.1.2), exposure to PFOA resulted in increases in the incidence of hepatocellular hypertrophy in male and female rats and increases in the incidence of papillary urothelium hyperplasia at the 16-week interim time point of the feeding study. In this 2-year feeding study of PFOA, increases in the incidence of hepatocyte hypertrophy, bile duct hyperplasia, hyperplasia of the renal papillary epithelium, and epithelial hyperplasia of the forestomach were observed in female rats at 300 or 1000 ppm. In the male rats, increases in the incidence of hepatocyte hypertrophy and bile duct hyperplasia were also observed. In addition, acinus hyperplasia was

also significantly increased in all the postweaning-only exposure groups in the chronic study and in the groups at 300 ppm in the perinatal and postweaning study, and this lesion was considered to be potentially preneoplastic.

Filgo et al. (2015) (see Section 3), described non-neoplastic lesions in four strains of mice (age, 18 months) after exposure to PFOA during gestation. Significant increases in the incidence of non-neoplastic liver lesions were observed in CD-1 mice after PFOA exposure, including oval cell hyperplasia, Ito cell hypertrophy, and centrilobular hepatocyte hypertrophy. Several non-neoplastic changes were observed after PFOA exposure in PPARa-KO mice, but not in 129/Sv WT mice, including significant dose-related increases in the incidence of both bile duct hyperplasia and bile duct inclusion bodies (hyaline droplets). In addition, the incidence of centrilobular hepatocyte hypertrophy and of haematopoietic cell proliferation was significantly increased by PFOA exposure in PPARa-KO mice, but not in 129/Sv WT mice. The incidence of Ito cell hypertrophy was reduced by PFOA treatment in PPARα-KO mice (Filgo et al., <u>2015</u>). [The Working Group noted that this study demonstrated that PPARa is not essential for the liver effects of PFOA in mice, because these were observed in PPARa-KO mice.]

Male Sprague-Dawley rats were exposed orally to PFOA (300 ppm) in the diet for 1, 7, or 28 days. The results showed that the hepatic bromodeoxyuridine (BrdU) labelling index was increased by approximately three-fold 1 day after the start of treatment. The increase in the hepatic BrdU labelling index versus the control group was largest after 8 days (approximately five-fold compared with controls). The labelling index values were near the background levels for male rats after 2 and 29 days, even though they were statistically significantly higher than those for the respective controls. The administration of PFOA to rats led to hepatomegaly, characterized by hypertrophy and hyperplasia, as a result of early increases in cell proliferation, which would ultimately lead to liver tumour formation. (Elcombe et al., 2010). [The Working Group noted that technical problems occurred with fixation and immunostaining for BrdU, so a second study was conducted to evaluate cell proliferation. The results were those of both studies together.]

Male Sprague-Dawley rats were treated intravenously via a tail vein with methylpalmitate, a Kupffer cell inhibitor, 24 hours before a single oral dose of PFOA (100 mg/kg) and were killed 24 hours later. PFOA significantly increased the BrdU labelling index in the liver, and pre-treatment with methylpalmitate reduced PFOAinduced labelling by 57% (Alsarra et al., 2006).

Thottassery et al. (1992) investigated the role of adrenal hormones in the hepatomegaly induced by PFOA. Male Sprague-Dawley rats underwent adrenalectomy and received a single dose of PFOA at 150 mg/kg by oral gavage. In intact rats and in rats that had undergone adrenalectomy, PFOA caused increases in ornithine decarboxylase activity and significantly reduced hepatic DNA levels (by 30%; n = 4-10; P < 0.05) (Thottassery et al., 1992). [The Working Group noted that the reduced hepatic DNA concentration was inversely proportional to the hepatomegaly observed. In rats that had undergone adrenalectomy, however, hepatomegaly was mostly caused by hyperplasia.]

Thirty-six adult Sprague-Dawley rats received ethane dimethyl sulfonate to eliminate Leydig cells, then were treated with PFOA at 0, 25 or 50 mg/kg per day by oral gavage for 9 consecutive days. The number of proliferating cell nuclear antigen (PCNA)-positive Leydig cells in testes sections was reduced after 21 days, indicating that PFOA exposure may reduce the proliferation of stem Leydig cells (Lu et al., 2019).

Male C57BL/6 mice were fed a low-fat control diet or a high-fat diet (HFD) for 16 weeks to model normal and steatotic livers, respectively. After 16 weeks on these diets, the mice were treated with PFOA (1 mg/kg bw per day) for 2,

8, or 16 weeks. PFOA induced hepatocyte hypertrophy, regardless of the diet, as indicated by histological examination. In rats consuming the low-fat control diet, PFOA induced an increase of 3.30-fold in hepatocyte DNA synthesis compared with vehicle-treated control diet-fed mice by week 2. HFD and PFOA had a synergistic effect on the BrdU labelling index by week 2. By week 8, HFD significantly increased DNA synthesis in hepatocytes compared with the vehicle-treated control diet-fed group. The control diet + PFOA and the HFD + PFOA groups also showed elevated DNA synthesis levels of 11.23-fold and 15.72-fold, respectively, compared with the vehicle-treated control diet-fed group (Li et al., 2019c). The Working Group noted that, taken together, these results suggest that pre-existing NAFLD enhanced PFOA-stimulated hepatocyte hyperplasia only at the early time point studied. The Working Group also noted that this study shows that PFOA activates PPARa, CAR, and PXR and that PFOA reverses HFD-induced steatosis and reduces the size of adipose tissue depots.]

Gestational intrauterine exposure to PFOA at 5 mg/kg in female Kunming mice was used to characterize the potential effects of prenatal PFOA exposure on the cerebral cortex cells of offspring at PND21. PFOA-treated PND21 mice demonstrated increased levels of nerve growth factor (NGF) in serum and cortex cells. In addition, PFOA-exposed cerebral cortex cells showed higher NGF and PCNA expression, and exposure to PFOA and an NGF-specific inhibitor downregulated the expression (Qin et al., 2018). In contrast, in other studies in PFOA-exposed Kunming mice, increases in apoptotic markers were observed in the liver (Liu et al., 2015b) and uterus (Li et al., 2018c; Zhang et al., 2021b).

Adult male and female cynomolgus monkeys (*S. cynomolgus* or *M. fascicularis*) were treated with PFOS (potassium salt) at 0, 0.03, 0.15, or 0.75 mg/kg per day for 182 days and were monitored for 1 year after the treatment. Hepatocellular hypertrophy and lipid vacuolation were present

at term in the group at 0.75 mg/kg per day. The treatment had no significant effect on cell proliferation in the liver, pancreas, or testes after 182 days of treatment, as determined using the PCNA immunohistochemistry cell labelling index (Seacat et al., 2002).

Male and female Sprague-Dawley rats were exposed to PFOS (potassium salt) at a dietary concentration of 0.5, 2, 5, or 20 ppm for up to 104 weeks (Butenhoff et al., 2012b). A 20-ppm dose recovery group received 20 ppm PFOS in the diet for up to 53 weeks, after which it was fed control diet. The male rats at 20 ppm that were killed during week 53 had an increased incidence and severity of centrilobular hepatocytic hypertrophy and vacuolation, whereas the female rats had only centrilobular hypertrophy and the changes were less severe. These alterations were also observed in rats killed at the scheduled end of the study. There were no statistically significant increases in hepatocellular S-phase labelling index (cell proliferation index), as measured using BrdU immunohistochemistry, in any of the experimental groups.

Hepatocellular hypertrophy, increased cell proliferation, and reduced liver apoptotic index were observed in Sprague-Dawley rats fed PFOS (potassium salt; 20 or 100 ppm) for 28 days, and this was reported to be mediated by the nuclear receptors PPARa, CAR, and PXR (see also Section 4.8) (Elcombe et al., 2012c). A subsequent study showed that after a recovery period of 84 days after treatment, increased liver proliferative index, reduced liver apoptotic index, and a lower number of hepatocellular glycogen vacuoles were observed (Elcombe et al., 2012a).

Han et al. (2018b) treated male Sprague-Dawley rats daily by gavage with PFOS (1 or 10 mg/kg) for 28 days. PFOS exposure triggered Kupffer cell activation and significantly upregulated the expression of PCNA, c-Jun, c-MYC, and cyclin D1 in the liver (Han et al., 2018b). [The Working Group noted that this study showed inflammatory cell infiltration; see Section 4.2.6.] Groups of 25 pregnant Sprague-Dawley rats were given daily oral doses of PFOS (potassium salt) at 0.1, 0.3, or 1.0 mg/kg per day from GD0 to PND20. The mean number of Ki-67⁺ thyroid follicular epithelial cells in female fetal thyroids from the 1.0 mg/kg per day group was higher by 2.1-fold than for the control group (P < 0.05) (Chang et al., 2009).

The effects of PFOS on GJIC in vivo were studied using Sprague-Dawley rats (n = 4-6) exposed to PFOS (5 mg/kg per day) orally for 3 days or 3 weeks. GJIC was significantly reduced in the livers of PFOS-treated rats, with an EC₅₀ of 30 µM (15 mg/L) after 3 days of exposure, and the magnitude of inhibition was the same for the longer exposure period of 21 days (no statistical significance was shown) (Hu et al., 2002).

Male Sprague-Dawley rats were treated with PFOS (0, 5, or 10 mg/kg per day) by gavage for 7 days and injected intraperitoneally with ethane dimethyl sulfonate the next day to eliminate Leydig cells, so that the Leydig cell regeneration process could be investigated. PFOS pre-treatment significantly lowered the serum testosterone level and reduced the number of regenerated Leydig cells (Mao et al., 2021). [The Working Group noted that transverse sections of the testes immunohistochemically stained for CYP11A1 and 11 β -HSD1 were used to analyse cell number. In addition, in a 3D seminiferous tubule culture system, PFOS inhibited stem Leydig cell proliferation and differentiation, as well as the hedgehog signalling pathway.]

Qu et al. (2016) showed alterations in testicular ER expression, together with reduced proliferation and increased apoptosis of germ cells, which might be involved in the PFOS-induced testicular toxicity. In male ICR mice, oral administration of PFOS (0–10 mg/kg bw) for 4 weeks caused reduced sperm count, testosterone level, and CRTC2/StAR expression, and damage to the testicular interstitium, paralleled by increases in phosphorylated PKA, CREB, and p38 in the testes (Qiu et al., 2021). PFOS impaired normal placental angiogenesis in female CD-1 mice by disrupting the lncRNA Xist/miR-429/VEGF-A pathway, adversely affecting fetal development (<u>Chen et al., 2018</u>).

(ii) Non-human mammalian systems in vitro

The 3T3-L1 preadipocyte culture system has been used to test numerous compounds that influence adipocyte differentiation or function. Cells were treated with PFOA (5-100 µM) or PFOS $(50-300 \mu M)$, the PPARa agonist WY-14 643, or the PPARy agonist rosiglitazone. The cells were assessed morphometrically and biochemically for number, size, and lipid content. There was a significant concentration-related increase in cell number and reduced cell size after exposure to PFOA or PFOS (Watkins et al., 2015). However, Fischer rat thyroid line-5 (FRTL-5) cells exposed to 10⁵ nM PFOA or PFOS showed significant inhibition of cell proliferation. In particular, the percentage of proliferating FRTL-5 cells was 14.2% of the total number of cells in the medium. and this was reduced to 7.5% by PFOA and to 3.9% by PFOS (<u>Coperchini et al., 2015</u>).

PND4 neonatal ovaries from CD-1 mice were cultured in control medium (dimethyl sulfoxide < 0.01% final concentration) or PFOA (50 µM or 100 μ M). The results showed that exposure to PFOA at 50 µM for 96 hours increased the number of secondary follicles, the expression of Ki-67 and the protein and gene expression of Ccna2, Ccnb2, Ccne1, Ccnd1, Ccnd2, and Ccnd3. PFOA also induced the expression of the Hippo pathway components Mst1/2, Lats1, Mob1b, Yap1, and Taz, as well as the downstream Hippo pathway targets Areg, Amotl2, and Cyr61, although it reduced the expression of the anti-apoptotic gene Birc5. Inhibition of the Hippo pathway effector YAP1 with verteporfin resulted in the attenuation of PFOA-induced follicular growth and proliferation (Clark et al., 2022). [The Working Group noted that PFOA can disrupt the Hippo pathway, leading to changes in the cell cycle, increased cell growth, and enhanced follicle development.]

Rat liver epithelial cells were cultured with PFOA at 10 µM, 50 µM, or 100 µM for 38 weeks and compared with passage-matched control cells. PFOA-treated cells showed increases in MMP-9 secretion and cell migration, and they developed more and larger colonies in soft agar. Microarray data showed Myc pathway activation at 50 µM and 100 µM, associated with Myc upregulation and PFOA-induced morphological transformation. Western blotting confirmed that PFOA caused significant increases in c-MYC protein expression in a time- and concentration-related manner. The tumour invasion indicators MMP-2 and MMP-9, the cell-cycle regulator cyclin D1, and the oxidative stress protein GST were all significantly upregulated at 100 µM (Qu et al., 2023).

The role of Wnt/ β -catenin signalling in PFOS-induced neurotoxicity has also been investigated. C17.2 neural stem cells (mouse-derived multipotent neural stem cells isolated from the cerebellum) were treated with PFOS at 12.5, 25, 50, 100, or 200 nM for \leq 48 hours (Dong et al., 2016). The CCK-8 assay was used to count the cells and indicated that PFOS exposure impaired the proliferation of the cells in a dose-dependent and time-dependent (at 50 nM) manner. Furthermore, flow cytometry analyses of the cell-cycle distribution revealed a decrease in the number of cells in S phase. The protein levels of cyclin D1 and PCNA were significantly reduced after PFOS exposure for 12 hours, compared with the control group. Gene expression of Myc and Cox2 and survival were significantly impaired in a dose- and time-dependent manner after PFOS exposure, suggesting the involvement of β -catenin signalling (Dong et al., 2016).

In ovine primary theca cells, PFOS at 50 ng/mL was not cytotoxic after 24 hours and had no effect on GJIC (<u>Gingrich et al., 2021</u>).

Synopsis

[The Working Group noted that the above studies reported some evidence that PFOA increased cell proliferation in human primary cells. Transcriptomic analyses in human primary cells suggested that PFOA modulated gene signalling pathways involved in cell proliferation. PFOA induced cell proliferation, migration, or invasion in human cell lines and cell proliferation or hyperplasia in multiple tissues in rodents, including in PPARa-null mice.

Similarly, PFOS seemed to increase cell proliferation, migration, or invasion in human cell lines and cell proliferation or hyperplasia in the liver of rats.]

4.2.11 Multiple key characteristics of carcinogens

See <u>Tables 4.29</u> to <u>4.31</u>.

Data from transcriptomic and metabolomic studies were assessed for quality with respect to study design parameters, quality control of the raw data, data processing criteria, and differential analysis and information content regarding the 10 KCs of carcinogens. The results of the studies were mapped/associated to the KCs. To establish the association, the Working Group mapped genes, metabolites, or enriched pathways to each KC, based on the known association of the transcript or metabolite with a biological process underlying the KC.

(a) Humans

(i) Exposed humans

<u>Chang et al. (2022)</u> used a meet-in-the-middle approach to investigate the interrelationships between serum PFOA and PFOS concentrations, maternal metabolomic perturbations, and fetal growth. The authors showed an association of maternal serum PFOA level with reduced fetal growth in a population of 313 African-American women. Changes in amino acid, lipid and fatty acid, and bile acid metabolism were associated with PFOA, and to a lesser extent PFOS, exposure. Uric acid was suggested to be a potential intermediate biomarker of the early response to PFOA exposure and to predict reduced fetal growth (see Table 4.30). This study was in good agreement with a previous cross-sectional study performed in the US adult population (NHANES 2009–2014; n = 4917), which showed associations of PFOA and PFOS exposure with serum uric acid and gout (Scinicariello et al., 2020).

Rhee et al. (2023) identified several metabolites that were significantly associated with both *n*-PFOA and *n*-PFOS in eight nested case– control serum metabolomic profiling studies as part of the Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial. The strongest associations were observed for sphingolipids, fatty acid metabolites, and androgenic steroids. [The Working Group noted that sphingolipids, derivatives of the amino alcohol sphingosine, are biologically active components of cell membranes that play a significant role in intracellular signal transduction, regulate cellular processes (e.g. proliferation, maturation, and apoptosis), and are involved in cellular stress responses.] The associations of specific metabolites with *n*-PFOS remained significant after the model was adjusted for serum *n*-PFOA, but the *n*-PFOA–metabolite associations were substantially attenuated after adjustment for *n*-PFOS. [The Working Group noted that differences in the associations of PFOA or PFOS levels with metabolites in patients with cancer and controls were not discussed in this study.]

A metabolome-wide association study using non-targeted ultra-high-resolution mass spectrometry identified metabolites associated with serum PFOA and PFOS levels in 115 children aged 8 years (Kingsley et al., 2019). In this cross-sectional study, serum PFOA and PFOS concentrations correlated with lipid metabolism and with arginine, proline, aspartate, asparagine, and butanoate metabolism. In addition,

End-point	Location, setting, study design	Sample, tissue, or cell type No. of study participants	Methods	Type of exposure	Results	Covariates adjusted for	Potentially related KCs	Comments	Reference
MWAS with pathway enrichment analysis	Emory University African American Vaginal, Oral, and Gut Microbiome in Pregnancy Study, Atlanta, Georgia, USA Prospective birth cohort study	Serum 313 pregnant African- American women at 8–14 wk of gestation	Isotope dilution calibration Untargeted, high-resolution metabolomic profiling using hydrophilic interaction LC with positive ESI and reverse- phase (C18) chromatography with negative ESI Multiple linear regression models	General population PFOA, Q1: < LOD- 0.45; Q4: 1.07-4.42 ng/mL PFOS, Q1: < LOD- 1.44; Q4: 3.24-12.4 ng/mL	Significant association was found between serum PFOA (OR, 1.20; 95% CI, 0.94–1.49) and small- for-gestational age birth; and correlations with level 1 biomarkers (uric acid and ferulic acid) and level 2 biomarkers (unsaturated fatty acid C18:1, parent bile acid, and bile acid– glycine conjugate) PFOS correlated with 2-hexyl-3-phenyl-2- propenal and parental bile acid	Maternal age, education, BMI, parity, tobacco use, marijuana use, and infant sex	KC5; lipid metabolism		<u>Chang</u> <u>et al.</u> (2022)

Table 4.29 Omics data relevant to multiple key characteristics of carcinogens in humans exposed to PFOA or PFOS

Table 4.29	(continued)								
End-point	Location, setting, study design	Sample, tissue, or cell type No. of study participants	Methods	Type of exposure	Results	Covariates adjusted for	Potentially related KCs	Comments	Reference
MWAS	One of 8 nested case– control studies conducted within the PLCO cohort, 1993–2001, collected at 10 screening centres across the USA Cross-sectional	Serum 3647 participants (1818 patients in the cancer study and 1829 controls)	UHPLC-MS/ MS, multiple linear regression modelling and meta- analysis using DerSimonian and Laird random effects models were used to combine findings from the various studies. Bonferroni- corrected significance threshold applied	General population. PFOA: < 2.9– > 6.5 µg/L PFOA: < 19.1– > 47.12 µg/L	38 <i>n</i> -PFOA-associated metabolites, including 17 lipids, two nucleotides (uric acid, 3-methylcytidine), one carbohydrate (1,5-anhydrosorbitol), and one cofactor/ vitamin 51 <i>n</i> -PFOS-associated metabolites, including 20 lipids (sphingolipids, fatty acid metabolites, and bile acid metabolites), five cofactors/ vitamins, three amino acids, two nucleotides (guanosine, 3-methylcytidine), and one carbohydrate (D-glucose)	Estimated glomerular filtration rate, educational attainment, BMI, sex	KC4, KC10, lipid metabolism	Study used semi-targeted metabolomic analyses, measuring relative metabolite levels, which made direct comparison between populations difficult. Use of non-fasted serum samples may lead to short-term changes in metabolite levels and introduce bias. The majority of participants were postmenopausal women.	<u>Rhee</u> <u>et al.</u> (2023)
MWAS	Cincinnati, Ohio, USA HOME Study, a prospective pregnancy and birth cohort Cross-sectional	Serum 115 children aged 8 yr	Non-targeted, high-resolution metabolomic profiling using LC and Fourier transform high- resolution MS	General population PFOA, mean ± SD, 2.6 ± 1.0 ng/mL PFOS, 4.4 ± 3.2 ng/mL	Serum PFOA and PFOS concentrations correlated with TCA cycle, pyrimidine and purine metabolism, changes in de novo fatty acid biosynthesis	Age, sex, race/ ethnicity	KC10, lipid metabolism	Results based on enriched pathways in network-based metabolome- wide correlation analysis.	<u>Kingsley</u> <u>et al.</u> (2019)

End-point	Location, setting, study design	Sample, tissue, or cell type No. of study participants	Methods	Type of exposure	Results	Covariates adjusted for	Potentially related KCs	Comments	Reference
MWAS	CHDS (1959– 1967), Oakland, California, USA Cross-sectional	Maternal perinatal serum 397 participants	C18 chromatography coupled with quadruple orbitrap MS Metabolic features were annotated with xMSannotator using the Human Metabolome Database	General population PFOA median (25th–75th percentile), 0.4; 0.25–0.6 ng/mL PFOS, 33.9; 16.05–61 ng/mL	301 metabolites were significantly associated with serum PFOS concentration. Pathway enrichment analyses: glycine, threonine, alanine, and serine metabolism, and urea cycle/amino group metabolism, carnitine shuttle, lysine metabolism, and branched- chain amino acid metabolism	Total cholesterol (continuous variable), age (continuous variable), and <i>p</i> , <i>p</i> '- DDE level (continuous variable)	KC10	Study included 50 women whose daughters developed breast cancer, but the association with metabolomic changes was not discussed. Linear regression model used.	<u>Hu et al.</u> (2019)
Transcriptomics	C8 Health Project (2005– 2006), Mid- Ohio Valley and Parkersburg, West Virginia, USA Cross-sectional	Serum and whole blood 290 participants	TaqMan Low- density array (quantitative RT-PCR), solid- phase extraction, followed by reverse-phase HPLC-MS/MS	General population PFOA GM (95% CI): 40.9 (33.7–49.5) ng/ mL in men; 25.5 (20.7–31.2) ng/ mL in women PFOS: 8.3 (7.3–9.45) ng/ mL in men; 5.5 (20.7–31.2) ng/ mL in women	In men, inverse associations of PFOA level were identified with ABCG1, NPC1, and PPARA transcripts; no associations with PFOS In women, an inverse association between the NR1H2 (LXRB) transcript with PFOA and a positive association of PFOA with NCEH1 expression were identified; NCEH1 and PPARA expression were positively correlated with the layed of DFOS	Age, sex, BMI, household family income, smoking status	KC8	Adjusted linear regression model.	Fletcher et al. (2013)

Table 4.29 (continued)											
End-point	Location, setting, study design	Sample, tissue, or cell type No. of study participants	Methods	Type of exposure	Results	Covariates adjusted for	Potentially related KCs	Comments	Reference		
Transcriptomics	Postmenopausal Norwegian women (age 48– 62 years), part of Norwegian Women And Cancer Study Cross-sectional	Serum and whole blood RNA 270 healthy participants	HPLC-QTOF- MS and AB Human Genome Survey Microarray V2.0	General population PFOA median (range), 4.4 (0.79–0.21) ng/ mL PFOS median range, 19 (5.7–84) ng/mL	Higher levels of PFOS were associated with the TCA cycle pathway ↓ NNT, PDHB, SDHD, SDHC, SUCLA2, IDH3A, MDH1, and SUCLG2 expression ↑ ACO2	None	KC10	No covariates adjusted for	<u>Rylander</u> <u>et al.</u> (2011)		

BMI, body mass index; CHDS, Child Health and Development Studies; CI, confidence interval; *p,p'*-DDE, dichlorodiphenyldichloroethylene; ESI, electrospray ionization; GM, geometric mean; HOME, Health Outcomes and Measures of the Environment; HPLC-MS/MS, high-performance liquid chromatography-tandem mass spectrometry; HPLC-QTOF-MS, high-performance liquid chromatography-quadrupole time-of-flight mass spectrometry; KC, key characteristic of carcinogens; LC, liquid chromatography; LOD, limit of detection; MS, mass spectrometry; MWAS, metabolome-wide association study; OR, odds ratio; PFOA, perfluorooctanoic acid; *n*-PFOA, linear isomer of perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PLCO, Prostate, Lung, Colorectal and Ovarian Cancer Screening Trial; Q, quartile; RT-PCR, reverse transcriptionpolymerase chain reaction; SD, standard deviation; TCA, tricarboxylic acid; UHPLC-MS/MS, ultra-high performance liquid chromatography-tandem mass spectrometry; USA, United States of America; wk, weeks(s); yr, year(s). in reversed-phase chromatography with ESI operated in negative mode (C18-negative mode), tyrosine, galactose, vitamin A (retinol), and lysine metabolism pathways, de novo fatty acid biosynthesis, tricarboxylic acid (TCA) cycle, pyrimidine, and purine metabolism pathways were enriched for both PFOA and PFOS. Thus, this study indicated that the serum levels of PFOA and PFOS were associated with a common set of biological pathways related to energy production and catabolism (Kingsley et al., 2019).

Hu et al. (2019) investigated the metabolite profiles associated with the serum PFOS levels of 397 participants in the Child Health and Development Studies (CHDS) cohort. Using a linear regression model, the authors identified 301 metabolites that were significantly associated with the serum PFOS concentration. Pathway enrichment analysis showed that these metabolites were associated with glycine, threonine, alanine, and serine metabolism; the urea cycle/ amino group metabolism; the carnitine shuttle; lysine metabolism, which generates carnitine; and branched-chain amino acid metabolism (i.e. valine, leucine, and isoleucine degradation). This cluster of pathways suggested a mechanistic link of PFOS with energy production and lipid regulation. Metabolites involved in the urea cycle and amino group metabolism, such as β -alanine, creatinine, pipecolate, lysine, arginine, creatine, and adrenochrome, were increased in the samples with high PFOS levels; and homocysteine and betaine negatively correlated with the PFOS concentration in serum. [The Working Group noted that there were associations between PFOA or PFOS levels and several common metabolites across multiple human studies, identified using a metabolomic approach.]

<u>Guo et al. (2022b)</u> summarized the application of non-targeted metabolomics in epidemiological studies that assessed metabolite and metabolic pathway alterations associated with exposure to PFOA (11 studies) and PFOS (10 studies) in a scoping review that included human studies involving children and adolescents (three studies), non-pregnant adults (five studies), and pregnant women (three studies). Alterations in tryptophan metabolism and the urea cycle were associated with PFOA or PFOS exposure in multiple studies. Lipid metabolites involved in glycerophospholipid metabolism, which is critical for biological membrane function, and fatty acids and carnitines, which are relevant to the energy supply pathway of fatty acid oxidation, were also associated with PFOA and PFOS exposure. Secondary significant metabolome changes associated with PFOA and PFOS exposure included the components of the TCA cycle, which is involved in energy generation, and purine and pyrimidine metabolism, which are cellular energy pathways. [The Working Group noted that there were commonalities in the associations of PFOA or PFOS level with non-targeted metabolomic alterations. Because of the cross-sectional nature of the studies, the authors were unable to rule out the temporality of the observed associations and minimize confounding bias and measurement errors. The studies had limitations in power, with sample sizes of < 1000 participants and relatively low PFOA and PFOS exposure levels.]

Fletcher et al. (2013) conducted transcriptional profiling of 13 genes in whole-blood samples from 290 out of 69 000 participants in the C8 Health Project that was carried out during 2005-2006 to examine the potential health effects of PFAS on residents of the Mid-Ohio Valley, USA, who lived in six contaminated water districts surrounding a chemical plant. The data showed inverse associations of serum PFOA concentration with the NR1H2 (LXRB), NPC1, and ABCG1 genes, which are involved in cholesterol transport; a positive association between serum PFOS level and the cholesterol mobilization-related NCEH1 gene, and a negative association with the NR1H3 gene, which is involved in cholesterol transport, were also identified. Moreover, the authors noted sex-specific differences in the expression of genes related to

cholesterol mobilization and transport and serum PFOA and PFOS levels. Inverse associations of PFOA level with *ABCG1*, *NPC1*, and *PPARA* transcripts was shown in male participants, but no associations of PFOS with any transcript. In women, the authors showed an inverse association of *NR1H2* (*LXRB*), but a positive association of *NCEH1* gene expression, with the PFOA level. The levels of the *NCEH1* and *PPARA* transcripts also positively correlated with the level of PFOS in women.

In another study, the applicability of peripheral blood transcriptomics for exploration of the effects of PFOA and PFOS exposure on 270 healthy postmenopausal Norwegian women (age 48-62 years) was investigated (<u>Rylander et al.</u>, <u>2011</u>). The authors identified two significantly dysregulated gene sets related to the TCA cycle in the "PFOS high" group (> 30 ng/mL; n = 42), compared with the "PFOS low" group (< 30 ng/mL; n = 228), but no significantly enriched genes in the tested sets were detected in the groups with different PFOA levels. Eight key genes (NNT, PDHB, SDHD, SDHC, SUCLA2, IDH3A, MDH1, and SUCLG2) were downregulated, and one, ACO2, was upregulated, in the group with higher level of PFOS. [The Working Group noted that the presence of PFOA and PFOS in the blood of all the participants may have reduced the chance of detecting differentially expressed single genes or metabolites.]

(ii) Human cells in vitro

See <u>Table 4.30</u>.

Buhrke et al. (2015) conducted a study to investigate the potential PFOA-mediated alterations in the transcriptome of human primary hepatocytes. PFOA exposure affected the PPARa pathway, influenced by substantial gene expression alterations, including upregulation of *PPARA*, *JUN*, and *FOS*, and downregulation of *ER1* and *HNF4a*, which is an important factor for liver development and embryogenesis. Activation of the PPARa network and the inhibition of copper-transporting ATPase 2 (ATP7B), sterol regulatory element-binding transcription factors 1 and 2 (SREBF 1 and SREBF 2), sterol regulatory element-binding protein cleavage-activating protein (SCAP), and insulin receptor (INSR) networks of differentially expressed genes was shown in 3D human primary hepatocyte spheroids treated with PFOA or PFOS for 14 days (Rowan-Carroll et al., 2021).

In addition, <u>Rosen et al. (2013)</u> treated human primary hepatocytes with 12 different perfluoroalkyl acids, including PFOA and PFOS, and showed that *CYP2B6*, *CYP3A4*, *PLIN2*, and *FABP1* were among the most upregulated genes.

Proteomic studies showed that the inhibition of GRP78, HSP27, CTSD, HNRNPC, HUWE1, UBQLN1, and hnRNPC, and the activation of PAF1, may be involved in the activation of p53, which triggered the apoptotic process in human hepatic L-02 cells treated with PFOA (Huang et al., 2013, 2014).

Li et al. (2023b) combined metabolomic and proteomic analyses to investigate the altered profiles in metabolite and protein levels in human primary hepatocytes exposed to PFOS at human exposure-relevant concentrations. The authors showed that an alteration in glycerophospholipid metabolism was the most significant lipid metabolism dysregulation induced by PFOS in hepatocytes and was associated with the intracellular transport process.

A transcriptomic analysis performed in HepaRG cells exposed to PFOS showed dose-dependent dysregulation of genes involved in a PPARα-regulated network, cholesterol biosynthesis, ATF4-activated genes in response to ERS, cytosolic tRNA aminoacylation, and amino acid transport across the plasma membrane (Louisse et al., 2023) (see also Section 4.2.8).

PFOA induced changes in the levels of lipid metabolites, arachidonic acid, myristic acid, and oleic acid in L-02 cells; precursors associated with nucleic acid synthesis (e.g. adenine and guanosine diphosphate) in DLD-1 cells; and

To do not not	T:	F	M.d. J.	D14	D-44-11	Commente	Deference
Ena-point	type or line	Exposure concentration or range, duration	метоаз	Results"	related KCs	Comments	Kelerence
Transcriptomics	Primary human hepatocytes	1, 25, or 100 μM PFOA, 24 h	Human genome GeneChips HG-U133 plus 2.0 (Affymetrix) Statistically significant dysregulated genes were identified by one- way ANOVA (<i>P</i> < 0.05)	43 genes were significantly dysregulated (FC, > 2; P < 0.05) after 1, 25–109, or 100–215 µM PFOA exposure The PPARα-dependent signalling pathway was identified by IPA to be the most significantly dysregulated pathway Gene expression: $\uparrow PPARA$, $\downarrow ESR1$, $\downarrow HNF4A$, $\uparrow JUN$, and FOS	KC8, KC10		<u>Buhrke et al.</u> (<u>2015)</u>
Gene expression	3D human primary hepatocytes	0.02–100 μM PFOA or 0.02– 100 μM PFOS, 14 days	RNA sequencing human TempO-Seq S1500 panel DESeq2 v1.30, FDR- adjusted <i>P</i> < 0.05 and FC > 1.5	PFOA and PFOS-induced changes in gene expression related to cholesterol biosynthesis and lipid metabolism, and PPARα activation network	KC8	Small number of genes analysed.	<u>Rowan-</u> Carroll et al. (2021)
Gene expression	Human primary hepatocytes	0–200 μM PFOA or 0–225 μM PFOS, 48 h	Custom 48-gene TaqMan low-density RT-PCR arrays; the 2- ^{AACt} method was used; dose-response data were evaluated using SAS jmp	CPT1A, ANGPTL4, PLIN2, and APOA2 were the most dose-dependently responsive genes to PFOA and PFOS exposure Upregulation of CYP2B6, CYP3A4, and FABP1 after PFOA or PFOS exposure	Lipid metabolism	Small number of genes were analysed; analysis performed in context of other PFAA.	<u>Rosen et al.</u> (2013)

Table 4.30 Omics data relevant to multiple key characteristics of carcinogens in human cells in vitro exposed to PFOA or PFOS
Table 4.30	(continued)						
End-point	Tissue, cell type or line	Exposure concentration or range, duration	Methods	Results ^a	Potentially related KCs	Comments	Reference
Proteomics	Human non-tumour hepatocytes (L-02)	0, 25, or 50 mg/L, 72 h PFOA	2D fluorescence DIGE coupled with ultrafleXtreme MALDI- TOF/TOF MS Results with <i>P</i> < 0.05, which is equal to a Mascot score > 33 were considered to be a positive identification	~1500 protein spots were detected in the DIGE gels at pH 4–7, and 28 protein spots were statistically significantly changed (1.5- fold increase or decrease, P < 0.05) 24 spots were downregulated and 4 were upregulated by PFOA The identified proteins were associated with cancer, cell death and survival, and cellular development networks; TP53 (p53), ERK1/2, and STAT3 were the key regulators of these networks	KC10	Small number of proteins detected.	<u>Huang et al.</u> (2013)
Proteomics	Human non-tumour hepatocytes (L-02)	0, 25, or 50 mg/L, 72 h PFOS	iTRAQ labelling and 2D nanoLC-MS/MS analysis. The acquired peak lists for all the MS/MS spectra were searched using the Mascot search engine (<u>Matrix Science, 2023</u>); IPA network analysis of proteins	~1300 proteins were identified and quantified at more than a 95% CI with an FDR < 0.99% 18 proteins were significantly differentially expressed (11 upregulated and 7 downregulated) in a dose- dependent manner by PFOS exposure Differentially expressed <i>HNRNPC</i> , <i>HUWE1</i> , <i>UBQLN1</i> , <i>RPL21</i> , and <i>PAF1</i> shown to be involved in p53 and c-myc networks, which are associated with DNA replication, recombination, and repair, RNA post- transcriptional modification, and the cell cycle	KC10	Small number of proteins detected.	<u>Huang et al.</u> (2014)

Table 4.30 (co	ntinued)						
End-point	Tissue, cell type or line	Exposure concentration or range, duration	Methods	Resultsª	Potentially related KCs	Comments	Reference
Metabolomics and proteomics	Primary human hepatocytes	10 μM PFOS, 48 h	UHPLC-MS/MS coupled with quadrupole-Orbitrap high-resolution MS for lipidomics and metabolomics; U3000nano RSLC nanoLC interfaced with a high-resolution MS system and XCalibur 4.3 software for proteomics; DEPs were defined using $\log_2 FC$, ≥ 1.0 ; $P < 0.05$	82 metabolites present at different concentrations were placed into 13 classes after PFOS treatment; these were involved in glycerophospholipid metabolism and phosphatidylethanolamine biosynthesis 55 proteins (27 upregulated and 28 downregulated) were significantly changed by PFOS treatment; intracellular transport, nuclear lumen, and ribonucleoprotein complex pathway were significantly upregulated; and amide transport and establishment of protein localization to organelle pathway were significantly downregulated after PFOS treatment	KC5, KC10		<u>Li et al.</u> (2023b)
Transcriptomics	HepaRG cells	6.25, 12.5, 25, 50, 100, 200, or 400 μM PFOS, 24 h	Whole-genome gene expression microarray and BMDExpress as the software tool; ANOVA was used, <i>P</i> < 0.05; Benjamini–Hochberg applied; FC filter, 1.0	18 Reactome gene sets were upregulated and 90 downregulated. 10 genes were selected that showed clear concentration- response curves for PFOS and were involved in diverse biological processes (ATF4, SLC7A11, YARS1, PDK4, ANGPTL4, LSS, HMGCR, OAT5, THRSP, and CXCL10	KC8, KC10		<u>Louisse</u> <u>et al. (2023)</u>

Table 4.30 (c	ontinued)						
End-point	Tissue, cell type or line	Exposure concentration or range, duration	Methods	Results ^a	Potentially related KCs	Comments	Reference
Metabolomics	A549 (lung), DLD-1 (intestine) and L-02 (liver) cells	0, 100, or 300 μM PFOA, 48 h	UHPLC with MS system and quadruple orbitrap MS with switching positive and negative mode electrospray ionization; Compound Discoverer 2.1 software and Optimal Scaling analysis using SPSS 20.0 software were used for metabolomic analysis; one-way ANOVA was conducted to compare the relative abundances of metabolites between PFOA-treated groups and the controls	Number of different metabolites (DM) L-02: 10 DLD-1: 12 A549: 67 Most DMs were changed in dose-dependent manner L-02: lipid metabolites: arachidonic acid, myristic acid, and oleic acid DLD-1: precursors associated with nucleic acid synthesis (e.g. adenine and GDP) A549: lipids, amino acids, and carbohydrates In A549 and L-02, PFOA induced the production of pro-inflammatory interleukins (IL-1β, IL-6, IL-8, and IL-13)	KC6, KC10	Small number of metabolites detected.	Zhang et al. (2021c)

PFOA and PFOS

_
~
()
_
<
1
\circ
-
~
\cap
×
ር)
~
Ť.
S
,
ω
J

Table 4.30 (cor	ntinued)						
End-point	Tissue, cell type or line	Exposure concentration or range, duration	Methods	Resultsª	Potentially related KCs	Comments	Reference
Transcriptomics and lipidomics	Primary human lymphocytes	50 μM PFOS, 72 h	mRNA sequencing, edgeR was used to identify DEGs and \log_2 FC, ≥ 1.0 , $P < 0.05$ applied; HPLC-triple quadrupole time-of- flight MS and XCMS software were used for lipidomic analysis	530 DEGs (247 upregulated and 283 downregulated) were found in lymphocytes after PFOS treatment PFOS exposure dysregulated genes (<i>BHLHE41</i> , <i>DCSTAMP</i> , <i>FCRLA</i> , <i>MYO7B</i> , <i>NOTCH3</i> , <i>NTRK2</i> , <i>RARRES2</i> , <i>SDC2</i> , <i>SORT1</i> , <i>SPIB</i> , and <i>SPP1</i>) and lipids that play important roles in immune functions, such as lymphocyte differentiation, inflammatory response, and immune response PFOS induced changes in 96 metabolites, including 37 lipids associated with glycerophospholipid, sphingolipid, glycerolipid metabolism; adipocytokine signalling pathway; regulation of autophagy, and arachidonic acid metabolism	KC7, KC10	Single concentration tested.	<u>Li et al.</u> (<u>2020c)</u>

Table 4.30 (continued)													
End-point	Tissue, cell type or line	Exposure concentration or range, duration	Methods	Results ^a	Potentially related KCs	Comments	Reference						
Transcriptomics and metabolomics	PrECs	10 nM PFOA or PFOS, 3–4 weeks	Single-cell RNA sequencing using 10X Cellranger pipeline and Seurat package (Satija Lab, 2023). Agilent GC/MS system was used for metabolite detection. AMDIS 2.71 (NIST, 2023) database and MetaboAnalyst software were used for metabolite analysis. Principal components analysis and uniform manifold approximation and projection were used for cluster visualization. GSEA was performed on normalized gene expression data using Molecular Signatures Database (Broad Institute, 2023).	UMAP plots revealed cell 5 clusters in the PFOS-, and 8 clusters in the PFOA-, exposed prostaspheres. Both chemicals induced changes in expression of the luminal keratin genes <i>KRT8/KRT18</i> The cells lacked stemness and showed basal keratin gene expression Significant enrichment of pathways involved in cell replication including, E2F targets, G2/M checkpoint, and mitotic spindle; increased TNF α via NF κ B pathway and k-RAS signalling; IL-2, IL-6, and TGF β inflammatory response; and metabolic pathways (glycolysis, oxidative phosphorylation) were shown in PFOA and PFOS-treated spheroids; the top enriched metabolites were involved in glycine and serine metabolism, with an enhancement of glucose metabolism through the Warburg effect; the top individual metabolites significantly induced by PFOA and PFOS exposure included glycerol, glutamic acid, citric acid, urea, serine, alanine, and glucose	KC8, KC9, KC10	Single concentration tested.	<u>Hu et al.</u> (2022)						

IARC MONOGRAPHS - 135

Table 4.30 (cor	ntinued)						
End-point	Tissue, cell type or line	Exposure concentration or range, duration	Methods	Results ^a	Potentially related KCs	Comments	Reference
Transcriptomics	Human Burkitt lymphoma cell line (Namalwa)	100 μM PFOA, 48 h	RNA sequencing, DEGs identified using generalized linear models that incorporated empirical Bayesian methods; canonical pathways identified using IPA	The four most significantly downregulated genes were <i>RAG1</i> , <i>RAG2</i> , <i>TCL1A</i> , and <i>TFRC</i> . PFOA affected two cellular processes related to immune function, B-cell development and primary immunodeficiency signalling RT-PCR confirmed that PFOA and PFOS caused a time-dependent reduction in <i>RAG1</i> and <i>RAG2</i>	KC7	Single concentration tested.	Janssen et al. (2022)
Metabolomics and proteomics	MLTC-1 Leydig cells	0.1, 1, or 10 μM PFOA, 48 h	Proteomic data acquisition using NanoLC-MS/MS and the MaxQuant software (Max Planck Institute of Biochemistry, 2023) were used for protein identification and quantification. UPLC system coupled to a quadruple orbitrap MS and SIMCA-P software (v14.0) were used for multivariate analysis of metabolites; MetaboAnalyst software were used for metabolic and molecular network analysis	The expression levels of 67 proteins were significantly changed in PFOA-treated cells ($P < 0.05$; FC, ≥ 1.5 in 10 µM PFOA) involved in lipid and fatty acid metabolic processes, catabolic processes, and steroid hormone regulation Metabolic pathway analysis of 17 DMs showed that these metabolites were involved in lipid and fatty acid, amino acid, and carbohydrate metabolism and steroidogenesis IPA showed that ERK1/2, p38 MAPK, and cAMP were key regulators of 18 proteins and 7 metabolites related to steroid hormone regulation, and that fatty acid and lipid metabolism were affected by PFOA exposure	KC10		<u>Huang et al.</u> (2022b)

644

End-point	Tissue, cell type or line	Exposure concentration or range, duration	Methods	Resultsª	Potentially related KCs	Comments	Reference
Transcriptomics and proteomics	Human breast cancer cells (SKBr3)	10 or 50 μM PFOA, 48 h	RNA sequencing LC-MS/MS analysis using Flex Binary UHPLC System connected to a Hybrid quadrupole–Orbitrap mass spectrometer; the FC thresholds (> 1.05 or < 0.95) and P < 0.05 were used to identify DEGs or DEPs	PFOA induced 1390 DEGs and 136 of the DEGs were associated with DEPs; 14 genes/proteins were associated with calcium metabolism cAMP signalling pathway was identified as a key network dysregulated by PFOA exposure; ADORA1 was suggested to be a target for PFOA binding that may have induced the Gi-cAMP- PKA pathway and reduced the concentration of cAMP Low concentrations of PFOA inhibited ADORA2A expression, whereas a high concentration (50 μM) induced its expression, causing opposite cellular	KC8, KC10	The suggestion that PFOA can bind to ADORA1 or ADORA2A receptors was not experimentally confirmed.	<u>Li et al.</u> (2022g)

ADORA1, adenosine A1 receptor; ANOVA, analysis of variance; cAMP, cyclic adenosine monophosphate; CI, confidence interval; DEGs, differentially expressed genes; DEPs, differentially expressed proteins; 2D DIGE, 2D fluorescence difference gel electrophoresis; 2D nanoLC-MS/MS, two dimensional nanoliquid chromatography-tandem mass spectrometry; DMs, differential metabolites; ESI, electrospray ionization; FC, fold change; FDR, false discovery rate; GC-MS, gas chromatography-mass spectrometry; GDP, guanosine diphosphate; GSEA, gene set enrichment analysis; IL, interleukin; IPA, Ingenuity Pathway Analysis; IQR, interquartile range; iTRAQ, isobaric tags for relative and absolute quantitation; KC, key characteristic of carcinogens; LC-MS/MS, liquid chromatography-tandem mass spectrometry; LOD, limit of detection; MALDI-TOF/TOF MS, matrix-assisted laser desorptionionization-time of flight/time-of-flight mass spectrometry; MS, mass spectrometry; NF-κB, nuclear factor kappa-light-chain-enhancer of activated B cells; PCA, principal components analysis; PFAA, perfluoroalkyl acids; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PKA, protein kinase A; PPAR, peroxisome proliferator-activated receptor; PrECs, primary normal human prostate epithelial cells; RT-PCR, reverse transcription-polymerase chain reaction; STAT, signal transducer and activator of transcription; UMAP, uniform manifold approximation and projection; UPLC, ultra-high performance liquid chromatography.

^a ↑, increase; ↓, decrease.

lipids, amino acids, and carbohydrates in A549 cells. Regardless of the cell type, A549 or L-02, PFOA induced the production of pro-inflammatory IL-1 β , IL-6, IL-8, and IL-13, as determined using a non-targeted metabolomic approach (Zhang et al., 2021c).

Li et al. (2020c), using a transcriptomic analysis in human primary lymphocytes, showed that PFOS exposure induced changes in the expression of genes and lipids that play important roles in immune function, such as in lymphocyte differentiation, the inflammatory response, and immune responses (see Sections 4.2.6 and 4.2.7).

To investigate carcinogenic mechanisms in the prostate after chronic exposure to PFOA and PFOS, Hu et al. (2022) conducted a study in primary normal human prostate epithelial cells in the form of a serial passage of prostasphere cultures that were treated with 10 nM PFOA or PFOS for 3-4 weeks. Exposure to either PFAS caused a significant increase in the total number of spheres, indicating augmentation of stem cell symmetric self-renewal. Transcriptome analysis using single-cell RNA sequencing showed that both chemicals induced changes in the expression of the luminal keratin genes KRT8/18, and the cells lacked stemness and basal keratin gene expression. Significant enrichment of pathways involved in cell replication, including E2F targets, G2M checkpoint and mitotic spindle, increased TNF-α via the NF-κBpathway, kRAS signalling, IL-2, IL-6, TGF β , the inflammatory response, and metabolic pathways (glycolysis and oxidative phosphorylation), were shown in PFOA and PFOS-treated spheroids. Metabolomic gas chromatography-MS analysis of PFOA- and PFOSexposed prostaspheres showed an enrichment in metabolites involved in glycine and serine metabolism, with enhancement of anaerobic glucose utilization through the Warburg effect. The top individual metabolites that were significantly induced by PFOA and PFOS exposure included glycerol, glutamic acid, citric acid, urea, serine, alanine, and glucose (Hu et al., 2022).

A transcriptomic-based approach was also used to investigate the carcinogenic potential of PFOA and PFOS in several other in vitro systems, including the bladder (<u>Ye et al., 2022</u>) and breast cancer (<u>Li et al., 2022f</u>), and their immunosuppressive properties in human B lymphoma cells (Janssen et al., 2022).

(b) Experimental systems

Non-human mammalian in vivo

See <u>Table 4.31</u>.

A recent study investigated the effects of PFOA on the hepatic transcriptome of PPAR $\alpha^{-/-}$ and wildtype mice fed an HFD and treated with PFOA at 0.05 mg/kg or 0.3 mg/kg bw per day for 20 weeks (Attema et al., 2022). In the wildtype mice, the largest effects were observed in the high-dose PFOA group, with a total of 788 genes being significantly changed. Although the overall effects of high-dose PFOA were substantially reduced in PPAR $\alpha^{-/-}$ mice, 294 genes were still significantly altered by high-dose PFOA in the absence of PPARa. Of the genes induced by high-dose PFOA in the wildtype mice, 88% were dependent on PPARa. This result was confirmed by gene set enrichment analysis. Significant positive enrichment was observed for pathways related to xenobiotic metabolism, steroid hormone biosynthesis, and omega-6 fatty acid metabolism in wildtype and PPARa^{-/-} mice. Also, in PPAR $\alpha^{-/-}$ mice, many of the genes that were significantly upregulated were regulated by the rodent-specific PXR agonist pregnenolone 16a-carbonitrile, as well as by the CAR agonist 1,4-bis (2-(3,5-dichloropyridyloxy)) benzene, suggesting that they are PXR and CAR target genes.

<u>Rosen et al. (2008a)</u> compared the transcript profiles of the livers of wildtype and PPARa-null mice exposed to PFOA and concluded that the majority of the genes were dependent on PPARa. The independent genes were involved in lipid homeostasis and xenobiotic metabolism. The expression of many of the identified xenobiotic

Table 4.31 Omics data relevant to multiple key characteristics of carcinogens in non-human mammalian systems in vivo exposed to PFOA and PFOS

End-point	Species, strain, sex	Tissue	Route, dose duration, dosing regimen	Methods	Results	Relevant KCs	Comments	Reference
Transcriptomics	Mouse, C57BL/6J, male (wildtype and PPARa ^{-/-})	Liver	PFOA in drinking- water, 0.05 or 0.3 mg/kg bw per day, 20 wk	RNA-Seq, Limma analysis with cut-off $P \le 0.001$; FC, > 1.5	Effects of PFOA were mostly PPARα-dependent. PXR and CAR could be involved in the absence of PPARα.	KC8, lipid metabolism		<u>Attema</u> <u>et al.</u> (2022)
Transcriptomics	Mouse, wildtype and PPARa-null 129S1/SvlmJ, male	Liver	PFOA by gavage, 0, 1, or 3 mg/kg bw per day, 7 days	Applied Biosystems Mouse Genome Survey Microarrays, a two-way ANOVA across dose ($P \le 0.03$); post-hoc <i>t</i> -test of the least-square means was used to evaluate individual treatment effects ($P \le 0.0025$)	In PFOA-treated mice, the changes in transcripts related to fatty acid metabolism, inflammation, xenobiotic metabolism, and cell cycle regulation were PPARα- independent. Involvement of other PPAR isoforms in fatty acid metabolism and inflammation suggested.	KC6, KC8, KC10, and lipid metabolism	In PPARα-null mice, the number of DEGs was ~5 times as low as that in wildtype mice, which may have led to misinterpretation of the data.	<u>Rosen</u> <u>et al.</u> (2008a)
Proteomics	Mouse, BALB/c, male and female	Liver	PFOA by gavage, 0, 0.05, 2.5, or 5 mg/kg bw per day, 28 days	iTRAQ labelling, FC, ≥ 1.5	Dose-dependent proteomic changes: mitochondrial dysfunction, oxidoreductase activity, peroxisome proliferator activity, ion 236 binding, and transferase activity.	KC5, KC8, KC10	Livers of 10 mice pooled per group (not biological replicates).	<u>Li et al.</u> (<u>2017b)</u>

End-point	Species, strain, sex	Tissue	Route, dose duration, dosing regimen	Methods	Results	Relevant KCs	Comments	Reference
Transcriptomics	ICR mice	Liver	PFOA by gavage, 0 or 10 mg/kg bw per day, 7 days	RNA sequencing, edgeR 3.0.8. log ₂ (FC) > 1 and <i>P</i> < 0.05	PFOA induced 2426 DEGs associated with fatty acid and lipid metabolism, oxidative stress, alterations of liver cell proliferation and apoptosis of hepatocytes, liver inflammation, necrosis, hepatic steatosis, and steatohepatitis.	KC5, KC6, KC8, KC10, and lipid metabolism	Single dose.	<u>Li et al.</u> (2022g)
Transcriptomics and lipidomics	Mouse, CD-1, female	Neonatal testes (PND1) and male offspring at PND63	PFOS by gavage, 0, 0.3, or 3 mg/kg bw per day, throughout gestation	RNA sequencing, edgeR package. log ₂ (FC) > 0.3 and FDR < 0.05; LC-MS/MS using Kinetex C18 column	56 (low dose) and 319 (high dose) DEGs were associated with lipid metabolism, oxidative stress, and cell junction signalling in testes. Levels of adrenic acid, docosahexaenoic acid, and eicosapentaenoic acid were reduced in testes at PND1 by PFOS treatment; and LOX-mediated 5-HETE and 15-HETE derived from arachidonic acid were increased. Male offspring at PND63 showed reductions in serum testosterone and endidumal snerm count	KC5, KC8, KC10, lipid metabolism		<u>Lai et al.</u> <u>(2017b)</u>

648

-

_

_

Table 4.31 (co	ontinued)							
End-point	Species, strain, sex	Tissue	Route, dose duration, dosing regimen	Methods	Results	Relevant KCs	Comments	Reference
Transcriptomics	Rat, Sprague- Dawley	Liver	PFOA by gavage, 0, 1, 3, 5, 10, or 15 mg/kg bw per day, 21 days	Affymetrix rat genome 230 2.0 GeneChip For upregulated genes change <i>P</i> -value < 0.0025; downregulated genes change <i>P</i> -value < 0.9975	> 500 genes significantly altered ($P < 0.0025$, FC > 2) after exposure to PFOA at any dose, with the largest number being at 10 mg/kg (813 genes) and 15 mg/kg (667 genes) PFOA. Dysregulated genes were associated with fatty acid synthesis and degradation, mitochondrial fatty acid β -oxidation (7 genes), apoptosis, cell communication and adhesion, growth and cell cycle, signal transduction and regulation of hormones.	KC10, KC6, KC5, lipid metabolism		<u>Guruge</u> <u>et al.</u> (2006)
Transcriptomics	Rat, Sprague- Dawley (Crl:CD(SD) IGS BR), male	Liver	PFOA or PFOS by gavage (10 mL/kg) for 1, 3, or 5 consecutive days The PFOA and PFOS groups received 20 and 10 mg/kg bw per day, respectively	Microarray, differentially expressed genes filtered at P < 0.05; pathway perturbations were visualized by DrugMatrix	PFOA and PFOS exhibited PPARα agonist-like effects on genes associated with fatty acid homeostasis. PFOA and PFOS exposure also resulted in the downregulation of cholesterol biosynthesis genes.	KC8, lipid metabolism		<u>Martin</u> <u>et al.</u> <u>(2007)</u>

End-point	Species, strain, sex	Tissue	Route, dose duration, dosing regimen	Methods	Results	Relevant KCs	Comments	Reference
Transcriptomics	Mouse, C57BL/6, male	Liver	PFOA HFD for 24 wk + 8 wk PFOA, gavage, 1 mg/kg bw per day	RNA sequencing, DESeq2; log ₂ (fFC) > 1; and a Benjamini– Hochberg- corrected <i>P</i> < 0.1	1233 and 835 DEGs were detected in PFOA- treated chow and HFD groups, respectively. 11 lipid metabolism related pathways were increased by PFOA and most were correlated with the "clearance" (oxidation, hydrolysis, catabolism) of lipids, causing significant downregulation of the hepatic steatosis pathway in HFD-fed mice. KEGG pathway analysis showed pathways including "PPAR signalling pathway," "Fatty acid degradation," "Biosynthesis of unsaturated fatty acids," "Peroxisome," and "Chemical carcinogenesis" were enriched by PFOA exposure, regardless of diet, leading to activation of the PPARα target genes <i>Cyp4a10, Lpl</i> , and <i>Cd36</i> , and the CAR target genes <i>Cyp2b10</i> and <i>Cyp3a11</i> , but inhibited Pparγ and Pparδ signalling.	KC8, KC10, lipid metabolism		<u>Li et al.</u> (2019c)

End-point	Species, strain, sex	Tissue	Route, dose duration, dosing regimen	Methods	Results	Relevant KCs	Comments	Reference
Transcriptomics	Mouse, athymic nude, male	Prostate tumour xenografts	PFOS and an HFD; gavage of 0 or 10 mg/kg bw per day, 40 days	RNA sequencing of RWPE-kRAS xenografts	Synergistic effect on xenograft growth of PFOS and HFD was observed. Genes involved in pyruvate metabolism, glycolysis pathways, PPARα network, and chromatin organization were significantly upregulated by PFOS in tumours from mice fed an HFD.	KC8, KC4	Single dose; histone modification analysis was not coherent.	<u>Imir et al.</u> (2021)
Transcriptomics	Mouse, CD- 1, female	Fetal liver, lung	PFOA Gavage, 0, 1, 3, 5, or 10 mg/kg bw per day on GD1-GD17	Affymetrix mouse 430 2.0 expression GeneChips, two-way ANOVA across dose $(P \le 0.05); t$ -test of the least square means was used as a post-hoc test to evaluate individual treatment effects $(P \le 0.0025)$	Clear dose-response effects in both the fetal liver and lung, with more extensive gene expression changes in liver. In the fetal liver, the DEGs were associated with lipid transport, ketogenesis, glucose metabolism, lipoprotein metabolism, cholesterol biosynthesis, steroid metabolism, bile acid biosynthesis, phospholipid metabolism, retinol metabolism, proteosome activation, and inflammation. These were associated with PPARa (except bile acid and glucose metabolism). Genes related to fatty acid catabolism were changed in both the fetal liver and	KC5, KC6, KC10, lipid metabolism		<u>Rosen</u> <u>et al.</u> (2007)

Table 4.31 (co	ntinued)							
End-point	Species, strain, sex	Tissue	Route, dose duration, dosing regimen	Methods	Results	Relevant KCs	Comments	Reference
Transcriptomics	Mouse, CD-1 female	Fetal liver, lung	PFOS Gavage, 0, 5, or 10 mg/kg bw per day, GD1–GD17	Affymetrix mouse 430 2.0 expression GeneChips, two-way ANOVA across dose $(P \le 0.05)$; <i>t</i> -test of the least square means was used as a post-hoc test to evaluate individual treatment effects $(P \le 0.0025)$	PFOS induced similar gene expression changes in liver and lung to PFOA (Rosen et al., 2007), which were primarily related to PPARα activation. In fetal lung: <i>Cyp4a14</i> , enoyl-coenzyme A hydratase (<i>Ehhadh</i>), and fatty acid binding protein 1 (<i>Fabp1</i>).	KC5, KC6, KC10, lipid metabolism		<u>Rosen</u> <u>et al.</u> (2009)
Transcriptomics	Mouse, CD-1	Fetal liver	PFOS Corn oil gavage, from mating to GD18.5, 0.3mg/kgbwper day (equivalent to human tolerable daily intake of 150 ng/kg bw per day)	RNA sequencing, edgeR 3.0.8. log ₂ (FC) > 1 and a Benjamini– Hochberg- corrected <i>P</i> < 0.05	PFOS activated the synthesis and metabolism of fatty acids and lipids, caused liver damage, and affected liver development in the fetus. Wnt/b- catenin, Rac, and TGF-b pathways activated.	KC8, KC10, lipid metabolism	1 pool (3 samples) per group.	<u>Lai et al.</u> (2017a)
Transcriptomics	Rat, Sprague- Dawley	Fetal liver	PFOS Gavage, 0 or 0.3 mg/kg bw per day, GD2–GD20	Affymetrix RAE 230A microarray, $P \le 0.05$	225 upregulated and 220 downregulated genes. Peroxisomal proliferation pathway was dysregulated, but no change in <i>Ppara</i> gene. <i>Cyp7a1</i> reduced.	KC8, lipid and bile acid metabolism	Single dose.	<u>Bjork</u> et al. (2008)

Table 4.31 (continued)								
End-point	Species, strain, sex	Tissue	Route, dose duration, dosing regimen	Methods	Results	Relevant KCs	Comments	Reference
Transcriptomics	Rat, Sprague- Dawley	Blood	PFOS Gavage, 0, 2.5, or 5 mg/kg bw per day, 28 days	RNA sequencing, DEGseq R package. log ₂ (FC) > 1/ < -1 and Q-value < 0.001	DEGs in blood of treated rats were associated with spliceosome, B-cell receptor signalling pathway, acute myeloid leukaemia, protein processing in the endoplasmic reticulum, NF-κB signalling pathway, and Fc gamma R-mediated phagocytosis.	KC6, KC8, KC10	No transcriptomic data for the liver or kidney.	<u>Wang</u> <u>et al.</u> (2023b)
Transcriptomics and metabolomics	Rat, Sprague- Dawley	Livers of rat mothers	PFOS Gavage, 0, 0.03, or 0.3 mg/kg bw per day, during pregnancy (GD1–GD18)	RNA sequencing, DESeq and Q-values < 0.05; UPLC/MS Progenesis QI software, OPLS- DA model, FC > 1.5 or \leq 0.66, and VIP \geq 1	DEGs were related to several metabolic pathways, such as PPAR signalling, ovarian steroid synthesis, arachidonic acid metabolism, insulin resistance, cholesterol metabolism, unsaturated fatty acid synthesis, and bile acid secretion. Untargeted metabolomics identified 164 and 158 DMs These were enriched with respect to a-linolenic acid metabolism, glycolysis/ gluconeogenesis, glycerolipid metabolism, glucagon signalling pathway, and glycine, serine, and threonine metabolism.	KC8, lipid metabolism		<u>Yu et al.</u> (2023)

Table 4.31 (co	ntinued)							
End-point	Species, strain, sex	Tissue	Route, dose duration, dosing regimen	Methods	Results	Relevant KCs	Comments	Reference
Transcriptomics	Mouse, BALB/c, male	Spleen	PFOA Gavage, 0, 0.4, 2, or 10 mg/kg bw per day, 28 days	RNA sequencing, weighted gene co-expression network analysis	7043 DEGs, with enrichment in cell cycle, autoimmunity, and anaemia in the spleen after PFOA.	KC6, KC10	Only one dose (10 mg/kg) was investigated.	<u>Guo et al.</u> (2021b)
Transcriptomics	Mouse, wildtype and PPARα-null 129S1/SvlmJ, male	Liver	PFOS Gavage, PFOS 0, 3, or 10 mg/kg bw per day, 7 days	Applied Biosystems Mouse Genome Survey Affymetrix 430 2.0 GeneChips, two-way ANOVA across doses $(P \le 0.03)$; <i>t</i> -test of the least square means was used as a post-hoc test to evaluate individual treatment effects $(P \le 0.0025)$	The PPARa-dependent DEGs were associated with lipid metabolism, peroxisome biogenesis, proteasome activation, and inflammation; PPARa-independent DEGs were related to lipid metabolism and xenobiotic metabolism. Modest activation of CAR, and possibly PPAR γ and/ or PPAR β/δ was noted.	KC6, KC8, lipid metabolism		<u>Rosen</u> <u>et al.</u> (2010)

ANOVA, analysis of variance; CAR, constitutive androstane receptor; DEG, differentially expressed gene; DMs, differential metabolites; FC, fold change; FDR, false discovery rate; GD, gestational day; HETE, hydroxyeicosatetraenoic acid; iTRAQ, isobaric tags for relative and absolute quantitation; HFD, high-fat diet; KC, key characteristic of carcinogens; KEGG, Kyoto Encyclopedia of Genes and Genomes; LC-MS/MS, liquid chromatography-tandem mass spectrometry; LOX, LOX, lipoxygenase; OPLS-DA, orthogonal partial least squares discriminant analysis; PFOA, perfluorooctanoic acid; PFOS, perfluorooctanesulfonic acid; PND, postnatal day; PPAR, peroxisome proliferator-activated receptor; PXR, pregnane X receptor; UPLC-MS, ultra-high performance liquid chromatography-mass spectrometry; VIP, variable importance in the projection; wk, week(s).

metabolism genes is known to be under the control of the nuclear receptor CAR and the transcription factor NRF2 (<u>Rosen et al., 2008a</u>).

Exposure of male and female Balb/c mice to PFOA at 0.05, 0.5, or 2.5 mg/kg per day for 28 days induced dose-dependent changes in proteins associated with mitochondrial dysfunction, oxidoreductase activity, peroxisome proliferator activity, ion binding, and transferase activity (Li et al., 2017a). In another study, ICR mice treated with PFOA for 7 days showed transcriptomic changes associated with the metabolism of many fatty acids and lipids, and particularly fatty acid β -oxidation, long-chain fatty acid transport, and the acyl-CoA metabolic process (Li et al., 2022g).

Lai et al. (2017b) conducted transcriptomic and targeted lipidomic analyses using neonatal testes in an effort to identify molecular targets and lipid markers associated with in utero PFOS exposure at doses of 0.3 or 3 µg/g bw, corresponding to the general population and occupational exposure levels, respectively. Analysis of male offspring at PND63 showed significant reductions in serum testosterone and epididymal sperm count. After PFOS exposure, the levels of adrenic acid and docosahexaenoic acid in the testes were significantly reduced by the low and high PFOS concentrations, respectively. Exposure to PFOS significantly induced the generation of 5-HETE and 15-HETE from arachidonic acid by LOX in the testes. Pathway analysis of the transcriptomic data highlighted that PFOS exposure induced changes in redox responses and oxidation-reduction processes in neonatal testes (Lai et al., 2017b).

<u>Guruge et al. (2006)</u> showed that male Sprague-Dawley rats exposed to PFOA at 1, 3, 5, 10, or 15 mg/kg bw per day for 21 days were characterized by a moderately dose-dependent number of significantly dysregulated genes. The upregulated genes are involved in the metabolism of lipids, cell communication, adhesion, growth, apoptosis, hormone regulatory pathways, proteolysis and peptidolysis, and signal transduction. The downregulated genes are related to the transport of lipids, inflammation, cell adhesion, apoptosis, the regulation of hormones, metabolism, and G-protein-coupled receptor protein signalling pathways.

Martin et al. (2007), using transcriptomic analysis, showed that PFOA and PFOS exposure resulted in the downregulation of cholesterol biosynthesis genes and alterations to thyroid hormone metabolism genes. These effects were associated with a decrease in serum cholesterol and serum thyroid hormone depletion, respectively, in the livers of male rats after 1, 3 and 5 days of exposure.

NAFLD could be considered to be a risk factor and potentiate the toxic carcinogenic effects of chemicals. Based on these considerations, <u>Li et al. (2019c)</u> investigated the hepatic effects of PFOA in mice in which NAFLD had been induced. PFOA activated xenobiotic nuclear receptors, inflammation, and cell proliferation in the livers of mice fed an HFD. Transcriptomic analysis showed that PFOA activated PPARa, CAR, and PXR in the livers of mice fed a control diet or an HFD, but reduced the severity of hepatic steatosis and hepatic triglyceride levels, enhanced lipid oxidation pathways, and attenuated HFD-induced hepatic fibrosis.

<u>Imir et al. (2021)</u> investigated the impact of metabolic alterations induced by an HFD combined with PFOS exposure on prostate tumour progression by analysing prostate RWPE-kRAS xenograft tumour growth in vivo. PFOS exposure of athymic nude male that were fed an HFD-induced RWPE-kRAS xenograft tumour growth and caused alterations in metabolites associated with glucose metabolism via the Warburg effect, the transfer of acetyl groups into mitochondria, and the TCA cycle, and in particular pyruvate and acetyl-CoA. Gene set enrichment analysis identified genes involved in pyruvate metabolism and glycolysis pathways to be significantly upregulated by PFOS exposure in tumours in mice fed an HFD. These data

indicate that metabolic alterations induced by HFD combined with PFOS exposure may play a significant role in prostate tumour growth and progression.

Rosen et al. (2007, 2009) conducted a transcriptomic microarray analysis of the lungs and livers of fetuses from pregnant CD-1 mice exposed to PFOA and PFOS to investigate the mechanism whereby they induce developmental toxicity. The expression of genes related to fatty acid catabolism was altered in both the fetal liver and lung. In the fetal liver, exposure to PFOA or PFOS caused significant alterations in the expression of genes associated with lipid transport, ketogenesis, glucose metabolism, lipoprotein metabolism, cholesterol biosynthesis, steroid metabolism, bile acid biosynthesis, phospholipid metabolism, oxidative phosphorylation, retinol metabolism, proteosome activation, and inflammation. Interestingly, PFOA and PFOS altered the expression of genes related to lipid metabolism, inflammation, and xenobiotic metabolism in both wildtype and PPARa-null CD-1 mice, which was consistent with modest activation of CAR, and possibly PPARy and/or PPAR β/δ (Rosen et al., 2008b, 2010).

Similarly, prenatal PFOS exposure induced transcriptomic changes that may activate the synthesis and metabolism of fatty acids and lipids, leading to liver damage and interference with liver development in the fetuses of CD-1 mice (Lai et al., 2017a) and Sprague-Dawley rats (Bjork et al., 2008).

A transcriptomic analysis of blood samples from rats exposed to PFOS identified differentially expressed genes that were associated with the spliceosome, the B-cell receptor signalling pathway, acute myeloid leukaemia, protein processing in the ER pathway, NF- κ B signalling pathway, and Fc gamma R-mediated phagocytosis (Wang et al., 2023b).

Using transcriptome sequencing combined with non-targeted metabolomic assays, <u>Yu et al.</u> (2023) identified differentially expressed genes in the livers of Sprague-Dawley rats given PFOS at 0.03 or 0.3 mg/kg bw per day that were related to several metabolic pathways, such as PPAR signalling, ovarian steroid synthesis, arachidonic acid metabolism, insulin resistance, cholesterol metabolism, unsaturated fatty acid synthesis, and bile acid secretion. Non-targeted metabolomics identified 164 and 158 metabolites present at different concentrations in 0.03 and 0.3 mg/kg bw per day exposure groups, respectively, which could be associated with α -linolenic acid metabolism, glycolysis/gluconeogenesis, glycerolipid metabolism, glucagon signalling pathway, and glycine, serine, and threonine metabolism.

Yu et al. (2016) conducted a high-throughput targeted metabolomic study of 278 metabolites to investigate the effects of PFOA exposure for 28 days on the brains and livers of male Balb/c mice. This study aimed to link the metabolic profiles of the livers and brains of mice exposed to PFOA with alterations in the transcriptome and proteome, and PFOA-induced hepatomegaly and neurobehavioural effects. PFOA treatment induced metabolic changes in the brain and liver that were associated with the metabolism of amino acids, lipids, and carbohydrates. The energy and lipid metabolism pathways were more susceptible to PFOA exposure. Lipidomic data in mice exposed to PFOA suggested that the β -oxidation and biosynthesis of fatty acids and inflammation are involved in PFOA-induced hepatomegaly. An iTRAQ labelling quantitative proteomic technology was used for the global characterization of the liver proteome in mice exposed to PFOS at 1.0, 2.5, or 5.0 mg/kg bw for 24 hours (Tan et al., 2012). Seventy-one of 1038 unique detectable proteins were significantly dysregulated in the mouse livers after PFOS exposure, and these were involved in lipid metabolism, transport, biosynthetic processes, and the response to a stimulus. Long-chain acyl-CoA synthetase, acyl-CoA oxidase 1, bifunctional enzyme, 3-ketoacyl-CoA thiolase A, CYPs, and GSTs were identified as key enzymes that regulate

peroxisomal β -oxidation and the metabolism of xenobiotic compounds that were affected by PFOS exposure.

In the study by <u>Guo et al. (2021b)</u>, male mice were fed diet containing PFOA at a dose of 0, 0.4, 2, or 10 mg/kg per day for 28 days to investigate the splenic atrophy induced by PFOA. The authors demonstrated that mice exposed to PFOA reduced spleen weight and relative spleen weight and lower iron levels in the spleen and serum. Weighted gene co-expression network analysis of 7043 genes showed enrichment in those involved in cell cycle, autoimmunity, and anaemia in the spleen of PFOA-treated mice. PFOA exposure resulted in an increase in the ratio of the total number of macrophages to M1 macrophages in the spleen, the phagocytic ability of macrophages, and the levels of cytokines such as TNF- α , IL-1 β , and IL-6. These findings suggest that overactivation of macrophages may play an important role in the splenic atrophy induced by PFOA exposure.

[The Working Group noted that these transcriptomic alterations provide an insight into PFOA and PFOS exposure in relation to the KCs of carcinogens. In addition, the Working Group also noted that transcriptomic, metabolomic, and lipidomic data collected after PFOA or PFOS exposure provide information associated with KCs, such as inflammation, cell proliferation, stress responses, and lipid metabolism.]

Synopsis

[The Working Group noted that the above studies showed some evidence that human exposure to PFOA and PFOS alters pathways related to nutrient and energy supply. Metabolomic analyses in exposed humans have suggested that PFOA and PFOS increase the activities of glycolytic pathways. Transcriptomic analyses in human primary cells have shown alterations in cell proliferation pathways, and transcriptomic data in experimental systems have shown alterations in lipid metabolism pathways.]

4.3 Evaluation of high-throughput in vitro screening data

See <u>Tables 4.32</u> to <u>4.35</u>.

An analysis of the in vitro bioactivities of PFOA and PFOS was informed by data from high-throughput screening assays generated by the Toxicology in the 21st Century (Tox21) and Toxicity Forecaster (ToxCast) research programmes of the government of the USA (Thomas et al., 2018). PFOA, its ammonium salt APFO, PFOS, and its salt potassium perfluorooctanesulfonate, were among the thousands of chemicals tested in the large assay battery of the Tox21 and ToxCast research programmes of the US EPA and the US NIH. Detailed information about the chemicals tested, assays used, and associated procedures for data analysis is publicly available (US EPA, 2021). A supplementary table (Annex 6, Supplementary material for Section 4.3, Evaluation of high-throughput in vitro toxicity screening data, online only, available from: <u>https://publications.iarc.who.int/636</u>) provides a summary of the findings, including the assay names, the corresponding KCs, the resulting "hit calls", both positive and negative, and any reported caution flags for PFOA. The results were generated with the software "kc-hits" (key characteristics of carcinogens - highthroughput screening discovery tool) (Reisfeld et al., 2022; available from: https://gitlab.com/ i1650/kc-hits); the evaluations discussed in the present monograph were performed using the US EPA ToxCast and Tox21 assay data and the curated mapping of KCs to assays available at the time. The findings and interpretation of these high-throughput assays for PFOA and PFOS are discussed below. [The Working Group noted that for PFOA, its ammonium salt, PFOS, and its potassium salt, the chemical purity quality control rating was "Unknown/Inconclusive", so the results should be interpreted with caution.]

			PFOA
		Active (no flag)	Active with flag, or inactive
APFO	Active (no flag)	13	1
	Active with flag or inactive	2	270

Table 4.32 Number of assays available for the evaluation of high-throughput in vitro screening data for PFOA

APFO, ammonium perfluorooctanoate; PFOA, perfluorooctanoic acid.

4.3.1 PFOA

After mapping against the KCs of carcinogens, the ToxCast/Tox21 database contained 289 assay end-points in which PFOA was tested, and 288 assay end-points in which APFO was tested (<u>US EPA, 2023a</u>). Of these, PFOA and APFO were found to be active and without caution flags in 15 and 14 assay end-points, respectively, relevant to the KCs of carcinogens, with 13 assay end-points active without flags for either (see <u>Tables 4.32</u> and <u>4.33</u>).

PFOA and APFO were active in two assay end-points mapped to KC5, "induces oxidative stress". Both of these assays were performed in the HepG2 cell line, and both were related to transcription factor activity mapping to endogenous human NRF2, which regulates antioxidant response elements.

In addition, PFOA and APFO were both active in nine assay end-points mapped to KC8, "modulates receptor-mediated effects", with a tenth assay end-point for which PFOA was active with < 50% efficacy, and APFO was active without flags. Three assay end-points were for ER, one for antagonist activity in a human embryonic kidney cell line (HEK293T), and two for inducible changes in transcription in a human liver cancer cell line (HepG2). One assay using a human liver cancer cell line (HepaRG) was active for CYP2B6 induction, which is considered to be a marker for PXR-mediated metabolism. The remaining assays indicated activity for PPARa and PPARy, PPRE, and PXR response element, all of which are indicative of nuclear receptor activation.

Finally, PFOA and APFO were both active in three assays mapped to KC10, "alters cell proliferation, cell death, or nutrient supply", with a fourth assay end-point for which APFO was active with less than 50% efficacy, and PFOA perfluorooctanoate was active without flags in two of them. Two assay end-points measured cell viability and one measured reduced mitochondrial membrane potential, all in HepG2 cells. The fourth assay end-point was only active without flags for the effect of PFOA on cell viability in a human cervical cell line (ME-180).

4.3.2 PFOS

After mapping against the KCs of carcinogens, the ToxCast/Tox21 database contained 292 assay end-points in which PFOS was tested and 289 assay end-points in which potassium perfluorooctanesulfonate was tested (<u>US EPA, 2023b</u>). Of these, PFOS and potassium perfluorooctanesulfonate were found to be active and without caution flags in 32 and 31 assay end-points, respectively, relevant to the KCs of carcinogens, with 24 assay end-points active without flags for either (see <u>Tables 4.34</u> and <u>4.35</u>), the results of which are summarized below.

PFOS and potassium perfluorooctanesulfonate were active in two assay end-points mapped to KC2, "is genotoxic", with a third assay end-point being active with flags for PFOS and without flags for potassium perfluorooctanesulfonate. The first two of these were in HepG2 cell lines and related to p53 activation, but at different time points (24 and 72 hours, respectively),

Assay ID	Assay name	PFOA	L	APFO)
		AC ₅₀ (μM)	Flags	AC ₅₀ (μM)	Flags
KC5: Induc	ces oxidative stress				
1110	TOX21_ARE_BLA_agonist_ratio	29.06		43.91	
97	ATG_NRF2_ARE_CIS_up	47.89		116	
KC8: Modu	lates receptor-mediated effects				
786	TOX21_ERa_BLA_Antagonist_ratio	51.93		46.87	
117	ATG_ERa_TRANS_up	7.561		44.7	
75	ATG_ERE_CIS_up	10.36		162.6	
132	ATG_PPARa_TRANS_up	21.83		14.23	
134	ATG_PPARg_TRANS_up	124.1		114.7	
719	NVS_NR_hPPARg	23.35	a	26.36	
969	LTEA_HepaRG_CYP2B6_up	23.76		5.641	
142	ATG_RXRb_TRANS_up	65.13		37.67	
103	ATG_PXRE_CIS_up	35.28		40.99	
102	ATG_PPRE_CIS_up	116.2		86.49	
KC10: Alte	rs cell proliferation, cell death, or nutrient	supply			
2066	TOX21_HRE_BLA_Agonist_viability	23.43		48.35	b
64	ATG_AP_1_CIS_up	139.3		124.7	
51	APR_HepG2_MitoMembPot_72h_dn	116.1		111.2	
45	APR_HepG2_CellLoss_72h_dn	123.2		114.2	с

Table 4.33 High-throughput in vitro screening data for PFOA

 AC_{50} , concentration that elicits a half-maximal response; APFO, ammonium perfluorooctanoate; ID, identifier; KC, key characteristic of carcinogens; PFOA, perfluorooctanoic acid.

^a Less than 50% efficacy.

^b Only the highest concentration above baseline, active, < 50% efficacy.

° Unspecified flag.

and the third assay was related to DNA repair in a chicken lymphoblast cell line (DT40). Additionally, activity for PFOS was reported for one active and unflagged assay end-point for KC4, "induces epigenetic alterations". This was also performed in HepG2 cells and was related to transcription factor activity of the cis-acting elements in the reporter *Pax* genes. The same assay end-points were reported to be active, but only for one concentration above baseline for potassium perfluorooctanesulfonate.

PFOS was active in two assays mapped to KC5, "induces oxidative stress". Both of these assays were performed in HepG2 cells. One indicated transcriptional activation of the NRF2 promoter, which regulates antioxidant response elements, and this was also active for potassium

perfluorooctanesulfonate. The other indicated transcriptional activation that induces metallothioneins, and was reported to be active, but only for one concentration above baseline for potassium perfluorooctanesulfonate. Potassium perfluorooctanesulfonate was also reported to be active for a third assay end-point indicating increased stress kinases in HepG2 cell line, in which PFOS was not tested.

In addition, PFOS was active without flags in 10 assays mapped to KC8, "modulates receptor-mediated effects", seven of which were also active, without flags, for potassium perfluorooctanesulfonate. Three assays for PFOS were active for ER/PR, including progesterone antagonism in a kidney cell line (HEK293T) and two for transcriptional activation in a human liver cancer cell

Table 4.34 Number of assays available for the evaluation of high-throughput in vitro screening data for PFOS data

		I	PFOS
		Active (no flag)	Active with flag or inactive
KPFOS	Active (no flag)	24	7
	Active with flag or inactive	8	249

KPFOS, potassium perfluorooctanesulfonate; PFOS, perfluorooctanesulfonic acid.

Table 4.35 High-throughput in vitro screening data for PFOS

Assay ID	Assay name	PFOS	6	KPFC	S
		AC ₅₀ (μM)	Flags	AC ₅₀ (μM)	Flags
KC2: Is genote	oxic				
2131	TOX21_DT40_657	67.02	а	65.28	
60	APR_HepG2_p53Act_72h_up	5.435		111	
40	APR_HepG2_p53Act_24h_up	109.3		123.6	
KC4: Induces	epigenetic alterations				
100	ATG_Pax6_CIS_up	76.27		231.1	b
KC5: Induces	oxidative stress				
97	ATG_NRF2_ARE_CIS_up	10.22		30.7	
91	ATG_MRE_CIS_up	41.98		165.9	b
62	APR_HepG2_StressKinase_72h_up	NA		112	
KC8: Modulat	es receptor-mediated effects				
804	TOX21_TR_LUC_GH3_Antagonist	86.54	а	65.05	
2127	TOX21_PR_BLA_Antagonist_ratio	35.5		63.53	a, c
75	ATG_ERE_CIS_up	150.3		32.63	
117	ATG_ERa_TRANS_up	23.05		35.92	b
132	ATG_PPARa_TRANS_up	58.85		88.24	
719	NVS_NR_hPPARg	20.9		20.28	
963	LTEA_HepaRG_CYP1A1_up	21.51		NA	
102	ATG_PPRE_CIS_up	179.7		179.5	
135	ATG_PXR_TRANS_up	18.01		14.01	
103	ATG_PXRE_CIS_up	11.32		30.38	
134	ATG_PPARg_TRANS_up	167		73.43	
KC10: Alters of	ell proliferation, cell death, or nutrient supply				
2066	TOX21_HRE_BLA_Agonist_viability	41.58		26.33	
1195	TOX21_PPARd_BLA_Agonist_viability	57.77		32.44	
1121	TOX21_FXR_BLA_antagonist_viability	31.45		31.86	
2120	TOX21_ERb_BLA_Antagonist_viability	30.28		42.46	
2116	TOX21_ERb_BLA_Agonist_viability	25.45		42.18	
1188	TOX21_FXR_BLA_agonist_viability	30.13		29.07	d
2128	TOX21_PR_BLA_Antagonist_viability	32.85		55.94	
2082	TOX21_RT_HEK293_FLO_40hr_viability	27.66	с	29.09	
2080	TOX21_RT_HEK293_FLO_32hr_viability	27.93	с	28.81	
2078	TOX21_RT_HEK293_FLO_24hr_viability	28.14	с	29.16	

Assay ID	Assay name	PFOS	6	KPFOS	
		AC ₅₀ (μM)	Flags	AC ₅₀ (μM)	Flags
2077	TOX21_RT_HEK293_FLO_16hr_viability	29.81	c, d	23.96	
2124	TOX21_PR_BLA_Agonist_viability	27.65		40.5	
64	ATG_AP_1_CIS_up	18.83		23.71	
55	APR_HepG2_NuclearSize_72h_dn	8.097		105.1	
54	APR_HepG2_MitoticArrest_72h_up	113.8		107.5	
49	APR_HepG2_MitoMass_72h_dn	104.3		112.9	
45	APR_HepG2_CellLoss_72h_dn	110.6		110.9	
1326	TOX21_p53_BLA_p4_viability	81.43		45.36	
29	APR_HepG2_MitoMass_24h_dn	114.2		120.2	
251	BSK_hDFCGF_Proliferation_down	10		60	b, f
112	ATG_TGFb_CIS_up	107.6		101.5	

Table 4.35 (continued)

 AC_{50} , concentration that elicits a half-maximal response; ID, identifier KC, key characteristic; KPFOS, potassium perfluorooctanesulfonate; NA, not tested or not available; PFOS, perfluorooctanesulfonic acid.

^a Only the highest concentration above baseline, active.

^b Only one concentration above baseline, active.

^c Efficacy, < 50%.

^d Borderline active.

^e Multiple points above baseline, inactive.

f Hit-call potentially confounded by overfitting.

line (HepG2), although for potassium perfluorooctanesulfonate two of these had activity flags. Multiple assays were active for other nuclear receptors, including transcriptional activation of PPAR α in HepG2 cells, PPAR γ in HepG2 cells and a biochemical (cell-free, using extracted gene-proteins) assay, CYP1A1 induction in HepaRG cells, and the transcriptional activation of PXR in HepG2 cells. Additionally, thyroid hormone modulation was indicated in one assay end-point performed in a rat pituitary gland cell line (GH3) for potassium perfluorooctanesulfonate, but only at the highest concentration of PFOS.

Finally, PFOS and potassium perfluorooctanesulfonate were active without flags in 17 and 18 assay end-points, respectively, mapped to KC10, "alters cell proliferation, cell death, or nutrient supply", 14 of which overlapped. Many of these assay end-points measured cell viability in ME-180, HEK293T, HepG2, and HCT116 cells. Reduced proliferation, which is also indicative of a loss of viability under pro-inflammatory conditions, was detected in one assay for PFOS using human foreskin fibroblasts, with potassium perfluorooctanesulfonate being active at only one concentration above baseline. Reduced nuclear size 72 hours after treatment, reduced mitochondrial mass after 24 and 72 hours, and increased mitotic arrest after 72 hours were also detected in HepG2 cells. Increased transactivation of TGF β in HepG2 cells was also detected in one assay.

References

Abbott BD, Wood CR, Watkins AM, Tatum-Gibbs K, Das KP, Lau C (2012). Effects of perfluorooctanoic acid (PFOA) on expression of peroxisome proliferator-activated receptors (PPAR) and nuclear receptor-regulated genes in fetal and postnatal CD-1 mouse tissues. *Reprod Toxicol.* 33(4):491–505. doi:10.1016/j. reprotox.2011.11.005 PMID:22154759

- Abdellatif A, Al-Tonsy AH, Awad ME, Roberfroid M, Khan MN (2003–2004). Peroxisomal enzymes and 8-hydroxydeoxyguanosine in rat liver treated with perfluorooctanoic acid. *Dis Markers*. 19(1):19–25. doi:<u>10.1155/2003/135859</u> PMID:<u>14757943</u>
- Abe T, Takahashi M, Kano M, Amaike Y, Ishii C, Maeda K, et al. (2017). Activation of nuclear receptor CAR by an environmental pollutant perfluorooctanoic acid. *Arch Toxicol.* 91(6):2365–74. doi:<u>10.1007/s00204-016-1888-3</u> PMID:<u>27832320</u>
- Abraham K, Mielke H, Fromme H, Völkel W, Menzel J, Peiser M, et al. (2020). Internal exposure to perfluoroalkyl substances (PFASs) and biological markers in 101 healthy 1-year-old children: associations between levels of perfluorooctanoic acid (PFOA) and vaccine response. *Arch Toxicol.* 94(6):2131–47. doi:10.1007/ s00204-020-02715-4 PMID:32227269
- Abraham K, Monien BH (2022). Transdermal absorption of ${}^{13}C_4$ -perfluorooctanoic acid (${}^{13}C_4$ -PFOA) from a sunscreen in a male volunteer what could be the contribution of cosmetics to the internal exposure of perfluoroalkyl substances (PFAS)? *Environ Int.* 169:107549. doi:10.1016/j.envint.2022.107549 PMID:36191486
- Abudayyak M, Öztaş E, Özhan G (2021a). Assessment of perfluorooctanoic acid toxicity in pancreatic cells. *ToxicolIn Vitro*. 72:105077. doi:<u>10.1016/j.tiv.2021.105077</u> PMID:<u>33422635</u>
- Abudayyak M, Öztaş E, Özhan G (2021b). Determination of perflourooctanoic acid toxicity in a human hepatocarcinoma cell line. *J Health Pollut*. 11(31):210909. doi:<u>10.5696/2156-9614-11.31.210909</u> PMID:<u>34434601</u>
- Ahmad S, Wen Y, Irudayaraj JMK (2021). PFOA induces alteration in DNA methylation regulators and SARS-CoV-2 targets *Ace2* and *Tmprss2* in mouse lung tissues. *Toxicol Rep.* 8:1892–8. doi:10.1016/j. toxrep.2021.11.014 PMID:34853776
- Aimuzi R, Luo K, Chen Q, Wang H, Feng L, Ouyang F, et al. (2019). Perfluoroalkyl and polyfluoroalkyl substances and fetal thyroid hormone levels in umbilical cord blood among newborns by prelabor caesarean delivery. *Environ Int.* 130:104929. doi:10.1016/j. envint.2019.104929 PMID:31228788
- Aimuzi R, Luo K, Huang R, Huo X, Nian M, Ouyang F, et al.; Shanghai Birth Cohort Study (2020). Perfluoroalkyl and polyfluroalkyl substances and maternal thyroid hormonesinearlypregnancy. *EnvironPollut*.264:114557. doi:10.1016/j.envpol.2020.114557 PMID:32388293
- Ait Bamai Y, Goudarzi H, Araki A, Okada E, Kashino I, Miyashita C, et al. (2020). Effect of prenatal exposure to per- and polyfluoroalkyl substances on childhood allergies and common infectious diseases in children up to age 7 years: the Hokkaido study on environment and children's health. *Environ Int.* 143:105979. doi:10.1016/j.envint.2020.105979 PMID:32717646

- Alam MN, Han X, Nan B, Liu L, Tian M, Shen H, et al. (2021). Chronic low-level perfluorooctane sulfonate (PFOS) exposure promotes testicular steroidogenesis through enhanced histone acetylation. *Environ Pollut.* 284:117518. doi:10.1016/j.envpol.2021.117518 PMID:34261222
- Albrecht PP, Torsell NE, Krishnan P, Ehresman DJ, Frame SR, Chang SC, et al. (2013). A species difference in the peroxisome proliferator-activated receptor α-dependent response to the developmental effects of perfluorooctanoic acid. *Toxicol Sci.* 131(2):568–82. doi:<u>10.1093/toxsci/kfs318</u> PMID:<u>23143925</u>
- Alharthy SA, Hardej D (2021). The role of transcription factor Nrf2 in the toxicity of perfluorooctane sulfonate (PFOS) and perfluorooctanoic acid (PFOA) in C57BL/6 mouse astrocytes. *Environ Toxicol Pharmacol.* 86:103652. doi:10.1016/j.etap.2021.103652 PMID:33812015
- Alsarra IA, Brockmann WG, Cunningham ML, Badr MZ (2006). Hepatocellular proliferation in response to agonists of peroxisome proliferator-activated receptor alpha: a role for Kupffer cells? *J Carcinog.* 5(1):26. doi:10.1186/1477-3163-5-26 PMID:17129391
- Ambroz A, Rossner P Jr, Rossnerova A, Honkova K, Milcova A, Pastorkova A, et al. (2022). Oxidative stress and antioxidant response in populations of the Czech Republic exposed to various levels of environmental pollutants. *Int J Environ Res Public Health*. 19(6):3609. doi:10.3390/ijerph19063609 PMID:35329296
- Amstutz VH, Cengo A, Gehres F, Sijm DTHM, Vrolijk MF (2022). Investigating the cytotoxicity of per- and polyfluoroalkyl substances in HepG2 cells: a structure– activity relationship approach. *Toxicology*. 480:153312. doi:<u>10.1016/j.tox.2022.153312</u> PMID:<u>36075290</u>
- Andersen ME, Mallick P, Clewell HJ 3rd, Yoon M, Olsen GW, Longnecker MP (2021). Using quantitative modeling tools to assess pharmacokinetic bias in epidemiological studies showing associations between biomarkers and health outcomes at low exposures. *Environ Res.* 197:111183. doi:10.1016/j. envres.2021.111183 PMID:33887277
- Andersson AG, Lundgren A, Xu Y, Nielsen C, Lindh CH, Pineda D, et al. (2023). High exposure to perfluoroalkyl substances and antibody responses to SARS-CoV-2 mRNA vaccine – an observational study in adults from Ronneby, Sweden. *Environ Health Perspect*. 131(8):87007. doi:10.1289/EHP11847 PMID:37578904
- Arbuckle TE, MacPherson S, Foster WG, Sathyanarayana S, Fisher M, Monnier P, et al. (2020). Prenatal perfluoroalkyl substances and newborn anogenital distance in a Canadian cohort. *Reprod Toxicol*. 94:31–9. doi:<u>10.1016/j.</u> <u>reprotox.2020.03.011</u> PMID:<u>32283250</u>
- Assali NS, Brinkman CR 3rd (1972). Pathophysiology of gestation. Volume I: maternal disorders. Cambridge (MA), USA: Academic Press; p.278.

- Attema B, Janssen AWF, Rijkers D, van Schothorst EM, Hooiveld GJEJ, Kersten S (2022). Exposure to low-dose perfluorooctanoic acid promotes hepatic steatosis and disrupts the hepatic transcriptome in mice. *Mol Metab.* 66:101602. doi:10.1016/j.molmet.2022.101602 PMID:36115532
- AustinME,KasturiBS,BarberM,KannanK,MohanKumar PS, MohanKumar SM (2003). Neuroendocrine effects of perfluorooctane sulfonate in rats. *Environ Health Perspect*. 111(12):1485–9. doi:<u>10.1289/ehp.6128</u> PMID:<u>12948888</u>
- Bach CC, Vested A, Jørgensen KT, Bonde JP, Henriksen TB, Toft G (2016). Perfluoroalkyl and polyfluoroalkyl substances and measures of human fertility: a systematic review. *Crit Rev Toxicol*. 46(9):735–55. doi:<u>10.1080/10408444.2016.1182117</u> PMID:<u>27268162</u>
- Badr MZ, Birnbaum LS (2004). Enhanced potential for oxidative stress in livers of senescent rats by the peroxisome proliferator-activated receptor alpha agonist perfluorooctanoic acid. *Mech Ageing Dev.* 125(1):69–75. doi:10.1016/j.mad.2003.10.006 PMID:14706239
- Barrett ES, Chen C, Thurston SW, Haug LS, Sabaredzovic A, Fjeldheim FN, et al. (2015). Perfluoroalkyl substances and ovarian hormone concentrations in naturally cycling women. *Fertil Steril*. 103(5):1261–70. e3. doi:10.1016/j.fertnstert.2015.02.001 PMID:25747128
- Bartell SM, Calafat AM, Lyu C, Kato K, Ryan PB, Steenland K (2010). Rate of decline in serum PFOA concentrations after granular activated carbon filtration at two public water systems in Ohio and West Virginia. *Environ Health Perspect*. 118(2):222–8. doi:<u>10.1289/ehp.0901252</u> PMID:<u>20123620</u>
- Barton KE, Zell-Baran LM, DeWitt JC, Brindley S, McDonough CA, Higgins CP, et al. (2022). Crosssectional associations between serum PFASs and inflammatory biomarkers in a population exposed to AFFF-contaminated drinking water. *Int J Hyg Environ Health*. 240:113905. doi:10.1016/j.ijheh.2021.113905 PMID:35065522
- Basini G, Bussolati S, Torcianti V, Grasselli F (2022). Perfluorooctanoic acid (PFOA) induces redox status disruption in swine granulosa cells. *Vet Sci.* 9(6):254. doi:<u>10.3390/vetsci9060254</u> PMID:<u>35737306</u>
- Bassler J, Ducatman A, Elliott M, Wen S, Wahlang B, Barnett J, et al. (2019). Environmental perfluoroalkyl acid exposures are associated with liver disease characterized by apoptosis and altered serum adipocytokines. *Environ Pollut*. 247(April):1055–63. doi:<u>10.1016/j.</u> <u>envpol.2019.01.064</u> PMID:<u>30823334</u>
- Bastos Sales L, Kamstra JH, Cenijn PH, van Rijt LS, Hamers T, Legler J (2013). Effects of endocrine disrupting chemicals on in vitro global DNA methylation and adipocyte differentiation. *Toxicol In Vitro*. 27(6):1634–43. doi:10.1016/j.tiv.2013.04.005 PMID:23603478

- Beesoon S, Martin JW (2015). Isomer-specific binding affinity of perfluorooctanesulfonate (PFOS) and perfluorooctanoate (PFOA) to serum proteins. *Environ Sci Technol.* 49(9):5722–31. doi:<u>10.1021/es505399w</u> PMID:<u>25826685</u>
- Beesoon S, Webster GM, Shoeib M, Harner T, Benskin JP, Martin JW (2011). Isomer profiles of perfluorochemicals in matched maternal, cord, and house dust samples: manufacturing sources and transplacental transfer. *Environ Health Perspect*. 119(11):1659–64. doi:10.1289/ehp.1003265 PMID:21757419
- Behnisch PA, Besselink H, Weber R, Willand W, Huang J, Brouwer A (2021). Developing potency factors for thyroid hormone disruption by PFASs using TTR-TRβ CALUX* bioassay and assessment of PFASs mixtures in technical products. *Environ Int*. 157:106791. doi:10.1016/j.envint.2021.106791 PMID:34364217
- Behr A-C, Kwiatkowski A, Ståhlman M, Schmidt FF, Luckert C, Braeuning A, et al. (2020a). Impairment of bile acid metabolism by perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS) in human HepaRG hepatoma cells. *Arch Toxicol.* 94(5):1673–86. doi:10.1007/s00204-020-02732-3 PMID:32253466
- Behr AC, Lichtenstein D, Braeuning A, Lampen A, Buhrke T (2018). Perfluoroalkylated substances (PFAS) affect neither estrogen and androgen receptor activity nor steroidogenesis in human cells in vitro. *Toxicol Lett.* 291:51–60. doi:10.1016/j.toxlet.2018.03.029 PMID:29601859
- Behr AC, Plinsch C, Braeuning A, Buhrke T (2020b). Activation of human nuclear receptors by perfluoroalkylated substances (PFAS). *Toxicol In Vitro*. 62:104700. doi:10.1016/j.tiv.2019.104700 PMID:31676336
- Benninghoff AD, Bisson WH, Koch DC, Ehresman DJ, Kolluri SK, Williams DE (2011). Estrogen-like activity of perfluoroalkyl acids in vivo and interaction with human and rainbow trout estrogen receptors in vitro. *Toxicol Sci.* 120(1):42–58. doi:<u>10.1093/toxsci/kfq379</u> PMID:<u>21163906</u>
- Benskin JP, De Silva AO, Martin LJ, Arsenault G, McCrindle R, Riddell N, et al. (2009). Disposition of perfluorinated acid isomers in Sprague-Dawley rats; part 1: single dose. *Environ Toxicol Chem.* 28(3):542– 54. doi:10.1897/08-239.1 PMID:18939892
- Berg V, Nøst TH, Hansen S, Elverland A, Veyhe A-S, Jorde R, et al. (2015). Assessing the relationship between perfluoroalkyl substances, thyroid hormones and binding proteins in pregnant women; a longitudinal mixed effects approach. *Environ Int.* 77:63–9. doi:10.1016/j.envint.2015.01.007 PMID:25647630
- Bernstein AS, Kapraun DF, Schlosser PM (2021). A model template approach for rapid evaluation and application of physiologically based pharmacokinetic models for use in human health risk assessments: a case study on per- and polyfluoroalkyl substances.

Toxicol Sci. 182(2):215–28. doi:<u>10.1093/toxsci/kfab063</u> PMID:<u>34077538</u>

- Berntsen HF, Bjørklund CG, Audinot JN, Hofer T, Verhaegen S, Lentzen E, et al. (2017). Time-dependent effects of perfluorinated compounds on viability in cerebellar granule neurons: dependence on carbon chain length and functional group attached. *Neurotoxicology*. 63:70–83. doi:10.1016/j. neuro.2017.09.005 PMID:28919516
- Bérubé R, LeFauve MK, Heldman S, Chiang Y-TT, Birbeck J, Westrick J, et al. (2023). Adipogenic and endocrine disrupting mixture effects of organic and inorganic pollutant mixtures. *Sci Total Environ*. 876:162587. doi:<u>10.1016/j.scitotenv.2023.162587</u> PMID:<u>36871739</u>
- Biegel LB, Hurtt ME, Frame SR, O'Connor JC, Cook JC (2001). Mechanisms of extrahepatic tumor induction by peroxisome proliferators in male CD rats. *Toxicol Sci.* 60(1):44–55. doi:<u>10.1093/toxsci/60.1.44</u> PMID:<u>11222872</u>
- Bijland S, Rensen PC, Pieterman EJ, Maas AC, van der Hoorn JW, van Erk MJ, et al. (2011). Perfluoroalkyl sulfonates cause alkyl chain length-dependent hepatic steatosis and hypolipidemia mainly by impairing lipoprotein production in APOE*3-Leiden CETP mice. *Toxicol Sci.* 123(1):290–303. doi:<u>10.1093/toxsci/kfr142</u> PMID:<u>21705711</u>
- Bjork JA, Butenhoff JL, Wallace KB (2011). Multiplicity of nuclear receptor activation by PFOA and PFOS in primary human and rodent hepatocytes. *Toxicology*. 288(1–3):8–17. doi:<u>10.1016/j.tox.2011.06.012</u> PMID: 21723365
- Bjork JA, Lau C, Chang SC, Butenhoff JL, Wallace KB (2008). Perfluorooctane sulfonate-induced changes in fetal rat liver gene expression. *Toxicology*. 251(1–3):8–20. doi:10.1016/j.tox.2008.06.007 PMID:18692542
- Bjork JA, Wallace KB (2009). Structure–activity relationships and human relevance for perfluoroalkyl acid-induced transcriptional activation of peroxisome proliferation in liver cell cultures. *Toxicol Sci.* 111(1):89– 99. doi:10.1093/toxsci/kfp093 PMID:19407336
- Blake BE, Cope HA, Hall SM, Keys RD, Mahler BW, McCord J, et al. (2020). Evaluation of maternal, embryo, and placental effects in CD-1 mice following gestational exposure to perfluorooctanoic acid (PFOA) or hexafluoropropylene oxide dimer acid (HFPO-DA or GenX). *Environ Health Perspect*. 128(2):27006. doi:10.1289/EHP6233 PMID:32074459
- Blake BE, Pinney SM, Hines EP, Fenton SE, Ferguson KK (2018). Associations between longitudinal serum perfluoroalkyl substance (PFAS) levels and measures of thyroid hormone, kidney function, and body mass index in the Fernald Community Cohort. *Environ Pollut*. 242(Pt A):894–904. doi:10.1016/j.envpol.2018.07.042 PMID:30373035

- Blanc M, Rüegg J, Scherbak N, Keiter SH (2019). Environmental chemicals differentially affect epigenetic-related mechanisms in the zebrafish liver (ZF-L) cell line and in zebrafish embryos. *Aquat Toxicol.* 215:105272. doi:10.1016/j.aquatox.2019.105272 PMID:31442592
- Bloom MS, Kannan K, Spliethoff HM, Tao L, Aldous KM, Vena JE (2010). Exploratory assessment of perfluorinated compounds and human thyroid function. *Physiol Behav.* 99(2):240–5. doi:10.1016/j.physbeh.2009.02.005 PMID:19419669
- Bogan AA, Dallas-Yang Q, Ruse MDJ Jr, Maeda Y, Jiang G, Nepomuceno L, et al. (2000). Analysis of protein dimerization and ligand binding of orphan receptor HNF4alpha. J Mol Biol. 302(4):831–51. doi:10.1006/ jmbi.2000.4099 PMID:10993727
- Bogdanska J, Borg D, Sundström M, Bergström U, Halldin K, Abedi-Valugerdi M, et al. (2011). Tissue distribution of ³⁵S-labelled perfluorooctane sulfonate in adult mice after oral exposure to a low environmentally relevant dose or a high experimental dose. *Toxicology.* 284(1–3):54–62. doi:10.1016/j.tox.2011.03.014 PMID:21459123
- Bonzo JA, Ferry CH, Matsubara T, Kim JH, Gonzalez FJ (2012). Suppression of hepatocyte proliferation by hepatocyte nuclear factor 4α in adult mice. *J Biol Chem.* 287(10):7345–56. doi:<u>10.1074/jbc.M111.334599</u> PMID:<u>22241473</u>
- Brede E, Wilhelm M, Göen T, Müller J, Rauchfuss K, Kraft M, et al. (2010). Two-year follow-up biomonitoring pilot study of residents' and controls' PFC plasma levels after PFOA reduction in public water system in Arnsberg, Germany. *Int J Hyg Environ Health.* 213(3):217–23. doi:10.1016/j.ijheh.2010.03.007 PMID:20488749
- Brieger A, Bienefeld N, Hasan R, Goerlich R, Haase H (2011). Impact of perfluorooctanesulfonate and perfluorooctanoic acid on human peripheral leukocytes. *Toxicol In Vitro*. 25(4):960–8. doi:10.1016/j. tiv.2011.03.005 PMID:21397682
- Broad Institute (2023). MSigDB. Molecular Signatures Database. USA: Broad Institute, Massachusetts Institute of Technology, and Regents of the University of California. Available from: <u>https://www.gseamsigdb.org/gsea/msigdb/index.jsp</u>.
- Broedbaek K, Siersma V, Henriksen T, Weimann A, Petersen M, Andersen JT, et al. (2015). Urinary markers of nucleic acid oxidation and cancer in type 2 diabetes. *Redox Biol.* 4:34–9. doi:<u>10.1016/j.redox.2014.11.010</u> PMID:<u>25498965</u>
- Buhrke T, Kibellus A, Lampen A (2013). In vitro toxicological characterization of perfluorinated carboxylic acids with different carbon chain lengths. *Toxicol Lett.* 218(2):97–104. doi:10.1016/j.toxlet.2013.01.025 PMID:23391484
- Buhrke T, Krüger E, Pevny S, Rößler M, Bitter K, Lampen A (2015). Perfluorooctanoic acid (PFOA) affects distinct molecular signalling pathways in human primary

hepatocytes. *Toxicology*. 333:53–62. doi:<u>10.1016/j</u>. tox.2015.04.004 PMID:<u>25868421</u>

- Butenhoff J, Costa G, Elcombe C, Farrar D, Hansen K, Iwai H, et al. (2002). Toxicity of ammonium perfluorooctanoate in male cynomolgus monkeys after oral dosing for 6 months. *Toxicol Sci.* 69(1):244–57. doi:<u>10.1093/</u> <u>toxsci/69.1.244</u> PMID:<u>12215680</u>
- Butenhoff JL, Chang SC, Olsen GW, Thomford PJ (2012b). Chronic dietary toxicity and carcinogenicity study with potassium perfluorooctanesulfonate in Sprague-Dawley rats. *Toxicology*. 293(1–3):1–15. doi:10.1016/j. tox.2012.01.003 PMID:22266392
- Butenhoff JL, Kennedy GL Jr, Chang SC, Olsen GW (2012a). Chronic dietary toxicity and carcinogenicity study with ammonium perfluorooctanoate in Sprague-Dawley rats. *Toxicology*. 298(1–3):1–13. doi:10.1016/j. tox.2012.04.001 PMID:22531602
- Butenhoff JL, Kennedy GL Jr, Hinderliter PM, Lieder PH, Jung R, Hansen KJ, et al. (2004). Pharmacokinetics of perfluorooctanoate in cynomolgus monkeys. *Toxicol Sci.* 82(2):394–406. doi:<u>10.1093/toxsci/kfh302</u> PMID:<u>15470233</u>
- Butenhoff JL, Kennedy GL, Jung R, Chang SC (2014). Evaluation of perfluorooctanoate for potential genotoxicity. *Toxicol Rep.* 1:252–70. doi:10.1016/j. toxrep.2014.05.012 PMID:28962243
- Byrne SC, Miller P, Seguinot-Medina S, Waghiyi V, Buck CL, von Hippel FA, et al. (2018). Exposure to perfluoroalkyl substances and associations with serum thyroid hormones in a remote population of Alaska natives. *Environ Res.* 166:537–43. doi:10.1016/j. envres.2018.06.014 PMID:29958161
- Cadiou S, Bustamante M, Agier L, Andrusaityte S, Basagaña X, Carracedo A, et al. (2020). Using methylome data to inform exposome-health association studies: an application to the identification of environmental drivers of child body mass index. *Environ Int.* 138:105622. doi:<u>10.1016/j.envint.2020.105622</u> PMID:<u>32179316</u>
- Caron-Beaudoin É, Ayotte P, Laouan Sidi EA, Gros-Louis McHugh N, Lemire M; Community of Lac Simon; Community of Winneway–Long Point First Nation; CSSS Tshukuminu Kanani of Nutashkuan; Community of Unamen Shipu (2019). Exposure to perfluoroalkyl substances (PFAS) and associations with thyroid parameters in First Nation children and youth from Quebec. *Environ Int.* 128:13–23. doi:10.1016/j. envint.2019.04.029 PMID:31029975
- Casals-Casas C, Desvergne B (2011). Endocrine disruptors: from endocrine to metabolic disruption. *Annu Rev Physiol.* 73(1):135–62. doi:10.1146/annurevphysiol-012110-142200 PMID:21054169
- Caserta D, Ciardo F, Bordi G, Guerranti C, Fanello E, Perra G, et al. (2013). Correlation of endocrine disrupting chemicals serum levels and white blood cells gene expression of nuclear receptors in a population

of infertile women. *Int J Endocrinol*. 2013:510703. doi:<u>10.1155/2013/510703</u> PMID:<u>23710174</u>

- Caverly Rae JM, Frame SR, Kennedy GL, Butenhoff JL, Chang SC (2014). Pathology review of proliferative lesions of the exocrine pancreas in two chronic feeding studies in rats with ammonium perfluorooctanoate. *Toxicol Rep.* 1:85–91. doi:<u>10.1016/j.toxrep.2014.04.005</u> PMID:<u>28962229</u>
- Çelik A, Eke D, Ekinci SY, Yıldırım S (2013). The protective role of curcumin on perfluorooctane sulfonate-induced genotoxicity: single cell gel electrophoresis and micronucleus test. *Food Chem Toxicol.* 53:249–55. doi:10.1016/j.fct.2012.11.054 PMID:23246701
- Chang CJ, Barr DB, Ryan PB, Panuwet P, Smarr MM, Liu K, et al. (2022). Per- and polyfluoroalkyl substance (PFAS) exposure, maternal metabolomic perturbation, and fetal growth in African American women: a meet-in-the-middle approach. *Environ Int*. 158:106964. doi:10.1016/j.envint.2021.106964 PMID:34735953
- Chang CJ, Barr DB, Zhang Q, Dunlop AL, Smarr MM, Kannan K, et al. (2021). Associations of single and multiple per- and polyfluoroalkyl substance (PFAS) exposure with vitamin D biomarkers in African American women during pregnancy. *Environ Res.* 202:111713. doi:10.1016/j.envres.2021.111713 PMID:34284018
- Chang S, Allen BC, Andres KL, Ehresman DJ, Falvo R, Provencher A, et al. (2017). Evaluation of serum lipid, thyroid, and hepatic clinical chemistries in association with serum perfluorooctanesulfonate (PFOS) in cynomolgus monkeys after oral dosing with potassium PFOS. *Toxicol Sci.* 156(2):387–401. doi:10.1093/toxsci/ kfw267 PMID:28115654
- Chang S-C, Ehresman DJ, Bjork JA, Wallace KB, Parker GA, Stump DG, et al. (2009). Gestational and lactational exposure to potassium perfluorooctanesulfonate (K+PFOS) in rats: toxicokinetics, thyroid hormone status, and related gene expression. *Reprod Toxicol.* 27(3–4):387–99. doi:10.1016/j.reprotox.2009.01.005 PMID:19429409
- Chang S-C, Noker PE, Gorman GS, Gibson SJ, Hart JA, Ehresman DJ, et al. (2012). Comparative pharmacokinetics of perfluorooctanesulfonate (PFOS) in rats, mice, and monkeys. *Reprod Toxicol.* 33(4):428–40. doi:10.1016/j.reprotox.2011.07.002 PMID:21889587
- Chang SC, Thibodeaux JR, Eastvold ML, Ehresman DJ, Bjork JA, Froehlich JW, et al. (2008). Thyroid hormone status and pituitary function in adult rats given oral doses of perfluorooctanesulfonate (PFOS). *Toxicology*. 243(3):330–9. doi:<u>10.1016/j.tox.2007.10.014</u> PMID:<u>18063289</u>
- Chao MR, Evans MD, Hu CW, Ji Y, Møller P, Rossner P, et al. (2021). Biomarkers of nucleic acid oxidation a summary state-of-the-art. *Redox Biol.* 42:101872. doi:10.1016/j.redox.2021.101872 PMID:33579665

- Charazac A, Hinault C, Dolfi B, Hautier S, Decondé Le Butor C, Bost F, et al. (2022). Low doses of PFOA promote prostate and breast cancer cells growth through different pathways. *Int J Mol Sci.* 23(14):7900. doi:10.3390/ijms23147900 PMID:35887249
- Chen G, Xu LL, Huang YF, Wang Q, Wang BH, Yu ZH, et al. (2018). Prenatal exposure to perfluorooctane sulfonate impairs placental angiogenesis and induces aberrant expression of LncRNA xist. *Biomed Environ Sci.* 31(11):843–7. PMID:<u>30558704</u>
- Chen J, Miao Y, Gao Q, Cui Z, Xiong B (2021). Exposure to perfluorooctane sulfonate in vitro perturbs the quality of porcine oocytes via induction of apoptosis. *Environ Pollut*. 284:117508. doi:<u>10.1016/j.envpol.2021.117508</u> PMID:<u>34261219</u>
- Chen JC, Baumert BO, Li Y, Li Y, Pan S, Robinson S, et al. (2023a). Associations of per- and polyfluoroalkyl substances, polychlorinated biphenyls, organochlorine pesticides, and polybrominated diphenyl ethers with oxidative stress markers: a systematic review and meta-analysis. *Environ Res.* 239(Pt 1):117308. doi:10.1016/j. envres.2023.117308 PMID:37813138
- Chen L, Zhu M, Liu Y, Yang Z, Li H, Mu H, et al. (2023b). Perfluorobutanesulfonate exposure induces metabolic disturbances in different regions of mouse gut. *Sci Total Environ*. 866:161261. doi:<u>10.1016/j.</u> <u>scitotenv.2022.161261</u> PMID:<u>36587682</u>
- Chen Q, Huang R, Hua L, Guo Y, Huang L, Zhao Y, et al. (2018a). Prenatal exposure to perfluoroalkyl and poly-fluoroalkyl substances and childhood atopic dermatitis: a prospective birth cohort study. *Environ Health*. 17(1):8. doi:10.1186/s12940-018-0352-7 PMID:29343261
- Chen T, Zhang L, Yue JQ, Lv ZQ, Xia W, Wan YJ, et al. (2012). Prenatal PFOS exposure induces oxidative stress and apoptosis in the lung of rat off-spring. *Reprod Toxicol*. 33(4):538–45. doi:<u>10.1016/j.reprotox.2011.03.003</u> PMID: <u>21440054</u>
- Chen X, Nie X, Mao J, Zhang Y, Yin K, Jiang S (2018b). Perfluorooctanesulfonate induces neuroinflammation through the secretion of TNF-α mediated by the JAK2/STAT3 pathway. *Neurotoxicology*. 66:32–42. doi:<u>10.1016/j.neuro.2018.03.003</u> PMID:<u>29526747</u>
- Chen Y, Hu W, Huang C, Hua S, Wei Q, Bai C, et al. (2016). Subchronic perfluorooctanesulfonate (PFOS) exposure induces elevated mutant frequency in an in vivo λ transgenic medaka mutation assay. *Sci Rep.* 6(1):38466. doi:<u>10.1038/srep38466</u> PMID:<u>27929129</u>
- Chen Y, Zhou L, Xu J, Zhang L, Li M, Xie X, et al. (2017). Maternal exposure to perfluorooctanoic acid inhibits luteal function via oxidative stress and apoptosis in pregnant mice. *Reprod Toxicol*. 69:159–66. doi:<u>10.1016/j.</u> <u>reprotox.2017.02.010</u> PMID:<u>28219760</u>
- Chen YM, Guo LH (2009). Fluorescence study on site-specific binding of perfluoroalkyl acids to human serum albumin. *Arch Toxicol.* 83(3):255–61. doi:<u>10.1007/</u> <u>s00204-008-0359-x</u> PMID:<u>18854981</u>

- Cheng W, Yu Z, Feng L, Wang Y (2013). Perfluorooctane sulfonate (PFOS) induced embryotoxicity and disruption of cardiogenesis. *Toxicol In Vitro*. 27(5):1503–12. doi:10.1016/j.tiv.2013.03.014 PMID:23562911
- Cheng X, Klaassen CD (2008). Perfluorocarboxylic acids induce cytochrome P450 enzymes in mouse liver through activation of PPAR-alpha and CAR transcription factors. *Toxicol Sci.* 106(1):29–36. doi:10.1093/ toxsci/kfn147 PMID:18648086
- Cheng X, Maher J, Lu H, Klaassen CD (2006). Endocrine regulation of gender-divergent mouse organic anion-transporting polypeptide (Oatp) expression. *Mol Pharmacol.* 70(4):1291–7. doi:<u>10.1124/mol.106.025122</u> PMID:<u>16807376</u>
- Cheng X, Wei Y, Zhang Z, Wang F, He J, Wang R, et al. (2022). Plasma PFOA and PFOS levels, DNA methylation, and blood lipid levels: a pilot study. *Environ Sci Technol.* 56(23):17039–51. doi:<u>10.1021/acs.est.2c04107</u> PMID:<u>36374530</u>
- Chiu WA, Lynch MT, Lay CR, Antezana A, Malek P, Sokolinski S, et al. (2022). Bayesian estimation of human population toxicokinetics of PFOA, PFOS, PFHxS, and PFNA from studies of contaminated drinking water. *Environ Health Perspect*. 130(12):127001. doi:10.1289/ <u>EHP10103</u> PMID:36454223
- Choi EM, Suh KS, Rhee SY, Oh S, Woo JT, Kim SW, et al. (2016). Perfluorooctanoic acid induces mitochondrial dysfunction in MC3T3–E1 osteoblast cells. J Environ Sci Health Part A Tox Hazard Subst Environ Eng. 52(3):281–9. doi:10.1080/10934529.2016.1253402 PMID:27901621
- Choi S-K, Kim J-H, Park J-K, Lee K-M, Kim E, Jeon WB (2013). Cytotoxicity and inhibition of intercellular interaction in N2a neurospheroids by perfluoro-octanoic acid and perfluorooctanesulfonic acid. *Food Chem Toxicol.* 60(October):520–9. doi:10.1016/j. fct.2013.07.070 PMID:23948355
- Chou HC, Wen LL, Chang CC, Lin CY, Jin L, Juan SH (2017). From the cover: l-carnitine via PPARγ- and sirt1-dependent mechanisms attenuates epithelialmesenchymal transition and renal fibrosis caused by perfluorooctanesulfonate. *Toxicol Sci.* 160(2):217–29. doi:10.1093/toxsci/kfx183 PMID:28973641
- Chou W-C, Lin Z (2019). Bayesian evaluation of a physiologically based pharmacokinetic (PBPK) model for perfluorooctane sulfonate (PFOS) to characterize the interspecies uncertainty between mice, rats, monkeys, and humans: development and performance verification. *Environ Int.* 129:408–22. doi:<u>10.1016/j. envint.2019.03.058</u> PMID:<u>31152982</u>
- Christensen BT, Calkins MM (2023). Occupational exposure to per- and polyfluoroalkyl substances: a scope review of the literature from 1980–2021. *J Expo Sci Environ Epidemiol.* 33(5):673–86. doi:<u>10.1038/s41370-023-00536-y</u> PMID:<u>36977833</u>

- Clarity C, Trowbridge J, Gerona R, Ona K, McMaster M, Bessonneau V, et al. (2021). Associations between polyfluoroalkyl substance and organophosphate flame retardant exposures and telomere length in a cohort of women firefighters and office workers in San Francisco. *Environ Health.* 20(1):97. doi:10.1186/s12940-021-00778-z PMID:34454526
- Clark KL, George JW, Hua G, Davis JS (2022). Perfluorooctanoic acid promotes proliferation of the human granulosa cell line HGrC1 and alters expression of cell cycle genes and Hippo pathway effector YAP1. *Reprod Toxicol*. 110(June):49–59. doi:10.1016/j. reprotox.2022.03.011 PMID:35346789
- Conley JM, Lambright CS, Evans N, Medlock-Kakaley E, Hill D, McCord J, et al. (2022). Developmental toxicity of Nafion byproduct 2 (NBP2) in the Sprague-Dawley rat with comparisons to hexafluoropropylene oxidedimer acid (HFPO-DA or GenX) and perfluorooctane sulfonate (PFOS). *Environ Int.* 160:107056. doi:10.1016/ j.envint.2021.107056 PMID:34952357
- Convertino M, Church TR, Olsen GW, Liu Y, Doyle E, Elcombe CR, et al. (2018). Stochastic pharmacokinetic-pharmacodynamic modeling for assessing the systemic health risk of perfluorooctanoate (PFOA). *Toxicol Sci.* 163(1):293–306. doi:<u>10.1093/toxsci/kfy035</u> PMID:<u>29462473</u>
- Coperchini F, Croce L, Pignatti P, Ricci G, Gangemi D, Magri F, et al. (2021b). The new generation PFAS C6O4 does not produce adverse effects on thyroid cells in vitro. *J Endocrinol Invest.* 44(8):1625–35. doi:10.1007/ <u>\$40618-020-01466-4</u> PMID:<u>33315184</u>
- Coperchini F, Croce L, Ricci G, Magri F, Rotondi M, Imbriani M, et al. (2021a). Thyroid disrupting effects of old and new generation PFAS. *Front Endocrinol* (*Lausanne*). 11:612320. doi:<u>10.3389/fendo.2020.612320</u> PMID:<u>33542707</u>
- Coperchini F, Pignatti P, Lacerenza S, Negri S, Sideri R, Testoni C, et al. (2015). Exposure to perfluorinated compounds: in vitro study on thyroid cells. *Environ Sci Pollut Res Int.* 22(3):2287–94. doi:<u>10.1007/s11356-014-3480-9</u> PMID:<u>25182428</u>
- Corsini E, Avogadro A, Galbiati V, dell'Agli M, Marinovich M, Galli CL, et al. (2011). In vitro evaluation of the immunotoxic potential of perfluorinated compounds (PFCs). *Toxicol Appl Pharmacol.* 250(2):108–16. doi:10.1016/j.taap.2010.11.004 PMID:21075133
- Corsini E, Sangiovanni E, Avogadro A, Galbiati V, Viviani B, Marinovich M, et al. (2012). In vitro characterization of the immunotoxic potential of several perfluorinated compounds (PFCs). *Toxicol Appl Pharmacol.* 258(2):248–55. doi:<u>10.1016/j.taap.2011.11.004</u> PMID: <u>22119708</u>
- Corton JC, Peters JM, Klaunig JE (2018). The PPARadependent rodent liver tumor response is not relevant to humans: addressing misconceptions. *Arch*

Toxicol. 92(1):83–119. doi:<u>10.1007/s00204-017-2094-7</u> PMID:<u>29197930</u>

- Costa G, Sartori S, Consonni D (2009). Thirty years of medical surveillance in perfluooctanoic acid production workers. *J Occup Environ Med.* 51(3):364–72. doi:10.1097/JOM.0b013e3181965d80 PMID:19225424
- Costello E, Rock S, Stratakis N, Eckel SP, Walker DI, Valvi D, et al. (2022). Exposure to per- and polyfluoroalkyl substances and markers of liver injury: a systematic review and meta-analysis. *Environ Health Perspect*. 130(4):46001. doi:10.1289/EHP10092 PMID:35475652
- Crawford L, Halperin SA, Dzierlenga MW, Skidmore B, Linakis MW, Nakagawa S, et al. (2023). Systematic review and meta-analysis of epidemiologic data on vaccine response in relation to exposure to five principal perfluoroalkyl substances. *Environ Int.* 172:107734. doi:10.1016/j.envint.2023.107734 PMID:36764183
- Crawford NM, Fenton SE, Strynar M, Hines EP, Pritchard DA, Steiner AZ (2017). Effects of perfluorinated chemicals on thyroid function, markers of ovarian reserve, and natural fertility. *Reprod Toxicol.* 69:53–9. doi:10.1016/j.reprotox.2017.01.006 PMID:28111093
- Crebelli R, Caiola S, Conti L, Cordelli E, De Luca G, Dellatte E, et al. (2019). Can sustained exposure to PFAS trigger a genotoxic response? A comprehensive genotoxicity assessment in mice after subacute oral administration of PFOA and PFBA. *Regul Toxicol Pharmacol.* 106:169–77. doi:10.1016/j.yrtph.2019.05.005 PMID:31071380
- Croce L, Coperchini F, Tonacchera M, Imbriani M, Rotondi M, Chiovato L (2019). Effect of long- and short-chain perfluorinated compounds on cultured thyroid cells viability and response to TSH. *J Endocrinol Invest*. 42(11):1329–35. doi:<u>10.1007/s40618-019-01062-1</u> PMID:31102255
- Cui J, Wang P, Yan S, Liang Y, Liu D, Ren S (2022). Perfluorooctane sulfonate induces dysfunction of human umbilical vein endothelial cells via ferroptosis pathway. *Toxics*. 10(9):503. doi:<u>10.3390/toxics10090503</u> PMID:<u>36136468</u>
- Cui L, Liao CY, Zhou QF, Xia TM, Yun ZJ, Jiang GB (2010). Excretion of PFOA and PFOS in male rats during a subchronic exposure. *Arch Environ Contam Toxicol.* 58(1):205–13. doi:<u>10.1007/s00244-009-9336-5</u> PMID:<u>19468665</u>
- Cui L, Zhou QF, Liao CY, Fu JJ, Jiang GB (2009). Studies on the toxicological effects of PFOA and PFOS on rats using histological observation and chemical analysis. *Arch Environ Contam Toxicol*. 56(2):338–49. doi:10.1007/s00244-008-9194-6 PMID:18661093
- Cui Q, Pan Y, Wang J, Liu H, Yao B, Dai J (2020). Exposure to per- and polyfluoroalkyl substances (PFASs) in serum versus semen and their association with male reproductive hormones. *Environ Pollut*. 266(Pt 2):115330. doi:10.1016/j.envpol.2020.115330 PMID:32781340

- Cui R, Li C, Wang J, Dai J (2019). Induction of hepatic miR-34a by perfluorooctanoic acid regulates metabolism-related genes in mice. *Environ Pollut*. 244:270–8. doi:10.1016/j.envpol.2018.10.061 PMID:30342367
- D'Amico R, Gugliandolo E, Cordaro M, Fusco R, Genovese T, Peritore AF, et al. (2022). Toxic effects of endocrine disruptor exposure on collagen-induced arthritis. *Biomolecules*. 12(4):12. doi:<u>10.3390/biom12040564</u> PMID:<u>35454153</u>
- D'eon JC, Mabury SA (2011). Is indirect exposure a significant contributor to the burden of perfluorinated acids observed in humans? *Environ Sci Technol.* 45(19):7974– 84. doi:<u>10.1021/es200171y</u> PMID:<u>21630688</u>
- D'Eon JC, Simpson AJ, Kumar R, Baer AJ, Mabury SA (2010). Determining the molecular interactions of perfluorinated carboxylic acids with human sera and isolated human serum albumin using nuclear magnetic resonance spectroscopy. *Environ Toxicol Chem.* 29(8):1678–88. doi:10.1002/etc.204 PMID:20821619
- Dallaire R, Dewailly E, Pereg D, Dery S, Ayotte P (2009). Thyroid function and plasma concentrations of polyhalogenated compounds in Inuit adults. *Environ Health Perspect.* 117(9):1380–6. doi:<u>10.1289/ehp.0900633</u> PMID:<u>19750101</u>
- Dalsager L, Christensen N, Halekoh U, Timmermann CAG, Nielsen F, Kyhl HB, et al. (2021). Exposure to perfluoroalkyl substances during fetal life and hospitalization for infectious disease in childhood: a study among 1503 children from the Odense Child Cohort. *Environ Int.* 149:106395. doi:10.1016/j. envint.2021.106395 PMID:33508532
- Davidsen N, Ramhøj L, Lykkebo CA, Kugathas I, Poulsen R, Rosenmai AK, et al. (2022). PFOS-induced thyroid hormone system disrupted rats display organ-specific changes in their transcriptomes. *Environ Pollut.* 305:119340. doi:10.1016/j.envpol.2022.119340 PMID:35460815
- de Cock M, de Boer MR, Lamoree M, Legler J, van de Bor M (2014). Prenatal exposure to endocrine disrupting chemicals in relation to thyroid hormone levels in infants – a Dutch prospective cohort study. *Environ Health*. 13(1):106. doi:<u>10.1186/1476-069X-13-106</u> PMID: 25495114
- De Guise S, Levin M (2021). Suppression of Th2 cytokines as a potential mechanism for reduced antibody response following PFOA exposure in female B6C3F₁ mice. *Toxicol Lett.* 351:155–62. doi:<u>10.1016/j.</u> toxlet.2021.09.002 PMID:<u>34517056</u>
- De Silva AO, Benskin JP, Martin LJ, Arsenault G, McCrindle R, Riddell N, et al. (2009). Disposition of perfluorinated acid isomers in Sprague-Dawley rats; part 2: subchronic dose. *Environ Toxicol Chem*. 28(3):555–67. doi:10.1897/08-254.1 PMID:18939893
- De Toni L, Di Nisio A, Rocca MS, Pedrucci F, Garolla A, Dall'Acqua S, et al. (2022). Comparative evaluation of the effects of legacy and new generation perfluoralkyl

substances (PFAS) on thyroid cells *in vitro*. *Front Endocrinol* (*Lausanne*). 13:915096. doi:10.3389/ fendo.2022.915096 PMID:35813651

- Derakhshan A, Kortenkamp A, Shu H, Broeren MAC, Lindh CH, Peeters RP, et al. (2022). Association of perand polyfluoroalkyl substances with thyroid homeostasis during pregnancy in the SELMA study. *Environ Int.* 167:107420. doi:10.1016/j.envint.2022.107420 PMID:35870378
- Dewitt JC, Copeland CB, Strynar MJ, Luebke RW (2008). Perfluorooctanoic acid-induced immunomodulation in adult C57BL/6J or C57BL/6N female mice. *Environ Health Perspect*. 116(5):644–50. doi:<u>10.1289/ehp.10896</u> PMID:<u>18470313</u>
- DeWitt JC, Williams WC, Creech NJ, Luebke RW (2016). Suppression of antigen-specific antibody responses in mice exposed to perfluorooctanoic acid: role of PPARa and T- and B-cell targeting. *J Immunotoxicol*. 13(1):38– 45. doi:10.3109/1547691X.2014.996682 PMID:25594567
- Dhingra R, Winquist A, Darrow LA, Klein M, Steenland K (2017). A study of reverse causation: examining the associations of perfluorooctanoic acid serum levels with two outcomes. *Environ Health Perspect*. 125(3):416–21. doi:10.1289/EHP273 PMID:27529882
- Di Nisio A, Pannella M, Vogiatzis S, Sut S, Dall'Acqua S, Rocca MS, et al. (2022). Impairment of human dopaminergic neurons at different developmental stages by perfluoro-octanoic acid (PFOA) and differential human brain areas accumulation of perfluoroalkyl chemicals. *Environ Int.* 158:106982. doi:10.1016/j. envint.2021.106982 PMID:34781208
- Di Nisio A, Sabovic I, Valente U, Tescari S, Rocca MS, Guidolin D, et al. (2019). Endocrine disruption of androgenic activity by perfluoroalkyl substances: clinical and experimental evidence. *J Clin Endocrinol Metab.* 104(4):1259–71. doi:10.1210/jc.2018-01855 PMID:30403786
- Ding J, Dai Y, Zhang J, Wang Z, Zhang L, Xu S, et al. (2023). Associations of perfluoroalkyl substances with adipocytokines in umbilical cord serum: a mixtures approach. *Environ Res.* 216(Pt 3):114654. doi:<u>10.1016/j.envres.2022.114654</u> PMID:<u>36309220</u>
- Dong GH, Liu MM, Wang D, Zheng L, Liang ZF, Jin YH (2011). Sub-chronic effect of perfluorooctanesulfonate (PFOS) on the balance of type 1 and type 2 cytokine in adult C57BL6 mice. *Arch Toxicol.* 85(10):1235–44. doi:10.1007/s00204-011-0661-x PMID:21327619
- Dong GH, Wang J, Zhang YH, Liu MM, Wang D, Zheng L, et al. (2012b). Induction of p53-mediated apoptosis in splenocytes and thymocytes of C57BL/6 mice exposed to perfluorooctane sulfonate (PFOS). *Toxicol Appl Pharmacol.* 264(2):292–9. doi:10.1016/j. taap.2012.08.010 PMID:22959464
- Dong G-H, Zhang Y-H, Zheng L, Liang Z-F, Jin Y-H, He Q-C (2012a). Subchronic effects of perfluorooctanesulfonate exposure on inflammation in adult

male C57BL/6 mice. *Environ Toxicol*. 27(5):285–96. doi:<u>10.1002/tox.20642</u> PMID:<u>20737580</u>

- Dong GH, Zhang YH, Zheng L, Liu W, Jin YH, He QC (2009). Chronic effects of perfluorooctanesulfonate exposure on immunotoxicity in adult male C57BL/6 mice. *Arch Toxicol.* 83(9):805–15. doi:<u>10.1007/s00204-009-0424-0</u> PMID:<u>19343326</u>
- Dong H, Curran I, Williams A, Bondy G, Yauk CL, Wade MG (2016). Hepatic miRNA profiles and thyroid hormone homeostasis in rats exposed to dietary potassium perfluorooctanesulfonate (PFOS). *Environ Toxicol Pharmacol.* 41:201–10. doi:10.1016/j.etap.2015.12.009 PMID:26724606
- Dong Z, Wang J, Qiu T, Wu J, An Y, Shi X, et al. (2022). Perfluorooctane sulfonate induces mitochondrial calcium overload and early hepatic insulin resistance via autophagy/detyrosinated alpha-tubulin-regulated IP3R2–VDAC1–MICU1 interaction. *Sci Total Environ*. 825:153933. doi:<u>10.1016/j.scitotenv.2022.153933</u> PMID: 35192817
- Dos Santos RS, Medina-Gali RM, Babiloni-Chust I, Marroqui L, Nadal A (2022). In vitro assays to identify metabolism-disrupting chemicals with diabetogenic activity in a human pancreatic β -cell model. *Int J Mol Sci.* 23(9):5040. doi:<u>10.3390/ijms23095040</u> PMID:<u>35563431</u>
- Dourson ML, Gadagbui B, Onyema C, McGinnis PM, York RG (2019). Data derived extrapolation factors for developmental toxicity: a preliminary research case study with perfluorooctanoate (PFOA). *Regul Toxicol Pharmacol.* 108:104446. doi:10.1016/j. yrtph.2019.104446 PMID:31425727
- Dreyer AF, Jensen RC, Glintborg D, Schmedes AV, Brandslund I, Nielsen F, et al. (2020). Perfluoroalkyl substance exposure early in pregnancy was negatively associated with late pregnancy cortisone levels. *J Clin Endocrinol Metab.* 105(8):dgaa292. doi:<u>10.1210/clinem/ dgaa292</u> PMID:<u>32436946</u>
- Du G, Hu J, Huang H, Qin Y, Han X, Wu D, et al. (2013). Perfluorooctane sulfonate (PFOS) affects hormone receptor activity, steroidogenesis, and expression of endocrine-related genes in vitro and in vivo. *Environ Toxicol Chem.* 32(2):353–60. doi:<u>10.1002/etc.2034</u> PMID:<u>23074026</u>
- Du J, Wang S, You H, Jiang R, Zhuang C, Zhang X (2016). Developmental toxicity and DNA damage to zebrafish induced by perfluorooctane sulfonate in the presence of ZnO nanoparticles. *Environ Toxicol.* 31(3):360–71. doi:10.1002/tox.22050 PMID:25258305
- Du Y, Cai Z, Zhou G, Liang W, Man Q, Wang W (2022). Perfluorooctanoic acid exposure increases both proliferation and apoptosis of human placental trophoblast cells mediated by ER stress-induced ROS or UPR pathways. *Ecotoxicol Environ Saf.* 236(May):113508. doi:10.1016/j.ecoeny.2022.113508 PMID:35427876

- Du Y, Xu T, Luo D, Wang Y, Yin H, Liu C, et al. (2023). Perfluorooctane sulfonate-induced apoptosis in kidney cells by triggering the NOX4/ROS/JNK axis and antagonism of cannabidiol. *Environ Toxicol.* 38(7):1651–64. doi:10.1002/tox.23794 PMID:36988283
- Dusek J, Skoda J, Holas O, Horvatova A, Smutny T, Linhartova L, et al. (2019). Stilbene compound trans-3,4,5,4'-tetramethoxystilbene, a potential anti-cancer drug, regulates constitutive androstane receptor (Car) target genes, but does not possess proliferative activity in mouse liver. *Toxicol Lett.* 313:1–10. doi:10.1016/j. toxlet.2019.05.024 PMID:31170421
- Dzierlenga AL, Robinson VG, Waidyanatha S, DeVito MJ, Eifrid MA, Gibbs ST, et al. (2020). Toxicokinetics of perfluorohexanoic acid (PFHxA), perfluoro-octanoic acid (PFOA) and perfluorodecanoic acid (PFDA) in male and female Hsd:Sprague-Dawley SD rats following intravenous or gavage administration. *Xenobiotica.* 50(6):722–32. doi:10.1080/00498254.2019 .1683776 PMID:31680603
- EFSA (2018). Risk to human health related to the presence of perfluorooctane sulfonic acid and perfluorooctanoic acid in food. EFSA Panel on Contaminants in the Food Chain (CONTAM). *EFSA J*. 16(12): e05194. <u>https://efsa. onlinelibrary.wiley.com/doi/10.2903/j.efsa.2018.5194</u> PMID:<u>32625773</u>
- EFSA (2020). Risk to human health related to the presence of perfluoroalkyl substances in food. EFSA Panel on Contaminants in the Food Chain. *EFSA J.* 18(9):6223. <u>https://efsa.onlinelibrary.wiley.com/doi/</u> <u>epdf/10.2903/j.efsa.2020.6223</u> PMID:<u>32994824</u>
- Eick SM, Goin DE, Cushing L, DeMicco E, Park JS, Wang Y, et al. (2021). Mixture effects of prenatal exposure to per- and polyfluoroalkyl substances and polybrominated diphenyl ethers on maternal and newborn telomere length. *Environ Health*. 20(1):76. doi:10.1186/ s12940-021-00765-4 PMID:34193151
- Eke D, Çelik A (2016). Curcumin prevents perfluorooctane sulfonate-induced genotoxicity and oxidative DNA damage in rat peripheral blood. *Drug Chem Toxicol.* 39(1):97–103. doi:<u>10.3109/01480545.2015.1041</u> <u>601 PMID:25950456</u>
- Eke D, Çelik A, Yilmaz MB, Aras N, Kocatürk Sel S, Alptekin D (2017). Apoptotic gene expression profiles and DNA damage levels in rat liver treated with perfluorooctane sulfonate and protective role of curcumin. *Int J Biol Macromol.* 104(Pt A):515–20. doi:10.1016/j. ijbiomac.2017.06.075 PMID:28634058
- Elcombe CR, Elcombe BM, Foster JR, Chang SC, Ehresman DJ, Butenhoff JL (2012b). Hepatocellular hypertrophy and cell proliferation in Sprague-Dawley rats from dietary exposure to potassium perfluorooctanesulfonate results from increased expression of xenosensor nuclear receptors PPARα and CAR/PXR. *Toxicology*. 293(1–3):16–29. doi:10.1016/j. tox.2011.12.014 PMID:22245121

- Elcombe CR, Elcombe BM, Foster JR, Chang SC, Ehresman DJ, Butenhoff JL (2012c). Hepatocellular hypertrophy and cell proliferation in Sprague-Dawley rats from dietary exposure to potassium perfluorooctanesulfonate results from increased expression of xenosensor nuclear receptors PPARγ and CAR/PXR. *Reprod Toxicol.* 33(4):626. doi:10.1016/j.reprotox.2011.11.108 PMID:22245121
- Elcombe CR, Elcombe BM, Foster JR, Chang SC, Ehresman DJ, Noker PE, et al. (2012a). Evaluation of hepatic and thyroid responses in male Sprague-Dawley rats for up to eighty-four days following seven days of dietary exposure to potassium perfluorooctane-sulfonate. *Toxicology*. 293(1–3):30–40. doi:10.1016/j. tox.2011.12.015 PMID:22239858
- Elcombe CR, Elcombe BM, Foster JR, Farrar DG, Jung R, Chang SC, et al. (2010). Hepatocellular hypertrophy and cell proliferation in Sprague-Dawley rats following dietary exposure to ammonium perfluorooctanoate occurs through increased activation of the xenosensor nuclear receptors PPARa and CAR/PXR. *Arch Toxicol.* 84(10):787–98. doi:<u>10.1007/s00204-010-0572-2</u> PMID:<u>20614104</u>
- Elcombe CR, Wolf CR, Westwood AL (2013). Compositions comprising perfluorooctanoic acid. Pub. No. US 2013/0029928. US Patent Application Publication. Available from: <u>https://tera.org/Alliance%20for%20</u> <u>Risk/Workshop%20Xll/Prelim%20Case%20Refs/</u> <u>Elcombe%20et%20al.,%202013+%20copy.pdf</u>, accessed January 2025.
- Elumalai S, Karunakaran U, Won KC, Chung SM, Moon JS (2023). Perfluorooctane sulfonate-induced oxidative stress contributes to pancreatic β-cell apoptosis by inhibiting cyclic adenosine monophosphate pathway: prevention by pentoxifylline. *Environ Pollut*. 320:120959. doi:<u>10.1016/j.envpol.2022.120959</u> PMID:<u>36621715</u>
- Emerce E, Çetin Ö (2018). Genotoxicity assessment of perfluoroalkyl substances on human sperm. *Toxicol Ind Health.* 34(12):884–90. doi:<u>10.1177/0748233718799191</u> PMID:<u>30326792</u>
- Endirlik BU, Eken A, Canpınar H, Öztürk F, Gürbay A (2022). Perfluorooctanoic acid affects mouse brain and liver tissue through oxidative stress. *Arh Hig Rada Toksikol*. 73(2):148–57. doi:<u>10.2478/aiht-2022-73-3629</u> PMID:<u>35792765</u>
- Eriksen KT, Raaschou-Nielsen O, Sørensen M, Roursgaard M, Loft S, Møller P (2010). Genotoxic potential of the perfluorinated chemicals PFOA, PFOS, PFBS, PFNA and PFHxA in human HepG2 cells. *Mutat Res.* 700(1–2):39–43. doi:10.1016/j.mrgentox.2010.04.024 PMID:20451658
- Escher P, Wahli W (2000). Peroxisome proliferator-activated receptors: insight into multiple cellular functions. *Mutat Res.* 448(2):121–38. doi:10.1016/ <u>S0027-5107(99)00231-6</u> PMID:10725467

- ESCODD (European Standards Committee on Oxidative DNA Damage) (2002). Comparative analysis of baseline 8-oxo-7,8-dihydroguanine in mammalian cell DNA, by different methods in different laboratories: an approach to consensus. *Carcinogenesis*. 23(12):2129–33. doi:10.1093/carcin/23.12.2129 PMID:12507938
- Estefanía González-Alvarez M, Severin A, Sayadi M, Keating AF (2022). PFOA-induced ovotoxicity differs between lean and obese mice with impacts on ovarian reproductive and DNA damage sensing and repair proteins. *Toxicol Sci*. 190(2):173–88. doi:<u>10.1093/toxsci/ kfac104</u> PMID:<u>36214631</u>
- Etzel TM, Braun JM, Buckley JP (2019). Associations of serum perfluoroalkyl substance and vitamin D biomarker concentrations in NHANES, 2003–2010. *Int J Hyg Environ Health*. 222(2):262–9. doi:<u>10.1016/j.</u> <u>ijheh.2018.11.003</u> PMID:<u>30503928</u>
- Evans N, Conley JM, Cardon M, Hartig P, Medlock-Kakaley E, Gray LEJ Jr (2022). In vitro activity of a panel of per- and polyfluoroalkyl substances (PFAS), fatty acids, and pharmaceuticals in peroxisome proliferator-activated receptor (PPAR) alpha, PPAR gamma, and estrogen receptor assays. *Toxicol Appl Pharmacol.* 449:116136. doi:10.1016/j.taap.2022.116136 PMID:35752307
- Fåbrega F, Kumar V, Schuhmacher M, Domingo JL, Nadal M (2014). PBPK modeling for PFOS and PFOA: validation with human experimental data. *Toxicol Lett.* 230(2):244–51. doi:<u>10.1016/j.toxlet.2014.01.007</u> PMID:<u>24440341</u>
- Fair PA, Driscoll E, Mollenhauer MA, Bradshaw SG, Yun SH, Kannan K, et al. (2011). Effects of environmentally-relevant levels of perfluorooctane sulfonate on clinical parameters and immunological functions in B6C3F1 mice. *J Immunotoxicol.* 8(1):17–29. doi:10.310 9/1547691X.2010.527868 PMID:21261439
- Feldstein AE, Wieckowska A, Lopez AR, Liu YC, Zein NN, McCullough AJ (2009). Cytokeratin-18 fragment levels as noninvasive biomarkers for nonalcoholic steatohepatitis: a multicenter validation study. *Hepatology*. 50(4):1072–8. doi:10.1002/hep.23050 PMID:19585618
- Feng X, Wang X, Cao X, Xia Y, Zhou R, Chen L (2015). Chronic exposure of female mice to an environmental level of perfluorooctane sulfonate suppresses estrogen synthesis through reduced histone H3K14 acetylation of the StAR promoter leading to deficits in follicular development and ovulation. *Toxicol Sci.* 148(2):368–79. doi:10.1093/toxsci/kfv197 PMID:26358002
- Fenton SE, Reiner JL, Nakayama SF, Delinsky AD, Stanko JP, Hines EP, et al. (2009). Analysis of PFOA in dosed CD-1 mice. Part 2. Disposition of PFOA in tissues and fluids from pregnant and lactating mice and their pups. *Reprod Toxicol.* 27(3–4):365–72. doi:<u>10.1016/j.</u>reprotox.2009.02.012 PMID:<u>19429407</u>

- Fernández Freire P, Pérez Martin JM, Herrero O, Peropadre A, de la Peña E, Hazen MJ (2008). In vitro assessment of the cytotoxic and mutagenic potential of perfluorooctanoic acid. *Toxicol In Vitro*. 22(5):1228–33. doi:10.1016/j.tiv.2008.04.004 PMID:18499391
- Filgo AJ, Quist EM, Hoenerhoff MJ, Brix AE, Kissling GE, Fenton SE (2015). Perfluorooctanoic acid (PFOA)induced liver lesions in two strains of mice following developmentalexposures: PPARaisnotrequired. *Toxicol Pathol.* 43(4):558–68. doi:10.1177/0192623314558463 PMID:25398757
- Fletcher T, Galloway TS, Melzer D, Holcroft P, Cipelli R, Pilling LC, et al. (2013). Associations between PFOA, PFOS and changes in the expression of genes involved in cholesterol metabolism in humans. *Environ Int.* 57–58:2–10. doi:<u>10.1016/j.envint.2013.03.008</u> PMID: <u>23624243</u>
- Florentin A, Deblonde T, Diguio N, Hautemaniere A, Hartemann P (2011). Impacts of two perfluorinated compounds (PFOS and PFOA) on human hepatoma cells: cytotoxicity but no genotoxicity? *Int J Hyg Environ Health*. 214(6):493–9. doi:<u>10.1016/j.ijheh.2011.05.010</u> PMID:<u>21676652</u>
- Forsthuber M, Kaiser AM, Granitzer S, Hassl I, Hengstschläger M, Stangl H, et al. (2020). Albumin is the major carrier protein for PFOS, PFOA, PFHxS, PFNA and PFDA in human plasma. *Environ Int.* 137:105324. doi:<u>10.1016/j.envint.2019.105324</u> PMID:<u>32109724</u>
- Franco ME, Sutherland GE, Fernandez-Luna MT, Lavado R (2020). Altered expression and activity of phase I and II biotransformation enzymes in human liver cells by perfluorooctanoate (PFOA) and perfluorooctane sulfonate (PFOS). *Toxicology*. 430:152339. doi:10.1016/j. tox.2019.152339 PMID:31809754
- Franken C, Koppen G, Lambrechts N, Govarts E, Bruckers L, Den Hond E, et al. (2017). Environmental exposure to human carcinogens in teenagers and the association with DNA damage. *Environ Res.* 152:165–74. doi:10.1016/j.envres.2016.10.012 PMID:27771571
- Franko J, Meade BJ, Frasch HF, Barbero AM, Anderson SE (2012). Dermal penetration potential of perfluorooctanoic acid (PFOA) in human and mouse skin. J Toxicol Environ Health A. 75(1):50–62. doi:10.1080/15 287394.2011.615108 PMID:22047163
- Fromme H, Mosch C, Morovitz M, Alba-Alejandre I, Boehmer S, Kiranoglu M, et al. (2010). Pre- and postnatal exposure to perfluorinated compounds (PFCs). *Environ Sci Technol.* 44(18):7123–9. doi:10.1021/ es101184f PMID:20722423
- Fu J, Gao Y, Cui L, Wang T, Liang Y, Qu G, et al. (2016). Occurrence, temporal trends, and half-lives of perfluoroalkyl acids (PFAAs) in occupational workers in China. *Sci Rep.* 6(1):38039. doi:<u>10.1038/srep38039</u> PMID:<u>27905562</u>

- Fuentes S, Colomina MT, Rodriguez J, Vicens P, Domingo JL (2006). Interactions in developmental toxicology: concurrent exposure to perfluorooctane sulfonate (PFOS) and stress in pregnant mice. *Toxicol Lett.* 164(1):81–9. doi:10.1016/j.toxlet.2005.11.013 PMID:16384669
- Fujii Y, Niisoe T, Harada KH, Uemoto S, Ogura Y, Takenaka K, et al. (2015). Toxicokinetics of perfluoroalkyl carboxylic acids with different carbon chain lengths in mice and humans. *J Occup Health*. 57(1):1– 12. doi:10.1539/joh.14-0136-OA PMID:25422127
- Gallo E, Barbiellini Amidei C, Barbieri G, Fabricio ASC, Gion M, Pitter G, et al. (2022). Perfluoroalkyl substances and thyroid stimulating hormone levels in a highly exposed population in the Veneto Region. *Environ Res.* 203:111794. doi:10.1016/j.envres.2021.111794 PMID:34358507
- Gao B, He X, Liu W, Zhang H, Saito N, Tsuda S (2015b). Distribution of perfluoroalkyl compounds in rats: indication for using hair as bioindicator of exposure. *J Expo Sci Environ Epidemiol*. 25(6):632–8. doi:<u>10.1038/jes.2014.54</u> PMID:<u>25118134</u>
- Gao Y, Fu J, Cao H, Wang Y, Zhang A, Liang Y, et al. (2015a). Differential accumulation and elimination behavior of perfluoroalkyl acid isomers in occupational workers in a manufactory in China. *Environ Sci Technol.* 49(11):6953–62. doi:<u>10.1021/acs.est.5b00778</u> PMID:<u>25927957</u>
- Gao Y, Guo X, Wang S, Chen F, Ren X, Xiao H, et al. (2020). Perfluorooctane sulfonate enhances mRNA expression of PPARγ and ap2 in human mesenchymal stem cells monitored by long-retained intracellular nanosensor. *Environ Pollut*. 263(Pt B):114571. doi:10.1016/j. envpol.2020.114571 PMID:32320904
- Garoche C, Boulahtouf A, Grimaldi M, Chiavarina B, Toporova L, den Broeder MJ, et al. (2021). Interspecies differences in activation of peroxisome proliferator-activated receptor γ by pharmaceutical and environmental chemicals. *Environ Sci Technol.* 55(24):16489–501. doi:10.1021/acs.est.1c04318 PMID:34843233
- Gebbink WA, Berger U, Cousins IT (2015). Estimating human exposure to PFOS isomers and PFCA homologues: the relative importance of direct and indirect (precursor) exposure. *Environ Int.* 74:160–9. doi:10.1016/j.envint.2014.10.013 PMID:25454233
- Genuis SJ, Birkholz D, Ralitsch M, Thibault N (2010). Human detoxification of perfluorinated compounds. *Public Health*. 124(7):367–75. doi:<u>10.1016/j.puhe.2010</u>. <u>03.002</u> PMID:<u>20621793</u>
- Giménez-Bastida JA, Surma M, Zieliński H (2015). In vitro evaluation of the cytotoxicity and modulation of mechanisms associated with inflammation induced by perfluorooctanesulfonate and perfluorooctanoic acid in human colon myofibroblasts CCD-18Co. *Toxicol In Vitro*. 29(7):1683–91. doi:10.1016/j.tiv.2015.07.001 PMID:26142696

- Gingrich J, Pu Y, Upham BL, Hulse M, Pearl S, Martin D, et al. (2021). Bisphenol S enhances gap junction intercellular communication in ovarian theca cells. *Chemosphere*. 263(January):128304. doi:10.1016/j. chemosphere.2020.128304 PMID:33155548
- Glynn A, Berger U, Bignert A, Ullah S, Aune M, Lignell S, et al. (2012). Perfluorinated alkyl acids in blood serum from primiparous women in Sweden: serial sampling during pregnancy and nursing, and temporal trends 1996–2010. *Environ Sci Technol.* 46(16):9071–9. doi:10.1021/es301168c PMID:22770559
- Goecke CM, Jarnot BM, Reo NV (1992). A comparative toxicological investigation of perfluorocarboxylic acids in rats by fluorine-19 NMR spectroscopy. *Chem Res Toxicol.* 5(4):512–9. doi:<u>10.1021/tx00028a009</u> PMID:<u>1391617</u>
- Goeden HM, Greene CW, Jacobus JA (2019). A transgenerational toxicokinetic model and its use in derivation of Minnesota PFOA water guidance. *J Expo Sci Environ Epidemiol.* 29(2):183–95. doi:<u>10.1038/s41370-018-</u> <u>0110-5</u> PMID:<u>30631142</u>
- Gogola J, Hoffmann M, Ptak A (2019). Persistent endocrine-disrupting chemicals found in human follicular fluid stimulate the proliferation of granulosa tumor spheroids via GPR30 and IGF1R but not via the classic estrogen receptors. *Chemosphere*. 217(February):100–10. doi:10.1016/j. chemosphere.2018.11.018 PMID:30414542
- Goines S, Dick JE (2022). Investigating the cytotoxic redox mechanism of PFOS within Hep G2 by hyper-spectral-assisted scanning electrochemical micros-copy. *Analyst (Lond)*. 147(19):4356–64. doi:10.1039/ D2AN00904H PMID:36043461
- Gomis MI, Vestergren R, MacLeod M, Mueller JF, Cousins IT (2017). Historical human exposure to perfluoroalkyl acids in the United States and Australia reconstructed from biomonitoring data using population-based pharmacokinetic modelling. *Environ Int.* 108:92–102. doi:10.1016/j.envint.2017.08.002 PMID:28818713
- Gomis MI, Vestergren R, Nilsson H, Cousins IT (2016). Contribution of direct and indirect exposure to human serum concentrations of perfluorooctanoic acid in an occupationally exposed group of ski waxers. *Environ Sci Technol.* 50(13):7037–46. doi:<u>10.1021/acs.est.6b01477</u> PMID:<u>27304840</u>
- Gong X, Yang C, Hong Y, Chung ACK, Cai Z (2019). PFOA and PFOS promote diabetic renal injury in vitro by impairing the metabolisms of amino acids and purines. *Sci Total Environ*. 676:72–86. doi:<u>10.1016/j.</u> <u>scitotenv.2019.04.208</u> PMID:<u>31029902</u>
- Gonzalez FJ, Shah YM (2008). PPARalpha: mechanism of species differences and hepatocarcinogenesis of peroxisome proliferators. *Toxicology*. 246(1 August):2–8. doi:10.1016/j.tox.2007.09.030 PMID:18006136

- Goodrich JM, Calkins MM, Caban-Martinez AJ, Stueckle T, Grant C, Calafat AM, et al. (2021). Per- and poly-fluoroalkyl substances, epigenetic age and DNA methylation: a cross-sectional study of firefighters. *Epigenomics*. 13(20):1619–36. doi:<u>10.2217/epi-2021-0225</u> PMID:<u>34670402</u>
- Gorrochategui E, Pérez-Albaladejo E, Casas J, Lacorte S, Porte C (2014). Perfluorinated chemicals: differential toxicity, inhibition of aromatase activity and alteration of cellular lipids in human placental cells. *Toxicol Appl Pharmacol.* 277(2):124–30. doi:10.1016/j. taap.2014.03.012 PMID:24680846
- Goudarzi H, Araki A, Itoh S, Sasaki S, Miyashita C, Mitsui T, et al. (2017b). The association of prenatal exposure to perfluorinated chemicals with glucocorticoid and androgenic hormones in cord blood samples: the Hokkaido Study. *Environ Health Perspect*. 125(1):111–8. doi:10.1289/EHP142 PMID:27219028
- Goudarzi H, Miyashita C, Okada E, Kashino I, Chen CJ, Ito S, et al. (2017a). Prenatal exposure to perfluoroalkyl acids and prevalence of infectious diseases up to 4 years of age. *Environ Int.* 104:132–8. doi:10.1016/j. envint.2017.01.024 PMID:28392064
- Grandjean P, Andersen EW, Budtz-Jørgensen E, Nielsen F, Mølbak K, Weihe P, et al. (2012). Serum vaccine antibody concentrations in children exposed to perfluorinated compounds. *JAMA*. 307(4):391–7. doi:<u>10.1001/</u> jama.2011.2034 PMID:22274686
- Grandjean P, Heilmann C, Weihe P, Nielsen F, Mogensen UB, Timmermann A, et al. (2017). Estimated exposures to perfluorinated compounds in infancy predict attenuated vaccine antibody concentrations at age 5 years. *J Immunotoxicol*. 14(1):188–95. doi:10.1080/1547 691X.2017.1360968 PMID:28805477
- Granum B, Haug LS, Namork E, Stølevik SB, Thomsen C, Aaberge IS, et al. (2013). Pre-natal exposure to perfluoroalkyl substances may be associated with altered vaccine antibody levels and immune-related health outcomes in early childhood. *J Immunotoxicol*. 10(4):373–9. doi:1 0.3109/1547691X.2012.755580 PMID:23350954
- Griesbacher T, Pommer V, Schuligoi R, Tiran B, Peskar BA (2008). Anti-inflammatory actions of perfluorooctanoic acid and peroxisome proliferator-activated receptors (PPAR) alpha and gamma in experimental acute pancreatitis. *Int Immunopharmacol.* 8(2):325–9. doi:10.1016/j.intimp.2007.08.005 PMID:18182248
- Griffith FD, Long JE (1980). Animal toxicity studies with ammonium perfluorooctanoate. *Am Ind Hyg Assoc J.* 41(8):576–83. doi:<u>10.1080/15298668091425301</u> PMID:<u>6773404</u>
- Guerrero-Preston R, Goldman LR, Brebi-Mieville P, Ili-Gangas C, Lebron C, Witter FR, et al. (2010). Global DNA hypomethylation is associated with in utero exposure to cotinine and perfluorinated alkyl compounds. *Epigenetics*. 5(6):539–46. doi:10.4161/epi.5.6.12378 PMID:20523118

- Guo C, Zhao Z, Zhao K, Huang J, Ding L, Huang X, et al. (2021a). Perfluorooctanoic acid inhibits the maturation rate of mouse oocytes cultured in vitro by triggering mitochondrial and DNA damage. *Birth Defects Res.* 113(14):1074–83. doi:10.1002/bdr2.1899 PMID:33871176
- Guo H, Zhang H, Sheng N, Wang J, Chen J, Dai J (2021b). Perfluorooctanoic acid (PFOA) exposure induces splenic atrophy via overactivation of macrophages in male mice. J Hazard Mater. 407:124862. doi:<u>10.1016/j.</u> jhazmat.2020.124862 PMID:<u>33360190</u>
- Guo J, Zhang J, Wang Z, Zhang L, Qi X, Zhang Y, et al. (2021c). Umbilical cord serum perfluoroalkyl substance mixtures in relation to thyroid function of newborns: findings from Sheyang Mini Birth Cohort Study. *Chemosphere*. 273:129664. doi:10.1016/j. chemosphere.2021.129664 PMID:33493812
- Guo P, Furnary T, Vasiliou V, Yan Q, Nyhan K, Jones DP, et al. (2022b). Non-targeted metabolomics and associations with per- and polyfluoroalkyl substances (PFAS) exposure in humans: a scoping review. *Environ Int.* 162:107159. doi:10.1016/j.envint.2022.107159 PMID:35231839
- Guo XX, He QZ, Li W, Long DX, Pan XY, Chen C, et al. (2017). Brain-derived neurotrophic factor mediated perfluorooctane sulfonate induced-neurotoxicity via epigenetics regulation in SK-N-SH cells. *Int J Mol Sci.* 18(4):893. doi:10.3390/ijms18040893 PMID:28441774
- Guo Y, Yuan J, Ni H, Ji J, Zhong S, Zheng Y, et al. (2022a). Perfluorooctanoic acid-induced developmental cardiotoxicity in chicken embryo: roles of miR-490-5p. *Environ Pollut.* 312:120022. doi:<u>10.1016/j.envpol.2022.120022</u> PMID:<u>36028080</u>
- Guruge KS, Hikono H, Shimada N, Murakami K, Hasegawa J, Yeung LW, et al. (2009). Effect of perfluorooctane sulfonate (PFOS) on influenza A virus-induced mortality in female B6C3F₁ mice. *J Toxicol Sci.* 34(6):687–91. doi:10.2131/jts.34.687 PMID:19952504
- Guruge KS, Yeung LW, Yamanaka N, Miyazaki S, Lam PK, Giesy JP, et al. (2006). Gene expression profiles in rat liver treated with perfluorooctanoic acid (PFOA). *Toxicol Sci.* 89(1):93–107. doi:<u>10.1093/toxsci/kfj011</u> PMID:<u>16221955</u>
- Gyllenhammar I, Benskin JP, Sandblom O, Berger U, Ahrens L, Lignell S, et al. (2018). Perfluoroalkyl acids (PFAAs) in serum from 2–4-month-old infants: influence of maternal serum concentration, gestational age, breast-feeding, and contaminated drinking water. *Environ Sci Technol.* 52(12):7101–10. doi:10.1021/acs. est.8b00770 PMID:29758986
- Hallberg I, Persson S, Olovsson M, Sirard MA, Damdimopoulou P, Rüegg J, et al. (2021). Perfluorooctane sulfonate (PFOS) exposure of bovine oocytes affects early embryonic development at human-relevant levels in an in vitro model. *Toxicology*. 464:153028. doi:<u>10.1016/j.tox.2021.153028</u> PMID:<u>34762985</u>

- Hamm MP, Cherry NM, Chan E, Martin JW, Burstyn I (2010). Maternal exposure to perfluorinated acids and fetal growth. *J Expo Sci Environ Epidemiol*. 20(7):589–97. doi:10.1038/jes.2009.57 PMID:19865074
- Han JS, Jang S, Son HY, Kim YB, Kim Y, Noh JH, et al. (2020). Subacute dermal toxicity of perfluoroalkyl carboxylic acids: comparison with different carbonchain lengths in human skin equivalents and systemic effects of perfluoroheptanoic acid in Sprague-Dawley rats. *Arch Toxicol.* 94(2):523–39. doi:10.1007/s00204-019-02634-z PMID:31797001
- Han R, Hu M, Zhong Q, Wan C, Liu L, Li F, et al. (2018a). Perfluorooctane sulphonate induces oxidative hepatic damage via mitochondria-dependent and NF-κB/TNF-α-mediated pathway. *Chemosphere*. 191:1056–64. doi:10.1016/j.chemosphere.2017.08.070 PMID:28939271
- Han R, Zhang F, Wan C, Liu L, Zhong Q, Ding W (2018b). Effect of perfluorooctane sulphonate-induced Kupffer cell activation on hepatocyte proliferation through the NF- κ B/TNF- α /IL-6-dependentpathway.*Chemosphere*. 200:283–94. doi:10.1016/j.chemosphere.2018.02.137 PMID:29494909
- Han X, Alam MN, Cao M, Wang X, Cen M, Tian M, et al. (2022). Low levels of perfluorooctanoic acid exposure activates steroid hormone biosynthesis through repressing histone methylation in rats. *Environ Sci Technol.* 56(9):5664–72. doi:<u>10.1021/acs.est.1c08885</u> PMID:<u>35438966</u>
- Han X, Hinderliter PM, Snow TA, Jepson GW (2004). Binding of perfluorooctanoic acid to rat liver-form and kidney-form alpha2u-globulins. *Drug Chem Toxicol.* 27(4):341–60. doi:<u>10.1081/DCT-200039725</u> PMID:<u>15573471</u>
- Han X, Kemper RA, Jepson GW (2005). Subcellular distribution and protein binding of perfluorooctanoic acid in rat liver and kidney. *Drug Chem Toxicol*. 28(2):197–209. doi:10.1081/DCT-52547 PMID:15865261
- Han X, Snow TA, Kemper RA, Jepson GW (2003). Binding of perfluorooctanoic acid to rat and human plasma proteins. *Chem Res Toxicol*. 16(6):775–81. doi:10.1021/ <u>tx034005w</u> PMID:12807361
- Harada K, Inoue K, Morikawa A, Yoshinaga T, Saito N, Koizumi A (2005). Renal clearance of perfluorooctane sulfonate and perfluorooctanoate in humans and their species-specific excretion. *Environ Res.* 99(2):253–61. doi:10.1016/j.envres.2004.12.003 PMID:16194675
- Harada KH, Hashida S, Kaneko T, Takenaka K, Minata M, Inoue K, et al. (2007). Biliary excretion and cerebrospinal fluid partition of perfluorooctanoate and perfluorooctane sulfonate in humans. *Environ Toxicol Pharmacol.* 24(2):134–9. doi:10.1016/j.etap.2007.04.003 PMID:21783801
- Harlow SD, Hood MM, Ding N, Mukherjee B, Calafat AM, Randolph JF Jr, et al. (2021). Per- and polyfluoroalkyl substances and hormone levels during the menopausal

transition. *J Clin Endocrinol Metab.* 106(11):e4427–37. doi:<u>10.1210/clinem/dgab476</u> PMID:<u>34181018</u>

- Hayhurst GP, Lee YH, Lambert G, Ward JM, Gonzalez FJ (2001). Hepatocyte nuclear factor 4alpha (nuclear receptor 2A1) is essential for maintenance of hepatic gene expression and lipid homeostasis. *Mol Cell Biol.* 21(4):1393–403. doi:10.1128/MCB.21.4.1393-1403.2001 PMID:11158324
- Heffernan AL, Cunningham TK, Drage DS, Aylward LL, Thompson K, Vijayasarathy S, et al. (2018). Perfluorinated alkyl acids in the serum and follicular fluid of UK women with and without polycystic ovarian syndrome undergoing fertility treatment and associations with hormonal and metabolic parameters. *Int J Hyg Environ Health*. 221(7):1068–75. doi:10.1016/j. ijheh.2018.07.009 PMID:30037723
- Hinderliter PM, DeLorme MP, Kennedy GL (2006). Perfluorooctanoic acid: relationship between repeated inhalation exposures and plasma PFOA concentration in the rat. *Toxicology*. 222(1–2):80–5. doi:10.1016/j. tox.2006.01.029 PMID:16513238
- Hines EP, White SS, Stanko JP, Gibbs-Flournoy EA, Lau C, Fenton SE (2009). Phenotypic dichotomy following developmental exposure to perfluorooctanoic acid (PFOA) in female CD-1 mice: low doses induce elevated serum leptin and insulin, and overweight in mid-life. *Mol Cell Endocrinol.* 304(1–2):97–105. doi:10.1016/j. mce.2009.02.021 PMID:19433254
- Højsager FD, Andersen M, Juul A, Nielsen F, Möller S, Christensen HT, et al. (2022). Prenatal and early postnatal exposure to perfluoroalkyl substances and bone mineral content and density in the Odense child cohort. *Environ Int*. 167:107417. doi:10.1016/j. envint.2022.107417 PMID:35914335
- Houck KA, Friedman KP, Feshuk M, Patlewicz G, Smeltz M, Clifton MS, et al. (2023). Evaluation of 147 perfluoroalkyl substances for immunotoxic and other (patho) physiological activities through phenotypic screening of human primary cells. *ALTEX*. 40(2):248–70. doi:10.14573/altex.2203041 PMID:36129398
- Houck KA, Patlewicz G, Richard AM, Williams AJ, Shobair MA, Smeltz M, et al. (2021). Bioactivity profiling of per- and polyfluoroalkyl substances (PFAS) identifies potential toxicity pathways related to molecular structure. *Toxicology*. 457:152789. doi:10.1016/j. tox.2021.152789 PMID:33887376
- Hu Q, Franklin JN, Bryan I, Morris E, Wood A, DeWitt JC (2012). Does developmental exposure to perflurooctanoic acid (PFOA) induce immunopathologies commonly observed in neurodevelopmental disorders? *Neurotoxicology.* 33(6):1491–8. doi:10.1016/j. neuro.2012.10.016 PMID:23137609
- Hu Q, Strynar MJ, DeWitt JC (2010). Are developmentally exposed C57BL/6 mice insensitive to suppression of TDAR by PFOA? *J Immunotoxicol*. 7(4):344–9. doi:1 0.3109/1547691X.2010.520045 PMID:20954796

- Hu W, Jones PD, Upham BL, Trosko JE, Lau C, Giesy JP (2002). Inhibition of gap junctional intercellular communication by perfluorinated compounds in rat liver and dolphin kidney epithelial celllines in vitro and Sprague-Dawley rats in vivo. *Toxicol Sci.* 68(2):429–36. doi:10.1093/toxsci/68.2.429 PMID:12151638
- Hu WY, Lu R, Hu DP, Imir OB, Zuo Q, Moline D, et al. (2022). Per- and polyfluoroalkyl substances target and alter human prostate stem-progenitor cells. *Biochem Pharmacol*. 197:114902. doi:<u>10.1016/j.bcp.2021.114902</u> PMID:<u>34968493</u>
- Hu X, Li S, Cirillo PM, Krigbaum NY, Tran V, Jones DP, et al. (2019). Metabolome wide association study of serum poly and perfluoroalkyl substances (PFASs) in pregnancy and early postpartum. *Reprod Toxicol.* 87:70–8. doi:10.1016/j.reprotox.2019.05.058 PMID:31121237
- Hu XZ, Hu DC (2009). Effects of perfluorooctanoate and perfluorooctane sulfonate exposure on hepatoma Hep G2 cells. *Arch Toxicol.* 83(9):851–61. doi:<u>10.1007/</u> <u>s00204-009-0441-z</u> PMID:<u>19468714</u>
- Huang H, Wang Q, He X, Wu Y, Xu C (2019b). Association between polyfluoroalkyl chemical concentrations and leucocyte telomere length in US adults. *Sci Total Environ*.653:547–53.doi:<u>10.1016/j.scitotenv.2018.10.400</u> PMID:<u>30414584</u>
- Huang J, Ren H, Chen A, Li T, Wang H, Jiang L, et al. (2022a). Perfluorooctane sulfonate induces suppression of testosterone biosynthesis via Sertoli cell-derived exosomal/miR-9–3p downregulating StAR expression in Leydig cells. *Environ Pollut*. 301:118960. doi:10.1016/j. envpol.2022.118960 PMID:35150797
- Huang MC, Dzierlenga AL, Robinson VG, Waidyanatha S, DeVito MJ, Eifrid MA, et al. (2019a). Toxicokinetics of perfluorobutane sulfonate (PFBS), perfluorohexane-1-sulphonic acid (PFHxS), and perfluorooctane sulfonic acid (PFOS) in male and female Hsd:Sprague-Dawley SD rats after intravenous and gavage administration. *Toxicol Rep.* 6:645–55. doi:10.1016/j. toxrep.2019.06.016 PMID:31334035
- Huang MC, Dzierlenga AL, Robinson VG, Waidyanatha S, DeVito MJ, Eifrid MA, et al. (2021). Corrigendum to "Toxicokinetics of perfluorobutane sulfonate (PFBS), perfluorohexane-1-sulphonic acid (PFHxS), and perfluorooctane sulfonic acid (PFOS) in male and female Hsd:Sprague-Dawley SD rats after intravenous and gavage administration" [Toxicol. Rep. (2019) 6:645–655]. *Toxicol Rep.* 8:365. doi:10.1016/j. toxrep.2021.02.001 PMID:33665134
- Huang Q, Luo L, Han X, Li F, Zhang X, Tian M (2022b). Low-dose perfluorooctanoic acid stimulates steroid hormone synthesis in Leydig cells: integrated proteomics and metabolomics evidence. *J Hazard Mater*. 424(Pt D):127656. doi:10.1016/j.jhazmat.2021.127656 PMID:<u>34774353</u>
- Huang Q, Zhang J, Martin FL, Peng S, Tian M, Mu X, et al. (2013). Perfluorooctanoic acid induces apoptosis through the p53-dependent mitochondrial pathway in human hepatic cells: a proteomic study. *Toxicol Lett.* 223(2):211–20. doi:10.1016/j.toxlet.2013.09.002 PMID:24035753
- Huang Q, Zhang J, Peng S, Du M, Ow S, Pu H, et al. (2014). Proteomic analysis of perfluorooctane sulfonate-induced apoptosis in human hepatic cells using the iTRAQ technique. *J Appl Toxicol.* 34(12):1342–51. doi:10.1002/jat.2963 PMID:24301089
- Huang T, Zhang Y, Zhang W, Lin T, Chen L, Yang B, et al. (2020). Attenuation of perfluorooctane sulfonate-in-duced steatohepatitis by grape seed proanthocyanidin extract in mice. *BioMed Res Int.* 2020:8818160. doi:10.1155/2020/8818160 PMID:33457418
- Hundley SG, Sarrif AM, Kennedy GL Jr (2006). Absorption, distribution, and excretion of ammonium perfluorooctanoate (APFO) after oral administration to various species. *Drug Chem Toxicol*. 29(2):137–45. doi:10.1080/01480540600561361 PMID:16707323
- Husøy T, Caspersen IH, Thépaut E, Knutsen H, Haug LS, Andreassen M, et al. (2023). Comparison of aggregated exposure to perfluorooctanoic acid (PFOA) from diet and personal care products with concentrations in blood using a PBPK model – results from the Norwegian biomonitoring study in EuroMix. *Environ Res.* 239(Pt 2):117341. doi:10.1016/j.envres.2023.117341 PMID:<u>37839534</u>
- Imir OB, Kaminsky AZ, Zuo QY, Liu YJ, Singh R, Spinella MJ, et al. (2021). Per- and polyfluoroalkyl substance exposure combined with high-fat diet supports prostate cancer progression. *Nutrients*. 13(11):3902. doi:10.3390/nu13113902 PMID:34836157
- Impinen A, Longnecker MP, Nygaard UC, London SJ, Ferguson KK, Haug LS, et al. (2019). Maternal levels of perfluoroalkyl substances (PFASs) during pregnancy and childhood allergy and asthma related outcomes and infections in the Norwegian Mother and Child (MoBa) cohort. *Environ Int*. 124:462–72. doi:10.1016/j. envint.2018.12.041 PMID:30684804
- Impinen A, Nygaard UC, Lødrup Carlsen KC, Mowinckel P, Carlsen KH, Haug LS, et al. (2018). Prenatal exposure to perfluoralkyl substances (PFASs) associated with respiratory tract infections but not allergy- and asthma-related health outcomes in childhood. *Environ Res.* 160:518–23. doi:10.1016/j.envres.2017.10.012 PMID:29106950
- Inoue K, Ritz B, Andersen SL, Ramlau-Hansen CH, Høyer BB, Bech BH, et al. (2019). Perfluoroalkyl substances and maternal thyroid hormones in early pregnancy; findings in the Danish National Birth Cohort. *Environ Health Perspect*. 127(11):117002. doi:<u>10.1289/EHP5482</u> PMID:<u>31714155</u>

- Ishibashi H, Hirano M, Kim EY, Iwata H (2019). In vitro and in silico evaluations of binding affinities of perfluoroalkyl substances to Baikal seal and human peroxisome proliferator-activated receptor α. *Environ Sci Technol.* 53(4):2181–8. doi:<u>10.1021/acs.est.8b07273</u> PMID:<u>30649875</u>
- Ishida K, Tsuyama Y, Sanoh S, Ohta S, Kotake Y (2017). Perfluorooctane sulfonate induces neuronal vulnerability by decreasing GluR2 expression. *Arch Toxicol.* 91(2):885–95. doi:<u>10.1007/s00204-016-1731-x</u> PMID:<u>27155986</u>
- Itoh S, Araki A, Mitsui T, Miyashita C, Goudarzi H, Sasaki S, et al. (2016). Association of perfluoroalkyl substances exposure in utero with reproductive hormone levels in cord blood in the Hokkaido Study on Environment and Children's Health. *Environ Int.* 94:51–9. doi:10.1016/j. envint.2016.05.011 PMID:27209000
- Itoh S, Araki A, Miyashita C, Yamazaki K, Goudarzi H, Minatoya M, et al. (2019). Association between perfluoroalkyl substance exposure and thyroid hormone/ thyroid antibody levels in maternal and cord blood: the Hokkaido Study. *Environ Int.* 133(Pt A):105139. doi:10.1016/j.envint.2019.105139 PMID:31518930
- Iwabuchi K, Senzaki N, Mazawa D, Sato I, Hara M, Ueda F, et al. (2017). Tissue toxicokinetics of perfluoro compounds with single and chronic low doses in male rats. *J Toxicol Sci.* 42(3):301–17. doi:<u>10.2131/jts.42.301</u> PMID:<u>28496036</u>
- Iwai H, Yamashita K (2006). A fourteen-day repeated dose oral toxicity study of APFO in rats. *Drug Chem Toxicol*. 29(3):323–32. doi:<u>10.1080/01480540600653143</u> PMID:<u>16777710</u>
- Jabeen M, Fayyaz M, Irudayaraj J (2020). Epigenetic modifications, and alterations in cell cycle and apoptosis pathway in A549 lung carcinoma cell line upon exposure to perfluoroalkyl substances. *Toxics*. 8(4):112. doi:10.3390/toxics8040112 PMID:33238432
- Jacquet N, Maire MA, Landkocz Y, Vasseur P (2012b). Carcinogenic potency of perfluorooctane sulfonate (PFOS) on Syrian hamster embryo (SHE) cells. *Arch Toxicol.* 86(2):305–14. doi:<u>10.1007/s00204-011-0752-8</u> PMID:<u>22057587</u>
- Jacquet N, Maire MA, Rast C, Bonnard M, Vasseur P (2012a). Perfluorooctanoic acid (PFOA) acts as a tumor promoter on Syrian hamster embryo (SHE) cells. *Environ Sci Pollut Res Int*. 19(7):2537–49. doi:<u>10.1007/</u> <u>s11356-012-0968-z</u> PMID:<u>22828883</u>
- Jain RB (2013). Association between thyroid profile and perfluoroalkyl acids: data from NHANES 2007–2008. *Environ Res.* 126:51–9. doi:<u>10.1016/j.envres.2013.08.006</u> PMID:<u>24053974</u>
- Jain RB, Ducatman A (2022). Associations between the concentrations of α -klotho and selected perfluoroalkyl substances in the presence of eGFR based kidney function and albuminuria: data for US adults aged 40–79

years. *Sci Total Environ*. 838(Pt 1):155994. doi:<u>10.1016/j.</u> scitotenv.2022.155994 PMID:<u>35595139</u>

- Jandacek RJ, Rider T, Keller ER, Tso P (2010). The effect of olestra on the absorption, excretion and storage of 2,2',5,5' tetrachlorobiphenyl; 3,3',4,4' tetrachlorobiphenyl; and perfluorooctanoic acid. *Environ Int.* 36(8):880–3. doi:<u>10.1016/j.envint.2009.06.010</u> PMID:<u>19616301</u>
- Janssen AWF, Louisse J, Rijkers D, Pinckaers NET, Hoekstra SA, Hoogenboom RLAP, et al. (2022). Perfluoroalkyl substances (PFASs) decrease the expression of recombination-activating genes (RAG1 and RAG2) in human B lymphoma Namalwa cells. *Arch Toxicol*. 97(2):457–68. doi:<u>10.1007/s00204-022-03405-z</u> PMID:<u>36326898</u>
- Jensen RC, Glintborg D, Gade Timmermann CA, Nielsen F, Kyhl HB, Frederiksen H, et al.; The Odense Child Cohort (2020). Prenatal exposure to perfluorodecanoic acid is associated with lower circulating concentration of adrenal steroid metabolites during mini puberty in human female infants. *Environ Res.* 182:109101. doi:10.1016/j.envres.2019.109101 PMID:32069767
- Jensen RC, Glintborg D, Timmermann CAG, Nielsen F, Boye H, Madsen JB, et al.; The Odense Child Cohort (2022). Higher free thyroxine associated with PFAS exposure in first trimester. *Environ Res.* 212(Pt D):113492. doi:10.1016/j.envres.2022.113492 PMID:35597289
- Jernbro S, Rocha PS, Keiter S, Skutlarek D, Färber H, Jones PD, et al. (2007). Perfluorooctane sulfonate increases the genotoxicity of cyclophosphamide in the micronucleus assay with V79 cells. Further proof of alterations in cell membrane properties caused by PFOS. *Environ Sci Pollut Res Int.* 14(2):85–7. doi:10.1065/ espr2007.01.384 PMID:17455816
- Ji K, Kim S, Kho Y, Paek D, Sakong J, Ha J, et al. (2012). Serum concentrations of major perfluorinated compounds among the general population in Korea: dietary sources and potential impact on thyroid hormones. *Environ Int.* 45:78–85. doi:<u>10.1016/j.</u> envint.2012.03.007 PMID:<u>22580293</u>
- Jin H, Zhang Y, Jiang W, Zhu L, Martin JW (2016). Isomerspecific distribution of perfluoroalkyl substances in blood. *Environ Sci Technol*. 50(14):7808–15. doi:<u>10.1021/</u> acs.est.6b01698 PMID:27295125
- Joensen UN, Bossi R, Leffers H, Jensen AA, Skakkebaek NE, Jørgensen N (2009). Do perfluoroalkyl compounds impair human semen quality? *Environ Health Perspect*. 117(6):923–7. doi:10.1289/ehp.0800517 PMID:19590684
- Joensen UN, Veyrand B, Antignac J-P, Blomberg Jensen M, Petersen JH, Marchand P, et al. (2013). PFOS (perfluorooctanesulfonate) in serum is negatively associated with testosterone levels, but not with semen quality, in healthy men. *Hum Reprod.* 28(3):599–608. doi:<u>10.1093/</u> <u>humrep/des425</u> PMID:<u>23250927</u>

- Johnson JD, Gibson SJ, Ober RE (1984). Cholestyramineenhanced fecal elimination of carbon-14 in rats after administration of ammonium [¹⁴C]perfluorooctanoate or potassium [¹⁴C]perfluorooctanesulfonate. *Fundam Appl Toxicol.* 4(6):972–6. doi:<u>10.1016/0272-0590(84)90235-5</u> PMID:<u>6519377</u>
- Jones LE, Ghassabian A, Lawrence DA, Sundaram R, Yeung E, Kannan K, et al. (2022). Exposure to perfluoroalkyl substances and neonatal immunoglobulin profiles in the upstate KIDS study (2008–2010). *Environ Pollut*. 308:119656. doi:<u>10.1016/j.envpol.2022.119656</u> PMID:<u>35787426</u>
- Kamei Y, Suganami T, Kohda T, Ishino F, Yasuda K, Miura S, et al. (2007). Peg1/Mest in obese adipose tissue is expressed from the paternal allele in an isoform-specific manner. *FEBS Lett.* 581(1):91–6. doi:<u>10.1016/j.febslet.2006.12.002</u> PMID:<u>17182038</u>
- Kamendulis LM, Hocevar JM, Stephens M, Sandusky GE, Hocevar BA (2022). Exposure to perfluorooctanoic acid leads to promotion of pancreatic cancer. *Carcinogenesis*. 43(5):469–78. doi:<u>10.1093/carcin/ bgac005</u> PMID:<u>35022659</u>
- Kamendulis LM, Wu Q, Sandusky GE, Hocevar BA (2014). Perfluorooctanoic acid exposure triggers oxidative stress in the mouse pancreas. *Toxicol Rep.* 1:513–21. doi:<u>10.1016/j.toxrep.2014.07.015</u> PMID:<u>28962265</u>
- Kang H, Lee HK, Moon HB, Kim S, Lee J, Ha M, et al. (2018). Perfluoroalkyl acids in serum of Korean children: occurrences, related sources, and associated health outcomes. *Sci Total Environ*. 645:958–65. doi:10.1016/j.scitotenv.2018.07.177 PMID:30032090
- Kang JS, Choi JS, Park JW (2016). Transcriptional changes in steroidogenesis by perfluoroalkyl acids (PFOA and PFOS) regulate the synthesis of sex hormones in H295R cells. *Chemosphere*. 155:436–43. doi:<u>10.1016/j.</u> <u>chemosphere.2016.04.070</u> PMID:<u>27139122</u>
- Kasten-Jolly J, Lawrence DA (2022). Perfluorooctanesulfonate (PFOS) and perfluorooctanoic acid (PFOA) modify *in vitro* mitogen- and antigen-induced human peripheral blood mononuclear cell (PBMC) responses. *J Toxicol Environ Health A*. 85(17):715–37. doi:10.1080/ 15287394.2022.2075816 PMID:35611390
- Katakura M, Kudo N, Tsuda T, Hibino Y, Mitsumoto A, Kawashima Y (2007). Rat organic anion transporter 3 and organic anion transporting polypeptide 1 mediate perfluorooctanoic acid transport. *J Health Sci*. 53(1):77– 83. doi:10.1248/jhs.53.77
- Kato K, Wong LY, Chen A, Dunbar C, Webster GM, Lanphear BP, et al. (2014). Changes in serum concentrations of maternal poly- and perfluoroalkyl substances over the course of pregnancy and predictors of exposure in a multiethnic cohort of Cincinnati, Ohio pregnant women during 2003–2006. *Environ Sci Technol.* 48(16):9600–8. doi:10.1021/es501811k PMID:25026485

- Kato S, Itoh S, Yuasa M, Baba T, Miyashita C, Sasaki S, et al. (2016). Association of perfluorinated chemical exposure in utero with maternal and infant thyroid hormone levels in the Sapporo cohort of Hokkaido Study on the Environment and Children's Health. *Environ Health Prev Med.* 21(5):334–44. doi:10.1007/s12199-016-0534-2 PMID:27137816
- Kaur K, Lesseur C, Chen L, Andra SS, Narasimhan S, Pulivarthi D, et al. (2023). Cross-sectional associations of maternal PFAS exposure on SARS-CoV-2 IgG antibody levels during pregnancy. *Environ Res.* 219:115067. doi:10.1016/j.envres.2022.115067 PMID:36528042
- Kawamoto K, Oashi T, Oami K, Liu W, Jin Y, Saito N, et al. (2010). Perfluorooctanoic acid (PFOA) but not perfluorooctane sulfonate (PFOS) showed DNA damage in comet assay on *Paramecium caudatum*. J Toxicol Sci. 35(6):835–41. doi:10.2131/jts.35.835 PMID:21139333
- Kawashima Y, Suzuki S, Kozuka H, Sato M, Suzuki Y (1994). Effects of prolonged administration of perfluorooctanoic acid on hepatic activities of enzymes which detoxify peroxide and xenobiotic in the rat. *Toxicology*. 93(2–3):85–97. doi:<u>10.1016/0300-483X(94)</u> 90070-1 PMID:7974521
- Keil DE, Mehlmann T, Butterworth L, Peden-Adams MM (2008). Gestational exposure to perfluorooctane sulfonate suppresses immune function in B6C3F₁ mice. *Toxicol Sci.* 103(1):77–85. doi:<u>10.1093/toxsci/kfn015</u> PMID:<u>18252804</u>
- Kennedy GL Jr (1985). Dermal toxicity of ammonium perfluorooctanoate. *ToxicolApplPharmacol*.81(2):348– 55. doi:10.1016/0041-008X(85)90172-3 PMID:4060160
- Khalil N, Ebert JR, Honda M, Lee M, Nahhas RW, Koskela A, et al. (2018). Perfluoroalkyl substances, bone density, and cardio-metabolic risk factors in obese 8–12 year old children: a pilot study. *Environ Res.* 160:314–21. doi:10.1016/j.envres.2017.10.014 PMID:29040951
- Khansari MR, Yousefsani BS, Kobarfard F, Faizi M, Pourahmad J (2017). In vitro toxicity of perfluorooctane sulfonate on rat liver hepatocytes: probability of distructive binding to CYP 2E1 and involvement of cellular proteolysis. *Environ Sci Pollut Res Int*. 24(29):23382–8. doi:10.1007/s11356-017-9908-2 PMID:28842823
- Kim H, Hong MW, Bae YH, Lee SJ (2021). Epigenetic toxicity and cytotoxicity of perfluorooctanoic acid and its effects on gene expression in embryonic mouse hypothalamus cells. *Arh Hig Rada Toksikol*. 72(3):182– 90. doi:10.2478/aiht-2021-72-3555 PMID:34587672
- Kim HY, Kim K-N, Shin CH, Lim Y-H, Kim JI, Kim B-N, et al. (2020). The relationship between perfluoroalkyl substances concentrations and thyroid function in early childhood: a prospective cohort study. *Thyroid.* 30(11):1556–65. doi:<u>10.1089/thy.2019.0436</u> PMID:<u>32368952</u>
- Kim JH, Park HY, Jeon JD, Kho Y, Kim SK, Park MS, et al. (2016b). The modifying effect of vitamin C on the association between perfluorinated compounds and

insulin resistance in the Korean elderly: a double-blind, randomized, placebo-controlled crossover trial. *Eur J Nutr.* 55(3):1011–20. doi:<u>10.1007/s00394-015-0915-0</u> PMID:<u>25939797</u>

- Kim MJ, Moon S, Oh BC, Jung D, Ji K, Choi K, et al. (2018). Association between perfluoroalkyl substances exposure and thyroid function in adults: a meta-analysis. *PLoS One*. 13(5):e0197244. doi:<u>10.1371/journal.</u> pone.0197244 PMID:<u>29746532</u>
- Kim S, Choi K, Ji K, Seo J, Kho Y, Park J, et al. (2011). Trans-placental transfer of thirteen perfluorinated compounds and relations with fetal thyroid hormones. *Environ Sci Technol.* 45(17):7465–72. doi:10.1021/ es202408a PMID:21805959
- Kim SJ, Heo SH, Lee DS, Hwang IG, Lee YB, Cho HY (2016a). Gender differences in pharmacokinetics and tissue distribution of 3 perfluoroalkyl and polyfluoroalkyl substances in rats. *Food Chem Toxicol*. 97:243–55. doi:10.1016/j.fct.2016.09.017 PMID:27637925
- Kim WK, Lee SK, Jung J (2010). Integrated assessment of biomarker responses in common carp (*Cyprinus carpio*) exposed to perfluorinated organic compounds. *J Hazard Mater.* 180(1-3):395–400. doi:10.1016/j. jhazmat.2010.04.044 PMID:20452121
- Kimura O, Fujii Y, Haraguchi K, Kato Y, Ohta C, Koga N, et al. (2017). Uptake of perfluorooctanoic acid by Caco-2 cells: involvement of organic anion transporting polypeptides. *Toxicol Lett.* 277:18–23. doi:<u>10.1016/j.toxlet.2017.05.012</u> PMID:<u>28552774</u>
- Kimura O, Fujii Y, Haraguchi K, Kato Y, Ohta C, Koga N, et al. (2020). Effects of perfluoroalkyl carboxylic acids on the uptake of sulfobromophthalein via organic anion transporting polypeptides in human intestinal Caco-2 cells. *Biochem Biophys Rep.* 24:100807. doi:10.1016/j. bbrep.2020.100807 PMID:32964147
- Kingsley SL, Kelsey KT, Butler R, Chen A, Eliot MN, Romano ME, et al. (2017). Maternal serum PFOA concentration and DNA methylation in cord blood: a pilot study. *Environ Res.* 158:174–8. doi:10.1016/j. envres.2017.06.013 PMID:28645023
- Kingsley SL, Walker DI, Calafat AM, Chen A, Papandonatos GD, Xu Y, et al. (2019). Metabolomics of childhood exposure to perfluoroalkyl substances: a cross-sectional study. *Metabolomics*. 15(7):95. doi:<u>10.1007/s11306-019-1560-z</u> PMID:<u>31227916</u>
- Kjeldsen LS, Bonefeld-Jørgensen EC (2013). Perfluorinated compounds affect the function of sex hormone receptors. *Environ Sci Pollut Res Int*. 20(11):8031–44. doi:10.1007/s11356-013-1753-3 PMID:23764977
- Klaunig JE, Babich MA, Baetcke KP, Cook JC, Corton JC, David RM, et al. (2003). PPARalpha agonist-induced rodent tumors: modes of action and human relevance. *Crit Rev Toxicol*. 33(6):655–780. doi:<u>10.1080/713608372</u> PMID:<u>14727734</u>

- Knox SS, Jackson T, Javins B, Frisbee SJ, Shankar A, Ducatman AM (2011). Implications of early menopause in women exposed to perfluorocarbons. *J Clin Endocrinol Metab.* 96(6):1747–53. doi:<u>10.1210/jc.2010-2401 PMID:21411548</u>
- Kobayashi S, Azumi K, Goudarzi H, Araki A, Miyashita C, Kobayashi S, et al. (2017). Effects of prenatal perfluoroalkyl acid exposure on cord blood IGF2/H19 methylation and ponderal index: the Hokkaido Study. *J Expo Sci Environ Epidemiol*. 27(3):251–9. doi:10.1038/ jes.2016.50 PMID:27553991
- Kobayashi S, Sata F, Ikeda-Araki A, Miyashita C, Itoh S, Goudarzi H, et al. (2021). Associations among maternal perfluoroalkyl substance levels, fetal sex-hormone enzymatic gene polymorphisms, and fetal sex hormone levels in the Hokkaido study. *Reprod Toxicol*. 105:221– 31. doi:10.1016/j.reprotox.2021.09.003 PMID:34536543
- Kraugerud M, Zimmer KE, Ropstad E, Verhaegen S (2011). Perfluorinated compounds differentially affect steroidogenesis and viability in the human adreno-cortical carcinoma (H295R) in vitro cell assay. *Toxicol Lett.* 205(1):62–8. doi:10.1016/j.toxlet.2011.05.230 PMID:21641976
- Kristensen SL, Ramlau-Hansen CH, Ernst E, Olsen SF, Bonde JP, Vested A, et al. (2013). Long-term effects of prenatal exposure to perfluoroalkyl substances on female reproduction. *Hum Reprod.* 28(12):3337–48. doi:<u>10.1093/humrep/det382</u> PMID:<u>24129614</u>
- Ku MS, Pan WC, Huang YT, Hsieh WS, Hsu YH, Chen PC, et al. (2022). Associations between prenatal exposure to perfluoroalkyl substances, hypomethylation of MEST imprinted gene and birth outcomes. *Environ Pollut.* 304:119183. doi:<u>10.1016/j.envpol.2022.119183</u> PMID:<u>35331797</u>
- Kudo N, Katakura M, Sato Y, Kawashima Y (2002). Sex hormone-regulated renal transport of perfluorooctanoic acid. *Chem Biol Interact.* 139(3):301–16. doi:10.1016/S0009-2797(02)00006-6 PMID:11879818
- Kudo N, Sakai A, Mitsumoto A, Hibino Y, Tsuda T, Kawashima Y (2007). Tissue distribution and hepatic subcellular distribution of perfluorooctanoic acid at low dose are different from those at high dose in rats. *Biol Pharm Bull.* 30(8):1535–40. doi:<u>10.1248/bpb.30.1535</u> PMID:<u>17666816</u>
- Kummu M, Sieppi E, Koponen J, Laatio L, Vähäkangas K, Kiviranta H, et al. (2015). Organic anion transporter 4 (OAT 4) modifies placental transfer of perfluorinated alkyl acids PFOS and PFOA in human placental ex vivo perfusion system. *Placenta*. 36(10):1185–91. doi:<u>10.1016/j.placenta.2015.07.119</u> PMID:<u>26303760</u>
- Kung Y-P, Lin C-C, Chen M-H, Tsai M-S, Hsieh W-S, Chen P-C (2021). Intrauterine exposure to per- and polyfluoroalkyl substances may harm children's lung function development. *Environ Res.* 192:110178. doi:10.1016/j.envres.2020.110178 PMID:32991923

- Kuslikis BI, Vanden Heuvel JP, Peterson RE (1992). Lack of evidence for perfluorodecanoyl- or perfluorooctanoyl-coenzyme A formation in male and female rats. *J Biochem Toxicol*. 7(1):25–9. doi:<u>10.1002/jbt.2570070106</u> PMID:1588571
- Kvalem HE, Nygaard UC, Lødrup Carlsen KC, Carlsen KH, Haug LS, Granum B (2020). Perfluoroalkyl substances, airways infections, allergy and asthma related health outcomes implications of gender, exposure period and study design. *Environ Int.* 134:105259. doi:10.1016/j.envint.2019.105259 PMID:31733527
- La Rocca C, Tait S, Guerranti C, Busani L, Ciardo F, Bergamasco B, et al. (2015). Exposure to endocrine disruptors and nuclear receptors gene expression in infertile and fertile men from Italian areas with different environmental features. *Int J Environ Res Public Health.* 12(10):12426–45. doi:10.3390/ijerph121012426 PMID:26445054
- Lai KP, Lee JC, Wan HT, Li JW, Wong AY, Chan TF, et al. (2017a). Effects of in utero PFOS exposure on transcriptome, lipidome, and function of mouse testis. *Environ Sci Technol.* 51(15):8782–94. doi:<u>10.1021/acs. est.7b02102</u> PMID:<u>28654245</u>
- Lai KP, Li JW, Cheung A, Li R, Billah MB, Chan TF, et al. (2017b). Transcriptome sequencing reveals prenatal PFOS exposure on liver disorders. *Environ Pollut*. 223:416–25. doi:10.1016/j.envpol.2017.01.041 PMID:28131474
- Lau C, Thibodeaux JR, Hanson RG, Rogers JM, Grey BE, Stanton ME, et al. (2003). Exposure to perfluorooctane sulfonate during pregnancy in rat and mouse. II: postnatal evaluation. *Toxicol Sci*. 74(2):382–92. doi:<u>10.1093/</u> <u>toxsci/kfg122</u> PMID:<u>12773772</u>
- Lebeaux RM, Doherty BT, Gallagher LG, Zoeller RT, Hoofnagle AN, Calafat AM, et al. (2020). Maternal serum perfluoroalkyl substance mixtures and thyroid hormone concentrations in maternal and cord sera: the HOME Study. *Environ Res.* 185:109395. doi:<u>10.1016/j.</u> <u>envres.2020.109395</u> PMID:<u>32222633</u>
- Lee HG, Lee YJ, Yang JH (2012). Perfluorooctane sulfonate induces apoptosis of cerebellar granule cells via a ROS-dependent protein kinase C signalling pathway. *Neurotoxicology*. 33(3):314–20. doi:10.1016/j. neuro.2012.01.017 PMID:22326494
- Lee J-K, Lee S, Baek M-C, Lee B-H, Lee H-S, Kwon TK, et al. (2017). Association between perfluorooctanoic acid exposure and degranulation of mast cells in allergic inflammation. *J Appl Toxicol.* 37(5):554–62. doi:10.1002/jat.3389 PMID:27682001
- Lee YY, Wong CK, Oger C, Durand T, Galano JM, Lee JC (2015). Prenatal exposure to the contaminant perfluorooctane sulfonate elevates lipid peroxidation during mouse fetal development but not in the pregnant dam. *Free Radic Res.* 49(8):1015–25. doi:10.3109/10715762.20 15.1027199 PMID:25787935

- Lefebvre DE, Curran I, Armstrong C, Coady L, Parenteau M, Liston V, et al. (2008). Immunomodulatory effects of dietary potassium perfluorooctane sulfonate (PFOS) exposure in adult Sprague-Dawley rats. *J Toxicol Environ Health A*. 71(23):1516–25. doi:10.1080/15287390802391943 PMID:18923994
- Leter G, Consales C, Eleuteri P, Uccelli R, Specht IO, Toft G, et al. (2014). Exposure to perfluoroalkyl substances and sperm DNA global methylation in Arctic and European populations. *Environ Mol Mutagen*. 55(7):591–600. doi:10.1002/em.21874 PMID:24889506
- Lewis RC, Johns LE, Meeker JD (2015). Serum biomarkers of exposure to perfluoroalkyl substances in relation to serum testosterone and measures of thyroid function among adults and adolescents from NHANES 2011– 2012. *Int J Environ Res Public Health*. 12(6):6098–114. doi:10.3390/ijerph120606098 PMID:26035660
- Li A, Koch Z, Ideker T (2022b). Epigenetic aging: biological age prediction and informing a mechanistic theory of aging. *J Intern Med.* 292(5):733–44. doi:<u>10.1111/</u> joim.13533 PMID:<u>35726002</u>
- Li C, Jiang L, Qi Y, Zhang D, Liu X, Han W, et al. (2023b). Integration of metabolomics and proteomics reveals the underlying hepatotoxic mechanism of perfluorooctane sulfonate (PFOS) and 6:2 chlorinated polyfluoroalkyl ether sulfonic acid (6:2 Cl-PFESA) in primary human hepatocytes. *Ecotoxicol Environ Saf*. 249:114361. doi:10.1016/j.ecoenv.2022.114361 PMID:36508832
- Li C, Liu X, Liu Q, Li S, Li Y, Hu H, et al. (2017b). Protection of taurine against PFOS-induced neurotoxicity in PC12 Cells. *Adv Exp Med Biol*. 975(Pt 2):907–16. doi:10.1007/978-94-024-1079-2_72 PMID:28849510
- Li C-H, Ren X-M, Guo L-H (2019b). Adipogenic activity of oligomeric hexafluoropropylene oxide (perfluorooctanoic acid alternative) through peroxisome proliferator-activated receptor γ pathway. *Environ Sci Technol.* 53(6):3287–95. doi:<u>10.1021/acs.est.8b06978</u> PMID:<u>30785727</u>
- Li C-H, Ren X-M, Ruan T, Cao L-Y, Xin Y, Guo L-H, et al. (2018b). Chlorinated polyfluorinated ether sulfonates exhibit higher activity toward peroxisome proliferator-activated receptors signalling pathways than perfluorooctanesulfonate. *Environ Sci Technol.* 52(5):3232–9. doi:10.1021/acs.est.7b06327 PMID:29389105
- Li D, Zhang L, Zhang Y, Guan S, Gong X, Wang X (2019a). Maternal exposure to perfluorooctanoic acid (PFOA) causes liver toxicity through PPAR-α pathway and lowered histone acetylation in female offspring mice. *Environ Sci Pollut Res Int*. 26(18):18866–75. doi:10.1007/ s11356-019-05258-z PMID:31062244
- Li J, Cao H, Feng H, Xue Q, Zhang A, Fu J (2020e). Evaluation of the estrogenic/antiestrogenic activities of perfluoroalkyl substances and their interactions with the human estrogen receptor by combining in vitro assays and in silico modeling. *Environ Sci*

Technol. 54(22):14514–24. doi:<u>10.1021/acs.est.0c03468</u> PMID:<u>33111528</u>

- Li J, Quan X, Lei S, Chen G, Hong J, Huang Z, et al. (2022c). LncRNA MEG3 alleviates PFOS induced placental cell growth inhibition through its derived miR-770 targeting PTX3. *Environ Pollut*. 293:118542. doi:10.1016/j.envpol.2021.118542 PMID:34801623
- Li J, Quan X, Lei S, Huang Z, Wang Q, Xu P (2021a). PFOS inhibited normal functional development of placenta cells via *PPARy* signalling. *Biomedicines*. 9(6):677. doi:10.3390/biomedicines9060677 PMID:34203907
- Li J, Quan XJ, Chen G, Hong JW, Wang Q, Xu LL, et al. (2020b). PFOS-induced placental cell growth inhibition is partially mediated by lncRNA H19 through interacting with miR-19a and miR-19b. *Chemosphere*. 261:127640. doi:10.1016/j.chemosphere.2020.127640 PMID:32738709
- Li K, Sun J, Yang J, Roberts SM, Zhang X, Cui X, et al. (2017a). Molecular mechanisms of perfluorooctanoateinduced hepatocyte apoptosis in mice using proteomic techniques. *Environ Sci Technol.* 51(19):11380–9. doi:10.1021/acs.est.7b02690 PMID:28885018
- Li L, Guo Y, Ma S, Wen H, Li Y, Qiao J (2024). Association between exposure to per- and perfluoroalkyl substances (PFAS) and reproductive hormones in human: a systematic review and meta-analysis. *Environ Res.* 241:117553. doi:10.1016/j.envres.2023.117553 PMID:37931739
- Li L, Xu ZS, Song GW (2009). Study on the Langmuir aggregation of fluorinated surfactants on protein. *J Fluor Chem.* 130(2):225–30. doi:10.1016/j. jfluchem.2008.10.007
- Li Q-Q, Liu J-J, Su F, Zhang Y-T, Wu L-Y, Chu C, et al. (2022d). Chlorinated polyfluorinated ether sulfonates and thyroid hormone levels in adults: isomers of C8 Health Project in China. *Environ Sci Technol.* 56(10):6152–61. doi:<u>10.1021/acs.est.1c03757</u> PMID:<u>35380809</u>
- Li R, Guo C, Lin X, Chan TF, Lai KP, Chen J (2020c). Integrative omics analyses uncover the mechanism underlying the immunotoxicity of perfluorooctanesulfonate in human lymphocytes. *Chemosphere*. 256:127062. doi:10.1016/j.chemosphere.2020.127062 PMID:32434090
- Li R, Guo C, Lin X, Chan TF, Su M, Zhang Z, et al. (2022g). Integrative omics analysis reveals the protective role of vitamin C on perfluorooctanoic acid-induced hepatoxicity. *J Adv Res.* 35:279–94. doi:10.1016/j. jare.2021.04.003 PMID:35024202
- Li R, Guo C, Tse WKF, Su M, Zhang X, Lai KP (2020d). Metabolomic analysis reveals metabolic alterations of human peripheral blood lymphocytes by perfluorooctanoic acid. *Chemosphere*. 239:124810. doi:10.1016/j. chemosphere.2019.124810 PMID:31520980
- Li W, He QZ, Wu CQ, Pan XY, Wang J, Tan Y, et al. (2015). PFOS disturbs BDNF-ERK-CREB signalling in association with increased microRNA-22 in SH-SY5Y cells.

BioMed Res Int. 2015:302653. doi:<u>10.1155/2015/302653</u> PMID:<u>26649298</u>

- Li X, Bao C, Ma Z, Xu B, Ying X, Liu X, et al. (2018c). Perfluorooctanoic acid stimulates ovarian cancer cell migration, invasion via ERK/NF-κB/MMP-2/-9 pathway. *Toxicol Lett.* 294(15 September):44–50. doi:<u>10.1016/j.toxlet.2018.05.009</u> PMID:<u>29753068</u>
- Li X, Chen L, Zhou H, Wang J, Zhao C, Pang X (2022f). PFOA regulate adenosine receptors and downstream concentration-response cAMP–PKA pathway revealed by integrated omics and molecular dynamics analyses. *Sci Total Environ*. 803:149910. doi:<u>10.1016/j.</u> <u>scitotenv.2021.149910</u> PMID:<u>34500266</u>
- Li X, Liu H, Wan H, Li Y, Xu S, Xiao H, et al. (2023a). Sex-specific associations between legacy and novel per- and polyfluoroalkyl substances and telomere length in newborns in Wuhan, China: mixture and single pollutant associations. *Sci Total Environ*. 857(Pt 3):159676. doi:10.1016/j.scitotenv.2022.159676 PMID:36283531
- Li X, Wang Z, Klaunig JE (2019c). The effects of perfluorooctanoate on high fat diet induced non-alcoholic fatty liver disease in mice. *Toxicology*. 416:1–14. doi:<u>10.1016/j.</u> tox.2019.01.017 PMID:<u>30711707</u>
- Li Y, Andersson A, Xu Y, Pineda D, Nilsson CA, Lindh CH, et al. (2022a). Determinants of serum half-lives for linear and branched perfluoroalkyl substances after long-term high exposure a study in Ronneby, Sweden. *Environ Int.* 163:107198. doi:10.1016/j. envint.2022.107198 PMID:35447437
- Li Y, Barregard L, Xu Y, Scott K, Pineda D, Lindh CH, et al. (2020a). Associations between perfluoroalkyl substances and serum lipids in a Swedish adult population with contaminated drinking water. *Environ Health*. 19(1):33. doi:10.1186/s12940-020-00588-9 PMID:32169067
- Li Y, Cheng Y, Xie Z, Zeng F (2017c). Perfluorinated alkyl substances in serum of the southern Chinese general population and potential impact on thyroid hormones. *Sci Rep.* 7(1):43380. doi:<u>10.1038/srep43380</u> PMID:<u>28240244</u>
- Li Y, Fletcher T, Mucs D, Scott K, Lindh CH, Tallving P, et al. (2018a). Half-lives of PFOS, PFHxS and PFOA after end of exposure to contaminated drinking water. *Occup Environ Med.* 75(1):46–51. doi:<u>10.1136/oemed-2017-104651</u> PMID:<u>29133598</u>
- Li Y, Ramdhan DH, Naito H, Yamagishi N, Ito Y, Hayashi Y, et al. (2011). Ammonium perfluorooctanoate may cause testosterone reduction by adversely affecting testis in relation to PPARa. *Toxicol Lett.* 205(3):265–72. doi:10.1016/j.toxlet.2011.06.015 PMID:21712084
- Li Y, Xu Y, Fletcher T, Scott K, Nielsen C, Pineda D, et al. (2021b). Associations between perfluoroalkyl substances and thyroid hormones after high exposure through drinking water. *Environ Res.* 194:110647. doi:10.1016/j.envres.2020.110647 PMID:33358873

- Li Z, Lin Z, Ji S, Lai KP, Wan HT, Wong CKC, et al. (2022e). Perfluorooctanesulfonic acid exposure altered hypothalamic metabolism and disturbed male fecundity. *Sci Total Environ*. 844:156881. doi:<u>10.1016/j.scitotenv.2022.156881</u> PMID:<u>35753445</u>
- Liang H, Wang Z, Miao M, Tian Y, Zhou Y, Wen S, et al. (2020). Prenatal exposure to perfluoroalkyl substances and thyroid hormone concentrations in cord plasma in a Chinese birth cohort. *Environ Health*. 19(1):127. doi:10.1186/s12940-020-00679-7 PMID:33243245
- Liao Y, Dong S, Kiyama R, Cai P, Liu L, Shen H (2013). *Flos lonicerae* extracts and chlorogenic acid protect human umbilical vein endothelial cells from the toxic damage of perfluorooctane sulphonate. *Inflammation*. 36(3): 767–79.doi:<u>10.1007/s10753-013-9603-5</u>PMID:<u>23392856</u>
- Liao Y, Wang J, Huang QS, Fang C, Kiyama R, Shen H, et al. (2012). Evaluation of cellular response to perfluorooctane sulfonate in human umbilical vein endothelial cells. *Toxicol In Vitro*. 26(3):421–8. doi:<u>10.1016/j.</u> <u>tiv.2012.01.011</u> PMID:<u>22269386</u>
- Lin CY, Chen PC, Lo SC, Torng PL, Sung FC, Su TC (2016). The association of carotid intima-media thickness with serum level of perfluorinated chemicals and endothelium-platelet microparticles in adolescents and young adults. *Environ Int*. 94:292–9. doi:<u>10.1016/j.</u> <u>envint.2016.06.004</u> PMID:<u>27288966</u>
- Lin CY, Lee HL, Chen CW, Wang C, Sung FC, Su TC (2022). Global DNA methylation mediates the association between serum perfluorooctane sulfonate and carotid intima-media thickness in young and middle-aged Taiwanese populations. *Ecotoxicol Environ Saf.* 241:113782. doi:10.1016/j.ecoenv.2022.113782 PMID:35753273
- Lin CY, Lee HL, Hwang YT, Su TC (2020b). The association between total serum isomers of per- and polyfluoroalkyl substances, lipid profiles, and the DNA oxidative/nitrative stress biomarkers in middle-aged Taiwanese adults. *Environ Res.* 182:109064. doi:10.1016/j.envres.2019.109064 PMID:31884197
- Lin C-Y, Wen LL, Lin L-Y, Wen T-W, Lien G-W, Chen C-Y, et al. (2011). Associations between levels of serum perfluorinated chemicals and adiponectin in a young hypertension cohort in Taiwan. *Environ Sci Technol.* 45(24):10691–8. doi:10.1021/es201964x PMID:22047537
- Lin CY, Wen LL, Lin LY, Wen TW, Lien GW, Hsu SH, et al. (2013). The associations between serum perfluorinated chemicals and thyroid function in adolescents and young adults. *J Hazard Mater.* 244–245:637–44. doi:10.1016/j.jhazmat.2012.10.049 PMID:23177245
- Lin T, Zhang Y, Ding X, Huang T, Zhang W, Zou W, et al. (2020a). Perfluorooctanoic acid induces cytotoxicity in spermatogonial GC-1 cells. *Chemosphere*. 260:127545. doi:<u>10.1016/j.chemosphere.2020.127545</u> PMID:<u>32653749</u>

- Lindeman B, Maass C, Duale N, Gützkow KB, Brunborg G, Andreassen Å (2012). Effects of per- and polyfluorinated compounds on adult rat testicular cells following in vitro exposure. *Reprod Toxicol.* 33(4):531–7. doi:10.1016/j.reprotox.2011.04.001 PMID:21530637
- Liu C, Chang VW, Gin KY (2014a). Oxidative toxicity of perfluorinated chemicals in green mussel and bioaccumulation factor dependent quantitative structure-activity relationship. *Environ Toxicol Chem*. 33(10):2323–32. doi:10.1002/etc.2679 PMID:24995545
- Liu C, Chang VW, Gin KY, Nguyen VT (2014b). Genotoxicity of perfluorinated chemicals (PFCs) to the green mussel (*Perna viridis*). *Sci Total Environ*. 487:117– 22. doi:10.1016/j.scitotenv.2014.04.017 PMID:24784736
- Liu CY, Chen PC, Lien PC, Liao YP (2018a). Prenatal perfluorooctyl sulfonate exposure and Alu DNA hypomethylation in cord blood. *Int J Environ Res Public Health*. 15(6):1066. doi:<u>10.3390/ijerph15061066</u> PMID:<u>29795014</u>
- Liu D, Tang B, Nie S, Zhao N, He L, Cui J, et al. (2023a). Distribution of per- and poly-fluoroalkyl substances and their precursors in human blood. *J Hazard Mater.* 441:129908. doi:<u>10.1016/j.jhazmat.2022.129908</u> PMID:<u>36115093</u>
- Liu H, Chen Q, Lei L, Zhou W, Huang L, Zhang J, et al. (2018b). Prenatal exposure to perfluoroalkyl and polyfluoroalkyl substances affects leukocyte telomere length in female newborns. *Environ Pollut*. 235:446–52. doi:10.1016/j.envpol.2017.12.095 PMID:29310088
- Liu H, Huang Y, Pan Y, Cheng R, Li X, Li Y, et al. (2023c). Associations between per and polyfluoroalkyl ether sulfonic acids and vitamin D biomarker levels in Chinese newborns. *Sci Total Environ*. 866:161410. doi:10.1016/j.scitotenv.2023.161410 PMID:36621489
- Liu H, Pan Y, Jin S, Li Y, Zhao L, Sun X, et al. (2020b). Associations of per-/polyfluoroalkyl substances with glucocorticoids and progestogens in newborns. *Environ Int.* 140:105636. doi:<u>10.1016/j.envint.2020.105636</u> PMID:<u>32474218</u>
- Liu H, Pan Y, Jin S, Sun X, Jiang Y, Wang Y, et al. (2021). Associations between six common per- and polyfluoroalkyl substances and estrogens in neonates of China. *J Hazard Mater.* 407:124378. doi:<u>10.1016/j.</u> jhazmat.2020.124378 PMID:<u>33139105</u>
- Liu L, Liu W, Song J, Yu H, Jin Y, Oami K, et al. (2009). A comparative study on oxidative damage and distributions of perfluorooctane sulfonate (PFOS) in mice at different postnatal developmental stages. *J Toxicol Sci*. 34(3):245–54. doi:10.2131/jts.34.245 PMID:19483379
- Liu M, Zhang G, Meng L, Han X, Li Y, Shi Y, et al. (2022b). Associations between novel and legacy per- and polyfluoroalkyl substances in human serum and thyroid cancer: a case and healthy population in Shandong Province, East China. *Environ Sci Technol.* 56(10):6144– 51. doi:10.1021/acs.est.1c02850 PMID:34618433

- Liu RC, Hurtt ME, Cook JC, Biegel LB (1996). Effect of the peroxisome proliferator, ammonium perfluorooctanoate (C8), on hepatic aromatase activity in adult male Crl:CD BR (CD) rats. *Fundam Appl Toxicol*. 30(2):220–8. doi:10.1006/faat.1996.0059 PMID:8812269
- Liu W, Irudayaraj J (2020). Perfluorooctanoic acid (PFOA) exposure inhibits DNA methyltransferase activities and alters constitutive heterochromatin organization. *Food Chem Toxicol.* 141:111358. doi:<u>10.1016/j.fct.2020.111358</u> PMID:<u>32315686</u>
- Liu W, Qin H, Pan Y, Luo F, Zhang Z (2019). Low concentrations of perfluorooctane sulfonate repress osteogenic and enhance adipogenic differentiation of human mesenchymal stem cells. *Toxicol Appl Pharmacol.* 367:82–91. doi:10.1016/j.taap.2019.02.001 PMID:30738844
- Liu W, Xu C, Sun X, Kuang H, Kuang X, Zou W, et al. (2015b). Grape seed proanthocyanidin extract protects against perfluorooctanoic acid-induced hepatotoxicity by attenuating inflammatory response, oxidative stress and apoptosis in mice. *Toxicol Res (Camb)*. 5(1):224–34. doi:10.1039/C5TX00260E PMID:30090339
- Liu W, Yang B, Wu L, Zou W, Pan X, Zou T, et al. (2015a). Involvement of NRF2 in perfluorooctanoic acid-induced testicular damage in male mice. *Biol Reprod.* 93(2):41. doi:10.1095/biolreprod.115.128819 PMID:26108789
- Liu X, Chen D, Wang B, Xu F, Pang Y, Zhang L, et al. (2020a). Does low maternal exposure to per- and poly-fluoroalkyl substances elevate the risk of spontaneous preterm birth? A nested case-control study in China. *Environ Sci Technol.* 54(13):8259–68. doi:<u>10.1021/acs.est.0c01930</u> PMID:<u>32510220</u>
- Liu Y, Eliot MN, Papandonatos GD, Kelsey KT, Fore R, Langevin S, et al. (2022a). Gestational perfluoroalkyl substance exposure and DNA methylation at birth and 12 years of age: a longitudinal epigenome-wide association study. *Environ Health Perspect*. 130(3):37005. doi:10.1289/EHP10118 PMID:35266797
- Liu Y, Li J, Ding H, Ge D, Wang J, Xu C (2022c). Perfluorooctane sulfonate (PFOS) triggers migration and invasion of esophageal squamous cell carcinoma cells via regulation of Zeb1. *Drug Chem Toxicol*. 45(6):2804–13. doi:10.1080/01480545.2021.1991775 PMID:34732098
- Liu Y, Wang J, Wei Y, Zhang H, Liu Y, Dai J (2008). Molecular characterization of cytochrome P450 1A and 3A and the effects of perfluorooctanoic acid on their mRNA levels in rare minnow (*Gobiocypris rarus*) gills. *Aquat Toxicol.* 88(3):183–90. doi:10.1016/j. aquatox.2008.04.008 PMID:18508136
- Liu Y, Yu G, Zhang R, Feng L, Zhang J (2023b). Early life exposure to low-dose perfluorooctane sulfonate disturbs gut barrier homeostasis and increases the risk of intestinal inflammation in offspring. *Environ*

Pollut. 329:121708. doi:<u>10.1016/j.envpol.2023.121708</u> PMID:<u>37100370</u>

- Loccisano AE, CampbellJLJr, Andersen ME, ClewellHJ3rd (2011). Evaluation and prediction of pharmacokinetics of PFOA and PFOS in the monkey and human using a PBPK model. *Regul Toxicol Pharmacol.* 59(1):157–75. doi:10.1016/j.yrtph.2010.12.004 PMID:21168463
- Loccisano AE, Longnecker MP, Campbell JL Jr, Andersen ME, Clewell HJ 3rd (2013). Development of PBPK models for PFOA and PFOS for human pregnancy and lactation life stages. *J Toxicol Environ Health A*. 76(1):25–57. doi:10.1080/15287394.2012.722523 PMID:23151209
- Loft S, Olsen A, Møller P, Poulsen HE, Tjønneland A (2013). Association between 8-oxo-7,8-dihydro-2'-deoxyguanosine excretion and risk of postmenopausal breast cancer: nested case-control study. *Cancer Epidemiol Biomarkers Prev.* 22(7):1289–96. doi:10.1158/1055-9965.EPI-13-0229 PMID:23658396
- Loft S, Svoboda P, Kasai H, Tjønneland A, Vogel U, Møller P, et al. (2006). Prospective study of 8-oxo-7,8-dihydro-2'-deoxyguanosine excretion and the risk of lung cancer. *Carcinogenesis*. 27(6):1245–50. doi:<u>10.1093/</u> <u>carcin/bgi313</u> PMID:<u>16364924</u>
- Loft S, Svoboda P, Kawai K, Kasai H, Sørensen M, Tjønneland A, et al. (2012). Association between 8-oxo-7,8-dihydroguanine excretion and risk of lung cancer in a prospective study. *Free Radic Biol Med*. 52(1):167–72. doi:10.1016/j.freeradbiomed.2011.10.439 PMID:22044660
- Logeshwaran P, Sivaram AK, Surapaneni A, Kannan K, Naidu R, Megharaj M (2021). Exposure to perfluorooctanesulfonate (PFOS) but not perflurorooctanoic acid (PFOA) at ppb concentration induces chronic toxicity in *Daphnia carinata. Sci Total Environ.* 769:144577. doi:10.1016/j.scitotenv.2020.144577 PMID:33482550
- Long M, Ghisari M, Bonefeld-Jørgensen EC (2013). Effects of perfluoroalkyl acids on the function of the thyroid hormone and the aryl hydrocarbon receptor. *Environ Sci Pollut Res Int*. 20(11):8045–56. doi:<u>10.1007/s11356-013-1628-7 PMID:23539207</u>
- Looker C, Luster MI, Calafat AM, Johnson VJ, Burleson GR, Burleson FG, et al. (2014). Influenza vaccine response in adults exposed to perfluorooctanoate and perfluorooctanesulfonate. *Toxicol Sci.* 138(1):76–88. doi:10.1093/toxsci/kft269 PMID:24284791
- López-Arellano P, López-Arellano K, Luna J, Flores D, Jiménez-Salazar J, Gavia G, et al. (2019). Perfluorooctanoic acid disrupts gap junction intercellular communication and induces reactive oxygen species formation and apoptosis in mouse ovaries. *Environ Toxicol.* 34(1):92–8. doi:10.1002/tox.22661 PMID:30277307
- López-Doval S, Salgado R, Fernández-Pérez B, Lafuente A (2015). Possible role of serotonin and neuropeptide Y on the disruption of the reproductive axis activity by

perfluorooctane sulfonate. *Toxicol Lett*. 233(2):138–47. doi:<u>10.1016/j.toxlet.2015.01.012</u> PMID:<u>25623392</u>

- Lopez-Espinosa MJ, Carrizosa C, Luster MI, Margolick JB, Costa O, Leonardi GS, et al. (2021). Perfluoroalkyl substances and immune cell counts in adults from the Mid-Ohio Valley (USA). *Environ Int*. 156:106599. doi:10.1016/j.envint.2021.106599 PMID:33993002
- Lopez-Espinosa M-J, Mondal D, Armstrong B, Bloom MS, Fletcher T (2012). Thyroid function and perfluoroalkyl acids in children living near a chemical plant. *Environ Health Perspect*. 120(7):1036–41. doi:10.1289/ ehp.1104370 PMID:22453676
- Lopez-Espinosa M-J, Mondal D, Armstrong BG, Eskenazi B, Fletcher T (2016). Perfluoroalkyl substances, sex hormones, and insulin-like growth factor-1 at 6–9 years of age: a cross-sectional analysis within the C8 Health Project. *Environ Health Perspect*. 124(8):1269– 75. doi:10.1289/ehp.1509869 PMID:26794451
- Lou I, Wambaugh JF, Lau C, Hanson RG, Lindstrom AB, Strynar MJ, et al. (2009). Modeling single and repeated dose pharmacokinetics of PFOA in mice. *Toxicol Sci.* 107(2):331–41. doi:<u>10.1093/toxsci/kfn234</u> PMID:<u>19005225</u>
- Louisse J, Fragki S, Rijkers D, Janssen A, van Dijk B, Leenders L, et al. (2023). Determination of in vitro hepatotoxic potencies of a series of perfluoroalkyl substances (PFASs) based on gene expression changes in HepaRG liver cells. *Arch Toxicol*. 97(4):1113–31. doi:10.1007/s00204-023-03450-2 PMID:36864359
- Louisse J, Rijkers D, Stoopen G, Janssen A, Staats M, Hoogenboom R, et al. (2020). Perfluorooctanoic acid (PFOA), perfluorooctane sulfonic acid (PFOS), and perfluorononanoic acid (PFNA) increase triglyceride levels and decrease cholesterogenic gene expression in human HepaRG liver cells. *Arch Toxicol*. 94(9):3137–55. doi:<u>10.1007/s00204-020-02808-0</u> PMID:<u>32588087</u>
- Loveless SE, Finlay C, Everds NE, Frame SR, Gillies PJ, O'Connor JC, et al. (2006). Comparative responses of rats and mice exposed to linear/branched, linear, or branched ammonium perfluorooctanoate (APFO). *Toxicology*. 220(2–3):203–17. doi:10.1016/j. tox.2006.01.003 PMID:16448737
- Loveless SE, Hoban D, Sykes G, Frame SR, Everds NE (2008). Evaluation of the immune system in rats and mice administered linear ammonium perfluorooctanoate. *Toxicol Sci.* 105(1):86–96. doi:10.1093/toxsci/ kfn113 PMID:18559402
- Lu H, Zhang H, Gao J, Li Z, Bao S, Chen X, et al. (2019). Effects of perfluorooctanoic acid on stem Leydig cell functions in the rat. *Environ Pollut*. 250:206–15. doi:<u>10.1016/j.envpol.2019.03.120</u> PMID:<u>30999198</u>
- Lu L, Guo L, Li M, Kang T, Cheng S, Miao W (2016a). Investigation of perfluorooctanoic acid induced DNA damage using electrogenerated chemiluminescence associated with charge transfer in DNA. *Anal Bioanal*

Chem. 408(25):7137–45. doi:<u>10.1007/s00216-016-</u> <u>9559-z</u> PMID:<u>27108285</u>

- Lu L, Xu L, Kang T, Cheng S (2012). Investigation of DNA damage treated with perfluorooctane sulfonate (PFOS) on ZrO2/DDAB active nano-order film. *Biosens Bioelectron*. 35(1):180–5. doi:<u>10.1016/j.bios.2012.02.043</u> PMID:<u>22459586</u>
- Lu Y, Luo B, Li J, Dai J (2016c). Perfluorooctanoic acid disrupts the blood-testis barrier and activates the TNFα/p38 MAPK signalling pathway in vivo and in vitro. *Arch Toxicol.* 90(4):971-83. doi:<u>10.1007/s00204-015-1492-y</u> PMID:<u>25743374</u>
- Lu Y, Pan Y, Sheng N, Zhao AZ, Dai J (2016b). Perfluorooctanoic acid exposure alters polyunsaturated fatty acid composition, induces oxidative stress and activates the AKT/AMPK pathway in mouse epididymis. *Chemosphere*. 158:143–53. doi:10.1016/j. chemosphere.2016.05.071 PMID:27262104
- Lu Y, Wang J, Guo X, Yan S, Dai J (2017). Perfluorooctanoic acid affects endocytosis involving clathrin light chain A and microRNA-133b-3p in mouse testes. *Toxicol Appl Pharmacol.* 318:41–8. doi:<u>10.1016/j.taap.2017.01.014</u> PMID:<u>28126411</u>
- Luebker DJ, Hansen KJ, Bass NM, Butenhoff JL, Seacat AM (2002). Interactions of fluorochemicals with rat liver fatty acid-binding protein. *Toxicology*. 176(3):175–85. doi:10.1016/S0300-483X(02)00081-1 PMID:12093614
- Luebker DJ, York RG, Hansen KJ, Moore JA, Butenhoff JL (2005). Neonatal mortality from in utero exposure to perfluorooctanesulfonate (PFOS) in Sprague-Dawley rats: dose-response, and biochemical and pharamacokinetic parameters. *Toxicology*. 215(1-2):149–69. doi:10.1016/j.tox.2005.07.019 PMID:16129535
- Luo K, Liu X, Nian M, Wang Y, Qiu J, Yu H, et al.; Shanghai Birth Cohort (2021). Environmental exposure to perand polyfluoroalkyl substances mixture and male reproductive hormones. *Environ Int.* 152:106496. doi:10.1016/j.envint.2021.106496 PMID:33744484
- Luo Z, Shi X, Hu Q, Zhao B, Huang M (2012). Structural evidence of perfluorooctane sulfonate transport by human serum albumin. *Chem Res Toxicol*. 25(5):990–2. doi:<u>10.1021/tx300112p</u> PMID:<u>22482699</u>
- Lv QY, Wan B, Guo LH, Yang Y, Ren XM, Zhang H (2015). In vivo immunotoxicity of perfluorooctane sulfonate in BALB/c mice: identification of T-cell receptor and calcium-mediated signalling pathway disruption through gene expression profiling of the spleen. *Chem Biol Interact.* 240:84–93. doi:<u>10.1016/j.cbi.2015.07.015</u> PMID:<u>26300304</u>
- Lv Z, Li G, Li Y, Ying C, Chen J, Chen T, et al. (2013). Glucose and lipid homeostasis in adult rat is impaired by early-life exposure to perfluorooctane sulfonate. *Environ Toxicol.* 28(9):532–42. doi:<u>10.1002/tox.20747</u> PMID:<u>23983163</u>

- LvZ, WuW, GeS, Jia R, Lin T, Yuan Y, et al. (2018). Naringin protects against perfluorooctane sulfonate-induced liver injury by modulating NRF2 and NF-κB in mice. *Int Immunopharmacol.* 65:140–7. doi:10.1016/j. intimp.2018.09.019 PMID:30316072
- Ma S, Sun Y, Zheng X, Yang Y (2021). Gastrodin attenuates perfluorooctanoic acid-induced liver injury by regulating gut microbiota composition in mice. *Bioengineered*. 12(2):11546–56. doi:<u>10.1080/21655979.2</u> 021.2009966 PMID:<u>34866532</u>
- Ma X, Ren X, Zhang X, Griffin N, Liu H, Wang L (2023). Rutin ameliorates perfluorooctanoic acid-induced testicular injury in mice by reducing oxidative stress and improving lipid metabolism. *Drug Chem Toxicol.* 46(6):1223-34. <u>https://www.tandfonline.com/doi/ full/10.1080/01480545.2022.2145483?scroll=top&ne edAccess=truedoi:10.1080/01480545.2022.2145483</u> PMID:36373176
- Ma Y, Yang J, Wan Y, Peng Y, Ding S, Li Y, et al. (2018). Low-level perfluorooctanoic acid enhances 3T3–L1 preadipocyte differentiation via altering peroxisome proliferator activated receptor gamma expression and its promoter DNA methylation. *J Appl Toxicol*. 38(3):398–407. doi:10.1002/jat.3549 PMID:29094436
- Ma Z, Liu X, Li F, Wang Y, Xu Y, Zhang M, et al. (2016). Perfluorooctanoic acid induces human Ishikawa endometrial cancer cell migration and invasion through activation of ERK/mTOR signalling. *Oncotarget*. 7(41):66558–68. doi:<u>10.18632/oncotarget.11684</u> PMID: <u>27589685</u>
- Macon MB, Villanueva LR, Tatum-Gibbs K, Zehr RD, Strynar MJ, Stanko JP, et al. (2011). Prenatal perfluorooctanoic acid exposure in CD-1 mice: low-dose developmental effects and internal dosimetry. *Toxicol Sci.* 122(1):134–45. doi:<u>10.1093/toxsci/kfr076</u> PMID:<u>21482639</u>
- Maddalon A, Cari L, Iulini M, Alhosseini MN, Galbiati V, Marinovich M, et al. (2023a). Impact of endocrine disruptors on peripheral blood mononuclear cells in vitro: role of gender. *Arch Toxicol.* 97(12):3129–50. doi:10.1007/s00204-023-03592-3 PMID:37676302
- Maddalon A, Pierzchalski A, Kretschmer T, Bauer M, Zenclussen AC, Marinovich M, et al. (2023b). Mixtures of per- and poly-fluoroalkyl substances (PFAS) reduce the in vitro activation of human T cells and basophils. *Chemosphere*. 336:139204. doi:10.1016/j. chemosphere.2023.139204 PMID:37315852
- Maestri L, Negri S, Ferrari M, Ghittori S, Fabris F, Danesino P, et al. (2006). Determination of perfluorooctanoic acid and perfluorooctanesulfonate in human tissues by liquid chromatography/single quadrupole mass spectrometry. *Rapid Commun Mass Spectrom*. 20(18):2728–34. doi:10.1002/rcm.2661 PMID:16915561
- Magnifico MC, Xhani M, Sprovera B, Buttari B, Abballe G, Desideri F, et al. (2022). Nitro-oxidative stress and mitochondrial dysfunction in human cell lines exposed

to the environmental contaminants PFOA and BPA. *Front Biosci (Landmark Ed)*. 27(10):292. doi:<u>10.31083/j.</u> <u>fbl2710292</u> PMID:<u>36336871</u>

- Maisonet M, Calafat AM, Marcus M, Jaakkola JJK, Lashen H (2015). Prenatal exposure to perfluoroalkyl acids and serum testosterone concentrations at 15 years of age in female ALSPAC study participants. *Environ Health Perspect*. 123(12):1325–30. doi:10.1289/ehp.1408847 PMID:26034840
- Maloney EK, Waxman DJ (1999). trans-Activation of PPARα and PPARγ by structurally diverse environmental chemicals. Toxicol Appl Pharmacol. 161(2):209– 18. doi:10.1006/taap.1999.8809 PMID:10581215
- Mamsen LS, Björvang RD, Mucs D, Vinnars M-T, Papadogiannakis N, Lindh CH, et al. (2019). Concentrations of perfluoroalkyl substances (PFASs) in human embryonic and fetal organs from first, second, and third trimester pregnancies. *Environ Int*. 124:482– 92. doi:10.1016/j.envint.2019.01.010 PMID:30684806
- Mandard S, Müller M, Kersten S (2004). Peroxisome proliferator-activated receptor alpha target genes. *Cell Mol Life Sci.* 61(4):393–416. doi:<u>10.1007/s00018-003-3216-3</u> PMID:<u>14999402</u>
- Manzano-Salgado CB, Granum B, Lopez-Espinosa MJ, Ballester F, Iñiguez C, Gascón M, et al. (2019). Prenatal exposure to perfluoroalkyl substances, immune-related outcomes, and lung function in children from a Spanish birth cohort study. *Int J Hyg Environ Health.* 222(6):945–54. doi:10.1016/j.ijheh.2019.06.005 PMID:31262703
- Mao B, Li C, Wen Z, Li H, Wang Y, Chen L, et al. (2021). Short-term perfluorooctane sulfonate exposure impairs Leydig cell regeneration in the adult rat testis via targeting hedgehog signalling. *Ecotoxicol Environ Saf.* 214:112121. doi:10.1016/j.ecoenv.2021.112121 PMID:33721578
- Mao Z, Xia W, Wang J, Chen T, Zeng Q, Xu B, et al. (2013). Perfluorooctane sulfonate induces apoptosis in lung cancer A549 cells through reactive oxygen species-mediated mitochondrion-dependent pathway. *J Appl Toxicol*. 33(11):1268–76. <u>https://analyticalsciencejournals.onlinelibrary.wiley.com/doi/epdf/10.1002/jat.2785</u> PMID:22976841
- Mario T, Yvonne D, Veronica S, Alejandro D, Juan RM, Diana F, et al. (2022). Effects of perfluorooctanoic acid in oxidative stress generation, DNA damage in cumulus cells, and its impact on in vitro maturation of porcine oocytes. *Environ Toxicol.* 37(6):1394–403. doi:10.1002/ tox.23492 PMID:35187785
- Marques E, Pfohl M, Wei W, Tarantola G, Ford L, Amaeze O, et al. (2022). Replacement per- and polyfluoroalkyl substances (PFAS) are potent modulators of lipogenic and drug metabolizing gene expression signatures in primary human hepatocytes. *Toxicol Appl Pharmacol.* 442:115991. doi:10.1016/j.taap.2022.115991 PMID:35337807

- Martin MT, Brennan RJ, Hu W, Ayanoglu E, Lau C, Ren H, et al. (2007). Toxicogenomic study of triazole fungicides and perfluoroalkyl acids in rat livers predicts toxicity and categorizes chemicals based on mechanisms of toxicity. *Toxicol Sci.* 97(2):595–613. doi:<u>10.1093/toxsci/ kfm065</u> PMID:<u>17383973</u>
- Mashayekhi V, Tehrani KH, Hashemzaei M, Tabrizian K, Shahraki J, Hosseini MJ (2015). Mechanistic approach for the toxic effects of perfluorooctanoic acid on isolated rat liver and brain mitochondria. *Hum Exp Toxicol.* 34(10):985–96. doi:<u>10.1177/0960327114565492</u> PMID:<u>25586001</u>
- Masi M, Maddalon A, Iulini M, Linciano P, Galbiati V, Marinovich M, et al. (2022). Effects of endocrine disrupting chemicals on the expression of RACK1 and LPS-induced THP-1 cell activation. *Toxicology.* 480:153321. doi:<u>10.1016/j.tox.2022.153321</u> PMID:<u>36113621</u>
- Maso L, Trande M, Liberi S, Moro G, Daems E, Linciano S, et al. (2021). Unveiling the binding mode of perfluorooctanoic acid to human serum albumin. *Protein Sci*. 30(4):830–41. doi:10.1002/pro.4036 PMID:33550662
- Matkowskyj KA, Bai H, Liao J, Zhang W, Li H, Rao S, et al. (2014). Aldoketoreductase family 1B10 (AKR1B10) as a biomarker to distinguish hepatocellular carcinoma from benign liver lesions. *Hum Pathol.* 45(4):834–43. doi:10.1016/j.humpath.2013.12.002 PMID:24656094
- Matrix Science (2023). Mascot search engine. Boston (MA), USA: Matrix Science. Available from: <u>http://www.matrixscience.com</u>.
- Max Planck Institute of Biochemistry (2023). MaxQuant software v1.6.10.43. Martinsried, Germany: Max Planck Institute of Biochemistry. Available from: <u>http://maxquant.org/</u>.
- Mazumder N-U-S, Hossain MT, Jahura FT, Girase A, Hall AS, Lu J, et al. (2023). Firefighters' exposure to per-and polyfluoroalkyl substances (PFAS) as an occupational hazard: a review. *Front Mater.* 10:1143411. doi:<u>10.3389/</u> <u>fmats.2023.1143411</u> PMID:<u>38074949</u>
- McCoy JA, Bangma JT, Reiner JL, Bowden JA, Schnorr J, Slowey M, et al. (2017). Associations between perfluorinated alkyl acids in blood and ovarian follicular fluid and ovarian function in women undergoing assisted reproductive treatment. *Sci Total Environ*. 605–606:9– 17. doi:10.1016/j.scitotenv.2017.06.137 PMID:28651210
- McDonough CA, Ward C, Hu Q, Vance S, Higgins CP, DeWitt JC (2020). Immunotoxicity of an electrochemically fluorinated aqueous film-forming foam. *Toxicol Sci.* 178(1):104–14. doi:<u>10.1093/toxsci/kfaa138</u> PMID:<u>32898269</u>
- Messina P, Prieto G, Dodero V, Ruso JM, Schulz P, Sarmiento F (2005a). Ultraviolet-circular dichroism spectroscopy and potentiometric study of the interaction between human serum albumin and sodium perfluorooctanoate. *Biopolymers*. 79(6):300–9. doi:10.1002/bip.20353 PMID:16113999

- Messina PV, Prieto G, Ruso JM, Sarmiento F (2005b). Conformational changes in human serum albumin induced by sodium perfluorooctanoate in aqueous solutions. *JPhys Chem B*. 109(32):15566–73. doi:10.1021/ jp051655y PMID:16852974
- Midgett K, Peden-Adams MM, Gilkeson GS, Kamen DL (2015). In vitro evaluation of the effects of perfluorooctanesulfonic acid (PFOS) and perfluorooctanoic acid (PFOA) on IL-2 production in human T-cells. *J Appl Toxicol.* 35(5):459–65. doi:10.1002/jat.3037 PMID:25056757
- Midic U, Goheen B, Vincent KA, VandeVoort CA, Latham KE (2018). Changes in gene expression following longterm in vitro exposure of *Macaca mulatta* trophoblast stem cells to biologically relevant levels of endocrine disruptors. *Reprod Toxicol.* 77:154–65. doi:10.1016/j. reprotox.2018.02.012 PMID:29505797
- Minata M, Harada KH, Kärrman A, Hitomi T, Hirosawa M, Murata M, et al. (2010). Role of peroxisome proliferator-activated receptor-alpha in hepatobiliary injury induced by ammonium perfluorooctanoate in mouse liver. *Ind Health.* 48(1):96–107. doi:10.2486/indhealth.48.96 PMID:20160413
- Mitro SD, Sagiv SK, Fleisch AF, Jaacks LM, Williams PL, Rifas-Shiman SL, et al. (2020). Pregnancy per- and polyfluoroalkyl substance concentrations and postpartum health in Project Viva: a prospective cohort. *J Clin Endocrinol Metab.* 105(9):e3415–26. doi:<u>10.1210/ clinem/dgaa431</u> PMID:<u>32620010</u>
- Miura R, Araki A, Miyashita C, Kobayashi S, Kobayashi S, Wang SL, et al. (2018). An epigenome-wide study of cord blood DNA methylations in relation to prenatal perfluoroalkylsubstance exposure: the Hokkaido study. *Environ Int.* 115:21–8. doi:<u>10.1016/j.envint.2018.03.004</u> PMID:<u>29544137</u>
- Modaresi SMS, Wei W, Emily M, DaSilva NA, Slitt AL (2022). Per- and polyfluoroalkyl substances (PFAS) augment adipogenesis and shift the proteome in murine 3T3–L1 adipocytes. *Toxicology*. 465:153044. doi:<u>10.1016/j.tox.2021.153044</u> PMID:<u>34800597</u>
- Mogensen UB, Grandjean P, Heilmann C, Nielsen F, Weihe P, Budtz-Jørgensen E (2015b). Structural equation modeling of immunotoxicity associated with exposure to perfluorinated alkylates. *Environ Health*. 14(1):47. doi:<u>10.1186/s12940-015-0032-9</u> PMID:<u>26041029</u>
- Mogensen UB, Grandjean P, Nielsen F, Weihe P, Budtz-Jørgensen E (2015a). Breastfeeding as an exposure pathway for perfluorinated alkylates. *Environ Sci Technol.* 49(17):10466–73. doi:<u>10.1021/acs.est.5b02237</u> PMID:<u>26291735</u>
- Mollenhauer MA, Bradshaw SG, Fair PA, McGuinn WD, Peden-Adams MM (2011). Effects of perfluorooctane sulfonate (PFOS) exposure on markers of inflammation in female B6C3F₁ mice. *J Environ Sci Health Part A Tox Hazard Subst Environ Eng.* 46(2):97–108. doi:10. 1080/10934529.2011.532418 PMID:21170772

- Mondal D, Weldon RH, Armstrong BG, Gibson LJ, Lopez-Espinosa M-J, Shin H-M, et al. (2014). Breastfeeding: a potential excretion route for mothers and implications for infant exposure to perfluoroalkyl acids. *Environ Health Perspect*. 122(2):187–92. doi:10.1289/ ehp.1306613 PMID:24280536
- Morimura K, Cheung C, Ward JM, Reddy JK, Gonzalez FJ (2006). Differential susceptibility of mice humanized for peroxisome proliferator-activated receptor alpha to Wy-14,643-induced liver tumorigenesis. *Carcinogenesis*. 27(5):1074–80. doi:<u>10.1093/carcin/</u> bgi329 PMID:<u>16377806</u>
- Mortensen AS, Letcher RJ, Cangialosi MV, Chu S, Arukwe A (2011). Tissue bioaccumulation patterns, xenobiotic biotransformation and steroid hormone levels in Atlantic salmon (*Salmo salar*) fed a diet containing perfluoroactane sulfonic or perfluorooctane carboxylic acids. *Chemosphere*. 83(8):1035–44. doi:10.1016/j. chemosphere.2011.01.067 PMID:21354591
- Murase W, Kubota A, Ikeda-Araki A, Terasaki M, Nakagawa K, Shizu R, et al. (2023). Effects of perfluorooctanoic acid (PFOA) on gene expression profiles via nuclear receptors in HepaRG cells: comparative study with in vitro transactivation assays. *Toxicology*. 494:153577. doi:10.1016/j.tox.2023.153577 PMID:37302725
- Naderi M, Seyedabadi M, Amiri FT, Mohammadi E, Akbari S, Shaki F (2023). Taurine protects against perfluorooctanoic acid-induced hepatotoxicity via inhibition of oxidative stress, inflammatory, and apoptotic pathways. *Toxicol Res (Camb)*. 12(1):124–32. doi:10.1093/toxres/tfad005 PMID:36866213
- Nakagawa H, Hirata T, Terada T, Jutabha P, Miura D, Harada KH, et al. (2008). Roles of organic anion transporters in the renal excretion of perfluorooctanoic acid. *Basic Clin Pharmacol Toxicol*. 103(1):1–8. doi:10.1111/ j.1742-7843.2007.00155.x PMID:18373647
- Nakagawa H, Terada T, Harada KH, Hitomi T, Inoue K, Inui K, et al. (2009). Human organic anion transporter hOAT4 is a transporter of perfluorooctanoic acid. *Basic Clin Pharmacol Toxicol*. 105(2):136–8. doi:<u>10.1111/</u> j.1742-7843.2009.00409.x PMID:<u>19371258</u>
- Nakagawa T, Ramdhan DH, Tanaka N, Naito H, Tamada H, Ito Y, et al. (2012). Modulation of ammonium perfluorooctanoate-induced hepatic damage by genetically different PPARα in mice. *Arch Toxicol.* 86(1):63– 74. doi:10.1007/s00204-011-0704-3 PMID:21499893
- Nakamura R, Takeuchi R, Kuramochi K, Mizushina Y, Ishimaru C, Takakusagi Y, et al. (2007). Chemical properties of fatty acid derivatives as inhibitors of DNA polymerases. *Org Biomol Chem.* 5(24):3912–21. doi:<u>10.1039/b710944j</u> PMID:<u>18043794</u>
- Nakamura T, Ito Y, Yanagiba Y, Ramdhan DH, Kono Y, Naito H, et al. (2009). Microgram-order ammonium perfluorooctanoate may activate mouse peroxisome proliferator-activated receptor α , but not human

PPARa. *Toxicology*. 265(1–2):27–33. doi:<u>10.1016/j</u>. tox.2009.09.004 PMID:<u>19751795</u>

- National Center for Health Statistics (2023). NHANES 2007–2008 Laboratory Data Overview. USA: National Center for Health Statistics. Available from: <u>https://wwwn.cdc.gov/nchs/nhanes/continuousnhanes/overviewlab.aspx?BeginYear=2007</u>.
- NCBI (2023). ACOX1 acyl-CoA oxidase 1 [*Homo sapiens* (human)]. Bethesda (MD), USA: National Library of Medicine, National Center for Biotechnology Information. Available from: <u>https://www.ncbi.nlm.nih.gov/gene/51</u>.
- Nielsen G, Heiger-Bernays WJ, Schlezinger JJ, Webster TF (2022). Predicting the effects of per- and poly-fluoroalkyl substance mixtures on peroxisome proliferator-activated receptor alpha activity in vitro. *Toxicology.* 465:153024. doi:10.1016/j.tox.2021.153024 PMID:34743024
- Nilsson S, Thompson J, Mueller JF, Bräunig J (2022). Apparent half-lives of chlorinated-perfluorooctane sulfonate and perfluorooctane sulfonate isomers in aviation firefighters. *Environ Sci Technol*. 56(23):17052– 60. doi:<u>10.1021/acs.est.2c04637</u> PMID:<u>36367310</u>
- Nishimura Y, Moriya K, Kobayashi S, Ikeda-Araki A, Sata F, Mitsui T, et al. (2022). Association of exposure to prenatal perfluoroalkyl substances and estrogen receptor 1 polymorphisms with the second to fourth digit ratio in school-aged children: the Hokkaido study. *Reprod Toxicol.* 109:10–8. doi:10.1016/j. reprotox.2022.02.002 PMID:35217143
- NIST (2023). Automated mass spectral deconvolution and identification system for GC/MS Mass Spectrometry Data Center. AMDIS 2.71 database. Gaithersburg (MD), USA: National Institute of Standards and Technology. Available from: https://chemdata.nist.gov/.
- Nøst TH, Vestergren R, Berg V, Nieboer E, Odland JØ, Sandanger TM (2014). Repeated measurements of perand polyfluoroalkyl substances (PFASs) from 1979 to 2007 in males from northern Norway: assessing time trends, compound correlations and relations to age/birth cohort. *Environ Int.* 67:43–53. doi:10.1016/j. envint.2014.02.011 PMID:24657493
- NTP (2019). Toxicity studies of perfluoroalkyl sulfonates administered by gavage to Sprague Dawley (Hsd:Sprague Dawley SD) rats (revised). *Toxic Rep Ser.* (96):NTP-TOX-96. PMID:<u>33533754</u>
- NTP (2020). NTP technical report on the toxicology and carcinogenesis studies of perfluorooctanoic acid (CASRN 335-67-1) administered in feed to Sprague-Dawley (Hsd:Sprague-Dawley[®] SD[®]) rats (revised). *Natl Toxicol Program Tech Rep Ser.* (598). Revised February 2023. <u>https://www.ncbi.nlm.nih.gov/books/ NBK560147/</u>
- Oda Y, Nakayama S, Harada KH, Koizumi A (2007). Negative results of *umu* genotoxicity test of fluorotelomer alcohols and perfluorinated alkyl acids. *Environ*

Health Prev Med. 12(5):217–9. doi:<u>10.1265/ehpm.12.217</u> PMID:<u>21432084</u>

- OECD (2002). Co-operation on existing chemicals hazard of perfluorooctane sulfonate (PFOS) and its salts. Report ENV/JM/RD(2002)17/FINAL. JT00135607. Paris, France: Organisation for Economic Cooperation and Development. Available from: <u>https://www.regulations.gov/document/EPA-HQ-OPPT-2005-0015-0016</u>, accessed January 2025.
- Ojo AF, Peng C, Ng JC (2022a). Genotoxicity assessment of per- and polyfluoroalkyl substances mixtures in human liver cells (HepG2). *Toxicology*. 482:153359. doi:<u>10.1016/j.tox.2022.153359</u> PMID:<u>36341878</u>
- Ojo AF, Peng C, Ng JC (2022b). Combined effects of mixed per- and polyfluoroalkyl substances on the Nrf2–ARE pathway in ARE reporter–HepG2 cells. *J Hazard Mater.* 421:126827. doi:<u>10.1016/j.jhazmat.2021.126827</u> PMID:<u>34388917</u>
- Ojo AF, Xia Q, Peng C, Ng JC (2021). Evaluation of the individual and combined toxicity of perfluoroalkyl substances to human liver cells using biomarkers of oxidativestress. *Chemosphere*.281:130808.doi:10.1016/j. chemosphere.2021.130808 PMID:34022600
- Okada E, Sasaki S, Saijo Y, Washino N, Miyashita C, Kobayashi S, et al. (2012). Prenatal exposure to perfluorinated chemicals and relationship with allergies and infectious diseases in infants. *Environ Res.* 112:118–25. doi:10.1016/j.envres.2011.10.003 PMID:22030285
- Olsen GW, Burris JM, Burlew MM, Mandel JH (2003b). Epidemiologicassessmentofworkerserumperfluorooctanesulfonate (PFOS) and perfluorooctanoate (PFOA) concentrations and medical surveillance examinations. *J Occup Environ Med.* 45(3):260–70. doi:10.1097/01. jom.0000052958.59271.10 PMID:12661183
- Olsen GW, Burris JM, Ehresman DJ, Froehlich JW, Seacat AM, Butenhoff JL, et al. (2007). Half-life of serum elimination of perfluorooctanesulfonate, perfluorohexanesulfonate, and perfluorooctanoate in retired fluorochemical production workers. *Environ Health Perspect.* 115(9):1298–305. doi:10.1289/ehp.10009 PMID:17805419
- Olsen GW, Hansen KJ, Stevenson LA, Burris JM, Mandel JH (2003a). Human donor liver and serum concentrations of perfluorooctanesulfonate and other perfluorochemicals. *Environ Sci Technol*. 37(5):888–91. doi:10.1021/es020955c PMID:12666917
- Olsen GW, Lange CC, Ellefson ME, Mair DC, Church TR, Goldberg CL, et al. (2012). Temporal trends of perfluoroalkyl concentrations in American Red Cross adult blood donors, 2000–2010. *Environ Sci Technol.* 46(11):6330–8. doi:10.1021/es300604p PMID:22554481
- Olsen GW, Zobel LR (2007). Assessment of lipid, hepatic, and thyroid parameters with serum perfluorooctanoate (PFOA) concentrations in fluorochemical production workers. *Int Arch Occup Environ Health*. 81(2):231–46. doi:10.1007/s00420-007-0213-0 PMID:17605032

- Omoike OE, Pack RP, Mamudu HM, Liu Y, Strasser S, Zheng S, et al. (2021). Association between per and polyfluoroalkyl substances and markers of inflammation and oxidative stress. *Environ Res.* 196:110361. doi:10.1016/j.envres.2020.110361 PMID:33131681
- Ophaug RH, Singer L (1980). Metabolic handling of perfluorooctanoic acid in rats. *Proc Soc Exp Biol Med.* 163(1):19–23. doi:<u>10.3181/00379727-163-40715</u> PMID:<u>7352143</u>
- Orbach SM, Ehrich MF, Rajagopalan P (2018). Highthroughput toxicity testing of chemicals and mixtures in organotypic multi-cellular cultures of primary human hepatic cells. *Toxicol In Vitro*. 51:83–94. doi:<u>10.1016/j.tiv.2018.05.006</u> PMID:<u>29751030</u>
- Oseguera-López I, Pérez-Cerezales S, Ortiz-Sánchez PB, Mondragon-Payne O, Sánchez-Sánchez R, Jiménez-Morales I, et al. (2020). Perfluorooctane sulfonate (PFOS) and perfluorohexane sulfonate (PFHxS) alters protein phosphorylation, increase ROS levels and DNA fragmentation during in vitro capacitation of boar spermatozoa. *Animals (Basel)*. 10(10):1934. doi:10.3390/ ani10101934 PMID:33096732
- Osemwegie O, Butler L, Subbiah S, Smith E (2021). Effects of in vitro exposure of perfluorooctanoic acid and monocrotophos on astroglia SVG p12 cells. *J Appl Toxicol*. 41(9):1380–9. doi:10.1002/jat.4129 PMID:33569802
- Ouidir M, Mendola P, Buck Louis GM, Kannan K, Zhang C, Tekola-Ayele F (2020). Concentrations of persistent organic pollutants in maternal plasma and epigenome-wide placental DNA methylation. *Clin Epigenetics*. 12(1):103. doi:<u>10.1186/s13148-020-00894-6</u> PMID:<u>32653021</u>
- Owumi S, Bello T, Oyelere AK (2021a). *N*-acetyl cysteine abates hepatorenal toxicities induced by perfluorooctanoic acid exposure in male rats. *Environ Toxicol Pharmacol.* 86:103667. doi:<u>10.1016/j.etap.2021.103667</u> PMID:<u>33933708</u>
- Owumi SE, Akomolafe AP, Imosemi IO, Odunola OA, Oyelere AK (2021b). *N*-acetyl cysteine co-treatment abates perfluorooctanoic acid-induced reproductive toxicity in male rats. *Andrologia*. 53(5):e14037. doi:<u>10.1111/and.14037</u> PMID:<u>33724529</u>
- Pan D, Shao Y, Song Y, Huang D, Liu S, Zeng X, et al. (2022). Association between maternal per- and polyfluoroalkyl substance exposure and newborn telomere length: effect modification by birth seasons. *Environ Int.* 161:107125. doi:10.1016/j.envint.2022.107125
 PMID:35183942
- Pan X, Qin P, Liu R, Yu W, Dong X (2018). Effects of carbon chain length on the perfluoroalkyl acids-induced oxidative stress of erythrocytes in vitro. J Agric Food Chem. 66(25):6414–20. doi:<u>10.1021/acs.jafc.8b02197</u> PMID:<u>29860827</u>
- Pan Y, Cui Q, Wang J, Sheng N, Jing J, Yao B, et al. (2019). Profiles of emerging and legacy per-/polyfluoroalkyl substances in matched serum and semen samples:

new implications for human semen quality. *Environ Health Perspect*. 127(12):127005. doi:<u>10.1289/EHP4431</u> PMID:<u>31841032</u>

- Panaretakis T, Shabalina IG, Grandér D, Shoshan MC, DePierre JW (2001). Reactive oxygen species and mitochondria mediate the induction of apoptosis in human hepatoma HepG2 cells by the rodent peroxisome proliferator and hepatocarcinogen, perfluorooctanoic acid. *Toxicol Appl Pharmacol.* 173(1):56–64. doi:10.1006/ taap.2001.9159 PMID:11350215
- Papadopoulou E, Stratakis N, Basagaña X, Brantsæter AL, Casas M, Fossati S, et al. (2021). Prenatal and postnatal exposure to PFAS and cardiometabolic factors and inflammation status in children from six European cohorts. *Environ Int.* 157:106853. doi:10.1016/j. envint.2021.106853 PMID:34500361
- Park MH, Gutiérrez-García AK, Choudhury M (2019). Mono-(2-ethylhexyl) phthalate aggravates inflammatory response via sirtuin regulation and inflammasome activation in RAW 264.7 cells. *Chem Res Toxicol*. 32(5):935-42. doi:<u>10.1021/acs.chemrestox.9b00101</u> PMID:<u>31016965</u>
- Park SJ, Sim KH, Shrestha P, Yang JH, Lee YJ (2021). Perfluorooctane sulfonate and bisphenol A induce a similar level of mast cell activation via a common signalling pathway, Fyn-Lyn-Syk activation. *Food Chem Toxicol*. 156:112478. doi:10.1016/j.fct.2021.112478 PMID:34363875
- Parolini M, Colombo G, Valsecchi S, Mazzoni M, Possenti CD, Caprioli M, et al. (2016). Potential toxicity of environmentally relevant perfluorooctane sulfonate (PFOS) concentrations to yellow-legged gull *Larus michahellis* embryos. *Environ Sci Pollut Res Int*. 23(1):426–37. doi:10.1007/s11356-015-5248-2 PMID:26310703
- Peden-Adams MM, Keller JM, Eudaly JG, Berger J, Gilkeson GS, Keil DE (2008). Suppression of humoral immunity in mice following exposure to perfluorooctane sulfonate. *Toxicol Sci.* 104(1):144–54. doi:10.1093/ toxsci/kfn059 PMID:18359764
- Perera NLD, Betancourt J, Miksovska J, O'Shea KE (2023). Detail study on the interaction between perfluorooctanoic acid (PFOA) with human hemoglobin (Hb). *Curr Res Toxicol.* 5:100130. doi:10.1016/j. crtox.2023.100130 PMID:37822784
- Pérez F, Nadal M, Navarro-Ortega A, Fàbrega F, Domingo JL, Barceló D, et al. (2013). Accumulation of perfluoroalkyl substances in human tissues. *Environ Int.* 59:354– 62. doi:<u>10.1016/j.envint.2013.06.004</u> PMID:<u>23892228</u>
- Perkins RG, Butenhoff JL, Kennedy GL Jr, Palazzolo MJ (2004). 13-week dietary toxicity study of ammonium perfluorooctanoate (APFO) in male rats. *Drug Chem Toxicol*. 27(4):361–78. doi:<u>10.1081/DCT-200039773</u> PMID:<u>15573472</u>
- Peropadre A, Fernández Freire P, Hazen MJ (2018). A moderate exposure to perfluorooctanoic acid causes persistent DNA damage and senescence in human

epidermal HaCaT keratinocytes. *Food Chem Toxicol*. 121:351–9.doi:<u>10.1016/j.fct.2018.09.020</u>PMID:<u>30217483</u>

- Petersen KU, Hærvig KK, Flachs EM, Bonde JP, Lindh C, Hougaard KS, et al. (2022). Per- and polyfluoroalkyl substances (PFAS) and male reproductive function in young adulthood; a cross-sectional study. *Environ Res.* 212(Pt A):113157. doi:10.1016/j.envres.2022.113157 PMID:35318009
- Petersen KU, Larsen JR, Deen L, Flachs EM, Hærvig KK, Hull SD, et al. (2020). Per- and polyfluoroalkyl substances and male reproductive health: a systematic review of the epidemiological evidence. *J Toxicol Environ Health B Crit Rev.* 23(6):276–91. doi:10.1080/1 0937404.2020.1798315 PMID:32741292
- Petersen MS, Halling J, Jørgensen N, Nielsen F, Grandjean P, Jensen TK, et al. (2018). Reproductive function in a population of young Faroese men with elevated exposure to polychlorinated biphenyls (PCBs) and perfluorinated alkylate substances (PFAS). *Int J Environ Res Public Health.* 15(9):1880. doi:10.3390/ijerph15091880 PMID:30200252
- Petro EML, D'Hollander W, Covaci A, Bervoets L, Fransen E, De Neubourg D, et al. (2014). Perfluoroalkyl acid contamination of follicular fluid and its consequence for in vitro oocyte developmental competence. *Sci Total Environ*. 496:282–8. doi:<u>10.1016/j.scitotenv.2014.07.028</u> PMID:<u>25089690</u>
- Pierozan P, Cattani D, Karlsson O (2020). Perfluorooctane sulfonate (PFOS) and perfluorooctanoic acid (PFOA) induce epigenetic alterations and promote human breast cell carcinogenesis in vitro. *Arch Toxicol*. 94(11):3893–906. doi:10.1007/s00204-020-02848-6 PMID:32700164
- Pierozan P, Karlsson O (2018). PFOS induces proliferation, cell-cycle progression, and malignant phenotype in human breast epithelial cells. *Arch Toxicol*. 92(2):705– 16. doi:10.1007/s00204-017-2077-8 PMID:29063134
- Pierpont TM, Limper C, Elmore J, Redko A, Anannya O, Imbiakha B, et al. (2023). Effects of PFOS and cyclophosphamide exposure on immune homeostasis in mice. *Immunobiology*. 228(3):152356. doi:<u>10.1016/j. imbio.2023.152356</u> PMID:<u>36857909</u>
- Pilkerton CS, Hobbs GR, Lilly C, Knox SS (2018). Rubella immunity and serum perfluoroalkyl substances: sex and analytic strategy. *PLoS One*. 13(9):e0203330. doi:<u>10.1371/journal.pone.0203330</u> PMID:<u>30248109</u>
- Pinney SM, Fassler CS, Windham GC, Herrick RL, Xie C, Kushi LH, et al. (2023). Exposure to perfluoroalkyl substances and associations with pubertal onset and serum reproductive hormones in a longitudinal study of young girls in Greater Cincinnati and the San Francisco Bay Area. *Environ Health Perspect*. 131(9):97009. doi:10.1289/EHP11811 PMID:37751325
- Pirali B, Negri S, Chytiris S, Perissi A, Villani L, La Manna L, et al. (2009). Perfluorooctane sulfonate and perfluorooctanoic acid in surgical thyroid specimens of

patients with thyroid diseases. *Thyroid*. 19(12):1407–12. doi:<u>10.1089/thy.2009.0174</u> PMID:<u>20001722</u>

- Pizzurro DM, Seeley M, Kerper LE, Beck BD (2019). Interspecies differences in perfluoroalkyl substances (PFAS) toxicokinetics and application to healthbased criteria. *Regul Toxicol Pharmacol*. 106:239–50. doi:<u>10.1016/j.yrtph.2019.05.008</u> PMID:<u>31078680</u>
- Poothong S, Thomsen C, Padilla-Sanchez JA, Papadopoulou E, Haug LS (2017). Distribution of novel and well-known poly-and perfluoroalkyl substances (PFASs) in human serum, plasma, and whole blood. *Environ Sci Technol*. 51(22):13388–96. doi:<u>10.1021/acs.</u> est.7b03299 PMID:<u>29056041</u>
- Porter AK, Kleinschmidt SE, Andres KL, Reusch CN, Krisko RM, Taiwo OA, et al. (2022). Antibody response to COVID-19 vaccines among workers with a wide range of exposure to per- and polyfluoroalkyl substances. *Environ Int.* 169:107537. doi:10.1016/j. envint.2022.107537 PMID:36183490
- Poteser M, Hutter H-P, Moshammer H, Weitensfelder L (2020). Perfluoroctanoic acid (PFOA) enhances NOTCH-signalling in an angiogenesis model of placental trophoblast cells. *Int J Hyg Environ Health*. 229(August):113566. doi:10.1016/j.ijheh.2020.113566 PMID:32485599
- Preston EV, Webster TF, Oken E, Claus Henn B, McClean MD, Rifas-Shiman SL, et al. (2018). Maternal plasma per- and polyfluoroalkyl substance concentrations in early pregnancy and maternal and neonatal thyroid function in a prospective birth cohort: Project Viva (USA). *Environ Health Perspect.* 126(2):027013. doi:10.1289/EHP2534 PMID:29488882
- Qazi MR, Hassan M, Nelson BD, Depierre JW, Abedi-Valugerdi M (2013). Sub-acute, moderate-dose, but not short-term, low-dose dietary pre-exposure of mice to perfluorooctanoate aggravates concanavalin A-induced hepatitis. *Toxicol Lett.* 219(1):1–7. doi:10.1016/j.toxlet.2013.02.017 PMID:23458897
- Qazi MR, Nelson BD, Depierre JW, Abedi-Valugerdi M (2010). 28-Day dietary exposure of mice to a low total dose (7 mg/kg) of perfluorooctanesulfonate (PFOS) alters neither the cellular compositions of the thymus and spleen nor humoral immune responses: does the route of administration play a pivotal role in PFOSinduced immunotoxicity? *Toxicology*. 267(1–3):132–9. doi:10.1016/j.tox.2009.10.035 PMID:19900501
- Qazi MR, Xia Z, Bogdanska J, Chang SC, Ehresman DJ, Butenhoff JL, et al. (2009). The atrophy and changes in the cellular compositions of the thymus and spleen observed in mice subjected to short-term exposure to perfluorooctanesulfonate are high-dose phenomena mediated in part by peroxisome proliferator-activated receptor-alpha (PPARalpha). *Toxicology*. 260(1–3):68– 76. doi:10.1016/j.tox.2009.03.009 PMID:19464571

- Qi Q, Niture S, Gadi S, Arthur E, Moore J, Levine KE, et al. (2023). Per- and polyfluoroalkyl substances activate UPR pathway, induce steatosis and fibrosis in liver cells. *Environ Toxicol.* 38(1):225–42. doi:<u>10.1002/tox.23680</u> PMID:<u>36251517</u>
- Qian Y, Ducatman A, Ward R, Leonard S, Bukowski V, Lan Guo N, et al. (2010). Perfluorooctane sulfonate (PFOS) induces reactive oxygen species (ROS) production in human microvascular endothelial cells: role in endothelial permeability. *J Toxicol Environ Health A*. 73(12):819–36. doi:<u>10.1080/15287391003689317</u> PMID:<u>20391123</u>
- Qin C, Lu YX, Borch T, Yang LL, Li YW, Zhao HM, et al. (2022a). Interactions between extracellular DNA and perfluoroalkylacids (PFAAs) decrease the bioavail-ability of PFAAs in pakchoi (*Brassica chinensis* L.). *J Agric Food Chem.* 70(46):14622–32. doi:10.1021/acs.jafc.2c04597 PMID:36375011
- Qin H, Niu Y, Luan H, Li M, Zheng L, Pan Y, et al. (2022c). Effects of legacy and emerging per- and polyfluoroalkyl substances on PPAR $\alpha/\beta/\gamma$ regulation and osteogenic/ adipogenic differentiation. *Environ Int*. 170:107584. doi:10.1016/j.envint.2022.107584 PMID:36265359
- Qin X, Xie G, Wu X, Xu X, Su M, Yang B (2018). Prenatal exposure to perfluorooctanoic acid induces nerve growth factor expression in cerebral cortex cells of mouse offspring. *Environ Sci Pollut Res Int.* 25(19):18914– 20. doi:10.1007/s11356-018-2117-9 PMID:29717428
- Qin Y, Gu T, Ling J, Luo J, Zhao J, Hu B, et al. (2022b). PFOS facilitates liver inflammation and steatosis: an involvement of NLRP3 inflammasome-mediated hepatocyte pyroptosis. *J Appl Toxicol.* 42(5):806–17. doi:10.1002/jat.4258 PMID:34687223
- Qiu L, Wang H, Dong T, Huang J, Li T, Ren H, et al. (2021). Perfluorooctane sulfonate (PFOS) disrupts testosterone biosynthesis via CREB/CRTC2/StAR signalling pathway in Leydig cells. *Toxicology*. 449:152663. doi:10.1016/j.tox.2020.152663 PMID:33359577
- Qiu Z, Qu K, Luan F, Liu Y, Zhu Y, Yuan Y, et al. (2020). Binding specificities of estrogen receptor with perfluorinated compounds: a cross species comparison. *Environ Int.* 134:105284. doi:<u>10.1016/j.envint.2019.105284</u> PMID:<u>31707300</u>
- Qu JH, Lu CC, Xu C, Chen G, Qiu LL, Jiang JK, et al. (2016). Perfluorooctane sulfonate-induced testicular toxicity and differential testicular expression of estrogen receptor in male mice. *Environ Toxicol Pharmacol.* 45:150–7. doi:<u>10.1016/j.etap.2016.05.025</u> PMID:<u>27310206</u>
- Qu W, Yan Y, Gerrish K, Scappini E, Tucker CJ, Dixon D, et al. (2023). Chronic PFOA exposure in vitro causes acquisition of multiple tumor cell characteristics in rat liver cells. *Toxicol In Vitro*. 89:105577. doi:<u>10.1016/j.</u> <u>tiv.2023.105577</u> PMID:<u>36849026</u>

- Quist EM, Filgo AJ, Cummings CA, Kissling GE, Hoenerhoff MJ, Fenton SE (2015). Hepatic mitochondrial alteration in CD-1 mice associated with prenatal exposures to low doses of perfluorooctanoic acid (PFOA). *Toxicol Pathol.* 43(4):546–57. doi:10.1177/0192623314551841 PMID:25326589
- Rakhshandehroo M, Sanderson LM, Matilainen M, Stienstra R, Carlberg C, de Groot PJ, et al. (2007). Comprehensive analysis of PPARalpha-dependent regulation of hepatic lipid metabolism by expression profiling.*PPARRes*.2007:26839.doi:<u>10.1155/2007/26839</u> PMID:<u>18288265</u>
- Rashid F, Ahmad S, Irudayaraj JMK (2020b). Effect of perfluorooctanoic acid on the epigenetic and tight junction genes of the mouse intestine. *Toxics.* 8(3):64. doi:10.3390/toxics8030064 PMID:32872178
- Rashid F, Ramakrishnan A, Fields C, Irudayaraj J (2020a). Acute PFOA exposure promotes epigenomic alterations in mouse kidney tissues. *Toxicol Rep.* 7:125–32. doi:10.1016/j.toxrep.2019.12.010 PMID:31938689
- Raymer JH, Michael LC, Studabaker WB, Olsen GW, Sloan CS, Wilcosky T, et al. (2012). Concentrations of perfluorooctane sulfonate (PFOS) and perfluorooctanoate (PFOA) and their associations with human semen quality measurements. *Reprod Toxicol.* 33(4):419–27. doi:10.1016/j.reprotox.2011.05.024 PMID:21736937
- Reardon AJF, Khodayari Moez E, Dinu I, Goruk S, Field CJ, Kinniburgh DW, et al.; APrON Study (2019). Longitudinal analysis reveals early-pregnancy associations between perfluoroalkyl sulfonates and thyroid hormone status in a Canadian prospective birth cohort. *Environ Int.* 129:389–99. doi:10.1016/j. envint.2019.04.023 PMID:31150980
- Reisfeld B, de Conti A, El Ghissassi F, Benbrahim-Tallaa L, Gwinn W, Grosse Y, et al. (2022). kc-hits: a tool to aid in the evaluation and classification of chemical carcinogens. *Bioinformatics*. 38(10):2961–2. doi:<u>10.1093/</u> <u>bioinformatics/btac189</u> PMID:<u>35561175</u>
- Reistad T, Fonnum F, Mariussen E (2013). Perfluoroalkylated compounds induce cell death and formation of reactive oxygen species in cultured cerebellar granule cells. *Toxicol Lett.* 218(1):56–60. doi:10.1016/j. toxlet.2013.01.006 PMID:23340305
- Remy S, Govarts E, Wens B, De Boever P, Den Hond E, Croes K, et al. (2016). Metabolic targets of endocrine disrupting chemicals assessed by cord blood transcriptome profiling. *Reprod Toxicol*. 65:307–20. doi:<u>10.1016/j.</u> <u>reprotox.2016.08.018</u> PMID:<u>27589886</u>
- Ren XM, Qin WP, Cao LY, Zhang J, Yang Y, Wan B, et al. (2016). Binding interactions of perfluoroalkyl substances with thyroid hormone transport proteins and potential toxicological implications. *Toxicology*. 366–367(367):32–42. doi:10.1016/j.tox.2016.08.011 PMID:27528273

- Ren XM, Zhang YF, Guo LH, Qin ZF, Lv QY, Zhang LY (2015). Structure–activity relations in binding of perfluoroalkyl compounds to human thyroid hormone T3 receptor. Arch Toxicol. 89(2):233–42. doi:<u>10.1007/</u> <u>s00204-014-1258-y</u> PMID:<u>24819616</u>
- Rhee J, Loftfield E, Albanes D, Layne TM, Stolzenberg-Solomon R, Liao LM, et al. (2023). A metabolomic investigation of serum perfluorooctane sulfonate and perfluorooctanoate. *Environ Int.* 180:108198. doi:10.1002/jat.3207 PMID:26178269
- Rigden M, Pelletier G, Poon R, Zhu J, Auray-Blais C, Gagnon R, et al. (2015). Assessment of urinary metabolite excretion after rat acute exposure to perfluorooctanoic acid and other peroxisomal proliferators. *Arch Environ Contam Toxicol*. 68(1):148–58. doi:10.1007/ s00244-014-0058-y PMID:25015730
- Robinson SL, Zeng X, Guan W, Sundaram R, Mendola P, Putnick DL, et al. (2021). Perfluorooctanoic acid (PFOA) or perfluorooctane sulfonate (PFOS) and DNA methylation in newborn dried blood spots in the Upstate KIDS cohort. *Environ Res.* 194:110668. doi:10.1016/j.envres.2020.110668 PMID:33387539
- Rosen MB, Abbott BD, Wolf DC, Corton JC, Wood CR, Schmid JE, et al. (2008a). Gene profiling in the livers of wild-type and PPARalpha-null mice exposed to perfluorooctanoic acid. *Toxicol Pathol.* 36(4):592–607. doi:10.1177/0192623308318208 PMID:18467677
- Rosen MB, Das KP, Rooney J, Abbott B, Lau C, Corton JC (2017). PPARα-independent transcriptional targets of perfluoroalkyl acids revealed by transcript profiling. *Toxicology*. 387:95–107. doi:10.1016/j.tox.2017.05.013 PMID:28558994
- Rosen MB, Das KP, Wood CR, Wolf CJ, Abbott BD, Lau C (2013). Evaluation of perfluoroalkyl acid activity using primary mouse and human hepatocytes. *Toxicology.* 308:129–37. doi:<u>10.1016/j.tox.2013.03.011</u> PMID:<u>23567314</u>
- Rosen MB, Lee JS, Ren H, Vallanat B, Liu J, Waalkes MP, et al. (2008b). Toxicogenomic dissection of the perfluorooctanoic acid transcript profile in mouse liver: evidence for the involvement of nuclear receptors PPAR alpha and CAR. *Toxicol Sci.* 103(1):46–56. doi:10.1093/toxsci/kfn025 PMID:18281256
- Rosen MB, Schmid JE, Das KP, Wood CR, Zehr RD, Lau C (2009). Gene expression profiling in the liver and lung of perfluorooctane sulfonate-exposed mouse fetuses: comparison to changes induced by exposure to perfluorooctanoic acid. *Reprod Toxicol*. 27(3–4):278– 88. doi:10.1016/j.reprotox.2009.01.007 PMID:19429403
- Rosen MB, Schmid JR, Corton JC, Zehr RD, Das KP, Abbott BD, et al. (2010). Gene expression profiling in wild-type and PPARα-null mice exposed to perfluorooctane sulfonate reveals PPARα-independent effects. *PPAR Res.* 2010:794739. doi:10.1155/2010/794739 PMID:20936131

- Rosen MB, Thibodeaux JR, Wood CR, Zehr RD, Schmid JE, Lau C (2007). Gene expression profiling in the lung and liver of PFOA-exposed mouse fetuses. *Toxicology*. 239(1–2):15–33. doi:10.1016/j.tox.2007.06.095 PMID:17681415
- Rosen Vollmar AK, Lin EZ, Nason SL, Santiago K, Johnson CH, Ma X, et al. (2023). Per- and polyfluoroalkyl substances (PFAS) and thyroid hormone measurements in dried blood spots and neonatal characteristics: a pilot study. *J Expo Sci Environ Epidemiol*. 33(5):737–47. doi:10.1038/s41370-023-00603-4 PMID:37730931
- Rosenmai AK, Ahrens L, le Godec T, Lundqvist J, Oskarsson A (2018). Relationship between peroxisome proliferator-activated receptor alpha activity and cellular concentration of 14 perfluoroalkyl substances in HepG2 cells. *JApplToxicol*. 38(2):219–26. doi:<u>10.1002/</u> jat.3515 PMID:<u>28857218</u>
- Rosenmai AK, Nielsen FK, Pedersen M, Hadrup N, Trier X, Christensen JH, et al. (2013). Fluorochemicals used in food packaging inhibit male sex hormone synthesis. *Toxicol Appl Pharmacol.* 266(1):132–42. doi:10.1016/j. taap.2012.10.022 PMID:23142464
- Rosenmai AK, Taxvig C, Svingen T, Trier X, van Vugt-Lussenburg BM, Pedersen M, et al. (2016). Fluorinated alkyl substances and technical mixtures used in food paper-packaging exhibit endocrine-related activity in vitro. *Andrology*. 4(4):662–72. doi:<u>10.1111/andr.12190</u> PMID:<u>27152447</u>
- Rowan-Carroll A, Reardon A, Leingartner K, Gagné R, Williams A, Meier MJ, et al. (2021). High-throughput transcriptomic analysis of human primary hepatocyte spheroids exposed to per- and polyfluoroalkyl substances (PFAS) as a platform for relative potency characterization. *Toxicol Sci.* 181(2):199–214. doi:10.1093/toxsci/kfab039 PMID:33772556
- Ruggiero MJ, Miller H, Idowu JY, Zitzow JD, Chang SC, Hagenbuch B (2021). Perfluoroalkyl carboxylic acids interact with the human bile acid transporter NTCP. *Livers*. 1(4):221–9. doi:<u>10.3390/livers1040017</u> PMID:<u>34738093</u>
- Russell MH, Waterland RL, Wong F (2015). Calculation of chemical elimination half-life from blood with an ongoing exposure source: the example of perfluorooctanoic acid (PFOA). *Chemosphere*. 129:210–6. doi:10.1016/j.chemosphere.2014.07.061 PMID:25149361
- Rylander C, Dumeaux V, Olsen KS, Waaseth M, Sandanger TM, Lund E (2011). Using blood gene signatures for assessing effects of exposure to perfluoroalkyl acids (PFAAs) in humans: the NOWAC postgenome study. *Int J Mol Epidemiol Genet*. 2(3):207–16. <u>https://pmc.ncbi.</u> <u>nlm.nih.gov/articles/PMC3166148/</u> PMID:21915359
- Saejia P, Lirdprapamongkol K, Svasti J, Paricharttanakul NM (2019). Perfluorooctanoic acid enhances invasion of follicular thyroid carcinoma cells through NF-κB and matrix metalloproteinase-2 activation. *Anticancer*

Res. 39(5):2429–35. doi:<u>10.21873/anticanres.13360</u> PMID:<u>31092435</u>

- Sakai G, Hirao-Suzuki M, Koga T, Kobayashi T, Kamishikiryo J, Tanaka M, et al. (2022). Perfluorooctanoic acid (PFOA) as a stimulator of estrogen receptor-negative breast cancer MDA-MB-231 cell aggressiveness: evidence for involvement of fatty acid 2-hydroxylase (FA2H) in the stimulated cell migration. *J Toxicol Sci.* 47(4):159–68. doi:10.2131/jts.47.159 PMID:35370244
- Salgado R, Pereiro N, López-Doval S, Lafuente A (2015). Initial study on the possible mechanisms involved in the effects of high doses of perfluorooctane sulfonate (PFOS) on prolactin secretion. *Food Chem Toxicol*. 83:10–6. doi:10.1016/j.fct.2015.05.013 PMID:26032630
- Salihović S, Dickens AM, Schoultz I, Fart F, Sinisalu L, Lindeman T, et al. (2020a). Simultaneous determination of perfluoroalkyl substances and bile acids in human serum using ultra-high-performance liquid chromatography-tandem mass spectrometry. Anal Bioanal Chem. 412(10):2251–9. doi:10.1007/s00216-019-02263-6 PMID:31760452
- Salihović S, Kärrman A, Lind L, Lind PM, Lindström G, van Bavel B (2015). Perfluoroalkyl substances (PFAS) including structural PFOS isomers in plasma from elderly men and women from Sweden: results from the Prospective Investigation of the Vasculature in Uppsala Seniors (PIVUS). *Environ Int.* 82:21–7. doi:10.1016/j. envint.2015.05.003 PMID:26001496
- Salihović S, Lind L, Larsson A, Lind PM (2020b). Plasma perfluoroalkyls are associated with decreased levels of proteomic inflammatory markers in a cross-sectional study of an elderly population. *Environ Int*. 145:106099. doi:10.1016/j.envint.2020.106099 PMID:32916415
- Salimi A, Nikoosiar Jahromi M, Pourahmad J (2019). Maternal exposure causes mitochondrial dysfunction in brain, liver, and heart of mouse fetus: an explanation for perfluorooctanoic acid induced abortion and developmental toxicity. *Environ Toxicol.* 34(7):878–85. doi:<u>10.1002/tox.22760</u> PMID:<u>31037826</u>
- Sarzo B, Ballesteros V, Iñiguez C, Manzano-Salgado CB, Casas M, Llop S, et al. (2021). Maternal perfluoroalkyl substances, thyroid hormones, and *DIO* genes: a Spanish cross-sectional study. *Environ Sci Technol.* 55(16):11144–54. doi:10.1021/acs.est.1c01452 PMID:34314170
- Satija Lab (2023). Seurat Package. R toolkit for single cell genomics. New York, USA: Satija Lab, New York Genome Center. Available from: <u>https://satijalab.org/ seurat/</u>.
- Savitz DA, Stein CR, Bartell SM, Elston B, Gong J, Shin H-M, et al. (2012). Perfluorooctanoic acid exposure and pregnancy outcome in a highly exposed community. *Epidemiology*. 23(3):386–92. doi:10.1097/ EDE.0b013e31824cb93b PMID:22370857

- Scharmach E, Buhrke T, Lichtenstein D, Lampen A (2012). Perfluorooctanoic acid affects the activity of the hepatocyte nuclear factor 4 alpha (HNF4α). *Toxicol Lett.* 212(2):106–12. doi:10.1016/j.toxlet.2012.05.007 PMID:22609092
- Schlezinger JJ, Hyötyläinen T, Sinioja T, Boston C, Puckett H, Oliver J, et al. (2021). Perfluorooctanoic acid induces liver and serum dyslipidemia in humanized PPARα mice fed an American diet. *Toxicol Appl Pharmacol.* 426:115644. doi:<u>10.1016/j.taap.2021.115644</u> PMID:34252412
- Schlezinger JJ, Puckett H, Oliver J, Nielsen G, Heiger-Bernays W, Webster TF (2020). Perfluorooctanoic acid activates multiple nuclear receptor pathways and skews expression of genes regulating cholesterol homeostasis in liver of humanized PPARα mice fed an American diet. *Toxicol Appl Pharmacol*. 405:115204. doi:<u>10.1016/j.</u> <u>taap.2020.115204</u> PMID:<u>32822737</u>
- Schnellmann RG (1990). The cellular effects of a unique pesticide sulfluramid (*N*-ethylperfluorooctane sulphonamide) on rabbit renal proximal tubules. *Toxicol In Vitro.* 4(1):71–4. doi:10.1016/0887-2333(90)90012-I PMID:20702287
- Scinicariello F, Buser MC, Balluz L, Gehle K, Murray HE, Abadin HG, et al. (2020). Perfluoroalkyl acids, hyperuricemia and gout in adults: analyses of NHANES 2009–2014. *Chemosphere*. 259:127446. doi:10.1016/j. chemosphere.2020.127446 PMID:32590180
- Seacat AM, Thomford PJ, Hansen KJ, Clemen LA, Eldridge SR, Elcombe CR, et al. (2003). Sub-chronic dietary toxicity of potassium perfluorooctanesulfonate in rats. *Toxicology*. 183(1–3):117–31. doi:10.1016/S0300-<u>483X(02)00511-5</u> PMID:12504346
- Seacat AM, Thomford PJ, Hansen KJ, Olsen GW, Case MT, Butenhoff JL (2002). Subchronic toxicity studies on perfluorooctanesulfonate potassium salt in cynomolgus monkeys. *Toxicol Sci.* 68(1):249–64. doi:10.1093/ toxsci/68.1.249 PMID:12075127
- Seals R, Bartell SM, Steenland K (2011). Accumulation and clearance of perfluorooctanoic acid (PFOA) in current and former residents of an exposed community. *Environ Health Perspect*. 119(1):119–24. doi:10.1289/ ehp.1002346 PMID:20870569
- Selano J, Richardson V, Washington J, Mazur C (2019). Characterization of non-radiolabeled thyroxine (T₄) uptake in cryopreserved rat hepatocyte suspensions: pharmacokinetic implications for PFOA and PFOS chemical exposure. *Toxicol In Vitro*. 58:230–8. doi:10.1016/j.tiv.2019.03.022 PMID:30930230
- Shabalina IG, Panaretakis T, Bergstrand A, DePierre JW (1999). Effects of the rodent peroxisome proliferator and hepatocarcinogen, perfluorooctanoic acid, on apoptosis in human hepatoma HepG2 cells. *Carcinogenesis*. 20(12):2237–46. doi:10.1093/carcin/20.12.2237 PMID:10590214

- Shalaby AM, Albakkosh AM, Shalaby RH, Alabiad MA, Elshamy AM, Alorini M, et al. (2023). Lemongrass essential oil attenuates perfluorooctane sulfonate-induced jejunal mucosal injury in rat: a histological, immunohistochemical, and biochemical study. *Microsc Microanal.* 29(2):841–57. doi:10.1093/micmic/ozad009 PMID:37749735
- Shan G, Wang Z, Zhou L, Du P, Luo X, Wu Q, et al. (2016). Impacts of daily intakes on the isomeric profiles of perfluoroalkyl substances (PFASs) in human serum. *Environ Int.* 89–90:62–70. doi:10.1016/j. envint.2016.01.002 PMID:26826363
- Shan G, Ye M, Zhu B, Zhu L (2013). Enhanced cytotoxicity of pentachlorophenol by perfluorooctane sulfonate or perfluorooctanoic acid in HepG2 cells. *Chemosphere*. 93(9):2101–7. doi:10.1016/j.chemosphere.2013.07.054 PMID:23972907
- Shane HL, Baur R, Lukomska E, Weatherly L, Anderson SE (2020). Immunotoxicity and allergenic potential induced by topical application of perfluorooctanoic acid (PFOA) in a murine model. *Food Chem Toxicol.* 136:111114. doi:10.1016/j.fct.2020.111114 PMID:31904477
- Shao X, Ji F, Wang Y, Zhu L, Zhang Z, Du X, et al. (2018). Integrative chemical proteomics-metabolomics approach reveals acaca/acacb as direct molecular targets of PFOA. *Anal Chem.* 90(18):11092–8. doi:10.1021/acs. analchem.8b02995 PMID:30134650
- Shao X, Zhao B, Wang B, Zhao B, Zhu Y, Yuan Z, et al. (2019). Neuroprotective effects of blueberry anthocyanins against perfluorooctanoic sulfonate on planarian *Dugesia japonica*. *Ecotoxicol Environ Saf*. 175:39–47. doi:<u>10.1016/j.ecoenv.2019.03.023</u> PMID:<u>30884343</u>
- Sharma S, Kelly TK, Jones PA (2010). Epigenetics in cancer. *Carcinogenesis*. 31(1):27–36. doi:<u>10.1093/carcin/bgp220</u> PMID:<u>19752007</u>
- Sheng N, Li J, Liu H, Zhang A, Dai J (2016). Interaction of perfluoroalkyl acids with human liver fatty acidbinding protein. *Arch Toxicol*. 90(1):217–27. doi:<u>10.1007/</u> <u>s00204-014-1391-7</u> PMID:<u>25370009</u>
- Shi L, Zheng J, Yan S, Li Y, Wang Y, Liu X, et al. (2020). Exposure to perfluorooctanoic acid induces cognitive deficits via altering gut microbiota composition, impairing intestinal barrier integrity, and causing inflammation in gut and brain. J Agric Food Chem. 68(47):13916–28. doi:10.1021/acs.jafc.0c05834 PMID:33151664
- Shin H-M, Vieira VM, Ryan PB, Steenland K, Bartell SM (2011). Retrospective exposure estimation and predicted versus observed serum perfluorooctanoic acid concentrations for participants in the C8 Health Project. *Environ Health Perspect*. 119(12):1760–5. doi:10.1289/ehp.1103729 PMID:21813367
- Shipley JM, Hurst CH, Tanaka SS, DeRoos FL, Butenhoff JL, Seacat AM, et al. (2004). *trans*-Activation of PPARalpha and induction of PPARalpha

target genes by perfluorooctane-based chemicals. *Toxicol Sci.* 80(1):151–60. doi:<u>10.1093/toxsci/kfh130</u> PMID:<u>15071170</u>

- Shrestha S, Bloom MS, Yucel R, Seegal RF, Wu Q, Kannan K, et al. (2015). Perfluoroalkyl substances and thyroid function in older adults. *Environ Int.* 75:206–14. doi:10.1016/j.envint.2014.11.018 PMID:25483837
- SIB (2024). Cellosaurus L-02 (CVCL_6926). The Cellosaurus [online database]. Geneva, Switzerland: Swiss Institute of Bioinformatics. Available from: <u>https://www.cellosaurus.org/CVCL_6926</u>, accessed January 2025.
- Simmon VF, Marx N (1978). In-vitro microbiological mutagenicity assays of 3M Company compounds T-2247 CoC and T-2248 CoC. Final report. SRI Project LSC-442-016. Prepared by Simmon VF for SRI International, USA. EPA AR 226-0134. US Environmental Protection Agency.
- Singh TS, Lee S, Kim HH, Choi JK, Kim SH (2012). Perfluorooctanoic acid induces mast cell-mediated allergic inflammation by the release of histamine and inflammatory mediators. *Toxicol Lett.* 210(1):64–70. doi:10.1016/j.toxlet.2012.01.014 PMID:22322153
- Sivaram AK, Logeshwaran P, Surapaneni A, Shah K, Crosbie N, Rogers Z, et al. (2021). Evaluation of cyto-genotoxicity of perfluorooctane sulfonate (PFOS) to Allium cepa. Environ Toxicol Chem. 40(3):792–8. doi:10.1002/etc.4905 PMID:33074584
- Slotkin TA, MacKillop EA, Melnick RL, Thayer KA, Seidler FJ (2008). Developmental neurotoxicity of perfluorinated chemicals modeled in vitro. *Environ Health Perspect*. 116(6):716–22. doi:10.1289/ehp.11253 PMID:18560525
- Smith MT, Guyton KZ, Gibbons CF, Fritz JM, Portier CJ, Rusyn I, et al. (2016). Key characteristics of carcinogens as a basis for organizing data on mechanisms of carcinogenesis. *Environ Health Perspect*. 124(6):713–21. doi:10.1289/ehp.1509912 PMID:26600562
- Smutny T, Bernhauerova V, Smutna L, Tebbens JD, Pavek P (2022). Expression dynamics of pregnane X receptor-controlled genes in 3D primary human hepatocyte spheroids. *Arch Toxicol.* 96(1):195–210. doi:10.1007/s00204-021-03177-y PMID:34689256
- Soltani M, Pourfarzam M, Sharifabad AH, Neisiani AK, Mousavi MK, Aliomrani M (2023). Effect of pretreatment with a synbiotic on perfluorooctanoic acid-induced liver damage after sub-acute oral exposure in C57BL/6J mice. *Toxicol Appl Pharmacol*. 459:116360. doi:<u>10.1016/j.taap.2022.116360</u> PMID:<u>36572227</u>
- Son H-Y, Lee S, Tak E-N, Cho H-S, Shin H-I, Kim S-H, et al. (2009). Perfluorooctanoic acid alters T lymphocyte phenotypes and cytokine expression in mice. *Environ Toxicol.* 24(6):580–8. doi:<u>10.1002/tox.20459</u> PMID:<u>19051282</u>

- Song M, Kim Y-J, Park Y-K, Ryu J-C (2012). Changes in thyroid peroxidase activity in response to various chemicals. *J Environ Monit*. 14(8):2121–6. doi:<u>10.1039/ c2em30106g</u> PMID:<u>22699773</u>
- Song P, Li D, Wang X, Zhong X (2018). Effects of perfluorooctanoic acid exposure during pregnancy on the reproduction and development of male offspring mice. *Andrologia*. 50(8):e13059. doi:<u>10.1111/and.13059</u> PMID:<u>29862542</u>
- Sonkar R, Kay MK, Choudhury M (2019). PFOS modulates interactive epigenetic regulation in first-trimester human trophoblast cell line HTR-8/SV_{neo}. *Chem Res Toxicol.* 32(10):2016–27. doi:10.1021/acs. <u>chemrestox.9b00198</u> PMID:31508952
- Sonthithai P, Suriyo T, Thiantanawat A, Watcharasit P, Ruchirawat M, Satayavivad J (2016). Perfluorinated chemicals, PFOS and PFOA, enhance the estrogenic effects of 17β-estradiol in T47D human breast cancer cells. *J Appl Toxicol*. 36(6):790–801. doi:<u>10.1002/jat.3210</u> PMID:<u>26234195</u>
- Sørli JB, Låg M, Ekeren L, Perez-Gil J, Haug LS, Da Silva E, et al. (2020). Per- and polyfluoroalkyl substances (PFASs) modify lung surfactant function and pro-in-flammatory responses in human bronchial epithe-lial cells. *Toxicol In Vitro*. 62:104656. doi:10.1016/j. tiv.2019.104656 PMID:31536757
- Souders CL 2nd, Sanchez CL, Malphurs W, Aristizabal-Henao JJ, Bowden JA, Martyniuk CJ (2021). Metabolic profiling in human SH-SY5Y neuronal cells exposed to perfluorooctanoic acid (PFOA). *Neurotoxicology*. 85:160–72. doi:10.1016/j.neuro.2021.05.009 PMID: 34029635
- Specht IO, Hougaard KS, Spanò M, Bizzaro D, Manicardi GC, Lindh CH, et al. (2012). Sperm DNA integrity in relation to exposure to environmental perfluoroalkyl substances – a study of spouses of pregnant women in three geographical regions. *Reprod Toxicol.* 33(4):577– 83. doi:10.1016/j.reprotox.2012.02.008 PMID:22449571
- Spliethoff HM, Tao L, Shaver SM, Aldous KM, Pass KA, Kannan K, et al. (2008). Use of newborn screening program blood spots for exposure assessment: declining levels of perfluorinated compounds in New York State infants. *Environ Sci Technol*. 42(14):5361–7. doi:10.1021/es8006244 PMID:18754394
- Sprengel J, Behnisch PA, Besselink H, Brouwer A, Vetter W (2021). In vitro human cell-based TTR-TRβ CALUX assay indicates thyroid hormone transport disruption of short-chain, medium-chain, and long-chain chlorinated paraffins. *Arch Toxicol.* 95(4):1391–6. doi:10.1007/ s00204-021-02994-5 PMID:33555371
- Starkov AA, Wallace KB (2002). Structural determinants of fluorochemical-induced mitochondrial dysfunction. *Toxicol Sci.* 66(2):244–52. doi:<u>10.1093/toxsci/66.2.244</u> PMID:<u>11896291</u>

- Starling AP, Liu C, Shen G, Yang IV, Kechris K, Borengasser SJ, et al. (2020). Prenatal exposure to per- and polyfluoroalkyl substances, umbilical cord blood DNA methylation, and cardio-metabolic indicators in newborns: the Healthy Start Study. *Environ Health Perspect*. 128(12):127014. doi:<u>10.1289/EHP6888</u> PMID:<u>33356526</u>
- Steenland K, Fletcher T, Stein CR, Bartell SM, Darrow L, Lopez-Espinosa M-J, et al. (2020). Review: evolution of evidence on PFOA and health following the assessments of the C8 Science Panel. *Environ Int*. 145:106125. doi:10.1016/j.envint.2020.106125 PMID:32950793
- Stein CR, McGovern KJ, Pajak AM, Maglione PJ, Wolff MS (2016). Perfluoroalkyl and polyfluoroalkyl substances and indicators of immune function in children aged 12–19 y: National Health and Nutrition Examination Survey. *Pediatr Res.* 79(2):348–57. doi:10.1038/ pr.2015.213 PMID:26492286
- Steves AN, Turry A, Gill B, Clarkson-Townsend D, Bradner JM, Bachli I, et al. (2018). Per- and polyfluoroalkyl substances impact human spermatogenesis in a stem-cell-derived model. *Syst Biol Reprod Med*. 64(4):225–39. doi:10.1080/19396368.2018.1481465 PMID:29911897
- Stiehl A (1974). Bile salt sulphates in cholestasis. *Eur J Clin Invest*. 4(1):59–63. doi:<u>10.1111/j.1365-2362.1974.</u> <u>tb00373.x</u> PMID:<u>4819836</u>
- Su S, Billy LJ, Chang S, Gonzalez FJ, Patterson AD, Peters JM (2022a). The role of mouse and human peroxisome proliferator-activated receptor-α in modulating the hepatic effects of perfluorooctane sulfonate in mice. *Toxicology*. 465:153056. doi:<u>10.1016/j.tox.2021.153056</u> PMID:<u>34861291</u>
- Su Y, Santucci-Pereira J, Dang NM, Kanefsky J, Rahulkannan V, Hillegass M, et al. (2022b). Effects of pubertal exposure to butyl benzyl phthalate, perfluorooctanoic acid, and zeranol on mammary gland development and tumorigenesis in rats. *Int J Mol Sci.* 23(3):1398. doi:10.3390/ijms23031398 PMID:35163327
- Suh KS, Choi EM, Kim YJ, Hong SM, Park SY, Rhee SY, et al. (2017). Perfluorooctanoic acid induces oxidative damage and mitochondrial dysfunction in pancreatic β -cells. *Mol Med Rep.* 15(6):3871–8. doi:10.3892/mmr.2017.6452 PMID:28440430
- Suo C, Fan Z, Zhou L, Qiu J (2017). Perfluorooctane sulfonate affects intestinal immunity against bacterial infection. *Sci Rep.* 7(1):5166. doi:10.1038/s41598-017-04091-z PMID:28701769
- Taibl KR, Schantz S, Aung MT, Padula A, Geiger S, Smith S, et al. (2022). Associations of per- and polyfluoroalkyl substances (PFAS) and their mixture with oxidative stress biomarkers during pregnancy. *Environ Int.* 169:107541. doi:10.1016/j.envint.2022.107541 PMID:36191484
- Takacs ML, Abbott BD (2007). Activation of mouse and human peroxisome proliferator-activated receptors (alpha, beta/delta, gamma) by perfluorooctanoic acid

and perfluorooctane sulfonate. *Toxicol Sci*. 95(1):108– 17. doi:<u>10.1093/toxsci/kf1135</u> PMID:<u>17047030</u>

- Takagi A, Sai K, Umemura T, Hasegawa R, Kurokawa Y (1991). Short-term exposure to the peroxisome proliferators, perfluorooctanoic acid and perfluorodecanoic acid, causes significant increase of 8-hydroxydeoxyguanosine in liver DNA of rats. *Cancer Lett.* 57(1):55– 60. doi:10.1016/0304-3835(91)90063-N PMID:2025879
- Tan F, Jin Y, Liu W, Quan X, Chen J, Liang Z (2012). Global liver proteome analysis using iTRAQ labeling quantitative proteomic technology to reveal biomarkers in mice exposed to perfluorooctane sulfonate (PFOS). *Environ Sci Technol.* 46(21):12170–7. doi:10.1021/es3027715 PMID:23046066
- Tan Y, Taibl KR, Dunlop AL, Barr DB, Panuwet P, Yakimavets V, et al. (2023). Association between a mixture of per- and polyfluoroalkyl substances (PFAS) and inflammatory biomarkers in the Atlanta African American Maternal–Child Cohort. *Environ Sci Technol.* 57(36):13419–28. doi:10.1021/acs.est.3c04688 PMID:37649345
- Tang L, Yu J, Zhuge S, Chen H, Zhang L, Jiang G (2022). Oxidative stress and Cx43-mediated apoptosis are involved in PFOS-induced nephrotoxicity. *Toxicology.* 478:153283. doi:10.1016/j.tox.2022.153283 PMID:35934180
- Taylor BK, Dadia N, Yang CB, Krishnan S, Badr M (2002). Peroxisome proliferator-activated receptor agonists inhibit inflammatory edema and hyperalgesia. *Inflammation*. 26(3):121–7. doi:10.1023/A:1015500531113 PMID:12083418
- Taylor KD, Woodlief TL, Ahmed A, Hu Q, Duncker PC, DeWitt JC (2023). Quantifying the impact of PFOA exposure on B-cell development and antibody production. *Toxicol Sci.* 194(1):101–8. doi:10.1093/toxsci/ kfad043 PMID:37162486
- Thomas RS, Paules RS, Simeonov A, Fitzpatrick SC, Crofton KM, Casey WM, et al. (2018). The US Federal Tox21 Program: a strategic and operational plan for continued leadership. *Altern Anim Exp.* 35(2):163–8. doi:10.14573/altex.1803011 PMID:29529324
- Thompson J, Lorber M, Toms LL, Kato K, Calafat AM, Mueller JF (2010). Use of simple pharmacokinetic modeling to characterize exposure of Australians to perfluorooctanoic acid and perfluorooctane sulfonic acid. *Environ Int.* 36(4):390–7. doi:<u>10.1016/j.</u> <u>envint.2010.02.008</u> PMID:<u>20236705</u>
- Thottassery J, Winberg L, Youssef J, Cunningham M, Badr M (1992). Regulation of perfluorooctanoic acid-induced peroxisomal enzyme activities and hepatocellular growth by adrenal hormones. *Hepatology*. 15(2):316–22. doi:10.1002/hep.1840150223 PMID:1735536
- Tian M, Peng S, Martin FL, Zhang J, Liu L, Wang Z, et al. (2012). Perfluorooctanoic acid induces gene promoter hypermethylation of glutathione-S-transferase Pi in

human liver L02 cells. *Toxicology*. 296(1–3):48–55. doi:10.1016/j.tox.2012.03.003 PMID:22425687

- Tien PT, Lin HJ, Tsai YY, Lim YP, Chen CS, Chang CY, et al. (2020). Perfluorooctanoic acid in indoor particulate matter triggers oxidative stress and inflammation in corneal and retinal cells. *Sci Rep.* 10(1):15702. doi:10.1038/s41598-020-72600-8 PMID:32973190
- Timmermann CAG, Jensen KJ, Nielsen F, Budtz-Jørgensen E, van der Klis F, Benn CS, et al. (2020). Serum perfluoroalkyl substances, vaccine responses, and morbidity in a cohort of Guinea-Bissau children. *Environ Health Perspect*. 128(8):87002. doi:10.1289/ <u>EHP6517</u> PMID:32772733
- Timmermann CAG, Pedersen HS, Weihe P, Bjerregaard P, Nielsen F, Heilmann C, et al. (2022). Concentrations of tetanus and diphtheria antibodies in vaccinated Greenlandic children aged 7–12 years exposed to marine pollutants, a cross sectional study. *Environ Res.* 203:111712. doi:10.1016/j.envres.2021.111712 PMID:34343554
- Tirona RG, Lee W, Leake BF, Lan L-B, Cline CB, Lamba V, et al. (2003). The orphan nuclear receptor HNF4alpha determines PXR- and CAR-mediated xenobiotic induction of CYP3A4. *Nat Med.* 9(2):220–4. doi:<u>10.1038/</u> <u>nm815</u> PMID:<u>12514743</u>
- Toft G, Jönsson BAG, Bonde JP, Nørgaard-Pedersen B, Hougaard DM, Cohen A, et al. (2016). Perfluorooctane sulfonate concentrations in amniotic fluid, biomarkers of fetal Leydig cell function, and cryptorchidism and hypospadias in Danish boys (1980–1996). *Environ Health Perspect*. 124(1):151–6. doi:10.1289/ehp.1409288 PMID:26046833
- Tojima H, Kakizaki S, Yamazaki Y, Takizawa D, Horiguchi N, Sato K, et al. (2012). Ligand dependent hepatic gene expression profiles of nuclear receptors CAR and PXR. *Toxicol Lett.* 212(3):288–97. doi:<u>10.1016/j.toxlet.2012.06.001</u> PMID:<u>22698814</u>
- Torres L, Redko A, Limper C, Imbiakha B, Chang S, August A (2021). Effect of perfluorooctanesulfonic acid (PFOS) on immune cell development and function in mice. *Immunol Lett.* 233:31–41. doi:<u>10.1016/j.</u> <u>imlet.2021.03.006</u> PMID:<u>33722553</u>
- Tsai M-S, Lin C-Y, Lin C-C, Chen M-H, Hsu SHJ, Chien K-L, et al. (2015). Association between perfluoroalkyl substances and reproductive hormones in adolescents and young adults. *Int J Hyg Environ Health*. 218(5):437– 43. doi:10.1016/j.ijheh.2015.03.008 PMID:25921520
- Tsang H, Cheung T-Y, Kodithuwakku SP, Chai J, Yeung WSB, Wong CKC, et al. (2013). Perfluorooctanoate suppresses spheroid attachment on endometrial epithelial cells through peroxisome proliferator-activated receptor alpha and down-regulation of Wnt signalling. *Reprod Toxicol.* 42(December):164–71. doi:10.1016/j. reprotox.2013.08.001 PMID:23978332

- Tucker DK, Macon MB, Strynar MJ, Dagnino S, Andersen E, Fenton SE (2015). The mammary gland is a sensitive pubertal target in CD-1 and C57Bl/6 mice following perinatal perfluorooctanoic acid (PFOA) exposure. *Reprod Toxicol.* 54:26–36. doi:10.1016/j. reprotox.2014.12.002 PMID:25499722
- Upham BL, Deocampo ND, Wurl B, Trosko JE (1998). Inhibition of gap junctional intercellular communication by perfluorinated fatty acids is dependent on the chain length of the fluorinated tail. *Int J Cancer*. 78(4):491–5. doi:10.1002/(SICI)1097-0215(19981109) 78:4<491::AID-IJC16>3.0.CO;2-9 PMID:9797139
- Upham BL, Park JS, Babica P, Sovadinova I, Rummel AM, Trosko JE, et al. (2009). Structure-activity-dependent regulation of cell communication by perfluorinated fatty acids using in vivo and in vitro model systems. *Environ Health Perspect*. 117(4):545–51. doi:<u>10.1289/ ehp.11728 PMID:19440492</u>
- US EPA (1979). Mutagenicity evaluation of T-2014 CoC in the Ames Salmonella/microsome plate test. Final report. Prepared by Litton Bionetics Inc., USA. Washington (DC), USA: US Environmental Protection Agency.
- US EPA (1995a). Mutagenicity test with T-6342 in the Salmonella–*Escherichia coli*/mammalian-microsome reverse mutation assay. Laboratory No. 17073-0-409. Prepared by Lawlor TE for Corning Hazleton Inc., USA. EPA AR 226–0436. Washington (DC), USA: US Environmental Protection Agency (as cited in SIAR, 2006).
- US EPA (1995b). Mutagenicity test on T-6342 in an in vivo mouse micronucleus assay. Study No. 17073-0-455. Prepared by Murli H for Corning Hazleton Inc., USA. EPA AR 226-0435. Washington (DC), USA: US Environmental Protection Agency.
- US EPA (1996f). Mutagenicity test with T-6564 in the Salmonella–*Escherichia coli*/mammalian-microsome reverse mutation assay with a confirmatory assay. Final report. CHV study No. 17750-0-409R. EPA AR 226-0432. Prepared by Lawlor TE for Corning Hazleton Inc., USA. Washington (DC), USA: US Environmental Protection Agency (as cited in SIAR, 2006).
- US EPA (1996a). Mutagenicity test on T-6342 measuring chromosome aberrations in human whole blood lymphocytes with a confirmatory assay with multiple harvests. Study No. 17073-0-449CO. Prepared by Murli H for Corning Hazleton Inc., USA. EPA AR 26-0433. Washington (DC), USA: US Environmental Protection Agency.
- US EPA (1996b). Mutagenicity test on T-6564 measuring chromosomal aberrations in Chinese hamster ovary (CHO) cells with a confirmatory assay with multiple harvests. Study No. 17750-0-437CO. Prepared by Murli H for Corning Hazleton Inc., USA. EPA AR 226-0431.

Washington (DC), USA: US Environmental Protection Agency.

- US EPA (1996c). Mutagenicity test on T-6342 measuring chromosomal aberrations in Chinese hamster ovary (CHO) cells: with a confirmatory assay with multiple harvests. Study No. 17073-0-437CO. Prepared by Murli H for Corning Hazleton Inc., USA. EPA AR 226-0434. Washington (DC), USA: US Environmental Protection Agency.
- US EPA (1996d). Mutagenicity test on T-6564 in an in vivo mouse micronucleus assay. Study number 17750-0-455. Prepared by Murli H for Corning Hazleton Inc., USA. EPA AR 226-0430. Washington (DC), USA: US Environmental Protection Agency.
- US EPA (1996e). Mutagenicity test on T-6295 in an in vivo mouse micronucleus assay. Final report. CHV Study No. 17403-0-455. Prepared by Murli H for Corning Hazleton Inc., USA. Washington (DC), USA: US Environmental Protection Agency.
- US EPA (1999a). Salmonella–*Escherichia coli*/mammalian-microsome reverse mutation assay with PFOS. Final report. Covance study No. 20784-0-409. Prepared by Mecchi MS for Covance Laboratories, USA. EPA AR 226-0133. Washington (DC), USA: US Environmental Protection Agency. Available from: <u>https://cdn.toxicdocs.org/xj/xjJK9</u> N7gNLR78LzpX4jXYVKOg/xjJK9N7gNLR78 LzpX4jXYVKOg.pdf, accessed January 2025.
- US EPA (1999b). Chromosomal aberrations in human whole blood lymphocytes with PFOS. Final report. Covance study No. 2784-0-499. Prepared by Murli H for Corning Hazleton Inc., USA. EPA-AR-226-0131. US Environmental Protection Agency.
- US EPA (1999c). Unscheduled DNA synthesis in rat liver primary cell cultures with PFOS. Final report. Covance Study No. 207840447. Prepared by Cifone MA for Covance Laboratories, USA. EPA AR 226-0132. Washington (DC), USA: US Environmental Protection Agency.
- US EPA (2002). CHO/HGPRT forward mutation assay ISO (T6.889.7). Report No. 01-7019-G1. Prepared by Sadhu D for Toxicon Corporation, USA. EPA AR 226-1101 (as cited in SIAR, 2006). US Environmental Protection Agency.
- US EPA (2003). Perfluorooctanoic acid: toxicokinetics in the rat. DuPont 7473. AR-226-149. Prepared by Kemper R, Jepson G, for Haskell Laboratory for Health and Environmental Sciences. Washington (DC), USA: United States Environmental Protection Agency.
- US EPA (2021). CompTox Chemicals Dashboard. Washington (DC), USA: US Environmental Protection Agency. Available from: <u>https://www.epa.gov/ chemical-research/comptox-chemicals-dashboard</u>, accessed January 2025.

- US EPA (2023a). Perfluorooctanoic acid. 335-67-1. DTXSID8031865. CompTox Chemicals Dashboard V2.5.2 [online database]. Washington (DC), USA: US Environmental Protection Agency. Available from: <u>https://comptox.epa.gov/dashboard/chemical/details/</u> DTXSID8031865
- US EPA (2023b). Perfluorooctanesulfonic acid. 1763-23-1. DTXSID3031864. CompTox Chemicals Dashboard V2.5.2 [online database]. Washington (DC), USA: US Environmental Protection Agency. Available from: <u>https://comptox.epa.gov/dashboard/chemical/details/</u> DTXSID3031864
- van den Dungen MW, Murk AJ, Kampman E, Steegenga WT, Kok DE (2017a). Association between DNA methylation profiles in leukocytes and serum levels of persistent organic pollutants in Dutch men. *Environ Epigenet*. 3(1):dvx001. doi:<u>10.1093/eep/dvx001</u> PMID:29492303
- van den Dungen MW, Murk AJ, Kok DE, Steegenga WT (2017b). Persistent organic pollutants alter DNA methylation during human adipocyte differentiation. *Toxicol In Vitro*. 40:79–87. doi:<u>10.1016/j.tiv.2016.12.011</u> PMID:<u>28011154</u>
- van den Dungen MW, Rijk JCW, Kampman E, Steegenga WT, Murk AJ (2015). Steroid hormone related effects of marine persistent organic pollutants in human H295R adrenocortical carcinoma cells. *Toxicol In Vitro*. 29(4):769–78. doi:<u>10.1016/j.tiv.2015.03.002</u> PMID:<u>25765474</u>
- van Gerwen M, Alpert N, Alsen M, Ziadkhanpour K, Taioli E, Genden E (2020). The impact of smoking on the association between perfluoroalkyl acids (PFAS) and thyroid hormones: a National Health and Nutrition Examination Survey analysis. *Toxics*. 8(4):116. doi:10.3390/toxics8040116 PMID:33316920
- Vanden Heuvel JP, Kuslikis BI, Peterson RE (1992). Covalent binding of perfluorinated fatty acids to proteins in the plasma, liver and testes of rats. *Chem Biol Interact.* 82(3):317–28. doi:10.1016/0009-2797(92)90003-4 PMID:1606626
- Vanden Heuvel JP, Kuslikis BI, Van Rafelghem MJ, Peterson RE (1991). Tissue distribution, metabolism, and elimination of perfluorooctanoic acid in male and female rats. *J Biochem Toxicol*. 6(2):83–92. doi:<u>10.1002/</u> jbt.2570060202 PMID:<u>1941903</u>
- Vanden Heuvel JP, Thompson JT, Frame SR, Gillies PJ (2006). Differential activation of nuclear receptors by perfluorinated fatty acid analogs and natural fatty acids: a comparison of human, mouse, and rat peroxisome proliferator-activated receptor-alpha, -beta, and -gamma, liver X receptor-beta, and retinoid X receptor-alpha. *Toxicol Sci.* 92(2):476–89. doi:10.1093/toxsci/kfl014 PMID:16731579
- Verner M-A, Loccisano AE, Morken N-H, Yoon M, Wu H, McDougall R, et al. (2015). Associations of perfluoroalkyl substances (PFAS) with lower birth

weight: an evaluation of potential confounding by glomerular filtration rate using a physiologically based pharmacokinetic model (PBPK). *Environ Health Perspect*. 123(12):1317–24. doi:<u>10.1289/ehp.1408837</u> PMID:<u>26008903</u>

- Verner M-A, Ngueta G, Jensen ET, Fromme H, Völkel W, Nygaard UC, et al. (2016). A simple pharmacokinetic model of prenatal and postnatal exposure to perfluoroalkyl substances (PFASs). *Environ Sci Technol.* 50(2):978–86. doi:<u>10.1021/acs.est.5b04399</u> PMID:<u>26691063</u>
- Vested A, Ramlau-Hansen CH, Olsen SF, Bonde JP, Kristensen SL, Halldorsson TI, et al. (2013). Associations of in utero exposure to perfluorinated alkyl acids with human semen quality and reproductive hormones in adult men. *Environ Health Perspect*. 121(4):453–8. doi:10.1289/ehp.1205118 PMID:23360585
- Vestergren R, Cousins IT, Trudel D, Wormuth M, Scheringer M (2008). Estimating the contribution of precursor compounds in consumer exposure to PFOS and PFOA. *Chemosphere*. 73(10):1617–24. doi:10.1016/j. chemosphere.2008.08.011 PMID:18834614
- Vriens A, Nawrot TS, Janssen BG, Baeyens W, Bruckers L, Covaci A, et al. (2019). Exposure to environmental pollutants and their association with biomarkers of aging: a multipollutant approach. *Environ Sci Technol.* 53(10):5966–76. doi:<u>10.1021/acs.est.8b07141</u> PMID:<u>31041867</u>
- Wan C, Han R, Liu L, Zhang F, Li F, Xiang M, et al. (2016). Role of miR-155 in fluorooctane sulfonate-induced oxidative hepatic damage via the Nrf2-dependent pathway. *Toxicol Appl Pharmacol*. 295:85–93. doi:10.1016/j.taap.2016.01.023 PMID:26844784
- Wan HT, Zhao YG, Leung PY, Wong CK (2014). Perinatal exposure to perfluorooctane sulfonate affects glucose metabolism in adult offspring. *PLoS One*. 9(1):e87137. doi:<u>10.1371/journal.pone.0087137</u> PMID:<u>24498028</u>
- Wan HT, Zhao YG, Wong MH, Lee KF, Yeung WS, Giesy JP, et al. (2011). Testicular signalling is the potential target of perfluorooctanesulfonate-mediated subfertility in male mice. *Biol Reprod.* 84(5):1016–23. doi:10.1095/ biolreprod.110.089219 PMID:21209418
- Wan YJ, Li YY, Xia W, Chen J, Lv ZQ, Zeng HC, et al. (2010). Alterations in tumor biomarker GSTP gene methylation patterns induced by prenatal exposure to PFOS. *Toxicology*. 274(1-3):57-64. doi:10.1016/j. tox.2010.05.006 PMID:20621739
- Wan Ibrahim WN, Tofighi R, Onishchenko N, Rebellato P, Bose R, Uhlén P, et al. (2013). Perfluorooctane sulfonate induces neuronal and oligodendrocytic differentiation in neural stem cells and alters the expression of PPARy in vitro and in vivo. *Toxicol Appl Pharmacol*. 269(1):51– 60. doi:10.1016/j.taap.2013.03.003 PMID:23500012
- Wang C, Nie X, Zhang Y, Li T, Mao J, Liu X, et al. (2015c). Reactive oxygen species mediate nitric oxide production through ERK/JNK MAPK signalling in HAPI

microglia after PFOS exposure. *Toxicol Appl Pharmacol*. 288(2):143–51. doi:<u>10.1016/j.taap.2015.06.012</u> PMID: <u>26086160</u>

- Wang C, Ruan T, Liu J, He B, Zhou Q, Jiang G (2015d). Perfluorooctyl iodide stimulates steroidogenesis in H295R cells via a cyclic adenosine monophosphate signalling pathway. *Chem Res Toxicol.* 28(5):848–54. doi:10.1021/tx5004563 PMID:25871633
- Wang F, Liu W, Jin Y, Wang F, Ma J (2015b). Prenatal and neonatal exposure to perfluorooctane sulfonic acid results in aberrant changes in miRNA expression profile and levels in developing rat livers. *Environ Toxicol*. 30(6):712–23. doi:10.1002/tox.21949 PMID:24420840
- Wang F, Liu W, Ma J, Yu M, Jin Y, Dai J (2012b). Prenatal and neonatal exposure to perfluorooctane sulfonic acid results in changes in miRNA expression profiles and synapse associated proteins in developing rat brains. *Environ Sci Technol*. 46(12):6822–9. doi:10.1021/ es3008547 PMID:22594572
- WangH, DuH, YangJ, JiangH, OK, XuL, etal. (2019). PFOS, PFOA, estrogen homeostasis, and birth size in Chinese infants. *Chemosphere*. 221:349–55. doi:<u>10.1016/j.</u> <u>chemosphere.2019.01.061</u> PMID:<u>30641376</u>
- Wang H, Li W, Yang J, Wang Y, Du H, Han M, et al. (2023a). Gestational exposure to perfluoroalkyl substances is associated with placental DNA methylation and birth size. *Sci Total Environ*. 858(Pt 1):159747. doi:<u>10.1016/j.</u> <u>scitotenv.2022.159747</u> PMID:<u>36309289</u>
- Wang J, Pan Y, Cui Q, Yao B, Wang J, Dai J (2018a). Penetration of PFASs across the blood cerebrospinal fluid barrier and its determinants in humans. *Environ Sci Technol*. 52(22):13553–61. doi:<u>10.1021/acs.</u> <u>est.8b04550</u> PMID:<u>30362723</u>
- Wang J, Zhang Y, Zhang W, Jin Y, Dai J (2012a). Association of perfluorooctanoic acid with HDL cholesterol and circulating miR-26b and miR-199-3p in workers of a fluorochemical plant and nearby residents. *Environ Sci Technol.* 46(17):9274–81. doi:<u>10.1021/es300906q</u> PMID:<u>22862179</u>
- Wang L, Wang Y, Liang Y, Li J, Liu Y, Zhang J, et al. (2014b). PFOS induced lipid metabolism disturbances in BALB/c mice through inhibition of low-density lipoproteins excretion. *Sci Rep.* 4(1):4582. doi:<u>10.1038/</u> <u>srep04582</u> PMID:<u>24694979</u>
- Wang P, Liu D, Yan S, Liang Y, Cui J, Guo L, et al. (2022b). The role of ferroptosis in the damage of human proximal tubule epithelial cells caused by perfluorooctane sulfonate. *Toxics*. 10(8):10. doi:<u>10.3390/toxics10080436</u> PMID:<u>36006114</u>
- Wang Q, Chen W, Zhang B, Gao Z, Zhang Q, Deng H, et al. (2022a). Perfluorooctanoic acid induces hepatocellular endoplasmic reticulum stress and mitochondrial-mediated apoptosis in vitro via endoplasmic reticulum-mitochondria communication. *Chem Biol Interact.* 354:109844. doi:<u>10.1016/j.cbi.2022.109844</u> PMID:<u>35123991</u>

- Wang T, Zhao X, Liu T, Zhang J, Qiu J, Li M, et al. (2023b). Transcriptional investigation of the toxic mechanisms of perfluorooctane sulfonate in rats based on an RNA-seq approach. *Chemosphere*. 329:138629. doi:10.1016/j.chemosphere.2023.138629 PMID:<u>37030344</u>
- Wang X, Bai Y, Tang C, Cao X, Chang F, Chen L (2018b). Impact of perfluorooctane sulfonate on reproductive ability of female mice through suppression of estrogen receptor α-activated kisspeptin neurons. *Toxicol Sci.* 165(2):475–86. doi:10.1093/toxsci/kfy167 PMID:29939337
- Wang X, Liu L, Zhang W, Zhang J, Du X, Huang Q, et al. (2017). Serum metabolome biomarkers associate low-level environmental perfluorinated compound exposure with oxidative /nitrosative stress in humans. *Environ Pollut*. 229:168–76. doi:10.1016/j. envpol.2017.04.086 PMID:28599201
- Wang X, Zhao G, Liang J, Jiang J, Chen N, Yu J, et al. (2013). PFOS-induced apoptosis through mitochondrion-dependent pathway in human-hamster hybrid cells. *Mutat Res.* 754(1-2):51–7. doi:10.1016/j. mrgentox.2013.04.004 PMID:23618922
- Wang Y, Aimuzi R, Nian M, Zhang Y, Luo K, Zhang J (2021b). Perfluoroalkyl substances and sex hormones in postmenopausal women: NHANES 2013–2016. *Environ Int.* 149:106408. doi:10.1016/j.envint.2021. 106408 PMID:33548847
- Wang Y, Wang L, Li J, Liang Y, Ji H, Zhang J, et al. (2014a). The mechanism of immunosuppression by perfluorooctanoic acid in BALB/c mice. *Toxicol Res (Camb)*. 3(3):205–13. doi:<u>10.1039/c3tx50096a</u>
- Wang Y, Wang L, Liang Y, Qiu W, Zhang J, Zhou Q, et al. (2011). Modulation of dietary fat on the toxicological effects in thymus and spleen in BALB/c mice exposed to perfluorooctane sulfonate. *Toxicol Lett.* 204(2–3):174–82. doi:10.1016/j.toxlet.2011.04.029 PMID:21571052
- Wang Y, Zhang X, Wang M, Cao Y, Wang X, Liu Y, et al. (2015a). Mutagenic effects of perfluorooctanesulfonic acid in *gpt* delta transgenic system are mediated by hydrogen peroxide. *Environ Sci Technol.* 49(10):6294– 303. doi:10.1021/acs.est.5b00530 PMID:25875360
- Wang Z, Li C, Shao Y, Xue W, Wang N, Xu X, et al. (2021a). Antioxidant defense system responses, lysosomal membrane stability and DNA damage in earthworms (*Eisenia fetida*) exposed to perfluorooctanoic acid: an integrated biomarker approach to evaluating toxicity. *RSC Advances*. 11(43):26481–92. doi:10.1039/ D1RA04097A PMID:35479973
- Wang Z, Shi R, Ding G, Yao Q, Pan C, Gao Y, et al. (2022d). Association between maternal serum concentration of perfluoroalkyl substances (PFASs) at delivery and acute infectious diseases in infancy. *Chemosphere*. 289:133235. doi:10.1016/j.chemosphere.2021.133235
 PMID:<u>34896425</u>

- Wang Z, Yao J, Guo H, Sheng N, Guo Y, Dai J (2022c). Comparative hepatotoxicity of a novel perfluoroalkyl ether sulfonic acid, nafion byproduct 2 (H-PFMO2OSA), and legacy perfluorooctane sulfonate (PFOS) in adult male mice. *Environ Sci Technol.* 56(14):10183–92. doi:10.1021/acs.est.2c00957 PMID:35786879
- Watkins AM, Wood CR, Lin MT, Abbott BD (2015). The effects of perfluorinated chemicals on adipocyte differentiation in vitro. *Mol Cell Endocrinol.* 400:90–101. doi:10.1016/j.mce.2014.10.020 PMID:25448844
- Watkins DJ, Wellenius GA, Butler RA, Bartell SM, Fletcher T, Kelsey KT (2014). Associations between serum perfluoroalkyl acids and LINE-1 DNA methylation. *Environ Int.* 63:71–6. doi:<u>10.1016/j.envint.2013.10.018</u> PMID:<u>24263140</u>
- Weaver YM, Ehresman DJ, Butenhoff JL, Hagenbuch B (2010). Roles of rat renal organic anion transporters in transporting perfluorinated carboxylates with different chain lengths. *Toxicol Sci.* 113(2):305–14. doi:10.1093/ toxsci/kfp275 PMID:19915082
- Webster GM, Rauch SA, Marie NS, Mattman A, Lanphear BP, Venners SA (2016). Cross-sectional associations of serum perfluoroalkyl acids and thyroid hormones in US adults: variation according to TPOAb and iodine status (NHANES 2007–2008). *Environ Health Perspect.* 124(7):935–42. doi:10.1289/ehp.1409589 PMID:26517287
- Webster GM, Venners SA, Mattman A, Martin JW (2014). Associations between perfluoroalkyl acids (PFASs) and maternal thyroid hormones in early pregnancy: a population-based cohort study. *Environ Res.* 133:338– 47. doi:10.1016/j.envres.2014.06.012 PMID:25019470
- Wei KN, Wang XJ, Zeng ZC, Gu RT, Deng SZ, Jiang J, et al. (2021). Perfluorooctane sulfonate affects mouse oocyte maturation in vitro by promoting oxidative stress and apoptosis induced by mitochondrial dysfunction. *Ecotoxicol Environ Saf.* 225:112807. doi:10.1016/j. ecoenv.2021.112807 PMID:34562787
- Weiss JM, Andersson PL, Lamoree MH, Leonards PEG, van Leeuwen SPJ, Hamers T (2009). Competitive binding of poly- and perfluorinated compounds to the thyroid hormone transport protein transthyretin. *Toxicol Sci.* 109(2):206–16. doi:<u>10.1093/toxsci/kfp055</u> PMID:<u>19293372</u>
- Wen LL, Chen YT, Lee YG, Ko TL, Chou HC, Juan SH (2021). Perfluorooctane sulfonate induces autophagy-associated apoptosis through oxidative stress and the activation of extracellular signal-regulated kinases in renal tubular cells. *PLoS One*. 16(1):e0245442. doi:<u>10.1371/journal.pone.0245442</u> PMID:<u>33471797</u>
- Wen LL, Lin CY, Chou HC, Chang CC, Lo HY, Juan SH (2016). Perfluorooctanesulfonate mediates renal tubular cell apoptosis through PPARgamma inactivation. *PLoS One*. 11(5):e0155190. doi:<u>10.1371/journal.</u> <u>pone.0155190</u> PMID:<u>27171144</u>

- Wen LL, Lin LY, Su TC, Chen PC, Lin CY (2013). Association between serum perfluorinated chemicals and thyroid function in US adults: the National Health and Nutrition Examination Survey 2007–2010. J Clin Endocrinol Metab. 98(9):E1456–64. doi:10.1210/ jc.2013-1282 PMID:23864701
- Wen Y, Chen J, Li J, Arif W, Kalsotra A, Irudayaraj J (2020). Effect of PFOA on DNA methylation and alternative splicing in mouse liver. *Toxicol Lett.* 329:38–46. doi:<u>10.1016/j.toxlet.2020.04.012</u> PMID:<u>32320774</u>
- Wen Y, Rashid F, Fazal Z, Singh R, Spinella MJ, Irudayaraj J (2022). Nephrotoxicity of perfluorooctane sulfonate (PFOS)-effect on transcription and epigenetic factors. *Environ Epigenet*. 8(1):c010. doi:<u>10.1093/eep/dvac010</u> PMID:<u>35633893</u>
- Wielsøe M, Long M, Ghisari M, Bonefeld-Jørgensen EC (2015). Perfluoroalkylated substances (PFAS) affect oxidative stress biomarkers in vitro. *Chemosphere*. 129:239–45. doi:10.1016/j.chemosphere.2014.10.014 PMID:25455676
- Wilson J, Berntsen HF, Zimmer KE, Verhaegen S, Frizzell C, Ropstad E, et al. (2016). Do persistent organic pollutants interact with the stress response? Individual compounds, and their mixtures, interaction with the glucocorticoid receptor. *Toxicol Lett.* 241:121–32. doi:10.1016/j.toxlet.2015.11.014 PMID:26599974
- Winquist A, Lally C, Shin H-M, Steenland K (2013). Design, methods, and population for a study of PFOA health effects among highly exposed mid-Ohio valley community residents and workers. *Environ Health Perspect*. 121(8):893–9. doi:<u>10.1289/ehp.1206450</u> PMID: <u>23735518</u>
- Wisely GB, Miller AB, Davis RG, Thornquest AD Jr, Johnson R, Spitzer T, et al. (2002). Hepatocyte nuclear factor 4 is a transcription factor that constitutively binds fatty acids. *Structure*. 10(9):1225–34. doi:10.1016/ <u>S0969-2126(02)00829-8</u> PMID:12220494
- Wolf CJ, Schmid JE, Lau C, Abbott BD (2012). Activation of mouse and human peroxisome proliferator-activated receptor-alpha (PPARα) by perfluoroalkyl acids (PFAAs): further investigation of C4–C12 compounds. *Reprod Toxicol.* 33(4):546–51. doi:<u>10.1016/j.reprotox.</u> <u>2011.09.009</u> PMID:<u>22107727</u>
- Wolf CJ, Takacs ML, Schmid JE, Lau C, Abbott BD (2008a). Activation of mouse and human peroxisome proliferator-activated receptor alpha by perfluoroalkyl acids of different functional groups and chain lengths. *Toxicol Sci.* 106(1):162–71. doi:10.1093/toxsci/kfn166 PMID:18713766
- Wolf DC, Moore T, Abbott BD, Rosen MB, Das KP, Zehr RD, et al. (2008b). Comparative hepatic effects of perfluorooctanoic acid and WY14,643 in PPAR-alpha knockout and wild-type mice. *Toxicol Pathol.* 36(4):632–9. doi:10.1177/0192623308318216 PMID:18467680

- Wong F, MacLeod M, Mueller JF, Cousins IT (2014). Enhanced elimination of perfluorooctane sulfonic acid by menstruating women: evidence from population-based pharmacokinetic modeling. *Environ Sci Technol.* 48(15):8807–14. doi:10.1021/es500796y PMID:24943117
- Woodcroft MW, Ellis DA, Rafferty SP, Burns DC, March RE, Stock NL, et al. (2010). Experimental characterization of the mechanism of perfluorocarboxylic acids' liver protein bioaccumulation: the key role of the neutral species. *Environ Toxicol Chem.* 29(8):1669–77. doi:10.1002/etc.199 PMID:20821618
- Worley RR, Moore SM, Tierney BC, Ye X, Calafat AM, Campbell S, et al. (2017a). Per- and polyfluoroalkyl substances in human serum and urine samples from a residentially exposed community. *Environ Int*. 106:135– 43. doi:10.1016/j.envint.2017.06.007 PMID:28645013
- Worley RR, Yang X, Fisher J (2017b). Physiologically based pharmacokinetic modeling of human exposure to perfluorooctanoic acid suggests historical non drinking-water exposures are important for predicting current serum concentrations. *Toxicol Appl Pharmacol.* 330:9–21. doi:<u>10.1016/j.taap.2017.07.001</u> PMID:<u>28684146</u>
- Wu L-L, Gao H-W, Gao N-Y, Chen F-F, Chen L (2009). Interaction of perfluorooctanoic acid with human serum albumin. *BMC Struct Biol.* 9(1):31. doi:10.1186/1472-6807-9-31 PMID:19442292</jrn>
- Xiao C, Grandjean P, Valvi D, Nielsen F, Jensen TK, Weihe P, et al. (2020). Associations of exposure to perfluoroalkyl substances with thyroid hormone concentrations and birth size. *J Clin Endocrinol Metab.* 105(3):735–45. doi:10.1210/clinem/dgz147 PMID:31665456
- Xie MY, Sun XF, Wu CC, Huang GL, Wang P, Lin ZY, et al. (2023). Glioma is associated with exposure to legacy and alternative per- and polyfluoroalkyl substances. *J Hazard Mater.* 441:129819. doi:<u>10.1016/j.</u> jhazmat.2022.129819 PMID:<u>36084455</u>
- Xie X, Weng X, Liu S, Chen J, Guo X, Gao X, et al. (2021). Perfluoroalkyl and polyfluoroalkyl substance exposure and association with sex hormone concentrations: results from the NHANES 2015–2016. *Environ Sci Eur.* 33(1):69. doi:<u>10.1186/s12302-021-00508-9</u> PMID:<u>36061407</u>
- Xu B, Chen X, Mao Z, Chen M, Han X, Du G, et al. (2013b). Perfluorooctane sulfonate disturbs Nanog expression through miR-490-3p in mouse embryonic stem cells. *PLoS One*. 8(10):e74968. doi:10.1371/journal. pone.0074968 PMID:24098361
- Xu B, Ji X, Chen X, Yao M, Han X, Chen M, et al. (2015). Effect of perfluorooctane sulfonate on pluripotency and differentiation factors in mouse embryoid bodies. *Toxicology.* 328:160–7. doi:<u>10.1016/j.tox.2014.12.010</u> PMID:<u>25510869</u>

- Xu C, Jiang Z-Y, Liu Q, Liu H, Gu A (2017). Estrogen receptor beta mediates hepatotoxicity induced by perfluorooctane sulfonate in mouse. *Environ Sci Pollut Res Int.* 24(15):13414–23. doi:10.1007/s11356-017-8943-3 PMID:28386898
- Xu D, Li C, Wen Y, Liu W (2013a). Antioxidant defense system responses and DNA damage of earthworms exposed to perfluorooctane sulfonate (PFOS). *Environ Pollut*. 174:121–7. doi:10.1016/j.envpol.2012.10.030 PMID:23257261
- Xu J, Shimpi P, Armstrong L, Salter D, Slitt AL (2016). PFOS induces adipogenesis and glucose uptake in association with activation of Nrf2 signalling pathway. *Toxicol Appl Pharmacol.* 290:21–30. doi:10.1016/j. taap.2015.11.002 PMID:26548598
- Xu M, Wan J, Niu Q, Liu R (2019). PFOA and PFOS interact with superoxide dismutase and induce cytotoxicity in mouse primary hepatocytes: a combined cellular and molecular methods. *Environ Res.* 175:63– 70. doi:10.1016/j.envres.2019.05.008 PMID:31103794
- Xu Y, Fletcher T, Pineda D, Lindh CH, Nilsson C, Glynn A, et al. (2020a). Serum half-lives for short-and longchain perfluoroalkyl acids after ceasing exposure from drinking water contaminated by firefighting foam. *Environ Health Perspect*. 128(7):77004. doi:10.1289/ EHP6785 PMID:32648786
- Xu Y, Jurkovic-Mlakar S, Li Y, Wahlberg K, Scott K, Pineda D, et al. (2020b). Association between serum concentrations of perfluoroalkyl substances (PFAS) and expression of serum microRNAs in a cohort highly exposed to PFAS from drinking water. *Environ Int.* 136:105446. doi:10.1016/j.envint.2019.105446 PMID:31926437
- Xu Y, Lindh CH, Fletcher T, Jakobsson K, Engström K (2022). Perfluoroalkyl substances influence DNA methylation in school-age children highly exposed through drinking water contaminated from fire-fighting foam: a cohort study in Ronneby, Sweden. *Environ Epigenet.* 8(1):c004. doi:<u>10.1093/eep/dvac004</u> PMID:<u>35308102</u>
- Yahia D, Haruka I, Kagashi Y, Tsuda S (2016). 8-Hydroxy-2'-deoxyguanosine as a biomarker of oxidative DNA damage induced by perfluorinated compounds in TK6 cells. *Environ Toxicol.* 31(2):192–200. doi:10.1002/ tox.22034 PMID:25113910
- Yan S, Wang J, Dai J (2015b). Activation of sterol regulatory element-binding proteins in mice exposed to perfluorooctanoic acid for 28 days. *Arch Toxicol.* 89(9):1569–78. doi:10.1007/s00204-014-1322-7 PMID:25092180
- Yan S, Wang J, Zhang W, Dai J (2014). Circulating microRNA profiles altered in mice after 28 d exposure to perfluorooctanoic acid. *Toxicol Lett.* 224(1):24–31. doi:<u>10.1016/j.toxlet.2013.10.017</u> PMID:<u>24459700</u>
- Yan S, Zhang H, Guo X, Wang J, Dai J (2017). High perfluorooctanoic acid exposure induces autophagy blockage and disturbs intracellular vesicle fusion in the

liver. Arch Toxicol. 91(1):247–58. doi:<u>10.1007/s00204-</u> <u>016-1675-1</u> PMID:<u>26879310</u>

- Yan S, Zhang H, Wang J, Zheng F, Dai J (2015a). Perfluorooctanoic acid exposure induces endoplasmic reticulum stress in the liver and its effects are ameliorated by 4-phenylbutyrate. *Free Radic Biol Med.* 87:300–11. doi:10.1016/j.freeradbiomed.2015.06.043 PMID:26159507
- Yang B, Zou W, Hu Z, Liu F, Zhou L, Yang S, et al. (2014). Involvement of oxidative stress and inflammation in liver injury caused by perfluorooctanoic acid exposure in mice. *BioMed Res Int*. 2014:409837. doi:10.1155/2014/409837 PMID:24724082
- Yang CH, Glover KP, Han X (2010). Characterization of cellular uptake of perfluorooctanoate via organic anion-transporting polypeptide 1A2, organic anion transporter 4, and urate transporter 1 for their potential roles in mediating human renal reabsorption of perfluorocarboxylates. *Toxicol Sci.* 117(2):294–302. doi:10.1093/toxsci/kfq219 PMID:20639259
- Yang J, Wang C, Nie X, Shi S, Xiao J, Ma X, et al. (2015). Perfluorooctane sulfonate mediates microglial activation and secretion of TNF-α through Ca²⁺-dependent PKC-NF-κB signalling. *Int Immunopharmacol.* 28(1): 52–60. doi:10.1016/j.intimp.2015.05.019 PMID:26004316
- Yang M, Lee Y, Gao L, Chiu K, Meling DD, Flaws JA, et al. (2022). Perfluorooctanoic acid disrupts ovarian steroidogenesis and folliculogenesis in adult mice. *Toxicol Sci.* 186(2):260–8. doi:<u>10.1093/toxsci/kfac005</u> PMID:<u>35104888</u>
- Yang M, Li L-Y, Qin X-D, Ye X-Y, Yu S, Bao Q, et al. (2021). Perfluorooctanesulfonate and perfluorooctanoate exacerbate airway inflammation in asthmatic mice and in vitro. *Sci Total Environ*. 766:142365. doi:<u>10.1016/j. scitotenv.2020.142365</u> PMID:<u>33601665</u>
- Yang Q, Abedi-Valugerdi M, Xie Y, Zhao XY, Möller G, Nelson BD, et al. (2002b). Potent suppression of the adaptive immune response in mice upon dietary exposure to the potent peroxisome proliferator, perfluorooctanoic acid. *Int Immunopharmacol.* 2(2–3):389–97. doi:10.1016/S1567-5769(01)00164-3 PMID:11811941
- Yang Q, Guo X, Chen Y, Zhang W, Ren J, Wang J, et al. (2020). Blood levels of perfluoroalkyl substances (PFASs), elements and their associations with metabolic syndrome (MetS) in Chinese male adults mediated by metabolic-related risk factors. *Sci Total Environ*. 742:140595. doi:10.1016/j.scitotenv.2020.140595 PMID:32629270
- Yang Q, Wang W, Liu C, Wang Y, Sun K (2016). Effect of PFOS on glucocorticoid-induced changes in human decidual stromal cells in the first trimester of pregnancy. *Reprod Toxicol*. 63:142–50. doi:<u>10.1016/j.</u> <u>reprotox.2016.06.003</u> PMID:<u>27288094</u>
- Yang Q, Xie Y, Alexson SE, Nelson BD, DePierre JW (2002a). Involvement of the peroxisome proliferator-activated receptor alpha in the immunomodulation

caused by peroxisome proliferators in mice. *Biochem Pharmacol.* 63(10):1893–900. doi:<u>10.1016/S0006-2952</u> (02)00923-1 PMID:<u>12034374</u>

- Yang Q, Xie Y, Depierre JW (2000). Effects of peroxisome proliferators on the thymus and spleen of mice. *Clin Exp Immunol.* 122(2):219–26. doi:<u>10.1046/j.1365-2249.2000.01367.x</u> PMID:<u>11091278</u>
- Yang Q, Xie Y, Eriksson AM, Nelson BD, DePierre JW (2001). Further evidence for the involvement of inhibition of cell proliferation and development in thymic and splenic atrophy induced by the peroxisome proliferator perfluoroctanoic acid in mice. *Biochem Pharmacol.* 62(8):1133–40. doi:10.1016/S0006-2952(01)00752-3 PMID:11597582
- Yang YD, Tian R, Lu N (2023). Binding of serum albumin to perfluorooctanoic acid reduced cytotoxicity. Sci Total Environ. 876:162738. doi:<u>10.1016/j.</u> <u>scitotenv.2023.162738</u> PMID:<u>36906033</u>
- Yao PL, Ehresman DJ, Rae JM, Chang SC, Frame SR, Butenhoff JL, et al. (2014). Comparative in vivo and in vitro analysis of possible estrogenic effects of perfluorooctanoic acid. *Toxicology*. 326:62–73. doi:<u>10.1016/j.</u> tox.2014.10.008 PMID:<u>25456267</u>
- Yao Q, Gao Y, Zhang Y, Qin K, Liew Z, Tian Y (2021). Associations of paternal and maternal per- and polyfluoroalkyl substances exposure with cord serum reproductive hormones, placental steroidogenic enzyme and birth weight. *Chemosphere*. 285:131521. doi:10.1016/j. chemosphere.2021.131521 PMID:34273704
- Yao Q, Shi R, Wang C, Han W, Gao Y, Zhang Y, et al. (2019). Cord blood per- and polyfluoroalkyl substances, placental steroidogenic enzyme, and cord blood reproductive hormone. *Environ Int*. 129:573–82. doi:<u>10.1016/j.envint.2019.03.047</u> PMID:<u>31174145</u>
- Yao X, Zhong L (2005). Genotoxic risk and oxidative DNA damage in HepG2 cells exposed to perfluorooctanoic acid. *Mutat Res.* 587(1–2):38–44. doi:10.1016/j. mrgentox.2005.07.010 PMID:16219484
- Yarahalli Jayaram V, Baggavalli S, Reddy D, Sistla S, Malempati R (2020). Effect of endosulfan and bisphenol A on the expression of SUMO and UBC9. *Drug Chem Toxicol*. 43(6):637–44. doi:<u>10.1080/01480545.2018.1526</u> <u>179 PMID:30426790</u>
- Ye S, Liu Q, Huang K, Jiang X, Zhang X (2022). The comprehensive analysis based study of perfluorinated compounds – environmental explanation of bladder cancer progression. *Ecotoxicol Environ Saf.* 229:113059. doi:10.1016/j.ecoenv.2021.113059 PMID:34894427
- Yeh MM, Bosch DE, Daoud SS (2019). Role of hepatocyte nuclear factor 4-alpha in gastrointestinal and liver diseases. World J Gastroenterol. 25(30):4074–91. doi:<u>10.3748/wjg.v25.i30.4074</u> PMID:<u>31435165</u>
- Yeung LW, Guruge KS, Taniyasu S, Yamashita N, Angus PW, Herath CB (2013a). Profiles of perfluoroalkyl substances in the liver and serum of patients with liver cancer and cirrhosis in Australia. *Ecotoxicol Environ*

Saf. 96:139–46. doi:<u>10.1016/j.ecoenv.2013.06.006</u> PMID:<u>23849467</u>

- Yeung LW, Robinson SJ, Koschorreck J, Mabury SA (2013b). Part II. A temporal study of PFOS and its precursors in human plasma from two German cities in 1982–2009. *Environ Sci Technol.* 47(8):3875–82. doi:10.1021/es4004153 PMID:23484930
- Ylinen M, Hanhijärvi H, Jaakonaho J, Peura P (1989). Stimulation by oestradiol of the urinary excretion of perfluorooctanoic acid in the male rat. *Pharmacol Toxicol.* 65(4):274–7. doi:<u>10.1111/j.1600-0773.1989.</u> <u>tb01172.x</u> PMID:<u>2587510</u>
- Ylinen M, Kojo A, Hanhijärvi H, Peura P (1990). Disposition of perfluorooctanoic acid in the rat after single and subchronic administration. *Bull Environ Contam Toxicol.* 44(1):46–53. doi:<u>10.1007/BF01702360</u> PMID:<u>2306537</u>
- Yu G, Wang J, Liu Y, Luo T, Meng X, Zhang R, et al. (2023). Metabolic perturbations in pregnant rats exposed to low-dose perfluorooctanesulfonic acid: an integrated multi-omics analysis. *Environ Int.* 173:107851. doi:10.1016/j.envint.2023.107851 PMID:36863164
- Yu M, Hazelton WD, Luebeck GE, Grady WM (2020). Epigenetic aging: more than just a clock when it comes to cancer. *Cancer Res.* 80(3):367–74. doi:10.1158/0008-5472.CAN-19-0924 PMID:31694907
- Yu N, Wei S, Li M, Yang J, Li K, Jin L, et al. (2016). Effects of perfluorooctanoic acid on metabolic profiles in brain and liver of mouse revealed by a high-throughput targeted metabolomics approach. *Sci Rep.* 6(1):23963. doi:10.1038/srep23963 PMID:27032815
- Yu WG, Liu W, Jin YH, Liu XH, Wang FQ, Liu L, et al. (2009). Prenatal and postnatal impact of perfluorooctane sulfonate (PFOS) on rat development: a cross-foster study on chemical burden and thyroid hormone system. *Environ Sci Technol*. 43(21):8416–22. doi:10.1021/es901602d PMID:19924978
- Yuan Y, Ding X, Cheng Y, Kang H, Luo T, Zhang X, et al. (2020). PFOA evokes extracellular Ca²⁺ influx and compromises progesterone-induced response in human sperm. *Chemosphere*. 241:125074. doi:10.1016/j. chemosphere.2019.125074 PMID:31627108
- Zarei MH, Hosseini Shirazi SF, Aghvami M, Pourahmad J (2018). Perfluorooctanesulfonate (PFOS) induces apoptosis signalling and proteolysis in human lymphocytes through ROS mediated mitochondrial dysfunction and lysosomal membrane labialization. *Iran J Pharm Res.* 17(3):995–1007. <u>https://pmc.ncbi.nlm.nih.gov/articles/PMC6094418/</u> PMID:30127822
- Zeng HC, Zhang L, Li YY, Wang YJ, Xia W, Lin Y, et al. (2011). Inflammation-like glial response in rat brain induced by prenatal PFOS exposure. *Neurotoxicology*. 32(1):130–9. doi:<u>10.1016/j.neuro.2010.10.001</u> PMID: <u>20937303</u>

- Zeng HC, Zhu BQ, Wang YQ, He QZ (2021). ROS-triggered autophagy is involved in PFOS-induced apoptosis of human embryo liver L-02 cells. *BioMed Res Int.* 2021:6625952. doi:10.1155/2021/6625952 PMID:33880372
- Zhang H, Cui R, Guo X, Hu J, Dai J (2016b). Low dose perfluorooctanoate exposure promotes cell proliferation in a human non-tumor liver cell line. J Hazard Mater. 313(5 August):18–28. doi:10.1016/j. jhazmat.2016.03.077 PMID:27045622
- Zhang H, Xu C, Shi C, Zhang J, Qian T, Wang Z, et al. (2021a). Hypermethylation of heparanase 2 promotes colorectal cancer proliferation and is associated with poor prognosis. *J Transl Med.* 19(1):98. doi:10.1186/s12967-021-02770-0 PMID:33663522
- Zhang H, Zhang C, Xu D, Wang Q, Xu D (2023b). Effects of subchronic exposure of perfluorooctane sulfonate on cognitive function of mice and its mechanism. *Environ Pollut.* 329:121650. doi:<u>10.1016/j.envpol.2023.121650</u> PMID:<u>37062406</u>
- Zhang J, Liang J, Zhu H, Li C, Wu Q (2013d). PFOS and PCB 153 have direct adverse effects on neonatal testis modeled using a coculture of primary gonocyte and Sertoli cells. *Environ Toxicol*. 28(6):322–31. doi:<u>10.1002/</u> tox.20723 PMID:<u>21544924</u>
- Zhang J, Shao X, Zhao B, Zhai L, Liu N, Gong F, et al. (2020a). Neurotoxicity of perfluorooctanoic acid and post-exposure recovery due to blueberry anthocyanins in the planarians *Dugesia japonica*. *Environ Pollut*. 263(PtB):114471. doi:10.1016/j.envpol.2020.114471 PMID:32268227
- Zhang L, Krishnan P, Ehresman DJ, Smith PB, Dutta M, Bagley BD, et al. (2016a). Perfluorooctane sulfonate-choline ion pair formation: a potential mechanism modulating hepatic steatosis and oxidative stress in mice. *Toxicol Sci.* 153(1):186–97. doi:10.1093/toxsci/kfw120 PMID:27413108
- Zhang L, Louie A, Rigutto G, Guo H, Zhao Y, Ahn S, et al. (2023e). A systematic evidence map of chronic inflammation and immunosuppression related to per- and polyfluoroalkyl substance (PFAS) exposure. *Environ Res.* 220:115188. doi:10.1016/j.envres.2022.115188
 PMID:36592815
- Zhang L, Ren XM, Guo LH (2013c). Structure-based investigation on the interaction of perfluorinated compounds with human liver fatty acid binding protein. *Environ Sci Technol.* 47(19):11293–301. doi:10.1021/es4026722 PMID:24006842
- Zhang L, Ren X-M, Wan B, Guo L-H (2014b). Structuredependent binding and activation of perfluorinated compounds on human peroxisome proliferator-activated receptor γ. *Toxicol Appl Pharmacol.* 279(3):275– 83. doi:10.1016/j.taap.2014.06.020 PMID:24998974
- Zhang N, Wang WS, Li WJ, Liu C, Wang Y, Sun K (2015b). Reduction of progesterone, estradiol and hCG secretion by perfluorooctane sulfonate via induction of apoptosis

in human placental syncytiotrophoblasts. *Placenta*. 36(5):575–80. doi:<u>10.1016/j.placenta.2015.02.008</u> PMID:<u>25748801</u>

- Zhang P, Qi C, Ma Z, Wang Y, Zhang L, Hou X (2022a). Perfluorooctanoic acid exposure in vivo perturbs mitochondrial metabolic during oocyte maturation. *Environ Toxicol.* 37(12):2965–76. doi:<u>10.1002/tox.23652</u> PMID:<u>36029293</u>
- Zhang Q, Wang J, Chen C, Kong Y, Yan H, Duan J, et al. (2019). Perfluorooctanoic acid induces migration and invasion and inhibits apoptosis through the PI3K/AKT signalling pathway in human rhabdomyosarcoma cells. Oncol Rep. 42(4):1558–68. doi:10.3892/or.2019.7265 PMID:31524277
- Zhang R, Yao Y, Tu L, Luan T, Chen B (2021c). Non-targeted metabolomics of multiple human cells revealing differential toxic effects of perfluorooctanoic acid. *J Hazard Mater.* 409:125017. doi:10.1016/j.jhazmat.2020.125017 PMID:33421881
- Zhang T, Qin X (2014). Assessment of fetal exposure and maternal elimination of perfluoroalkyl substances. *Environ Sci Process Impacts*. 16(8):1878–81. doi:<u>10.1039/</u> <u>C4EM00129J</u> PMID:<u>24882725</u>
- Zhang T, Sun H, Lin Y, Qin X, Zhang Y, Geng X, et al. (2013a). Distribution of poly- and perfluoroalkyl substances in matched samples from pregnant women and carbon chain length related maternal transfer. *Environ Sci Technol.* 47(14):7974–81. doi:10.1021/ es400937y PMID:23777259
- Zhang T, Sun H, Qin X, Gan Z, Kannan K (2015a). PFOS and PFOA in paired urine and blood from general adults and pregnant women: assessment of urinary elimination. *Environ Sci Pollut Res Int.* 22(7):5572–9. doi:10.1007/s11356-014-3725-7 PMID:25367642
- Zhang Y, Beesoon S, Zhu L, Martin JW (2013b). Biomonitoring of perfluoroalkyl acids in human urine and estimates of biological half-life. *Environ Sci Technol.* 47(18):10619–27. doi:10.1021/es401905e PMID:23980546
- Zhang Y, Cao X, Chen L, Qin Y, Xu Y, Tian Y, et al. (2020b). Exposure of female mice to perfluorooctanoic acid suppresses hypothalamic kisspeptin-reproductive endocrine system through enhanced hepatic fibroblast growth factor 21 synthesis, leading to ovulation failure and prolonged dioestrus. J Neuroendocrinol. 32(5):e12848. doi:10.1111/jne.12848 PMID:32307816
- Zhang Y, Jiang W, Fang S, Zhu L, Deng J (2014a). Perfluoroalkyl acids and the isomers of perfluorooctanesulfonate and perfluorooctanoate in the sera of 50 new couples in Tianjin, China. *Environ Int*. 68:185–91. doi:<u>10.1016/j.envint.2014.03.022</u> PMID:<u>24747327</u>
- Zhang Y, Mustieles V, Sun Y, Oulhote Y, Wang YX, Messerlian C (2022b). Association between serum perand polyfluoroalkyl substances concentrations and common cold among children and adolescents in the

United States. *Environ Int.* 164:107239. doi:<u>10.1016/j.</u> envint.2022.107239 PMID:<u>35447424</u>

- Zhang Y, Mustieles V, Wang YX, Sun Q, Coull B, Sun Y, et al. (2023d). Red blood cell folate modifies the association between serum per- and polyfluoroalkyl substances and antibody concentrations in US adoles-cents. *Environ Sci Technol*. 57(6):2445–56. doi:10.1021/ acs.est.2c07152 PMID:36715557
- Zhang Y, Zhang L, Bao J, Liu L, Wang X (2021b). Perfluorooctanoic acid exposure in early pregnancy induces oxidative stress in mice uterus and liver. *Environ Sci Pollut Res Int*. 28(46):66355–65. doi:10.1007/ s11356-021-15453-6 PMID:34331232
- Zhang YH, Wang J, Dong GH, Liu MM, Wang D, Zheng L, et al. (2013d). Mechanism of perfluorooctanesul-fonate (PFOS)-induced apoptosis in the immuno-cyte. *J Immunotoxicol*. 10(1):49–58. doi:10.3109/1547 691X.2012.691123 PMID:22953760
- Zhang YM, Dong XY, Fan LJ, Zhang ZL, Wang Q, Jiang N, et al. (2017b). Poly- and perfluorinated compounds activate human pregnane X receptor. *Toxicology*. 380:23–9. doi:<u>10.1016/j.tox.2017.01.012</u> PMID:<u>28115241</u>
- Zhang Y-Z, Zeng X-W, Qian ZM, Vaughn MG, Geiger SD, Hu L-W, et al. (2017a). Perfluoroalkyl substances with isomer analysis in umbilical cord serum in China. *Environ Sci Pollut Res Int*. 24(15):13626–37. doi:10.1007/s11356-017-8954-0 PMID:28391466
- Zhang Z, Tian J, Liu W, Zhou J, Zhang Y, Ding L, et al. (2023a). Perfluorooctanoic acid exposure leads to defect in follicular development through disrupting the mitochondrial electron transport chain in granulosa cells. *Sci Total Environ*. 905:166954. doi:10.1016/j. scitotenv.2023.166954 PMID:37722425
- Zhang Z, Wen R, Zhang W, Ge N, Bo J (2023c). Efficacy and mechanism of lipoic acid in the treatment of reproductive injury caused by perfluorooctanoic acid. *Exp Ther Med.* 25(3):116. doi:<u>10.3892/etm.2023.11815</u> PMID:<u>36815965</u>
- Zhao B, Lian Q, Chu Y, Hardy DO, Li XK, Ge RS (2011b). The inhibition of human and rat 11β-hydroxysteroid dehydrogenase 2 by perfluoroalkylated substances. *J Steroid Biochem Mol Biol*. 125(1–2):143–7. doi:<u>10.1016/j.</u> jsbmb.2010.12.017 PMID:21237268
- Zhao C, Wang S, Zhai Y, Wang M, Tang Y, Li H, et al. (2023). Direct inhibition of human and rat 11β-hydroxysteroid dehydrogenase 2 by per- and polyfluoroalkyl substances: structure–activity relationship and in silico docking analysis. *Toxicology*. 488:153484. doi:<u>10.1016/j.</u> tox.2023.153484 PMID:<u>36878351</u>
- Zhao G, Wang J, Wang X, Chen S, Zhao Y, Gu F, et al. (2011a). Mutagenicity of PFOA in mammalian cells: role of mitochondria-dependent reactive oxygen species. *Environ Sci Technol.* 45(4):1638–44. doi:10.1021/es1026129 PMID:21194205

- Zhao H, Xie J, Wu S, Sánchez OF, Zhang X, Freeman JL, et al. (2022). Pre-differentiation exposure of PFOA induced persistent changes in DNA methylation and mitochondrial morphology in human dopaminergic-like neurons. *Environ Pollut.* 308:119684. doi:10.1016/j.envpol.2022.119684 PMID:35764183
- Zhao W, Zitzow JD, Ehresman DJ, Chang SC, Butenhoff JL, Forster J, et al. (2015). Na+/taurocholate cotransporting polypeptide and apical sodium-dependent bile acid transporter are involved in the disposition of perfluoroalkyl sulfonates in humans and rats. *Toxicol Sci.* 146(2):363–73. doi:10.1093/toxsci/kfv102 PMID:26001962
- Zhao Y, Tan YS, Haslam SZ, Yang C (2010). Perfluorooctanoic acid effects on steroid hormone and growth factor levels mediate stimulation of peripubertal mammary gland development in C57BL/6 mice. *Toxicol Sci.* 115(1):214–24. doi:<u>10.1093/toxsci/kfq030</u> PMID:<u>20118188</u>
- Zheng L, Dong GH, Jin YH, He QC (2009). Immunotoxic changes associated with a 7-day oral exposure to perfluorooctanesulfonate (PFOS) in adult male C57BL/6 mice. *Arch Toxicol*. 83(7):679–89. doi:10.1007/s00204-008-0361-3 PMID:19015834
- Zheng L, Dong GH, Zhang YH, Liang ZF, Jin YH, He QC (2011). Type 1 and Type 2 cytokines imbalance in adult male C57BL/6 mice following a 7-day oral exposure to perfluorooctanesulfonate (PFOS). *J Immunotoxicol*. 8(1):30–8. doi:10.3109/1547691X.2010.537287 PMID: 21299352
- Zheng XQ, Shi YJ, Lu Y-L, Xu AB (2016). Growth inhibition and DNA damage in the earthworm (*Eisenia fetida*) exposed to perfluorooctane sulphonate and perfluorooctanoic acid. *Chem Ecol.* 32(2):103–16. doi: 10.1080/02757540.2015.1116524
- Zhong SQ, Chen ZX, Kong ML, Xie YQ, Zhou Y, Qin XD, et al. (2016). Testosterone-mediated endocrine function and TH1/TH2 cytokine balance after prenatal exposure to perfluorooctane sulfonate: by sex status. *Int J Mol Sci.* 17(9):1509. doi:10.3390/ijms17091509 PMID:27626407
- Zhou Y, Hu L-W, Qian ZM, Chang J-J, King C, Paul G, et al. (2016). Association of perfluoroalkyl substances exposure with reproductive hormone levels in adolescents: by sex status. *Environ Int.* 94:189–95. doi:10.1016/j. envint.2016.05.018 PMID:27258660
- Zhou Y, Hu L-W, Qian ZM, Geiger SD, Parrish KL, Dharmage SC, et al. (2017). Interaction effects of polyfluoroalkyl substances and sex steroid hormones on asthma among children. *Sci Rep.* 7(1):899. doi:<u>10.1038/</u> <u>s41598-017-01140-5</u> PMID:<u>28420867</u>
- Zhou Y, Li H, Lin C, Mao Y, Rao J, Lou Y, et al. (2020). Perfluorooctanoic acid (PFOA) inhibits the gap junction intercellular communication and induces apoptosis in human ovarian granulosa cells.

Reprod Toxicol. 98(December):125–33. doi:<u>10.1016/j.</u> reprotox.2020.09.005 PMID:<u>32971237</u>

- Zhou YT, Li R, Li SH, Ma X, Liu L, Niu D, et al. (2022). Perfluorooctanoic acid (PFOA) exposure affects early embryonic development and offspring oocyte quality via inducing mitochondrial dysfunction. *Environ Int.* 167:107413. doi:10.1016/j.envint.2022.107413 PMID:35863238
- Zhou Z, Shi Y, Vestergren R, Wang T, Liang Y, Cai Y (2014). Highly elevated serum concentrations of perfluoroalkyl substances in fishery employees from Tangxun lake, China. *Environ Sci Technol*. 48(7):3864–74. doi:10.1021/ es4057467 PMID:24588690
- Zhu J, Qian W, Wang Y, Gao R, Wang J, Xiao H (2015). Involvement of mitogen-activated protein kinase and NF-κB signalling pathways in perfluorooctane sulfonic acid-induced inflammatory reaction in BV2 microglial cells. *J Appl Toxicol*. 35(12):1539–49. doi:<u>10.1002/</u> jat.3119 PMID:<u>25677194</u>
- Zhu TR, Cao J, Hong JW, Li J (2020). [Effects of PFOS on inflammatory factors in human placental trophoblast cells]. *Zhonghua Lao Dong Wei Sheng Zhi Ye Bing Za Zhi*. 38(7):481–4. doi:10.3760/cma.j.cn121094-20190917-00380 PMID:32746565 [Chinese]
- Zhu Y, Qin XD, Zeng XW, Paul G, Morawska L, Su MW, et al. (2016). Associations of serum perfluoroalkyl acid levels with T-helper cell-specific cytokines in children: by gender and asthma status. *Sci Total Environ*. 559:166– 73. doi:10.1016/j.scitotenv.2016.03.187 PMID:27060656
- Zhu Y, Shin H-M, Jiang L, Bartell SM (2022). Retrospective exposure reconstruction using approximate Bayesian computation: a case study on perfluorooctanoic acid and preeclampsia. *Environ Res.* 209:112892. doi:10.1016/j. envres.2022.112892 PMID:35149111
- Zota AR, Geller RJ, Romano LE, Coleman-Phox K, Adler NE, Parry E, et al. (2018). Association between persistent endocrine-disrupting chemicals (PBDEs, OH-PBDEs, PCBs, and PFASs) and biomarkers of inflammation and cellular aging during pregnancy and postpartum. *Environ Int.* 115:9–20. doi:<u>10.1016/j.envint.2018.02.044</u> PMID:<u>29533840</u>
- Zou W, Liu W, Yang B, Wu L, Yang J, Zou T, et al. (2015). Quercetin protects against perfluorooctanoic acid-induced liver injury by attenuating oxidative stress and inflammatory response in mice. *Int Immunopharmacol.* 28(1):129–35. doi:<u>10.1016/j.intimp.2015.05.043</u> PMID: <u>26054880</u>

5. SUMMARY OF DATA REPORTED

5.1 Exposure characterization

Perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS) are per- and polyfluoroalkyl substances (PFAS) with a carbon chain length of eight carbons. The carbon-fluorine bond is one of the strongest bonds known in nature, making PFOA and PFOS extremely resistant to degradation in the natural environment. Both PFOA and PFOS exist as linear and branched isomers, and their salts exhibit different physicochemical properties to those of the pure acid form. PFOA and PFOS and their respective salts will be in an acid-base equilibrium in aqueous solutions such as in the human body and are present mainly as their conjugate bases perfluorooctanoate and perfluorooctane sulfonate, respectively. All isomeric forms and their salts should be considered as included within the definition of the agents reviewed in the present monograph.

The production of PFOA and PFOS began in the 1940s and steadily increased until the late 1990s, and companies located in the USA, Europe, and Japan were responsible for most of the manufacturing. However, in the early 2000s, there was a geographical shift in the production of PFOA and PFOS to other parts of the world (primarily in emerging Asian economies) and a shift towards production of other PFAS.

PFOA and PFOS have unique properties (e.g. hydrophobicity and oleophobicity, surface-active properties, chemical stability, and thermal resistance). They may be present in products as main ingredients, or as unreacted raw materials, undesired reaction by-products, or cross-contaminants along the production and supply chains. Ammonium perfluorooctanoate (APFO) - a salt of PFOA - has been used extensively to manufacture fluoropolymers, such as polytetrafluoroethylene (PTFE). Applications for fluoropolymers, as well as direct uses for PFOA, include household products with non-stick coatings; textiles for outdoor or personal protection applications; personal care products; seals and gaskets; coatings for cables and wires; electronics, solar panels and electrolyte fuel cells; carpets; cleaning and impregnating agents; construction materials; and surface coatings for conferring stain, oil and water resistance on carpets, textiles, leather products, and paper or cardboard for food and feed packaging.

With some applications that overlap those of PFOA, such as waxes, carpets, and food and feed packaging, PFOS has additionally been used in the semiconductor industry; as a hydraulic fluid additive; as an etchant and antireflective coating in photolithography processes; in the fabrication of imaging devices; as a mist suppressant in electroplating operations; in building and construction materials, including paints and varnishes; in insulation; in dyes and ink; and in wetting, levelling, and dispersing agents. PFOS has been used extensively in class B firefighting foams known as aqueous film-forming foams (AFFFs).

PFOA and PFOS occur in the whole ecosystem, including air, water, dust, soil, and food, but levels vary greatly in different geographical regions due to pollution sources such as industrial sites, firefighter-training areas, waste deposits, and contaminated wastewater. The transport of PFOA and PFOS in air and surface water leads to their deposition in oceans, soil, and groundwater.

Foods are contaminated with PFOA and PFOS through atmospheric deposition and uptake from water and soil, including from use of biosolids as fertilizer. Animal-based foods are contaminated through water, feed, soil, and air. The highest concentrations have been measured in fish, seafood, and eggs.

Occupationally exposed populations have some of the highest exposure to PFOA and PFOS, with the leading route of exposure consisting of inhalation, as well as potentially dermal absorption and ingestion of dust. Biomonitoring data indicate exposure in diverse occupational settings, with the highest levels in primary manufacturing (up to median values of thousands of nanograms per millilitre of serum) and lower levels in secondary manufacturing, public safety, and services. Not all occupations have been characterized for PFOA and PFOS exposure. Measures in the work environment such as air frequently indicate that concentrations of PFOA and PFOS are higher in facilities manufacturing or using PFAS-laden products than in other occupational environments.

PFOA and PFOS are detected in blood samples in all populations worldwide who have been tested. The general population in non-polluted communities is mainly exposed to PFOA and PFOS via the diet and drinking-water. Additional exposure via consumer products and building materials may occur. In communities located in the proximity of polluted sites, the general population is mainly exposed via drinking-water. Biomonitoring in general populations mainly in North America and Europe has shown serum concentrations in the low nanograms per millilitre range and that concentrations have decreased since the early 2000s. Median concentrations in serum samples collected in contaminated communities have been measured in the hundreds of nanograms per millilitre range.

The term "precursor compounds" refers to PFAS known to break down or transform into PFOA or PFOS in the environment or biota, including in humans. Although estimates vary according to exposure scenario, it has been estimated that a substantial proportion of the body burden of PFOA and PFOS may originate from exposure to precursors. Direct exposure to PFOA and PFOS may decline as a result of regulation or voluntary efforts; however, production and use of precursors may contribute to ongoing exposure.

International, national, and regional authorities have developed occupational exposure thresholds for PFOA, PFOS, and/or related compounds, restrictions on the use of PFOA and PFOS in consumer products, and regulatory standards or guidance values for these PFAS in environmental media. PFOA and its salts and PFOS and derivatives are listed in Annex A (elimination) and Annex B (restriction), respectively, in the Stockholm Convention on Persistent Organic Pollutants. Drinking-water is a major focus for the regulation of PFOA and PFOS. Additional restrictions, regulations, and guidance values continue to be developed and have generally become more stringent over time.

5.2 Cancer in humans

More than three dozen studies were available for the evaluation of the carcinogenicity of PFOA and PFOS in humans; this represents a substantial increase over the number available during the previous evaluation of PFOA in Volume

110 of the IARC Monographs (Some Chemicals used as Solvents and in Polymer Manufacture). Most of these were cohort studies (including nested case-control and case-cohort studies), but there were also some population-based or hospital-based case-control studies. The studies were conducted within three different types of populations: (i) workers exposed to high levels of PFOA and/or PFOS during employment at industrial plants manufacturing or using these chemicals; (ii) general populations of residents exposed to high environmental levels of PFOA and/or PFOS, primarily through drinking-water near sites contaminated by chemical production or use; and (iii) populations exposed to background levels of these compounds primarily through food and drinking-water. The studies were conducted mainly in the USA and Europe, although several studies were carried out in China. Exceptionally, the Working Group performed an ecological analysis of the association between average serum concentrations of PFOA and the rates of orchiectomies for a set of 21 municipalities in the Veneto region of northern Italy, where drinking-water had been heavily contaminated by pollution from a local chemical plant, described below. Orchiectomies were found to be a highly reliable surrogate for testicular cancer in this region.

Despite the overall large number of available studies, for most cancer types there were fewer than 10 studies that examined risk for the type. The most informative studies for the evaluation were large cohort and nested case–control studies from all three exposure scenarios described above. There were three occupational cohorts from the USA: at a PFOA-manufacturing plant in Minnesota, a fluoropolymer-manufacturing plant (using PFOA) in West Virginia, and a fluorochemical plant in Alabama where PFOS was extensively used. There was an additional small occupational cohort of workers in another fluorochemical plant (with mainly PFOA exposure) in Veneto, Italy. This plant was the source of the contamination in that area. The most informative occupational cohort was from the facility in West Virginia. The Mid-Ohio Valley (West Virginia and Ohio) general population (part of the C8 Science Panel Cohort and exposed to high background levels of PFOA), a prospective cohort based on the Prostate, Lung, Colorectal, and Ovarian (PLCO) Cancer Screening Trial participants, and other prospective cohorts of populations exposed to background levels of PFOA or PFOS exposure were also considered highly informative.

An important consideration in the evaluation of the human cancer evidence was the quality of the exposure assessment methodology used in the studies. The highest-quality studies used pre-diagnostic, targeted serum analyses for individual PFAS compounds, based on samples collected at least several years before cancer diagnosis. These were features of most of the nested case-control and case-cohort studies. One potential concern was the fact that most studies in the general population relied on a single time point measurement of PFOA or PFOS in serum, and it was unclear how representative such single time point exposure measures were for long-term exposure assessment in relation to cancer. This concern was allayed by the Working Group's evaluation of the possible impact of such exposure misclassification, given the long serum half-life of the compounds in humans and the resulting high correlation between repeated time point measures of exposure available for two of the studies. The Working Group therefore concluded that only minor bias towards the null would probably result from this source of exposure uncertainty, at least over a 5-8-year time period.

The main concern across the set of studies related to the potential for co-exposure to other potentially carcinogenic PFAS compounds (e.g. PFOA and PFOS together, or with other PFAS compounds). In the most informative studies, the researchers adjusted statistically for the effects of the other compounds, or the Working Group concluded that exposure to one of the compounds –either PFOA or PFOS – was predominant (this was generally the case for the occupational and high-environmental-exposure studies). Other types of confounding were not a major concern in the set of studies, particularly for kidney and testicular cancer, as relatively few strong risk factors are known, and correlations of these risk factors with occupational or environmental exposures to PFOA or PFOS are anticipated to be low. In addition, for some other cancer types, e.g. breast, estimates were well adjusted for important potential confounders.

Among the occupational cohort studies, in the Minnesota and Alabama cohorts, findings were mostly null, but the studies and case numbers were small and there were limitations related to potential survivor bias and/or weaknesses in exposure assessment, which would be expected to cause bias downwards or towards the null. The Veneto occupational cohort was small and showed some evidence (albeit weak) of positive findings for a few cancer sites.

5.2.1 PFOA

The cancer sites with the strongest evidence of an association with PFOA were kidney and testis. For kidney cancer, two independent studies were considered most informative: the set of three partly overlapping studies of workers and residents in West Virginia and Ohio (the Mid-Ohio Valley Study), and the general population casecontrol study nested within the PLCO cohort. The studies set in the Mid-Ohio Valley consistently showed increased risk of kidney cancer related to PFOA exposure. A clear increase in risk of renal cell carcinoma (which accounts for 80-90% of kidney cancers) with indication of an exposure-response relationship was seen in the PLCO cohort, which had much lower exposure than the Mid-Ohio Valley study. In contrast, no increase in kidney cancer incidence

or mortality was seen in the occupational studies in Minnesota or in the International TFE (tetrafluoroethylene) cohort, which were considered to be less informative because they were small, were subject to survivor effects, and/or had exposure assessment limitations. Two other prospective studies in general population cohorts provided equivocal evidence for renal cell carcinoma: in the Multiethnic Cohort (MEC), which was considered informative, positive findings were seen only in White participants, but not overall or in African-American, Japanese-American, Latino, or Native Hawaiian participants; and in the Lifelink subcohort of the Cancer Prevention Study II cohort, for which there were concerns about survivor bias, positive findings were seen only among women exposed to PFOA. Taken together, the body of epidemiological evidence indicated that a positive association between PFOA and renal cell carcinoma is credible, but positive findings have not been consistently observed among the most informative studies, and chance, bias, and confounding by other PFAS in some of the studies could not be ruled out with reasonable confidence. For other subtypes of kidney cancer, no conclusions could be drawn about an association with PFOA.

For cancer of the testis, the most informative studies for the evaluation of PFOA were the set of Mid-Ohio Valley studies, a study of Air Force servicemen with exposure levels similar to those in the general population of the USA, and the ecological analysis of orchiectomies in relation to average serum PFOA concentrations, conducted in the Veneto region, Italy, among municipalities with different levels of PFOA contamination. A positive finding was observed in the Mid-Ohio Valley study and in the Veneto ecological analysis, but not in the Air Force study overall. Mortality studies were deemed to be less informative, because of the high survivability of testicular cancer and the unknown impact of determinants of survival. In summary, there were indications in two independent populations for

an increased risk of testicular cancer associated with PFOA serum concentrations in residents exposed at high levels. In the third informative study, a null association was seen, but exposure levels were at background in this population, which meant that a low exposure contrast existed in the population, making a positive effect, if present, difficult to detect, and did not preclude effects at higher levels of exposure. Overall, the Working Group concluded that a positive association between PFOA and testicular cancer is credible; however, chance and/or bias could not be ruled out as explanations for these findings, given the small number of cases in the few available studies, concerns about co-exposure to other PFAS compounds, and the fact that one of the positive studies was of ecological design.

For breast cancer, most epidemiological studies gave generally null results for all types of breast cancer combined. However, the epidemiological studies with prospective serum samples for PFOA showed a slightly elevated but uncertain association with PFOA. The two most informative studies were null overall but were the only prospective studies that examined postmenopausal breast cancer cases by estrogen receptor/progesterone receptor (ER/PR) status. Both found nonlinear positive associations with ER-negative and PR-negative postmenopausal breast cancer. The statistical power was low in studies examining associations with specific tumour subtypes or stratified by levels of endogenous hormone levels (pre- or postmenopausal cancer), limiting the ability to identify causal associations. Moreover, there were few data on risk of breast cancer above background levels of PFOA exposure. Overall, the available epidemiological evidence was not considered consistent enough to permit a conclusion about the presence of a causal association between exposure to PFOA and breast cancer.

For other cancer types, there was little consistent evidence of an association with PFOA, and the results were considered inconclusive regarding the presence or absence of a causal association.

5.2.2 PFOS

For PFOS, there were fewer available studies than for PFOA. The evidence was suggestive but sparse or inconsistent for three cancer sites: the testis, thyroid gland, and breast. For breast cancer, there was little evidence of an association between PFOS exposure and all types of breast cancer combined. However, the two most informative studies, one from France and one from the USA, which were the only prospective studies to examine the association by hormone receptor breast tumour subtype, found an imprecise but increased risk of hormone receptor-positive breast cancers associated with higher levels of PFOS. However, there were null findings among postmenopausal women in two cohorts from China and the USA, for which there was no stratification by receptor status (most postmenopausal breast cancers are hormone receptor-positive). Given the inconsistencies across studies, the Working Group considered that the available evidence on risk of breast cancer conferred by PFOS exposure was inconclusive.

For testicular cancer, the only informative studies were conducted among the Air Force servicemen in the USA and in the population exposed to contaminated drinking-water near a military airfield in Ronneby, Sweden. In the Air Force study, overall a positive but imprecise association was observed for PFOS exposure, after controlling for exposure to PFOA and other PFAS compounds. For the Ronneby study (in which PFOS levels were much higher than PFOA levels), a positive association was observed for testicular cancer, but co-exposure to perfluorohexanesulfonic acid (PFHxS) was a concern. For thyroid cancer, some positive evidence related to PFOS

exposure came from the less-informative occupational studies; among women in the Ronneby Register cohort study in Sweden; and in a hospital-based case-control study in New York, USA, in which exposure was at background levels. But in a well-conducted population-based study conducted among women in Finland who were exposed at background levels, findings for PFOS were null after adjusting for other PFAS compounds. There was evidence of an inverse association in two case-control studies in China that were considered less informative. For kidney cancer, there were several informative studies, but the findings were largely null. Overall, the evidence for all cancer types was considered to be inconclusive for PFOS exposure.

5.3 Cancer in experimental animals

5.3.1 PFOA

Treatment with PFOA caused an increase in the incidence of an appropriate combination of benign and malignant neoplasms, in both sexes of a single species, in a well-conducted study that complied with Good Laboratory Practice (GLP).

PFOA was administered by oral administration (in feed) in one well-conducted study that complied with GLP, in male and female Sprague-Dawley rats. In males, there was a significant increase in the incidence of hepatocellular adenoma (includes multiple), with a significant positive trend. There was a significant positive trend in the incidence of hepatocellular carcinoma. There was a significant positive trend in the incidence of hepatocellular adenoma or carcinoma (combined), with the incidence being significantly increased. There was a significant positive trend in the incidence of acinar cell adenoma of the pancreas (includes multiple), with the incidence being significantly increased. There was a significant positive trend in the incidence of acinar cell adenomas or adenocarcinoma (combined) of the pancreas, with the incidence being significantly increased. In females, there was a significant increase in the incidence of adenocarcinoma of the uterus. There was significant positive trend in incidence of pancreatic acinar cell adenoma or adenocarcinoma (combined).

In another well-conducted study that complied with GLP, PFOA was administered in the feed of male and female Sprague-Dawley rats. PFOA increased the incidence of testicular Leydig cell adenoma in males.

In a non-GLP study on oral administration (in feed) in male Sprague-Dawley rats only, PFOA increased the incidence of hepatocellular adenoma, Leydig cell tumours, and pancreatic acinar cell adenoma. In a study in female CD-1 mice treated by gavage, there was a positive trend in the incidence of liver haemangiosarcoma in females. PFOA was shown to promote hepatocarcinogenesis in two feeding studies in male Wistar rats and two feeding studies in rainbow trout.

5.3.2 PFOS

Treatment with PFOS caused an increase in the incidence of an appropriate combination of benign and malignant neoplasms in one sex (female) of a single species (rat) in a well-conducted study that complied with GLP.

PFOS was administered by oral administration (in feed) in one study that complied with GLP, in male and female Sprague-Dawley rats. In males, there was a significant positive trend in the incidence of hepatocellular adenoma, with the incidence being significantly increased. In females, there was a significant positive trend and significant increase in the incidence of hepatocellular adenoma, and hepatocellular adenoma or carcinoma (combined).

PFOS was also shown to promote hepatocarcinogenesis in one feeding study in male and female rainbow trout.
5.4 Mechanistic evidence

5.4.1 PFOA

Regarding the absorption, distribution, metabolism, and excretion of PFOA, data were available from studies in humans and from experimental systems. Studies in experimental animals demonstrated high bioavailability after oral exposure, which was presumed to be similar in humans. Absorption via dermal and inhalation routes has been demonstrated in rodents; in humans, there is some evidence that these exposure routes may also be relevant. On the basis of its structure and physicochemical properties, PFOA is unlikely to readily diffuse across cellular membranes; membrane transporters mediate tissue distribution and cell uptake. PFOA can bind to specific proteins, including albumin in serum and liver-type fatty acid-binding protein (L-FABP). Partitioning of PFOA to the liver and kidney can differ across species. There is no evidence in humans or experimental animals that PFOA is biotransformed; PFOA is eliminated by excretion. PFOA undergoes enterohepatic recirculation. Biliary and urinary excretion are the major elimination pathways in humans, with women of reproductive age also eliminating PFOA via blood loss during menstruation, placental transfer to the fetus, and lactational transfer to infants. Urinary excretion is predominant in rodents. In humans, half-lives are in the order of years; half-lives in experimental animals range from hours to months. The basis of species differences in distribution and elimination is not well understood.

There was consistent and coherent evidence that PFOA exhibits key characteristics of carcinogens.

PFOA induces epigenetic alterations. Consistent and coherent evidence came from numerous studies in exposed humans showing that exposure to PFOA alters DNA methylation. Several studies using umbilical cord and peripheral blood leukocytes, or dried blood spots from exposed humans, showed associations between blood PFOA and gene-specific methylation. A robust human epigenome-wide association study showed persistence of PFOAassociated 5'-C-phosphate-G-3' dinucleotide (CpG) methylation between birth and adolescence. This study was of great importance as it investigated developmental reprogramming that may influence human cancer susceptibility. In additional studies in exposed humans, alterations were found in the expression of cancer-related microRNAs (miRNAs) in relation to PFOA exposure. There were no data in primary human cells. Consistent and coherent evidence from experimental systems, both in vivo and in vitro, suggested that PFOA induced changes in DNA methylation, histone modifications, or miRNA expression in multiple tissues, including the liver or kidney.

PFOA is immunosuppressive. Consistent and coherent evidence from multiple well-conducted studies in different populations of exposed humans, including children and adults, demonstrated that exposure to PFOA is associated with increased risk of infectious disease and decreased vaccine response to diverse antigens. These findings were corroborated by consistent and coherent evidence from studies in primary human cells showing that PFOA decreases the production of cytokines and reduces lymphoproliferation. Additionally, consistent and coherent evidence from multiple studies in rodents has demonstrated that PFOA administration alters antibody responses to T-cell dependent antigens. In some studies in rodents, alterations in leukocyte populations were reported.

PFOA induces oxidative stress. The few available studies in exposed humans were not informative. There was consistent and coherent evidence in human primary cells that PFOA exposure increases reactive oxygen species (ROS) production, alters antioxidant function, or increases markers of lipid peroxidation. Consistent and coherent evidence from experimental systems showed induction of oxidative stress by PFOA, including increased levels of oxidatively damaged DNA in cell lines or 8-oxo-2'-deoxyguanosine (8-oxodG) in the urine and liver in rodents. Several studies in experimental systems showed that biomarkers of oxidative stress induced by PFOA were reduced by co-treatment with antioxidants.

PFOA modulates receptor-mediated effects. Data were available for peroxisome proliferatoractivated receptors alpha and gamma (PPARa, PPARy), constitutive androstane receptor/pregnane X receptor (CAR/PXR), hepatocyte nuclear factor 4 alpha (HNF4a), aryl hydrocarbon receptor (AHR), estrogen, androgen, thyroid, progesterone, and glucocorticoid pathways. In exposed humans, the data were suggestive of an association between PFOA exposure and modulation of thyroid, androgen, and progesterone pathways. Data for the remaining receptor pathways in exposed humans were sparse or absent. Consistent and coherent evidence in human primary cells showed that PFOA modulates the action of PPARa and CAR/PXR. Data for human primary cells suggested that PFOA modulates the action of both estrogen and PPARy. Data for the remaining receptor pathways in exposed human primary cells were sparse. There was consistent and coherent evidence from numerous studies performed in experimental systems, including human cell lines, that exposure to PFOA modulates the activity of PPARa and CAR/PXR, as well as PPARy. There was suggestive evidence that PFOA alters serum estradiol and testosterone concentrations in rodents. There was a paucity of information for PFOA in other receptor pathways in experimental systems.

PFOA alters cell proliferation, cell death, or nutrient supply. The evidence in PFOA-exposed humans was suggestive on the basis of highthroughput metabolomic analyses showing alterations in pathways related to nutrient and energy supply. Evidence from primary human cells suggested that PFOA increases cell proliferation. Transcriptomic analyses from primary human cells suggested that PFOA modulates gene signalling pathways involved in cell proliferation and oncogenesis. Metabolomic analyses from primary human cells suggested that PFOA increases activity in glycolytic pathways. Consistent and coherent evidence in multiple experimental systems showed that PFOA induces cell proliferation, migration, or invasion in human cell lines and cell proliferation or hyperplasia in multiple tissues in rodents, including in PPARα-null mice.

There was suggestive evidence that PFOA is genotoxic. A single study in exposed humans reported increased levels of DNA strand breaks; results from other studies using less-relevant end-points were mixed. The results of studies in human primary cells were negative. Evidence in experimental systems suggested that PFOA causes DNA damage. Available studies in rodents in vivo showed largely negative results for DNA damage and micronucleus assays.

There was suggestive evidence that PFOA induces chronic inflammation. Data in exposed humans were not informative. In most studies in human primary cells, decreased production of pro-inflammatory markers occurred after PFOA exposure. The results of several studies in rodents suggested that PFOA induces small increases in severity or incidence of chronic inflammation in the stomach, liver, or pancreas. The results of studies of inflammatory markers in experimental systems were mixed, with results differing depending on the model, tissue, and assay.

There was a paucity of data for the following key characteristics: is electrophilic or metabolized to an electrophile, alters DNA repair or genomic instability, or causes immortalization.

PFOA and its ammonium salt were tested in the Toxicology Testing in the 21st Century (Tox21) and Toxicity Forecaster (ToxCast) research programmes. However, the analytical purity quality control rating for these data was labelled "unknown/inconclusive".

5.4.2 PFOS

Regarding the absorption, distribution, metabolism, and excretion of PFOS, data were available from studies in humans and from experimental systems. Studies in experimental animals demonstrated high bioavailability after oral exposure, which was presumed to be similar in humans. Absorption via dermal and inhalation routes has been demonstrated in rodents; in humans, these exposure routes may also be relevant. On the basis of its structure and physicochemical properties, PFOS is unlikely to readily diffuse across cellular membranes; membrane transporters mediate tissue distribution and cell uptake. PFOS can bind to specific proteins, including albumin in serum and L-FABP. Partitioning of PFOS to the liver can differ across species. There is no evidence in humans or experimental animals that PFOS is biotransformed; PFOS is eliminated by excretion. PFOS undergoes enterohepatic recirculation. Biliary and urinary excretion are the major elimination pathways in humans, with women of reproductive age also eliminating PFOS via blood loss during menstruation, placental transfer to the fetus, and lactational transfer to infants. Urinary excretion is predominant in rodents. In humans, half-lives are on the order of years; half-lives in experimental animals range from weeks to months. The basis of species differences in distribution and elimination is not well understood.

There was consistent and coherent evidence that PFOS exhibits key characteristics of carcinogens.

PFOS induces epigenetic alterations. Consistent and coherent evidence from numerous studies in exposed humans showed that exposure to PFOS alters DNA methylation. Several studies using umbilical cord and peripheral blood leukocytes, or dried blood spots from exposed humans, showed associations between blood PFOS and gene-specific methylation. A robust human epigenome-wide association study showed persistence of PFOS-associated CpG methylation between birth and adolescence. This study was of great importance as it investigated developmental reprogramming that may influence human cancer susceptibility. In additional studies in exposed humans, alterations were found in the expression of cancer-related miRNAs in relation to PFOS exposure. There were no data in primary human cells. Consistent and coherent evidence from studies in experimental systems, both in vivo and in vitro, suggested that PFOS induced changes in DNA methylation, histone modifications, or miRNA expression in multiple tissues, including the liver or kidney.

PFOS is immunosuppressive. Consistent and coherent evidence from multiple well-conducted studies in different populations of exposed humans, including children and adults, demonstrated that exposure to PFOS is associated with increased risk of infectious disease and decreased vaccine response to diverse antigens. These findings were corroborated by consistent and coherent evidence from studies in primary human cells showing that PFOS decreases production of cytokines and reduces lymphoproliferation. Additionally, consistent and coherent evidence from multiple studies in rodents has demonstrated that PFOS administration alters antibody responses to T-cell dependent antigens. In some studies in rodents, alterations in leukocyte populations were reported. One study in mice showed that PFOS increased morbidity and mortality after influenza A infection.

PFOS induces oxidative stress. There was suggestive evidence in exposed humans that PFOS induces oxidative stress, with several studies showing associations between PFOS and various oxidative stress markers in serum or urine. Two of three studies that measured urinary 8-oxodG with high specificity gave positive results. There was consistent and coherent evidence in human primary cells that PFOS exposure increases ROS production, alters antioxidant function, or increases markers of lipid peroxidation. One study in human primary cells showed that biomarkers of oxidative stress induced by PFOS were reduced by co-treatment with antioxidants. There was consistent and coherent evidence from experimental systems that PFOS induces oxidative stress. In cell lines, PFOS increased levels of ROS production. PFOS increased markers of lipid peroxidation and altered antioxidant function in rodent tissues. Several studies in experimental systems showed that biomarkers of oxidative stress induced by PFOS were reduced by co-treatment with antioxidants.

PFOS modulates receptor-mediated effects. Data were available for PPARa, PPARy, CAR/PXR, HNF4a, AHR, estrogen, androgen, thyroid, progesterone, and glucocorticoid pathways. In exposed humans, the data were suggestive of an association between PFOS exposure and modulation of thyroid, estrogen, androgen, progesterone, and glucocorticoid pathways. Data for the remaining receptor pathways in exposed humans were sparse or absent. Consistent and coherent evidence in human primary cells showed that PFOS modulates the action of PPARa and CAR/PXR. Data from human primary cells suggested that PFOS modulates the PPARy pathway. Data for the remaining receptor pathways in human primary cells were sparse. Consistent and coherent evidence came from numerous studies in experimental systems, including human cell lines, and showed that exposure to PFOS modulates the activity of PPARa and CAR/PXR. Consistent and coherent evidence from experimental systems showed that PFOS modulates the androgen and thyroid pathways. Evidence from human cell lines and receptor assays suggested that PFOS modulates the PPARy pathway. Several studies in experimental systems suggested that PFOS modulates the estrogen pathway. There was a paucity of information for PFOS in other receptor pathways in experimental systems.

PFOS alters cell proliferation, cell death, or nutrient supply. The evidence in PFOS-exposed humans was suggestive on the basis of highthroughput metabolomic analyses showing alterations in pathways related to nutrient and energy supply. Evidence from primary human cells suggested that PFOS increases cell proliferation, migration, or invasion. Transcriptomic analyses from primary human cells suggested that PFOS modulates gene signalling pathways involved in cell proliferation and oncogenesis. Metabolomic analyses from primary human cells suggested that PFOS increases activity in glycolytic pathways. Consistent and coherent evidence in multiple experimental systems showed that PFOS induces cell proliferation, migration, or invasion in human cell lines and cell proliferation or hyperplasia in multiple tissues in rodents.

There was suggestive evidence that PFOS is genotoxic. Results from the few studies in exposed humans were mixed. The results of studies in human primary cells were negative. Evidence in experimental systems suggested that PFOS causes DNA damage. Studies in rodents showed mixed results for DNA damage and micronucleus assays.

There was suggestive evidence that PFOS induces chronic inflammation. Data in exposed humans were not informative. In most studies in human primary cells, decreased production of pro-inflammatory markers occurred after PFOS exposure. The results of studies in rodents suggested that PFOS increases inflammation. The results of studies of inflammatory markers in other experimental systems were mixed, with results differing depending on the model, tissue, and assay.

There was a paucity of data for the following key characteristics: is electrophilic or metabolized to an electrophile, alters DNA repair or genomic instability, or causes immortalization. PFOS and its potassium salt were tested in the assay battery of the Tox21 and ToxCast research programmes. However, the analytical purity quality control rating for these data was labelled "unknown/inconclusive".

6. EVALUATION AND RATIONALE

6.1 Cancer in humans

There is *limited evidence* in humans for the carcinogenicity of perfluorooctanoic acid (PFOA). Positive associations have been observed between PFOA and renal cell carcinoma and cancer of the testis.

There is *inadequate evidence* in humans regarding the carcinogenicity of perfluorooc-tanesulfonic acid (PFOS).

6.2 Cancer in experimental animals

There is *sufficient evidence* in experimental animals for the carcinogenicity of PFOA.

There is *limited evidence* in experimental animals for the carcinogenicity of PFOS.

6.3 Mechanistic evidence

There is *strong evidence* that PFOA exhibits multiple key characteristics of carcinogens in exposed humans, in human primary cells, and in experimental systems.

There is *strong evidence* that PFOS exhibits multiple key characteristics of carcinogens in exposed humans, in human primary cells, and in experimental systems.

6.4 Overall evaluation

PFOA is *carcinogenic to humans* (Group 1). PFOS is *possibly carcinogenic to humans* (Group 2B).

6.5 Rationale

6.5.1 PFOA

The Group 1 evaluation for PFOA is based on the combination of sufficient evidence for cancer in experimental animals and strong mechanistic evidence of key characteristics of carcinogens in exposed humans. The evidence for cancer in experimental animals was *sufficient* because exposure to PFOA caused an increase in the incidence of an appropriate combination of benign and malignant neoplasms in both sexes of a single species (rat) in one study that complied with GLP. The mechanistic evidence was strong in exposed humans because PFOA induces epigenetic alterations and is immunosuppressive. In exposed humans, PFOA induces epigenetic alterations in the form of gene-specific methylation and cancer-related miRNAs. These effects are supported by evidence of epigenetic alterations in multiple experimental systems. In exposed humans, PFOA is immunosuppressive, increasing risk of infectious disease and decreasing vaccine response to diverse antigens. These effects are supported by evidence

of immunosuppression in human primary cells and experimental systems. In addition, in human primary cells and experimental systems, PFOA induces oxidative stress and modulates receptor-mediated effects. Additionally, in experimental systems, PFOA alters cell proliferation, cell death, or nutrient supply.

Also, for PFOA, the evidence for cancer in humans was found to be *limited* for renal cell carcinoma and cancer of the testis. Despite the increase in the number of available human cancer studies since the previous evaluation by the IARC Monographs, the results were somewhat inconsistent across the studies. For renal cell carcinoma, positive findings were observed in three studies conducted in partly overlapping occupationally and environmentally exposed populations and in a fourth population with background exposure. However, positive findings were not observed overall in two other background-exposed populations. For testicular cancer, there were two studies with positive findings: one cohort study and a second ecological study that had limitations. For other cancer types, there were only sporadic positive findings in the informative studies (e.g. breast), and for all these other cancer types, the evidence was inadequate.

6.5.2 PFOS

The Group 2B evaluation for PFOS is based on strong mechanistic evidence. There is strong evidence that PFOS exhibits multiple key characteristics of carcinogens in exposed humans, human primary cells, and experimental systems. There is strong evidence that PFOS in exposed humans induces epigenetic alterations in the form of gene-specific methylation and cancer-related miRNAs. These effects are supported by evidence of epigenetic alterations in multiple experimental systems. In exposed humans, PFOS is immunosuppressive, increasing risk of infectious disease and decreasing vaccine response to diverse antigens. These effects are supported by evidence of immunosuppression in human primary cells and experimental systems. In human primary cells and experimental systems, PFOS modulates receptor-mediated effects and induces oxidative stress. Additionally, in experimental systems, PFOS alters cell proliferation, cell death, or nutrient supply.

In addition, the evidence for cancer in experimental animals was *limited*. Exposure to PFOS caused an increase in the incidence of an appropriate combination of benign and malignant neoplasms in one sex (female) of a single species (rat) in a study that complied with GLP. The evidence regarding cancer in humans was found to be *inadequate*, because among the relatively few available studies, positive findings were seen only sporadically and inconsistently for a few cancer sites (i.e. breast, testis, and thyroid).

LIST OF ABBREVIATIONS

2-AAF	2-acetylaminofluorene
ACGIH	American Conference of Governmental Industrial Hygienists
ACOX	acyl-coenzyme A oxidase
ACS	American Cancer Society
ADME	absorption, distribution, metabolism, or excretion
AFB ₁	aflatoxin B ₁
AFFF	aqueous film-forming foam
AHR	aryl hydrocarbon receptor
AOR	adjusted odds ratios
APFO	ammonium perfluorooctanoate
AR	androgen receptor
ARE	antioxidant responsive element
ASTM	American Society for Testing and Materials
ATBC	Alpha-Tocopherol, Beta-Carotene Cancer Prevention
ATP	adenosine triphosphate
ATSDR	Agency for Toxic Substances and Disease Registry
AUC	area under the curve
BMI	body mass index
BrdU	bromodeoxyuridine
bw	body weight
cAMP	cyclic adenosine monophosphate
CAR	constitutive androstane receptor
CAT	catalase
CBD	cannabidiol
CEPA	Canadian Environmental Protection Act
CERCLA	Comprehensive Environmental Response, Compensation, and Liability Act
CGA	chlorogenic acid
CHDS	Child Health and Development Studies
СНО	Chinese hamster ovary
CI	confidence interval
CIOB	Chemicals in Our Bodies
CNBCSP	Chinese National Breast Cancer Screening Program
CoA	coenzyme A
ConA	concanavalin A

CONTAM	European Food Safety Authority Panel on Contaminants in the Food Chain
CpG	5'-C-phosphate-G-3'-dinucleotide
CPS	Cancer Prevention Study
CRP	C-reactive protein
CS	collagen sandwich
CTS	California Teachers Study
CV	coefficient of variation
СҮР	cytochrome P450
DCF	2',7'-dichlorofluorescein
DCF-DA	2',7'-dichlorofluorescein diacetate
DCFH-DA	2',7'-dichlorodihydrofluorescein diacetate
DEN	diethylnitrosamine
DFTJ	Dongfeng-Tongji
dG	2'-deoxyguanosine
DHEA	dehydroepiandrosterone
diPAP	polyfluoroalkyl phosophate diester
DMBA	7,12-dimethylbenz[a]anthracene
DMR	differentially methylated region
DMSO	dimethyl sulfoxide
DNMT	DNA methyltransferase
DoD	Department of Defense
DoDSR	Department of Defense Serum Repository
DOX	doxycycline
DTH	delayed-type hypersensitivity
EC	European Commission
EC ₅₀	half-maximal effective concentration
ECA	Environment and Childhood Asthma
ECCC	Environment and Climate Change Canada
ECF	electrochemical fluorination
ECHA	European Chemicals Agency
EFSA	European Food Safety Authority
eGFR	estimated glomerular filtration rate
ELISA	enzyme-linked immunosorbent assay
EPCRA	Emergency Planning and Community Right-to-Know Act
ER	estrogen receptor
ERS	endoplasmic reticulum stress
ESI	electrospray ionization
N-EtFOSAA	N-ethyl-perfluorooctane sulfonamido acetic acid
EU	European Union
EWAS	epigenome-wide association analysis study
FDR	false discovery rate
FMC	Finnish Maternity Cohort
Fpg	formamidopyrimidine-DNA glycosylase
FSANZ	Food Standards Australia New Zealand
FT3	free triiodothyronine
FT4	free thyroxine
FTOH	fluorotelomer alcohol
GBCA	Genetic and Biomarkers study for Childhood Asthma
GC	gas chromatography
GDP	guanosine diphosphate

GF	glomerular function
GFR	glomerular filtration rate
GGT	gamma-glutamyl transferase
GJIC	gap junctional intercellular communication
GLP	Good Laboratory Practice
GM	geometric mean
GM-CSF	granulocyte-macrophage colony-stimulating factor
GPx	glutathione peroxidase
GRULAC	Group of Latin America and the Caribbean
GSH	glutathione
GSPE	grape seed proanthocyanidin extract
GSR	glutathione reductase
GSSG	oxidized glutathione
GST	glutathione S-transferase
vH2AX	phosphorylated H2A histone family member X
HBM	Human Biomonitoring
HBM4EU	Human Biomonitoring for Europe
HCC	henatocellular carcinoma
HETE	hydroxyeicosatetraenoic acid
HED	high fat diet
HCE	hepatocyte growth factor
Hib	Hagmothilus influenza type b
HMC	human mast cell
HMVEC	human miast cen
HO_1	ham ovygenase 1
HOME	Health Outcomes and Measures of the Environment
HDIC	high performance liquid chrometography
HP LC	hazard ratio
HIVEC	human umbilical vain and thalial call
IAPC	International Agency for Desearch on Cancor
IC	half maximal inhibitory concentration
ICC	
IFIN-γ I=	imeneron gamma
Ig	Immunoglobulin Illin die Kide Deredenwerst Sterke
IKIDS	
IL	interleukin
INHAND	International Harmonization of Nomenclature and Diagnostic Criteria for Lesions in Rats and Mice
INOS	inducible nitric oxide synthase
INSR	insulin receptor
i.p.	intraperitoneal
IQR	interquartile range
IRR	incidence rate ratio
IsoF	isoturan
IsoP	isoprostane
TTRC	Interstate Technology and Regulatory Council
i.v.	intravenous
JEM	job-exposure matrix
KC	key characteristic of carcinogens
KEEP	Korean Elderly Environmental Panel
KLH	keyhole limpet haemocyanin

LC ₅₀	median lethal concentration
LC-MS/MS	liquid chromatography-tandem mass spectrometry
L-FABP	liver fatty acid-binding protein
LH	luteinizing hormone
LINE-1	long interspersed nuclear element 1
LLE	liquid–liquid extraction
LMIC	low- or middle-income country
LOD	limit of detection
LOQ	limit of quantification
LOX	lipoxygenase
LPS	lipopolysaccharide
LRTI	lower respiratory tract infection
LTL	leukocyte telomere length
LWBC	Laizhou Wan (Bay) birth cohort
MCP-1	monocyte chemoattractant protein
MDA	malondialdehyde
MDL	method detection limit
MEC	Multiethnic Cohort
МНС	major histocompatibility complex
miRNA	microRNA
mRNA	messenger RNA
MLQ	method limit of quantification
MNNG	N-methyl-N'-nitro-N-nitrosoguanidine
MRL	minimum reporting level
MRP	multidrug resistance protein
MS	mass spectrometry
MS/MS	tandem mass spectrometry
MTT	3-[4,5-dimethylthiazol-2-yl]-2,5 diphenyl tetrazolium bromide
NAC	N-acetylcysteine
NADPH	nicotinamide adenine dinucleotide phosphate
NAFLD	non-alcoholic fatty liver disease
NAR	naringin; 4',5,7-trihydroxyflavonone-7-rhamnoglucoside
NASEM	National Academies of Sciences, Engineering, and Medicine
NDAA	National Defense Authorization Act
NDI	National Death Index
NESHAP	National Emissions Standards for Hazardous Air Pollutants
NeuroP	neuroprostane
NGF	nerve growth factor
NHANES	National Health and Nutrition Examination Survey
NHL	non-Hodgkin lymphoma
NHMRC	National Health and Medical Research Council
NICHD	National Institute of Child Health and Human Development
NK	natural killer
NO	nitric oxide
NOx	nitrogen oxides
8-NO ₂ Gua	8-nitrosoguanine
NOAEL	no observed adverse effect level
NPDES	National Pollutant Discharge Elimination System
NTCP	Na+/taurocholate cotransporting polypeptide
NTP	National Toxicology Program

OAT	organic acid transporter
OATP	organic anion transporting polypeptide
OCC	Odense Child Cohort
OCM	organotypic culture model
8-OHdG	8-hydroxy-2'-deoxyguanosine
OR	odds ratio
8-oxodG	8-oxo-2'-deoxyguanosine
PAH	polycyclic aromatic hydrocarbon
PanIN	pancreatic intraepithelial neoplasia
PBDE	polybrominated diphenyl ether
РВМС	peripheral blood mononuclear cells
PBPK	physiologically based pharmacokinetic
РСВ	polychlorinated biphenyl
PCDD	2,3,7,8-substituted polychlorinated dibenzodioxin
PCDF	2.3.7.8-substituted polychlorinated dibenzofuran
PCNA	proliferating cell nuclear antigen
РСО	palmitovl coenzyme A oxidase
PCR	polymerase chain reaction
PFAA	perfluoroalkyl acid
PFAS	perfluoroalkyl and polyfluoroalkyl substances
PFHpS	perfluoroheptane sulfonate
PFHxS	perfluorohexanesulfonic acid
PFNA	perfluorononanoic acid
PFOA	perfluorooctanoic acid
PFOS	perfluorooctanesulfonic acid
PGF	prostaglandin
РНА	phytohaemagglutinin
РКА	protein kinase A
РКС	protein kinase C
PLCO	Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial
PM	particulate matter
PM.	particulate matter with diameter $< 2.5 \mu m$
POP	persistent organic pollutant
POSE	perfluorooctane sulfonyl fluoride
PPAR	peroxisome proliferator-activated receptor
ppb	parts per billion
PPE	personal protective equipment
ppm	parts per million
PR	progesterone recentor
PTFE	polytetrafluoroethylene
PUF	polyurethane foam
PWS	public water systems
PXR	pregnane X recentor
aPCR	quantitative polymerase chain reaction
QuEChERS	quick easy cheap effective rugged safe
RACK-1	receptor for activated C kinase 1
RBC	red blood cell
RCC	renal cell carcinoma
RD	human embryonal rhabdomyosarcoma cell line
REACH	Registration Evaluation Authorisation and Restriction of Chemicals
KLIIGH	registration, Evaluation, ruthorisation and restriction of Chemicals

RL	reporting limit
RNA	ribonucleic acid
tRNA	transfer RNA
RNS	reactive nitrogen species
ROS	reactive oxygen species
RONS	reactive oxygen and nitrogen species
RPE	retinal pigment epithelial
RR	rate ratio
RT-PCR	reverse transcription-polymerase chain reaction
RSV	respiratory syncytial virus
SCAP	sterol regulatory element-binding protein cleavage-activating protein
SD	standard deviation
SEER	Surveillance. Epidemiology, and End Results
SEM	systemic evidence man
SHBG	sex hormone-hinding globulin
SIR	standardized incidence ratio
SMBCS	Shevang Mini Birth Cohort Study
SMR	standardized mortality ratio
SNILLD	Significant New Lise Pulse
SOD	superovide disputese
SOD	superoxide distinutase
SPE	show red blood cell
SKDC St A D	streep red blood cell
SIAK	short taum avecaura limit
SIEL T	balf life
TAD	
TAD	total administered dose
TBARS	thiobarbituric acid-reactive substance
TCA	tricarboxylic acid
TCDD	2,3,7,8-tetrachlorodibenzo-p-dioxin
TCR	T-cell receptor
TDAR	T-cell-dependent antibody response
TDI	tolerable daily intake
TFE	tetrafluoroethylene
TGCT	testicular germ cell tumour
TH	thyroid hormone
TK	toxicokinetic
TL	telomere length
TNF-a	tumour necrosis factor alpha
TNP	trinitrophenyl
TPOAb	thyroid peroxidase antibody
TR	thyroid hormone receptor
TRI	Toxics Release Inventory
TSH	thyroid-stimulating hormone
TSCA	Toxic Substances Control Act
TT3	total triiodothyronine
TT4	total thyroxine
TTR	transthyretin
TWA	time-weighted average
TWI	tolerable weekly intake

UBA	Umweltbundesamt (German Environment Agency)
UCMR 3	Third Unregulated Contaminant Monitoring Rule
UK	United Kingdom
UNEP	United Nations Environment Programme
UPR	unfolded protein response
US	United States
USA	United States of America
US EPA	United States Environmental Protection Agency
US FDA	United States Food and Drug Administration
$V_{ m d}$	volume of distribution
vP	very persistent
WBC	white blood cell
WT	wildtype

ANNEX 1. SUPPLEMENTARY MATERIAL FOR SECTION 1, EXPOSURE CHARACTERIZATION

These supplementary online-only tables are available from: <u>https://publications.iarc.who.int/636</u>.

Please report any errors to imo@iarc.who.int.

The following tables were produced in draft form by the Working Group and were subsequently fact-checked but not edited:

Table S1.11	Occurrence of PFOA and PFOS in drinking-water, surface water, groundwater, snow, and ice
Table S1.15	Occupational exposure to PFOA and PFOS measured in biological matrices
Table S1.22	Exposure assessment review and critique for epidemiological studies on cancer in humans exposed to PFOA and PFOS
Table S1.23	Exposure assessment review and critique for mechanistic studies on cancer and exposure to PFOA and PFOS

The following tables were produced in draft form by the Working Group, fact-checked, and edited:

- Table S1.13Occurrence of PFOA in foodTable S1.14Occurrence of PFOA in food
- Table S1.14Occurrence of PFOS in food

ANNEX 2. ACTIONS AND REGULATIONS FOR THE ELIMINATION OF PFAS WORLDWIDE

In the European Union (EU), the Stockholm Convention on Persistent Organic Pollutants (POPs) places restrictions on perfluorooctanesulfonic acid (PFOS) and has targeted global elimination of perfluorooctanoic acid (PFOA); these mandates are implemented through the Persistent Organic Pollutants (POPs) Regulation (ECHA, 2023c). The first POP regulation (European Commission, EC) 850/2004 was published in 2004, but only in the 2019 recast (POP Regulation (EU) 2019/1021) was PFOS included in Annex 1 of the regulation. Annex 1 is dedicated to substances that should be allowed to be manufactured and used only as closed-system site-limited intermediates if an annotation to that effect is expressly entered in the relevant Annex and if the manufacturer demonstrates to the Member State concerned that the substance is manufactured and used only under strictly controlled conditions (EU, 2019). In the scope of the REACH (Registration, Evaluation, Authorisation and Restriction of Chemicals) regulation, PFOA and ammonium perfluorooctanoate (APFO) are identified as Substances of Very High Concern under Article 57 I and (d) (ECHA, 2023a). These two per- and polyfluoroalkyl substances (PFAS) already have a harmonized classification and labelling under the Classification, Labelling and Packaging (CLP) Regulation (ECHA, 2023c). In 2020, the European Commission published its Chemicals Strategy for Sustainability, which

states that the use of PFAS is to be phased out in the EU, unless the use is proven to be essential for society (European Commission, 2023; OECD, 2023). A proposal submitted in January 2023, by Germany, Denmark, the Netherlands, Norway, and Sweden that would further restrict the manufacture, placing on the market, and use of a broader group of more than 10 000 PFAS is under consideration by the scientific Committee for Risk Assessment (RAC) and the Committee for Socio-Economic Analysis (SEAC) of the European Chemicals Agency (ECHA) (ECHA, 2023c).

In Canada, both PFOS and its salts and PFOA and its precursors are listed on the Canadian Environmental Protection Act (CEPA) Schedule 1 – List of Toxic Substances. The 2012 Prohibition of Certain Toxic Substances Regulation from Environment and Climate Change Canada (ECCC) restricts the manufacture, use, sale, or import of products containing PFAS (including PFOA and PFOS), such as aqueous film-forming foam (AFFF) and personal care products. In 2022, changes were proposed to this regulation that would remove some exemptions and accommodations (ECCC, 2023).

Under the Emergency Planning and Community Right-to-Know Act (EPCRA) in the United States of America (USA), the Toxics Release Inventory (TRI) collects information reported by companies manufacturing, processing, or otherwise using 100 lbs or more of PFOA or PFOS or other listed PFAS. Additional reporting is required under the Toxic Substances Control Act (TSCA), and there are multiple recent Significant New Use Rules (SNURs) restrictions. Discharge of PFOA and PFOS is controlled under the Clean Water Act National Pollutant Discharge Elimination System (NPDES) permitting system of the United States Environmental Protection Agency (US EPA). In 2022, the US EPA proposed that PFOA and PFOS be designated as hazardous substances under the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) (Office of the Federal Register, 2022). Several states in the USA have also designated PFAS as a hazardous substance and proposed or enacted restrictions on use or limits in products and in the environment (ITRC, 2023a).

In 2020, Australia issued a National PFAS Position Statement that was added to the 2018 Intergovernmental Agreement on a National Framework for Responding to PFAS Contamination. This Position Statement describes a national stance on an intentional phase-out of PFAS from use (Australian Government, 2020).

In 2016, the United States Food and Drug Administration (US FDA) revoked regulations authorizing the remaining uses of PFOA and PFOS in food packaging (81 FR 5, 4 January 2016 and 81 FR 83672, 22 November 2016; US FDA, <u>2022</u>). Since then, multiple states in the USA have enacted bans on PFAS in food packaging materials (ITRC, 2023b). There are also state reporting requirements for PFOS and PFOA in children's products, regulations on PFAS in carpets, and regulations broadly governing the sale of products containing PFAS (ITRC 2023a, b). At present, the most comprehensive law regulating PFAS in consumer products in the USA has been established in Maine. This law includes reporting requirements for manufacturers of products to which certain PFAS (including PFOS and PFOA) are intentionally added, as well as bans

on the sale of carpets, rugs, and fabric treatments containing PFAS (<u>Maine DEP, 2023</u>). However, other states, such as Minnesota, have proposed banning all "nonessential use" of PFAS in products. If enacted, this would prohibit the sale of certain consumer products containing PFAS, including carpets, cleaning products, cookware, cosmetics, dental floss, fabric treatments, juvenile products, menstruation products, textile furnishings, ski wax, and upholstered furniture (<u>MNPCA, 2023</u>).

The class B firefighting foam known as AFFF is the subject of numerous legislative efforts. In the EU, a 2022 proposal to the ECHA to restrict the use of PFAS in firefighting foams is moving forward, with combined opinions from the RAC and SEAC committees to review by the European Commission. This proposal would implement a gradual ban on PFAS in foams in the EU, if adopted, and could reduce PFAS emissions into the environment by around 13 200 tonnes over 30 years (ECHA, 2023b). In Canada, the Prohibition of Certain Toxic Substances Regulations currently allow for certain uses of AFFF, including AFFF that contains PFOA, AFFF contaminated with PFOS in military vessels or firefighting vehicles returning from foreign operations, and AFFF containing residual levels of PFOS (\leq 10 ppm) (<u>ECCC, 2017</u>). Legislation enacted under the 2019-2022 National Defense Authorization Act (NDAA) requires the United States Department of Defense to take certain actions, including surveying technology for AFFF replacement to be included on the approved list for military use (ITRC, 2023b); this list is also used by commercial airports in the USA. Numerous states in the USA have also enacted partial bans on the sale and distribution of AFFF, as well as AFFF collection or buy-back programmes and restrictions on firefighting training with AFFF (ITRC, 2023b). In Australia, some states, including Queensland, South Australia, and New South Wales, have implemented phased restrictions on AFFF, including banning the use of AFFF during training or demonstration and restrictions on the use or sale of AFFF products (<u>NSW EPA, 2023</u>).

References

- Australian Government (2020). Intergovernmental Agreement on a National Framework for Responding to PFAS Contamination. 7 February 2020. Australian Governments. Available from: <u>https://federation.gov.</u> <u>au/about/agreements/intergovernmental-agreementnational-framework-responding-pfas-contamination,</u> accessed 5 September 2023.
- ECCC (2017). Aqueous film-forming foam and the Prohibition of Certain Toxic Substances Regulations. Ontario, Canada: Environment and Climate Change Canada. Available from: <u>https://www.canada.ca/en/ environment-climate-change/services/managementtoxic-substances/list-canadian-environmentalprotection-act/perfluorooctane-sulfonate/ film-forming-foam-prohibition-toxic-substances. html, accessed 26 March, 2023.</u>
- ECCC (2023). Prohibition of Certain Toxic Substances Regulations, 2012: overview. Ontario, Canada: Environment and Climate Change Canada. Available from: https://www.canada.ca/en/environment-climatechange/services/management-toxic-substances/ prohibition-regulations.html, accessed 5 September 2023
- ECHA (2023a). Candidate list of substances of very high concern for authorisation. Helsinki, Finland: European Chemicals Agency. Available from: <u>https://echa.europa. eu/candidate-list-table</u>, accessed 24 March 2023.
- ECHA (2023b). ECHA's committees: EU-wide PFAS ban in firefighting foams warranted. Helsinki, Finland: European Chemicals Agency. Available from: <u>https:// echa.europa.eu/-/echa-s-committees-eu-wide-pfasban-in-firefighting-foams-warranted</u>, accessed 5 September 2023.
- ECHA (2023c). Per- and polyfluoroalkyl substances (PFAS). Helsinki, Finland: European Chemicals Agency. Available from: <u>https://echa.europa.eu/en/web/guest/hot-topics/perfluoroalkyl-chemicals-pfas</u>, accessed 25 March 2023.
- EU (2019). Regulation (EU) 2019/1021 of the European Parliament and of the Council of 20 June 2019 on persistent organic pollutants (recast). Off J Euro Un. L169/45. Available from: <u>https://eur-lex.europa.eu/legal-con</u> tent/EN/TXT/PDF/?uri=CELEX:32019R1021.
- European Commission (2023). Chemicals strategy. The EU's chemicals strategy for sustainability towards a toxic-free environment. Available from: <u>https://</u>

<u>environment.ec.europa.eu/strategy/chemicals-</u> <u>strategy_en</u>, accessed January 2024.

- ITRC (2023a). 8. Basis of regulations. PFAS Technical and Regulatory Guidance Document and Fact Sheets PFAS-1: Washington (DC), USA: Interstate Technology & Regulatory Council, PFAS Team. Available from: <u>https://pfas-1.itrcweb.org/8-basis-of-regulations/#8_2</u>, accessed 11 November 2023.
- ITRC (2023b). PFAS regulatory programs summary table. 8. Basis of regulations. PFAS Technical and Regulatory Guidance Document and Fact Sheets PFAS-1. Washington (DC), USA: Interstate Technology & Regulatory Council, PFAS Team. Available in list of external data tables at: <u>https://pfas-1.itrcweb.org/</u> <u>external-data-tables/</u>, accessed 11 November 2023.
- Maine DEP (2023). PFAS in products. Augusta (ME), USA: Maine Department of Environmental Protection. Available from: <u>https://www.maine.gov/dep/spills/</u> <u>topics/pfas/PFAS-products/#:~:text=Beginning%20</u> <u>January%201%2C%202023%2C%20fabric,in%20</u> <u>the%20State%20of%20Maine</u>, accessed November 2023.
- MNPCA (2023). PFAS ban. St Paul (MN), USA: Minnesota Pollution Control Agency. Available from: <u>https:// www.pca.state.mn.us/get-engaged/pfas-ban</u>, accessed November 2023.
- NSW EPA (2023). Regulation of PFAS firefighting foams. Parramatta, Australia: New South Wales Environment Protection Agency. Available from: <u>https://www.epa.nsw.gov.au/your-environment/</u> <u>contaminated-land/regulation-of-pfas-firefighting-</u> foams#:~:text=The%20Protection%20of%20the%20 <u>Environment,relevant%20authorities%20and%20</u> <u>exempt%20entities</u>, accessed 5 September 2023.
- OECD (2023). Portal on per and poly fluorinated chemicals – country information. Available from: <u>https:// www.oecd.org/chemicalsafety/portal-perfluorinatedchemicals/countryinformation/european-union.htm</u>, accessed 18 March 2023.
- Office of the Federal Register (2022). Designation of perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS) as CERCLA Hazardous Substances. EPA proposed rule. 40 CFR Parts 302. Docket Nos. EPA-HQ-OLEM-2019-0341 and FRL-7204-02-OLEM. *Federal Register*. Vol. 87, No. 171, 6 September 2022. Available from: https://www.govinfo.gov/content/pkg/FR-2022-09-06/pdf/2022-18657.pdf, accessed 6 September 2023.
- US FDA (2022). Per- and polyfluoroalkyl substances (PFAS). Silver Spring (MD), USA: United States Food and Drug Administration. Available from: <u>https://www.fda.gov/food/environmental-contaminants-food/and-polyfluoroalkyl-substances-pfas</u>, accessed 25 March 2023.

ANNEX 3. SUPPLEMENTARY ANALYSES USED IN REVIEWING EVIDENCE ON CANCER IN HUMANS

A3.1 How well does the PFOA concentration in a single serum sample represent long-term exposure in a population with low exposure?

Introduction

Several of the epidemiology studies on perfluoroalkyl and polyfluoroalkyl substance(s) (PFAS) and cancer were cohort studies in the general population, or nested case-control studies within such cohorts, and used a single serum sample per participant to assess exposure. There was little information on how well the PFAS measurement in a single serum sample (typically at baseline) represents longer-term exposure, which is important for studying chronic diseases. In this analysis, summary statistics from two cohorts with repeated measurements of serum perfluorooctanoic acid (PFOA) for the participants in the control groups were used to evaluate the potential impacts of using a single serum sample to represent chronic exposure for each participant.

Methods

The first study, by <u>Rhee et al. (2023a)</u>, was a nested case–control study on prostate cancer, with 675 cases and 675 controls from within the Prostate, Lung, Colorectal, and Ovarian Cancer (PLCO) cohort. In this study, three repeat serum samples were collected from 60 control participants at baseline, 1 years, and 5 years (in 1996, 1997, and 2001). The second study, by Purdue et al. (2023), was a nested case-control study on testicular cancer among Air Force servicemen, with 530 cases and 530 controls. Purdue et al. (2023) had available a second prediagnostic serum sample from 187 case-control pairs. Of these, summary statistics for repeat samples were available from 84 controls for which the dates of first and repeat sampling were the furthest apart (collected \geq 4.7 years apart) (mean for years of sampling, 1999 and 2007). Serum PFOA concentrations in these populations were similar to those in the general US population as measured by the US National Health and Nutrition Examination Survey (NHANES). The actual analysis by Rhee et al. (2023a) and the main analysis by Purdue et al. (2023) used only the single (or first) sample for each subject.

The summary statistics from these repeat samples are posted on a National Cancer Institute GitHub project (NCI, 2024). Summary statistics for the repeated serum PFOA measurements for the controls in two cohorts were used to generate plausible serum concentrations for each participant at each time point, taking within-subject correlations into account. Five data sets of controls were generated for each cohort, with

Parameter ^b	Sai	mple 1 (T0)	Sa	mple 2 (T1)	Sa	mple 3 (T5)	All sam	ples (T0, T	`1, T5)
	Mean	Median	SD	Mean	Median	SD	Mean	Median	SD	Mean	Median	SD
Observed PFOA concentration (ng/mL)	3.88	3.63	1.8	3.87	3.53	1.99	4.69	4.53	2.43	4.15	NR	1.91
Simulated PFOA concentration (ng/mL)	3.85	3.88	1.82	3.82	3.81	1.96	4.63	4.61	2.4	4.1	4.09	1.81

Table A3.1 Descriptive statistics for participants with repeated samples (*n* = 60) from the study by <u>Rhee et al. (2023a)</u>^a

NR, not reported; PFOA, perfluorooctanoic acid; SD, standard deviation.

^a An analysis of prostate cancer in the PLCO study. The data reported here are posted in the National Cancer Institute GitHub project (NCL

<u>2024</u>).

^b Spearman correlations: T0-T1 observed, 0.78; simulated, 0.77; T1-T5 observed, 0.60; simulated, 0.60; T0-T5 observed, 0.62; simulated, 0.60.

the same number of controls as in the original studies for each simulated data set. The statistics used to generate simulated data were the mean and standard deviation (SD) for the PFOA serum concentrations among controls at each time point, as well as the Spearman correlation between each set of samples (three correlations for <u>Rhee et al. (2023a)</u> for three samples, one for <u>Purdue et al. (2023)</u> for two samples).

The samples appeared to have an approximately normal distribution, judging by the small differences between means and medians. From the three samples from the PLCO study used in Rhee et al. (2023a), the means and medians were 3.88 ng/mL and 3.63 ng/mL, 3.87 ng/mL and 3.53 ng/mL, and 4.69 ng/mL and 4.53 ng/mL, respectively (SDs, 1.8, 1.99, and 2.43) (Table A3.1). For the samples from Air Force servicemen in Purdue et al. (2023), the mean and median for the first sample were 6.8 ng/mL and 6.1 ng/mL (SD, 3.0), respectively, while for the second sample they were 5.5 ng/mL and 5.1 ng/mL (SD, 2.3), respectively (Table A3.2). For normally distributed data, Pearson and Spearman correlations are similar (de Winter et al., 2016), and we used Spearman correlation coefficients between samples to generate the simulated data, as an approximation of the Pearson correlations (Rhee et al., 2023a; ρ for sample T0–T1, 0.78; ρ for samples T1–T5, 0.60; ρ for samples T0–T5, 0.62); (<u>Purdue et al., 2023</u>; ρ for samples 1 and 2, 0.32) (<u>Table A3.1</u>, <u>Table A3.2</u>).

The distributions were generated using an R package (mvtnorm library) for generating multivariate normal samples with known means, standard deviations, and (Pearson) correlations between different sets of samples (<u>Genz and Bretz, 2009</u>). The mean across the five simulations for each control was then used to represent the simulated data for each sample.

Having generated simulated serum PFOA concentrations for each control at each time point (three time points for <u>Rhee et al., 2023a</u>, two time points for <u>Purdue et al., 2023</u>), mean concentrations across samples for each control in each study were used as an estimate of long-term exposure. For the controls in each study, the first samples were then compared with the long-term average exposure, the latter taken as the "true" exposure and the former as the "misclassified" exposure.

Exposures were categorized into quintiles, as in the original published analysis of <u>Rhee</u> <u>et al. (2023a)</u>, and used to determine the extent of misclassification across exposure categories using serum concentrations at the first time point versus the long-term exposure. "True" or longterm exposure values were then also generated

Table A3.2 Descriptive statistics for participants with repeated samples (*n* = 84) from the study by <u>Purdue et al. (2023)</u> ^a

Parameter ^b	Sample 1			Sample 2			Both samples
	Mean	Median	SD	Mean	Median	SD	Mean ^c
Observed PFOA concentration (ng/mL)	6.8	6.1	3.0	5.5	5.1	2.3	6.1
Simulated PFOA concentration (ng/mL)	6.8	6.8	2.9	5.5	5.5	2.3	6.1

PFOA, perfluorooctanoic acid; SD, standard deviation.

^a A study of testicular cancer in the United States Air Force cohort. The data reported here are posted in the National Cancer Institute GitHub project (<u>NCI, 2024</u>).

^b Spearman correlation between samples 1 and 2: observed, 0.32; simulation, 0.30.

^c Restricted to those with a second sample collected > 4.7 years (the median for controls sampled twice) after the first sample. The mean time between first and second samples for these 84 subjects was 7.8 years.

for a set of hypothetical cases for each study, such that there was a monotonic increasing trend across quintiles, and an approximate rate ratio of 1.5 for the highest versus the lowest quintile, using long-term exposure. The misclassification rates used for the hypothetical cases were the same as those observed in the controls (i.e. assuming non-differential exposure misclassification) to simulate "misclassified" exposure at the first time point for the hypothetical cases. Finally, epidemiological effect estimates (odds ratios) were computed across quintiles for cases and controls, using the long-term ("true") versus first sample ('misclassified') data.

Results

The simulated data corresponded well with the observed means and standard deviations for the original data, and the Spearman correlations between repeated samples in each study from the simulated data closely resembled the same correlation from the observed data. For example, for <u>Rhee et al. (2023a)</u>, the observed Spearman correlations (ρ) between samples T0–T1, T1–T5, and T0–T5 were 0.78, 0.60, and 0.62, respectively, while the Spearman correlations in the simulated data were 0.77, 0.60, and 0.60 (<u>Table A3.1</u>). The Spearman correlations between first and second samples for <u>Purdue et al. (2023)</u> in the simulated and observed data were 0.30 and 0.32, respectively (<u>Table A3.2</u>).

Comparing long-term "true" exposure (the mean across samples) with "misclassified" exposure (for the first sample alone), epidemiological results were quite similar, with a relatively small bias to the null when using only a single serum sample per participant (bias to the null is expected for non-differential misclassification, see <u>Weinberg et al., 1994</u>). For the data from <u>Rhee</u> et al. (2023a), the odds ratios (ORs) by quintile, using the long-term average, or "true", data were 1.00, 1.14, 1.29, 1.43, and 1.57 (*P* for trend, 0.007), while the odds ratios by quintile using the first serum sample only were 1.00, 1.12, 1.24, 1.35, and 1.42 (P for trend, 0.007) respectively, indicating only a slight bias to the null (Table A3.3). Similarly for the data from <u>Purdue et al. (2023)</u>, odds ratios by quintile using the long-term average were 1.00, 1.12, 1.23, 1.35, and 1.47 (P for trend, 0.005), whereas odds ratios by quintile using the first sample only were 1.00, 1.07, 1.13, 1.17, and 1.31 (P for trend, 0.02), again indicating only a slight bias towards the null (Table A3.4).

Table A3.3 Hypothetical "true" (long-term) and "misclassified" (first sample only) for cases (*n* = 675) and controls (*n* = 675) with a positive exposure–response relation for PFOA, based on the PLCO data in <u>Rhee et al. (2023a)</u>

Analysis	Exposure metric	PFOA quintile (ng/mL)	Cases	Controls	Odds ratio
True	Mean (T0, T1, T5) ^a	≤ 3.45	105	135	1.00
		> 3.45 to ≤ 3.90	120	135	1.14
		$> 3.90 \text{ to} \le 4.31$	135	135	1.29
		> 4.31 to ≤ 4.77	150	135	1.43
		> 4.77	165	135	1.57
				,	Frend-test <i>P</i> value, 0.007
Misclassified	T0 only	≤ 3.45	177	210	1.00
		> 3.45 to ≤ 3.90	123	130	1.12
		$> 3.90 \text{ to} \le 4.31$	145	138	1.24
		> 4.31 to ≤ 4.77	124	109	1.35
		> 4.77	105	88	1.42
				,	Frend-test <i>P</i> value, 0.007

OR, odds ratio; PFOA, perfluorooctanoic acid; PLCO, Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial.

^a The quintile cut-points used here stem from the data generated via multivariate normal distributions based on summary statistics from the controls in each study (posted in the National Cancer Institute GitHub project; <u>NCI</u>, 2024) with repeat samples, and differ from the cut-points in the original papers, which are based on all cases and controls in the original studies.

Table A3.4 Hypothetical "true" (long-term) and "misclassified" (first sample only) cases and controls with positive exposure–response relation for PFOA, based on the Air Force data in <u>Purdue et al. (2023)</u>

Analysis	Exposure metric	PFOA quintile (ng/mL)	Cases	Controls	Odds ratio
True	Mean (samples 1, 2) ^a	≤ 5.38	86	106	1.00
		> 5.38 to ≤ 5.93	96	106	1.12
		> 5.93 to ≤ 6.39	106	106	1.23
		> 6.39 to ≤ 6.81	116	106	1.35
		> 6.80	126	106	1.47
		Totals	530	530	Trend-test <i>P</i> value, 0.005
Misclassified	Sample 1 only	≤ 5.38	58	70	1.00
		> 5.38 to ≤ 5.93	47	53	1.07
		> 5.93 to ≤ 6.39	61	65	1.13
		> 6.39 to ≤ 6.81	70	72	1.17
		> 6.80	294	270	1.31
		Totals	530	530	Trend-test P value, 0.02

^a The quintile cut-points used here stem from the data generated via multivariate normal distributions based on summary statistics from the controls in each study (posted in the National Cancer Institute GitHub project; <u>NCI, 2024</u>) with repeat samples, and differ from the cut-points in the original papers, which are based on all cases and controls in the original studies.

Discussion

Single (first) samples represented rather well the mean of repeated samples taken an average of 5 and 8 years apart in two cohort studies of populations with background levels of exposure to PFOA (Spearman correlations, 0.87 and 0.83, for PLCO and Air Force data respectively). Others have demonstrated that changes in PFOA exposure estimates after correcting for measurement error cause little change in epidemiological findings in a high-exposure population in which the rank order of exposure among participants changes little with modest group-level misclassification (Avanasi et al., 2016). The same results were found here, for individual-level misclassification in low exposure populations more typical of the general population. Individual serum PFOA concentrations changed somewhat over time in these two cohorts, but the reported within-subject correlations were high, so the relative ranking of exposure remained approximately the same over time. This implied that tests for trend in disease risk in studies relying on only one serum sample might not differ markedly from those using a more accurate estimate of long-term exposure, i.e. the average for repeated samples over time. Such relative rankings might remain relatively constant in the case of the legacy PFAS, such as PFOA, because these chemicals have relatively long half-lives, and because local sources in the environment where the participants live (e.g. drinking-water, consumer products, and diet) may be relatively stable over time, at least during the decade of serum sampling represented by these two cohorts, despite longerterm secular trends in environmental levels and serum concentrations.

The limitations of these findings were the restriction to only a few repeated samples over relatively short time periods, before the production of PFOA and perfluorooctanesulfonic acid (PFOS) was phased out in the USA. However, another study with five repeated samples collected over almost 40 years in a general population sample in Norway also reported a relatively high within-subject correlation between samples over time (Nøst et al., 2014). Another limitation here was that all the data in the studies by <u>Rhee et al.</u> (2023a) and Purdue et al. (2023) came from men. However, while there are important differences between men and women, e.g. women's serum levels of PFAS change during pregnancy and after menopause (Dhingra et al., 2017; Steenland et al., 2018), there is no a priori reason to think that the findings regarding the consistency of relative rankings across time would be radically different for men and women. Other limitations included the assumption of normality, the use of Spearman instead of Pearson coefficients, and a relatively small number of simulations. However, it is not expected that changing these assumptions would have a substantial impact on the findings of the present analysis.

A3.2 Summary of the Working Group's ecological analysis of PFOA and orchiectomy among men aged 15–54 years in 21 municipalities of the Veneto region in Italy, 1997–2014

A3.2.1 Background on PFOA exposure in the Veneto region

The Trissino factory in the Veneto region, Italy, produced PFOA from 1968 to 2014 (Girardi and Merler, 2019). When PFOA production started in 1968, production was estimated to be about 12 tonnes per year in 1968–1970 and then increased over time until the 2000s, when the annual production of PFOA and its ammonium salt was on average 250 tonnes, peaking at 460 tonnes in 2007. PFOS production also occurred at the site but at much lower volumes, with an average of 36.6 tonnes per year in 2001–2011, peaking at 88 tonnes in 2004 (Girardi and Merler, 2019). As reported by Pitter et al. (2020):

"...based on general information on production practices, it is believed that the plant produced long-chain PFAS only, particularly PFOA and PFOS, from 1968 until 2001. PFOA reached the highest concentrations both in drinking-water and serum, consistent with previous reports from the Mid-Ohio Valley (Frisbee et al., 2009). PFBA and PFBS were found in high concentrations in drinking-water but were detected only in a minority of serum samples at relatively low concentration, whereas PFOS and PFHxS, which were scarcely represented in drinking-water, were detected in almost 100% of serum samples. This discrepancy may be explained by the exposure to PFOS and PFHxS from other sources, as demonstrated for the general population".

Releases of PFOA from the factory resulted in contamination of ground and surface water used for drinking in the region. The groundwater contamination plume extended over an area of 190 km² and affected both public waterworks and private wells. The municipalities in the area of maximum exposure (referred to as the "red area") are further divided into "red area A", which includes municipalities served by the contaminated waterworks that are also located on the groundwater contamination plume; and "red area B", which includes municipalities served by the contaminated waterworks but not located on the groundwater contamination plume. Initially, the red area was composed of 21 municipalities, with 126 000 inhabitants. In 2018, nine additional municipalities were added, some of which were only partially supplied by the contaminated waterworks. Currently, the red area is 595 km² wide and has a total population of approximately 140 000 people.

community since 2015 (Ingelido et al., 2018; Pitter et al., 2020). In 2015-2016, Ingelido et al. (2018) measured PFOA and PFOS and other PFAS in the serum of 257 individuals, aged 20-51 years, residing in municipalities in the affected areas (Altavilla, Brendola, Creazzo, Lonigo, Montecchio Maggiore, Sarego, and Sovizzo) and in 250 individuals living in uncontaminated areas. In each area, participants were selected and stratified by sex and age. Each participant had resided in an area for at least 10 years. Serum levels of PFOA were much higher in the contaminated areas (median, 13.77 ng/g; maximum, 754.50 ng/g) than in uncontaminated areas (median, 1.64 ng/g; maximum, 27.88 ng/g); similarly, PFOS levels were higher in the exposed group (median, 8.69 ng/g; maximum, 70.27 ng/g) than in the non-exposed (median, 5.84 ng/g; maximum, 118.58 ng/g). The Spearman correlation for PFOA and PFOS in serum was 0.743 in the exposed and 0.619 in the unexposed. In 2015–2016, Pitter et al. (2020) conducted a larger study of 18 345 participants aged 14-39 years at recruitment; 63.5% agreed to participate in the surveillance programme; serum results for people who had lived in the red area for < 1 year were excluded. The PFAS with the highest serum concentrations were PFOA (median, 44.4 ng/mL; interquartile range, IQR, 19.3-84.9 ng/mL), PFOS (median, 3.9 ng/mL; IQR, 2.6-5.8 ng/mL), and PFHxS (median, 3.9 ng/mL; IQR, 1.9–7.4 ng/mL). Within the red areas, median PFOA levels varied by community, ranging from 10.9 ng/mL in Terrazo to 73.3 ng/mL in Asigiliano-Veneto.

Biomonitoring has been conducted in this

Individuals in this contaminated area of the Veneto region are exposed to a mixture of PFAS, including PFOA and PFOS; but information on production volumes, water levels, and biomonitoring data are consistent with PFOA being the PFAS present at highest concentrations throughout the region (Ingelido et al., 2018; Mastrantonio et al., 2018; Girardi and Merler, 2019; Pitter et al., 2020; Giglioli et al., 2023). Taken as a whole, these data provided extensive evidence for PFOA contamination in the region, with both water and biomonitoring data showing differences in concentrations within the region. While other PFAS, notably PFOS and PFHxS, are correlated with PFOA, they are present at levels that are substantially lower than those of PFOA.

A3.2.2 Working Group analysis of orchiectomy data

The Working Group conducted an ecological analysis comparing biomonitoring data from Pitter et al. (2020) with data from an investigation on the frequency of orchiectomies in 21 municipalities in this region between 1997 and 2014 (Sistema Epidemiologico Regionale, 2016, summarized in English by Saugo et al., 2024). Orchiectomy was used as a proxy for diagnosis of testicular cancer [sensitivity and positive predictive values of 91.7% (95% CI, 88.0-95.4%) and 92.8% (95% CI, 89.3-96.2%), respectively, in this region]. Orchiectomies were ascertained using information in hospital discharge records, including address of residence, which included the main medical procedures from hospital stays and were completed for the purpose of reimbursement from the Italian national health system. As shown in Table A3.5 below, standardized incidence ratios (SIRs) for orchiectomy were estimated for each of the 21 municipalities separately by comparing the observed orchiectomies (n = 70, overall) versus expected numbers based on rates in the region overall that were standardized on age by 5-year age groups from 15 to 54 years (Sistema Epidemiologico Regionale, <u>2016</u>). A strong correlation was observed between median serum PFOA concentration and the rate of orchiectomy by municipality (Spearman correlation, 0.57; P = 0.006). The Working Group also conducted a Poisson regression of observed orchiectomy counts regressed on median PFOA levels across the 21 municipalities. The Poisson regression was done using the log of expected

events as an offset and correcting for dispersion. The rate ratio for each unit (ng/mL) increase of PFOA was 1.018 (95% CI, 1.006–1.031; P = 0.003).

The SAS code used in this analysis is presented in <u>Table A3.6</u>.

A3.3 Working Group meta-analysis of studies on kidney cancer to estimate rate ratio per unit (linear) increase in serum PFOA concentration

The Working Group conducted a meta-analysis that included estimates from the studies of Steenland and Woskie (2012), Barry et al. (2013), Vieira et al. (2013), Shearer et al. (2021), Rhee et al. (2023b), and Winquist et al. (2023). The studies by <u>Barry et al. (2013)</u>, <u>Vieira et al.</u> (2013), and Steenland and Woskie (2012) were included, although they overlap to an unknown extent, under the assumptions that: (i) they are largely independent; and (ii) the mortality rate ratio in Steenland and Woskie (2012) is roughly equivalent to what would have been obtained for an incidence rate ratio. The kidney cancer results from <u>Raleigh et al. (2014)</u> were not included, given that the exposure assessment in this study was based on air measurements, nor were those from Consonni et al. (2013), in which there were no serum data to permit the pooling of a comparable cumulative dose-response estimate with the other studies, or from Mastrantonio et al. (2018), because of its ecological design and lack of data on serum levels.

The Working Group used the approach of the meta-analysis by <u>Bartell and Vieira (2021)</u>. This approach uses categorical rate ratios based on contrasting the upper category (usually quartiles) with the referent, together with the assumed midpoints of the upper category and referent, to regress the log of the rate ratios on the midpoints to obtain a single linear continuous coefficient that estimates the change in log rate

Municipality (red area A or B)	Serum PFOA concentrations, by municipality ^a		Orchiectomy data, by municipality ^b		
	n (%) of samples	Median serum PFOA concentration (ng/mL)	Observed N	SIR	95% CI
Albaredo D'Adige (B)	767 (4.2%)	29	1	0.34	0.01-1.90
Alonte (A)	346 (1.9%)	62.6	1	1.13	0.03-6.27
Arcole (B)	899 (5.0%)	29.5	2	0.58	0.07-2.11
Asigliano Veneto (A)	161 (0.9%)	73.3	1	2.15	0.05-11.98
Bevilacqua (B)	216 (1.2%)	56.2	1	0.97	0.02-5.43
Bonavigo (B)	279 (1.5%)	29.8	1	0.87	0.02-4.85
Boschi Sant'Anna (B)	206 (1.1%)	38.4	0	0	0.00-3.77
Brendola (A)	1007 (5.6%)	41	6	1.60	0.59-3.48
Cologna Veneta (A)	1208 (6.7%)	53.9	2	0.44	0.05-1.60
Legnago (B)	2945 (16.3%)	22.2	11	0.83	0.42-1.49
Lonigo (A)	2569 (14.2%)	61.8	16	1.84	1.05-2.98
Minerbe (B)	628 (3.5%)	55.2	3	1.18	0.24-3.46
Montagnana (A)	1146 (6.3%)	67.6	8	1.54	0.67-3.04
Noventa Vicentina (A)	1410 (7.8%)	46.4	3	0.62	0.13-1.80
Pojana Maggiore (A)	767 (4.2%)	67.5	3	1.18	0.24-3.46
Pressana (A)	365 (2.0%)	58.8	1	0.69	0.02-3.82
Roveredo Di Guà (A)	263 (1.4%)	55.8	0	0	0.00-3.61
Sarego (A)	1124 (6.2%)	47.5	3	0.81	0.17-2.38
Terrazzo (B)	288 (1.6%)	10.9	0	0	0.00-2.51
Veronella (B)	778 (4.3%)	48.2	4	1.58	0.43-4.04
Zimella (A)	750 (4.1%)	49.9	3	1.10	0.23-3.21

Table A3.5 Data used by the Working Group for an ecological analysis of PFOA and orchiectomy among men aged 15–54 years in 21 municipalities of the Veneto region, Italy, 1997–2014

CI, confidence interval; PFOA, perfluorooctanoic acid; SIR, standardized incidence ratio.

^a Data from <u>Pitter et al. (2020)</u>.

^b Data from Sistema Epidemiologico Regionale (2016).

ratio per unit of (linear) PFOA. In addition, for Steenland and Woskie (2012), Barry et al. (2013), and Vieira et al. (2013), which used cumulative exposure, the Working Group divided the midpoints of exposure by the assumed average duration of exposure. In the case of Vieira et al. (2013), there were 10 years of cumulative exposure for cases and controls, so the Working Group divided by 10. In the case of Barry et al. (2013), the Working Group used the average duration of follow-up, which was 33 years (the average length of follow-up in the study), as the divisor. In the case of Steenland and Woskie (2012), the average length of follow-up was 30 years, so the Working Group divided the cumulative exposure by 30. The studies by <u>Steenland and Woskie</u> (2012), <u>Barry et al.</u> (2013), and <u>Rhee et al.</u> (2023b) did not have midpoints for the upper categories. For the studies by <u>Steenland and Woskie</u> (2012) and <u>Barry et al.</u> (2013), we multiplied the upper cut-point by 4, based on the observed midpoint in <u>Vieira et al.</u> (2013) (who studied a similar population), being about 4 times the lower level of the uppermost category. For <u>Rhee et al.</u> (2023b), the Working Group multiplied the upper cut-point by 2.5, based on the observed midpoint for the two other general population studies by <u>Shearer</u>

Table A3.6 SAS code used in the Working Group analysis of orchiectomy data

data one;

input medianpfoa sir numsamples obs exp estgeomean; *estgeomean comes from Pitter Table 2 regression; logexp=log(exp); Inmedianpfoa=log(medianpfoa); lnestgeomean=log(estgeomean); cards; 29 0.34 767 1 2.94 28.3 62.6 1.13 346 1 0.89 36.6 29.5 0.58 899 2 3.43 31.5 73.3 2.15 161 1 0.47 28.5 56.2 0.97 216 1 1.03 34.4 29.8 0.87 279 1 1.15 29.4 38.4 0 206 0 0.80 32.9 41 1.6 1007 6 3.76 24.5 53.9 0.44 1208 2 4.51 37.4 22.2 0.83 2945 11 13.19 20.4 61.8 1.84 2569 16 8.71 38.4 55.2 1.18 628 3 2.53 46.3 67.6 1.54 1146 8 5.18 39.7 46.4 0.62 1410 3 4.87 29.0 67.5 1.18 767 3 2.53 39.6 58.8 0.69 365 1 1.46 40.6 55.8 0 263 0 0.83 37.4 47.5 0.81 1124 3 3.68 32.4 10.9 0 288 0 1.19 10.9 48.2 1.58 778 4 2.54 48.0 49.9 1.1 750 3 2.73 36.6 *proc univariate plot; *var sir medianpfoa estgeomean; proc corr spearman; var medianpfoa sir; run;

proc freq; tables medianpfoa; run;

proc genmod; model obs=medianpfoa / dist=poisson link=log offset=logexp pscale; *best AIC; proc genmod; model obs=lnmedianpfoa / dist=poisson link=log offset=logexp pscale; run; proc genmod; model obs=estgeomean / dist=poisson link=log offset=logexp pscale; run; proc genmod; model obs=lnestgeomean / dist=poisson link=log offset=logexp pscale; run;

et al. (2021) and <u>Winquist et al. (2023)</u> for the upper category.

Once the continuous linear coefficient for each study was obtained, the Working Group then used an R package (metagen) to calculate random weights (inverse variance weights, where the variance is the sum of the within and between variance across studies) using the formulae from restricted maximum likelihood (REML) (<u>Veroniki et al., 2016</u>). The Working Group used random weights, given the high heterogeneity of the linear coefficient across studies (*I*² value, 0.91)

The meta-analysis described above gave the result for an increase in the rate ratio per increase of 10 ng/mL in PFOA as 1.15 (95% CI, 0.97–1.37), with an I^2 value of 91%.

We then also conducted, as a sensitivity analysis, a meta-analysis of <u>Winquist et al.</u> (2023), <u>Shearer et al.</u> (2021), <u>Rhee et al.</u> (2023b), and <u>Barry et al.</u> (2013), to avoid the overlapping nature of <u>Barry et al.</u> (2013) with <u>Steenland</u> and <u>Woskie</u> (2012), and <u>Vieira et al.</u> (2013), and choosing <u>Barry et al.</u> (2013) because it was an incidence study that also had the best exposure estimation.

This sensitivity analysis gave the result for an increase in the rate ratio per increase of 10 ng/mL PFOA as 1.21 (95% CI, 0.94–1.57) with an I^2 value of 95%.

As a general limitation to the meta-analysis, we noted the assumption of a linear exposureresponse relation, although we know that, in studies with continuous exposure coefficients (Barry et al., 2013; Shearer et al., 2021; Winquist et al., 2023; Rhee et al., 2023b), a log-linear model (i.e. log-transformed PFOA) seemed to fit the data better than did a linear model (i.e. untransformed PFOA). Other main limitations were: (i) the estimate of the linear coefficient using assumed midpoints of only two categories (uppermost and lowest); (ii) the use of average duration of exposure to transform cumulative exposure in Barry et al. and Viera et al. to an assumed average exposure; and (iii) the assumption in the studies by Rhee et al. (2023b), Shearer et al. (2021), and Winquist et al. (2023) that a single PFOA measurement is a good estimate of long-term lifetime average exposure (beyond a 5-8-year duration, discussed Section A3.1 of the present Annex). Given these limitations, as well as the high heterogeneity across studies with different strengths and weaknesses, the Working Group chose to not rely primarily on the metaanalysis of exposure-response relations to determine the hazard identification for kidney cancer in humans.

The R code used for these estimations is presented in Table A3.7.

Table A3.7 R code^a used for the Working Group's meta-analysis of kidney cancer to estimate rate ratio per unit (linear) increase in serum PFOA concentration

library(meta) #####Outcome: Summary RR based on 10 ng/mL increase #function to get increase per unit trendp = function(datalist){ lapply(datalist, function(df) { se1 = (log(df\$upper)-log(df\$RR))/qnorm(.975) # se of log RR for each dose category se2 = (log(df\$RR) - log(df\$lower))/qnorm(.975)se = (se1 + se2) / 2scores = 0:(length(se)-1) $if(se[1] == 0) \{$ $lm1 = lm(log(RR) \sim 0 + scores, weights = 1/se^2, data=df, subset=se>0)$ $lm2 = lm(log(RR) \sim 0 + mids, weights = 1 / se^2, data=df, subset=se>0)$ p1 = summary(lm1)\$coef[1,4] p2 = summary(lm2)\$coef[1,4] slope = summary(lm2)\$coef[1,1] se = summary(lm2)\$coef[1,2] } else { $lm1 = lm(log(RR) \sim scores, weights = 1 / se^2, data=df)$ $lm2 = lm(log(RR) \sim mids, weights = 1 / se^2, data=df)$ p1 = summary(lm1)\$coef[2,4] p2 = summary(lm2)\$coef[2,4] slope = summary(lm2)\$coef[2,1] se = summary(lm2)\$coef[2,2] return(c(p1,p2,slope,se)) }) ##### # Kidney/PFOA (Including Rhee overall) #per 10 ng/mL serum

}

```
kidney = list(
 shearer = data.frame(
 stlab = c("Shearer et al., 2020","",""),
 labs = c("0-4 ng/mL","4-5.5 ng/mL", "5.5-7.3 ng/mL", "7.3-27.2 ng/mL"),
 cutpoints = c(0, 4.0, 5.5, 7.3), # max given as 27.2; sub in after lapply
 RR = c(1.0, 1.47, 1.24, 2.63),
 lower = c(1, 0.77, 0.64, 1.33),
 upper = c(1, 2.80, 2.41, 5.20)),
 vieira = data.frame(
 stlab = c("Vieira et al., 2013","","",""),
 labs = c("0-3.7 ng/mL-yr", "3.8-88 ng/mL-yr", "89-197 ng/mL-yr", "198-599 ng/mL-yr", "600-4679 ng/mL-yr"), #this are
categories cut-points taken from Table S1 (cumulative over 10 years)
 cutpoints = c(0, 3.8, 89, 198, 600) / 10, \# max is given as 4679, sub in after lapply #divided by 10 because is 10 cumulative
exposure
 RR = c(1, 0.8, 1.2, 2.0, 2.1),
 lower = c(1, 0.4, 0.7, 1.3, 1.1),
 upper = c(1, 1.5, 2.0, 3.2, 4.2)),
 barry = data.frame(
 stlab = c("Barry et al., 2013","",""),
 labs = c("0-219 ng/mL-yr", "219-812 ng/mL-yr", "812-5358 ng/mL-yr", ">5358 ng/mL-yr", ">5358 ng/mL-yr", "219-812 ng/mL-yr", "812-5358 ng/mL-yr", ">5358 ng/mL-yr", "219-812 ng/mL-yr", "812-5358 ng/mL-yr", ">5358 ng/mL-yr", "219-812 ng/mL-yr", "812-5358 ng/mL-yr", ">5358 ng/mL-yr", ">5358 ng/mL-yr", "219-812 ng/mL-yr", "812-5358 ng/mL-yr", ">5358 ng/mL-yr", ">5358 ng/mL-yr", "</a>
 cutpoints = c(0, 219, 812, 5358) / 33, # ng/mL-yr / av age diag (divided by 33 because this is average length follow-up)
```

Table A3.7 (continued)

```
#cutpoints are reported in the meta-analysis by Bartell and Vieira (2021), in the R code within the supplement material but not
#in the original publication
RR = c(1, 1.23, 1.48, 1.58),
 lower = c(1, 0.70, 0.84, 0.88),
 upper= c(1, 2.17, 2.60, 2.84)),
 steenland = data.frame(
 stlab = c("Steenland and Woskie, 2012","",""),
 labs = c("0-904 ng/L-years", "904-1520 ng/mL-yr","1520-2700 ng/mL-yr", ">2700 ng/mL-yr"),
 cutpoints = c(0, 904, 1520, 2700) / 30, # Divided by 30 because of average follow-up
 RR = c(1.07, 1.37, 0.005, 2.66), # mortality; 3rd RR is 0 but cannot log
 lower = c(0.02, 0.28, 0.005, 1.15),
 upper= c(3.62, 3.99, 1.42, 5.24)),
 rhee = data.frame(
 stlab = c("Rhee et al., 2023","","",""),
 labs = c("0-3.27 \text{ ng/mL}", "3.27-4.47 \text{ ng/mL}", "4.47-6.22 \text{ ng/mL}", ">6.22 \text{ ng/mL}"),
 cutpoints = c(0, 3.27, 4.47, 6.22), #
 RR = c(1.0, 1.26, 1.26, 1.04),
 lower = c(1, 0.80, 0.78, 0.60),
 upper = c(1, 1.97, 2.05, 1.81)),
 winquist = data.frame(
 stlab = c("Winquist et al., 2023","",""),
 labs = c("0-3.9 ng/mL","3.9-5.2 ng/mL", "5.2-7.3 ng/mL", ">7.3"),
 cutpoints = c(0, 3.9, 5.2, 7.3), #
 RR = c(1.0, 0.93, 0.83, 1.20),
 lower = c(1, 0.56, 0.49, 0.71),
 upper = c(1, 1.56, 1.40, 2.04))
#calculate midpoints of the time-averaged serum PFOA categories within each study
kidney2 = lapply(kidney, function(df) {
cp = df$cutpoints
l = length(cp) + 1
cp[l] = 2.5 * cp[l-1] # assume max is 2.5*last cutpoint
 df$mids = apply(rbind(cp[-1],cp[-1]),2,mean)
return(df)
})
#in this we assumed that maximum is 2.5*last cutpoints, but in reality for some studies maximum is reported
kidney2$vieira$mids[5] = mean(c(600,4679)) / 10
kidney2shearer$mids[4] = mean(c(7.3,27.2))
kidney2$winquist$mids[4] = mean(c(7.3,54))
Vieira
kidney2$barry$mids[4] = mean(c(5358,21432))/33 ##assumed a maximum 4 times the highest cutpoint as more similar to Vieira
#apply ktrend function
(ktrend = trendp(kidney2))
# get RR and CI per 10 ng/mL increase in serum PFOA in each study
lapply(ktrend, function(df) round(exp(10^{*}df[3] + 10^{*}c(0, -1, 1)^{*}qnorm(.975)^{*}df[4]),2))
klogRR = c(ktrend$vieira[3], ktrend$barry[3], ktrend$shearer[3], ktrend$steenland[3], ktrend$rhee[3], ktrend$winquist[3])
kse = c(ktrend$vieira[4], ktrend$barry[4], ktrend$shearer[4], ktrend$steenland[4], ktrend$rhee[4], ktrend$winquist[4])
(m2 = metagen(klogRR,kse)) # meta-analysis for kidney
round(exp(10*c(m2$TE.fixed,m2$lower.fixed,m2$upper.fixed)),2)
```

round(exp(10*c(m2\$TE.random,m2\$lower.random,m2\$upper.random)),2)

Table A3.7 (continued)

m2\$pval.Q m2\$tau2 m2\$I2

#sensitivity taking out steenland and vieira
get RR and CI per 10 ng/mL increase in serum PFOA in each study
lapply(ktrend, function(df) round(exp(10*df[3] + 10*c(0,-1,1)*qnorm(.975)*df[4]),2))
klogRR3 = c(ktrend\$barry[3], ktrend\$shearer[3], ktrend\$rhee[3], ktrend\$winquist[3])
kse3 = c(ktrend\$barry[4], ktrend\$shearer[4], ktrend\$rhee[4], ktrend\$winquist[4])

(m3 = metagen(klogRR3,kse3)) # meta-analysis for kidney round(exp(10*c(m3\$TE.fixed,m3\$lower.fixed,m3\$upper.fixed)),2) round(exp(10*c(m3\$TE.random,m3\$lower.random,m3\$upper.random)),2)

m3\$pval.Q m3\$tau2 m3\$I2

^a Note that, if the code is copied from this document and pasted directly to R, the user may need to retype the quotation marks for the code to run correctly.

References

- Avanasi R, Shin HM, Vieira VM, Savitz DA, Bartell SM (2016). Impact of exposure uncertainty on the association between perfluorooctanoate and preeclampsia in the C8 health project population. *Environ Health Perspect.* 124(1):126–32. doi:<u>10.1289/ehp.1409044</u> PMID:26090912
- Barry V, Winquist A, Steenland K (2013). Perfluorooctanoic acid (PFOA) exposures and incident cancers among adults living near a chemical plant. *Environ Health Perspect.* 121(11–12):1313–8. doi:<u>10.1289/ehp.1306615</u> PMID:<u>24007715</u>
- Bartell SM, Vieira VM (2021). Critical review on PFOA, kidney cancer, and testicular cancer. J Air Waste Manag Assoc. 71(6):663–79. doi:10.1080/10962247.202 1.1909668 PMID:33780327
- Consonni D, Straif K, Symons JM, Tomenson JA, van Amelsvoort LG, Sleeuwenhoek A, et al. (2013). Cancer risk among tetrafluoroethylene synthesis and polymerization workers. *Am J Epidemiol*. 178(3):350–8. doi:<u>10.1093/aje/kws588</u> PMID:<u>23828249</u>
- de Winter JCF, Gosling SD, Potter J (2016). Comparing the Pearson and Spearman correlation coefficients across distributions and sample sizes: A tutorial using simulations and empirical data. *Psychol Methods*. 21(3):273– 90. doi:10.1037/met0000079 PMID:27213982
- Dhingra R, Winquist A, Darrow LA, Klein M, Steenland K (2017). A study of reverse causation: examining the associations of perfluorooctanoic acid serum levels with

two outcomes. *Environ Health Perspect*. 125(3):416–21. doi:<u>10.1289/EHP273</u> PMID:<u>27529882</u>

- Frisbee SJ, Brooks AP Jr, Maher A, Flensborg P, Arnold S, Fletcher T, et al. (2009). The C8 health project: design, methods, and participants. *Environ Health Perspect*. 117(12):1873–82. doi:<u>10.1289/ehp.0800379</u> PMID:<u>20049206</u>
- Genz A, Bretz F (2009). Computation of multivariate normal and t probabilities. Series Lecture Notes in Statistics. Heidelberg, Germany: Springer-Verlag.
- Giglioli S, Colombo L, Azzellino A (2023). Cluster and multivariate analysis to study the diffuse contamination of emerging per- and polyfluoroalkyl substances (PFAS) in the Veneto Region plain (north-eastern Italy). *Chemosphere*. 319:137916. doi:10.1016/j. chemosphere.2023.137916 PMID:36706810
- Girardi P, Merler E (2019). A mortality study on male subjects exposed to polyfluoroalkyl acids with high internal dose of perfluorooctanoic acid. *Environ Res.* 179(PtA):108743. doi:10.1016/j.envres.2019.108743 PMID:31542491
- Ingelido AM, Abballe A, Gemma S, Dellatte E, Iacovella N, De Angelis G, et al. (2018). Biomonitoring of perfluorinated compounds in adults exposed to contaminated drinking water in the Veneto Region, Italy. *Environ Int*. 110:149–59. doi:10.1016/j.envint.2017.10.026 PMID: 29108835
- Mastrantonio M, Bai E, Uccelli R, Cordiano V, Screpanti A, Crosignani P (2018). Drinking water contamination from perfluoroalkyl substances (PFAS): an ecological mortality study in the Veneto Region, Italy. *Eur J*

Public Health. 28(1):180–5. doi:<u>10.1093/eurpub/ckx066</u> PMID:<u>28541558</u>

- NCI (2024). Per- and polyfluoroalkyl substances (PFAS). Summary statistics. DCEG GitHub Organization. Bethesda (MD), USA: National Cancer Institute. Available from: <u>https://github.com/NCI-DCEG/PFAS</u> <u>cancer summary stat</u>.
- Nøst TH, Vestergren R, Berg V, Nieboer E, Odland JØ, Sandanger TM (2014). Repeated measurements of perand polyfluoroalkyl substances (PFASs) from 1979 to 2007 in males from northern Norway: assessing time trends, compound correlations and relations to age/birth cohort. *Environ Int*. 67:43–53. doi:<u>10.1016/j. envint.2014.02.011</u> PMID:<u>24657493</u>
- Pitter G, Da Re F, Canova C, Barbieri G, Zare Jeddi M, Daprà F, et al. (2020). Serum levels of perfluoroalkyl substances (PFAS) in adolescents and young adults exposed to contaminated drinking water in the Veneto Region, Italy: a cross-sectional study based on a health surveillance program. *Environ Health Perspect*. 128(2):27007. doi:10.1289/EHP5337 PMID:32068468
- Purdue MP, Rhee J, Denic-Roberts H, McGlynn KA, Byrne C, Sampson J, et al. (2023). A nested case-control study of serum per- and polyfluoroalkyl substances and testicular germ cell tumors among US Air Force servicemen. *Environ Health Perspect*. 131(7):77007. doi:10.1289/EHP12603 PMID:37458713
- Raleigh KK, Alexander BH, Olsen GW, Ramachandran G, Morey SZ, Church TR, et al. (2014). Mortality and cancer incidence in ammonium perfluorooctanoate production workers. *Occup Environ Med.* 71(7):500–6. doi:10.1136/oemed-2014-102109 PMID:24832944
- Rhee J, Barry KH, Huang WY, Sampson JN, Hofmann JN, Silverman DT, et al. (2023a). A prospective nested case-control study of serum concentrations of per- and polyfluoroalkyl substances and aggressive prostate cancer risk. *Environ Res.* 228:115718. doi:<u>10.1016/j.envres.2023.115718</u> PMID:<u>36958379</u>
- Rhee J, Chang VC, Cheng I, Calafat AM, Botelho JC, Shearer JJ, et al. (2023b). Serum concentrations of perand polyfluoroalkyl substances and risk of renal cell carcinoma in the Multiethnic Cohort Study. *Environ Int.* 180:108197. doi:<u>10.1016/j.envint.2023.108197</u> PMID: <u>37741007</u>
- Saugo M, Ioverno E, Olivieri A, Bertola F, Pasinato A, Ducatman A (2024). PFOA and testis cancer in the Veneto Region (Italy). *Environ Health*. 23(1):33. doi:10.1186/s12940-024-01064-4 PMID:38549149

- Shearer JJ, Callahan CL, Calafat AM, Huang WY, Jones RR, Sabbisetti VS, et al. (2021). Serum concentrations of per- and polyfluoroalkyl substances and risk of renal cell carcinoma. *J Natl Cancer Inst.* 113(5):580–7. doi:10.1093/jnci/djaa143 PMID:32944748
- Sistema Epidemiologico Regionale (2016). Ricognizione epidemiologica iniziale sulle orchiectomie per tumore del testicolo rilevate nell'area interessata dalla contaminazione idropotabile da PFAS. Veneto, Italy: Sistema Epidemiologico Regionale. Available from: https://sian.aulss9.veneto.it/index.cfm?method=mys. apridoc&iddoc=809. [Italian]
- Steenland K, Barry V, Savitz D (2018). Serum perfluorooctanoic acid and birthweight: an updated meta-analysis with bias analysis. *Epidemiology*. 29(6):765–76. doi:<u>10.1097/EDE.0000000000000903</u> PMID:<u>30063543</u>
- Steenland K, Woskie S (2012). Cohort mortality study of workers exposed to perfluorooctanoic acid. Am J Epidemiol. 176(10):909–17. doi:<u>10.1093/aje/kws171</u> PMID:<u>23079607</u>
- Veroniki AA, Jackson D, Viechtbauer W, Bender R, Bowden J, Knapp G, et al. (2016). Methods to estimate the between-study variance and its uncertainty in metaanalysis. *Res Synth Methods*. 7(1):55–79. doi:10.1002/ jrsm.1164 PMID:26332144
- Vieira VM, Hoffman K, Shin HM, Weinberg JM, Webster TF, Fletcher T (2013). Perfluorooctanoic acid exposure and cancer outcomes in a contaminated community: a geographic analysis. *Environ Health Perspect.* 121(3):318–23. doi:<u>10.1289/ehp.1205829</u> PMID:23308854
- Weinberg CR, Umbach DM, Greenland S (1994). When will nondifferential misclassification of an exposure preserve the direction of a trend? *Am J Epidemiol*. 140(6):565–71. doi:<u>10.1093/oxfordjournals.aje.a117283</u> PMID:<u>8067350</u>
- Winquist A, Hodge JM, Diver WR, Rodriguez JL, Troeschel AN, Daniel J, et al. (2023). Case-cohort study of the association between PFAS and selected cancers among participants in the American Cancer Society's Cancer Prevention Study II LifeLink cohort. *Environ Health Perspect*. 131(12):127007. doi:10.1289/EHP13174 PMID:38088576
ANNEX 4. SUPPLEMENTARY MATERIAL FOR SECTION 2, CANCER IN HUMANS

These supplementary online-only tables are available from: <u>https://publications.iarc.who.int/636</u>.

Please report any errors to imo@iarc.who.int.

The following tables were produced in draft form by the Working Group and were subsequently fact-checked but not edited:

Table S2.5	Epidemiological studies on exposure to PFOA or PFOS and cancers of the digestive tract
Table S2.6	Epidemiological studies on exposure to PFOA or PFOS and cancers of the brain, haematolymphoid system, and other cancers
Table S2.7	Epidemiological studies on exposure to PFOA or PFOS and cancers of all sites combined

ANNEX 5. SUPPLEMENTARY MATERIAL FOR SECTIONS 4.1 AND 4.2, MECHANISTIC EVIDENCE

These supplementary online-only tables are available from: <u>https://publications.iarc.who.int/636</u>.

Please report any errors to imo@iarc.who.int.

The following tables were produced in draft form by the Working Group and were subsequently fact-checked but not edited:

Table S4.1	Calculations to derive dissociation constants, by the Working Group
Table S4.23	End-points relevant to modulation of receptor-mediated effects in humans exposed to PFOA or PFOS
Table S4.24	End-points relevant to modulation of receptor-mediated effects in human cells in vitro exposed to PFOA or PFOS
Table S4.25	End-points relevant to modulation of receptor-mediated effects in experi mental systems in vivo exposed to PFOA or PFOS
Table S4.26	End-points relevant to modulation of receptor-mediated effects (endocrine) in experimental systems in vivo exposed to PFOA and PFOS

ANNEX 6. SUPPLEMENTARY MATERIAL FOR SECTION 4.3, EVALUATION OF HIGH-THROUGHPUT IN VITRO TOXICITY SCREENING DATA

These supplementary online-only tables (available from: <u>https://www.publications.iarc.who.</u> int/636) contain summaries of the findings (including the assay name, the corresponding key characteristic, the resulting "hit calls" both positive and negative, and any reported caution flags) for those chemicals evaluated in the present volume that have been tested in high-throughput screening assays performed by the United States Environmental Protection Agency (US EPA) and the United States National Institutes of Health. The results were generated by the Working Group using the software "kc-hits" (key characteristics of carcinogens – high-throughput screening discovery tool) available from <u>https://gitlab.com/i1650/kc-hits.git</u> (Reisfeld et al., 2022), using the US EPA Toxicology in the 21st Century (Tox21) and Toxicity Forecaster (ToxCast) assay data and the curated mapping of key characteristics to assays available at the time of the evaluations performed for *IARC Monographs* Volume 135. Data were available for perfluorooctanoic acid (PFOA), ammonium perfluorooctanoate (APFO), perfluorooctanesulfonic acid (PFOS), and potassium perfluorooctanesulfonate.

Please report any errors to imo@iarc.who.int.

- 1. Perfluorooctanoic acid (PFOA): ToxCast/Tox21 assay results mapped to the key characteristics of carcinogens
- 2. Ammonium perfluorooctanoate (APFO): ToxCast/Tox21 assay results mapped to the key characteristics of carcinogens
- 3. Perfluorooctanesulfonic acid (PFOS): ToxCast/Tox21 assay results mapped to the key characteristics of carcinogens
- 4. Potassium perfluorooctanesulfonate: ToxCast/Tox21 assay results mapped to the key characteristics of carcinogens

Reference

Reisfeld B, de Conti A, El Ghissassi F, Benbrahim-Tallaa L, Gwinn W, Grosse Y, et al. (2022). kc-hits: a tool to aid in the evaluation and classification of chemical carcinogens. *Bioinformatics*. 38(10):2961–2. doi:<u>10.1093/bioinformatics/btac189</u> PMID:<u>35561175</u>

SUMMARY OF FINAL EVALUATIONS

Summary of final evaluations for Volume 135

Agent	Evidence stream			Overall evaluation
	Cancer in humans	Cancer in experimental animals	Mechanistic evidence	
Perfluorooctanoic acid (PFOA)	Limited	Sufficient	Strong ^a	Group 1
Perfluorooctanesulfonic acid (PFOS)	Inadequate	Limited	<i>Strong</i> ^a	Group 2B

^a The mechanistic evidence was *strong* in exposed humans because PFOA and PFOS were found to induce epigenetic alterations and to be immunosuppressive.



This volume of the *IARC Monographs* provides evaluations of the carcinogenicity of two agents, perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS), and their corresponding isomers and salts.

PFOA and PFOS are per- and polyfluoroalkyl substances (PFAS) that are extremely resistant to degradation. First produced in the 1940s, PFOA has extensive uses, including in fluoropolymer manufacture and applications; in surface coatings conferring stain-, oil-, and water-resistance on household products, carpets, textiles, leather products, and food and feed packaging; in electrics and electronics; and in construction materials. With some similar uses to those of PFOA, PFOS additionally has applications in aqueous film-forming foams used in firefighting; in the fabrication of imaging devices and semiconductors; in photolithography and electroplating; and in insulation, dyes, and ink.

PFOA and PFOS occur ubiquitously in the environment, with high levels at pollution sources such as industrial sites and in firefighter-training areas and waste deposits. They may also be present in contaminated food, especially fish, seafood, and eggs. Occupationally exposed populations can have high levels of exposure, mainly via inhalation. The general population in contaminated areas is mainly exposed via drinking-water, and the general population in communities that are not near pollution sources is mainly exposed via diet and drinking-water.

An *IARC Monographs* Working Group reviewed evidence from epidemiological studies, cancer bioassays in experimental animals, and mechanistic studies to assess the carcinogenic hazard to humans of exposure to these agents and concluded that:

- PFOA is carcinogenic to humans (Group 1);
- **PFOS** is possibly carcinogenic to humans (Group 2B).